BELLSOUTH® / CLEC Agreement

Customer Name: e.spire Communications, Inc.

e.spire Communications, Inc.	2
Amendment 4/4/01	492
e.spire Amendment to add Virtual Collocation	494
Notices Amendment - e.spire Communications, Inc.	511
XMC 8State Rates, Port/Loop Combos & Ordinarily Combined UNE Combos	512

Note: This page is not part of the actual signed contract/amendment, but is present for record keeping purposes only.

of

e.spire

BellSouth Standard Interconnection Agreement

Agreement Effective Date: January 1, 2000	Agreement Expiration Date: December 31, 2002
OCN:	GAC:
CIC (if applicable):	ACNA:
Negotiator: Pat Finlen	Negotiator Tel No: (404) 927-8389
Location of Executive Summary:	Location of Interconnection Agreement:
t:\hendrix\finlen\espire\renegotiation\cmag	t:\hendrix\\finlen\espire\renegotiation\cmag*.*

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
Terms/Conditions PartA	1	6/15/99		Y	N		Interduction
	2	6/15/99		Y	N		Interpretation and Construction
	3	6/15/99		Y	Y	Н	Effective Date is date of execttion
	4	6/15/99		Y	Y	Н	Term is 3 years beginning 1/1/00
	5	6/15/99		Y	N		Resale
	6	6/15/99		Y	N		Unbundled Network Elemets
	7	6/15/99		Y	N		Interconnection
	8	6/15/99		Y	N		Collocation
	9	6/15/99		Y	N		Numbers and Number Portability
	10	6/15/99		Y	Y	Н	OSS Rates for entire agreement are in this section
	11	6/15/99		Y	N		Billing
	12	6/15/99		Y	N		Rights of Way, Conduits and Pole Attachments
	13	6/15/99		Y	N		Directory Listing
	14	6/15/99		Y	N		Parity
	15	6/15/99		Y	N		BFT/NBR Process
	16	6/15/99		Y	N		Local Dialing Parity
	17	6/15/99		Y	N		Law Enforcement and Civil Porcess

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
	18	6/15/99		Y	N		Performance Measurements
	19	6/15/99		Y	N		Liability and Indemnification
	20	6/15/99		Y	N		Disclaimer of Representation and Warranties
	21	6/15/99		Y	N		Intellectual Property Rights and Indemnification
	22	6/15/99		Y	N		Treatment of Proprietar and Confidiential Information
	23	6/15/99		Y	N		Assigments
	24	6/15/99		Y	Y	Н	Escalation Procedures - Each Party hereto shall provide the other party hereto with the names and telephone numbers or pagers of their respective managers up to the Vice Presidential level for the escalation of unresolved matters relating to their performance of their duties under this Agreement. Each Party shall supplement and update such information as necessary to facilitate prompt resolution of such matters. Each Party further agrees to establish an automatic internal escalation procedure relating to unresolved disputes arising under this Agreement.
	25	6/15/99		Y	N		Expedite Procedures
	26	6/15/99		Y	N		Resolution of Disputes - In the event of a Dispute between the Parties relating to this Agreement, and upon the written request of either Party, each of the Parties shall appoint within ten (10) calendar days after a Party's receipt of such request, a designated

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							representative who has authority to settle the Dispute and who is at a higher level of management than the persons with direct responsibility for administration of this Agreement.
	27		Y				Taxes - Formerly Section 13
	28			Y	N		Network Maintenance and Management
	29			Y	N		Changes in Subscriber Carrier Selection - Comply with 47CFR § 64.11000
	30			Y	N		Force Majeure
	31			Y	N		Year 2000 Compliance
	32			Y	N		Binding Effect
	33			Y	N		Consent
	34			Y	N		Modification of Agreement
	35			Y	N		Waivers
	36			Y	N		Expenses
	37			Y	N		Relationship of Parties
	38			Y	N		Third Party Beneficiaries
	39			Y	N		Cooperation of preventing End User Fraud
	40			Y	N		Good Faith Performance
	41			Y	N		Independent Contractors
	42			Y	N		Subcontrating
	43			Y	N		Serverability
	44			Y	N		Survial of Obligations
	45			Y	Y	Н	Customer Inquiries - Each Party is to refer all questions to the other Party. Each Party shall ensure that each of their representatives who receive inquiries regarding the other Party's

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							services: (i) provide the numbers described in Section 45.1 to callers who inquire about the other Party's services or products, and (ii) do not in any way disparage or discriminate against the other Party or its products or services.
	46			Y	Y	L	Compliance with Applicable Law
	47			Y	Y	L	Labor Relation
	48			Y	Y	L	Compliance with the Communications Law enforcement act of 1994
	49			Y	Y	L	Additional Fair Competition Requirements
	50		N				Governing Law - formerly Section 18
	51		N				Arm's Length Negotiations - formerly Section 19
	52			Y	N		Nonexclusive Dealings
	53		N				Notices - Formerly Section 20
	54		N				Rule of Construction - Formerly Section 21
	55		N				Headings of no force or effect - Formerly Section 22
	56			Y	N		Multiple Counterparts
	57		N				Filing of Agreement - Formerly Section 25
	58		N				Entire Agreement - Formerly Section 26
Terms/Conditions Part B							Added Numerous Telecommunication Definitions
Terms/Conditions Part C							
1-Resale	1	6/8/99		Y	N		
	2	6/8/99		Y	N		Lists type of services available for Resale

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
	3	6/8/99		Y	Y	Н	Section 3.3 BellSouth shall not be required to provide to e.spire Resale Services at a wholesale rate when those services are offered at a special promotional rate if: (a) Such promotions involve rates that will be in effect for not more than ninety (90) days; and (b) Such promotional offerings are not used to evade the wholesale rate obligation; for example, by making available a series of ninety (90) day promotional rates.
	4	6/8/99	N				J \ / J I
	5	6/8/99		Y	Y	Н	BellSouth shall use best efforts to provide e.spire forty-five (45) days advance notice via Internet posting of changes to the prices, terms or conditions of services available for Resale. To the extent that revisions occur between the time BellSouth notifies e.spire of changes under this Agreement and the time the changes are scheduled to be implemented,

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							BellSouth will notify e.spire of such revisions consistent with its internal notification process; provided that, e.spire shall not utilize any notice given under this subsection to market resold offerings of that service in advance of BellSouth. In addition, upon request BellSouth shall furnish e.spire with copies of publicly available service descriptions regarding the Resale Services. Notwithstanding the foregoing, e.spire shall not utilize any such BellSouth service descriptions as part of its own sales or marketing efforts.
	6	6/8/99	N				Formerly Section 5
	7	6/8/99	N				Formerly Section 6
	8	6/8/99	N				Formerly Section 7
	9	6/8/99	N				Formerly Section 8
	10	6/8/99		Y	Y	Н	Functionally Required to Support Resale Services
	11	6/8/99		Y	Y	Н	Resale of Customer Specifice Arrangements
	12	6/8/99	N				Formerly Section 9
	13	6/8/99	N				Formerly Section 10
	14	6/8/99	N				Formerly Section 11

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
	15	6/8/99	N				Formerly Section 12
	16	6/8/99	N				Formerly Section 13
	Exhibit A	6/8/99		Y	N		See OSS in GT&Cs
	Exhibit B	6/8/99		Y	Y	Y	Modified to Resale Short Term Promotions in KY and TN
	Exhibit C	6/8/99	N				
	Exhibit D	6/8/99	N				
	Exhibit E	6/8/99	N				
	Exhibit F	6/8/99	N				
	Exhibit G	6/8/99	N				
	Exhibit H	6/8/99	N				
2-Network Elements & Other Services	1	6/8/99					
	2	6/8/99		Y	N		Loop Definition now complies with UNE Remand Order Section 2.2.3 Loop cutover language - Intervals for loop conversions shall be as follows: (1) for single loop conversions per location, the conversion shall be completed within fifteen (15) minutes; (2) for up to ten (10) loop conversions per location, the conversion of all loops shall be completed within sixty (60) minutes, and each individual loop conversion shall be completed within fifteen (15) minutes; (3) for loop conversions not exceeding thirty (30) loops per location and not determined complex or exceptionally large, the conversion

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							of all loops shall be completed within one hundred and twenty (120) minutes: and (4) all loops above a thirty loop quantity, or ten (10) loop quantity and determined as complex (a cut that requires more operation than a single cut point), will be negotiated by e.spire and BellSouth prior to the due date.
							Section 2.5 Loop Conditioning language added
	3	6/8/99					
	4	6/8/99					
	5	6/8/99					
	6	6/8/99					Section 6.2.1 - Modified Definition of Subloop
							The subloop network element is defined as any portion of the loop that is technically feasible to access at terminals in BellSouth's outside plant, including inside wire owned and controlled by BellSouth, if any. An accessible terminal is any point on the loop where technicians can access the wire or fiber within the cable without removing a splice case to reach the wire or fiber within. Such points may include, but are not limited to, the pole or pedestal, the network interface device, the minimum point of entry, the single point of interconnection, the main distribution frame, the remote terminal, and the

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							feeder/distribution interface. 6.2.1.2 Technical feasibility. Subject to applicable and effective FCC rules and orders, if the Parties are unable to reach agreement, pursuant to voluntary negotiations, as to whether it is technically feasible, or whether sufficient space is available, to unbundle the subloop at the point where a carrier requests, BellSouth shall have the burden of demonstrating to the Commission, pursuant to state arbitration proceedings under section 252 of the Act, that there is not sufficient space available, or that it is not technically feasible, to unbundle the subloop at the point requested. 6.2.1.3. Best practices. Once any state commission has determined that it is technically feasible to unbundle subloops at a designated point, BellSouth shall have the burden of demonstrating, pursuant to state arbitration proceedings under section 252 of the Act, that it is not technically feasible, or that sufficient space is not available, to unbundle its own loops at such a point. 6.2.1.4. Subloop access via collocation. Where requested by e.spire, BellSouth shall provide access to the subloop in accordance

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							with the FCC's collocation rules, 47 C.F.R. §§ 51.321-323.
							6.2.1.5. Single point of interconnection. Subject to applicable and effective FCC rules and orders, BellSouth shall provide a single point of interconnection at multi-unit premises that is suitable for use by multiple carriers. This obligation is in addition to BellSouth's obligation to provide nondiscriminatory access to subloops at any technically feasible point. If the Parties are unable to negotiate terms and conditions regarding a single point of interconnection, issues in dispute, including compensation due BellSouth under forward-looking pricing principles, shall be resolved under the dispute resolution processes set forth in this Agreement.
	7	6/8/99		Y	Y	Н	Switching definition modified to comply with UNE Remand Order
	8	6/8/99		Y	Y	Н	Transport definition modified to omply with UNE Remand Order
	9	6/8/99	Y				
	10	6/8/99	Y				
	11	6/8/99	Y				
	12	6/8/99	Y				
	13	6/8/99	Y				
	14	6/8/99	Y				
	15	6/8/99	Y				

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
	16	6/8/99	Y				
	17	New		Y	Y	Н	Combination Language Section 17.2 - EELs Section 17.5 - Special access Conversions Section 17.6 - Loop Port Combo's
	18	6/8/99		Y	N		True Up is for TN Only
		6/8/99					
		6/8/99					
	Exhibit A	6/8/99					
	Exhibit B	6/8/99					
	Exhibit C	1Q2000 6/5/00		Y	N		Added the following to each section - Where the state Commission has adopted rates for the rate elements containedherein, it is the intent of the Parties to reflect such rates in thisExhibit and to apply the same consistent with applicable FCC and Commissionrules and orders.
3-Local Interconnection	1	6/8/99		Y	N		Rewrote Section 1.2 for clarification - NXX assignment must comply with industry standards.
	2	6/8/99		Y	Y	Н	Section 2.3 Cost of two way trunking will be based on percentage of orginating traffic.
	3	6/8/99	Y				
	4	6/8/99		Y	N		Added langauge to include "Equal in Quality" based on FCC Rules
	5	6/8/99	Y				
	6	6/8/99		Y	Y	Н	The parties will compensate for ISP traffic

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							The Parties will compensate each other on a mutual and reciprocal basis for the transport and termination of Local Traffic at the following rates:
							1/1/00 – 12/31/00 \$.00200 per MOU 1/1/01 – 12/31/01 \$.00175 per MOU 1/1/02 – 12/31/02 \$.00150 per MOU
							Section 6.9.1 - Definition of Switched Access Traffic - Switched Access Traffic is defined as telephone calls requiring local transmission or switching services for the purpose of the origination or termination of Telephone Toll Service. Switched Access Traffic includes the following types of traffic: Feature Group A, Feature Group B, Feature Group C, Feature Group D, toll free access (e.g., 800/877/888), 900 access, and their successors or similar Switched Exchange Access Services. The
							Parties have been unable to agree as to whether "Voice-Over-Internet Protocol" transmissions ("VOIP") which cross LATA boundaries constitute Switched Access Traffic. Notwithstanding the foregoing, and without waiving any rights with respect to either Party's position as to the jurisdictional nature of VOIP, the Parties agree to abide by any effective and applicable FCC rules and orders

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							regarding the nature of such traffic and the compensation payable by the Parties for such traffic, if any.
	7	6/8/99	N				
	8	6/8/99		Y	Y	Н	OSS Rates are in General Terms and Conditions
	Exhibit A	6/8/99					Contains new NCUC Rates
4-Physical Collocation	1	6/8/99	N				
-	2	6/8/99	N				
	3	6/8/99		Y	N		Section 3.3 Added clarifying language on shared collocation
							Deleted Adjacent Collocation Section
	4	6/8/99	N				
	5	6/8/99		Y	N		Section 5.9 Added clarifying Language on Interference or Impairment
	6	6/8/99		Y	N		Section 6.9 Added clarifying Language on Virtual Collocation Transition
	7	6/8/99	N				
	8	6/8/99	N				
	9	6/8/99	N				
	10	6/8/99	N				
	11	6/8/99	N				
	12	6/8/99	N				
	13	6/8/99	N				
	14	6/8/99	N				
	Exhibit A	6/8/99		Y	Y	Н	Include recently Ordered NCUC rates

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
	Exhibit B	6/8/99	N				
5-Access to Numbers &		6/8/99	N				
Number Portability	1						
	2	6/8/99	N				
	3	6/8/99	N				Renumbered Section to Section 2
	4	6/8/99	N				Renumbered Section to Section 2
	5	6/8/99	N				Renumbered Section to Section 2
	6	6/8/99	N				Now Section 3
	7	6/8/99	N				Now Section 4
	Exhibit A	6/8/99		Y	Y		Contains new NCUC Rates
6-Ordering/Provisioning	1	6/8/99		Y	Y	Н	Section 1.2 - If BellSouth begins working on an order which is scheduled to be completed during standard hours, but, due solely to BellSouth's delay, completes the work after standard hours, no such additional charges shall apply. If e.spire requests such provisioning services outside of normal hours of operation, BellSouth shall quote within three (3) Business Days of the request, a rate for such services in accordance with BellSouth's FCC No. 1 tariff. If e.spire accepts BellSouth's quote, BellSouth shall provide the requested services. If BellSouth agrees to provide expanded standard coverage hours to any other Telecommunications Carrier, e.spire shall be able immediately to avail itself of the same expanded hours on the same terms as made available to such other Telecommunications

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
							Carrier.
	2	6/8/99		Y	Y	M	Sections 2.2.1 - 2.3.5 Interfaces, CSR Information, Flow Through
	3	6/8/99		Y	Y	Н	Section 3.4 Contact Numbers - The Parties agree to provide one another with toll-free contact numbers for the purpose of ordering, provisioning and maintenance of services. BellSouth shall provide single points of contact ("SPOC") for the provisioning of Resale Services (LCSC) and UNEs (UNE Center) ordered by e.spire. Pre-ordering and ordering shall be available via an electronic interface seven (7) days a week, 24 hours a day. BellSouth shall provide access to assistance for technical issues such as connectivity and passwords related to LENS, TAG and TAFI, and to the "EDI Central Group" for technical problems with EDI. Assistance will be available by telephone during normal business hours and through other contacts on nights, weekends and holidays. Section 3.8 - Ordering and Provisioning Information
	Exhibit A	New		Y	Y	L	Disaster Recover Plan
7-Billing & Billing Accuracy Certification	1	6/8/99	N	•	1	2	Distance Record Final

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia-	Deviation	Deviation Affect	If Compliance	If Deviation, enter Paragraph No. And Brief Description of Deviation.
			tion		Compliance Y/N	Item, Priority H/M/L	If different by state, note here also.
	2	6/8/99	N				
	3	6/8/99	N				
	4	6/8/99	N				
	5	6/8/99	N				
	6	6/8/99	N				
	7	6/8/99	N				
	Exhibit A	6/8/99		Y	Y		Contains new NCUC Rates
8-ROW/Conduits/PoleAtt	1	6/8/99	N				
9-Perf Measurement	Scope	1Q2000 3/6/00	N				
	Reporting	1Q2000 3/6/00	N				
	Modifications	1Q2000 3/6/00	N				
	Enforcement Mechanisms	1Q2000 3/6/00	N				
	Exhibit A	1Q2000 3/6/00	N				
	Exhibit B	1Q2000 3/6/00	N				
	Exhibit C	1Q2000 3/6/00	N				
	Exhibit D	1Q2000 3/6/00	N				
	Exhibit E	1Q2000 3/6/00	N				
Attachment 10	Implementation Schedule	6/8/99	N				

of

e.spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	No Devia- tion	Deviation	Deviation Affect Compliance Y/N	If Compliance Item, Priority H/M/L	If Deviation, enter Paragraph No. And Brief Description of Deviation. If different by state, note here also.
Attachment 11	BAPCO	New		Y	Y	Н	Bapco Agreement
Attachment 12	BFR/NBR	New		Y	Y	Н	Details on BFR/NBR Process
				-	·		

AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS INC. AND e.spire COMMUNICATIONS, INC.

Table of Contents

1.	INTRODUCTION	1
2.	INTERPRETATION AND CONSTRUCTION	2
3.	EFFECTIVE DATE	3
4.	TERM OF THE AGREEMENT	3
5.	RESALE	4
6.	UNBUNDLED NETWORK ELEMENTS	5
7.	INTERCONNECTION	5
8.	COLLOCATION	
9.	NUMBERS AND NUMBER PORTABILITY	5
10.	OPERATIONAL SUPPORT SYSTEMS	5
11.	BILLING	7
12.	RIGHTS OF WAY, CONDUITS AND POLE ATTACHMENTS	7
13.	DIRECTORY LISTINGS	
14.	PARITY	8
15.	BONA FIDE REQUEST/NEW BUSINESS REQUEST PROCESS FOR FURTHER UNBUNDLIN	\G8
16.	LOCAL DIALING PARITY	8
17.	LAW ENFORCEMENT AND CIVIL PROCESS	9
18.	PERFORMANCE MEASUREMENTS	9
19.	LIABILITY AND INDEMNIFICATION	10
20.	DISCLAIMER OF REPRESENTATIONS AND WARRANTIES	12
21.	INTELLECTUAL PROPERTY RIGHTS AND INDEMNIFICATION	12
22.	TREATMENT OF PROPRIETARY AND CONFIDENTIAL INFORMATION	13
23.	ASSIGNMENTS	15
24.	ESCALATION PROCEDURES	15
25.	EXPEDITE PROCEDURES	16
26.	RESOLUTION OF DISPUTES	16
27.	TAXES	16

Table of Contents

28.	NETWORK MAINTENANCE AND MANAGEMENT	19
29.	CHANGES IN SUBSCRIBER CARRIER SELECTION	19
30.	FORCE MAJEURE	20
31.	YEAR 2000 COMPLIANCE	20
32.	BINDING EFFECT	20
33.	CONSENT	21
34.	MODIFICATION OF AGREEMENT	21
35.	WAIVERS	21
36.	EXPENSES	22
37.	RELATIONSHIP OF PARTIES	22
38.	THIRD PARTY BENEFICIARIES	22
39.	COOPERATION ON PREVENTING END USER FRAUD	22
40.	GOOD FAITH PERFORMANCE	22
41.	INDEPENDENT CONTRACTORS	22
42.	SUBCONTRACTING	23
43.	SEVERABILITY	23
44.	SURVIVAL OF OBLIGATIONS	23
45.	CUSTOMER INQUIRIES	23
46.	COMPLIANCE WITH APPLICABLE LAW	24
47.	LABOR RELATIONS	24
48. ("C	COMPLIANCE WITH THE COMMUNICATIONS LAW ENFORCEMENT ACT OF 1994 ALEA")	24
49.	ADDITIONAL FAIR COMPETITION REQUIREMENTS	25
50.	GOVERNING LAW	25
51.	ARM'S LENGTH NEGOTIATIONS	25
52.	NONEXCLUSIVE DEALINGS	25
53.	NOTICES	25
54.	RULE OF CONSTRUCTION	26
55.	HEADINGS OF NO FORCE OR EFFECT	26

Table of Contents

56.	MULTIPLE COUNTERPARTS	26
57.	FILING OF AGREEMENT	27
58.	ENTIRE AGREEMENT	27
59	SIGNATURE	2.7

AGREEMENT

THIS AGREEMENT is made by and between BellSouth Telecommunications, Inc., ("BellSouth"), a Georgia corporation, and e.spire Communications, Inc., Delaware corporation, on behalf of its operating affiliates identified in Part C hereof collectively, "e.spire" and shall be deemed effective as of the Effective Date specified in Section 3 hereof. This Agreement may refer to either BellSouth or e.spire or both as a "Party" or "Parties."

WITNESSETH

WHEREAS, BellSouth is an Incumbent Local Exchange Carrier ("ILEC") as defined by the Telecommunications Act of 1996 ("the FTA"), authorized to provide Telecommunications Services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee; and

WHEREAS, e.spire is a Competitive Local Exchange Carrier ("CLEC") authorized to provide Telecommunications Services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee; and

WHEREAS, both Parties wish to interconnect their facilities and exchange local traffic for termination, and e.spire wishes to purchase unbundled Network Elements and resell BellSouth's Telecommunications Services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee, specifically by means of exercising their rights and fulfilling their obligations pursuant to Sections 251 and 252 of the Communications Act of 1934, as amended by the Federal Telecommunications Act of 1996 ("the Act").

NOW THEREFORE, in consideration of the mutual agreements contained herein, and other good and valuable consideration, the sufficiency of which are hereby acknowledged, BellSouth and e.spire agree as follows:

1. INTRODUCTION

- 1.1 This Agreement sets forth the terms, conditions and prices under which BellSouth agrees to provide to e.spire (a) services for Resale (hereinafter referred to as Resale Services), (b) unbundled Network Elements, (c) Interconnection, (d) Collocation, and (e) all other features and arrangements described in this Agreement.
- 1.2 The Network Elements, and Resale Services provided pursuant to this Agreement may be connected by e.spire to other Network Elements, Resale Services or Access Services provided by BellSouth, or to any network components or services provided by e.spire itself or by any other vendor or Telecommunications Carrier. Subject to the requirements of this Agreement, e.spire may at any time add,

- delete, relocate or modify the Resale Services or Network Elements purchased hereunder.
- 1.3 BellSouth and e.spire may fulfill the requirements imposed upon them by this Agreement by themselves or may cause their agents to take action to fulfill such responsibilities.
- 1.4 This Agreement includes and incorporates herein the Attachments to this Agreement, and all Appendices, Exhibits, Schedules, Addenda and Amendments hereto.

2. INTERPRETATION AND CONSTRUCTION

- 2.1 Capitalized terms used in this Agreement shall have the respective meanings specified in Part B hereof, or As Defined by the Act.
- 2.2 The definitions in Part B hereof shall apply equally to both the singular and plural forms of the terms defined. Whenever the context may require, any pronoun used in this Agreement shall include the corresponding masculine, feminine and neuter forms. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation" throughout this Agreement. The words "shall" and "will" are used interchangeably throughout this Agreement and the use of either connotes a mandatory obligation. The use of one or the other shall not mean a different degree of right or obligation for either Party.
- 2.3 References herein to Articles, Sections, Exhibits, Attachments, Appendices, and Schedules shall be deemed to be references to Articles and Sections of, and Exhibits, Attachments, Appendices and Schedules to, this Agreement unless the context shall otherwise require.
- 2.4 The headings of the Articles, Sections, Exhibits, Attachments, Appendices and Schedules are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 2.5 Unless the context shall otherwise require, any reference to any agreement, other instrument (including BellSouth, e.spire or any third party offerings, guides or practices), statute, regulation, rule or Tariff is to such agreement, instrument, statute, regulation, rule or tariff as amended and supplemented from time to time (and, in the case of a statute, regulation, rule or Tariff, to any successor provision).
- 2.6 Subject to the terms set forth in Attachment 2 regarding rates and charges, and the Resale Discount set forth in Attachment 1, each Party hereby incorporates by reference those provisions of its Tariffs that govern the provision of any of the services or facilities provided hereunder. However, if any provision of this Agreement and any applicable Tariff cannot be reasonably construed or interpreted to avoid conflict, the provision contained in this Agreement shall prevail. If any provision contained in the main body of this Agreement and any

Attachment, Schedule, Appendix or Exhibit hereto cannot reasonably be construed or interpreted to avoid conflict, the provision contained in the main body of this Agreement shall prevail. The fact that a condition, right, obligation, or other term appears in this Agreement but not in any such Tariff shall not be interpreted as, or be deemed grounds for finding of a conflict for purposes of this Section 2 or in a tariff and not in this Agreement.

2.7 Technical references that describe the practices, procedures and specifications for certain services (and the applicable interfaces relating thereto) are listed in Attachment 2 and other relevant Attachments hereto to assist the Parties in meeting their respective responsibilities hereunder.

3. EFFECTIVE DATE

This Agreement becomes effective on the date when executed by both Parties (the "Effective Date").

4. TERM OF THE AGREEMENT

- 4.1 The term of this Agreement shall be three (3) years, beginning on January 1, 2000 and ending on December 31, 2002, and shall apply to the state(s) of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee. If, as of the expiration of this Agreement, a Subsequent Agreement (as defined in Section 4.2 below) has not been executed by the Parties, this Agreement shall continue on a month-to-month basis while a Subsequent Agreement is being negotiated. The Parties' rights and obligations with respect to this Agreement after expiration shall be as set forth in Sections 4.3 and 4.4 below.
- 4.2 The Parties agree that by no later than one hundred and eighty (180) days prior to the expiration of this Agreement, they shall commence negotiations with regard to the rates, terms, and conditions of any service arrangement described herein to be effective beginning on the expiration date of this Agreement ("Subsequent Agreement").
- 4.3 If, within one hundred and thirty-five (135) days of commencing the negotiation referred to in Section 4.2, above, the Parties are unable to satisfactorily negotiate new resale and/or local interconnection terms, conditions and prices, either Party may petition the Commission to establish appropriate local interconnection and/or resale arrangements pursuant to 47 U.S.C. 252. The Parties agree that, in such event, they shall encourage the Commission to issue its order regarding the appropriate local Interconnection and/or Resale arrangements no later than the expiration date of this Agreement. The Parties further agree that in the event the Commission does not issue its order prior to the expiration date of this Agreement, or if the Parties continue beyond the expiration date of this Agreement to negotiate the Services herein without Commission intervention, the terms, conditions and prices ultimately ordered by the Commission, or negotiated by the Parties, will be effective retroactive to the day following the expiration date

- of this Agreement. Until the Subsequent Agreement becomes effective, the Parties shall continue to Interconnect, exchange traffic, provide Resale Services and Network Elements, pursuant to the terms and conditions of this Agreement.
- 4.4 Notwithstanding the foregoing, in the event that as of the date of expiration of this Agreement and conversion of this Agreement to a month-to-month term, the Parties have not entered into a Subsequent Agreement and either no arbitration proceeding has been filed in accordance with Section 4.3 above, or the Parties have not mutually agreed (where permissible) to extend the arbitration window for petitioning the applicable Commission(s) for resolution of those terms upon which the Parties have not agreed, then either Party may terminate this Agreement upon sixty (60) days notice to the other Party. In the event that BellSouth terminates this Agreement as provided above, BellSouth shall continue to offer services to e.spire pursuant to the terms, conditions and rates set forth in BellSouth's Statement of Generally Available Terms (SGAT) to the extent an SGAT has been approved by the applicable Commission(s). If any state Commission has not approved a BellSouth SGAT, then upon BellSouth's termination of this Agreement as provided herein, BellSouth will continue to provide services to e.spire pursuant to BellSouth's then current standard interconnection agreement. In the event that the SGAT or BellSouth's standard interconnection agreement becomes effective as between the Parties, the Parties may continue to negotiate a Subsequent Agreement, and the terms of such Subsequent Agreement shall be effective retroactive to the day following expiration of this Agreement.
- 4. 5 Subject to Section 1 of Attachment 7 of this Agreement, where termination would affect the services(s) provided by the purchaser thereof to its End Users, the Network Elements and Resale Services provided hereunder are vital to e.spire (or, if applicable, to BellSouth) and must be continued without interruption. When e.spire provides or retains another vendor to provide such comparable Network Elements or Resale Services, BellSouth and e.spire agree to cooperate in an orderly and efficient transition of affected facilities and services to e.spire or such other vendor. The Parties shall coordinate as necessary to ensure that the levels and quality of the Network Elements and Resale Services is not degraded. Each Party will exercise its best efforts to effect an orderly and efficient transition to the replacement provider of facilities and/or services.

5. RESALE

BellSouth shall make available to e.spire for resale each of the Telecommunications Services it provides at retail to subscribers that are not Telecommunications Carriers on a nondiscriminatory basis, without unreasonable or discriminatory restrictions, and at a discount reflecting its avoided costs, pursuant to the rates, terms and conditions contained in Attachment 1 hereto and in accordance with the Act, including Sections 251(b)(1), 251(c)(4) and 252(d)(3), and all applicable FCC and State Commission rules, regulations, orders and policies (hereafter, "Requirements").

6. UNBUNDLED NETWORK ELEMENTS

BellSouth shall provide e.spire nondiscriminatory access to unbundled Network Elements at just, reasonable and nondiscriminatory rates, terms and conditions for the provision of Telecommunications Services pursuant to the rates, terms and conditions in Attachment 2 hereto in accordance with the Act, including Sections 251(c)(3) and 252(d)(1), and all applicable FCC and State Commission Requirements.

7. INTERCONNECTION

BellSouth shall provide for the Interconnection of e.spire's facilities and equipment with BellSouth's network at any technically feasible point within BellSouth's network at just, reasonable and nondiscriminatory rates, terms and conditions, pursuant to the terms and conditions of Attachment 3 hereto and the prices in Attachment 3 hereto and in accordance with the Act, including Sections 251(a), 251(b)(5), 251(c)(2), 252(d)(1), 252(d)(2), and all applicable FCC and State Commission Requirements.

8. COLLOCATION

BellSouth shall permit Physical Collocation of e.spire's equipment at the premises of BellSouth as necessary for Interconnection and access to unbundled Network Elements on a just, reasonable, and nondiscriminatory basis, pursuant to the rates, terms and conditions set forth in Attachment 4 hereto and in accordance with the Act, including Section 251(c)(6), and all applicable FCC and State Commission Requirements. BellSouth shall permit Virtual Collocation pursuant to the terms of BellSouth's FCC Tariff No. 1.

9. NUMBERS AND NUMBER PORTABILITY

BellSouth shall provide e.spire with nondiscriminatory access to telephone numbers and Local Number Portability pursuant to the terms and conditions set forth in Attachment 5 hereto, and in accordance with the Act, including Sections 251(b)(2), 251(b)(3), 251(e), and applicable FCC and State Commission Requirements.

10. OPERATIONAL SUPPORT SYSTEMS

BellSouth shall provide e.spire with nondiscriminatory access to BellSouth's Operations Support Systems ("OSS") functions for pre-ordering, ordering, provisioning, maintenance and repair, and billing of Resale Services and unbundled Network Elements, pursuant to the terms and conditions of Attachment 6, and consistent with the Act, including Sections 251(c)(3), 251(d)(1), and 252(d)(1), applicable State statutes, all applicable FCC and State Commission Requirements.

BellSouth has developed and made available the following mechanized systems by which e.spire may submit LSRs electronically.

LENS Local Exchange Navigation System

EDI Electronic Data Interface

EDI-PC Electronic Data Interface – Personal Computer

TAG Telecommunications Access Gateway

RoboTAGTM Enhanced TAG system*

LSRs submitted by means of one of these interactive interfaces will incur an OSS electronic ordering charge as specified in the table below. An individual LSR will be identified for billing purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (mail, fax, courier, etc.) will incur a manual order charge as specified in the table below:

	AL, GA, LA, MS, SC	FL, KY, NC, TN
OPERATIONAL SUPPORT SYSTEMS		
OSS LSR charge, per LSR received from the	\$3.50	\$3.50
CLEC by one of the OSS interactive interfaces		
	SOMEC	SOMEC
Incremental charge per LSR received from the	See applicable rate	\$19.99
CLEC by means other than one of the OSS	element	
interactive interfaces		SOMAN

Note: In addition to the OSS charges, applicable discounted service order and related discounted charges apply per the tariff.

Denial/Restoral OSS Charge

In the event e.spire provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and, therefore will be billed as one LSR per location.

Cancellation OSS Charge

e.spire will incur an OSS charge for an accepted LSR that is later canceled by e.spire.

Note: Supplements or clarifications to a previously billed LSR will not incur another OSS charge.

Network Elements and Other Services Manual Additive

The Commissions in Alabama, Georgia, Louisiana, Mississippi and South Carolina have ordered incremental manual non-recurring charges (NRC) for Network Elements and Other Services ordered by means other than one of the interactive interfaces. These ordered Network Elements and Other Services manual additive NRCs will apply in these states, rather than the charge per LSR.

^{*}RoboTAG shall be available at a price to be negotiated by the Parties.

Threshold Billing Plan

The Parties agree that e.spire will incur the mechanized rate for all LSRs, both mechanized and manual, if the percentage of mechanized LSRs to total LSRs meets or exceeds the threshold percentages shown below:

Year	Ratio: Mechanized/Total LSRs
2000	80%
2001	90%

The threshold plan will be discontinued in 2002.

BellSouth will track the total LSR volume for each CLEC for each quarter. At the end of that time period, a Percent Electronic LSR calculation will be made for that quarter based on the LSR data tracked in the LCSC. If this percentage exceeds the threshold volume, all of that CLECs' future manual LSRs will be billed at the mechanized LSR rate. To allow time for obtaining and analyzing the data and updating the billing system, this billing change will take place on the first day of the second month following the end of the quarter (e.g. May 1 for 1Q, Aug 1 for 2Q, etc.). There will be no adjustments to the amount billed for previously billed LSRs.

11. BILLING

BellSouth shall provide intercarrier billing arrangements to e.spire pursuant to the rates, terms and conditions of Attachment 7 hereto, and in accordance with the Act, and applicable FCC and State Commission Requirements.

12. RIGHTS OF WAY, CONDUITS AND POLE ATTACHMENTS

BellSouth shall offer e.spire non-discriminatory access to its poles, ducts, conduits and rights of way, pursuant to terms and conditions set forth in Attachment 8 hereto, and in accordance with the Act, including Sections 224, and 251(b)(4), applicable State statutes, and all applicable FCC and State Commission Requirements. Such access to rights-of-way, conduit and pole attachments shall be at least equal in quality to, and at rates, terms and conditions at least as favorable, as comparable arrangements provided by BellSouth to itself, its Subsidiaries or Affiliates, or to any other Telecommunications Service Provider.

13. DIRECTORY LISTINGS

BellSouth shall make available to e.spire's End Users White Page Directory Listing in accordance with the Act, and all applicable FCC and State Commission Requirements. BellSouth shall include e.spire subscriber listings in BellSouth's Directory Assistance database(s) at no charge provided e.spire does not charge for such listings. BellSouth and e.spire shall formulate appropriate procedures regarding lead time, timeliness, format and content of listing information.

14. PARITY

When e.spire purchases, pursuant to Attachment 1 of this Agreement, Telecommunications Services from BellSouth for the purposes of Resale to End Users, BellSouth shall provide said services so that the services are equal in quality, subject to the same conditions, and provided within the same provisioning time intervals that BellSouth provides to its Affiliates, Subsidiaries and End Users to the extent technically feasible. The quality of a Network Element provided pursuant to Attachment 2 of this Agreement, as well as the quality of the access to such Network Element provided by BellSouth to e.spire, shall be at least equal in quality to that which BellSouth provides to itself. The quality of the Interconnection between the networks of BellSouth and e.spire shall be at a level that is equal to that which BellSouth provides itself, a Subsidiary, an Affiliate, or any other. Person. The Interconnection facilities shall be designed to meet the same technical criteria and service standards that are used within BellSouth's network and shall extend to a consideration of service quality as perceived by end users and service quality as perceived by e.spire. "Equal in quality" in each instance also means the same interface specifications, provisioning, installation, maintenance, testing and repair intervals for the same services, or facilities.

15. BONA FIDE REQUEST/NEW BUSINESS REQUEST PROCESS FOR FURTHER UNBUNDLING

BellSouth shall, upon request of e.spire, provide e.spire access to its Network Elements at any technically feasible point for the provision of e.spire's Telecommunications Services where such access is necessary and failure to provide access would impair the ability of e.spire to provide services that it seeks to offer. Any request by e.spire for access to a Network Element, Interconnection option, or for the provisioning of any service or product that is not already available, shall be treated as a Bona Fide Request/New Business Request, and shall be submitted to BellSouth pursuant to the Bona Fide Request/New Business Request process set forth in Attachment 12 hereto.

16. LOCAL DIALING PARITY

BellSouth shall provide local dialing parity As Described by the Act and required by FCC rules, regulations and policies. e.spire End Users shall not have to dial any greater number of digits than BellSouth End Users to complete the same call. In addition, e.spire End Users shall experience at least the same service quality as BellSouth End Users in terms of post-dial delay, call completion rate and transmission quality.

17. LAW ENFORCEMENT AND CIVIL PROCESS

- 17.1 <u>Intercept Devices</u>. Local and federal law enforcement agencies periodically request information or assistance from Local Exchange Carriers. When either Party receives a request associated with an End User of the other Party, the receiving Party will refer such request to the appropriate Party, unless the request directs the receiving Party to attach a pen register, trap-and-trace or form of intercept on the Party's own facilities, in which case that Party will comply with any valid request, to the extent the receiving Party is able to do so.
- 17.2 Subpoenas. e.spire agrees that BellSouth will respond to subpoenas and court ordered requests delivered directly to BellSouth for the purpose of providing call detail records when the targeted telephone numbers belong to e.spire end users. Billing for such requests will be generated by BellSouth and directed to the law enforcement agency initiating the request. e.spire agrees that in cases where e.spire receives subpoenas or court ordered requests for call detail records for targeted telephone numbers belonging to e.spire end users, e.spire will advise the law enforcement agency initiating the request to redirect the subpoena or court ordered request to BellSouth. Billing for call detail information will be generated by BellSouth and directed to the law enforcement agency initiating the request. e.spire will provide e.spire end user and/or other customer information that is available to e.spire in response to subpoenas and court orders for their own customer records. BellSouth will redirect subpoenas and court ordered requests for e.spire end user and/or other customer information to e.spire for the purpose of providing this information to the law enforcement agency.
- 17.3 <u>Law Enforcement Emergencies</u>. If a Party receives a request from a law enforcement agency to implement at its switch a temporary number change, temporary disconnect, or one-way denial of outbound calls for an End User of the other Party, the receiving Party will comply so long as it is a valid emergency request. Neither Party shall be held liable for any claims or damages arising from compliance with such requests.

18. PERFORMANCE MEASUREMENTS

The Parties agree that the services offered and rendered by BellSouth pursuant to this Agreement shall be provisioned at parity to the service levels and intervals for which BellSouth performs such services for itself, its Affiliates or any other Person or Telecommunications Carrier. The Parties further agree that the service level specified for each item addressed by the Performance Measurements set forth in Attachment 9 shall be at parity. BellSouth agrees to meet the performance standard of parity as measured by the relevant Performance Measurements for each reporting period during the term of this Agreement and any extension thereof. In addition to the service quality measurements described in Attachment 9, no later than third quarter of 2000, BellSouth shall also provide to e.spire a disaggregation report for the provisioning and maintenance of (1)

enhanced extended links ("EELs") and (2) Frame Relay resale. These additional disaggregation reports shall be provided to e.spire as a separate, e.spire-specific report published each month on BellSouth's PMAP website rather than as part of the standard service quality measurements.

19. LIABILITY AND INDEMNIFICATION

- 19.1 <u>BellSouth Liability</u>. BellSouth shall take financial responsibility for its own actions in causing, or its lack of action in preventing, unbillable or uncollectible e.spire revenues.
- 19.2 <u>e.spire Liability</u>. In the event that e.spire consists of two (2) or more separate entities as set forth in the preamble to this Agreement, all such entities shall be jointly and severally liable for the obligations of e.spire under this Agreement.
- 19.3 <u>Liability for Acts or Omissions of Third Parties</u>. Neither BellSouth nor e.spire shall be liable for any act or omission of another telecommunications company providing a portion of the services provided under this Agreement.
- 19.4 Limitation of Liability.
- 19.4.1 <u>Limitations in Tariffs</u>. A Party may, in its sole discretion, provide in its tariffs and contracts with its Customer and third parties that relate to any service, product or function provided or contemplated under this Agreement, that to the maximum extent permitted by Applicable Law, such Party shall not be liable to Customer or third Party for (i) any loss relating to or arising out of this Agreement, whether based in contract, tort or otherwise, that exceeds the amount such Party would have charged that applicable person for the service, product or function that gave rise to such loss and (ii) Consequential Damages. To the extent that a Party elects not to place in its tariffs or contracts such limitations of liability, and the other Party incurs a loss as a result thereof, such Party shall indemnify and reimburse the other Party for that portion of the loss that would have been limited had the first Party included in its tariffs and contracts the limitations of liability that such other Party included in its own tariffs at the time of such loss.
- 19.4.2 With respect to any claim or suit, whether based in contract, tort or any other theory of legal liability, by e.spire, any e.spire Customer or by any other Person or entity, for damages associated with any of the services provided by BellSouth pursuant to or in connection with this Agreement, including but not limited to the installation, provision, preemption, termination, maintenance, repair or restoration of service, and subject to the provisions of the remainder of this Section, BellSouth's liability shall be limited to an amount equal to the proportionate charge for the service provided pursuant to this Agreement for the period during which the service was affected. Notwithstanding the foregoing, claims for damages by e.spire, any e.spire Customer or any other Person or entity, resulting from the gross negligence or willful misconduct of BellSouth, shall not be subject to such limitation of liability.

- 19.4.3 With respect to any claim or suit, whether based in contract, tort or any other theory of legal liability, by BellSouth, any BellSouth Customer or by any other Person or entity, for damages associated with any of the services provided by e.spire pursuant to or in connection with this Agreement, including but not limited to the installation, provision, preemption, termination, maintenance, repair or restoration of service, and subject to the provisions of the remainder of this Section, e.spire's liability shall be limited to an amount equal to the proportionate charge for the service provided pursuant to this Agreement for the period during which the service was affected. Notwithstanding the foregoing, claims for damages by BellSouth, any BellSouth Customer or any other Person or entity resulting from the gross negligence or willful misconduct of e.spire, shall not be subject to such limitation of liability.
- 19.5 Neither Party shall be liable for damages to the other Party's terminal location, POI or the other Party's Customers' premises resulting from the furnishing of a service, including, but not limited to, the installation and removal of equipment and associated wiring, except to the extent the damage is caused by such Party's gross negligence or willful misconduct or by a company's failure to properly ground a local loop after disconnection.
- 19.6 Except to the extent caused by gross negligence or willful misconduct neither Party shall be responsible or liable for indirect, incidental, or consequential damages, including, but not limited to, economic loss or lost business or profits, damages arising from the use or performance of equipment or software, or the loss of use of software or equipment, or accessories attached thereto, delay, error, or loss of data. In connection with this limitation of liability, each Party recognizes that the other Party may, from time to time, provide advice, make recommendations, or supply other analyses related to the Services, or facilities described in this Agreement, and, while each Party shall use diligent efforts in this regard, the Parties acknowledge and agree that this limitation of liability shall apply to provision of such advice, recommendations, and analyses.
- 19.7 <u>Indemnification for Certain Claims</u>. The Party providing services hereunder, its Affiliates and its parent company, shall be indemnified, defended and held harmless by the Party receiving services hereunder against any claim, loss or damage arising from the receiving Party's use of the services provided under this Agreement pertaining to (1) claims for libel, slander or invasion of privacy arising from the content of the receiving company's own communications, or (2) any claim, loss or damage claimed by the Customer of the Party receiving services arising from such Party's use or reliance on the providing company's services, actions, duties, or obligations arising out of this Agreement. Notwithstanding the foregoing, to the extent that a claim, loss or damage is caused by the gross negligence or willful misconduct of a supplying Party, the receiving Party shall have no obligation to indemnify, defend and hold harmless the supplying Party hereunder.

20. DISCLAIMER OF REPRESENTATIONS AND WARRANTIES

EXCEPT AS EXPRESSLY PROVIDED UNDER THIS AGREEMENT, NEITHER PARTY HERETO MAKES OR RECEIVES ANY WARRANTY, EXPRESS OR IMPLIED, WITH RESPECT TO THE SERVICES, FACILITIES, FUNCTIONS AND PRODUCTS PROVIDED UNDER OR CONTEMPLATED BY THIS AGREEMENT, AND THE PARTIES DISCLAIM THE IMPLIED WARRANTIES OF MERCHANTABILITY AND OF FITNESS FOR A PARTICULAR PURPOSE.

21. INTELLECTUAL PROPERTY RIGHTS AND INDEMNIFICATION

- 21.1 <u>No License</u>. No patent, copyright, trademark or other proprietary right is licensed, granted or otherwise transferred by this Agreement.
- 21.2 <u>Publicity</u>. Each Party is strictly prohibited from any use, including but not limited to use in sales, marketing or advertising of Telecommunications Services, of the other Party's name, service mark, trademark or logo. Either Party may reference the name of the other party in connection with factual statements in response to questions from Customers or potential Customers regarding the source of the underlying service.
- 21.3 Ownership of Intellectual Property. Any intellectual property which originates from or is developed by a Party shall remain in the exclusive ownership of that Party. Except for a limited license to use patents or copyrights to the extent necessary for the Parties to use any facilities or equipment (including software) or to receive any service solely as provided under this Agreement, no license patent, copyright, trademark or trade secret, or other proprietary or intellectual property right now or hereafter owned, controlled or licensable by a Party, is granted to the other Party or shall be implied or arise by estoppel. It is the responsibility of each Party to ensure at no additional cost to the other Party that it has obtained any necessary licenses in relation to intellectual property of third Parties used in its network that may be required to enable the other Party to use any facilities or equipment (including software), to receive any service, or to perform its respective obligations under this Agreement.
- 21.4 <u>Indemnification.</u> The Party providing a service pursuant to this Agreement will defend the Party receiving such service or data provided as a result of such service against claims of copyright, trademarks or trade secrets, or other potential intellectual property infringement arising solely from the use by the receiving Party of such service and will indemnify the receiving Party for any damages awarded based solely on such claims. Such indemnification shall not, however, extend to claims for patent infringement to the extent the alleged infringement results from:
- 21.4.1 Modification of the service by someone other than the providing Party and/or its subcontractors, where there would be no such infringement or violation in the absence of such modification; or

- 21.4.2 The combination, operation or use of the service with any product, data or apparatus not provided by the providing Party and/or its subcontractors, where there would be no such infringement or violation in the absence of such combination, operation or use; or
- 21.4.3 conformance to specifications of the indemnitee which would necessarily result in infringement.
- 21.5 <u>Claim of Infringement</u>. In the event that use of any facilities or equipment (including software), becomes, or in reasonable judgment of the Party who owns the affected network is likely to become, the subject of a claim, action, suit, or proceeding based on intellectual property infringement, then said Party shall promptly and at its sole expense, but subject to the limitations of liability set forth below: (i) modify or replace the applicable facilities or equipment (including software) while maintaining its form and function, or (ii) obtain a license sufficient to allow such use to continue.
- 21.5.1 In the event (i) or (ii) are commercially unreasonable, then said Party may terminate, upon reasonable notice, this contract with respect to use of, or services provided through use of, the affected facilities or equipment (including software), but solely to the extent required to avoid the infringement claim.
- 21.5.2 Promptly after receipt of notice of any claim or the commencement of any action for which a Party may seek indemnification pursuant to this Section, such Party ("Indemnified Party") shall promptly give written notice to the other Party ("Indemnifying Party") of such claim or action, but the failure to so notify the Indemnifying Party shall not relieve the Indemnifying Party of any liability it may have to the Indemnified Party except to the extent the Indemnifying Party has actually been prejudiced thereby. The Indemnifying Party shall be obligated to assume the defense of such claim, at its own expense. The Indemnified Party shall cooperate with the Indemnifying Party's reasonable requests for assistance or information relating to such claim, at the Indemnifying Party's expense. The Indemnified Party shall have the right to participate in the investigation and defense of such claim or action, with separate counsel chosen and paid for by the Indemnified Party.
- 21.6 <u>Exclusive Remedy</u>. The foregoing shall constitute the Parties' sole and exclusive remedies and obligations with respect to a third party claim of intellectual property infringement arising out of the conduct of business under this Agreement.

22. TREATMENT OF PROPRIETARY AND CONFIDENTIAL INFORMATION

22.1 For the purposes of this Agreement, "Confidential Information" means confidential or proprietary technical or business information given by the Discloser to the Recipient. All Confidential Information shall be in writing or other tangible form and clearly marked with a confidential, private or proprietary legend. In addition, by way of example and not limitation, all orders for Facilities

- and Services provided herein, placed by either Party pursuant to this Agreement, and, information that would constitute Customer Proprietary Network Information ("CPNI") of either Party's Customers pursuant to the Act and the rules and regulations of the FCC, and Recorded Usage Data, whether disclosed by one Party to the other or otherwise acquired by one Party from the other in the course of the performance of this Agreement, shall be deemed Confidential Information of either Party for all purposes under this Agreement.
- 22.2 For a period of five (5) years from the receipt of Confidential Information from the Discloser, except as otherwise specified in this Agreement, the Recipient agrees (a) to use it only for the purpose of performing under this Agreement, (b) to hold it in confidence and disclose it to no one other than its employees having a need to know for the purpose of performing under this Agreement, and (c) to safeguard it from unauthorized use or disclosure using at least the same degree of care with which the Recipient safeguards its own Confidential Information. If the Recipient wishes to disclose the Discloser's Confidential Information to a third-party agent or consultant, such agent or consultant must have agreed in writing to comply with and be bound by the terms of this Section 22. Each Party expressly commits, without limitation, that no Confidential Information of the other Party will be made available to any employee, agent or consultant with retail sales, marketing or sales management-related functions other than as expressly provided in this Agreement.
- 22.3 The Recipient may make copies of Confidential Information only as reasonably necessary to perform its obligations under this Agreement. All such copies will be subject to the same restrictions and protections as the original and will bear the same copyright and proprietary rights notices as are contained on the original.
- The Recipient agrees to return all Confidential Information in tangible form received from the Discloser, including any copies made by the Recipient within thirty (30) days after a written request is delivered to the Recipient, or to destroy all such Confidential Information if directed to do so by Discloser except for Confidential Information that the Recipient reasonably requires to perform its obligations under this Agreement. If either Party becomes aware that it has lost or made an unauthorized disclosure of the other Party's Confidential Information, it will notify such other party immediately and use reasonable efforts to retrieve the lost or wrongfully disclosed information.
- 22.5 The Recipient will have no obligation to safeguard Confidential Information: (a) which was in the possession of the Recipient free of restriction prior to its receipt from the Disclosure; (b) after it becomes publicly known or available through no breach of this Agreement by the Recipient; (c) after it is rightfully acquired by the Recipient free of restrictions on its disclosure; or (d) after it is independently developed by personnel of the Recipient to whom the Discloser's Confidential Information had been previously disclosed. In addition, either Party will have the right to disclose Confidential Information to any mediator, arbitrator, state or federal regulatory body, or court, in the conduct of any mediation, arbitration or approval of this Agreement, so long as, in the absence of an applicable protective

order, the Discloser has been promptly notified by the Recipient and so long as the Recipient undertakes all lawful measures to avoid disclosing such information until Discloser has had reasonable time to negotiate a protective order or confidentiality agreement, as applicable, with any such mediator, arbitrator, state or regulatory body or court, and complies with any protective order that covers the Confidential Information.

- 22.6 The Parties acknowledge that an individual End User may simultaneously seek to become or be a Customer of both Parties. Nothing in this Agreement is intended to limit the ability of either Party to use customer specific information lawfully obtained from End Users or sources other than the Disclosing Party.
- 22.7 Each Party's obligations to safeguard Confidential Information disclosed prior to expiration or termination of this Agreement will survive such expiration or termination of the Agreement.
- 22.8 Each Party agrees that the Discloser may be irreparably injured by a disclosure of Confidential Information in breach of this Agreement by the Recipient or its representatives, and the Discloser shall be entitled to seek equitable relief, including injunctive relief and specific performance, in the event of any breach or threatened breach of the confidentiality provisions of this Agreement. Such remedies shall not be deemed to be the exclusive remedies for a breach of this Agreement, but shall be in addition to all other remedies available at law or in equity.

23. ASSIGNMENTS

Neither Party hereto may assign or otherwise transfer its rights or obligations under this Agreement, except with the prior written consent of the other Party hereto, which consent shall not be unreasonably withheld; provided, however, that, so long as the performance of any assignee is guaranteed by the assignor: (i) either Party may assign its rights and delegate its benefits, duties and obligations under this Agreement, without the consent of the other Party, to any Affiliate of such Party and (ii) either Party may assign its rights and delegate its benefits, duties and obligations under this Agreement, without the consent of the other, to any person or entity that obtains control of all or substantially all of such assigning Party's assets, by stock purchase, asset purchase, merger, foreclosure, or otherwise. Each Party shall notify the other in writing of any such assignment. Nothing in this Section is intended to impair the right of either Party to utilize subcontractors.

24. ESCALATION PROCEDURES

Each Party hereto shall provide the other party hereto with the names and telephone numbers or pagers of their respective managers up to the Vice Presidential level for the escalation of unresolved matters relating to their performance of their duties under this Agreement. Each Party shall supplement and update such information as necessary to facilitate prompt resolution of such matters. Each Party further agrees to establish an

automatic internal escalation procedure relating to unresolved disputes arising under this Agreement.

25. EXPEDITE PROCEDURES

Each Party shall promptly establish a nondiscriminatory procedure for expediting installation and repair of facilities provided pursuant to this Agreement.

26. RESOLUTION OF DISPUTES

Except as otherwise provided herein, any dispute, controversy or claim (individually and collectively, a "Dispute") arising under this Agreement shall be resolved in accordance with the procedures set forth in this Section. In the event of a Dispute between the Parties relating to this Agreement, and upon the written request of either Party, each of the Parties shall appoint within ten (10) calendar days after a Party's receipt of such request, a designated representative who has authority to settle the Dispute and who is at a higher level of management than the persons with direct responsibility for administration of this Agreement. The designated representatives shall meet as often as they reasonably deem necessary in order to discuss the Dispute and negotiate in good faith in an effort to resolve such Dispute. The specific format for such discussions will be left to the discretion of the designated representatives; however, all reasonable requests for relevant information made by one Party to the other Party shall be honored. If the Parties are unable to resolve issues related to a Dispute within thirty (30) days after a Party's request is made for appointment of designated representatives as set forth above, either Party may seek relief from the appropriate state regulatory agency of any Dispute upon which the Parties hereto are unable to reach agreement or may seek such other relief to which it is entitled to under Applicable Law. Notwithstanding the foregoing, in no event shall the Parties permit the pendency of a Dispute to disrupt service to any e.spire or BellSouth End User, unless such service is damaging or interfering with customer services or network operations.

27. TAXES

- 27.1 <u>Definition</u>. For purposes of this Section, the terms "taxes" and "fees" shall include but not limited to federal, state or local sales, use, excise, gross receipts or other taxes or tax-like fees of whatever nature and however designated (including tariff surcharges and any fees, charges or other payments, contractual or otherwise, for the use of public streets or rights of way, whether designated as franchise fees or otherwise) imposed, or sought to be imposed, on or with respect to the services furnished hereunder or measured by the charges or payments therefore, excluding any taxes levied on income.
- 27.2 Taxes and Fees Imposed Directly On Either Providing Party or Purchasing Party.

- 27.2.1 Taxes and fees imposed on the providing Party, which are not permitted or required to be passed on by the providing Party to its customer, shall be borne and paid by the providing Party.
- 27.2.2 Taxes and fees imposed on the purchasing Party, which are not required to be collected and/or remitted by the providing Party, shall be borne and paid by the purchasing Party.
- 27.3 <u>Taxes and Fees Imposed on Purchasing Party But Collected And Remitted By</u> Providing Party.
- 27.3.1 Taxes and fees imposed on the purchasing Party shall be borne by the purchasing Party, even if the obligation to collect and/or remit such taxes or fees is placed on the providing Party.
- 27.3.2 To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.
- 27.3.3 If the purchasing Party determines that in its opinion any such taxes or fees are not payable, the providing Party shall not bill such taxes or fees to the purchasing Party if the purchasing Party provides written certification, reasonably satisfactory to the providing Party, stating that it is exempt or otherwise not subject to the tax or fee, setting forth the basis therefor, and satisfying any other requirements under applicable law. If any authority seeks to collect any such tax or fee that the purchasing Party has determined and certified not to be payable, or any such tax or fee that was not billed by the providing Party, the purchasing Party may contest the same in good faith, at its own expense. In any such contest, the purchasing Party shall promptly furnish the providing Party with copies of all filings in any proceeding, protest, or legal challenge, all rulings issued in connection therewith, and all correspondence between the purchasing Party and the taxing authority.
- 27.3.4 In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.
- 27.3.5 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.
- 27.3.6 Notwithstanding any provision to the contrary, the purchasing Party shall protect, indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other charges or payable expenses (including reasonable attorney fees) with

- respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.
- 27.3.7 Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a taxing authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.
- 27.4 Taxes and Fees Imposed on Providing Party But Passed On To Purchasing Party.
- 27.4.1 Taxes and fees imposed on the providing Party, which are permitted or required to be passed on by the providing Party to its customer, shall be borne by the purchasing Party.
- 27.4.2 To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.
- 27.4.3 If the purchasing Party disagrees with the providing Party's determination as to the application or basis for any such tax or fee, the Parties shall consult with respect to the imposition and billing of such tax or fee. Notwithstanding the foregoing, the providing Party shall retain ultimate responsibility for determining whether and to what extent any such taxes or fees are applicable, and the purchasing Party shall abide by such determination and pay such taxes or fees to the providing Party. The providing Party shall further retain ultimate responsibility for determining whether and how to contest the imposition of such taxes and fees; provided, however, that any such contest undertaken at the request of the purchasing Party shall be at the purchasing Party's expense.
- 27.4.4 In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.
- 27.4.5 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.
- 27.4.6 Notwithstanding any provision to the contrary, the purchasing Party shall protect indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other reasonable charges or payable expenses (including reasonable attorney fees) with respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.

- 27.4.7 Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a taxing authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.
- 27.5 <u>Mutual Cooperation</u>. In any contest of a tax or fee by one Party, the other Party shall cooperate fully by providing records, testimony and such additional information or assistance as may reasonably be necessary to pursue the contest. Further, the other Party shall be reimbursed for any reasonable and necessary out-of-pocket copying and travel expenses incurred in assisting in such contest.

28. NETWORK MAINTENANCE AND MANAGEMENT

- 28.1 The Parties shall work cooperatively to implement this Agreement. The Parties shall exchange appropriate information (e.g., maintenance contact numbers, network information, information required to comply with law enforcement and other security agencies of the Government, etc.) as reasonably required to implement and perform this Agreement.
- 28.2 Each Party hereto shall design, maintain and operate their respective networks as necessary to ensure that the other Party hereto receives service quality which is consistent with generally accepted industry standards at least at parity with the network service quality given to itself, its Affiliates, its End Users or any other Telecommunications Carrier.
- 28.3 Neither Party shall use any service or facility provided under this Agreement in a manner that impairs the quality of service to other Telecommunications Carriers' or to either Party's End Users. Each Party will provide the other Party notice of any such impairment at the earliest practicable time.
- 28.4 BellSouth agrees to provide e.spire prior notice consistent with applicable FCC rules and the Act of changes in the information necessary for the transmission and routing of services using BellSouth's facilities or networks, as well as other changes that affect the interoperability of those respective facilities and networks. This Agreement is not intended to limit BellSouth's ability to upgrade its network through the incorporation of new equipment, new software or otherwise so long as such upgrades are not inconsistent with BellSouth's obligations to e.spire under the terms of this Agreement.

29. CHANGES IN SUBSCRIBER CARRIER SELECTION

29.1 Both Parties hereto shall apply all of the principles set forth in 47 C.F.R. § 64.1100 to the process for End User selection of a primary Local Exchange Carrier. BellSouth shall not require a disconnect order from an e.spire Customer or another LEC in order to process an e.spire order for Resale Service for an e.spire End User. Until the FCC or the Commission adopts final rules and

procedures regarding a Customer's selection of a primary Local Exchange Carrier, unless already done so, e.spire shall deliver to BellSouth a Blanket Representation of Authorization that applies to all orders submitted by e.spire under this Agreement that require a primary Local Exchange Carrier change. Both Parties hereto shall retain on file all applicable documentation of authorization, including letters of authorization, relating to their End User's selection as its primary Local Exchange Carrier, which documentation shall be available for inspection by the other Party hereto upon reasonable request during normal business hours.

29.2 If an End User denies authorizing a change in his or her primary Local Exchange Carrier selection to a different local exchange carrier ("Unauthorized Switching"), the Party receiving the End User complaint shall switch or caused to be switched that End User back to his preferred carrier in accordance with Applicable Law.

30. FORCE MAJEURE

In the event performance of this Agreement, or any obligation hereunder, is either directly or indirectly prevented, restricted, or interfered with by reason of fire, flood, earthquake or like acts of God, wars, revolution, civil commotion, explosion, acts of public enemy, embargo, acts of the government in its sovereign capacity, labor difficulties, including without limitation, strikes, slowdowns, picketing, or boycotts, unavailability of equipment from vendor, changes requested by Customer, or any other circumstances beyond the reasonable control and without the fault or negligence of the Party affected, the Party affected, upon giving prompt notice to the other Party, shall be excused from such performance on a day-to-day basis to the extent of such prevention, restriction, or interference (and the other Party shall likewise be excused from performance of its obligations on a day-to-day basis until the delay, restriction or interference has ceased); provided however, that the Party so affected shall use diligent efforts to avoid or remove such causes of non-performance and both Parties shall proceed whenever such causes are removed or cease.

31. YEAR 2000 COMPLIANCE

Each Party warrants that it has implemented a program the goal of which is to ensure that all software, hardware and related materials (collectively called "Systems") delivered, connected with the other Party or supplied in the furtherance of the terms and conditions specified in this Agreement: (i) will record, store, process and display calendar dates falling on or after January 1, 2000, in the same manner, and with the same functionality as such software records, stores, processes and calendar dates falling on or before December 31, 1999; and (ii) shall include without limitation date data century recognition, calculations that accommodate same century and multicentury formulas and date values, and date data interface values that reflect the century.

32. BINDING EFFECT

This Agreement shall be binding on and inure to the benefit of the respective successors and permitted assigns of the Parties.

33. CONSENT

Where consent, approval or mutual agreement is required of a Party, it shall not be unreasonably withheld or delayed, unless otherwise expressly provided herein.

34. MODIFICATION OF AGREEMENT

- 34.1 BellSouth shall make available, pursuant to 47 USC § 252(i) and the FCC rules and regulations regarding such availability, to e.spire, at the same rates, and the same terms and conditions, any interconnection, service, or network element provided under any other agreement filed and approved pursuant to 47 USC § 252. The adopted interconnection, service, or network element and agreement shall apply to the same states as such other agreement and for the identical term of such other agreement.
- 34.2 If e.spire changes its name or makes changes to its company structure or identity due to a merger, acquisition, transfer or any other reason, it is the responsibility of e.spire to notify BellSouth of said change and request that an amendment to this Agreement, if necessary, be executed to reflect said change.
- 34.3 Execution of this Agreement by either Party does not confirm or infer that the executing Party agrees with any decision(s) issued pursuant to the Telecommunications Act of 1996 and the consequences of those decisions on specific language in this Agreement. Neither Party waives its rights to appeal or otherwise challenge any such decision(s) and each Party reserves all of its rights to pursue any and all legal and/or equitable remedies, including appeals of any such decision(s).
- In the event that any effective legislative, regulatory, judicial or other legal action materially affects any material terms of this Agreement, or the ability of e.spire or BellSouth to perform any material terms of this Agreement, e.spire or BellSouth may, on thirty (30) days' written notice require that such terms be renegotiated, and the Parties shall renegotiate in good faith such mutually acceptable new terms as may be required. In the event that such new terms are not renegotiated within ninety (90) days after such notice, the Dispute shall be referred to the Dispute Resolution procedure set forth in Section 26.

35. WAIVERS

Except as otherwise provided in this Agreement, no amendment or waiver of any provision of this Agreement, and no consent to any default under this Agreement, will be effective unless the same is in writing and signed by an authorized representative of the Party against whom such amendment, waiver or consent is claimed. A failure or delay of either Party to enforce any of the provisions hereof, to exercise any option which is herein provided, or to require performance of any of the provisions hereof shall in no way be construed to be a waiver of such provisions or options, and each Party, notwithstanding such failure, shall have the right thereafter to insist upon the specific performance of any and all of the provisions of this Agreement.

36. EXPENSES

Except as specifically set out in this Agreement, or as otherwise required by a regulatory agency with jurisdiction, each Party shall be solely responsible for its own expenses involved in all activities related to the subject of this Agreement.

37. RELATIONSHIP OF PARTIES

This Agreement shall not establish, be interpreted as establishing, or be used by either Party to establish, or to represent their relationship as any form of agency, partnership or joint venture. Neither Party shall have any authority to bind the other or to act as an agent for the other unless written authority, separate form this Agreement, is provided. Nothing in this Agreement shall be construed as providing for the sharing of profits or losses arising out of the efforts of either or both of the Parties. Nothing herein shall be construed as making either Party responsible or liable for the obligations and undertakings of the other Party.

38. THIRD PARTY BENEFICIARIES

This Agreement does not provide, and shall not be construed to provide, third parties with any benefit, remedy, claim, liability, reimbursement, cause of action, or other privilege.

39. COOPERATION ON PREVENTING END USER FRAUD

The Parties agree to cooperate fully with one another to investigate, minimize, prevent, and take corrective action in cases of fraud.

40. GOOD FAITH PERFORMANCE

In the performance of their obligations under this Agreement the Parties will act in good faith and consistently with the intent of the Act. Where notice, approval or similar action by a Party is permitted or required by any provision of this Agreement (including without limitation, the obligation of the Parties to further negotiate the resolution of new or open issues under this Agreement), such action will not be unreasonably delayed, withheld or conditioned.

41. INDEPENDENT CONTRACTORS

Each Party is an independent contractor, and has and hereby retains the right to exercise full control of and supervision over its own performance of its obligations under this Agreement, and retains full control over the employment, direction, compensation and discharge of its employees assisting in the performance of such obligations. Each Party shall be solely responsible for all matters relating to payment of such employees, including compliance with social security taxes, withholding taxes and all other regulations governing such matters. Subject to the limitations on liability and except as otherwise provided in this Agreement, each Party shall be responsible for (i) its own acts and performance of all obligations imposed by Applicable Law in connection with its activities, legal status and property, real or personal and, (ii) the acts of its own Affiliates,

employees, agents and contractors during the performance of the Party's obligations hereunder.

42. SUBCONTRACTING

If any obligation is performed through a subcontractor, each Party shall remain fully responsible for the performance of this Agreement in accordance with its terms, including any obligations either Party performs through subcontractors, and each Party shall be solely responsible for payments due the Party's subcontractors. No contract, subcontract or other Agreement entered into by either Party with any third party in connection with the provision of any facilities or services provided herein, shall provide for any indemnity, guarantee or assumption of liability by, or other obligation of, the other Party to this Agreement with respect to such arrangement, except as consented to in writing by the other Party. No subcontractor shall be deemed a third party beneficiary for any purposes under this Agreement. Any subcontractor who gains access to CPNI or Confidential Information covered by this Agreement shall be required by the subcontracting Party to protect such CPNI or Confidential Information to the same extent that the subcontracting Party is required to protect the same under the terms of this Agreement.

43. SEVERABILITY

If any term, condition or provision of this Agreement is held to be invalid or unenforceable for any reason, such invalidity or unenforceability shall not invalidate the entire Agreement, unless such construction would be unreasonable. The Agreement shall be construed as if it did not contain the invalid or unenforceable provision or provisions, and the rights and obligations of each Party shall be construed and enforced accordingly. Provided, however, that in the event such invalid or unenforceable provision or provisions are essential elements of this Agreement and substantially impair the rights or obligations of either Party, the Parties shall promptly negotiate a replacement provision or provisions. If impasse is reached, the Parties will resolve said impasse under the dispute resolution procedures set forth in Section 26.

44. SURVIVAL OF OBLIGATIONS

Any liabilities or obligations of a Party for acts or omissions prior to the cancellation or termination of this Agreement, and any obligation of a Party under the provisions regarding indemnification, Confidential Information, limitations on liability, and any other provisions of this Agreement which, by their terms are contemplated to survive (or to be performed after) termination of this Agreement, shall survive cancellation or termination thereof.

45. CUSTOMER INQUIRIES

Each Party shall refer all questions regarding the other Party's services or products directly to the other Party at a telephone number specified by that Party.

45.2 Each Party shall ensure that each of their representatives who receive inquiries regarding the other Party's services: (i) provide the numbers described in Section 45.1 to callers who inquire about the other Party's services or products, and (ii) do not in any way disparage or discriminate against the other Party or its products or services.

46. COMPLIANCE WITH APPLICABLE LAW

- 46.1 Each Party shall comply at its own expense with all applicable federal, state, and local statutes, laws, rules, regulations, codes, final and nonappealable orders, decisions, injunctions, judgments, awards and decrees that relate to its obligations under this Agreement. Nothing in this Agreement shall be construed as requiring or permitting either Party to contravene any mandatory requirement of Applicable Law, and nothing herein shall be deemed to prevent either Party from recovering its cost or otherwise billing the other Party for compliance with the Order to the extent required or permitted by the term of such Order.
- 46.2 Each Party shall be responsible for obtaining and keeping in effect all approvals from, and rights granted by, governmental authorities, building and property owners, other carriers, and any other persons that may be required in connection with the performance of its obligations under this Agreement. Each Party shall reasonably cooperate with the other Party in obtaining and maintaining any required approvals and rights for which such Party is responsible.

47. LABOR RELATIONS

Each Party shall be responsible for labor relations with its own employees. Each Party agrees to notify the other Party as soon as practicable whenever such Party has knowledge that a labor dispute concerning its employees is delaying or threatens to delay such Party's timely performance of its obligations under this Agreement and shall endeavor to minimize impairment of service to the other Party (by using its management personnel to perform work or by other means) in the event of a labor dispute to the extent permitted by Applicable Law.

48. COMPLIANCE WITH THE COMMUNICATIONS LAW ENFORCEMENT ACT OF 1994 ("CALEA")

Each Party represents and warrants that any equipment, facilities or services provided to the other Party under this Agreement comply with CALEA. Each Party shall indemnify and hold the other Party harmless from any and all penalties imposed upon the other Party for such other Party's noncompliance, and shall at the non-compliant Party's sole cost and expense, modify or replace any equipment, facilities or services provided to the other Party under this Agreement to ensure that such equipment, facilities and services fully comply with CALEA.

49. ADDITIONAL FAIR COMPETITION REQUIREMENTS

- 49.1 In the event that either Party transfers facilities or other assets to an Affiliate which are necessary to comply with its obligations under this Agreement, the obligations hereunder shall survive and transfer to such Affiliate.
- 49.2 BellSouth shall allow local exchange customers of e.spire to select BellSouth for the provision of intraLATA toll services on a nondiscriminatory basis; provided, however, that prior to establishment of BellSouth as the intraLATA toll carrier for e.spire local exchange customers, the Parties shall negotiate a billing and collections agreement on commercially reasonable terms whereby e.spire shall bill the customer on BellSouth's behalf and shall collect from the customer and remit to BellSouth intraLATA toll revenues. e.spire agrees to bill its customers on BellSouth's behalf for both presubscribed and "dial around" intraLATA toll traffic. The Parties shall exchange customer record data on a timely basis as necessary to bill such customers for intraLATA toll usage.
- 49.3 BellSouth shall not use information derived from providing services or facilities to e.spire to create a lead or other information base for a "winback" sales program.

50. GOVERNING LAW

This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Georgia, without regard to its conflict of laws principles.

51. ARM'S LENGTH NEGOTIATIONS

This Agreement was executed after arm's length negotiations between the undersigned Parties and reflects the conclusion of the undersigned that this Agreement is in the best interests of all Parties.

52. NONEXCLUSIVE DEALINGS

This Agreement does not prevent either Party from providing or purchasing services or facilities to or from any other Person, nor does it obligate either Party to provide or purchase any services or facilities not specifically provided herein.

53. NOTICES

53.1 Every notice, consent, approval, or other communications required or contemplated by this Agreement shall be in writing and shall be delivered in person or given by postage prepaid mail, address to:

BellSouth Telecommunications, Inc.

CLEC Account Team 9th Floor 600 North 19th Street Birmingham, Alabama 35203

and

General Attorney - COU Suite 4300 675 W. Peachtree St. Atlanta, GA 30375

e.spire Communications, Inc.

James C. Falvey Senior Vice President - Regulatory Affairs Suite 100 131 National Business Parkway Annapolis Junction, Maryland 20701

or at such other address as the intended recipient previously shall have designated by written notice to the other Party.

- Where specifically required, notices shall be by certified or registered mail.

 Unless otherwise provided in this Agreement, notice by mail shall be effective on the date it is officially recorded as delivered by return receipt or equivalent, and in the absence of such record of delivery, it shall be presumed to have been delivered the fifth day, or next business day after the fifth day, after it was deposited in the mails.
- 53.3 BellSouth shall provide e.spire notice via Internet posting of price changes and of changes to the terms and conditions of services available for resale.

54. RULE OF CONSTRUCTION

No rule of construction requiring interpretation against the drafting Party hereof shall apply in the interpretation of this Agreement.

55. HEADINGS OF NO FORCE OR EFFECT

The headings of Articles and Sections of this Agreement are for convenience of reference only, and shall in no way define, modify or restrict the meaning or interpretation of the terms or provisions of this Agreement.

56. MULTIPLE COUNTERPARTS

This Agreement may be executed multiple counterparts, each of which shall be deemed an original, but all of which shall together constitute but one and the same document.

57. FILING OF AGREEMENT

Upon execution of this Agreement BellSouth shall file with the appropriate state Commission pursuant to the requirements of Section 252 of the Act. Each Party shall cooperate with the other and with any applicable regulatory agency to obtain regulatory approval of this Agreement. If the regulatory agency imposes any filing or public interest notice fees regarding the filing or approval of the Agreement, e.spire shall be responsible for the costs of publishing the required notice and the filing fee shall be borne by both Parties equally.

58. ENTIRE AGREEMENT

This Agreement and its Attachments, incorporated herein by this reference, sets forth the entire understanding and supersedes prior Agreements between the Parties relating to the subject matter contained herein and merges all prior discussions between them, and neither Party shall be bound by any definition, condition, provision, representation, warranty, covenant or promise other than as expressly stated in this Agreement or as is contemporaneously or subsequently set forth in writing and executed by a duly authorized officer or representative of the Party to be bound thereby.

The following services are included as options for purchase by e.spire. e.spire shall elect said services by written request to its Account Manager if applicable.

Optional Daily Usage File (ODUF) Enhanced Optional Daily Usage File (EODUF) Access Daily Usage File (ADUF) Line Information Database (LIDB) Storage Centralized Message Distribution Service (CMDS) Calling Name (CNAM)

59. SIGNATURE

Signatures transmitted by the Parties by facsimile shall have the same effect as original signatures as of the date transmitted by the executing Party.

IN WITNESS WHEREOF, the Parties have executed this Agreement the day and year above first written.

BellSouth Telecommunications, Inc.	e.spire Communications, Inc.
Signature on File	Signature on File
Signature	Signature
Jerry D. Hendrix	JAMES C. FALVEY
Name	Name
Sr. Director	SENIOR VICE PREIDENT,
	REGULATORY AFFAIRS
Title	Title
7/25/00	JULY 25, 2000
Date	Date

Definitions

- 1 "Act" means the Communications Act of 1934 (47 U.S.C. 151 et seq.), as amended by the Telecommunications Act of 1996, and as from time to time interpreted and implemented in the duly authorized rules and regulations of the FCC or a State Commission within its state of jurisdiction.
- 2 "ADSL" or "Asymmetrical Digital Subscriber Line" means a transmission technology which transmits an asymmetrical digital signal of up to 6 Mbps to the End User and up to 640 Kbps from the End User.
- 3 "Affiliate" is defined as a person that (directly or indirectly) owns or controls, is owned or controlled by, or is under common ownership or control with, another person. For purposes of this paragraph, the term "own" means to own an equity interest (or equivalent thereof) of more than 10 percent.
- 4 "Agreement" refers to this Interconnection Agreement between e.spire and BellSouth and all Attachments, Appendices, Exhibits, Schedules and Addenda or Amendments hereto.
- 5 "AMA" means the Automated Message Accounting structure inherent in switch technology that initially records telecommunication message information. AMA format is contained in the Automated Message Accounting document, published by Telcordia as GR-1100-CORE which defines the industry standard for message recording.
- 6 "Applicable Law" means all laws, regulations, and orders applicable to each Party's performance of its obligations hereunder.
- 7 "As Defined in the Act" means as specifically defined by the Act and as from time to time interpreted in the duly authorized rules and regulations of the FCC or a State Commission within its state of jurisdiction.
- 8 "As Described in the Act" means as described in or required by the Act and as from time to time interpreted in the duly authorized rules and regulations of the FCC or the Commission.
- 9 "Automatic Location Identification" or "ALI" means a feature by which the service address associated with the calling party's listed telephone number identified by ANI as defined herein, is forwarded to the PSAP for display. Additional telephones with the same number as the calling party's, including secondary locations and off-premise extensions will be identified with the service address of the calling party's listed number.
- 10 "Automatic Number Identification" or "ANI" means a signaling parameter which refers to the number transmitted through a network identifying the calling party.
 - 11 "Bellcore" now known as "Telcordia"
 - 12 "Bill Date" means the date that a bill is issued by a party.
 - 13 "Bona Fide Request" as defined in Attachment 12.

- 14 "Business Day" means a day on which banking institutions are required to be open for business in New York.
- 15 "Calling Party Number" or "CPN" is a Common Channel Signaling ("CCS") parameter which refers to the number transmitted through a network identifying the calling party.
- 16 "Carrier Identification Code" or "CIC" is a three-digit or four digit or five digit number that identifies a specific Interexchange Carrier.
- 17 "Central Office Switch" means a switch used to provide Telecommunications Services, including, but not limited to:
- 17.1 "End Office Switches" which are used to terminate Customer station Loops for the purpose of interconnection to each other and to trunks; and
- 17.2 "Tandem Office Switches" or "Tandems" which are used to connect and switch trunk circuits between and among other Central Office Switches.
- 18 "Centralized Message Distribution System" or "CMDS" is the Telcordia (formerly BellCore) administered national system, based in Kansas City, Missouri, used to exchange Exchange Message Interface (EMI) formatted data among host companies.
- 19 "CLASS Features" means certain CCIS-based features available to Customers including but not limited to: Automatic Call Back; Call Trace; Caller Identification and related blocking features; Distinctive Ringing/Call Waiting; Selective Call Forward; and Selective Call Rejection.
- 20 "Collocation" is As Described in the Act and FCC Rules and Orders, and as further defined in Attachment 4 hereto.
 - 21 "Commercial Mobile Radio Service" or "CMRS" is As Defined in the Act.
- 22 "Commission" is defined as the appropriate regulatory agency in each of BellSouth's nine state region, Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee.
- 23 "Common Channel Signaling" or "CCS" means the signaling system, developed for use between switching systems with stored-program control, in which all of the signaling information for one or more groups of trunks is transmitted over a dedicated high-speed data link rather than on a per-trunk basis. The current industry standard for common carrier network signaling is SS7.1.33 "Competitive Local Exchange Carrier" or "CLEC" means any Local Exchange Carrier other than BellSouth, operating as such in BellSouth's certificated territory.
 - 24 "Confidential Information" is As Defined in Part A hereof.
- 25 "Cross Connection" means a jumper cable or similar connection provided pursuant to Collocation at the digital signal cross connect, Main Distribution Frame or other suitable frame or panel between (i) the Collocating Party's equipment and (ii) the equipment or facilities of the housing party.

- 26 "Customer" or "End User" means a third-party residence or business that subscribes to Telecommunications Services provided by either of the Parties.
- 27 "Customer of Record" means the entity responsible for placing applications for service; requesting additions, rearrangements, maintenance or discontinuance of service; and payment in full of charges incurred such as non-recurring, monthly recurring, toll, directory assistance, etc.
 - 28 "Customer Proprietary Network Information" or "CPNI" is As Defined in the Act.
- 29 "Customer Specific Arrangement" or "CSA" means a service arrangement negotiated with an individual customer that includes rates, terms or conditions that differ from those included in BellSouth's intrastate retail services tariff.
- 30 "Daily Usage File" or "DUF" is the compilation of messages or copies of messages in standard Exchange Message Interface (EMI) format exchanged from BellSouth to an CLEC.
- 31 "Dark Fiber" is as defined in Attachment 2 of this Agreement. 1.45 "Data Management System" or "DMS" means a system of manual procedures and computer processes used to create, store and update the data required to provide the Selecting Routing ("SR") and ALI features.
- 32 "Demarcation Point" means a point on a property or premises where the Customer's service is located as determined by the applicable LEC. This point is where network access recurring charges and the LEC's responsibility stop and beyond which Customer responsibility begins.
- 33 "Deposit" means assurance provided by a customer in the form of cash, surety bond or bank letter of credit.
 - 34 "Dialing Parity" is As Defined in the Act.
- 35 "Digital Signal Level" means one of several transmission rates in the time-division multiplex hierarchy.
- 36 "Digital Signal Level 0" or "DS0" means the 64 Kbps zero-level signal in the timedivision multiplex hierarchy.
- 37 "Digital Signal Level 1" or "DS1" means the 1.544 Mbps first-level signal in the time-division multiplex hierarchy. In the time-division multiplexing hierarchy of the telephone network, DS1 is the initial level of multiplexing.
- 38 "Digital Signal Level 3" or "DS3" means the 44.736 Mbps third-level in the time-division multiplex hierarchy. In the time-division multiplexing hierarchy of the telephone network, DS3 is defined as the third level of multiplexing.
 - 39 "Dispute" is As Defined in Part A hereof.

- 40 "End User Customer Location" means the physical location of the premises where an End User makes use of the Telecommunications Services.
 - 41 "Exchange Access" is As Defined in the Act.
- 42 "Exchange Area" means an area, defined by the Commission, for which a distinct local rate schedule is in effect.
- 43 "Exchange Message Interface" or "EMI" is the nationally administered standard format for the exchange of data among the Exchange Carriers within the telecommunications industry.
 - 44 "FCC" means the Federal Communications Commission.
- 45 "FCC Regulations" means the effective rules, regulations, requirements, orders and policies adopted or issued by the FCC, as each may be revised from time to time.
 - 46 "Feeder" is As Defined in Attachment 2.
 - 47 "Fiber-Meet" or "Mid-Span Meet" as defined in Attachment 3 of this Agreement.
 - 48 "Grandfathered Services" is As Defined in Attachment 1 of this Agreement
 - 49 "Hazardous Substances" is As Defined in Attachment 4 of this Agreement.
- 50 "HDSL" or "High-Bit Rate Digital Subscriber Line" means a transmission technology which transmits up to a DS1-level signal, using any one of the following line codes: 2 Binary / 1 Quartenary ("2B1Q"), Carrierless AM/PM, Discrete Multitone ("DMT"), or 3 Binary / 1 Octel ("3B1O").
- 51 "Incumbent Local Exchange Carrier" or "ILEC" is As Defined in the Act. For purposes of this Agreement, BellSouth is an Incumbent Local Exchange Carrier.
- 52 "Independent Telephone Company" or "ITC" means any entity other than BellSouth which, with respect to its operations within the states covered by this Agreement, is an Incumbent Local Exchange Carrier.
- 53 "Information Service" means the offering of a capability for generating, acquiring, storing, transforming, processing, retrieving, utilizing, or making available information via telecommunications, and includes electronic publishing, but does not include any use of any such capability for the management, control, or operation of a telecommunications system or the management of a telecommunications service.
- 54 "Inside Wire" or "Inside Wiring" means all wire, cable, terminals, and associated equipment or materials on the Customer's side of the Rate Demarcation Point.
 - 55 "Integrated Digital Loop Carrier" is as described in Attachment 2 of this Agreement.
- 56 Intercompany Settlements (ICS) is the revenue associated with charges billed by a company other than the company in whose service area such charges were incurred. ICS on a

national level includes third number and credit card calls and is administered by Telcordia (formerly BellCore)'s Calling Card and Third Number Settlement System (CATS). Included is traffic that originates in one Regional Bell Operating Company's (RBOC) territory and bills in another RBOC's territory.

- 57 "Interconnection" is As Described in the Act.
- 58 "Interexchange Carrier" or "IXC" means a carrier that provides, directly or indirectly, interLATA or intraLATA Telephone Toll Services.
- 59 "Interim Number Portability" or "INP" is As Described in the Attachment 5 of this Agreement.
 - 60 "InterLATA" is As Defined in the Act.
- 61 "IntraLATA Toll Traffic" means all basic intraLATA message services calls other than Local Traffic.
- 62 "Integrated Services Digital Network" (ISDN) means a switched network service that provides end-to-end digital connectivity for the simultaneous transmission of voice and data, as set forth in Attachment 2.
- 63 "Line Information Data Base(s)" or "LIDB" is as described in Attachment 2 of this Agreement.
 - 64 "Local Access and Transport Area" or "LATA" is As Defined in the Act.
 - 65 "Local Exchange Carrier" or "LEC" is As Defined in the Act.
- 66 "Local Interconnection" is defined as 1) the delivery of local traffic to be terminated on each Party's local network so that end users of either Party have the ability to reach end users of the other Party without the use of any access code or substantial delay in the processing of the call; 2) the LEC network features, functions, and capabilities set forth in this Agreement; and 3) Service Provider Number Portability sometimes referred to as temporary telephone number portability to be implemented pursuant to the terms of this Agreement.
 - 67 "Local Loop" element is As Defined in Attachment 2 of this Agreement.
- 68 "Local Number Portability" or "LNP" means the ability of users of Telecommunications Services to retain, at the same location, existing telephone numbers without impairment of quality, reliability, or convenience when switching from one Telecommunications Carrier to another.
 - 69 "Local Traffic" is as defined in Attachment 3 of this Agreement.
- 70 "Main Distribution Frame" means the distribution frame of the Housing Party used to interconnect cable pairs and line and trunk equipment terminals on a switching system.

- 71 "MECAB" means the Multiple Exchange Carrier Access Billing (MECAB) document prepared by the Billing Committee of the Ordering and Billing Forum ("OBF"), which functions under the auspices of the Carrier Liaison Committee ("CLC") of the Alliance for Telecommunications Industry Solutions ("ATIS"). The MECAB document, published by Telcordia as Special Report SR-BDS-000983, contains the recommended guidelines for the billing of an Exchange Access service provided by two or more LECs, or by one LEC in two or more states, within a single LATA.
- 72 "MECOD" means the Multiple Exchange Carriers Ordering and Design (MECOD) Guidelines for Access Services Industry Support Interface, a document developed by the Ordering/Provisioning Committee under the auspices of OBF. The MECOD document, published by Telcordia as Special Report SR-STS-002643, establishes methods for processing orders for Exchange Access service which is to be provided by two or more LECs.
- 73 "Meet-Point Billing" means the process whereby each Party bills the appropriate tariffed rate for its portion of a jointly provided Switched Exchange Access Service as agreed to in the Agreement for Switched Access Meet Point Billing.
 - 74 "NECA" refers to the National Exchange Carriers Association.
- 75 "Network Element" is defined to mean a facility or equipment used in the provision of a telecommunications service. Such term may include, but is not limited to, features, functions, and capabilities that are provided by means of such facility or equipment, including but not limited to, subscriber numbers, databases, signaling systems, and information sufficient for billing and collection or used in the transmission, routing, or other provision of a telecommunications service. BellSouth offers access to the Network Elements, unbundled loops; network interface device; sub-loop elements; local switching; transport; tandem switching; operator systems; signaling; access to call-related databases; dark fiber as set forth in Attachment 2 of this Agreement. 1.110 "Network Interface Device" or "NID" is as described in Attachment 2 of this Agreement.
- 76 "Non-Intercompany Settlement System" or "NICS" is the Telcordia (formerly BellCore) system that calculates non-intercompany settlements amounts due from one company to another within the same RBOC region. It includes credit card, third number and collect messages.
- 77 "North American Numbering Plan" or "NANP" means the numbering plan used in the United States that also serves Canada, Bermuda, Puerto Rico and certain Caribbean Islands. The NANP format is a 10-digit number that consists of a 3-digit NPA code (commonly referred to as the area code), followed by a 3-digit NXX code and 4-digit line number.
- 78 "Numbering Plan Area" or "NPA" also is sometimes referred to as an area code. There are two general categories of NPAs, "Geographic NPAs" and "Non-Geographic NPAs." A Geographic NPA is associated with a defined geographic area, and all telephone numbers bearing such NPA are associated with services provided within that geographic area. A Non-Geographic NPA, also known as a "Service Access Code" or "SAC Code," is typically associated with a specialized telecommunications service which may be provided across multiple

geographic NPA areas (e.g., 800, 900, 700, 500 and 888 are examples of Non-Geographic NPAs).

- 79 "NXX Code" or "End Office Code" means the three digit switch entity indicator (i.e., the first three digits of a seven digit telephone number).
- 80 "OBF" means the "Ordering and Billing Forum", which functions under the auspices of the Carrier Liaison Committee (CLC) of the Alliance for Telecommunications Industry Solutions (ATIS).
 - 81 "OCN" refers to an Operating Company Number.
 - 82 "Optical Carrier Level 3" or "OC3" is As Defined in Attachment 2.
 - 83 "Optical Carrier Level 12" or "OC12" is As Defined in Attachment 2.
 - 84 "Optical Line Terminating Multiplexor" or "OLTM" is As Defined in Attachment 2.
 - 85 "Party" means either BellSouth or e.spire, and "Parties" means BellSouth and e.spire.
- 86 "Percent of Interstate Usage" or "PIU" is defined as a factor to be applied to terminating access services minutes of use to obtain those minutes that should be rated as interstate access services minutes of use. The numerator includes all interstate "non-intermediary" minutes of use, including interstate minutes of use that are forwarded due to service provider number portability less any interstate minutes of use for Terminating Party Pays services, such as 800 Services. The denominator includes all "non-intermediary", local, interstate, intrastate, toll and access minutes of use adjusted for service provider number portability less all minutes attributable to terminating Party pays services.
- 87 "Percent Local Usage" or "PLU" is defined as a factor to be applied to intrastate terminating minutes of use. The numerator shall include all "non-intermediary" local minutes of use adjusted for those minutes of use that only apply local due to Service Provider Number Portability. The denominator is the total intrastate minutes of use including local, intrastate toll, and access, adjusted for Service Provider Number Portability less intrastate terminating Party pays minutes of use.
 - 88 "Performance Measurements" is as described in Attachment 9 hereto.
 - 89 "Person" is As Defined in the Act.
 - 90 "Physical Collocation" is As Defined in the Act.
 - 91 "PIC" means Primary or Presubscribed Interexchange Carrier.
- 92 "Public Safety Answering Point" or "PSAP" means an answering location for 9-1-1 calls originating in a given area. A PSAP may be designated as Primary or Secondary, which refers to the order in which calls are directed for answering. Primary PSAPs respond first. Secondary PSAPs receive calls on a transfer basis only, and generally serve as a centralized answering location for a particular type of emergency call. PSAPs are staffed by employees of

Service Agencies such as police, fire or emergency medical agencies or by employees of a common bureau serving a group of such entities.

- 93 "Publisher" means BellSouth's White Pages and Yellow Pages Directories publisher(s), i.e., currently "BAPCO."
- 94 "Rate Center" means the specific geographic point which has been designated by a given LEC as being associated with a particular NPA-NXX code which has been assigned to the LEC for its provision of Telephone Exchange Service. The Rate Center is the finite geographic point identified by a specific V&H coordinate, which is used by that LEC to measure, for billing purposes, distance sensitive transmission services associated with the specific Rate Center. Rate Centers will be identical for each Party until such time as e.spire is permitted by an appropriate regulatory body to create its own Rate Centers within an area.
- 95 "Reciprocal Compensation" is As Described in the Act, and refers to the payment arrangement for transport and termination of Local Traffic specified in Attachment 3.
- 96 "Resale" means an activity wherein a certificated CLEC subscribes to the Telecommunications Services of BellSouth and then reoffers those telecommunications services to the public (with or without "adding value").
- 97 "Resale Service Area" means the area, as defined in a state Commission approved certificate of operation, within which a CLEC may offer resold local exchange telecommunications service.
- 98 "Resale Services" means the BellSouth local services provided to e.spire for Resale pursuant to the terms of Attachment 1 hereto.
- 99 "Revenue Accounting Office" (RAO) Status Company is a local exchange company/alternate local exchange company that has been assigned a unique RAO code. Message data exchanged among RAO status companies is grouped (i.e. packed) according to From/To/Bill RAO combinations.
- 100 "Serving Wire Center" or "SWC" is as described in Attachment 3 of this Agreement.
- 101 "Shared Tenant Service" or "STS" is as defined in BellSouth's appropriate General Subscriber Service Tariff.
- 102 "Shared Transport" or "Common Transport" is as defined in Attachment 2 of this Agreement.
- 103 "Signal Transfer Points" ("STPs") are signaling message switches that interconnect Signaling Links to route signaling messages between switches and databases. STPs enable the exchange of Signaling System 7 ("SS7") messages between switching elements, database elements and STPs. STPs provide access to various BellSouth and third party network elements such as local switching and databases.

- 104 "Signaling Links" are dedicated transmission paths carrying signaling messages between carrier switches and signaling networks. Signal Link Transport is a set of two or four dedicated 56 kbps transmission paths between e.spire designated Signaling Points of Interconnection that provide a diverse transmission path and cross connect to a BellSouth Signal Transfer Point.
- 105 "Subsidiary" means a corporation or other legal entity owned or controlled by a Party.
 - "Switched Access Traffic" is as described in Attachment 3 of this Agreement.
- 107 "Synchronous Optical Network" or "SONET" means an optical interface standard that allows inter-networking of transmission products from multiple vendors. The base rate is 51.84 Mbps (OC-I/STS-1) and higher rates are direct multiples of the base rate, up to 13.22 Gpbs.
- 108 "Tariff" means any applicable federal or state tariff of a Party, that is filed and effective with the FCC or Commission, each as may be amended by the Party from time to time, under which a Party offers a particular service, facility, or arrangement. A Tariff shall not include BellSouth's "Statement of Generally Available Terms and Conditions for Interconnection, Unbundled Network Elements, Ancillary Services and Resale of Telecommunications Services" which has been approved or is pending approval by the Commission pursuant to Section 252(f) of the Communications Act of 1934, 47 U.S.C. § 252(f).
 - "Tax" is As Defined in Part A hereof.
 - "Technically Feasible Point" is As Described in the Act.
 - "Telecommunications" is As Defined in the Act.
- 112 "Telecommunications Act of 1996" means Public Law 104-104 of the United States Congress effective February 8, 1996 as amended, and any rules, and regulations promulgated thereunder.
 - "Telecommunications Carrier" is As Defined in the Act.
 - "TRS" refers to Telecommunications Relay Service.
 - "Telecommunications Service" is As Defined in the Act.
 - "Telcordia" is the former Bell Communications Research, Inc.
 - "Telephone Exchange Service" is As Defined in the Act.
 - "Telephone Toll Service" is As Defined in the Act.
- 119 "Toll Traffic" means traffic that is originated by a Customer of one Party on that Party's network and terminates to a Customer of the other Party on that Party's network and is not Local Traffic or ancillary traffic. Toll Traffic may be either "IntraLATA Toll Traffic" or

"InterLATA Toll Traffic," depending on whether the originating and terminating points are within the same LATA.

- 120 "Transit Traffic" means any traffic that originates from or terminates at e.spire's network, "transits" BellSouth's network substantially unchanged, and terminates to or originates from a third carrier's network, as the case may be. "Transit Traffic Service" provides e.spire with the ability to use its connection to a BellSouth Tandem for the delivery of calls which originate or terminate with e.spire and terminate to or originate from a carrier other than BellSouth, such as another CLEC, an ILEC other than BellSouth. In these cases, neither the originating nor terminating End User is an End User of BellSouth. This service is provided through BellSouth's Tandems.
 - "Transport" network element is as defined in Attachment 2 of this Agreement.
 - "Transport and Termination" is as described in the Act.
 - 123 "V&H Coordinates" means vertical and horizontal coordinates.
 - "Virtual Collocation" is As Defined in the Act.
- 125 "Voice Grade" means either an analog signal of 300 to 3000 Hz or a digital signal of 56/64 kilobits per second. When referring to digital voice grade service (a 56/64 kbps channel), the term "DS-0" may also be used.
- 126 "White Pages Directories" means directories or the portion of co-bound directories which include a list in alphabetical order by name of the telephone numbers and addresses of telecommunication company customers.

SCHEDULE OF E.SPIRE OPERATING SUBSIDIARIES

American Communication Services of Atlanta, Inc.

American Communication Services of Baton Rouge, Inc.

American Communication Services of Birmingham, Inc.

American Communication Services of Charleston, Inc.

American Communication Services of Columbia. Inc.

American Communication Services of Columbus, Inc.

American Communication Services of Greenville, Inc.

American Communication Services of Jackson, Inc.

American Communication Services of Jacksonville, Inc

American Communication Services of Knoxville, Inc.

American Communication Services of Lexington, Inc.

American Communication Services of Louisiana, Inc.

American Communication Services of Louisville, Inc.

American Communication Services of Mobile, Inc.

American Communication Services of Montgomery, Inc.

American Communication Services of Shreveport, Inc.

American Communication Services of Spartanburg, Inc.

American Communication Services of Tampa, Inc.

Attachment 1

Resale

RESALE

1. Discount Rates

e.spire shall be permitted to purchase all Telecommunications Services that BellSouth provides at retail to subscribers that are not Telecommunications Carriers at a wholesale discount rate off of the retail rate for the Telecommunications Service. The wholesale discount shall be as set forth in Exhibit A to this Attachment, attached hereto and incorporated herein by this reference. Such discount shall reflect the costs avoided or avoidable by BellSouth when selling a service for wholesale purposes as established by the applicable state Commissions.

2. Resale at Wholesale Rates – Section 251(c)(4)

- At the request of e.spire, BellSouth shall make available to e.spire for resale at wholesale rates all Telecommunications Services that BellSouth provides at retail to subscribers who are not Telecommunications Carriers, as required by Section 251(c)(4) of the Act (the "Resale Services") and applicable FCC orders and rules, at the terms, conditions and limitations set forth in this Agreement. Resale Services shall include, but not be limited to, the following categories of Telecommunications Services as long as they continue to be provided by BellSouth:
 - (i) Local Service Residence, as described in the applicable Tariff;
 - (ii) Local Service Business, as described in the applicable Tariff;
 - (iii) Message Toll Service, as described in the applicable Tariff;
 - (iv) PBX Trunk, as described in the applicable Tariff;
 - (v) ISDN Services, as described in the applicable Tariff;
 - (vi) Centrex Service, as described in the applicable Tariff;
 - (vii) Private Line Services, as described in the applicable Tariff;
 - (viii) IntraLATA Inbound Services, as described in the applicable Tariff;
 - (ix) Customer Owned Pay Telephone Access Line Services, as described in the applicable Tariff; and
 - (x) Frame Relay Service, as described in the applicable Tariff.

The Resale Services shall be made available to e.spire at the discount rates set forth in this Attachment to the Agreement.

2.2 Other Services

- 2.2.1. BellSouth may, at its sole discretion, and as agreed to by e.spire, make available to e.spire under this Agreement services other than Telecommunications Services for resale at rates, terms and conditions agreed upon by the Parties.
- 2.2.2. "Grandfathered Services" include any Telecommunications Services, which BellSouth offers to existing subscribers of applicable retail services, but not to new subscribers. BellSouth agrees to make Grandfathered Services available to e.spire for resale to any End User of BellSouth that subscribes to a Grandfathered Service from BellSouth at the time of its selection of e.spire as its service provider; provided that if such Grandfathered Services are provided under a Shared Tenant Service arrangement, such Grandfathered Services shall be available for resale by e.spire to all existing and future tenants of the premises covered by the Shared Tenant Service arrangement. If a local Telecommunications Service is subsequently classified as a Grandfathered Service by BellSouth, BellSouth agrees to continue to sell such Grandfathered Service to e.spire for resale to e.spire's Customers that subscribe to such Grandfathered Service at the time it is so classified by BellSouth, on the same terms and conditions that BellSouth sales the service to its own end users.
- 2.2.3. Excepting the application of the wholesale discount specified in Exhibit A hereto, each Party acknowledges that Resale Services shall be available to e.spire on the same basis as offered by BellSouth to itself or to any Subsidiary, Affiliate, or any other Person to which BellSouth directly provides the Resale Services, including BellSouth's retail Customers and other resellers of BellSouth's Telecommunications Services, provided that such Resale Services shall be provided (i) only in those service areas in which such Resale Services (or any feature or capability thereof) are offered by BellSouth as an incumbent LEC to any other Person, and (ii) to the same extent as BellSouth's retail Telecommunications Services are subject to the availability of facilities.

3. General Provisions

3.1 BellSouth shall make available Telecommunications Services for resale at the rates set forth herein to this Attachments and subject to the exclusions and limitations set forth in Exhibit B to this Agreement. Unless otherwise set forth in this Attachment, neither Party hereby waives its rights to appeal or otherwise challenge any decision regarding resale that resulted in the discount rates contained in Exhibit A or the exclusions and limitations contained in Exhibit B.

The Parties reserve the right to pursue any and all legal and/or equitable remedies, including appeals of any decisions. If such appeals or challenges result in changes in the discount rates or exclusions and limitations, the Parties agree that appropriate modifications to this Agreement will be made promptly to make its terms consistent with the outcome of the appeal.

- 3.2 e.spire may purchase Resale services from BellSouth for its own use in operating its business. The resale discount will apply to those services under the following conditions:
 - 3.2.1 e.spire must resell services to other End Users.
 - 3.2.2 e.spire must order services through resale interfaces, i. e., the Local Carrier Service Center (LCSC) and/or appropriate Resale Account Teams pursuant to Section 3 of the General Terms and Conditions.
 - 3.2.3 e.spire cannot be a CLEC for the single purpose of selling to itself.
- 3.3 BellSouth shall not be required to provide to e.spire Resale Services at a wholesale rate when those services are offered at a special promotional rate if:
 - (a) Such promotions involve rates that will be in effect for not more than ninety (90) days; and
 - (b) Such promotional offerings are not used to evade the wholesale rate obligation; for example, by making available a series of ninety (90) day promotional rates.
- 3.4 The provision of services by BellSouth to e.spire does not constitute a joint undertaking for the furnishing of any service.
- e.spire will be the Customer of Record for all services purchased from BellSouth. Except as specified herein, BellSouth will take orders from, bill and expect payment from e.spire for all services.
- 3.6 e.spire will be BellSouth's single point of contact for all services purchased pursuant to this Agreement. BellSouth shall have no contact with the End User except to the extent provided for herein.
- 3.7 BellSouth will continue to bill the End User for any services that the End User specifies it wishes to receive directly from BellSouth.
- 3.8 BellSouth maintains the right to serve directly any End User within the service area of e.spire. BellSouth will continue to directly market its own telecommunications products and services and in doing so may establish independent relationships with End Users of e.spire.

- 3.9 Neither Party shall interfere with the right of any person or entity to obtain service directly from the other Party.
- 3.10 Current telephone numbers may normally be retained by the End User. However, neither Party nor the End User has a property right to the telephone number or any other call number designation associated with services furnished by BellSouth, and no right to require the continuance of service through any particular Central Office. BellSouth reserves the right to change such numbers, or the Central Office designation associated with such numbers, or both, whenever such change is reasonably necessary to the conduct of business.
- 3.11 For the purpose of the resale of BellSouth's Telecommunications Services by e.spire, BellSouth will provide e.spire with an on line access to telephone numbers for reservation on a first come first serve basis. Such reservations of telephone numbers, on a pre-ordering basis shall be for a period of nine (9) days. e.spire acknowledges that there may be instances where there is a shortage of telephone numbers in a particular Common Language Location Identifier Code (CLLIC) and in such instances BellSouth may request on a nondiscriminatory basis that e.spire cancel its reservations of numbers. e.spire shall comply with such request.
- 3.12 Further, upon e.spire's request, and for the purpose of the resale of BellSouth's telecommunications services by e.spire, BellSouth will reserve up to 100 telephone numbers per CLLIC, for e.spire's sole use. Such telephone number reservations shall be valid for ninety (90) days from the reservation date. e.spire acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth shall use its best efforts to reserve for a ninety (90) day period a sufficient quantity of e.spire's reasonable need in that particular CLLIC.
- 3.13 BellSouth may provide any Telecommunications Service it offers to its End Users or facility for which a charge is not established herein, as long as it is offered on the same terms to e.spire at rates that reflect the resale discount expressed in Exhibit A hereto.
- 3.14 Service is furnished subject to the condition that it will not be used for any unlawful purpose.
- 3.15 Service will be discontinued if any law enforcement agency advises that the service being used is in violation of the law.
- 3.16 BellSouth can refuse service when it has grounds to believe that service will be used in violation of the law.
- 3.17 BellSouth accepts no responsibility to any person for any unlawful act committed by e.spire or its End Users as part of providing service to e.spire for purposes of resale or otherwise.

- 3.18 The Parties will cooperate fully with law enforcement agencies with subpoenas and court orders as specified in Section 17 of the General Terms and Conditions of this Agreement.
- 3.19 The characteristics and methods of operation of any circuits, facilities or equipment provided by any person or entity other than BellSouth shall not:
 - 3.19.1. Interfere with or impair service over any facilities of BellSouth, its Affiliates, or its connecting and concurring carriers involved in its service;
 - 3.19.2. Cause damage to BellSouth's plant;
 - 3.19.3. Impair the privacy of any communications; or
 - 3.19.4. Create hazards to any BellSouth employees or the public.
- 3.20 e.spire assumes the responsibility of notifying BellSouth regarding less than standard operations with respect to services provided by e.spire.
- 3.21 Facilities and/or equipment utilized by BellSouth to provide service to e.spire remain the property of BellSouth.
- 3.22 White Page Directory Listings will be provided in accordance with regulations set forth in Section A6 of the General Subscriber Services Tariff and will be available for resale.
- 3.23 BellSouth shall provide electronic access to customer record information to e.spire; provided that e.spire has the appropriate Letter(s) of Authorization. BellSouth shall provide customer record information in accordance with the terms of Attachment 6 (OSS, O&P) hereto.
- 3.24 Where available to BellSouth's end users, BellSouth shall provide the following Telecommunications Services at a discount to allow for voice mail services:
 - Simplified Message Desk Interface Enhanced ("SMDI-E")
 - Simplified Message Desk Interface ("SMDI") Message Waiting Indicator ("MWI") stutter dialtone and message waiting light feature capabilities
 - Call Forward on Busy/Don't Answer ("CF-B/DA")
 - Call Forward on Busy ("CF/B")
 - Call Forward Don't Answer ("CF/DA")
- 3.25 BellSouth messaging services set forth in BellSouth's Messaging Service Information Package shall be made available for resale at rates charged to BellSouth End Users, but without the wholesale discount.

- 3.26 BellSouth's Inside Wire Maintenance Service Plan shall be made available for resale at rates, terms and conditions offered to BellSouth End Users, but without the wholesale discount.
- 3.27 Recovery of charges associated with implementing Number Portability through monthly charges assessed to end users has been authorized by the FCC. This end user line charge will be billed to Resellers of BellSouth's telecommunications services and will be as filed in the BellSouth FCC No. 1 tariff. This charge will not be discounted.
- 3.28 BellSouth shall refer all questions regarding any e.spire service or product directly to e.spire. BellSouth shall use its best efforts to ensure that all BellSouth representatives who receive inquiries regarding e.spire services do not in any way disparage or discriminate against e.spire or its products or services
- 3.29 The same quality standards that BellSouth requires of its employees when contacting BellSouth End Users (e.g., honesty, respect and courtesy) shall apply when its employees are in contact with e.spire End Users.

4. BellSouth's Provision of Services to e.spire

- 4.1 e.spire agrees that its resale of BellSouth services shall be as follows:
 - 4.1.1. The resale of telecommunications services shall be limited to users and uses conforming to the class of service restrictions.
 - 4.1.2. Hotel and Hospital PBX services are the only telecommunications services available for resale to Hotel/Motel and Hospital end users, respectively. Similarly, Access Line Service for Customer Provided Coin Telephones is the only local service available for resale to Independent Payphone Provider (IPP) customers. Shared Tenant Service customers can only be sold those local exchange access services available in BellSouth's A23 Shared Tenant Service Tariff in the states of Florida, Georgia, North Carolina and South Carolina, and in A27 in the states of Alabama, Kentucky, Louisiana, Mississippi and Tennessee.
 - 4.1.3. BellSouth reserves the right to periodically audit services purchased by e.spire to establish authenticity of use. Such audit shall not occur more than once in a calendar year. e.spire shall make any and all records and data available to BellSouth or BellSouth's auditors on a reasonable basis. BellSouth shall bear the cost of said audit.
- 4.2 Resold services can only be used in the same manner as specified in BellSouth's Tariffs. Resold services are subject to the same terms and conditions as are specified for such services when furnished to an individual End User of BellSouth in the appropriate section of BellSouth's Tariffs. Specific tariff features, e.g. a

- usage allowance per month, shall not be aggregated across multiple resold services.
- 4.3 e.spire may resell services only within the specific resale service area as defined in its certificate.
- 4.4 Telephone numbers transmitted via any resold service feature are intended solely for the use of the End User of the feature. Resale of this information is prohibited.
- 4.5 e.spire may provide both flat and measured rate service on the same business premise to the same subscribers (End Users) only in accordance with Section A2 of BellSouth's General Subscriber Tariff.

5. New Resale Services; Changes in Provision of Resale Services

BellSouth shall use best efforts to provide e.spire forty-five (45) days advance notice via Internet posting of changes to the prices, terms or conditions of services available for Resale. To the extent that revisions occur between the time BellSouth notifies e.spire of changes under this Agreement and the time the changes are scheduled to be implemented, BellSouth will notify e.spire of such revisions consistent with its internal notification process; provided that, e.spire shall not utilize any notice given under this subsection to market resold offerings of that service in advance of BellSouth. In addition, upon request BellSouth shall furnish e.spire with copies of publicly available service descriptions regarding the Resale Services. Notwithstanding the foregoing, e.spire shall not utilize any such BellSouth service descriptions as part of its own sales or marketing efforts.

6. Maintenance of Services

- 6.1 e.spire will adhere to the reasonable and nondiscriminatory procedures established by BellSouth regarding maintenance and installation of service.
- 6.2 Services resold under BellSouth's Tariffs and facilities and equipment provided by BellSouth shall be maintained by BellSouth.
- e.spire or its end users may not rearrange, move, disconnect, remove or attempt to repair any facilities owned by BellSouth, other than by connection or disconnection to any interface means used, except with the written consent of BellSouth.
- e.spire accepts responsibility to notify BellSouth of situations that arise that may result in a service problem.

- e.spire will be BellSouth's single point of contact for all repair calls on behalf of e.spire's End Users. The parties agree to provide one another with toll-free contact numbers for such purposes.
- 6.6 BellSouth will bill e.spire for handling troubles that are found not to be in BellSouth's network pursuant to its standard time and material charges. The standard time and material charges will be no more than what BellSouth charges to its retail customers for the same services.
- 6.7 BellSouth reserves the right to contact e.spire's End Users, if deemed necessary, for maintenance purposes.

7. Establishment of Service

- 7.1 After receiving certification as a Local Exchange Company from the appropriate regulatory agency, unless it has already done so, e.spire will provide the appropriate BellSouth service center the necessary documentation to enable BellSouth to establish a master account for e.spire's resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable. When necessary deposit requirements are met, BellSouth will begin taking orders for the resale of service.
- 7.2 Service orders will be in a standard format designated by BellSouth. All Local Service Requests ("LSRs") submitted for products and services under this Attachment will be subject to the OSS charges set forth in the General Terms and Conditions of this Agreement.
- 7.3 When notification is received from e.spire that a current End User of BellSouth will subscribe to e.spire's service, standard service order intervals for the appropriate class of service will apply.
- 7.4 BellSouth will not require End User confirmation prior to establishing service for e.spire's End User customer. e.spire must, however, be able to demonstrate End User authorization upon request.
- 7.5 e.spire will be the single point of contact with BellSouth for all subsequent ordering activity resulting in additions or changes to resold services except that BellSouth will accept a request directly from the End User for conversion of the End User's service from e.spire to BellSouth or will accept a request from another CLEC for conversion of the End User's service from e.spire to the other LEC. BellSouth will notify e.spire within five (5) business days via US mail that such a request has been processed.

- 7.6 If BellSouth determines that an unauthorized change in local service to e.spire has occurred, BellSouth will reestablish service with the appropriate local service provider and will assess e.spire as the CLEC initiating the unauthorized change, the unauthorized change charge described in BellSouth FCC. Tariff No. 1, Section 13 or applicable state tariff. Appropriate nonrecurring charges, as set forth in Section A4 of the General Subscriber Service Tariff, will also be assessed to e.spire. These charges will be refunded if e.spire provides satisfactory proof of authorization.
- 7.7 BellSouth reserves the right to secure a deposit not to exceed two (2) estimated months billing. Any such security deposit may be held during the continuance of the service as security for the payment of any and all amounts accruing for the service.
 - 7.7.1 The fact that a security deposit has been made in no way relieves from complying with BellSouth's regulations as to advance payments and the prompt payment of bills on presentation nor does it constitute a waiver or modification of the regular practices of BellSouth providing for the discontinuance of service for non-payment of any sums due BellSouth.
 - 7.7.2 BellSouth reserves the right to increase the security deposit requirements when, in its sole judgment, circumstances so warrant and gross monthly billing has increased beyond the level initially used to determine the security deposit.
 - 7.7.3 In the event that e.spire defaults on its account, service to e.spire will be terminated and any security deposits held will be applied to its account.
 - 7.7.4 Interest on a security deposit shall accrue and be paid in accordance with the terms in the appropriate BellSouth tariff.
- 7.8 Orders to switch services "as is" shall be treated as a change of service and shall *not* be treated as a disconnection and subsequent reconnection of service.
- 7.9 BellSouth shall provide e.spire notification of disconnects, updated and delivered once daily, via an electronic process known as OUTPLOC.

8. Payment And Billing Arrangements

- 8.1 BellSouth shall bill e.spire on a current basis all applicable charges and credits.
- 8.2 Payment of all charges will be the responsibility of e.spire. e.spire shall make payment to BellSouth for all services billed. BellSouth is not responsible for payments not received by e.spire from e.spire's End User. BellSouth will not become involved in billing disputes that may arise between e.spire and its End

- User. Payments made to BellSouth as payment on account will be credited to an accounts receivable master account and not to an End User's account.
- 8.3 BellSouth will render bills each month on established bill days for each of e.spire's accounts.
- 8.4 BellSouth will bill e.spire, in advance charges for all services to be provided during the ensuing billing period except charges associated with service usage, which charges will be billed in arrears. Charges will be calculated on an individual End User account level, including, if applicable, any charge for usage or usage allowances. BellSouth will also bill e.spire, and e.spire will be responsible for and remit to BellSouth, all government mandated surcharges applicable to resold services including but not limited to 911 and E911 charges, telecommunications relay charges (TRS), and franchise fees.
- 8.5 The payment will be due by the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by BellSouth.
 - 8.5.1. If the payment due date falls on a Sunday or on a holiday which is observed on a Monday, the payment due date shall be the first non-holiday day following such Sunday or holiday. If the payment due date falls on a Saturday or on a holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-holiday day preceding such Saturday or holiday. If payment is not received by the payment due date, a late payment penalty, as set forth in Section 8.7 following, shall apply.
 - 8.5.2. If e.spire requests multiple billing media or additional copies of bills, BellSouth will provide these at an appropriate charge to e.spire.
- 8.6 Billing Disputes
 - 8.6.1 Any billing disputes shall be handled in accordance with Section 26 of the General Term and Conditions and Attachment 7 of this Agreement.
- 8.7 Upon proof of tax exempt certification from e.spire, the total amount billed to e.spire will not include any taxes due from the end user to reflect the tax exempt certification and local tax laws. e.spire will be solely responsible for the computation, tracking, reporting, and payment of taxes applicable to e.spire's end user.
- 8.8 If any portion of the payment is received by BellSouth after the payment due date as set forth preceding, or if any portion of the payment is received by BellSouth in funds that are not immediately available to BellSouth, then a late payment penalty shall be due to BellSouth. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor and will be applied on a per bill basis. The late factor shall be as set forth in Section A2 of

the General Subscriber Services Tariff and Section B2 of the Private Line Service Tariff.

- 8.9 Any switched access charges properly billed to interexchange carriers for access to the resold local exchange lines will be billed by, and due to, BellSouth. e.spire shall bill access charge components properly billed to End Users.
- 8.10 BellSouth will not perform billing and collection services for e.spire as a result of the execution of this Agreement. All requests for billing services should be referred to the appropriate entity or operational group within BellSouth.
- 8.11 Pursuant to 47 CFR Section 51.617, BellSouth will bill espire directly the end user common line charges in the amount identical to the end user common line charges BellSouth bills its end users.
- 8.12 In general, BellSouth will not become involved in disputes between e.spire and e.spire's End User customers over resold services. If a dispute does arise that cannot be settled without the involvement of BellSouth, e.spire shall contact the designated Service Center for resolution. BellSouth will make every effort to assist in the resolution of the dispute and will work with e.spire to resolve the matter in as timely a manner as possible. e.spire may be required to submit documentation to substantiate the claim.

9. Discontinuance of Service

- 9.1 The procedures for discontinuing service to an End User are as follows:
 - 9.1.1. Where possible, BellSouth will deny service to e.spire's End User on behalf of, and at the request of, e.spire. Upon restoration of the End User's service, restoral charges will apply and will be the responsibility of e.spire.
 - 9.1.2. At the request of e.spire, BellSouth will disconnect -an e.spire End User customer.
 - 9.1.3. All requests by e.spire for denial or disconnection of an End User for nonpayment must be in writing.
 - 9.1.4. e.spire will be made solely responsible for notifying the End User of the proposed disconnection of the service.
 - 9.1.5. BellSouth will continue to process calls made to the Annoyance Call Center and will advise e.spire when it is determined that annoyance calls are originated from one of their End User's locations. BellSouth shall be indemnified, defended and held harmless by e.spire and/or the End User against any claim, loss or damage arising from providing this information

- to e.spire. It is the responsibility of e.spire to take the corrective action necessary with its End Users who make annoying calls. Failure to do so will result in BellSouth's disconnecting the End User's service.
- 9.1.6. Use of Facilities. When an End User of e.spire elects to discontinue service from e.spire and to transfer service to another LEC, including BellSouth, BellSouth shall have the right to reuse the facilities provided to e.spire for retail or Resale service or as, unbundled Loops or unbundled Ports for that End User. In addition, BellSouth may disconnect and reuse facilities when the facility is in a denied state, and BellSouth has received an order to establish new service, or transfer service from an End User or an End User's CLEC, at the same address served by the denied facility.
 - 9.1.6.1. The foregoing applies when BellSouth has received a new order from the End User or the End User's new LEC for a retail service or Resale service or for a UNE which the End User or the End User's new LEC has indicated constitutes a transfer of service from the LEC to another provider (i.e., the order is not for a new line or an additional line).
 - 9.1.6.2. The order for retail service, Resale service, unbundled Loop and/or Port can be for either Exchange or private line service.
- 9.2 The procedures for discontinuing service to e.spire are as follows:
 - 9.2.1. BellSouth reserves the right to suspend or terminate service for nonpayment or in the event of prohibited, unlawful or improper use of the facilities or service, abuse of the facilities, or any other violation or noncompliance by e.spire of the rules and regulations of BellSouth's Tariffs.
 - 9.2.2. If payment of account is not received by the bill day in the month after the original bill day, BellSouth may provide written notice to e.spire, that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition BellSouth may, at the same time, give thirty days notice to the person designated by e.spire to receive notices of noncompliance, and discontinue the provision of existing services to e.spire at any time thereafter.
 - 9.2.3. In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
 - 9.2.4. If BellSouth does not discontinue the provision of the services involved on the date specified in the thirty days notice and e.spire's noncompliance continues, nothing contained herein shall preclude BellSouth's right to discontinue the provision of the services to e.spire without further notice.

- 9.2.5. If payment is not received or arrangements made for payment by the date given in the written notification, e.spire's services will be discontinued. Upon discontinuance of service on a e.spire's account, service to e.spire's End Users will be denied. BellSouth will also reestablish service at the request of the End User or e.spire upon payment of the appropriate connection fee and subject to BellSouth's normal application procedures. e.spire is solely responsible for notifying the End User of the proposed disconnection of the service.
- 9.2.6. If within fifteen days after an End User's service has been denied no contact has been made in reference to restoring service, the End User's service will be disconnected.

10. Functionality Required to Support Resale Service.

- 10.1 LEC Assigned Telephone Calling Card Numbers. Effective as of the date of an End User's subscription to e.spire's service, BellSouth shall block the LEC-assigned telephone line calling card number (including area code) ("TLN") from the Line Identification Database ("LIDB"), unless otherwise agreed by e.spire in the Implementation Plan.
- 10.2 Telephone Assistance Programs. Telephone Assistance Programs shall be available for Resale as indicated in Exhibit B to this Attachment. Upon conversion to e.spire's Resale Service of an existing Telecommunications Assistance Program Customer, no exchange of qualification documentation is necessary.
- 10.3 9-1-1 Services. BellSouth shall provide to e.spire "911" emergency call routing services in accordance with the terms of Attachment 2 to the Agreement.
- 10.4 Special Services. If BellSouth makes a notation on the Customer Service Record ("CSR") of End Users who qualify for certain services available to physically challenged individuals (e.g., special discounts) ("Special Services"), BellSouth shall provide such data to e.spire on the CSR made available to BellSouth for its End Users. For usage by an e.spire End User of a Telephone Relay Service ("TRS"), BellSouth shall provide e.spire with all billing information furnished to BellSouth by the provider of the TRS.
- 10.5 TTY/TDD. BellSouth shall cooperate with e.spire to provide services and equipment necessary to serve TTY/TDD customers at rates, terms and conditions set forth in a separate agreement to be negotiated between the Parties.

11. Resale of Customer Specific Arrangements

11.1 CSAs shall be available for resale at the wholesale discount set forth in Exhibit A of this Attachment; provided, however, that in the event the Commission establishes a specific discount for CSAs such discount shall apply thereafter. e.spire may resell a CSA to the end user for whom the CSA was constructed or to end users similarly situated to the specific end user for whom the CSA was constructed. Customers shall be deemed to be similarly situated when the quantity of use; time of use; manner of service; and costs of rendering the service are the same. In cases where e.spire resells an existing CSA, no termination or rollover charges shall apply to the assignment of the CSA to e.spire provided that e.spire assumes the obligations set forth within the CSA. Notwithstanding the foregoing, BellSouth may impose a single service order charge (not to exceed the level of tariffed service order charges for comparable services) to recover the cost of changing the billing name on the account.

12. Line Information Database (LIDB)

- 12.1 BellSouth will store in its Line Information Database (LIDB) records relating to service only in the BellSouth region. The LIDB Storage Agreement is included in this Attachment as Exhibit C.
- 12.2 BellSouth will provide LIDB Storage upon written request to e.spire Account Manager stating requested activation date.

13. RAO Hosting

- 13.1 The RAO Hosting Agreement is included in this Attachment as Exhibit D. Rates for BellSouth's Centralized Message Distribution System (CMDS) are as set forth in Exhibit H of this Attachment.
- 13.2 BellSouth will provide RAO Hosting upon written request to its Account Manager stating requested activation date.

14. Optional Daily Usage File (ODUF)

- 14.1 The Optional Daily Usage File (ODUF) Agreement with terms and conditions is included in this Attachment as Exhibit E. Rates for ODUF are as set forth in Exhibit H of this Attachment.
- 14.2 BellSouth will provide Optional Daily Usage File (ODUF) service upon written request to its Account Manager stating requested activation date.

15. Enhanced Optional Daily Usage File (EODUF)

- 15.1 The Enhanced Optional Daily Usage File (EODUF) service Agreement with terms and conditions is included in this Attachment as Exhibit F. Rates for EODUF are as set forth in Exhibit H of this Attachment.
- 15.2 BellSouth will provide Enhanced Optional Daily Usage File (EODUF) service upon written request to its Account Manager stating requested activation date.

16. Calling Name Delivery (CNAM) Database Service

- 16.1 Calling Name Delivery (CNAM) Database Service Agreement is included in this Attachment as <u>Exhibit F</u>. Rates for CNAM are as set forth in <u>Exhibit H</u> of this Attachment.
- 16.2 BellSouth will provide Calling Name Delivery (CNAM) Database service upon written request to its Account Manager stating requested activation date.

EXHIBIT A

APPLICABLE DISCOUNTS

The Telecommunications Services available for purchase by e.spire for the purposes of resale to e.spire End Users shall be available at the following discount off of the retail rate.

DISCOUNT*

<u>STATE</u>	RESIDENCE	BUSINESS	CSAs***
ALABAMA	16.3%	16.3%	
FLORIDA	21.83%	16.81%	
GEORGIA	20.3%	17.3%	
KENTUCKY	16.79%	15.54%	
LOUISIANA	20.72%	20.72%	9.05%
MISSISSIPPI	15.75%	15.75%	
NORTH CAROLINA	21.5%	17.6%	
SOUTH CAROLINA	14.8%	14.8%	8.98%
TENNESSEE**	16%	16%	

- * When a CLEC provides Resale service in a cross boundary area (areas that are part of the local serving area of another state's exchange) the rates, regulations and discounts for the tariffing state will apply. Billing will be from the serving state.
- ** In Tennessee, if CLEC provides its own operator services and directory services, the discount shall be 21.56%. CLEC must provide written notification to BellSouth within 30 days prior to providing its own operator services and directory services to qualify for the higher discount rate of 21.56%.
- *** Unless noted in this column, the discount for Business will be the applicable discount rate for CSAs.

EXCLUSIONS AND LIMITATIONS ON SERVICES AVAILABLE FOR RESALE

-

	Type of	AL			FL	GA		KY		LA	
Service		Resale? Discount?		Resale? Discount?		Resale? Discount?		Resale? Discount?		Resale Discoun	
1	Grandfathered Services (Note 1)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
2	Contract Service Arrangements	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
3	Promotions - > 90 Days(Note 2)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
4	Promotions - < 90 Days (Note 2)	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
5	Lifeline/Link Up Services	Yes	Yes	Yes	Yes	Yes	Yes	Note 3	Note 3	Yes	Yes
6	911/E911 Services	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
7	N11 Services	Yes	Yes	Yes	Yes	Yes	Yes	No	No	No	No
8	AdWatch SM Svc (See Note 6)	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
9	MemoryCall® Service	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
10	Mobile Services	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
11	Federal Subscriber Line Charges	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
12	Non-Recurring Charges	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
13	End User Line Charge – Number Portability	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
	Type of		MS		NC		SC		TN	1	

	Type of		MS		NC		SC	TN	
	Service	Resale? Discount?		Resale?	Resale? Discount?		Resale? Discount?		Discount?
1	Grandfathered Services (Note 1)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
2	Contract Service Arrangements	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
3	Promotions - > 90 Days	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Note 2
4	Promotions - < 90 Days (Note 2)	Yes	No	Yes	No	Yes	No	Yes	No
5	Lifeline/Link Up Services	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Note 3
6	911/E911 Services	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
7	N11 Services	No	No	No	No	Yes	Yes	Yes	Yes

8	AdWatch SM Svc (See Note 6)	Yes	No	Yes	No	Yes	No	Yes	No
9	MemoryCall [®] Service	Yes	No	Yes	No	Yes	No	Yes	No
10	Mobile Services	Yes	No	Yes	No	Yes	No	Yes	No
	Federal Subscriber Line Charges	Yes	No	Yes	No	Yes	No	Yes	No
12	Non-Recurring Charges	Yes	No						
13	End User Line Charge – Number Portability	Yes	No	Yes	No	Yes	No	Yes	No

Applicable Notes:

- **1 Grandfathered services** can be resold only to existing subscribers of the grandfathered service.
- Where available for resale, **promotions** will be made available only to end users who would have qualified for the promotion had it been provided by BellSouth directly.
- Lifeline/Link Up services may be offered only to those subscribers who meet the criteria that BellSouth currently applies to subscribers of these services. In Kentucky, e.spire is responsible for funding its own Lifeline and Link Up benefit. In Tennessee, e.spire shall purchase BellSouth's Message Rate Service at the stated tariff rate, less the wholesale discount. e.spire must further discount the wholesale Message Rate Service to Lifeline customers with a discount which is no less than the minimum discount that BellSouth now provides. e.spire is responsible for recovering the Subscriber Line Charge from the National Exchange Carriers Association interstate toll settlement pool just as BellSouth does today. The maximum rate that e.spire may charge for Lifeline Service shall be capped at the flat retail rate offered by BellSouth.
- **<u>4</u>** Some of BellSouth's local exchange and toll telecommunications services are not available in certain central offices and areas.
- <u>5</u> AdWatchSM Service is tariffed as BellSouth[®] AIN Virtual Number Call Detail Service.

LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

I. SCOPE

A. This Agreement sets forth the terms and conditions pursuant to which BST agrees to store in its LIDB certain information at the request of the Local Exchange Company and pursuant to which BST, its LIDB customers and Local Exchange Carrier shall have access to such information. Local Exchange Carrier understands that BST provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of Local Exchange Carrier, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained in the attached Addendum(s) are hereby made a part of this Agreement as if fully incorporated herein.

- B. LIDB is accessed for the following purposes:
 - 1. Billed Number Screening
 - 2. Calling Card Validation
 - 3. Fraud Control
- C. BST will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BST's LIDB, provided that such information is included in the LIDB query. BST will establish fraud alert thresholds and will notify the Local Exchange Company of fraud alerts so that the Local Exchange Company

may take action it deems appropriate. Local Exchange Company understands and agrees BST will administer all data stored in the LIDB, including the data provided by Local Exchange Company pursuant to this Agreement, in the same manner as BST's data for BST's end user customers. BST shall not be responsible to Local Exchange Company for any lost revenue which may result from BST's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BST in its sole discretion from time to time.

Local Exchange Company understands that BST currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearing houses. Local Exchange Company further understands that these billing and collection customers of BST query BST's LIDB to determine whether to accept various billing options from end users. Additionally, Local Exchange Company understands that presently BST has no method to differentiate between BST's own billing and line data in the LIDB and such data which it includes in the LIDB on Local Exchange Company's behalf pursuant to this Agreement. Therefore, until such time as BST can and does implement in its LIDB and its supporting systems the means to differentiate Local Exchange Company's data from BST's data and the parties to this Agreement execute appropriate amendments hereto, the following terms and conditions shall apply:

(a) The Local Exchange Company agrees that it will accept responsibility for telecommunications services billed by BST for its billing and collection customers for Local Exchange Customer's end user accounts which are resident in LIDB pursuant to this Agreement. Local Exchange Company authorizes BST to place such charges on Local

Exchange Company's bill from BST and agrees that it shall pay all such charges. Charges for which Local Exchange Company hereby takes responsibility include, but are not limited to, collect and third number calls.

- (b) Charges for such services shall appear on a separate BST bill page identified with the name of the entity for which BST is billing the charge.
- (c) Local Exchange Company shall have the responsibility to render a billing statement to its end users for these charges, but Local Exchange Company's obligation to pay BST for the charges billed shall be independent of whether Local Exchange Company is able or not to collect from the Local Exchange Company's end users.
- (d) BST shall not become involved in any disputes between Local Exchange Company and the entities for which BST performs billing and collection.

 BellSouth will not issue adjustments for charges billed on behalf of an entity to Local Exchange Company. It shall be the responsibility of the Local Exchange Company and the other entity to negotiate and arrange for any appropriate adjustments.

II. TERM

This Agreement will be effective as of ________, 2000, and will continue in effect for one year, and thereafter may be continued until terminated by either party upon thirty (30) days written notice to the other party.

III. FEES FOR SERVICE AND TAXES

- A. The Local Exchange Company will not be charged a fee for storage services provided by BST to the Local Exchange Company, as described in Section I of this Agreement.
- B. Sales, use and all other taxes (excluding taxes on BST's income) determined by BST or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by the Local Exchange Company. The Local Exchange Company shall have the right to have BST contest with the imposing jurisdiction, at the Local Exchange Company's expense, any such taxes that the Local Exchange Company deems are improperly levied.

IV. INDEMNIFICATION

To the extent not prohibited by law, each party will indemnify the other and hold the other harmless against any loss, cost, claim, injury, or liability relating to or arising out of negligence or willful misconduct by the indemnifying party or its agents or contractors in connection with the indemnifying party's provision of services, provided, however, that any indemnity for any loss, cost, claim, injury or liability arising out of or relating to errors or omissions in the provision of services under this Agreement shall be limited as otherwise specified in this Agreement. The indemnifying party under this Section agrees to defend any suit brought against the other party for any such loss, cost, claim, injury or liability. The indemnified party agrees to notify the other party promptly, in writing, of any written claims, lawsuits, or demands for which the other party is responsible under this Section and to cooperate in every reasonable way to facilitate defense or settlement of claims. The

indemnifying party shall not be liable under this Section for settlement by the indemnified party of any claim, lawsuit, or demand unless the defense of the claim, lawsuit, or demand has been tendered to it in writing and the indemnifying party has unreasonably failed to assume such defense.

V. LIMITATION OF LIABILITY

Neither party shall be liable to the other party for any lost profits or revenues or for any indirect, incidental or consequential damages incurred by the other party arising from this Agreement or the services performed or not performed hereunder, regardless of the cause of such loss or damage.

VI. MISCELLANEOUS

- A. It is understood and agreed to by the parties that BST may provide similar services to other companies.
- B. All terms, conditions and operations under this Agreement shall be performed in accordance with, and subject to, all applicable local, state or federal legal and regulatory tariffs, rulings, and other requirements of the federal courts, the U. S. Department of Justice and state and federal regulatory agencies. Nothing in this Agreement shall be construed to cause either party to violate any such legal or regulatory requirement and either party's obligation to perform shall be subject to all such requirements.
 - C. The Local Exchange Company agrees to submit to BST all advertising, sales

promotion, press releases, and other publicity matters relating to this Agreement wherein BST's corporate or trade names, logos, trademarks or service marks or those of BST's affiliated companies are mentioned or language from which the connection of said names or trademarks therewith may be inferred or implied; and the Local Exchange Company further agrees not to publish or use advertising, sales promotions, press releases, or publicity matters without BST's prior written approval.

- D. This Agreement constitutes the entire agreement between the Local Exchange Company and BST which supersedes all prior agreements or contracts, oral or written representations, statements, negotiations, understandings, proposals and undertakings with respect to the subject matter hereof.
- E. Except as expressly provided in this Agreement, if any part of this Agreement is held or construed to be invalid or unenforceable, the validity of any other Section of this Agreement shall remain in full force and effect to the extent permissible or appropriate in furtherance of the intent of this Agreement.
- F. Neither party shall be held liable for any delay or failure in performance of any part of this Agreement for any cause beyond its control and without its fault or negligence, such as acts of God, acts of civil or military authority, government regulations, embargoes, epidemics, war, terrorist acts, riots, insurrections, fires, explosions, earthquakes, nuclear accidents, floods, strikes, power blackouts, volcanic action, other major environmental disturbances, unusually severe weather conditions, inability to secure products or services of other persons or transportation facilities, or acts or omissions of transportation common carriers.

G. This Agreement shall be deemed to be a contract made under the laws of the State of Georgia, and the construction, interpretation and performance of this Agreement and all transactions hereunder shall be governed by the domestic law of such State.

RESALE ADDENDUM TO LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

	This is a Resale Addendum to the Line I	Information Data Base Storage Agreement
dated _	1, 2000	0, between BellSouth Telecommunications,
Inc. ("	("BST"), and Local Exchange Company ('Local Exchange Company"), effective the
	_ day of, 2000.	

I. GENERAL

This Addendum sets forth the terms and conditions for Local Exchange Company's provision of billing number information to BST for inclusion in BST's LIDB. BST will store in its LIDB the billing number information provided by Local Exchange Company, and BST will provide responses to on-line, call-by-call queries to this information for purposes specified in Section I.B. of the Agreement.

II. DEFINITIONS

- A. Billing number a number used by BST for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.
- B. Line number a ten digit number assigned by BST that identifies a telephone line associated with a resold local exchange service, or with a SPNP arrangement.
- C. Special billing number a ten digit number that identifies a billing account established by BST in connection with a resold local exchange service or with a SPNP arrangement.
 - D. Calling Card number a billing number plus PIN number assigned by BST.
- E. PIN number a four digit security code assigned by BST which is added to a billing number to compose a fourteen digit calling card number.
- F. Toll billing exception indicator associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by the Local Exchange Company.
- G. Billed Number Screening refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.
- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- I. Billing number information information about billing number or Calling Card number as assigned by BST and toll billing exception indicator provided to BST by the Local Exchange Company.

III. RESPONSIBILITIES OF PARTIES

- A. BST will include billing number information associated with resold exchange lines or SPNP arrangements in its LIDB. The Local Exchange Company will request any toll billing exceptions via the Local Service Request (LSR) form used to order resold exchange lines, or the SPNP service request form used to order SPNP arrangements.
- B. Under normal operating conditions, BST shall include the billing number information in its LIDB upon completion of the service order establishing either the resold local exchange service or the SPNP arrangement, provided that BST shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BST's reasonable control. BST will store in its LIDB an unlimited volume of the working telephone numbers associated with either the resold local exchange lines or the SPNP arrangements. For resold local exchange lines or for SPNP arrangements, BST will issue line-based calling cards only in the name of Local Exchange Company. BST will not issue line-based calling cards in the name of Local Exchange Company's individual end users. In the event that Local Exchange Company wants to include calling card numbers assigned by the Local Exchange Company in the BST LIDB, a separate agreement is required.
- C. BST will provide responses to on-line, call-by-call queries to the stored information for the specific purposes listed in the next paragraph.
- D. BST is authorized to use the billing number information to perform the following functions for authorized users on an on-line basis:

- 1. Validate a 14 digit Calling Card number where the first 10 digits are a line number or special billing number assigned by BST, and where the last four digits (PIN) are a security code assigned by BST.
- 2. Determine whether the Local Exchange Company has identified the billing number as one which should not be billed for collect or third number calls, or both.

RAO Hosting

- 1. RAO Hosting, Calling Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to e.spire by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this Agreement, including such revisions as may be made from time to time by BellSouth.
- 2. e.spire shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- 3. Applicable compensation amounts will be billed by BellSouth to e.spire on a monthly basis in arrears. Amounts due from one Party to the other (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- 4. e.spire must have its own unique RAO code. Requests for establishment of RAO status where BellSouth is the selected Centralized Message Distribution System (CMDS) interfacing host, require written notification from e.spire to the BellSouth RAO Hosting coordinator at least eight (8) weeks prior to the proposed effective date. The proposed effective date will be mutually agreed upon between the Parties with consideration given to time necessary for the completion of required Telcordia (formerly BellCore) functions. BellSouth will request the assignment of an RAO code from its connecting contractor, currently Telcordia (formerly BellCore), on behalf of e.spire and will coordinate all associated conversion activities.
- 5. BellSouth will receive messages from e.spire that are to be processed by BellSouth, another LEC or CLEC in the BellSouth region or a LEC outside the BellSouth region.
- 6. BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from e.spire.
- 7. All data received from e.spire that is to be processed or billed by another LEC or CLEC within the BellSouth region will be distributed to that LEC or CLEC in accordance with the agreement(s) which may be in effect between BellSouth and the involved LEC or CLEC.
- 8. All data received from e.spire that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the agreement(s) which may be in effect between BellSouth and its connecting contractor (currently Telcordia (formerly BellCore)).
- 9. BellSouth will receive messages from the CMDS network that are destined to be processed by e.spire and will forward them to e.spire on a daily basis.

- 10. Transmission of message data between BellSouth and e.spire will be via CONNECT:Direct.
- 11. All messages and related data exchanged between BellSouth and e.spire will be formatted in accordance with accepted industry standards for EMI formatted records and packed between appropriate EMI header and trailer records, also in accordance with accepted industry standards.
- e.spire will ensure that the recorded message detail necessary to recreate files provided to BellSouth will be maintained for back-up purposes for a period of three (3) calendar months beyond the related message dates.
- 13. Should it become necessary for e.spire to send data to BellSouth more than sixty (60) days past the message date(s), e.spire will notify BellSouth in advance of the transmission of the data. If there will be impacts outside the BellSouth region, BellSouth will work with its connecting contractor and e.spire to notify all affected Parties.
- In the event that data to be exchanged between the two Parties should become lost or destroyed, both Parties will work together to determine the source of the problem. Once the cause of the problem has been jointly determined and the responsible Party (BellSouth or e.spire) identified and agreed to, the company responsible for creating the data (BellSouth or e.spire) will make every effort to have the affected data restored and retransmitted. If the data cannot be retrieved, the responsible Party will be liable to the other Party for any resulting lost revenue. Lost revenue may be a combination of revenues that could not be billed to the end users and associated access revenues. Both Parties will work together to estimate the revenue amount based upon historical data through a method mutually agreed upon. The resulting estimated revenue loss will be paid by the responsible Party to the other Party within three (3) calendar months of the date of problem resolution, or as mutually agreed upon by the Parties.
- 15. Should an error be detected by the EMI format edits performed by BellSouth on data received from e.spire, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify e.spire of the error condition. e.spire will correct the error(s) and will resend the entire pack to BellSouth for processing. In the event that an out-of-sequence condition occurs on subsequent packs, e.spire will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth.
- In association with message distribution service, BellSouth will provide e.spire with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 17. In no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this agreement.
- 18. RAO Compensation

- 18.1 Rates for message distribution service provided by BellSouth for e.spire are as set forth in Exhibit A to this Attachment.
- 18.2 Rates for data transmission associated with message distribution service are as set forth in Exhibit A to this Attachment .
- Data circuits (private line or dial-up) will be required between BellSouth and e.spire for the purpose of data transmission. Where a dedicated line is required, e.spire will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. e.spire will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to e.spire. Additionally, all message toll charges associated with the use of the dial circuit by e.spire will be the responsibility of e.spire. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties.
- All equipment, including modems and software, that is required on the e.spire end for the purpose of data transmission will be the responsibility of e.spire.

19. Intercompany Settlements Messages

- 19.1 This Section addresses the settlement of revenues associated with traffic originated from or billed by e.spire as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one Bell operating territory and bills in another Bell operating territory is included. Traffic that originates and bills within the same Bell operating territory will be settled on a local basis between e.spire and the involved company(ies), unless that company is participating in NICS.
- 19.2 Both traffic that originates outside the BellSouth region by e.spire and is billed within the BellSouth region, and traffic that originates within the BellSouth region and is billed outside the BellSouth region by e.spire, is covered by this Agreement (CATS). Also covered is traffic that either is originated by or billed by e.spire, involves a company other than e.spire, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).
- 19.3 Once e.spire is operating within the BellSouth territory, revenues associated with calls originated and billed within the BellSouth region will be settled via Telcordia (formerly BellCore)'s, its successor or assign, NICS system.
- BellSouth will receive the monthly NICS reports from Telcordia (formerly BellCore), its successor or assign, on behalf of e.spire. BellSouth will distribute copies of these reports to e.spire on a monthly basis.

- 19.5 BellSouth will receive the monthly Calling Card and Third Number Settlement System (CATS) reports from Telcordia (formerly BellCore), its successor or assign, on behalf of e.spire. BellSouth will distribute copies of these reports to e.spire on a monthly basis.
- 19.6 BellSouth will collect the revenue earned by e.spire from the Bell operating company in whose territory the messages are billed (CATS), less a per message billing and collection fee of five cents (\$0.05), on behalf of e.spire. BellSouth will remit the revenue billed by e.spire to the Bell operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf on e.spire. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to e.spire via a monthly Carrier Access Billing System (CABS) miscellaneous bill.
- 19.7 BellSouth will collect the revenue earned by e.spire within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of e.spire. BellSouth will remit the revenue billed by e.spire within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit issued to e.spire via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

BellSouth and e.spire agree that monthly netted amounts of less than fifty dollars (\$50.00) will not be settled.

Optional Daily Usage File

- Upon written request from e.spire, BellSouth will provide the Optional Daily Usage File (ODUF) service to e.spire pursuant to the terms and conditions set forth in this section.
- The e.spire shall furnish all relevant information required by BellSouth for the provision of the Optional Daily Usage File.
- The Optional Daily Usage Feed will contain billable messages that were carried over the BellSouth Network and processed in the BellSouth Billing System, but billed to a e.spire customer.

Charges for delivery of the Optional Daily Usage File will appear on the e.spire's monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- The Optional Daily Usage Feed will contain both rated and unrated messages. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of the e.spire will be the responsibility of the e.spire. If, however, the e.spire should encounter significant volumes of errored messages that prevent processing by the e.spire within its systems, BellSouth will work with the e.spire to determine the source of the errors and the appropriate resolution.
- 6 The following specifications shall apply to the Optional Daily Usage Feed.
 - 6.1 Usage To Be Transmitted
 - 6.1.1 The following messages recorded by BellSouth will be transmitted to the e.spire:
 - message recording for per use/per activation type services (examples: Three Way Calling, Verify, Interrupt, Call Return, ETC.)
 - measured billable Local

- Directory Assistance messages
- intraLATA Toll
- WATS & 800 Service
- N11
- Information Service Provider Messages
- Operator Services Messages
- Operator Services Message Attempted Calls (UNE only)
- Credit/Cancel Records
- Usage for Voice Mail Message Service
- 6.1.2 Rated Incollects (originated in BellSouth and from other companies) can also be on Optional Daily Usage File. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.
- 6.1.3 BellSouth will perform duplicate record checks on records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to e.spire.
- 6.1.4 In the event that e.spire detects a duplicate on Optional Daily Usage File they receive from BellSouth, e.spire will drop the duplicate message (e.spire will not return the duplicate to BellSouth).

6.2 Physical File Characteristics

- 6.2.1 The Optional Daily Usage File will be distributed to e.spire via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a variable block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- 6.2.2 Data circuits (private line or dial-up) may be required between BellSouth and e.spire for the purpose of data transmission. Where a dedicated line is required, e.spire will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. e.spire will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to e.spire. Additionally, all message toll charges associated with the use of the dial circuit by e.spire will be the responsibility of e.spire. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including modems and software, that is required on e.spire end for the purpose of data transmission will be the responsibility of e.spire.

6.3 Packing Specifications

- 6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 6.3.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to e.spire which BellSouth RAO that is sending the message. BellSouth and e.spire will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by e.spire and resend the data as appropriate.

THE DATA WILL BE PACKED USING ATIS EMI RECORDS.

6.4 Pack Rejection

6.4.1 e.spire will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. e.spire will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to e.spire by BellSouth.

6.5 Control Data

e.spire will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate e.spire received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by e.spire for reasons stated in the above section.

6.6 Testing

6.6.1 Upon request from e.spire, BellSouth shall send test files to e.spire for the Optional Daily Usage File. The parties agree to review and discuss the file's content and/or format. For testing of usage results, BellSouth shall request that e.spire set up a production (LIVE) file. The live test may consist of e.spire's employees making test calls for the types of services e.spire requests on the Optional Daily Usage File. These test calls are logged by e.spire, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be completed within 30 calendar days from the date on which the initial test file was sent.

Enhanced Optional Daily Usage File

- Upon written request from e.spire, BellSouth will provide the Enhanced Optional Daily Usage File (EODUF) service to e.spire pursuant to the terms and conditions set forth in this section. EODUF will only be sent to existing ODUF subscribers who request the EODUF option.
- The e.spire shall furnish all relevant information required by BellSouth for the provision of the Enhanced Optional Daily Usage File.
- The Enhanced Optional Daily Usage File (EODUF) will provide usage data for local calls originating from resold Flat Rate Business and Residential Lines.

Charges for delivery of the Enhanced Optional Daily Usage File will appear on the e.spire's monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- 4 All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of the e.spire will be the responsibility of the e.spire. If, however, the e.spire should encounter significant volumes of errored messages that prevent processing by the e.spire within its systems, BellSouth will work with the e.spire to determine the source of the errors and the appropriate resolution.
- 6 The following specifications shall apply to the Optional Daily Usage Feed.
 - 6.1 Usage To Be Transmitted
 - 6.1.1 The following messages recorded by BellSouth will be transmitted to e.spire:

Customer usage data for flat rated local call originating from CLEC end user lines (1FB or 1FR). The EODUF record for flat rate messages will include:

Date of Call

From Number

To Number

Connect Time

Conversation Time

Method of Recording

From RAO

Rate Class

Message Type

Billing Indicators

Bill to Number

- 6.1.2 BellSouth will perform duplicate record checks on EODUF records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to e.spire.
- 6.1.3 In the event that e.spire detects a duplicate on Enhanced Optional Daily Usage File they receive from BellSouth, e.spire will drop the duplicate message (e.spire will not return the duplicate to BellSouth).

6.2 Physical File Characteristics

- 6.2.1 The Enhanced Optional Daily Usage Feed will be distributed to e.spire over their existing Optional Daily Usage File (ODUF) feed. The EODUF messages will be intermingled among e.spire's Optional Daily Usage File (ODUF) messages. The EODUF will be a variable block format (2476) with an LRECL of 2472. The data on the EODUF will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays).
- Data circuits (private line or dial-up) may be required between BellSouth and espire for the purpose of data transmission. Where a dedicated line is required, espire will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. espire will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to

attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to e.spire. Additionally, all message toll charges associated with the use of the dial circuit by e.spire will be the responsibility of e.spire. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including modems and software, that is required on e.spire end for the purpose of data transmission will be the responsibility of e.spire.

6.3 Packing Specifications

- 6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 6.3.2 The Operating Company Number (OCN), From Revenue Accounting Office (RAO), and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to e.spire which BellSouth RAO that is sending the message. BellSouth and e.spire will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by e.spire and resend the data as appropriate.

THE DATA WILL BE PACKED USING ATIS EMI RECORDS.

CALLING NAME DELIVERY (CNAM) DATABASE SERVICES

1.00 DEFINITIONS

For the purpose of this Attachment, the following terms shall be defined as:

CALLING NAME DELIVERY DATABASE SERVICE (CNAM) - The ability to associate a name with the calling party number, allowing the end user subscriber (to which a call is being terminated) to view the calling party's name before the call is answered. This service also provides e.spire the opportunity to load and store its subscriber names in the BellSouth CNAM SCPs.

CALLING PARTY NUMBER (CPN) - The number of the calling party that is delivered to the terminating switch using common channel signaling system 7 (CCS7) technology, and that is contained in the Initial Address Message (IAM) portion of the CCS7 call setup.

COMMON CHANNEL SIGNALING SYSTEM 7 (CCS7) - A network signaling

technology in which all signaling information between two or more nodes is transmitted over high-speed data links, rather than over voice circuits.

SERVICE CONTROL POINTS (SCPs) - The real-time data base systems that contain the names to be provided in response to queries received from CNAM SSPs.

SERVICE MANAGEMENT SYSTEM (SMS) - The main operations support system of CNAM DATABASE SERVICE. CNAM records are loaded into the SMS, which in turn downloads into the CNAM SCP.

SERVICE SWITCHING POINTS (SSPs) - Features of computerized switches in the telephone network that determine that a terminating line has subscribed to CNAM service, and then communicate with CNAM SCPs in order to provide the name associated with the calling party number.

SUBSYSTEM NUMBER (SSN) - The address used in the Signaling Connection Control Part (SCCP) layer of the SS7 protocol to designate an application at an end signaling point. A SSN for CNAM at the end office designates the CNAM application within the end office. BellSouth uses the CNAM SSN of 232.

2.0 ATTACHMENT

- 2.01 This Attachment contains the terms and conditions where BellSouth will provide to the e.spire access to the BellSouth CNAM SCP for query or record storage purposes.
- 2.02 e.spire shall submit to BellSouth a notice of its intent to access and utilize BellSouth CNAM Database Services pursuant to the terms and conditions of this Attachment. Said notice shall be in writing, no less than 60 days prior to e.spire's access to BellSouth's CNAM Database Services and shall be addressed to e.spire's Account Manager.

3.00 PHYSICAL CONNECTION AND COMPENSATION

- 3.01 BellSouth's provision of CNAM Database Services to e.spire requires interconnection from e.spire to BellSouth CNAM Service Control Points (SCPs). Such interconnections shall be established pursuant to Attachment 3 of this Agreement. The appropriate charge for access to and use of the BellSouth CNAM Database service shall be as set forth in this Attachment.
- 3.02 In order to formulate a CNAM query to be sent to the BellSouth CNAM SCP, e.spire_shall provide its own CNAM SSP. e.spire's CNAM SSPs must be compliant with TR-NWT-001188, "CLASS Calling Name Delivery Generic Requirements".
- 3.03 If e.spire elects to access the BellSouth CNAM SCP via a third party CCS7 transport provider, the third party CCS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia (formerly BellCore)'s CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points (LSTPs) serving the BellSouth CNAM SCPs that e.spire desires to query.

3.04 Out-Of-Region Customers. If the customer queries the BellSouth CNAM SCP via a third party national SS7 transport provider, the third party SS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and BellCore's CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish SS7 interconnection at one or more of the BellSouth Gateway Signal Transfer Points (STPs). The payment of all costs associated with the transport of SS7 signals via a third party will be established by mutual agreement of the parties and writing shall, by this reference become an integral part of this Agreement.

4.00 CNAM RECORD INITIAL LOAD AND UPDATES

- 4.01 The mechanism to be used by e.spire for initial CNAM record load and/or updates shall be determined by mutual agreement. The initial load and all updates shall be provided by e.spire in the BellSouth specified format and shall contain records for every working telephone number that can originate phone calls. It is the responsibility of e.spire to provide accurate information to BellSouth on a current basis.
- 4.02 Updates to the SMS shall occur no less than once a week, reflect service order activity affecting either name or telephone number, and involve only record additions, deletions or changes.
- 4.03 e.spire CNAM records provided for storage in the BellSouth CNAM SCP shall be available, on a SCP query basis only, to all parties querying the BellSouth CNAM SCP. Further, CNAM service shall be provided by each party consistent with state and/or federal regulation.

BELLSOUTH/e.spire RATES ODUF/EDOUF/CMDS/CNAM

		RATES BY STATE								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
ODUF/EODUF/CMDS										
ODUF: Recording, per message	N/A	\$0.0002	\$0.008	\$0.008	\$0.0008611	\$0.00019	\$0.0001179	\$0.008	\$0.0002862	\$0.008
ODUF: Message Processing, per message	N/A	\$0.0033	\$0.004	\$0.004	\$0.0032357	\$0.0024	\$0.0032089	\$0.004	\$0.0032344	\$0.004
EODUF: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
CMDS: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
ODUF: Message Processing, per magnetic tape provisioned	N/A	\$55.19	\$54.95	\$54.95	\$55.68	\$47.30	\$54.62	\$54.95	\$54.72	\$54.95
EODUF: Message Processing, per magnetic tape provisioned	N/A	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30
ODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.00004	\$0.001	\$0.001	\$0.0000365	\$0.00003	\$0.0000354	\$0.001	\$0.0000357	\$0.001
EODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364
CMDS: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001
CALLING NAME (CNAM) QUERY SERVICE										
CNAM (Database Owner), Per Query	N/A	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016
CNAM (Non-Database Owner), Per Query *	N/A	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01
NRC, applicable when e●spire uses the Character Based User Interface (CHUI) method to transmit										
the names to the BellSouth CNAM database	N/A	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00
* Volume and term arrangements are also available.										

NOTES:

If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the parties upon request by either party.

Attachment 2

Network Elements and Other Services

TABLE OF CONTENTS

1.	<u>INTRODUCTION</u>	3
2.	UNBUNDLED LOOPS	4
3.	INTEGRATED DIGITAL LOOP CARRIERS	89
4.	NETWORK INTERFACE DEVICE	99
5.	UNBUNDLED LOOP CONCENTRATION (ULC) SYSTEM	10
6.	SUB-LOOP ELEMENTS	10
7.	LOCAL SWITCHING	14
8.	TRANSPORT	20
10.	OPERATOR SYSTEMS	28
11.	<u>SIGNALING</u>	33
12.	SIGNALING TRANSFER POINTS (STPS)	35
13.	SERVICE CONTROL POINTS/DATABASES	39
13.8	CALLING NAME (CNAM) DATABASE SERVICE	44
14.	DARK FIBER	46
15.	SS7 NETWORK INTERCONNECTION	46
16.	BASIC 911 AND E911	50
17. 1.		
17.2	OPERATIONAL SUPPORT SYSTEMS (OSS)	57
1.00	DEFINITIONS	
2.0	ATTACHMENT	65
	PHYSICAL CONNECTION AND COMPENSATION	
	CNAM RECORD INITIAL LOAD AND UPDATES	
	HIBIT A – LIDB STORAGE AGREEMENT	
EXI	HIBIT B – CNAM DATABASE SERVICES	EXHIBIT B
EXI	HIBIT C – RATES	EXHIBIT C

ACCESS TO NETWORK ELEMENTS AND OTHER SERVICES

1. Introduction

- 1.1 Network Element is defined to mean a facility or equipment used in the provision of a telecommunications service. Such term may include, but is not limited to, features, functions, and capabilities that are provided by means of such facility or equipment, including but not limited to, subscriber numbers, databases, signaling systems, and information sufficient for billing and collection or used in the transmission, routing, or other provision of a telecommunications service.

 BellSouth offers access to the Network Elements, unbundled loops; network interface device; sub-loop elements; local switching; transport; tandem switching; operator systems; signaling; access to call-related databases; dark fiber as set forth in this Attachment.
- 1.2 BellSouth shall, upon request of e.spire, and to the extent technically feasible, provide to e.spire access to its network elements for the provision of e.spire's telecommunications service. If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.
- 1.3 e.spire may purchase network elements and other services from BellSouth for the purpose of combining such network elements in any manner e.spire chooses to provide telecommunication services to its intended users, including recreating existing BellSouth services. With the exception of the sub-loop elements which are located outside of the central office, BellSouth shall deliver the network elements purchased by e.spire for combining to the designated e.spire collocation space. The network elements shall be provided as set forth in this Attachment.
- 1.4 BellSouth will provide the following combined network elements for purchase by e.spire. The rate of the following combined network elements is the sum of the individual element prices as set forth in this Attachment. Order Coordination as defined in Section 2 of Attachment 2 of this Agreement is available for each of these combinations:
 - SL2 loop and cross connect
 - Port and cross connect
 - Port and cross connect and common (shared) transport
 - Port and vertical features
 - SL2 Loop with loop concentration
 - Port and common (shared) transport
 - SL2 Loop and LNP
- 1.5 BellSouth shall comply with the requirements as set forth in the technical references within Attachment 2 to the extent that they are consistent with the greater of BellSouth's actual performance or applicable industry standards.

- In the event that any final and nonappealable legislative, regulatory, judicial or other legal action modifies or redefines the "Network Elements" in a manner which materially affects the terms of this Attachment or the Network Elements and/or prices set forth herein, either Party may, on thirty (30) days written notice, require renegotiation of such terms, and the Parties shall renegotiate in good faith such new terms in accordance with such legislative, regulatory, judicial or other legal action. In the event such new terms are not renegotiated within ninety (90) days after the notice for renegotiation, either Party may petition the Commission for resolution of the dispute between the Parties. Each Party reserves the right to seek judicial review of any Commission ruling concerning this Attachment.
- 1.7 e.spire will adopt and adhere to the reasonable and nondiscriminatory standards contained in the applicable CLEC Work Center Operational Understanding Agreement regarding maintenance and installation of service.
- 1.8 BellSouth will provide reasonable and nondiscriminatory access to Network Elements on an unbundled basis, pursuant to the terms, conditions and rates set forth in this attachment, and in accordance with all effective rules and decisions of the FCC and the Commission.

2. <u>Unbundled Loops</u>

- 2.1.1 BellSouth shall provide nondiscriminatory access, in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act, to the local loop on an unbundled basis pursuant to the following terms and conditions and at the rates approved by the Commission and set forth in this Attachment.
- 2.2 Definition
- 2.2.1 The local loop network element is defined as a transmission facility between a distribution frame (or its equivalent) in an incumbent LEC central office and the loop demarcation point at an end-user customer premises, including inside wire owned by the incumbent LEC. The local loop network element includes all features, functions, and capabilities of such transmission facility. Those features, functions, and capabilities include, but are not limited to, Dark Fiber as described in Section 14 of this Attachment, attached electronics (except those electronics used for the provision of advanced services, such as Digital Subscriber Line Access Multiplexers), and loop conditioning. The local loop includes, but is not limited to, DSl, DS3, fiber, and other high capacity loops. Unless otherwise requested, all loops will be provisioned with a Network Interface Device ("NID").
- 2.2.2 The provisioning of service to a CLEC will require cross-office cabling and cross-connections within the central office to connect the loop to a local switch or to

other transmission equipment in collocation space. These cross-connects are a separate element and are not considered a part of the loop.

BellSouth Order Coordination referenced in Attachment 2 includes two types: "Order Coordination" and "Order Coordination - Time Specific."

"Order Coordination" refers to standard BellSouth service order coordination involving SL2 voice loops and all digital loops. Order coordination for physical conversions will be scheduled at BellSouth's discretion during normal working hours on the committed due date and e.spire advised.

"Order Coordination – Time Specific" refers to service order coordination in which e.spire requests a specific time for a service order conversion to take place. Loops on a single service order of 14 or more loops will be provisioned on a project basis. This is a chargeable option for any coordinated order and is billed in addition to the OC charge. e.spire may specify a time between 8:00 a.m. and 5:00 p.m. (location time) Monday through Friday (excluding holidays). If e.spire specifies a time outside this window, or selects a time or quantity of loops that requires BellSouth technicians to work outside normal work hours, overtime charges will apply in addition to the OC and OC-TS charges. Overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances.

Where facilities are available, BellSouth will install loops within a 5-7 business days interval. For orders of 14 or more loops, the installation will be handled on a project basis and the intervals will be set by the BellSouth project manager for that order. Some loops require a Service Inquiry (SI) to determine if facilities are available prior to issuing the order. The interval for the SI process is 3-5 business days and is separate from the installation interval. For expedite requests by e.spire, expedite charges will apply for intervals less than 5 days. The charges outlined in BST's FCC # 1 Tariff, Section 5.1.1, will apply. If e.spire cancels an order for network elements and other services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with FCC #1 Tariff, Section 5.4.

If e.spire modifies an order after being sent a Firm Order Confirmation (FOC) from BellSouth, any costs incurred by BellSouth to accommodate the modification will be reimbursed by e.spire.

2.2.3 Intervals for loop conversions shall be as follows: (1) for single loop conversions per location, the conversion shall be completed within fifteen (15) minutes; (2) for up to ten (10) loop conversions per location, the conversion of all loops shall be completed within sixty (60) minutes, and each individual loop conversion shall be completed within fifteen (15) minutes; (3) for loop conversions not exceeding thirty (30) loops per location and not determined complex or exceptionally large, the conversion of all loops shall be completed within one hundred and twenty (120) minutes: and (4) all loops above a thirty loop quantity, or ten (10) loop

quantity and determined as complex (a cut that requires more operation than a single cut point), will be negotiated by e.spire and BellSouth prior to the due date.

2.2.4 BellSouth will offer Unbundled Voice Loops (UVL) in two different service levels - Service Level One (SL1) and Service Level Two (SL2). SL1 loops will be non-designed, will not have test points, and will not come with any Order Coordination (OC) or engineering information/circuit make-up data. Upon issuance of an order in the service order system, SL1 loops will be activated on the due date in the same manner and time frames that BellSouth normally activates POTS-type loops for its customers. If e.spire requests work to be done for SL1s that requires BellSouth technicians to work outside normal work hours, overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances.

SL2 loops shall have test points, will be designed with a Design Layout Record provided to e.spire, and will be provided with Order Coordination. The OC feature will allow e.spire to coordinate the installation of the loop with the disconnect of an existing customer's service and/or number portability service. In these cases, BellSouth will perform the order conversion with standard order coordination at its discretion during normal work hours.

- 2.2.5 BellSouth will also offer Unbundled Digital Loops (UDL). They will be designed, will be provisioned with test points (where appropriate), and will come standard with Order Coordination and a Design Layout Record (DLR).
- 2.2.6 As a chargeable option on all loops except UVL-SL1, BellSouth will offer Order Coordination Time Specific (OC-TS). This will allow e.spire the ability to specify the time that the coordinated conversion takes place. The OC-TS charge for orders due on the same day at the same location will be applied on a per Local Service Request (LSR) basis.
- 2.2.7 e.spire will be responsible for testing and isolating troubles on the loops. Once e.spire has isolated a trouble to the BellSouth provided loop, e.spire will issue a trouble to BellSouth on the loop. BellSouth will take the actions necessary to repair the loop if a trouble actually exists. BellSouth will repair these loops in the same time frames that BellSouth repairs similarly situated loops to its customers.
- 2.2.8 If e.spire reports a trouble on SL1 loops and no trouble actually exists, BellSouth will charge e.spire for any dispatching and testing (both inside and outside the CO) required by BellSouth in order to confirm the loop's working status.
- 2.2.9 If e.spire reports a trouble on SL2 loops and no trouble actually exists, BellSouth will charge e.spire for any dispatching and testing, (outside the CO) required by BellSouth in order to confirm the loop's working status.
- 2.4 Technical Requirements

- 2.4.1 To the extent available within BST's Network at a particular location, BellSouth will offer loops capable of supporting telecommunications services such as: POTS, Centrex, basic rate ISDN, analog PBX, voice grade private line, ADSL, HDSL, DS1 and digital data (up to 64 kb/s). Additional services may include digital PBXs, primary rate ISDN, xDSL, and Nx 64 kb/s. If a requested loop type is not available, then the CLEC can use the Special Construction process to request that BellSouth place facilities or otherwise modify facilities in order to meet the CLEC's request.
- 2.4.1.1 The loop will support the transmission, signaling, performance and interface requirements of the services described in 2.3.1 above. It is recognized that the requirements of different services are different, and that a number of types or grades of loops are required to support these services. Services provided over the loop by e.spire will be consistent with industry standards and BST's TR73600.
- 2.4.1.2 In some instances, e.spire will require access to a copper twisted pair loop unfettered by any intervening equipment (e.g., filters, load coils, range extenders, etc.), so that e.spire can use the loop for a variety of services by attaching appropriate terminal equipment at the ends. e.spire will determine the type of service that will be provided over the loop. In some cases, e.spire may be required to pay additional charges for the removal of certain types of equipment. BellSouth's Special Construction process will be used to determine the costs and feasibility of these activities.

In cases in which e.spire has requested that BellSouth remove equipment from the BellSouth loop, BellSouth will no longer be expected to maintain and repair the loop to the standards specified for that loop type in the TR73600 and other standards referenced in this Agreement.

e.spire, in performance of its obligations pursuant to the preceding Section, shall maintain records that will reflect that pursuant to e.spire's request BellSouth has removed certain equipment from BellSouth provided loops and as such the loop may not perform within the technical specifications associated with that loop type. e.spire will not report to BellSouth troubles on said loops where the loops are not performing within the technical specifications of that loop type.

In addition, e.spire recognizes there may be instances where a loop modified in this manner may be subjected to normal network configuration changes that may cause the circuit characteristics to be changed and may create an outage of the service that e.spire has placed on the loop. If this occurs, BellSouth will work cooperatively with e.spire to restore the circuit to its previous modified status as quickly as possible. e.spire will pay the Time and Materials costs associated with BellSouth's work efforts needed to bring the loop back to its previous modified status.

- 2.4.2 The loop shall be provided to e.spire in accordance with the following Technical References:
 - BellSouth's TR73600, Unbundled Local Loop Technical Specification
- 2.4.2.1 Telcordia (formerly BellCore) TR-NWT-000057, Functional Criteria for Digital Loop Carrier Systems, Issue 2, January 1993.
- 2.4.2.2 Telcordia (formerly BellCore) TR-NWT-000393, Generic Requirements for ISDN Basic Access Digital Subscriber Lines.
- 2.4.2.3 ANSI T1.102 1993, American National Standard for Telecommunications Digital Hierarchy Electrical Interfaces.
- 2.4.2.4 ANSI T1.403 1989, American National Standard for Telecommunications Carrier to Customer Installation, DS1 Metallic Interface Specification.

2.5 Loop Conditioning

- 2.5.1 Subject to applicable and effective FCC rules and orders, BellSouth shall condition loops, as requested by e.spire, whether or not BellSouth offers advanced services to the End User on that loop.
- 2.5.2 Loop conditioning is defined as the removal from the loop of any devices that may diminish the capability of the loop to deliver high-speed switched wireline telecommunications capability, including xDSL service. Such devices include, but are not limited to, bridge taps, low pass filters, and range extenders.
- 2.5.3 BellSouth shall recover the cost of loop conditioning requested by e.spire at the non-recurring cost-based rates set forth in this Attachment.
- 2.5.4 To the extent technically feasible and using testing equipment that exists within BellSouth's network, BellSouth shall test and report trouble for all the features, functions, and capabilities of conditioned loops, and may not restrict testing to voice-transmission only.

3. Integrated Digital Loop Carriers

3.1.1 Where BellSouth uses Integrated Digital Loop Carrier (IDLC) systems to provide the local loop and BellSouth has a suitable alternate facility available, BellSouth will make arrangements to permit e.spire to order a contiguous local loop. To the extent it is technically feasible, these arrangements will provide e.spire with the capability to serve end users at a level that is at parity with the level of service BellSouth provides its customers. If no alternate facility is available, BellSouth will utilize its Special Construction (SC) process to determine the additional costs

required to provision the loop facilities. e.spire will then have the option of paying the one-time SC rates to place the loop facilities or e.spire may chose some other method of providing service to the end-user (e.g., Resale, private facilities, etc.).

4. Network Interface Device

4.1 Definition

4.1.1 The NID is defined as any means of interconnection of end-user customer premises wiring to BellSouth's distribution plant, such as a cross connect device used for that purpose. BellSouth shall permit e.spire to connect its own loop facilities to on-premises wiring through BellSouth's network interface device, or at any other technically feasible point.

4.2 Technical Requirements

- 4.2.1 The Network Interface Device shall provide a clean, accessible point of connection for the inside wiring and for the Distribution Media and shall maintain a connection to ground that meets the requirements set forth below.
- 4.2.2 The NID shall be capable of transferring electrical analog or digital signals between the customer's inside wiring and the Distribution Media.
- 4.2.3 All NID posts or connecting points shall be in place, secure, usable and free of any rust or corrosion. The protective ground connection shall exist and be properly installed. The ground wire will also be free of rust or corrosion and have continuity relative to ground.
- 4.2.4 The NID shall be capable of withstanding all normal local environmental variations.
- 4.2.5 Where feasible, the NID shall be physically accessible to e.spire designated personnel. In cases where entrance to the end user's premises is required to give access to the NID, e.spire shall obtain entrance permission directly from the end user.
- 4.2.6 BellSouth shall offer the NID as a stand-alone component. Additionally, e.spire may connect its loop to any spare capacity on the BellSouth NID. Where necessary to comply with an effective Commission order, BellSouth will allow e.spire to disconnect the BellSouth loop from the BellSouth NID in order to connect e.spire's loop to the BellSouth NID. In these cases, e.spire accepts all liability associated with this process and it is e.spire's responsibility to make sure the disconnected BellSouth loop is properly grounded.

4.3 Interface Requirements

- 4.3.1 The NID shall be equal to or better than all of the requirements for NIDs set forth in the following technical references:
- 4.3.1.1 Telcordia (formerly BellCore) Technical Advisory TA-TSY-000120 "Customer Premises or Network Ground Wire";
- 4.3.1.2 Telcordia (formerly BellCore) Generic Requirement GR-49-CORE "Generic Requirements for Outdoor Telephone Network Interface Devices";
- 4.3.1.3 Telcordia (formerly BellCore) Technical Requirement TR-NWT-00239 "Indoor Telephone Network Interfaces";
- 4.3.1.4 Telcordia (formerly BellCore) Technical Requirement TR-NWT-000937 "Generic Requirements for Outdoor and Indoor Building Entrance".

5. <u>Unbundled Loop Concentration (ULC) System</u>

- 5.1.1 BellSouth will provide to e.spire loop concentration (ULC). Loop concentration systems in the central office concentrate the signals transmitted over local loops onto a digital loop carrier system. The concentration device is placed inside a BellSouth central office. BellSouth will offer ULC with a TR008 interface or a TR303 interface.
- 5.1.2 ULC will be offered in two sizes. System A will allow up to 96 BellSouth loops to be concentrated onto multiple DS1s. The high speed connection from the concentrator will be at the electrical DS1 level and may connect to e.spire at e.spire's collocation site. System B will allow up to 192 BellSouth loops to be concentrated onto multiple DS1s. System A may be upgraded to a System B. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). All DS1 interfaces will terminate to the CLEC's collocation space. ULC service is offered with or without concentration and with or without protection. A Line Interface element will be required for each loop that is terminated onto the ULC system. Rates for ULC are as set forth in this Attachment.

6. <u>Sub-loop Elements</u>

Where facilities permit and subject to applicable and effective FCC rules and orders, BellSouth shall offer access to its Unbundled Sub Loop (USL), Unbundled Sub Loop Concentration (USLC) System and Unbundled Network Terminating Wire (UNTW) elements. BellSouth shall provide nondiscriminatory access, in accordance with § 51.311 and section 251(c)(3) of the Act, to the subloop on an unbundled basis pursuant to the following terms and conditions and at the rates set forth in this Attachment.

- 6.2 Unbundled Sub-Loop (USL)
- 6.2.1 Definition
- 6.2.1.1 The subloop network element is defined as any portion of the loop that is technically feasible to access at terminals in BellSouth's outside plant, including inside wire owned and controlled by BellSouth, if any. An accessible terminal is any point on the loop where technicians can access the wire or fiber within the cable without removing a splice case to reach the wire or fiber within. Such points may include, but are not limited to, the pole or pedestal, the network interface device, the minimum point of entry, the single point of interconnection, the main distribution frame, the remote terminal, and the feeder/distribution interface.
- 6.2.1.2 Technical feasibility. Subject to applicable and effective FCC rules and orders, if the Parties are unable to reach agreement, pursuant to voluntary negotiations, as to whether it is technically feasible, or whether sufficient space is available, to unbundle the subloop at the point where a carrier requests, BellSouth shall have the burden of demonstrating to the Commission, pursuant to state arbitration proceedings under section 252 of the Act, that there is not sufficient space available, or that it is not technically feasible, to unbundle the subloop at the point requested.
- 6.2.1.3. Best practices. Once any state commission has determined that it is technically feasible to unbundle subloops at a designated point, BellSouth shall have the burden of demonstrating, pursuant to state arbitration proceedings under section 252 of the Act, that it is not technically feasible, or that sufficient space is not available, to unbundle its own loops at such a point.
- 6.2.1.4. *Subloop access via collocation*. Where requested by e.spire, BellSouth shall provide access to the subloop in accordance with the FCC's collocation rules, 47 C.F.R. §§ 51.321-323.
- 6.2.1.5. Single point of interconnection. Subject to applicable and effective FCC rules and orders, BellSouth shall provide a single point of interconnection at multi-unit premises that is suitable for use by multiple carriers. This obligation is in addition to BellSouth's obligation to provide nondiscriminatory access to subloops at any technically feasible point. If the Parties are unable to negotiate terms and conditions regarding a single point of interconnection, issues in dispute, including compensation due BellSouth under forward-looking pricing principles, shall be resolved under the dispute resolution processes set forth in this Agreement.
- 6.2.2 Requirements for All Unbundled Sub-Loops
- 6.2.2.1 Unbundled Sub-Loops shall be capable of carrying all signaling messages or tones needed to provide telecommunications services.

Unbundled Sub-Loop shall support functions associated with provisioning, maintenance and testing of the Unbundled Sub-Loop. In these scenarios, e.spire would be required to place a cross-box, remote terminal (RT), or other similar device and deliver a cable to the BellSouth remote terminal or cross-box. This cable would be connected, by a BST technician, to a cross-connect panel within the BellSouth RT/cross-box. e.spire's cable pairs can then be connected to BST's USL within the BST cross-box by the BST technician.

- 6.2.3 Interface Requirements
- 6.2.3.1 Unbundled Sub-Loop shall be equal to or better than each of the applicable interface requirements set forth in the following technical references:
- 6.2.3.2 Telcordia (formerly BellCore) TR-NWT-000049, "Generic Requirements for Outdoor Telephone Network Interface Devices," Issued December 1,1994.
- 6.3 Unbundled Sub-Loop Concentration System (USLC)
- 6.3.1 BellSouth will provide e.spire with the ability to concentrate its sub-loops onto multiple DS1s back to the BellSouth Central Office. The DS1s will then be terminated into e.spire's collocation space. TR-008 and TR303 interface standards are available.
- USLC, using the Lucent Series 5 equipment, will be offered in two different systems. System A will allow up to 96 of e.spire's sub-loops to be concentrated onto multiple DS1s. System B will allow an additional 96 of e.spire's sub-loops to be concentrated onto multiple DS1s. One System A may be supplemented with one System B and they both must be physically located in a single Series 5 dual channel bank. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). The DS1 level facility that connects the RT site with the serving wire center is known as a Feeder Interface. All DS1 Feeder Interfaces will terminate to the CLEC's collocation space within the SWC that serves the RT where the CLEC's sub-loops are connected. USLC service is offered with or without concentration and with or without a protection DS1.
- In these scenarios e.spire would be required to place a cross-box, remote terminal (RT), or other similar device and deliver a cable to the BellSouth remote terminal. This cable would be connected, by a BellSouth technician, to a cross-connect panel within the BellSouth RT/cross-box and would allow e.spire's sub-loops to then be placed on the ULSC and transported to their collocation space at a DS1 level.
- 6.4 Unbundled Network Terminating Wire (UNTW)

6.4.1 BellSouth will provide e.spire with access to its Unbundled Network Terminating Wire (UNTW) pursuant to the following terms and conditions at rates as set forth in this Attachment, and in a manner consistent with applicable and effective FCC rules and decisions, including, but not limited to 47 C.F.R § 51.319.

6.5 Definition

UNTW is twisted copper wire that extends from BellSouth's point-of-entry into a multi-dwelling unit (MDU) complex or multi-tenant unit (MTU) complex to the point of demarcation at the end-users location. The UNTW will not include a Network Interface Device (NID).

6.6 Requirements

- 6.6.1 BellSouth will offer spare pairs that are available to an end-users premises to e.spire. Available spare pairs are defined as pairs that are not being utilized by BellSouth or by a third party to provide an End User with working service at the time of e.spire's request for UNTW. If BellSouth has relinquished the first pair to e.spire and the End User decides to change local service providers to BellSouth, e.spire will relinquish the first pair back to BellSouth.
- Notwithstanding the foregoing, should BellSouth subsequently require the use of additional pair(s) to provide for the activation of additional lines in an End User's premises in response to a request from such End User and no additional pairs are available, e.spire agrees to surrender its spare pair(s) upon request by BellSouth.
- 6.6.3 If an End User of e.spire desires to receive local exchange service from a service provider who is not a Party to this Agreement, and such third party service provider needs access to the BellSouth UNTW to provide local exchange service to the End User, then e.spire agrees to surrender the requisite number of its inactive spare pair(s) if no other spare pair is available and upon request by BellSouth.
- 6.6.4 If e.spire has placed NTW at a location and an End User desires to receive local exchange service from BellSouth and BellSouth needs access to e.spire's NTW to provide local exchange service to the End User, then e.spire agrees to surrender the requisite number of its spare pair(s) upon request by BellSouth.
- In new construction, where possible, both Parties may at their option and with the property owner's agreement install their own NTW. In existing construction, BellSouth shall not be required to install new or additional NTW beyond existing NTW to provision the services of e.spire.

6.7 <u>Technical Requirements</u>

6.7.1 In these scenarios, BellSouth will connect the requested UNTW pairs to a cross-connect panel designed for CLEC access to BellSouth's NTW. e.spire will be required to place a cross-box, terminal, or other similar device and deliver a cable to this cross-connect panel. e.spire will then connect their cable to the cross-connect panel to access the requested UNTW pairs.

7. Switching

BellSouth agrees to offer access to local switching pursuant to the following terms and conditions and at the rates set forth in this Attachment.

7.1 Definition

- 7.1.1 Subject to applicable and effective FCC rules and orders, BellSouth shall provide nondiscriminatory access, in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act, to local circuit switching capability and local tandem switching capability on an unbundled basis, except as set forth in FCC Rule 51.319(c)(1)(B), to e.spire for the provision of a telecommunications service. Subject to applicable and effective FCC rules and orders, BellSouth shall be required to provide nondiscriminatory access in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act to packet switching capability on an unbundled basis to e.spire for the provision of a telecommunications service as described in Rule 51.319(c)(3)(B).
- 7.1.1.1 Local Circuit Switching Capability, including Tandem Switching Capability. The local circuit switching capability network element is defined as:
 - 1. Line-side facilities, which include, but are not limited to, the connection between a loop termination at a main distribution frame and a switch line card;
 - 2. Trunk-side facilities, which include, but are not limited to, the connection between trunk termination at a trunk-side cross-connect panel and a switch trunk card; and
 - 3. All features, functions and capabilities of the switch, which include, but are not limited to:
 - a. The basic switching function of connecting lines to lines, lines to trunks, trunks to lines, and trunks to trunks, as well as the same basic capabilities made available to the incumbent LEC's customers, such as a telephone number, white page listing and dial tone, and
 - b. All other features that the switch is capable of providing, including but not limited to, customer calling, customer local area signaling service features, and Centrex, as well as any technically feasible customized routing functions provided by the switch.

- 7.1.1.2 Subject to applicable and effective FCC rules and orders, notwithstanding BellSouth's general duty to unbundle local circuit switching, BellSouth shall not be required to unbundle local circuit switching for e.spire in cases where e.spire intends to use such facilities to serve End Users with four or more voice grade (DS0) equivalents or lines, BellSouth provides nondiscriminatory, unrestricted, cost-based access to the enhanced extended link ("EEL") throughout Density Zone 1, and BellSouth's local circuit switches are located in:
 - 1. The top 50 Metropolitan Statistical Areas as set forth in Appendix B of the *Third Report and Order and Fourth Further Notice of Proposed Rulemaking* in CC Docket No. 96-98, and
 - 2. Density Zone I, as defined in FCC Rule 69.123, as of January 1, 1999.
- 7.1.1.3 Local Tandem Switching Capability. The tandem switching capability network element is defined as:
 - (A) Trunk-connect facilities, which include, but are not limited to, the connection between trunk termination at a cross connect panel and switch trunk card;
 - (B) The basic switch trunk function of connecting trunks to trunks; and
 - (C) The functions that are centralized in tandem switches (as distinguished from separate end office switches), including but not limited, to call recording, the routing of calls to operator services, and signaling conversion features.

Tandem Switching is the function that establishes a communications path between two switching offices through a third switching office (the Tandem switch).

- 7.1.1.4 Packet Switching Capability. The packet switching capability network element is defined as the basic packet switching function of routing or forwarding packets, frames, cells or other data units based on address or other routing information contained in the packets, frames, cells or other data units, and the functions that are performed by Digital Subscriber Line Access Multiplexers, including but not limited to:
 - The ability to terminate copper customer loops (which includes both a low band voice channel and a high-band data channel, or solely a data channel);
 - 2. The ability to forward the voice channels, if present, to a circuit switch or multiple circuit switches;
 - 3. The ability to extract data units from the data channels on the loops, and
 - 4. The ability to combine data units from multiple loops onto one or more trunks connecting to a packet switch or packet switches.

- 7.1.1.5 Subject to applicable and effective FCC rules and orders, BellSouth shall provide nondiscriminatory access to unbundled packet switching capability only in cases where each of the following conditions are satisfied:
 - 1. BellSouth has deployed digital loop carrier systems, including but not limited to, integrated digital loop carrier or universal digital loop carrier systems; or has deployed any other system in which fiber optic facilities replace copper facilities in the distribution section (*e.g.*, end office to remote terminal, pedestal or environmentally controlled vault);
 - 2. There are no spare copper loops capable of supporting the xDSL services the requesting carrier seeks to offer;
 - 3. BellSouth has not permitted a requesting carrier to deploy a Digital Subscriber Line Access Multiplexer at the remote terminal, pedestal or environmentally controlled vault or other interconnection point, nor has the requesting carrier obtained a virtual collocation arrangement at these subloop interconnection points as defined by FCC Rule 51.319(b); and
 - 4. BellSouth has deployed packet switching capability for its own use.
- 7.1.2 A featureless port is one that has a line port, switching functionality, and an interoffice port. A featured port is a port that includes all features then capable or a number of then capable features specifically requested by e.spire. Any features that are not currently then capable but are technically feasible through the switch can be requested through the BFR process.
- Where required to do so in order to comply with an effective Commission order, BellSouth will provide to e.spire purchasing local BellSouth switching and reselling BellSouth local exchange service under Attachment 1, selective routing of calls to a requested directory assistance services platform or operator services platform. e.spire customers may use the same dialing arrangements as BellSouth customers, but obtain a e.spire branded service.
- 7.2 <u>Technical Requirements</u>
- 7.2.1 The requirements set forth in this Section apply to Local Switching, but not to the Packet Switching function of Local Switching.
- 7.2.1.1 Local Switching shall be equal to or better than the requirements for Local Switching set forth in Telcordia (formerly BellCore)'s Local Switching Systems General Requirements (FR-NWT-000064).
- 7.2.1.2 When applicable, BellSouth shall route calls to the appropriate trunk or lines for call origination or termination.

- 7.2.1.3 Subject to this section, BellSouth shall route calls on a per line or per screening class basis to (1) BellSouth platforms providing Network Elements or additional requirements (2) Operator Services platforms, (3) Directory Assistance platforms, and (4) Repair Centers. Any other routing requests by e.spire will be made pursuant to the Bona Fide Request/ New Business Request Process as set forth in General Terms and Conditions.
- 7.2.1.4 BellSouth shall provide unbranded recorded announcements and call progress tones to alert callers of call progress and disposition.
- 7.2.1.5 BellSouth shall activate service for an e.spire customer or network interconnection on any of the Local Switching interfaces. This includes provisioning changes to change a customer from BellSouth's services to e.spire's services without loss of switch feature functionality as defined in this Agreement.
- 7.2.1.6 BellSouth shall perform routine testing (e.g., Mechanized Loop Tests (MLT) and test calls such as 105, 107 and 108 type calls) and fault isolation on a mutually agreed upon schedule.
- 7.2.1.7 BellSouth shall repair and restore any equipment or any other maintainable component that may adversely impact Local Switching.
- 7.2.1.8 BellSouth shall control congestion points such as those caused by radio station call-ins, and network routing abnormalities. All traffic shall be restricted in a non discriminatory manner.
- 7.2.1.9 BellSouth shall perform manual call trace and permit customer originated call trace.
- 7.2.1.10 Special Services provided by BellSouth will include the following:
- 7.2.1.10.1 Telephone Service Prioritization;
- 7.2.1.10.2 Related services for handicapped;
- 7.2.1.10.3 Soft dial tone where required by law; and
- 7.2.1.10.4 Any other service required by law.
- 7.2.1.11 BellSouth shall provide Switching Service Point (SSP) capabilities and signaling software to interconnect the signaling links destined to the Signaling Transfer Point Switch (STP). These capabilities shall adhere to Telcordia (formerly BellCore) specifications TCAP (GR-1432-CORE), ISUP(GR-905-CORE), Call Management (GR-1429-CORE), Switched Fractional DS1 (GR-1357-CORE), Toll Free Service (GR-1428-CORE), Calling Name (GR-1597-CORE), Line Information Database (GR-954-CORE), and Advanced Intelligent Network (GR-2863-CORE).

7.2.1.12 BellSouth shall provide interfaces to adjuncts through Telcordia (formerly BellCore) standard interfaces. These adjuncts can include, but are not limited to, the Service Circuit Node and Automatic Call Distributors. 7.2.1.13 BellSouth shall provide performance data regarding a customer line, traffic characteristics or other measurable elements to e.spire, upon a reasonable request from e.spire. CLEC will pay BellSouth for all costs incurred to provide such performance data through the Business Opportunity Request process. 7.2.1.14 BellSouth shall offer Local Switching that provides feature offerings at parity to those provided by BellSouth to itself or any other Party. Such feature offerings shall include but are not limited to: 7.2.1.14.1 Basic and primary rate ISDN; 7.2.1.14.2 Residential features; 7.2.1.14.3 Customer Local Area Signaling Services (CLASS/LASS); 7.2.1.14.4 CENTREX (including equivalent administrative capabilities, such as customer accessible reconfiguration and detailed message recording); and 7.2.1.14.5 Advanced intelligent network triggers supporting e.spire and BellSouth service applications. BellSouth shall offer to e.spire all AIN triggers in connection with its SMS/SCE offering which are supported by BellSouth for offering AIN-based services. Triggers that are currently available are: Off-Hook Immediate 7.2.1.14.5.1 7.2.1.14.5.2 Off-Hook Delay 7.2.1.14.5.3 **Termination Attempt** 7.2.1.14.5.4 6/10 Public Office Dialing Plan 7.2.1.14.5.5 Feature Code Dialing 7.2.1.14.5.6 Customer Dialing Plan 7.2.1.14.6 When the following triggers are supported by BellSouth, BellSouth will make these triggers available to e.spire: Private EAMF Trunk 7.2.1.14.6.1 7.2.1.14.6.2 Shared Interoffice Trunk (EAMF, SS7) 7.2.1.14.6.3 N11

7.2.1.14.6.4 **Automatic Route Selection** 7.2.1.15 Where capacity exists, BellSouth shall assign each e.spire customer line the class of service designated by e.spire (e.g., using line class codes or other switch specific provisioning methods), and shall route directory assistance calls from e.spire customers to e.spire directory assistance operators at e.spire's option. 7.2.1.16 Where capacity exists, BellSouth shall assign each e.spire customer line the class of services designated by e.spire (e.g., using line class codes or other switch specific provisioning methods) and shall route operator calls from e.spire customers to e.spire operators at e.spire's option. For example, BellSouth may translate 0- and 0+ intraLATA traffic, and route the call through appropriate trunks to an e.spire Operator Services Position System (OSPS). Calls from Local Switching must pass the ANI-II digits unchanged. 7.2.1.17 Local Switching shall be offered in accordance with the requirements of the following technical references: 7.2.1.17.1 Telcordia (formerly BellCore) GR-1298-CORE, AIN Switching System Generic Requirements, as implemented in BellSouth's switching equipment; 7.2.1.17.2 Telcordia (formerly BellCore) GR-1299-CORE, AIN Switch-Service Control Point (SCP)/Adjunct Interface Generic Requirements; 7.2.1.17.3 Telcordia (formerly BellCore) TR-NWT-001284, AIN 0.1 Switching System Generic Requirements; 7.2.1.17.4 Telcordia (formerly BellCore) SR-NWT-002247, AIN Release 1 Update. 7.2.2 **Interface Requirements** 7.2.2.1 BellSouth shall provide the following interfaces to loops: 7.2.2.2 Standard Tip/Ring interface including loopstart or groundstart, on-hook signaling (e.g., for calling number, calling name and message waiting lamp); 7.2.2.3 Coin phone signaling; 7.2.2.4 Basic Rate Interface ISDN adhering to appropriate Telcordia (formerly BellCore) Technical Requirements: 7.2.2.5 Two-wire analog interface to PBX; 7.2.2.5.1 Four-wire analog interface to PBX; 7.2.2.6 Four-wire DS1 interface to PBX or customer provided equipment (e.g. computers and voice response systems);

- 7.2.2.7 Primary Rate ISDN to PBX adhering to ANSI standards Q.931, Q.932 and appropriate Telcordia (formerly BellCore) Technical Requirements;
- 7.2.2.8 Switched Fractional DS1 with capabilities to configure Nx64 channels (where N = 1 to 24); and
- 7.2.2.9 Loops adhering to Telcordia (formerly BellCore) TR-NWT-08 and TR-NWT-303 specifications to interconnect Digital Loop Carriers.
- 7.2.2.10 BellSouth shall provide access to the following but not limited to:
- 7.2.2.11 SS7 Signaling Network or Multi-Frequency trunking if requested by e.spire;
- 7.2.2.12 Interface to e.spire operator services systems or Operator Services through appropriate trunk interconnections for the system; and
- 7.2.2.13 Interface to e.spire directory assistance services through the e.spire switched network or to Directory Assistance Services through the appropriate trunk interconnections for the system; and 950 access or other e.spire required access to interexchange carriers as requested through appropriate trunk interfaces.

8. Interoffice Transmission Facilities

BellSouth shall provide nondiscriminatory access, in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act, to interoffice transmission facilities on an unbundled basis to espire for the provision of a telecommunications service.

- 8.1 Interoffice transmission facility network elements include:
 - Dedicated transport, defined as BellSouth's transmission facilities, including all technically feasible capacity-related services including, but not limited to, DS1, DS3 and OCn levels, dedicated to a particular customer or carrier, that provide telecommunications between wire centers or switches owned by BellSouth, or between wire centers and switches owned by BellSouth and e.spire;
 - 2. Dark Fiber transport, defined as BellSouth's optical transmission facilities without attached multiplexing, aggregation or other electronics;
 - 3. Shared transport, defined as transmission facilities shared by more than one carrier, including BellSouth, between end office switches, between end office switches and tandem switches, and between tandem switches, in BellSouth's network..
- 8.1.1 BellSouth shall:

- 1. Provide e.spire exclusive use of interoffice transmission facilities dedicated to a particular customer or carrier, or shared use of the features, functions, and capabilities of interoffice transmission facilities shared by more than one customer or carrier;
- 2. Provide all technically feasible transmission facilities, features, functions, and capabilities that e.spire could use to provide telecommunications services;
- 3. Permit, to the extent technically feasible, e.spire to connect such interoffice facilities to equipment designated by e.spire, including but not limited to, e.spire's collocated facilities; and
- 4. Permit, to the extent technically feasible, e.spire to obtain the functionality provided by BellSouth's digital cross-connect systems in the same manner that BellSouth provides such functionality to interexchange carriers.

8.2 <u>Technical Requirements of Common (Shared) Transport</u>

- 8.2.1 Common (Shared) Transport provided on DS1 or VT1.5 circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Central Office to Central Office ("CO to CO") connections in the appropriate industry standards.
- 8.2.2 Common (Shared) Transport provided on DS3 circuits, STS-1 circuits, and higher transmission bit rate circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CO to CO connections in the appropriate industry standards.
- 8.2.3 BellSouth shall be responsible for the engineering, provisioning, and maintenance of the underlying equipment and facilities that are used to provide Common (Shared) Transport.
- 8.2.4 At a minimum, Common (Shared) Transport shall meet all of the requirements set forth in the following technical references (as applicable for the transport technology being used):
- 8.2.4.1 ANSI T1.101-1994, American National Standard for Telecommunications Synchronization Interface Standard Performance and Availability;
- 8.2.4.2 ANSI T1.102-1993, American National Standard for Telecommunications Digital Hierarchy Electrical Interfaces;
- 8.2.4.3 ANSI T1.102.01-199x, American National Standard for Telecommunications Digital Hierarchy VT1.5;

8.2.4.4 ANSI T1.105-1995, American National Standard for Telecommunications -Synchronous Optical Network (SONET) - Basic Description including Multiplex Structure, Rates and Formats; ANSI T1.105.01-1995, American National Standard for Telecommunications -8.2.4.5 Synchronous Optical Network (SONET) - Automatic Protection Switching; 8.2.4.6 ANSI T1.105.02-1995, American National Standard for Telecommunications -Synchronous Optical Network (SONET) - Payload Mappings; 8.2.4.7 ANSI T1.105.03-1994, American National Standard for Telecommunications -Synchronous Optical Network (SONET) - Jitter at Network Interfaces; 8.2.4.8 ANSI T1.105.03a-1995, American National Standard for Telecommunications -Synchronous Optical Network (SONET): Jitter at Network Interfaces - DS1 Supplement; 8.2.4.9 ANSI T1.105.05-1994, American National Standard for Telecommunications -Synchronous Optical Network (SONET) - Tandem Connection; 8.2.4.10 ANSI T1.105.06-199x, American National Standard for Telecommunications -Synchronous Optical Network (SONET) - Physical Layer Specifications; 8.2.4.11 ANSI T1.105.07-199x, American National Standard for Telecommunications -Synchronous Optical Network (SONET) - Sub STS-1 Interface Rates and Formats: 8.2.4.12 ANSI T1.105.09-199x, American National Standard for Telecommunications -Synchronous Optical Network (SONET) - Network Element Timing and Synchronization; 8.2.4.13 ANSI T1.106-1988, American National Standard for Telecommunications -Digital Hierarchy - Optical Interface Specifications (Single Mode); 8.2.4.14 ANSI T1.107-1988, American National Standard for Telecommunications -Digital Hierarchy - Formats Specifications; 8.2.4.15 ANSI T1.107a-1990 - American National Standard for Telecommunications -Digital Hierarchy - Supplement to Formats Specifications (DS3 Format Applications); 8.2.4.16 ANSI T1.107b-1991 - American National Standard for Telecommunications -Digital Hierarchy - Supplement to Formats Specifications; 8.2.4.17 ANSI T1.117-1991, American National Standard for Telecommunications -Digital Hierarchy - Optical Interface Specifications (SONET) (Single Mode -Short Reach);

8.2.4.18 ANSI T1.403-1989, Carrier to Customer Installation, DS1 Metallic Interface Specification; 8.2.4.19 ANSI T1.404-1994, Network-to-Customer Installation - DS3 Metallic Interface Specification; 8.2.4.20 ITU Recommendation G.707, Network node interface for the synchronous digital hierarchy (SDH); 8.2.4.21 ITU Recommendation G.704, Synchronous frame structures used at 1544, 6312, 2048, 8488 and 44736 kbit/s hierarchical levels; 8.2.4.22 Telcordia (formerly BellCore) FR-440 and TR-NWT-000499, Transport Systems Generic Requirements (TSGR): Common Requirements; 8.2.4.23 Telcordia (formerly BellCore) GR-820-CORE, Generic Transmission Surveillance: DS1 & DS3 Performance; 8.2.4.24 Telcordia (formerly BellCore) GR-253-CORE, Synchronous Optical Network Systems (SONET); Common Generic Criteria; 8.2.4.25 Telcordia (formerly BellCore) TR-NWT 000507, Transmission, Section 7, Issue 5 (Telcordia (formerly BellCore), December 1993). (A module of LSSGR, FR-NWT-000064.); Telcordia (formerly BellCore) TR-NWT-000776, Network Interface Description 8.2.4.26 for ISDN Customer Access; 8.2.4.27 Telcordia (formerly BellCore) TR-INS-000342, High-Capacity Digital Special Access Service-Transmission Parameter Limits and Interface Combinations, Issue 1 February 1991; 8.2.4.28 Telcordia (formerly BellCore) ST-TEC 000052, Telecommunications Transmission Engineering Textbook, Volume 2: Facilities, Third Edition, Issue I May 1989; 8.2.4.29 Telcordia (formerly BellCore) ST-TEC-000051, Telecommunications Transmission Engineering Textbook Volume 1: Principles, Third Edition. Issue 1 August 1987. 8.3 **Dedicated Transport** 8.3.1. BellSouth shall offer Dedicated Transport in each of the following ways: 8.3.1.1 As capacity on a shared facility. 8.3.1.2 As a circuit (e.g., DS0, DS1 or DS3) dedicated to e.spire.

- When Dedicated Transport is provided as a system it shall include:
- 8.3.2.1 Transmission equipment such as multiplexers, line terminating equipment, amplifiers, and regenerators;
- 8.3.2.2 Inter-office transmission facilities such as optical fiber, copper twisted pair, and coaxial cable.
- 8.3.3 Unbundled Local Channel
- 8.3.3.1 The Unbundled Local Channel is the dedicated transmission path between e.spire's Point of Presence and the BellSouth Serving Wire Center.
- 8.3.3.2 BellSouth currently offers Unbundled Local Channels for switched traffic. Rates for these elements are listed in this Attachment. For those states that do not contain rates in this Attachment for DS1 and DS3 switched Local Channels, the rates in the applicable State Access Tariff will apply as interim rates. When final rates are developed, these interim rates will be subject to true-up, and the Parties will amend the Agreement to reflect the new rates.
- 8.3.3.3 BellSouth currently offers Unbundled Local Channels for non-switched traffic at DS1 and DS3 levels at rates as set forth in Exhibit C to this Attachment.
- 8.3.4 <u>Technical Requirements</u>

This Section sets forth technical requirements for all Dedicated Transport.

- 8.3.4.1 When BellSouth provides Dedicated Transport as a circuit or a system, the entire designated transmission circuit or system (*e.g.*, DS0, DS1,DS3) shall be dedicated to e.spire designated traffic.
- 8.3.4.2 BellSouth shall offer Dedicated Transport in all technologies that become available including, but not limited to, DS1 and DS3 transport systems, SONET (or SDH) Bi-directional Line Switched Rings, SONET (or SDH) Unidirectional Path Switched Rings, and SONET (or SDH) point-to-point transport systems (including linear add-drop systems), at all available transmission bit rates. While SONET Ring facilities are not available in every application, they are typically available in the major metropolitan areas.
- 8.3.4.3 For DS1 or VT1.5 circuits, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Customer Interface to Central Office ("CI to CO") connections in the appropriate industry standards.
- Where applicable, for DS3 circuits, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the appropriate industry standards.

8.3.4.5	BellSouth shall offer the following interface transmission rates for Dedicated Transport:
8.3.4.5.1	DS0 Equivalent;
8.3.4.5.2	DS1 (Extended SuperFrame - ESF and D4 channel bank shall be provided);
8.3.4.5.3	DS3 where applicable (M13 multiplexer shall be provided);
8.3.4.5.4	SDH Standard interface rates in accordance with International Telecommunications Union (ITU) Recommendation G.707 and Plesiochronous Digital Hierarchy (PDH) rates per ITU Recommendation G.704.
8.3.4.6	When Dedicated Transport is provided as a system, BellSouth shall design the system according to our network infrastructure to allow for the termination points specified by e.spire.
8.3.5	At a minimum, Dedicated Transport shall meet each of the requirements set forth in the following technical references:
8.3.5.1	ANSI T1.231-1993 - American National Standard for Telecommunications - Digital Hierarchy - Layer 1 In-Service Digital Transmission performance monitoring.
8.3.5.1.1	ANSI T1.102-1993, American National Standard for Telecommunications - Digital Hierarchy - Electrical Interfaces;
8.3.5.1.2	ANSI T1.106-1988, American National Standard for Telecommunications - Digital Hierarchy - Optical Interface Specifications (Single Mode);
8.3.5.1.3	ANSI T1.107-1988, American National Standard for Telecommunications - Digital Hierarchy - Formats Specifications;
8.3.5.1.4	ANSI T1.107a-1990 - American National Standard for Telecommunications - Digital Hierarchy - Supplement to Formats Specifications (DS3 Format Applications);
8.3.5.1.5	ANSI T1.107b-1991 - American National Standard for Telecommunications - Digital Hierarchy - Supplement to Formats Specifications;
8.3.5.1.6	Telcordia (formerly BellCore) FR-440 and TR-NWT-000499, Transport Systems Generic Requirements (TSGR): Common Requirements;
8.3.5.1.7	Telcordia (formerly BellCore) GR-820-CORE, Generic Transmission Surveillance: DS1 & DS3 Performance;
8.3.5.1.8	Telcordia (formerly BellCore) TR-NWT 000507, Transmission, Section 7, Issue 5 (Telcordia (formerly BellCore), December 1993). (A module of LSSGR, FR-NWT-000064.);

- 8.3.5.1.9 Telcordia (formerly BellCore) TR-INS-000342, High-Capacity Digital Special Access Service-Transmission Parameter Limits and Interface Combinations, Issue 1 February 1991;
- 8.3.5.1.10 Telcordia (formerly BellCore) ST-TEC 000052, Telecommunications
 Transmission Engineering Textbook, Volume 2: Facilities, Third Edition, Issue I
 May 1989;
- 8.3.5.1.11 Telcordia (formerly BellCore) ST-TEC-000051, Telecommunications
 Transmission Engineering Textbook Volume 1: Principles, Third Edition. Issue 1
 August 1987;

9. Tandem Switching

- 9.1 Tandem Switching is as defined in Section 7.1.1.3 of this Attachment.
- 9.2 Technical Requirements
- 9.2.1 Tandem Switching shall have the same capabilities or equivalent capabilities as those described in Bell Communications Research TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90. The requirements for Tandem Switching include, but are not limited to the following:
- 9.2.1.1 Tandem Switching shall provide signaling to establish a tandem connection;
- 9.2.1.2 Tandem Switching will provide screening as jointly agreed to by e.spire and BellSouth;
- 9.2.1.3 Tandem Switching shall provide Advanced Intelligent Network triggers supporting AIN features where such routing is not available from the originating end office switch, to the extent such Tandem switch has such capability;
- 9.2.1.4 Tandem Switching shall provide access to Toll Free number portability database as designated by e.spire;
- 9.2.1.5 Tandem Switching shall provide all trunk interconnections discussed under the "Network Interconnection" section (e.g., SS7, MF, DTMF, DialPulse, PRI-ISDN, DID, and CAMA-ANI (if appropriate for 911));
- 9.2.1.6 Tandem Switching shall provide connectivity to PSAPs where 911 solutions are deployed and the tandem is used for 911; and
- 9.2.1.7 Where appropriate, Tandem Switching shall provide connectivity to transit traffic to and from other carriers.
- 9.2.2 Tandem Switching shall accept connections (including the necessary signaling and trunking interconnections) between end offices, other tandems, IXCs, ICOs, CAPs and CLEC switches.

9.2.3 Tandem Switching shall provide local tandeming functionality between two end offices including two offices belonging to different CLEC's (e.g., between a CLEC end office and the end office of another CLEC). 9.2.4 Tandem Switching shall preserve CLASS/LASS features and Caller ID as traffic is processed. 9.2.5 Tandem Switching shall record billable events and send them to the area billing centers designated by e.spire. Tandem Switching will provide recording of all billable events as jointly agreed to by e.spire and BellSouth. 9.2.6 Upon a reasonable request from e.spire, BellSouth shall perform routine testing and fault isolation on the underlying switch that is providing Tandem Switching and all its interconnections. The results and reports of the testing shall be made immediately available to e.spire. 9.2.7 BellSouth shall maintain e.spire's trunks and interconnections associated with Tandem Switching at least at parity to its own trunks and interconnections. 9.2.8 BellSouth shall control congestion points and network abnormalities. All traffic will be restricted in a non discriminatory manner. 9.2.9 Selective Call Routing through the use of line class codes is not available through the use of tandem switching. Selective Call Routing through the use of line class codes is an end office capability only. Detailed primary and overflow routing plans for all interfaces available within BellSouth switching network shall be mutually agreed to by e.spire and BellSouth. 9.2.10 Tandem Switching shall process originating toll-free traffic received from e.spire local switch. 9.2.11 In support of AIN triggers and features, Tandem Switching shall provide SSP capabilities when these capabilities are not available from the Local Switching Network Element, to the extent such Tandem Switch has such capability. 9.3 **Interface Requirements** 9.3.1 Tandem Switching shall provide interconnection to the E911 PSAP where the underlying Tandem is acting as the E911 Tandem. 9.3.2 Tandem Switching shall interconnect, with direct trunks, to all carriers with which BellSouth interconnects. 9.3.3 BellSouth shall provide all signaling necessary to provide Tandem Switching with no loss of feature functionality. 9.3.4 Tandem Switching shall interconnect with e.spire's switch, using two-way trunks, for traffic that is transiting via BellSouth network to interLATA or intraLATA

carriers. At e.spire's request, Tandem Switching shall record and keep records of traffic for billing.

- 9.3.5 Tandem Switching shall provide an alternate final routing pattern for e.spire traffic overflowing from direct end office high usage trunk groups.
- 9.4 Tandem Switching shall meet or exceed (i.e., be more favorable to e.spire) each of the requirements for Tandem Switching set forth in the following technical references:
- 9.4.1 Bell Communications Research TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90;
- 9.4.2 GR-905-CORE covering CCSNIS;
- 9.4.3 GR-1429-CORE for call management features; and GR-2863-CORE and Telcordia (formerly BellCore) GR-2902-CORE covering CCS AIN interconnection

10. Operator Systems

BellSouth agrees to offer access to operator systems pursuant to the terms and conditions following and at the rates set forth in this Attachment.

10.1 Definition

Operator Systems is the Network Element that provides operator and automated call handling and billing, special services, end user telephone listings and optional call completion services. The Operator Systems, Network Element provides two types of functions: Operator Service functions and Directory Assistance Service functions, each of which are described in detail below.

10.2 <u>Operator Service</u>

10.2.1 Definition

Operator Service provides: (1) operator handling for call completion (for example, collect, third number billing, and manual credit card calls), (2) operator or automated assistance for billing after the end user has dialed the called number (for example, credit card calls); and (3) special services including but not limited to Busy Line Verification and Emergency Line Interrupt (ELI), Emergency Agency Call, Operator-assisted Directory Assistance, and Rate Quotes.

11.2.2 Requirements

10.2.2.1 When e.spire requests BellSouth to provide Operator Services, the following requirements apply:

10.2.2.1.1	BellSouth shall complete 0+ and 0- dialed local calls.
10.2.2.1.2	BellSouth shall complete 0+ intraLATA toll calls.
10.2.2.1.3	BellSouth shall complete calls that are billed to e.spire end user's calling card that can be validated by BellSouth.
10.2.2.1.4	BellSouth shall complete person-to-person calls.
10.2.2.1.5	BellSouth shall complete collect calls.
10.2.2.1.6	BellSouth shall provide the capability for callers to bill to a third party and complete such calls.
10.2.2.1.7	BellSouth shall complete station-to-station calls.
10.2.2.1.8	BellSouth shall process emergency calls.
10.2.2.1.9	BellSouth shall process Busy Line Verify and Emergency Line Interrupt requests.
10.2.2.1.10	BellSouth shall process emergency call trace, as they do for their End users prior to the Effective Date. Call must originate from a 911 provider.
10.2.2.1.11	BellSouth shall process operator-assisted directory assistance calls.
10.2.2.2	BellSouth shall adhere to equal access requirements, providing e.spire local end users the same IXC access as provided to BellSouth end users.
10.2.2.3	BellSouth shall exercise at least the same level of fraud control in providing Operator Service to e.spire that BellSouth provides for its own operator service.
10.2.2.4	BellSouth shall perform Billed Number Screening when handling Collect, Person to-Person, and Billed-to-Third-Party calls.
10.2.2.5	BellSouth shall direct customer account and other similar inquiries to the customer service center designated by e.spire.
10.2.2.6	BellSouth shall provide a feed of customer call records in "EMI" format to e.spire in accordance with CLECODUF standards specified in Attachment 7.
10.2.3	Interface Requirements
	With respect to Operator Services for calls that originate on local switching capability provided by or on behalf of e.spire, the interface requirements shall conform to the then current established system interface specifications for the platform used to provide Operator Service and the interface shall conform to industry standards.

10.3 <u>Directory Assistance Service</u>

10.3.1 Definition

Directory Assistance Service provides local end user telephone number listings with the option to complete the call at the callers direction separate and distinct from local switching.

10.3.2 Requirements

10.3.2.1 Directory Assistance Service shall provide up to two listing requests per call. If available and if requested by e.spire's end user, BellSouth shall provide caller-optional directory assistance call completion service at rates contained in this Attachment to one of the provided listings, equal to that which BellSouth provides its end users. If not available, e.spire may request such requirement pursuant to the Bona Fide Request/New Business Process as set forth in General Terms and Conditions.

10.3.2.2 <u>Directory Assistance Service Updates</u>

- 10.3.2.2.1 BellSouth shall update end user listings changes daily. These changes include:
- 10.3.2.2.1.1 New end user connections: BellSouth will provide service to e.spire that is equal to the service it provides to itself and its end users;
- 10.3.2.2.1.2 End user disconnections: BellSouth will provide service to e.spire that is equal to the service it provides to itself and its end users; and
- 10.3.2.2.1.3 End user address changes: BellSouth will provide service to e.spire that is equal to the service it provides to itself and its end users;
- These updates shall also be provided for non-listed and non-published numbers for use in emergencies.

10.4 Branding for Operator Call Processing and Directory Assistance

- 10.4.1 The BellSouth Operator Systems Branding Feature provides a definable announcement to e.spire end users using Directory Assistance (DA)/Operator Call Processing (OCP) prior to placing them in queue or connecting them to an available operator or automated operator system. This feature allows e.spire to have its calls custom branded with e.spire name on whose behalf BellSouth is providing Directory Assistance and/or Operator Call Processing. Rates for Custom Branding, Operator Call Process and Directory Assistance are set forth in this Attachment.
- BellSouth offers four service levels of branding to e.spire when ordering Directory Assistance and/or Operator Call Processing.
- 10.4.2.1 Service Level 1 BellSouth Branding

10.4.2.2 Service Level 2 - Unbranded 10.4.2.3 Service Level 3 - Custom Branding 10.4.2.4 Service Level 4 - Self Branding (applicable only to e.spire for Resale or use with an Unbundled Port when routing to an operator service provider other than BellSouth). 10.4.3 For Resellers and Use with an Unbundled Port 10.4.3.1 BellSouth Branding is the Default Service Level. 10.4.3.2 Unbranding, Custom Branding, and Self Branding require e.spire to order selective routing for each originating BellSouth end office identified by e.spire. Rates for Selective Routing are set forth in this Attachment. 10.4.3.3 Customer Branding and Self Branding require e.spire to order dedicated trunking from each BellSouth end office identified by e.spire, to either the BellSouth Traffic Operator Position System (TOPS) or e.spire Operator Service Provider. Rates for trunks are set forth in applicable BellSouth tariffs. 10.4.3.4 Unbranding - Unbranded Directory Assistance and/or Operator Call Processing calls ride common trunk groups provisioned by BellSouth from those end offices identified by e.spire to the BellSouth TOPS. These calls are routed to "No Announcement." 10.4.4 For Facilities Based Carriers 10.4.4.1 All Service Levels require e.spire to order dedicated trunking from their end office(s) point of interface to the BellSouth TOPS Switches. Rates for trunks are set forth in applicable BellSouth tariffs. 10.4.4.2 Customized Branding includes charges for the recording of the branding

Directory Assistance customized branding uses:

NAV equipment for which e.spire requires service

- the recording of the name;
- the front-end loading of the Digital Recorded Announcement Machine (DRAM) in each TOPS switch.

announcement and the loading of the audio units in each TOPS Switch, IVS and

Operator Call Processing customized branding uses:

- the recording of the name;
- the front-end loading of the DRAM in the TOPS Switch;

- the back-end loading in the audio units in the Automated Alternate Billing System (AABS) in the Interactive Voice Subsystem (IVS);
- the 0- automation loading for the audio units in the Enhanced Billing and Access Service (EBAS) in the Network Applications Vehicle (NAV).
- 10.4.4.3 BellSouth will provide to e.spire purchasing local BellSouth switching and reselling BellSouth local exchange service, selective routing of calls to a requested directory assistance services platform or operator services platform. e.spire end users may use the same dialing arrangements as BellSouth end users, but obtain a e.spire branded service.

10.5 <u>Directory Assistance Database Service (DADS)</u>

- 10.5.1 BellSouth shall make its Directory Assistance Database Service (DADS) available solely for the expressed purpose of providing Directory Assistance type services to e.spire end users. The term "end user" denotes any entity which obtains Directory Assistance type services for its own use from a DADS customer. Directory Assistance type service is defined as Voice Directory Assistance (DA Operator assisted and Electronic Directory Assistance (Data System assisted)). e.spire agrees that Directory Assistance Database Service (DADS) will not be used for any purpose which violates federal or state laws, statutes, regulatory orders or tariffs. Except for the permitted users, e.spire agrees not to disclose DADS to others and shall provide due care in providing for the security and confidentiality of DADS. Further, e.spire authorizes the inclusion of e.spire Subscriber listings in the BellSouth Directory Assistance products.
- BellSouth shall provide e.spire initially with a base file of subscriber listings which reflect all listing change activity occurring since e.spire's most recent update via magnetic tape, and subsequently using electronic connectivity such as Network Data Mover to be developed mutually by e.spire and BellSouth. e.spire agrees to assume the costs associated with CONNECT: Direct TM connectivity, which will vary depending upon volume and mileage.
- BellSouth will require approximately one month after receiving an order to prepare the Base File. BellSouth will provide daily updates which will reflect all listing change activity occurring since CLEC's most recent update. BellSouth shall provide updates to e.spire on a Business, Residence, or combined Business and Residence basis. e.spire agrees that the updates shall be used solely to keep the information current. Delivery of Daily Updates will commence the day after e.spire receives the Base File.
- BellSouth is authorized to include e.spire Subscriber List Information in its Directory Assistance Database Service (DADS) and its Directory Publishers Database Service (DPDS). Any other use by BellSouth of e.spire Subscriber List Information is not authorized and with the exception of a request for DADS or DPDS, BellSouth shall refer any request for such information to e.spire.

10.5.5 Rates for DADS are as set forth in this Attachment.

10.6 Direct Access to Directory Assistance Service

- 10.6.1 Direct Access to Directory Assistance Service (DADAS) will provide e.spire's directory assistance operators with the ability to search all available BellSouth's subscriber listings using the Directory Assistance search format. Subscription to DADAS will allow e.spire to utilize its own switch, operator workstations and optional audio subsystems.
- 10.6.2 BellSouth will provide DADAS from its DA location. e.spire will access the DADAS system via a telephone company provided point of availability. e.spire has the responsibility of providing the physical links required to connect to the point of availability. These facilities may be purchased from the telephone company as rates and charges billed separately from the charges associated with this offering.
- A specified interface to each e.spire subsystem will be provided by BellSouth. Interconnection between e.spire system and a specified BellSouth location will be pursuant to the use of e.spire owned or e.spire leased facilities and shall be appropriate sized based upon the volume of queries being generated by e.spire.
- The specifications for the three interfaces necessary for interconnection are available in the following documents:
- 10.6.4.1 DADAS to Subscriber Operator Position System—Northern Telecom Document CSI-2300-07; Universal Gateway/ Position Message Interface Format Specification
- DADAS to Subscriber Switch—Northern Telecom Document Q210-1 Version A107; NTDMS/CCIDAS System Application Protocol; and AT&T Document 250-900-535 Operator Services Position System Listing Service and Application Call Processing Data Link Interface Specification
- 10.6.4.3 DADAS to Audio Subsystem (Optional)—Directory One Call Control to Audio Response Unit system interface specifications are available through Northern Telecom as a licensed access protocol—Northern Telecom Document 355-004424 and Gateway/Interactive Voice subsystem Protocol Specification
- 10.6.5 Rates for DADAS are as set forth in this Attachment.

11. <u>Signaling</u>

BellSouth agrees to offer access to signaling and access to BellSouth's signaling databases subject to compatibility testing and at the rates set forth in this Attachment. BellSouth may provide mediated access to BellSouth signaling

systems and databases. Available signaling elements include signaling links, signal transfer points and service control points. Signaling functionality will be available with both A-link and B-link connectivity.

11.1 <u>Definition of Signaling Link Transport</u>

Signaling Link Transport is a set of two or four dedicated 56 Kbps. transmission paths between CLEC-designated Signaling Points of Interconnection (SPOI) that provides appropriate physical diversity.

11.2 Technical Requirements

- 11.2.1 Signaling Link Transport shall consist of full duplex mode 56 kbps transmission paths.
- Of the various options available, Signaling Link Transport shall perform in the following two ways:
- 11.2.2.1 As an "A-link" which is a connection between a switch or SCP and a home Signaling Transfer Point Switch (STP) pair; and
- 11.2.2.2 As a "B-link" which is a connection between two STP pairs in different company networks (e.g., between two STP pairs for two Competitive Local Exchange Carriers (CLECs)).
- 11.2.3 Signaling Link Transport shall consist of two or more signaling link layers as follows:
- 11.2.3.1 An A-link layer shall consist of two links.
- 11.2.3.2 A B-link layer shall consist of four links.
- 11.2.4 A signaling link layer shall satisfy a performance objective such that:
- 11.2.4.1 There shall be no more than two minutes down time per year for an A-link layer; and
- There shall be negligible (less than 2 seconds) down time per year for a B-link layer.
- 11.2.5 A signaling link layer shall satisfy interoffice and intraoffice diversity of facilities and equipment, such that:
- 11.2.5.1 No single failure of facilities or equipment causes the failure of both links in an A-link layer (i.e., the links should be provided on a minimum of two separate physical paths end-to-end); and

- 11.2.5.2 No two concurrent failures of facilities or equipment shall cause the failure of all four links in a B-link layer (i.e., the links should be provided on a minimum of three separate physical paths end-to-end).
- 11.3 <u>Interface Requirements</u>
- There shall be a DS1 (1.544 Mbps) interface at the e.spire-designated SPOIs. Each 56 kbps transmission path shall appear as a DS0 channel within the DS1 interface.

12. <u>Signaling Transfer Points (STPs)</u>

- 12.1 <u>Definition</u> Signaling Transfer Points is a signaling network function that includes all of the capabilities provided by the signaling transfer point switches (STPs) and their associated signaling links which enable the exchange of SS7 messages among and between switching elements, database elements and signaling transfer point switches
- 12.2 Technical Requirements
- 12.2.1 STPs shall provide access to Network Elements connected to BellSouth SS7 network. These include:
- 12.2.1.1 BellSouth Local Switching or Tandem Switching;
- 12.2.1.2 BellSouth Service Control Points/DataBases;
- 12.2.1.3 Third-party local or tandem switching;
- 12.2.1.4 Third-party-provided STPs.
- The connectivity provided by STPs shall fully support the functions of all other Network Elements connected to BellSouth SS7 network. This explicitly includes the use of BellSouth SS7 network to convey messages which neither originate nor terminate at a signaling end point directly connected to BellSouth SS7 network (*i.e.*, transient messages). When BellSouth SS7 network is used to convey transient messages, there shall be no alteration of the Integrated Services Digital Network User Part (ISDNUP) or Transaction Capabilities Application Part (TCAP) user data that constitutes the content of the message.
- 12.2.3 If a BellSouth tandem switch routes calling traffic, based on dialed or translated digits, on SS7 trunks between an e.spire local switch and third party local switch, BellSouth SS7 network shall convey the TCAP messages that are necessary to provide Call Management features (Automatic Callback, Automatic Recall, and Screening List Editing) between e.spire local STPs and the STPs that provide connectivity with the third party local switch, even if the third party local switch is not directly connected to BellSouth STPs.

- 12.2.4 STPs shall provide all functions of the MTP as defined in Telcordia (formerly BellCore) ANSI Interconnection Requirements. This includes:
- 12.2.4.1 Signaling Data Link functions, as defined in Telcordia (formerly BellCore) ANSI Interconnection Requirements,
- 12.2.4.2 Signaling Link functions, as defined in Telcordia (formerly BellCore) ANSI Interconnection Requirements, and
- 12.2.4.3 Signaling Network Management functions, as defined in Telcordia (formerly BellCore) ANSI Interconnection Requirements.
- STPs shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as defined in Telcordia (formerly BellCore) ANSI Interconnection Requirements. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. In cases where the destination signaling point is a e.spire or third party local or tandem switching system directly connected to BellSouth SS7 network, BellSouth shall perform final GTT of messages to the destination and SCCP Subsystem Management of the destination. In all other cases, BellSouth shall perform intermediate GTT of messages to a gateway pair of STPs in an SS7 network connected with BellSouth SS7 network, and shall not perform SCCP Subsystem Management of the destination. If BellSouth performs final GTT to a e.spire database, then e.spire agrees to provide BellSouth with the Destination Point Code for the e.spire database.
- 12.2.6 STPs shall provide on a non-discriminatory basis all functions of the OMAP commonly provided by STPs, as specified in the reference in Section 12.4.5 of this Attachment. All OMAP functions will be on a "where available" basis and can include:
- 12.2.6.1 MTP Routing Verification Test (MRVT) and
- 12.2.6.2 SCCP Routing Verification Test (SRVT).
- In cases where the destination signaling point is a BellSouth local or tandem switching system or database, or is an e.spire or third party local or tandem switching system directly connected to the BellSouth SS7 network, STPs shall perform MRVT and SRVT to the destination signaling point. In all other cases, STPs shall perform MRVT and SRVT to a gateway pair of STPs in an SS7 network connected with the BellSouth SS7 network. This requirement shall be superseded by the specifications for Internetwork MRVT and SRVT if and when these become approved ANSI standards and available capabilities of BellSouth STPs, and if mutually agreed upon by e.spire and BellSouth.
- 12.2.8 STPs shall be on parity with BellSouth.

- 12.2.9 SS7 Advanced Intelligent Network (AIN) Access
- When technically feasible and upon request by e.spire, SS7 Access shall be made available in association with switching. SS7 AIN Access is the provisioning of AIN 0.1 triggers in an equipped BellSouth local switch and interconnection of the BellSouth SS7 network with the e.spire SS7 network to exchange TCAP queries and responses with an e.spire SCP.
- 12.2.9.2 SS7 AIN Access shall provide e.spire SCP access to BellSouth local switch in association with switching via interconnection of BellSouth SS7 and e.spire SS7 Networks. BellSouth shall offer SS7 access through its STPs. If BellSouth requires a mediation device on any part of its network specific to this form of access, BellSouth must route its messages in the same manner. The interconnection arrangement shall result in the BellSouth local switch recognizing the e.spire SCP as at least at parity with BellSouth's SCPs in terms of interfaces, performance and capabilities.

12.3 <u>Interface Requirements</u>

- 12.3.1 BellSouth shall provide the following STPs options to connect e.spire or e.spire-designated local switching systems or STPs to BellSouth SS7 network:
- 12.3.1.1 An A-link interface from e.spire local switching systems; and,
- 12.3.1.2 A B-link interface from e.spire local STPs.
- Each type of interface shall be provided by one or more sets (layers) of signaling links.
- The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling for interconnecting e.spire local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and e.spire will work jointly to establish mutually acceptable SPOIs.
- BellSouth CO shall provide intraoffice diversity between the SPOIs and BellSouth STPs, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and e.spire will work jointly to establish mutually acceptable SPOIs.
- BellSouth shall provide MTP and SCCP protocol interfaces that shall conform to all sections relevant to the MTP or SCCP in the following specifications:

- 12.3.5.1 Telcordia (formerly BellCore) GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP);
- 12.3.5.2 Telcordia (formerly BellCore) GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).

12.3.6 <u>Message Screening</u>

- BellSouth shall set message screening parameters so as to accept valid messages from e.spire local or tandem switching systems destined to any signaling point within BellSouth's SS7 network where the e.spire switching system has a legitimate signaling relation.
- 12.3.6.2 BellSouth shall set message screening parameters so as to pass valid messages from e.spire local or tandem switching systems destined to any signaling point or network accessed through BellSouth's SS7 network where the e.spire switching system has a legitimate signaling relation.
- BellSouth shall set message screening parameters so as to accept and pass/send valid messages destined to and from e.spire from any signaling point or network interconnected through BellSouth's SS7 network where the e.spire SCP has a legitimate signaling relation.
- 12.4 STPs shall be equal to or better than all of the requirements for STPs set forth in the following technical references:
- 12.4.1 ANSI T1.111-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP);
- 12.4.2 ANSI T1.111A-1994 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP) Supplement;
- 12.4.3 ANSI T1.112-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Signaling Connection Control Part (SCCP);
- 12.4.4 ANSI T1.115-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Monitoring and Measurements for Networks;
- 12.4.5 ANSI T1.116-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Operations, Maintenance and Administration Part (OMAP);
- 12.4.6 ANSI T1.118-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Intermediate Signaling Network Identification (ISNI);

- 12.4.7 Telcordia (formerly BellCore) GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP); and
- 12.4.8 Telcordia (formerly BellCore) GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).

13. Service Control Points/DataBases

13.1 Definition

- Databases are the Network Elements that provide the functionality for storage of, access to, and manipulation of information required to offer a particular service and/or capability. Databases include, but are not limited to: Local Number Portability, LIDB, Toll Free Number Database, Automatic Location Identification/Data Management System, Calling Name Database, access to Service Creation Environment and Service Management System (SCE/SMS) application databases and Directory Assistance.
- 13.1.2 A Service Control Point (SCP) is a specific type of Database functionality deployed in a Signaling System 7 (SS7) network that executes service application logic in response to SS7 queries sent to it by a switching system also connected to the SS7 network. Service Management Systems provide operational interfaces to allow for provisioning, administration and maintenance of subscriber data and service application data stored in SCPs.

13.2 <u>Technical Requirements for SCPs/Databases</u>

Requirements for SCPs/Databases within this section address storage of information, access to information (e.g. signaling protocols, response times), and administration of information (e.g., provisioning, administration, and maintenance). All SCPs/Databases shall be provided to e.spire in accordance with the following requirements.

- BellSouth shall provide physical access to SCPs through the SS7 network and protocols with TCAP as the application layer protocol.
- BellSouth shall provide physical interconnection to databases via industry standard interfaces and protocols (e.g. SS7, ISDN and X.25).
- The reliability of interconnection options shall be consistent with requirements for diversity and survivability.
- 13.2.4 <u>Database Availability</u>

Call processing databases shall have a maximum unscheduled availability of 30 minutes per year. Unavailability due to software and hardware upgrades shall be scheduled during minimal usage periods and only be undertaken upon proper notification to providers which might be impacted. Any downtime associated with the provision of call processing related databases will impact all service providers, including BellSouth, equally.

The operational interface provided by BellSouth shall complete Database transactions (i.e., add, modify, delete) for e.spire customer records stored in BellSouth databases within 3 days, or sooner where BellSouth provisions its own customer records within a shorter interval.

13.3 Local Number Portability Database

13.3.1 Definition

The Permanent Number Portability (PNP) database supplies routing numbers for calls involving numbers that have been ported from one local service provider to another. PNP is currently being worked in industry forums. The results of these forums will dictate the industry direction of PNP. BellSouth agrees to provide access to the PNP database at rates, terms and conditions as set forth by BellSouth and in accordance with an effective FCC or Commission directive.

13.4 <u>Line Information Database (LIDB)</u>

BellSouth will store in its LIDB only records relating to service in the BellSouth region. The LIDB Storage Agreement is included in this Attachment.

13.4.1 Definition

The Line Information Database (LIDB) is a transaction-oriented database accessible through Common Channel Signaling (CCS) networks. It contains records associated with end user Line Numbers and Special Billing Numbers. LIDB accepts queries from other Network Elements and provides appropriate responses. The query originator need not be the owner of LIDB data. LIDB queries include functions such as screening billed numbers that provides the ability to accept Collect or Third Number Billing calls and validation of Telephone Line Number based non-proprietary calling cards. The interface for the LIDB functionality is the interface between BellSouth CCS network and other CCS networks. LIDB also interfaces to administrative systems.

13.4.2 Technical Requirements

BellSouth will offer to e.spire any additional capabilities that are developed for LIDB during the life of this Agreement.

- BellSouth shall process e.spire's Customer records in LIDB at least at parity with BellSouth customer records, with respect to other LIDB functions. BellSouth shall indicate to e.spire what additional functions (if any) are performed by LIDB in the BellSouth network.
- Within two (2) weeks after a request by e.spire, BellSouth shall provide e.spire with a list of the customer data items which e.spire would have to provide in order to support each required LIDB function. The list shall indicate which data items are essential to LIDB function, and which are required only to support certain services. For each data item, the list shall show the data formats, the acceptable values of the data item and the meaning of those values.
- BellSouth shall provide LIDB systems for which operating deficiencies that would result in calls being blocked, shall not exceed 30 minutes per year.
- BellSouth shall provide LIDB systems for which operating deficiencies that would not result in calls being blocked shall not exceed 12 hours per year.
- 13.4.2.5 BellSouth shall provide LIDB systems for which the LIDB function shall be in overload no more than 12 hours per year.
- All additions, updates and deletions of e.spire data to the LIDB shall be solely at the direction of e.spire. Such direction from e.spire will not be required where the addition, update or deletion is necessary to perform standard fraud control measures (e.g., calling card auto-deactivation).
- 13.4.2.7 BellSouth shall provide priority updates to LIDB for e.spire data upon e.spire's request (e.g., to support fraud detection), via password-protected telephone card, facsimile, or electronic mail within one hour of notice from the established BellSouth contact.
- 13.4.2.8 BellSouth shall provide LIDB systems such that no more than 0.01% of e.spire customer records will be missing from LIDB, as measured by e.spire audits. BellSouth will audit e.spire records in LIDB against DBAS to identify record mismatches and provide this data to a designated e.spire contact person to resolve the status of the records and BellSouth will update system appropriately. BellSouth will refer record of mis-matches to e.spire within one business day of audit. Once reconciled records are received back from e.spire, BellSouth will update LIDB the same business day if less than 500 records are received before 1:00PM Central Time. If more than 500 records are received, BellSouth will contact e.spire to negotiate a time frame for the updates, not to exceed three business days.
- BellSouth shall perform backup and recovery of all of e.spire's data in LIDB including sending to LIDB all changes made since the date of the most recent backup copy, in at least the same time frame BellSouth performs backup and recovery of BellSouth data in LIDB for itself. Currently, BellSouth performs

backups of the LIDB for itself on a weekly basis and when a new software release is scheduled, a backup is performed prior to loading the new release.

- 13.4.2.10 BellSouth shall provide e.spire with LIDB reports of data which are missing or contain errors, as well as any misrouted errors, within a reasonable time period as negotiated between e.spire and BellSouth.
- 13.4.2.11 BellSouth shall prevent any access to or use of e.spire data in LIDB by BellSouth personnel that are outside of established administrative and fraud control personnel, or by any other Party that is not authorized by e.spire in writing.
- BellSouth shall provide e.spire performance of the LIDB Data Screening function, which allows a LIDB to completely or partially deny specific query originators access to LIDB data owned by specific data owners, for Customer Data that is part of an NPA-NXX or RAO-0/1XX wholly or partially owned by e.spire at least at parity with BellSouth Customer Data. BellSouth shall obtain from e.spire the screening information associated with LIDB Data Screening of e.spire data in accordance with this requirement. BellSouth currently does not have LIDB Data Screening capabilities. When such capability is available, BellSouth shall offer it to e.spire under the Bona Fide Request/New Business Process as set forth in General Terms and Conditions.
- BellSouth shall accept queries to LIDB associated with e.spire customer records, and shall return responses in accordance with industry standards.
- 13.4.2.14 BellSouth shall provide mean processing time at the LIDB within 0.50 seconds under normal conditions as defined in industry standards.
- 13.4.2.15 BellSouth shall provide processing time at the LIDB within 1 second for 99% of all messages under normal conditions as defined in industry standards.
- 13.4.3 Interface Requirements

BellSouth shall offer LIDB in accordance with the requirements of this subsection.

- 13.4.3.1 The interface to LIDB shall be in accordance with the technical references contained within.
- 13.4.3.2 The CCS interface to LIDB shall be the standard interface described herein.
- 13.4.3.3 The LIDB Data Base interpretation of the ANSI-TCAP messages shall comply with the technical reference herein. Global Title Translation shall be maintained in the signaling network in order to support signaling network routing to the LIDB.

13.5 <u>Toll Free Number Database</u>

The Toll Free Number Database is a SCP that provides functionality necessary for toll free (e.g., 800 and 888) number services by providing routing information and additional so-called vertical features during call set-up in response to queries from SSPs. BellSouth shall provide the Toll Free Number Database in accordance with the following:

13.5.1 Technical Requirements

- 13.5.1.1 BellSouth shall make BellSouth Toll Free Number Database available for e.spire to query with a toll-free number and originating information.
- 13.5.1.2 The Toll Free Number Database shall return carrier identification and, where applicable, the queried toll free number, translated numbers and instructions as it would in response to a query from a BellSouth switch.
- The SCP shall also provide, at e.spire's option, such additional feature as described in SR-TSV-002275 (BOC Notes on BellSouth Networks, SR-TSV-002275, Issue 2, (Telcordia (formerly BellCore), April 1994)) as are available to BellSouth. These may include but are not limited to:
- 13.5.1.3.1 Network Management;
- 13.5.1.3.2 Customer Sample Collection; and
- 13.5.1.3.3 Service Maintenance

13.6 <u>Automatic Location Identification/Data Management System (ALI/DMS)</u>

The ALI/DMS Database contains end user information (including name, address, telephone information, and sometimes special information from the local service provider or end user) used to determine to which Public Safety Answering Point (PSAP) to route the call. The ALI/DMS database is used to provide more routing flexibility for E911 calls than Basic 911. BellSouth shall provide the Emergency Services Database in accordance with the following:

13.6.1 Technical Requirements

- 13.6.1.1 BellSouth shall offer e.spire a data link to the ALI/DMS database or permit e.spire to provide its own data link to the ALI/DMS database. BellSouth shall provide error reports from the ALI/DMS database to e.spire immediately after e.spire inputs information into the ALI/DMS database. Alternately, e.spire may utilize BellSouth, to enter end user information into the data base on a demand basis, and validate end user information on a demand basis.
- 13.6.1.2 The ALI/DMS database shall contain the following end user information:
- 13.6.1.2.1 Name;

13.6.1.2.2 Address: 13.6.1.2.3 Telephone number; and 13.6.1.2.4 Other information as appropriate (e.g., whether a end user is blind or deaf or has another disability). 13.6.1.3 When BellSouth is responsible for administering the ALI/DMS database in its entirety, ported number NXXs entries for the ported numbers should be maintained unless e.spire requests otherwise and shall be updated if e.spire requests, provided e.spire supplies BellSouth with the updates. 13.6.1.4 When Remote Call Forwarding (RCF) is used to provide number portability to the local end user and a remark or other appropriate field information is available in the database, the shadow or "forwarded-to" number and an indication that the number is ported shall be added to the customer record. If BellSouth is responsible for configuring PSAP features (for cases when the 13.6.1.5 PSAP or BellSouth supports an ISDN interface) it shall ensure that CLASS Automatic Recall (Call Return) is not used to call back to the ported number. Although BellSouth currently does not have ISDN interface, BellSouth agrees to comply with this requirement once ISDN interfaces are in place. 13.6.2 **Interface Requirements** The interface between the E911 Switch or Tandem and the ALI/DMS database for e.spire end users shall meet industry standards. 13.7 Calling Name (CNAM) Database Service. The Agreement for Calling Name (CNAM) with standard pricing is included as Exhibit B to this Attachment. e.spire must provide to its account manager a written request with a requested activation date to activate this service. If e.spire is interested in requesting CNAM with volume and term pricing, e.spire must contact its account manager to request a separate CNAM volume and term Agreement. 13.8 SCPs/Databases shall be equal to or better than all of the requirements for SCPs/Databases set forth in the following technical references:

13.8.1

13.8.2

GR-246-CORE, Bell Communications Research Specification of Signaling System Number 7, ISSUE 1 (Telcordia (formerly BellCore), December 199);

GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting

Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP). (Telcordia (formerly BellCore), March 1994);

13.8.3 GR-954-CORE, CCS Network Interface Specification (CCSNIS) Supporting Line Information Database (LIDB) Service 6, Issue 1, Rev. 1 (Telcordia (formerly BellCore), October 1995); 13.8.4 GR-1149-CORE, OSSGR Section 10: System Interfaces, Issue 1 (Telcordia (formerly BellCore), October 1995) (Replaces TR-NWT-001149); 13.8.5 Telcordia (formerly BellCore) GR-1158-CORE, OSSGR Section 22.3: Line Information Database 6, Issue (Telcordia (formerly BellCore), October 1995); 13.8.6 Telcordia (formerly BellCore) GR-1428-CORE, CCS Network Interface Specification (CCSNIS) Supporting Toll Free Service (Telcordia (formerly BellCore), May 1995); and 13.8.7 BOC Notes on BellSouth Networks, SR-TSV-002275, ISSUE 2, (Telcordia (formerly BellCore), April 1994). 13.9 Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access. 13.9.1 BellSouth's Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access shall provide e.spire the capability that will allow e.spire and other third parties to create service applications in a BellSouth Service Creation Environment and deploy those applications in a BellSouth SMS to a BellSouth SCP. The third party service applications interact with AIN triggers provisioned on a BellSouth SSP. 13.9.2 BellSouth's SCE/SMS AIN Access shall provide access to SCE hardware, software, testing and technical support (e.g., help desk, system administrator) resources available to e.spire. Scheduling procedures shall provide e.spire equivalent priority to these resources 13.9.3 BellSouth SCP shall partition and protect e.spire service logic and data from unauthorized access, execution or other types of compromise. 13.9.4 When e.spire selects SCE/SMS AIN Access, BellSouth shall provide training, documentation, and technical support to enable e.spire to use BellSouth's SCE/SMS AIN Access to create and administer applications. Training, documentation, and technical support will address use of SCE and SMS access and administrative functions, but will not include support for the creation of a specific service application. 13.9.5 When e.spire selects SCE/SMS AIN Access, BellSouth shall provide for a secure,

dial-in, ISDN).

controlled access environment in association with its internal use of AIN components. e.spire access will be provided via remote data connection (e.g.,

When e.spire selects SCE/SMS AIN Access, BellSouth shall allow e.spire to download data forms and/or tables to BellSouth SCP via BellSouth SMS without intervention from BellSouth (*e.g.*, service customization and end user subscription).

14. <u>DARK FIBER</u>

BellSouth agrees to offer access to Dark Fiber pursuant to the terms and conditions following and at the rates set forth in this Attachment.

Dark Fiber is defined as BellSouth's optical transmission facilities without attached multiplexing, aggregation or other electronics. e.spire may utilize Dark Fiber for either loops or transport. Dark Fiber does not have electronics on either end of the Dark Fiber segment. It may be strands of optical fiber existing in aerial or underground structure. No regeneration or optical amplification will be included with this element.

14.2 <u>Requirements</u>

- 14.2.1 BellSouth shall make available Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. BellSouth shall offer all Dark Fiber to e.spire pursuant to the prices set forth in this Attachment.
- e.spire may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- 14.2.3 BellSouth shall use its best efforts to provide to e.spire information regarding the location, availability and performance of Dark Fiber within ten (10) business days for a records based answer and twenty (20) business days for a field based answer, after receiving a request from e.spire ("Request"). Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation").
- 14.2.4 BellSouth shall use its best efforts to make Dark Fiber available to e.spire within thirty (30) business days after it receives written confirmation from e.spire that the Dark Fiber previously deemed available by BellSouth is wanted for use by e.spire. This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable e.spire to connect or splice e.spire provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.

15. <u>SS7 Network Inter</u>connection

15.1.1 Definition

SS7 Network Interconnection is the interconnection of e.spire local Signaling Transfer Point Switches (STP) and e.spire local or tandem switching systems with BellSouth STPs. This interconnection provides connectivity that enables the exchange of SS7 messages among BellSouth switching systems and databases (DBs), e.spire local or tandem switching systems, and other third-party switching systems directly connected to the BellSouth SS7 network.

15.1.2 Technical Requirements

- 15.1.2.1 SS7 Network Interconnection shall provide connectivity to all components of the BellSouth SS7 network. These include:
- 15.1.2.1.1 BellSouth local or tandem switching systems;
- 15.1.2.1.2 BellSouth DBs; and
- 15.1.2.1.3 Other third-party local or tandem switching systems.
- 15.1.2.2 The connectivity provided by SS7 Network Interconnection shall fully support the functions of BellSouth switching systems and DBs and e.spire or other third-party switching systems with A-link access to the BellSouth SS7 network.

If traffic is routed based on dialed or translated digits between an e.spire local switching system and a BellSouth or other third-party local switching system, either directly or via a BellSouth tandem switching system, then it is a requirement that the BellSouth SS7 network convey via SS7 Network Interconnection the TCAP messages that are necessary to provide Call Management services (Automatic Callback, Automatic Recall, and Screening List Editing) between the e.spire local STPs and BellSouth or other third-party local switch.

- When the capability to route messages based on Intermediate Signaling Network Identifier (ISNI) is generally available on BellSouth STPs, the BellSouth SS7 Network shall also convey TCAP messages using SS7 Network Interconnection in similar circumstances where the BellSouth switch routes traffic based on a Carrier Identification Code (CIC).
- 15.1.2.4 SS7 Network Interconnection shall provide all functions of the MTP as specified in ANSI T1.111. This includes:
- 15.1.2.4.1 Signaling Data Link functions, as specified in ANSI T1.111.2;
- 15.1.2.4.2 Signaling Link functions, as specified in ANSI T1.111.3; and
- 15.1.2.4.3 Signaling Network Management functions, as specified in ANSI T1.111.4.
- 15.1.2.5 SS7 Network Interconnection shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as specified in ANSI T1.112. In particular, this includes Global Title Translation (GTT) and SCCP Management

procedures, as specified in T1.112.4. Where the destination signaling point is a BellSouth switching system or DB, or is another third-party local or tandem switching system directly connected to the BellSouth SS7 network, SS7 Network Interconnection shall include final GTT of messages to the destination and SCCP Subsystem Management of the destination. Where the destination signaling point is an e.spire local or tandem switching system, SS7 Network Interconnection shall include intermediate GTT of messages to a gateway pair of e.spire local STPs, and shall not include SCCP Subsystem Management of the destination.

- 15.1.2.6 SS7 Network Interconnection shall provide all functions of the Integrated Services Digital Network User Part (ISDNUP), as specified in ANSI T1.113.
- 15.1.2.7 SS7 Network Interconnection shall provide all functions of the TCAP, as specified in ANSI T1.114.
- 15.1.2.8 If and when Internetwork MTP Routing Verification Test (MRVT) and SCCP Routing Verification Test (SRVT) become approved ANSI standards and available capabilities of BellSouth STPs, SS7 Network Interconnection shall provide these functions of the OMAP.
- 15.1.2.9 SS7 Network Interconnection shall be equal to or better than the following performance requirements:
- 15.1.2.9.1 MTP Performance, as specified in ANSI T1.111.6;
- 15.1.2.9.2 SCCP Performance, as specified in ANSI T1.112.5; and
- 15.1.2.9.3 ISDNUP Performance, as specified in ANSI T1.113.5.
- 15.1.3 Interface Requirements
- 15.1.3.1 BellSouth shall offer the following SS7 Network Interconnection options to connect e.spire or e.spire-designated local or tandem switching systems or STPs to the BellSouth SS7 network:
- 15.1.3.1.1 A-link interface from e.spire local or tandem switching systems; and
- 15.1.3.1.2 B-link interface from e.spire STPs.
- The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where the BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling links for interconnecting e.spire local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and e.spire will work jointly to establish mutually acceptable SPOI.

- BellSouth CO shall provide intraoffice diversity between the SPOIs and the BellSouth STP, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and e.spire will work jointly to establish mutually acceptable SPOI.
- 15.1.3.4 The protocol interface requirements for SS7 Network Interconnection include the MTP, ISDNUP, SCCP, and TCAP. These protocol interfaces shall conform to the following specifications:
- Telcordia (formerly BellCore) GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP);
- 15.1.3.4.2 Telcordia (formerly BellCore) GR-1428-CORE, CCS Network Interface Specification (CCSNIS) Supporting Toll Free Service;
- 15.1.3.4.3 Telcordia (formerly BellCore) GR-1429-CORE, CCS Network Interface Specification (CCSNIS) Supporting Call Management Services; and
- 15.1.3.4.4 Telcordia (formerly BellCore) GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).
- 15.1.3.5 BellSouth shall set message screening parameters to block accept messages from e.spire local or tandem switching systems destined to any signaling point in the BellSouth SS7 network with which the e.spire switching system has a legitimate signaling relation.
- 15.1.4 SS7 Network Interconnection shall be equal to or better than all of the requirements for SS7 Network Interconnection set forth in the following technical references:
- 15.1.4.1 ANSI T1.110-1992 American National Standard Telecommunications Signaling System Number 7 (SS7) General Information;
- 15.1.4.2 ANSI T1.111-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP);
- 15.1.4.3 ANSI T1.111A-1994 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP) Supplement;
- 15.1.4.4 ANSI T1.112-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Signaling Connection Control Part (SCCP);
- 15.1.4.5 ANSI T1.113-1995 American National Standard for Telecommunications Signaling System Number 7 (SS7) Integrated Services Digital Network (ISDN) User Part;

- 15.1.4.6 ANSI T1.114-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Transaction Capabilities Application Part (TCAP);
- 15.1.4.7 ANSI T1.115-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Monitoring and Measurements for Networks;
- 15.1.4.8 ANSI T1.116-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Operations, Maintenance and Administration Part (OMAP);
- 15.1.4.9 ANSI T1.118-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Intermediate Signaling Network Identification (ISNI);
- Telcordia (formerly BellCore) GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP);
- 15.1.4.11 Telcordia (formerly BellCore) GR-954-CORE, CCS Network Interface Specification (CCSNIS) Supporting Line Information Database (LIDB) Service;
- 15.1.4.12 Telcordia (formerly BellCore) GR-1428-CORE, CCS Network Interface Specification (CCSNIS) Supporting Toll Free Service;
- 15.1.4.13 Telcordia (formerly BellCore) GR-1429-CORE, CCS Network Interface Specification (CCSNIS) Supporting Call Management Services; and,
- 15.1.4.14 Telcordia (formerly BellCore) GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).

16. Basic 911 and E911

If e.spire orders network elements and other services, then e.spire is also responsible for providing E911 to its end users. BellSouth agrees to offer access to the 911/E911 network pursuant to the following terms and conditions set forth in this Attachment.

16.1 Definition

Basic 911 and E911 is an additional requirement that provides a caller access to the applicable emergency service bureau by dialing a 3-digit universal telephone number (911).

16.2 Requirements

- Basic 911 Service Provisioning. For Basic 911 service, BellSouth will provide to e.spire a list consisting of each municipality that subscribes to Basic 911 service. The list will also provide, if known, the E911 conversion date for each municipality and, for network routing purposes, a ten-digit directory number representing the appropriate emergency answering position for each municipality subscribing to 911. e.spire will be required to arrange to accept 911 calls from its end users in municipalities that subscribe to Basic 911 service and translate the 911 call to the appropriate 10-digit directory number as stated on the list provided by BellSouth. e.spire will be required to route that call to BellSouth at the appropriate tandem or end office. When a municipality converts to E911 service, e.spire will be required to discontinue the Basic 911 procedures and being using E911 procedures.
- 16.2.2 E911 Service Provisioning. For E911 service, e.spire will be required to install a minimum of two dedicated trunks originating from the e.spire serving wire center and terminating to the appropriate E911 tandem. The dedicated trunks shall be, at a minimum, DS-0 level trunks configured either as a 2-wire analog interface or as part of a digital (1.544 Mb/s) interface. Either configuration shall use CAMAtype signaling with multifrequency ("MF") pulsing that will deliver automatic number identification ("ANI") with the voice portion of the call. If the user interface is digital, MF pulses, as well as other AC signals, shall be encoded per the u-255 Law convention. e.spire will be required to provide BellSouth daily updates to the E911 database. e.spire will be required to forward 911 calls to the appropriate E911 tandem, along with ANI, based upon the current E911 end office to tandem homing arrangement as provided by BellSouth. If the E911 tandem trunks are not available, e.spire will be required to route the call to a designated 7-digit local number residing in the appropriate Public Service Answering Point ("PSAP"). This call will be transported over BellSouth's interoffice network and will not carry the ANI of the calling party. e.spire shall be responsible for providing BellSouth with complete and accurate data for submission to the 911/E911 database for the purpose of providing 911/E911 to its end users.
- 16.2.3 <u>Rates.</u> Charges for 911/E911 service are borne by the municipality purchasing the service. BellSouth will impose no charge on e.spire beyond applicable charges for BellSouth trunking arrangements.
- 16.2.4 Basic 911 and E911 functions provided to e.spire shall be at least at parity with the support and services that BellSouth provides to its end users for such similar functionality.

<u>Detailed Practices and Procedures</u>. The detailed practices and procedures contained in the E911 Local Exchange Carrier Guide For Facility-Based Providers as amended from time to time during the term of this Agreement will determine the appropriate practices and procedures for BellSouth and e.spire to follow in providing 911/E911 services.

17. Combinations

17.1 For purposes of this Section, references to "Already Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.

17.2 **EELs**

- Where necessary to comply with an effective FCC and/or State Commission order, or as otherwise mutually agreed by the Parties, BellSouth shall offer access to loop and transport combinations, also known as the Enhanced Extended Link ("EEL") as defined in Section 17.3 below.
- Subject to Section 17.2.3 below, BellSouth will provide access to the EEL in the combinations set forth in 17.3 following. This offering is intended to provide connectivity from an end user's location through that end user's SWC to e.spire's POP serving wire center. The circuit must be connected to e.spire's switch for the purpose of provisioning telecommunications services, including telephone exchange service, to e.spire's end-user customers. Except as provided for in paragraph 22 of the FCC's Supplemental Order Clarification, released June 2, 2000, in CC Docket No. 96-98 ("June 2, 2000 Order"), the EEL will be connected to e.spire's facilities in e.spire's collocation space at the POP SWC. e.spire may purchase BellSouth's access facilities between e.spire's POP and e.spire's collocation space at the POP SWC.
- 17.2.3 BellSouth shall provide EEL combinations to e.spire in the state of Georgia regardless of whether or not such EELs are Already Combined. In all other states, BellSouth shall make available to e.spire those EEL combinations described in Section 4.3 below only to the extent such combinations are Already Combined.
- 17.2.4 BellSouth will make available EEL combinations to e.spire in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs, regardless of whether or not such EELs are Already Combined.
- 17.2.5 Additionally, BellSouth shall make available to e.spire a combination of an unbundled loop and tariffed special access interoffice facilities. To the extent e.spire will require multiplexing functionality in connection with such combination, BellSouth will provide access to multiplexing within the central office pursuant to the terms, conditions and rates set forth in its Access Services Tariffs. The combination of an unbundled loop and tariffed special access interoffice facilities and any associated tariffed services, including but not limited to multiplexing, shall not be eligible for conversion to UNEs as described in Section 17.5 below. Where

multiplexing functionality is required in connection with loop and transport combinations, such multiplexing will be provided at the rates and on the terms set forth in this Agreement.

17.3	EEL Combinations
17.3.1	DS1 Interoffice Channel + DS1 Channelization + 2-wire VG Local Loop
17.3.2	DS1 Interoffice Channel + DS1 Channelization + 4-wire VG Local Loop
17.3.3	DS1 Interoffice Channel + DS1 Channelization + 2-wire ISDN Local Loop
17.3.4	DS1 Interoffice Channel + DS1 Channelization + 4-wire 56 kbps Local Loop
17.3.5	DS1 Interoffice Channel + DS1 Channelization + 4-wire 64 kbps Local Loop
17.3.6	DS1 Interoffice Channel + DS1 Local Loop
17.3.7	DS3 Interoffice Channel + DS3 Local Loop
17.3.8	STS-1 Interoffice Channel + STS-1 Local Loop
17.3.9	DS3 Interoffice Channel + DS3 Channelization + DS1 Local Loop
17.3.10	STS-1 Interoffice Channel + DS3 Channelization + DS1 Local Loop
17.3.11	2-wire VG Interoffice Channel + 2-wire VG Local Loop
17.3.12	4wire VG Interoffice Channel + 4-wire VG Local Loop
17.3.13	4-wire 56 kbps Interoffice Channel + 4-wire 56 kbps Local Loop
17.3.14	4-wire 64 kbps Interoffice Channel + 4-wire 64 kbps Local Loop

17.4 Other Network Element Combinations

In the state of Georgia, BellSouth shall make available to e.spire, in accordance with Section 17.6 below: (1) combinations of network elements other than EELs that are Already Combined; and (2) combinations of network elements other than EELs that are not Already Combined but that BellSouth ordinarily combines in its network. In all other states, BellSouth shall make available to e.spire, in accordance with Section 17.6 below, combinations of network elements other than EELs only to the extent such combinations are Already Combined.

17.5 Special Access Service Conversions

- e.spire may not convert special access services to combinations of loop and transport network elements, whether or not e.spire self-provides its entrance facilities (or obtains entrance facilities from a third party), unless e.spire uses the combination to provide a "significant amount of local exchange service" (as described in Section 17.5.2 below), in addition to exchange access service, to a particular customer.
- 17.5.2 For the purpose of special access conversions, a "significant amount of local exchange service" is as defined in the FCC's June 2, 2000 Order. The Parties agree to incorporate by reference paragraph 22 of the June 2, 2000 Order. When e.spire requests conversion of special access circuits, e.spire will self-certify to BellSouth in the manner specified in paragraph 29 of the June 2, 2000 Order that the circuits to be converted qualify for conversion. In addition there may be extraordinary circumstances where e.spire is providing a significant amount of local exchange service, but does not qualify under any of the three options set forth in paragraph 22 of June 2, 2000 Order. In such case, e.spire may petition the FCC for a waiver of the local usage options set forth in the June 2, 2000 Order. If a waiver is granted, then upon e.spire's request the Parties shall amend this Agreement to the extent necessary to incorporate the terms of such waiver for such extraordinary circumstance.
- 17.5.3 Upon request for conversions of up to 15 circuits from special access to EELs, BellSouth shall perform such conversions within seven (7) days from BellSouth's receipt of a valid, error free service order from e.spire. Requests for conversions of fifteen (15) or more circuits from special access to EELs will be provisioned on a project basis. Conversions should not require the special access circuit to be disconnected and reconnected because only the billing information or other administrative information associated with the circuit will change when e.spire requests a conversion. The Access Service Request process will be used for conversion requests.
- BellSouth may, at its sole expense, and upon thirty (30) days notice to e.spire, audit e.spire's records not more than once in any twelve month period, unless an audit finds non-compliance with the local usage options referenced in the June 2, 2000 Order, in order to verify the type of traffic being transmitted over combinations of loop and transport network elements. If, based on its audits, BellSouth concludes that e.spire is not providing a significant amount of local exchange traffic over the combinations of loop and transport network elements, BellSouth may file a complaint with the appropriate Commission, pursuant to the dispute resolution process set forth in this Agreement. In the event that BellSouth prevails, BellSouth may convert such combinations of loop and transport network elements to special access services and may seek appropriate retroactive reimbursement from e.spire.

- 17.6 Rates
- 17.6.1 Georgia
- 17.6.1.1 The non-recurring and recurring rates for the EEL Combinations of network elements set forth in 17.3, whether Already Combined or new, are as set forth in this Attachment.
- 17.6.1.2 On an interim basis, for combinations of loop and transport network elements not set forth in Section 17.3, where the elements are not Already Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.
- 17.6.1.3 To the extent that e.spire seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, e.spire, at its option, can request that such rates be determined pursuant to the Bona Fide Request/New Business Request (NBR) process set forth in this Agreement.
- 17.6.2 All Other States
- 17.6.2.1 Subject to Section 17.2.3 and 17.4 preceding, all other states, the rates for (1) Already Combined EEL combinations set forth in Section 17.3, and (2) other combinations of network elements that are Already Combined in the network will be the sum of the recurring rates for the individual network elements plus a nonrecurring charge as specified in Exhibit A hereto.
- 17.6.2.2 Rates for new EEL combinations in Density Zone 1 in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs shall be as set forth in Exhibit A hereto; provided, however, that to the extent a rate is not established in Exhibit A, the rate shall be the sum of the recurring and nonrecurring charges for the individual network elements as set forth in Exhibit A to this Attachment, unless otherwise established by the Commission.

17.7 Port/Loop Combinations

- 17.7.1 At e.spire's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in Section 17.7.4 below, that are Already Combined in BellSouth's network except as specified in Sections 17.7.1.1 and 17.7.1.2 below, consistent with the requirements of 47 C.F.R. 315(b) and all applicable FCC and Commission rules and policies.
- 17.7.1.1 BellSouth shall not provide access to combinations of unbundled port and loop network elements in locations where, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.

- 17.7.1.2 In accordance with effective and applicable FCC rules, BellSouth shall not provide unbundled circuit switching in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, of the Atlanta, Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs to e.spire if e.spire's customer has 4 or more DS0 equivalent lines.
- 17.7.2 In Georgia, BellSouth shall provide combinations of port and loop network elements to e.spire regardless of whether or not such combinations are Already Combined except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Section 17.7.1.1 above. In all other states, and subject to Sections 17.7.1.1 and 17.7.1.2 above, BellSouth shall provide combinations of port and loop network elements to e.spire only to the extent such elements are Already Combined.
- 17.7.3 Rates for Combinations of Loop and Port Network Elements
- 17.7.3.1 Rates for combinations of loop and port network elements, as set forth in Section 17.7.4, are provided in Exhibit A of this Attachment.
- 17.7.3.2 Rates for Circuit Switching
- 17.7.3.2.1 Rates for circuit switching, where BellSouth is not required, pursuant to Sections 17.7.1.1 and 17.7.1.2, to provide circuit switching are as set forth in Exhibit A of this Attachment.
- 17.7.4 Combination Offerings
- 17.7.4.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.4.2 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.4.3 2-wire CENTREX port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.4.4. 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.4.5 2-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

17.7.4.6 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

18. Rates

18.1. General

The prices that e.spire shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit C to this Attachment.

18.2. Operations Support Systems ("OSS")

BellSouth shall provide nondiscriminatory access in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act to OSS on an unbundled basis to e.spire for the provision of a telecommunications service. OSS functions consist of preordering, ordering, provisioning, maintenance and repair, and billing functions supported by BellSouth's databases and information. Subject to applicable and effective FCC rules and orders, BellSouth, as part of its duty to provide access to the pre-ordering function, shall provide e.spire with nondiscriminatory access to the same detailed information about the loop that is available to its own retail unit. The rate associated with BellSouth's provision of loop qualification information shall be as set forth in this Attachment.

All Local Service Requests ("LSRs") submitted for products and services under this Attachment will be subject to the OSS charges set forth in the General Terms and Conditions of this Agreement.

18.3 True-up

This section applies only to Tennessee.

The interim prices for Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:

1. The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by a final order (including any appeals) of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body

having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions of Section 16 of the General Terms and Conditions and Attachment 1 of the Agreement.

- 2. The Parties may continue to negotiate toward final prices, but in the event that no such Agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in Section 16 of the General Terms and Conditions and Attachment 1 of the Agreement, so long as they file the resulting Agreement with the Commission as a "negotiated Agreement" under Section 252(e) of the Act.
- 3. A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
 - (a) BellSouth and CLEC is entitled to be a full Party to the proceeding;
 - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
 - (c) It shall include as an issue the geographic deaveraging of network element and other services prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

18.4 Geographic Deaveraging

The Parties acknowledge that this Agreement contains geographically deaveraged rates consistent with 47 C.F.R. 51.507(f) for all states except North Carolina. Upon adoption of geographically deaveraged rates by the North Carolina Utilities Commission, the Parties shall amend this Agreement to include such rates. Prior thereto, rates for the state of North Carolina shall be as set forth in this Agreement.

EXHIBIT A

LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

I. SCOPE

- A. This Agreement sets forth the terms and conditions pursuant to which BST agrees to store in its LIDB certain information at the request of the Local Exchange Company and pursuant to which BST, its LIDB customers and Local Exchange Carrier shall have access to such information. Local Exchange Carrier understands that BST provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of Local Exchange Carrier, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained in the attached Addendum(s) are hereby made a part of this Agreement as if fully incorporated herein.
 - B. LIDB is accessed for the following purposes:
 - 1. Billed Number Screening
 - 2. Calling Card Validation
 - 3. Fraud Control
- C. BST will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BST's LIDB, provided that such information is included in the LIDB query. BST will establish fraud alert thresholds and will notify the Local Exchange Company of fraud alerts so that the Local Exchange Company may take action it deems appropriate. Local Exchange Company understands and agrees BST will administer all data stored in the LIDB, including the data provided by Local Exchange Company pursuant to this Agreement, in the same manner as BST's data for BST's end user customers. BST shall not be responsible to Local Exchange Company for any lost revenue which may result from BST's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BST in its sole discretion from time to time.

Local Exchange Company understands that BST currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearing houses. Local Exchange Company further understands that these billing and collection customers of BST query BST's LIDB to determine whether to accept

various billing options from end users. Additionally, Local Exchange Company understands that presently BST has no method to differentiate between BST's own billing and line data in the LIDB and such data which it includes in the LIDB on Local Exchange Company's behalf pursuant to this Agreement. Therefore, until such time as BST can and does implement in its LIDB and its supporting systems the means to differentiate Local Exchange Company's data from BST's data and the Parties to this Agreement execute appropriate amendments hereto, the following terms and conditions shall apply:

- (a) The Local Exchange Company agrees that it will accept responsibility for telecommunications services billed by BST for its billing and collection customers for Local Exchange Customer's end user accounts which are resident in LIDB pursuant to this Agreement. Local Exchange Company authorizes BST to place such charges on Local Exchange Company's bill from BST and agrees that it shall pay all such charges. Charges for which Local Exchange Company hereby takes responsibility include, but are not limited to, collect and third number calls.
- (b) Charges for such services shall appear on a separate BST bill page identified with the name of the entity for which BST is billing the charge.
- (c) Local Exchange Company shall have the responsibility to render a billing statement to its end users for these charges, but Local Exchange Company's obligation to pay BST for the charges billed shall be independent of whether Local Exchange Company is able or not to collect from the Local Exchange Company's end users.
 - (d) BST shall not become involved in any disputes between Local Exchange Company and the entities for which BST performs billing and collection. BellSouth will not issue adjustments for charges billed on behalf of an entity to Local Exchange Company. It shall be the responsibility of the Local Exchange Company and the other entity to negotiate and arrange for any appropriate adjustments.

II. TERM

This Agreement will be effective as of _______, 199___, and will continue in effect for one year, and thereafter may be continued until terminated by either Party upon thirty (30) days written notice to the other Party.

III. FEES FOR SERVICE AND TAXES

C:\WINDOWS\TEMP\ATT2CLEA. DOC 7/13/00 Version

- A. The Local Exchange Company will not be charged a fee for storage services provided by BST to the Local Exchange Company, as described in Section I of this Agreement.
- B. Sales, use and all other taxes (excluding taxes on BST's income) determined by BST or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by the Local Exchange Company. The Local Exchange Company shall have the right to have BST contest with the imposing jurisdiction, at the Local Exchange Company's expense, any such taxes that the Local Exchange Company deems are improperly levied.

IV. INDEMNIFICATION

To the extent not prohibited by law, each Party will indemnify the other and hold the other harmless against any loss, cost, claim, injury, or liability relating to or arising out of negligence or willful misconduct by the indemnifying Party or its agents or contractors in connection with the indemnifying Party's provision of services, provided, however, that any indemnity for any loss, cost, claim, injury or liability arising out of or relating to errors or omissions in the provision of services under this Agreement shall be limited as otherwise specified in this Agreement. The indemnifying Party under this Section agrees to defend any suit brought against the other Party for any such loss, cost, claim, injury or liability. The indemnified Party agrees to notify the other Party promptly, in writing, of any written claims, lawsuits, or demands for which the other Party is responsible under this Section and to cooperate in every reasonable way to facilitate defense or settlement of claims. The indemnifying Party shall not be liable under this Section for settlement by the indemnified Party of any claim, lawsuit, or demand unless the defense of the claim, lawsuit, or demand has been tendered to it in writing and the indemnifying Party has unreasonably failed to assume such defense.

V. LIMITATION OF LIABILITY

Neither Party shall be liable to the other Party for any lost profits or revenues or for any indirect, incidental or consequential damages incurred by the other Party arising from this Agreement or the services performed or not performed hereunder, regardless of the cause of such loss or damage.

VI. MISCELLANEOUS

- A. It is understood and agreed to by the Parties that BST may provide similar services to other companies.
- B. All terms, conditions and operations under this Agreement shall be performed in accordance with, and subject to, all applicable local, state or federal legal and regulatory tariffs, rulings, and other requirements of the

C:\WINDOWS\TEMP\ATT2CLEA. DOC 7/13/00 Version

federal courts, the U. S. Department of Justice and state and federal regulatory agencies. Nothing in this Agreement shall be construed to cause either Party to violate any such legal or regulatory requirement and either Party's obligation to perform shall be subject to all such requirements.

- C. The Local Exchange Company agrees to submit to BST all advertising, sales promotion, press releases, and other publicity matters relating to this Agreement wherein BST's corporate or trade names, logos, trademarks or service marks or those of BST's affiliated companies are mentioned or language from which the connection of said names or trademarks therewith may be inferred or implied; and the Local Exchange Company further agrees not to publish or use advertising, sales promotions, press releases, or publicity matters without BST's prior written approval.
- D. This Agreement constitutes the entire Agreement between the Local Exchange Company and BST which supersedes all prior Agreements or contracts, oral or written representations, statements, negotiations, understandings, proposals and undertakings with respect to the subject matter hereof.
- E. Except as expressly provided in this Agreement, if any part of this Agreement is held or construed to be invalid or unenforceable, the validity of any other Section of this Agreement shall remain in full force and effect to the extent permissible or appropriate in furtherance of the intent of this Agreement.
- F. Neither Party shall be held liable for any delay or failure in performance of any part of this

 Agreement for any cause beyond its control and without its fault or negligence, such as acts of God, acts of civil or
 military authority, government regulations, embargoes, epidemics, war, terrorist acts, riots, insurrections, fires,
 explosions, earthquakes, nuclear accidents, floods, strikes, power blackouts, volcanic action, other major
 environmental disturbances, unusually severe weather conditions, inability to secure products or services of other
 persons or transportation facilities, or acts or omissions of transportation common carriers.
- G. This Agreement shall be deemed to be a contract made under the laws of the State of Georgia, and the construction, interpretation and performance of this Agreement and all transactions hereunder shall be governed by the domestic law of such State.

FACILITIES BASED ADDENDUM TO LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

This is a Facilities Based Addend	tum to the Line Information Data Base Storage Agreement dated
, 1	99, between BellSouth Telecommunications, Inc. ("BST"), and
	("Local Exchange Company"), effective the day of
, 199	

I. GENERAL

This Addendum sets forth the terms and conditions for Local Exchange Company's provision of billing number information to BST for inclusion in BST's LIDB. BST will store in its LIDB the billing number information provided by Local Exchange Company, and BST will provide responses to on-line, call-by-call queries to this information for purposes specified in Section I.B. of the Agreement.

II. DEFINITIONS

- A. Billing number a number that the Local Exchange Company creates for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.
- B. Line number a ten digit number that identifies a telephone line administered by the Local Exchange Company.
- C. Special billing number a ten digit number that identifies a billing account established by the Local Exchange Company.
 - D. Calling Card number a billing number plus PIN number.
- E. PIN number a four digit security code assigned by the Local Exchange Company which is added to a billing number to compose a fourteen digit calling card number.
- F. Toll billing exception indicator associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by the Local Exchange Company.
- G. Billed Number Screening refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.
- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.

C:\WINDOWS\TEMP\ATT2CLEA. DOC 7/13/00 Version

I. Billing number information - information about billing number, Calling Card number and toll billing exception indicator provided to BST by the Local Exchange Company.

III. RESPONSIBILITIES OF PARTIES

- A. The Local Exchange Company will provide its billing number information to BST's LIDB each business day by a method that has been mutually agreed upon by both Parties.
- B. BST will store in its LIDB the billing number information provided by the Local Exchange Company. Under normal operating conditions, BST shall include the Local Exchange Company's billing number information in its LIDB no later than two business days following BST's receipt of such billing number information, provided that BST shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BST's reasonable control. BST will store in its LIDB an unlimited volume of the Local Exchange Company's working telephone numbers.
- C. BST will provide responses to on-line, call-by-call queries to the stored information for the specific purposes listed in the next paragraph.
- D. BST is authorized to use the billing number information provided by the Local Exchange Company to perform the following functions for authorized users on an on-line basis:
- 1. Validate a 14 digit Calling Card number where the first 10 digits are a line number or special billing number assigned by the Local Exchange Company, and where the last four digits (PIN) are a security code assigned by the Local Exchange Company.
- 2. Determine whether the Local Exchange Company or the subscriber has identified the billing number as one which should not be billed for collect or third number calls, or both.
- E. The Local Exchange Company will provide its own billing number information to BST for storage and to be used for Billed Number Screening and Calling Card Validation. The Local Exchange Company will arrange and pay for transport of updates to BST.

IV. COMPLIANCE

Unless expressly authorized in writing by the Local Exchange Company, all billing number information provided pursuant to this Addendum shall be used for no purposes other than those set forth in this Addendum.

EXHIBIT B

CALLING NAME DELIVERY (CNAM) DATABASE SERVICES

1.00 DEFINITIONS

For the purpose of this Attachment, the following terms shall be defined as:

CALLING NAME DELIVERY DATABASE SERVICE (CNAM) - The ability to associate a name with the calling party number, allowing the end user subscriber (to which a call is being terminated) to view the calling party's name before the call is answered. This service also provides e.spire the opportunity to load and store its subscriber names in the BellSouth CNAM SCPs.

CALLING PARTY NUMBER (CPN) - The number of the calling party that is delivered to the terminating switch using common channel signaling system 7 (CCS7) technology, and that is contained in the Initial Address Message (IAM) portion of the CCS7 call setup.

COMMON CHANNEL SIGNALING SYSTEM 7 (CCS7) - A network signaling technology in which all signaling information between two or more nodes is transmitted over high-speed data links, rather than over voice circuits.

SERVICE CONTROL POINTs (SCPs) - The real-time data base systems that contain the names to be provided in response to queries received from CNAM SSPs.

SERVICE MANAGEMENT SYSTEM (SMS) - The main operations support system of CNAM DATABASE SERVICE. CNAM records are loaded into the SMS, which in turn downloads into the CNAM SCP.

SERVICE SWITCHING POINTs (SSPs) - Features of computerized switches in the telephone network that determine that a terminating line has subscribed to CNAM service, and then communicate with CNAM SCPs in order to provide the name associated with the calling party number.

SUBSYSTEM NUMBER (SSN) - The address used in the Signaling Connection Control Part (SCCP) layer of the SS7 protocol to designate an application at an end signaling point. A SSN for CNAM at the end office designates the CNAM application within the end office. BellSouth uses the CNAM SSN of 232.

2.0 ATTACHMENT

C:\WINDOWS\TEMP\ATT2CLEA. DOC 7/13/00 Version

- 2.01 This Attachment contains the terms and conditions where BellSouth will provide to the e.spire access to the BellSouth CNAM SCP for query or record storage purposes.
- 2.02 e.spire shall submit to BellSouth a notice of its intent to access and utilize BellSouth CNAM Database Services pursuant to the terms and conditions of this Attachment. Said notice shall be in writing, no less than 60 days prior to e.spire's access to BellSouth's CNAM Database Services and shall be addressed to e.spire's Account Manager.

3.00 PHYSICAL CONNECTION AND COMPENSATION

- 3.01 BellSouth's provision of CNAM Database Services to e.spire requires interconnection from e.spire to BellSouth CNAM Service Control Points (SCPs). Such interconnections shall be established pursuant to Attachment 3 of this Agreement. The appropriate charge for access to and use of the BellSouth CNAM Database service shall be as set forth in this Attachment.
- 3.02 In order to formulate a CNAM query to be sent to the BellSouth CNAM SCP, e.spire e.spire shall provide its own CNAM SSP. e.spire's CNAM SSPs must be compliant with TR-NWT-001188, "CLASS Calling Name Delivery Generic Requirements".
- 3.03 If e.spire elects to access the BellSouth CNAM SCP via a third party CCS7 transport provider, the third party CCS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia (formerly BellCore)'s CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points (LSTPs) serving the BellSouth CNAM SCPs that e.spire desires to query.

3.04 Out-Of-Region Customers

If the customer queries the BellSouth CNAM SCP via a third party national SS7 transport provider, the third party SS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Bellcore's CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish SS7 interconnection at one or more of the BellSouth Gateway Signal Transfer Points (STPs). The payment of all costs associated with the transport of SS7 signals via a third party will be established by mutual agreement of the Parties and writing shall, by this reference become an integral part of this Agreement.

4.00 CNAM RECORD INITIAL LOAD AND UPDATES

4.01 The mechanism to be used by e.spire for initial CNAM record load and/or updates shall be determined by mutual agreement. The initial load and all updates shall be provided by e.spire in the BellSouth specified format and shall contain records for every working

- telephone number that can originate phone calls. It is the responsibility of e.spire to provide accurate information to BellSouth on a current basis.
- 4.02 Updates to the SMS shall occur no less than once a week, reflect service order activity affecting either name or telephone number, and involve only record additions, deletions or changes.
- 4.03 e.spire CNAM records provided for storage in the BellSouth CNAM SCP shall be available, on a SCP query basis only, to all Parties querying the BellSouth CNAM SCP. Further, CNAM service shall be provided by each Party consistent with state and/or federal regulation.

		HICOC	AL		^	L/V		MC	NC	SC	TN
NIDs	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	50	IN
	(all times) man manth	LINDAY	NIA	£4.00	NIA	£4.00	NIA	NIA	#0.50	NIA	
	(all types), per month	UNDAX	NA	\$1.08	NA	\$1.80	NA	NA	\$0.52	NA	\$0.56
Insi	allation of 2-Wire/4Wire CLEC NID	UNDAX		# 70.00							
+	NRC - 1st	UNDAX	NA	\$70.32	NA	NA	NA	NA	NA	NA	NA
44	NRC - Add'l	UNDAX	NA	\$54.35	NA	NA	NA	NA	NA	NA	NA
	to NID Cross Connect, 2-Wire or 4-Wire, NRC	UNDC2	NA	\$6.15	NA	NA	NA	NA	NA	NA	NA
NID	per 2-Wire Analog VG Loop, Per Month	UNDAX	\$1.18	NA	\$1.10	NA	\$1.09	\$1.22	\$1.01	\$1.13	NA
Ш	NRC - 1st	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.36	NA
Ш	NRC - Add'l	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.36	NA
Ш	NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
	NRC - Disconnect Charge - Add'l	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
T	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NID	per 4-Wire Analog VG Loop, Per Month	UNDAX	\$1.30	NA	\$1.21	NA	\$1.22	\$1.34	\$1.14	\$1.25	NA
\prod	NRC - 1st	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
\Box	NRC - Add'l	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
11	NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
T	NRC - Disconnect Charge - Add'l	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
+	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA
+	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
+	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA NA	NA	NA
NIL	per 2-Wire ISDN Digital VG Loop, Per Month	UNDAX	\$1.18	NA NA	\$1.10	NA NA	\$1.08	\$1.22	\$1.01	\$1.13	NA NA
1110	NRC - 1st	UNDAX	\$1.44	NA NA	\$2.10	NA NA	\$2.02	\$2.84	\$1.42	\$1.36	NA NA
++	NRC - Add'l	UNDAX	\$1.44	NA NA	\$2.10	NA NA	\$2.02	\$2.84	\$1.42	\$1.36	NA NA
++	NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA NA	Ψ2.10 NA	NA NA	\$2.02	\$2.84	Ψ1.42 NA	\$1.30 NA	NA NA
++	NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA NA	NA NA	NA NA	\$2.01	\$2.84	NA NA	NA NA	NA NA
++	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$1.44	NA NA	\$18.94	NA NA	\$18.14	\$2.84	\$26.94	\$44.42	NA NA
++	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$12.97	NA NA	\$8.42	NA NA	\$8.06	\$25.52 \$11.34		\$13.55	NA NA
++									\$12.76		
- LUE	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA ©4.40	NA	\$11.41	\$16.06	NA C4 04	NA C4.40	NA
NID	per 2-Wire Asymmetrical Dig Subscriber Line (ADSL) Loop, Per Mo.	UNDAX	\$1.18	NA	\$1.10	NA	\$1.09	\$1.22	\$1.01	\$1.13	NA
+	NRC - 1st	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.36	NA
+	NRC - Add'l	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.36	NA
ш	NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
$+\!\!+\!\!\!+$	NRC - Disconnect Charge - Add'l	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
$+\!\!+\!\!\!+$	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA
$\bot\!\!\!\!\bot$	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
$\perp \! \! \perp \! \! \! \! \! \perp$	NRC - Incremental Charge - Manual Service Order - Disconnect -1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NID	per 2-Wire High Bit Rate Dig Subscriber Line (HDSL) Loop	UNDAX	\$1.18	NA	\$1.10	NA	\$1.09	\$1.22	\$1.01	\$1.13	NA
Ш	NRC - 1st	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.36	NA
\perp	NRC - Add'l	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.36	NA
	NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
⊥∏	NRC - Disconnect Charge - Add'l	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA
\prod	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
\Box	NRC - Incremental Charge - Manual Service Order - Disconnect -1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NIC	per 4-Wire High Bit Rate Dig Subscriber Line (HDSL) Loop	UNDAX	\$1.30	NA	\$1.21	NA	\$1.21	\$1.34	\$1.14	\$1.25	NA
ΤĪ	NRC - 1st	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
+	NRC - Add'l	UNDAX	\$1.44	NA NA	\$2.10	NA NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
+	NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA	NA NA	NA	\$2.01	\$2.84	NA	NA	NA
		UNDAX	\$1.44	NA	NA NA	NA NA	\$2.01	\$2.84	NA NA	NA NA	NA
++	NRC - Disconnect Charge - Add'l	UNUAX									

		OTHER SERVICE								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NID per 4-Wire 56 Kbps Dig Grade Loop	UNDAX	\$1.30	NA	\$1.21	NA	\$1.21	\$1.34	\$1.14	\$1.25	NA
NRC - 1st	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
NRC - Add'l	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Disconnect Charge - Add'l	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NID per 4-Wire 64 Kbps Dig Grade Loop	UNDAX	\$1.30	NA	\$1.21	NA	\$1.21	\$1.34	\$1.14	\$1.25	NA
NRC - 1st	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
NRC - Add'l	UNDAX	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$1.42	\$1.35	NA
NRC - Disconnect Charge - 1st	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Disconnect Charge - Add'l	UNDAX	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Incremental Charge - Manual Svc Ord - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA
NRC - Incremental Charge - Manual Svc Ord - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
NRC - Incremental Charge - Manual Svc Ord - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NID per 2-Wire Unbundled Copper Loop, per month	UNDAX	\$1.55	\$1.55	\$1.55	\$1.55	\$1.55	\$1.55	\$1.55	\$1.55	\$1.55
NRC - 1st	UNDAX	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60
NRC - Add'l	UNDAX	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60	\$5.60
NRC - Disconnect Charge - 1st	UNDAX	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Disconnect Charge - Add'l	UNDAX	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Svc. Ord - 1st	SOMAN	\$47.00	\$47.00	\$47.00	\$47.00	\$47.00	\$47.00	\$47.00	\$47.00	\$47.00
NRC - Incremental Charge - Manual Svc. Ord - Add'l	SOMAN	\$21.00	\$21.00	\$21.00	\$21.00	\$21.00	\$21.00	\$21.00	\$21.00	\$21.00
NRC - Incremental Charge - Manual Svc. Ord Disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
Nonrecurring Charge - customer transfer, feature additions, changes (1)		\$5.00	NA	NA	NA	NA	\$5.00	NA	NA	NA
LOOP, EXCLUDING NID										
2-Wire Analog VG Loop (Standard), per month	TBD	NA	NA	NA	\$18.20	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$86.08	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$58.57	NA	NA	NA	NA	NA
2-Wire Analog VG Loop (Customized), per month	TBD	NA	NA	NA	\$21.41	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$236.75	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$177.10	NA	NA	NA	NA	NA
4-Wire Analog VG Loop (Standard), per month	TBD	NA	NA	NA	\$26.38	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$457.14	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$348.83	NA	NA	NA	NA	NA
2-Wire ISDN Digital Grade Loop (Standard), per month	TBD	NA	NA	NA	\$29.65	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$541.28	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$431.61	NA	NA	NA	NA	NA
2-Wire ADSL Loop (Standard), per month	TBD	NA	NA	NA	\$10.63	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$713.50	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$609.44	NA	NA	NA	NA	NA
2-Wire HDSL Loop (Standard), per month	TBD	NA	NA	NA	\$7.40	NA	NA	NA	NA	NA
NRC - 1st		NA	NA	NA	\$713.50	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA NA	\$609.44	NA	NA.	NA	NA	NA
4-Wire HDSL Loop (Standard), per month	TBD	NA NA	NA	NA NA	\$9.70	NA	NA.	NA.	NA NA	NA.
NRC - 1st		NA	NA	NA	\$748.93	NA	NA	NA	NA	NA
NRC - Add'l		NA	NA	NA	\$646.17	NA	NA	NA	NA	NA
		1					1	i		
		1	1		1	L	1	1		

	AND	OTHER SERVICE	S							
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
LOOP, INCLUDING NID										
2-Wire Analog VG Loop										
RC - Statewide, per month	UEAL2	NA	NA	NA	NA	NA	NA	\$16.71	NA	\$18.00
RC - Zone 1, per month (Note 2)	TBD	NA	\$13.75	NA	NA	NA	NA	TBD	NA	\$15.54
RC - Zone 2, per month (Note 2)	TBD	NA	\$20.13	NA	NA	NA	NA	TBD	NA	\$19.55
RC - Zone 3, per month (Note 2)	TBD	NA	\$44.40	NA	NA	NA	NA	TBD	NA	\$28.02
RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - 1st	UEAL2	NA	\$140.00	NA	NA	NA	NA	\$86.50	NA	\$58.50
NRC - Add'l	UEAL2	NA	\$42.00	NA	NA	NA	NA	\$27.80	NA	\$31.00
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA	\$55.00	NA	NA	NA	NA	\$55.00	NA	\$55.00
2-Wire Analog VG Loop-SL1										
RC - Statewide, per month	UEAL2	NA	NA	NA	NA	NA	NA	\$15.88	NA	NA
RC - Zone 1, per month (Note 2)	TBD	\$15.24	\$13.75	\$14.21	\$14.79	\$14.96	\$15.58	TBD	\$18.48	\$15.92
RC - Zone 2, per month (Note 2)	TBD	\$24.75	\$20.13	\$16.41	\$27.68	\$25.69	\$20.65	TBD	\$27.87	\$20.79
RC - Zone 3, per month (Note 2)	TBD	\$44.85	\$44.40	\$26.08	\$47.78	\$52.47	\$29.51	TBD	\$36.91	\$27.18
RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	\$38.94	NA	NA	NA
NRC - 1st	UEAL2	\$59.03	\$80.00	\$42.54	NA	\$40.69	\$59.25	\$57.99	\$70.44	\$78.93
NRC - Add'l	UEAL2	\$43.14	\$55.00	\$31.33	NA	\$29.96	\$43.67	\$42.37	\$44.05	\$50.98
NRC - Disconnect Charge - 1st	UEAL2	\$15.21	NA	NA	NA	\$16.48	\$16.35	NA	NA	NA
NRC - Disconnect Charge - Add'l	UEAL2	\$3.22	NA	NA	NA	\$3.36	\$4.06	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.22	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NRC - Manual Order Coordination - 1st	TBD	NA	NA	NA	NA	NA	NA	\$61.38	NA	NA
NRC - Manual Order Coordination - addl	TBD	NA	NA	NA	NA	NA	NA	\$61.38	NA	NA
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	TBD	NA	NA	NA	NA	NA	NA	\$45.34	NA	NA
NRC - Loop Make-Up	UEANM	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
2-Wire Analog VG Loop-SL2 w/loop or ground start signaling										
RC - Statewide, per month	UEAL2	NA	NA	NA	NA	NA	NA	\$19.50	NA	NA
RC - Zone 1, per month (Note 2)	TBD	\$17.95	\$13.75	\$16.84	\$17.27	\$17.65	\$18.35	TBD	\$21.57	\$15.92
RC - Zone 2, per month (Note 2)	TBD	\$29.16	\$20.13	\$19.45	\$32.32	\$30.32	\$24.33	TBD	\$32.53	\$20.79
RC - Zone 3, per month (Note 2)	TBD	\$52.84	\$44.40	\$30.92	\$55.78	\$61.93	\$34.77	TBD	\$43.08	\$27.18
RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	\$45.88	NA	NA	NA
NRC - 1st	UEAL2	\$145.46	\$140.00	\$104.17	NA	\$99.69	\$144.01	\$142.97	\$178.12	\$192.97
NRC - Add'l	UEAL2	\$108.40	\$42.00	\$78.10	NA	\$74.73	\$107.70	\$106.56	\$128.80	\$140.72
NRC - Disconnect Charge - 1st	UEAL2	\$40.31	NA	NA	NA	\$28.73	\$40.98	NA	NA	NA
NRC - Disconnect Charge - Add'l	UEAL2	\$26.01	NA	NA	NA	\$18.87	\$26.95	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA 055.00	NA To 1 00	NA	\$11.41	\$26.95	NA 0.45.04	NA 0.45,40	NA 055.00
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
2-Wire Analog VG Loop-SL2 w/ reverse battery signaling	UEAR2	NA	NIA	NA	NIA	NIA	NA	£40.50	NIA	NIA
RC - Statewide, per month			NA C40.75		NA C47.07	NA C47.05		\$19.50	NA CO4.57	NA ©45.00
RC - Zone 1, per month (Note 2)	TBD	\$17.95	\$13.75	\$16.84	\$17.27	\$17.65	\$18.35	TBD	\$21.57	\$15.92
RC - Zone 2, per month (Note 2)	TBD	\$29.16	\$20.13	\$19.45	\$32.32	\$30.32	\$24.33	TBD TBD	\$32.53	\$20.79
RC - Zone 3, per month (Note 2)	TBD TBD	\$52.84	\$44.40	\$30.92	\$55.78	\$61.93	\$34.77	1	\$43.08	\$27.18
RC - Zone 4, per month (Note 2)		NA \$1.45.46	NA (\$4.40.00)	NA C104.47	NA NA	NA COO CO	\$45.88	NA	NA ************************************	NA ©400.07
NRC - 1st NRC - Add'l	UEAR2 UEAR2	\$145.46 \$108.40	\$140.00 \$42.00	\$104.17 \$78.10	NA NA	\$99.69 \$74.73	\$144.01 \$107.70	\$142.97 \$106.56	\$178.12 \$128.80	\$192.97 \$140.72
NRC - Disconnect Charge - 1st	UEAR2	\$40.31	\$42.00 NA	\$76.10 NA	NA NA	\$28.73	\$40.98	NA	NA	\$140.72 NA
NRC - Disconnect Charge - 1st	UEAR2	\$26.01	NA NA	NA NA	NA NA	\$18.87	\$26.95	NA NA	NA NA	NA NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA NA	\$18.94	NA NA	\$18.14	\$25.52	\$29.64	\$44.42	NA NA
T T Intro-inciding Charge - Manual Service Order - 15t	SOIVIAIN	φ21.31	INA	φ10.54	INA	φ10.1 4	φ20.02	φ ∠ ઝ.υ 4	φ 44.4 2	INA

			OTHER SERVICE			1616					
DE	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
<u> </u>	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$26.95	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOCL	\$45.99	\$55.00	\$34.22	NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
2-1	Nire Analog VG Loop (Standard)										
	RC - Statewide, per month	UEAL2	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - Zone 1, per month (Note 2)	TBD	NA	NA	NA	\$14.79	NA	NA	NA	NA	NA
	RC - Zone 2, per month (Note 2)	TBD	NA	NA	NA	\$27.68	NA	NA	NA	NA	NA
	RC - Zone 3, per month (Note 2)	TBD	NA	NA	NA	\$47.78	NA	NA	NA	NA	NA
	RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 1st	UEAL2	NA	NA	NA	\$86.08	NA	NA	NA	NA	NA
	NRC - Add'l	UEAL2	NA	NA	NA	\$58.57	NA	NA	NA	NA	NA
	NRC - Loop Make-up	UEANM	NA	NA	NA	TBD	NA	NA	NA	NA	NA
	NRC - Manual Order Coordination	UEAMC	NA	NA	NA	TBD	NA	NA	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA	NA	NA	\$55.00	NA	NA	NA	NA	NA
2-\	Wire Analog VG Loop (Customized), w/ loop or ground start signaling										
	RC - Statewide, per month	UEAL2	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - Zone 1, per month (Note 2)	TBD	NA	NA	NA	\$17.27	NA	NA	NA	NA	NA
	RC - Zone 2, per month (Note 2)	TBD	NA	NA	NA	\$32.32	NA	NA	NA	NA	NA
	RC - Zone 3, per month (Note 2)	TBD	NA	NA	NA	\$55.78	NA	NA	NA	NA	NA
	RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 1st	UEAL2	NA	NA	NA	\$236.75	NA	NA	NA	NA	NA
	NRC - Add'l	UEAL2	NA	NA	NA	\$177.10	NA	NA	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA NA	NA	NA	\$55.00	NA	NA	NA NA	NA NA	NA
2-1	Wire Analog VG Loop (Customized), w/ reverse battery signaling	00002	147	1471	107	ψου.σσ	14/	1471	1471	10/	1471
H2.	RC - Statewide, per month	UEAR2	NA	NA	NA	NA	NA	NA	NA	NA	NA
-	RC - Zone 1, per month (Note 2)	TBD	NA NA	NA NA	NA	\$17.27	NA	NA NA	NA NA	NA NA	NA NA
\vdash	RC - Zone 2, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$32.32	NA NA	NA NA	NA NA	NA NA	NA NA
-	RC - Zone 3, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$55.78	NA NA	NA NA	NA NA	NA NA	NA NA
-	RC - Zone 4, per month (Note 2)	TBD	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA
-	NRC - 1st	UEAR2	NA NA	NA NA	NA NA	\$236.75	NA NA	NA NA	NA NA	NA NA	NA NA
-	NRC - Add'l		NA NA		NA NA			NA NA	NA NA	NA NA	NA NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	UEAR2 OCOSL	NA NA	NA NA	NA NA	\$177.10 \$55.00	NA NA	NA NA	NA NA	NA NA	NA NA
4.		UCUSL	NA	INA	INA	\$55.00	NA	INA	NA	NA	INA
4-1	Wire Analog VG Loop	115014	NIA	N10	NIA.	NIA	NIA	NI A	607.40	NIA.	N10
Н-	RC - Statewide, per month	UEAL4	NA COA OA	NA TO 4.00	NA Coo.oo	NA	NA COARC	NA Too oo	\$27.49	NA COO 47	NA ©45.00
	RC - Zone 1, per month (Note 2)	TBD	\$24.01	\$24.26	\$22.26	NA	\$24.36	\$22.38	TBD	\$29.47	\$15.92
	RC - Zone 2, per month (Note 2)	TBD	\$39.00	\$35.51	\$25.70	NA	\$41.85	\$29.67	TBD	\$44.44	\$20.79
-	RC - Zone 3, per month (Note 2)	TBD	\$70.67	\$78.35	\$40.85	NA NA	\$85.47	\$42.40	TBD	\$58.85	\$27.18
-	RC - Zone 4, per month (Note 2)	TBD	NA C202.70	NA C144.00	NA COOC OF	NA	NA ©400.40	\$55.96	NA COOR 47	NA Cana an	NA ©E0.E0
-	NRC - 1st	UEAL4	\$293.70	\$141.00	\$206.95	NA	\$198.10	\$289.06	\$288.47	\$383.39	\$58.50
4	NRC - Add'l	UEAL4	\$241.76	\$43.00	\$170.57	NA	\$163.26	\$238.19	\$237.45	\$286.77	\$31.00
4	NRC - Disconnect Charge - 1st	UEAL4	\$108.96	NA	NA	NA	\$74.27	\$108.14	NA	NA	NA
	NRC - Disconnect Charge - Add'l	UEAL4	\$57.01	NA	NA	NA	\$39.44	\$57.28	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
4-\	Nire Analog VG Loop (Standard)										1
	RC - Statewide, per month	UEAL4	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - Zone 1, per month (Note 2)	TBD	NA	NA	NA	\$20.92	NA	NA	NA	NA	NA
	RC - Zone 2, per month (Note 2)	TBD	NA	NA	NA	\$39.14	NA	NA	NA	NA	NA
	RC - Zone 3, per month (Note 2)	TBD	NA	NA	NA	\$67.56	NA	NA	NA	NA	NA
	RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA

DE	ESCRIPTION	USOC	OTHER SERVICE AL	FL	GA	KY	LA	MS	NC	sc	TN
1	NRC - 1st	UEAL4	NA NA	NA NA	NA NA	\$457.14	NA NA	NA NA	NA NA	NA NA	NA NA
-	NRC - Add'l	UEAL4	NA NA	NA NA	NA NA	\$348.83	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA NA	NA NA	NA NA	\$55.00	NA NA	NA NA	NA NA	NA NA	NA NA
2.1	Wire ISDN Digital Grade Loop	UCUSL	INA	INA	INA	\$55.00	INA	INA	INA	INA	INA
2-1	RC - Statewide, per month	U1L2X	NA	NA	NA	NA	NA	NA	\$24.98	NA	NA
		TBD	\$23.23	\$32.34	\$21.89	\$23.66	\$21.15	\$21.86	\$24.96 TBD	\$26.68	\$15.92
	RC - Zone 1, per month (Note 2)	TBD	\$23.23	\$32.34 \$47.35	\$21.89	\$44.28	\$36.22	\$21.86	TBD	\$40.24	\$15.92
	RC - Zone 2, per month (Note 2) RC - Zone 3, per month (Note 2)	TBD	\$68.38		\$25.27 \$40.17	\$44.28 \$76.42	\$36.22 \$74.19		TBD	\$40.24 \$53.29	\$20.79
		TBD		\$104.47				\$41.40	NA NA		
-	RC - Zone 4, per month (Note 2)	U1L2X	NA \$331.85	NA \$306.00	NA \$233.38	NA NA	NA \$223.27	\$54.64 \$326.38	\$325.91	NA \$423.04	NA \$58.50
-	NRC - 1st	U1L2X	\$255.87			NA NA		\$326.38		\$423.04	\$31.00
	NRC - Add'l NRC - Disconnect Charge - 1st	U1L2X		\$283.00	\$180.35	NA NA	\$172.63		\$251.31 NA		\$31.00 NA
			\$108.95	NA	NA NA	NA NA	\$74.27	\$108.14	NA NA	NA NA	NA NA
-	NRC - Disconnect Charge - Add'l	U1L2X SOMAN	\$57.01	NA NA	\$18.94	NA NA	\$39.44	\$57.27	\$26.94	NA \$44.42	NA NA
-	NRC - Incremental Charge - Manual Service Order - 1st		\$27.37				\$18.14	\$25.52			
-	NRC - Incremental Charge - Manual Service Order - Add'l NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN SOMAN	\$12.97	NA NA	\$8.42	NA NA	\$8.06	\$11.34	\$12.76	\$13.55	NA NA
-			\$17.77	NA #FF.00	NA Co. L. CO.		\$11.41	\$16.06	NA C45.04	NA ************************************	
2.	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	\$55.00	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
2-1	Wire ISDN Digital Grade Loop (Standard) RC - Statewide, per month	U1L2X	NA	NA	NA	NA	NIA.	NA	NA	NA	NA
		TBD	NA NA				NA NA	NA NA	NA NA	NA NA	NA NA
-	RC - Zone 1, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$23.66 \$44.28		NA NA	NA NA	NA NA	NA NA
-	RC - Zone 2, per month (Note 2)	TBD	NA NA	NA NA	NA NA		NA NA	NA NA	NA NA	NA NA	NA NA
	RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$76.42 NA	NA NA	NA NA	NA NA	NA NA	NA NA
		U1L2X	NA NA	NA NA			NA NA	NA NA		NA NA	NA NA
-	NRC - 1st NRC - Add'l	U1L2X	NA NA	NA NA	NA NA	\$541.28 \$431.61	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA NA		NA NA		NA NA	NA NA	NA NA	NA NA	NA NA
		UCUSL	INA	NA	INA	\$55.00	NA	INA	NA	INA	NA
2-1	Wire Asymmetrical Dig Subscriber Line (ADSL) Compatible Loop RC - Statewide, per month	UAL2X	NA	NA	NA	NA	NA	NA	\$14.60	NA	\$18.46
-	1 1 '1	TBD	\$12.09	\$12.78	\$11.23	NA NA	\$11.90	\$10.87	\$14.60 TBD	\$17.10	\$18.46
	RC - Zone 1, per month (Note 2) RC - Zone 2, per month (Note 2)	TBD	\$12.09	\$12.78		NA NA		\$10.87	TBD	\$17.10	
		TBD	\$19.64		\$12.97 \$20.62	NA NA	\$20.43	\$14.40	TBD		\$20.05
	RC - Zone 3, per month (Note 2) RC - Zone 4, per month (Note 2)	TBD	\$35.59 NA	\$41.29 NA	\$20.62 NA	NA NA	\$41.73 NA	\$20.58	NA NA	\$34.15 NA	\$28.74 NA
	NRC - 1st	UAL2X	\$514.21	\$113.85	\$359.73	NA NA	\$343.13	\$504.82	\$504.90	\$600.61	\$640.79
	NRC - Add'l	UAL2X UAL2X	\$464.58	\$99.61	\$359.73	NA NA	\$343.13	\$504.82 \$456.24	\$456.17	\$507.33	\$541.94
	NRC - Disconnect Charge - 1st	UAL2X UAL2X	\$464.58 \$106.65	\$99.61 NA	\$325.15 NA	NA NA	\$72.54	\$456.24	NA	\$507.33 NA	\$541.94 NA
	NRC - Disconnect Charge - 1st NRC - Disconnect Charge - Add'l	SOMAN	\$56.98	NA NA	NA NA	NA NA	\$39.42	\$57.25	NA NA	NA NA	NA NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA NA	\$18.94	NA NA	\$18.14	\$25.52	\$26.94	\$44.42	NA NA
+	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$12.97	NA NA	\$8.42	NA NA	\$8.06	\$11.34	\$12.76	\$13.55	NA NA
+	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA NA	Φ0.42 NA	NA NA	\$11.41	\$16.06	NA	\$13.55 NA	NA NA
+	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	NA NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
2.1	Wire ADSL Loop (Standard)	UCUSL	Ф45.99	\$55.00	φ34.ZZ	INA	φ32.77	Φ45.27	ф45.34	Ф45.43	φ55.00
2-1	RC - Statewide, per month	UAL2X	NA	NA	NA	NA	NA	NA	NA	NA	NA
-	RC - Statewide, per month RC - Zone 1, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$8.79	NA NA	NA NA	NA NA	NA NA	NA NA
+	RC - Zone 1, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$16.46	NA NA	NA NA	NA NA	NA NA	NA NA
+	RC - Zone 2, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$16.46	NA NA	NA NA	NA NA	NA NA	NA NA
+	RC - Zone 4, per month (Note 2)	TBD	NA NA	NA NA	NA NA	\$28.40 NA	NA NA	NA NA	NA NA	NA NA	NA NA
+	NRC - 1st	UAL2X	NA NA	NA NA	NA NA	\$713.50	NA NA	NA NA	NA NA	NA NA	NA NA
╁	NRC - Add'l	UAL2X UAL2X	NA NA	NA NA	NA NA	\$609.44	NA NA	NA NA	NA NA	NA NA	NA NA
+	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA NA	NA NA	NA NA		NA NA	NA NA	NA NA	NA NA	NA NA
2 1	Wire High Bit Rate Dig Subscriber Line (HDSL) Compatible Loop	UCUSL	INA	INA	INA	\$55.00	INA	INA	INA	INA	INA
2-1	RC - Statewide, per month	UHL2X	NA	NA	NA	NA	NA	NA	\$11.98	NA	\$13.46
-	RC - Statewide, per month RC - Zone 1, per month (Note 2)	TBD	\$9.41	\$9.80	\$7.88	\$6.29	\$8.97	\$8.50	\$11.98 TBD	\$12.21	\$13.46 \$11.62
	I Inc - Zone 1, per month (Note 2)	ואט	Ф 9.4 1	Φ 3 .80	φ1.00	⊅ 0.∠9	φο.97	υC.οφ	עמו	φ12.Z1	∠0.11¢

-	CODITION		OTHER SERVICE			L/V		140	NO		TN
DE	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	RC - Zone 2, per month (Note 2)	TBD	\$15.29	\$14.35	\$9.09	\$11.78	\$15.41	\$11.26	TBD	\$18.41	\$14.62
	RC - Zone 3, per month (Note 2)	TBD	\$27.70	\$31.65	\$14.46	\$20.33	\$31.48	\$16.10	TBD	\$24.39	\$20.96
	RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	\$21.25	NA	NA	NA
	NRC - 1st	UHL2X	\$514.21	\$113.85	\$359.73	NA	\$343.13	\$504.82	\$504.90	\$600.61	\$640.79
	NRC - Add'l	UHL2X	\$464.58	\$99.61	\$325.15	NA	\$310.03	\$456.24	\$456.17	\$507.33	\$541.94
	NRC - Disconnect Charge - 1st	UHL2X	\$106.65	NA	NA	NA	\$72.54	\$105.86	NA	NA	NA
	NRC - Disconnect Charge - Add'l	UHL2X	\$56.98	NA	NA	NA	\$39.42	\$57.25	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
2-V	Vire HDSL Loop (Standard)							·		·	
	RC - Statewide, per month	UHL2X	NA	NA	NA	NA	NA	NA	NA	NA	NA
H	RC - Zone 1, per month (Note 2)	TBD	NA	NA	NA	\$6.29	NA	NA	NA	NA	NA
H	RC - Zone 2, per month (Note 2)	TBD	NA	NA	NA.	\$11.78	NA NA	NA.	NA NA	NA NA	NA
Н	RC - Zone 3, per month (Note 2)	TBD	NA.	NA	NA	\$20.33	NA	NA NA	NA NA	NA NA	NA
Н	RC - Zone 4, per month (Note 2)	TBD	NA NA	NA NA	NA NA	Ψ20.33 NA	NA NA	NA NA	NA NA	NA NA	NA NA
H	NRC - 1st	UHL2X	NA NA	NA NA	NA NA	\$713.50	NA NA	NA NA	NA NA	NA NA	NA NA
H	NRC - Add'l	UHL2X	NA NA	NA NA	NA NA	\$609.44	NA NA	NA NA	NA NA	NA NA	NA NA
H	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA NA	NA NA	NA NA	\$55.00	NA NA	NA NA	NA NA	NA NA	NA NA
4.1		UCUSL	INA	INA	INA	\$55.00	NA	INA	NA	NA	INA
4-V	Vire High Bit Rate Dig Subscriber Line (HDSL) Compatible Loop RC - Statewide, per month	LILII AV	NA	NA	NA	NIA	NIA	NA	£42.07	NA	£47.04
\vdash		UHL4X				NA	NA O40.07		\$13.97		\$17.91
\Box	RC - Zone 1, per month (Note 2)	TBD	\$11.52	\$14.75	\$10.39	NA	\$12.97	\$10.36	TBD	\$16.21	\$15.46
$oxed{\sqcup}$	RC - Zone 2, per month (Note 2)	TBD	\$18.71	\$21.59	\$12.00	NA	\$21.76	\$13.73	TBD	\$24.45	\$19.46
$oxed{\sqcup}$	RC - Zone 3, per month (Note 2)	TBD	\$33.90	\$47.64	\$19.07	NA	\$44.44	\$19.62	TBD	\$32.38	\$27.88
Ш	RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	\$25.90	NA	NA	NA
Ш	NRC - 1st	UHL4X	\$541.13	\$116.91	\$378.86	NA	\$361.45	\$531.21	\$531.35	\$625.11	\$666.70
	NRC - Add'l	UHL4X	\$491.50	\$101.71	\$344.28	NA	\$328.35	\$482.63	\$482.62	\$532.78	\$568.86
	NRC - Disconnect Charge - 1st	UHL4X	\$106.65	NA	NA	NA	\$72.54	\$105.86	NA	NA	NA
Ш	NRC - Disconnect Charge - Add'l	UHL4X	\$56.98	NA	NA	NA	\$39.42	\$57.25	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.06	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect -1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
4-V	Vire HDSL Loop (Standard)										i
	RC - Statewide, per month	UHL4X	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - Zone 1, per month (Note 2)	TBD	NA	NA	NA	\$7.68	NA	NA	NA	NA	NA
	RC - Zone 2, per month (Note 2)	TBD	NA	NA	NA	\$14.38	NA	NA	NA	NA	NA
П	RC - Zone 3, per month (Note 2)	TBD	NA	NA	NA	\$24.82	NA	NA	NA	NA	NA
	RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	NRC - 1st	UHL4X	NA	NA	NA	\$748.93	NA	NA	NA	NA	NA
H	NRC - Add'l	UHL4X	NA	NA	NA	\$646.17	NA	NA	NA	NA	NA
H	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	NA	NA	NA	\$55.00	NA NA	NA	NA NA	NA.	NA
4-V	Vire DS1 Digital Loop					******		1			
HŤ	RC - Statewide, per month	USLXX	NA	NA	NA	NA	NA	NA	\$62.78	NA	TBD
H	RC - Zone 1, per month (Note 2)	TBD	\$51.74	\$64.69	\$55.53	\$50.26	\$56.32	\$50.99	TBD	\$59.61	TBD
H	RC - Zone 2, per month (Note 2)	TBD	\$84.05	\$94.71	\$64.13	\$94.06	\$96.73	\$67.58	TBD	\$89.90	TBD
HH	RC - Zone 3, per month (Note 2)	TBD	\$152.29	\$208.93	\$101.93	\$162.34	\$197.57	\$96.58	TBD	\$119.06	TBD
+	RC - Zone 4, per month (Note 2)	TBD	\$152.29 NA	\$206.93 NA	NA	\$162.34 NA	NA	\$127.47	NA NA	NA	NA NA
Н	NRC - 1st	USLXX	\$610.13	\$540.00	\$429.98	\$849.80	\$410.38	\$127.47	\$714.84	\$715.77	TBD
H - H	NRC - 1st NRC - Add'l										TBD
H		USLXX	\$380.26	\$465.00	\$268.18	\$523.27	\$255.48	\$373.90	\$421.47	\$421.50	
	NRC - Disconnect Charge - 1st	USLXX	\$134.77	NA	NA	NA	\$92.35	\$133.53	NA	NA	NA

BELLSOUTH/e.spire RATES NETWORK ELEMENTS

			OTHER SERVICE	S							
DI	ESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	NRC - Disconnect Charge - Add'l	USLXX	\$55.97	NA	NA	NA	\$38.44	\$56.25	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$42.19	\$43.77	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$49.18	\$55.00	\$34.52	\$55.00	\$33.05	\$48.17	\$48.31	\$48.47	NA
4-	Wire 56 Kbps Dig Grade Loop										
	RC - Statewide, per month	UDL56	NA	NA	NA	NA	NA	NA	\$32.67	NA	\$42.23
	RC - Zone 1, per month (Note 2)	TBD	\$27.33	\$39.08	\$25.75	NA	\$27.50	\$25.61	TBD	\$34.26	\$36.45
	RC - Zone 2, per month (Note 2)	TBD	\$44.40	\$57.21	\$29.74	NA	\$47.24	\$33.94	TBD	\$51.67	\$45.87
	RC - Zone 3, per month (Note 2)	TBD	\$80.45	\$126.22	\$47.27	NA	\$96.48	\$48.51	TBD	\$68.43	\$65.75
	RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	\$64.02	NA	NA	NA
	NRC - 1st	UDL56	\$498.05	\$654.72	\$348.55	NA	\$333.28	\$489.00	\$489.04	\$602.73	\$643.00
	NRC - Add'l	UDL56	\$343.70	\$428.45	\$241.20	NA	\$230.50	\$337.93	\$337.51	\$393.50	\$421.26
	NRC - Disconnect Charge - 1st	UDL56	\$129.62	NA	NA	NA	\$87.99	\$128.36	NA	\$44.06	NA
	NRC - Disconnect Charge - Add'l	UDL56	\$64.25	NA	NA	NA	\$44.24	\$64.35	NA	\$13.55	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	NA	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
4-	Wire 64 Kbps Dig Grade Loop		7.0.00	400.00	4 0		4 0=	V 10121	V 1010 1	4.01.0	40000
ĦŤ	RC - Statewide, per month	UDL64	NA	NA	NA	NA	NA	NA	\$32.67	\$41.70	\$42.23
+	RC - Zone 1, per month (Note 2)	TBD	\$27.33	\$39.08	\$25.75	NA	\$27.50	\$25.61	TBD	\$34.26	\$36.45
+	RC - Zone 2, per month (Note 2)	TBD	\$44.40	\$57.21	\$29.74	NA	\$47.24	\$33.94	TBD	\$51.67	\$45.87
+	RC - Zone 3, per month (Note 2)	TBD	\$80.45	\$126.22	\$47.27	NA NA	\$96.48	\$48.51	TBD	\$68.43	\$65.75
	RC - Zone 4, per month (Note 2)	TBD	NA NA	NA NA	NA NA	NA NA	NA	\$64.02	NA	NA NA	NA
H	NRC - 1st	UDL64	\$498.05	\$654.72	\$348.55	NA NA	\$333.28	\$489.00	\$489.04	\$602.73	\$643.00
H	NRC - Add'I	UDL64	\$343.70	\$428.45	\$241.20	NA.	\$230.50	\$337.93	\$337.51	\$393.50	\$421.26
H	NRC - Disconnect Charge - 1st	UDL64	\$129.62	NA NA	NA NA	NA NA	\$87.99	\$128.36	NA NA	\$44.06	NA NA
+	NRC - Disconnect Charge - Add'l	UDL64	\$64.25	NA	NA	NA NA	\$44.24	\$64.35	NA.	\$13.55	NA.
+	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA.	\$18.94	NA.	\$18.14	\$25.52	\$26.94	NA	NA.
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA NA	NA NA
+	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA NA	NA	NA NA	\$11.41	\$16.06	NA NA	NA NA	NA
	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	OCOSL	\$45.99	\$55.00	\$34.22	NA NA	\$32.77	\$45.27	\$45.34	\$45.43	\$55.00
2-	Wire Unbundled Copper Loop (18kft or less) Note 3	00000	Ψ-0.00	ψ55.00	Ψ04.22	14/5	Ψ02.11	ψ+3.21	ψ+0.0+	ψ+3.+3	ψ55.00
	RC - Statewide, per month	UCLPB	\$15.11	\$18.00	\$13.97	\$11.89	\$21.00	NA	\$19.00	\$20.81	\$12.16
+	RC - Zone 1, per month (Note 2)	TBD	TBD	\$18.60	\$19.80	TBN	\$18.80	\$16.85	TBD	\$18.90	\$19.85
-	RC - Zone 2, per month (Note 2)	TBD	TBD	\$27.23	\$22.86	TBN	\$25.85	\$22.34	TBD	\$28.50	\$24.98
-	RC - Zone 3, per month (Note 2)	TBD	TBD	\$60.07	\$36.34	TBN	\$39.14	\$31.92	TBD	\$37.75	\$35.81
+	RC - Zone 4, per month (Note 2)	TBD	NA NA	NA	NA	NA NA	NA	\$42.13	NA	NA	NA
+	NRC - 1st	UCLPB	\$514.21	\$340.00	\$395.16	\$713.50	\$340.00	\$504.82	\$504.90	\$600.61	\$270.01
+	NRC - Add'l	UCLPB	\$464.58	\$340.00	\$217.39	\$609.44	\$340.00	\$456.24	\$456.17	\$507.33	\$270.01
+	NRC - Disconnect Charge - 1st	UCLPB	5464.56 NA	\$300.00 NA	Φ217.39 NA	λ609.44 NA	\$72.54	\$105.86	NA	νA	\$74.54
+	NRC - Disconnect Charge - 1st NRC - Disconnect Charge - Add'l	UCLPB	NA NA	NA NA	NA NA	NA NA	\$39.42	\$57.25	NA NA	NA NA	\$39.14
+	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$47.00	\$47.00	\$18.94	\$47.00	\$18.14	\$25.52	\$26.94	\$25.52	NA
\vdash	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$21.00	\$21.00	\$8.42	\$21.00	\$8.06	\$11.34	\$12.76	\$47.00	NA NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$21.00 NA	\$21.00 NA	\$142.27	\$21.00 NA	\$6.06 NA	δ11.34 NA	\$12.76 NA	NA	NA NA
+	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA NA	\$37.86	\$17.77	\$11.41	\$16.06	NA NA	\$21.00	NA NA
╬	NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	\$17.77	\$16.00	\$37.86	\$17.77 NA	\$11.41	\$16.06	\$45.34	\$45.43	\$34.29
2	· · ·	UCLIVIC	\$10.00	\$10.00	Ф30.40	INA	Φ3∠.//	Φ45.∠1	φ40.34	Ф40.43	Ф34.29
2-	Wire Unbundled Copper Loop (>18kft) Note 3	UCL2L	\$40.00	\$35.00	\$41.61	\$40.00	\$37.00	\$45.00	\$35.00	\$40.00	\$35.00
	RC - Statewide, per month	TBD	\$40.00 TBD						\$35.00 TBD		
+	RC - Zone 1, per month (Note 2)			\$18.60	\$19.80	TBN	\$18.80	\$16.85		\$18.90	\$19.85
ш	RC - Zone 2, per month (Note 2)	TBD	TBD	\$27.23	\$22.86	TBN	\$25.85	\$22.34	TBD	\$28.50	\$24.98

		OTHER SERVICE								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
RC - Zone 3, per month (Note 2)	TBD	TBD	\$60.07	\$36.34	TBN	\$39.14	\$31.92	TBD	\$37.75	\$35.81
RC - Zone 4, per month (Note 2)	TBD	NA	NA	NA	NA	NA	\$42.13	NA	NA	NA
NRC - 1st	UCL2L	\$514.21	\$340.00	\$395.16	\$713.50	\$340.00	\$504.82	\$504.90	\$600.61	\$270.01
NRC - Add'l	UCL2L	\$464.58	\$300.00	\$217.39	\$609.44	\$300.00	\$456.24	\$456.17	\$507.33	\$234.63
NRC - Disconnect Charge - 1st	UCL2L	NA	NA	NA	NA	\$72.54	\$105.86	NA	NA	\$74.54
NRC - Disconnect Charge - Add'l	UCL2L	NA	NA	NA	NA	\$39.42	\$57.25	NA	NA	\$39.14
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$47.00	\$47.00	\$18.94	\$47.00	\$18.14	\$25.52	\$26.94	\$25.52	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$21.00	\$21.00	\$8.42	\$21.00	\$8.06	\$11.34	\$12.76	\$47.00	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	NA	\$142.27	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	\$17.77	NA	\$37.86	\$17.77	\$11.41	\$16.06	NA	\$21.00	NA
NRC - Incremental Charge - Manual Order Coordination - per loop	UCLMC	\$16.00	\$16.00	\$36.46	NA	\$32.77	\$45.27	\$45.34	\$45.43	\$34.29
DS3 Unbundled Local Loop			*******	***************************************		***	*	V 1010 1	¥ 101 10	***************************************
DS3 Unbundled Local Loop - per mile	1L5ND	\$43.96	\$40.01	\$29.96	\$43.69	\$38.98	\$54.39	\$32.53	\$56.71	\$30.53
DS3 Unbundled Local Loop- per Facility Termination	UE3PX	\$456.18	\$470.83	\$392.61	\$436.95	\$497.08	\$427.81	\$387.01	\$510.30	\$400.21
NRC - Facility Termination - 1st	UE3PX	\$973.58	\$770.47	\$770.96	\$1,091.00	\$709.14	\$975.22	\$964.04	\$1,091.00	\$726.16
NRC - Facility Termination - Add'l	UE3PX	\$547.59	\$436.40	\$437.71	\$661.23	\$402.63	\$549.17	\$542.73	\$654.13	\$411.64
NRC - Facility Termination - Disconnect - 1st	UE3PX	\$132.56	\$108.95	\$111.07	NA	\$102.16	\$134.07	NA NA	NA	\$103.36
NRC - Facility Termination - Disconnect - Add'l	UE3PX	\$129.07	\$106.01	\$108.14	NA	\$99.46	\$130.59	NA NA	NA NA	\$100.59
NRC - Incremental ChargeManual Svc Order - 1st	SOMAC	\$70.10	NA NA	\$54.64	\$93.12	\$50.25	\$68.62	\$69.34	\$92.52	\$53.03
NRC - Incremental ChargeManual Svc Order - Add'l	SOMAC	\$70.10	NA.	\$54.64	\$93.12	\$50.25	\$68.62	\$69.34	\$92.52	\$53.03
NRC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAC	\$30.09	NA NA	\$22.77	NA	\$20.94	\$28.59	\$29.76	NA	\$22.95
NRC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAC	\$30.09	NA.	\$22.77	NA NA	\$20.94	\$28.59	\$29.76	NA NA	\$22.95
STS-1 Unbundled Local Loop	OOWAG	ψ30.03	19/3	ΨΖΖ.ΓΓ	14/3	Ψ20.54	Ψ20.00	Ψ23.70	INA	Ψ22.00
STS-1 Unbundled Local Loop - per mile	1L5ND	\$43.96	\$40.01	\$29.96	\$43.69	\$38.98	\$54.39	\$32.53	\$56.71	\$30.53
STS-1 Unbundled Local Loop- per Facility Termination	UDLS1	\$456.18	\$470.83	\$392.61	\$436.95	\$497.08	\$427.81	\$387.01	\$510.30	\$400.21
NRC - STS-1 - Facility Termination - 1st	UDLS1	\$973.58	\$770.47	\$770.96	\$1,091	\$709.14	\$975.22	\$964.04	\$1.091	\$726.16
NRC - STS-1 - Facility Termination - Add'l	UDLS1	\$547.59	\$436.40	\$437.71	\$661.23	\$402.63	\$549.17	\$542.73	\$654.13	\$411.64
NRC - STS-1 - Facility Termination - Disconnect - 1st	UDLS1	\$132.56	\$108.95	\$111.07	NA NA	\$102.16	\$134.07	NA NA	NA NA	\$103.36
NRC - STS-1 - Facility Termination - Disconnect - Add'l	UDLS1	\$129.07	\$106.01	\$108.14	NA	\$99.46	\$130.59	NA NA	NA NA	\$100.59
NRC - STS-1 - Incremental ChargeManual Svc Order - 1st	SOMAC	\$70.10	NA	\$54.64	\$93.12	\$50.25	\$68.62	\$69.34	\$92.52	\$53.03
NRC - STS-1 - Incremental ChargeManual Svc Order - Add'l	SOMAC	\$70.10	NA NA	\$54.64	\$93.12	\$50.25	\$68.62	\$69.34	\$92.52	\$53.03
NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	SOMAC	\$30.09	NA NA	\$22.77	NA	\$20.94	\$28.59	\$29.76	NA	\$22.95
NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	SOMAC	\$30.09	NA.	\$22.77	NA NA	\$20.94	\$28.59	\$29.76	NA NA	\$22.95
Unbundled Loop Modification - Note 3	OOWAG	ψ30.03	19/3	ΨΖΖ.ΓΓ	14/3	Ψ20.54	Ψ20.00	Ψ23.70	INA	ΨΖΖ.55
Load Coil/Equipment Removal per pair - Loops up to 18kft	ULM2L	\$80.55	\$80.55	\$80.55	\$80.55	\$80.55	\$80.55	\$80.55	\$80.55	\$80.55
Load Coil/Equipment Removal per pair - Loops > 18kft - 1st	ULM2G	\$880.08	\$880.08	\$880.08	\$880.08	\$880.08	\$880.08	\$880.08	\$880.08	\$880.08
Load Coil/Equipment Removal per pair - Loops > 18kft - Add'l	ULM2G	\$27.30	\$27.30	\$27.30	\$27.30	\$27.30	\$27.30	\$27.30	\$27.30	\$27.30
Bridged Tap Removal per pair unloaded	ULMBT	\$121.14	\$121.14	\$121.14	\$121.14	\$121.14	\$121.14	\$121.14	\$121.14	\$121.14
Bridged Tap Kernovai per pair difloaded	OLIVID I	Φ121.14	φ1Z1.14	\$121.14	Φ121.14	Φ121.14	φ1Z1.14	\$121.14	φ1Z1.14	\$121.14
Loop Make Un Comice Insuins Note 2	 									-
Loop Make-Up Service Inquiry - Note 3	LIMICED	#000 7F	#000 7F	#000 7F	#000 7F	#000 75	#000 7F	#000 7F	#000 7F	#000 75
Per Service Inquiry	UMKLP	\$233.75	\$233.75	\$233.75	\$233.75	\$233.75	\$233.75	\$233.75	\$233.75	\$233.75
Unbundled Sub-Loops	<u> </u>	1	 	 			 	-	 	
Sub-Loop Analog										
Loop Distribution per 2-Wire Analog VG Loop (Including NID), per month	USBN2	NA	\$8.57	\$9.12	\$10.83	BFR	NA	NA	NA	\$9.79
NRC - Set-Up per Cross Box location - CLEC Feeder Facility set-up	USBSA	TBN	TBD	TBD	TBD	TBN	TBN	TBN	TBN	TBD
NRC - Set-Up per Cross Box location - per 25 pair panel set-up	USBSB	TBN	TBD	TBD	TBD	TBN	TBN	TBN	TBN	TBD
NRC - 1st	USBN2	TBN	\$78.28	\$207.01	\$459.85	TBN	TBN	TBN	TBN	\$586.00
NRC - Add'l	USBN2	TBN	\$58.33	\$171.32	\$352.89	TBN	TBN	TBN	TBN	\$255.00
NRC - Disconnect Charge - 1st	USBN2	TBN	NA	NA	NA	TBN	TBN	TBN	TBN	NA
NRC - Disconnect Charge - Add'l	USBN2	TBN	NA	NA	NA	TBN	TBN	TBN	TBN	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBN	NA	\$18.94	NA	TBN	TBN	TBN	TBN	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBN	NA	\$8.42	NA	TBN	TBN	TBN	TBN	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBN	NA	NA	NA	TBN	TBN	TBN	TBN	NA

	AND OTHER SERVICES										
DE	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	TBN	TBD	TBD	TBD	TBN	TBN	TBN	TBN	TBD
Lor	op Distribution per 2-Wire Analog VG Loop (Excluding NID), per month	TBD	NA	NA	NA	\$9.95	NA	NA	NA	NA	\$9.23
	NRC - Set-Up per Cross Box location - CLEC Feeder Facility set-up	USBSA	NA	NA	NA	\$9.95	NA	NA	NA	NA	TBD
	NRC - Set-Up per Cross Box location - per 25 pair panel set-up	USBSB	NA	NA	NA	\$9.95	NA	NA	NA	NA	TBD
	NRC - 1st	TBD	NA	NA	NA	\$459.85	NA	NA	NA	NA	\$587.00
	NRC - Add'l	TBD	NA	NA	NA	\$352.89	NA	NA	NA	NA	\$255.00
	NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	TBN	TBD	TBN	TBN	TBN	TBN	TBN	TBN	TBD
Lor	op Distribution per 4-Wire Analog VG Loop (Incl NID), per month	USBN4	TBN	\$11.29	TBN	TBN	TBN	TBN	TBN	TBN	TBD
	NRC - Set-Up per Cross Box location - CLEC Feeder Facility set-up	USBSA	TBN	TBD	TBN	TBN	TBN	TBN	TBN	TBN	TBD
	NRC - Set-Up per Cross Box location - per 25 pair panel set-up	USBSB	TBN	TBD	TBN	TBN	TBN	TBN	TBN	TBN	TBD
	NRC - 1st	USBN4	TBN	\$112.07	TBN	TBN	TBN	TBN	TBN	TBN	TBD
	NRC - Add'l	USBN4	TBN	\$92.11	TBN	TBN	TBN	TBN	TBN	TBN	TBD
	NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	TBN	TBD	TBN	TBN	TBN	TBN	TBN	TBN	TBD
Su	b-Loop-Intrabuilding Network Cable (INC) (riser cable), 2W analog, per month	USBR2									
	NRC - Set-Up per Building Equipment Room - CLEC Feeder Facility set-up	USBSC	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	NRC - Set-Up per Building Equipment Room - per 25 pair panel set-up	USBSD	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Ш	NRC - 1st	USBR2	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	NRC - Add'l	USBR2	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	NRC - Disconnect Charge - 1st	USBR2	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	NRC - Disconnect Charge - Add'l	USBR2	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Ш	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Sul	b-Loop-Intrabuilding Network Cable (INC) (riser cable), 4W analog, per month	USBR4	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Ш	NRC - Set-Up per Building Equipment Room - CLEC Feeder Facility set-up	USBSC	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Ш	NRC - Set-Up per Building Equipment Room - per 25 pair panel set-up	USBSD	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Ш	NRC - 1st	USBR4	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Щ	NRC - Add'l	USBR4	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Щ	NRC - Disconnect Charge - 1st	USBR4	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Щ	NRC - Disconnect Charge - Add'l	USBR4	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Ш	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Ш	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Щ	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Щ	NRC - Incremental Charge - Manual Order Coordination - per loop	USBMC	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
Un	bundled Network Terminating Wire										
\bot	UNTW Pair, per pair, per month	UENPP	TBN	\$0.67	\$1.56	\$1.24	NA	NA	NA	NA	\$1.31
\bot	Site Visit Survey, per MDU/MTU Complex, NRC	UENVS	TBN	\$225.00	\$225.00	\$225.00	NA	NA	NA	NA	\$225.00
\dashv	Site Visit Set-Up – Terminal Preparation, per terminal	LIENIOO	TDN	#00.00	#00.00	#00.00	TDN	TDN	TDN	TDN	#00.0C
\dashv	NRC - 1st terminal	UENSS	TBN	\$98.00	\$98.00	\$98.00	TBN	TBN	TBN TBN	TBN	\$98.00
\dashv	NRC - Add'l terminal	UENSS	TBN	\$65.00	\$65.00	\$65.00	TBN	TBN	1	TBN	\$65.00
\dashv	Access Terminal Provisioning & 1st 25 pair panel (SPOI), per terminal, NRC Existing Access Terminal Provisioning, 2nd 25 pair panel, per terminal, NRC	UEN1T UEN2T	TBN TBN	\$110.00 \$35.00	\$110.00 \$35.00	\$110.00 \$35.00	TBN TBN	TBN TBN	TBN TBN	TBN TBN	\$110.00 \$35.00
\dashv	UNTW Pair Provisioning, per pair, NRC	UENPP	TBN	\$35.00	\$35.00	\$35.00	TBN	TBN	TBN	TBN	\$35.00
+	Service Visit for Provisioning, per request, per premises, NRC	UENSV	TBN	\$9.00 \$55.00	\$9.00 \$55.00	\$9.00 \$55.00	TBN	TBN	TBN	TBN	\$9.00 \$55.00
$+\!+\!$	Manual Service Order, NRC	MOCLA	TBN	\$55.00 \$45.00	\$55.00 \$45.00	\$45.00	TBN	TBN	TBN	TBN	\$45.00
Su	b-Loop Concentration - Channelization Sys (Outside CO)	IVIOULA	IDIN	ψ45.00	ψ45.00	ψ40.00	IDN	IDIN	IDIN	IDIN	ψ+3.00
+ 3	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	TBD	\$18.94	TBD	BFR	BFR	BFR	BFR	TBD
+	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	TBD	\$8.42	TBD	BFR	BFR	BFR	BFR	TBD
TR	008 - System A (96 channel capacity - channels 1-96), per month	UCT8A	NA NA	\$792.49	\$724.79	\$757.00	NA NA	NA NA	NA NA	NA NA	\$683.78
	NRC - 1st	UCT8A	NA NA	\$640.93	\$632.36	\$633.94	NA	NA NA	NA NA	NA NA	\$634.31

Version 1Q00:6/5/00 7/7/00 Filefolder

		OTHER SERVICE								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
TR008 - System B (96 channel capacity - channels 97-192), per month			\$155.32	\$92.91	\$95.60	NA	NA	NA	NA	\$102.12
NRC - 1st	UCT8B	NA	\$640.93	\$632.36	\$633.94	NA	NA	NA	NA	\$634.31
NRC - Add'l	UCT8B	NA	\$315.03	\$310.82	\$311.60	NA	NA	NA	NA	\$311.78
TR303 - System A (96 channel capacity - channels 1-96), per month			\$835.72	\$764.42	\$799.95	NA	NA	NA	NA	\$726.87
NRC - 1st	UCT3A	NA	\$640.93	\$632.36	\$633.94	NA	NA	NA	NA	\$634.31
NRC - Add'l	UCT3A	NA	\$315.03	\$310.82	\$311.60	NA	NA	NA	NA	\$311.78
TR303 - System B (96 channel capacity - channels 97-192), per month	UCT3B	NA	\$198.55	\$132.54	\$138.55	NA	NA	NA	NA	\$145.21
NRC - 1st	UCT3B	NA	\$640.93	\$632.36	\$633.94	NA	NA	NA	NA	\$634.31
NRC - Add'l	UCT3B	NA	\$315.03	\$310.82	\$311.60	NA	NA	NA	NA	\$311.78
DS1 Feeder Interface, per month	UCTFS	NA	\$78.43	\$72.12	\$77.02	NA	NA	NA	NA	\$76.73
NRC 1st	UCTFS	NA	\$422.74	\$425.74	\$418.13	NA	NA	NA	NA	\$418.37
NRC Add'I	UCTFS	NA	\$200.74	\$198.06	\$198.56	NA	NA	NA	NA	\$198.67
Channel Interface - 2 Wire Voice - Loop Start , per month	TBD	NA.	\$2.62	\$2.38	\$2.68	NA NA	NA NA	NA NA	NA NA	\$2.61
NRC 1st	TBD	NA NA	\$42.39	\$41.82	\$41.92	NA NA	NA NA	NA NA	NA NA	\$41.95
NRC Add'l	TBD	NA NA	\$42.15	\$41.58	\$41.69	NA NA	NA NA	NA NA	NA NA	\$41.71
Channel Interface - 2 Wire ISDN, per month	ULCC1	NA NA	\$10.49	\$9.53	\$10.72	NA NA	NA NA	NA NA	NA NA	\$10.43
NRC 1st	ULCC1	NA NA	\$10.49	\$9.53 \$41.82	\$41.92	NA NA	NA NA	NA NA	NA NA	\$10.43
NRC Add'I	ULCC1	NA NA	\$42.39 \$42.15	\$41.82 \$41.58	\$41.92 \$41.69	NA NA	NA NA	NA NA	NA NA	\$41.95 \$41.71
	TBD	NA NA	\$42.15 \$15.59		\$41.69 \$15.94	NA NA	NA NA	NA NA	NA NA	
Channel Interface - 2 Wire Voice - Ground Start or Reverse Battery, per month	TBD	NA NA		\$14.17		NA NA	NA NA	NA NA	NA NA	\$15.51
. NRC 1st			\$42.39	\$41.82	\$41.92					\$41.95
NRC Add'l	TBD	NA	\$42.15	\$41.58	\$41.69	NA	NA	NA	NA	\$41.71
Channel Interface - 4 Wire Voice, per month	ULCC4	NA	\$9.30	\$8.45	\$9.50	NA	NA	NA	NA	\$9.26
NRC 1st	ULCC4	NA	\$42.39	\$41.82	\$41.92	NA	NA	NA	NA	\$41.95
NRC Add'l	ULCC4	NA	\$42.15	\$41.58	\$41.69	NA	NA	NA	NA	\$41.71
Test Circuit, per month		NA	\$45.46	\$41.30	\$46.44	NA	NA	NA	NA	\$45.22
NRC 1st	UCTTC	NA	\$42.39	\$41.82	\$41.92	NA	NA	NA	NA	\$41.95
NRC Add'I	UCTTC	NA	\$42.15	\$41.58	\$41.69	NA	NA	NA	NA	\$41.71
Channel Interface - Digital 56Kbps, per month	ULCC5	NA	\$13.78	\$12.51	\$14.08	NA	NA	NA	NA	\$13.71
NRC 1st	ULCC5	NA	\$42.39	\$41.82	\$41.92	NA	NA	NA	NA	\$41.95
NRC Add'l	ULCC5	NA	\$42.15	\$41.58	\$41.69	NA	NA	NA	NA	\$41.71
Channel Interface - Digital 64Kbps, per month	ULCC6	NA	\$13.78	\$12.51	\$14.08	NA	NA	NA	NA	\$13.71
NRC 1st	ULCC6	NA	\$42.39	\$41.82	\$41.92	NA	NA	NA	NA	\$41.95
NRC Add'l	ULCC6	NA	\$42.15	\$41.58	\$41.69	NA	NA	NA	NA	\$41.71
Loop Concentration System (Inside C.O.)										
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	TBD	\$18.94	TBD	\$18.14	\$25.52	TBD	\$44.06	TBD
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	TBD	\$8.42	TBD	\$8.06	\$11.34	TBD	\$13.55	TBD
Loop Channelization System - Digital Loop Carrier	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
RC - Loop Channelization System - Digital Loop Carrier	TBD	NA	NA	NA	NA	NA	NA	\$315.16	NA	NA
NRC-1st	TBD	NA	NA	NA	NA	NA	NA	\$426.48	NA	NA
NRC- Addl	TBD	NA	NA	NA	NA	NA	NA	\$103.42	NA	NA
NRC- Incremental Cost - Manaul Service Order- 1st	TBD	NA	NA	NA	NA	NA	NA	\$42.19	NA	NA
NRC- Incremental Cost - Manaul Service Order- Addl	TBD	NA	NA	NA	NA	NA	NA	\$12.76	NA	NA
TR008 -System A (96 channel capacity - channels 1-96), per month	UCT8A	\$327.44	\$400.33	\$316.63	\$394.00	\$308.74	\$454.79	\$375.96	\$399.21	\$380.06
NRC - 1st	UCT8A	\$1,115.10	\$1,128.75	\$1,111.95	\$1,116.15	\$1,117.20	\$1,115.10	\$1,113.00	\$1,119.30	\$1,114.05
NRC - Add'l	UCT8A	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA	NA	NA NA
TR008 -System B (96 channel capacity - channels 97-192), per month	UCT8B	\$67.41	\$70.48	\$65.27	\$72.21	\$76.58	\$73.30	\$65.98	\$71.91	\$68.71
NRC - 1st	UCT8B	\$464.57	\$470.41	\$463.37	\$465.11	\$465.64	\$464.71	\$463.74	\$466.38	\$464.21
NRC - Add'l	UCT8B	NA	NA	Ψ 4 03.37	NA	NA	NA	NA	NA	NA
TR303 - System A (96 channel capacity - channels 1-96), per month	UCT3A	\$375.18	\$450.24	\$362.87	\$445.14	\$385.97	\$506.70	\$422.68	\$450.13	\$428.73
NRC - 1st	UCT3A	\$1,115.10	\$1,128.75	\$1,111.95	\$1,116.15	\$1,117.20	\$1,115.10	\$1,113.00	\$1,119.30	\$1,114.05
NRC - Add'I	UCT3A	NA	λ1,126.75 NA	NA	NA	\$1,117.20 NA	λ1,115.10 NA	NA	NA	\$1,114.05 NA
	UCT3B	\$111.30	\$118.76	\$110.02	\$121.45	\$129.05	\$123.52	\$111.17	\$121.16	\$115.79
TR303 - System B (96 channel capacity - channels 97-192), per month	UCISB	\$111.30	\$118.76	\$110.02	\$1Z1.45	\$129.05	\$123.52	φ111.1 <i>/</i>	\$121.10	\$115.79

Version 1Q00:6/5/00 7/7/00 Filefolder

DESCRIPTION		USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC - 1st		UCT3B	\$464.57	\$470.41	\$463.37	\$465.11	\$465.64	\$464.71	\$463.74	\$466.38	\$464.21
NRC - Add'l		UCT3B	NA	NA	NA NA	NA	NA NA	NA	NA NA	NA NA	NA NA
DS1 Interface, per month		UCTCO	\$6.42	\$6.47	\$6.15	\$403.20	\$7.35	\$6.99	\$6.27	\$6.79	\$6.49
NRC 1st		UCTCO	\$367.70	\$372.32	\$366.72	\$132.18	\$368.54	\$367.80	\$367.04	\$369.13	\$367.41
NRC Add'l		UCTCO	\$132.03	\$133.69	\$130.63	\$132.18	\$132.33	\$132.07	\$131.79	\$132.54	\$131.92
Channel Interface - 2 Wire Voice - Loop Start , per month		TBD	\$2.55	\$2.66	\$2.44	\$2.79	\$2.91	\$2.77	\$0.89	\$2.69	\$2.58
NRC 1st		TBD	\$35.77	\$36.23	\$35.68	\$35.82	\$35.86	\$35.78	\$35.73	\$35.91	\$35.74
NRC Add'l		TBD	\$35.55	\$36.02	\$35.48	\$35.62	\$35.66	\$35.37	\$35.49	\$35.71	\$35.54
Channel Interface - 2 Wire ISDN, per month		ULCC1	\$10.19	\$10.67	\$9.76	\$11.18	\$11.66	\$11.10	\$9.95	\$10.76	\$10.30
NRC 1st		ULCC1	\$35.77	\$36.23	\$35.68	\$35.82	\$35.86	\$35.78	\$35.71	\$35.91	\$35.74
NRC Add'l		ULCC1	\$35.55	\$36.02	\$35.48	\$35.62	\$35.66	\$35.37	\$35.51	\$35.71	\$35.54
Channel Interface - 2 Wire Voice - Ground Start or Revers	se Battery, per month	TBD	\$15.15	\$15.85	\$14.51	\$16.62	\$17.33	\$16.46	\$14.80	\$16.01	\$15.32
. NRC 1st		TBD	\$35.77	\$36.23	\$35.68	\$35.82	\$35.86	\$35.78	\$35.71	\$35.91	\$35.74
NRC Add'l		TBD	\$35.55	\$36.02	\$35.48	\$35.62	\$35.66	\$35.37	\$35.51	\$35.71	\$35.54
Channel Interface - 4 Wire Voice, per month		ULCC4	\$9.04	\$9.44	\$8.65	\$9.91	\$10.34	\$9.83	\$8.82	\$9.55	\$9.13
NRC 1st		ULCC4	\$35.77	\$36.23	\$35.68	\$35.82	\$35.86	\$35.78	\$35.71	\$35.91	\$35.74
NRC Add'I		ULCC4	\$35.55	\$36.02	\$35.48	\$35.62	\$35.66	\$35.37	\$35.51	\$35.71	\$35.54
Test Circuit, per month		UCTTC	\$44.16	\$46.14	\$42.30	\$48.43	\$50.53	\$47.85	\$43.13	\$46.66	\$44.65
NRC 1st		UCTTC	\$35.77	\$36.23	\$35.68	\$35.82	\$35.86	\$35.78	\$35.71	\$35.91	\$35.74
NRC Add'I		UCTTC	\$35.55	\$36.02	\$35.48	\$35.62	\$35.66	\$35.37	\$35.51	\$35.71	\$35.54
Channel Interface - Digital 56Kbps, per month		ULCC5	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC 1st		ULCC5	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC Add'I		ULCC5	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Channel Interface - Digital 64Kbps, per month		ULCC6	TBD	TBD	TBD	TBD	TBD TBD	TBD	TBD	TBD	TBD
NRC 1st		ULCC6	TBD	TBD	TBD	TBD		TBD	TBD	TBD	TBD
NRC Add'I		ULCC6	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
DARK FIBER		1L5DF	\$59.84	\$55.35	\$44.22	\$64.64	\$65.29	\$70.35	\$49.88	\$72.45	\$52.67
Per four fiber strands, per route mile or fraction thereof, p		1L5DF		\$1,715.61	\$44.22				\$49.88		\$1,672.44
NRC - Per each four-fiber dark fiber arrangement - 1st		1L5DF	\$2,518.66	. ,	\$1,355.29	\$2,304.00 \$740.93	\$1,685.19 \$580.11	\$2,389.99	\$733.08	\$2,406.00	\$1,672.44
INKC - Per each four-liber dark liber afrangement - Add	uı	ILOUF	\$835.08	\$622.68	\$273.69	\$740.93	\$580.11	\$804.32	\$733.08	\$765.30	\$509.09
NOTES:											-
1 In states where a specific NRC for customer transfer, for											
is not stated, the applicable NRC from the appropriate											-
2 Effective May 1, 2000 statewide rates will be replaced Zone where available. Until approximately December											
BellSouth billing systems have been developed to hand											
structure, BellSouth will bill at the Zone 1 Deaveraged											
December 31, 2000 or such time that the billing system											
handle the new zone rate structure, BellSouth will begin											
interconnection agreement.	in billing pursuant to OLEO-15										
3 All rates are interim and subject to true-up.											
Where the state Commission has adopted rates for the	e rate elements										1
containedherein, it is the intent of the Parties to reflect											
4 apply the same consistent with applicable FCC and Co	ommissionrules and orders.										1

		OTHER SERVI								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
LOCAL EXCHANGE SWITCHING (PORTS)										
2-Wire Analog Line Port (Res., Bus.), per month										
2- wire voice unbundled port - residence	UEPRL	\$2.07	2.00 - Note	1 1.85 - Note 1	2.61 - Note 1	\$2.20	\$2.11	\$2.19	\$2.35	1.90 - Note 1
2-wire voice unbundled port with caller ID - residence	UEPRC	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled port outgoing only - residence	UEPRO	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled area plus port with caller ID - residence	UEPRM	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA	\$2.00	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence						*				
(LW8)	UEPAJ	NA	NA	NA	NA	NA	NA	NA	\$2.35	NA
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
(F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
(TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence									1	******
(TACSR)	UEPAM	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence		1		1,7,				1		ψσσ
(1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence		1.0.								ψσσ
(2MR)	UEPAO	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-wire voice unbundled res, low usage line port with Caller ID (LUM)	UEPAP	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2 mile veise unbundied ree, iem deage mile pert mili ediler ib (2em)	02.7	Ψ2.01	Ψ2.00	\$1.55	Ψ2.01	\$2.20	Ψ2	Ψ2.00	Ψ2.00	ψσσ
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCX									
ESONE NOMBERT (NEGOTIES ONE FERT ON)	LITI OX									-
2-wire voice unbundled port without Caller ID	UEPBL	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled port without caller ID 2-wire voice unbundled port with unbundled port with Caller+E484 ID	UEPBC	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled outgoing only port	UEPBO	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled area plus port with Caller ID	UEPBM	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled area plus port with Caller ID 2-wire voice unbundled incoming only port with Caller ID	UEPB1	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-wire voice unbundled Incoming only port with Galler ID 2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	NA	NA	NA	NA	\$2.20	NA	NA	NA	\$1.90 NA
2-wire voice unbundled CA bus Area Calling Port with Caller ID (B0C) 2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	UEPAB	NA NA	NA NA	NA NA	NA NA	\$2.20 ΝΑ	NA NA	NA NA	\$2.35	NA NA
	UEPAD	INA	INA	INA	INA	INA	INA	INA	φ2.35	INA
2-wire voice unbundled TN Bus 2-Way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-wire voice unbundled TN Bus 2-Way Area Calling Port Standard Option	UEPAC	INA	INA	INA	NA	NA	INA	NA	NA	\$1.90
(TACC2)	UEPAD	NA	NIA	NIA	NIA	NA	NIA	NIA	NIA	¢4.00
2-wire voice unbundled TN Bus 2-WAY Collierville and Memphis Local Calling Port	UEPAD	INA	NA	NA	NA	NA	NA	NA	NA	\$1.90
(B2F)	UEPAE	NA	NIA	NA	NIA	NA	NA	NA	NA	¢4.00
(DZF)	UEPAE	INA	NA	INA	NA	NA	INA	NA	NA	\$1.90
LOCAL MUMBER PORTABILITY (PEOUIDEC ONE PER PORT)	LNPCX	-					-			+
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCX	-					-			+
Non Requiring Charges (NRC) 4st (Residence)			+				 		 	1
Non-Recurring Charges (NRC) - 1st (Residence)			1				.		.	DOT COOT
O universaries make madical and residence	LIEDDI	#04.00	#00.0C	047.46	#07.70	# 40.46	#00.05	#04.00	#04.00	BST GSST
2- wire voice unbundled port - residence	UEPRL	\$21.93	\$38.00	\$17.16	\$37.78	\$16.43	\$22.98	\$21.60	\$24.98	A4.3.1
	LIEBBO	001.00	#00.00	047.40	007.70	040.40	#0C CC	00101	00100	BST GSST
2-wire voice unbundled port with caller ID - residence	UEPRC	\$21.93	\$38.00	\$17.16	\$37.78	\$16.43	\$22.98	\$24.04	\$24.98	A4.3.1
	LIEDDO	004.05	000.05	0.7.46	007.70	040.46	000.05	0040:	004.05	BST GSST
2-wire voice unbundled port outgoing only - residence	UEPRO	\$21.93	\$38.00	\$17.16	\$37.78	\$16.43	\$22.98	\$24.04	\$24.98	A4.3.1
	LIEBBA	00:	00	0.1- :-	00===	0.46 :-	005	00:5:		BST GSST
2-wire voice unbundled area plus port with caller ID - residence	UEPRM	\$21.93	\$38.00	\$17.16	\$37.78	\$16.43	\$22.98	\$24.04	\$24.98	A4.3.1
2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA	\$38.00	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA

Version 1Q00:6/5/00

7/7/00 Filefolder

		OTHER SERVIC								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	UEPAJ	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	BST GSST A4.3.1
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	UEPAL	NA NA	NA NA	NA	NA	NA	NA	NA	NA	BST GSST A4.3.1
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	UEPAM	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	BST GSS
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										BST GSS
(1MF2X) 2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	BST GSS
(2MR)	UEPAO	NA	NA	NA	NA	NA	NA	NA	NA	A4.3.1 BST GSS
2-wire voice unbundled Res Low Usage Line Port with Caller+E563 ID (LUM)	UEPAP	\$21.93	\$38.00	\$17.16	\$37.78	\$16.43	\$22.98	\$24.04	\$24.98	A4.3.1
NRC - Add'l (Residence)										
2- wire voice unbundled port - residence -	UEPRL	\$21.93	\$15.00	\$17.16	\$37.78	\$16.43	\$22.98	\$21.60	\$24.98	BST GSST A4.3.1
2-wire voice unbundled port with caller ID - residence	UEPRC	\$21.93	\$15.00	\$17.16	\$37.78	\$16.43	\$22.98	\$9.08	\$24.98	BST GSST A4.3.1
2-wire voice unbundled port outgoing only - residence	UEPRO	\$21.93	\$15.00	\$17.16	\$37.78	\$16.43	\$22.98	\$9.08	\$24.98	BST GSST A4.3.1
2-wire voice unbundled area plus port with caller ID - residence	UEPRM	\$21.93	\$15.00	\$17.16	\$37.78	\$16.43	\$22.98	\$9.08	\$24.98	BST GSST A4.3.1
2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA	\$15.00	NA						
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	UEPAJ	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	BST GSST A4.3.1
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	UEPAL	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA	BST GSST A4.3.1
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence ((TACSR)	UEPAM	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	BST GSST A4.3.1
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	UEPAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	BST GSST A4.3.1
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence [(2MR)	UEPAO	NA NA				NA NA	NA NA	NA NA	NA NA	BST GSS1 A4.3.1
			NA	NA	NA					BST GSS1
2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)	UEPAP	\$21.93	\$15.00	\$17.16	\$37.78	\$16.43	\$22.98	\$9.08	\$24.98	A4.3.1
NRC - 1st (Business)										DOT OCC
2-wire Voice Unbundled Port without Caller ID	UEPBL	\$21.93	\$38.00	\$17.16	\$37.55	\$16.43	\$22.98	\$21.60	\$24.98	BST GSST A4.3.1
2-wire voice unbundled port with Caller ID	UEPBC	\$21.93	\$38.00	\$17.16	\$37.55	\$16.43	\$22.98	\$24.04	\$24.98	BST GSS1 A4.3.1
2-wire voice unbundled outgoing only port	UEPBO	\$21.93	\$38.00	\$17.16	\$37.55	\$16.43	\$22.98	\$24.04	\$24.98	BST GSS1 A4.3.1
2-wire voice unbundled Area Plus Port with Caller ID	UEPBM	\$21.93	\$38.00	\$17.16	\$37.55	\$16.43	\$22.98	\$24.04	\$24.98	BST GSST A4.3.1

Version 1Q00:6/5/00 7/7/00 Filefolder

			OTHER SERVICE								
DES	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
											BST GSST
	2-wire voice unbundled Incoming only Port with Caller ID	UEPB1	\$21.93	\$38.00	\$17.16	\$37.55	\$16.43	\$22.98	\$24.04	\$24.98	A4.3.1
	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
	2-wire voice unbundled SC Bus Area Calling Port with Caller ID+E587 (LMB)	UEPAB	NA	NA	NA	NA	NA	NA	NA	\$24.98	NA
											BST GSST
	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	A4.3.1
											BST GSST
	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	A4.3.1
	2-wire voice unbundled TN Bus 2-way Collierville and Memphis Local Calling Port										BST GSST
$\sqcup \!\!\!\! \perp$	(B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	A4.3.1
											BST GSST
$\sqcup \sqcup$											A4.3.1
	NDO ALUM (Duelle e e)	LIEDDI	004.00	0.15.00	017.10	007.55	# 10.10	000.00	# 0.00	00400	BST GSST
$\sqcup \sqcup$	NRC - Add'l (Business)	UEPBL	\$21.93	\$15.00	\$17.16	\$37.55	\$16.43	\$22.98	\$9.08	\$24.98	A4.3.1
	Outro veiso veiso veiso de de controlle de College ID	LIEDDI	CO4.00	#45.00	04740	07.55	# 40.40	# 00.00	#04.00	#04.00	BST GSST
++	2-wire voice unbundled port without Caller ID	UEPBL	\$21.93	\$15.00	\$17.16	\$37.55	\$16.43	\$22.98	\$21.60	\$24.98	A4.3.1 BST GSST
	2 wire waise we have all and most with College ID	UEPBC	CO4.00	#45.00	04740	07.55	# 40.40	# 00.00	#0.00	#04.00	
++	2-wire voice unbundled port with Caller ID	UEPBC	\$21.93	\$15.00	\$17.16	\$37.55	\$16.43	\$22.98	\$9.08	\$24.98	A4.3.1 BST GSST
	2 wire waise walled autoring orbinal	UEPBO	CO4.00	#45.00	04740	07.55	# 40.40	\$22.98	#0.00	#04.00	A4.3.1
++	2-wire voice unbundled outgoing only port	UEPBU	\$21.93	\$15.00	\$17.16	\$37.55	\$16.43	\$22.98	\$9.08	\$24.98	BST GSST
	2 wire waise walled Area Blue Bort with Calley ID	UEPBM	CO4.00	#45.00	047.40	07.55	# 40.40	# 00.00	#0.00	# 04.00	A4.3.1
+++	2-wire voice unbundled Area Plus Port with Caller ID	UEPDIVI	\$21.93	\$15.00	\$17.16	\$37.55	\$16.43	\$22.98	\$9.08	\$24.98	BST GSST
	2-wire voice unbundled incoming only port with Caller ID	UEPB1	\$21.93	\$15.00	\$17.16	\$37.55	¢40.40	\$22.98	\$9.08	\$24.98	A4.3.1
++	2-wire voice unbundled incoming only port with Caller ID 2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	\$21.93 NA	\$15.00 NA	\$17.16 NA	\$37.55 NA	\$16.43 \$16.43	\$22.98 NA	\$9.08 NA	\$24.98 NA	NA NA
H +	2-wire voice unburidled LA Bus Area Calling Port with Caller ID (BOC) 2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	UEPAB	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	\$24.98	NA NA
H +	2-wife voice dribultuled SC bus Area Calling Fort with Caller ID (LIND)	ULFAB	INA	INA	INA	INA	INA	INA	INA	\$24.90	BST GSST
	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	UEPAC	NA	NA	NA	NA	NA	NA	NA	NA	A4.3.1
++	2-wire voice dribundled TN bus 2-way Area Calling For Economy Option (TACCT)	OLIAC	INA	INA	INA	INA	INA	INA	INA	INA	BST GSST
	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	UEPAD	NA	NA	NA	NA	NA	NA	NA	NA	A4.3.1
H	2-wire voice unbundled TN Bus 2-way Collierville and Memphis Local Calling Port	OLIAD	INA	INA	INA	INA	INA	INA	INA	INA	BST GSST
	(B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	A4.3.1
H	(SEI)	OLITAL	10/	1471	1471	10/1	1471	1471	1471	1471	711.0.1
H +	NRC - Disconnect Charge - 1st										
++	2- wire voice unbundled port - residence		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
++	2-wire voice unburidled port with caller ID - residence		\$6.21	NA NA	NA NA	NA NA	\$4.38	\$6.56	NA NA	NA	NA NA
${\sf H}{\sf H}$	2-wire voice unbundled port with ealer 15 residence		\$6.21	NA NA	NA NA	NA NA	\$4.38	\$6.56	NA NA	NA NA	NA NA
H +	2-wire voice unbundled area plus port with caller ID - residence		\$6.21	NA NA	NA NA	NA NA	\$4.38	\$6.56	NA NA	NA NA	NA NA
H +	2-wire voice unbundled Florida area calling with caller ID - residence		NA NA	NA NA	NA NA	NA NA	NA	NA	NA NA	NA NA	NA.
++	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)		NA.	NA NA	NA NA	NA	\$4.38	NA NA	NA NA	NA	NA
${\sf H}{\sf H}$	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)		NA NA	NA NA	NA NA	NA NA	\$4.38	NA NA	NA NA	NA NA	NA NA
${\sf H}{\sf H}$	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence					,	\$, .	. */ (1 171
	(LW8)		NA	NA	NA	NA	NA	NA	NA	NA	NA
H +	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence			1							1
	(F2R)		NA	NA	NA	NA	NA	NA	NA	NA	NA
H +	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence		1	1 1				1			1
	(TACER)		NA	NA	NA	NA	NA	NA	NA	NA	NA
\Box	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence			1 1	1						1
	(TACSR)		NA	NA	NA	NA	NA	NA	NA	NA	NA
H	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										1
	(1MF2X)		NA	NA	NA	NA	NA	NA	NA	NA	NA
_	·										

Version 1Q00:6/5/00 7/7/00 Filefolder

		AND	OTHER SERVIC	ES							
DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(2MR)		NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	· · ·										
	2-wire voice unbundled port without Caller ID		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	2-wire voice unbundled port with Caller ID		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	2-wire voice unbundled outgoing only Port		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	2-wire voice unbundled Area Plus Port with Caller ID		\$6.21	NA.	NA.	NA NA	\$4.38	\$6.56	NA NA	NA.	NA
+++	2-wire voice unbundled Incoming only Port with Caller ID		\$6.21	NA.	NA.	NA	\$4.38	\$6.56	NA NA	NA NA	NA.
++	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)		NA	NA	NA NA	NA NA	\$4.38	NA	NA NA	NA NA	NA NA
+++	2-wire voice unbundles SC Bus Area Calling Port with Caller ID (LMB)		NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA
++	2-wire voice unbullules 3C bus Alea Calling Fort with Caller ID (LIVID)		INA	INA	INA	INA	INA	INA	INA	INA	INA
	2 wire waise web and led TN Due 2 west Area Calling Dark Frances (Continue (TA CCA)		NA	NA	NA	NA	NA	NA	NA	NA	NA
+++	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)		INA	INA	INA	INA	INA	INA	INA	INA	INA
	0 · · · · · · · · · · · · · · · · · · ·										
++	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)		NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice unbundled TN Bus 2-Way Collierville and Memphis Local Calling Port				l						l
\vdash	(B2F)		NA	NA	NA	NA	NA	NA	NA	NA	NA
HH			1								
$\sqcup \!\!\!\! \perp$	NRC - Disconnect Charge - Add'I					ļ				ļ	ļ
$\sqcup \!\!\!\! \perp$	2- wire voice unbundled port - residence		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	2-wire voice unbundled port with caller ID - residence		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	2-wire voice unbundled port outgoing only - residence		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	2-wire voice unbundled area plus port with caller ID - residence		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
	2-wire voice unbundled Florida area calling with caller ID - residence		NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)		NA	NA	NA	NA	\$4.38	NA	NA	NA	NA
	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)		NA	NA	NA	NA	\$4.38	NA	NA	NA	NA
	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence										
	(LW8)		NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(F2R)		NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										1
	(TACER)		NA	NA	NA	NA	NA	NA	NA	NA	NA
+++	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence										
	(TACSR)		NA	NA	NA	NA	NA	NA	NA	NA	NA
++	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence		14/-1	14/4	14/3	19/3	14/-1	14/3	19/3	1973	19/3
	(1MF2X)		NA	NA	NA	NA	NA	NA	NA	NA	NA
+++	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence		14/3	14/3	11/3	19/3	11//	11/3	19/3	19/3	19/3
	(2MR)		NA	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
++	2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)		\$6.21	NA NA	NA NA	NA NA	\$4.38	\$6.56	NA NA	NA NA	NA NA
++	2-wire voice disputibled has now osage fille for with caller in (FOM)		Φ0.∠1	INA	INA	INA	Φ4.30	Φ0.00	INA	INA	INA
++	2 wire vaice unbundled port without Caller ID		Ø6 04	NI A	NI A	NI A	¢4.00	\$6.50	NI A	NI A	NIA.
++	2-wire voice unbundled port without Caller ID		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
H	2-wire voice unbundled port with Caler ID		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
HH	2-wire voice unbundled outgoing only port		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
$\sqcup \!\!\!\! \perp$	2-wire voice unbundled Area Plus Port with Caller ID		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
$\sqcup \!\!\!\! \perp$	2-wire voice unbundled incoming only port with Caller ID		\$6.21	NA	NA	NA	\$4.38	\$6.56	NA	NA	NA
\coprod	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)		NA	NA	NA	NA	\$4.38	NA	NA	NA	NA
\coprod	2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)		NA	NA	NA	NA	NA	NA	NA	NA	NA
$I I I^{-}$	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option				_	_				_	_
	(TACC1)		NA	NA	NA	NA	NA	NA	NA	NA	NA
\Box											
Ш	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)		NA	NA	NA	NA	NA	NA	NA	NA	NA

Version 1Q00:6/5/00 7/7/00 Filefolder

			AND	OTHER SERVIC	ES							
D	ESC	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
		2-wire voice unbundled TN Bus 2-way Collierville and Memphis Local Calling Port										
		(B2F)		NA	NA	NA	NA	NA	NA	NA	NA	NA
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$44.42	NA
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$14.63	NA
		NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$10.39	\$16.06	NA	NA	NA
		NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$1.44	NA	NA	NA	NA	NA	NA	NA	NA
Α	II av	ailable features, per month	UEPVF	\$5.55	NA	NA	NA	\$8.28	\$6.75	NA	\$6.29	NA
		NRC - 1st (all types)		\$24.72	NA	NA	NA	NA	\$21.42	NA	\$36.24	NA
H	1	NRC - Add'l (all types)		\$24.72	NA	NA	NA	NA	\$21.42	NA	\$36.24	NA
		NRC - Disconnect Charge - 1st		\$18.41	NA	NA	NA	NA	\$19.68	NA	NA	NA
		NRC - Disconnect Charge - Add'l		\$18.41	NA	NA	NA	NA	\$19.68	NA	NA	NA
H	+	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA NA	NA NA	NA	NA.	\$25.52	NA NA	\$44.42	NA
H	+	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA NA	NA NA	NA NA	NA NA	\$11.34	NA NA	\$14.63	NA NA
+	+	NRC - Incremental Charge - Manual Service Order - Add 1	SOMAN	\$17.77	NA NA	NA NA	NA NA	NA NA	\$16.06	NA NA	Ψ14.03 NA	NA NA
+	+	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$1.44	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA
-	hres	e available feature, per month	UEPVF	δ1.44 NA	NA NA	NA NA	NA NA	\$8.28	\$3.31	NA NA	\$3.03	NA NA
Н"	mee	NRC - 1st (all types)	UEFVF	NA NA	NA NA	NA NA	NA NA	\$8.28 NA	\$3.31	NA NA	\$4.53	NA NA
+	+	NRC - 1st (all types)		NA NA	NA NA	NA NA	NA NA	NA NA	\$3.06	NA NA	\$4.53 \$4.53	NA NA
H	-	NRC - Disconnect Charge - 1st		NA NA	NA NA	NA NA	NA NA	NA NA	\$8.20	NA NA	ъ4.55 NA	NA NA
-	-	NRC - Disconnect Charge - 1st NRC - Disconnect Charge - Add'l		NA NA	NA NA	NA NA	NA NA	NA NA	\$8.20	NA NA	NA NA	NA NA
H	+		SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	\$25.52	NA NA	\$44.42	NA NA
H	_	NRC - Incremental Charge - Manual Service Order - 1st										
Щ.	-	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	\$11.34	NA	\$14.63	NA
Щ	_	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	NA	NA	NA	NA	\$16.06	NA	NA	NA
Щ	_	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
H.					<u>.</u>							
4-	-Wir	e Analog VG Port, per month	UEP4A	NA	\$9.14	\$8.47	NA	\$10.13	\$9.60	\$8.69	\$2.28	NA
Ш	_	NRC - 1st	UEP4A	NA	\$5.86	\$17.16	NA	\$16.43	\$22.98	\$21.69	\$3.50	NA
Ш	_	NRC - Add'l	UEP4A	NA	\$5.86	\$17.16	NA	\$16.43	\$22.98	\$21.69	\$3.50	NA
Ш		NRC - Disconnect Charge - 1st	BFR	NA	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
Ш		NRC - Disconnect Charge - Add'l	BFR	NA	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	\$18.94	NA	\$18.14	\$25.52	\$26.85	NA	NA
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	\$8.42	NA	\$8.06	\$11.34	\$12.67	NA	NA
		NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	NA	NA	NA	\$8.94	\$16.06	NA	NA	NA
2-	-Wir	e DID Port, per month	UEPP2	\$12.08	TBD	\$11.35	NA	\$13.12	\$14.63	\$12.36	\$12.08	\$12.68
$\prod_{i=1}^{n}$												BST GSST
Ш		NRC - 1st	UEPP2	\$50.00	TBD	\$61.91	NA	\$59.28	\$83.09	\$81.84	\$50.00	A4.3.1
Π^{-}										_	_	BST GSST
Ш		NRC - Add'l	UEPP2	\$18.00	TBD	\$61.91	NA	\$59.28	\$83.09	\$81.84	\$50.00	A4.3.1
Ш		NRC - Disconnect Charge - 1st	UEPP2	NA	NA	NA	NA	\$9.20	\$13.48	NA	NA	NA
		NRC - Disconnect Charge - Add'l	UEPP2	NA	NA	NA	NA	\$9.20	\$13.48	NA	NA	NA
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	NA	NA
П		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA	NA
		NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	NA	NA	NA	\$10.39	\$16.07	NA	NA	NA
4-	-Wir	e DS1 Port w/DID capability, per month	UEPDD	\$130.23	\$125.00	\$120.80	NA	\$149.27	\$146.46	\$123.65	\$130.23	\$120.00
П												To be
		NRC - 1st	UEPDD	\$50.00	\$112.00	\$89.44	NA	\$85.63	\$117.81	\$116.59	\$60.00	negotiated
П												To be
Ш		NRC - Add'I	UEPDD	\$18.00	\$91.00	\$52.46	NA	\$50.23	\$71.18	\$69.92	\$60.00	negotiated
IT	1	NRC - Disconnect Charge - 1st	UEPDD	NA	NA	NA NA	NA	\$8.82	\$12.94	NA	NA	NA
H	1	NRC - Disconnect Charge - Add'l	UEPDD	NA NA	NA NA	NA	NA	\$8.82	\$12.94	NA NA	NA	NA.
டட			02,00	1 17 1	1471	1471	1471	Ψ0.02	Ψ12.01	14/1	1473	1 1 1 1

Version 1Q00:6/5/00

7/7/00 Filefolder
PORTS

			OTHER SERVIC								
DE	DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	NA	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	NA	NA	NA	NA	\$10.39	\$16.06	NA	NA	NA
2-\	2-Wire ISDN Port(2) (3), per month	U1PMA	\$16.42	\$13.00	\$13.47	\$12.33	\$23.33	\$51.91	\$24.50	\$33.74	\$1.90
											BST GSST
	NRC - 1st	U1PMA	\$63.24	\$88.00	\$47.37	\$90.48	\$45.35	\$63.59	\$62.29	\$65.79	A4.3.1
											BST GSST
	NRC - Add'l	U1PMA	\$63.24	\$66.00	\$47.37	\$84.53	\$45.35	\$63.59	\$62.29	\$65.79	A4.3.1
	NRC - Disconnect Charge - 1st	U1PMA	\$5.69	NA	NA	NA	\$4.31	\$7.04	NA	NA	NA
	NRC - Disconnect Charge - Add'l	U1PMA	\$5.69	NA	NA	NA	\$4.31	\$7.04	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$56.19	NA	\$39.98	NA	\$38.29	\$53.87	\$55.30	\$67.52	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$56.19	NA	\$39.98	NA	\$38.29	\$53.87	\$55.30	\$67.52	NA
+	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$12.97	NA	NA	NA NA	\$6.65	\$11.34	NA	NA	NA
+	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$12.97	NA NA	NA.	NA NA	\$6.65	\$11.34	NA	NA	NA
	NRC - User Profile per B Channel (4)	U1UMA	NA NA	NA.	NA.	\$5.61	NA	NA NA	NA NA	NA	NA
2.1	2-Wire ISDN Port(2) (3) including all available features, per month	U1PMA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	\$38.68	NA NA
	NRC - 1st	U1PMA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$106.40	NA NA
+	NRC - Add'l	U1PMA	NA NA	NA NA	NA NA	NA	NA	NA	NA	\$106.40	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$67.52	NA NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$67.52	NA NA
2 1	2-Wire ISDN Port(2) (3) including three available features, per month	U1PMA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$36.01	NA NA
2-1	NRC - 1st	U1PMA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$70.32	NA NA
	NRC - Add'l	U1PMA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$70.32	NA NA
				NA NA	NA NA						NA NA
	NRC - Incremental Charge - Manual Service Order - 1st NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA			NA NA	NA	NA	NA	\$67.52	1
٠.	· · · · · · · · · · · · · · · · · · ·	SOMAN	NA Outpoons	NA	NA Outpooling	NA	NA Outstand	NA DOLLO DA	NA 0.170.75	\$67.52	NA Tabasasa
4-1	4-Wire ISDN DS1 Port, per month	UEPEX	\$186.02	NA	\$163.16	NA	\$194.72	\$213.21	\$179.75	\$214.79	\$308.00
	NIDO 4 4	===:/									To be
	NRC - 1st	UEPEX	\$244.85	NA	\$186.80	NA	\$181.89	\$244.12	\$241.63	\$278.37	negotiated
	NIDO ALIII	HEDEV	004405		0400.00		* 404.00	001110	0044.00	0070.07	To be
	NRC - Add'l	UEPEX	\$244.85	NA	\$186.80	NA	\$181.89	\$244.12	\$241.63	\$278.37	negotiated
	NRC - Disconnect Charge - 1st	UEPEX	\$51.19	NA	NA	NA	\$27.11	\$53.32	NA	NA	NA
	NRC - Disconnect Charge - Add'l	UEPEX	\$51.19	NA	NA	NA	\$27.11	\$53.32	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$54.75	NA	\$37.88	NA	\$33.18	\$51.03	\$53.89	\$65.48	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$54.75	NA	\$37.88	NA	\$33.18	\$51.03	\$53.89	\$65.48	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$11.53	NA	NA	NA	\$7.73	\$8.51	NA	NA	NA
	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$11.53	NA	NA	NA	\$7.73	\$8.51	NA	NA	NA
4-	4-Wire ISDN DS1 Port including all available features, per month	UEPEX	NA	NA	NA	\$275.48	NA	NA	NA	\$251.00	NA
	NRC - 1st	UEPEX	NA	NA	NA	\$181.27	NA	NA	NA	\$311.73	NA
	NRC - Add'l	UEPEX	NA	NA	NA	\$116.42	NA	NA	NA	\$311.73	NA
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	NA	NA	NA	NA	NA	\$65.48	NA
	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	\$65.48	NA
2-1	2-Wire Analog Line Port (PBX), per month										
	2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.18	\$2.35	\$1.90
	LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
	LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
	LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
	LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS	UEPLD	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
	TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	UEPT2	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
	TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	UEPTO	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
1	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING						* -			*	
	PORT	UEPA2	\$2.07	NA	NA	NA	NA	NA	NA	NA	NA

Version 1Q00:6/5/00 7/7/00 Filefolder

	AND	OTHER SERVICE	ES							
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA										
CALLING PORT	UEPL2	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE	OLI LD	Ψ2.07	Ψ2.00	ψ1.00	Ψ2.01	ΨΣ.ΣΟ	Ψ2.11	Ψ2.00	Ψ2.00	Ψ1.00
CALLING PORT	UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
	UEP12	INA	INA	INA	INA	INA	INA	INA	INA	\$1.90
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING	LIEDTO									04.00
PORT	UEPTO	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPXA	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPXB	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD										
CAPABLE PORT	UEPXE	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING		, ,	*	*		•	·		*	
PORT WITHOUT LUD	UEPXF	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA NA	NA NA	NA	\$2.61	NA NA	NA NA	NA NA	NA NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA NA	NA NA	NA NA	\$2.61	NA NA	NA NA	NA NA	NA NA	NA NA
	ОЕРАП	INA	INA	INA	\$2.01	INA	INA	INA	INA	INA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT	==									
WITHOUT LUD	UEPXJ	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL										
CALLING PORT	UEPXK	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ADMINISTRATIVE CALLING PORT	UEPXL	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ROOM CALLING PORT	URPXM	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
			V	¥e	V	*	4	¥=	V=.00	******
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL	OLI XII	INA	INA	INA	INA	INA	INA	INA	INA	ψ1.50
DIACOUNT ROOM CALLING PORT	LIEDVO	¢0.07	#0.00	Ф4 ОГ	CO C4	#0.00	CO 44	#0.00	ФО О Б	£4.00
	UEPXO	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL	==./=									
DISCOUNT CALLING PORT	UEPXP	NA	NA	NA	NA	\$2.20	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY										
CALLING PORT	UEPXQ	NA	NA	NA	NA	NA	\$2.11	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL										
CALLING PORT	UEPXR	NA	NA	NA	NA	NA	\$2.11	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$2.07	\$2.00	\$1.85	\$2.61	\$2.20	\$2.11	\$2.00	\$2.35	\$1.90
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS										
CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	\$2.35	NA
	02.7								ψ2.00	
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$1.90
2-WIRE VOICE UNBUNDLED 1-BX COLLECTION TO CALLING TOKEN 2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV	UEFAU	INA	INA	INA	INA	INA	INA	INA	INA	\$1.90
CALLING PORT	UEPXV	NA	NIA	NIA	NIA	NA	NIA	NA	NIA	\$1.90
CALLING FOR I	UEPAV	INA	NA	NA	NA	INA	NA	INA	NA	\$1.90
UNBUNDLED LOOP BILLING USOC (REQUIRES ONE PER PORT)	UEPLX		ļ		ļ	ļ		ļ		↓
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCP									
		1								
NRC - 1st	UEPPC	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$21.60	\$24.36	NA
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	UEPPO	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
LINE SIDE GRADGIADEED OO I WARD FOX INDIAN - DOSINESS	OL: 1 O	Ψ∠ 1.33	ψ56.00	ψ17.10	ψ50.47	ψ10.43	ΨΖΖ.30	Ψ ∠ 4.04	Ψ ∠ 4.30	INA

Version 1Q00:6/5/00 7/7/00 Filefolder

DESCRIPTION		OTHER SERVIC			I/V		МС	NC	66	TNI
DESCRIPTION DESCRIPTION DESCRIPTION DESCRIPTION DESCRIPTION DE LA PROPRIE DE LA PROPRIED DE LA PROPRIE DE LA PROPRIED DE LA PROPRIE DE LA PROP	USOC	AL	FL	GA	KY	LA	MS	NC 101101	SC	TN
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS	UEPLD	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	UEPT2	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	UEPTO	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING										
PORT	UEPA2	\$21.93	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA										l
CALLING PORT	UEPL2	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE										
CALLING PORT	UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING										
PORT	UEPTO	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPXA	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPXB	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	UEPXC	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXD	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD FOR I	UEFAD	φ <u>∠</u> 1.33	φ30.00	φ17.10	φ30.47	φ10.43	φ∠∠.30	φ ∠ 4.04	φ∠4.30	INA
CAPABLE PORT	HEDVE	004.00	# 00.00	047.40	COO 47	040.40	# 00.00	004.04	# 04.00	
	UEPXE	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING										1
PORT WITHOUT LUD	UEPXF	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPXG	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT										İ
WITHOUT LUD	UEPXJ	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL										Ĭ
CALLING PORT	UEPXK	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ADMINISTRATIVE CALLING PORT	UEPXL	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ROOM CALLING PORT	URPXM	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
			*			* -	•	•	*	
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL	OLI AIT	1471	10/	1471	1471	1471	14/1	147.	1471	10.
DIACOUNT ROOM CALLING PORT	UEPXO	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL	OLI XO	Ψ21.93	ψ30.00	Ψ17.10	ψ30.47	ψ10. 4 3	Ψ22.90	Ψ24.04	Ψ24.50	INA
DISCOUNT CALLING PORT	UEPXP	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY	ULFAF	INA	INA	INA	INA	\$10.43	INA	INA	INA	INA
	LIEDYO	NIA	NIA	NIA	NIA	NIA	# 00.00	NIA.	NIA	NIA
CALLING PORT	UEPXQ	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL	LIEDVO		l	l			***			1
CALLING PORT	UEPXR	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	\$21.93	\$38.00	\$17.16	\$36.47	\$16.43	\$22.98	\$24.04	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS				ĺ						1
CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	\$24.36	NA
				_						1
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV										
CALLING PORT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Add'l										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	UEPRD	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$21.60	\$24.36	NA
1 12 TITLE VOICE GROUNDELED COMMINATION 2-WAT I DA TRONK - RESIDENCE	OLIND	Ψ2 1.30	ψ10.00	ψ17.10	ψυυ.τι	ψ10.70	ΨΖΖ.30	Ψ21.00	ΨΔ-1.00	IΝΛ

Version 1Q00:6/5/00 7/7/00 Filefolder

AND OTHER SERVICES DESCRIPTION LISOC AL EL GA KY LA MS NC SC TN											
DESCRIPTION		USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX T	RUNK - BUSINESS	UEPPC	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUS	SINESS	UEPPO	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUS	SINESS	UEPP1	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS		UEPLD	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS		UEPT2	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
TN OUTWARD CALLING PLAN PBX TRUNK - BUSINES	SS	UEPTO	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA.
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION P		020	Ψ21.00	ψ10.00	ψινιιο	ψου. 17	Ψ10.10	Ψ22.00	Ψ0.00	Ψ2 1.00	1471
PORT	_	UEPA2	\$21.93	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION P		OLITIE	Ψ21.00	101	14/1	14/1	10/1	1471	14/	1471	147.
CALLING PORT		UEPL2	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORT		UEPLD	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION P		OLILD	Ψ21.93	ψ15.00	Ψ17.10	ψ50.47	ψ10.43	Ψ22.90	ψ9.00	Ψ24.30	INA
CALLING PORT		UEPT2	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX		UEP12	INA	INA	INA	INA	INA	INA	INA	INA	INA
I PORT		UEPTO	NA	NA	NA	NA	NA	NA	NA	NA	NIA
1 1 1 2											NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION P		UEPXA	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HO		UEPXB	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS	-	UEPXC	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWIT		UEPXD	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWIT											
CAPABLE PORT		UEPXE	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY											
PORT WITHOUT LUD		UEPXF	NA	NA	NA	\$36.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AR		UEPXG	NA	NA	NA	\$37.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIL	JM CALLING PORT	UEPXH	NA	NA	NA	\$38.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA	A CALLING PORT										
WITHOUT LUD		UEPXJ	NA	NA	NA	\$39.47	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA I	LOCAL OPTIONAL										
CALLING PORT		UEPXK	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOS	PITAL ECONOMY										
ADMINISTRATIVE CALLING PORT		UEPXL	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOS	PITAL ECONOMY										
ROOM CALLING PORT		URPXM	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
					•	*		•		*	
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX	HOTEL/HOSPITAL										
ECONOMY ADMINIATRATIVE CALLING PORTTENNES	SEE CALLING PORT	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX											
DIACOUNT ROOM CALLING PORT		UEPXO	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX			+= : : : :	Ţ.3.00	Ţ .	+=====	+ . 5	ţ	Ţ	+=	1
DISCOUNT CALLING PORT		UEPXP	NA	NA	NA	NA	\$16.43	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI		JE1741	14/3	14/3	14/7	14/3	ψ10.70	14/3	14/7	14/7	13/3
CALLING PORT		UEPXQ	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI		JLI AG	14/7	14/7	IVA	11/7	14/7	Ψ22.30	147	147	INC
CALLING PORT		UEPXR	NA	NA	NA	NA	NA	\$22.98	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXN		UEPXS	\$21.93	\$15.00	\$17.16	\$36.47	\$16.43	\$22.98	\$9.05	\$24.36	NA NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING FBXIII		ULFAS	φ21.33	φ13.00	φ17.10	φ30.41	φ10. 4 3	φ ∠ ∠.30	φσ.υυ	φ24.30	INA
CALLING PORT		UEPXT	NA	NA	NA	NA	NA	NA	NA	\$24.36	NA
I DALLING FOR I		UEPAI	INA	INA	INA	INA	INA	INA	INA	Φ24.30	INA
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & ME	EMBUIS CALLING BODT	HEDVII	NI A	NI A	N/A	N/A	NI A	NI A	NI A	NI A	NI A
		UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE		LIEDW/			NIA.			NIA			
CALLING PORT		UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	NA
									İ	1	l

Version 1Q00:6/5/00 7/7/00 Filefolder

			OTHER SERVIC	ES							
DES	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	NRC - Disconnect Charge - 1st										
	2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
+	LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS		\$6.21	NA NA	NA.	NA NA	\$3.77	\$6.56	NA NA	NA NA	NA.
++	TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS		\$6.21	NA NA	NA.	NA NA	\$3.77	\$6.56	NA NA	NA NA	NA
++	TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS		\$6.21	NA NA	NA NA	NA NA	\$3.77	\$6.56	NA NA	NA NA	NA
++	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING		φυ.Ζ1	INA	INA	INA	φ3.77	φ0.50	INA	INA	INA
	PORT		\$6.21	NA	NA	NA	NA	NA	NA	NA	NA
++	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA		\$6.21	INA	INA	INA	INA	INA	INA	INA	INA
	CALLING PORT		NIA	210	N1.0	N10	ΦO 77	NIA.	N10	N1.0	
			NA 00.04	NA	NA	NA	\$3.77	NA 00.50	NA	NA	NA
	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE										
	CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING										
	PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
Ш	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD										
	CAPABLE PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
+	2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING		ψ0.Z ·				ψο	ψ0.00			1
	PORT WITHOUT LUD		NA	NA	NA	NA	NA	NA	NA	NA	NA
+	2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT		NA	NA NA	NA.	NA NA	NA.	NA.	NA NA	NA NA	NA
++	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT		NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	2-WIRE VOICE UNBUNDLED 1-BA KENTUCKY AREA CALLING PORT		INA	INA	INA	INA	INA	INA	INA	INA	INA
	WITHOUT LUD		NIA	NA	NIA	NA	NIA	NIA	NIA	NIA	NA
	2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL		NA	INA	NA	INA	NA	NA	NA	NA	NA NA
			NIA	NIA	NIA	NIA	ФО 77	NIA	NIA	NIA	NIA.
	CALLING PORT		NA	NA	NA	NA	\$3.77	NA	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
	ADMINISTRATIVE CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
	ROOM CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
	ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
	DIACOUNT ROOM CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL										
	DISCOUNT CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY										
	CALLING PORT		NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
H	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL										1
	CALLING PORT		NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
++	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT		\$6.21	NA NA	NA	NA NA	\$3.77	\$6.56	NA NA	NA NA	NA.
++	2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS		Ψ0.21	14/1	1473	14/1	ψ0.77	ψ0.00	14/1	14/1	1373
	CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
++	O/LENTO I OTT		13/3	14/-1	14/-1	11/1	11/1	19/3	14/1	11/-1	13/3
	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT		NA	NA	NA	NA	NA	NIA	NA	NA	NA
	12-WINE VOICE UNDUNDLED PDA COLLIER VILLE & WEWPHIS CALLING PORT		INA	INA	NA	INA	NA	NA	INA	INA	INA

Version 1Q00:6/5/00 7/7/00 Filefolder

DESCRIPTION	USOC	OTHER SERVIC AL	FL	GA	KY	LA	MS	NC	sc	TN
	0300	AL	FL	GA	N1	LA	IVIS	NC	30	IN
CALLING PORT		NIA	NIA	NIA	NIA	NIA	NIA	NIA	NIA	NIA
CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
NIDO DI LOLI ALIII										
NRC - Disconnect Charge - Add'l										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING										
PORT		\$6.21	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA										
CALLING PORT		NA	NA	NA	NA	\$3.77	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE		Ψ0.21	14/1	1471	1471	ψο	ψ0.00	14/1	1473	1471
CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING		INA	INA	INA	INA	INA	INA	INA	INA	INA
		NIA	NIA	NA	NA	NIA	NIA	NIA	NA	NIA
		NA \$6.21	NA NA	NA NA	NA NA	NA CO 77	NA CC FC	NA	NA NA	NA NA
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT						\$3.77	\$6.56	NA		NA
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD										
CAPABLE PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING										
PORT WITHOUT LUD		NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT										
WITHOUT LUD		NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL										
CALLING PORT		NA	NA	NA	NA	\$3.77	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY										
ADMINISTRATIVE CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY		*****				44	40.00			
ROOM CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		Ψ0.21	14/1	1171	14/1	ΨΟ	Ψ0.00	14/1	14/1	14/1
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL										
ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL		INA	INA	INA	INA	INA	INA	INA	INA	INA
DIACOUNT ROOM CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL		Φ0.∠1	INA	INA	INA	φ3.11	Φ0.00	INA	INA	INA
		# 0.04	N/A	NI A	N1 A	60.77	#0.50	N/A	N. A	N. A
DISCOUNT CALLING PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY										
CALLING PORT		NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL										
CALLING PORT		NA	NA	NA	NA	NA	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT		\$6.21	NA	NA	NA	\$3.77	\$6.56	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS										
CALLING PORT		NA	NA	NA	NA	NA	NA	NA	NA	NA

Version 1Q00:6/5/00 7/7/00 Filefolder

	ANI	D OTHER SERVICI	ES							
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
		N10		NIA		NIA.	N10	NIA.		
		NA	NA	NA	NA	NA	NA	NA	NA	NA
		NA	NA	NA	NA	NA	NA	NA	NA	NA
CALLING FOR I		INA	INA	INA	INA	INA	INA	INA	INA	INA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$26.94	\$41.86	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$12.97	NA NA	\$8.42	NA NA	\$8.06	\$11.34	\$12.76	\$14.46	NA NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA NA	NA	NA NA	\$8.94	\$16.06	NA	NA	NA NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$0.48	NA NA	NA NA	NA NA	Ψ0.94 NA	NA	NA NA	NA NA	NA NA
INTO - Inclemental charge - Mandal Service Order - Disconnect - Add 1	SOMAN	Ψ0.40	INA	INA	INA	INA	INA	INA	INA	INA
2-Wire Analog Line Port (PBX) including all available features, per month	UEPPC	NA	NA	NA	NA	NA	NA	NA	\$8.67	NA
NRC - 1st	UEPPC	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA	\$60.60	NA NA
NRC - Add'l	UEPPC	NA NA	NA	NA	NA NA	NA	NA NA	NA	\$60.60	NA NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA	NA	NA.	NA NA	NA NA	NA	\$41.86	NA NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA NA	NA	NA	NA	NA	NA NA	NA	\$14.46	NA NA
2-Wire Analog Line Port (PBX) including three available features, per month	UEPPC	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA	\$5.38	NA NA
NRC - 1st	UEPPC	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$28.89	NA NA
NRC - Add'l	UEPPC	NA NA	NA	NA	NA NA	NA	NA NA	NA	\$28.89	NA NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA	\$41.86	NA NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$14.46	NA NA
111.0 Introductional Charge Manda Colvice Class 7.001	0017117	14/1	1471	14/1	10/	1471	147.	1471	ψ11.10	107
2-Wire Analog Hunting, per line per month	HTGUX	See features	NA	NA	\$0.29	NA	See features	NA	See features	NA
NRC - 1st	HTGUX	See features	NA	NA.	\$2.14	NA NA	See features	NA	See features	NA.
NRC - Add'I	HTGUX	See features	NA	NA NA	\$2.14	NA NA	See features	NA	See features	
Coin Port, per month	ттоох	\$2.34	NA	\$2.05	\$3.04	\$2.50	\$2.32	NA	\$2.77	\$1.90
		Ψ2.01	1471	Ψ2.00	ψο.σ τ	Ψ2.00	Ψ2.02	1471	Ψ2.77	BST GSST
NRC - 1st		\$21.93	NA	\$17.16	\$40.71	\$16.43	\$22.98	NA	\$24.75	A4.3.1
		Ψ21.00	1471	ψ17.10	Ψ10.71	ψ10.10	Ψ22.00	10.	Ψ2 1.7 0	BST GSST
NRC - Add'l		\$21.93	NA	\$17.16	\$40.71	\$16.43	\$22.98	NA	\$24.75	A4.3.1
NRC - Disconnect Charge - 1st		\$5.21	NA	NA	NA	\$4.15	\$6.56	NA	NA NA	NA
NRC - Disconnect Charge - Add'l		\$5.21	NA	NA	NA	\$4.15	\$6.56	NA	NA	NA.
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$25.93	NA	\$18.94	NA	\$18.14	\$25.52	NA	\$43.48	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$12.97	NA	\$8.42	NA	\$8.06	\$11.34	NA	\$14.57	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$16.33	NA	NA	NA	\$9.86	\$16.06	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$0.48	NA	NA	NA	NA	NA	NA	NA	NA
4- Wire Coin Port, per month		NA	NA	NA	NA	NA	NA	\$2.59	NA	NA
NRC - 1st		NA	NA	NA	NA	NA	NA	\$21.60	NA	NA
NRC - Add'l		NA	NA	NA	NA	NA	NA	\$21.60	NA	NA
NRC - Disconnect Charge - 1st		NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Disconnect Charge - Add'l		NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st		NA	NA	NA	NA	NA	NA	\$26.94	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l		NA	NA	NA	NA	NA	NA	\$12.76	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st		NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l		NA	NA	NA	NA	NA	NA	NA	NA	NA
VERTICAL FEATURES		1								
			No add'l		No add'l					
Local Switching Features offered with Port, Per month	N/A	NA	charge	NA	charge	\$8.28	NA	NA	See above	NA
Three-Way Calling, per month		\$1.12	NA	NA	NA	NA	\$1.32	\$0.89	\$1.10	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA

Version 1Q00:6/5/00

7/7/00 Filefolder

		OTHER SERVIC								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Customer Changeable Speed Calling, per month		\$0.08	NA	NA	NA	NA	\$0.0755	\$0.17	\$0.1247	NA
NRC NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Waiting		\$0.03	NA	NA	NA	NA	\$0.033	\$0.09	\$0.0665	NA
INRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA NA	\$0.5466	NA NA	NA NA	NA
Remote Activation of Call Fordwarding, per month		\$0.18	NA NA	NA	NA NA	NA NA	\$0.4859	\$0.85	\$0.3743	NA NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect			NA NA	NA NA						NA NA
		\$0.55			NA	NA	\$0.5466	NA	NA On soon	
Cancel Call Waiting, per month		\$0.01	NA	NA	NA	NA	\$0.0082	\$0.01	\$0.0099	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Automatic Callback, per month		\$0.29	NA	NA	NA	NA	\$0.9977	\$0.66	\$0.8015	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Automatic Recall, per month		\$0.28	NA	NA	NA	NA	\$0.3164	\$0.29	\$0.3102	NA
I INRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA NA	NA.	NA NA	NA NA	\$0.5466	NA NA	NA NA	NA NA
Calling Number Delivery, per month		\$0.22	NA	NA	NA NA	NA NA	\$0.1817	\$0.33	\$0.3272	NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Calling Number Delivery Blocking, per month		\$1.17	NA	NA	NA	NA	\$0.9913	\$0.02	\$0.3684	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Customer Originated Trace, per month		\$0.14	NA	NA	NA	NA	\$0.1918	\$0.14	\$0.1402	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Selective Call Rejection, per month		\$0.13	NA	NA	NA	NA	\$0.1721	\$0.13	\$0.1528	NA
I INRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA.	NA	NA.	NA NA	\$0.5466	NA NA	NA.	NA NA
Selective Call Forwarding, per month		\$0.05	NA NA	NA NA	NA NA	NA NA	\$0.3460	\$0.28	\$0.1287	NA NA
NRC			NA NA	NA NA	NA NA	NA NA				NA NA
		\$1.03					\$1.02	\$1.51	\$1.51	
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Selective Call Acceptance, per month		\$0.29	NA	NA	NA	NA	\$0.4010	\$0.33	\$0.3283	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Multiline Hunt Service (Rotary)										ł
Service per line, (in addition to port), per month		\$0.11	NA	NA	NA	NA	\$0.1271	\$0.14	\$0.1301	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Forwarding Variable, per month		\$0.05	NA	NA	NA	NA	\$0.0474	\$0.10	\$0.0768	NA
NRC		\$1.03	NA.	NA	NA.	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect		\$0.55	NA NA	NA NA	NA NA	NA NA	\$0.5466	NA	NA NA	NA NA
Call Forwarding Busy Line, per month	+	\$0.03	NA NA	NA NA	NA NA	NA NA	\$0.0279	\$0.08	\$0.0603	NA NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$1.02			NA NA
	_							\$1.51	\$1.51	
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Forwarding Don't Answer All Calls, per month		\$0.03	NA	NA	NA	NA	\$0.0308	\$0.09	\$0.0655	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Remote Call Forwarding, per month		\$1.36	NA	NA	NA	NA	\$1.47	\$0.95	\$1.41	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Wareho Touristano	i	+					+5.00			

Version 1Q00:6/5/00 7/7/00 Filefolder

		OTHER SERVIC								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Call Transfer, per month		\$0.12	NA	NA	NA	NA	\$0.1404	\$0.14	\$0.1392	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call Hold, per month		\$0.03	NA	NA	NA	NA	\$0.0190	\$0.15	\$0.0677	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC – Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Toll Restricted Service, per month		\$0.04	NA	NA	NA	NA	\$0.0387	\$0.10	\$0.0743	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Message Waiting Indicator – Stutter Dial Tone, per month		\$0.03	NA	NA	NA	NA	\$0.0356	\$0.03	\$0.0318	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Anonymous Call Rejection, per month		\$0.93	NA	NA	NA	NA	\$0.9519	\$1.29	\$1.13	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Shared Call Appearances of a DN, per month		\$0.41	NA	NA	NA	NA	\$0.5015	\$0.29	\$0.3513	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Multiple Call Appearances, per month		\$0.09	NA	NA	NA	NA	\$0.0932	\$0.07	\$0.0891	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
ISDN Bridged Call Exclusion, per month		\$0.00	NA	NA	NA	NA	\$0.0013	\$0.0011	\$0.0013	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
Call by Call Access, per month		\$28.29	NA NA	NA	NA	NA	\$50.89	\$19.83	\$0.3621	NA
NRC		\$28.94	NA	NA	NA	NA	\$28.61	\$33.33	\$33.36	NA
NRC - Disconnect		\$5.22	NA NA	NA	NA	NA	\$5.16	NA	NA	NA
Privacy Release, per month		\$0.01	NA NA	NA	NA	NA NA	\$0.0030	\$0.0041	\$0.0116	NA
NRC		\$1.03	NA	NA	NA	NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA NA	NA	NA	NA	\$0.5466	NA NA	NA NA	NA
Multi Appearance Directory Number Calls, per month		\$0.10	NA NA	NA NA	NA NA	NA NA	\$0.1115	\$0.13	\$0.1048	NA.
NRC		\$1.03	NA NA	NA	NA	NA NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA	NA	NA	NA	\$0.5466	NA NA	NA NA	NA
Make Set Busy, per month		\$0.01	NA NA	NA NA	NA	NA	\$0.0013	\$0.0020	\$0.0101	NA
NRC		\$1.03	NA NA	NA	NA	NA NA	\$1.02	\$1.51	\$1.51	NA
NRC - Disconnect		\$0.55	NA NA	NA	NA	NA NA	\$0.5466	NA NA	NA NA	NA NA
Teen Service (Res. Dist. Alerting Service), per month		\$0.15	NA	NA	NA	NA	\$0.1071	\$0.26	\$0.2149	NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect		\$0.55	NA NA	NA NA	NA NA	NA NA	\$0.5466	NA	NA	NA NA
Code Restriction and Diversion, per month		\$0.04	NA NA	NA	NA	NA NA	\$0.0464	\$0.09	\$0.0708	NA NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect		\$0.55	NA NA	NA NA	NA NA	NA NA	\$0.5466	NA	NA	NA NA
Call Park, per month		\$0.04	NA NA	NA NA	NA NA	NA NA	\$0.0443	\$0.09	\$0.0694	NA NA
NRC		\$1.03	NA NA	NA	NA NA	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect		\$0.55	NA NA	NA NA	NA NA	NA NA	\$0.5466	NA	NA	NA NA
Automatic Line, per month		\$0.09	NA NA	NA NA	NA NA	NA NA	\$0.5466	\$0.14	\$0.1179	NA NA
NRC		\$1.03	NA NA	NA NA	NA NA	NA NA	\$1.02	\$1.51	\$1.51	NA NA
NRC - Disconnect		\$1.03	NA NA	NA NA	NA NA	NA NA	\$0.5466	\$1.51 NA	NA	NA NA
I IAVO - DISCOILIECE		\$0.55	INA	INA	INA	INA	\$0.5466	INA	INA	INA
2-WIRE ISDN BRI FEATURES		1					 			
Shared Primary Number-First Appr On Each Add'l Terminal	DS1FJ	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Secondary Only Dn (Shared/Non-Shared) First Appearance	LLDSF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Version 1Q00:6/5/00	LLDSF	100	100	טטו	100	100	100	100	טטו	
7/7/00 Filefolder										

7/7/00 Filefolder

		OTHER SERVIC								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Shared Secondary Only Dn-First Appr On Each Add'l Term	DS1F1	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Shared Non-ISDN DN	DOE	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Privacy Release	DS1FU	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Manual Exclusion	DS1FM	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable-Voice Or Voice/Data	LLNCV	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable – Data	LLOCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable – Feature Button – Voice	GJXCF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Variable – Feature Button – Data	LLPCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Busy Line – Voice Or Voice/Data	LLQCV	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Busy Line – Data	LLRCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Frwdng Busy Line–Prgrmmbl–Voice Or Voice/Data	M6AVA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Busy Line – Programmable - Data	M6ADF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Don't Answer – Voice Or Voice/Data	LLSCV	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Don't Answer – Data	LLUCD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwdng Don't Answer–Prgrmmble Voice Or Voice/Data	M6BVA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Don't Answer – Programmable - Data	M6BDF	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Frwdng Multiple Simultaneous – Voice Or Voice/Data	M6CV5	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Forwarding Multiple Simultaneous – Data	M6CD5	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Conference, Drop, Hold And Transfer	DS1FN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Six-Way Conference, Drop, Hold And Transfer	LLY6P	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Multi-Line Hunt Group – Voice Or Voice/Data	HTG	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Multi-Line Hunt Group – Data	HTGSD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Speed Calling	LLZSU	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Visual Message Waiting Indicator	LLAVP	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Audible Message Waiting Indicator	MWW	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Additional Call Appearance, PDN Or DN	DS1FG	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Tracing	NST	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Return	NSS	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Preferred Call Forwarding	NCE	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Block	NSY	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Repeat Dialing	NSQ	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Agencies/Law Enforcement	NOB	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub Customers	NOBNN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For General Public	NOBPC	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub, And Non-Listed Customer	NOBPP	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub Customers	NOBNP	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Per Line Blocking For Non-Pub Customers	NOBNR	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Return Denial Of, Per Activation	BCR	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Repeat Dialing, Denial Of, Per Activation	BRD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Automatic Line/Direct Connect	M6GN9	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Make Set Busy	M6MPD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Selective Call Acceptance	M6K16	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Park/Call Retrieve	M6HP6	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Call Transfer System Exception	M6QTD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Make Set Busy – Intragroup	M6MGD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
All Customized Code Restrictions	CREX+	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Additional Listings	CLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Additional Listing No Rate	FLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Cross Reference Listing	LLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Non-Pub Listing No Rate	NP3	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Non-List Listing	NLT	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Non-List Listing No Rate	NLE	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Alternate Call Listing	FNA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Manual Service Order Charge	SOMAN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD

Version 1Q00:6/5/00 7/7/00 Filefolder

AND	OTHER SERVIC	ES							
USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
SRG++	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
	\$0.01	NA	NA	NA	NA	\$0.0105	\$0.0107	\$0.0138	NA
	\$1.03	NA	NA	NA	NA	\$1.02	\$1.47	\$1.47	NA
	\$0.55	NA	NA	NA	NA	\$0.5466	NA	NA	NA
	NA	NA	NA	NA	NA	NA	NA		
	\$1.03	NA	NA	NA			\$1.51	\$1.51	NA
	\$0.55	NA	NA	NA		\$0.5466	NA	NA	NA
	NA	NA	NA	NA	NA	NA	NA		
	\$2.88	NA	NA	NA	NA	\$2.84	\$5.42	\$1.36	NA
	\$0.96	NA	NA	NA	NA	\$0.95	\$0.95	\$0.71	NA
	\$4.80	NA	NA	NA	NA	\$4.73	\$1.89	\$7.35	NA
	\$0.96	NA	NA	NA	NA	\$0.95	NA	\$0.95	NA
	\$2.88	NA	NA	NA	NA	\$2.84	NA	NA	NA
N/A		\$0.0175	\$0.0016333		\$0.0021			\$0.0019295	\$0.0019
			NA		NA			NA	NA
N/A	\$0.0002	NA	\$0.0001564	NA	\$0.0002	\$0.0001927	NA	\$0.0002581	NA
N/A	\$0.00063								\$0.000676
		NA	\$0.0002126	NA	\$0.0003	\$0.0002834	NA	\$0.0004034	NA
	-						-		
	-						-		
	†						†		
	1						1		
	USOC SRG++	USOC AL SRG++ TBD \$0.01 \$1.03 \$0.55 NA \$1.03 \$0.55 NA \$2.88 \$0.96 \$4.80 \$0.96 \$2.88 N/A \$0.0018 N/A NA N/A \$0.0002	\$RG++ TBD TBD \$0.01 NA \$1.03 NA \$0.55 NA NA NA \$1.03 NA \$1.03 NA \$1.03 NA \$0.55 NA NA NA \$0.96 NA \$4.80 NA \$2.88 NA \$0.96 NA \$2.88 NA \$0.96 NA \$2.88 NA \$0.96 NA \$1.03 NA \$0.006 NA \$1.03 NA NA \$2.88 NA \$2.88 NA \$3.096 NA \$3.096 NA \$3.096 NA \$3.096 NA \$3.096 NA \$3.096 NA \$3.096 NA \$3.096 NA \$3.096 NA \$3.0006 NA \$3.0006 NA \$3.0006 NA NA \$3.0006 NA NA \$3.0006 NA	USOC AL FL GA SRG++ TBD TBD TBD \$0.01 NA NA NA \$1.03 NA NA NA NA NA NA NA \$0.55 NA NA NA NA NA NA NA NA NA NA NA \$2.88 NA NA NA \$0.96 NA NA NA \$0.96 NA NA NA \$2.88 NA NA NA N/A \$0.0018 \$0.0175 \$0.0016333 N/A NA \$0.0005 NA N/A \$0.0002 NA \$0.0001564	USOC AL FL GA KY	USOC AL FL GA KY LA	USOC AL FL GA KY LA MS	USOC	USOC AL FL GA KY LA MS NC SC SRG++ TBD
	ANL	OTHER SERVIC	ES	1		1	1		
--	-------	--------------	---	---	---	---	---	---	-------------
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc
INTEROFFICE TRANSPORT									
Common (Shared) Transport									
Common (Shared) Transport per mile per mou	N/A	\$0.00001	\$0.000012	\$0.000008	\$0.0000049	\$0.000083	\$0.0000091	\$0.00001	\$0.0000121
Common (Shared) Transport Facilities Termination per mou	N/A	\$0.00045	\$0.0005	\$0.0004152	\$0.000426	\$0.00047	\$0.0004281	\$0.00034	\$0.0004672
Interoffice Channel - Dedicated Transport - VG		70.000	***************************************	***************************************	***************************************	***************************************	***************************************	***************************************	***********
Interoffice Channel - Dedicated Transport - 2-Wire VG - per mile	1L5XX	\$0.03390	NA	\$0.0222	\$0.03	\$0.0384	\$0.0323	\$0.0282	\$0.0373
Interoffice Channel - Dedicated Transport - 2-Wire VG - facility termination per		70.0000		4 0.00	40.00	40.000	¥ 0.00	4 0.0000	40.00.0
	U1TV2	\$18.49	NA	\$17.07	\$27.66	\$19.10	\$21.33	\$18.00	\$21.42
NRC - 1st	U1TV2	\$144.27	NA	\$79.61	\$142.31	\$104.23	\$144.77	\$137.48	\$136.44
NRC - Add'l	U1TV2	\$54.15	NA	\$36.08	\$56.21	\$39.91	\$56.06	\$52.58	\$51.37
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$40.34	NA	\$18.94	\$37.21	\$26.20	\$36.86	\$38.07	\$39.63
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$40.54	NA	\$18.94	\$37.21	\$26.20	\$36.86	\$38.07	\$39.63
Interoffice Channel - Dedicated Transport - DS0 - 56/64 KBPS				*	•	* -	•	*	,
Interoffice Channel - Dedicated Transport - DS0 - per mile per month	1L5XX	\$0.0339	\$0.0252	\$0.0222	\$0.03	\$0.0384	\$0.0323	\$0.0282	\$0.0373
Interoffice Channel - Dedicated Transport - DS0 - facility termination per month	U1TD6	\$17.81	\$21.33	\$16.45	\$26.95	\$18.37	\$20.64	\$17.40	\$20.71
NRC - 1st	U1TD6	\$144.27	\$137.15	\$79.61	\$142.31	\$104.23	\$144.77	\$137.48	\$136.44
NRC - Add'l	U1TD6	\$54.15	\$64.45	\$36.08	\$56.21	\$39.91	\$56.06	\$52.58	\$51.37
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$40.34	NA	\$18.94	\$37.21	\$26.20	\$36.86	\$38.07	\$39.63
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$40.34	NA	\$18.94	\$37.21	\$26.20	\$36.86	\$38.07	\$39.63
Interoffice Channel - Dedicated Transport - DS1				*	•	* -	•	*	,
Interoffice Channel - Dedicated Transport - DS1- per mile per month	1L5XX	\$0.69	\$0.6013	\$0.4523	\$0.45	\$0.7831	\$0.6598	\$0.5753	\$0.7598
Interoffice Channel - Dedicated Transport - DS1 facility termination per month	U1TF1	\$79.69	\$99.79	\$78.47	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98
NRC - 1st	U1TF1	\$223.59	\$45.91	\$147.07	\$298.18	\$160.49	\$222.81	\$217.17	\$216.27
NRC - Add'l	U1TF1	\$168.60	\$44.18	\$111.75	\$231.23	\$123.03	\$168.92	\$163.75	\$162.70
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$40.34	NA	\$18.94	NA	\$26.20	\$36.83	\$38.07	\$39.63
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$40.34	NA	\$18.94	NA	\$26.20	\$36.86	\$38.07	\$39.63
Interoffice Channel - Dedicated Transport - DS3									
Interoffice Channel - Dedicated Transport - DS3 - per mile per month	1L5XX	\$11.93	\$10.25	\$7.07	\$12.06	\$16.15	\$15.02	\$12.98	\$19.14
Interoffice Channel - Dedicated Transport - DS3 - facility termination per month	U1TF3	\$736.60	\$994.83	\$743.41	\$1,112.02	\$1,131.09	\$744.38	\$720.38	\$904.49
NRC - 1st	U1TF3	\$877.36	\$884.71	\$878.95	\$858.75	\$883.62	\$812.30	\$794.94	\$856.96
NRC - Add'l	U1TF3	\$540.46	\$552.81	\$542.61	\$524.95	\$545.50	\$596.55	\$579.55	\$522.20
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$101.69	NA	\$98.49	\$94.57	\$99.02	\$92.05	\$91.26	\$99.09
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$101.69	NA	\$98.49	\$94.57	\$101.69	\$92.05	\$91.26	\$99.09
Interoffice Channel - Dedicated Transport - STS-1									
Interoffice Channel - Dedicated Transport - STS-1 - per mile per month	1L5XX	\$11.93	\$10.25	\$7.07	\$12.06	\$16.15	\$13.48	\$11.62	\$19.14
Interoffice Channel - Dedicated Transport - STS-1 - facility termination per month	U1TFS	\$733.93	\$966.49	\$733.72	\$1,088.67	\$1,114.68	\$692.52	\$814.72	\$944.40
NRC - 1st	U1TFS	\$858.02	\$868.23	\$856.62	\$858.75	\$861.17	\$858.15	\$857.29	\$861.20
NRC - Add'l	U1TFS	\$524.50	\$530.74	\$523.64	\$524.94	\$526.42	\$524.58	\$524.05	\$526.44
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$94.49	\$95.61	\$94.34	\$94.57	\$94.84	\$94.50	\$94.41	\$94.84
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$94.49	\$95.61	\$94.34	\$94.57	\$94.84	\$94.50	\$94.41	\$94.84
Local Channel - Dedicated Transport									
Local Channel - Dedicated Transport - 2-Wire VG									
Monthly Recurring	ULDV2	\$14.61	\$18.02	\$13.91	\$22.26	\$14.94	\$17.83	\$14.82	\$16.83
NRC - 1st	ULDV2	\$572.46	\$477.33	\$382.95	\$597.14	\$401.17	\$565.31	\$553.80	\$554.00
NRC - Add'l	ULDV2	\$92.07	\$124.32	\$62.40	\$110.52	\$66.35	\$93.30	\$86.69	\$88.58
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$45.12	NA	\$18.94	\$41.46	\$29.54	\$41.57	\$42.17	\$43.75
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$18.73	NA	\$8.42	NA	\$19.46	\$27.39	\$12.76	\$13.55
Local Channel - Dedicated Transport - 4-Wire VG									
Monthly Recurring	ULDD6	\$15.77	\$19.01	\$14.99	\$23.38	\$16.21	\$19.03	\$15.87	\$18.05
NRC - 1st	ULDD6	\$581.14	\$77.33	\$368.44	\$585.15	\$407.11	\$573.83	\$562.23	\$562.46
NRC - Add'l	ULDD6	\$95.21	\$124.32	\$64.05	\$98.53	\$68.61	\$96.40	\$92.67	\$91.57
NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$45.12	NA	\$18.94	\$98.53	\$29.54	\$41.57	\$42.17	\$43.64
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$18.73	NA	\$8.42	\$11.99	\$19.46	\$27.39	\$12.76	\$13.55

Version 1Q00:6/5/00 7/7/00 Filefolder

TRANSPORT

			AND	OTHER SERVIC	ES							
DE	SCI	RIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
Loc	cal (Channel - Dedicated Transport - DS1										
		Monthly Recurring	TMECS	\$35.52	\$44.35	\$38.36	\$43.80	\$43.80	\$38.91	\$35.68	\$37.20	\$40.27
		NRC - 1st	TMECS	\$549.85	\$246.50	\$356.15	\$538.95	\$396.86	\$588.53	\$534.48	\$534.81	\$343.71
		NRC - Add'l	TMECS	\$475.02	\$230.49	\$312.89	\$464.94	\$342.92	\$501.32	\$462.69	\$462.81	\$277.86
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$91.22	NA	\$44.22	\$87.71	\$61.82	\$81.30	\$86.15	\$87.99	\$23.51
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	NA	NA	NA	NA	NA	NA	\$1.77	\$3.11	\$21.75
Loc		Channel - Dedicated Transport - DS3										
		DS3 - per mile per month	1L5NC	\$34.21	\$30.65	\$23.06	\$34.00	\$30.34	NA	NA	\$44.13	\$23.76
		DS3 - Facility Termination per month	ULDF3	\$536.23	\$598.84	\$531.90	\$635.09	\$669.01	NA	\$498.87	\$582.93	\$607.28
		NRC - 1st	ULDF3	\$877.36	\$884.71	\$878.95	\$858.75	\$883.62	\$858.15	\$562.25	\$856.96	\$877.70
		NRC - Add'l	ULDF3	\$540.46	\$552.81	\$542.61	\$524.95	\$545.50	\$524.58	\$527.88	\$522.20	\$540.32
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$101.69	NA	\$98.49	NA	\$99.02	NA	\$56.25	NA	\$102.75
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$101.69	NA	\$98.49	NA	\$99.02	NA	\$56.25	NA	\$102.75
Loc	cal (Channel - Dedicated Transport - STS-1										
		STS-1 - per mile per month	1L5NC	\$24.82	\$27.61	\$19.93	\$30.04	\$29.89	\$38.98	\$24.39	\$29.97	\$25.11
		STS-1 - Facility Termination per month	ULDFS	\$502.62	\$681.61	\$516.91	\$610.64	\$693.02	\$531.39	\$555.92	\$556.66	\$615.65
		NRC - 1st	ULDFS	\$1,084.17	\$1,097.06	\$1,082.37	\$1,085.09	\$1,088.15	\$1,084.33	\$1,083.24	\$1,088.19	\$1,085.73
		NRC - Add'l	ULDFS	\$682.02	\$690.14	\$680.91	\$682.61	\$684.53	\$682.13	\$681.44	\$684.56	\$683.01
		NRC - Incremental Charge - Manual Service Order - 1st	SOMAC	\$96.08	\$97.23	\$95.93	\$96.17	\$96.44	\$96.10	\$96.00	\$96.44	\$96.22
		NRC - Incremental Charge - Manual Service Order - Add'l	SOMAC	\$96.08	\$97.23	\$95.93	\$96.17	\$96.44	\$96.10	\$96.00	\$96.44	\$96.22
CH	ANI	NELIZATION										
	DS3	3 Channelization (DS3 to DS1)										
	per	Channelized System per month	MQ3	\$210.87	\$213.22	\$173.51	\$236.32	\$245.84	\$229.30	\$226.81	\$204.07	\$225.59
		NRC - 1st	MQ3	\$355.25	\$280.12	\$284.43	\$425.41	\$259.76	\$356.80	\$351.95	\$423.77	\$265.08
		NRC - Add'l	MQ3	\$245.86	\$196.07	\$199.98	\$303.33	\$182.64	\$247.40	\$243.76	\$295.21	\$185.94
		NRC -1sr - Disconnect	MQ3	\$78.43	\$64.06	\$66.76	NA	\$60.96	\$79.94	\$77.90	NA	\$61.09
		NRC -Add'l - Disconnect	MQ3	\$63.70	\$52.60	\$55.25	NA	\$50.46	\$65.20	\$63.32	NA	\$50.31
		NRC - Channel System - Incremental Cost - Manual Svc. Order -1st	SOMAC	\$28.44	NA	\$21.61	\$41.47	\$19.74	\$26.95	\$28.13	\$43.41	\$21.71
		NRC - Channel System - Incremental Cost - Manual Svc. Order - Add'l	SOMAC	\$13.47	NA	\$9.61	NA	\$8.77	\$11.98	\$13.33	\$15.36	\$10.46
		NRC - Channel System - Incremenatl Cost - Manual Svc. Order - Disconnect - 1st	SOMAC	\$18.46	NA	\$13.61	NA	\$12.43	\$16.97	\$18.26	NA	\$14.21
		NRC - Channel System - Incremenatl Cost - Manual Svc. Order - Disconnect - Add	SOMAC	\$1.50	NA	NA	NA	NA	NA	\$1.48	NA	\$1.46
		Interface per month	1PQE1	\$4.53	\$6.31	\$7.13	\$8.52	\$7.55	\$5.58	\$4.61	\$9.69	\$3.91
		NRC - 1st	1PQE1	\$15.85	\$13.39	\$13.45	\$15.86	\$12.29	\$15.85	\$15.76	\$15.54	\$12.61
		NRC - Add'l	1PQE1	\$11.35	\$9.59	\$9.63	\$11.36	\$8.80	\$11.35	\$11.28	\$11.13	\$9.03
		hannelization (DS1 to DS0)	1101	0.100.50	* 400.00	0407.07	#	****	0.1.10.07	* 477.70	0470.04	* 405.04
+		Channelized System per month	MQ1 MQ1	\$139.58 \$269.98	\$163.88 \$208.64	\$137.97 \$212.01	\$200.01 \$302.82	\$209.87	\$146.87	\$177.72	\$179.81	\$165.21 \$197.21
		NRC - 1st NRC - Add'I	MQ1	\$269.98 \$163.04	\$208.64 \$126.61	\$212.01	\$302.82 \$184.20	\$193.63 \$118.37	\$271.52 \$164.56	\$267.19 \$161.43	\$304.00 \$178.92	\$197.21
		NRC - Add I NRC -1sr - Disconnect	MQ1	\$34.88			\$184.20 NA	\$26.44	\$36.38	\$34.55	\$178.92 NA	\$25.66
+		NRC -1sr - Disconnect NRC -Add'l - Disconnect	MQ1	\$34.88	\$26.42 \$15.95	\$28.95 \$18.43	NA NA	\$26.44 \$16.83	\$22.82	\$34.55	NA NA	\$15.81
++		NRC - Add 1 - Disconnect NRC - Channel System - Incremental Cost - Manual Svc. Order -1st	SOMAC	\$28.44	NA	\$10.43	\$41.47	\$19.74	\$26.95	\$21.14	\$43.41	\$21.71
+		NRC - Channel System - Incremental Cost - Manual Svc. Order -1st	SOMAC	\$13.47	NA NA	\$9.61	\$41.47 \$11.99	\$8.77	\$11.98	\$13.33	\$15.36	\$10.46
+	_	NRC - Channel System - Incremental Cost - Manual Svc. Order - Add i	SOMAC	\$18.46	NA NA	\$13.61	\$11.99 NA	\$12.43	\$16.97	\$18.26	NA	\$10.46
+		NRC - Channel System - Incremental Cost - Manual Svc. Order - Disconnect -Add	SOMAC	\$1.50	NA NA	NA	NA NA	NA	NA	\$1.48	NA NA	\$1.46
D.S		hannization Interfaces	001111110	ψ1.00	1471	1471	177	1471	1471	Ψ1.10	1471	ψ1.13
ŤĬ		er OCU-DP(data) card per month(2.4-64kbps)	1D1DD	\$2.61	\$3.13	\$2.65	\$2.94	\$3.12	\$2.86	\$2.88	\$3.36	\$2.46
11		NRC - 1st	1D1DD	\$15.85	\$13.39	\$13.45	\$15.86	\$12.29	\$15.85	\$15.76	\$15.54	\$12.61
11		NRC - Add'l	1D1DD	\$11.35	\$9.59	\$9.63	\$11.36	\$8.80	\$11.35	\$11.28	\$11.13	\$9.03
11		er VG card per month	1D1VG	\$1.26	\$1.78	\$1.48	\$1.40	\$1.62	\$1.45	\$1.64	\$1.93	\$1.25
		NRC - 1st	1D1VG	\$15.85	\$13.39	\$13.45	\$15.86	\$12.29	\$15.85	\$15.76	\$15.54	\$12.61
		NRC - Add'l	1D1VG	\$11.35	\$9.59	\$9.63	\$11.36	\$8.80	\$11.35	\$11.28	\$11.13	\$9.03
┰┪												
		FIBER										
Versio		200:6/5/00										

7/7/00 Filefolder

TRANSPORT

	DES	CRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
П	Per f	our fiber strands, per route mile or fraction thereof, per month	1L5DF	\$59.84	\$55.35	\$44.22	\$64.64	\$65.29	\$70.35	\$49.88	\$72.45	\$52.67
		NRC - Per each four-fiber dark fiber arrangement - 1st	1L5DF	\$2,518.66	\$1,715.61	\$1,355.29	\$2,304.00	\$1,685.19	\$2,389.99	\$2,277.00	\$2,406.00	\$1,672.44
П		NRC - Per each four-fiber dark fiber arrangement - Add'l	1L5DF	\$835.08	\$622.68	\$273.69	\$740.93	\$580.11	\$804.32	\$733.08	\$765.30	\$509.09
		Where the state Commission has adopted rates for the rate elements containedherein, it is the intent of the Parties to reflect such rates in thisExhibit and to apply the same consistent with applicable FCC and Commissionrules and orders.										

Version 1Q00:6/5/00 7/7/00 Filefolder

TRANSPORT

· · · · · · · · · · · · · · · · · · ·	Α	ND OTHER SERV	ICES		1	•				
DESCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
UNBUNDLED LOOP COMBINATIONS										
Unbundled Loop/Port Combinations (Notes 4 & 5)										
								1		
UNBUNDLED LOOP BILLING USOC (REQUIRES ONE PER PORT)	UEPLX	This USOC to	be used for l	Jnbundled Loop v	vhen ordering	Loop/Port Cor	nbination			
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCX	This USOC to	be used for L	ocal Number Por	tability when o	rdering Loop/	Port Combinat	ions		
					l	I		Ī		
Zone 1 / Top 8 MSAs in BellSouth Region										
Currently Combined										
Customers with less than 4 DS0 Equivalent										
2-Wire Voice Grade Loop with 2-Wire Line Port										
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	TBD	\$16.55	NA	\$12.59	NA	\$16.60	\$16.71	NA	\$20.71	NA
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	TBD	\$25.51	NA	\$14.26	NA	\$26.69	\$21.45	NA	\$29.35	NA
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	TBD	\$44.44	NA NA	\$21.62	NA NA	\$51.85	\$29.75	NA NA	\$37.68	NA NA
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)	TBD	NA NA	NA NA	NA NA	NA NA	NA	\$38.59	NA NA	NA	NA
RC - 2- Wire Voice Grade Loop - Zone 1	UEPLX	\$19.04	\$17.00	\$10.80	\$20.00	\$14.05	\$14.59	\$14.27	\$17.02	\$18.00
RC - 2- Wire Voice Grade Loop - Zone 2	UEPLX	\$19.04 NA	NA	\$10.80	NA	\$24.14	\$19.33	NA	\$25.66	NA
RC - 2- Wire Voice Grade Loop - Zone 3	UEPLX	NA NA	NA NA	\$19.83	NA NA	\$49.30	\$27.63	NA NA	\$33.99	NA NA
RC - 2- Wire Voice Grade Loop - Zone 4	UEPLX	NA NA	NA NA	NA	NA NA	NA	\$36.47	NA NA	NA	NA NA
ING - 2- Wile Voice Glade Loop - Zolle 4	OLILA	INA	INA	INA	INA	INA	φ30.47	INA	INA	INA
RC - Exchange Port - 2-Wire Line Port	TBD	\$2.07	\$2.00	\$1.79	\$2.61	\$2.20	\$2.11	\$2.19	\$2.35	\$1.90
NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, with change	USACC	\$10.00	\$10.00	\$2.01	\$10.00	\$10.00	\$10.00	\$2.77	\$10.00	\$10.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, with change	USACC	\$10.00	\$10.00	\$0.3108000	\$10.00	\$10.00	\$10.00	\$0.40	\$10.00	\$10.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add i, with change	USACC USAC2	\$10.00	\$10.00	\$2.01	\$10.00	\$10.00	\$10.00	\$0.40	\$10.00	\$10.00
	USAC2	\$10.00	\$10.00	\$0.3108000	\$10.00	\$10.00	\$10.00	\$2.77	\$10.00	\$10.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, no change			· ·							
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge,										
Electronic, per LSR received from the CLEC by one of the OSS interactive	COMEC	#2.50	#2.50	#2.50	#2.50	#2.50	#2.50	60.50	#2.50	ድረ ደር
interfaces NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost -	SOMEC	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	COMAN	NIA	NA	# 22.67	NI A	NIA	NA	£40.40	NA	NIA
Manual Svc.Order vs. Electronic - 1st NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost -	SOMAN	NA	INA	\$33.67	NA	NA	NA	\$40.18	NA	NA
Manual Svc.Order vs. Electronic - Add'l	0014411	NIA	NIA	67.00		NIA.	NI A	DO 45		NI A
	SOMAN	NA	NA	\$7.88	NA	NA	NA	\$9.45	NA	NA
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic	0014411	040.00	# 40.00	N1.0	# 40.00	# 40.00	# 40.00		# 40.00	# 40.00
Ivianual Svc.Older vs. Electionic	SOMAN	\$19.99	\$19.99	NA	\$19.99	\$19.99	\$19.99	NA	\$19.99	\$19.99
NDC 2 Wire Voice Crade Lean/Line Part Combination Cube accurate Database			1					 		
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Electronic	TBD	NIA	NA	NIA	NIA	NA	NA	61.42	NA	NA
Update - Electronic NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database	IBD	NA	INA	NA	NA	NA NA	INA	\$1.42	INA	NA NA
Update - Manual Service Order	TBD	NA	NA	NA	NA	NA	NA	\$10.27	NA	NA
Opuate - Manual Service Order	IBD	INA	INA	INA	INA	INA	INA	φ1U.2 <i>I</i>	INA	INA
2 Wire Vaice Crade Lean with 2 Wire DID Trunk Part			-					-		
2- Wire Voice Grade Loop with 2 - Wire DID Trunk Port RC- 2 Wire Voice Grade Loop with 2 - Wire Line Port	TDD	NA	NIA	NIA	NIA	NIA	NIA	\$22.70	NIA	NIA
	TBD	_	NA NA	NA NA	NA NA	NA NA	NA NA	\$23.79	NA NA	NA NA
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - 1st	TBD	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	\$13.26	NA NA	NA NA
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - Addl	TBD	NA	NA	NA	NA	NA	NA	\$8.39	NA	NA
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - Incremental Cost- Manual	TOD							# 50.00		
Service Order - 1st	TBD	NA	NA	NA	NA	NA	NA	\$53.89	NA	NA
NRC- 2- Wire Voice Grade Loop with 2- Wire Line Port - Incremental Cost- Manual	TDD	NIA						644.04		
Service Order - Addl	TBD	NA	NA	NA	NA	NA	NA	\$11.34	NA	NA
O Mine IODN Divided Overdad and width Co. 1. IODN Divide Divide			1					-		
Version 1000:6/5/00 Digital Grade Loop with 2-wire ISDN Digital Port								L		

7/7/00 Filefolder

LOOP-PORT COMBOS

		Al	ND OTHER SERV	CES	1	•	•	•	•	1	
DES	SCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
	RC - 2-Wire ISDN Digital Grade Loop	USL2X	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$18.32
	RC - Exchange Port - 2-Wire ISDN Line Side Port	UEPPB	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$15.72
	RC- 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port	TBD	NA	NA	NA NA	NA	NA	NA	\$43.45	NA	NA
+++	in a management of the state of								ψ10110		
	NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - 1st conversion	USACB	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$117.23
++	141.0 2 Wile lebit bighar Grade 200p/2 Wile lebit bighar of the conversion	00/100	ψ171.00	Ψ17 1.00	Ψ171.00	ψ17 1.00	ψ17-1.00	Ψ17-1.00	ψ17 1.00	Ψ17 1.00	ψ117.20
	NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Add'l conversion	USACB	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$117.23
++	NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Non Feature	OUNOD	ψ174.00	ψ174.00	ψ174.00	ψ174.00	ψ174.00	ψ174.00	ψ174.00	ψ17-4.00	ψ117.23
	Subsequent Activity	USASB	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$212.88
+++	Oubsequent Activity	OOAOD	Ψ200.13	Ψ200.13	Ψ200.13	Ψ200.13	Ψ200.13	Ψ200.13	\$200.13	Ψ200.13	Ψ212.00
H	4-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port		+								
++	RC - 4-Wire ISDN Digital Grade Loop	USL4P	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$61.74
+++	RC - Exchange Port - 4-Wire ISDN Digital Trunk Port	UEPPP	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$73.62
+++	NRC - 4-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Trunk Port	ULFFF	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$179.01	\$73.02
	Combination - 1st conversion	USACP	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$328.53
++	NRC - 4-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Trunk Port	USACP	Φ 4 01.51	Φ401.51	Ψ 4 01.31	Φ 4 01.51	Φ401.51	Φ401.51	ֆ 4 01.51	Φ401.51	\$320.33
	Combination - Add'l conversion	USACP	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$328.53
+++	Combination - Add reconversion Combination - Subsequent Channel Activity - Per Channel	USASP	\$36.92	\$36.92	\$36.92		\$36.92	\$36.92	\$36.92	\$36.92	\$28.39
+++	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port	USASP	\$30.92	\$36.92	\$30.92	\$36.92	\$30.92	\$36.92	\$36.92	\$36.92	\$28.39
	Combination - Subsequent Inward/2-way Telephone Numbers	DD7T0	04.47	04.47	04.47	04.47	04.47	04.47	04.47	04.47	# 0.04
\vdash		PR7TG	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$0.94
	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port	DD=TD	000.47	000.47	#00 4 7	000.47	000.47	000.47	000.47	000.47	
$\sqcup \!\!\!\! \perp \!\!\!\! \perp$	Combination - Subsequent Outward Telephone numbers	PR7TP	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$22.36
	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port										
$\sqcup \sqcup \sqcup$	Combination - Subsequent Inward Telephone Numbers	PR7ZT	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$44.71
	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port										
$\sqcup \! \! \perp \! \! \perp$	Combination - Subsequent Service Order Per Order	USASP	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$189.76
$\sqcup \sqcup$											
$\sqcup \! \! \perp \! \! \perp$	4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port										
	RC - 4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port	TBD	NA	NA	NA	NA	NA	NA	\$241.72	NA	NA
											1
$\sqcup \sqcup \sqcup$	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - 1st	TBD	NA	NA	NA	NA	NA	NA	\$481.51	NA	NA
											İ
$\sqcup \sqcup \sqcup$	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Addl	TBD	NA	NA	NA	NA	NA	NA	\$481.51	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										İ
$\sqcup \sqcup \sqcup$	Subsequent Channel Activation - Per Channel	TBD	NA	NA	NA	NA	NA	NA	\$36.92	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										Ĭ
	Subsequent Inward/2way Telephone Numbers	TBD	NA	NA	NA	NA	NA	NA	\$1.17	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										Ï
	Subsequent Outward Telephone Numbers	TBD	NA	NA	NA	NA	NA	NA	\$28.17	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										İ
Ш	Subsequent Inward Telephone Numbers	TBD	NA	NA	NA	NA	NA	NA	\$56.33	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										1
Ш	Subsequent Service Order Per Order	TBD	NA	NA	NA	NA	NA	NA	\$255.25	NA	NA
Ш			1								
Ш	4 - Wire DS1 Digital Loop with 4 - Wire DID Trunk Port										
Ш	RC - 4 - Wire DS1 Digital Loop with 4 - Wire DID Trunk Port	TBD	NA	NA	NA	NA	NA	NA	\$186.23	NA	NA
Ш	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DID Trunk Port - 1st	TBD	NA	NA	NA	NA	NA	NA	\$490.38	NA	NA
											1
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port - Addl	TBD	NA	NA	NA	NA	NA	NA	\$490.38	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DIDTrunk Port - Subsequent										1
\coprod	Channel Activation - Per Channel	TBD	NA	NA	NA	NA	NA	NA	\$146.91	NA	NA

Version 1Q00:6/5/00 7/7/00 Filefolder

LOOP-PORT COMBOS

		Ai	ND OTHER SERV	ICES							
DE	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
T I	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										
	Subsequent Telephone Numbers	TBD	NA	NA	NA	NA	NA	NA	\$120.96	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										
	Subsequent Signaling Changes	TBD	NA	NA	NA	NA	NA	NA	\$29.65	NA	NA
	NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -										
+	Subsequent Service Order Per Order	TBD	NA	NA	NA	NA	NA	NA	\$127.63	NA	NA
+	Customers with 4 or more DS0 Equivalent										
\pm	2-Wire Voice Grade Loop with 2-Wire Line Port	TBD	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
\pm	2-vviie voice Grade Loop with 2-vviie Line i ort	100	Note 3	Note 3	Note 5	Note 5	Note 5	Note 5	Note 3	Note 5	Note 5
	All Other Loop/Port Combinations	TBD	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	The other Edoph of Combinations	100	1511	1511	1511	1514	1511	1511	1514	1511	1511
No	t Currently Combined										
	Customers with less than 4 DS0 Equivalent										
	2-Wire Voice Grade Loop with 2-Wire Line Port										
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	TBD	NA	NA	NA	NA	NA	NA NA	NA	NA	NA
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - 2- Wire Voice Grade Loop	UEPLX	\$19.04	\$17.00	\$12.55	\$20.00	\$19.35	\$21.26	\$14.27	\$22.49	\$18.00
	RC - Exchange Port - 2-Wire Line Port	TBD	\$14.00	\$14.00	\$1.79	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, with change	USACC	Note 3	Note 3	\$2.01	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, with change	USACC	Note 3	Note 3	\$0.3108000	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, no change	USAC2	Note 3	Note 3	\$2.01	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, no change	USAC2	Note 3	Note 3	\$0.3108000	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost -										
	Manual Svc.Order vs. Electronic - 1st	TBD	Note 3	Note 3	\$33.67	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost -										Ĭ
	Manual Svc.Order vs. Electronic - Add'l	TBD	Note 3	Note 3	\$7.88	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port										
	RC - 2-Wire ISDN Digital Grade Loop	USL2X	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$19.08	\$18.32
	RC - Exchange Port - 2-Wire ISDN Line Side Port	UEPPB	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$24.37	\$15.72
	NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - 1st conversion	USACB	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$117.23
	, ,										
\perp	NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Add'l conversion	USACB	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$174.35	\$117.23
	NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Port - Non Feature										
+	Subsequent Activity	USASB	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$286.15	\$212.88
+	4-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port		+	 			 				
+	RC - 4-Wire ISDN Digital Grade Loop RC - 4-Wire ISDN Digital Grade Loop	USL4P	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$62.71	\$61.74
+	RC - 4-Wire ISDN Digital Grade Loop RC - Exchange Port - 4-Wire ISDN Digital Trunk Port	USL4P UEPPP	\$179.01	\$179.01	\$62.71 \$179.01	\$179.01	\$62.71	\$62.71 \$179.01	\$179.01	\$62.71 \$179.01	\$61.74 \$73.62
+	NRC - 4-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Trunk Port	UEPPP	φ1/9.U1	φ1/9.UI	φ1/9.UI	φ1/9.UT	φ1/9.U1	φ1/9.01	φ1/9.UT	φ1/9.01	Φ13.0∠
	Combination - 1st conversion	USACP	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$328.53
+	NRC - 4-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Trunk Port	USAUF	υ 4 01.31	φ 4 σ1.31	φ 4 σ1.31	η 4 ο1.51	φ 4 01.51	φ401.31	φ 4 01.51	φ401.51	φ3∠0.33
	Combination - Add'l conversion	USACP	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$481.51	\$328.53
+	Combination - Add Conversion Combination - Subsequent Channel Activity - Per Channel	USASP	\$36.92	\$36.92	\$36.92	\$36.92	\$36.92	\$36.92	\$36.92	\$36.92	\$28.39
+	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port	55/101	ψ00.02	Ψ00.02	ψ00.02	ψ00.02	ψ00.02	ψ00.0 <u>2</u>	₩00.0 <u>2</u>	₩00.0 <u>2</u>	Ψ20.00
	Combination - Subsequent Inward/2-way Telephone Numbers	PR7TG	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$1.17	\$0.94
	252	110710	ψ1.17	ψ1.17	ψ1.17	Ψ1.11	Ψ1.17	ψ1.17	Ψ1.17	Ψ1.17	Ψ0.54

Version 1Q00:6/5/00 7/7/00 Filefolder

LOOP-PORT COMBOS

		AN	ID OTHER SERV	ICES	•	•			,	•	
DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port										
	Combination - Subsequent Outward Telephone numbers	PR7TP	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$28.17	\$22.36
	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port										
	Combination - Subsequent Inward Telephone Numbers	PR7ZT	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$56.33	\$44.71
	NRC - 4-Wire ISDN Digital Grade Loop/4-wire ISDN Digital Trunk Port										
	Combination - Subsequent Service Order Per Order	USASP	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$255.25	\$189.76
	All Other Loop/Port Combinations	TBD	TBN	TBN	Note 2	TBN	TBN	TBN	TBN	TBN	TBN
C	customers with 4 or more DS0 Equivalent										
	2-Wire Voice Grade Loop with 2-Wire Line Port	TBD	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	All Other Loop/Port Combinations	TBD	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN	TBN
	ner MSAs in BellSouth Region	<u> </u>		-				<u> </u>	-		
Curr	rently Combined										
\coprod	2-Wire Voice Grade Loop with 2-Wire Line Port										
\coprod	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
$\sqcup \bot$	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
	RC - 2- Wire Voice Grade Loop	UEPLX	\$19.04	\$17.00	\$12.55	\$20.00	\$19.35	\$21.26	\$14.27	\$22.49	\$18.00
	RC - Exchange Port - 2-Wire Line Port	TBD	\$2.07	\$2.00	\$1.79	\$2.61	\$2.20	\$2.11	\$2.19	\$2.35	\$1.90
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, with change	USACC	\$10.00	\$10.00	\$2.01	\$10.00	\$10.00	\$10.00	\$2.77	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, with change	USACC	\$10.00	\$10.00	\$0.3108000	\$10.00	\$10.00	\$10.00	\$0.40	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, no change	USAC2	\$10.00	\$10.00	\$2.01	\$10.00	\$10.00	\$10.00	\$2.77	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, no change	USAC2	\$10.00	\$10.00	\$0.3108000	\$10.00	\$10.00	\$10.00	\$0.40	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge,										
	Electronic, per LSR received from the CLEC by one of the OSS interactive										
++	interfaces	SOMEC	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost -	TDD	NIA		#00.07	NIA	N14	N1.0	040.40	NIA	N. 1.0
++	Manual Svc.Order vs. Electronic - 1st NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost -	TBD	NA	NA	\$33.67	NA	NA	NA	\$40.18	NA	NA
	Manual Svc.Order vs. Electronic - Add'l	TBD	NA	NA	\$7.88	NA	NIA	NA	\$9.45	NA	NA
+++	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost -	IBD	NA	INA	\$7.88	NA	NA	INA	\$9.45	NA	INA
	Manual Svc.Order vs. Electronic	SOMAN	\$19.99	\$19.99	NA	\$19.99	\$19.99	\$19.99	NA	\$19.99	\$19.99
++	IVIATION SYS. CICCHOTHO	JOINAIN	ψ13.33	ψ13.33	INA	ψ13.33	ψ13.33	Ψ13.55	INA	ψ13.33	ψ13.33
++	All Other Loop/Port Combinations	TBD	TBN	TBN	Note 2	TBN	TBN	TBN	TBN	TBN	TBN
++	Pari Other Ecopit of Combinations	טטו	IDIN	אום ו	14016 2	ווטו	1014	וטוז	ווטו	ווטו	אום ו
Not	Currently Combined		+								
1.131	2-Wire Voice Grade Loop with 2-Wire Line Port		+								-
++	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
++	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	TBD	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	TBD	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
H +	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)	TBD	NA	NA.	NA NA	NA.	NA NA	NA	NA NA	NA NA	NA NA
HT	RC - 2- Wire Voice Grade Loop	UEPLX	\$19.04	\$17.00	\$12.55	\$20.00	\$19.35	\$21.26	\$14.27	\$22.49	\$18.00
HT	RC - Exchange Port - 2-Wire Line Port	TBD	\$14.00	\$14.00	\$1.79	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00
ĦĦ	V				,		,				,
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, with change - Res	UEPRL	\$90.00	\$90.00	\$59.70	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00
H				*	****	*	*	*	*	*	*
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l. w/change - Res	UEPRL	\$41.50	\$41.50	\$59.70	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
Version	orl 1Q00.6/5/00										

7/7/00 Filefolder

LOOP-PORT COMBOS

		AN	ND OTHER SERV	ICES		,	•		•	•	
DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, with change - Bus	UEPBL	\$90.00	\$90.00	\$59.70	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l. w/change - Bus	UEPBL	\$41.50	\$41.50	\$59.70	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, no change -Res	UEPRL	\$90.00	\$90.00	\$59.70	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, no change - Res	UEPRL	\$41.50	\$41.50	\$59.70	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, no change - Bus	UEPBL	\$90.00	\$90.00	\$59.70	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, no change - Bus	UEPBL	\$41.50	\$41.50	\$59.70	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st	TBD	Note 3	Note 3	\$33.67	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l	TBD	Note 3	Note 3	\$7.88	Note 3	Note 3	Note 3	Note 3	Note 3	Note 3
Ш	All Other Loop/Port Combinations	TBD	TBN	TBN	Note 2	TBN	TBN	TBN	TBN	TBN	TBN
MADE	ET RATES (INCLUDING ALL VERTICAL FEATURES)		+								-
	rently Combined										
1	2-Wire Analog Line Port (Res., Bus.), per month	TBD	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00
	2-Wire Analog Loop, per month	UEPLX	\$19.04	\$17.00	NA	\$20.00	\$19.35	\$21.26	NA	\$22.49	\$18.00
	NRC	TBD	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50	\$41.50
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive										
	interfaces	SOMEC	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Incremental Manual Service Order	SOMAN	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
	NRC - Incremental Manual Service Order Disconnect	TBD	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
Nat	Companies Compliand										
NOt	Currently Combined 2-Wire Analog Line Port (Res., Bus.), per month	TBD	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00	\$14.00
++	2-Wire Analog Line Fort (Res., Bus.), per month	UEPLX	\$14.00	\$17.00	NA	\$20.00	\$19.35	\$21.26	\$14.00 NA	\$22.49	\$18.00
++	NRC	TBD	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00	\$90.00
++	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USASC	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
+	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge,	00/100	\$10.00	\$10.00	ψ.σ.σσ	ψ.ο.σσ	\$10.00	\$10.00	\$10.00	ψ.σ.σσ	Ψ.σ.σσ
	Electronic, per LSR received from the CLEC by one of the OSS interactive										
Ш	interfaces	SOMEC	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - Incremental Manual Service Order	SOMAN	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
ШΠ	NRC - Incremental Manual Service Order Disconnect	TBD	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
NO.	TES:										
	In the absence of ordered rates by a State Commission, the rates for Currently Combined combinations of loop and port network elements will be the sum of the stand alone recurring rates of the UNEs which make up the combinations.										
			1			l .	l	l	l	l	

Version 1Q00:6/5/00 7/7/00 Filefolder

LOOP-PORT COMBOS

	AND OTHER SERVICES										
DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
2	For Georgia, on an interim basis, for those currently combined port/loop combinations defined by the Georgia Public Service Commission as not currently combined, the non-recurring and recurring rates for such UNE combinations shall be the sum of the stand										
1	Where BellSouth is not required to provide combinations of loop/port network elements, the rates for the 2-wire voice grade loop with 2-wire line port combination will be as follows: the recurring charges will be the sum of the standalone UNE loop rates										
4	Usage and Common Transport rates associated with the stand-alone UNE port elements will apply to all combinations of loop/port network elements.										
	The Extended Area Calling Plans set forth in the stand-alone UNE Port rates section will apply to combinations of the loop/port network elements.										
	Effective May 1, 2000 statewide rates will be replaced by Deaveraged Loop Rates by Zone where available. Until approximately December 31, 2000 or until such time that BellSouth billing systems have been developed to handle the new zone rate structure, BellSouth will bill at the Zone 1 Deaveraged Loop rate level only. After December 31, 2000 or such time that the billing systems have been developed to handle the new zone rate structure, BellSouth will begin billing bursuant to CLEC-1's interconnection agreement.										
	Where the state Commission has adopted rates for the rate elements containedherein, it is the intent of the Parties to reflect such rates in thisExhibit and to apply the same consistent with applicable FCC and Commissionrules and orders.										

Version 1Q00:6/5/00 7/7/00 Filefolder

LOOP-PORT COMBOS

EELS

BELLSOUTH/e.spire RATES NETWORK ELEMENTS AND OTHER SERVICES

	A	ND OTHER SERV	ICES							
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Unbundled Loop / Transport Combinations										
Enhanced Extended Link ("EEL")										
DEDICATED TRANSPORT - ALREADY COMBINED										
Local Loop - 2-wire VG - per month										
Statewide	UEAL2	\$22.43	\$17.00	\$17.89	\$23.35	\$22.84	\$25.05	\$15.88	\$26.25	\$26.02
Zone 1 (Note 1)	TBD	NA	NA	\$15.40	NA	NA	NA	NA	NA	NA
Zone 2 (Note 1)	TBD	NA	NA	\$17.78	NA	NA	NA	NA	NA	NA
Zone 3 (Note 1)	TBD	NA	NA	\$28.26	NA	NA	NA	NA	NA	NA
Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
Local Loop - 4-wire VG - per month										
Statewide	UEAL4	\$30.00	\$30.00	\$26.58	NA	\$31.52	\$30.55	\$27.49	\$35.86	\$18.00
Zone 1 (Note 1)	TBD	NA	NA	\$22.88	NA	NA	NA	NA	NA	NA
Zone 2 (Note 1)	TBD	NA	NA	\$26.42	NA	NA	NA	NA	NA	NA
Zone 3 (Note 1)	TBD	NA	NA	\$41.99	NA	NA	NA	NA	NA	NA
Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
Local Loop - 56kbps - per month										
Statewide	UDL56	\$34.15	\$48.33	\$29.92	NA	\$35.58	\$34.95	\$32.67	\$41.70	\$42.23
Zone 1 (Note 1)	TBD	NA	NA	\$26.44	NA	NA	NA	NA	NA	NA
Zone 2 (Note 1)	TBD	NA	NA	\$30.53	NA	NA	NA	NA	NA	NA
Zone 3 (Note 1)	TBD	NA	NA	\$48.53	NA	NA	NA	NA	NA	NA
Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
Local Loop - 64 kbps - per month										
Statewide	UDL64	\$34.15	\$48.33	\$29.22	NA	\$35.58	\$34.95	\$32.67	\$41.70	\$42.23
Zone 1 (Note 1)	TBD	NA	NA	\$26.44	NA	NA	NA	NA	NA	NA
Zone 2 (Note 1)	TBD	NA	NA	\$30.53	NA	NA	NA	NA	NA	NA
Zone 3 (Note 1)	TBD	NA	NA	\$48.53	NA	NA	NA	NA	NA	NA
Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
Local Loop - DS1 - per month										
Statewide	USLXX	\$64.65	\$80.00	\$60.88	\$67.96	\$72.86	\$69.59	\$62.78	\$72.55	TBD
Zone 1 (Note 1)	TBD	NA	NA	\$52.40	NA	NA	NA	NA	NA	NA
Zone 2 (Note 1)	TBD	NA	NA	\$60.51	NA	NA	NA	NA	NA	NA
Zone 3 (Note 1)	TBD	NA	NA	\$96.18	NA	NA	NA	NA	NA	NA
Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	NA	NA	NA	NA
1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2										
Local Loop - DS3 - per Mile	1L5ND	\$43.96	\$40.01	\$29.96	\$43.69	\$38.98	\$54.39	\$32.53	\$56.71	\$30.53
Local Loop - DS3 - per Facility Termination	UE3PX	\$456.18	\$470.83	\$392.61	\$436.95	\$497.08	\$427.81	\$387.01	\$510.30	\$400.21
		1	ļ	ļ	ļ				1	ļ
Local Loop - STS-1 - per Mile	1L5ND	\$43.96	\$40.01	\$29.96	\$43.69	\$38.98	\$54.39	\$32.53	\$56.71	\$30.53
Local Loop - STS-1 - per Facility Termination	UDLS1	\$456.18	\$470.83	\$392.61	\$436.95	\$497.08	\$427.81	\$387.01	\$510.30	\$400.21
		1							1	
Local Channel - Dedicated - 2-Wire VG per month	ULDV2	\$14.61	\$18.02	\$16.28	\$22.26	\$14.94	\$17.83	\$14.82	\$16.83	\$19.02
				L			***			
Local Channel - Dedicated - 4-Wire VG per month	ULDV4	\$15.77	\$19.01	\$17.18	\$23.38	\$16.21	\$19.03	\$15.87	\$18.05	\$20.14

Local Channel - Dedicated - DS1 per month	TMECS	\$35.52	\$44.35	\$38.57	\$43.80	\$43.80	\$38.91	\$35.68	\$37.20	\$40.27
Local Channel - Dedicated - DS3 - per mile per month	1L5NC	\$34.21	\$30.65	\$23.06	\$34.00	\$30.34	NA	NA	\$44.13	\$23.76
Local Channel - Dedicated - DS3 - Facility Termination per month	ULDF3	\$536.23	\$598.84	\$531.90	\$635.09	\$669.01	\$526.67	\$498.87	\$582.93	\$607.28

7/7/00 Filefolder

		Al	ND OTHER SERV	ICES							
DESCRIPT	TION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Local C	Channel - Dedicated - STS-1 - per mile per month	1L5NC	\$24.82	\$27.61	\$19.93	\$30.04	\$29.89	\$38.98	\$24.39	\$29.97	\$25.11
Local C	Channel - Dedicated - STS-1 - Facility Termination per month	ULDS1	\$502.62	\$681.61	\$516.91	\$610.64	\$693.02	\$531.39	\$555.92	\$556.66	\$615.65
	fice Channel - Dedicated - 2-Wire VG - per mile per month	1L5XX	\$0.03	NA	\$0.02	\$0.03	\$0.04	\$0.03	\$0.0282	\$0.04	\$0.02
Interof	fice Channel - Dedicated - 2-Wire VG - Facility Termination per month	U1TV2	\$18.49	NA	\$17.07	\$27.66	\$19.10	\$21.33	\$18.00	\$21.42	\$18.33
1.1	for Observat Dedicated DOO Follows are all assessed.	41.57/7	#0.04	#0.00	#0.00	#0.00	#0.04	#0.00	#0.0000	# 0.04	00.47
	fice Channel - Dedicated - DS0 - 56kbps - per mile per month	1L5XX	\$0.04	\$0.03	\$0.02	\$0.03	\$0.04	\$0.03	\$0.0282	\$0.04	\$0.17
Interof	fice Channel - Dedicated - DS0 - 56 kbps - Facility Termination per month	U1TD5	\$17.81	21.33	\$16.45	\$26.95	\$18.37	\$20.64	\$17.40	20.71	\$17.74
1-1	For Observat Dedicated DOO Office and a state of the second	41.57/7	CO 04	#0.00	#0.00	#0.00	#0.04	60.00	# 0.00	# 0.04	DO 47
	fice Channel - Dedicated - DS0 - 64kbps - per mile per month	1L5XX U1TD6	\$0.04	\$0.03	\$0.02	\$0.03	\$0.04	\$0.03 \$20.64	\$0.03	\$0.04 20.71	\$0.17 \$17.74
Interol	fice Channel - Dedicated - DS0 - 64 kbps - Facility Termination per month	01106	\$17.81	21.33	\$16.45	\$26.95	\$18.37	\$20.64	\$17.40	20.71	\$17.74
lataraff	in Channel Dedicated DC4 normals normanth	1L5XX	\$0.69	\$0.60	\$0.31	\$0.45	\$0.78	\$0.66	\$0.57530	\$0.76	\$0.35
	ice Channel - Dedicated - DS1 - per mile per month ice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	\$79.69	\$99.79	\$63.39	\$0.45 \$55.05	\$93.40	\$74.40	\$0.57530 \$71.29	\$94.98	\$75.83
interoil	ice Channel - Dedicated - DST - Facility Termination per month	UTIFT	\$79.69	\$99.79	\$63.39	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
Intoroff	ice Channel - Dedicated - DS3 - per mile per month	1L5XX	\$11.93	\$10.25	\$6.46	\$12.06	\$16.15	\$13.48	\$12.98	\$19.14	\$6.88
	ice Channel - Dedicated - DS3 - per mile per month	U1TF3	736.6	994.83	\$717.60	\$1,112.02	\$1,131.09	\$686.84	\$720.38	\$19.14	\$840.61
meron	ice Channer - Dedicated - Doo - Facility Termination per month	UIIFS	730.0	994.03	φ/1/.0U	φ1,112.02	φ1,131.09	φ000.04	Φ1 ZU.36	φ904.49	Φ04U.01
-			+		-	 	 			 	
l	in Ohannah Badisatad OTO 4 annualla manualta	41.57/7	044.00	£40.05	#7.07	# 40.00	040.45	# 40.40	644.00	040.44	#0.00
	ice Channel - Dedicated - STS-1 - per mile per month	1L5XX	\$11.93	\$10.25	\$7.07	\$12.06	\$16.15	\$13.48	\$11.62	\$19.14	\$6.88
Interoff	ice Channel - Dedicated - STS-1 - Facility Termination per month	U1TFS	\$733.93	\$966.49	\$733.72	\$1,088.67	\$1,114.68	\$692.52	\$814.72	\$944.40	\$838.65
D00.0		1100	0040.07	0010.00	000001	#	0045.04	*	# 000 04	000407	0005.50
	hannelized System per month	MQ3	\$210.87	\$213.22	\$202.91	\$236.32	\$245.84	\$229.30	\$226.81	\$204.07	\$225.59
DS	3 Interface per month (DS1 COCI)	1PQE1	\$4.53	\$6.31	\$0.67	\$8.52	\$7.55	\$5.58	\$4.61	\$9.69	\$3.91
D04 0	h 1 O 1	1404	£400.50	# 400.00	£407.07	# 000 04	# 000 07	04.40.07	6477.70	#470.04	6405.04
	hannelized System per month	MQ1	\$139.58	\$163.88	\$137.97	\$200.01	\$209.87	\$146.87	\$177.72	\$179.81	\$165.21
	J-DP(data) interface card per month (2.4-64kbs)	1D1DD	\$2.61	\$3.13	\$1.06	\$2.94	\$3.12	\$2.86	\$2.88	\$3.36	\$2.46
VG	interface card per month (DS0)	1D1VG	\$1.26	\$1.78	\$2.20	\$1.40	\$1.62	\$1.45	\$1.64	\$1.93	\$1.25
NDO	All Foliation LINE Combination II Control As In II Comments of Change										
	All Existing UNE Combination "Switch As Is" Conversion Charge	1111000	05400	A00 70	A 74.04	05400	A 54.00	AF100	**	# 54.00	05440
	C - "Switch As Is" Conversion Charge - 1st	UNCCC	\$54.03	\$63.73	\$71.04	\$54.09	\$54.23	\$54.09	\$114.00	\$54.26	\$54.13
	C - "Switch As Is" Conversion Charge - Add'l	UNCCC	\$32.11	\$33.10	\$39.60	\$32.16	\$32.24	\$32.16	\$32.10	\$32.25	\$32.17
	C rates above, if not ordered, are subject to true-up.)		1								
	ced Extended Link ("EEL")										
	VG Loop/DS1 Interoffice Channel - Dedicated Transport EEL										
	re analog voice grade loop SL2 and DS1 ded interoffice transport with										
	nnelization										
Zon		TBD	\$196.90	TBD	\$99.22	NA	\$208.13	\$229.90	NA	\$264.80	NA
Zon		TBD	\$208.11	TBD	\$101.60	NA	\$220.80	\$235.88	NA	\$275.76	NA
Zon		TBD	\$231.79	TBD	\$112.08	NA	\$252.41	\$246.32	NA	\$286.31	NA
Zon	e 4		NA	NA	NA	NA	NA	\$257.43	NA	NA	NA
	re VG Loop per month, statewide	MQ3	\$22.43	\$17.00	NA	\$23.35	NA	NA	\$15.88	NA	\$26.02
	re VG Loop per month, Zone 1 (Note 1)	TBD	NA	NA	\$15.40	NA	\$17.65	\$18.35	NA	\$21.57	NA
	re VG Loop per month, Zone 2 (Note 1)	TBD	NA	NA	\$17.78	NA	\$30.32	\$24.33	NA	\$32.53	NA
	re VG Loop per month, Zone 3 (Note 1)	TBD	NA	NA	\$28.26	NA	\$61.93	\$34.77	NA	\$43.08	NA
2-wi	re VG Loop per month, Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	\$45.88	NA	NA	NA
			1			1					
	Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$0.69	\$0.60	\$0.31	\$0.45	\$0.78	\$0.66	\$0.5753	\$0.76	\$0.35
	Interoffice Channel - Dedicated Transport EEL - Facility Termination per month	U1TF1	\$79.69	\$99.79	\$63.39	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
	Channelization System per system per month	MQ1	\$139.58	\$163.88	\$137.97	\$200.01	\$209.87	\$146.87	\$177.72	\$179.81	\$165.21
	Channelization Interface -VG per month	1PQE1	\$4.53	\$6.31	\$2.20	\$8.52	\$7.55	\$5.58	\$4.61	\$9.69	\$3.91
F	Per additional circuit in same DS1, Recurring - Zone 1	TBD	\$19.21	NA	\$17.60	NA	\$19.07	\$18.35	NA	\$23.33	NA

7/7/00 Filefolder EELS

		ND OTHER SERV								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Per additional circuit in same DS1, Recurring - Zone 2	TBD	\$30.42	NA	\$19.98	NA	\$31.74	\$24.33	NA	\$34.29	NA
Per additional circuit in same DS1, Recurring - Zone 3	TBD	\$54.10	NA	\$30.46	NA	\$63.35	\$34.77	NA	\$44.84	NA
Per additional circuit in same DS1, Recurring - Zone 4	TBD	NA	NA	NA	NA	NA	\$45.88	NA	NA	NA
NRC - Switch As Is - EEL- 1st	UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
NRC - Switch As Is - EEL - Add'I	UNCCC	\$13.33	\$15.48	\$11.27	\$15.48	\$11.10	\$13.33	\$15.48	\$28.35	\$15.48
NRC - Switch As Is - EEL - Disconnect - 1st	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Disconnect - Add'I	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
NRC - Switch As Is - EEL- Manual vs. Elect - Add'l	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56
			Orlando,							
			Miami, Ft			New		Greensboro		
INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC		NashvilleTN
NRC - 2-wire VG Loop - 1st	SOMAC	NA	\$195.00	\$157.33	NA	\$190.74	NA	\$57.99	NA	\$247.97
NRC - 2-wire VG Loop - Add'l	SOMAC	NA	\$97.00	\$120.74	NA	\$134.43	NA	\$42.37	NA	\$195.72
NRC - Interoffice Channel - DS1- Facility Termination - 1st	SOMAC	NA	\$45.91	\$166.01	NA	\$186.69	NA	\$217.17	NA	\$195.68
NRC - Interoffice Channel - DS1- Facility Termination - Add'l	SOMAC	NA	\$44.18	\$130.69	NA	\$149.23	NA	\$163.75	NA	\$156.47
NRC - DS1 Channelization System - 1st	SOMAC	NA	\$235.06	\$240.96	NA	\$220.07	NA	\$301.74	NA	\$222.87
NRC - DS1 Channelization System - Add'l	SOMAC	NA	\$142.56	\$148.03	NA	\$135.20	NA	\$182.57	NA	\$135.80
NRC - DS1 Channelization System - VG Interface - 1st	SOMAC	NA	\$13.39	\$13.45	NA	\$12.29	NA	\$15.76	NA	\$12.61
NRC - DS1 Channelization System - VG Interface - Add'l	SOMAC	NA	\$9.59	\$9.63	NA	\$8.80	NA	\$11.28	NA	\$9.03
4-wire VG Loop/DS1 Interoffice Channel - Dedicated Transport EEL										
4-wire analog voice grade loop and DS1 ded interoffice transport with										
channelization										
Zone 1	TBD	\$204.34	NA	\$101.17	NA	\$216.32	\$235.35	NA	\$274.14	NA
Zone 2	TBD	\$129.33	NA	\$110.71	NA	\$233.81	\$242.64	NA	\$289.11	NA
Zone 3	TBD	\$251.00	NA	\$126.28	NA	\$277.43	\$255.37	NA	\$303.52	NA
Zone 4	TBD	NA	NA	NA	NA	NA	\$268.93	NA	NA	NA
4-wire VG Loop, per month, statewide	UEAL4	\$30.00	\$30.00	\$26.58	NA	NA	NA	\$27.49	NA	\$18.00
4-wire VG Loop, per month, Zone 1 (Note 1)	TBD	NA	NA	\$22.88	NA	\$24.36	\$22.38	NA	\$29.47	NA
4-wire VG Loop, per month, Zone 2 (Note 1)	TBD	NA.	NA	\$26.42	NA NA	\$41.85	\$29.67	NA NA	\$44.44	NA NA
4-wire VG Loop, per month, Zone 3 (Note 1)	TBD	NA	NA	\$41.99	NA	\$85.47	\$42.40	NA	\$58.85	NA NA
4-wire VG Loop, per month, Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA NA	\$55.96	NA	NA	NA
DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$0.69	\$0.60	\$0.31	\$0.45	\$0.78	\$0.66	\$0.5753	\$0.76	\$0.35
DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per monti		\$79.69	\$99.79	\$63.39	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
DS1 Channelization System per system per month	MQ1	\$139.58	\$163.88	\$137.97	\$200.01	\$209.87	\$146.87	\$177.72	\$179.81	\$165.21
DS1 Channelization Interface -VG per month	1D1VG	\$1.26	\$1.78	\$2.20	\$1.40	\$1.62	\$1.45	\$1.64	\$1.93	\$1.25
Per additional circuit in same DS1, Recurring - Zone 1	TBD	26.65	NA	\$24.93	NA	\$27.26	\$22.38	NA	\$32.67	NA
Per additional circuit in same DS1, Recurring - Zone 2	TBD	41.64	NA	\$28.37	NA	\$44.75	\$29.67	NA	\$47.64	NA
Per additional circuit in same DS1, Recurring - Zone 3	TBD	54.1	NA	\$43.52	NA	\$88.37	\$42.40	NA	\$62.05	NA
Per additional circuit in same DS1, Recurring - Zone 4	TBD	NA	NA	NA	NA		\$55.96	NA	NA	NA
NRC - Switch As Is - EEL- 1st	UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
NRC - Switch As Is - EEL - Add'I	UNCCC	\$13.33	\$15.48	\$11.27	\$15.48	\$11.10	\$13.33	\$15.48	\$28.35	\$15.48
NRC - Switch As Is - EEL - Disconnect - 1st	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Disconnect - Add'I	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
NRC - Switch As Is - EEL- Manual vs. Elect - Add'l	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56
			Orlando,			i i				
			Miami. Ft			New		Greensboro		
INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC		NashvilleTN
NRC 4-wireVG Loop - 1st	SOMAC	NA	\$141.00	\$260.11	NA	\$334.69	NA	\$288.47	NA	\$113.50
NRC 4-wireVG Loop - Add'l	SOMAC	NA.	\$43.00	\$213.21	NA NA	\$243.53	NA	\$237.45	NA NA	\$86.00
part the temperature and t	00.00		ψ.σ.σσ	Ψ= . Ο. Ε ι		Ψ= .0.00		Ψ=07.10	. 1/ 1	, 400.00

Version 1Q00:6/5/00 7/7/00 Filefolder

EELS

		1A	ND OTHER SERVI	CES							
DE	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	NRC - DS1 - Interoffice Channel - Facility Termination - 1st	SOMAC	NA	\$45.91	\$166.01	NA	\$186.69	NA	\$217.17	NA	\$195.68
	NRC - DS1 - Interoffice Channel - Facility Termination - Add'l	SOMAC	NA	\$44.18	\$130.69	NA	\$149.23	NA	\$163.75	NA	\$156.47
	NRC - DS1 Channelization System - 1st	SOMAC	NA	\$235.06	\$240.96	NA	NA	NA	\$301.74	NA	\$222.87
	NRC - DS1 Channelization System - Add'l	SOMAC	NA	\$142.56	\$148.03	NA	NA	NA	\$182.57	NA	\$135.80
	NRC - DS1 Channelization System - Interface VG - 1st	SOMAC	NA	\$13.39	\$13.45	NA	\$12.29	NA	\$15.76	NA	\$12.61
\vdash	Three Deventurion Dystem internace version	00.110		ψ.σ.σσ	ψ.σσ		V.2.2 0		ψ10.11 C		ψ·2.σ·
	NRC - DS1 Channelization System - Interface VG - Add'I	SOMAC	NA	\$9.59	\$9.63	NA	\$8.80	NA	\$11.28	NA	\$9.03
H-	4-wire 56 kbps Loop/DS1 Interoffice Channel - Dedicated Transport EEL	SOMAC	INA	ф9.59	Ф 9.03	INA	φο.ου	INA	\$11.20	INA	\$9.03
-	DS0 digital 56 or 64 kbps loop and DS1 ded interoffice transport with										
-	channelization		0007.00		* 100.10		0010.10	0000 50	110	#070.00	
	Zone 1	TBD	\$207.66	NA	\$109.12	NA	\$219.46	\$238.58	NA	\$278.93	NA
Щ.	Zone 2	TBD	\$224.73	NA	\$113.21	NA	\$239.20	\$246.91	NA	\$296.34	NA
	Zone 3	TBD	\$280.78	NA	\$131.21	NA	\$288.44	\$261.48	NA	\$313.10	NA
	Zone 4	TBD	NA	NA	NA	NA	NA	\$276.99	NA	NA	NA
	4-wire 56 kbps Loop, per month, statewide	UNCD5	NA	NA	NA	NA	NA	NA	\$32.67	NA	\$42.23
П	4-wire 56 kbps Loop, per month, Zone 1 (Note 1)	TBD	NA	NA	\$26.44	NA	\$27.50	\$25.61	NA	\$34.26	NA
	4-wire 56 kbps Loop, per month, Zone 2 (Note 1)	TBD	NA	NA	\$30.53	NA	\$47.24	\$33.94	NA	\$51.67	NA
	4-wire 56 kbps Loop, per month, Zone 3 (Note 1)	TBD	NA	NA	\$48.53	NA	\$96.48	\$48.51	NA	\$68.43	NA
	4-wire 56 kbps Loop, per month, Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	\$64.02	NA	NA	NA
	DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$0.69	\$0.60	\$0.31	\$0.45	\$0.78	\$0.66	\$0.5753	\$0.76	\$0.35
\vdash	DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month	UNCB1	\$79.69	\$99.79	\$63.39	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
	DS1 Channelization System per system per month	UNCN1	\$139.58	\$163.88	\$137.97	\$200.01	\$209.87	\$146.87	\$177.72	\$179.81	\$165.21
	DS1 Channelization Interface - OCU-DP per month	UNC1D	\$4.53	\$6.31	\$2.20	\$8.52	\$7.55	\$5.58	\$4.61	\$9.69	\$3.91
	Per additional circuit in same DS1, Recurring - Zone 1	TBD	\$29.97	NA	\$28.42	NA	\$30.40	\$28.48	NA	\$37.46	NA
H	Per additional circuit in same DS1, Recurring - Zone 2	TBD	\$47.04	NA NA	\$32.41	NA NA	\$50.14	\$36.81	NA NA	\$54.87	NA NA
H	Per additional circuit in same DS1, Recurring - Zone 3	TBD	\$73.31	NA NA	\$49.94	NA NA	\$99.38	\$51.38	NA NA	\$71.63	NA NA
+	Per additional circuit in same DS1, Recurring - Zone 3 Per additional circuit in same DS1, Recurring - Zone 4	TBD	NA	NA NA	NA	NA NA	NA	\$66.89	NA NA	NA	NA NA
		UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
+	NRC - Switch As Is - EEL- 1st	UNCCC		\$15.48	\$12.97	\$15.48	\$12.70		\$15.48	\$28.87	\$15.48
-	NRC - Switch As Is - EEL - Add'I		\$13.33					\$13.33			
-	NRC - Switch As Is - EEL - Disconnect - 1st	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
	NRC - Switch As Is - EEL - Disconnect - Add'l	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
Щ.	NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
Щ.	NRC - Switch As Is - EEL- Manual vs. Elect - Add'l	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56
				Orlando,							
				Miami, Ft			New		Greensboro		
	INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC		NashvilleTN
	NRC - 4-wire 56 kbps Loop - 1st	SOMAC	NA	\$709.72	\$401.71	NA	\$483.59	NA	\$489.04	NA	\$698.42
	NRC - 4-wire 56 kbps Loop - Add'l	SOMAC	NA	\$483.45	\$283.84	NA	\$315.57	NA	\$337.51	NA	NA
	NRC - DS-1 Interoffice Channel - Facility Termination - 1st	SOMAC	NA	\$45.91	\$166.01	NA	\$186.69	NA	\$217.17	NA	\$195.68
	,										
	NRC - DS-1 Interoffice Channel - Facility Termination - Add'l	SOMAC	NA	\$44.18	\$130.69	NA	\$149.23	NA	\$163.75	NA	\$156.47
	NRC- New - DS1 Channelization System	COMING	14/	Ψ11.10	Ψ100.00	10/1	Ψ110.20	107	ψ100.70	10.0	ψ100.17
\vdash	NRC - DS1 Channelization System - 1st	SOMAC	NA	\$238.43	\$302.82	NA	\$297.96	NA	\$338.55	NA	\$222.87
H	NRC - DS1 Channelization System - Add'l	SOMAC	NA NA	\$145.55	\$184.20	NA NA	\$181.39	NA NA	\$200.06	NA NA	\$135.80
\vdash	NRC - DS1 Channelization System - Add1 NRC - DS1 Channelization Interface OCU-DP card per month(2.4-64kbps) - 1st	SOMAC	NA NA	\$13.39	\$13.45	NA NA	\$12.29	NA NA	\$15.76	NA NA	\$133.60
+	NRC - DS1 Channelization Interface OCU-DP card per month(2.4-64kbps) - 1st NRC - DS1 Channelization Interface OCU-DP card per month(2.4-64kbps) - Add'l	SOMAC	NA NA	\$9.59	\$9.63	NA NA	\$8.80	NA NA	\$15.76	NA NA	\$9.03
\vdash	4-wire 64 kbps Loop/DS1 Interoffice Channel - Dedicated Transport EEL	SUIVIAU	INA	φσ.υσ	φ3.03	INA	φο.ου	INA	φ11.20	INA	φ9.03
\vdash			+				 				-
	4-wire analog voice grade loop and DS1 ded interoffice transport with										
\vdash	channelization		055151		# 105 :=		001- :-	0000		0077	
\mathbf{H}	Zone 1	TBD	\$204.34	NA	\$109.12	NA	\$219.46	\$238.58	NA	\$278.93	NA
Ш	Zone 2	TBD	\$219.33	NA	\$113.21	NA	\$239.20	\$246.91	NA	\$296.34	NA
Ш	Zone 3	TBD	\$251.00	NA	\$131.21	NA	\$288.44	\$261.48	NA	\$313.10	NA

Version 1Q00:6/5/00 7/7/00 Filefolder

EELS

		<u>ID OTHER SERV</u>				1 1		1		
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
Zone 4	TBD	NA	NA	NA	NA	NA	\$276.99	NA	NA	NA
4-wire 64 kbps Loop, per month, statewide	UDL64	NA	\$48.33	NA	NA	NA	NA	\$32.67	NA	\$42.23
4-wire 64 kbps Loop, per month, Zone 1 (Note 1)	TBD	NA	NA	\$26.44	NA	\$27.50	\$25.61	NA	\$34.26	NA
4-wire 64 kbps Loop, per month, Zone 2 (Note 1)	TBD	NA	NA	\$30.53	NA	\$47.24	\$33.94	NA	\$51.67	NA
4-wire 64 kbps Loop, per month, Zone 3 (Note 1)	TBD	NA	NA	\$48.53	NA	\$96.48	\$48.51	NA	\$68.43	NA
4-wire 64 kbps Loop, per month, Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	\$64.02	NA	NA	NA
DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$0.69	\$0.60	\$0.31	\$0.45	\$0.78	\$0.66	\$0.5753	\$0.76	\$0.35
DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per monti	u1TF1	\$79.69	\$99.79	\$63.39	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
DS1 Channelization System per system per month	MQ1	\$139.58	\$163.88	\$137.97	\$200.01	\$209.87	\$146.87	\$177.72	\$179.81	\$165.21
DS1 Channelization Interface - OCU-DP per month	1D1DD	\$2.61	\$3.13	NA	\$2.94	\$3.12	\$2.86	\$2.88	\$3.36	\$2.46
Per additional circuit in same DS1, Recurring - Zone 1	TBD	\$29.97	NA	\$28.42	NA	\$30.40	\$28.48	NA	\$37.46	NA
Per additional circuit in same DS1, Recurring - Zone 2	TBD	\$47.04	NA	\$32.41	NA NA	\$50.14	\$36.81	NA NA	\$54.87	NA NA
Per additional circuit in same DS1, Recurring - Zone 3	TBD	\$73.31	NA NA	\$49.94	NA	\$99.38	\$51.38	NA NA	\$71.63	NA NA
Per additional circuit in same DS1, Recurring - Zone 4	100	NA NA	NA	NA NA	NA NA	NA NA	\$66.89	NA	NA NA	NA NA
NRC - Switch As Is - EEL- 1st	UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
NRC - Switch As Is - EEL - Add'I	UNCCC	\$13.33	\$15.48	\$11.27	\$15.48	\$11.10	\$13.33	\$15.48	\$28.35	\$15.48
NRC - Switch As Is - EEL - Disconnect - 1st	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Disconnect - Add'l	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
INC - Switch As is - EEL - Disconnect - Add i	UNCCC	\$15.21	\$13.92	\$12.01	\$13.92	\$12.00	φ15.21	\$13.92	IDA	\$13.92
NIDO Cuitab As la EEL Maguelui Elist Ast	0.01400	Ø50 40	ΦE4.04	¢45.40	# E4 04	£40.70	PEE 11	фг.4.04	PEC 54	фг.4.04
NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
NRC - Switch As Is - EEL- Manual vs. Elect - Add'l	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56
			Orlando,							
			Miami, Ft			New		Greensboro		
INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC		NashvilleTN
NRC - 4-wire 64 kbps Loop - 1st	SOMAC	NA	\$709.72	\$401.71	NA	\$483.59	NA	\$489.04	NA	\$698.42
NRC - 4-wire 64 kbps Loop - Add'l	SOMAC	NA	\$483.45	\$283.84	NA	\$315.57	NA	\$337.51	NA	NA
NRC - DS1- Interoffice Channel - Facility Termination - 1st	SOMAC	NA	\$45.91	\$166.01	NA	\$186.69	NA	\$217.17	NA	\$195.68
NRC - DS1- Interoffice Channel - Facility Termination - Add'I	SOMAC	NA	\$44.18	\$130.69	NA	\$149.23	NA	\$163.75	NA	\$156.47
NRC - DS1- Interoffice Channel - Facility Termination - Add'l NRC - DS1 Channelization System - 1st	SOMAC SOMAC	NA NA	\$44.18 \$238.43	\$130.69 \$331.77	NA NA	\$149.23 \$297.96	NA NA	\$163.75 \$338.55	NA NA	\$156.47 \$222.87
NRC - DS1 Channelization System - 1st	SOMAC	NA	\$238.43	\$331.77	NA	\$297.96	NA	\$338.55	NA	\$222.87
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l	SOMAC SOMAC SOMAC	NA NA	\$238.43 \$145.55	\$331.77 \$202.63	NA NA	\$297.96 \$181.39	NA NA	\$338.55 \$200.06	NA NA	\$222.87 \$135.80
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1	SOMAC SOMAC SOMAC	NA NA NA	\$238.43 \$145.55 \$13.39	\$331.77 \$202.63 \$13.45	NA NA NA	\$297.96 \$181.39 \$12.29	NA NA NA	\$338.55 \$200.06 \$15.76	NA NA NA	\$222.87 \$135.80 \$12.61
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A	SOMAC SOMAC SOMAC	NA NA NA	\$238.43 \$145.55 \$13.39 \$9.59	\$331.77 \$202.63 \$13.45 \$9.63	NA NA NA	\$297.96 \$181.39 \$12.29	NA NA NA	\$338.55 \$200.06 \$15.76 \$11.28	NA NA NA	\$222.87 \$135.80 \$12.61
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC	NA NA NA NA	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02	\$331.77 \$202.63 \$13.45	NA NA NA NA	\$297.96 \$181.39 \$12.29 \$8.80	NA NA NA NA	\$338.55 \$200.06 \$15.76 \$11.28	NA NA NA	\$222.87 \$135.80 \$12.61 \$9.03
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX	NA NA NA NA \$14.61 \$0.69	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31	NA NA NA NA \$22.26	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78	NA NA NA NA	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753	NA NA NA NA \$16.83	\$222.87 \$135.80 \$12.61 \$9.03
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1	NA NA NA NA \$14.61 \$0.69 \$79.69	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39	NA NA NA NA \$22.26 \$0.45 \$55.05	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40	NA NA NA NA \$17.83 \$0.66 \$74.40	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29	NA NA NA NA \$16.83 \$0.76 \$94.98	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX	NA NA NA NA \$14.61 \$0.69	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31	NA NA NA NA \$22.26	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78	NA NA NA NA \$17.83	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753	NA NA NA NA \$16.83	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG	NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20	NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC	NA NA NA NA S14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$11.78	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97	NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG	NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20	NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC	NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78 \$16.86 \$15.48	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97	NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.87	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'I NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A 2-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL 2-wire VG Local Channel per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month DS1 Channelization System per system per month DS1 Channelization Interface -VG per month NRC - Switch As Is - EEL - 1st NRC - Switch As Is - EEL - Add'I	SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC	NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78 \$16.86 \$15.48	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27	NA NA NA NA S22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC UNCCC	NA NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78 \$16.86 \$15.48	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27	NA NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$15.48 \$13.92 \$13.92	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$13.92	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX MQ1 1D1VG UNCCC UNCCC UNCCC UNCCC SOMAC	NA NA NA NA NA S14.61 \$0.69 \$79.69 \$139.58 \$11.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$16.86 \$15.48 \$11.78 \$16.86 \$15.48	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27	NA NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$42.70	NA NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$15.21 \$55.41	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$17.72 \$1.64 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31	NA NA NA NA S16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA TBA	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC UNCCC	NA NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27	NA NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$15.48 \$13.92 \$13.92	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$13.92	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48
NRC - DS1 Channelization System - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX MQ1 1D1VG UNCCC UNCCC UNCCC UNCCC SOMAC	NA NA NA NA NA S14.61 \$0.69 \$79.69 \$139.58 \$11.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Orlando ,	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27	NA NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$42.70 \$14.77	NA NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$15.21 \$55.41	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$13.92 \$177.56	NA NA NA NA S16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA TBA	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'I NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A 2-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL 2-wire VG Local Channel per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month DS1 Channelization System per system per month DS1 Channelization Interface -VG per month NRC - Switch As Is - EEL - 1st NRC - Switch As Is - EEL - Add'I NRC - Switch As Is - EEL - Disconnect - 1st NRC - Switch As Is - EEL - Disconnect - Add'I NRC - Switch As Is - EEL - Manual vs. Elect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - Add'I	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX MQ1 1D1VG UNCCC UNCCC UNCCC UNCCC SOMAC	NA NA NA NA NA S14.61 \$0.69 \$79.69 \$139.58 \$11.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$17.88 \$15.48 \$15.48 \$15.48 \$13.92 \$	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27	NA NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$12.66 \$42.70 \$14.77	NA NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$15.21 \$55.41	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Greensboro	NA NA NA NA S16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA TBA	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48 \$13.92 \$13.92 \$13.92 \$17.56
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A 2-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL 2-wire VG Local Channel per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month DS1 Channelization System per system per month NS1 Channelization Interface -VG per month NS2 - Switch As Is - EEL - 1st NS3 - Switch As Is - EEL - Add'l NS4 - Switch As Is - EEL - Disconnect - 1st NS5 - Switch As Is - EEL - Disconnect - Add'l NS6 - Switch As Is - EEL - Manual vs. Elect - 1st NS6 - Switch As Is - EEL - Manual vs. Elect - 1st NS6 - Switch As Is - EEL - Manual vs. Elect - 1st NS7 - Switch As Is - EEL - Manual vs. Elect - 1st NS6 - Switch As Is - EEL - Manual vs. Elect - 1st NS7 - Switch As Is - EEL - Manual vs. Elect - 1st NS7 - Switch As Is - EEL - Manual vs. Elect - 1st NS7 - Switch As Is - EEL - Manual vs. Elect - 1st NS7 - Switch As Is - EEL - Manual vs. Elect - 1st NS8 - Switch As Is - EEL - Manual vs. Elect - 1st NS7 - Switch As Is - EEL - Manual vs. Elect - 1st NS8 - Switch As Is - EEL - Manual vs. Elect - 1st NS8 - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC UNCCC UNCCC SOMAC SOMAC	NA NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43 \$19.15	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$163.88 \$1.78 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Orlando, Miami, Ft Laud FL	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27 \$12.61 \$12.61 \$45.46 \$15.72	NA NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$42.70 \$14.77 New Orleans LA	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$15.21 \$55.41 \$19.16	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Greensboro Charlotte NC	NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA TBA \$56.54 \$19.02	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48 \$13.92 \$13.92 \$13.92 \$17.56
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A 2-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL 2-wire VG Local Channel per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month DS1 Channelization System per system per month DS1 Channelization Interface - VG per month NRC - Switch As Is - EEL - 1st NRC - Switch As Is - EEL - Add'l NRC - Switch As Is - EEL - Disconnect - 1st NRC - Switch As Is - EEL - Disconnect - Add'l NRC - Switch As Is - EEL - Manual vs. Elect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - Add'l INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP: NRC - 2-wire VG - Local Channel - 1st	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX MQ1 1D1VG UNCCC UNCCC UNCCC UNCCC SOMAC SOMAC SOMAC	NA NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$11.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43 \$19.15	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$16.86 \$15.48 \$11.78 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Orlando, Miami, Ft Laud FL \$477.33	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27 \$12.61 \$45.46 \$15.72	NA NA NA NA NA \$22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$42.70 \$14.77 New Orleans LA	NA NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$15.21 \$55.41 \$19.16	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$17.72 \$1.64 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Greensboro Charlotte NC \$553.80	NA NA NA NA NA \$16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA TBA \$56.54 \$19.02	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A 2-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL 2-wire VG Local Channel per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month DS1 Channelization System per system per month DS1 Channelization Interface -VG per month NRC - Switch As Is - EEL - 1st NRC - Switch As Is - EEL - Disconnect - 1st NRC - Switch As Is - EEL - Disconnect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - Add'l INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP: NRC - 2-wire VG - Local Channel - 1st NRC - 2-wire VG - Local Channel - Add'l	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC UNCCC SOMAC SOMAC SOMAC	NA NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43 \$19.15	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Orlando, Miami, Ft Laud FL \$477.33 \$124.32	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27 \$12.61 \$45.46 \$15.72	NA NA NA NA NA NA S22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48 \$13.92 \$51.31 \$17.56	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$42.70 \$14.77 New Orleans LA \$430.71 \$74.41	NA NA NA NA S17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$55.41 \$19.16	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$51.31 \$17.56 Greensboro Charlotte NC \$553.80 \$86.69	NA NA NA NA \$16.83 \$0.76 \$94.98 \$17.93 \$28.87 \$28.35 TBA TBA \$56.54 \$19.02	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A 2-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL 2-wire VG Local Channel per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month DS1 Channelization System per system per month NS1 Channelization Interface - VG per month NRC - Switch As Is - EEL - 1st NRC - Switch As Is - EEL - Add'l NRC - Switch As Is - EEL - Disconnect - 1st NRC - Switch As Is - EEL - Disconnect - Add'l NRC - Switch As Is - EEL - Manual vs. Elect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - Add'l INC - Switch As Is - EEL - Manual vs. Elect - Add'l NRC - Switch As Is - EEL - Manual vs. Elect - Add'l NRC - Switch As Is - EEL - Manual vs. Elect - Add'l NRC - Switch As Is - EEL - Manual vs. Elect - Add'l NRC - Switch As Is - EEL - Manual vs. Elect - Add'l NRC - Switch As Is - EEL - Manual vs. Elect - Add'l	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC UNCCC SOMAC SOMAC SOMAC SOMAC	NA NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43 \$19.15	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$16.88 \$1.78 \$16.86 \$15.48 \$13.92 \$51.31 \$17.56 Orlando, Miami, Ft Laud FL \$477.33 \$124.32 \$45.91	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$12.97 \$11.27 \$12.61 \$45.46 \$15.72	NA NA NA NA NA NA NA NA NA NA NA NA NA N	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$12.66 \$42.70 \$14.77 New Orleans LA \$430.71 \$74.41 \$186.69	NA NA NA NA \$17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$15.21 \$15.21 \$15.21 \$15.21	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$51.31 \$17.56 Greensboro Charlotte NC \$553.80 \$86.69 \$217.17	NA NA NA NA NA S16.83 \$0.76 \$94.98 \$179.81 \$1.93 \$28.87 \$28.35 TBA TBA \$56.54 \$19.02	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48 \$13.92 \$13.92 \$13.92 \$17.56 NashvilleTN \$287.79 \$39.50 \$195.68
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - 1 NRC - DS1 Channelization Sys. Interface OCU-DP card per month(2.4-64kbps) - A 2-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL 2-wire VG Local Channel per month DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month DS1 Channelization System per system per month DS1 Channelization Interface -VG per month NRC - Switch As Is - EEL - 1st NRC - Switch As Is - EEL - Disconnect - 1st NRC - Switch As Is - EEL - Disconnect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - 1st NRC - Switch As Is - EEL - Manual vs. Elect - Add'l INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP: NRC - 2-wire VG - Local Channel - 1st NRC - 2-wire VG - Local Channel - Add'l	SOMAC SOMAC SOMAC SOMAC SOMAC ULDV2 1L5XX U1TF1 MQ1 1D1VG UNCCC UNCCC UNCCC SOMAC SOMAC SOMAC	NA NA NA NA NA \$14.61 \$0.69 \$79.69 \$139.58 \$1.26 \$14.37 \$13.33 \$15.21 \$15.21 \$56.43 \$19.15	\$238.43 \$145.55 \$13.39 \$9.59 \$18.02 \$0.60 \$99.79 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56 Orlando, Miami, Ft Laud FL \$477.33 \$124.32	\$331.77 \$202.63 \$13.45 \$9.63 \$16.28 \$0.31 \$63.39 \$137.97 \$2.20 \$12.97 \$11.27 \$12.61 \$45.46 \$15.72	NA NA NA NA NA NA S22.26 \$0.45 \$55.05 \$200.01 \$1.40 \$16.86 \$15.48 \$13.92 \$51.31 \$17.56	\$297.96 \$181.39 \$12.29 \$8.80 \$14.94 \$0.78 \$93.40 \$209.87 \$1.62 \$12.70 \$11.10 \$12.66 \$42.70 \$14.77 New Orleans LA \$430.71 \$74.41	NA NA NA NA S17.83 \$0.66 \$74.40 \$146.87 \$1.45 \$15.41 \$13.33 \$15.21 \$55.41 \$19.16	\$338.55 \$200.06 \$15.76 \$11.28 \$14.82 \$0.5753 \$71.29 \$177.72 \$1.64 \$16.86 \$15.48 \$13.92 \$51.31 \$17.56 Greensboro Charlotte NC \$553.80 \$86.69	NA NA NA NA \$16.83 \$0.76 \$94.98 \$17.93 \$28.87 \$28.35 TBA TBA \$56.54 \$19.02	\$222.87 \$135.80 \$12.61 \$9.03 \$19.02 \$0.35 \$75.83 \$165.21 \$1.25 \$16.86 \$15.48 \$13.92 \$13.92 \$51.31 \$17.56

7/7/00 Filefolder EELS

DESCRIPTION	USOC	D OTHER SERV	FL	GA	KY	LA	MS	NC	SC	TN
NRC - DS1 Channelization System - Add'l	SOMAC	NA NA	\$142.56	\$148.03	NA NA	\$135.20	NA NA	\$182.57	NA	\$135.80
NRC - DS1 Channelization VG Interface - 1st	SOMAC	NA NA	\$13.39	\$13.45	NA NA	\$12.29	NA NA	\$15.76	NA NA	\$12.61
NRC - DS1 Channelization VG Interface - Add'l	SOMAC	NA NA	\$9.59	\$9.63	NA NA	\$8.80	NA NA	\$11.28	NA NA	\$9.03
4-wire VG Local Channel/DS1 Interoffice Channel - Dedicated Transport EEL	OOWING	1973	ψ3.55	ψ5.00	11/3	ψ0.00	14/3	ψ11.20	1973	ψ5.05
4-wire VG Local Channel per month	ULDV4	\$15.77	\$19.01	\$17.18	\$23.38	\$16.21	\$19.03	\$15.87	\$18.05	\$20.14
DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$0.69	\$0.60	\$0.31	\$0.45	\$0.78	\$0.66	\$0.5753	\$0.76	\$0.35
DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month	1 U1TF1	\$79.69	\$99.79	\$63.39	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$75.83
DS1 Channelization System per system per month	MQ1	\$139.58	\$163.88	\$137.97	\$200.01	\$209.87	\$146.87	\$177.72	\$179.81	\$165.21
201 Chamicinzation Gystom per Gystom per month	WIQ I	Ψ100.00	ψ100.00	Ψ107.07	Ψ200.01	Ψ200.07	ψ110.07	Ψ177.72	ψ170.01	ψ100.21
DS1 Channelization Interface -VG per month	1D1VG	\$4.53	\$6.31	\$2.20	\$8.52	\$7.55	\$5.58	\$4.61	\$9.69	\$3.91
NRC - Switch As Is - EEL- 1st	UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
NRC - Switch As Is - EEL - Ist	UNCCC	\$13.33	\$15.48	\$12.97	\$15.48	\$11.10	\$13.33	\$15.48	\$28.35	\$15.48
NRC - Switch As Is - EEL - Add 1	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.46
NRC - Switch As Is - EEL - Disconnect - Add'l	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
NRC - Switch As Is - EEL- Manual vs. Elect - Ist	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56
INC - Switch As is - EEL- Walldal Vs. Elect - Add I	SOIVIAC	\$19.15	Orlando,	\$15.72	\$17.50	Φ14.77	\$19.10	\$17.50	\$19.02	\$17.50
			Miami, Ft			New		Greensboro		
INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC		NashvilleTN
NRC - 4-wire Local Channel - VG - 1st	SOMAC	NA	\$77.33	\$387.38	NA	\$433.31	NA	\$562.23	NA	\$287.94
NRC - 4-wire Local Channel - VG - 1st NRC - 4-wire Local Channel - VG - Add'I	SOMAC	NA NA	\$124.32	\$72.47	NA NA	\$88.07	NA NA	\$92.67	NA NA	\$54.18
NRC - DS1 - Facility Termination - 1st	SOMAC	NA NA	\$45.91	\$166.01	NA NA	\$186.69	NA NA	\$217.17	NA NA	\$195.68
NRC - DS1 - Facility Termination - 1st	SOMAC	NA NA	\$44.18	\$130.69	NA NA	\$149.23	NA NA	\$163.75	NA NA	\$195.66
NRC - DS1 - Facility Termination - Add1	SOMAC	NA NA	\$235.06	\$240.96	NA NA	\$220.07	NA NA	\$301.74	NA NA	\$222.87
NRC - DS1 Channelization System - 1st NRC - DS1 Channelization System - Add'l	SOMAC	NA NA	\$142.56	\$148.03	NA NA	\$135.20	NA NA	\$182.57	NA NA	\$135.80
NRC - DS1 Channelization System Interface VG - 1st	SOMAC	NA NA	\$142.56	\$146.03	NA NA	\$135.20	NA NA	\$152.57	NA NA	\$133.60
NRC - DS1 Channelization System Interface vG - 1st	SOMAC	NA NA	\$9.59	\$9.63	NA NA	\$8.80	NA NA	\$11.28	NA NA	\$9.03
DS1 Loop/DS1 Interoffice Channel - Dedicated Transport EEL	SOWAC	INA	φ9.59	φ9.03	INA	φο.ου	INA	\$11.20	INA	\$9.03
Zone 1	TBD	NA	NA	\$115.79	NA	\$149.72	\$125.39	NA	\$154.59	NA
Zone 2	TBD	NA	NA NA	\$123.90	NA NA	\$190.13	\$141.98	NA NA	\$184.88	NA NA
Zone 3	TBD	NA NA	NA NA	\$159.57	NA NA	\$290.97	\$170.98	NA NA	\$214.04	NA NA
Zone 4	TBD	NA NA	NA NA	NA	NA NA	φ290.97 NA	\$201.87	NA NA	η214.04 NA	NA NA
2016 4	100	INA	INA	INA	INA	INA	φ201.07	INA	INA	INA
DS1 Loop, per month, statewide	USLXX	\$64.65	\$80.00	NA	\$67.96	NA	NA	\$62.78	NA	TBD
DS1 Loop, per month, Statewide DS1 Loop, per month, Zone 1 (Note 1)	TBD	NA	NA	\$52.40	NA	\$56.32	\$50.99	NA	\$59.61	NA
DS1 Loop, per month, Zone 1 (Note 1)	TBD	NA NA	NA NA	\$60.51	NA NA	\$96.73	\$67.58	NA NA	\$89.90	NA NA
DS1 Loop, per month, Zone 3 (Note 1)	TBD	NA NA	NA NA	\$96.18	NA NA	\$197.57	\$96.58	NA NA	\$119.06	NA NA
Do i Loop, per monar, zone o (Note 1)	TDD	INA	INA	ψ90.10	INA	ψ197.37	ψ90.00	INA	ψ113.00	INA
DS1 Loop, per month, Zone 4 (Note 1)	TBD	NA	NA	NA	NA	NA	\$127.47	NA	NA	NA
DS1 Loop, per month, Zone 4 (Note 1) DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$0.69	\$0.60	\$0.31	\$0.45	\$0.78	\$0.66	\$0.5753	\$0.76	\$0.35
DS1 Interoffice Channel - Dedicated Transport EEL - Per Mile per month DS1 Interoffice Channel - Dedicated Transport EEL - Facility Termination per month	1L5XX 1 U1TF1	\$79.69	\$99.79	\$63.39	\$0.45 \$55.05	\$0.78	\$74.40	\$0.5753 \$71.29	\$94.98	\$75.83
Per additional circuit in same DS3 - Zone 1	TBD	\$79.69 NA	ъ99.79 NA	\$63.07	\$55.05 NA	φ93.40 NA	NA	\$71.29 NA	ъ94.96 NA	\$75.63 NA
Per additional circuit in same DS3 - Zone 2	TBD	NA NA	NA NA	\$61.18	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
Per additional circuit in same DS3 - Zone 3	TBD	NA NA	NA NA	\$96.85	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
Per additional circuit in same DS3 - Zone 4	TBD	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
1 ST Additional official in Same DOS - ZONE 4	יטטי	14/7	INC	147	1477	14/7	14/7	14/7	14/7	11/4
NRC - Switch As Is - EEL- 1st	UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
NRC - Switch As Is - EEL - Add'I	UNCCC	\$13.33	\$15.48	\$11.27	\$15.48	\$11.10	\$13.33	\$15.48	\$28.35	\$15.48
NRC - Switch As Is - EEL - Disconnect - 1st	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Disconnect - Add'l	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
NRC - Switch As Is - EEL- Manual vs. Elect - Add'l	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56
parts of the state	55.111.15	ψ.σ.ισ	Ψου	Ψ.J.12	Ψου	Ψ. 1.77	ψ.σ.ισ	ψου	Ų.J.∪L	Ψ.7.00

Version 1Q00:6/5/00 7/7/00 Filefolder

EELS

		ID OTHER SERV								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
			Orlando,						ł	
			Miami, Ft			New		Greensboro	ł	
INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC	ł	NashvilleTN
NRC - DS1 Loop - 1st	SOMAC	NA	NA	\$448.92	NA	NA	NA	\$714.84	NA	NA
NRC - DS1 Loop - Add'l	SOMAC	NA	NA	\$276.60	NA	NA	NA	\$421.47	NA	NA
NRC - DS1 Interoffice Channel - Facility Termination - 1st	SOMAC	NA	\$45.91	\$166.01	NA	\$186.69	NA	\$217.17	NA	\$195.68
NRC - DS1Interoffice Channel - Facility Termination - Add'l	SOMAC	NA	\$44.18	\$130.69	NA	\$149.23	NA	\$163.75	NA	\$156.47
DS1 Loop/DS3 Interoffice Channel - Dedicated Transport EEL										1
Zone 1	TBD	NA	NA	\$973.58	NA	NA	NA	NA	NA	NA
Zone 2	TBD	NA	NA	\$981.69	NA	NA	NA	NA	NA	NA
Zone 3	TBD	NA	NA	\$1.017.36	NA	NA	NA	NA	NA	NA
Zone 4	TBD	NA.	NA.	NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA
Zone 4	100	INA	INA	INA	INA	INA	INA	INA	INA	INA
DS1 Loop, per month, statewide	USLXX	\$64.65	\$80.00	NA	\$67.96	\$72.86	\$69.59	\$62.78	\$72.55	TBD
	TBD		δ60.00 NA	\$52.40	_					_
DS1 Loop, per month, Zone 1 (Note 1)	TBD	NA NA	NA NA	*	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
DS1 Loop, per month, Zone 2 (Note 1)		NA NA		\$60.51						
DS1 Loop, per month, Zone 3 (Note 1)	TBD TBD	NA NA	NA NA	\$96.18	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
DS1 Loop, per month, Zone 4 (Note 1)				NA 00.40						
DS3 Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$11.93	\$10.25	\$6.46	\$12.06	\$16.15	\$13.48	\$12.98	\$19.14	\$6.88
DS3 Interoffice Channel - Dedicated Transport EEL - Facility Termination per mont		736.6	994.83	\$717.60	\$1,112.02	\$1,131.09	\$686.84	\$720.38	\$904.49	\$840.61
DS3 Channelization System per system per month	MQ3	\$210.87	\$213.22	\$202.91	\$236.32	\$245.84	\$229.30	\$226.81	\$204.07	\$225.59
DS3 Channelization Interface -DS1 per month	1PQE1	\$4.53	\$6.31	\$0.67	\$8.52	\$7.55	\$5.58	\$4.61	\$9.69	\$3.91
NRC - Switch As Is - EEL- 1st	UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
NRC - Switch As Is - EEL - Add'l	UNCCC	\$13.33	\$15.48	\$11.27	\$15.48	\$11.10	\$13.33	\$15.48	\$28.35	\$15.48
NRC - Switch As Is - EEL - Disconnect - 1st	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Disconnect - Add'l	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
NRC - Switch As Is - EEL- Manual vs. Elect - Add'l	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56
			Orlando,						ł	
			Miami, Ft			New		Greensboro	ł	
INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC	ł	NashvilleTN
NRC - DS1 Loop - 1st	SOMAC	NA	NA	\$53.46	NA	NA	NA	\$714.84	NA	NA
NRC - DS1 Loop - Add'l	SOMAC	NA	NA	\$319.54	NA NA	NA	NA	\$421.47	NA	NA
NRC - DS3 - Interoffice Channel - Facility Termination - 1st	SOMAC	NA	\$879.42	\$959.44	NA NA	\$882.49	NA	\$794.94	NA	\$905.50
NRC - DS3 - Interoffice Channel - Facility Termination - Add'l	SOMAC	NA	\$542.41	\$623.26	NA NA	\$573.28	NA	\$579.55	NA NA	\$565.26
NRC - DS3 Channelization System - 1st	SOMAC	NA NA	\$408.24	\$453.17	NA NA	\$413.85	NA NA	\$428.07	NA NA	\$423.18
NRC - DS3 Channelization System - Add'l	SOMAC	NA NA	\$301.27	\$320.09	NA NA	\$292.33	NA NA	\$298.37	NA NA	\$298.48
NRC - DS3 Channelization System DS1 Interface - 1st	SOMAC	NA NA	\$13.39	\$13.45	NA NA	\$12.29	NA NA	\$15.76	NA NA	\$12.61
NRC - DS3 Channelization System DS1 Interface - Add'l	SOMAC	NA NA	\$9.59	\$9.63	NA NA	\$8.80	NA NA	\$11.28	NA NA	\$9.03
DS-1 Local Channel/ DS-3 Interoffice Channel - Dedicated Transport EEL	SOIVIAG	INA	φσ.υσ	φ3.03	INA	φο.ου	INA	φ11.20	INA	φ3.03
DS1 Local Channel per month	TMECS	\$35.52	\$44.35	\$38.57	\$43.80	\$43.80	\$38.91	\$35.68	\$37.20	\$40.27
									*	
DS3 Interoffice Channel - Dedicated Transport EEL - Per Mile per month	1L5XX	\$11.93	\$10.25	\$6.46	\$12.06	\$16.15	\$13.48	\$12.98	\$19.14	\$6.88
DS3 Interoffice Channel - Dedicated Transport EEL - Facility Termination per mont		\$736.60	\$994.83	\$717.60	\$1,112.02	\$1,131.09	\$686.84	\$720.38	\$904.49	\$840.61
DS3 Channelization System per system per month	MQ3	\$210.87	\$213.22	\$202.91	\$2.94	\$3.12	\$2.86	\$2.88	\$3.36	\$2.46
DS3 Channelization Interface -DS1 per month	1PQE1	\$4.53	\$6.31	\$0.67	\$1.40	\$1.62	\$1.45	\$1.64	\$1.93	\$1.25
NRC - Switch As Is - EEL- 1st	UNCCC	\$14.37	\$16.86	\$12.97	\$16.86	\$12.70	\$15.41	\$16.86	\$28.87	\$16.86
NRC - Switch As Is - EEL - Add'l	UNCCC	\$13.33	\$15.48	\$11.27	\$15.48	\$11.10	\$13.33	\$15.48	\$28.35	\$15.48
NRC - Switch As Is - EEL - Disconnect - 1st	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Disconnect - Add'I	UNCCC	\$15.21	\$13.92	\$12.61	\$13.92	\$12.66	\$15.21	\$13.92	TBA	\$13.92
NRC - Switch As Is - EEL - Manual vs. Elect - 1st	SOMAC	\$56.43	\$51.31	\$45.46	\$51.31	\$42.70	\$55.41	\$51.31	\$56.54	\$51.31
NRC - Switch As Is - EEL- Manual vs. Elect - Add'l	SOMAC	\$19.15	\$17.56	\$15.72	\$17.56	\$14.77	\$19.16	\$17.56	\$19.02	\$17.56

Version 1Q00:6/5/00 7/7/00 Filefolder

т.				JUINER SERV								
ш	DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
					Orlando,							
					Miami, Ft			New		Greensboro		
		INTERIM NRCs FOR NEW EEL SUBJECT TO TRUE-UP:			Laud FL			Orleans LA		Charlotte NC		NashvilleTN
		NRC -DS1 Local Channel - 1st	SOMAC	NA	\$246.50	\$400.37	NA	\$434.53	NA	\$534.48	NA	\$377.96
		NRC -DS1 Local Channel - Add'l	SOMAC	NA	\$230.49	\$312.89	NA	\$341.09	NA	\$462.69	NA	\$277.31
		NRC- DS3 Interoffice Channel - Facility Termination - 1st	SOMAC	NA	\$884.71	977.44	NA	982.64	NA	\$794.94	NA	980.45
		NRC- DS3 Interoffice Channel - Facility Termination - Add'l	SOMAC	NA	\$552.81	641.1	NA	644.52	NA	\$579.55	NA	643.07
		NRC - DS3 Channelization System - 1st	SOMAC	NA	\$344.18	\$386.41	NA	\$352.89	NA	\$476.24	NA	\$362.09
		NRC - DS3 Channelization System - Add'l	SOMAC	NA	\$248.67	\$264.84	NA	\$241.87	NA	\$321.89	NA	\$248.17
		NRC - DS3 Channelization System DS1 Interface - 1st	SOMAC	NA	\$13.39	\$13.45	NA	\$12.29	NA	\$15.76	NA	\$12.61
		NRC - DS3 Channelization System DS1 Interface - Add'l	SOMAC	NA	\$9.59	\$9.63	NA	\$8.80	NA	\$11.28	NA	\$9.03
П	N	otes:										
П		Effective May 1, 2000 statewide rates will be replaced by Deaveraged Loop Rates										
		by Zone where available. Until approximately December 31, 2000 or until such										
		time that BellSouth billing systems have been developed to handle the new zone										
		rate structure, BellSouth will bill at the Zone 1 Deaveraged Loop rate level only.										
		After December 31, 2000 or such time that the billing systems have been										
		developed to handle the new zone rate structure, BellSouth will begin billing										
		pursuant to CLEC-1's interconnection agreement. The status of the rates shown										
	1	by state is as follows:										
Ħ	T	Where the state Commission has adopted rates for the rate elements										
		containedherein, it is the intent of the Parties to reflect such rates in thisExhibit and										
		to apply the same consistent with applicable FCC and Commissionrules and			1							
		orders.										
ш		0.00.0.										

Version 1Q00:6/5/00 7/7/00 Filefolder

DESCRIPTION		Al	ND OTHER SERV	CES							
Recovery of incremental OSS costs, per CLP, per month TBD S0.002 S0.008 S0.008 S0.0008	DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
RC - OSS CLEC Daily Usage File: Recounting, Per Message TBD \$0,0002 \$0,0008 \$0,0001779 \$0,0000179 \$0,0000000179 \$0,0000000179 \$0,0000000179 \$0,0000000179 \$0,0000000179 \$0,00000000179 \$0,00000000179 \$0,0000000000000000000000000000000000	Operational Support Systems										
RC - CSS OLEC Cally Usage Price Message Processing, Per Message TSD \$50.0033 \$0.0004 \$0.0002395 \$50.00024 \$0.0003289 \$0.000329 \$0.000324 \$0.0003289 \$0.000324 \$0.0003289 \$0.0003289 \$0.000324 \$0.0003289 \$0.000	Recovery of incremental OSS costs, per CLP, per month	TBD	NA	NA	NA	NA	NA	NA	\$305.00	NA	NA
RC - OSS CRIE C Day Usage File Message Direction Per Algoretic Type RG - OSS CRIE C Day Usage File Message Direction (CONNECT DIRECT). Per TBD 30,0000 50,0001 50,0000365 50,00	RC - OSS OLEC Daily Usage File: Recording, Per Message	TBD	\$0.0002	\$0.008	\$0.0001275	\$0.0008611	\$0.00019	\$0.0001179	\$0.0003	\$0.0002862	\$0.008
RC - OSS CILE C Daily Usage File: Data Transmission (CONNECT DIRECT), Per BD \$0.00004 \$0.0000143 \$0.0000038 \$0.00000300 \$0.0000387 \$0.0001 RC - RC - RC - RC - RC - RC - RC - RC	RC- OSS OLEC Daily Usage File: Message Processing, Per Message		\$0.0033	\$0.004	\$0.0062548	\$0.0032357	\$0.0024	\$0.0032089	\$0.0032	\$0.0032344	\$0.004
Roceast Daily Usage File (ADUP)	RC - OSS OLEC Daily Usage File: Message Distribution, Per Magnetic Tape	TBD	\$55.19	\$54.95	\$28.25	\$55.68	\$47.3000	\$54.62	\$54.61	\$54.72	\$54.95
RC - ADUF, Message Processing, per message TBD \$0.004 \$0.0014 \$0.0	RC - OSS OLEC Daily Usage File: Data Transmission (CONNECT:DIRECT), Per	TBD	\$0.00004	\$0.001	\$0.0000434	\$0.0000365	\$0.0000300	\$0.0000354	\$0.00004	\$0.0000357	\$0.001
RC - ADUF, Message Destribution, per Magnetice Tape provisioned TBD \$54.95 \$											1
RC - ADUF, Data Transmison (CONNECT-CIRECT), per message TBD \$0.001	RC - ADUF, Message Processing, per message										
Enhanced Optional Daily Usage File (EDDUF)											
Enhanced Optional Daily Usage File: Message Processing, Per Message TBD \$0.004		TBD	\$0.001	\$0.001	\$0.0000434	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001
Enhanced Optional Daily Usage File: Message Processing, per magnetic lape TED \$47,30											
Enhanced Optional Daily Usage File: Data Transmision (CONNECT DIRECT), per TBD \$0,0000364											
NA NA NA NA NA NA NA NA										4	
BXX Access Fen Digit Screening Sev. WIXX No. Delivery N/A NA NA NA NA NA NA N	Enhanced Optional Daily Usage File: Data Transmision (CONNECT:DIRECT), per	TBD	\$0.0000364	\$0.0000364	NA	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364
BXX Access Fen Digit Screening Sev. WIXX No. Delivery N/A NA NA NA NA NA NA N											
BXX Access Fen Digit Screening Svc. WBXX No. Delivery											
Der query		N/A	\$0.0005	NA	\$0.0004868	NA	\$0.0005305	\$0.0005321	\$0.00050	\$0.0005227	NA
Tof 8XX Numbers, with Optional Complex Features, per query											
BAX Access Fen Digit Screening Svc. WiPOTS No. Delivery N/A NA NA NA NA NA NA NA NA NA NA NA NA NA											
per query		N/A	NA	NA	NA	\$0.0011	NA	NA	\$0.00431	NA	\$0.004
With Optional Complex Features, per query N/A		11/A				* • • • • • • • • • • • • • • • • • • •			# 0.0000		00.004
BXX Access Ten Digit Screening Svc. W/800 No. Delivery N/A NA NA NA NA NA NA NA NA NA NA NA NA NA											
Der message		N/A	NA	NA	NA	\$0.0011	NA	NA	\$0.00431	NA	\$0.004
Interest Numbers, wiOptional Complex Features, per message		NI/A	NIA	NIA	NIA	NIA	NIA.	NIA	NIA.	NIA	N10
BXX Access Ten Digit Screening Svc. W/POTS No. Delivery											
per message		N/A	NA	NA	NA	NA	NA	NA	NA	NA	NA
with Optional Complex Features, per message	<u> </u>	NI/A	NIA	NIA	NIA	NIA	NIA	NIA	NIA	NIA	NIA.
Reservation Charge per 8XX number reserved											
NRC - 1st		IN/A	INA	INA	INA	INA	INA	INA	INA	INA	INA
NRC - Add NRC - Add S0.76 \$1.19 \$0.73 \$0.96 \$0.9683 \$0.50		NIOD1V	¢7 12	NΑ	¢6 57	\$10.0E	¢6.20	¢0.46	\$7.05	¢6.20	\$20.00
NRC - Incremental Charge - Manual Service Order - 1st SOMAN \$27.37 NA \$18.94 NA \$18.14 \$25.52 \$26.94 \$27.84 NA NRC - Incremental Charge - Manual Service Order - Add'l SOMAN NA NA NA NA NA NA NA											
NRC - Incremental Charge - Manual Service Order - Add' SOMAN NA NA NA NA NA NA NA NA NA NA NA NA N											
NRC - 1st											
NRC - 1st		OOMAIN	14/3	14/4	INA	14/-1	14/4	14/3	14/3	14/4	14/4
NRC - Add NRC - Disconnect Charge - 1st	` '	NI/A	\$15.88	NΑ	\$12.81	\$30.50	¢12.27	\$17.04	\$23.82	\$22.63	\$67.50
NRC - Disconnect Charge - 1st											
NRC - Disconnect Charge - Add' N/A \$0.97 NA NA NA \$0.73 \$0.96 NA NA NA NA NA NA NA NA NA NA NA NA NA											
NRC - Incremental Charge - Manual Service Order - 1st SOMAN \$27.37 NA \$18.94 NA \$18.14 \$25.52 \$41.35 NA NA NA NRC - Incremental Charge - Manual Service Order - Add'l SOMAN NA NA NA NA NA NA NA											
NRC - Incremental Charge - Manual Service Order - Add'l SOMAN NA NA NA NA NA NA NA											
NRC - Incremental Charge - Manual Service Order - Disconnect SOMAN \$17.75 NA NA NA \$11.40 \$16.05 NA NA NA NA NA NA NA N											
NRC - 1st											
NRC - 1st		SOMAN	\$17.75	INA	INA	INA	\$11. 4 0	\$10.05	INA	INA	INA
NRC - Add NRC - Disconnect Charge - 1st		NOETV	¢15.00	NIA	¢10.01	\$20 E0	¢40.07	¢17.04	¢22.02	¢22.62	\$67.F0
NRC - Disconnect Charge - 1st											
NRC - Disconnect Charge - Add'l											
NRC - Incremental Charge - Manual Service Order - 1st SOMAN \$27.37 NA \$18.94 NA \$18.14 \$25.52 \$41.35 NA NA NRC - Incremental Charge - Manual Service Order - Add'l SOMAN NA	NDC Disconnect Charge Add!										
NRC - Incremental Charge - Manual Service Order - Add'l SOMAN NA NA NA NA NA NA NA NA NA NA NA NA N										1	
NRC - Incremental Charge - Manual Service Order - Disconnect SOMAN \$17.75 NA NA NA \$11.40 \$16.05 NA NA NA	INKC - Incremental Charge - Manual Service Order - 1St										
		SOMAN	\$17.75	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
Customized Area of Service per 8XX Number	•	Nocov	#5.00	NIA	0.4.40	\$0.07	# 4.07	#5.00	#5.00	\$5.04	#0.00
NRC - 1st											
NRC - Addi' N8FCX \$2.85 NA \$2.23 \$3.49 \$2.14 \$2.81 \$2.82 \$1.50 \$2.70	Version 1Q00:0/5/00	Norux	\$2.85	NA	\$2.23	\$3.49	\$2.14	\$2.81	\$2.82	\$2.82	\$1.50

DESCRIPTION		A	ND OTHER SERV	ICES	_						
MileC - Incremental Charge - Manual Service Order - Add	DESCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
Neith-cite Nei	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NPEC. 1et NPEC. Additionated Charger - Manual Service Order - 1st SOURCE - Additionated Charger - Manual Service Order - 1st SOURCE - Additionated Charger - Manual Service Order - Manual Service Order - Manual Service Ord		SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRF-AddII NRF-											
NRC-Incomental Charge - Minual Service Order - Set											
NRC - Incomendal Charge - Manual Service Order - AddT											
Charge Charge per request											
NRC - 1st	<u> </u>	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - AddII											
NRC - Incremental Charge - Manual Service Order - 1st SOMAN \$27.37 NA \$18.04 NA \$18.14 \$25.52 \$26.94 \$27.84 NA NRC - Incremental Charge - Manual Service Order - Add1 SOMAN NA NA NA NA NA NA NA											
RRC - Incremental Charge - Manual Service Order - AddT											
Call Handling and Destination Features											
NRC - 1st		SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Addf		NOEDV	ФE CO	NIA	¢4.70	ФС 07	£4.07	ድር ዕ ር	ሲር ር ጋ	ФГ C4	#2.00
LIDB Common Transport per query											
LIDB Common Transport per query	INIC - Add I	NOFDA	INA	INA	\$4.46	\$6.97	\$4.27	\$3.63	INA	\$5.64	\$3.00
LIDB Common Transport per query	LINE INFORMATION DATABASE ACCESS (LIDB)										
LIDB Validation per query		OQT	\$0.00004	\$0.0003	\$0.0000338	\$0.00006	\$0.0000418	\$0.0000446	\$0.0003	\$0.0000442	\$0.0003
NRC - Incremental Charge - Electronic Service Order TBD NA NA NA NA NA NA S62.26 NA NA NA NA NA NA NA S62.26 NA NA NA NA NA NA NA NA NA NA NA S62.26 NA NA NA NA NA NA NA NA NA NA NA NA NA		OQU	\$0.041003	\$0.041003	\$0.0105974	\$0.00938	\$0.0103774	\$0.0142132	\$0.013400	\$0.0141003	\$0.041003
NRC - Incremental Charge - Manual Service Order - 1st SOMAN SCB-93 NA S18.94 NA S18.14 \$25.52 \$26.94 \$27.84 \$91.00	LIDB Originating Point Code Establishment or Change - NRC	N/A	\$64.36	NA	\$50.30	\$107.60	\$48.17	\$63.63	\$91.00	\$61.62	NA
NRC - Incremental Charge - Manual Service Order - AddT SOMAN NA NA NA NA NA NA NA		TBD	NA	NA	NA	NA	NA	NA	\$62.26	NA	NA
CCS7 Signaling Connection, per link (A link) per month		SOMAN			\$18.94		\$18.14	\$25.52	\$26.94	\$27.84	\$91.00
CCS7 Signaling Connection, per link (A link) per month	NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	\$27.84	NA
CCS7 Signaling Connection, per link (A link) per month											
NRC - Disconnect											
NRC - Disconnect											
NRC - Incremental Charge - Manual Service Order - Disconnect SOMAN \$25,93 NA \$18,94 NA \$18,14 \$25,52 NA NA NA NA NA NA NA N											
NRC - Incremental Charge - Manual Service Order - Disconnect SOMAN \$16.31 NA NA \$11.40 \$16.05 NA NA NA \$10.05 NA NA NA \$10.05 NRC SOMAN \$16.31 NRC - Disconnect SOMAN \$16.31 NRC - Disconnect SOMAN \$13.96 \$354.95 \$126.34 \$169.72 \$510.00 \$277.07 \$510.00 NRC - Disconnect SOMAN \$1335.70 NA NA \$10.10 \$134.06 NA \$42.95 NA NA \$10.10 \$134.06 NA \$42.95 NA NA NA \$10.10 \$134.06 NA \$42.95 NA NA NA \$10.10 \$134.06 NA \$42.95 NA NA NA \$10.10 \$134.06 NA \$42.95 NA NA NA NA \$10.10 \$134.06 NA \$42.95 NA NA NA NA NA NA NA N											
CC37 Signaling Connection, per link (B link) (also known as D link) per month \$18.79 \$5.00 \$17.05 \$16.31 \$19.48 \$21.58 \$155.00 \$21.79 Not available NRC - Disconnect \$131.96 \$354.95 \$126.34 \$169.72 \$510.00 \$277.07 \$510.00 NRC - Disconnect \$133.70 NA											
NRC NRC - Disconnect \$1371.98 \$400.00 \$131.96 \$354.95 \$126.34 \$169.72 \$510.00 \$277.07 \$510.00 NRC - Disconnect \$135.70 NA NA NA NA \$101.10 \$134.08 NA \$42.95 NA NRC - Incremental Charge - Manual Service Order - Disconnect SOMAN \$25.93 NA \$18.94 NA \$118.14 \$25.52 NA NA NA NA NA NA NA N		SOMAN									
NRC - Disconnect											
NRC - Incremental Charge - Manual Service Order											
NRC - Incremental Charge - Manual Service Order - Disconnect SOMAN \$16.31 NA NA \$11.40 \$16.05 NA NA NA CCS7 Signaling Termination, per STP port per month \$148.72 \$113.00 \$133.99 \$174.08 \$161.12 \$132.88 \$156.33 \$355.00 \$162.00		COMAN									
CCS7 Signaling Termination, per STP port per month											
CCS7 Signaling Usage, per ISUP message \$0.00004 \$0.00001 \$0.0000354 \$0.0000430 \$0.0000456 \$0.00004 \$0.0000452 \$0.000023 \$ (applicable when measurement and billing capability exists.)		SOMAN									
(applicable when measurement and billing capability exists.)											
CCS7 Signaling Usage, per TCAP message \$0.0001 \$0.00004 \$0.0000870 \$0.000102042 \$0.0001052 \$0.0001115 \$0.00009 \$0.0001108 \$0.00005 \$0.000102042 \$0.0001052 \$0.0001115 \$0.00009 \$0.0001108 \$0.00005 \$0.000102042 \$0.0001052 \$0.0001105 \$0.00001052 \$0.0001105 \$0.00005 \$0.000102042 \$0.0001052 \$0.0001105 \$0.00005 \$0.0001052 \$0.0001105 \$0.00005 \$0.0001052 \$0.0001052 \$0.0001052 \$0.0001105 \$0.00005 \$0.0001052			ψ0.00004	ψ0.00001	\$0.0000334	ψ0.000037693	ψ0.0000430	ψ0.0000430	\$0.00004	ψ0.0000432	ψ0.000023
(applicable when measurement and billing capability exists.)			\$0,0001	\$0,00004	\$0,0000870	\$0.000102042	\$0,0001052	\$0,0001115	\$0,00009	\$0,0001108	\$0,00005
CCS7 Signaling Usage Surrogate, per link per LATA per mo (9) \$376.12 \$64.00 \$340.67 \$329.98 \$406.71 \$406.53 \$338.98 \$396.55 \$395.00			ψ0.0001	ψ0.00001	φο.σσσσστο	ψ0.000102012	ψ0.0001002	ψ0.0001110	ψ0.00000	ψο.σσστισσ	ψ0.00000
CCS7 Signaling Point Code, Establishment or Change, per STP affected NRC \$62.00 \$62			\$376.12	\$64.00	\$340.67	\$329.98	\$406.71	\$406.53	\$338.98	\$396.55	\$395.00
NRC \$62.00 \$62.			40.0		40.000	40=0.00	V 10011	*	4000.00	4000.00	700000
Operator Provided Call Handling per min - Using BST LIDB			\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00	\$62.00
Operator Provided Call Handling per min - Using BST LIDB											
Call Completion Access Termination Charge per call attempt											
Operator Provided Call Handling per min - Using Foreign LIDB N/A \$1.25 \$1.00 \$1.02 \$1.6249 \$0.96 \$1.24 \$1.24 \$1.25 NA Call Completion Access Termination Charge per call attempt N/A \$0.08 NA NA NA NA NA NA NA N											
Call Completion Access Termination Charge per call attempt											
Operator Provided Call Handling, per call N/A NA NA NA NA NA NA NA NA NA NA NA NA NA											
Fully Automated Call Handling per call - Using BST LIDB N/A \$0.11 \$0.10 \$0.0776409 \$0.0856 \$0.10 \$0.1072884 \$0.11 \$0.1115808 \$0.15 Fully Automated Call Handling per call - Using Foreign LIDB N/A \$0.13 \$0.10 \$0.0976984 \$0.1071 \$0.12 \$0.1253666 \$0.12 \$0.1293459 \$0.15 Professional recording of name (OCP alone) USOD1 \$7,000.00											
Fully Automated Call Handling per call - Using Foreign LIDB											
Professional recording of name (OCP alone)											
Professional recording of name (DA and OCP alone) USOD1 \$7,000.00											
7/7/00 5% - 6-14	Version 1Q00:6/5/00	USUDI	φτ,υυυ.υυ	φ,,υυυ.υυ	φ1,000.00	φ1,000.00	φ1,000.00	φ1,000.00	φ1,000.00	φ,,υυυ.υυ	φ1,000.00

	A	ND OTHER SERV	ICES	1 1		1				
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
DRAM or front-end loading, per TOPS switch	USOD2	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00
AABS or back-end loading, per IVS	USOD2	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00
EBAS or 0- automation loading, per NAV shelf	USOD2	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00
Recording Charge per Branded Announcement – Disconnect – Initial	N/A	\$9.61	NA	NA	NA	NA	NA	NA	NA	NA
Recording Charge per Branded Announcement – Disconnect – Subsequent	N/A	\$9.61	NA	NA	NA	NA	NA	NA	NA	NA
									1	
INWARD OPERATOR SERVICES									1	
Verification, per minute	N/A	\$1.16	NA	\$0.921083	NA	\$0.86	\$1.14	\$1.15	\$1.15	NA
Verification and Emergency Interrupt, per minute	N/A	\$1.16	NA	\$0.921083	NA	\$0.86	\$1.14	\$1.15	\$1.15	NA
Verification, per call	VIL	NA	\$0.80	NA	\$1.00	NA	NA	\$0.54	NA	\$0.90
Verification and Emergency Interrupt, per call	N/A	NA	\$1.00	NA	\$1.111	NA	NA	\$0.65	NA	\$1.95
DIRECTORY ASSISTANCE SERVICES										
Directory Assist Call Completion Access Svc (DACC), per call attempt	N/A	\$0.10	\$0.10	\$0.10	\$0.10	\$0.10	\$0.10	\$0.062	\$0.10	\$0.10
Call Completion Access Term charge per completed call	N/A	NA	NA	NA	NA	NA	NA	NA	\$0.08	NA
Number Services Intercept per query	N/A	\$0.0235	\$0.01	\$0.0097497	\$0.0086	\$0.02	\$0.0188268	\$0.0110	\$0.0124036	\$0.15
Number Services Intercept per Intercept Query Update	N/A	NA	NA	NA	\$0.0055	NA	NA	NA	NA	NA
Directory Assistance Access Service Calls, per call		\$0.275	\$0.275	\$0.275	\$0.275	\$0.275	\$0.275	\$0.260000	\$0.275	\$0.275
Professional recording of name (DA alone)		\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00	\$3,000.00
Professional recording of name (DA and OCP alone)		\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00	\$7,000.00
DRAM or front-end loading, per TOPS switch		\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00	\$250.00
AABS or back-end loading, per IVS		\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00	\$225.00
EBAS or 0- automation loading, per NAV shelf		\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00	\$270.00
Recording Charge per Branded Announcement – Disconnect – Initial	N/A	\$9.61	NA	NA	NA	NA	NA	NA	NA	NA
Recording Charge per Branded Announcement – Disconnect – Subsequent	N/A	\$9.61	NA	NA	NA	NA	NA	NA	NA	NA
Directory Transport										
Directory Transport - Local Channel DS1, per month	N/A	\$35.52	\$43.64	\$38.36	\$36.32	\$43.83	\$38.91	\$35.68	\$37.20	\$133.81
NRC - 1st	N/A	\$503.57	\$242.45	\$356.15	\$637.46	\$339.69	\$494.83	\$534.48	\$534.81	\$868.97
NRC - Add'l	N/A	\$442.84	\$226.44	\$312.89	\$546.94	\$298.29	\$435.28	\$462.69	\$462.81	\$486.83
NRC - Disconnect Charge - 1st	N/A	\$46.28	NA	NA	NA	\$33.02	\$46.85	NA	NA	NA
NRC - Disconnect Charge - Add'l	N/A	\$32.18	NA	NA	NA	\$23.32	\$33.02	NA	NA	NA
NRC - Incremental Charge-Manual Svc Order - NRC - 1st	SOMAN	\$61.99	NA	\$44.22	NA	\$42.34	\$59.58	\$86.15	\$87.99	NA
NRC - Incremental Charge-Manual Svc Order - NRC -addl	TBD	NA	NA	NA	NA	NA	NA	\$1.77	NA	NA
NRC - Incremental Charge-Manual Svc Order - NRC-Disconnect	SOMAN	\$29.27	NA	NA	NA	\$19.48	\$27.41	NA	\$3.11	NA
Directory Transport - Dedicated DS1 Level Interoffice per mile per mo	N/A	\$0.6923	\$0.6013	\$0.4523	\$0.45	\$0.78	\$0.6598	\$0.5753	\$0.7598	\$23.00
Directory Transport - Dedicated DS1 Level Interoffice per facility termination per mo		\$79.69	\$99.79	\$78.47	\$55.05	\$93.40	\$74.40	\$71.29	\$94.98	\$90.00
NRC - 1st	N/A	\$198.15	\$45.91	\$147.07	\$298.18	\$140.49	\$196.28	\$217.17	\$216.27	\$100.49
NRC - Add'l	N/A	\$148.18	\$44.18	\$111.75	\$231.18	\$106.69	\$147.31	\$163.75	\$162.70	\$100.49
NRC - Disconnect Charge - 1st	N/A	\$25.44	NA	NA	NA	\$20.00	\$26.56	NA	NA	NA
NRC - Disconnect Charge - Add'l	N/A	\$20.42	NA	NA	NA	\$16.34	\$21.61	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$38.07	\$39.63	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$27.37	NA	NA	NA	\$18.14	\$25.52	\$38.07	\$39.63	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA
Switched Common Transport per DA Access Service per call	N/A	\$0.0003	\$0.0003	\$0.0002906	\$0.000175	\$0.0003274	\$0.0002997	\$0.00020	\$0.000327	NA
Switched Common Transport per DA Access Service per call per mile	N/A	\$0.00003	\$0.00001	\$0.0000186	\$0.000004	\$0.0000175		\$0.00003	\$0.0000303	NA
Access Tandem Switching per DA Access Service per call	N/A	\$0.0023	\$0.00055	\$0.0019152	\$0.000783	\$0.0025257	\$0.0023713	\$0.0021	\$0.0024809	NA
DA Interconnection, per DA Access Service Call	N/A	\$0.00269	NA	\$0.00269	NA	NA	NA	\$0.00	\$0.000269	NA
Directory Transport-Installation NRC, per trunk or signaling connection	N/A									
NRC - 1st	N/A	\$260.69	\$206.06	\$204.23	\$501.98	\$195.54	\$257.73	NA	\$407.81	NA
NRC - Add'l	N/A	\$5.95	\$4.71	\$4.42	\$13.32	\$4.23	\$5.85	NA	\$11.00	NA
NRC - Disconnect Charge - 1st	N/A	\$173.46	NA	NA	NA	NA	NA	NA	NA	NA

	<i>F</i>	ND OTHER SERV	ICES	1		1	1			
DESCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	NA	NA	\$44.22	NA	\$130.05	\$171.49	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	NA	NA	NA	NA	\$4.23	\$5.85	NA	NA	NA
NRC - Manual Service Order - 1st	TBD	NA	NA	NA	NA	NA	NA	\$407.53	NA	NA
NRC - Manual Service Order - Add'l	TBD	NA	NA	NA	NA	NA	NA	\$10.98	NA	NA
			1					,		
Directory Assistance Database Service (DADS)										
Directory Assistance Database Service charge per listing	N/A	\$0.0446	\$0.001	\$0.0445	\$0.0193	\$0.0443	\$0.0447	\$0.04460	\$0.0444	NA
Directory Assistance Database Service, per month	DBSOF	\$128.55	\$100.00	\$95.50	\$120.76	\$90.54	\$126.17	\$126.26	\$127.23	NA
			,	*						
Direct Access to Directory Assistance Service (DADAS)										
Direct Access to Directory Assistance Service, per month	DBSDS	\$7,055.00	\$5,000.00	\$5,254.00	\$7,235.01	\$4,982.00	\$6,926.00	\$6,930.00	\$6,983.00	NA
Direct Access to Directory Assistance Service, per query	DBSDA	\$0.0472685	\$0.01	\$0.0469016	\$0.0052	\$0.0460	\$0.0461336	\$0.0456	\$0.0468212	NA
Direct Access to Directory Assistance Service, svc estab charge	DBSDE									
NRC	DBSDE	\$1,118.00	\$820.00	\$788.24	\$1,186.94	\$786.82	\$1,097.00	\$1,164.00	\$1,173.00	NA
NRC - Disconnect	DBSDE	\$81.83	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Charge Manual Service Order - 1st	SOMAN	NA	NA	NA	NA	\$57.23	\$80.52	NA	NA	NA
AIN (Note 4)										TBD
AIN, per message	CAM	NA	\$0.00004	NA	NA	NA	NA	NA	NA	NA
AIN - BellSouth AIN SMS Access Service	CAM								NA	NA
Service Establishment Charge, per state, initial set-up										
NRC	CAMSE	\$197.49	NA	\$90.25	NA	\$153.31	\$174.03	\$294.77	\$296.16	NA
NRC - Disconnect	CAMSE	\$114.22	NA	NA	NA	\$78.06	\$135.96	NA	NA	NA
Port Connection - Dial/Shared Access										
NRC	CAMDP	\$64.05	NA	\$29.66	NA	\$50.07	\$53.47	\$86.94	\$87.29	NA
NRC - Disconnect	CAMDP	\$27.04	NA	NA	NA	\$18.61	\$37.70	NA	NA	NA
Port Connection - ISDN Access										
NRC	CAM1P	\$64.05	NA	\$29.66	NA	\$50.07	\$53.47	\$86.94	\$87.29	NA
NRC - Disconnect	CAM1P	\$27.04	NA	NA	NA	\$18.61	\$37.70	NA	NA	NA
User ID Codes - per User ID Code										
NRC	CAMAU	\$141.84	NA	\$84.43	NA	\$104.95	\$129.83	\$200.83	\$202.08	NA
NRC - Disconnect	CAMAU	\$70.05	NA	NA	NA	\$48.95	\$79.91	NA	NA	NA
Security Card per User ID Code, initial or replacement										
NRC	CAMRC	\$142.13	NA	\$35.44	NA	\$125.33	\$131.54	\$172.05	\$172.26	NA
NRC - Disconnect	CAMRC	\$35.26	NA	NA	NA	\$24.40	\$45.77	NA	NA	NA
Storage, per unit (100Kb)	N/A	\$0.0026	NA	\$0.0023	NA	\$0.0029	\$0.0029	\$0.0023	\$0.0028	NA
Session per minute	N/A	\$0.0892	NA	\$0.0795604	NA	\$0.10	\$0.0975650	\$0.0791	\$0.0942966	NA
C0. Performed Session, per minute					NA	\$1.97	\$2.09	\$2.08	\$2.07	NA
AIN - BellSouth AIN Toolkit Service										
AIN, Service Creation Tools	CAMBP	NA	TBD	NA	NA	NA	NA	NA	NA	NA
Service Establishment Charge, per state, initial set-up										
NRC	BAPSC	\$192.69	NA	\$86.74	NA	\$153.25	\$169.31	\$290.05	\$291.41	NA
NRC - Disconnect	BAPSC	\$114.22	NA	NA	NA	\$78.05	\$135.96	NA	NA	NA
Training Session, per customer										
NRC	BAPVX	\$8,363.00	NA	\$8,348.00	NA	\$8,315.00	\$8,379.00	\$8,363.00	\$8,333.00	NA
NRC - Disconnect	BAPVX	NA	NA	NA	NA	NA	NA	NA	NA	NA
Trigger Access Charge, per trigger, per DN, Term. Attempt						1	1			
NRC	BAPTT	\$49.64	NA	\$19.13	NA	\$41.08	\$39.30	\$72.76	\$73.02	NA
NRC - Disconnect	BAPTT	\$27.04	NA	NA	NA	\$18.60	\$37.70	NA	NA	NA
Trigger Access Charge, per trigger per DN, Off-Hook Delay					<u> </u>					
NRC	BAPTD	\$49.64	NA	\$114.80	NA	\$41.08	\$39.30	\$72.76	\$73.02	NA
NRC - Disconnect	BAPTD	\$27.04	NA	NA	NA	\$18.60	\$37.70	NA	NA	NA
Trigger Access Charge, per trigger, per DN, Off-Hook Immediate										
**************************************					-			•		

	A	ND OTHER SERV	CES		•	1				
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC	BAPTM	\$49.64	NA	\$19.13	NA	\$41.08	\$39.30	\$72.76	\$73.02	NA
NRC - Disconnect	BAPTM	\$27.04	NA	NA	NA	\$18.60	\$37.70	NA	NA	NA
Trigger Access Charge, per trigger, per DN, 10-Digit PODP										
NRC	BAPTO	\$117.98	NA	\$70.06	NA	\$92.99	\$106.90	\$149.95	\$150.25	NA
NRC - Disconnect	BAPTO	\$37.90	NA	NA	NA	\$26.73	\$48.44	NA	NA	NA
Trigger Access Charge, per trigger, per DN, CDP										
NRC	BAPTC	\$117.98	NA	\$70.06	NA	\$92.99	\$106.90	\$149.95	\$150.25	NA
NRC - Disconnect	BAPTC	\$37.90	NA	NA	NA	\$26.73	\$48.44	NA	NA	NA
Trigger Access Charge, per trigger, per DN, Feature Code										
NRC	BAPTF	\$117.98	NA	\$70.06	NA	\$92.99	\$106.90	\$149.95	\$150.25	NA
NRC - Disconnect	BAPTF	\$37.90	NA	NA	NA	\$26.73	\$48.44	NA	NA	NA
Query Charge, per query		\$0.024	NA	\$0.0209223	NA NA	\$0.03	\$0.0256138	\$0.02	\$0.0250662	NA
Type 1 Node Charge, per AIN Toolkit Subscription, per node, per query		\$0.006	NA	\$0.0053137	NA	\$0.0065	\$0.0065161	\$0.005	\$0.0062979	NA
SCP Storage Charge, per SMS Access Acct, per 100 Kb	N/A	\$1.63	NA	\$1.46	NA	\$1.79	\$1.79	\$1.45	\$1.73	NA
Monthly Report - per AIN Toolkit Service Subscription	BAPMS	\$16.00	NA	\$15.96	NA NA	\$15.89	\$16.01	\$15.98	\$15.93	NA NA
I INRC	BAPMS	\$44.56	NA	\$22.64	NA	\$34.61	\$44.02	\$71.80	\$72.15	NA
NRC - Disconnect	BAPMS	\$31.84	NA	NA NA	NA	\$21.97	\$31.28	NA	NA NA	NA
Special Study - per AIN Toolkit Service Subscription	BAPLS	\$0.10	NA	\$0.0861109	NA	\$0.08	\$0.0810536	\$0.08	\$0.0872769	NA
NRC	BAPLS	\$47.74	NA	\$22.64	NA	\$37.77	\$47.21	\$47.20	\$47.35	NA
NRC - Disconnect	BAPLS	\$15.90	NA	NA	NA	NA	NA	NA	NA	NA
Call Event Report - per AIN Toolkit Service Subscription	BAPDS	\$15.90	NA	\$15.87	NA	\$15.81	\$15.93	\$15.90	\$15.84	NA
NRC	BAPDS	\$44.56	NA	\$22.64	NA	\$34.61	\$44.02	\$71.80	\$72.15	NA
NRC - Disconnect	BAPDS	\$31.84	NA	NA	NA	\$21.97	\$31.28	NA	NA	NA
Call Event special Study - per AIN Toolkit Service Subscription	BAPES	\$0.003	NA	\$0.0028704	NA	\$0.0026	\$0.0027018	\$0.003	\$0.0029092	NA
NRC	BAPES	\$47.74	NA	\$22.64	NA	\$37.77	\$47.21	\$47.20	\$47.35	NA
NRC - Disconnect	BAPES	\$15.90	NA	NA	NA	\$37.77	NA	NA	NA	NA
CALLING NAME (CNAM) QUERY SERVICE										
CNAM (Database Owner), Per Query	N/A	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016	\$0.016
CNAM (Non-Database Owner), Per Query *	N/A	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01	\$0.01
NRC, applicable when CLEC-1 uses the Character Based User Interface (CHUI)	N/A	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00	\$595.00
* Volume and term arrangements are also available.										
SELECTIVE ROUTING (Note 5)										
Per Line or PBX Trunk, each		NA	NA	NA	\$10.00 (Interim	NA	NA	NA	NA	TBD
INRC		NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	TBD
Customized routing per unique line class code, per request, per switch		INA	INA	INA	INA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC	USRCR	\$230.60	\$229.65	\$180.62	\$229.65	\$229.65	\$227.99	\$229.65	\$226.22	\$229.65
NRC - Incremental Charge - Manual Service Order	00.10.1	\$25.93	NA NA	\$18.94	NA NA	NA NA	\$253.51	NA NA	\$27.84	NA NA
		V		Ţ			4=====		4 =	
VIRTUAL COLLOCATION										
NRC - Virtual Collocation - Application Cost - Manual	TBD	NA	NA	NA	NA	NA	NA	\$3,622.00	NA	NA
NRC - Virtual Collocation - Cable Installation Cost per Cable - Manual	TBD	NA	NA	NA	NA	NA	NA	\$2,305.00	NA	NA
RC - Virtual Collocation - Floor space per square feet	TBD	NA	NA	NA	NA	NA	NA	\$3.45	NA	NA
RC - Virtual Collocation - Floor space power, per ampere	TBD	NA	NA	NA	NA	NA	NA	\$6.65	NA	NA
RC - Virtual Collocation - Cable support structure, per entrance cable	TBD	NA	NA	NA	NA	NA	NA	\$18.66	NA	NA
2-wire Cross-Connect										
RC	UEAC2	\$0.28	\$0.524	\$0.30	\$0.31	\$0.26	\$0.3996	\$0.09	\$0.3648	\$0.30
NRC - 1st	UEAC2	\$30.76	\$11.57	\$12.60	\$54.21	\$23.04	\$30.93	\$41.78	\$41.50	\$19.20
NRC - Add'l	UEAC2	\$29.40	\$11.57	\$12.60	\$51.07	\$22.11	\$29.59	\$39.23	\$38.94	\$19.20
NRC - 1st - Manual Service Order	TBD	NA	NA	NA	NA	NA	NA	\$4.75	NA	NA
NRC - Add'l - Manual Service Order	TBD	NA 010.75	NA	NA	NA	NA ©0.40	NA *10.70	\$4.75	NA	NA
Version 1000:6:5/00	UEAC2	\$12.75	NA	NA	NA	\$9.48	\$12.76	NA	NA	NA

		Ar	ID OTHER SERV	ICES				1			
DE	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
	NRC - Disconnect - Add'l	UEAC2	\$11.38	NA	NA	NA	\$8.54	\$11.43	NA	NA	NA
4-v	vire Cross-Connect										
	RC	UEAC4	\$0.56	\$0.524	\$0.50	\$0.62	\$0.52	\$0.7992	\$0.18	\$0.7297	\$0.50
	NRC - 1st	UEAC4	\$66.71	\$11.57	\$12.60	\$54.23	\$23.23	\$31.17	\$41.91	\$41.56	\$19.20
	NRC - Add'l	UEAC4	\$50.43	\$11.57	\$12.60	\$50.96	\$22.24	\$29.77	\$39.25	\$38.90	\$19.20
	NRC - 1st - Manual Service Order	TBD	NA	NA	NA	NA	NA	NA	\$4.73	NA	NA
	NRC - Add'l - Manual Service Order	TBD	NA	NA	NA	NA	NA	NA	\$4.73	NA	NA
	NRC - Disconnect - 1st	UEAC4	\$12.82	NA	NA	NA	\$9.53	\$12.83	NA	NA	NA
	NRC - Disconnect - Add'l	UEAC4	\$11.39	NA	NA	NA	\$8.55	\$11.43	NA	NA	NA
2-f	iber Cross-Connect										
	RC	CNC2F	\$12.10	NA	\$15.64	\$15.64	\$19.13	\$15.64	\$15.99	\$15.06	\$15.64
	NRC - 1st	CNC2F	\$55.46	NA	\$41.56	\$41.56	\$41.07	\$41.56	\$67.34	\$69.28	\$41.56
	NRC - Add'l	CNC2F	\$39.18	NA	\$29.82	\$29.82	\$29.63	\$29.82	\$48.55	\$48.89	\$29.82
	NRC - Disconnect - 1st	CNC2F	\$16.83	NA	NA	NA	\$12.84	\$12.96	NA	NA	NA
	NRC - Disconnect - Add'l	CNC2F	\$13.27	NA	NA	NA	\$10.29	\$10.34	NA	NA	NA
4-f	iber Cross-Connect		, ,								
ΠŤ	RC	CNC4F	\$21.75	NA	\$28.11	\$28.11	\$34.38	\$28.11	\$28.74	\$27.08	\$28.11
	NRC - 1st	CNC4F	\$66.71	NA	\$50.53	\$50.53	\$49.81	\$50.53	\$82.35	\$84.07	\$50.53
	NRC - Add'l	CNC4F	\$50.43	NA	\$38.78	\$38.78	\$38.37	\$38.78	\$63.56	\$63.68	\$38.78
	NRC - Disconnect - 1st	CNC4F	\$21.86	NA	NA	NA	\$16.75	\$16.97	NA	NA	NA
	NRC - Disconnect - Add'l	CNC4F	\$18.31	NA	NA	NA	\$14.20	\$14.35	NA	NA	NA
DS	1 Cross-Connects		7.0.0				*****	***************************************			
	RC	TBD	NA	NA	NA	NA	NA	NA	\$0.97	NA	NA
	NRC - 1st	TBD	NA.	NA	NA NA	NA	NA	NA	\$71.02	NA	NA.
	NRC - Add'l	TBD	NA	NA	NA NA	NA	NA	NA	\$51.08	NA	NA
H	NRC - Manual Service Order - 1st	TBD	NA	NA	NA	NA	NA	NA	\$4.70	NA	NA
	NRC - Manual Service Order - Add'l	TBD	NA.	NA	NA NA	NA	NA	NA	\$4.70	NA	NA
DS	33 Cross-Connects	100	147	10/1	147.	147	1471	1471	ψ1.70	10/	100
1	RC	TBD	NA	NA	NA	NA	NA	NA	\$12.33	NA	NA
H	NRC - 1st	TBD	NA	NA	NA	NA	NA	NA	\$69.84	NA	NA
	NRC - Add'l	TBD	NA	NA	NA	NA	NA	NA	\$49.43	NA	NA
H	NRC - Manual Service Order - 1st	TBD	NA	NA NA	NA NA	NA	NA	NA	\$4.70	NA NA	NA.
H	NRC - Manual Service Order - Add'l	TBD	NA	NA NA	NA NA	NA	NA	NA	\$4.70	NA NA	NA.
	Title Manual Colvido Clasi Addi	100	1471	1471	1471	1471	1471	1471	ψ1.70	10/	147
lf r	no rate is identified in the contract, the rate for the specific service or function will be as s	et forth in applicab	le BellSouth ta	riff or as negot	iated by the na	rties unon reque	st hy either na	irtv			
Η".		ot fortif in applicat	T DOILEGGGGT TO	I as neger	lated by the pa	nico apon roque	I by cirrior pa	 			
	1 BellSouth and CLEC shall negotiate rates for this offering. If agreement is not										
	reached within sixty (60) days of the Effective Date, either party may petition the										
	Florida PSC to settle the disputed charge or charges. (FL)										
	2 This rate element is for those states w/o separate rates for 800 calls with 800 No.										
	Delivery vs. POTS No. Delivery and calls with Optional Complex Features vs. w/o										
	Optional Complex Features.										
	3 This charge is only applicable where signaling usage measurement or billing										
	capability does not exist.		1							1	1
	4 Prices for AIN to be determined upon development of mediation device. (TN)										
	5 Price for Line Class Codes for Selective Routing shall be determined by the TRA.										
	(TN)		1	ĺ			1		ĺ	ĺ	I
+	Where the state Commission has adopted rates for the rate elements		1	1			1	Ì	1	1	
	containedherein, it is the intent of the Parties to reflect such rates in this Exhibit and		1							1	1
	to apply the same consistent with applicable FCC and Commissionrules and		1							1	1
	orders.		1				1				1
	uluels.		1	1	l		1	l	1	1	L

Version 1Q00:6/5/00 7/7/00 Filefolder

OSS-8XX-DATABASES

Attachment 3

Local Interconnection

TABLE OF CONTENTS

1.	NETWORK INTERCONNECTION	3
2.	INTERCONNECTION TRUNKING AND ROUTING	8
3.	NETWORK DESIGN AND MANAGEMENT FOR INTERCONNECTION	9
4.	PARITY IN ORDERING AND PROVISIONING	11
5.	LOCAL DIALING PARITY	12
6.	INTERCONNECTION COMPENSATION	12
7.	FRAME RELAY SERVICE	19
8	OPERATIONAL SUPPORT SYSTEMS (OSS) RATES	22.

Network Interconnection

The Parties shall provide interconnection with each other's networks for the transmission and routing of telephone exchange service (local) and exchange access (intraLATA toll and switched access) on the following terms:

1. NETWORK INTERCONNECTION

- 1.1 Interconnection is available to both Parties through: (1) delivery of a Party's facilities to a collocation arrangement or Fiber Meet arrangement as defined in this Agreement; or (2) interconnection via purchase of facilities from the other Party. Interconnection may be provided by the Parties at any other technically feasible point. Requests to BellSouth for interconnection at other points may be made through the Bona Fide Request/New Business Request process set out in Attachment 12.
- 1.2 e.spire must establish, at a minimum, a single Point of Presence, Interface, and Interconnection with BellSouth within the LATA for the delivery of traffic originated by e.spire. Each party hereto is free to define its own local calling area, subject to state commission approval where required. If e.spire chooses to interconnect at a single Point of Interconnection within a LATA, the interconnection must be at a BellSouth Access Tandem. Furthermore, for LATAs served by multiple Access Tandems, e.spire must establish trunks from the Point of Interconnection to the remaining Access Tandems where e.spire's NXXs are homed. It is e.spire's responsibility to enter its own NPA/NXX access homing arrangements into the LERG. In order for e.spire to home its NPA/NXXs on a BellSouth tandem, e.spire's NPA/NXX must be assigned within the Exchange Rate Center areas served by that BellSouth Tandem as specified by BellSouth. Any new Rate Centers established by either Party within a BellSouth tandem serving area must be approved by the Commission and defined in the Local Exchange Routing Guide ("LERG"). The specified association between BellSouth tandems and Exchange Rate Center areas will be defined in the LERG. A "Homing" arrangement is defined by a "Final" Trunk Group between the BellSouth Tandem and e.spire 's End Office switch. A "Final" Trunk Group is the last choice telecommunications path between the Tandem and End Office switch. It is e.spire's responsibility to enter its own NPA/NXX access and/or local tandem "homing" arrangements into the national Local Exchange Routing Guide (LERG).

- 1.3 A <u>Point of Presence (POP)</u> is the physical location (a structure where the environmental, power, air conditioning, etc. specifications for a Party's terminating equipment can be met) at which a Party establishes itself for obtaining access to the other Party's network. The POP is the physical location within which the Point of Interfaces occur.
- 1.4 A <u>Point of Interface</u> is the physical telecommunications interface between BellSouth and e.spire's interconnection functions. It establishes the technical interface and point of operational responsibility. The primary function of the Point of Interface is to serve as the terminus for the interconnection service. The Point of Interface has the following main characteristics:
 - 1. It is a cross-connect point to allow connection, disconnection, transfer or restoration of service.
 - 2. It is a point where BellSouth and e.spire can verify and maintain specific performance objectives.
 - 3. It is specified according to the interface offered in the tariff or local interconnection agreement (for example: for DS1 service the FCC # 1 tariff specifies that the interface meets the technical specifications detailed in Generic Requirements GR-342-CORE, Issue 1, December 1995.)
 - 4. The Parties provide their own equipment (CPE) to interface with the DS0, DS1, DS3, STS1 and/or OCn circuits on the customer premises.
- 1.5 The **Point of Interconnection** is the point at which the originating Party delivers its originated traffic to the terminating Party's first point of switching on the terminating Party's common (shared) network for call transport and termination. Points of Interconnection are available at either Access Tandems, Local Tandems, or End Offices as described in this Agreement. e.spire's requested Point of Interconnection will also be used for the receipt and delivery of transit traffic at BellSouth Access and Local Tandems. Points of Interconnection established at the BellSouth Local Tandem shall not be used to interconnect for the exchange of Switched Access Traffic.
- e.spire, at its option, shall establish Points of Presence and Points of Interface for the delivery of traffic originated by e.spire to BellSouth. The Point of Interface may not necessarily be established at the Point of Interconnection.
- 1.7 BellSouth shall designate the Points of Presence and Points of Interface for the delivery of traffic originated by BellSouth to e.spire for call transport and termination by e.spire.

1.8 Interconnection via Purchase of Facilities

1.8.1 Either Party may purchase Local Channel facilities from the Party's specified Point of Interface to its designated serving wire center. The Parties agree that charges for such Local Channel facilities are as set forth in Exhibit A to this Attachment. If a nonrecurring or recurring rate is not identified in Exhibit A for a Local Channel, the rate shall be as set forth in the appropriate Party's intrastate or

interstate tariff for switched access services as filed and effective with the appropriate Commission.

Additionally, either Party may purchase Dedicated Transport facilities from its designated serving wire center to the other Party's first point of switching. The Parties agree that charges for such Dedicated Transport facilities are as set forth in Exhibit A to this Attachment. If a nonrecurring or recurring rate is not identified in Exhibit A for Dedicated Transport, the rate shall be as set forth in the appropriate Party's intrastate or interstate tariff for switched access services services as filed and effective with the appropriate Commission.

- 1.8.2 For the purposes of this Attachment, Local Channel is defined as a switched transport facility between a Party's Point of Presence and its designated serving wire center.
- 1.8.3 For the purposes of this Attachment, Serving Wire Center is defined as the wire center owned by one Party from which the other Party would normally obtain dial tone for its Point of Presence.
- 1.8.4 For the purposes of this Attachment, Dedicated Transport is defined as a switch transport facility between a Party's designated serving wire center and the first point of switching on the other Party's common (shared) network.
- 1.9 BellSouth **Multiple Tandem Access** (**MTA**) provides for LATA wide BellSouth transport and termination of e.spire-originated local and intraLATA toll traffic by establishing a Point of Interconnection at a BellSouth access tandem with routing through multiple BellSouth access tandems as required. However, e.spire must still establish Points of Interconnection at all BellSouth access tandems where e.spire NXXs are "homed". MTA shall be provisioned in accordance with BellSouth's reasonable and nondiscriminatory Ordering Guidelines.
- 1.9.1 MTA does not include switched access traffic that transits the BellSouth network to an Interexchange Carrier (IXC). Switched Access traffic will be delivered to and by IXCs based on e.spire's NXX Access Tandem homing arrangement as specified by e_spire in the national Local Exchange Routing Guide (LERG).
- 1.9.2 For e.spire-originated local and intraLATA toll traffic that BellSouth transports but is destined for termination by a third party network (transit traffic), BellSouth MTA is required if multiple BellSouth access tandems are necessary to deliver the call to the third Party network.
- 1.9.3 To the extent e.spire does not purchase MTA in a calling area that has multiple access tandems serving the calling area as defined by BellSouth, e.spire must establish Points of Interconnection to every access tandem in the calling area in order to serve the entire calling area. To the extent e.spire does not purchase MTA and provides intraLATA toll service to its customers, it may be necessary for it to establish a Point of Interconnection to additional BellSouth access tandems that serve end offices outside the local calling area.

- 1.10 **Local Tandem Interconnection**. This interconnection arrangement allows e.spire to establish a Point of Interconnection at BellSouth local tandems for: (1) the delivery of e.spire -originated local traffic transported and terminated by BellSouth to BellSouth end offices within the local calling area as defined in BellSouth's GSST, section A3 served by those BellSouth local tandems, and (2) for local transit traffic transported by BellSouth for third party network providers who have also established Points of Interconnection at those BellSouth local tandems.
- 1.10.1 If e.spire opts for local tandem interconnection when a specified local calling area is served by more than one BellSouth local tandem, e.spire must designate a "home" local tandem for each of its assigned NPA/NXXs and establish trunk connections to such local tandems. Additionally, e.spire may choose to establish a Point of Interconnection at the BellSouth local tandems where it has no codes homing but is not required to do so. e.spire may deliver local traffic to a "home" BellSouth local tandem that is destined for other BellSouth or third party network provider end offices subtending other BellSouth local tandems in the same local calling area where e.spire does not choose to establish a Point of Interconnection. It is e.spire's responsibility to enter its own NPA/NXX local tandem homing arrangements into the Local Exchange Routing Guide (LERG) either directly or via a vendor in order for other third party network providers to determine appropriate traffic routing to e.spire's codes. Likewise, e.spire shall obtain its routing information from the LERG.
- 1.10.2 Notwithstanding establishing Points of Interconnection to BellSouth's local tandems, e.spire must also establish Points of Interconnection to BellSouth access tandems within the LATA on which e.spire has NPA/NXXs homed for the delivery of Interexchange Carrier Switched Access (SWA) and toll traffic, and traffic to Type 2A CMRS connections located at the access tandems. BellSouth cannot switch SWA traffic through more than one BellSouth access tandem. SWA, Type 2A CMRS or toll traffic routed to the local tandem in error will not be backhauled to the BellSouth access tandem for completion. (Type 2A CMRS interconnection is defined in BellSouth's A35 General Subscriber Services Tariff.)
- 1.10.3 BellSouth's provisioning of local tandem interconnection assumes that e.spire has executed the necessary local interconnection agreements with the other third party network providers subtending those local tandems as required by the Act.

1.11 Fiber Meet

1.11.1 "Fiber-Meet" is an interconnection arrangement whereby the Parties physically interconnect their networks via an optical fiber interface (as opposed to an electrical interface) at a mutually agreed upon location, at which one Party's facilities, provisioning, and maintenance responsibility begins and the other Party's responsibility ends (i.e. Point Of Interface).

- 1.11.2 If e.spire elects to establish a Point of Interconnection with BellSouth pursuant to a Fiber Meet, e.spire and BellSouth shall jointly engineer and operate a Synchronous Optical Network ("SONET") transmission system by which they shall interconnect networks and routing of local traffic via a Local Channel facility at either the DS0, DS1, or DS3 level and shall be ordered via an Access Services Request ("ASR") in the initial phase of this offering. The Parties shall work jointly to determine the specific transmission system. However, e.spire's SONET transmission must be compatible with BellSouth's equipment in the serving wire center and the Data Communications Channel (DCC) must be turned off.
- 1.11.3 BellSouth shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the BellSouth Interconnection Wire Center ("BIWC").
- e.spire shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the e.spire Interconnection Wire Center (""e.spire Wire Center").
- 1.11.5 BellSouth shall designate a Point of Interface outside the BIWC as a Fiber Meet point, and shall make all necessary preparations to receive, and to allow and enable e.spire to deliver, fiber optic facilities into the Point of Interface with sufficient spare length to reach the fusion splice point at the Point of Interface. BellSouth shall, wholly at its own expense, procure, install, and maintain the fusion splicing point in the Point of Interface. A Common Language Location Identification ("CLLI") code will be established for each Point of Interface. The code established must be a building type code. All orders shall originate from the Point of Interface (i.e., Point of Interface to e.spire, Point of Interface to BellSouth).
- e.spire shall deliver and maintain such strands wholly at its own expense. Upon verbal request by e.spire, BellSouth shall allow e.spire access to the Fiber Meet entry point for maintenance purposes as promptly as possible and in a reasonable and nondiscriminatory manner.
- 1.11.7 The Parties shall jointly coordinate and undertake maintenance of the SONET transmission system. Each Party shall be responsible for maintaining the components of their own SONET transmission system.
- 1.11.8 Each Party will be responsible for (i) providing its own transport facilities to the Fiber Meet, and (ii) the cost to build-out its facilities to such Fiber Meet.
- 1.11.9 Neither Party shall charge the other for its portion of the Fiber Meet facility used exclusively for non-transit local traffic (i.e. the Local Channel). Charges incurred for other services including dedicated transport facilities to the Point of Interconnection if applicable will apply. Charges for Switched and Special Access Services shall be billed in accordance with the applicable Access Service

tariff (*i.e.*, the providing Party's Interstate or Intrastate Access Services Tariff as filed and in effect with the FCC or appropriate Commission).

2. INTERCONNECTION TRUNKING AND ROUTING

- 2.1 BellSouth and e.spire shall establish interconnecting trunk groups and trunking configurations between networks including the establishment of one-way or two-way trunks in accordance with applicable, reasonable and nondiscriminatory requirements of *BellSouth Call Transport & Termination Service For Facility Based CLECs section of the Facility Based CLEC Activation Requirements Customer Guide* as it is revised from time to time.
- Any e.spire interconnection request that deviates from the reasonable and nondiscriminatory standard trunking configurations as described in the *BellSouth Call Transport & Termination Service For Facility Based CLECs section of the Facility Based CLEC Activation Requirements Customer Guide* that affects traffic delivered to e.spire from a BellSouth switch that requires special BellSouth switch translations and other network modifications will require e.spire to submit a Bona Fide Request/New Business Request via the Bona Fide Request/New Business Request Process set forth in General Terms and Conditions.
 - 2.3 All terms and conditions, as well as charges, both non-recurring and recurring, associated with interconnecting trunk groups between BellSouth and e.spire not addressed in Exhibit A shall be as set forth in the appropriate Party's intrastate or interstate tariff for switched access services as filed and in effect with the FCC or appropriate state Commission. For two-way trunking that carries the Parties' local and intraLATA toll traffic only, excluding trunking that carries Transit Traffic, the Parties shall be compensated for the recurring charges for transport facilities and nonrecurring charges for facility additions based on the percentage of the total traffic originated by each Party. BellSouth shall determine the applicable percentages twice per year based on the previous 6 months' minutes of use billed by each Party. The Parties shall be compensated for nonrecurring charges for initial facilities based on the joint forecasts for circuits required by each Party. Each Party shall be responsible for ordering and paying for any facilities for twoway trunks carrying its transit traffic. Furthermore, each Party shall be responsible for the compensation for transport facilities for two-way trunking that it orders for its traffic but utilizes unidirectionally.
- 2.4 The Parties shall utilize direct end office trunking under the following conditions:
 - (1) Tandem Exhaust If a tandem through which the Parties are interconnected is unable to, or is forecasted to be unable to support additional traffic loads for any period of time, the Parties will mutually agree on an end office trunking plan that will alleviate the tandem capacity shortage and ensure completion of traffic between e.spire's and BellSouth's subscribers.

(2) Traffic Volume –To the extent either Party has the capability to measure the amount of traffic between a e.spire switching center and a BellSouth end office, either Party shall install and retain direct end office trunking sufficient to handle actual or reasonably forecasted traffic volumes, whichever is greater, between a e.spire switching center and a BellSouth end office where the traffic exceeds or is forecasted to exceed a single DS1 of local traffic per month. Either Party will install additional capacity between such points when overflow traffic between e.spire's switching center and BellSouth's end office exceeds or is forecasted to exceed a single DS1 of local traffic per month. In the case of one way trunking, additional trunking shall only be required by the Party whose trunking has achieved the preceding usage threshold.

Mutual Agreement - The Parties may install direct end office trunking upon mutual agreement in the absence of the conditions (1) or (2) above and agreement will not unreasonably be withheld.

- 2.5 Switched Access traffic will be delivered to and by IXCs based on e.spire's NXX Access Tandem homing arrangement as specified by e.spire in the national Local Exchange Routing Guide (LERG).
- 2.6 All trunk groups will be provisioned as Signaling System 7 (SS7) capable where technically feasible.
- 2.7 Subject to Section 2.8 below, the standard interval used for the provision of local interconnection trunk groups shall in no event be longer than forty-five (45) working days from the receipt of an error-free ASR for the establishment of new local interconnection trunk groups comprised of 96 or fewer new trunks, or ten (10) working days from the receipt of an error-free ASR for orders of 96 or fewer trunks for additions to existing local interconnection trunk groups.
- 2.8 For orders that comprise a major project, the implementation and intervals shall be jointly planned and coordinated. Major projects are those that require the coordination and execution of multiple orders or related activities between and among BellSouth and e.spire work groups, including but not limited to the initial establishment of interconnection or transit trunk groups in a service area, NXX code moves, re-homes, facility grooming or network rearrangements. Major projects also include orders for more than 96 new or additional trunks.

3. NETWORK DESIGN AND MANAGEMENT FOR INTERCONNECTION

3.1 <u>Network Management and Changes</u>. Both Parties will work cooperatively with each other to install and maintain the most effective and reliable interconnected telecommunications networks, including but not limited to, the exchange of toll-free maintenance contact numbers and escalation procedures. Both Parties agree

to provide public notice of changes in the information necessary for the transmission and routing of services using its local exchange facilities or networks, as well as of any other changes that would affect the interoperability of those facilities and networks.

- Interconnection Technical Standards. The interconnection of all networks will be based upon accepted industry/national guidelines for transmission standards and traffic blocking criteria. Interconnecting facilities shall conform, at a minimum, to the telecommunications industry standard of DS-1 pursuant to BellCore Standard No. TR-NWT-00499. Signal transfer point, Signaling System 7 ("SS7") connectivity is required at each interconnection point. BellSouth will provide out-of-band signaling using Common Channel Signaling Access Capability where technically and economically feasible, in accordance with the technical specifications set forth in the BellSouth Guidelines to Technical Publication, TR-TSV-000905. Facilities of each Party shall provide the necessary on-hook, off-hook answer and disconnect supervision and shall hand off calling number ID (Calling Party Number) when technically feasible.
- 3.3 Quality of Interconnection. The local interconnection for the transmission and routing of telephone exchange service and exchange access that each Party provides to each other will be at least equal in quality to what it provides to itself and any subsidiary or affiliate, where technically feasible, or to any other Party to which each Party provides local interconnection.
- 3.4 <u>Network Management Controls.</u> Both Parties will work cooperatively and in good faith to exchange applicable information and to apply sound network management principles by invoking appropriate network management controls, *e.g.*, call gapping, to alleviate or prevent trunk blocking and network congestion.
- Common Channel Signaling. Both Parties will provide LEC-to-LEC Common Channel Signaling ("CCS") to each other, where available, in conjunction with all traffic in order to enable full interoperability of CLASS features and functions except for call return. All CCS signaling parameters will be provided, including automatic number identification ("ANI"), originating line information ("OLI") calling company category, charge number, etc. All privacy indicators will be honored, and each Party will cooperate with each other on the exchange of Transactional Capabilities Application Part ("TCAP") messages to facilitate full interoperability of CCS-based features between the respective networks.
- 3.6 <u>Forecasting Requirements.</u>
- 3.6.1 The Parties shall exchange technical descriptions and forecasts of their interconnection and traffic requirements in sufficient detail necessary to establish the interconnections required to assure traffic completion to and from all customers in their respective designated service areas. In order for BellSouth to provide as accurate reciprocal trunking forecasts as possible to e.spire, e.spire must timely inform BellSouth of any known or anticipated events that may affect BellSouth reciprocal trunking requirements. If e.spire refuses to provide such

information, BellSouth shall provide reciprocal trunking forecasts based only on existing trunk group growth and BellSouth's annual estimated percentage of BellSouth subscriber line growth.

- 3.6.2 Both Parties shall meet every six months or at otherwise mutually agreeable intervals for the purpose of exchanging non-binding forecast of its traffic and volume requirements for the interconnection and network elements provided under this Agreement, in the form and in such detail as agreed by the Parties. The Parties agree that each forecast provided under this Section shall be deemed "Confidential Information" in the General Terms and Conditions Part A of this Agreement.
- 3.6.3 The trunk forecast should include trunk requirements for all of the interconnecting trunk groups for the current year plus the next two future years. The forecast meeting between the two companies may be a face-to-face meeting, video conference or audio conference. It may be held regionally or geographically. Ideally, these forecast meetings should be held at least semi-annually, or more often if the forecast is no longer usable. Updates to a forecast or portions thereof should be made whenever the Party providing the forecast deems that the latest trunk requirements exceed the original quantities by 24 trunks or 10%, whichever is greater. Either Party should notify the other Party if they have measurements indicating that a trunk group is exceeding its designed call carrying capacity and is impacting other trunk groups in the network. Also, either Party should notify the other Party if they know of situations in which the traffic load is expected to increase significantly and thus affect the interconnecting trunk requirements as well as the trunk requirements within the other Party's network. The Parties agree that the forecast information provided under this Section shall be deemed "Confidential Information" as set forth in the General Terms and Conditions of this Agreement.
- 3.6.4 For a non-binding trunk forecast, agreement between the two Parties on the trunk quantities and the timeframe of those trunks does not imply any liability for failure to perform if the trunks are not available for use at the required time.
- 3.6.5 <u>Signaling Call Information</u>. BellSouth and e.spire will send and receive 10 digits for local traffic. Additionally, BellSouth and e.spire will exchange the proper call information, i.e. originated call company number and destination call company number, CIC, and OZZ, including all proper translations for routing between networks and any information necessary for billing.

4. PARITY IN ORDERING AND PROVISIONING

Each Party shall provide interconnection ordering and provisioning services to the other Party that are Equal in Quality to the ordering and provisioning services the Parties provide themselves. "Equal in Quality" shall have the meaning accorded in Section 51.305(a)(3) of the FCC's Rules, 47 C.F.R. § 51.305(a)(3).

Reasonable and nondiscriminatory procedures for ordering and provisioning BellSouth interconnection services are set forth in the *BellSouth Call Transport & Termination Service For Facility Based CLECs section of the Facility Based CLEC Activation Requirements Customer Guide*.

5. LOCAL DIALING PARITY

Each Party shall provide local dialing parity, meaning that each Party's customers will not have to dial any greater number of digits than the other Party's customers to complete the same call. In addition, under equivalent interconnection arrangements, e.spire local service customers will experience at least the same quality as BellSouth local service customers regarding post-dial delay, call completion rate and transmission quality.

6. INTERCONNECTION COMPENSATION

- 6.1 Compensation for Call Transportation and Termination for Local Traffic and Inter-Carrier Compensation for ISP-Bound Traffic
- 6.1.1 Local Traffic is defined as any telephone call that originates in one exchange and terminates in either the same exchange, or a corresponding Extended Area Service ("EAS") exchange.
- 6.1.2 The Parties will compensate each other on a mutual and reciprocal basis for the transport and termination of Local Traffic at the following rates:

1/1/00 - 12/31/00	\$.00200 per MOU
1/1/01 - 12/31/01	\$.00175 per MOU
1/1/02 = 12/31/02	\$ 00150 per MOU

The Parties recognize and agree that they negotiated these annual rates together as a complete rate structure to apply over the full three-year term of this Agreement and that the parties would not have mutually agreed to accept a single annual rate in any single year. Nothing in this Paragraph shall limit e.spire's rights pursuant to Paragraph 6.1.3.3.

6.1.3 The Parties have been unable to agree upon whether dial up calls to Information Service Providers ("ISPs") should be considered Local Traffic for purposes of this Agreement. Dial-up Calls are defined as calls to an ISP that are dialed by using a local dialing pattern (7 or 10 digits) by the calling party (hereinafter referred to as "ISP-bound traffic"). However, without prejudice to either Party's position concerning the application of reciprocal compensation to ISP-bound traffic, the Parties agree for purposes of this Agreement only to compensate each other for ISP-

bound traffic at the same per minute of use rates set forth in Paragraph 6.1.2. It is expressly understood and agreed that this inter-carrier compensation mechanism for ISP-bound traffic is being established: (1) in consideration for a waiver and release by each party for any and all claims for reciprocal compensation for ISP-bound traffic exchanged between the parties prior to December 31, 1999, which is hereby given; and (2) subject to the terms and conditions in section 6.1.4.

- 6.1.3.1 The Parties recognize and agree that the FCC, courts of competent jurisdiction, or state commissions with jurisdiction over the Parties will issue subsequent decisions on ISP-bound traffic ("Subsequent Decisions"). Notwithstanding any provision in this Agreement to the contrary, the inter-carrier compensation mechanism established in section 6.1.3 shall continue at the rates set forth in section 6.1.2 for the full term of this Agreement without regard to such Subsequent Decisions, except as provided for in sections 6.1.3.2 and 6.1.3.3.
- 6.1.3.2 To the extent such Subsequent Decisions render the inter-carrier compensation mechanism for ISP-bound traffic set forth in section 6.1.3 in violation of applicable federal or state law, the Parties agree to amend this Agreement within thirty (30) days of the effective date of any such Subsequent Decision to conform the inter-carrier compensation mechanism set forth in section 6.1.3 with such Subsequent Decision. In the event of such an amendment, there will no true-up for compensation paid prior to the amendment.
- Nothing herein shall preclude e.spire from exercising its rights under this Agreement or Section 252(i) of the 1996 Act and applicable FCC regulations to elect rates, terms, and conditions with respect to the payment of reciprocal compensation from any other approved interconnection agreement executed by BellSouth under which BellSouth is paying reciprocal compensation for ISP-bound traffic other than on an interim basis. The Parties recognize and agree that this provision is intended to ensure that e.spire is treated in the same manner with respect to the payment of reciprocal compensation for ISP-bound traffic as the competing local exchange carrier from whose interconnection agreement e.spire seeks to elect rates, terms, and conditions. Accordingly, e.spire agrees that it will not seek to elect reciprocal compensation rates, terms, or conditions from another interconnection agreement unless those rates, terms, and conditions apply to ISP-bound traffic (other than on an interim basis), either by the express terms of that agreement, by voluntary action by BellSouth, or pursuant to an effective state commission or court order.
- 6.1.4 The Parties recognize and agree that the compensation for the transport and termination of Local Traffic set forth in section 6.1.2 and the inter-carrier compensation mechanism for ISP-bound traffic set forth in section 6.1.3 are intended to allow each Party to recover costs associated with such traffic. Accordingly, the Parties recognize and agree that such compensation will not be billed and shall not be paid for a call placed by an end user customer, or placed on behalf of an end user

customer, to establish or maintain a network connection if: (1) such call is not recognized by industry practice to constitute traffic (voice or data) which results from a telephone call; (2) the end user customer does not control the dialed number destination and content of that call; and (3) the primary purpose of that call is to generate the payment of reciprocal compensation as a result of establishing or maintaining the network connection.

- Neither Party shall represent switched access traffic as Local Traffic for purposes of payment of reciprocal compensation.
- Unidentifiable traffic. e.spire shall utilize its NPA/NXXs in such a way and will provide the necessary information so that BellSouth shall be able to distinguish Local from IntraLATA Toll traffic for BellSouth originated traffic. e.spire end users' assigned NPA/NXX line numbers shall be physically located in the BellSouth rate center with which the NPA/NXX has been associated. Whenever BellSouth delivers traffic to e.spire for termination on the e.spire 's network, if BellSouth cannot determine, because of the manner in which e.spire has utilized its NXX codes whether the traffic is local or toll, BellSouth will charge the applicable rates for originating intrastate network access service as reflected in BellSouth's Intrastate Access Service Tariff. BellSouth will make appropriate billing adjustments if e.spire can provide sufficient information for BellSouth to determine whether said traffic is local or toll.
- 6.4 Percent Local Use. Each Party will report to the other a Percentage Local Usage ("PLU"). The application of the PLU will determine the amount of local minutes to be billed to the other Party. For purposes of developing the PLU, each Party shall consider every local call and every long distance call, excluding transit traffic. By the first of January, April, July and October of each year, BellSouth and e.spire shall provide a positive report updating the PLU. Notwithstanding the foregoing, where the terminating company has message recording technology that identifies the traffic terminated, such information, in lieu of the PLU factor, shall at the company's option be utilized to determine the appropriate local usage compensation to be paid.
- 6.5 Percentage Interstate Usage. For combined interstate and intrastate e.spire traffic terminated by BellSouth over the same facilities, e.spire will be required to provide a projected Percentage Interstate Usage ("PIU") to BellSouth. All jurisdictional report requirements, rules and regulations for Interexchange Carriers specified in BellSouth's Intrastate Access Services Tariff will apply to e.spire. After interstate and intrastate traffic percentages have been determined by use of PIU procedures, the PLU factor will be used for application and billing of local interconnection. Notwithstanding the foregoing, where the terminating company has message recording technology that identifies the traffic terminated, such information, in lieu of the PLU factor, shall at the company's option be utilized to determine the appropriate local usage compensation to be paid.

Audits. On thirty (30) days written notice, each Party must provide the other the ability and opportunity to conduct an annual audit to ensure the proper billing of traffic. BellSouth and e.spire shall retain records of call detail for a minimum of nine months from which a PLU and/or PIU can be ascertained. The audit shall be accomplished during normal business hours at an office designated by the Party being audited. Audit requests shall not be submitted more frequently than one (1) time per calendar year. Audits shall be performed by a mutually acceptable independent auditory paid for by the Party requesting the audit. The PLU and/or PIU shall be adjusted based upon the audit results and shall apply to the usage for the quarter the audit was completed, to the usage for the quarter prior to the completion of the audit, and to the usage for the two quarters following the completion of the audit. If, as a result of an audit, either Party is found to have overstated the PLU and/or PIU by twenty percentage points (20%) or more, that Party shall reimburse the auditing Party for the cost of the audit.

6.7 **Rate True-up**

This section applies only to Tennessee.

- 6.7.1 The interim prices for Unbundled Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:
- 6.7.2 The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by an effective order of the Commission which order meets the criteria of Section 6.7.4 below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions of Section 16 of the General Terms and Conditions and Attachment 1 of the Agreement.
- 6.7.3 The Parties may continue to negotiate toward final prices, but in the event that no such agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in the General Terms and Conditions of the Agreement, so long as they file the resulting Agreement with the Commission as a "negotiated agreement" under Section 252(e) of the Act.
- A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a

final order in any other Commission proceeding which meets the following criteria:

- (a) BellSouth and CLEC is entitled to be a full Party to the proceeding;
- (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,(c) It shall include as an issue the geographic deaveraging of unbundled element prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.
- 6.8 Compensation for IntraLATA Toll Traffic
- 6.8.1 <u>IntraLATA Toll Traffic</u>. IntraLATA Toll Traffic is defined as any traffic that originates and terminates within a single LATA, excluding Local Traffic or EAS.
- 6.8.2 Compensation for intraLATA toll traffic. For terminating its intraLATA toll traffic on the other company's network, the originating Party will pay the terminating Party's current intrastate or interstate, whichever is appropriate, terminating switched access tariff rates as set forth in the terminating Party's Intrastate or Interstate Access Services Tariff as filed and in effect with the FCC or state Commission. The appropriate charges will be determined by the routing of the call. If one Party is the other Party's end user's presubscribed interexchange carrier or if one Party's end user uses the other Party as an interexchange carrier on a 101XXXXX basis, the originating Party will charge the other Party the appropriate originating switched access tariff rates as set forth in the originating Party's Intrastate or Interstate Access Services Tariff as filed and in effect with the FCC or appropriate state Commission.
- 6.8.3 <u>Compensation for 800 Traffic</u>. Each Party shall compensate the other pursuant to the appropriate switched access charges, including the database query charge as set forth in the providing Party's tariff as filed and in effect with the FCC or appropriate state Commission.
- 6.8.4 Records for 800 Billing. Each Party will provide to the other the appropriate records necessary for billing intraLATA 800 customers. The records provided will be in a standard EMI format for a fee of \$0.013 per record.
- 6.8.5 <u>800 Access Screening</u>. Should e.spire require 800 Access Ten Digit Screening Service from BellSouth, it shall have signaling transfer points connecting directly to BellSouth's local or regional signaling transfer point for service control point database query information. e.spire shall utilize SS7 signaling links, ports and usage as set forth in Attachment 2. e.spire will not utilize switched access FGD service. 800 Access Ten Digit Screening Service is an originating service that is provided via 800 Switched Access Service trunk groups from BellSouth's SS7 equipped end office or access tandem providing an IXC identification function

and delivery of a call to the IXC based on the dialed ten digit number. The terms and conditions for this service are set out in BellSouth's Intrastate Access Services Tariff as amended.

- 6.9 Mutual Provision of Switched Access Service
- 6.9.1 Switched Access Traffic. Switched Access Traffic is defined as telephone calls requiring local transmission or switching services for the purpose of the origination or termination of Telephone Toll Service. Switched Access Traffic includes the following types of traffic: Feature Group A, Feature Group B, Feature Group C, Feature Group D, toll free access (e.g., 800/877/888), 900 access, and their successors or similar Switched Exchange Access Services. The Parties have been unable to agree as to whether "Voice-Over-Internet Protocol" transmissions ("VOIP") which cross LATA boundaries constitute Switched Access Traffic. Notwithstanding the foregoing, and without waiving any rights with respect to either Party's position as to the jurisdictional nature of VOIP, the Parties agree to abide by any effective and applicable FCC rules and orders regarding the nature of such traffic and the compensation payable by the Parties for such traffic, if any.
- 6.9.2 When BellSouth and e.spire provide an access service connection between an interexchange carrier ("IXC") and each other, each Party will provide its own access services to the IXC on a multi-bill, multi-tariff meet-point basis. Each Party will bill its own access services rates to the IXC with the exception of the interconnection charge. The interconnection charge will be billed by the Party providing the end office function. The Parties will use the Multiple Exchange Carrier Access Billing (MECAB) system to establish meet point billing for all applicable traffic including traffic terminated to ported numbers via INP and nongeographic NPAs. Thirty (30)-day billing periods will be employed for these arrangements. The recording Party agrees to provide to the initial billing company, at no charge, the switched access detailed usage data within no more than sixty (60) days after the recording date. The initial billing company will provide the switched access summary usage data to all subsequent billing companies within 10 days of rendering the initial bill to the IXC. Each company will notify the other when it is not feasible to meet these requirements so that the customers may be notified for any necessary revenue accrual associated with the significantly delayed recording or billing. As business requirements change data reporting requirements may be modified as necessary by mutual agreement of the Parties.
- 6.9.3 In the event that either Party fails to provide switched access detailed usage data to the other Party within 90 days after the recording date and the receiving Party is unable to bill and/or collect access revenues due to the sending Party's failure to provide such data within said time period, then the Party failing to send the data as specified herein shall be liable to the other Party in an amount equal to the unbillable or uncollectible revenues. Each company will provide complete documentation to the other to substantiate any claim of unbillable access revenues. A negotiated settlement will be agreed upon between the companies.

- 6.9.4 Each company will retain for a minimum period of sixty (60) days, access message detail sufficient to recreate any data which is lost or damaged by their company or any third party involved in processing or transporting data.
- 6.9.5 In the event of a loss of data, both Parties shall cooperate to reconstruct the lost data and shall make best efforts to do so within 48 hours. If such reconstruction is not possible, the Parties shall use a reasonable estimate of the lost data, based on twelve (12) months of prior usage data; provided that if twelve (12) months of prior usage data is not available, the Parties shall base the estimate on as much prior usage data that is available; and further provided, however, that if reconstruction is required prior to the availability of at least three (3) months of prior usage data, the Parties shall defer such reconstruction until three (3) months of prior usage data is available. If the estimated billing is not accepted for payment by the affected Access Services Customer(s), the responsible Party shall be liable to the other Party for any resulting lost revenue. Lost revenue is revenue that could not be billed to Access Service customers. Lost revenue will be calculated by subtracting the amount actually paid by the affected Access Services Customer(s) from the estimated billing derived pursuant to the process set forth in this section.
- 6.9.6 Each company also agrees to process the recreated data within forty-eight (48) hours of receipt at its data processing center.
- 6.9.7 All claims should be filed with the other company within 120 days of the receipt of the date of the unbillable usage.
- 6.9.8 The Initial Billing Company shall keep records of its billing activities relating to jointly-provided Intrastate and Interstate access services in sufficient detail to permit the Subsequent Billing Company to, by formal or informal review or audit, to verify the accuracy and reasonableness of the jointly-provided access billing data provided by the Initial billing Company. Each company agrees to cooperate in such formal or informal reviews or audits and further agrees to jointly review the findings of such reviews or audits in order to resolve any differences concerning the findings thereof.
- 6.10 **Transit Traffic Service**. Each Party shall provide tandem switching and transport services for the other's transit traffic. Transit traffic is traffic originating on one carrier's network that is switched and transported by a second carrier's network and terminates on a third carrier's network. Rates for local transit traffic shall be the applicable call transport and termination charges for Local Traffic, as set forth in Exhibit A to this Attachment. Rates for intraLATA toll and Switched Access transit traffic shall be the applicable call transport and termination rates as set forth in the providing Party's Intrastate or Interstate switched access tariff as filed and in effect with the FCC or appropriate state Commission. Wireless Type 1 traffic shall not be treated as transit traffic from a routing or billing perspective. Wireless Type 2A traffic shall not be treated as transit traffic from a routing or

billing perspective until either Party and the Wireless carrier have the capability to properly meet-point-bill in accordance with MECAB guidelines.

- 6.10.1 The delivery of traffic originated by e.spire which transits the BellSouth network and is transported to another carrier's network is excluded from any BellSouth billing guarantees and will be delivered at the rates as set forth in Exhibit A to this Attachment. e.spire is responsible for establishing the necessary agreements or the placement of valid orders with the terminating carrier for the receipt of this traffic through the BellSouth network. BellSouth will not be liable for any compensation to the terminating carrier as a result of providing the transit function. Further, e.spire agrees to compensate BellSouth for any charges or costs for the delivery of transit traffic to a connecting carrier on behalf of e.spire for which a valid contract or order has not been established. Additionally, the Parties agree that any billing to a third party or other telecommunications carrier under this section shall be pursuant to MECAB procedures.
- 6.10.2 Except for as provided in 6.10.3, transit charges shall only be assessed on the originating carrier and shall not be assessed on the terminating carrier.
- 6.10.3 Transit charges associated with the provisioning of toll free services (e.g., 800/888/877) shall be assessed upon the terminating carrier and shall not be imposed on the originating carrier.

7. FRAME RELAY SERVICE

- 7.1 In addition to the Local Interconnection services set forth above, BellSouth will offer a network to network Interconnection arrangement between BellSouth's and e.spire's frame relay switches as set forth below. The following provisions will apply only to Frame Relay Service and Exchange Access Frame Relay Service in those states in which e.spire is certified and providing Frame Relay Service as a Local Exchange Carrier and where traffic is being exchanged between e.spire and BellSouth Frame Relay Switches in the same LATA.
- 7.2 The Parties agree to establish two-way Frame Relay facilities between their respective Frame Relay Switches to the mutually-agreed upon Frame Relay Service point(s) of interconnection ("POIs") within the LATA.
- 7.2.1 Upon the request of either Party, such interconnection will be established where BellSouth and e.spire have Frame Relay Switches in the same LATA. Where there are multiple Frame Relay switches in the central office of a Party, an interconnection with any one of the switches will be considered an interconnection with all of the switches at that central office for purposes of routing packet traffic.
- 7.2.2 The Parties agree to provision local and IntraLATA Frame Relay Service and Exchange Access Frame Relay Service (both intrastate and interstate) over Frame Relay Trunks between the respective Frame Relay switches and the POIs.

- 7.2.3 The Parties agree to assess each other reciprocal charges for the facilities that each provides to the other according to the Percent Local Circuit Use ("PLCU") factor PLCU, determined as follows:
 - (i) Frame Relay framed packet data is transported within Virtual Circuits ("VC"). For the purposes of calculating the PLCU, if all the data packets transported within a VC remain within the LATA, then consistent with the local definitions in this Agreement, the traffic on that VC is local ("Local VC").
 - (ii) If the originating and terminating locations of the two way packet data traffic are not in the same LATA, the traffic on that VC is interLATA.
 - (iii) The PLCU shall be determined by dividing the total number of Local VCs, by the total number of VCs on each Frame Relay facility at the end of the reporting period. The Parties agree to renegotiate the method for determining PLCU, at either Parties' request, and within 90 days, if either Party notifies the other that it has found that this method does not adequately represent the PLCU.
 - (iv) If there are no VCs on a facility when it is billed, the PLCU will be zero.
- 7.3 BellSouth will provide the Frame Relay Trunk(s) between the Parties' respective Frame Relay Switches. The Parties will be compensated as follows: BellSouth will invoice, and e.spire will pay, the total non-recurring and recurring charges for the trunk facility. e.spire will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed charges for the trunk facility by one-half of e.spire's PLCU.
- 7.3.1 If e.spire requests interconnection outside the serving area in which the POI is located, then e.spire may, at its option, purchase facilities and transport between Frame Relay Network Serving areas, as defined in Section A40 of the General Subscriber Service Tariff, according to the rates, terms and conditions of the applicable tariff General Subscriber Service Tariff for that state.
- Each Party will provide a Frame Relay network-to-network interface ("NNI") port to the other Party for each trunk facility provided pursuant to 7.2, above.

 Compensation for NNI ports shall be based upon the NNI rates set forth in the BellSouth F.C.C Tariff No. 1. Pursuant to that tariff, e.spire may select a month-to-month or term rate structure for the NNI ports BellSouth provides to e.spire. Whatever rate structure e.spire selects shall be deemed to be the same rate structure that applies to the NNI port e.spire provides to BellSouth. There shall be no termination liability to either party for the local portion of the NNI port as determined by the e.spire PLCU at the time of termination.
- 7.5 Compensation for the NNI ports shall be calculated as follows:

- 7.5.1 For NNI ports provided by BellSouth to e.spire, BellSouth will invoice, and e.spire will pay, the total non-recurring and recurring charges for the NNI port. e.spire will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed non-recurring and recurring charges for the NNI port by one-half of e.spire's PLCU.
- 7.5.2 For NNI ports provided by e.spire to BellSouth, e.spire will invoice, and BellSouth will pay, the total non-recurring and recurring charges for the NNI port. BellSouth will then invoice, and e.spire will pay, an amount determined as follows: e.spire's combined interLATA and local usage will be calculated by subtracting one-half of e.spire's PLCU factor from one hundred percent. The difference will then be multiplied by the total charges initially billed by e.spire for the NNI port. BellSouth will then invoice, and e.spire will pay, this amount to BellSouth.
- A Permanent Virtual Circuit ("PVC") is a logical channel from a frame relay network interface (e.g., NNI or User Network Interface) to another frame relay network interface. A PVC is created when a Data Link Channel Identifier ("DLCI") is mapped together with another DLCI. Neither Party will charge the other Party any DLCI or Committed Information Rate ("CIR") charges for the PVC from its Frame Relay switch to its own subscriber's premises.
- 7.7 For the PVC between the e.spire and BellSouth Frame Relay switches, compensation for the DLCI and CIR charges are based upon the rates in the BellSouth FCC Tariff No. 1. Compensation for PVC and CIR rate elements shall be calculated as follows:
- 7.7.1 For PVCs between the BellSouth Frame Relay switch and the e.spire Frame Relay switch, BellSouth will invoice, and e.spire will pay, the total non-recurring and recurring DLCI and CIR charges. If the VC is a Local VC, e.spire will invoice and BellSouth will pay, 100% of the DLCI and CIR charges initially billed by BellSouth for that PVC. If the VC is not local, no compensation will be paid to e.spire for the PVC.
- 7.7.2 Each Party will compensate the other Party for any applicable Feature Change or Transfer of Service Charges as set forth in BellSouth's Tariff F.C.C. No. 1. A.6.3. The Parties agree to limit the sum of the CIR for the VCs on a given NNI port to not more than two times the port speed.
- 7.8 Except as expressly provided herein, this Agreement does not address or alter in any way either Party's provision of Exchange Access Frame Relay Service or interLATA Frame Relay Service. All charges by each Party to the other for carriage of Exchange Access Frame Relay Service or interLATA Frame Relay Service are included in the BellSouth access tariffs.

- 7.9 Until such time as BellSouth obtains authority to provide in-region, interLATA service, e.spire will identify and report its PLCU to BellSouth on a quarterly basis.
- 7.10 Either Party may request a review or audit of the various service components, including but not limited to a Party's determination of its PLCU, consistent with the provisions of section E2 of the BellSouth State Access Services tariffs or Section 2 of the BellSouth FCC No.1 Tariff.
- 7.11 If during the term of this Agreement, BellSouth obtains authority to provide inregion, interLATA service, the Parties shall renegotiate the provisions of 7.3, 7.5, 7.7 and 7.9 to account for BellSouth's PLCU. In the event the parties are unable to reach agreement within one hundred eighty (180) days of the date BellSouth receives interLATA authority, the matter shall be resolved pursuant to the dispute resolution provisions set forth in the Interconnection Agreement.

8. OPERATIONAL SUPPORT SYSTEMS (OSS) RATES

All Local Service Requests ("LSRs") or Access Service Requests ("ASRs") submitted for products and services under this Attachment will be subject to the OSS charges set forth in the General Terms and Conditions of this Agreement.

BELLSOUTH/e.spire RATES LOCAL INTERCONNECTION

Direct Local Interconnection, per mou (same as End Office Switching in FL, 8, LA)			RATES BY STATE										
Contract Contract	DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN		
Direct Local Interconnection, per more more famour as Entire Office Sentitiving in FL & LV NA \$0,0005 NA NA \$0,0	LOCAL INTERCONNECTION (CALL TRANSPORT AND TERMINATION)												
Transfer Setting per mod	End Office Switching, per mou	N/A	\$0.0018		\$0.0016333	\$0.002562		\$0.0023771	\$0.004	\$0.0019295	\$0.0019		
Tandem Lose International process Transport of the process Transport							\$0.00209				NA		
Transfer Local Infections (Local Infection) per more) includes and office satisfancy alternated (Local Internation), per more (Local Internation), per mor											\$0.000676		
Multiple Transfert Swotning, per most agopties to initial student only, effective 1099 NA \$0.00155 NA NA \$0.0030 NA NA NA NA NA NA NA N		N/A									NA		
Local intermediary, per moli (agolese to france only)											NA		
All serms and conditions, as well as changes, both non-recurring and recurring, associated with section E.G. of the perspectable BISS Table (Park Park Park Park Park Park Park Park											NA		
### Access Tarrif											NA		
Rates Rate											BST State		
Transcript intermediary Change, pier mou" N/A \$0,00016 N/A N/A \$0,00016 N/A											Access Tariff		
This charge is applicable only to intermediary traffic and is applied in addition to applicable											Rates		
### Simbling and/or interconnection changes. Simbling and/or interconnection changes.		N/A	\$0.0015	NA	NA	\$0.001096	NA	NA	NA	NA	NA		
INTEROPTICE TRANSPORT Common (Shared) Transport Per lile per mou N/A \$0.00004 \$0.000004 \$0.000005 \$0.000005 \$0.000005 \$0.000004 \$0.000012 \$0.000012 \$0.000006 \$0.000005 \$0.000006 \$0											İ		
Common (Shared) Transport Emberd Transport Embed Transport Embed													
Common (Shared) Transport per mile per mou N/A \$0,000018 \$0,0000018 \$0,0000018 \$0,0000018 \$0,0000018 \$0,0000018 \$0,0000018 \$0,000012 \$0,000018 \$0,0000													
Common (Shared) Transport Facilities Termination per moru NA \$0.00045 \$0.000472 \$0.000472 \$0.000472 \$0.000472 \$0.000472 \$0.0004672													
Interoffice Transport - Decidated - Wiley WG - per mile											\$0.00004		
Interoffice Transport - Dedicated - 2-Wire VG - per mile		N/A	\$0.00045	\$0.0005	\$0.0004152	\$0.000426	\$0.00047	\$0.0004281	\$0.00036	\$0.0004672	\$0.00036		
Interoffice Transport - Dedicated - 2-Wire VG - facilities termination per month UEA \$18.49 NA \$17.07 NA \$19.10 \$21.33 NA \$21.42 NRC - fadril UEA \$18.47 NA \$79.61 NA													
NRC - 1st											NA		
NRC - AddT											NA		
NRC - Disconnect Charge - 1st											NA		
NRC - Disconnect Charge - AddT											NA		
NRC - Incremental Charge - Manual Service Order - 1st											NA		
NRC - Incremental Charge - Manual Service Order - AddT UEA \$12,97 NA \$18,94 NA \$18,14 \$25,52 NA \$39,93											NA		
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st UEA \$12.97 NA NA NA \$8.06 \$11.34 NA NA NA NRC - Incremental Charge - Manual Service Order - Disconnect - 1st UDL \$27.37 NA NA \$8.06 \$11.34 NA NA NA NA NA NRC - Incremental Charge - Manual Service Order - Disconnect - 1st UDL \$27.37 NA NA NA \$8.06 \$11.34 NA NA NA NA NA NA NA N											NA		
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'											NA		
Interoffice Transport - Dedicated - DS0 - 58/64 KBPS UDL \$0.0329 NA \$0.0322 SA \$0.0323 SA \$0.03											NA		
Interoffice Transport - Dedicated - DS0 - per mile per month		UEA	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA		
Interoffice Transport - Dedicated - DS0 - facilities termination per month													
NRC - 1st											\$1.90		
NRC - Add' NRC - Disconnect Charge - 1st											\$38.37		
NRC - Disconnect Charge - 1st											TBD		
NRC - Disconnect Charge - Add*											TBD		
NRC - Incremental Charge - Manual Service Order - 1st											NA		
NRC - Incremental Charge - Manual Service Order - Add'											NA		
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st											NA		
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l											NA		
Interoffice Transport - Dedicated - DS1 - per mile per month											NA		
Interoffice Transport - Dedicated - DS1 - per mile per month		UDL	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA		
Interoffice Transport - Dedicated - DS1 - facilities termination per month													
NRC - 1st											\$23.00		
NRC - Add' NRC - Add' NRC - Disconnect Charge - 1st											\$90.00		
NRC - Disconnect Charge - 1st											\$100.49		
NRC - Disconnect Charge - Add' NRC - Incremental Charge - Manual Service Order - 1st USL \$20.42 NA NA \$18.94 NA \$18.14 \$25.52 NA \$39.63 NRC - Incremental Charge - Manual Service Order - Add' NRC - Incremental Charge - Manual Service Order - Disconnect - 1st USL \$27.37 NA \$18.94 NA \$18.14 \$25.52 NA \$39.63 NRC - Incremental Charge - Manual Service Order - Disconnect - 1st USL \$12.97 NA NA NA NA \$8.06 \$11.34 NA NA NRC - Incremental Charge - Manual Service Order - Disconnect - Add' USL \$12.97 NA NA NA NA \$8.06 \$11.34 NA NA NRC - Incremental Charge - Manual Service Order - Disconnect - Add' USL \$12.97 NA NA NA NA \$8.06 \$11.34 NA NA Interoffice Transport - Dedicated - DS3 - per mile per month UE3 NA NA NA NA NA NA \$15.02 \$175.00 \$40.00 Interoffice Transport - Dedicated - DS3 - facilities termination per month UE3 NA NA NA NA NA NA \$744.38 \$1,200.00 \$600.00 NA NA NA NA NA NA NA											\$100.49		
NRC - Incremental Charge - Manual Service Order - 1st USL \$27.37 NA \$18.94 NA \$18.14 \$25.52 NA \$39.63 NRC - Incremental Charge - Manual Service Order - Add'l USL \$27.37 NA \$18.94 NA \$18.14 \$25.52 NA \$39.63 NRC - Incremental Charge - Manual Service Order - Disconnect - 1st USL \$12.97 NA NA NA \$8.06 \$11.34 NA NA NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l USL \$12.97 NA NA NA NA NA NA Interoffice Transport - Dedicated - DS3 USL \$12.97 NA NA NA NA NA NA Interoffice Transport - Dedicated - DS3 USL NA											NA		
NRC - Incremental Charge - Manual Service Order - Add'l USL \$27.37 NA \$18.94 NA \$18.14 \$25.52 NA \$39.63 NRC - Incremental Charge - Manual Service Order - Disconnect - 1st USL \$12.97 NA NA NA \$8.06 \$11.34 NA NA NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l USL \$12.97 NA NA </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>NA</td>											NA		
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st USL \$12.97 NA NA NA \$8.06 \$11.34 NA NA NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l USL \$12.97 NA NA NA \$8.06 \$11.34 NA NA Interoffice Transport - Dedicated - DS3 Interoffice Transport - Dedicated - DS3 - per mile per month UE3 NA											NA		
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l USL \$12.97 NA NA NA \$8.06 \$11.34 NA NA Interoffice Transport - Dedicated - DS3 - per mile per month UE3 NA											NA		
Interoffice Transport - Dedicated - DS3 UE3 NA NA <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>NA</td>											NA		
Interoffice Transport - Dedicated - DS3 - per mile per month UE3 NA NA NA NA NA \$15.02 \$175.00 \$40.00 Interoffice Transport - Dedicated - DS3 - facilities termination per month UE3 NA NA NA NA NA NA NA \$744.38 \$1,200.00 \$600.00		USL	\$12.97	NA	NA	NA	\$8.06	\$11.34	NA	NA	NA		
Interoffice Transport - Dedicated - DS3 - facilities termination per month UE3 NA NA NA NA NA NA \$744.38 \$1,200.00 \$600.00			+		L				L				
											NA		
NRC - 1st UE3 NA NA NA NA S686,74 \$67.19 \$67.19											NA		
NRC - Add'I UE3 NA NA NA NA NA \$477.76 \$67.19 \$67.19											NA NA		

BELLSOUTH/e.spire RATES LOCAL INTERCONNECTION

						RATES BY STAT	ΓE			
NRC - Disconnect Charge - 1st	UE3	NA	NA	NA	NA	NA	\$125.56	NA	NA	NA
NRC - Disconnect Charge - Add'l	UE3	NA	NA	NA	NA	NA	\$118.79	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	UE3	NA	NA	NA	NA	NA	\$64.97	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	UE3	NA	NA	NA	NA	NA	\$64.97	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	UE3	NA	NA	NA	NA	NA	\$27.08	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	UE3	NA	NA	NA	NA	NA	\$27.08	NA	NA	NA
Local Channel - Dedicated										1
Local Channel - Dedicated - 2-Wire VG	N/A	\$14.61	NA	\$13.91	NA	\$14.94	\$17.83	NA	\$16.83	NA
NRC - 1st	N/A	\$494.65	NA	\$362.95	NA	\$347.49	\$487.62	NA	\$554.00	NA
NRC - Add'l	N/A	\$84.44	NA	\$62.40	NA	\$59.75	\$84.35	NA	\$88.58	NA
NRC - Disconnect Charge - 1st	N/A	\$77.81	NA	NA	NA	\$53.68	\$77.69	NA	NA	NA
NRC - Disconnect Charge - Add'l	N/A	\$7.63	NA	NA	NA	\$6.60	\$8.95	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	N/A	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	NA	\$43.75	NA
NRC - Incremental Charge - Manual Service Order - Add'l	N/A	\$18.73	NA	\$8.42	NA	\$8.06	\$11.34	NA	\$13.55	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	N/A	\$17.75	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
Local Channel - Dedicated - 4-Wire VG	N/A	\$15.77	NA	\$14.99	NA	\$16.21	\$19.03	NA	\$18.05	NA
NRC - 1st	N/A	\$502.43	NA	\$368.44	NA	\$352.75	\$495.25	NA	\$562.46	NA
NRC - Add'l	N/A	\$86.68	NA	\$64.05	NA	\$61.33	\$86.56	NA	\$91.57	NA
NRC - Disconnect Charge - 1st	N/A	\$78.71	NA	NA	NA	\$54.36	\$78.58	NA	NA	NA
NRC - Disconnect Charge - Add'l	N/A	\$8.53	NA	NA	NA	\$7.28	\$9.84	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	N/A	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	NA	\$43.75	NA
NRC - Incremental Charge - Manual Service Order - Add'l	N/A	\$18.73	NA	\$8.42	NA	\$8.06	\$11.34	NA	\$13.55	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	N/A	\$17.75	NA	NA	NA	\$11.40	\$16.05	NA	NA	NA
Local Channel - Dedicated - DS1	N/A	\$35.52	\$44.35	\$38.36	NA	\$43.80	\$38.91	NA	\$37.20	\$133.8
NRC - 1st	N/A	\$503.57	\$246.50	\$356.15	NA	\$348.56	\$494.83	NA	\$534.81	\$868.9
NRC - Add'l	N/A	\$442.84	\$230.49	\$312.89	NA	\$300.30	\$435.28	NA	\$462.81	\$486.8
NRC - Disconnect Charge - 1st	N/A	\$46.28	NA	NA	NA	\$24.15	\$46.85	NA	NA	NA
NRC - Disconnect Charge - Add'l	N/A	\$32.18	NA	NA	NA	\$21.31	\$33.02	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	N/A	\$61.95	NA	\$44.22	NA	\$42.34	\$59.58	NA	\$87.99	NA
NRC - Incremental Charge - Manual Service Order - Add'l	N/A	NA	NA	NA	NA	NA	NA	NA	\$3.11	NA
NRC - Incremental Charge - Manual Service Order - Disconnect	N/A	\$29.27	NA	NA	NA	\$19.48	\$27.41	NA	NA	NA
Local Channel - Dedicated - DS3	N/A	NA	NA	NA	NA	NA	\$533.33	NA	NA	NA
NRC - 1st	N/A	NA	NA	NA	NA	NA	\$526.67	NA	NA	NA
NRC - Add'l	N/A	NA	NA	NA	NA	NA	\$493.71	NA	NA	NA
NRC - Disconnect Charge - 1st	N/A	NA	NA	NA	NA	NA	\$42.41	NA	NA	NA
NRC - Disconnect Charge - Add'l	N/A	NA	NA	NA	NA	NA	\$40.87	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	N/A	NA	NA	NA	NA	NA	\$31.49	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	N/A	NA	NA	NA	NA	NA	\$31.49	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	N/A	NA	NA	NA	NA	NA	\$25.35	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	N/A	NA	NA	NA	NA	NA	\$25.35	NA	NA	NA

NOTES:

If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the parties upon request by either party.

Attachment 4 Physical Collocation

PHYSICAL COLLOCATION

1. SCOPE OF ATTACHMENT

- 1.1 <u>Scope of Attachment.</u> The rates, terms, and conditions contained within this Attachment shall only apply when e.spire is occupying the collocation space as a sole occupant or as a Host pursuant to Section 4.
- 1.2 <u>Right to Occupy.</u> Subject to Section 4 of this Attachment, BellSouth hereby grants to e.spire a right to occupy that certain area designated by BellSouth within a BellSouth Premises, of a size which is specified by e.spire and agreed to by BellSouth (hereinafter "Collocation Space").

Premises is defined as BellSouth's central offices and serving wire centers, as well as all buildings or similar structures owned or leased by BellSouth that house its network facilities, and all structures that house BellSouth facilities on public rights-of-way, including but not limited to vaults containing loop concentrators or similar structures.

To the extent this Attachment does not include all the necessary rates, terms and conditions for BellSouth Premises other than BellSouth Central Offices, the Parties will negotiate said rates, terms, and conditions in accordance with the Bona Fide Request process specified in Attachment 12 of this Agreement. Notwithstanding the foregoing, BellSouth shall consider in its designation for cageless collocation any unused space within the BellSouth Premises. The size specified by e.spire may contemplate a request for space sufficient to accommodate e.spire's growth within a two year period unless otherwise agreed to by the Parties.

- 1.2.1 <u>Space Reclamation.</u> In the event of space exhaust within a BellSouth Premises, e.spire may be required to release space to BellSouth to be allocated to other physical collocation applicants when 100% of the total amount of space is not being utilized by the end of the second year of operation.
- Use of Space. e.spire shall use the Collocation Space for the purposes of installing, maintaining and operating e.spire's equipment (to include testing and monitoring equipment) used or useful primarily to interconnect with BellSouth services and facilities, including access to unbundled network elements, for the provision of Telecommunications Services. Pursuant to Section 5 following, e.spire may at its option, place e.spire owned or leased entrance facilities to the Collocation Space. In addition to, and not in lieu of, interconnection to BellSouth services and facilities, e.spire may connect to other interconnectors within the designated BellSouth Premises (including to its other Virtual or Physical Collocation arrangements) through co-carrier cross connect facilities designated by e.spire pursuant to Section 5 following. The Collocation Space may be used for any purpose consistent with FCC Rules and Orders regarding use of the

- ILECs premises and generic state Commission Orders that are not in conflict therewith.
- 1.4 <u>Rates and Charges.</u> e.spire agrees to pay the rates and charges identified at Exhibit A attached hereto.

2. SPACE NOTIFICATION

- 2.1 <u>Availability of Space</u>. Upon submission of an application pursuant to Section <u>6</u>, BellSouth will permit e.spire to physically collocate, pursuant to the terms of this Attachment, at any BellSouth Premises, unless there is no space available due to space limitations or no space available due to technical infeasibility. BellSouth will respond to an application within ten (10) business days as to whether space is available or not available within a BellSouth Premises.
- 2.2 <u>Reporting.</u> Upon request from e.spire, BellSouth will provide a written report specifying the amount of Collocation Space available at Premises requested, the number of collocators present at the Premises, any modifications in the use of the space since the last report or the Premises requested and the measures BellSouth is taking to make additional space available for collocation arrangements.
 - 2.2.1 The request from e.spire must be written and must include the Premises and Common Language Location Identification (CLLI) code of the Premises. If applicable, information regarding central office premises and CLLI code is located in the National Exchange Carriers Association (NECA) Tariff FCC No. 4.
 - 2.2.2 BellSouth will respond to a request for such a report for a particular Premises location within ten (10) business days of receipt of such request. BellSouth will make best efforts to respond in ten (10) business days to such a request when the request includes up to and including five (5) Premises locations within the same state. The response time for requests of more than five (5) shall be negotiated between the Parties, but BellSouth shall use best efforts to respond within thirty (30) days. If BellSouth cannot meet the specified response times, BellSouth shall notify e.spire and inform e.spire of the time frame under which it can respond.
- 2.3 <u>Denial of Application.</u> After notifying e.spire that BellSouth has no available space in the requested Central Office ("Denial of Application"), BellSouth will allow e.spire, upon request, to tour the entire Central Office within ten (10) business days of such Denial of Application. In order to schedule said tour within ten (10) business days, the request for a tour of the Central Office must be received by BellSouth within five (5) business days of e.spire's receipt of the Denial of Application.
- 2.4 <u>Filing of Petition for Waiver.</u> Upon Denial of Application BellSouth will timely file a petition with the Commission pursuant to 47 U.S.C. § 251(c)(6), and the FCC's

rules. Such petition shall include detailed floor plans or diagrams of the BellSouth Premises.

- Waiting List. On a first come first served basis, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. BellSouth will notify the Telecommunications Carriers on the waiting list when space becomes available according to how much space becomes available and the position of Telecommunications Carrier on said waiting list. Upon request, BellSouth will advise e.spire as to its position on the list.
- 2.6 <u>Public Notification.</u> BellSouth will maintain on its Interconnection Services website a notification document that will indicate all Central Office Premises that are without available space. BellSouth shall update such document within ten (10) business days of the Denial of Application date. BellSouth will also post a document on its Interconnection Services website that contains a general notice where space has become available in a Central Office previously on the space exhaust list. BellSouth shall allocate said available space pursuant to the waiting list referenced in Section 2.5.
- 2.7 <u>State Agency Procedures.</u> Notwithstanding the foregoing, should any state regulatory agency impose a procedure different than procedures set forth in this section, that procedure shall supersede the requirements set forth herein, provided that such state agency procedures are not in conflict with FCC rules and Orders.

3. COLLOCATION OPTIONS

3.1 <u>Cageless</u>. BellSouth shall allow e.spire to collocate e.spire's equipment and facilities without requiring the construction of a cage or similar structure and without requiring the creation of a separate entrance to the Collocation Space. BellSouth shall allow e.spire to have direct access to its equipment and facilities but may require e.spire to use a central entrance to the BellSouth Premises. BellSouth shall make cageless collocation available in single bay increments pursuant to Section 7.

Except where e.spire's equipment requires special technical considerations (e.g., special cable racking, isolated ground plane), BellSouth shall assign cageless Collocation Space in conventional equipment rack lineups where technically feasible on a space-available basis, in accordance with the FCC's Rules. For equipment requiring special technical considerations, e.spire must provide the equipment layout, including spatial dimensions for such equipment pursuant to generic requirements contained in BellCore (Telcordia) GR-63-Core and shall be responsible for constructing all special technical requirements associated with such equipment pursuant to Section 6.5 following.

3.2 <u>Caged Enclosures</u>. BellSouth shall authorize the enclosure of e.spire's equipment and facilities at e.spire's option. e.spire, at its sole expense, must arrange with a BellSouth certified contractor to construct a collocation

arrangement enclosure in accordance with BellSouth's guidelines and specifications ("BellSouth Guidelines") provided that such BellSouth Guidelines are reasonable, nondiscriminatory, and consistent with applicable laws, rules and Orders. BellSouth will provide BellSouth Guidelines upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard enclosure specification, e.spire and e.spire's BellSouth certified contractor must comply with such applicable and enforceable building code requirements to the same extent BellSouth complies with such building code requirements. e.spire's BellSouth certified contractor shall be responsible for filing and receiving any and all necessary permits and/or licenses for such The Certified Vendor shall bill e.spire directly for all work performed for e.spire pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the Certified Vendor. e.spire must provide the local BellSouth building contact with two Access Keys used to enter the locked enclosure. Except in case of emergency, BellSouth will not access e.spire's locked enclosure prior to notifying e.spire.

- 3.2.1 BellSouth has the right to review e.spire's plans and specifications prior to allowing construction to start. BellSouth has the right to inspect the enclosure after construction to make sure it is designed and constructed according to BellSouth's Guidelines and to require e.spire to remove or correct at e.spire's cost any structure that does not meet these standards.
- 3.3 Shared Collocation. e.spire may allow other Telecommunications Carriers to sublease, license or otherwise share e.spire's caged collocation arrangement pursuant to terms and conditions agreed to by e.spire ("Host") and other Telecommunications Carriers ("Guests") and pursuant to this section except where BellSouth Premises is located within a leased space and BellSouth is not authorized by the lessor to offer such an option. e.spire shall provide written notification within ten (10) business days of execution of any such agreement. Such notification shall identify the Guest and shall include appropriate contact information for the Guest.
 - 3.3.1 A shared collocation cage is a caged collocation space shared by two or more competitive LECs pursuant to terms and conditions agreed to by the competitive LECs. In making shared cage arrangements available, BellSouth may not increase the cost of site preparation or nonrecurring charges above the cost for provisioning such a cage of similar dimensions and material to a single collocating party. In addition, the incumbent must prorate the charge for site conditioning and preparation undertaken by the incumbent to construct the shared collocation cage or condition the space for collocation use, regardless of how many carriers actually collocate in that cage, by determining the total charge for site preparation and allocating that charge to a collocating carrier based on the percentage of the total space utilized by that carrier. BellSouth must make shared collocation available in single-bay increments or their equivalent, i.e., a competing carrier can purchase space in increments small enough to collocate a single rack, or bay, of equipment.

- 3.3.2 e.spire shall be the sole interface and responsible party to BellSouth for the purpose of submitting applications for initial and additional equipment placements of Guest; for assessment of rates and charges contained within this Attachment; and for the purposes of ensuring that the safety and security requirements of this Attachment are fully complied with by the Guest, its employees and agents. The initial Guest application shall require the assessment of an Application Fee, as set forth in Exhibit A. Notwithstanding the foregoing, Guest may arrange directly with BellSouth for the provision of the interconnecting facilities between BellSouth and Guest and for the provisions of the services and access to unbundled network elements.
- 3.3.3 e.spire shall indemnify and hold harmless BellSouth from any and all claims, actions, causes of action, of whatever kind or nature arising out of the presence of e.spire's Guests in the Collocation Space, according to the terms of Section 19 of the General Terms and Conditions.

e.spire and e.spire's Guest(s) shall maintain, throughout the term of this Agreement, the insurance coverage required under Section 8 of this Attachment and shall name BellSouth as an additional insured in all such policies as set forth in Section 8 hereof. In the event e.spire's Guest(s) fails to maintain such insurance as specified and/or to name BellSouth as an additional insured, e.spire shall indemnify and hold harmless BellSouth from any and all claims, actions, and causes of action, of whatever kind or nature, arising out of the presence of e.spire's Guest(s) in the Collocation Space.

4. OCCUPANCY

- 4.1 <u>Commencement Date.</u> The "Commencement Date" shall be the day e.spire's equipment becomes operational as described in Section 4.2, following.
- 4.2 Occupancy. BellSouth will notify e.spire in writing that the Collocation Space is e.spire must place operational telecommunications ready for occupancy. equipment in the Collocation Space and connect with BellSouth's network within one hundred eighty (180) days after receipt of such notice. e.spire must notify BellSouth in writing that collocation equipment installation is complete and is operational with BellSouth's network. BellSouth may, at its option, not accept orders for interconnected service until receipt of such notice. If e.spire fails to place operational telecommunications equipment in the Collocation Space within 180 calendar days and such failure continues for a period of thirty (30) days after receipt of written notice from BellSouth, then and in that event e.spire's right to occupy the Collocation Space terminates and BellSouth shall have no further obligations to e.spire with respect to said Collocation Space. Termination of e.spire's rights to the Collocation Space pursuant to this paragraph shall not operate to release e.spire from its obligation to reimburse BellSouth for all costs reasonably incurred by BellSouth in preparing the Collocation Space, but rather such obligation shall survive this Attachment. For purposes of this paragraph, e.spire's telecommunications equipment will be deemed operational when crossconnected to BellSouth's network for the purpose of service provision.

Version 7/7/00

Termination. Except where otherwise agreed to by the Parties, e.spire may 4.3 terminate occupancy in a particular Collocation Space upon thirty (30) days' prior written notice to BellSouth. Upon termination of such occupancy, e.spire at its expense shall remove its equipment and other property from the Collocation Space. e.spire shall have thirty (30) days from the termination date to complete such removal, including the removal of all equipment and facilities of e.spire's Guests; provided, however, that e.spire shall continue payment of monthly fees to BellSouth until such date as e.spire has fully vacated the Collocation Space. Should e.spire fail to vacate the Collocation Space within thirty (30) days from the termination date, BellSouth shall have the right to remove the equipment and other property of e.spire at e.spire's expense and with no liability for damage or injury to e.spire's property unless caused by the gross negligence or intentional misconduct of BellSouth. Upon expiration of e.spire's Collocation Arrangement(s) pursuant to this Attachment, e.spire shall surrender the Collocation Space to BellSouth in the same condition as when first occupied by the e.spire except for ordinary wear and tear. e.spire shall be responsible for the cost of removing any enclosure, together with all support structures (e.g., racking, conduits), of an Adjacent Collocation arrangement at the termination of occupancy and restoring the grounds to their original condition.

5. USE OF COLLOCATION SPACE

5.1 Equipment Type. BellSouth shall permit the collocation of any type of equipment used or useful for interconnection or access to unbundled network elements. Whenever BellSouth objects to collocation of equipment by e.spire for the purposes within the scope of section 251(c)(6) of the Act, BellSouth shall prove to the state commission that the equipment will not be actually used by e.spire for the purpose of obtaining interconnection or access to unbundled network elements.

BellSouth may not object to the collocation of equipment on the grounds that the equipment does not comply with safety or engineering standards that are more stringent than the safety or engineering standards than BellSouth applies to its own equipment. BellSouth may not object to the collocation of equipment on the ground that the equipment fails to comply with National Equipment and Building Specifications performance standards.

If BellSouth denies collocation of e.spire's equipment, citing safety standards, BellSouth must provide to e.spire within five (5) business days of the denial, a list of all equipment that BellSouth locates within the central office premises in question, together with an affidavit attesting that all of that equipment meets or exceeds the safety standard that BellSouth contends e.spire's equipment fails to meet.

Equipment used for interconnection and access to unbundled network elements includes, but is not limited to: (1) Transmission equipment including, but not limited to, optical terminating equipment and multiplexers, (2) Equipment being collocated to terminate basic transmission facilities pursuant to §§ 64.1401 and 64.1402 of Title 47 of the Code of Federal Regulations as of August 1, 1996, and

(3) Digital subscriber line access multiplexers, routers, asynchronous transfer mode multiplexers, and remote switching modules.

Nothing in this section requires BellSouth to permit collocation of equipment used solely for switching or solely to provide enhanced services; provided, however, that BellSouth may not place any limitations on the ability of e.spire to use all the features, functions, and capabilities of equipment collocated pursuant to the above, including, but not limited to, switching and routing features and functions and enhanced services functionalities.

- 5.1.1 e.spire shall not use the Collocation Space for marketing purposes nor shall it place any identifying signs or markings in the area surrounding the Collocation Space or on the grounds of the BellSouth Premises.
- 5.1.2 e.spire shall place a plaque or other identification affixed to e.spire's equipment necessary to identify e.spire's equipment, including a list of emergency contacts with telephone numbers.
- 5.2 Entrance Facilities. e.spire may elect to place e.spire owned or e.spire leased fiber entrance facilities into the Collocation Space. BellSouth will designate the point of interconnection in close proximity to the Premises building housing the Collocation Space, such as an entrance manhole or a cable vault which are physically accessible by both parties. e.spire will provide and place cable at the point of interconnection of sufficient length to be pulled through conduit and into the splice location. e.spire will provide and install a sufficient length of fire retardant riser cable, to which the entrance cable will be spliced, which will extend from the splice location to the e.spire's equipment in the Collocation Space. In the event e.spire utilizes a non-metallic, riser-type entrance facility, a splice will not be required. e.spire must contact BellSouth for instructions prior to placing the entrance facility cable in the manhole. e.spire is responsible for maintenance of the entrance facilities.
 - 5.2.1 Dual Entrance. BellSouth will provide at least two interconnection points at each BellSouth Premises where there are at least two such interconnection points available and where capacity exists. Upon receipt of a request for Physical Collocation under this Attachment, BellSouth shall provide e.spire with information regarding BellSouth's capacity to accommodate dual entrance facilities. If conduit in the serving manhole(s) is available and is not reserved for another purpose for utilization within 12 months of the receipt of an application for collocation. BellSouth will make the requested conduit space available for installing a second entrance facility to e.spire's arrangement. The location of the serving manhole(s) will be determined at the reasonable and nondiscriminatory discretion of BellSouth. Where dual entrance is not available due to lack of capacity, BellSouth will so state in the Application BellSouth shall not deny a collocation application solely for the reason that dual entrance facilities are not available.

- 5.2.2 <u>Shared Use.</u> e.spire may utilize spare capacity on an existing interconnector entrance facility for the purpose of providing an entrance facility to another e.spire collocation arrangement within the same BellSouth Premises. e.spire must arrange with BellSouth for BellSouth to splice the utilized entrance facility capacity to e.spire provided riser cable.
- 5.3 Splicing in the Entrance Manhole. Should e.spire request a splice to occur in the entrance manhole(s), BellSouth shall grant such a request, provided that BellSouth will not unreasonably withhold approval of requests to make such a splice where technically feasible. When the request for a splice is granted to e.spire by BellSouth, e.spire shall ensure its employees or agents entering and/or performing work in the entrance manhole(s) are trained and comply with BellSouth procedures and OSHA requirements regarding access to manholes and that BellSouth personnel are notified and present for all entrances and work performed in the entrance manhole(s). Manhole covers shall be properly closed and secured at the conclusion of entry and/or work. Advance notification to BellSouth shall occur at a minimum of 48 hours prior to desired entry for normal work activities and at a minimum of 2 hours prior to desired entry in an out of service condition.
- 5.4 Demarcation Point. BellSouth will designate the point(s) of interconnection between e.spire's equipment and/or network and BellSouth's network. BellSouth shall designate the closest demarcation point available. Each party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For 2-wire and 4-wire connections to BellSouth's network, the demarcation point shall be a common block on the BellSouth designated conventional distributing frame. e.spire shall be responsible for providing, and e.spire's BellSouth Certified Vendor shall be responsible for installing and properly labeling/stenciling, the common block, and necessary cabling. For all other terminations BellSouth shall designate a demarcation point on a per arrangement basis. e.spire or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Subsection 5.5, following, and may self-provision cross-connects that may be required within the collocation space to activate service requests. At e.spire's option, a Point of Termination (POT) bay or frame may be placed in the Collocation Space.
- 5.5 <u>e.spire's Equipment and Facilities.</u> e.spire, or if required by this Attachment, e.spire's BellSouth Certified Vendor, is solely responsible for the design, engineering, installation, testing, provisioning, monitoring, maintenance and repair of the equipment and facilities used by e.spire. Such equipment and facilities may include but are not limited to cable(s); equipment; and point of termination connections.
- 5.6 <u>Co-Carrier Cross-connect.</u> In addition to, and not in lieu of, obtaining interconnection with, or access to, BellSouth telecommunications services, unbundled network elements, and facilities, e.spire may directly connect to other Interconnectors within the designated BellSouth Premises (including to its other virtual or physical collocated arrangements) through facilities owned by e.spire or through BellSouth facilities designated by e.spire, at e.spire's option. Such

connections to other carriers may be made using either optical or electrical facilities. e.spire may deploy such optical or electrical connections directly between its own facilities and the facilities of other interconnector(s) without being routed through BellSouth equipment. BellSouth shall permit such connections with third-party carriers at any time during the term of this agreement.

- 5.6.1 If e.spire requests a co-Carrier cross-connect after the initial installation, e.spire must submit an application with a Subsequent Application Fee. e.spire must use a Certified Vendor to place the co-Carrier cross connect, except in cases where the e.spire equipment and the equipment of the other interconnector are located within contiguous collocation spaces. In cases where e.spire's equipment and the equipment of the other interconnector are located in contiguous collocation spaces, e.spire will have the option to deploy the co-Carrier cross connects between the sets of equipment. Where cable support structure exists for such connection there will be a recurring charge per linear foot of support structure used. When cable support structures do not exist and must be constructed a non-recurring charge for the individual case will be assessed.
- Right to Access Collocation Space. From time to time BellSouth may require access to the Collocation Space. BellSouth retains the right to access such space for the purpose of making BellSouth equipment and building modifications (e.g., running, altering or removing racking, ducts, electrical wiring, HVAC, and cables). BellSouth will give reasonable notice to e.spire when access to the Collocation Space is required. e.spire may elect to be present whenever BellSouth performs work in the Collocation Space. The Parties agree that e.spire will not bear any of the expense associated with this work.
- Access. Pursuant to Section 11, e.spire shall have unescorted access to the Collocation Space twenty-four (24) hours a day, seven (7) days a week. e.spire agrees to provide the name and one of the following: social security number, date of birth, or drivers license number of each employee, contractor, or agent provided with Access Keys or cards ("Access Keys") prior to the issuance of said Access Keys. Access Keys shall not be duplicated under any circumstances. e.spire agrees to be responsible for all Access Keys and for the return of all said Access Keys in the possession of e.spire employees, contractors, Guests or agents after termination of the employment relationship, contractual obligation with e.spire or upon the termination of this Attachment or the termination of occupancy of an individual collocation arrangement.
 - 5.8.1 Lost or Stolen Access Keys. e.spire shall notify BellSouth in writing immediately in the case of lost or stolen Access Keys. e.spire will reimburse BellSouth for the cost of replacing Access Key(s) lost or stolen. Should it become necessary for BellSouth to re-key buildings as a result of a lost Access Key(s) or for failure to return an Access Key(s), e.spire shall pay for all reasonable costs directly associated with the re-keying.

5.9 Interference or Impairment. Notwithstanding any other provisions of this Attachment, equipment and facilities placed in the Collocation Space shall not interfere with or impair service provided by BellSouth or by any other Interconnector located in the Premises; shall not endanger or damage the facilities of BellSouth or of any other Interconnector, the Collocation Space, or the Premises; shall not compromise the privacy of any communications carried in, from, or through the Premises; and shall not create an unreasonable risk of injury or death to any individual or to the public.

BellSouth shall not treat equipment deployed by e.spire as interfering with or impairing service provided by BellSouth or another interconnector solely on the basis that such equipment is of a different type that has not previously been utilized in a BellSouth Premise.

If BellSouth determines in a reasonable and nondiscriminatory manner that any equipment or facilities of violates the provisions of this paragraph, BellSouth shall give written notice to e.spire, which notice shall direct e.spire to cure the violation within forty-eight (48) hours of e.spire's actual receipt of written notice or, at a minimum, to commence curative measures within twenty-four (24) hours and to exercise reasonable diligence to complete such measures as soon as possible thereafter. After receipt of the notice, the Parties agree to consult immediately and, if necessary, to inspect the arrangement. The Parties will act in good faith and in a cooperative manner to determine or isolate the source of the interference or impairment.

- Personalty and its Removal. Subject to requirements of this Attachment, e.spire may place or install in or on the Collocation Space such facilities and equipment, including storage for and spare equipment, as it deems desirable for the conduct of business; provided that such equipment is telecommunications equipment, does not violate floor loading requirements, imposes or could impose or contains or could contain environmental conditions or hazards. Personal property, facilities and equipment placed by e.spire in the Collocation Space shall not become a part of the Collocation Space, even if nailed, screwed or otherwise fastened to the Collocation Space, but shall retain their status as personalty and may be removed by e.spire at any time. Any damage caused to the Collocation Space by e.spire's employees, agents or representatives during the removal of such property shall be promptly repaired by e.spire at its expense.
- 5.11 <u>Alterations.</u> In no case shall e.spire or any person acting on behalf of e.spire make any rearrangement, modification, improvement, addition, repair, or other alteration to the Collocation Space or the BellSouth Central Office without the written consent of BellSouth, which consent shall not be unreasonably withheld. The cost of any such specialized alterations shall be paid by e.spire.
- 5.12 <u>Janitorial Service.</u> e.spire is responsible for keeping its Caged Collocation Space free of debris and in good order. e.spire may elect to contract with a BellSouth certified vendor for the general upkeep and cleaning of the Caged Collocation Space. e.spire shall arrange directly with the certified vendor for such janitorial services. BellSouth shall provide a list of such contractors on a site-specific basis upon request.

6. ORDERING AND PREPARATION OF COLLOCATION SPACE

- 6.1 <u>Application for Space.</u> e.spire shall submit an application document when e.spire, or e.spire's Guest(s), as defined in Section 3.3, desires to request or modify the use of the Collocation Space.
 - 6.1.1 <u>Initial Application.</u> For e.spire's or e.spire's Guest(s) initial equipment placement, e.spire shall submit to BellSouth a complete and accurate Application and Inquiry document (Bona Fide Application), together with payment of the Application Fee as stated in Exhibit A. The Bona Fide Application shall contain a detailed description and schematic drawing of the equipment to be placed in e.spire's Collocation Space(s) and an estimate of the amount of square footage required.
 - Subsequent Application Fee. In the event e.spire or e.spire's Guest(s) 6.1.2 desire to modify the use of the Collocation Space, e.spire shall complete an Application document detailing all information regarding the modification to the Collocation Space together with payment of the minimum Subsequent Application Fee as stated in Exhibit A. minimum Subsequent Application Fee shall be considered a partial payment of the applicable Subsequent Application Fee, which shall be calculated as set forth below. BellSouth shall determine what modifications, if any, to the Premises are required to accommodate the change requested by e.spire in the Application. Such necessary modifications to the Premises may include but are not limited to, floor loading changes, changes necessary to meet HVAC requirements, changes to power plant requirements, and equipment additions. The fee paid by e.spire for its request to modify the use of the Collocation Space shall be dependent upon the modification requested. subsequent application does not require provisioning or construction work by BellSouth, no Subsequent Application Fee will be required and the pre-paid fee shall be refunded to e.spire. The fee for an application where the modification requested has limited effect (e.g., does not require capital expenditure by BellSouth) shall be the Subsequent Application Fee as set forth in Exhibit A. All other modifications shall require a Subsequent Application Fee assessed at the applicable application fee. In the event such modifications require the assessment of a full Application Fee as set forth in Exhibit A, the outstanding balance shall be due by e.spire within thirty (30) calendar days following e.spire's receipt of a bill or invoice from BellSouth.
- 6.2 Application Response. In addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond within ten (10) business days of receipt of an Application whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a comprehensive written response within thirty (30) business days of receipt of a complete application. When multiple applications are submitted within a fifteen (15) business day window, BellSouth will respond to the applications as soon as

possible on a non-discriminatory basis, but no later than the following: within thirty (30) business days for applications 1-5; within thirty-six (36) business days for applications 6-10; within forty-two (42) business days for applications 11-15; within forty-five (45) business days for more than 15 applications.

The Application Response will detail whether the amount of space requested is available or if the amount of space requested is not available, the amount of space that is available. The response will also include the configuration of the space.

When BellSouth's response includes an amount of space less than that requested by e.spire or differently configured, e.spire must amend its application to reflect the actual space available prior to submitting a Bona Fide Firm Order. Such amendment by e.spire will not require a supplemental application fee.

- 6.3 Bona Fide Firm Order. e.spire shall indicate its intent to proceed with equipment installation in a BellSouth Central Office by submitting a Bona Fide Firm Order to A Bona Fide Firm Order requires e.spire to complete the BellSouth. Application/Inquiry process described in Subsection 6.1, preceding, and submit the Expanded Interconnection Bona Fide Firm Order document (BSTEI-1P-F) indicating acceptance of the written application response provided by BellSouth ("Bona Fide Firm Order") and all appropriate fees. The Bona Fide Firm Order must be received by BellSouth no later than thirty (30) calendar days after BellSouth's response to e.spire's Application/Inquiry. If e.spire makes changes to its application in light of BellSouth's written Application Response, BellSouth will be required to re-evaluate and respond to the change(s). In this event, BellSouth's provisioning interval will not start until the re-evaluation and response to the change(s) is complete and the Bona Fide Firm Order is received by BellSouth and all appropriate fees and duties have been executed. If BellSouth needs to reevaluate e.spire's application as a result of changes requested by e.spire to e.spire's original application, then BellSouth will charge e.spire a fee based upon the additional engineering hours required to do the reassessment.
 - 6.3.1 BellSouth will establish a firm order date, per request, based upon the date BellSouth is in receipt of a Bona Fide Firm Order. BellSouth will acknowledge the receipt of e.spire's Bona Fide Firm Order within five (5) business days of receipt indicating that the Bona Fide Firm Order has been received. A BellSouth response to a Bona Fide Firm Order will include a Firm Order Confirmation containing the firm order date.
 - 6.3.2 BellSouth will permit one accompanied site visit to e.spire's designated collocation arrangement location after receipt of the Bona Fide Firm Order and prior to completing the Security Training requirements without charge to e.spire.
 - 6.3.3 Space preparation for the Collocation Space will not begin until BellSouth receives the Bona Fide Firm Order and all applicable fees.
 - 6.3.4 e.spire must submit to BellSouth the completed Access Control Request Form (RF-2906-C) for all employees or agents requiring access to the

BellSouth Central Office a minimum of thirty (30) calendar days prior to the date e.spire desires access to the Collocation Space.

- 6.4 Construction and Provisioning Intervals. BellSouth shall complete construction and provisioning intervals per request on an individual case basis. Excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will use best efforts to complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of 90 business days from receipt of a complete and accurate Bona Fide Firm Order. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will use best efforts to complete construction of all other collocation space ("extraordinary conditions") within 130 business days of the receipt of a complete and accurate Bona Fide Firm Order. Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement.
 - 6.4.1 <u>Joint Planning Meeting.</u> Unless otherwise agreed to by the Parties, a joint planning meeting or other method of joint planning between BellSouth and e.spire will commence within a maximum of fifteen (15) business days from BellSouth's receipt of a Bona Fide Firm Order and the payment of the agreed upon fees. At such meeting, the Parties will agree to the preliminary design of the Collocation Space and the equipment configuration requirements as reflected in the Application and affirmed in the Bona Fide Firm Order. The Collocation Space Completion time period will be provided to e.spire during the joint planning meeting or as soon as possible thereafter. BellSouth will complete all design work following the joint planning meeting.
 - 6.4.2 <u>Permits.</u> Each Party or its agents will diligently pursue filing for the permits required for the scope of work to be performed by that Party or its agents within seven (7) business days of the completion of finalized construction designs and specifications.
 - 6.4.3 Acceptance Walk Through. e.spire and BellSouth will complete an acceptance walk through of each Collocation Space requested from BellSouth by e.spire. BellSouth will correct any deviations to e.spire's original or jointly amended requirements within five (5) business days after the walk through, unless the Parties jointly agree upon a different time frame.
- 6.5 <u>Use of Certified Vendor.</u> e.spire shall select a vendor which has been approved as a BellSouth Certified Vendor to perform all engineering and installation work required in the Collocation Space, or utilize its own employees to perform such work provided that e.spire has been certified by BellSouth to perform such work. In some cases, e.spire must select separate BellSouth Certified Vendors for transmission equipment, switching equipment and power equipment. BellSouth

shall provide e.spire with a list of Certified Vendors and a statement of the criteria to qualify Certified Vendor, upon request. BellSouth shall not unreasonably withhold approval of any contractor proposed by e.spire that meets the standard BellSouth criteria. The Certified Vendor(s) shall be responsible for installing e.spire's equipment and components, installing co-carrier cross connects, extending power cabling to the BellSouth power distribution frame, performing operational tests after installation is complete, and notifying BellSouth's equipment engineers and e.spire upon successful completion of installation. The Certified Vendor shall bill e.spire directly for all work performed for e.spire pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the Certified Vendor. BellSouth shall consider certifying e.spire or any vendor proposed by e.spire.

- Alarm and Monitoring. BellSouth shall place environmental alarms in the Central Office for the protection of BellSouth equipment and facilities. e.spire shall be responsible for placement, monitoring and removal of environmental and equipment alarms used to service e.spire's Collocation Space. Upon request, BellSouth will provide e.spire with applicable tariffed services(s) to facilitate remote monitoring of collocated equipment by e.spire. Both parties shall use best efforts to notify the other of any verified environmental hazard known to that party. The parties agree to utilize and adhere to the Environmental Hazard Guidelines identified as Exhibit B attached hereto.
- 6.7 <u>Basic Telephone Service.</u> Upon request of e.spire, BellSouth will provide basic telephone service to the Collocation Space under the rates, terms and conditions of the current tariff offering for the service requested.
- 6.8 Space Preparation. BellSouth shall pro-rate the costs of any renovation or upgrade to Central Office space or support mechanisms which is required to accommodate Physical Collocation. e.spire's pro rated share will be calculated by multiplying such cost by a percentage equal to the amount of square footage occupied by e.spire divided by the total Central Office square footage receiving renovation or upgrade. For this section, support mechanisms provided by BellSouth may include, but not be limited to heating/ventilation/air conditioning (HVAC) equipment, HVAC duct work, cable support structure, fire wall(s), mechanical upgrade, asbestos abatement, or ground plane addition. Such renovation or upgrade will be evaluated and the charges assessed on a per Central Office basis. BellSouth will reimburse e.spire in an amount equal to e.spire reasonable, demonstrative and mitigated expenditures incurred as a direct result of delays to the completion and turnover dates caused by BellSouth.
- 6.9 <u>Virtual Collocation Transition.</u> BellSouth offers Virtual Collocation pursuant to the rates, terms and conditions set forth in its F.C.C. Tariff No. 1. For the interconnection to BellSouth's network and access to BellSouth unbundled network elements, e.spire may purchase 2-wire and 4-wire Cross-Connects as set forth in Exhibit A, and e.spire may place within its Virtual Collocation arrangements the telecommunications equipment set forth in Section 5.1.

In the event Physical Collocation Space was previously denied at a location due to technical reasons or space limitations, and that Physical Collocation Space

has subsequently become available, e.spire may transition its Virtual Collocation arrangement to a Physical Collocation arrangement and pay the appropriate non-recurring fees for Physical Collocation and for the rearrangement or reconfiguration of services terminated in the Virtual Collocation arrangement.

In the event that BellSouth knows when additional space for Physical Collocation may become available at the location requested by e.spire, such information will be provided to e.spire in BellSouth's written denial of Physical Collocation. To the extent that (i) Physical Collocation Space becomes available to e.spire within one-hundred and eighty (180) days of BellSouth's written denial of e.spire's request for Physical Collocation, and (ii) e.spire was not informed in the written denial that Physical Collocation Space would become available within such one-hundred and one-hundred and eighty (180) days, then e.spire may transition its Virtual Collocation arrangement to a Physical Collocation arrangement and will receive a credit for any nonrecurring charges previously paid for such Virtual Collocation.

- 6.10 <u>Cancellation.</u> If, at anytime, e.spire cancels its order for the Collocation Space(s), e.spire will reimburse BellSouth for any expenses incurred up to the date that written notice of the cancellation is received. In no event will the level of reimbursement under this paragraph exceed the maximum amount e.spire would have otherwise paid for work undertaken by BellSouth if no cancellation of the order had occurred.
- 6.11 <u>Licenses.</u> e.spire, at its own expense, will be solely responsible for obtaining from governmental authorities, and any other appropriate agency, entity, or person, all rights, privileges, and licenses necessary or required to operate as a provider of telecommunications services to the public or to occupy the Collocation Space.

7. RATES AND CHARGES

- Non-recurring Fees. In addition to the Application Fee referenced in Section 6, preceding, e.spire shall remit payment of a Cable Installation Fee and one-half (1/2) of the estimated Space Preparation Fee, as applicable, coincident with submission of a Bona Fide Firm Order. The outstanding balance of the actual Space Preparation Fee shall be due thirty (30) calendar days following e.spire's receipt of a bill or invoice from BellSouth. Once the installation of the initial equipment arrangement is complete, a subsequent application fee may apply, as described in Subsection 7.4, when e.spire requests a modification to the arrangement.
- 7.2 <u>Documentation.</u> BellSouth shall provide documentation to establish the actual Space Preparation Fee. The Space Preparation Fee will be pro-rated as prescribed in Section 6, preceding.
- 7.3 <u>Cable Installation.</u> Cable Installation Fee(s) are assessed per entrance facility placed.

- 7.4 Floor Space. The floor space charge includes reasonable charges for lighting. heat, air conditioning, ventilation and other allocated expenses associated with maintenance of the Central Office but does not include amperage necessary to power e.spire's equipment. When the Collocation Space is enclosed, e.spire shall pay floor space charges based upon the number of square feet so enclosed. When the Collocation Space is not enclosed, e.spire shall pay floor space charges based upon the following floor space calculation: [(depth of the equipment lineup in which the rack is placed) + (0.5 x maintenance aisle depth) + (0.5 x wiring aisle depth)] X (width of rack and spacers)]. For purposes of this calculation, the depth of the equipment lineup shall consider the footprint of equipment racks plus any equipment overhang. BellSouth will assign unenclosed Collocation Space in conventional equipment rack lineups where feasible. In the event e.spire's collocated equipment requires special cable racking, isolated grounding or other treatment which prevents placement within conventional equipment rack lineups, e.spire shall be required to request an amount of floor space sufficient to accommodate the total equipment arrangement. Floor space charges are due beginning with the date on which BellSouth releases the Collocation Space for occupancy or on the date e.spire first occupies the Collocation Space, whichever is sooner.
- 7.5 <u>Power.</u> BellSouth shall supply 48 Volt (-48V) DC power for e.spire's Collocation Space within the BellSouth Premises and shall make available AC power at e.spire's option for Adjacent Arrangement collocation.
 - 7.5.1 Charges for -48V DC power will be assessed per ampere per month based upon the certified vendor engineered and installed power feed fused ampere capacity. Rates include redundant feeder fuse positions (A&B) and cable rack to e.spire's equipment or space enclosure. When obtaining power from a BellSouth Battery Distribution Fuse Bay, fuses and power cables (A&B) must be engineered (sized), and installed by e.spire's certified vendor. When obtaining power from a BellSouth Power Board, power cables (A&B) must be engineered (sized), and installed by e.spire's certified power vendor. e.spire's certified vendor must also provide a copy of the engineering power specification prior to the Commencement Date. In the event BellSouth shall be required to construct additional DC power plant or upgrade the existing DC power plant in a Central Office as a result of e.spire's request to collocate in that Central Office ("Power Plant Construction"), e.spire shall pay its pro-rata share of costs associated with the Power Plant Construction. determination of whether Power Plant Construction is necessary shall be within BellSouth's sole, but reasonable, discretion. BellSouth shall comply with all BellCore (Telcordia) and ANSI Standards regarding power cabling, including BellCore (Telcordia) Network Equipment Building System (NEBS) Standard GR-63-CORE. BellSouth will notify e.spire of the need for the Power Plant Construction and will estimate the costs associated with the Power Plant Construction if BellSouth were to perform the Power Plant Construction. The costs of power plant construction shall be pro-rated and shared among all who benefit from that construction. e.spire shall pay BellSouth one-half of its prorata share of the estimated Power Plant Construction costs prior to commencement

of the work. e.spire shall pay BellSouth the balance due (actual cost less one-half of the estimated cost) within thirty (30) days of completion of the Power Plant Construction. e.spire has the option to perform the Power Plant Construction itself; provided, however, that such work shall be performed by a BellSouth certified contractor and such contractor shall comply with BellSouth's guidelines and specifications. Where the Power Plant Construction results in construction of a new power plant room, upon termination of this Attachment e.spire shall have the right to remove its equipment from the power plant room, but shall otherwise leave the room intact. Where the Power Plant Construction results in an upgrade to BellSouth's existing power plant, upon termination of this Attachment, such upgrades shall become the property of BellSouth.

- 7.5.2 Charges for AC power will be assessed per breaker ampere per month based upon the certified vendor engineered and installed power feed fused ampere capacity. Rates include the provision of commercial and standby AC power. When obtaining power from a BellSouth Service Panel, fuses and power cables must be engineered (sized), and installed by e.spire's certified vendor. e.spire's certified vendor must also provide a copy of the engineering power specification prior to the Commencement Date. Charges for AC power shall be assessed pursuant to the rates specified in Exhibit A. AC power voltage and phase ratings shall be determined on a per location basis.
- 7.6 Security Escort. A security escort will be required whenever e.spire or its approved agent desires access to the entrance manhole or must have access to the Central Office Premises after the one accompanied site visit allowed pursuant to subsection 6.3.2 prior to completing BellSouth's Security Training requirements and/or prior to Space Acceptance. Rates for a security escort are assessed in one-half (1/2) hour increments according to the schedule appended hereto as Exhibit A.
- 7.8 Rate "True-Up." The Parties agree that the prices reflected as interim herein shall be "trued-up" (up or down) based on final prices either determined by further agreement or by final order, including any appeals, in a proceeding involving BellSouth before the regulatory authority for the state in which the services are being performed or any other body having jurisdiction over this agreement (hereinafter "Commission"). Under the "true-up" process, the interim price for each service shall be multiplied by the volume of that service purchased to arrive at the total interim amount paid for that service ("Total Interim Price"). The final price for that service shall be multiplied by the volume purchased to arrive at the total final amount due ("Total Final Price"). The Total Interim Price shall be compared with the Total Final Price. If the Total Final Price is more than the Total Interim Price, e.spire shall pay the difference to BellSouth. If the Total Final Price is less than the Total Interim Price, BellSouth shall pay the difference to e.spire. Each party shall keep its own records upon which a "true-up" can be based and any final payment from one party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of

- such "true-up," the Parties agree that the Commission shall be called upon to resolve such differences.
- 7.9 Other. If no rate is identified in the contract, the rate for the specific service or function will be negotiated by the parties upon request by either party. Payment of all other charges under this Attachment shall be due thirty (30) days after receipt of the bill (payment due date). e.spire will pay a late payment charge of one and one-half percent (1-1/2%) assessed monthly on any balance which remains unpaid after the payment due date.

8. INSURANCE

- 8.1 e.spire shall, at its sole cost and expense, procure, maintain, and keep in force insurance as specified in this Article VI and underwritten by insurance companies licensed to do business in the states applicable under this Attachment and having a BEST Insurance Rating of B ++ X (B ++ ten).
- 8.2 e.spire shall maintain the following specific coverage:
 - 8.2.1 Commercial General Liability coverage in the amount of ten million dollars (\$10,000,000.00) or a combination of Commercial General Liability and Excess/Umbrella coverage totaling not less than ten million dollars (\$10,000,000.00). BellSouth shall be named as an ADDITIONAL INSURED on ALL applicable policies as specified herein.
 - 8.2.2 Statutory Workers Compensation coverage and Employers Liability coverage in the amount of one hundred thousand dollars (\$100,000.00) each accident, one hundred thousand dollars (\$100,000.00) each employee by disease, and five hundred thousand dollars (\$500,000.00) policy limit by disease.
 - 8.2.3 e.spire may elect to purchase business interruption and contingent business interruption insurance, having been advised that BellSouth assumes no liability for loss of profit or revenues should an interruption of service occur.
- 8.3 All policies purchased by e.spire shall be deemed to be primary and not contributing to or in excess of any similar coverage purchased by BellSouth. All insurance must be in effect on or before the date equipment is delivered to BellSouth's Central Office and shall remain in effect for the term of this Attachment or until all e.spire's property has been removed from BellSouth's Central Office, whichever period is longer. If e.spire fails to maintain required coverage, BellSouth may pay the premiums thereon and seek reimbursement of same from e.spire.
- 8.4 e.spire shall submit certificates of insurance reflecting the coverage required pursuant to this Section a minimum of ten (10) days prior to the commencement of any work in the Collocation Space. Failure to meet this interval may result in construction and equipment installation delays. e.spire shall arrange for BellSouth to receive thirty (30) days advance notice of cancellation from e.spire's

insurance company. e.spire shall forward a certificate of insurance and notice of cancellation to BellSouth at the following address:

BellSouth Telecommunications, Inc. Attn.: Risk Management Coordinator 600 N. 19th Street, 18B3 Birmingham, Alabama 35203

9. MECHANICS LIENS

9.1 If any mechanics lien or other liens shall be filed against property of either party (BellSouth or e.spire), or any improvement thereon by reason of or arising out of any labor or materials furnished or alleged to have been furnished or to be furnished to or for the other party or by reason of any changes, or additions to said property made at the request or under the direction of the other party, the other party directing or requesting those changes shall, within thirty (30) days after receipt of written notice from the party against whose property said lien has been filed, either pay such lien or cause the same to be bonded off the affected property in the manner provided by law. The party causing said lien to be placed against the property of the other shall also defend, at its sole cost and expense, on behalf of the other, any action, suit or proceeding which may be brought for the enforcement of such liens and shall pay any damage and discharge any judgment entered thereon.

10. INSPECTIONS

10.1 BellSouth shall conduct an inspection of e.spire's equipment and facilities in the Collocation Space(s) prior to the activation of facilities between e.spire's equipment and equipment of BellSouth. BellSouth may conduct an inspection if e.spire adds equipment and may otherwise conduct routine inspections at reasonable intervals mutually agreed upon by the Parties. BellSouth shall provide e.spire with a minimum of forty-eight (48) hours or two (2) business days, whichever is greater, advance notice of all such inspections. All costs of such inspection shall be borne by BellSouth.

11. SECURITY AND SAFETY REQUIREMENTS

11.1 Only BellSouth employees, BellSouth Certified Vendors and authorized employees, authorized Guests, pursuant to Section 3.3, preceding, or authorized agents of e.spire will be permitted in the BellSouth Premises. e.spire shall provide its employees and agents with picture identification which must be worn and visible at all times while in the Collocation Space or other areas in or around the Premises. The photo Identification card shall bear, at a minimum, the employee's name and photo, and the e.spire name. BellSouth reserves the right to remove from its Premises any employee of e.spire not possessing identification issued by e.spire. e.spire shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth Premise. e.spire shall be solely responsible for ensuring that any Guest of e.spire is in compliance with all subsections of this Section 11.

- 11.1.1 e.spire will be required, at its own expense, to conduct a statewide investigation of criminal history records for each e.spire employee being considered for work on the BellSouth Premises, for the states/counties where the e.spire employee has worked and lived for the past five (5) years. Where state law does not permit statewide collection or reporting, an investigation of the applicable counties is acceptable.
- 11.1.2 e.spire will be required to administer to their personnel assigned to the BellSouth Premises security training either provided by BellSouth, or meeting criteria defined by BellSouth.
- 11.1.3 e.spire shall not assign to the BellSouth Premises any personnel with records of felony criminal convictions. e.spire shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions, without advising BellSouth of the nature and gravity of the offense(s).

BellSouth reserves the right to refuse building access to any e.spire personnel who have been identified to have misdemeanor criminal convictions.

- 11.1.4 For each e.spire employee requiring access to a BellSouth Premises pursuant to this agreement, e.spire shall furnish BellSouth, prior to an employee gaining such access, a certification that the aforementioned background check and security training were completed. The certification will contain a statement that no felony convictions were found and certifying that the security training was completed by the employee. If the employee's criminal history includes misdemeanor convictions, e.spire will disclose the nature of the convictions to BellSouth at that time. In the alternative, e.spire may certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions other than misdemeanor traffic violations.
- 11.1.5 At BellSouth's request, e.spire shall promptly remove from the BellSouth's Premises any employee of e.spire BellSouth does not wish to grant access to its Premises pursuant to any investigation conducted by BellSouth.
- 11.2 Notification to BellSouth. BST reserves the right to interview e.spire's employees, agents, or contractors. e.spire and its contractors shall cooperate fully with BellSouth's investigation into allegations of wrongdoing or criminal conduct committed by or involving e.spire's employees, agents, or contractors. Additionally, BellSouth reserves the right to bill e.spire for all costs associated with reasonable and nondiscriminatory investigations involving its employees, agents, or contractors if it can be reasonably established that e.spire's employees, agents, or contractors are responsible for the alleged act. BellSouth shall bill e.spire for BellSouth property which is stolen or damaged where an investigation determines the culpability of e.spire's employees, agents, or contractors. e.spire shall notify BellSouth in writing immediately in the event that the CLEC discovers one of its employees already working on the BellSouth

Premises is a possible security risk. BellSouth reserves the right to permanently remove from its Premises any employee of e.spire identified as posing a security risk to BellSouth or any other CLEC, or having violated BellSouth policies set forth in the BellSouth CLEC Security Training.

- 11.2.1 <u>Use of BellSouth Supplies by e.spire Employees.</u> Use of any BellSouth supplies by a e.spire employee, whether or not used routinely to provide telephone service (e.g., plug-in cards,) will be considered theft and will be handled accordingly. Costs associated with such unauthorized use of BellSouth property may be charged to e.spire as may be all associated investigative costs. At BellSouth's request, e.spire shall promptly and permanently remove from BellSouth's Central Office any employee of e.spire found to be in violation of this rule.
- 11.3 <u>Use of Official Lines by e.spire Employees.</u> Except for local calls necessary or, in the performance of their work, e.spire employees shall not use the telephones on BellSouth Central Office. Charges for unauthorized telephone calls made by a e.spire's employees may be charged to e.spire as may be all associated investigative costs. At BellSouth's request, e.spire shall promptly and permanently remove from BellSouth's Premises any employee of e.spire found to be in violation of this rule.
- 11.4 <u>Accountability.</u> Full compliance with the security requirements of this section shall in no way limit the accountability of any CLEC for the improper actions of its employees.

12. DESTRUCTION OF COLLOCATION SPACE

12.1 In the event a Collocation Space is wholly or partially damaged by fire, windstorm, tornado, flood or by similar causes to such an extent as to be rendered wholly unsuitable for e.spire's permitted use hereunder, then either party may elect within ten (10) days after such damage, to terminate this Attachment, and if either party shall so elect, by giving the other written notice of termination, both parties shall stand released of and from further liability under the terms hereof. If the Collocation Space shall suffer only minor damage and shall not be rendered wholly unsuitable for e.spire's permitted use, or is damaged and the option to terminate is not exercised by either party, BellSouth covenants and agrees to proceed promptly without expense to e.spire, except for improvements not the property of BellSouth, to repair the damage. BellSouth shall have a reasonable time within which to rebuild or make any repairs, and such rebuilding and repairing shall be subject to delays caused by storms, shortages of labor and materials, government regulations, strikes, walkouts, and causes beyond the control of BellSouth, which causes shall not be construed as limiting factors, but as exemplary only. e.spire may, at its own expense, accelerate the rebuild of its collocated space and equipment provided however that a certified vendor is used and the necessary space preparation has been completed. Rebuild of equipment must be performed by a BellSouth Certified Vendor. If e.spire's acceleration of the project increases the cost of the project, then those additional charges will be incurred by e.spire. Where allowed and where practical, e.spire may erect a temporary facility while BellSouth rebuilds or makes repairs. In all cases where the Collocation Space shall be rebuilt or

Version 7/7/00

repaired, e.spire shall be entitled to an equitable abatement of rent and other charges, depending upon the unsuitability of the Collocation Space for e.spire's permitted use, until such Collocation Space is fully repaired and restored and e.spire's equipment installed therein (but in not event shall the abatement period extend beyond thirty (30) days after the Collocation Space is fully repaired and restored). Where e.spire has placed an Adjacent Arrangement pursuant to section 3.4, e.spire shall have the sole responsibility to repair or replace said Adjacent Arrangement provided herein. Pursuant to this section, BellSouth will restore the associated services to the Adjacent Arrangement.

13. EMINENT DOMAIN

13.1 If the whole of a Collocation Space or Adjacent Arrangement shall be taken by any public authority under the power of eminent domain, then this Attachment shall terminate with respect to such collocation arrangement as of the day possession shall be taken by such public authority and rent and other charges for the Collocation Space or Adjacent Arrangement shall be paid up to that day with proportionate refund by BellSouth of such rent and charges as may have been paid in advance for a period subsequent to the date of the taking. If any part of the Collocation Space or Adjacent Arrangement shall be taken under eminent domain, BellSouth and e.spire shall each have the right to terminate this Attachment and declare the same null and void, by written notice of such intention to the other party within ten (10) days after such taking with respect to the affected Collocation arrangement.

14. NONEXCLUSIVITY

14.1 e.spire understands that this Attachment is not exclusive and that BellSouth may enter into similar agreements with other parties. Assignment of space pursuant to all such agreements shall be determined by space availability and made on a first come, first served basis.

EXHIBIT A: BELLSOUTH/e.spire RATES – ALABAMA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per Request	NA	\$7,124.00 Disconnect Charge \$1.73
PE1CA	Subsequent Application Fee (Note 1)	Per Request	NA	\$1600.00 Minimum
PE1BB	Space Preparation Fee (Note 2)			
	Mechanical / HVAC* Ground Bar*	Per ton (one ton minimum) Per Connection		\$2,400.00 \$720.00
	Project Management*	Per arrangement		\$1675.00
	Cable Racking / Fiber Duct	Per arrangement, square foot		ICB
	Frame / Aisle Lighting	Per arrangement, square foot		ICB
	Framework Ground Conductors	Per arrangement		ICB
	Extraordinary Modifications	Per arrangement		ICB
	Space Enclosure (Note 3) Requested Prior to 6/1/99			
PE1BW PE1CW	Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$189.86 \$19.29	NA NA
PE1PJ	Floor Space	Per square foot	\$3.85	NA
DEADD	Oakla Installatia	Day Calab	, h.i.a	Ф0 005 00
PE1BD	Cable Installation	Per Cable	NA	\$2,335.00 Disconnect Charge \$54.39
PE1PM	Cable Support Structure	Per entrance cable	\$23.23	NA
. =	1 Cabic Support Structure	. J. Silitarios sabio	Ψ25.20	1 1/ 1

EXHIBIT A: BELLSOUTH/e.spire RATES - ALABAMA PHYSICAL COLLOCATION (cont.)

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1PL	Power -48V DC Power 120V AC Power single phase* 240V AC Power single phase* 120V AC Power three phase* 277V AC Power three phase*	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$7.14 \$5.50 \$11.00 \$16.50 \$38.20	ICB ICB ICB ICB ICB
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2 PE1F4	Cross Connects (Note 4) 2-wire 4-wire DS-1 DS-3 2-fiber 4-fiber	Per Cross Connect	\$.28 \$.56 \$2.14 \$38.63 \$10.44 \$18.76	First / Additional \$30.76 / \$29.40 \$31.01 / \$29.58 \$60.81 / \$41.71 \$57.80 / \$39.81 \$73.00 / \$52.00 \$88.00 / \$67.00 Disconnect Charges
	2-wire 4-wire DS-1 DS-3			First / Additional \$12.75 / \$11.38 \$12.82 / \$11.39 \$12.85 / \$11.50 \$14.93 / \$11.76
	Co-Carrier Cross-Connect (Note 5)			
PE1ES Fiber	Fiber Arrangement	Cable Support Structure, per linear foot	\$0.06	NA
PE1DS Copper	Copper or Coaxial	(existing)	\$0.03	NA
		Cable Support Structure (new)	NA	ICB
PE1A1	Security Access System Security system* New Access Card Activation* Administrative change, existing card* Replace lost or stolen card*	Per Central Office Per Card Per Card	\$52.00	\$55.00 \$35.00 \$250.00
	Space Availability Report*	Per Central Office Requested		\$550.00

EXHIBIT A: BELLSOUTH/e.spire RATES - ALABAMA PHYSICAL COLLOCATION (cont.)

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
DEADE	POT Bay Arrangements Prior to 6/1/99	Per Cross Connect	\$0.00	
PE1PE PE1PF	2 Wire Cross-Connect 4 Wire Cross-Connect		\$0.08 \$0.17	NA NA
PE1PG	DS1 Cross-Connect		\$0.69	NA NA
PE1PH PE1B2 PE1B4	DS3 Cross-Connect 2 Fiber Cross-Connect 4 Fiber Cross-Connect		\$4.74 \$25.89 \$34.91	NA NA NA
AEH	Additional Engineering Fee (Note 6)	Per request, First half hour/Add'l Half hour		First / Additional Basic Time - \$31.00 / \$22.00 Overtime - \$37.00 / \$26.00
	Coourity Foodst			
PE1BT PE1OT PE1PT	Security Escort Basic Time Overtime Premium Time	Per 1/2 hour/Additional Half-hour	NA NA NA	\$43.47/\$25.82 \$55.25/\$32.79 \$67.03/\$39.76

EXHIBIT A: BELLSOUTH/e.spire RATES - ALABAMA PHYSICAL COLLOCATION (cont.)

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) Space Preparation Fee: The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers the costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. In the event e.spire opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to e.spire as prescribed in Section 7 of the Collocation Attachment.
- (3) **Space Enclosure:** For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (4) **Cross Connects:** The charges for cross connects are for orders placed electronically. Cross connect elements may also be ordered manually for which there is an additional charge per element.

		Disconnect Charge:
	First / Additional	First / Additional
2-wire	\$34.03 / \$32.67	\$14.48 / \$13.11
4-wire	\$34.28 / \$32.85	\$14.55 / \$13.12
DS-1	\$64.08 / \$44.98	\$14.58 / \$13.23
DS-3	\$61.07 / \$43.08	\$16.66 / \$13.49

- (5) Co-Carrier Cross-Connect: As stated in Section 1.2 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross connection requested, the recurring charges as stated in this Exhibit A shall apply.
- (6) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling e.spire requested modifications to requests in progress or augmentations to existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT A: BELLSOUTH/e.spire RATES - FLORIDA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per Request	\$15.53	\$3,248.00
PE1CA	Subsequent Application Fee (Note 1)	Per Request	NA	\$1600.00 Minimum
PE1BB	Space Preparation Fee (Note 2)			
	Mechanical / HVAC*	Per ton		\$2,400.00
	Ground Bar*	(one ton minimum) Per Connection		\$720.00
	Project Management*	Per arrangement		\$1675.00
	Cable Racking / Fiber Duct	Per arrangement, square foot		ICB
	Frame / Aisle Lighting	Per arrangement, square foot		ICB
	Framework Ground Conductors	Per arrangement		ICB
	Extraordinary Modifications	Per arrangement		ICB
	Space Enclosure (Note 3)			
	Requested Prior to 6/1/99			
PE1BW PE1BC PE1BF	Wire Cage Gypsum Board Cage Fire Rated Cage	Per first 100 sq. Ft. Per first 100 sq. Ft. Per first 100 sq. Ft.	\$41.99 \$84.10 \$99.73	NA NA NA
PE1CW PE1CC PE1CF	Wire Cage Gypsum Board Cage Fire Rated Cage	Per add'l 50 sq. Ft. Per add'l 50 sq. Ft. Per add'l 50 sq. Ft.	\$4.14 \$9.35 \$11.30	NA NA NA
1 2 101	File Nated Cage	1 01 444 1 50 54.1 1.	ψ11.00	TVA
PE1PJ	Floor Space	Per sq. Ft.	\$4.25	NA
PE1BD	Cable Installation	Per Cable	\$2.77	\$1,056.00
PE1PM	Cable Support Structure		\$22.94	NA

EXHIBIT A: BELLSOUTH/e.spire RATES - FLORIDA PHYSICAL COLLOCATION (cont.)

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1PL	Power -48V DC Power 120V AC Power single phase * 240V AC Power single phase* 120V AC Power three phase* 277V AC Power three phase*	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$7.14 \$5.50 \$11.00 \$16.50 \$38.20	ICB ICB ICB ICB ICB
PE12C PE14C	Cross Connects (Note 4) 2-wire 4-wire	Per Cross Connect	\$.0524 \$.0524	\$11.57 \$11.57
PE11S PE11X	DS-1/DCS DS-1/DSX		\$8.085 \$.4110	\$69.64 \$69.64
PE13S PE13X	DS-3/DCS DS-3/DSX		\$56.97 \$10.06	\$528.00 \$528.00
PE1F2	Optical Cross Connects		\$6.46	\$2,431.00
	Co-Carrier Cross-Connect (Note 5)			
PE1ES	Fiber Cable Support Structure, existing	Per linear foot	\$0.06	NA
PE1DS	Copper or Coaxial Cable Support Structure, existing	Per linear foot	\$0.03	NA
(TBD)	Cable Support Structure Construction, new	Per new construction	NA	ICB
PE1A2	Security Access System Security System* New Access Card Activation* Administrative change, existing card*	Per Central Office Per request-5 cards Per Card	\$95.00 NA	\$85.12 \$35.00
	Replace lost or stolen card*	Per Card		\$250.00
	Space Availability Report *	Per Central Office Requested		\$550.00
	POT Bay (Note 6)		NA	NA

EXHIBIT A: BELLSOUTH/e.spire RATES - FLORIDA PHYSICAL COLLOCATION (cont.)

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
AEH	Additional Engineering Fee (Note 7)	Per request, First half hour/Add'l half hour		First /Add'l Basic Time - \$31.00/\$22.00 Overtime - \$37.00/\$26.00
	Security Escort Basic Time Overtime Premium Time	Per ¼ hour Per ¼ hour Per ¼ hour	NA NA NA	\$10.89 \$13.64 \$16.40

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) **Space Preparation Fee**: The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. BellSouth will pro rate the total shared space preparation costs among the collocators at each location based on the amount of square footage occupied by each collocator. This charge may vary depending on the location and type of arrangement requested.
- (3) **Space Enclosure Fee**: For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (4) **Cross Connects**: Rates shown are the equivalent per cross connect rates based on the Florida PSC Ordered rates as follows:

Cross Connects	Per Cross Connect	<u>RC</u>	<u>NRC</u>
2-wire	Per 100 X-Connects	\$5.24	\$1,157.00
4-wire	Per 100 X-Connects	\$5.24	\$1,157.00
DS-1/DCS	Per 28 X-Connects	\$226.39	\$1,950.00
DS-1/DSX	Per 28 X-Connects	\$11.51	\$1,950.00
DS-3/DCS	Per Cross Connect	\$56.97	\$ 528.00
DS-3/DSX	Per Cross Connect	\$10.06	\$528.00
Optical Cross Connects	Per Cross Connect	\$6.46	\$2,431.00

EXHIBIT A: BELLSOUTH/e.spire RATES - FLORIDA PHYSICAL COLLOCATION (cont.)

- (5) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the direct connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the direct connection requested, the recurring charges as stated in this Exhibit A shall apply.
- (6) **POT Bays**: BellSouth's Florida specific rates were established in the Florida Public Service Commission Docket No. 960833. The Commission did not set permanent rates for <u>POT Bays</u>, given the assumption by the parties to the Proceeding that they will always provide their own POT Bays. It will be necessary for e.spire to provide its own POT Bays per BellSouth specifications and provide the necessary information from which BellSouth can inventory.
- (7) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling e.spire requested modifications to requests in progress or augmentations to existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT A: BELLSOUTH/e.spire RATES - GEORGIA PHYSICAL COLLOCATION

USOC	rked with an asterisk (*) are inter Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per Request	NA	\$3,850.00
PE1CA	Subsequent Application Fee (Note 1)	Per Request	NA	\$1600.00 Minimum
PE1BB	Space Preparation Fee (Note 2)	Per square foot	NA	\$100.00
PE1BW PE1CW	Space Enclosure (Note 3) Cages Prior to 6/1/99 Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$170.64 \$17.33	NA NA
PE1PJ PE1PK	Floor Space Zone A Zone B	Per square foot Per square foot	\$7.50 \$6.75	NA NA
PE1BD	Cable Installation	Per Cable	NA	\$2,750.00
PE1PM	Cable Support Structure	Per entrance cable	\$13.35	NA
PE1PL	Power -48V DC Power 120V AC Power single phase* 240V AC Power single phase* 120V AC Power three phase* 277V AC Power three phase*	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$7.14 \$5.50 \$11.00 \$16.50 \$38.20	ICB ICB ICB ICB ICB
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2 PE1F4	Cross Connects 2-wire 4-wire DS-1 DS-3 2-fiber 4-fiber	Per Cross Connect	\$.30 \$.50 \$8.00 \$72.00 \$10.29 \$18.50	First / Additional \$12.60 / \$12.60 \$12.60 / \$12.60 \$155.00 / \$27.00 \$155.00 / \$27.00 \$73.00 / \$52.00 \$88.00 / \$67.00

EXHIBIT A: BELLSOUTH/e.spire RATES - GEORGIA PHYSICAL COLLOCATION (cont.)

Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
Co-Carrier Cross-Connect (Note 4) Fiber Cable Support Structure, existing	Per linear foot	\$0.06	NA
Copper or Coaxial Cable Support Structure, existing	Per linear foot	\$0.03	NA
Cable Support Structure Construction, new	Per new construction	NA	ICB
Security Access System Security system* New Access Card Activation* Administrative change, existing card* Replace lost or stolen card*	Per Central Office Per Card Per Card Per Card	\$52.00	\$55.00 \$35.00 \$250.00
Space Availability Report*	Per Central Office Requested		\$550.00
POT Bay Arrangements Prior to 6/1/99 2 Wire Cross-Connect 4 Wire Cross-Connect DS1 Cross-Connect DS3 Cross-Connect 2 Fiber Cross-Connect 4 Fiber Cross-Connect	Per Cross Connect	\$0.40 \$1.20 \$1.20 \$8.00 \$25.53 \$34.43	NA NA NA NA NA
Additional Engineering Fee (Note 5)	Per request, First half hour/Add'l Half hour		First /Add'l Basic Time - \$31.00/\$22.00 Overtime - \$37.00/\$26.00
Security Escort Basic Time Overtime Premium Time	Per 1/2 hour/Additional Half-hour	NA NA NA	\$41.00/\$25.00 \$48.00/\$30.00 \$55.00/\$35.00
	Co-Carrier Cross-Connect (Note 4) Fiber Cable Support Structure, existing Copper or Coaxial Cable Support Structure, existing Cable Support Structure Construction, new Security Access System Security system* New Access Card Activation* Administrative change, existing card* Replace lost or stolen card* Space Availability Report* POT Bay Arrangements Prior to 6/1/99 2 Wire Cross-Connect 4 Wire Cross-Connect DS1 Cross-Connect 2 Fiber Cross-Connect 4 Fiber Cross-Connect 4 Fiber Cross-Connect Additional Engineering Fee (Note 5) Security Escort Basic Time Overtime	Co-Carrier Cross-Connect (Note 4) Fiber Cable Support Structure, existing Copper or Coaxial Cable Support Structure existing Cable Support Structure Construction, new Security Access System Security Access System Security system* New Access Card Activation* Administrative change, existing card* Replace lost or stolen card* Per Card Space Availability Report* Per Central Office Per Card Per Card P	Co-Carrier Cross-Connect (Note 4) Fiber Cable Support Structure, existing Copper or Coaxial Cable Support Structure existing Cable Support Structure Construction, new Security Access System Security system* New Access Card Activation* Administrative change, existing card* Replace lost or stolen card* Per Card Space Availability Report* Per Central Office Requested Per Card Per Card Per Card Per Card Per Card Per Card Per Card Per Card Per Card Per Card Administrative Construction Per Card Per Card Per

EXHIBIT A: BELLSOUTH/e.spire RATES - GEORGIA PHYSICAL COLLOCATION (cont.)

Note(s)

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) Space Preparation Fee: The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers a portion of costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. This is a set fee of \$100 per square foot as established by the Georgia Public Service Commission Order in Docket No. 7061-U. In the event e.spire opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to e.spire as prescribed in Section 7 of the Collocation Attachment.
- (3) **Space Enclosure Fee**: For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a one-time fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (4) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.
- (5) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling e.spire requested modifications to requests in progress or augmentations to existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT A: BELLSOUTH/e.spire RATES - KENTUCKY PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

pplication Fee absequent Application Fee ace Preparation Fee ote 2) Mechanical / HVAC* Ground Bar* Project Management* Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground	Per Request Per Request Per ton (one ton minimum) Per Connection Per arrangement Per arrangement, per square foot Per arrangement, per square foot	Recurring Rate (RC) NA NA	Non-Recurring Rate (NRC) \$9,926.72 \$1600.00 Minimum \$2,100.00 \$720.00 \$1,675.00 ICB
begin and the second of the se	Per Request Per ton (one ton minimum) Per Connection Per arrangement Per arrangement, per square foot Per arrangement,		\$1600.00 Minimum \$2,100.00 \$720.00 \$1,675.00
pace Preparation Fee of the 2) Mechanical / HVAC* Ground Bar* Project Management* Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground	Per ton (one ton minimum) Per Connection Per arrangement Per arrangement, per square foot Per arrangement,	NA	\$2,100.00 \$720.00 \$1,675.00
Mechanical / HVAC* Ground Bar* Project Management* Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground	(one ton minimum) Per Connection Per arrangement Per arrangement, per square foot Per arrangement,		\$720.00 \$1,675.00
Mechanical / HVAC* Ground Bar* Project Management* Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground	(one ton minimum) Per Connection Per arrangement Per arrangement, per square foot Per arrangement,		\$720.00 \$1,675.00
Project Management* Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground	Per Connection Per arrangement Per arrangement, per square foot Per arrangement,		\$1,675.00
Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground	Per arrangement, per square foot Per arrangement,		
Frame / Aisle lighting Framework Ground	per square foot Per arrangement,		ICB
Framework Ground	Per arrangement,		1
			ICB
Conductors	Per arrangement		ICB
Extraordinary Modifications	Per arrangement		ICB
pace Enclosure (Note 3)			1
ior to 6/1/99	Don first 400 on ft	#204.02	NIA
Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$201.02 \$20.42	NA NA
oor Space	Per square foot	\$5.00	NA
able Installation	Per Cable	NA	\$2,327.08
able Support Structure	Per entrance cable	\$24.23	NA
ower			
18V DC Power	Per amp	\$7.68	ICB
			ICB
			ICB ICB
201/ AC Dower three phace*	Per breaker amp	\$38.20	ICB
1 2 2	able Installation able Support Structure	por Space Per square foot Able Installation Per Cable Per entrance cable Per entrance cable Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	por Space Per square foot \$5.00 Able Installation Per Cable NA Able Support Structure Per entrance cable \$24.23 Abover 18V DC Power 20V AC Power single phase* 40V AC Power single phase* 20V AC Power three phase* 20V AC Powe

EXHIBIT A: BELLSOUTH/e.spire RATES - KENTUCKY PHYSICAL COLLOCATION (cont.)

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2 PE1F4	Cross Connects 2-wire 4-wire DS-1 DS-3 2-fiber 4-fiber	Per Cross Connect	\$.31 \$.62 \$1.92 \$39.94 \$13.28 \$23.87	First / Additional \$54.21/\$51.07 \$54.23/\$50.96 \$99.23/\$69.15 \$97.48/\$66.90 \$73.00/\$52.00 \$88.00/\$67.00
	Co-Carrier Cross-Connect (Note 5)			
PE1ES Fiber	Fiber Arrangement Cable Support Structure	Per linear foot (existing)	\$0.06	NA
PE1DS Copper	Copper or Coaxial Arrangement	Per linear foot (existing)	\$0.03	NA
TBD	Cable Support Structure Construction	Per new construction	NA	ICB
PE1A1	Security Access System Security system New Access Card Activation Administrative change, existing card Replace lost or stolen card	Per Central Office Per Card Per Card	\$52.00	\$55.00 \$35.00 \$250.00
TBD	Space Availability Report	Per Central Office Requested	NA	\$550.00
PE1PE PE1PF PE1PG PE1PH PE1B2 PE1B4	POT Bay Arrangements Prior to 6/1/99 2 Wire Cross-Connect 4 Wire Cross-Connect DS1 Cross-Connect DS3 Cross-Connect 2 Fiber Cross-Connect 4 Fiber Cross-Connect	Per Cross Connect	\$0.06 \$0.15 \$0.58 \$4.51 \$32.94 \$44.42	NA NA NA NA NA
PE1BT PE1OT PE1PT	Security Escort Basic Time Overtime Premium Time	Per 1/2 hour/Additional Half-hour	NA NA NA	\$56.09/\$31.99 \$67.75/\$39.00 \$79.41/\$46.01

EXHIBIT A: BELLSOUTH/e.spire RATES - KENTUCKY PHYSICAL COLLOCATION (cont.)

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
AEH	Additional Engineering Fee (Note 5)	Per request, First half hour/Add'l Half hour		First /Add'l Basic Time - \$31.00/\$22.00 Overtime - \$37.00/\$26.00

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) Space Preparation Fee: The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers the costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. In the event e.spire opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to e.spire as prescribed in Section 7 of the Collocation Attachment.
- (3) **Space Enclosure Fee**: For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a one-time fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (4) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.
- (5) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling e.spire requested modifications to requests in progress or augmentations to existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT A: BELLSOUTH/e.spire RATES – LOUISIANA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
			(RC)	Rate (NRC)
PE1BA	Application Fee	Per Request	NA	\$4,910.00
PE1CA	Subsequent Application Fee	Per Request	NA	\$1600.00
	(Note 1)			Minimum
PE1BB	Space Preparation Fee			
1 2 100	(Note 2)			
	Mechanical / HVAC*	Per ton		\$2,100.00
	Ground Bar*	(one ton minimum) Per Connection		\$720.00
	Ground Bar	1 CI COMMCCHOM		\$720.00
	Project Management*	Per arrangement		\$1,675.00
	Cable Racking/Fiber Duct	Per arrangement,		ICB
		per square foot		
	Frame / Aisle lighting	Per arrangement,		ICB
	Framework Cround	per square foot		IOD
	Framework Ground Conductors	Per arrangement		ICB
	Extraordinary Modifications	Per arrangement		ICB
	Space Enclosure (Note 3)			
	Prior to 6/1/99			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$197.55	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$20.07	NA
			•	
PE1PJ	Floor Space	Per square foot	\$4.01	NA
PE1BD	Cable Installation	Per Cable	NA	\$1,706.00
1 2 100	Cable installation	1 CI Gabic	IVA	Disconnect charge
_				\$36.00
PE1PM	Cable Support Structure	Per entrance cable	\$24.05	NA
PE1PL	Power			
	-48V DC Power	Per amp	\$7.15	ICB
	120V AC Power single phase*	Per breaker amp	\$5.50	ICB
	240V AC Power single phase*	Per breaker amp	\$11.00	ICB
	120V AC Power three phase* 277V AC Power three phase*	Per breaker amp Per breaker amp	\$16.50 \$38.20	ICB ICB
	·	,	·	

EXHIBIT A: BELLSOUTH/e.spire RATES - LOUISIANA PHYSICAL COLLOCATION (cont.)

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
			_	U
			(RC)	Rate (NRC)
	Cross Connects (Note 4)	Per Cross Connect		First / Additional
PE1P2	2-wire		\$.26	\$23.04/\$22.11
PE1P4	4-wire		\$.52	\$23.23/\$22.24
PE1P1	DS-1		\$2.03	\$43.61/\$30.60
PE1P3	DS-3		\$36.27	\$41.46/\$29.20
PE1F2	2-fiber		\$10.20	\$73.00/\$52.00
PE1F4	4-fiber		\$18.34	\$88.00/\$67.00
	4-11bei		Ψ10.0-	φου.σο/φο/1.σο
				Disconnect
				charges
				First / Additional
	2-wire			\$9.48/\$8.54
	4-wire			\$9.53/\$8.55
	DS-1			\$9.56/\$8.63
	DS-3			\$11.06/\$8.86
	Co-Carrier Cross-Connect			
	(Note 5)			
PE1ES	Fiber Arrangement Cable	Per linear foot	\$0.06	NA
_	Support Structure	(existing)	ψ0.00	1471
TIDCI	Support Structure	(CXIStillig)		
PE1DS	Copper or Coaxial	Per linear foot	\$0.03	NA
	Arrangement	(existing)	ψ0.03	IWA
Copper	Arrangement	(existing)		
TBD	Cable Support Structure	Per new	NA	ICB
	Construction	construction		
DEANA	Constitut Assess Constant			
PE1A1	Security Access System	Dan Cantual Office	Ф БО ОО	
	Security system*	Per Central Office	\$52.00	
	New Access Card Activation*	Per Card		\$55.00
	Administrative change,	Per Card		\$35.00
	existing card*			
	Replace lost or stolen card	Per Card		\$250.00
TBD	Space Availability Report*	Per Central Office		\$550.00
	•	Requested		
		-		

EXHIBIT A: BELLSOUTH/e.spire RATES - LOUISIANA PHYSICAL COLLOCATION (cont.)

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	POT Bay Arrangements Prior to 6/1/99	Per Cross Connect		
PE1PE	2 Wire Cross-Connect		\$0.0776	NA
PE1PF	4 Wire Cross-Connect		\$0.1552	NA
PE1PG	DS1 Cross-Connect		\$0.6406	NA
PE1PH	DS3 Cross-Connect		\$4.75	NA
PE1B2	2 Fiber Cross-Connect		\$25.30	NA
PE1B4	4 Fiber Cross-Connect		\$34.12	NA
	Security Escort			
PE1BT	Basic Time	Per 1/2	NA	\$32.35/\$19.95
PE1OT	Overtime	hour/Additional	NA	\$40.50/\$25.00
PE1PT	Premium Time	Half-hour	NA	\$48.66/\$30.05
AEH	Additional Engineering Fee	Per request,		First /Add'l
	(Note 6)	First half		Basic Time -
		hour/Add'l		\$31.00/\$22.00
		Half hour		Overtime -
				\$37.00/\$26.00

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) Space Preparation Fee: The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers the costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. In the event e.spire opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to e.spire as prescribed in Section 7 of the Collocation Attachment.

EXHIBIT A: BELLSOUTH/e.spire RATES - LOUISIANA PHYSICAL COLLOCATION (cont.)

- (3) Space Enclosure Fee: For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (4) **Cross Connects**: The charges for cross connects are for orders placed electronically. Cross connect elements may also be ordered manually for which there is an additional charge per element.

		Disconnect Charges
	First / Additional	First / Additional
2-wire	\$24.92/\$23.99	\$10.56/\$9.62
4-wire	\$25.11/\$24.12	\$10.61/\$9.63
DS-1	\$45.49/\$32.48	\$10.64/\$9.71
DS-3	\$43.34/\$31.08	\$12.14/\$9.94

- (5) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.
- (6) Additional Engineering Fee. BellSouth's additional engineering, and other labor costs associated with handling e.spire -requested modifications to requests in progress or augmentations to existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT A: BELLSOUTH/e.spire RATES – MISSISSIPPI PHYSICAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per Request	NA	\$6,993.00 Disconnect charge \$1.70
PE1CA	Subsequent Application Fee (Note 1)	Per Request	NA	\$1600.00 Minimum
PE1BB	Space Preparation Fee			
	(Note 2) Mechanical / HVAC*	Per ton (one ton minimum)		\$2,100.00
	Ground Bar*	Per Connection		\$720.00
	Project Management*	Per arrangement		\$1,675.00
	Cable Racking/Fiber Duct	Per arrangement, per square foot		ICB
	Frame / Aisle lighting	Per arrangement, per square foot		ICB
	Framework Ground Conductors	Per arrangement		ICB
	Extraordinary Modifications	Per arrangement		ICB
	Space Enclosure (Note 3) Prior to 6/1/99			
PE1BW PE1CW	Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$205.08 \$20.83	NA NA
PE1PJ	Floor Space	Per square foot	\$3.45	Disconnect charge \$53.24
PE1BD	Cable Installation	Per Cable	NA	\$2,419.00
PE1PM	Cable Support Structure	Per entrance cable	\$22.90	NA
PE1PL	Power			
	-48V DC Power	Per amp	\$6.93	ICB
	120V AC Power single phase*	Per breaker amp	\$5.50	ICB
	240V AC Power single phase*	Per breaker amp	\$11.00 \$16.50	ICB
	120V AC Power three phase* 277V AC Power three phase*	Per breaker amp Per breaker amp	\$16.50 \$38.20	ICB ICB

EXHIBIT A: BELLSOUTH/e.spire RATES - MISSISSIPPI PHYSICAL COLLOCATION (cont.)

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
			(RC)	Rate (NRC)
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2 PE1F4	Cross Connects (Note 4) 2-wire 4-wire DS-1 DS-3 2-fiber 4-fiber	Per Cross Connect	\$.3996 \$.7992 \$2.90 \$53.31 \$15.82 \$28.43	First / Additional \$30.93/\$29.59 \$31.17/\$29.77 \$60.42/\$41.68 \$57.45/\$39.81 \$73.00/\$52.00 \$88.00/\$67.00
	2-wire 4-wire DS-1 DS-3			Charges First / Additional \$12.76/\$11.43 \$12.83/\$11.43 \$12.87/\$11.54 \$14.92/\$11.80
	Co-Carrier Cross-Connect (Note 5)			
PE1ES Fiber	Fiber Arrangement Cable Support Structure	Per linear foot (existing)	\$0.06	NA
PE1DS Copper	Copper or Coaxial Arrangement	Per linear foot (existing)	\$0.03	NA
TBD	Cable Support Structure Construction	Per new construction	NA	ICB
PE1A1	Security Access System Security system* New Access Card Activation* Administrative change, existing card* Replace lost or stolen card	Per Central Office Per Card Per Card	\$52.00	\$55.00 \$35.00 \$250.00
TBD	Space Availability Report*	Per Central Office Requested		\$550.00

EXHIBIT A: BELLSOUTH/e.spire RATES - MISSISSIPPI PHYSICAL COLLOCATION (cont.)

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
			(RC)	Rate (NRC)
	POT Bay Arrangements	Per Cross Connect		
	Prior to 6/1/99			
PE1PE	2 Wire Cross-Connect		\$0.1195	NA
PE1PF	4 Wire Cross-Connect		\$0.2389	NA
PE1PG	DS1 Cross-Connect		\$0.9862	NA
PE1PH	DS3 Cross-Connect		\$5.81	NA
PE1B2	2 Fiber Cross-Connect		\$39.23	NA
PE1B4	4 Fiber Cross-Connect		\$52.91	NA
AEH	Additional Engineering Fee	Per request,		First /Add'l
	(Note 6)	First half		Basic Time -
		hour/Add'l		\$31.00/\$22.00
		Half hour		Overtime -
				\$37.00/\$26.00
	Security Escort			
PE1BT	Basic Time	Per 1/2	NA	\$42.87/\$25.54
PE1OT	Overtime	hour/Additional	NA	\$54.43/\$32.41
PE1PT	Premium Time	Half-hour	NA	\$65.99/\$39.28

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) **Space Preparation Fee**: The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers the costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. In the event e.spire opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to e.spire as prescribed in Section 7 of the Collocation Attachment.

EXHIBIT A: BELLSOUTH/e.spire RATES - MISSISSIPPI PHYSICAL COLLOCATION (cont.)

- (3) **Space Enclosure Fee**: For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (4) **Cross Connects**: The charges for cross connects are for orders placed electronically. Cross connect elements may also be ordered manually for which there is an additional charge per element.

		Disconnect Charges
	First / Additional	First / Additional
2-wire	\$33.58 / \$32.24	\$14.27 / \$12.94
4-wire	\$33.82 / \$32.42	\$14.34 / \$12.94
DS-1	\$63.07 / \$44.33	\$14.38 / \$13.05
DS-3	\$60.10 / \$42.46	\$16.43 / \$13.31

- (5) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.
- (6) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling e.spire -requested modifications to requests in progress or augmentations for existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT A: BELLSOUTH/e.spire RATES – NORTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate	Non-Recurring
			(RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,850.00
PE1CA	Subsequent Application Fee (Note	Per request	NA	\$1,600.00
	1)			Minimum
	Space Preparation Fee	D C	Φ1.57	
	Central Office Modification	Per sq. ft.	\$1.57	
	Common Systems Modification –	Per sq. ft.	\$3.26	
	Cageless	D	¢110.70	
	Common Systems Modification –	Per cage	\$110.79	
	Caged Power	Per nominal –48v	\$5.76	
	Fower	DC Amp	\$3.70	
		DC Allip		
	Space Enclosure (Note 2)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$102.76	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$10.44	NA
			, , , , , , , , , , , , , , , , , , ,	
PE1PJ	Floor Space	Per sq. ft.	\$3.45	NA
PE1BD	Cable Installation	Per cable	NA	\$2,305.00
PE1PM	Cable Support Structure	Per entrance cable	\$21.33	NA
	Power			
PE1PL	-48V DC Power	Per amp	\$6.65	ICB
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.50	ICB
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.00	ICB
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.50	ICB
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.20	ICB
	Cuasa Connecta (Nata 2)	Dan arous connect		First/Add'1
DE1D2	Cross Connects (Note 3)	Per cross connect	\$0.22	
PE1P2 PE1P4	2-wire 4-wire		\$0.32 \$0.64	\$41.78/\$39.23 \$41.91/\$39.25
PE1P4 PE1P1	DS-1		\$2.34	\$41.91/\$39.23 \$71.02/\$51.08
PE1P1 PE1P3	DS-1 DS-3		\$2.34 \$42.84	\$69.84/\$49.43
PE1F3	2-fiber		\$15.99	\$67.34/\$48.55
PE1F4	4-fiber		\$28.74	\$82.35/\$63.56
T L 1 L 4	+-110Cl		\$40.74	φο4.33/φ03.30

	NORTH CAROLINA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Co-Carrier Cross-Connect (Note				
	4)				
PE1ES	Fiber Cable Support Structure,	Per linear ft.	\$0.06	NA	
Fiber PE1DS	existing Copper or Coaxial Cable Support	Per linear ft.	\$0.03	NA	
Copper	Structure, existing	rei illicai it.	φ0.03	INA	
(TBD)	Cable Support Structure	Per new	NA	ICB	
()	Construction, new	construction	5.2		
PE1AX	Security Access System Security System*	Per premises	\$52.00		
	New Access Card Activation*	Per card		\$55.00	
PE1AA	Administrative change, existing	Per card		\$35.00	
	card*				
PE1AR	Replace lost or stolen card	Per card		\$250.00	
PE1SR	Space Availability Report*	Per premises		\$550.00	
	J 1	requested			
	DOT Days Assessment	D			
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect			
PE1PE	2-Wire Cross-Connect		\$0.10	NA	
PE1PF	4-Wire Cross-Connect		\$0.19	NA	
PE1PG	DS1 Cross-Connect		\$0.79	NA	
PE1PH	DS3 Cross-Connect		\$4.85	NA	
PE1B2	2 Fiber Cross-Connect		\$39.67	NA	
PE1B4	4 Fiber Cross-Connect		\$53.49	NA	
	Security Escort	Per half hr./Add'l			
	Security Escort	half hr.			
PE1BT	Basic Time	11411 111.	NA	\$42.92/\$25.56	
PE1OT	Overtime		NA	\$54.51/\$32.44	
PE1PT	Premium Time		NA	\$66.10/\$39.32	
AEH	Additional Engineering Fee (Note	Per request, first		First/Add'l	
ALII	5)	half hr/add'l half hr.		Basic Time	
	3)	nan m/add i nan ill.		\$31.00/\$22.00	
				Overtime	
				\$37.00/\$26.00	

EXHIBIT A: BELLSOUTH/e.spire RATES – NORTH CAROLINA PHYSICAL COLLOCATION (continued)

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) **Space Enclosure Fee**: The Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth Certified Contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the BellSouth Certified Contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (3) **Cross Connect:** The charges for cross connects are for orders placed electronically. Cross connect elements may also be ordered manually for which there is an additional charge per element.

	First/Additional
2-wire	\$46.53/\$43.98
4-wire	\$46.64/\$43.98
DS-1	\$75.72/\$55.78
DS-3	\$74.54/\$54.13

(4) **Co-Carrier Cross-Connect**. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Premises in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis as described in Section 5.6.1 of the Collocation Attachment. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.

(2)

EXHIBIT A: BELLSOUTH/e.spire RATES – SOUTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

Application Fee Subsequent Application Fee (Note 1) Space Preparation Fee (Note 2) Mechanical / HVAC*	Per Request Per Request	(RC) NA NA	\$4,850.00 \$1600.00 Minimum
(Note 1) Space Preparation Fee (Note 2)	Per Request	NA	
(Note 2)			IVIIIIIIIIIIIIII
Mechanical / HVAC"	Danton		#2.400.00
	Per ton (one ton minimum)		\$2,100.00
Ground Bar*	Per Connection Per arrangement		\$720.00
Project Management*			\$1,675.00
Cable Racking/Fiber Duct	per square foot Per arrangement,		ICB
Frame / Aisle lighting	per square foot Per arrangement		ICB
Framework Ground Conductors			ICB
Extraordinary Modifications	Per arrangement		ICB
Space Enclosure (Note 3) Prior to 6/1/99			
Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$224.60 \$22.81	NA NA
Floor Space	Per square foot	\$3.90	NA
Cable Installation	Per Cable	NA	\$2,217.00
Cable Support Structure	Per entrance cable	\$24.55	NA
Power -48V DC Power 120V AC Power single phase* 240V AC Power single phase* 120V AC Power three phase* 277V AC Power three phase*	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$7.09 \$5.50 \$11.00 \$16.50 \$38.20	ICB ICB ICB ICB ICB
	Project Management* Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground Conductors Extraordinary Modifications Space Enclosure (Note 3) Prior to 6/1/99 Welded Wire-mesh Welded Wire-mesh Welded Wire-mesh Cable Installation Cable Support Structure Power -48V DC Power 120V AC Power single phase* 240V AC Power single phase* 120V AC Power three phase*	Per arrangement Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground Conductors Extraordinary Modifications Space Enclosure (Note 3) Prior to 6/1/99 Welded Wire-mesh Welded Wire-mesh Welded Wire-mesh Welded Wire-mesh Teloor Space Cable Installation Per arrangement	Per arrangement Cable Racking/Fiber Duct Frame / Aisle lighting Framework Ground Conductors Extraordinary Modifications Space Enclosure (Note 3) Prior to 6/1/99 Welded Wire-mesh Welded Wire-mesh Floor Space Per square foot Per arrangement Per arrangement

EXHIBIT A: BELLSOUTH/e.spire RATES – SOUTH CAROLINA PHYSICAL COLLOCATION (cont.)

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2 PE1F4	Cross Connects (Note 4) 2-wire 4-wire DS-1 DS-3 2-fiber 4-fiber	Per Cross Connect Per Cross Connect Per Cross Connect Per Cross Connect Per Cross Connect Per Cross Connect	\$.3648 \$.7297 \$2.70 \$49.24 \$13.75 \$24.71	First / Additional \$41.50/\$38.94 \$41.56/\$38.90 \$70.79/\$50.78 \$69.60/\$49.14 \$73.00/\$52.00 \$88.00/\$67.00
	Co-Carrier Cross-Connect (Note 5)			
PE1ES Fiber	Fiber Arrangement Cable Support Structure	Per linear foot (existing)	\$0.06	NA
PE1DS Copper	Copper or Coaxial Arrangement	Per linear foot (existing)	\$0.03	NA
TBD	Cable Support Structure Construction	Per new construction	NA	ICB
PE1A1	Security Access System Security system* New Access Card Activation* Administrative change, existing card* Replace lost or stolen card	Per Central Office Per Card Per Card	\$52.00	\$55.00 \$35.00 \$250.00
TBD	Space Availability Report*	Per Central Office Requested		\$550.00
PE1PE PE1PF PE1PG PE1PH PE1B2 PE1B4	POT Bay Arrangements Prior to 6/1/99 2 Wire Cross-Connect 4 Wire Cross-Connect DS1 Cross-Connect DS3 Cross-Connect 2 Fiber Cross-Connect 4 Fiber Cross-Connect	Per Cross Connect	\$.1091 \$.2181 \$.9004 \$5.64 \$34.09 \$45.97	NA NA NA NA NA
PE1BT PE1OT PE1PT	Security Escort Basic Time Overtime Premium Time	Per 1/2 hour/Additional Half-hour	NA NA NA	\$43.00/\$25.57 \$54.62/\$32.46 \$66.24/\$39.35

EXHIBIT A: BELLSOUTH/e.spire RATES – SOUTH CAROLINA PHYSICAL COLLOCATION (cont.)

AEH	Additional Engineering Fee (Note 6)	Per request, First half hour/Add'l Half hour	First /Add'l Basic Time - \$31.00/\$22.00 Overtime - \$37.00/\$26.00
			ψ01:00/ψ20:00

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) **Space Preparation Fee:** The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers the costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. In the event e.spire opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to e.spire as prescribed in Section 7 of the Collocation Attachment.
- (3) **Space Enclosure Fee**: For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.
- (4) **Cross Connects**: The charges for cross connects are for orders placed electronically. Cross connect elements may also be ordered manually for which there is an additional charge per element.

	First / Additional
2-wire	\$46.66 / \$44.10
4-wire	\$46.68 / \$44.02
DS-1	\$75.88 / \$55.87
DS-3	\$74.69 / \$54.23

(5) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.

Version 7/7/00

EXHIBIT A: BELLSOUTH/e.spire RATES – SOUTH CAROLINA PHYSICAL COLLOCATION (cont.)

(6) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling e.spire - requested modifications to requests in progress or augmentations to existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT A: BELLSOUTH/e.spire RATES – TENNESSEE* PHYSICAL COLLOCATION

*All Rates are interim and subject to true-up.

*All Rates are interim and subject to true-up.					
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
PE1BA	Application Fee	Per Request	NA	\$3,850.00	
DE 404				* * * * * * * * * * * * * * * * * * *	
PE1CA	Subsequent Application Fee (Note 1)	Per Request	NA	\$1,600.00 Minimum	
PE1BB	Space Preparation Fee (Note 2)				
	Mechanical / HVAC	Per ton (one ton minimum)		\$2,100.00	
	Ground Bar	Per Connection		\$720.00	
	Project Management	Per arrangement		\$1,675.00	
	Cable Racking/Fiber Duct	Per arrangement, per square foot		ICB	
	Frame / Aisle lighting	Per arrangement, per square foot		ICB	
	Framework Ground Conductors	Per arrangement		ICB	
	Extraordinary Modifications	Per arrangement		ICB	
	Space Enclosure (Note 3) Prior to 6/1/99				
PE1BW PE1CW	Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$190.79 \$19.38	NA NA	
PE1PJ	Floor Space	Per square foot	\$7.50	NA	
	1 loor Space	i ei squaie ioot	Ψ1.50	IVA	
PE1BD	Cable Installation	Per Cable	NA	\$2,750.00	
PE1PM	Cable Support Structure	Per entrance cable	\$13.35	NA	
1 - 11 101	Cable Support Structure	1 Ci Cittalice Cable	ψ10.55	INA	
PE1PL	Power -48V DC Power	Per amp	\$5.00	ICB	
	120V AC Power single phase	Per breaker amp	\$5.50 \$11.00	ICB ICB	
	240V AC Power single phase 120V AC Power three phase	Per breaker amp Per breaker amp	\$11.00 \$16.50	ICB	
	277V AC Power three phase	Per breaker amp	\$38.20	ICB	

EXHIBIT A: BELLSOUTH/e.spire RATES – TENNESSEE* PHYSICAL COLLOCATION (cont.)

*Rates are interim and subject to true-up.

	re interim and subject to true-up.			
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1P2 PE1P4	Cross Connects 2-wire 4-wire	Per Cross Connect	\$.30 \$.50	First / Additional \$19.20/\$19.20 \$19.20/\$19.20
PE1P1 PE1P3	DS-1 DS-3		\$8.00 \$72.00	\$155.00/\$27.00 \$155.00/\$27.00
PE1F2 PE1F4	2-fiber 4-fiber		\$15.90 \$28.50	\$73.00/\$52.00 \$88.00/\$67.00
	Co-Carrier Cross-Connect (Note 4)			
PE1ES Fiber	Fiber cable support structure, existing	Per linear foot	\$0.06	NA
PE1DS Copper	Copper or Coaxial cable support structure, existing	Per linear foot	\$0.03	NA
TBD	Cable Support Structure Construction (new)	Per new construction	NA	ICB
PE1A1	Security Access System	Dan Cantual Office	Ф.Г.О.О.О.	
	Security system New Access Card Activation Administrative change, existing card	Per Central Office Per Card Per Card	\$52.00	\$55.00 \$35.00
	Replace lost or stolen card	Per Card		\$250.00
TBD	Space Availability Report	Per Central Office Requested		\$550.00
	POT Bay Arrangements Prior to 6/1/99			
PE1PE	2 Wire Cross-Connect	Per Cross Connect	\$0.40	NA
PE1PF PE1PG	4 Wire Cross-Connect DS1 Cross-Connect	Per Cross Connect Per Cross Connect	\$1.20 \$1.20	NA NA
PE1PG PE1PH	DS3 Cross-Connect	Per Cross Connect	\$8.00	NA NA
PE1B2 PE1B4	2 Fiber Cross-Connect 4 Fiber Cross-Connect	Per Cross Connect Per Cross Connect	\$39.30 \$53.00	NA NA
		1 3. C.CCS COMMON	\$20.00	

EXHIBIT A: BELLSOUTH/e.spire RATES – TENNESSEE* PHYSICAL COLLOCATION (cont.)

*Rates are interim and subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BT PE1OT PE1PT	Security Escort Basic Time Overtime Premium Time	Per 1/2 hour/Additional Half-hour	NA NA NA	\$41.00/\$25.00 \$48.00/\$30.00 \$55.00/\$35.00
AEH	Additional Engineering Fee (Note 5)	Per request, First half hour/Add'l Half hour		First /Add'I Basic Time - \$31.00/\$22.00 Overtime - \$37.00/\$26.00

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

- (1) Subsequent Application Fee: BellSouth requires the submission of an Application Fee for modifications to an existing arrangement. However, when the modifications do not require BellSouth to expend capital, BellSouth will assess the Subsequent Application Fee in lieu of the Application Fee. Proposed modifications that could result in assessment of a Subsequent Application Fee would cause BellSouth to analyze the following but are not limited to: floor loading changes, changes to HVAC requirements, power requirement changes which may result in a power plant upgrade, environmental or safety requirements, or equipment relocation. Should the Subsequent Application Fee not be included as part of this Attachment, e.spire will be assessed the full Application Fee for all subsequent activity for completed arrangements.
- (2) Space Preparation Fee: The Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers the costs associated with the shared physical collocation area within a Central Office, which include survey, engineering, design and modification costs for network, building and support systems. In the event e.spire opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to e.spire as prescribed in Section 7 of the Collocation Attachment.
- (3) Space Enclosure Fee: For cages requested prior to June 1, 1999, the Space Enclosure Construction Fee is a monthly recurring fee, assessed per enclosure, per location with a one-hundred (100) square foot minimum enclosure. It recovers costs associated with providing an optional equipment arrangement enclosure, which include architectural and engineering fees, materials, and installation costs. The cost for additional square feet is applicable only when ordered with the first 100 square feet and must be requested in fifty (50) square foot increments. e.spire may, at its option, arrange with a BellSouth certified contractor to construct the space enclosure in accordance with BellSouth's guidelines and specifications. In this event, the contractor shall directly bill e.spire for the space enclosure, and this fee shall not be applicable.

EXHIBIT A: BELLSOUTH/e.spire RATES – TENNESSEE* PHYSICAL COLLOCATION (cont.)

- (4) Co-Carrier Cross-Connect. As stated in Section 5 of the Collocation Attachment, e.spire may connect to other CLECs within the designated Central Office in addition to, and not in lieu of, interconnection to BellSouth services and facilities. Where BellSouth must construct a cable rack structure to house the co-Carrier cross-connection, construction charges will be applied on an individual case basis. BellSouth shall provide an estimate of these charges in the Application Response. Where an existing cable rack structure is in place and has sufficient capacity to accommodate the co-Carrier cross-connection requested, the recurring charges as stated in this Exhibit A shall apply.
- (5) Additional Engineering Fee: BellSouth's additional engineering, and other labor costs associated with handling e.spire -requested modifications to requests in progress or augmentations for existing arrangements shall be recovered as Additional Engineering charges, under provisions in BellSouth's F.C.C. Number 1 Tariff, Sections 13.1 and 13.2. Should Additional Engineering rates not be included, e.spire agrees not to make changes to collocation arrangement after a Bona Fide Firm Order is submitted.

EXHIBIT B Page 1 of 6

ENVIRONMENTAL AND SAFETY PRINCIPLES

The following principles provide basic guidance on environmental and safety issues when applying for and establishing Physical Collocation arrangements.

1. GENERAL PRINCIPLES

- 1.1 Compliance with Applicable Law. BellSouth and e.spire agree to comply with applicable federal, state, and local environmental and safety laws and regulations including U.S. Environmental Protection Agency (USEPA) regulations issued under the Clean Air Act (CAA), Clean Water Act (CWA), Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), Superfund Amendments and Reauthorization Act (SARA), the Toxic Substances Control Act (TSCA), and OSHA regulations issued under the Occupational Safety and Health Act of 1970, as amended and NFPA and National Electrical Codes (NEC) and the NESC ("Applicable Laws"). Each party shall notify the other if compliance inspections are conducted by regulatory agencies and/or citations are issued that relate to any aspect of this agreement.
- 1.2 <u>Notice</u>. BellSouth and e.spire shall provide notice to the other, including Material Safety Data Sheets (MSDSs), of known and recognized physical hazards or Hazardous Chemicals existing on site or brought on site. Each party is required to provide specific notice for known potential Imminent Danger conditions. e.spire should contact 1-800-743-6737 for BellSouth MSDS sheets.
- 1.3 Practices/Procedures. BellSouth may make available additional environmental control procedures for e.spire to follow when working at a BellSouth Premises (See Section 2, below). These practices/procedures will represent the regular work practices required to be followed by the employees and contractors of BellSouth for environmental protection. e.spire will require its contractors, agents and others accessing the BellSouth Premises to comply with these practices. Section 2 lists the Environmental categories where BST practices should be followed by CLEC when operating in the BellSouth Premises.
- 1.4 <u>Environmental and Safety Inspections</u>. BellSouth reserves the right to inspect the e.spire space with proper notification. BellSouth reserves the right to stop any e.spire work operation that imposes Imminent Danger to the environment, employees or other persons in the area or Facility.

EXHIBIT B Page 2 of 6

- 1.5 <u>Hazardous Materials Brought On Site</u>. Any hazardous materials brought into, used, stored or abandoned at the BellSouth Premises by e.spire are owned by e.spire. e.spire will indemnify BellSouth for claims, lawsuits or damages to persons or property caused by these materials. Without prior written BellSouth approval, no substantial new safety or environmental hazards can be created by e.spire or different hazardous materials used by e.spire at BellSouth Facility. e.spire must demonstrate adequate emergency response capabilities for its materials used or remaining at the BellSouth Facility.
- 1.6 <u>Spills and Releases.</u> When contamination is discovered at a BellSouth Premises, the party discovering the condition must notify BellSouth. All Spills or Releases of regulated materials will immediately be reported by e.spire to BellSouth.
- 1.7 Coordinated Environmental Plans and Permits. BellSouth and e.spire will coordinate plans, permits or information required to be submitted to government agencies, such as emergency response plans, spill prevention control and countermeasures (SPCC) plans and community reporting. If fees are associated with filing, BellSouth and e.spire will develop a cost sharing procedure. If BellSouth's permit or EPA identification number must be used, e.spire must comply with all of BellSouth's permit conditions and environmental processes, including environmental "best management practices (BMP)" (see Section 2, below) and/or selection of BST disposition vendors and disposal sites.
- 1.8 Environmental and Safety Indemnification. BellSouth and e.spire shall indemnify, defend and hold harmless the other party from and against any claims (including, without limitation, third-party claims for personal injury or death or real or personal property damage), judgments, damages, (including direct and indirect damages, and punitive damages), penalties, fines, forfeitures, costs, liabilities, interest and losses arising in connection with the violation or alleged violation of any Applicable Law or contractual obligation or the presence or alleged presence of contamination arising out of the acts or omissions of the indemnifying party, its agents, contractors, or employees concerning its operations at the Facility.

EXHIBIT B Page 3 of 6

2. CATEGORIES FOR CONSIDERATION OF ENVIRONMENTAL ISSUES

When performing functions that fall under the following Environmental categories on BellSouth's Premises, e.spire agrees to comply with the applicable sections of the current issue of BellSouth's Environmental and Safety Methods and Procedures (M&Ps), incorporated herein by this reference. e.spire further agrees to cooperate with BellSouth to ensure that e.spire's employees, agents, and/or subcontractors are knowledgeable of and satisfy those provisions of BellSouth's Environmental M&Ps which apply to the specific Environmental function being performed by e.spire, its employees, agents and/or subcontractors.

The most current version of reference documentation must be requested from BellSouth.

EXHIBIT B Page 4 of 6

2. <u>Categories for Consideration of Environmental Issues</u> (cont.)

ENVIRONMENTAL CATEGORIES	ENVIRONMENTAL ISSUES	ADDRESSED BY THE FOLLOWING DOCUMENTATION
Disposal of hazardous material or other regulated material (e.g., batteries, fluorescent tubes, solvents & cleaning materials)	Pollution liability insurance EVET approval of contractor	Std T&C 450 GU-BTEN-001BT, Chapter 4 Std T&C 660-3 GU-BTEN-001BT, Chapter 10
Emergency response	Hazmat/waste release/spill firesafety emergency	GU-BTEN-001BT, Chapter Building Emergency Operations Plan (EOP) (specific to Premises)
Contract labor/outsourcing for services with environmental implications to be performed on BellSouth Premises (e.g., disposition of hazardous material/waste; maintenance of storage tanks)	Performance of services in accordance with BST's environmental M&Ps Insurance	Std T&C 450 Std T&C 450-B (Contact E/S or your DEC/LDEC for copy of appropriate E/S M&Ps.) Std T&C 660
Transportation of hazardous material	Pollution liability insurance EVET approval of contractor	Std T&C 450 GU-BTEN-001BT, Chapter 4 Std T&C 660-3 GU-BTEN-001BT, Chapter 10
Maintenance/operations work which may produce a waste Other maintenance work	Protection of BST employees and equipment	Std T&C 450 GU-BTEN-001BT, Chapter 10 29CFR 1910.147 29CFR 1910 Subpart O

Version 7/7/00

Janitorial services	All waste removal and	P&SM Manager –				
	disposal must conform to all	Procurement				
	applicable federal, state and	GU-BTEN-001BT, Chapter				
	local regulations	4,				
		GU-BTEN-001BT, Chapter				
	All HazMat & Waste	3				
	Asbestos notification	BSP 010-170-001BS				
	protection of BST	(Hazcom)				
	employees and equipment					
Manhole cleaning	Pollution liability insurance	Std T&C 450				
		Std T&C 660-3				
	Manhole entry requirements	BSP 620-145-011PR				
		Issue A, August 1996				
	EVET approval of contractor	GU-BTEN-001BT, Chapter				
		10				
		RL9706008BT				
Removing or disturbing	Asbestos work practices	GU-BTEN-001BT, Chapter				
building materials that		3				
may contain asbestos						

EXHIBIT B Page 6 of 6

3. <u>DEFINITIONS</u>

<u>Generator</u>. Under RCRA, the person whose act produces a Hazardous Waste, as defined in 40 CFR 261, or whose act first causes a Hazardous Waste to become subject to regulation. The Generator is legally responsible for the proper management and disposal of Hazardous Wastes in accordance with regulations.

<u>Hazardous Chemical</u>. As defined in the U.S. Occupational Safety and Health (OSHA) hazard communication standard (29 CFR 1910.1200), any chemical which is a health hazard or physical hazard.

Hazardous Waste. As defined in section 1004 of RCRA.

<u>Imminent Danger</u>. Any conditions or practices at a facility which are such that a danger exists which could reasonably be expected to cause immediate death or serious harm to people or immediate significant damage to the environment or natural resources.

Spill or Release. As defined in Section 101 of CERCLA.

4. ACRONYMS

<u>DEC/LDEC</u> - Department Environmental Coordinator/Local Department Environmental Coordinator

GU-BTEN-001BT - BellSouth Environmental Methods and Procedures

EVET - Environmental Vendor Evaluation Team

<u>P&SM</u> - Property & Services Management

Std. T&C - Standard Terms & Conditions

NESC - National Electrical Safety Codes

Attachment 5

Access to Numbers and Number Portability

TABLE OF CONTENTS

1.	NON-DISCRIMINATORY ACCESS TO TELEPHONE NUMBERS	3
2.	LOCAL NUMBER PORTABILITY	3
3.	TRANSITION TO PERMANENT NUMBER PORTABILITY	9
4.	TRUE-UP	9
5.	OPERATIONAL SUPPORT SYSTEM (OSS) RATES	10

ACCESS TO NUMBERS and NUMBER PORTABILITY

1. Non-Discriminatory Access to Telephone Numbers

- 1.1 Nothing in this Agreement shall be construed to limit or otherwise adversely affect in any manner either Party's right to employ, or to request and be assigned, any Central Office (NXX) Codes pursuant to the Central Office Code Assignment Guidelines, as may be amended from time to time, or to establish, by Tariff or otherwise, Rate Center and Rating Points corresponding to such NXX Codes.
- 1.2 During the term of this Agreement, the Parties shall contact the applicable numbering resource administrator as determined by the FCC, for the assignment of numbering resources. In order to be assigned a Central Office Code, e.spire will be required to complete the Central Office Code (NXX) Assignment Request and Confirmation Form (Code Request Form) in accordance with Industry Numbering Committee's Central Office Code (NXX) Assignment Guidelines (INC 95-0407-008).
- 1.3 For the purposes of the resale of BellSouth's telecommunications services by e.spire, BellSouth will provide e.spire with on line access to telephone numbers for reservation on a first come first served basis. Such reservations of telephone numbers, on a pre-ordering basis shall be for a period of nine (9) days.
- 1.4 Further, upon e.spire's request and for the purposes of the resale of BellSouth's telecommunications services by e.spire, BellSouth will reserve up to 100 telephone numbers per Common Language Location Identifier Code (CLLIC), for e.spire 's sole use. Such telephone number reservations shall be transmitted to e.spire via electronic file transfer. Such reservations shall be valid for ninety (90) days from the reservation date. e.spire acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth shall use its best efforts to reserve for a ninety (90) day period a sufficient quantity for e.spire 's reasonable need in that particular CLLIC.

2. Local Number Portability

- 2.1 The Parties shall provide local number portability on a reciprocal basis to each other to the extent technically feasible, and in accordance with the applicable rules and regulations as prescribed from time to time by the FCC and/or the Commission.
- 2.2 Permanent Number Portability

- 2.2.1 <u>Deployment of LNP.</u> Local Number Portability ("LNP") is a permanent number portability solution that allows End Users to keep their existing Telephone Line Numbers ("TLNs") when switching LECs. The Parties shall implement and deploy the Location Routing Number ("LRN") solution for LNP in accordance with orders, rulings and policies regarding LNP issued by the FCC and the applicable State Commissions, including, without limitation, the FCC prescribed permanent LNP geographic deployment schedules.
- 2.2.2 <u>Description of LNP</u>. LNP uses the industry standard LRN that assigns a unique 10-digit number to each Wire Center. To support LNP, LRN data is stored, and LNP services are provisioned on Advanced Intelligent Network ("AIN") elements that replace the dialed TLN with the LRN so that LNP calls can be routed to the proper Wire Center for connection to the dialed party. To obtain the LRN data and properly provision LNP services, carriers must be connected to independently operated Regional Number Portability Administration Centers ("NPACs"), which will manage LNP services and provide LNP call routing data to carriers.
- 2.2.3 Once LNP is implemented, either Party may withdraw its Interim Number Portability ("INP") offerings (as described in Section 2.8 hereafter), subject to (i) provision of reasonable advance notice to the other Party; and (ii) coordination to allow the seamless and transparent conversion of INP Customers to LNP.
- 2.2.4 End User Line Charge Recovery of charges associated with implementing Number Portability through a monthly charge assessed to end users has been authorized by the FCC. This end user line charge will be as filed in the BellSouth FCC No. 1 Tariff and will be billed to e.spire where e.spire is a subscriber to local switching or where e.spire is a reseller of BellSouth telecommunications services. This charge will not be discounted.

2.3 <u>Interim Number Portability</u>

2.3.1 <u>Service Provider Number Portability</u>

2.3.1.1 Until the industry-wide permanent solution is implemented in an end office, BellSouth shall provide Service Provider Number Portability ("SPNP"). SPNP is an interim service arrangement whereby an end user who switches subscription of his local exchange service from BellSouth to a CLEC, or vice versa, is permitted to retain the use of his existing assigned telephone number, provided that the end user remains at the same location for his local exchange service or changes locations and service providers but stays within the same Rate Center for his Local Exchange Service.

- 2.3.1.2 SPNP is available through either remote call forwarding or direct inward dialing trunks, at the election of e.spire. Remote call forwarding (SPNP-RCF) is an existing switch-based BellSouth service that redirects calls within the telephone network. Direct inward dialing trunks (SPNP-DID) allow calls to be routed over a dedicated facility to the e.spire switch that serves the subscriber. SPNP-DID Service requires ordering consecutive telephone numbers in blocks of twenty. To order non-consecutive telephone numbers or telephone numbers in less than blocks of twenty, the NBR process must be used. SS7 Signaling is required for the provision of either of these services.
- 2.3.1.3 SPNP-DID is available from BellSouth on a per DS0, DS1, or DS3 basis. Where SPNP-DID is technically feasible and is provided on a DS1 or a DS3 basis, the applicable channelization rates are those specified in Section E6 in BellSouth's Intrastate Access Tariffs, incorporated herein by this reference. SPNP is available only for basic Local Exchange Service.
- 2.3.1.4 SPNP is available only where e.spire or BellSouth is currently providing, or will begin providing concurrent with provision of SPNP, basic Local Exchange Service to the affected End User. SPNP for a particular telephone number is available only from the Central Office originally providing Local Exchange Service to the End User. SPNP for a particular assigned telephone number will be disconnected when any End User, Commission, BellSouth, or e.spire initiated activity (e.g., a change in exchange boundaries) would normally result in a telephone number change had the End User retained his initial Local Exchange Service.
- 2.3.1.5 SPNP-RCF, as contemplated by this Agreement, is a telecommunications service whereby a call dialed to an SPNP-RCF equipped telephone number is automatically forwarded to an assigned seven- or ten- digit telephone number within the local calling area as defined in the Commission filed e.spire or BellSouth Local Exchange Tariff(s) of the Party porting the SPNP-RCF telephone number. The forwarded-to number shall be specified by the e.spire or BellSouth, as appropriate. The forwarding Party will provide identification of the originating telephone number, via SS7 signaling, to the receiving Party. Identification of the originating telephone number to the SPNP-RCF End User cannot be guaranteed, however. SPNP-RCF provides a single call path for the forwarding of no more than one simultaneous call to the receiving Party's specified forwarded-to number. Additional call paths for the forwarding

of multiple simultaneous calls are available on a per path basis at separate rates in addition to the rates for SPNP-RCF.

- 2.3.1.6 SPNP-DID service, as contemplated by this Agreement, provides trunk side access to End Office switches for direct inward dialing to the other company's premises equipment from the telecommunications network to lines associated with the other company's switching equipment and must be provided on all trunks in a group arranged for inward service. A SPNP-DID trunk termination charge, provided with SS7 Signaling only, applies for each trunk voice grade equivalent. In addition, direct facilities are required from the end office where a ported number resides to the end office serving the ported end user customer. The rates for a switched local channel and switched dedicated transport apply as contained in Attachment 2. Transport mileage will be calculated as the airline distance between the end office where the number is ported and the Point of Interface ("POI") using the V&H coordinate method. SPNP-DID must be established with a minimum configuration of two channels and one unassigned telephone number per switch, per arrangement for control purposes. Transport facilities arranged for SPNP-DID may not be mixed with any other type of trunk group, with no outgoing calls placed over said facilities. SPNP-DID will be provided only where such facilities are available and where the switching equipment of the ordering company is properly equipped. Where SPNP-DID service is required from more than one wire center or from separate trunk groups within the same wire center, such service provided from each wire center or each trunk group within the same wire center shall be considered a separate service. Only customer-dialed sent-paid calls will be completed to the first number of a SPNP-DID number group; however, there are no restrictions on calls completed to other numbers of a SPNP-DID number group. Interface group arrangements provided for terminating the switched transport at the Party's terminal location are as set forth in of BellSouth's Intrastate Access Services Tariff, § E6.1.3.A, as amended from time to time.
- 2.3.1.7 The calling Party shall be responsible for payment of the applicable charges for sent-paid calls to the SPNP number. For collect, third-Party, or other operator-assisted non-sent paid calls to the ported telephone number, BellSouth or e.spire shall be responsible for the payment of charges under the same terms and conditions for which the end user would have been liable for those charges. Either company may request that the other block collect and third company non-sent paid calls to the SPNP-assigned telephone number. If a company does not

request blocking, the other company will provide itemized local usage data for the billing of non-sent paid calls on the monthly bill of usage charges provided at the individual end user account level. The detail will include itemization of all billable usage. Each company shall have the option of receiving this usage data on a daily basis via a data file transfer arrangement. This arrangement will utilize the existing industry uniform standard, known as EMR standards, for exchange of billing data. Files of usage data will be created daily for the optional service. Usage originated and recorded in the sending BellSouth RAO will be provided in unrated or rated format, depending on processing system. e.spire usage originated elsewhere and delivered via CMDS to the sending BellSouth RAO shall be provided in rated format.

- 2.3.1.8 Each Party shall be responsible for obtaining authorization from the End User for the handling of the disconnection of the End User's service, the provision of new local service and the provision of SPNP services. Each Party shall be responsible for coordinating the provision of service with the other to assure that its switch is capable of accepting SPNP ported traffic. Each Party shall be responsible for providing equipment and facilities that are compatible with the other's service parameters, interfaces, equipment and facilities and shall be required to provide sufficient terminating facilities and services at the terminating end of an SPNP call to adequately handle all traffic to that location and shall be solely responsible to ensure that its facilities, equipment and services do not interfere with or impair any facility, equipment, or service of the other Party or any of its end users. In the event that either Party determines in its reasonable judgment that the other company will likely impair or is impairing, or interfering with any equipment, facility or service or any of its end users, that company may either refuse to provide SPNP service or may terminate SPNP service to the other Party after providing appropriate notice.
- 2.3.1.9 Each Party shall be responsible for providing an appropriate intercept announcement service for any telephone numbers subscribed to SPNP services for which it is not presently providing local exchange service or terminating to an end user. Where either Party chooses to disconnect or terminate any SPNP service, that Party shall be responsible for designating the preferred standard type of announcement to be provided.
- 2.3.1.10 Each Party shall be the other Party's single point of contact for all repair calls on behalf of each company's end user. Each

- Party reserves the right to contact the other company's customers if deemed necessary for maintenance purposes.
- 2.3.1.11 Neither Party shall be responsible for adverse effects on any service, facility or equipment from the use of SPNP services.

 End-to-end transmission characteristics may vary depending on the distance and routing necessary to complete calls over SPNP facilities and the fact that another carrier is involved in the provisioning of service. Therefore, end-to-end transmission characteristics cannot be specified by either Party for such calls. Neither Party shall be responsible to the other if any necessary change in protection criteria or in any of the facilities, operation, or procedures of either renders any facilities provided by the other company obsolete or renders necessary modification of the other Party's equipment.
- 2.3.1.12 For terminating IXC traffic ported to either Party which requires use of either Party's Tandem switching, the Tandem provider will bill the IXC Tandem switching and a pro rata portion of the transport, and the other Party will bill the IXC local switching, the carrier common line (CCL), the Interconnection Charge and a portion of the transport. If the Tandem provider is unable to provide the necessary access records to permit the other Party to bill the IXC directly for terminating access to ported numbers. then the Tandem provider will bill the IXC full terminating switched access charges at the tandem provider's rate and will compensate the other company at the tandem company's tariffed rates and remit the local switching, the Interconnection Charge, a pro rata portion of transport and CCL revenues to the other Party. If an intraLATA toll call is delivered, the delivering Party will pay terminating access rates to the other Party. This subsection does not apply in cases where SPNP-DID is utilized for number portability. 2.3.1.13 If, through a final and effective order, the Federal Communications Commission ("FCC") issues regulations pursuant to 47 U.S.C. § 251 to require number portability different than that provided pursuant to this section, BellSouth will comply with that order.
- 2.3.1.14 Charges for INP shall be as specified in Exhibit A, provided that interim rates will be replaced or trued-up in accordance with regulatory requirements.

2.4 INP Requirements

2.4.1 Either Party will exchange with the other SS7 TCAP messages as required for the implementation of Customer Local Area Signaling Services (CLASS) or other features available. 2.4.2 Either Party shall notify the

- other of any technical or capacity limitations that would prevent use of a requested INP implementation in a particular End Office or Wire Center.
- 2.4.3 Either Party shall pass all Calling Party Number ("CPN") or Automatic Number Identification ("ANI") information to and from the ported number, whenever technically feasible.
- 2.4.4 Unless approved by e.spire, BellSouth agrees not to issue Telephone Line Number ("TLN") based calling card numbers to End Users that port their numbers to e.spire.
- 2.4.5 BellSouth and e.spire shall cooperate in resolving all service calls involving the other Party's service, to avoid unnecessary service outages.

2.5 Number Portability Through NXX Migration

2.5.1 If the Parties mutually agree to use Local Exchange Routing Guide ("LERG") reassignment as the method to move an End User's telephone numbers from one Party's switch to the other Party's switch in a particular instance, the Parties shall enter into a separate written agreement that must address terms and conditions of the reassignment, including, but not limited to, ordering processes and specific implementation procedures for the reassignment of the appropriate NXX as shown in the LERG, to the new service providers switch, and any applicable rates.

3. Transition to Permanent Number Portability

Once a long-term database method of providing Local Number Portability (LNP) is implemented in an end office pursuant to Federal Communications Commission or State commission orders, rules or regulations, with advance written notice, both Parties must withdraw its Interim Number Portability (INP) offerings. The transition from existing INP arrangements to LNP shall occur within one hundred twenty (120) days from the date LNP is implemented in the end office serving the telephone number. Neither Party shall charge the other Party for conversion from INP to LNP. The Parties shall comply with any INP/LNP transition processes established by the FCC and State commissions and appropriate industry number portability work groups.

Notwithstanding the foregoing, the Parties acknowledge that the FCC has determined once LNP has been deployed pursuant to the FCC's orders, rules and regulations, that all local exchange carriers (LECs) have the duty to provide LNP. Therefore, either Party, at any time, may seek appropriate legal or regulatory relief concerning the transition from INP to LNP or other related issues.

4. True-up

This section applies only to Tennessee.

The interim prices for Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:

- 4.1 The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by a final order (including any appeals) of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions in the General Terms and Conditions and Attachment 1 of this Agreement.
- 4.2. The Parties may continue to negotiate toward final prices, but in the event that no such agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in the General Terms and Conditions and Attachment 1 of the Agreement, so long as they file the resulting Agreement with the Commission as a "negotiated Agreement" under Section 252(e) of the Act.
- 4.3. A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
 - (a) BellSouth and e.spire is entitled to be a full Party to the proceeding;
 - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
 - (c) It shall include as an issue the geographic deaveraging of network element prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

5. Operational Support System (OSS) Rates

All Local Service Requests ("LSRs") submitted for products and services under this Attachment will be subject to the OSS charges set forth in the General Terms and Conditions of this Agreement.

BELLSOUTH/e.spire RATES SERVICE PROVIDER NUMBER PORTABILITY

		RATES BY STATE								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF (1) (2)										
RCF, per number ported (Business Line), 10 paths	TNPBL	NA	NA	NA	NA	NA	NA	\$2.25	NA	NA
RCF, per number ported (Residence Line), 6 paths	TNPRL	NA	NA	NA	NA	NA	NA	\$1.15	NA	NA
RCF, per number ported (Business Line)	TNPBL	\$2.13	NA	\$2.03	NA	\$2.29	\$2.34	NA	\$2.17	\$1.50
NRC	TNPBL	\$0.65	NA	\$0.51	NA	\$0.49	\$0.6441	NA	\$0.7046	NA
NRC - Disconnect Charge	TNPBL	\$0.07	NA	NA	NA	\$0.05	\$0.0644	\$0.50	NA	NA
RCF, per number ported (Residence Line)	TNPRL	\$2.13	NA	\$2.03	NA	\$2.29	\$2.34	NA	\$2.17	\$1.25
NRC	TNPRL	\$0.65	NA	\$0.51	NA	\$0.49	\$0.6441	NA	\$0.7046	NA
NRC - Disconnect Charge	TNPRL	\$0.07	NA	NA	NA	\$0.05	\$0.0644	\$0.50	NA	NA
RCF, add'l capacity for simultaneous call forwarding, per additional path	N/A	\$0.32	NA	\$0.2836	NA	\$0.38	\$0.3838	NA	\$0.3854	\$0.50
· · · · · · · · · · · · · · · · · · ·	(++) Bus = TNPBD									
RCF, per service order, per location	Res = TNPRD									
NRC - 1st	TNP++	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	None	\$1.37	\$25.00
NRC - Add'l	TNP++	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	None	\$1.37	\$25.00
NRC - Disconnect - 1st	TNP++	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Disconnect - Add'l	TNP++	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	NA	NA	\$18.14	\$25.52	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$27.37	NA	NA	NA	\$18.14	\$25.52	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	\$44.70	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	\$44.70	NA
INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID										
DID per number ported, Residence - NRC	TNPDR	\$1.18	NA	\$0.93	NA	\$0.89	\$1.17	NA	\$2.25	NA
DID per number ported, Residence - NRC - Disconnect	TNPDR	\$1.18	NA	NA	NA	\$0.90	\$1.17	NA	NA	NA
DID per number ported, Business - NRC	TNPDB	\$1.18	NA	\$0.93	NA	\$0.89	\$1.17	NA	\$2.25	NA
DID per number ported, Business - NRC - Disconnect	TNPDB	\$1.18	NA	NA	NA	\$0.90	\$1.17	NA	NA	NA
DID per service order, per location										
NRC - 1st	TNPRD	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	NA	\$1.37	NA
NRC - Add'l	TNPRD	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	NA	\$1.37	NA
NRC - Disconnect - 1st	TNPRD	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	\$44.70	NA
NRC - Disconnect - Add'l	TNPRD	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	\$44.70	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$27.37	NA	NA	NA	\$18.14	\$25.52	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
DID, per trunk termination, Initial	TNPT2	\$11.84	NA	\$10.73	NA	\$12.46	\$13.78	NA	\$13.16	NA
DID, per trunk termination, Initial - NRC	TNPT2	\$173.73	NA	\$135.47	NA	\$129.69	\$171.68	NA	\$218.03	NA
DID, per trunk termination, Initial - Disconnect	TNPT2	\$50.43	NA	NA	NA	\$37.85	\$49.86	NA	NA	NA
DID, per trunk termination, Subsequent	TNPT2	\$11.84	NA	\$10.73	NA	\$12.46	\$13.78	NA	\$13.16	NA
DID, per trunk termination, Subsequent - NRC	TNPT2	\$51.35	NA	\$39.53	NA	\$37.85	\$50.69	NA	\$73.63	NA
DID, per trunk termination, Subsequent - Disconnect	TNPT2	\$25.00	NA	NA	NA	\$18.75	\$24.71	NA	NA	NA

NOTES:

If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

- 1 Until the FCC issues its order implementing a cost recovery mechanism for permanent number portability, the Company will track its costs of providing interim SPNP with sufficient detail to verify the costs. This will facilitate the Florida PSCs consideration of the recovery of these costs in Docket 950737-TP. (FL)
- 2 BellSouth and CLEC will each bear their own costs of providing remote call forwarding as an interim number portability option. (KY)

Attachment 6

Ordering and Provisioning

TABLE OF CONTENTS

1.	QUALITY OF ORDERING AND PROVISIONING	3
	<u> </u>	
2.	ACCESS TO OPERATIONAL SUPPORT SYSTEMS	. 4
		•
3	MISCELLANEOUS ORDERING AND PROVISIONING GUIDELINES	6
~•	middler areddd diwering / are i ita ridiaithig ddieleited iiiniiniiniiniiniiniinii	

ORDERING AND PROVISIONING

1. Quality of Ordering and Provisioning

- 1.1 BellSouth shall provide ordering and provisioning services to e.spire that are equal to the ordering and provisioning services BellSouth provides to itself or any other CLEC. Detailed guidelines for ordering and provisioning are set forth in BellSouth's Local Interconnection and Facility Based Ordering Guide and Resale Ordering Guide, as appropriate, and as they are amended from time to time during this Agreement. To the extent the Terms of this Attachment differ from BellSouth's Local Interconnection and Facilities Based Ordering Guide and Resale Guide, the rate terms and conditions of this Attachment shall take precedence.
- 1.2 BellSouth will perform provisioning services in the UNE Centers during the following normal hours of operation:

Monday - Friday - 8:00AM - 5:00PM Eastern time (excluding holidays)

(Resale/Network Element non coordinated, coordinated orders and order coordinated - Time Specific)

Saturday - 8:00 AM - 5:00 PM Eastern time (excluding holidays)
(Resale/Network Element non coordinated orders)

Field provisioning services shall be performed on the same schedule as set forth above; provided, however, that times shall be based upon location time rather than Eastern time.

All other e.spire requests for provisioning and installation services are considered outside of the normal hours of operation and may be performed subject to the application of overtime billing charges at the labor rates set forth in BellSouth's FCC No. 1 tariff.

If BellSouth begins working on an order which is scheduled to be completed during standard hours, but, due solely to BellSouth's delay, completes the work after standard hours, no such additional charges shall apply. If e.spire requests such provisioning services outside of normal hours of operation, BellSouth shall quote within three (3) Business Days of the request, a rate for such services in accordance with BellSouth's FCC No. 1 tariff. If e.spire accepts BellSouth's quote, BellSouth shall provide the requested services. If BellSouth agrees to provide expanded standard coverage hours to any other Telecommunications Carrier, e.spire shall be able immediately to avail itself of the same expanded hours on the same terms as made available to such other Telecommunications Carrier.

2. Access to Operational Support Systems

- 2.1 BellSouth shall provide e.spire with access to OSS pre-order functions at parity to that provided by BellSouth to itself, its Affiliates, or any other Telecommunications Carrier. Access to these support systems is available through a variety of means, including electronic interfaces. BellSouth also provides the option of placing orders manually (e.g., via facsimile) through the Local Carrier Service Center. The operations support systems available are:
- Pre-Ordering. Pre-ordering includes the activities undertaken by e.spire to gather and verify information necessary to formulate an accurate order for End Users. BellSouth provides electronic access to the following pre-ordering functions: service address validation, telephone number selection, including vanity number selection and reservations, service and feature availability at serving wire center, due date information, serving facilities information and Customer Service Record ("CSR") information. Access is provided through the Local Exchange Navigation System (LENS) and the Telecommunications Access Gateway (TAG). CSR information includes any and all customer specific information, including but not limited to, customer specific information in CRIS and RSAG. e.spire agrees not to view, copy, or otherwise obtain access to the CSR of any customer without that customer's permission and further agrees that e.spire will obtain access to CSR information only in strict compliance with applicable FCC Rules and Orders and other laws, rules, or regulations of the State in which the service is provided.
- 2.2.1 <u>Interfaces</u>. BellSouth shall make available the following interfaces to e.spire for access to pre-order functions: LENS; and TAG. Each such interface shall be available on a non-discriminatory basis in connection with pre-ordering for Resale services and UNES that are available electronically.
- 2.2.2 The Parties acknowledge that ordering requirements necessitate the use of current, real time pre-order information to accurately build service orders. Each pre-order interface shall be available except for downtime attributable to maintenance and upload, twenty-four (24) hours a day, seven (7) days a week.
- e.spire shall be permitted to reserve a number, including, without limitation, a vanity number, for up to thirty (30) days for End Users.
- 2.2.4 All CSR data exchanged must be in English text, and not only USOC or FID format, provided that such information is maintained in textual format by BellSouth. All other data shall be in a mutually agreed upon nomenclature.
- 2.2.5 Upon request, BellSouth shall provide e.spire with pre-order information in batch transmission to the extent available or provided to any other Telecommunications Carrier on the same terms and conditions and at the same rates.
- 2.2.6 Pre-ordering functions shall be provided at parity as measured by the Performance Measurement metrics included in Attachment 9 hereto.

- Service Ordering and Provisioning. BellSouth provides electronic options for the exchange of ordering and provisioning information. BellSouth provides an Electronic Data Interchange (EDI) arrangement for certain resale requests and certain network elements and other services. The EDI interface can be integrated with the TAG pre-ordering interface by e.spire. As an alternative to the EDI arrangement, BellSouth also provides ordering and provisioning capability through TAG that can be integrated with the TAG pre-ordering capability by e.spire. Also, as an alternative, BellSouth provides integrated pre-ordering, ordering, and provisioning capability through the LENS interface. Ordering and provisioning intervals shall be at provisioned at parity to what BellSouth provides to itself, its affiliates and/or other CLECs. Parity in performance shall be measured by the performance measurement metrics set forth in Attachment 9.
- 2.3.1 For generation of Resale service orders, ordering flows shall be available via such electronic interfaces for each of the following ordering functions: Conversion ("as is" or "with changes"); Change (features, listings, long distance); New Connect; Disconnect; From and To (change of premises with same service).
- 2.3.2 BellSouth shall provide to e.spire an electronic interface for transmitting of orders, and receiving Firm Order Confirmation ("FOC"), completion notices, Due-Date Jeopardies, and, as available, other provisioning data and information. BellSouth shall provide e.spire with a FOC for each Resale and UNE order. The FOC includes: purchase order number, telephone number, Local Service Request number, due date, and Service Order number.
- 2.3.3 BellSouth shall provision Resale Services and UNEs as prescribed in e.spire service order requests. Access to status on electronically-submitted Resale services and UNEs shall be provided via the electronic interfaces. Access to status on manually-submitted service order requests shall be provided manually or via the Purchase Order Number ("PON") report on the internet.
- 2.3.4 BellSouth shall provide notice of a lack of facilities availability at parity to that BellSouth provides to itself, its Affiliates, or any other Telecommunications Carrier.
- 2.3.5 Order Flow Through. "Order Flow Through" is defined as the process whereby espire's orders are transmitted electronically through the gateway and accepted into BellSouth's back office order systems without manual intervention. BellSouth shall provide Flow Through of electronic processes in a manner consistent with, at a minimum, at a level of quality equivalent to itself or to any CLEC with comparable systems.
- 2.4 <u>Service Trouble Reporting and Repair</u>. Service trouble reporting and repair allows e.spire to report and monitor service troubles and obtain repair services. BellSouth shall offer e.spire service trouble reporting in a non-discriminatory manner that provides e.spire the equivalent ability to report and monitor service troubles that BellSouth provides to itself. BellSouth also provides e.spire an estimated time to repair, an appointment time or a commitment time, as

appropriate, on trouble reports. BellSouth provides two options for electronic trouble reporting. For exchange services, BellSouth offers e.spire access to the Trouble Analysis Facilitation Interface (TAFI). For individually designed services, BellSouth provides electronic trouble reporting through an electronic communications gateway. If the CLEC requests BellSouth to repair a trouble after normal working hours, the CLEC will be billed the appropriate overtime charges associated with this request pursuant to BellSouth's tariffs.

- 2.5 <u>Migration of e.spire to New BellSouth Software Releases.</u> BellSouth will issue new software releases for its electronic interfaces as needed to improve operations and meet standards and regulatory requirements. When a new release is implemented, BellSouth will continue to support both the new release (N) and the prior release (N-1). When BellSouth makes the next release (N+1), BellSouth will eliminate support for the (N-1) release and support the two newest releases (N and N+1). Thus, BellSouth will always support the two most current releases.
- 2.5.1 BellSouth will issue documents to e.spire with sufficient notice to allow e.spire to make the necessary changes to their systems and operations to migrate to the newest release in a timely fashion.
- 2.5.2 With respect to any modification or discontinuation that materially affects e.spire's use of such interface, BellSouth shall provide e.spire with advance notice of such modification or discontinuation consistent with applicable FCC requirements.
- 2.6 <u>Rates.</u> Charges for use of Operational Support Systems shall be as set forth in the General Terms and Conditions of this Agreement.

3. <u>Miscellaneous Ordering and Provisioning Guidelines</u>

- Pending Orders. To ensure the most efficient use of facilities and resources, orders placed in the hold or pending status by e.spire will be held for a maximum of thirty (30) days from the date the order is placed on hold. After such time, if e.spire wishes to reinstate an order, e.spire may be required to submit a new service order. If an e.spire order is placed on hold by BellSouth then e.spire will not have to submit a new order.
- Single Point of Contact. e.spire will be the single point of contact with BellSouth for ordering activity for network elements and other services used by e.spire to provide services to its end users. BellSouth may accept an order directly from another CLEC, or BellSouth, acting with authorization of the affected end user. e.spire and BellSouth shall each execute a blanket letter of authorization with respect to customer orders. The Parties shall each be entitled to adopt their own internal processes for verification of customer authorization for orders, provided, however, that such processes shall comply with applicable state and federal law including, until superseded, the FCC guidelines and orders applicable to Presubscribed Interexchange Carrier (PIC) changes. Pursuant to such an order, BellSouth may disconnect any network element associated with the service to be

disconnected and being used by e.spire to provide service to that end user and reuse such network elements or facilities to enable such other LEC to provide service to the end user. BellSouth will notify e.spire that such an order has been processed, but will not be required to notify e.spire in advance of such processing. BellSouth will notify e.spire within (2) business days via OUTPLOC that such disconnect has been completed.

- 3.3 <u>Use of Facilities</u>. When a customer of e.spire elects to discontinue service and transfer service to another local exchange carrier, including BellSouth, BellSouth shall have the right to reuse the facilities provided to e.spire by BellSouth for retail or resale service, loop and/or port for that customer. In addition, BellSouth may disconnect and reuse facilities when the facility is in a denied state and BellSouth has received an order to disconnect or transfer the e.spire service.
- 3.3.1 Upon receipt of a service order, BellSouth will do the following:
- 3.3.1.1 Process disconnect and reconnect orders to provision the service which shall be due dated using nondiscriminatory procedures and intervals which are at parity to the provisioning intervals BellSouth provides itself or other CLECs;
- 3.3.1.2 Reuse the serving facility for the retail, resale service, or network element at the same location; and
- 3.3.1.3 Notify e.spire subsequent to the disconnect order being completed in accordance with Section 3.2 above.

3.4 Contact Numbers.

The Parties agree to provide one another with toll-free contact numbers for the purpose of ordering, provisioning and maintenance of services. BellSouth shall provide single points of contact ("SPOC") for the provisioning of Resale Services (LCSC) and UNEs (UNE Center) ordered by e.spire. Pre-ordering and ordering shall be available via an electronic interface seven (7) days a week, 24 hours a day.

BellSouth shall provide access to assistance for technical issues such as connectivity and passwords related to LENS, TAG and TAFI, and to the "EDI Central Group" for technical problems with EDI. Assistance will be available by telephone during normal business hours and through other contacts on nights, weekends and holidays.

- 3.5 <u>Disaster Recovery Plan.</u> BellSouth's Disaster Recovery Plan is as set forth in Exhibit A of this Attachment.
- <u>Subscription Functions.</u> In cases where BellSouth performs subscription functions for an inter-exchange carrier (i.e. PIC and LPIC changes via Customer Account Record Exchange (CARE)), BellSouth will provide the affected inter-exchange carriers with the Operating Company Number (OCN) of the local

provider for the purpose of obtaining end user billing account and other end user information required under subscription requirements.

- 3.7 <u>Cancellation Charges.</u> If e.spire cancels an order for network elements or other services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with FCC No. 1 Tariff, Section 5.4.
- 3.8 <u>Ordering and Provisioning Information:</u> BellSouth shall provide the following to e.spire upon request:
- 3.8.1 Design Layout Records ("DLRs") for designed unbundled Network Elements;
- 3.8.2 Advance information on the details and requirements for planning and implementation of NPA splits; and
- 3.8.3 Access to the Regional Street Address Guide ("RSAG") information via LENS or TAG pre-ordering.
- 3.9 BellSouth and e.spire shall establish mutually acceptable methods and procedures for handling all misdirected calls from e.spire End Users. All misdirected calls to BellSouth from e.spire End Users shall be given a recording (or a live statement) directing them to call an e.spire-designated toll free number. e.spire, on a reciprocal basis, shall refer all misdirected calls that e.spire receives from BellSouth End Users to a BellSouth-designated number. e.spire and BellSouth each shall be responsible for providing the other party with its current toll free number. The foregoing shall apply only when the Party receiving such call knows or has reason to know that the call is misdirected from an End User of the other Party hereto.
- 3.10 BellSouth shall provide order format specifications to e.spire for all available services, features, and functions and for ancillary data required by BellSouth to provision these services.
- 3.11 BellSouth shall provide e.spire with standard expected provisioning intervals for all unbundled Network Elements.
- 3.12 BellSouth shall not reconfigure any e.spire service rearrangements of any e.spire End User for Resale services, UNEs or Combinations, unless so directed by e.spire. Any e.spire End User that contacts BellSouth regarding a change to its e.spire service (excluding changes in its local service provider) shall be advised to contact e.spire. Any BellSouth End User that contacts e.spire regarding a change in BellSouth service (excluding changes in its local service provider) shall be advised to contact BellSouth.
- 3.13 The Parties shall provide a generic intercept referral message that includes any new telephone number of an End User for the same period of time that BellSouth currently provides such a message for its own End Users. The intercept message shall be similar in format to the intercept referral message currently provided by BellSouth for its own End Users.

- 3.14 BellSouth shall perform all pre-testing necessary to ensure the services ordered meet the specifications outlined in the technical service description provided by BellSouth for the service being ordered.
- Any written "leave behind" materials that BellSouth technicians provide to e.spire End Users shall be non-branded materials that do not identify the work being performed as being by BellSouth. These materials shall include, without limitation, non-branded forms for the Customer and non-branded "not at home" cards.
- 3.16 If an e.spire End User requests a change of service at the time of installation, BellSouth technicians shall direct them to contact e.spire directly and provide a toll-free number supplied by e.spire. When a BellSouth employee visits the premise of an e.spire End User, the BellSouth employee shall inform the Customer that he or she is acting on behalf of e.spire.
- 3.17 BellSouth shall provide telephone and/or facsimile notification of any e.spire end user service request and charges therefore not authorized on the e.spire service request, and obtain e.spire's approval prior to commencing work.
- 3.18 Each Party shall train and direct its employees who have contact with End Users of the other Party in the process of provisioning, maintenance or repair not to disparage the other Party or its services in any way to the other Party's End Users.
- When e.spire places an LSR, e.spire shall specify a requested Due Date, and BellSouth shall specify a Due Date based on the applicable intervals. In the event e.spire's requested date is less than the standard interval, e.spire shall contact BellSouth by telephone and the Parties shall negotiate an expedited Due Date. This situation shall be considered an expedited order for which expedite charges will apply in accordance with BellSouth FCC No. 1 Tariff. BellSouth shall not complete the order prior to the Due Date unless authorized by e.spire. If BellSouth misses the Due Date, BellSouth shall promptly notify e.spire of the revised installation Due Date. If e.spire requests that an order be expedited, BellSouth shall notify e.spire of the status of the order (i) by the end of the same Business Day when such expedite requests are made prior to noon; or (ii) by noon the following Business Day otherwise.
- 3.20 e.spire and BellSouth shall agree to escalation procedures and contacts for resolving questions and disputes related to ordering and provisioning procedures or to the processing of individual orders, subject ultimately to the dispute resolution provisions of this Agreement. The Parties shall use best efforts to notify each other of any modifications to these contacts within ten (10) days of any such modifications.
- 3.21 BellSouth shall transmit to e.spire a FOC or, in the alternative, notification of the lack of available facilities within time periods specified hereafter after BellSouth's receipt of a complete and correct order from e.spire, provided, however, that an order for complex services requiring a service inquiry shall be

deemed received for these purposes only after completion of the service inquiry. The FOC shall contain a commitment date, which shall be established on a nondiscriminatory basis with respect to installation dates for comparable orders at such time. If e.spire uses LENS, EDI, or any other electronic interface for the submission of the order, the FOC or notification shall be posted by BellSouth in such interface within twenty four (24) hours of receipt of the order. If e.spire does not use these interfaces, or these interfaces are not available for the service or UNE being ordered, BellSouth shall transmit the FOC or notification by telecopier to a toll-free number provided by e.spire within forty eight (48) hours of BellSouth's receipt of the order. When e.spire submits a complete and correct LSR for SPNP and an associated unbundled Loop simultaneously, BellSouth shall likewise issue a FOC for both the Loop and the SPNP simultaneously.

For Local Service Requests submitted via an electronic interface, BellSouth shall notify e.spire via the same electronic interface, of Rejections/Errors contained in any of the data element(s) field(s) contained on any e.spire Local Service Request. For Local Service Requests submitted manually, BellSouth shall notify e.spire by facsimile of such Rejections and Errors. BellSouth will notify e.spire of Rejections or Errors in 95% of mechanized orders within one (1) hour from BellSouth's receipt of the order. BellSouth will notify e.spire of Rejections or Errors in 85% of non-mechanized and partially mechanized orders within forty-eight (48) hours from BellSouth's receipt of the order.

Exhibit A Attachment 6 Page 1

BellSouth Disaster Recovery Plan

2000 BELLSOUTH

DISASTER RECOVERY PLANNING

For

CLECS

9

10

CONTENTS PAGE 1.0 Purpose 4 2.0 Single Point of Contact 4 3.0 Identifying the Problem 4 Site Control 5 3.1 3.2 **Environmental Concerns** 6 4.0 The Emergency Control Center (ECC) 6 5.0 Recovery Procedures 7 5.1 CLEC Outage 7 5.2 BellSouth Outage 7 5.2.1 Loss of Central Office 8 5.2.2 Loss of a Central Office with Serving Wire Center Functions 8 5.2.3 Loss of a Central Office with Tandem Functions 8 9 5.2.4 Loss of a Facility Hub 5.3 Combined Outage (CLEC and BellSouth Equipment 9

6.0 T1 Identification Procedures

7.0 Acronyms

1.0 PURPOSE

In the unlikely event of a disaster occurring that affects BellSouth's long-term ability to deliver traffic to a Competitive Local Exchange Carrier (CLEC), general procedures have been developed to hasten the recovery process. Since each location is different and could be affected by an assortment of potential problems, a detailed recovery plan is impractical. However, in the process of reviewing recovery activities for specific locations, some basic procedures emerge that appear to be common in most cases.

These general procedures should apply to any disaster that affects the delivery of traffic for an extended time period. Each CLEC will be given the same consideration during an outage and service will be restored as quickly as possible.

This document will cover the basic recovery procedures that would apply to every CLEC.

2.0 SINGLE POINT OF CONTACT

When a problem is experienced, regardless of the severity, the BellSouth Network Management Center (NMC) will observe traffic anomalies and begin monitoring the situation. Controls will be appropriately applied to insure the sanity of BellSouth's network; and, in the event that a switch or facility node is lost, the NMC will attempt to circumvent the failure using available reroutes.

BellSouth's NMC will remain in control of the restoration efforts until the problem has been identified as being a long-term outage. At that time, the NMC will contact BellSouth's Emergency Control Center (ECC) and relinquish control of the recovery efforts. Even though the ECC may take charge of the situation, the NMC will continue to monitor the circumstances and restore traffic as soon as damaged network elements are revitalized.

The telephone number for the BellSouth Network Management Center in Atlanta, as published in Telcordia's National Network Management Directory, is 404-321-2516.

3.0 IDENTIFYING THE PROBLEM

During the early stages of problem detection, the NMC will be able to tell which CLECs are affected by the catastrophe. Further analysis and/or first hand observation will determine if the disaster has affected CLEC equipment only; BellSouth equipment only or a combination. The initial restoration activity will be largely determined by the equipment that is affected.

Once the nature of the disaster is determined and after verifying the cause of the problem, the NMC will initiate reroutes and/or transfers that are jointly agreed upon by the affected CLECs' Network Management Center and the BellSouth NMC. The type and percentage of controls used will depend upon available network capacity. Controls necessary to stabilize the situation will be invoked and the NMC will attempt to re-establish as much traffic as possible.

For long term outages, recovery efforts will be coordinated by the Emergency Control Center (ECC). Traffic controls will continue to be applied by the NMC until facilities are re-established. As equipment is made available for service, the ECC will instruct the NMC to begin removing the controls and allow traffic to resume.

3.1 SITE CONTROL

In the total loss of building use scenario, what likely exists will be a smoking pile of rubble. This rubble will contain many components that could be dangerous. It could also contain any personnel on the premises at the time of the disaster. For these reasons, the local fire marshal with the assistance of the police will control the site until the building is no longer a threat to surrounding properties and the companies have secured the site from the general public.

During this time, the majority owner of the building should be arranging for a demolition contractor to mobilize to the site with the primary objective of reaching the cable entrance facility for a damage assessment. The results of this assessment would then dictate immediate plans for restoration, both short term and permanent.

In a less catastrophic event, i.e., the building is still standing and the cable entrance facility is usable, the situation is more complex. The site will initially be controlled by local authorities until the threat to adjacent property has diminished. Once the site is returned to the control of the companies, the following events should occur.

An initial assessment of the main building infrastructure systems (mechanical, electrical, fire & life safety, elevators, and others) will establish building needs. Once these needs are determined, the majority owner should lead the building restoration efforts. There may be situations where the site will not be totally restored within the confines of the building. The companies must individually determine their needs and jointly assess the cost of permanent restoration to determine the overall plan of action.

Multiple restoration trailers from each company will result in the need for designated space and installation order. This layout and control is required to maximize the amount of restoration equipment that can be placed at the site, and the priority of placements.

Care must be taken in this planning to insure other restoration efforts have logistical access to the building. Major components of telephone and building equipment will need to be removed and replaced. A priority for this equipment must also be jointly established to facilitate overall site restoration. (Example: If the AC switchgear has sustained damage, this would be of the highest priority in order to regain power, lighting, and HVAC throughout the building.)

If the site will not accommodate the required restoration equipment, the companies would then need to quickly arrange with local authorities for street closures, rights of way or other possible options available.

3.2 ENVIRONMENTAL CONCERNS

In the worse case scenario, many environmental concerns must be addressed. Along with the police and fire marshal, the state environmental protection department will be on site to monitor the situation.

Items to be concerned with in a large central office building could include:

- 1. Emergency engine fuel supply. Damage to the standby equipment and the fuel handling equipment could have created "spill" conditions that have to be handled within state and federal regulations.
- 2. Asbestos containing materials that may be spread throughout the wreckage. Asbestos could be in many components of building, electrical, mechanical, outside plant distribution, and telephone systems.
- 3. Lead and acid. These materials could be present in potentially large quantities depending upon the extent of damage to the power room.
- 4. Mercury and other regulated compounds resident in telephone equipment.
- 5. Other compounds produced by the fire or heat.

Once a total loss event occurs at a large site, local authorities will control immediate clean up (water placed on the wreckage by the fire department) and site access.

At some point, the companies will become involved with local authorities in the overall planning associated with site clean up and restoration. Depending on the clean up approach taken, delays in the restoration of several hours to several days may occur.

In a less severe disaster, items listed above are more defined and can be addressed individually depending on the damage.

In each case, the majority owner should coordinate building and environmental restoration as well as maintain proper planning and site control.

4.0 THE EMERGENCY CONTROL CENTER (ECC)

The ECC is located in the Colonnade Building in Birmingham, Alabama. During an emergency, the ECC staff will convene a group of pre-selected experts to inventory the damage and initiate corrective actions. These experts have regional access to BellSouth's personnel and equipment and will assume control of the restoration activity anywhere in the nine-state area.

In the past, the ECC has been involve with restoration activities resulting from hurricanes, ice storms and floods. They have demonstrated their capabilities during these calamities as well as

during outages caused by human error or equipment failures. This group has an excellent record of restoring service as quickly as possible.

During a major disaster, the ECC may move emergency equipment to the affected location, direct recovery efforts of local personnel and coordinate service restoration activities with the CLECs. The ECC will attempt to restore service as quickly as possible using whatever means is available; leaving permanent solutions, such as the replacement of damaged buildings or equipment, for local personnel to administer.

Part of the ECC's responsibility, after temporary equipment is in place, is to support the NMC efforts to return service to the CLECs. Once service has been restored, the ECC will return control of the network to normal operational organizations. Any long-term changes required after service is restored will be made in an orderly fashion and will be conducted as normal activity.

5.0 RECOVERY PROCEDURES

The nature and severity of any disaster will influence the recovery procedures. One crucial factor in determining how BellSouth will proceed with restoration is whether or not BellSouth's equipment is incapacitated. Regardless of who's equipment is out of service, BellSouth will move as quickly as possible to aid with service recovery; however, the approach that will be taken may differ depending upon the location of the problem.

5.1 CLEC OUTAGE

For a problem limited to one CLEC (or a building with multiple CLECs), BellSouth has several options available for restoring service quickly. For those CLECs that have agreements with other CLECs, BellSouth can immediately start directing traffic to a provisional CLEC for completion. This alternative is dependent upon BellSouth having concurrence from the affected CLECs.

Whether or not the affected CLECs have requested a traffic transfer to another CLEC will not impact BellSouth's resolve to re-establish traffic to the original destination as quickly as possible.

5.2 BELLSOUTH OUTAGE

Because BellSouth's equipment has varying degrees of impact on the service provided to the CLECs, restoring service from damaged BellSouth equipment is different. The outage will probably impact a number of Carriers simultaneously. However, the ECC will be able to initiate immediate actions to correct the problem.

A disaster involving any of BellSouth's equipment locations could impact the CLECs, some more than others. A disaster at a Central Office (CO) would only impact the delivery of traffic to and from that one location, but the incident could affect many Carriers. If the Central Office is a Serving Wire Center (SWC), then traffic from the entire area to those Carriers served from that switch would also be impacted. If the switch functions as an Access Tandem, or there is a tandem in the building, traffic from every CO to every CLEC could be interrupted. A disaster that destroys a facility hub could disrupt various traffic flows, even though the switching equipment may be unaffected.

The NMC would be the first group to observe a problem involving BellSouth's equipment. Shortly after a disaster, the NMC will begin applying controls and finding re-routes for the

completion of as much traffic as possible. These reroutes may involve delivering traffic to alternate Carriers upon receiving approval from the CLECs involved. In some cases, changes in translations will be required. If the outage is caused by the destruction of equipment, then the ECC will assume control of the restoration.

5.2.1 Loss of a Central Office

When BellSouth loses a Central Office, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Begin restoring service to CLECs and other customers.

5.2.2 Loss of a Central Office with Serving Wire Center Functions

The loss of a Central Office that also serves as a Serving Wire Center (SWC) will be restored as described in section 5.2.1.

5.2.3 Loss of a Central Office with Tandem Functions

When BellSouth loses a Central Office building that serves as an Access Tandem and as a SWC, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies;
- e) Re-direct as much traffic as possible to the alternate access tandem (if available) for delivery to those CLECs utilizing a different location as a SWC;
- f) Begin aggregating traffic to a location near the damaged building. From this location, begin re-establishing trunk groups to the CLECs for the delivery of traffic normally found on the direct trunk groups. (This aggregation point may be the alternate access tandem location or another CO on a primary facility route.)
- g) Begin restoring service to CLECs and other customers.

5.2.4 Loss of a Facility Hub

In the event that BellSouth loses a facility hub, the recovery process is much the same as above. Once the NMC has observed the problem and administered the appropriate controls, the ECC will assume authority for the repairs. The recovery effort will include

- a) Placing specialists and emergency equipment on notice;
- b) Inventorying the damage to determine what equipment and/or functions are lost;
- c) Moving containerized emergency equipment to the stricken area, if necessary;
- d) Reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Restoring service to CLECs and other customers. If necessary, BellSouth will aggregate the traffic at another location and build temporary facilities. This alternative would be viable for a location that is destroyed and building repairs are required.

5.3 COMBINED OUTAGE (CLEC AND BELLSOUTH EQUIPMENT)

In some instances, a disaster may impact BellSouth's equipment as well as the CLECs'. This situation will be handled in much the same way as described in section 5.2.3. Since BellSouth and the CLECs will be utilizing temporary equipment, close coordination will be required.

6.0 T1 IDENTIFICATION PROCEDURES

During the restoration of service after a disaster, BellSouth may be forced to aggregate traffic for delivery to a CLEC. During this process, T1 traffic may be consolidated onto DS3s and may become unidentifiable to the Carrier. Because resources will be limited, BellSouth may be forced to "package" this traffic entirely differently then normally received by the CLECs. Therefore, a method for identifying the T1 traffic on the DS3s and providing the information to the Carriers is required.

7.0 ACRONYMS

CO - Central Office (BellSouth)

DS3 - Facility that carries 28 T1s (672 circuits)

ECC - Emergency Control Center (BellSouth)

CLEC - Competitive Local Exchange Carrier

NMC - Network Management Center

SWC - Serving Wire Center (BellSouth switch)

T1 - Facility that carries 24 circuits

Hurricane Information

During a hurricane, BellSouth will make every effort to keep CLECs updated on the status of our network. Information centers will be set up throughout BellSouth Telecommunications. These centers are not intended to be used for escalations, but rather to keep the CLEC informed of network related issues, area damages and dispatch conditions, etc.

Hurricane-related information can also be found on line at http://www.interconnection.bellsouth.com/network/disaster/dis_resp.htm. Information concerning Mechanized Disaster Reports can also be found at this website by clicking on CURRENT MDR REPORTS or by going directly to http://www.interconnection.bellsouth.com/network/disaster/mdrs.htm.

BST Disaster Management Plan

BellSouth maintenance centers have geographical and redundant communication capabilities. In the event of a disaster removing any maintenance center from service another geographical center would assume maintenance responsibilities. The contact numbers will not change and the transfer will be transparent to the CLEC.

Attachment 7

Billing and Billing Accuracy Certification

TABLE OF CONTENTS

1.	PAYMENT AND BILLING ARRANGEMENTS	3
2.	BILLING AND BILLING ACCURACY CERTIFICATION	5
3.	BILLING DISPUTES	6
4.	RAO HOSTING	7
5.	OPTIONAL DAILY USAGE FILE	10
6.	ACCESS DAILY USAGE FILE	14
7.	ENHANCED OPTIONAL DAILY USAGE FILE	17
FYL	HRIT A RATES	

BILLING AND BILLING ACCURACY CERTIFICATION

1. PAYMENT AND BILLING ARRANGEMENTS

- Billing. BellSouth agrees to provide billing through the Carrier Access Billing System (CABS) and through the Customer Records Information System (CRIS) depending on the particular service(s) that e.spire requests. BellSouth will bill and record in accordance with this Agreement those charges e.spire incurs as a result of e.spire purchasing from BellSouth Network Elements and Other Services as set forth in this Agreement. BellSouth will format all bills in CBOS Standard or CLUB/EDI format, depending on the type of service ordered. For those services where standards have not yet been developed, BellSouth's billing format will change as necessary when standards are finalized by the industry forum.
- 1.1.1 For any service(s) BellSouth orders from e.spire, e.spire shall bill BellSouth in CABS format or in accordance with industry standards.
- 1.1.2 If either Party requests multiple billing media or additional copies of bills, the Billing Party will provide these at a reasonable cost.
- Master Account. After receiving certification as a local exchange company from the appropriate regulatory agency, e.spire will provide the appropriate BellSouth account manager the necessary documentation to enable BellSouth to establish a master account for Local Interconnection, Network Elements and Other Services, and/or resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA"), Carrier Identification Code (CIC), Group Access Code (GAC), Access Customer Name and Address (ACNA) and a tax exemption certificate, if applicable.
- Payment Responsibility. Payment of all charges will be the responsibility of e.spire. e.spire shall make payment to BellSouth for all services billed. BellSouth is not responsible for payments not received by e.spire from e.spire's customer. BellSouth will not become involved in billing disputes that may arise between e.spire and its customer. Payments made to BellSouth as payment on account will be credited to an accounts receivable master account and not to an end user's account.
- 1.4 <u>Payment Due</u>. The payment will be due by the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by BellSouth.

If the payment due date falls on a Sunday or on a Holiday which is observed on a Monday, the payment due date shall be the first non-Holiday day following such Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If payment is not received by the payment due date, a late payment penalty, as set forth in Section 1.7, below, shall apply.

- 1.5 <u>Tax Exemption</u>. Upon proof of tax exempt certification from e.spire, the total amount billed to e.spire will not include those taxes or fees for which the CLEC is exempt. e.spire will be solely responsible for the computation, tracking, reporting and payment of all taxes and like fees associated with the services provided to the end user of e.spire.
- Late Payment. If any portion of the payment is received by BellSouth after the payment due date as set forth preceding, or if any portion of the payment is received by BellSouth in funds that are not immediately available to BellSouth, then a late payment penalty shall be due to BellSouth. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor and will be applied on a per bill basis. The late factor shall be as set forth in Section A2 of the General Subscriber Service Tariff, Section B2 of the Private Line Service Tariff or Section E2 of the Intrastate Access Tariff, whichever BellSouth determines is appropriate.
- 1.7 <u>Discontinuing Service to e.spire</u>. The procedures for discontinuing service to e.spire are as follows:
- 1.7.1 BellSouth reserves the right to suspend or terminate service for nonpayment or in the event of prohibited, unlawful or improper use of BellSouth facilities or service or any other violation or noncompliance by e.spire of the rules and regulations contained in BellSouth's tariffs.
- 1.7.2 If payment of account is not received by the bill day in the month after the original bill day, BellSouth may provide written notice to e.spire that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition, BellSouth may, at the same time, give thirty days notice to e.spire at the billing address to discontinue the provision of existing services to e.spire at any time thereafter.
- 1.7.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
- 1.7.4 If BellSouth does not discontinue the provision of the services involved on the date specified in the thirty days notice and e.spire's noncompliance continues,

nothing contained herein shall preclude BellSouth's right to discontinue the provision of the services to e.spire without further notice.

- 1.7.5 If payment is not received or satisfactory arrangements made for payment by the date given in the written notification, e.spire's services will be discontinued. Upon discontinuance of service on e.spire's account, service to the e.spire's end users will be denied. BellSouth will reestablish service at the request of the end user or e.spire for BellSouth to reestablish service upon payment of the appropriate connection fee and subject to BellSouth's normal application procedures. e.spire is solely responsible for notifying the end user of the proposed service disconnection. If within fifteen days after an end user's service has been denied and no arrangements to reestablish service have been made consistent with this subsection, the end user's service will be disconnected.
- 1.8 Deposit Policy. When purchasing services from BellSouth, e.spire will be required to complete the BellSouth Credit Profile and provide information regarding credit worthiness. Based on the results of the credit analysis, the Company reserves the right to secure the account with a suitable form of security deposit. Such security deposit shall take the form of cash, an Irrevocable Letter of Credit (BellSouth form), Surety Bond (BellSouth form) or in its sole discretion some other form of security. Any such security deposit shall in no way release the customer from his obligation to make complete and timely payments of his bill. Such security shall be required prior to the inauguration of service. If, in the sole opinion of the Company, circumstances so warrant and/or gross monthly billing has increased beyond the level initially used to determine the level of security, the Company reserves the right to request additional security and/or file a Uniform Commercial Code (UCC1) security interest in e.spire's "accounts receivables and proceeds." Interest on a security deposit, if provided in cash, shall accrue and be paid in accordance with the terms in the appropriate BellSouth tariff.
- Rates. Rates for Optional Daily Usage File (ODUF), Enhanced Optional Daily Usage File (EODUF), Access Daily Usage File (ADUF), and Centralized Message Distribution Service (CMDS) are set out in Exhibit A to this Attachment. If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

2. BILLING AND BILLING ACCURACY CERTIFICATION

Upon request, BellSouth and e.spire will agree upon a billing quality assurance program for all billing elements covered in this Agreement that will eliminate the need for post-billing reconciliation. Appropriate terms for access to any BellSouth documents, systems, records, and procedures for the recording and billing of charges will be part of that program.

- As part of the billing quality assurance program, BellSouth and e.spire will develop standards, measurements, and performance requirements for a local billing measurements process. On a regular basis BellSouth will provide e.spire with mutually agreed upon performance measurement data that substantiates the accuracy, reliability, and integrity of the billing process for local billing. In return, e.spire will pay all bills received from BellSouth in full by the payment due date.
- 2.3 Local billing discrepancies will be addressed in an orderly manner via a mutually agreed upon billing exemption process.
- 2.3.1 Each Party agrees to notify the other Party upon identifying a billing discrepancy. The Parties shall endeavor to resolve any billing discrepancy within sixty (60) calendar days of the notification date. A mutually agreed upon escalation process will be established for resolving local billing discrepancies as part of the billing quality assurance program.
- 2.3.2 Closure of a specific billing period will occur by joint agreement of the Parties whereby the Parties agree that such billing period is closed to any further analysis and financial transactions except those resulting from regulatory mandates. Closure will take place within a mutually agreed upon time interval from the Bill Date. The month being closed represents those charges that were billed or should have been billed by the designated Bill Date.

3. BILLING DISPUTES

- 3.1 Where the Parties have not agreed upon a billing quality assurance program, billing disputes shall be handled pursuant to the terms of this section.
- 3.1.1 Each Party agrees to notify the other Party upon the discovery of a billing dispute. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the notification date.
- 3.2 If a Party disputes a charge and does not pay such charge by the payment due date, or if a payment or any portion of a payment is received by either Party after the payment due date, or if a payment or any portion of a payment is received in funds which are not immediately available to the other Party, then a late payment penalty shall be assessed. For bills rendered by either Party for payment, the late payment charge for both Parties shall be calculated based on the portion of the payment not received by the payment due date times the late factor as set forth in the following BellSouth tariffs: for services purchased from the General Subscribers Services Tariff for purposes of resale and for ports and non-designed loops, Section A2 of the General Subscriber Services Tariff; for services purchased from the Private Line Tariff for purposes of resale, Section B2 of the

Private Line Service Tariff; and for network elements and other services and local interconnection charges, Section E2 of the Access Service Tariff. In no event, however, shall interest be assessed by either Party on any previously assessed late payment charges. The Parties shall assess interest on previously assessed late payment charges only in a state where it has the authority pursuant to its tariffs.

4. RAO HOSTING

- 4.1 RAO Hosting, Calling Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to e.spire by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this Agreement, including such revisions as may be made from time to time by BellSouth.
- 4.2 e.spire shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- 4.3 Applicable compensation amounts will be billed by BellSouth to e.spire on a monthly basis in arrears. Amounts due from one Party to the other (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- e.spire must have its own unique RAO code. Requests for establishment of RAO status where BellSouth is the selected Centralized Message Distribution System (CMDS) interfacing host, require written notification from e.spire to the BellSouth RAO Hosting coordinator at least eight (8) weeks prior to the proposed effective date. The proposed effective date will be mutually agreed upon between the Parties with consideration given to time necessary for the completion of required Telcordia (formerly BellCore) functions. BellSouth will request the assignment of an RAO code from its connecting contractor, currently Telcordia (formerly BellCore), on behalf of e.spire and will coordinate all associated conversion activities.
- 4.5 BellSouth will receive messages from e.spire that are to be processed by BellSouth, another LEC or CLEC in the BellSouth region or a LEC outside the BellSouth region.
- 4.6 BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from e.spire.
- 4.7 All data received from e.spire that is to be processed or billed by another LEC or CLEC within the BellSouth region will be distributed to that LEC or CLEC in accordance with the Agreement(s) which may be in effect between BellSouth and the involved LEC or CLEC.

- 4.8 All data received from e.spire that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the agreement(s) which may be in effect between BellSouth and its connecting contractor (currently Telcordia (formerly BellCore)).
- 4.9 BellSouth will receive messages from the CMDS network that are destined to be processed by e.spire and will forward them to e.spire on a daily basis.
- 4.10 Transmission of message data between BellSouth and e.spire will be via CONNECT:Direct.
- 4.11 All messages and related data exchanged between BellSouth and e.spire will be formatted in accordance with accepted industry standards for EMI formatted records and packed between appropriate EMI header and trailer records, also in accordance with accepted industry standards.
- 4.12 e.spire will ensure that the recorded message detail necessary to recreate files provided to BellSouth will be maintained for back-up purposes for a period of three (3) calendar months beyond the related message dates.
- 4.13 Should it become necessary for e.spire to send data to BellSouth more than sixty (60) days past the message date(s), e.spire will notify BellSouth in advance of the transmission of the data. If there will be impacts outside the BellSouth region, BellSouth will work with its connecting contractor and e.spire to notify all affected Parties.
- In the event that data to be exchanged between the two Parties should become lost or destroyed, both Parties will work together to determine the source of the problem. Once the cause of the problem has been jointly determined and the responsible Party (BellSouth or e.spire) identified and agreed to, the company responsible for creating the data (BellSouth or e.spire) will make every effort to have the affected data restored and retransmitted. If the data cannot be retrieved, the responsible Party will be liable to the other Party for any resulting lost revenue. Lost revenue may be a combination of revenues that could not be billed to the end users and associated access revenues. Both Parties will work together to estimate the revenue amount based upon historical data through a method mutually agreed upon. The resulting estimated revenue loss will be paid by the responsible Party to the other Party within three (3) calendar months of the date of problem resolution, or as mutually agreed upon by the Parties.
- 4.15 Should an error be detected by the EMI format edits performed by BellSouth on data received from e.spire, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify e.spire of the error condition. e.spire will correct the error(s) and will resend the entire pack to BellSouth for

processing. In the event that an out-of-sequence condition occurs on subsequent packs, e.spire will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth.

- 4.16 In association with message distribution service, BellSouth will provide e.spire with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 4.17 In no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this Agreement.

4.18 RAO Compensation

- 4.18.1 Rates for message distribution service provided by BellSouth for e.spire are as set forth in Exhibit A to this Attachment.
- 4.18.2 Rates for data transmission associated with message distribution service are as set forth in Exhibit A to this Attachment.
- 4.18.3 Data circuits (private line or dial-up) will be required between BellSouth and e.spire for the purpose of data transmission. Where a dedicated line is required, e.spire will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. e.spire will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to e.spire. Additionally, all message toll charges associated with the use of the dial circuit by e.spire will be the responsibility of e.spire. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties.
- 4.18.4 All equipment, including modems and software, that is required on the e.spire end for the purpose of data transmission will be the responsibility of e.spire.

4.19 Intercompany Settlements Messages

4.19.1 This Section addresses the settlement of revenues associated with traffic originated from or billed by e.spire as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one Bell operating territory and bills in another Bell operating territory is included. Traffic that originates and bills within the same Bell operating territory will be settled on a local basis between e.spire and the involved company(ies), unless that company is participating in NICS.

- 4.19.2 Both traffic that originates outside the BellSouth region by e.spire and is billed within the BellSouth region, and traffic that originates within the BellSouth region and is billed outside the BellSouth region by e.spire, is covered by this Agreement (CATS). Also covered is traffic that either is originated by or billed by e.spire, involves a company other than e.spire, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).
- 4.19.3 Once e.spire is operating within the BellSouth territory, revenues associated with calls originated and billed within the BellSouth region will be settled via Telcordia (formerly BellCore)'s, its successor or assign, NICS system.
- 4.19.4 BellSouth will receive the monthly NICS reports from Telcordia (formerly BellCore), its successor or assign, on behalf of e.spire. BellSouth will distribute copies of these reports to e.spire on a monthly basis.
- 4.19.5 BellSouth will receive the monthly Calling Card and Third Number Settlement System (CATS) reports from Telcordia (formerly BellCore), its successor or assign, on behalf of e.spire. BellSouth will distribute copies of these reports to e.spire on a monthly basis.
- 4.19.6 BellSouth will collect the revenue earned by e.spire from the Bell operating company in whose territory the messages are billed (CATS), less a per message billing and collection fee of five cents (\$0.05), on behalf of e.spire. BellSouth will remit the revenue billed by e.spire to the Bell operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf on e.spire. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to e.spire via a monthly Carrier Access Billing System (CABS) miscellaneous bill.
- 4.19.7 BellSouth will collect the revenue earned by e.spire within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of e.spire. BellSouth will remit the revenue billed by e.spire within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit issued to e.spire via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

BellSouth and e.spire agree that monthly netted amounts of less than fifty dollars (\$50.00) will not be settled.

5. OPTIONAL DAILY USAGE FILE

- Upon written request from e.spire, BellSouth will provide the Optional Daily Usage File (ODUF) service to e.spire pursuant to the terms and conditions set forth in this section.
- The e.spire shall furnish all relevant information required by BellSouth for the provision of the Optional Daily Usage File.
- 5.3 The Optional Daily Usage Feed will contain billable messages that were carried over the BellSouth Network and processed in the BellSouth Billing System, but billed to a e.spire customer.

Charges for delivery of the Optional Daily Usage File will appear on the e.spire's monthly bills.

The charges are as set forth in Exhibit A to this Attachment.

- The Optional Daily Usage Feed will contain both rated and unrated messages. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of e.spire will be the responsibility of e.spire. If, however, e.spire should encounter significant volumes of errored messages that prevent processing by e.spire within its systems, BellSouth will work with e.spire to determine the source of the errors and the appropriate resolution.
- 5.6 The following specifications shall apply to the Optional Daily Usage Feed.
- 5.6.1 USAGE TO BE TRANSMITTED
- 5.6.1.1 The following messages recorded by BellSouth will be transmitted to the e.spire:
 - message recording for per use/per activation type services (examples: Three Way Calling, Verify, Interrupt, Call Return, ETC.)
 - measured billable Local
 - Directory Assistance messages
 - intraLATA Toll
 - WATS & 800 Service
 - N11
 - Information Service Provider Messages

- Operator Services Messages
- Operator Services Message Attempted Calls (Network Element only)
- Credit/Cancel Records
- Usage for Voice Mail Message Service
- Rated Incollects (originated in BellSouth and from other companies) can also be on Optional Daily Usage File. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.
- 5.6.1.3 BellSouth will perform duplicate record checks on records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to e.spire.
- 5.6.1.4 In the event that e.spire detects a duplicate on Optional Daily Usage File they receive from BellSouth, e.spire will drop the duplicate message (e.spire will not return the duplicate to BellSouth).

5.6.2 PHYSICAL FILE CHARACTERISTICS

- The Optional Daily Usage File will be distributed to e.spire via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a variable block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and e.spire for the purpose of data transmission. Where a dedicated line is required, e.spire will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. e.spire will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to e.spire. Additionally, all message toll charges associated with the use of the dial circuit by e.spire will be the responsibility of e.spire. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including

modems and software, that is required on e.spire end for the purpose of data transmission will be the responsibility of e.spire.

5.6.3 PACKING SPECIFICATIONS

- 5.6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to e.spire which BellSouth RAO that is sending the message. BellSouth and e.spire will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by e.spire and resend the data as appropriate.

The data will be packed using ATIS EMI records.

5.6.4 PACK REJECTION

e.spire will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. e.spire will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to e.spire by BellSouth.

5.6.5 CONTROL DATA

e.spire will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate e.spire received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by e.spire for reasons stated in the above section.

5.6.6 TESTING

5.6.6.1 Upon request from e.spire, BellSouth shall send test files to e.spire for the Optional Daily Usage File. The Parties agree to review and discuss the file's content and/or format. For testing of usage results, BellSouth shall request that e.spire set up a production (LIVE) file. The live test may consist of e.spire's employees making test calls for the types of services e.spire requests on the Optional Daily Usage File. These test calls are logged by e.spire, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be

completed within 30 calendar days from the date on which the initial test file was sent.

6. ACCESS DAILY USAGE FILE

- 6.1. Upon written request from e.spire, BellSouth will provide the Access Daily Usage File (ADUF) service to e.spire pursuant to the terms and conditions set forth in this section.
- The e.spire shall furnish all relevant information required by BellSouth for the provision of the Access Daily Usage File.
- The Access Daily Usage Feed will contain access messages associated with a port that e.spire has purchased from BellSouth.
- Charges for delivery of the Access Daily Usage File will appear on the e.spire's monthly bills. The charges are as set forth in Exhibit A to this Attachment. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of the e.spire will be the responsibility of the e.spire. If, however, the e.spire should encounter significant volumes of errored messages that prevent processing by the e.spire within its systems, BellSouth will work with the e.spire to determine the source of the errors and the appropriate resolution.
- 6.6 USAGE TO BE TRANSMITTED
- 6.6.1 The following messages recorded by BellSouth will be transmitted to e.spire:

Interstate and intrastate access records associated with a port.

Undetermined jurisdiction access records associated with a port.

When e.spire purchases Network Element ports from BellSouth and calls are made using these ports, BellSouth will handle the calls as follows:

Originating from Network Element and carried by Interexchange Carrier:

BellSouth will bill network element to CLEC and send access record to the CLEC via ADUF

Originating from network element and carried by BellSouth (e.spire is BellSouth's toll customer):

BellSouth will bill resale toll rates to e.spire and send toll record for the end user toll billing purposes via ODUF (Optional Daily Usage File). Access record will be sent to e.spire via ADUF.

Terminating on network element and carried by Interexchange Carrier:

BellSouth will bill network element to e.spire and send access record to e.spire.

Terminating on network element and carried by BellSouth:

BellSouth will bill network element to e.spire and send access record to e.spire.

- 6.6.3 BellSouth will perform duplicate record checks on records processed to the Access Daily Usage File. Any duplicate messages detected will be dropped and not sent to e.spire.
- In the event that e.spire detects a duplicate on the Access Daily Usage File they receive from BellSouth, e.spire will drop the duplicate message (e.spire will not return the duplicate to BellSouth).

6.6.5 PHYSICAL FILE CHARACTERISTICS

- 6.6.5.1 The Access Daily Usage File will be distributed to e.spire via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a fixed block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (210 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and e.spire for the purpose of data transmission. Where a dedicated line is required, e.spire will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. e.spire will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to e.spire. Additionally, all message toll charges associated with the use of the dial circuit by e.spire will be the responsibility of

e.spire. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software, that is required on e.spire end for the purpose of data transmission will be the responsibility of e.spire.

6.6.6 PACKING SPECIFICATIONS

- 6.6.6.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to espire which BellSouth RAO that is sending the message. BellSouth and espire will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by espire and resend the data as appropriate.

The data will be packed using ATIS EMI records.

6.6.7 PACK REJECTION

e.spire will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. e.spire will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to e.spire by BellSouth.

6.6.8 CONTROL DATA

e.spire will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate e.spire received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by e.spire for reasons stated in the above section.

6.6.9 TESTING

Upon request from e.spire, BellSouth shall send test files to e.spire for the Access Daily Usage File. Testing shall consist of actual calls made from live accounts. A call log shall be supplied along with test request information. The Parties agree to review and discuss the file's content and/or format.

7. ENHANCED OPTIONAL DAILY USAGE FILE

- 7.1 Upon written request from e.spire, BellSouth will provide the Enhanced Optional Daily Usage File (EODUF) service to e.spire pursuant to the terms and conditions set forth in this section. EODUF will only be sent to existing ODUF subscribers who request the EODUF option.
- e.spire shall furnish all relevant information required by BellSouth for the provision of the Enhanced Optional Daily Usage File.
- 7.3 The Enhanced Optional Daily Usage File (EODUF) will provide usage data for local calls originating from resold Flat Rate Business and Residential Lines.

Charges for delivery of the Enhanced Optional Daily Usage File will appear on the e.spire's monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- 7.4 All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of e.spire will be the responsibility of e.spire. If, however, e.spire should encounter significant volumes of errored messages that prevent processing by e.spire within its systems, BellSouth will work with e.spire to determine the source of the errors and the appropriate resolution.
- 7.6 The following specifications shall apply to the Optional Daily Usage Feed.
- 7.6.1 USAGE TO BE TRANSMITTED
- 7.6.1.1 The following messages recorded by BellSouth will be transmitted to e.spire:

Customer usage data for flat rated local call originating from CLEC end user lines (1FB or 1FR). The EODUF record for flat rate messages will include:

Date of Call
From Number
To Number
Connect Time
Conversation Time
Method of Recording
From RAO
Rate Class
Message Type
Billing Indicators

Bill to Number

- 7.6.1.2 BellSouth will perform duplicate record checks on EODUF records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to e.spire.
- 7.6.1.3 In the event that e.spire detects a duplicate on Enhanced Optional Daily Usage File they receive from BellSouth, e.spire will drop the duplicate message (e.spire will not return the duplicate to BellSouth).

7.6.2 PHYSICAL FILE CHARACTERISTICS

- 7.6.2.1 The Enhanced Optional Daily Usage Feed will be distributed to e.spire over their existing Optional Daily Usage File (ODUF) feed. The EODUF messages will be intermingled among e.spire's Optional Daily Usage File (ODUF) messages. The EODUF will be a variable block format (2476) with an LRECL of 2472. The data on the EODUF will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays).
- Data circuits (private line or dial-up) may be required between BellSouth and e.spire for the purpose of data transmission. Where a dedicated line is required, e.spire will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. e.spire will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to e.spire. Additionally, all message toll charges associated with the use of the dial circuit by e.spire will be the responsibility of e.spire. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software, that is required on e.spire end for the purpose of data transmission will be the responsibility of e.spire.

7.6.3 PACKING SPECIFICATIONS

- 7.6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 7.6.3.2 The Operating Company Number (OCN), From Revenue Accounting Office (RAO), and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to espire which BellSouth RAO that is sending the message. BellSouth and espire will use the invoice sequencing to control data

exchange. BellSouth will be notified of sequence failures identified by e.spire and resend the data as appropriate.

The data will be packed using ATIS EMI records.

BELLSOUTH/e.spire RATES ODUF/EDOUF/ADUF/CMDS

					F	RATES BY STAT	ГЕ			
DESCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
ODUF/EODUF/ADUF/CMDS										
ODUF: Recording, per message	N/A	\$0.0002	\$0.008	\$0.008	\$0.0008611	\$0.00019	\$0.0001179	\$0.008	\$0.0002862	\$0.008
ODUF: Message Processing, per message	N/A	\$0.0033	\$0.004	\$0.004	\$0.0032357	\$0.0024	\$0.0032089	\$0.004	\$0.0032344	\$0.004
EODUF: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
ADUF: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
CMDS: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
ODUF: Message Processing, per magnetic tape provisioned	N/A	\$55.19	\$54.95	\$54.95	\$55.68	\$47.30	\$54.62	\$54.95	\$54.72	\$54.95
EODUF: Message Processing, per magnetic tape provisioned	N/A	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30
ADUF: Message Processing, per magnetic tape provisioned	N/A	\$54.95	\$54.95	\$54.95	\$54.95	\$54.95	\$54.95	\$54.95	\$54.95	\$54.95
ODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.00004	\$0.001	\$0.001	\$0.0000365	\$0.00003	\$0.0000354	\$0.001	\$0.0000357	\$0.001
EODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364
ADUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001
CMDS: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001

NOTES:

If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the parties upon request by either party.

Attachment 8

Rights-of-Way, Conduits and Pole Attachments

Rights-of-Way, Conduits and Pole Attachments

Pursuant to terms and conditions negotiated between e.spire and BellSouth's Competitive Structure Provisioning Center and pursuant to 47 U.S.C. § 224, BellSouth will provide nondiscriminatory access to poles, ducts, conduit, and rights-of-way owned or controlled by BellSouth.

Attachment 9

Performance Measurements

TABLE OF CONTENTS

Serv	vice Performance Measurements And Enforcement Mechanisms	3
	Scope	
	Reporting	
3.	Modifications to Measurements	3
4.	Enforcement Mechanisms	4
EXI	HIBIT A	10
EXI	HIBIT B	79
EXI	HIBIT C	85
EXI	HIBIT D	101
FYI	HIRIT F	107

Service Performance Measurements And Enforcement Mechanisms

1. Scope

This Attachment includes Enforcement Measurements with corresponding Enforcement Mechanisms applicable to this Agreement.

2. Reporting

- 2.1 In providing services pursuant to this Agreement, BellSouth will report its performance to e.spire in accordance with BellSouth's Service Quality Measurements, which are contained in this Attachment as Exhibit A and in accordance with BellSouth's Enforcement Measurements, which are contained in this Attachment as Exhibit B.
- 2.2 BellSouth will make performance reports available to e.spire on a monthly basis. The reports will contain information collected in each performance category and will be available to e.spire through some electronic medium to be determined by BellSouth. BellSouth will also provide electronic access to the raw data underlying the performance measurements. Within thirty (30) days of execution of this Agreement, BellSouth will provide a detailed session of instruction to e.spire regarding access to the reports and to the raw data as well as the nature of the format of the data provided.

3. <u>Modifications to Measurements</u>

3.1 Service Quality Measurements

- 3.1.1 BellSouth will update the Service Quality Measurements contained in Exhibit A of this Attachment each calendar quarter. BellSouth will not delete any Service Quality Measurement without prior written consent of e.spire. e.spire may provide input to BellSouth regarding any suggested additions, deletions or other modifications to the Service Quality Measurements. BellSouth will provide notice of all changes to the Service Quality Measurements via BellSouth's internet website.
- 3.1.2 Notwithstanding the foregoing, BellSouth may, from time to time, be ordered by a regulatory or judicial body to modify or amend the Service Quality Measurements. BellSouth will make all such changes to the Service Quality Measurements pursuant to Section 34 of the General Terms and Conditions of this Agreement, incorporated herein by reference
- 3.1.3 Notwithstanding any other provision of this Agreement, in the event a dispute arises regarding the modification or amendment of the

Version 1Q00: 3/6/00 7/7/00 Version

Service Quality Measurements, the parties will refer the dispute to the Commission.

3.2 Enforcement Measurements and Statistical Test

- 3.2.1 In order for BellSouth to accurately administer the Enforcement Measurements contained in Exhibit B of this Attachment, the Enforcement Measurements shall be modified or amended only if BellSouth determines such modification or amendment is necessary. However, BellSouth will not delete any Enforcement Measurement without prior written consent of e.spire. BellSouth will notify e.spire of any such modification or amendment to the Enforcement Measurements via BellSouth's internet website.
- 3.2.2 Notwithstanding the foregoing, BellSouth may, from time to time, be ordered by a regulatory or judicial body to modify or amend the Enforcement Measurements and/or Statistical Test. BellSouth will make all such changes to the Enforcement Measurements and/or Statistical Test pursuant to Section 34 of the General Terms and Conditions of this Agreement, incorporated herein by reference.
- 3.2.3 Notwithstanding any other provision of this Agreement, in the event a dispute arises regarding the modification or amendment of the Enforcement Measurements and/or Statistical Test, the parties will refer the dispute to the Commission.

4. **Enforcement Mechanisms**

4.1 <u>Purpose</u>

This section establishes meaningful and significant enforcement mechanisms voluntarily provided by BellSouth to verify and maintain compliance between BellSouth and e.spire's operations as well as to maintain access to Operational Support System (OSS) functions. This section provides the terms and conditions for such self-effectuating enforcement mechanisms.

4.2 Effective Date

The enforcement mechanisms set forth in this section shall only become effective upon an effective FCC order, which has not been stayed, authorizing BellSouth to provide interLATA telecommunications services under section 271 of the Act within any state and shall apply to BellSouth's performance in each state in the BellSouth region.

4.3 Definitions

- 4.3.1 <u>Enforcement Measurement Elements</u> means the performance measurements set forth in Exhibit B, attached hereto and incorporated herein by this reference.
- 4.3.2 Enforcement Measurement Benchmark means a competitive level of performance negotiated by BellSouth used to compare the performance of BellSouth and e.spire where no analogous process, product or service is feasible. See Exhibit B.
- 4.3.3 <u>Enforcement Measurement Compliance</u> means comparing performance levels provided to BellSouth retail customers with performance levels provided by BellSouth to the CLEC customer, as set forth in Exhibit C, attached hereto and incorporated herein by this reference.
- 4.3.4 <u>Test Statistic and Balancing Critical Value</u> is the means by which enforcement will be determine using statistically valid equations. See Exhibit C.
- 4.3.5 <u>Cell</u> is the point (below the wire center level) at which like-to-like comparisons are made. For example, all BellSouth retail POTS services, for residential customers, requiring a dispatch in a particular wire center, at a particular point in time will be compared directly to e.spire resold services for residential customers, requiring a dispatch, in the same wire center, at a particular point in time. When determining compliance, these cells can have a positive or negative value. See Exhibit C.
- 4.3.6 <u>Affected Volume</u> means that proportion of the total e.spire volume or CLEC Aggregate volume for which remedies will be paid.
- 4.3.7 <u>Parity Gap</u> refers to the incremental departure from a compliant-level of service. (See Exhibit D). This is also referred to as "diff" in the Statistical paper (See Exhibit C).
- 4.3.8 <u>Tier-1 Enforcement Mechanisms</u> means self-executing liquidated damages paid directly to e.spire when BellSouth delivers non-compliant performance of any one of the Enforcement Measurement Elements for any month as calculated by BellSouth.
- 4.3.9 <u>Tier-2 Enforcement Mechanisms</u> means Assessments paid directly to a state Public Service Commission ("Commission") or its designee. Tier 2 Enforcement Mechanisms are triggered by three consecutive monthly failures in a quarter in which BellSouth performance is out of compliance or does not meet the benchmarks for the aggregate of all CLEC data as calculated by BellSouth for a particular Enforcement Measurement Element.

Version 1Q00: 3/6/00 7/7/00 Version

4.3.10 <u>Tier-3 Enforcement Mechanisms</u> means the voluntary suspension of additional marketing and sales of long distance services triggered by excessive repeat failures of those specific submeasures as defined in Exhibit D attached hereto and incorporated herein by this reference.

4.4 <u>Application</u>

- 4.4.1 The application of the Tier-1, Tier-2, and Tier-3 Enforcement Mechanisms does not foreclose other non-contractual legal and regulatory claims and remedies available to e.spire.
- 4.4.2 Proof of damages resulting from BellSouth's failure to maintain Enforcement Measurement Compliance would be difficult to ascertain and, therefore, liquidated damages are a reasonable approximation of any contractual damage. Liquidated damages under this provision are not intended to be a penalty.

4.5 Methodology

- 4.5.1 Tier-1 Enforcement Mechanisms will be triggered by BellSouth's failure to achieve Enforcement Measurement Compliance or Enforcement Measurement Benchmarks for the State for a given Enforcement Measurement Element in a given month based upon a test statistic and balancing critical value calculated by BellSouth utilizing BellSouth generated data. The method of calculation is attached hereto as Exhibit D and incorporated herein by this reference.
- 4.5.1.1 Tier-1 Enforcement Mechanisms apply on a per transaction basis for each negative cell and will escalate based upon the number of consecutive months that BellSouth has reported non-compliance.
- 4.5.1.2 Fee Schedule for Tier-1 Enforcement Mechanisms is shown in Table-1 attached hereto as Exhibit E and incorporated herein by this reference. Failures beyond Month 6 (as set forth in Table 1) will be subject to Month 6 fees.
- 4.5.2 Tier-2 Enforcement Mechanisms will be triggered by BellSouth's failure to achieve Enforcement Measurement Compliance or Enforcement Measurement Benchmarks for the State in a given calendar quarter based upon a statistically valid equation calculated by BellSouth utilizing BellSouth generated data. The method of calculation is attached hereto as Exhibit D and incorporated herein by reference.

- 4.5.2.1 Tier- 2 Enforcement Mechanisms apply, for an aggregate of all CLEC data generated by BellSouth, on a per transaction basis for each negative cell for a particular Enforcement Measurement Element.
- 4.5.2.2 Fee Schedule for Total Quarterly Tier-2 Enforcement Mechanisms is show in Table-2 attached hereto as Exhibit E and incorporated herein by this reference.
- 4.5.3 Tier-3 Enforcement Mechanisms will be triggered by BellSouth's failure to achieve Enforcement Measurement Compliance or Enforcement Measurement Benchmarks for a State in a given calendar quarter. The method of calculation for specified submeasures is identical to the method of calculation for Tier-2 Enforcement Mechanisms as described above. The specific submeasures which are the mechanism for triggering and removing a Tier-3 Enforcement Mechanisms are described in more detail in Exhibit D attached hereto and incorporated herein by this reference.

4.6 Payment of Tier-1 and Tier-2 Amounts

- 4.6.1 If BellSouth performance triggers an obligation to pay Tier-1 Enforcement Mechanisms to e.spire or an obligation to remit Tier-2 Enforcement Mechanisms to the Commission, BellSouth shall make payment in the required amount on or before the thirtieth (30th) day following the due date of the performance measurement report for the month in which the obligation arose.
- 4.6.2 For each day after the due date that BellSouth fails to pay e.spire the required amount, BellSouth will pay interest to e.spire at the maximum rate permitted by state law.
- 4.6.3 For each day after the due date that BellSouth fails to pay the Tier-2 Enforcement Mechanisms, BellSouth will pay the Commission an additional \$1,000 per day.
- 4.6.4 If e.spire disputes the amount paid to e.spire for Tier-1 Enforcement Mechanisms, e.spire shall submit a written claim to BellSouth within sixty (60) days after the date of the performance measurement report for which the obligation arose. BellSouth shall investigate all claims and provide e.spire written findings within thirty (30) days after receipt of the claim. If BellSouth determines e.spire is owed additional amounts, BellSouth shall pay e.spire such additional amounts within thirty (30) days after its findings along with interest paid at the maximum rate permitted by law.
- 4.6.5 At the end of each calendar year, BellSouth will have its independent auditing and accounting firm certify that the results of all Tier-1 and Tier-

2 Enforcement Mechanisms were paid and accounted for in accordance with Generally Accepted Account Principles (GAAP).

4.7 <u>Limitations of Liability</u>

- 4.7.1 BellSouth will not be responsible for e.spire acts or omissions that cause performance measures to be missed or fail, including but not limited to accumulation and submission of orders at unreasonable quantities or times or failure to submit accurate orders or inquiries. BellSouth shall provide e.spire with reasonable notice of such acts or omissions and provide e.spire any such supporting documentation.
- 4.7.2 BellSouth shall not be obligated for Tier-1, Tier-2 or Tier 3 Enforcement Mechanisms for non-compliance with a performance measure if such non-compliance was the result of an act or omission by e.spire that is in bad faith.
- 4.7.3 BellSouth shall not be obligated to pay Tier-1 Enforcement Mechanisms or Tier-2 Enforcement Mechanism for non-compliance with a performance measurement if such non-compliance was the result of any of the following: a Force Majeure event as set forth in the General Terms and Conditions of this Agreement; an act or omission by e.spire that is contrary to any of its obligations under its Interconnection Agreement with BellSouth; an act or omission by e.spire that is contrary to any of its obligations under the Act, Commission rule, or state law; an act or omission associated with third-party systems or equipment; or any occurrence that results from an incident reasonably related to the Y2K problem.
- 4.7.4 It is not the intent of the Parties that BellSouth be liable for both Tier-2 Enforcement Mechanisms and any other assessments or sanctions imposed by the Commission. e.spire will not oppose any effort by BellSouth to set off Tier-2 Enforcement Mechanisms from any additional assessment imposed by the Commission.
- 4.7.5 Payment of any Tier-1 or Tier-2 Enforcement Mechanisms shall not be considered as an admission against interest or an admission of liability or culpability in any legal, regulatory or other proceeding relating to BellSouth's performance. The payment of any Tier-1 Enforcement Mechanisms to e.spire shall release BellSouth for any liability associated with or related to the service performance measurement for the month for which the Enforcement Mechanisms was paid to e.spire.
- 4.7.6 e.spire acknowledges and argues that the Enforcement Mechanisms contained in this attachment have been provided by BellSouth on a completely voluntary basis in order to maintain compliance between

BellSouth and e.spire. Therefore, e.spire may not use the existence of this section or any payments of any Tier-1 or Tier-2 Enforcement Mechanisms under this section as evidence that BellSouth has not complied with or has violated any state or federal law or regulation.

4.8 <u>Enforcement Mechanism Caps</u>

4.8.1 BellSouth's liability for the payment of Tier-1 and Tier-2 Enforcement Mechanisms shall be collectively capped at \$625M per year for the entire BellSouth region as set forth below.

AL - \$54M	MS - \$44M	
FL - \$122M	NC - \$77M	
GA - \$131M	SC - \$47M	
KY - \$34M	TN - \$57M	
LA - \$59M		
Regional Total - \$625M		

4.8.2 If BellSouth's liability for the payment of Tier-1 and Tier-2 Enforcement Mechanisms exceed the caps referenced in this attachment, e.spire may commence a proceeding with the Commission to demonstrate why BellSouth should pay any amount in excess of the cap. e.spire shall have the burden of proof to demonstrate why, under the circumstances, BellSouth should have additional liability.

4.9 <u>Dispute Resolution</u>

4.9.1 Notwithstanding any other provision of this Agreement, any dispute regarding BellSouth's performance or obligations pursuant to this Attachment shall be resolved by the Commission.

EXHIBIT A

ORDERING

Report/Measurement:

O-7. Speed of Answer in Ordering Center

Definition:

Measures the average time a customer is in queue.

Exclusions:

None

Business Rules:

The clock starts when the appropriate option is selected (i.e. 1 for Resale Consumer, 2 for Resale Multiline, and 3 for UNE-LNP, etc.) and the call enters the queue for that particular group in the LCSC. The clock stops when a BST service representative in the LCSC answers the call. The speed of answer is determined by measuring and accumulating the elapsed time from the entry of a CLEC call into the BellSouth automatic call distributor (ACD) until the a service representative in BSTs Local Carrier Service Center (LCSC) answers the CLEC call.

Calculation:

(Total time in seconds to reach the LCSC) / (Total Number of Calls) in the Reporting Period.

Report Structure:

- CLEC Aggregate
- BST Aggregate (Combination of Residence Service Center and Business Service Center data under development)

Level of Disaggregation:

- CLEC Aggregate
- BST Aggregate (Combination of Residence Service Center and Business Service Center data under development)

Data Retained Relating to CLEC Experience:	Data Retained Relating to BST Performance:
Mechanized tracking through LCSC	Mechanized tracking through BST Retail center support
Automatic Call Distributor	systems

Retail Analog/Benchmark:

For CLEC, Speed of Answer in Ordering Center (LCSC) is comparable to Speed of Answer in BST Business Offices. See Appendix D

Revision Date: 02/16/00 (lg)

ORDERING – (LNP)

Report/Measurement:

LNP-8. Percent Rejected Service Requests

Definition:

Percent Rejected Service Request is the percent of total Local Service Requests (LSRs) which are rejected due to error or omission. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete, i.e., fatal rejects are excluded.

Exclusions:

- Service Requests canceled by the CLEC
- Fatal Rejects
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules:

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

<u>Fully Mechanized</u>: There are two types of "Rejects" in the Fully Mechanized category:

- A Fatal Reject occurs when a CLEC attempts to electronically submit an LSR (via EDI or TAG) but required fields are not populated correctly and the request is returned to the CLEC.
 - Fatal rejects are reported in a separate column, and for informational purposes ONLY. They are not considered in the calculation of the percent of total LSRs rejected or the total number of rejected LSRs.
- An Auto Clarification is a valid LSR which is electronically submitted (via EDI or TAG), but is rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

<u>Partially Mechanized</u>: A valid LSR which is electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Calculation

Percent Rejected Service Requests:

[(Number of Service Requests Rejected in the Reporting Period) / (Number of Service Requests Received in the Reporting Period)] x 100

Report Structure:

- Fully Mechanized, Partially Mechanized, Total Mechanized
- CLEC Specific
- CLEC Aggregate

Level of Disaggregation:

- Product Reporting Levels
 - LNP
 - UNE Loop with LNP
- Geographic Scope
 - > .State, Region

Retail Analog/Benchmark:

See Appendix D

Revision Date: 02/16/00 (lg)

ORDERING – (LNP)

Report/Measurement:

LNP-9. Reject Interval Distribution & Average Reject Interval

Definition:

Reject Interval is the average reject time from receipt of an LSR to the distribution of a Reject. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete, i.e., fatal rejects are excluded.

Exclusions:

- Service Requests canceled by CLEC
- Fatal Rejects
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules:

The Reject interval is determined for each rejected LSR processed during the reporting period. The Reject interval is the elapsed time from when BST receives LSR until that LSR is rejected back to the CLEC. Elapsed time for each LSR is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of rejected LSRs to produce the reject interval distribution.

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

<u>Fully Mechanized</u>: There are two types of "Rejects" in the Fully Mechanized category:

- A Fatal Reject occurs when a CLEC attempts to electronically submit an LSR but required fields are not populated correctly and the request is returned to the CLEC.
 - Fatal rejects are reported in a separate column, and for informational purposes ONLY. They are not considered in the calculation of the percent of total LSRs rejected or the number of rejected LSRs.
- An Auto Clarification is a valid LSR which is electronically submitted (via EDI or TAG), but rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

<u>Partially Mechanized</u>: A valid LSR which is electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Calculation:

Average Reject Interval:

 Σ [(Date & Time of Service Request Rejection) - (Date & Time of Service Request Receipt)] / (Total Number of Service Requests Rejected in Reporting Period)

Reject Interval Distribution:

 $[\Sigma \text{ (Service Requests Rejected in "X" minutes/hours)} / \text{ (Total Number of Service Requests Rejected in Reporting Period)}] X 100$

Report Structure:

- Fully Mechanized, Partially Mechanized, Total Mechanized
- CLEC Specific
- CLEC Aggregate

ORDERING – (LNP) - Reject Interval Distribution & Average Reject Interval – Continued)

Level of Disaggregation:

- Reported in intervals = 0 4 minutes, 4 8 minutes, 8 12 minutes, 12 60 minutes, 0 1 hours, 1 8 hours, 8 24 hours, >24 hours
- Product Reporting Levels
 - > LNP
 - UNE Loop with LNP
- Geographic Scope
 - > .State, Region
- Average Interval in Days

Retail Analog/Benchmark:

See Appendix D

Revision Date: 02/16/00 (lg)

ORDERING - (LNP)

Report/Measurement:

LNP-10. Firm Order Confirmation Timeliness Interval Distribution & Firm Order Confirmation Average Interval

Definition:

Interval for Return of a Firm Order Confirmation (FOC Interval) is the average response time from receipt of a valid LSR to distribution of a firm order confirmation.

Exclusions:

- Rejected LSRs (Clarifications or Fatal Rejects)
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules:

The Firm Order Confirmation interval is determined for each FOC'd LSR processed during the reporting period. The Firm Order Confirmation interval is the elapsed time from when BST receives an LSR until that LSR is confirmed back to the CLEC. Elapsed time for each LSR is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed to produce the Firm Order Confirmation timeliness interval distribution.

- <u>Mechanized</u> The elapsed time from receipt of a valid LSR until the LSR is processed and appropriate service orders are generated in SOCS without manual intervention.
- <u>Partially Mechanized</u> The elapsed time from receipt of an electronically submitted LSR which falls out for manual handling by the LCSC personnel until appropriate service orders are issued by a BST service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS).
- **Total Mechanized** Combination of Fully Mechanized and Partially Mechanized FOCs.

Calculation:

Average FOC Interval:

 Σ [(Date & Time of Firm Order Confirmation) - (Date & Time of Service Request Receipt)] / (Total number of Service Requests Confirmed in the Reporting Period)

FOC Interval Distribution:

 Σ [(Service Requests Confirmed in "X" minutes/hours in the Reporting Period) / (Total Service Requests Confirmed in the Reporting Period)] X 100

Report Structure:

- Fully Mechanized, Partially Mechanized, Total Mechanized
- CLEC Specific
- CLEC Aggregate

Level of Disaggregation:

- Reported in intervals = 0 15 minutes, 15 30 minutes, 30 45 minutes, 45 60 minutes, 90 120 minutes, 120 240 minutes, 4 8 hours, 8 12 hours, 12 16 hours, 16 20 hours, 20 24 hours, 24 48 hours, >48 hours
- Product Reporting Levels
 - > LNP
 - UNE Loop with LNP
- Geographic Scope
 - State, Region

Retail Analog/Benchmark:

See Appendix D

Revision Date: 02/16/00 (lg)

Provisioning Disaggregation

Product Reporting Levels

- Resale and Retail
 - ➤ Pots Residence
 - ➤ Pots Business
 - ➤ Design
 - ➤ PBX (Louisiana SQM)
 - > CENTREX (Louisiana SQM)
 - ➤ ISDN (Louisiana SQM) (NOTE: ISDN included in POTS for Georgia Only)
 - ➤ ESSX (Louisiana SQM)
- Unbundled Network Elements
 - ➤ UNE Design
 - ➤ UNE Non Design
 - ➤ UNE 2 Wire Loop (Louisiana SQM)
 - ➤ UNE Loop Other (Louisiana SQM)
 - ➤ Unbundled Ports (Louisiana SQM)
- Trunks
 - ➤ Local Interconnection Trunks
- Geographic Scope
 - ➤ State, Region and further geographic disaggregation as required by State Commission Order (e.g. Metropolitan Service Area MSA)

The following measure is the exception for all states:

Coordinated Customer Conversion

Which is disaggregated as follows:

UNE LOOPS with INP UNE LOOPS without INP

Report/Measurement:

P-1. Mean Held Order Interval & Distribution Intervals

Definition:

When delays occur in completing CLEC orders, the average period that CLEC orders are held for BST reasons, pending a delayed completion, should be no worse for the CLEC when compared to BST delayed orders.

Exclusions:

Order Activities of BST associated with internal or administrative use of local services.

Business Rules:

Mean Held Order Interval: This metric is computed at the close of each report period. The held order interval is established by first identifying all orders, at the close of the reporting interval, that both have not been reported as completed in SOCS and have passed the currently committed due date for the order. For each such order, the number of calendar days between the committed due date and the close of the reporting period is established and represents the held order interval for that particular order. The held order interval is accumulated by the standard groupings, unless otherwise noted, and the reason for the order being held. The total number of days accumulated in a category is then divided by the number of held orders within the same category to produce the mean held order interval. The interval is by calendar days with no exclusions for Holidays or Sundays.

CLEC Specific reporting is by type of held order (facilities, equipment, other), total number of orders held, and the total and average days.

<u>Held Order Distribution Interval</u>: This measure provides data to report total days held and identifies these in categories of >15 days and > 90 days. (orders counted in >90 days are also included in >15 days).

Calculation:

Mean Held Order Interval:

 Σ (Reporting Period Close Date – Committed Order Due Date) / (Number of Orders Pending and Past The Committed Due Date) for all orders pending and past the committed due date.

Held Order Distribution Interval:

(# of Orders Held for ≥90 days) / (Total # of Orders Pending But Not Completed) X 100 (# of Orders Held for ≥15 days) / (Total # of Orders Pending But Not Completed) X 100

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

Circuit breakout < 10, > = 10

PROVISIONING - Mean Held Order Interval & Distribution Intervals - Continued)

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
CLEC Order Number and PON (PON)	BST Order Number
 Order Submission Date (TICKET_ID) 	Order Submission Date
 Committed Due Date (DD) 	Committed Due Date
 Service Type(CLASS_SVC_DESC) 	Service Type
Hold Reason	Hold Reason
Total line/circuit count	Total line/circuit count
Geographic Scope	Geographic Scope
NOTE: Code in parentheses is the corresponding	
header found in the raw data file.	
Retail Analog/Benchmark:	
CLEC Residence Resale / BST Residence Retail	
CLEC Business Resale / BST Business Retail	
CLEC Non-UNE Design / BST Design	
Interconnection Trunks-CLEC / Interconnection Trunks	-BST
UNEs-(See Appendix D)	

Revision Date: 02/24/00 (taf)

Report/Measurement:

P-2. Average Jeopardy Notice Interval & Percentage of Orders Given Jeopardy Notices

When BST can determine in advance that a committed due date is in jeopardy, it will provide advance notice to the CLEC.

Exclusions:

- Orders held for CLEC end user reasons
- Orders submitted to BST through non-mechanized methods

Business Rules:

When BST can determine in advance that a committed due date is in jeopardy it will provide advance notice to the CLEC. The number of committed orders in a report period is the number of orders that have a due date in the reporting period.

Calculation:

Average Jeopardy Interval = Σ [(Date and Time of Scheduled Due Date on Service Order) - (Date and Time of Jeopardy Notice)]/[Number of Orders Notified of Jeopardy in Reporting Period).

Percent of Orders Given Jeopardy Notice = Σ [(Number of Orders Given Jeopardy Notices in Reporting Period) / (Number of Orders Confirmed (due) in Reporting Period)

Report Structure:

- CLEC Specific
- CLEC Aggregate

BST Aggregate	
Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
 CLEC Order Number and PON 	BST Order Number
 Date and Time Jeopardy Notice sent 	 Date and Time Jeopardy Notice sent
Committed Due Date	Committed Due Date
Service Type	Service type
NOTE: Code in parentheses is the corresponding header found in the raw data file.	
Retail Analog/Benchmark:	
95% > = 24 hours	

Revision Date: 01/05/00 (taf)

Report/Measurement:

P-3. Percent Missed Installation Appointments

Definition:

"Percent missed installation appointments" monitors the reliability of BST commitments with respect to committed due dates to assure that CLECs can reliably quote expected due dates to their retail customer as compared to BST.

Exclusions:

- Canceled Service Orders
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.)
- Disconnect (D) & From (F) orders
- End User Misses on Interconnection Trunks

Business Rules:

Percent Missed Installation Appointments is the percentage of total orders processed for which BST is unable to complete the service orders on the confirmed due dates. Missed Appointments caused by end-user reasons will be included and reported separately. A business day is any time period within the same date frame, which means there cannot be a cutoff time for commitments as certain types of orders are requested to be worked after standard business hours. Also, during Daylight Savings Time, field technicians are scheduled until 9PM in some areas and the customer is offered a greater range of intervals from which to select.

Calculation:

Percent Missed Installation Appointments = Σ (Number of Orders Not Complete by Committed Due Date in Reporting Period) / (Number of Orders Confirmed in Reporting Period) X 100

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Report explanation: The difference between End User MA and Total MA is the result of BST caused misses. Here, Total MA is the total % of orders missed either by BST or CLEC end user. The End User MA represents the percentage of orders missed by the CLEC or their end user.

Level of Disaggregation:

- Reported in categories of <10 lines/circuits; > = 10 lines/circuits
- Dispatch/No Dispatch

Retained Relating to BST Experience Report Month
Report Month
1
SST Order Number
Committed Due Date (DD)
Completion Date (CMPLTN DD)
tatus Type
tatus Notice Date
tandard Order Activity
Geographic Scope

Retail Analog/Benchmark:

CLEC Residence Resale / BST Residence Retail

CLEC Business Resale / BST Business Retail

CLEC Non-UNE Design / BST Design

Interconnection Trunks-CLEC / Interconnection Trunks -BST

UNEs-(See Appendix D)

Revision Date: 02/28/00 (taf)

Report/Measurement:

P-4. Average Completion Interval (OCI) & Order Completion Interval Distribution

Definition:

The "average completion interval" measure monitors the interval of time it takes BST to provide service for the CLEC or its' own customers. The "Order Completion Interval Distribution" provides the percentage of orders completed within certain time periods.

Exclusions:

- Canceled Service Orders
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.)
- D (Disconnect) and F (From) orders. (From is disconnect side of a move order when the customer moves to a new address).
- "L" Appointment coded orders (where the customer has requested a later than offered interval)

Business Rules:

The actual completion interval is determined for each order processed during the reporting period. The completion interval is the elapsed time from when BST issues a FOC or SOCS date time stamp receipt of an order from the CLEC to BST's actual order completion date. The clock starts when a valid order number is assigned by SOCS and stops when the technician or system completes the order in SOCS. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed.

The interval breakout for UNE and Design is: 0-5 = 0-4.99, 5-10 = 5-9.99, 10-15 = 10-14.99, 15-20 = 15-19.99 20-25 = 20-24.99, 25-30 = 25-29.99, >=30 = 30 and greater.

Calculation:

Average Completion Interval:

Σ [(Completion Date & Time) - (Order Issue Date & Time)] / Σ (Count of Orders Completed in Reporting period)

Order Completion Interval Distribution:

Σ (Service Orders Completed in "X" days) / (Total Service Orders Completed in Reporting Period) X 100

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

- ISDN Orders included in Non Design GA Only
- Dispatch/No Dispatch categories applicable to all levels except trunks.
- Residence & Business reported in day intervals = 0,1,2,3,4,5,5+
- UNE and Design reported in day intervals = 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, >=30
- All Levels are reported <10 line/circuits; >=10 line/circuits

(Average Completion Interval (OCI) & Order Completion Interval Distribution – Continued)

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
CLEC Company Name	BST Order Number
Order Number (PON)	Order Submission Date & Time
 Submission Date & Time (TICKET_ID) 	Order Completion Date & Time
 Completion Date (CMPLTN_DT) 	Service Type
• Service Type (CLASS_SVC_DESC)	Geographic Scope
Geographic Scope	
NOTE: Code in parentheses is the corresponding	
header found in the raw data file.	
Retail Analog/Benchmark	
CLEC Residence Resale / BST Residence Retail	
CLEC Business Resale / BST Business Retail	
CLEC Non-UNE Design / BST Design	
Interconnection Trunks-CLEC / Interconnection Tr	unks-BST
UNEs-(See Appendix D)	

Revision Date: 02/28/00 (taf)

Report/Measurement:

P-5. Average Completion Notice Interval

Definition:

The Completion Notice Interval is the elapsed time between the BST reported completion of work and the issuance of a valid completion notice to the CLEC.

Exclusions:

- · Non-mechanized Orders
- Cancelled Service Orders
- Order Activities of BST associated with internal or administrative use of local services
- D & F orders

Business Rules:

Measurement of interval of completion date and time by a field technician on dispatched orders, and 5PM start time on the due date for non-dispatched orders; to the release of a notice to the CLEC/BST of the completion status. The field technician notifies the CLEC the work was complete and then he enters the completion time stamp information in his computer. This information switches through to the SOCS systems either completing the order or rejecting the order to the Work Management Center (WMC). If the completion is rejected, it is manually corrected and then completed by the WMC. The notice is returned on each individual order submitted and as the notice is sent electronically, it can only be switched to those orders that were submitted by the CLEC electronically. The start time is the completion stamp either by the field technician or the 5PM due date stamp; the end time is the time stamp the notice was submitted to the CLEC/BST system.

Calculation:

 Σ (Date and Time of Notice of Completion) – (Date and Time of Work Completion) / (Number of Orders Completed in Reporting Period)

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

- Reporting intervals in Hours: 0-1, 1-2, 2-4, 4-8, 8-12, 12-24, > 24, plus Overall Average Hour Interval
- Reported in categories of <10 line/circuits; >= 10 line/circuits

Data Retained Relating to CLEC Experience Data Retained Relating to BST Experience • Report Month • Report Month • CLEC Order Number • BST Order Number • Work Completion Date • Work Completion Date • Work Completion Time • Work Completion Time • Completion Notice Availability Date • Completion Notice Availability Date • Completion Notice Availability Time • Completion Notice Availability Time • Service Type • Service Type • Activity Type Activity Type • Geographic Scope • Geographic Scope **NOTE:** Code in parentheses is the corresponding NOTE: Code in parentheses is the corresponding header header found in the raw data file. found in the raw data file.

Retail Analog/Benchmark:

CLEC Residence Resale / BST Residence Retail

CLEC Business Resale / BST Business Retail

CLEC Non-UNE Design / BST Design

Interconnection Trunks-CLEC / Interconnection Trunks-BST

UNEs – (See Appendix D)

Revision Date 02/24/00 (taf)

Report/Measurement:

P-6. Coordinated Customer Conversions

Definition:

This category measures the average time it takes BST to disconnect an unbundled loop from the BST switch and cross connect it to a CLEC's equipment. This measurement applies to service orders with and without INP, and where the CLEC has requested BST to provide a coordinated cutover.

Exclusions:

- Any order canceled by the CLEC will be excluded from this measurement.
- Delays due to CLEC following disconnection of the unbundled loop
- Unbundled Loops where there is no existing subscriber loop and loops where coordination in not requested.

Business Rules:

Where the service order includes INP, the interval includes the total time for the cutover including the translation time to place the line back in service on the ported line. The interval is calculated for the entire cutover time for the service order and then divided by items worked in that time to give the average per item interval for each service order.

Calculation:

 Σ [(Completion Date and Time for Cross Connection of an Coordinated Unbundled Loop)- (Disconnection Date and Time of an Coordinated Unbundled Loop)] / Total Number of Unbundled Loop with Coordinated Conversions (items) for the reporting period.

Report Structure:

- CLEC Specific
- CLEC Aggregate

Level of Disaggregation:

Reported in intervals <=5 minutes; >5,< =15 minutes; >15 minutes, plus Overall Average interval

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	No BST Analog Exists
CLEC Order Number	
• Committed Due Date (DD)	
 Service Type (CLASS_SVC_DESC) 	
Cutover Start Time	
Cutover Completion time	
 Portability start and completion times (INP orders) 	
 Total Conversions (Items) 	
NOTE: Code in parentheses is the corresponding header	
found in the raw data file.	

Retail Analog/Benchmark:

There is no retail analog for this measurement because it measures cutting loops to the CLEC.

Benchmark – See Appendix D

Revision Date: 02/28/00 (taf)

Report/Measurement:

P-7. % Provisioning Troubles within 30 days of Service Order Activity

Definition:

Percent Provisioning Troubles within 30 days of Installation measures the quality and accuracy of installation activities.

Exclusions:

- · Canceled Service Orders
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (R Orders, Test Orders, etc.)
- D & F orders

Business Rules:

Measures the quality and accuracy of completed orders. The first trouble report from a service order after completion is counted in this measure. Subsequent trouble reports are measured in Repeat Report Rate. Reports are calculated searching in the prior report period for completed service orders and following 30 days after completion for a trouble report.

D & F orders are excluded as there is no subsequent activity following a disconnect.

Calculation:

% Provisioning Troubles within 30 days of Service Order Activity = Σ (Trouble reports on all completed orders \leq 30 days following service order(s) completion) / (All Service Orders completed in the report calendar month) X 100

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

- Reported in categories of <10 line/circuits; > = 10 line/circuits
- Dispatch / No Dispatch

• Dispatcii / No Dispatcii	
Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
 CLEC Order Number and PON 	BST Order Number
 Order Submission Date(TICKET_ID) 	Order Submission Date
 Order Submission Time (TICKET_ID) 	Order Submission Time
Status Type	Status Type
 Status Notice Date 	Status Notice Date
 Standard Order Activity 	Standard Order Activity
 Geographic Scope 	Geographic Scope
NOTE: Code in parentheses is the corresponding	
header found in the raw data file.	

Retail Analog/Benchmark:

CLEC Residence Resale / BST Residence Retail

CLEC Business Resale / BST Business Retail

CLEC Non-UNE Design / BST Design

Interconnection Trunks-CLEC / Interconnection Trunks -BST

UNEs-(See Appendix D)

Revision Date: 02/28/00 (taf)

Report/Measurement:

P-8. Total Service Order Cycle Time (TSOCT)

Definition:

This report measures the total service order cycle time from receipt of a valid service order request to the completion of the service order.

Exclusions:

- Canceled Service Orders
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.)
- D (Disconnect) and F (From) orders. (From is disconnect side of a move order when the customer moves to a new address).
- "L" Appointment coded orders (where the customer has requested a later than offered interval)
- Orders with CLEC/Subscriber caused delays or CLEC/Subscriber requested due date changes.

Business Rules:

The interval is determined for each order processed during the reporting period. This measurement combines two reports: FOC (Firm Order Confirmation) with Average Order Completion Interval.

This interval starts with the receipt of a valid service order request and stops when the technician or system completes the order in SOCS. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed.

Calculation:

Total Service Order Cycle Time

 Σ (Date and Time of Service Request Receipt) – (Completion Date and Time of Service Order) (SOCS HIST-CD DATE) / (Count of Orders Completed in Reporting Period)

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

- Reported in categories of < 10 line/circuits; > = 10 line/circuits
- Dispatch/No Dispatch categories applicable to all levels except trunks.
- Intervals 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, > = 30 Days

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
 Report Month Interval for FOC CLEC Company Name Order Number (PON) Submission Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Geographic Scope 	 Report Month BST Order Number Order Submission Date & Time Order Completion Date & Time Service Type Geographic Scope
NOTE: Code in parentheses is the corresponding header found in the raw data file. Retail Analog/Benchmark	

Revision Date: 02/28/00 (taf)

See Appendix D

Report/Measurement:

P-9. Service Order Accuracy GEORGIA ONLY

Definition:

The "service order accuracy" measurement measures the accuracy and completeness of BST service orders by comparing what was ordered and what was completed.

Exclusions:

- Cancelled Service Orders
- Order Activities of BST associated with internal or administrative use of local services
- & F orders

Business Rules:

A manual sampling of service orders, completed during a monthly reporting period, is compared to the original account profile and the order that the CLEC sent to BST. An order is "completed without error" if all service attributes and account detail changes (as determined by comparing the original order) completely and accurately reflect the activity specified on the original order and any supplemental CLEC order.

Calculation:

Percent Service Order Accuracy = Σ (Orders Completed without Error) / Σ (Orders Completed in Reporting Period) x 100

Report Structure:

CLEC Aggregate

Level of Disaggregation:

- Reported in categories of <10 line/circuits; > = 10 line/circuits
- Dispatch / No Dispatch

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Being investigated at this time
 CLEC Order Number and PON 	
 Local Service Request (LSR) 	
 Order Submission Date 	
 Committed Due Date 	
Service Type	
Standard Order Activity	
NOTE: Code in parentheses is the corresponding header found in the raw data file.	

Retail Analog/Benchmark:

(Under Investigation)

Revision Date: 01/05/00 (taf)

Report/Measurement:

LNP – 10. Percent Missed Installation Appointments

Definition:

Percent Missed Installation Appointments monitors the reliability of BST commitments with respect to committed due dates to assure that CLECs can reliably quote expected due dates to their retail customer as compared to BST.

Exclusions:

- Canceled Service Orders
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules:

Percent Missed Installation Appointments (PMI) is the percentage of total orders processed for which BST is unable to complete the service orders on the committed due dates. Missed Appointments caused by end-user reasons will be included and reported in a separate category. A business day is any time period within the same date frame, which means there cannot be a cutoff time for commitments as certain types of orders are requested to be worked after standard business hours. Also, during Daylight Savings Time, field technicians are scheduled until 9PM in some areas and the customer is offered a greater range of intervals from which to select.

Calculation:

Percent Missed Installation Appointments:

[(Number of Orders Not Completed by Committed Due Date in Reporting Period) / (Number of Orders Completed in Reporting Period)] \times 100

Report Structure:

- Mechanized (service orders generated by LSRs submitted via EDI or TAG)
- CLEC Specific
- CLEC Aggregate

Report explanation: Total Missed Appointments is the total % of orders missed either by BST or the CLEC end user. End User MA represents the percentage of orders missed by the CLEC end user. The difference between End User Missed Appointments and Total Missed Appointments is the result of BST caused misses.

Level of Disaggregation:

- Product Reporting Levels
 - LNP
 - UNE Loop Associated w/LNP
 - Geographic Scope
 - State, Region

Retail Analog/Benchmark:

See Appendix D

Revision Date: 02/16/00 (taf)

PROVISIONING – (LNP)

Report/Measurement:

LNP-11. Average Disconnect Timeliness Interval & Disconnect Timeliness Interval Distribution

Definition:

Disconnect Timeliness is defined as the interval between the time the LNP Gateway receives the 'Number Ported' message from NPAC (signifying the CLEC 'Activate') until the time that the Disconnect service order for an LSR is completed in SOCS. This interval effectively measures BST responsiveness by isolating it from impacts that are caused by CLEC related activities.

Exclusions:

- •. Canceled Service Orders
- •. Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules:

The Disconnect Timeliness interval is determined for the last Disconnect service order processed on an LSR during the reporting period. The Disconnect Timeliness interval is the elapsed time from when BST receives the last 'Number Ported' message for an LSR from NPAC (signifying the CLEC 'Activate') until the last Disconnect service order is completed in SOCS. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the total number of selected disconnect orders which have been completed.

Calculation:

Average Disconnect Timeliness Interval:

 Σ [(Disconnect Service Order Completion Date & Time) - ('Number Ported' Message Received Date & Time)] / Σ (Total Number of Disconnect Service Orders Completed in Reporting Period)

Disconnect Timeliness Interval Distribution:

[Σ (Disconnect Service Orders Completed in "X" days) / (Total Disconnect Service Orders Completed in Reporting Period)] X 100

Report Structure:

- Mechanized (service orders generated by LSRs submitted via EDI or TAG)
- CLEC Specific
- CLEC Aggregate

Level of Disaggregation:

- Reported in day intervals = 0,1,2,3,4,5,>5 days
- Product Reporting Levels
 - >LNP
- Geographic Scope
 - ➤ State, Region

Retail Analog/Benchmark:

See Appendix D

Revision Date: 02/16/00 (taf)

Report/Measurement:

LNP-12. Total Service Order Cycle Time

Definition:

Total Service Order Cycle Time measures the interval from receipt of a valid service order request to the completion of the final service order associated with that service request.

Exclusions:

- Canceled Service Orders
- Order Activities of BST or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable
- "L" appointment coded orders (indicating the customer has requested a later than offered interval)
- "S" missed appointment coded orders (indicating subscriber missed reasons), except for "SP" codes (indicating subscriber prior due date requested).

Business Rules:

The interval is determined for each service request processed during the reporting period. This measurement combines two reports: FOC (Firm Order Confirmation) with Average Order Completion Interval.

This interval starts with the receipt of a valid service request and stops when the technician or system completes all the related service orders for the LSR in SOCS. Elapsed time for each service request is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of service requests completed to produce the total service order cycle time.

Calculation:

Average Total Service Order Cycle Time:

 Σ [(Service Order Completion Date & Time) - (Service Request Receipt Date & Time)] / Σ (Total Number Service Requests Completed in Reporting Period)

Total Service Order Cycle Time Interval Distribution:

[Σ (Total Number of Service Requests Completed in "X" minutes/hours) / (Total Number of Service Requests Received in Reporting Period)] X 100

Report Structure:

- Mechanized (service orders generated by LSRs submitted via EDI or TAG)
- CLEC Specific
- CLEC Aggregate
- "W" Appointment Code Only (Company Offered)

Level of Disaggregation:

- Reported in day intervals 0 5, 5 10, 10 15, 15 20, 20 25, 25 30, >30 days
- Product Reporting Levels
 - LNP
 - ➤ UNE Loop with LNP
- Geographic Scope
 - > State, Region

Retail Analog/Benchmark:

See Appendix D

Revision Date: 02/16/00 (taf)

Maintenance and Repair Level of Disaggregation

Product Reporting Levels

- Resale / Retail
 - ➤ Pots Residence
 - ➤ Pots Business
 - PBX (Louisiana SQM)
 - > ESSX (Louisiana SQM)
 - CENTREX (Louisiana SQM)
 - ➤ ISDN (Louisiana SQM) (NOTE: ISDN Troubles included in Non-Design Georgia Only)
 - Design
- Unbundled Network Elements
 - UNE Design
 - ➤ UNE Non Design
 - ➤ UNE 2 Wire Loop (Louisiana SQM)
 - ➤ UNE Loop Other (Louisiana SQM)
 - ➤ Unbundled Ports (Louisiana SQM)
 - ➤ UNE Other Non Design (Louisiana SQM)
- Trunks
 - ➤ Local Interconnection Trunks
- Dispatch/No Dispatch categories applicable to all product levels
- Geographic Scope

> State, Region and further geographic disaggregation as required by State Commission Order (e.g. Metropolitan Service Area – MSA)

MAINTENANCE & REPAIR

Report/Measurement:

M&R-1. Missed Repair Appointments

Definition:

The percent of trouble reports not cleared by the committed date and time.

Exclusions:

- Trouble tickets canceled at the CLEC request.
- BST trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules:

The negotiated commitment date and time is established when the repair report is received. The cleared time is the date and time that BST personnel clear the trouble and closes the trouble report in his Computer Access Terminal (CAT) or workstation. If this is after the Commitment time, the report is flagged as a "Missed Commitment" or a missed repair appointment. When the data for this measure is collected for BST and a CLEC, it can be used to compare the percentage of the time repair appointments are missed due to BST reasons. Note: Appointment intervals vary with force availability in the POTS environment. Specials and Trunk intervals are standard interval appointments of no greater than 24 hours.

Calculation:

Percentage of Missed Repair Appointments = Σ (Count of Customer Troubles Not Cleared by the Quoted Commitment Date and Time) / Σ (Total Trouble reports closed in Reporting Period) X 100

Report Structure:

- •. CLEC Specific
- •. CLEC Aggregate
- BST Aggregate

o. Boi Aggregate	
Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
 CLEC Company Name 	BST Company Code
 Submission Date & Time (TICKET_ID) 	Submission Date & Time
 Completion Date (CMPLTN_DT) 	Completion Date
 Service Type (CLASS_SVC_DESC) 	Service Type
 Disposition and Cause (CAUSE_CD & 	Disposition and Cause (Non-Design /Non-Special Only)
CAUSE_DESC)	Trouble Code (Design and Trunking Services)
Geographic Scope	Geographic Scope
NOTE: Code in parentheses is the corresponding	
header found in the raw data file.	

Retail Analog/Benchmark

- CLEC Residence-Resale / BST Residence-Retail
- CLEC Business-Resale / BST Business-Retail
- CLEC Design-Resale / BST Design-Retail
- CLEC PBX, Centrex, and ISDN Resale/ BST PBX, Centrex, and ISDN Retail
- CLEC Trunking-Resale / BST Trunking-Retail
- UNEs (See Appendix D)

Revision Date: 02/22/00 (see)

Report/Measurement:

M&R-2. Customer Trouble Report Rate

Definition:

Initial and repeated customer direct or referred troubles reported within a calendar month per 100 lines/ circuits in service.

Exclusions:

- Trouble tickets canceled at the CLEC request.
- BST trouble reports associated with administrative service.
- Customer provided Equipment (CPE) troubles or CLEC equipment troubles.

Business Rules:

Customer Trouble Report Rate is computed by accumulating the number of maintenance initial and repeated trouble reports during the reporting period. The resulting number of trouble reports are divided by the total "number of service" lines, ports or combination that exist for the CLEC's and BST respectively at the end of the report month.

Calculation:

Customer Trouble Report Rate = (Count of Initial and Repeated Trouble Reports in the Current Period) / (Number of Service Access Lines in service at End of the Report Period) X 100

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
CLEC Company Name	BST Company Code
 Ticket Submission Date & Time (TICKET_ID) 	 Ticket Submission Date & Time
 Ticket Completion Date (CMPLTN_DT) 	 Ticket Completion Date
 Service Type (CLASS_SVC_DESC) 	Service Type
 Disposition and Cause (CAUSE_CD & 	 Disposition and Cause (Non-Design / Non-Special
CAUSE_DESC)	Only)
 # Service Access Lines in Service at the end of 	 Trouble Code (Design and Trunking Services)
period	 # Service Access Lines in Service at the end of period
Geographic Scope	Geographic Scope
NOTE: Code in parentheses is the corresponding header found in the raw data file.	

Retail Analog/Benchmark:

CLEC Residence-Resale / BST Residence -Retail

CLEC Business-Resale / BST Business-Retail

CLEC Design-Resale / BST Design-Retail

CLEC PBX, Centrex and ISDN Resale/ BST PBX, Centrex, and ISDN Retail

 $CLEC\ Trunking-Resale\ /\ BST\ Trunking-Retail$

UNEs – (See Appendix D)

Revision Date: 02/22/00 (see)

Report/Measurement:

M&R-3. Maintenance Average Duration

Definition:

The Average duration of Customer Trouble Reports from the receipt of the Customer Trouble Report to the time the trouble report is cleared.

Exclusions:

- Trouble reports canceled at the CLEC request
- BST trouble reports associated with administrative service
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Troubles.
- Trouble reports greater than 10 days

Business Rules:

For Average Duration the clock starts on the date and time of the receipt of a correct repair request. The clock stops on the date and time the service is restored and the customer notified (when the technician completes the trouble ticket on his/her CAT or work system).

NOTE: Customer can be BST or CLEC

Calculation:

Maintenance Average Duration = Σ (Date and Time of Service Restoration) – (Date and Time Trouble Ticket was Opened) / Σ (Total Closed Troubles in the reporting period)

Report Structure:

- CLEC Specific
- BST Aggregate
- CLEC Aggregate

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
Total Tickets (LINE_NBR)	Total Tickets
CLEC Company Name	BST Company Code
 Ticket Submission Date & Time (TIME_ID) 	Ticket Submission Date
 Ticket Completion Date (CMPLTN_DT 	Ticket submission Time
Service Type (CLASS_SVC_DESC)	Ticket completion Date
 Disposition and Cause (CAUSE_CD & 	Ticket Completion Time
CAUSE_DESC)	Total Duration Time
Geographic Scope	Service Type
	 Disposition and Cause (Non – Design /Non-Special Only)
NOTE: Code in parentheses is the corresponding	 Trouble Code (Design and Trunking Services)
header found in the raw data file.	Geographic Scope

Retail Analog/Benchmark:

- CLEC Residence-Resale / BST Residence-Resale
- CLEC Business-Resale / BST Business-Retail
- CLEC Design-Resale / BST Design-Retail
- CLEC PBX, Centrex and ISDN Resale / BST PBX, Centrex and ISDN Retail
- CLEC Trunking-Resale /BST Trunking-Retail
- UNEs (See Appendix D)

Revision Date: 02/22/00 (see)

Report/Measurement:

M&R-4. Percent Repeat Troubles within 30 Days

Definition:

Trouble reports on the same line/circuit as a previous trouble report received within 30 calendar days as a percent of total troubles reported.

Exclusions:

- Trouble Reports canceled at the CLEC request
- BST Trouble Reports associated with administrative service
- Customer Provided Equipment (CPE) Troubles or CLEC Equipment Troubles.

Business Rules:

Includes Customer trouble reports received within 30 days of an original Customer trouble report.

Calculation:

Percent Repeat Troubles within $30 \text{ Days} = (\text{Count of Customer Troubles where more than one trouble report was logged for the same service line within a continuous <math>30 \text{ days}) / (\text{Total Trouble Reports Closed in Reporting Period}) X 100$

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

5 BST Aggregate	
Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
 Total Tickets (LINE_NBR) 	Total Tickets
 CLEC Company Name 	BST Company Code
• Ticket Submission Date & Time (TICKET_ID)	Ticket Submission Date
 Ticket Completion Date (CMPLTN_DT) 	Ticket Submission Time
 Total and Percent Repeat Trouble Reports 	Ticket Completion Date
within 30 Days (TOT_REPEAT)	Ticket Completion Time
 Service Type 	 Total and Percent Repeat Trouble Reports within 30 Days
 Disposition and Cause (CAUSE_CD & 	Service Type
CAUSE_DESC)	 Disposition and Cause (Non – Design/Non-Special only)
 Geographic Scope 	 Trouble Code (Design and Trunking Services)
NOTE: Code parentheses is the corresponding header format found in the raw data file.	Geographic Scope

Retail Analog/Benchmark:

CLEC Residence-Resale / BST Residence-Retail

CLEC Business-Resale / BST Business-Retail

 $CLEC\ Design-Resale\ /\ BST\ Design-Retail$

CLEC PBX, Centrex and ISDN Resale / BST PBX, Centrex and ISDN Retail

CLEC Trunking-Resale / BST Trunking-Retail

UNEs – Retail Analog (See Appendix D)

Revision date: 02/22/00 (see)

Report/Measurement:

M&R-5. Out of Service (OOS) > 24 Hours

Definition:

For Out of Service Troubles (no dial tone, cannot be called or cannot call out) the percentage of troubles cleared in excess of 24 hours. (All design services are considered to be out of service).

Exclusions:

- Trouble Reports canceled at the CLEC request
- BST Trouble Reports associated with administrative service
- Customer Provided Equipment (CPE) Troubles or CLEC Equipment Troubles.

Business Rules:

Customer Trouble reports that are out of service and cleared in excess of 24 hours. The clock begins when the trouble report is created in LMOS and the trouble is counted if the time exceeds 24 hours.

Calculation:

Out of Service (OOS) > 24 hours = (Total Troubles OOS > 24 Hours) / Total OOS Troubles in Reporting Period) X 100

Report Structure:

- CLEC Specific
- BST Aggregate
- CLEC Aggregate

CLEC Aggregate		
Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience	
Report Month	Report Month	
 Total Tickets 	Total Tickets	
 CLEC Company Name 	BST Company Code	
 Ticket Submission Date & Time (TICKET_ID) 	Ticket Submission Date	
 Ticket Completion Date (CMPLTN_DT 	Ticket Submission time	
 Percentage of Customer Troubles out of 	Ticket Completion Date	
 Service > 24 Hours (OOS>24_FLAG) 	Ticket Completion Time	
 Service type (CLASS_SVC_DESC) 	 Percent of Customer Troubles out of Service > 24 Hours 	
 Disposition and Cause (CAUSE_CD & 	Service type	
CAUSE-DESC)	 Disposition and Cause (Non – Design/Non-Special only) 	
 Geographic Scope 	Trouble Code (Design and Trunking Services)	
	Geographic Scope	
NOTE: Code in parentheses is the corresponding		
header found in the raw data file.		

Retail Analog/Benchmark:

- CLEC Residence-Resale / BST Residence- Retail
- CLEC Business-Resale / BST Business-Retail
- CLEC Design-Resale / BST Design-Retail
- CLEC PBX, Centrex and ISDN Resale / BST PBX, Centrex and ISDN Retail
- CLEC Trunking-Resale /BST Trunking- Retail
- UNEs Retail Analog (See Appendix D)

Revision Date: 02/22/00 (see)

Report/Measurement:

M&R-6. Average Answer Time – Repair Centers

Definition:

This measures the average time a customers is in Que.

Exclusions:

None

Business Rules:

This measure is designed to measure the time required for CLEC & BST from the time of the ACD choice to the time of being answered. The clock starts when the CLEC Rep makes a choice to be put in queue for the next repair attendant and the clock stops when the repair attendant answers the call.

(NOTE: The Column is a combined BST Residence and Business number)

Level of Disaggregation:

Region. CLEC/BST Service Centers and BST Repair Centers are regional.

Calculation:

Average Answer Time for BST's Repair Centers = (Time BST Repair Attendant Answers Call) – (Time of entry into queue until ACD Selection) / (Total number of calls by reporting period)

Report Structure:

- CLEC Aggregate
- BST Aggregate

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience	
CLEC Average Answer Time	BST Average Answer Time	

Retail Analog/Benchmark:

For CLEC, Average Answer Times in UNE Center and BRMC are comparable to the Average Answer Times in the BST Repair Centers.

See Appendix D

Revision Date: 02/22/00 (see)

Report/Measurement:

B-1. Invoice Accuracy

Definition:

This measure provides the percentage of accuracy of the billing invoices rendered to CLECs during the current month.

Exclusions:

Adjustments not related to billing errors (e.g., credits for service outage, special promotion credits, adjustments to satisfy the customer)

Business Rules:

The accuracy of billing invoices delivered by BST to the CLEC must enable them to provide a degree of billing accuracy comparative to BST bills rendered to retail customers BST. CLECs request adjustments on bills determined to be incorrect. The BellSouth Billing verification process includes manually analyzing a sample of local bills from each bill period. The bill verification process draws from a mix of different customer billing options and types of service. An end-to-end auditing process is performed for new products and services. Internal measurements and controls are maintained on all billing processes.

Calculation:

Invoice Accuracy = (Total Billed Revenues during current month) - (Billing Related Adjustments during current month) / Total Billed Revenues during current month X 100

Report Structure:

- **CLEC Specific**
- **CLEC Aggregate**
- **BST** Aggregate

Level of Disaggregation:

- Product / Invoice Type
 - Resale
 - UNE
 - Interconnection
- Geographic Scope
 - Region

Data Retained Relating to CLEC Experience:	Data Retained Relating to BST Performance:
Report Month	Report Month
Invoice Type	Retail Type
Total Billed Revenue	> CRIS
Billing Related Adjustments	➤ CABS
	Total Billed Revenue
	Billing Related Adjustments
Retail Analog/Renchmark	

Retail Analog/Benchmark

CLEC Invoice Accuracy is comparable to BST Invoice Accuracy See Appendix D

Report/Measurement:

B-2. Mean Time to Deliver Invoices

Definition:

This measure provides the mean interval for billing invoices

Exclusions:

Any invoices rejected due to formatting or content errors.

Business Rules:

Measures the mean interval for timeliness of billing records delivered to CLECs in an agreed upon format. CRIS-based invoices are measured in business days, and CABS-based invoices in calendar days.

Calculation:

Mean Time To Deliver Invoices = Σ _[(Invoice Transmission Date)– (Close Date of Scheduled Bill Cycle)] / (Count of Invoices Transmitted in Reporting Period)

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

- Product / Invoice Type
 - > Resale
 - > UNE
 - Interconnection
- Geographic Scope
 - Region

Data Retained Relating to CLEC Experience:	Data Retained Relating to BST Performance:	
Report Month	Report Month	
Invoice Type	Retail Type	
 Invoice Transmission Count 	> CRIS	
 Date of Scheduled Bill Close 	> CABS	
	Invoice Transmission Count	
	Date of Scheduled Bill Close	

Retail Analog/Benchmark:

CRIS-based invoices will be released for delivery within six (6) business days

CABS-based invoices will be released for delivery within eight (8) calendar days.

CLEC Average Delivery Intervals for both CRIS and CABS Invoices are comparable to BST Average delivery for both systems.

See Appendix D

Report/Measurement:

B-3. Usage Data Delivery Accuracy

Definition:

This measurement captures the percentage of recorded usage that is delivered error free and in an acceptable format to the appropriate Competitive Local Exchange Carrier (CLEC). These percentages will provide the necessary data for use as a comparative measurement for BellSouth performance. This measurement captures Data Delivery Accuracy rather than the accuracy of the individual usage recording.

Exclusions:

None

Business Rules:

The accuracy of the data delivery of usage records delivered by BST to the CLEC must enable them to provide a degree of accuracy comparative to BST bills rendered to their retail customers. If errors are detected in the delivery process, they are investigated, evaluated and documented. Errors are corrected and the data retransmitted to the CLEC.

Calculations:

Usage Data Delivery Accuracy = Σ [(Total number of usage data packs sent during current month) – (Total number of usage data packs requiring retransmission during current month)] / (Total number of usage data packs sent during current month) X 100

Report Structure:

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

- Geographic Scope
 - > Region

Data Retained Relating to CLEC Experience:	Data Retained Relating to BST Performance:
Report Month	Report Month
Record Type	Record Type
BellSouth Recorded	
Non BellSouth Recorded	

Retail Analog/Benchmark:

CLEC Usage Data Delivery Accuracy is comparable to BST Usage Data Delivery Accuracy See Appendix D

Report/Measurement:

B-4. Usage Data Delivery Completeness

Definition:

This measurement provides percentage of complete and accurately recorded usage data (usage recorded by BellSouth and usage recorded by other companies and sent to BST for billing) that is processed and transmitted to the CLEC within thirty (30) days of the message recording date. A parity measure is also provided showing completeness of BST messages processed and transmitted via CMDS. BellSouth delivers its own retail usage from recording location to billing location via CMDS as well as delivering billing data to other companies. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions:

None

Business Rules:

The purpose of these measurements is to demonstrate the level of quality of usage data delivered to the appropriate CLEC. Method of delivery is at the option of the CLEC.

Calculation:

Usage Data Delivery Completeness = Σ (Total number of Recorded usage records delivered during the current month that are within thirty (30) days of the message recording date) / Σ (Total number of Recorded usage records delivered during the current month) X 100

Report Structure

- CLEC Specific
- CLEC Aggregate
- BST Aggregate

Level of Disaggregation:

- Geographic Scope
 - Region

Data Retained Relating to CLEC Experience:	Data Retained Relating to BST Performance:	
Report Month	Report Monthly	
Record Type	Record Type	
BellSouth Recorded	· -	
Non BellSouth Recorded		

Retail Analog/Benchmark:

CLEC Usage Delivery Completeness is comparable to BST Usage Delivery Completeness See Appendix D

Report/Measurement:

B-5. Usage Data Delivery Timeliness

Definition:

This measurement provides a percentage of recorded usage data (usage recorded by BST and usage recorded by other companies and sent to BST for billing) that is delivered to the appropriate CLEC within six (6) calendar days from the receipt of the initial recording. A parity measure is also provided showing timeliness of BST messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions:

None

Business Rules:

The purpose of this measurement is to demonstrate the level of timeliness for processing and transmission of usage data delivered to the appropriate CLEC. The usage data will be mechanically transmitted or mailed to the CLEC data processing center once daily. The Timeliness interval of usage recorded by other companies is measured from the date BST receives the records to the date BST distributes to the CLEC. Method of delivery is at the option of the CLEC.

Calculation:

Usage Data Delivery Timeliness = Σ (Total number of usage records sent within six (6) calendar days from initial recording/receipt) / Σ (Total number of usage records sent) X 100

Report Structure:

- CLEC Aggregate
- CLEC Specific
- BST Aggregate

Level of Disaggregation:

- Geographic Scope
 - Region

Data Retained Relating to CLEC Experience:	Data Retained Relating to BST Performance:	
Report Month	Report Monthly	
Record Type	Record Type	
BellSouth Recorded		
Non-BellSouth Recorded		

Retail Analog/Benchmark:

CLEC Usage Data Delivery Timeliness is comparable to BST Usage Data Delivery Timeliness See Appendix D

Report/Measurement:

B-6. Mean Time to Deliver Usage

Definition:

This measurement provides the average time it takes to deliver Usage Records to a CLEC. A parity measure is also provided showing timeliness of BST messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions:

None

Business Rules:

The purpose of this measurement is to demonstrate the average number of days it takes BST to deliver Usage data to the appropriate CLEC. Usage data is mechanically transmitted or mailed to the CLEC data processing center once daily. Method of delivery is at the option of the CLEC.

Calculation:

 $\label{eq:mean_to_def} \mbox{Mean Time to Deliver Usage} = \Sigma _(\mbox{Record volume X estimated number of days to deliver the Usage Record)} \, / \, total \, record volume$

Report Structure:

- CLEC Aggregate
- CLEC Specific
- BST Aggregate

Level of Disaggregation:

• Geographic Scope

Region

Data Retained Relating to CLEC Experience:	Data Retained Relating to BST Performance:
Report Month	Report Monthly
 Record Type 	Record Type
BellSouth Recorded	
Non-BellSouth Recorded	

Retail Analog/Benchmark:

Mean Time to Deliver Usage to CLEC is comparable to Mean Time to Deliver Usage to BST See Appendix D

Report/Measurement:

OS-1. Speed to Answer Performance/Average Speed to Answer - Toll

Definition:

Measurement of the average time in seconds calls wait before answered by a toll operator.

Exclusions:

Calls abandoned by customers are not reflected in the average speed to answer but are reflected in the conversion tables where the percent answered within "X" seconds is determined.

Business Rules:

The call waiting measurement scan starts when the customer enters the queue and ends when a BST representative answers the call. The average speed to answer is determined by measuring and accumulating the seconds of wait time from the entry of a customer into the BST call management system queue until the customer is transferred to a BST representative. No distinction is made between CLEC customers and BST customers.

Calculation:

The Average Speed to Answer for toll is calculated by using data from monthly system measurement reports taken from the centralized call routing switches. The "total call waiting seconds" is a sub-component of this measure which BST systems calculate by monitoring the number of calls in queue throughout the day multiplied by the time (in seconds) between monitoring events. The "total calls served" is the other sub-component of this measure, which BST systems record as the total number of calls handled by Operator Services toll centers. Since calls abandoned are not reflected in the calculation, the percent answered within the required timeframe is determined by using conversion tables with input for the abandonment rate.

Report Structure:

Reported for the aggregate of BST and CLECs

> State

Level of Disaggregation:

None

Data Retained (on Aggregate Basis)

For the items below, BST's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP.

- Month
- Call Type (Toll)
- Average Speed of Answer

Retail Analog/Benchmark

Parity by Design

See Appendix D

Report/Measurement:

OS-2. Speed to Answer Performance/Percent Answered within "X" Seconds - Toll

Definition:

Measurement of the percent of toll calls that are answered in less than "X" seconds. The number of seconds represented by "X" is thirty, except where a different regulatory benchmark has been set against the Average Speed to Answer by a State Commission.

Exclusions:

Calls abandoned by customers are not reflected in the average speed to answer but are reflected in the conversion tables where the percent answered within "X" seconds is determined.

Business Rules:

The call waiting measurement scan starts when the customer enters the queue and ends when a BST representative answers the call. The average speed to answer is determined by measuring and accumulating the seconds of wait time from the entry of a customer into the BST call management system queue until the customer is transferred to a BST representative. No distinction is made between CLEC customers and BST customers.

Calculation:

The Percent Answered within "X" Seconds measurement for toll is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure:

- Reported for the aggregate of BST and CLECs
 - > State

Level of Disaggregation:

None

Data Retained (on Aggregate Basis)

For the items below, BST's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP.

- Month
- Call Type (Toll)
- Average Speed of Answer

Retail Analog/Benchmark

Parity by Design

See Appendix D

Report/Measurement:

OS-3. Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA)

Definition:

Measurement of the average time in seconds calls wait before answer by a DA operator.

Exclusions:

Calls abandoned by customers are not reflected in the average speed to answer but are reflected in the conversion tables where the percent answered within "X" seconds is determined.

Business Rules:

The call waiting measurement scan starts when the customer enters the queue and ends when a BST representative answers the call. The average speed to answer is determined by measuring and accumulating the seconds of wait time from the entry of a customer into the BST call management system queue until the customer is transferred to a BST representative. No distinction is made between CLEC customers and BST customers.

Calculation:

The Average Speed to Answer for DA is calculated by using data from monthly system measurement reports taken from the centralized call routing switches. The "total call waiting seconds" is a sub-component of this measure which BST systems calculate by monitoring the number of calls in queue throughout the day multiplied by the time (in seconds) between monitoring events. The "total calls served" is the other sub-component of this measure, which BST systems record as the total number of calls handled by Operator Services DA centers. Since calls abandoned are not reflected in the calculation, the percent answered within the required timeframe is determined by using conversion tables with input for the abandonment rate.

Report Structure:

- Reported for the aggregate of BST and CLECs
 - > State

Level of Disaggregation:

None

Data Retained (on Aggregate Basis)

For the items below, BST's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP.

- Month
- Call Type (DA)
- Average Speed of Answer

Retail Analog/Benchmark

Parity by Design

See Appendix D

Report/Measurement:

OS-4. Speed to Answer Performance/Percent Answered within "X" Seconds – Directory Assistance (DA)

Definition:

Measurement of the percent of DA calls that are answered in less than "X" seconds. The number of seconds represented by "X" is twenty, except where a different regulatory benchmark has been set against the Average Speed to Answer by a State Commission.

Exclusions:

Calls abandoned by customers are not reflected in the average speed to answer but are reflected in the conversion tables where the percent answered within "X" seconds is determined.

Business Rules:

The call waiting measurement scan starts when the customer enters the queue and ends when a BST representative answers the call. The average speed to answer is determined by measuring and accumulating the seconds of wait time from the entry of a customer into the BST call management system queue until the customer is transferred to a BST representative. No distinction is made between CLEC customers and BST customers.

Calculation:

The Percent Answered within "X" Seconds measurement for DA is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure:

- Reported for the aggregate of BST and CLECs
 - > State

Level of Disaggregation:

None

Data Retained (on Aggregate Basis)

For the items below, BST's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP.

- Month
- Call Type (DA)
- Average Speed of Answer

Retail Analog/Benchmark

Parity by Design

See Appendix D

E911

Report/Measurement:

E-1. Timeliness

Definition:

Measures the percentage of batch orders for E911 database updates (to CLEC resale and BST retail records) processed successfully within a 24-hour period.

Exclusions:

- Any resale order canceled by a CLEC
- Facilities-based CLEC orders

Business Rules:

The 24-hour processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Mechanical processing starts when SCC (BST's E911 vendor) receives E911 files containing batch orders extracted from BST's Service Order Communication System (SOCS). Processing stops when SCC loads the individual records to the E911 database. No distinctions are made between CLEC resale records and BST retail records.

Calculation:

E911 Timeliness = Σ (Number of batch orders processed within 24 hours ÷ Total number of batch orders submitted) X 100

Report Structure:

- Reported for the aggregate of CLEC resale updates and BST retail updates
 - > State
 - Region

Levels of Disaggregation:

None

Data Retained

- Report month
- Aggregate data

Retail Analog/Benchmark

Parity by Design

See Appendix D

E911

Report/Measurement:

E-2. Accuracy

Definition:

Measures the individual E911 telephone number (TN) record updates (to CLEC resale and BST retail records) processed successfully for E911 with no errors.

Exclusions:

- Any resale order canceled by a CLEC
- Facilities-based CLEC orders

Business Rules:

Accuracy is based on the number of records processed without error at the conclusion of the processing cycle. Mechanical processing starts when SCC (BST's E911 vendor) receives E911 files containing telephone number (TN) records extracted from BST's Service Order Communication System (SOCS). No distinctions are made between CLEC resale records and BST retail records.

Calculation:

E911 Accuracy = Σ (Number of record individual updates processed with no errors \div Total number of individual record updates) X 100

Report Structure:

- Reported for the aggregate of CLEC resale updates and BST retail updates
 - > State
 - > Region

Level of Disaggregation:

None

Data Retained

- Report month
- Aggregate data

Retail Analog/Benchmark

Parity by Design

See Appendix D

E911

Report/Measurement:

E-3. Mean Interval

Definition:

Measures the mean interval processing of E911 batch orders (to update CLEC resale and BST retail records).

Exclusions:

- Any resale order canceled by a CLEC
- Facilities-based CLEC orders

Business Rules:

The processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Data is posted in 4-hour increments up to and beyond 24 hours. No distinctions are made between CLEC resale records and BST retail records.

Calculation:

E911 Mean Interval = Σ (Date and time of batch order completion – Date and time of batch order submission) \div (Number of batch orders completed)

Report Structure:

- Reported for the aggregate of CLEC resale updates and BST retail updates
 - > State
 - > Region

Level of Disaggregation:

None

Data Retained (on Aggregate Basis)

- Report month
- Aggregate data

Retail Analog/Benchmark

Parity by Design

See Appendix D

TRUNK GROUP PERFORMANCE

Report/Measurement:

TGP-1. Trunk Group Performance-Aggregate

Definition:

A report of aggregate blocking information for CLEC trunk groups and BellSouth trunk groups.

Exclusions:

- Trunk Groups for which valid data is not available for an entire study period
- Duplicate trunk group information

Business Rules:

- Aggregate blocking results are created using the statistical analysis package and are output into Excel with separate table for each geographic area.
- For each geographic area, plots are generated for: a) the monthly blocking by hour for each affecting group (BellSouth or CLEC), and b) the difference between BellSouth blocking data and CLEC blocking data is calculated and plotted.
- The TCBH blocking is calculated by determining the monthly averaging blocking for each hour for each trunk. The hour with the highest usage is selected as the TCBH and the blocking for that hour is reported.
- Trunk Categorization: This report displays, over a reporting cycle, aggregate, weighted average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for two aggregate groups of selected trunk groups. These groups are CLEC affecting and BellSouth affecting trunk groups. In order to assign trunk groups to each aggregate group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups to that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows:

CLEC Affecting Categories:

	<u>Point A</u>	<u>Point B</u>
Category 1:	BellSouth End Office	BellSouth Access Tandem
Category 3:	BellSouth End Office	CLEC Switch
Category 4:	BellSouth Local Tandem	CLEC Switch
Category 5:	BellSouth Access Tandem	CLEC Switch
Category 10:	BellSouth End Office	BellSouth Local Tandem
Category 16:	BellSouth Tandem	BellSouth Tandem

BellSouth Affecting Category:

•	Point A	<u>Point B</u>
Category 9:	BellSouth End Office	BellSouth End Office

TRUNK GROUP PERFORMANCE - (Trunk Group Performance-Aggregate - Continued)

Calculation:

Monthly Weighted Average Blocking:

(Blocking data for each hour X number of valid measurement days within each week) $/ \Sigma$ (Total number of valid measurement days within each week)

Example:		Week 1	Week 2	Week 3	Week 4	Monthly
Hour						
1	Blocking	1%	0.5%	2%	1.5%	1.8%
	# Days	7	7	5	6	
2	Blocking	0%	0%	0.2%	0.3%	.1%
	# Days	7	5	5	7	
3	Blocking	1%	1%	0.5%	2%	1.1%
	# Days	7	7	7	7	
24	Blocking	1%	0.5%	2%	1.5%	1.2%
	# Days	7	7	5	6	

The monthly weighted average blocking for hour 1 for a particular trunk group is calculated as follows:

(1x5)+(0.5x5)+(2x4)+(1.5x4) = 1.2%

(5+5+4+4)

Aggregate Monthly Blocking:

(Monthly weighted average blocking value for each trunk group) X (number of trunks within each trunk group) $/ \Sigma$ (number of trunks in the aggregate group)

Example:	Trunk Group	Trunks in Service	Blocking Hour 1	Blocking Hour 2	Blocking Hour 3	Blocking Hour 4	BlockingHour 24
	A	24	3%	0%	1%	0%	0%
	В	144	2%	0%	1%	0.5%	0.5%
	C	528	0%	0.5%	1%	1%	1%
	D	316	1%	0%	1%	0.1%	0%
	E	940	1%	1%	4%	0%	0%
	Aggregate		0.8%	0.6%	2.4%	0.3%	0.3%

The aggregate weighted monthly blocking for hour 1 is calculated as follows:

(3x24)+(2x144)+(0x528)+(1x316)+(1x940) = 0.8%

(24+144+528+316+940)

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BST trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Report Structure:

CLEC Aggregate

> State

Level of Disaggregation:

Trunk Group

Truik Group				
Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience			
Report Month	Report Month			
Total Trunk Groups	Total Trunk Groups			
 Number of Trunk Groups by CLEC 	Aggregate Hourly average blocking			
 Hourly average blocking per trunk group 				

Retail Analog/Benchmark:

Any 2 hour period in 24 hours where CLEC blockage exceeds BST blockage by more than 0.5% = a miss using trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for BST.

TRUNK GROUP PERFORMANCE

Report/Measurement:

TGP-2. Trunk Group Performance-CLEC Specific

Definition:

A report of blocking information for CLEC trunk groups.

Exclusions:

- Trunk Groups for which valid data is not available for an entire study period
- Duplicate trunk group information

Business Rules:

- Aggregate blocking results are created using the statistical analysis package and are output into Excel with separate table for each geographic area.
- For each geographic area, plots are generated for the monthly blocking by hour
- The TCBH blocking is calculated by determining the monthly averaging blocking for each hour for each trunk. The hour with the highest usage is selected as the TCBH and the blocking for that hour is reported.
- Trunk Categorization: This report displays, over a reporting cycle, aggregate, weighted average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for CLEC trunk groups. In order to assign trunk groups to the CLEC group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups to that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows:

CLEC Affecting Categories:

	Point A	Point B	
Category 1:	BellSouth End Office	BellSouth Access Tandem	
Category 3:	BellSouth End Office	CLEC Switch	
Category 4:	BellSouth Local Tandem	CLEC Switch	
Category 5:	BellSouth Access Tandem	CLEC Switch	
Category 10:	BellSouth End Office	BellSouth Local Tandem	
Category 16:	BellSouth Tandem	BellSouth Tandem	

TRUNK GROUP PERFORMANCE - (Trunk Group Performance-CLEC Specific – Continued)

Calculation:

Monthly Weighted Average Blocking:

(Blocking data for each hour X number of valid measurement days within each week) / Σ (Total number of valid measurement days within each week)

Example:		Week 1	Week 2	Week 3	Week 4	<u>Monthly</u>
Hour						
1	Blocking	1%	0.5%	2%	1.5%	1.8%
	# Days	7	7	5	6	
2	Blocking	0%	0%	0.2%	0.3%	.1%
	# Days	7	5	5	7	
3	Blocking	1%	1%	0.5%	2%	1.1%
	# Days	7	7	7	7	5
24	Blocking	1%	0.5%	2%	1.5%	1.2%
	# Days	7	7	5	6	

The monthly weighted average blocking for hour 1 for a particular trunk group is calculated as follows:

(1x5)+(0.5x5)+(2x4)+(1.5x4) = 1.2%

(5+5+4+4)

Aggregate Monthly Blocking:

(Monthly weighted average blocking value for each trunk group) X (number of trunks within each trunk group) / Σ (number of trunks in the aggregate group)

Example:	Trunk	Trunks in	Blocking	Blocking	Blocking	Blocking	Blocking
	Group	Service	Hour 1	Hour 2	Hour 3	Hour 4	<u>Hour 24</u>
	A	24	3%	0%	1%	0%	0%
	В	144	2%	0%	1%	0.5%	0.5%
	C	528	0%	0.5%	1%	1%	1%
	D	316	1%	0%	1%	0.1%	0%
	Е	940	1%	1%	4%	0%	0%
	Aggregate		0.8%	0.6%	2.4%	0.3%	0.3%

The aggregate weighted monthly blocking for hour 1 is calculated as follows:

(3x24)+(2x144)+(0x528)+(1x316)+(1x940) = 0.8%

(24+144+528+316+940)

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BST trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Report Structure:

- CLEC Specific
- Trunk Group

Level of Disaggregation:

Trunk Group

Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report Month	Report Month
Total Trunk Groups	Total Trunk Groups
 Number of Trunk Groups by CLEC 	Aggregate Hourly average blocking
 Hourly average blocking per trunk group 	

Retail Analog/Benchmark:

Any 2 hour period in 24 hours where CLEC blockage exceeds BST blockage by more than 0.5% = a miss using trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for BST.

TRUNK GROUP PERFORMANCE

Report/Measurement:

TGP-3. Trunk Group Service Report

Definition:

A report of the percent blocking above the Measured Blocking Threshold (MBT) on all final trunk groups between CLEC Points of Termination and BST end offices or tandems.

Exclusions:

- Trunk groups for which valid traffic data is not available
- High use trunk groups

Business Rules:

Traffic trunking data measurements are validated and processed by the Total Network Data System/Trunking (TNDS/TK), a Telcordia (BellCore) supported application, on an hourly basis for Average Business Days (Monday through Friday). The traffic load sets, including offered load and observed blocking ratio (calls blocked divided by calls attempted), are averaged for a 20 day period, and the busy hour is selected. The busy hour average data for each trunk group is captured for reporting purposes. Although all trunk groups are available for reporting, the report highlight those trunk groups with blocking greater than the Measured Blocking Threshold (MBT) and the number of consecutive monthly reports that the trunk group blocking has exceeded the MBT. The MBT for CTTG is 2% and the MBT for all other trunk groups is 3%.

Calculation:

Measured blocking = (Total number of blocked calls) / (Total number of attempted calls) X 100

Report Structure:

- **BST** Aggregate
 - > CTTG
 - ➤ Local
- **CLEC** Aggregate
 - ➤ BST Administered CLEC Trunk
 - CLEC Administered CLEC Trunk
- **CLEC Specific**
 - ➤ BST Administered CLEC Trunk
 - CLEC Administered CLEC Trunk

Level of Disaggregation:

State

State	
Data Retained Relating to CLEC Experience	Data Retained Relating to BST Experience
Report month	Report month
Total trunk groups	Total trunk groups
Total trunk groups for which data is available	 Total trunk groups for which data is available
 Trunk groups with blocking greater than the 	 Trunk groups with blocking greater than the MBT
MBT	 Percent of trunk groups with blocking greater than the MBT
 Percent of trunk groups with blocking greater 	
than the MBT	
Retail Analog/Renchmark	

CLEC Trunk Blockage/BST Trunk Blockage

See Appendix D

TRUNK GROUP PERFORMANCE

Report/Measurement:

TGP-4. Trunk Group Service Detail

Definition:

A detailed list of all final trunk groups between CLEC Points of Presence and BST end offices or tandems, and the actual blocking performance when the blocking exceeds the Measured Blocking Threshold (MBT) for the trunk groups.

Exclusions:

- Trunk groups for which valid traffic data is not available
- High use trunk groups

Business Rules:

Traffic trunking data measurements are validated and processed by the Total Network Data System/Trunking (TNDS/TK), a Telcordia (BellCore) supported application, on an hourly basis for Average Business Days (Monday through Friday). The traffic load sets, including offered load and observed blocking ratio (calls blocked divided by calls attempted), are averaged for a 20 day period, and the busy hour is selected. The busy hour average data for each trunk group is captured for reporting purposes. Although all trunk groups are available for reporting, the report highlight those trunk groups with blocking greater than the Measured Blocking Threshold (MBT) and the number of consecutive monthly reports that the trunk group blocking has exceeded the MBT. The MBT for CTTG is 2% and the MBT for all other trunk groups is 3%.

Calculation:

Measured Blocking = (Total number of blocked calls) / (Total number of attempted calls) X 100

Report Structure:

- . BST Specific
 - Traffic Identity
 - > TGSN
 - > Tandem
 - End Office
 - Description
 - Observed Blocking
 - Busy Hour
 - Number Trunks
 - Valid study days
 - Number reports
 - Remarks

- CLEC Specific
 - > Traffic Identity
 - > TGSN
 - > Tandem
 - CLEC POT
 - Description
 - Observed Blocking
 - Busy Hour
 - Number Trunks
 - Valid study days
 - Number reports
 - Remarks

Level of Disaggregation:

State

Data Retained Relating to CLEC Experience

- Report month
- Total trunk groups
- Total trunk groups for which data is available
- Trunk groups with blocking greater than the MBT
- Percent of trunk groups with blocking greater than the MBT
- Traffic identity, TGSN, end points, description, busy hour, valid study days, number reports

Data Retained Relating to BST Experience

- Report month
- Total trunk groups
- Total trunk groups for which data is available
- Trunk groups with blocking greater than the MBT
- Percent of trunk groups with blocking greater than the MBT
- Traffic identity, TGSN, end points, description, busy hour, valid study days, number reports

Retail Analog/Benchmark:

CLEC Trunk Blockage/BST Blockage

See Appendix D

COLLOCATION

Report/Measurement:

C-1. Average Response Time

Definition:

Measures the average time (counted in business days) from the receipt of a complete and accurate collocation application (including receipt of application fees) to the date BellSouth responds in writing.

Exclusions:

- Requests to augment previously completed arrangements
- Any application cancelled by the CLEC

Business Rules:

The clock starts on the date that BST receives a complete and accurate collocation application accompanied by the appropriate application fee. The clock stops on the date that BST returns a response. The clock will restart upon receipt of changes to the original application request.

Calculation:

 $\label{eq:average} Average\ Response\ Time = \Sigma (Request\ Response\ Date) - (Request\ Submission\ Date) /\ Count\ of\ Responses\ Returned\ within\ Reporting\ Period.$

Report Structure:

- Individual CLEC (alias) aggregate
- Aggregate of all CLECs

Level of Disaggregation:

- State, Region and further geographic disaggregation as required by State Commission Order (e.g. Metropolitan Service Area MSA)
- Virtual
- Physical

Data Retained:

- Report period
- Aggregate data

Retail Analog/Benchmark:

See Appendix D

COLLOCATION

Report/Measurement:

C-2. Average Arrangement Time

Definition:

Measures the average time from the receipt of a complete and accurate Bona Fide firm order (including receipt of appropriate fee) to the date BST completes the collocation arrangement.

Exclusions:

- Any Bona Fide firm order cancelled by the CLEC
- Bona Fide firm orders to augment previously completed arrangements
- Time for BST to obtain permits
- Time during which the collocation contract is being negotiated

Business Rules:

The clock starts on the date that BST receives a complete and accurate Bona Fide firm order accompanied by the appropriate fee. The clock stops upon submission of the permit request and restarts upon receipt of the approved permit. Changes (affecting the provisioning interval or capital expenditures) that are submitted while provisioning is in progress may alter the completion date. The clock stops on the date that BST completes the collocation arrangement.

Calculation:

Average Arrangement Time = Σ (Date Collocation Arrangement is Complete) – (Date Order for Collocation Arrangement Submitted) / Total Number of Collocation Arrangements Completed during Reporting Period.

Report Structure:

- Individual CLEC (alias) aggregate
- Aggregate of all CLECs

Level of Disaggregation:

- State, Region and further geographic disaggregation as required by State Commission Order (e.g. Metropolitan Service Area MSA)
- Virtual
- Physical

Data Retained:

- Report period
- Aggregate data

Retail Analog/Benchmark:

See Appendix D

COLLOCATION

Report/Measurement:

C-3. Percent of Due Dates Missed

Definition:

Measures the percent of missed due dates for collocation arrangements.

Exclusions:

- Any Bona Fide firm order cancelled by the CLEC
- Bona Fide firm orders to augment previously completed arrangements
- Time for BST to obtain permits
- Time during which the collocation contract is being negotiated

Business Rules:

The clock starts on the date that BST receives a complete and accurate Bona Fide firm order accompanied by the appropriate fee. The clock stops on the date that BST completes the collocation arrangement.

Calculation:

% of Due Dates Missed = Σ (Number of Orders not completed w/i ILEC Committed Due Date during Reporting Period) / Number of Orders Completed in Reporting Period) X 100

Report Structure:

- Individual CLEC (alias) aggregate
- Aggregate of all CLECs

Level of Disaggregation:

- State, Region and further geographic disaggregation as required by State Commission Order (e.g. Metropolitan Service Area-MSA)
- Virtual
- Physical

Data Retained:

- Report period
- Aggregate data

Retail Analog/Benchmark:

90% ≤ Commit Date

Appendix A: Reporting Scope*

Standard Service Groupings	Pre-Order, Ordering ➤ Resale Residence ➤ Resale Business ➤ Resale Special ➤ Local Interconnection Trunks ➤ UNE ➤ UNE - Loops w/LNP
	Provisioning > UNE Non-Design > UNE Design > Local Interconnection Trunks > Resale Residence > Resale Business > Resale Design > BST Trunks > BST Residence Retail > BST Business Retail > BST Design Retail
	Maintenance and Repair Local Interconnection Trunks UNE Non-Design UNE Design Resale Residence Resale Business Resale Design BST Interconnection Trunks BST Residence Retail BST Business Retail BST Design Retail Local Interconnection Trunk Group Blockage BST CTTG Trunk Groups CLEC Trunk Groups

Appendix A: Reporting Scope*

Standard Service Order Activities These are the generic BST/CLEC service order activities which are included in the Pre-Ordering, Ordering, and Provisioning sections of this document. It is not meant to indicate specific reporting categories.	 New Service Installations Service Migrations Without Changes Service Migrations With Changes Move and Change Activities Service Disconnects (Unless noted otherwise)
Pre-Ordering Query Types: Maintenance Query Types:	 Address Telephone Number Appointment Scheduling Customer Service Record Feature Availability
Report Levels	 CLEC RESH CLEC MSA CLEC State CLEC Region Aggregate CLEC State Aggregate CLEC Region BST State BST Region

^{*} Scope is report, data source and system dependent, and, therefore, will differ with each report.

Appendix B: Glossary of Acronyms and Terms

A	ACD	Automatic Call Distributor - A service that provides status monitoring of agents in a call center and routes high volume incoming telephone calls to available agents while collecting management information on both callers and attendants.
	AGGREGATE	Sum total of all items in like category, e.g. CLEC aggregate equals the sum total of all CLECs' data for a given reporting level.
	ASR	Access Service Request - A request for access service terminating delivery of carrier traffic into a Local Exchange Carrier's network.
	ATLAS	Application for Telephone Number Load Administration System - The BellSouth Operations System used to administer the pool of available telephone numbers and to reserve selected numbers from the pool for use on pending service requests/service orders.
	ATLASTN	ATLAS software contract for Telephone Number
	AUTO CLARIFICATION	The number of LSRs that were electronically rejected from LESOG and electronically returned to the CLEC for correction.
	DH I DIG	
В	BILLING	The process and functions by which billing data is collected and by which account information is processed in order to render accurate and timely billing.
	BOCRIS	Business Office Customer Record Information System - A front-end presentation manager used by BellSouth organizations to access the CRIS database.
	BRC	Business Repair Center – The BellSouth Business Systems trouble receipt center which serves large business and CLEC customers.
	BST	BellSouth Telecommunications, Inc.
С	CKTID	A unique identifier for elements combined in a service configuration
	CLEC	Competitive Local Exchange Carrier
	CMDS	Centralized Message Distribution System - BellCore administered national system used to transfer specially formatted messages among companies.
	COFFI	Central Office Feature File Interface - A BellSouth Operations System database which maintains Universal Service Order Code (USOC) information based on current tariffs.

Appendix B: Glossary of Acronyms and Terms - Continued

	1						
C	COFIUSOC	COFFI software contract for feature/service information					
	CRIS	Customer Record Information System - The BellSouth proprietary corporate database and billing system for non-access customers and services.					
	CRSACCTS	CRIS software contract for CSR information					
	CSR	Customer Service Record					
	CTTG	Common Transport Trunk Group - Final trunk groups between BST & Independent end offices and the BST access tandems.					
D	DESIGN	Design Service is defined as any Special or Plain Old Telephone Service Order which requires BellSouth Design Engineering Activities					
	DISPOSITION & CAUSE	Types of trouble conditions, e.g. No Trouble Found, Central Office Equipment, Customer Premises Equipment, etc.					
	DLETH	Display Lengthy Trouble History - A history report that gives all activity on a line record for trouble reports in LMOS					
	DLR	Detail Line Record - All the basic information maintained on a line record in LMOS, e.g. name, address, facilities, features etc.					
	DOE	Direct Order Entry System - An internal BellSouth service order entry system used by BellSouth Service Representatives to input business service orders in BellSouth format.					
	DSAP	DOE (Direct Order Entry) Support Application - The BellSouth Operations System which assists a Service Representative or similar carrier agent in negotiating service provisioning commitments for non-designed services and UNEs.					
	DSAPDDI	DSAP software contract for schedule information					
E	E911	Provides callers access to the applicable emergency services bureau by dialing a 3-digit universal telephone number.					
	EDI	Electronic Data Interchange - The computer-to-computer exchange of inter and/or intra company business documents in a public standard format.					
F	FATAL REJECT	The number of LSRs that were electronically rejected from LEO, which checks to see of the LSR has all the required fields correctly populated					
	FLOW- THROUGH	In the context of this document, LSRs submitted electronically via the CLEC mechanized ordering process that flow through to the BST OSS without manual or human intervention.					
	FOC	Firm Order Confirmation - A notification returned to the CLEC confirming that the LSR has been received and accepted, including the specified commitment date.					

Appendix B: Glossary of Acronyms and Terms - Continued

G	1				
H	HAL	"Hands Off" Assignment Logic - Front end access and error resolution logic used in interfacing BellSouth Operations Systems such as ATLAS, BOCRIS, LMOS, PSIMS, RSAG and SOCS.			
	HALCRIS	HAL software contract for CSR information			
I	ISDN	Integrated Services Digital Network			
K					
L	LCSC	Local Carrier Service Center - The BellSouth center which is dedicated to handling CLEC LSRs, ASRs, and Preordering transactions along with associated expedite requests and escalations.			
	LEGACY SYSTEM	Term used to refer to BellSouth Operations Support Systems (see OSS)			
	LENS	Local Exchange Negotiation System - The BellSouth LAN/web server/OS application developed to provide both preordering and ordering electronic interface functions for CLECs.			
	LEO	Local Exchange Ordering - A BellSouth system which accepts the output of EDI, applies edit and formatting checks, and reformats the Local Service Requests in BellSouth Service Order format.			
	LESOG	Local Exchange Service Order Generator - A BellSouth system which accepts the service order output of LEO and enters the Service Order into the Service Order Control System using terminal emulation technology.			
	LMOS	Loop Maintenance Operations System - A BellSouth Operations System that stores the assignment and selected account information for use by downstream OSS and BellSouth personnel during provisioning and maintenance activities.			
	LMOS HOST	LMOS host computer			
	LMOSupd	LMOS updates			
	LNP	Local Number Portability - In the context of this document, the capability for a subscriber to retain his current telephone number as he transfers to a different local service provider.			
	LOOPS	Transmission paths from the central office to the customer premises.			
	LSR	Local Service Request – A request for local resale service or unbundled network elements from a CLEC.			
M	MAINTENANCE & REPAIR	The process and function by which trouble reports are passed to BellSouth and by which the related service problems are resolved.			
	MARCH	A BellSouth Operations System which accepts service orders, interprets the coding contained in the service order image, and constructs the specific switching system Recent Change command messages for input into end office switches.			

Appendix B: Glossary of Acronyms and Terms - Continued

N	NC	"No Circuits" - All circuits busy announcement							
0	OASIS	Obtain Availability Services Information System - A BellSouth front-end processor,							
		which acts as an interface between COFFI and RNS. This system takes the USOCs in							
		COFFI and translates them to English for display in RNS.							
	OASISBSN	OASIS software contract for feature/service							
	OASISCAR	OASIS software contract for feature/service							
	OASISLPC	OASIS software contract for feature/service							
	OASISMTN OASISNET	OASIS software contract for feature/service							
	OASISOCP	OASIS software contract for feature/service OASIS software contract for feature/service							
	OADIDOCI	OASIS SOftware contract for feature/service							
	ORDERING	The process and functions by which resale services or unbundled network elements a ordered from BellSouth as well as the process by which an LSR or ASR is placed with BellSouth.							
	OSPCM	Outside Plant Contract Management System - Provides Scheduling Information.							
	oss	Operations Support System - A support system or database which is used to mechanize the flow or performance of work. The term is used to refer to the overall system consisting of hardware complex, computer operating system(s), and application which is used to provide the support functions.							
	OUT OF SERVICE	Customer has no dial tone and cannot call out.							
P	POTS	Plain Old Telephone Service							
	PREDICTOR	The BellSouth Operations system which is used to administer proactive maintenance and rehabilitation activities on outside plant facilities, provide access to selected work groups (e.g. RRC & BRC) to Mechanized Loop Testing and switching system I/O ports, and provide certain information regarding the attributes and capabilities of outside plant facilities.							
	PREORDERING	The process and functions by which vital information is obtained, verified, or validated prior to placing a service request.							
	PROVISIONING	The process and functions by which necessary work is performed to activate a service requested via an LSR or ASR and to initiate the proper billing and accounting functions.							
	PSIMS	Product/Service Inventory Management System - A BellSouth database Operations System which contains availability information on switching system features and capabilities and on BellSouth service availability. This database is used to verify the availability of a feature or service in an NXX prior to making a commitment to the customer.							
	PSIMSORB	PSIMS software contract for feature/service							

Appendix B: Glossary of Acronyms and Terms – Continued

Q							
R	RNS	Regional Negotiation System - An internal BellSouth service order entry system used by BellSouth Consumer Services to input service orders in BellSouth format.					
	RRC	Residence Repair Center - The BellSouth Consumer Services trouble receipt center which serves residential customers.					
	RSAG	Regional Street Address Guide - The BellSouth database, which contains street addresses validated to be accurate with state and local governments.					
		RSAG software contract for address search					
	RSAGADDR	RSAG software contract for telephone number search					
	RSAGTN						
S	SOCS	Service Order Control System - The BellSouth Operations System which routes service order images among BellSouth drop points and BellSouth Operations Systems during the service provisioning process.					
	SOIR	Service Order Interface Record - any change effecting activity to a customer account by service order that impacts 911/E911.					
T	TAFI	Trouble Analysis Facilitation Interface - The BellSouth Operations System that supports trouble receipt center personnel in taking and handling customer trouble reports.					
	TAG	Telecommunications Access Gateway – TAG was designed to provide an electronic interface, or machine-to-machine interface for the bi-directional flow of information between BellSouth's OSSs and participating CLECs.					
	TN	Telephone Number					
	TOTAL MANUAL FALLOUT	The number of LSRs which are entered electronically but require manual entering into a service order generator.					
U	UNE	Unbundled Network Element					
V							
W	WTN	A unique identifier for elements combined in a service configuration					
X							
Y							
Z							
Σ		Sum of:					

Appendix C

BELLSOUTH'S AUDIT POLICY:

BellSouth currently provides many CLECs with certain audit rights as a part of their individual interconnection agreements. However, it is not reasonable for BellSouth to undergo an audit of the SQM for every CLEC with which it has a contract. BellSouth has developed a proposed Audit Plan for use by the parties to an audit. If requested by a Public Service Commission or by a CLEC exercising contractual audit rights, BellSouth will agree to undergo a comprehensive audit of the aggregate level reports for both BellSouth and the CLEC(s) for each of the next five (5) years (2000 – 2005), to be conducted by an independent third party. The results of that audit will be made available to all the parties subject to proper safeguards to protect proprietary information. This aggregate level audit includes the following specifications:

- 1. The cost shall be borne 50% by BellSouth and 50% by the CLEC or CLECs.
- 2. The independent third party auditor shall be selected with input from BellSouth, the PSC, if applicable, and the CLEC(s).
- 3. BellSouth, the PSC and the CLEC(s) shall jointly determine the scope of the audit.

BellSouth reserves the right to make changes to this audit policy as growth and changes in the industry dictate.

APPENDIX D							
BST SQM	Analogs and Benchmark MEASURES AND SUB-METRICS	RESALE	UNES				
Category	MILACONEO AND COD-MILITAGE	Retail	Retail Analogue	Benchmark*			
outogo.)		Analogue	r totali / ilialoguo	Donomian			
Pre-Ordering	Percent Response Received within "X" seconds		ity w/ retail where applicable.				
	OSS Interface Availability			99.5%			
Ordering	Percent Flow-Through Service Request						
	Residence			90%			
	Business			80%			
	• UNE			80%			
	Percent Rejected Service Request	Diagnosti		Diagnostic.			
		С					
	Reject Interval (Mechanized)	UD	UD	95% within 1 hrs			
	Reject Interval (Non-Mechanized and Partially Mechanized)	UD	UD	85% < 24 hrs			
	Firm Order Confirmation Timeliness (Mechanized)	UD	UD	95% within 4			
	(Non-Mechanized and Partially			hrs			
	Mechanized)			85% <48 Hrs			
	Speed of Answer in Ordering Center	X	X				
Provisioning	Mean Held Order Interval						
	Resale Residence	X					
	Resale Business	Х					
	Resale Design	X					
	Resale PBX	X					
	Resale Centrex	X					
	Resale IDSN	X					
	UNE Loop and Port Combos		Retail Residence and Business				
	UNE 2w Loop with NP – Non-Design		Retail Residence and Business				
	UNE 2w Loop without NP – Non-Design		Retail Residence and Business				
	UNE Loop Other with NP Non-Design		Retail Residence and Business				
	UNE Loop Other without NP Non-Design		Retail Residence and Business				
	UNE Other Non Design		Retail Residence and Business				
	UNE 2w Loop with NP – Design		Retail Residence and Business				
	UNE 2w Loop without NP – Design		Retail Residence and Business				
	UNE Loop Other with NP – Design		Retail Design				

Version 1Q00: 3/6/00 DC01/HEITJ/115887.1

APPENDIX D Analogs and Benchmarks						
BST SQM Category	MEASURES AND SUB-METRICS	RESALE Retail	UNES Retail Analogue	Benchmark*		
	LINE Land Office Pills (AND Decision	Analogue	Datail Daniera			
	UNE Loop Other without NP - Design		Retail Design			
	UNE Other Design Lead laterage restricts Trunks	X	Retail Design			
	Local Interconnection Trunks Average Jeopardy Notice Interval (Mechanized)	^				
	Resale Residence			95% >=24 Hrs		
				95% >=24 Hrs		
	Resale Business Resale Design			95% >=24 Hrs		
	Resale Design Resale PBX			95% >=24 Hrs		
	Resale PBA Resale Centrex			95% >=24 Hrs		
	Resale Centrex Resale IDSN			95% >=24 Hrs		
	UNE Loop and Port Combos			95% >=24 Hrs		
	UNE 2w Loop with NP – Non-Design			95% >=24 Hrs		
	UNE 2w Loop with NP – Non-Design UNE 2w Loop without NP – Non-Design			95% >=24 Hrs		
	UNE Loop Other with NP Non-Design			95% >=24 Hrs		
	UNE Loop Other without NP Non-Design			95% >=24 Hrs		
	UNE Other Non Design			95% >=24 Hrs		
	UNE 2w Loop with NP – Design			95% >=24 Hrs		
	UNE 2w Loop without NP – Design			95% >=24 Hrs		
	UNE Loop Other with NP – Design			95% >=24 Hr		
	UNE Loop Other without NP - Design			95% >=24 Hrs		
	UNE Other Design			95% >=24 Hr		
	Local Interconnection Trunks			95% >=24 Hrs		
	% of Orders given jeopardy notice (Mechanized)					
	Resale Residence	Х				
	Resale Business	Х				
	Resale Design	Х				
	Resale PBX	Х				
	Resale Centrex	Х				
	Resale IDSN	X				
	UNE Loop and Port Combos		Retail Residence and Business			
	UNE 2w Loop with NP – Non-Design		Retail Residence and Business			
	UNE 2w Loop without NP – Non-Design		Retail Residence and Business			
	UNE Loop Other with NP Non-Design		Retail Residence and Business			

	APPENDIX I			
BST SQM	Analogs and Benc MEASURES AND SUB-METRICS	nmarks RESALE	UNES	
Category	MEASURES AND SUB-METRICS	Retail Analogue	Retail Analogue	Benchmark*
	UNE Loop Other without NP Non-Design		Retail Residence and Business	
	UNE Other Non Design		Retail Residence and Business	
	UNE 2w Loop with NP – Design		Retail Residence and Business	
	UNE 2w Loop without NP – Design		Retail Residence and Business	
	UNE Loop Other with NP – Design		Retail Design	
	UNE Loop Other without NP - Design		Retail Design	
	UNE Other Design		Retail Design	
	Local Interconnection Trunks	Х		
	Percent Missed Installation Appointments			
	Resale Residence	X		
	Resale Business	X		
	Resale Design	X		
	Resale PBX	X		
	Resale Centrex	X		
	Resale IDSN	X		
	UNE Loop and Port Combos		Retail Residence and Business	
	UNE 2w Loop with NP – Non-Design		Retail Residence and Business	
	UNE 2w Loop without NP – Non-Design		Retail Residence and Business	
	UNE Loop Other with NP Non-Design		Retail Residence and Business	
	UNE Loop Other without NP Non-Design		Retail Residence and Business	
	UNE Other Non Design		Retail Residence and Business	
	UNE 2w Loop with NP – Design		Retail Residence and Business	
	UNE 2w Loop without NP – Design		Retail Residence and Business	
	UNE Loop Other with NP – Design		Retail Design	
	UNE Loop Other without NP – Design		Retail Design	
	UNE Other Design		Retail Design	
	Local Interconnection Trunks	X		
	Order Completion Interval			
	Resale Residence	X		
	Resale Business	X		
	Resale Design	X		
	Resale PBX	X		
	Resale Centrex	X		

	APPENDIX D			
	Analogs and Bench			
BST SQM	MEASURES AND SUB-METRICS	RESALE	<u>UNES</u>	
Category		Retail	Retail Analogue	Benchmark*
		Analogue		
	Resale IDSN	X		
	UNE Loop and Port Combos		Retail Residence and Business	
	UNE 2w Loop with NP – Non-Design		Retail Residence and Business	
	UNE 2w Loop without NP – Non-Design		Retail Residence and Business	
	UNE Loop Other with NP Non-Design		Retail Residence and Business	
	UNE Loop Other without NP Non-Design		Retail Residence and Business	
	UNE Other Non Design		Retail Residence and Business	
	UNE 2w Loop with NP – Design		Retail Residence and Business	
	UNE 2w Loop without NP – Design		Retail Residence and Business	
	UNE Loop Other with NP – Design		Retail Design	
	UNE Loop Other without NP - Design		Retail Design	
	UNE Other Design		Retail Design	
	Local Interconnection Trunks	X		
	Average Completion Notice Interval – Resale POTS (Mech)			
	Resale Residence	X		
	Resale Business	X		
	Resale Design	X		
	Resale PBX	X		
	Resale Centrex	X		
	Resale IDSN	X		
	UNE Loop and Port Combos		Retail Residence and Business	
	UNE 2w Loop with NP – Non-Design		Retail Residence and Business	
	UNE 2w Loop without NP – Non-Design		Retail Residence and Business	
	UNE Loop Other with NP Non-Design		Retail Residence and Business	
	UNE Loop Other without NP Non-Design		Retail Residence and Business	
	UNE Other Non Design		Retail Residence and Business	
	UNE 2w Loop with NP – Design		Retail Residence and Business	
	UNE 2w Loop without NP – Design		Retail Residence and Business	
	UNE Loop Other with NP – Design		Retail Design	
	UNE Loop Other without NP - Design		Retail Design	
	UNE Other Design		Retail Design	
	Local Interconnection Trunks	X		
	Percent Provisioning Troubles within 30 Days			

	APPENDIX			
DOT COM	Analogs and Benc		LINEO	T
BST SQM	MEASURES AND SUB-METRICS	RESALE Retail	UNES Retail Analogue	Benchmark*
Category		Analogue	Retail Arialogue	Delicilliaik
	Resale Residence	X		
	Resale Business	X		
	Resale Design	X		
	Resale PBX	X		
	Resale Centrex	Х		
	Resale IDSN	Х		
	UNE Loop and Port Combos		Retail Residence and Business	
	UNE 2w Loop with NP – Non-Design		Retail Residence and Business	
	UNE 2w Loop without NP – Non-Design		Retail Residence and Business	
	UNE Loop Other with NP Non-Design		Retail Residence and Business	
	UNE Loop Other without NP Non-Design		Retail Residence and Business	
	UNE Other Non Design		Retail Residence and Business	
	UNE 2w Loop with NP – Design		Retail Residence and Business	
	UNE 2w Loop without NP – Design		Retail Residence and Business	
	UNE Loop Other with NP – Design		Retail Design	
	UNE Loop Other without NP - Design		Retail Design	
	UNE Other Design		Retail Design	
	Local Interconnection Trunks	Х		
	Total Service Order Cycle Time	Diag.	Diagnostic	Diagnostic
Maintenance	Customer Trouble Report Rate			
	Resale Residence	X		
	Resale Business	X		
	Resale Design	X		
	Resale PBX	X		
	Resale Centrex	X		
	Resale IDSN	X		
	UNE Loop and Port Combos		Retail Residence and Business	
	UNE 2w Loop – Non-Design		Retail Residence and Business	
	UNE Loop Other - Non-Design		Retail Residence and Business	
	UNE Other Non Design		Retail Residence and Business	
	UNE 2w Loop – Design		Retail Residence and Business	
	UNE Loop Other – Design		Retail Design	
 I	UNE Other Design		Retail Design	

APPENDIX D **Analogs and Benchmarks BST SQM MEASURES AND SUB-METRICS** RESALE **UNES** Retail Analogue Retail Category Benchmark* Analogue Local Interconnection Trunks Χ **Percent Missed Repair Appointments** Resale Residence Χ X Resale Business X Resale Design X Resale PBX Χ Resale Centrex Resale IDSN Χ **UNE Loop and Port Combos** Retail Residence and Business Retail Residence and Business UNE 2w Loop - Non-Design Retail Residence and Business UNE Loop Other - Non-Design **UNE Other Non Design** Retail Residence and Business UNE 2w Loop - Design Retail Residence and Business UNE Loop Other - Design Retail Design Retail Design **UNE Other Design** • Local Interconnection Trunks Χ **Maintenance Average Duration** Χ Resale Residence **Resale Business** Χ Χ Resale Design Χ Resale PBX Χ Resale Centrex Χ Resale IDSN **UNE Loop and Port Combos** Retail Residence and Business Retail Residence and Business UNE 2w Loop - Non-Design UNE Loop Other - Non-Design Retail Residence and Business **UNE Other Non Design** Retail Residence and Business UNE 2w Loop - Design Retail Residence and Business UNE Loop Other - Design Retail Design **UNE Other Design** Retail Design Χ Local Interconnection Trunks **Percent Repeat Troubles within 30 Days** Resale Residence Χ

	APPENDIX								
Analogs and Benchmarks BST SQM MEASURES AND SUB-METRICS RESALE UNES October 1971 Particular Property of the control of the c									
Category		Retail	Retail Analogue	Benchmar					
3 ,		Analogue							
	Resale Business	X							
	Resale Design	Х							
	Resale PBX	Х							
	Resale Centrex	X							
	Resale IDSN	Х							
	UNE Loop and Port Combos		Retail Residence and Business						
	UNE 2w Loop – Non-Design		Retail Residence and Business						
	UNE Loop Other - Non-Design		Retail Residence and Business						
	UNE Other Non Design		Retail Residence and Business						
	UNE 2w Loop – Design		Retail Residence and Business						
	UNE Loop Other – Design		Retail Design						
	UNE Other Design		Retail Design						
	Local Interconnection Trunks	Х							
	Out of Service > 24hrs								
	Resale Residence	Х							
	Resale Business	Х							
	Resale Design	Х							
	Resale PBX	Х							
	Resale Centrex	Х							
	Resale IDSN	Х							
	UNE Loop and Port Combos		Retail Residence and Business						
	UNE 2w Loop – Non-Design		Retail Residence and Business						
	UNE Loop Other - Non-Design		Retail Residence and Business						
	UNE Other Non Design		Retail Residence and Business						
	UNE 2w Loop – Design		Retail Residence and Business						
	UNE Loop Other – Design		Retail Design						
	UNE Other Design		Retail Design						
	Local Interconnection Trunks	Х	- 1010m = 001g.						
	OSS Interface Availability								
	All systems except ECTA	X							
	ECTA			99.5%					
	OSS Response Interval and %								
	TAFI (Front End)	Х							

	APPENDIX D Analogs and Benchmarks			
BST SQM	MEASURES AND SUB-METRICS	RESALE	UNES	
Category		Retail	Retail Analogue	Benchmark*
.		Analogue	G	
	CRIS, DLETH, DLR, OSPCM, LMOS, LMOSUP, MARCH, Predictor,	PBD		
	SOCS, LNP (Parity by Design)			
	Average Answer Time – Repair Center	Х		
Billing	Invoice Accuracy	X		
	Mean Time To Deliver Invoices	Х		
	Usage Data Delivery Accuracy	Х		
	Usage Data Delivery Timeliness	Х		
	Usage Data Delivery Completeness	Х		
	Mean Time to Deliver Usage	Х		
Operator Services (Toll)	Average Speed to Answer	PBD		
. ,	% Answered in "X" Seconds	PBD		
Directory Assistance	Average Speed to Answer	PBD		
	% Answered in "X" Seconds	PBD		
E911	Timeliness	PBD		
	Accuracy	PBD		
	Mean Interval	PBD		
Trunk Group	Trunk Group Service Report (Percent Trunk Blockage)	X		
Performance	Any 2 hour period in 24 hours where CLEC blockage exceeds BST			
(Blockage)	blockage by more than 0.5% = a miss using trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for BST.			
	Trunk Group Service Report (Percent Trunk Blockage)	Х		
LNP	Average Disconnect Timeliness Interval			
LIVI	Percent Missed Installation Appointments		Retail Residence and Business	
	FOC Mechanized		. teta recordence and Edemoco	95% ≤4 hours
	% Reject Service Request	1	Diagnostic	5575 _ 1 115416
	Average Reject Interval Mechanized		Diagnostio	95% ≤1 hour
	TSOC		Diagnostic	3070 _ 1 11001
	% Flow Through	+	Diagnostio	80%

	APPENDIX Analogs and Bend			
BST SQM Category	MEASURES AND SUB-METRICS	RESALE Retail Analogue	Retail Analogue	Benchmark*
Customer Coordinated	Coordinated Customer Conversions – UNE Loop			95% <u><</u> 15min
Conversions	Coordinated Customer Conversions – LNP			95% <u><</u> 15 min
Collocation +	% of Due Dates Missed			90% < Commi
	Average Response Time		FL PSC is addressing this in generic docket	
+A contract with each CLEC required.	Average Arrangement Time		FL PSC is addressing this in generic docket	

Note 1: PBD = Parity by Design. UD = Under Development – Benchmarks will be replaced when Analogs are complete.

Note2: The retail analog for UNE Non-Design and UNE 2w Loops – Design is the average of Retail Residence Dispatch and Retail Business Dispatch transactions for the particular month. The retail analog for other UNE Design is Retail Design Dispatch.

Note3: Analogs and Benchmarks will be re-evaluated periodically, at least once a year, to validate applicability.

Appendix E Modification of Performance Measurements

In the event that the FCC or any State Commission adopts, orders, or imposes on BellSouth any standards, measurements, or performance requirements in addition to or different from the standards, measurements, and performance requirements contained in this attachment, the Parties shall amend this Attachment to incorporate such standards, measurements, or performance requirements at either Party's request in accordance with Section 35 of the General Terms and Conditions of this Agreement; provided, however, that if e•spire elects to retain the performance measurements set forth in this Attachment rather than to adopt the standards, measurements, or performance measurements so ordered or imposed, BellSouth will continue to provide to

e • spire the performance measurements set forth herein.

EXHIBIT B

VSEEMIII TIER-1 SUBMETRICS

- □ FOC Timeliness (Mechanized only)
- Reject Interval (Mechanized only)
- Order Completion Interval (Dispatch only) Resale POTS
- Order Completion Interval (Dispatch only) Resale Design
- □ Order Completion Interval (No Dispatch only) UNE Loop and Port Combos
- Order Completion Interval ('w' code orders, Dispatch only) UNE Loops
- □ Order Completion Interval (Dispatch only) IC Trunks
- Percent Missed Installation Appointments Resale POTS
- Percent Missed Installation Appointments Resale Design
- Percent Missed Installation Appointments UNE Loop and Port Combos
- Percent Missed Installation Appointments UNE Loops
- Percent Provisioning Troubles within 4 Days Resale POTS
- Percent Provisioning Troubles within 4 Days Resale Design
- Percent Provisioning Troubles within 4 Days UNE Loop and Port Combos
- Percent Provisioning Troubles within 4 Days UNE Loops
- Customer Trouble Report Rate Resale POTS
- Customer Trouble Report Rate Resale Design
- Customer Trouble Report Rate UNE Loop and Port Combos
- Customer Trouble Report Rate UNE Loops
- Percent Missed Repair Appointments Resale POTS
- Percent Missed Repair Appointments Resale Design
- Percent Missed Repair Appointments UNE Loop and Port Combos
- Percent Missed Repair Appointments UNE Loops
- Maintenance Average Duration Resale POTS
- Maintenance Average Duration Resale Design
- Maintenance Average Duration UNE Loop and Port Combos
- Maintenance Average Duration UNE Loops
- Maintenance Average Duration IC Trunks
- Percent Repeat Troubles within 30 Days Resale POTS
- Percent Repeat Troubles within 30 Days Resale Design
- Percent Repeat Troubles within 30 Days UNE Loop and Port Combos
- Percent Repeat Troubles within 30 Days UNE Loops
- Percent Trunk Blockage
- LNP Disconnect Timeliness
- LNP Percent Missed Installation Appointment
- Coordinated Customer Conversions for UNE Loops
- Coordinated Customer Conversions for LNP
- Percent Missed Collocation Due Dates

VSEEMIII TIER-2 SUBMETRICS

- Percent Response Received within "X" seconds Pre-Order OSS
- OSS Interface Availability
- Order Process Percent Flow-Through (Mechanized only)
- Order Completion Interval (Dispatch only) Resale POTS
- Order Completion Interval (Dispatch only) Resale Design
- □ Order Completion Interval (No Dispatch only) UNE Loop and Port Combos
- Order Completion Interval ('w' code orders, Dispatch only) UNE Loops
- □ Order Completion Interval (Dispatch only) IC Trunks
- Percent Missed Installation Appointments Resale POTS
- Percent Missed Installation Appointments Resale Design
- Percent Missed Installation Appointments UNE Loop and Port Combos
- Percent Missed Installation Appointments UNE Loops
- Percent Provisioning Troubles within 4 Days Resale POTS
- Percent Provisioning Troubles within 4 Days Resale Design
- Percent Provisioning Troubles within 4 Days UNE Loop and Port Combos
- Percent Provisioning Troubles within 4 Days UNE Loops
- Customer Trouble Report Rate Resale POTS
- Customer Trouble Report Rate Resale Design
- Customer Trouble Report Rate UNE Loop and Port Combos
- Customer Trouble Report Rate UNE Loops
- Percent Missed Repair Appointments Resale POTS
- Percent Missed Repair Appointments Resale Design
- Percent Missed Repair Appointments UNE Loop and Port Combos
- Percent Missed Repair Appointments UNE Loops
- Maintenance Average Duration Resale POTS
- Maintenance Average Duration Resale Design
- Maintenance Average Duration UNE Loop and Port Combos
- Maintenance Average Duration UNE Loops
- Maintenance Average Duration IC Trunks
- Percent Repeat Troubles within 30 Days Resale POTS
- □ Percent Repeat Troubles within 30 Days Resale Design
- Percent Repeat Troubles within 30 Days UNE Loop and Port Combos
- Percent Repeat Troubles within 30 Days UNE Loops
- Billing Timeliness
- Billing Accuracy
- Usage Data Delivery Timeliness
- Usage Data Delivery Accuracy
- Percent Trunk Blockage
- LNP Disconnect Timeliness
- LNP Percent Missed Installation Appointment
- Coordinated Customer Conversions for UNE Loops
- Coordinated Customer Conversions for LNP
- Percent Missed Collocation Due Dates

VSEEMIII TIER-3 SUBMETRICS

- Percent Missed Installation Appointments Resale POTS
- □ Percent Missed Installation Appointments Resale Design
- Percent Missed Installation Appointments UNE Loop and Port Combos
- Percent Missed Installation Appointments UNE Loops
- □ Percent Missed Repair Appointments Resale POTS
- Percent Missed Repair Appointments Resale Design
- Percent Missed Repair Appointments UNE Loop and Port Combos
- Percent Missed Repair Appointments UNE Loops
- Billing Timeliness
- Billing Accuracy
- Percent Trunk Blockage
- Percent Missed Collocation Due Dates

VSEEM III	MEASURES AND SUB-METRICS	RETAIL ANALOGUE	BENCH
		Resale (x) and UNEs	MARK
Pre-Ordering	Percent Response Received within "X" seconds	Retail Analogue + 4 sec	
	OSS Interface Availability	X	
Ordering	Percent Flow-Through Service Request (Fully Mechanized only)		90%
	Firm Order Confirmation Timeliness (Mechanized only)		95% <u><</u> 4
			hrs
	Reject Interval (Mechanized only)		95% <u><</u> 1
			hrs
Provisioning	Order Completion Interval (Dispatch only) – Resale POTS	Х	
_	Order Completion Interval (Dispatch only) - Resale Design	Х	
	Order Completion Interval (No Dispatch only) – UNE Loop & Port Combos	Retail Residence and Business	
	Order Completion Interval (Dispatch only) – UNE Loops	Design: Retail Design Dispatch 'w' Orders	
		Non-Design: Retail Res, Bus Dispatch 'w' Orders	
	Order Completion Interval (Dispatch only) – IC Trunks	Х	
	Percent Missed Installation Appointments – Resale POTS	Х	
	Percent Missed Installation Appointments – Resale Design	Х	
	Percent Missed Installation Appointments – UNE Loop and Port Combos	Retail Residence and Business	
	Percent Missed Installation Appointments – UNE Loops	Design: Retail Design 1	
		Non-Design: Retail Res, Bus ¹	
	Percent Provisioning Troubles within 4 Days - Resale POTS	Х	
	Percent Provisioning Troubles within 4 Days - Resale Design	Х	
	Percent Provisioning Troubles within 4 Days - UNE Loop and Port	Retail Residence and Business	
	Combos		
	Percent Provisioning Troubles within 4 Days - UNE Loops	Design: Retail Design 1	
		Non-Design: Retail Res, Bus 1	
Maintenance	Customer Trouble Report Rate – Resale POTS	Х	
	Customer Trouble Report Rate – Resale Design	Х	
	Customer Trouble Report Rate - UNE Loop and Port Combos	Retail Residence and Business	
	Customer Trouble Report Rate - UNE Loops	Design: Retail Design 1	
		Non-Design: Retail Res, Bus 1	
	Percent Missed Repair Appointments – Resale POTS	X	
	Percent Missed Repair Appointments - Resale Design	Х	
	Percent Missed Repair Appointments - UNE Loop and Port Combos	Retail Residence and Business	
	Percent Missed Repair Appointments - UNE Loops	Design: Retail Design 1	
		Non-Design: Retail Res, Bus ¹	

NOTES:

¹ The retail analog for UNE Non-Design is the average of all retail residence and retail business transactions for the particular month.

analog for UNE Design is calculated similarly using retail residence, business and design results.

² UD = Under Development

Maintenance			
Continued	Maintenance Average Duration – Resale POTS	X	
	Maintenance Average Duration – Resale Design	X	
	Maintenance Average Duration - UNE Loop and Port Combos	Retail Residence and Business	
	Maintenance Average Duration - UNE Loops	Design: Retail Design ¹ Non-Design: Retail Res, Bus ¹	
	Maintenance Average Duration – IC Trunks	X	
	Percent Repeat Troubles within 30 Days – Resale POTS	X	
	Percent Repeat Troubles within 30 Days – Resale Design	X	
	Percent Repeat Troubles within 30 Days - UNE Loop and Port Combos	Retail Residence and Business	
	Percent Repeat Troubles within 30 Days - UNE Loops	Design: Retail Design ¹ Non-Design: Retail Res, Bus ¹	
Billing	Invoice Accuracy	X	
	Mean Time To Deliver Invoices	X	
	Usage Data Delivery Accuracy	X	
	Usage Data Delivery Timeliness	X	
Trunk Blockage	Trunk Group Service Report (Percent Trunk Blockage)	X	
LNP	Average Disconnect Timeliness Interval		UD ²
	Percent Missed Installation Appointments		UD ²
CC	Coordinated Customer Conversions – UNE Loop		95% <u><</u> 15min
Conversions	Coordinated Customer Conversions – LNP		95% <u><</u> 15 min
Collocation	% of Due Dates Missed		<u><</u> 10%

NOTES:

¹ The retail analog for UNE Non-Design is the average of all retail residence and retail business transactions for the particular month.

analog for UNE Design is calculated similarly using retail residence, business and design results. $^2\,\mathrm{UD} = \mathrm{Under}\,\mathrm{Development}$

EXHIBIT C

Statistical Methods for BellSouth Performance Measure Analysis

I. Necessary Properties for a Test Methodology

The statistical process for testing if competing local exchange carriers (CLECs) customers are being treat equally with BellSouth (BST) customers involves more than just a mathematical formula. Three key elements need to be considered before an appropriate decision process can be developed. These are

- the type of data,
- the type of comparison, and
- the type of performance measure.

Once these elements are determined a test methodology should be developed that complies with the following properties.

- <u>Like-to-Like Comparisons</u>. When possible, data should be compared at appropriate levels, e.g. wire center, time of month, dispatched, residential, new orders. The testing process should:
 - Identify variables that may affect the performance measure.
 - Record these important confounding covariates.
 - Adjust for the observed covariates in order to remove potential biases and to make the CLEC and the ILEC units as comparable as possible.
- Aggregate Level Test Statistic. Each performance measure of interest should be summarized by one
 overall test statistic giving the decision maker a rule that determines whether a statistically significant
 difference exists. The test statistic should have the following properties.
 - The method should provide a single overall index, on a standard scale.
 - If entries in comparison cells are exactly proportional over a covariate, the aggregated index should be very nearly the same as if comparisons on the covariate had not been done.
 - The contribution of each comparison cell should depend on the number of observations in the cell.
 - Cancellation between comparison cells should be limited.
 - The index should be a continuous function of the observations.
- <u>Production Mode Process</u>. The decision system must be developed so that it does not require intermediate manual intervention, i.e. the process must be a "black box."
 - Calculations are well defined for possible eventualities.
 - The decision process is an algorithm that needs no manual intervention.
 - Results should be arrived at in a timely manner.
 - The system must recognize that resources are needed for other performance measure-related processes that also must be run in a timely manner.
 - The system should be auditable, and adjustable over time.
- <u>Balancing</u>. The testing methodology should balance Type I and Type II Error probabilities.
 - P(Type I Error) = P(Type II Error) for well defined null and alternative hypotheses.
 - The formula for a test's balancing critical value should be simple enough to calculate using standard mathematical functions, i.e. one should avoid methods that require computationally intensive techniques.

Little to no information beyond the null hypothesis, the alternative hypothesis, and the number of
observations should be required for calculating the balancing critical value.

In the following sections we describe appropriate testing processes that adhere as much as possible to the testing principles.

Measurement Types

The performance measures that will undergo testing are of three types:

- 1) means
- 2) proportions, and
- 3) rates

While all three have similar characteristics (a proportion is the average of a measure that takes on only the values of 0 or 1), a proportion or rate is derived from count data while a mean is generally an average of interval measurements.

II. Testing Methodology - The Truncated Z

Many covariates are chosen in order to provide deep comparison levels. In each comparison cell, a Z statistic is calculated. The form of the Z statistic may vary depending on the performance measure, but it should be distributed approximately as a standard normal, with mean zero and variance equal to one. Assuming that the test statistic is derived so that it is negative when the performance for the CLEC is worse than for the ILEC, a positive truncation is done – i.e. if the result is negative it is left alone, if the result is positive it is changed to zero. A weighted average of the truncated statistics is calculated where a cell weight depends on the volume of BST and CLEC orders in the cell. The weighted average is re-centered by the theoretical mean of a truncated distribution, and this is divided by the standard error of the weighted average. The standard error is computed assuming a fixed effects model.

Proportion Measures

For performance measures that are calculated as a proportion, in each adjustment cell, the truncated Z and the moments for the truncated Z can be calculated in a direct manner. In adjustment cells where proportions are not close to zero or one, and where the sample sizes are reasonably large, a normal approximation can be used. In this case, the moments for the truncated Z come directly from properties of the standard normal distribution. If the normal approximation is not appropriate, then the Z statistic is calculated from the hypergeometric distribution. In this case, the moments of the truncated Z are calculated exactly using the hypergeometric probabilities.

Rate Measures

The truncated Z methodology for rate measures has the same general structure for calculating the Z in each cell as proportion measures. For a rate measure, there are a fixed number of circuits or units for the CLEC, n_{2j} and a fixed number of units for BST, n_{1j} . Suppose that the performance measure is a "trouble rate." The modeling assumption is that the occurrence of a trouble is independent between units and the number of troubles in n circuits follows a Poisson distribution with mean λ n where λ is the probability of a trouble in 1 circuit and n is the number of circuits.

In an adjustment cell, if the number of CLEC troubles is greater than 15 and the number of BST troubles is greater than 15, then the Z test is calculated using the normal approximation to the Poisson. In this case, the moments of the truncated Z come directly from properties of the standard normal distribution. Otherwise, if there are very few troubles, the number of CLEC troubles can be modeled using a binomial distribution with n equal to the total number of troubles (CLEC plus BST troubles.) In this case, the moments for the truncated Z are calculated explicitly using the binomial distribution.

Mean Measures

For mean measures, an adjusted t statistic is calculated for each like-to-like cell which has at least 7 BST and 7 CLEC transactions. A permutation test is used when one or both of the BST and CLEC sample sizes is less than 6. Both the adjusted t statistic and the permutation calculation are described in the technical appendix.

APPENDIX TECHNICAL DESCRIPTION

We start by assuming that any necessary trimming of the data is complete, and that the data are disaggregated so that comparisons are made within appropriate classes or adjustment cells that define "like" observations.

NOTATION AND EXACT TESTING DISTRIBUTIONS

Below, we have detailed the basic notation for the construction of the truncated z statistic. In what follows the word "cell" should be taken to mean a like-to-like comparison cell that has both one (or more) ILEC observation and one (or more) CLEC observation.

L =the total number of occupied cells

i = 1,...,L; an index for the cells

 n_{1j} = the number of ILEC transactions in cell j

 n_{2j} = the number of CLEC transactions in cell j

 n_j = the total number transactions in cell j; n_{1j} + n_{2j}

 X_{1jk} = individual ILEC transactions in cell j; k = 1,..., n_{1j}

 X_{2jk} = individual CLEC transactions in cell j; k = 1,..., n_{2j}

 Y_{ik} = individual transaction (both ILEC and CLEC) in cell j

$$= \begin{cases} X_{1jk} & k = 1, K, n_{1j} \\ X_{2jk} & k = n_{1j} + 1, K, n_{j} \end{cases}$$

 $\Phi^{-1}(\cdot)$ = the inverse of the cumulative standard normal distribution function

For Mean Performance Measures the following additional notation is needed.

 $\overline{X}_{ij} =$ the ILEC sample mean of cell j

 \overline{X}_{ij} = the CLEC sample mean of cell j

 S_{1i}^2 = the ILEC sample variance in cell j

 S_{2j}^2 = the CLEC sample variance in cell j

 $y_{jk} =$ a random sample of size n_{2j} from the set of Y_{j1} , X_{jn_i} ; $k = 1, ..., n_{2j}$

 M_i = the total number of distinct pairs of samples of size n_{1i} and n_{2i} ;

$$= \begin{pmatrix} n_{j} \\ n_{1j} \end{pmatrix}$$

The exact parity test is the permutation test based on the "modified Z" statistic. For large samples, we can avoid permutation calculations since this statistic will be normal (or Student's t) to a good approximation. For small samples, where we cannot avoid permutation calculations, we have found that the difference between "modified Z" and the textbook "pooled Z" is negligible. We therefore propose to use the permutation test based on pooled Z for small samples. This decision speeds up the permutation computations considerably, because for each permutation we need only compute the sum of the CLEC sample values, and not the pooled statistic itself.

A permutation probability mass function distribution for cell j, based on the "pooled Z" can be written as

$$PM(t) = P(\sum_{k} y_{jk} = t) = \frac{\text{the number of samples that sum to t}}{M_{i}},$$

and the corresponding cumulative permutation distribution is

$$CPM(t) = P(\sum_k y_{jk} \le t) = \frac{\textit{the number of samples with sum } \le t}{M_j} \,.$$

For Proportion Performance Measures the following notation is defined

 a_{ij} the number of ILEC cases possessing an attribute of interest in cell j

a_{2j}= the number of CLEC cases possessing an attribute of interest in cell j

 a_j = the number of cases possessing an attribute of interest in cell j; a_{1j} + a_{2j}

The exact distribution for a parity test is the hypergeometric distribution. The hypergeometric probability mass function distribution for cell j is

$$HG(h) = P(H = h) = \begin{cases} \frac{\binom{n_{1j}}{h}\binom{n_{2j}}{a_j - h}}{\binom{n_j}{a_j}}, \max(0, a_j - n_{2j}) \le h \le \min(a_j, n_{1j}), \\ \binom{n_j}{a_j}, \min(0, a_j - n_{2j}) \le h \le \min(a_j, n_{2j}), \end{cases}$$

and the cumulative hypergeometric distribution is

$$CHG(x) = P(H \le x) = \begin{cases} 0 & x < max(0, a_{j} - n_{1j}) \\ \sum_{h=max(0, a_{j} - n_{1j})}^{x} HG(h), & max(0, a_{j} - n_{1j}) \le x \le min(a_{j}, n_{2j}). \\ 1 & x > min(a_{j}, n_{2j}) \end{cases}$$

For Rate Measures, the notation needed is defined as

 b_{1j} = the number of ILEC base elements in cell j

 b_{2i} = the number of CLEC base elements in cell j

 b_i = the total number of base elements in cell j; $b_{1j} + b_{2j}$

 $\vec{\mathbf{p}}_{l,j}$ = the ILEC sample rate of cell j; n_{lj}/b_{lj}

 \mathbf{r} = the CLEC sample rate of cell j; n_{2j}/b_{2j}

 q_j = the relative proportion of CLEC elements for cell j; b_{2j}/b_j

The exact distribution for a parity test is the binomial distribution. The binomial probability mass function distribution for cell j is

$$BN(k) = P(B = k) = \begin{cases} \binom{n_j}{k} q_j^k (1 - q_j)^{n_j - k}, & 0 \le k \le n_j \\ 0 & \text{otherwise} \end{cases},$$

and the cumulative binomial distribution is

$$CBN(x) = P(B \le x) = \begin{cases} 0 & x < 0 \\ \sum_{k=0}^{x} BN(k), & 0 \le x \le n_{j}. \\ 1 & x > n_{j} \end{cases}$$

CALCULATING THE TRUNCATED Z

The general methodology for calculating an aggregate level test statistic is outlined below.

1. Calculate cell weights, W_j. A weight based on the number of transactions is used so that a cell which has a larger number of transactions has a larger weight. The actual weight formulae will depend on the type of measure.

Mean Measure

$$W_{\rm j} = \sqrt{\frac{n_{\rm 1j}n_{\rm 2j}}{n_{\rm j}}}$$

Proportion Measure

$$\mathbf{W}_{j} = \sqrt{\frac{\mathbf{n}_{2j} \mathbf{n}_{1j}}{\mathbf{n}_{j}} \cdot \frac{\mathbf{a}_{j}}{\mathbf{n}_{j}} \cdot \left(1 - \frac{\mathbf{a}_{j}}{\mathbf{n}_{j}}\right)}$$

Rate Measure

$$W_{j} = \sqrt{\frac{b_{1j}b_{2j}}{b_{j}} \cdot \frac{n_{j}}{b_{j}}}$$

- 2. In each cell, calculate a Z value, Z_i. A Z statistic with mean 0 and variance 1 is needed for each cell.
 - If $W_i = 0$, set $Z_i = 0$.
 - Otherwise, the actual Z statistic calculation depends on the type of performance measure.

Mean Measure

$$Z_i = \Phi^{-1}(\alpha)$$

where α is determine by the following algorithm.

If $min(n_{1j}, n_{2j}) > 6$, then determine α as

$$\alpha = P(t_{n_1,-1} \le T_j),$$

that is, α is the probability that a t random variable with n_{1i} - 1 degrees of freedom, is less than

$$T_{j} = t_{j} + \frac{g}{6} \left(\frac{n_{1j} + 2n_{2j}}{\sqrt{n_{1j} n_{2j}(n_{1j} + n_{2j})}} \right) \left(t^{2} + \frac{n_{2j} - n_{1j}}{2n_{1j} + n_{2j}} \right),$$

where

$$t_{j} = \frac{\overline{X}_{1j} - \overline{X}_{2j}}{s_{1j} \sqrt{\frac{1}{n_{1i}} + \frac{1}{n_{2i}}}}$$

and the coefficient g is an estimate of the skewness of the parent population, which we assume is the same in all cells. It can be estimated from the ILEC values in the largest cells. This needs to be done only once for each measure. We have found that attempting to estimate this skewness parameter for each cell separately leads to excessive variability in the "adjusted" t. We therefore use a single compromise value in all cells.

Note, that t_j is the "modified Z" statistic. The statistic T_j is a "modified Z" corrected for the skewness of the ILEC data.

If $min(n_{1i}, n_{2i}) \leq 6$, and

- a) $M_i \le 1,000$ (the total number of distinct pairs of samples of size n_{1i} and n_{2i} is 1,000 or less).
 - Calculate the sample sum for all possible samples of size n_{2i}.
 - Rank the sample sums from smallest to largest. Ties are dealt by using average ranks.
 - Let R₀ be the rank of the observed sample sum with respect all the sample sums.

$$\alpha = 1 - \frac{R_0 - 0.5}{M_j}$$

b) $M_i > 1,000$

- Draw a random sample of 1,000 sample sums from the permutation distribution.
- Add the observed sample sum to the list. There is a total of 1001 sample sums. Rank the sample sums from smallest to largest. Ties are dealt by using average ranks.
- Let R_0 be the rank of the observed sample sum with respect all the sample sums.

$$\alpha = 1 - \frac{R_0 - 0.5}{1001}$$
.

Proportion Measure

$$Z_{j} = \frac{n_{j} a_{1j} - n_{1j} a_{j}}{\sqrt{\frac{n_{1j} n_{2j} a_{j} (n_{j} - a_{j})}{n_{j} - 1}}}.$$

Rate Measure

$$Z_{j} = \frac{n_{1j} - n_{j} q_{j}}{\sqrt{n_{j} q_{j} (1 - q_{j})}}.$$

3. Obtain a truncated Z value for each cell, Z_j^* . To limit the amount of cancellation that takes place between cell results during aggregation, cells whose results suggest possible favoritism are left alone. Otherwise the cell statistic is set to zero. This means that positive equivalent Z values are set to 0, and negative values are left alone. Mathematically, this is written as

$$Z_i^* = \min(0, Z_i).$$

- 4. Calculate the theoretical mean and variance of the truncated statistic under the null hypothesis of parity, $E(Z_j^*|H_0)$ and $Var(Z_j^*|H_0)$. In order to compensate for the truncation in step 3, an aggregated, weighted sum of the Z_j^* will need to be centered and scaled properly so that the final aggregate statistic follows a standard normal distribution.
 - If $W_j = 0$, then no evidence of favoritism is contained in the cell. The formulae for calculating $E(Z_j^* \mid H_0)$ and $Var(Z_j^* \mid H_0)$ cannot be used. Set both equal to 0.
 - If $\min(n_{1j}, n_{2j}) > 6$ for a mean measure, $\min\left\{a_{1j}\left(1 \frac{a_{1j}}{n_{1j}}\right), a_{2j}\left(1 \frac{a_{2j}}{n_{2j}}\right)\right\} > 9$ for a proportion measure, or $\min\left(n_{1j}, n_{2j}\right) > 15$ and $n_{i}q_{i}(1 q_{i}) > 9$ for a rate measure then

$$E(Z_{j}^{*} | H_{0}) = -\frac{1}{\sqrt{2\pi}}$$
, and

$$Var(Z_j^* | H_0) = \frac{1}{2} - \frac{1}{2\pi}.$$

• Otherwise, determine the total number of values for Z_j^* . Let z_{ji} and θ_{ji} , denote the values of Z_j^* and the probabilities of observing each value, respectively.

$$E(Z_{j}^{*}\,|\,\boldsymbol{H}_{0}) = \sum_{i} \boldsymbol{\theta}_{ji} \boldsymbol{z}_{ji}$$
 ,and

$$Var(Z_{j}^{*} | H_{0}) = \sum_{i} \theta_{ji} Z_{ji}^{2} - \left[E(Z_{j}^{*} | H_{0}) \right]^{2}.$$

The actual values of the z's and θ 's depends on the type of measure, and the sums in the equations are over all possible values of the index i.

Mean Measure

$$\begin{split} N_{j} &= \text{min}(M_{j}, 1,000), \ i = 1, \mathbb{K} \ , N_{j} \\ z_{ji} &= \text{min}\left\{0, 1 - \Phi^{-1}\left(\frac{R_{i} - 0.5}{N_{j}}\right)\right\} \ \text{where } R_{i} \text{ is the rank of sample sum i} \\ \theta_{j} &= \frac{1}{N_{j}} \end{split}$$

Proportion Measure

$$z_{ji} = \min \left\{ 0, \frac{n_{j} i - n_{1j} a_{j}}{\sqrt{\frac{n_{1j} n_{2j} a_{j} (n_{j} - a_{j})}{n_{j} - 1}}} \right\}, \quad i = \min(a_{j}, n_{2j}), K , \max(0, a_{j} - n_{1j})$$

$$\theta_{ii} = HG(i)$$

Rate Measure

$$z_{ji} = \min \left\{ 0, \frac{i - n_j q_j}{\sqrt{n_j q_j (1 - q_j)}} \right\}, \quad i = 0, \mathbb{K}, n_j$$

$$\theta_{ii} = BN(i)$$

5. Calculate the aggregate test statistic, Z^{T} .

$$Z^{T} = \frac{\sum_{j} W_{j} Z_{j}^{*} - \sum_{j} W_{j} E(Z_{j}^{*} | H_{0})}{\sqrt{\sum_{j} W_{j}^{2} Var(Z_{j}^{*} | H_{0})}}$$

The Balancing Critical Value

There are four key elements of the statistical testing process:

- 1. the null hypothesis, H₀, that parity exists between ILEC and CLEC services
- 2. the alternative hypothesis, H_a, that the ILEC is giving better service to its own customers
- 3. the Truncated Z test statistic, Z^{T} , and
- 4. a critical value, c

The decision rule is

• If $Z^T < c$ then accept H_a . • If $Z^T \ge c$ then accept H_0 .

There are two types of error possible when using such a decision rule:

¹ This decision rule assumes that a negative test statistic indicates poor service for the CLEC customer. If the opposite is true, then reverse the decision rule.

Type I Error: Deciding favoritism exists when there is, in fact, no favoritism. **Type II Error**: Deciding parity exists when there is, in fact, favoritism.

The probabilities of each type of each are:

Type I Error: $\alpha = P(Z^T < c \mid H_0)$. Type II Error: $\beta = P(Z^T \ge c \mid H_0)$.

We want a balancing critical value, c_B , so that $\alpha = \beta$.

It can be shown that.

$$c_{B} = \frac{\sum_{j} W_{j} M(m_{j}, se_{j}) - \sum_{j} W_{j} \frac{-1}{\sqrt{2\pi}}}{\sqrt{\sum_{j} W_{j}^{2} V(m_{j}, se_{j})} + \sqrt{\sum_{j} W_{j}^{2} \left(\frac{1}{2} - \frac{1}{2\pi}\right)}}.$$

where

$$\begin{split} M(\mu,\sigma) &= \mu \Phi(\tfrac{-\mu}{\sigma}) - \sigma \phi(\tfrac{-\mu}{\sigma}) \\ V(\mu,\sigma) &= (\mu^2 + \sigma^2) \Phi(\tfrac{-\mu}{\sigma}) - \mu \sigma \phi(\tfrac{-\mu}{\sigma}) - M(\mu,\sigma)^2 \end{split}$$

 $\Phi(\cdot)$ is the cumulative standard normal distribution function, and $\phi(\cdot)$ is the standard normal density function.

This formula assumes that Z_j is approximately normally distributed within cell j. When the cell sample sizes, n_{1j} and n_{2j} , are small this may not be true. It is possible to determine the cell mean and variance under the null hypothesis when the cell sample sizes are small. It is much more difficult to determine these values under the alternative hypothesis. Since the cell weight, W_j will also be small (see calculate weights section above) for a cell with small volume, the cell mean and variance will not contribute much to the weighted sum. Therefore, the above formula provides a reasonable approximation to the balancing critical value.

The values of m_j and se_j will depend on the type of performance measure.

Mean Measure

For mean measures, one is concerned with two parameters in each cell, namely, the mean and variance. A possible lack of parity may be due to a difference in cell means, and/or a difference in cell variances. One possible set of hypotheses that capture this notion, and take into account the assumption that transaction are identically distributed within cells is:

$$\begin{split} &H_{0}\!\!:\mu_{1j}=\mu_{2j},\,{\sigma_{1j}}^{2}={\sigma_{2j}}^{2}\\ &H_{a}\!\!:\mu_{2j}=\mu_{1j}+\delta_{j}\!\!\cdot\!\!\sigma_{1j},\,{\sigma_{2j}}^{2}=\lambda_{j}\!\!\cdot\!\!\sigma_{1j}^{2} & \delta_{j}>0,\,\lambda_{j}\geq1 \text{ and } j=1,\dots,L. \end{split}$$

Under this form of alternative hypothesis, the cell test statistic Z_j has mean and standard error given by

$$m_{j} = \frac{-\delta_{j}}{\sqrt{\frac{1}{n_{1j}} + \frac{1}{n_{2j}}}}$$
, and

$$se_{j} = \sqrt{\frac{\lambda_{j}n_{1j} + n_{2j}}{n_{1j} + n_{2j}}}$$

Proportion Measure

For a proportion measure there is only one parameter of interest in each cell, the proportion of transaction possessing an attribute of interest. A possible lack of parity may be due to a difference in cell proportions. A set of hypotheses that take into account the assumption that transaction are identically distributed within cells while allowing for an analytically tractable solution is:

$$H_0: \frac{p_{2j}(1-p_{1j})}{(1-p_{2j})p_{1j}} = 1$$

$$H_{a:} \ \frac{p_{2j}(1-p_{1j})}{(1-p_{2j})p_{1j}} = \psi_{j} \qquad \qquad \psi_{j} > 1 \ \text{and} \ j = 1,...,L.$$

These hypotheses are based on the "odds ratio." If the transaction attribute of interest is a missed trouble repair, then an interpretation of the alternative hypothesis is that a CLEC trouble repair appointment is ψ_j times more likely to be missed than an ILEC trouble.

Under this form of alternative hypothesis, the within cell asymptotic mean and variance of a_{1j} are given by²

$$E(a_{1j}) = n_j \pi_j^{(1)}$$

$$var(a_{1j}) = \frac{n_j}{\frac{1}{\pi_j^{(1)} + \frac{1}{\pi_j^{(2)}} + \frac{1}{\pi_j^{(3)}} + \frac{1}{\pi_j^{(4)}}}}$$

where

DC01/HEITJ/115887.1

² Stevens, W. L. (1951) Mean and Variance of an entry in a Contingency Table. *Biometrica*, **38**, 468-470. Version 1Q00: 3/6/00

$$\begin{split} &\pi_{\mathbf{j}}^{(1)} = f_{\mathbf{j}}^{(1)} \left(\mathbf{n}_{\mathbf{j}}^{2} + f_{\mathbf{j}}^{(2)} + f_{\mathbf{j}}^{(3)} - f_{\mathbf{j}}^{(4)} \right) \\ &\pi_{\mathbf{j}}^{(2)} = f_{\mathbf{j}}^{(1)} \left(-\mathbf{n}_{\mathbf{j}}^{2} - f_{\mathbf{j}}^{(2)} + f_{\mathbf{j}}^{(3)} + f_{\mathbf{j}}^{(4)} \right) \\ &\pi_{\mathbf{j}}^{(3)} = f_{\mathbf{j}}^{(1)} \left(-\mathbf{n}_{\mathbf{j}}^{2} + f_{\mathbf{j}}^{(2)} - f_{\mathbf{j}}^{(3)} + f_{\mathbf{j}}^{(4)} \right) \\ &\pi_{\mathbf{j}}^{(4)} = f_{\mathbf{j}}^{(1)} \left(\mathbf{n}_{\mathbf{j}}^{2} \left(\frac{2}{\psi_{\mathbf{j}}} - 1 \right) - f_{\mathbf{j}}^{(2)} - f_{\mathbf{j}}^{(3)} - f_{\mathbf{j}}^{(4)} \right) \\ &f_{\mathbf{j}}^{(1)} = \frac{1}{2\mathbf{n}_{\mathbf{j}}^{2} \left(\frac{1}{\psi_{\mathbf{j}}} - 1 \right)} \\ &f_{\mathbf{j}}^{(2)} = \mathbf{n}_{\mathbf{j}} \mathbf{n}_{\mathbf{1j}} \left(\frac{1}{\psi_{\mathbf{j}}} - 1 \right) \\ &f_{\mathbf{j}}^{(3)} = \mathbf{n}_{\mathbf{j}} \mathbf{a}_{\mathbf{j}} \left(\frac{1}{\psi_{\mathbf{j}}} - 1 \right) \\ &f_{\mathbf{j}}^{(4)} = \sqrt{\mathbf{n}_{\mathbf{j}}^{2} \left[4\mathbf{n}_{\mathbf{1j}} \left(\mathbf{n}_{\mathbf{j}} - \mathbf{a}_{\mathbf{j}} \right) \left(\frac{1}{\psi_{\mathbf{j}}} - 1 \right) + \left(\mathbf{n}_{\mathbf{j}} + \left(\mathbf{a}_{\mathbf{j}} - \mathbf{n}_{\mathbf{1j}} \right) \left(\frac{1}{\psi_{\mathbf{j}}} - 1 \right) \right)^{2}} \right] \end{split}$$

Recall that the cell test statistic is given by

$$Z_{j} = \frac{n_{j} a_{1j} - n_{1j} a_{j}}{\sqrt{\frac{n_{1j} n_{2j} a_{j} (n_{j} - a_{j})}{n_{j} - 1}}}.$$

Using the equations above, we see that Z_j has mean and standard error given by

$$m_{j} = \frac{n_{j}^{2} \pi_{j}^{(1)} - n_{1j} a_{j}}{\sqrt{\frac{n_{1j} n_{2j} a_{j} (n_{j} - a_{j})}{n_{j} - 1}}}, \text{ and}$$

$$se_{j} = \sqrt{\frac{n_{j}^{3}(n_{j} - 1)}{n_{1j} n_{2j} a_{j} (n_{j} - a_{j}) \left(\frac{1}{\pi_{j}^{(1)}} + \frac{1}{\pi_{j}^{(2)}} + \frac{1}{\pi_{j}^{(3)}} + \frac{1}{\pi_{j}^{(4)}}\right)}}.$$

Rate Measure

A rate measure also has only one parameter of interest in each cell, the rate at which a phenomenon is observed relative to a base unit, e.g. the number of troubles per available line. A possible lack of parity may be due to a difference in cell rates. A set of hypotheses that take into account the assumption that transaction are identically distributed within cells is:

$$H_0$$
: $r_{1j}=r_{2j}$
$$H_a\text{: }r_{2j}=\epsilon_i r_{1j} \qquad \qquad \epsilon_j>1 \text{ and } j=1,\ldots,L.$$

Given the total number of ILEC and CLEC transactions in a cell, n_j , and the number of base elements, b_{1j} and b_{2j} , the number of ILEC transaction, n_{1j} , has a binomial distribution from n_j trials and a probability of

$$q_j^* = \frac{r_{lj}b_{1j}}{r_{lj}b_{1j} + r_{2j}b_{2j}}.$$

Therefore, the mean and variance of n_{1j} , are given by

$$E(n_{1j}) = n_j q_j^*$$

$$var(n_{1j}) = n_j q_j^* (1 - q_j^*)$$

Under the null hypothesis

$$q_{j}^{*} = q_{j} = \frac{b_{1j}}{b_{i}},$$

but under the alternative hypothesis

$$q_{j}^{*} = q_{j}^{a} = \frac{b_{1j}}{b_{1j} + \varepsilon_{j}b_{2j}}.$$

Recall that the cell test statistic is given by

$$Z_{j} = \frac{n_{1j} - n_{j} q_{j}}{\sqrt{n_{i} q_{i} (1 - q_{i})}}.$$

Using the relationships above, we see that Z_i has mean and standard error given by

$$m_{j} = \frac{n_{j} \left(q_{j}^{a} - q_{j}\right)}{\sqrt{n_{j} q_{j} (1 - q_{j})}} = (1 - \varepsilon_{j}) \sqrt{\frac{n_{j} b_{1 j} b_{2 j}}{b_{1 j} + \varepsilon_{j} b_{2 j}}}, \text{ and }$$

$$se_{j} = \sqrt{\frac{q_{j}^{a}(1 - q_{j}^{a})}{q_{j}(1 - q_{j})}} = \sqrt{\epsilon_{j}} \frac{b_{j}}{b_{1j} + \epsilon_{j}b_{2j}}.$$

Determining the Parameters of the Alternative Hypothesis

In this appendix we have indexed the alternative hypothesis of mean measures by two sets of parameters, λ_j and δ_j . Proportion and rate measures have been indexed by one set of parameters each, ψ_j and ϵ_j respectively. While statistical science can be used to evaluate the impact of different choices of these parameters, there is not much that an appeal to statistical principles can offer in directing specific choices. Specific choices are best left to telephony experts. Still, it is possible to comment on some aspects of these choices:

• Parameter Choices for λ_j . The set of parameters λ_j index alternatives to the null hypothesis that arise because there might be greater unpredictability or variability in the delivery of service to a CLEC customer over that which would be achieved for an otherwise comparable ILEC customer. While concerns about differences in the variability of service are important, it turns out that the truncated Z testing which is being recommended here is relatively insensitive to all but very large values of the λ_j . Put another way, reasonable differences in the values chosen here could make very little difference in the balancing points chosen.

- Parameter Choices for δ_j . The set of parameters δ_j are much more important in the choice of the balancing point than was true for the λ_j . The reason for this is that they directly index differences in average service. The truncated Z test is very sensitive to any such differences; hence, even small disagreements among experts in the choice of the δ_j could be very important. Sample size matters here too. For example, setting all the δ_j to a single value $-\delta_j = \delta$ might be fine for tests across individual CLECs where currently in Louisiana the CLEC customer bases are not too different. Using the same value of δ for the overall state testing does not seem sensible, however, since the state sample would be so much larger.
- Parameter Choices for ψ_j or ε_j . The set of parameters ψ_j or ε_j are also important in the choice of the balancing point for tests of their respective measures. The reason for this is that they directly index increases in the proportion or rate of service performance. The truncated Z test is sensitive to such increases; but not as sensitive as the case of δ_j for mean measures. Sample size matters here as well. As with mean measures, using the same value of ψ or ε for the overall state testing does not seem sensible since the state sample would be so much larger.

The bottom line here is that beyond a few general considerations, like those given above, a principled approach to the choice of the alternative hypotheses to guard against, must come from elsewhere.

DECISION PROCESS

Once Z^T has been calculated, it is compared to the balancing critical value to determine if the ILEC is favoring its own customers over a CLEC's customers.

This critical value changes as the ILEC and CLEC transaction volume change. One way to make this transparent to the decision maker, is to report the difference between the test statistic and the critical value, $diff = Z^T - c_B$. If favoritism is concluded when $Z^T < c_B$, then the diff < 0 indicates favoritism.

This make it very easy to determine favoritism: a positive diff suggests no favoritism, and a negative diff suggests favoritism.

EXHIBIT D

BST VSEEM REMEDY PROCEDURE

TIER-1 CALCULATION FOR RETAIL ANALOGUES:

- 1. Calculate the overall test statistic for each CLEC; z^{T}_{CLEC1} (See Exhibit C)
- 2. Calculate the balancing critical value ($^{\text{C}}_{\text{B}_{\text{CLEC}1}}$) that is associated with the alternative hypothesis (for fixed parameters δ , ψ or ϵ). (See Exhibit C)
- 3. If the overall test statistic is equal to or above the balancing critical value, stop here. Otherwise, go to step 4.
- Calculate the Parity Gap by subtracting the value of step 2. from that of step 1.;
 Z^T_{CI FC1} B_{CLEC1}
- 5. Calculate the Volume Proportion using a linear distribution with slope of ¼. This can be accomplished by taking the absolute value of the Parity Gap from step 4. divided by 4; ABS((z^T_{CLEC1} B_{CLEC1}) / 4). All parity gaps equal or greater to 4 will result in a volume proportion of 100%.
- 6. Calculate the Affected Volume by multiplying the Volume Proportion from step 5. by the Total CLEC₁ Volume in the negatively affected cell; where the cell value is negative. (See Exhibit C)
- 7. Calculate the payment to e.spire by multiplying the result of step 6. by the appropriate dollar amount from the fee schedule.

So, e.spire payment = Affected Volume_{CLEC1} * \$\$ from Fee Schedule

Example: e.spire Missed Installation Appointments (MIA) for Resale POTS

	n _I	n _C	MIA_I	MIA_C	z^{T}_{CLEC1}	C_{B}	Parity Gap	Volume	Affected Volume
State	50000	600	9%	16%	-1.92	-0.21	1.71	Proportion 0.4275	volume
Cell					Z _{CLEC1}				
1		150	0.091	0.112	-1.994				64
2		75	0.176	0.098	0.734				
3		10	0.128	0.333	-2.619				4
4		50	0.158	0.242	-2.878				21
5		15	0.245	0.075	1.345				
6		200	0.156	0.130	0.021				
7		30	0.166	0.233	-0.600				13
8		20	0.106	0.127	-0.065				9
9		40	0.193	0.218	-0.918				17
10		10	0.160	0.235	-0.660				4
								-	133

where $n_{\text{\tiny I}} = ILEC$ observations and $n_{\text{\tiny C}} = e.spire$ observations

Payout for e.spire is (133 units) * (\$100/unit) = \$13,300 TIER-2 CALCULATION for RETAIL ANALOGUES:

- 1. Tier-2 is triggered by three monthly failures of any VSEEM submetric in the same quarter.
- 2. Calculate the overall test statistic for the CLEC Aggregate using all transactions from the calendar quarter; z^T_{CLECA}
- 3. Calculate the balancing critical value ($^{\text{C}}_{\text{B}_{\text{CLEC}1}}$) that is associated with the alternative hypothesis (for fixed parameters δ , ψ or ϵ). (See Exhibit C)
- 4. If the overall test statistic is equal to or above the balancing critical value for the calendar quarter, stop here. Otherwise, go to step 5.
- 5. Calculate the Parity Gap by subtracting the value of step 3. from that of step 2.; z^{T}_{CLECA} B_{CLECA}
- 6. Calculate the Volume Proportion using a linear distribution with slope of ¼. This can be accomplished by dividing the Parity Gap from step 5. by 4; ABS((z^T_{CLECA} B_{CLECA}) / 4). All parity gaps equal or greater to 4 will result in a volume proportion of 100%.
- 7. Calculate the Affected Volume by multiplying the Volume Proportion from step 6. by the Total CLEC_A Volume (CLEC Aggregate) in the negatively affected cell; where the cell value is negative (See Exhibit C).
- 8. Calculate the payment to State Designated Agency by multiplying the result of step 7. by the appropriate dollar amount from the fee schedule.

So, State Designated Agency payment = Affected Volume_{CLECA} * \$\$ from Fee Schedule

Example: CLEC-A Missed Installation Appointments (MIA) for Resale POTS

State	n _I	n _C	MIA_{l}	MIA_C	\mathbf{z}^{T}_{CLECA}	C_B	Parity Gap	Volume Proportion	Affected Volume
Quarter1	180000	2100	9%	16%	-1.92	-0.21	1.71	0.4275	VOIGITIE
Cell					Z _{CLECA}				
1		500	0.091	0.112	-1.994				214
2		300	0.176	0.098	0.734				
3		80	0.128	0.333	-2.619				34
4		205	0.158	0.242	-2.878				88
5		45	0.245	0.075	1.345				
6		605	0.156	0.130	0.021				
7		80	0.166	0.233	-0.600				34
8		40	0.106	0.127	-0.065				17

9	165	0.193	0.218	-0.918	71
10	80	0.160	0.235	-0.660	34
					492

where n_I = ILEC observations and n_C = CLEC-A observations

Payout for CLEC-A is (492 units) * (\$300/unit) = \$147,600

Tier-3

Tier-3 uses the monthly CLEC Aggregate results in a given State. Tier-3 is triggered when five of the twelve Tier-3 sub-metrics experience consecutive failures in a given calendar quarter. The table below displays a situation that would trigger a Tier-3 failure, and one that would not.

		TIER-3 FAILURE X = Miss			NOT A TIER-3 FAILURE X = Miss		
Process	Measures	Jan	Feb	Mar	Jan	Feb	Mar
Percent Missed Installation Appointments	Resale POTS	Х	Х	Х	X		
	Resale Design	X			X	X	Х
	UNE Loop & Port Combo		X				
	UNE Loops	Х	X	Х			
Percent Missed Repair Appointments	Resale POTS	Х	X	Х	X		Х
	Resale Design		Х	X		X	
	UNE Loop & Port Combo					Х	Х
	UNE Loops				X		
Billing	Billing Accuracy	Х	X	Х			
	Billing Timeliness				X	Х	Х
Trunk Blockage	Percent Trunk Blockage	Х	Х	Х			
Collocation	Percent Missed Collocation Due Dates						

Tier-3 is effective immediately after quarter results, and can only be lifted when two of the five failed sub-metrics show compliance for two consecutive months in the following quarter.

All tiers standalone, such that triggering Tier-3 will not cease payout of any Tier-1 or Tier-2 failures.

TIER-1 CALCULATION FOR BENCHMARKS:

- 1. For each CLEC, with five or more observations, calculate monthly performance results for the State.
- 2. CLECs having observations (sample sizes) between 5 and 30 will use Table I below:

TABLE I SMALL SAMPLE SIZE TABLE (95% Confidence)

Sample Size	Equivalent 90% Benchmark	Equivalent 95% Benchmark
5	60.00%	80.00%
6	66.67%	83.33%
7	71.43%	85.71%
8	75.00%	75.00%
9	66.67%	77.78%
10	70.00%	80.00%
11	72.73%	81.82%
12	75.00%	83.33%
13	76.92%	84.62%
14	78.57%	85.71%
15	73.33%	86.67%

Sample Size	Equivalent 90% Benchmark	Equivalent 95% Benchmark		
16	75.00%	87.50%		
17	76.47%	82.35%		
18	77.78%	83.33%		
19	78.95%	84.21%		
20	80.00%	85.00%		
21	76.19%	85.71%		
22	77.27%	86.36%		
23	78.26%	86.96%		
24	79.17%	87.50%		
25	80.00%	88.00%		
26	80.77%	88.46%		
27	81.48%	88.89%		
28	78.57%	89.29%		
29	79.31%	86.21%		
30	80.00%	86.67%		

- 3. If the percentage (or equivalent percentage for small samples) is equal to or below the benchmark standard, stop here. Otherwise, go to step 4.
- 4. Determine the Volume Proportion by taking the difference between the benchmark and the actual performance result.
- 5. Calculate the Affected Volume by multiplying the Volume Proportion from step 4. by the Total CLEC₁ Volume.
- 6. Calculate the payment to e.spire by multiplying the result of step 5. by the appropriate dollar amount from the fee schedule.

So, e.spire payment = Affected Volume_{CLEC1} * \$\$ from Fee Schedule

Example: e.spire Missed Installation Appointments (MIA) for UNE Loops

	n _c	Benchmark	MIA_C	Volume	Affected
				Proportion	Volume
State	600	9%	12%	.03	18

Payout for e.spire is (18 units) * (\$400/unit) = \$7,200

TIER-1 CALCULATION FOR BENCHMARKS (IN THE FORM OF A TARGET):

- For each, with five or more observations, CLEC calculate monthly performance results for the State.
- 2. CLECs having observations (sample sizes) between 5 and 30 will use Table I above.
- 3. Calculate the interval distribution based on the same data set used in step 1.
- 4. If the 'percent within' is equal to or exceeds the benchmark standard, stop here. Otherwise, go to step 5.
- 5. Determine the Volume Proportion by taking the difference between 100% and the actual performance result.
- 6. Calculate the Affected Volume by multiplying the Volume Proportion from step 5. by the Total CLEC₁ Volume.
- 7. Calculate the payment to e.spire by multiplying the result of step 6. by the appropriate dollar amount from the fee schedule.

So, e.spire payment = Affected Volume_{CLEC1} * \$\$ from Fee Schedule

Example: e.spire Reject Timeliness

	n _c	Benchmark	Reject Timeliness _C	Volume	Affected
				Proportion	Volume
State	600	95% within 1 hour	93% within 1 hour	.07	42

Payout for e.spire is (42 units) * (\$100/unit) = \$4,200

TIER-2 CALCULATIONS for BENCHMARKS:

Tier-2 calculations for benchmark measures are the same as the Tier-1 benchmark calculations except the CLEC Aggregate data having failed for three months in a given calendar quarter is being assessed.

EXHIBIT E

Table-1

<u>LIQUIDATED DAMAGES TABLE FOR TIER-1 MEASURES</u>

PER AFFECTED ITEM								
	Month 1	Month 2	Month3	Month4	Month 5	Month 6		
Ordering	\$40	\$50	\$60	\$70	\$80	\$90		
Provisioning	\$100	\$125	\$175	\$250	\$325	\$500		
Provisioning UNE (Coordinated Customer Conversions)	\$400	\$450	\$500	\$550	\$650	\$800		
Maintenance and Repair	\$100	\$125	\$175	\$250	\$325	\$500		
Maintenance and Repair UNE	\$400	\$450	\$500	\$550	\$650	\$800		
LNP	\$150	\$250	\$500	\$600	\$700	\$800		
IC Trunks	\$100	\$125	\$175	\$250	\$325	\$500		
Collocation	\$5,000	\$5,000	\$5,000	\$5,000	\$5,000	\$5,000		

Table-2 VOLUNTARY PAYMENTS FOR TIER-2 MEASURES

	Per Affected Item
OSS	\$20
Pre-Ordering	, -
Ordering	\$60
Provisioning	\$300
UNE Provisioning (Coordinated Customer Conversions)	\$875
Maintenance and Repair	\$300
UNE Maintenance and Repair	\$875
Billing	\$1.00
LNP	\$500
IC Trunks	\$500
Collocation	\$15,000

for

e•spire

BellSouth Standard Interconnection Agreement

Agreement Effective Date:	Agreement Expiration Date:
Account Manager:	Account Manager Tel No:

Attachment Name/Number	Section Number	Version Date	Planned Activities
Terms/Conditions PartA	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		
	13		
	14		
	15		
	16		
	17		
	18		
	19		
	20		
	21		
	22		
	23		
	24		
	25		
	26		
Terms/Conditions Part B			
1-Resale	1		

for

e•spire

BellSouth Standard Interconnection Agreement

Attachment	Section	Version	Planned Activities
Name/Number	Number	Date	
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		
	13		
	Exhibit A		
	Exhibit B		
	Exhibit C		
	Exhibit D		
	Exhibit E		
	Exhibit F		
	Exhibit G		
	Exhibit H		
2-Network Elements &	1		
Other Services			
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		

for

e•spire

BellSouth Standard Interconnection Agreement

Attachment	Section	Version	Planned Activities
Name/Number	Number	Date	
	11		
	12		
	13		
	14		
	15		
	16		
	17		
	Exhibit A		
	Exhibit B		
	Exhibit C		
3-Local Interconnection	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	Exhibit A		
4-Physical Collocation	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		

for

e•spire

BellSouth Standard Interconnection Agreement

Attachment	Section	Version	Planned Activities
Name/Number	Number	Date	
	13		
	14		
	Exhibit A		
	Exhibit B		
5-Access to Numbers &			
Number Portability	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	Exhibit A		
6-Ordering/Provisioning	1		
	2		
	3		
7-Billing & Billing Accuracy Certification			
Accuracy Certification	1		
	2		
	3		
	4		
	5		
	6		
	7		
	Exhibit A		
8-ROW/Conduits/PoleAtt	1		
9-Perf Measurement	Pre-Ordering		
	Ordering		
	Provisioning		
	Maint/Repair		

for

e•spire

BellSouth Standard Interconnection Agreement

Attachment Name/Number	Section Number	Version Date	Planned Activities
	Billing		
	Opr Svcs/DA		
	E911		
	Trunk Grp Perf		
	Collocation		
	Appendix A		
	Appendix B		
	Appendix C		

for e•spire

Agreement Effective Date:	Agreement Expiration Date:	
Account Manager:	Account Manager Tel No:	

Attachment	Section No.	Version	Planned Activities
Name		Date	
Terms/Conditions PartA	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		
	13		
	14		
	15		
	16		
	17		
	18		
	19		
	20		
	21		
	22		
	23		
	24		
	25		
	26		
Terms/Conditions Part B			
1-Resale	1		

for

e•spire

Attachment	Section No.	Version	Planned Activities
Name		Date	
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		
	13		
	Exhibit A		
	Exhibit B		
	Exhibit C		
	Exhibit D		
	Exhibit E		
	Exhibit F		
	Exhibit G		
	Exhibit H		
2-Network Elements & Other Services	1		
Other Bervices	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		

for

e•spire

Attachment	Section No.	Version	Planned Activities
Name		Date	
	11		
	12		
	13		
	14		
	15		
	16		
	17		
	Exhibit A		
	Exhibit B		
	Exhibit C		
3-Local Interconnection	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	Exhibit A		
4-Physical Collocation	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		

for

e•spire

Attachment	Section No.	Version	Planned Activities
Name		Date	
- 133555	13		
	14		
	Exhibit A		
	Exhibit B		
5-Access to Numbers &			
Number Portability	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	Exhibit A		
6-Ordering/Provisioning	1		
	2		
	3		
7-Billing & Billing			
Accuracy Certification	1		
	2		
	3		
	4		
	5		
	6		
	7		
	Exhibit A		
8-ROW/Conduits/PoleAtt	1		
9-Perf Measurement	Pre-Ordering		
	Ordering		
	Provisioning		
	Maint/Repair		

for

e•spire

Attachment	Section No.	Version	Planned Activities
Name		Date	
	Billing		
	Opr Svcs/DA		
	E911		
	Trunk Grp Perf		
	Collocation		
	Appendix A		
	Appendix B		
	Appendix C		

ATTACHMENT 11 BAPCO AGREEMENT

AGREEMENT

In consideration of the mutual promises contained herein, BellSouth Advertising & Publishing Corporation, a Georgia corporation ("BAPCO") and American Communications Services, Inc., a Delaware corporation ("CARRIER") agree as follows:

1. <u>RECITALS</u>. BAPCO is the publisher of alphabetical (or White Pages) and classified (or Yellow Pages) directories for certain communities in the southeastern region of the U.S (the "Directories"). CARRIFR provides, or intends to provide, local exchange telephone service in communities in which BAPCO publishes Directories. BAPCO and CARRIER hereby establish the terms by which BAPCO will include listings of CARRIER subscribers in such Directories and by which BAPCO will provide such Directories to CARRIER subscribers.

2. <u>CARRIER OBLIGATIONS</u>. CARRIER agrees as follows:

- (a) CARRIER shall provide to BAPCO, or its designee, at CARRIER's expense and at no charge, listing information concerning its subscribers (designating any who do not desire published listings), consisting of customer name, address, telephone number and all other information reasonably requested by BAPCO as set forth on Exhibit A in a mutually acceptable format for use by BAPCO or its affiliates in publishing Directories of whatever type and format and for other derivative purposes. Such subscriber listing information shall be provided in the format and on the schedule set forth in said Exhibit, or as otherwise mutually agreed between the parties from time to time.
- (b) CARRIER shall also provide directory delivery information to BAPCO as set forth in Exhibit A for all subscribers.
- (c) CARRIER shall advise BAPCO promptly of any directory-related inquiries, requests or complaints which it may receive from CARRIER subscribers and shall provide reasonable cooperation to BAPCO in response to or resolution of the same.
- (d) CARRIER shall respond promptly regarding corrections or queries raised by BAPCO to process listing changes requested by subscribers.

3. BAPCO OBLIGATIONS. BAPCO agrees as follows:

(a) BAPCO shall include one standard listing for each CARRIER subscriber per hunting group in BAPCO's appropriate local alphabetical Directory as published periodically by BAPCO unless nonlisted or nonpublished status is designated by subscribers. Such listings shall be interfiled with the listings of other local exchange telephone company subscribers and otherwise published in the manner of such other

listings according to BAPCO's generally applicable publishing policies and standards. BAPCO shall provide a process whereby CARRIER is afforded a reasonable opportunity to review and correct its subscribers' alphabetical listings in advance of publication.

- (b) Provided CARRIER establishes appropriate resale arrangements with BellSouth Telecommunications, Inc. ("BellSouth"), BAPCO shall publish additional listings, foreign listings and other alphabetical Directory listings offered by BellSouth, for CARRIER subscribers upon their request consistent with BAPCO's generally applicable policies in BAPCO's alphabetical Directories.
- (c) BAPCO will distribute its regularly published alphabetical and classified Directories to local CARRIER subscribers at no charge to CARRIER or CARRIER's subscribers in accordance with BAPCO's prevailing practices, including delivery following Directory publication and upon establishment of new CARRIER service, if a current Directory for that geographic area has not previously been provided. Such deliveries may include separate advertising materials accompanying the Directories.
- (d) BAPCO will include CARRIER information in the customer guide pages of its alphabetical Directories for communities where CARRIER provides local exchange telephone service at the time of publication in accordance with BAPCO's prevailing standards for the same. CARRIER will provide information requested by BAPCO for such purpose on a timely basis.
- (e) BAPCO shall make available at no charge to CARRIER or its subscribers one listing for CARRIER business customers per hunting group in one appropriate heading in BAPCO's appropriate local classified directory as published periodically by BAPCO. Such listings shall be published according to BAPCO's generally applicable publishing policies and standards.
- (f) BAPCO agrees to solicit, accept and publish directory advertising from business subscribers for CARRIER in communities for which BAPCO publishes classified Directories in the same manner and upon substantially the same terms as it solicits, accepts and publishes advertising from advertisers who are not CARRIER subscribers.
- (g) BAPCO will not provide information obtained from CARRIER concerning its subscribers to other local exchange telephone service providers without CARRIER approval, except as may be required in relation to publishing of Directories or as may be permitted by CARRIER for directory assistance or other purposes.
- (h) BAPCO and CARRIER acknowledge that mutual cooperation will be required to successfully serve the needs of their common customers and therefore agree to use reasonable efforts to provide such cooperation to achieve the highest quality of service for both parties' customers.

PUBLISHING POLICIES. BAPCO shall maintain full authority over its publishing schedules, policies, standards, and practices and over the scope and publishing schedules of its Directories. BAPCO shall periodically provide CARRIER with changes by BAPCO in the same which in BAPCO's judgment affect CARRIER's conduct in BAPCO's publishing of listings for CARRIER's subscribers.

5. LIABILITY AND INDEMNITY.

- BAPCO's liability to CARRIER for any errors or omissions in directories or for any default otherwise arising hereunder shall be limited to One Dollar (\$1) for errors or omissions in any subscriber listing in any directory published by BAPCO.
- Each party agrees to defend, indemnify and hold harmless the other from (b) all damages, claims, suits, losses or expenses, including without limitation costs and attorneys fees, to the extent of such party's relative fault, arising out of or resulting from any error, omission or act of such party hereunder. CARRIER agrees to limit its liability and that of BAPCO by contract with CARRIER's subscribers or by tariff to no more than the cost of service for any errors or omissions in any listings published hereunder for CARRIER subscribers. Each party shall notify in writing the other promptly of any claimed error or omission affecting this paragraph and of any claim or suit arising hereunder or relating to this Agreement and shall provide reasonable and timely cooperation in its resolution of the same. Without waiver of any rights hereunder, the indemnified party may at its expense undertake its own defense in any such claim or suit.
- 6. TERM. This Agreement shall be effective on the date of the last signature hereto for a term of two (2) years and shall relate to Directories published by BAPCO during such period. Thereafter, it shall continue in effect unless terminated by either party upon sixty days prior written notice.
- ASSIGNMENT. This Agreement shall be binding upon any successors or assigns of the parties during its Term.
- RELATIONSHIP OF THE PARTIES. This Agreement does not create any joint 8. venture, partnership or employment relationship between the parties or their employees, and the relationship between the parties shall be that of an independent contractor. There shall be no intended third party beneficiaries to this Agreement.

9. NONDISCLOSURE.

During the term of this Agreement it may be necessary for the parties to (a) provide each other with certain information ("Information") considered to be private or proprietary. The recipient shall protect such Information from distribution, disclosure or dissemination to anyone except its employees or contractors with a need to know such Information in conjunction herewith, except as otherwise authorized in writing. All such Information shall be in writing or other tangible form and clearly marked with a confidential or proprietary legend. Information conveyed orally shall be designated as

proprietary or confidential at the time or such oral conveyance and shall be reduced to writing within forty-five (45) days.

- (b) The parties will not have an obligation to protect any portion of Information which: (1) is made publicly available lawfully by a nonparty to this Agreement; (2) is lawfully obtained from any source other than the providing party; (3) is previously known without an obligation to keep it confidential; (4) is released by the providing party in writing; or (5) commencing two (2) years after the termination date of this Agreement if such Information is not a trade secret under applicable law.
- (c) Each party will make copies of the Information only as necessary for its use under the terms hereof, and each such copy will be marked with the same proprietary notices as appear on the originals. Each party agrees to use the Information solely in support of this Agreement and for no other purpose.
- 10. FORCE MAJEURE. Neither party shall be responsible to the other for any delay or failure to perform hereunder to the extent caused by fire, flood, explosion, war, strike, riot, embargo, governmental requirements, civic or military authority, act of God, or other similar cause beyond its reasonable control. Each party shall use best efforts to notify the other promptly of any such delay or failure and shall provide reasonable cooperation to ameliorate the effects thereof.
- 11. <u>PUBLICITY</u>. Neither party shall disclose the terms of this Agreement nor use the trade names or trademarks of the other without the prior express written consent of the other.

12. REPRESENTATIVES AND NOTICES.

- (a) Each party shall name one or more representatives for contacts between the parties which shall be authorized to act on its behalf. Such representatives may be changed from time to time upon written notice to the other party.
- (b) Notices required by law or under this Agreement shall be given in writing by hand delivery, certified or registered mail, or by facsimile followed by certified or registered mail, addressed to the named representatives of the parties with copies to:

If to BAPCO:

Director-LEC/BST Interface
BellSouth Advertising & Publishing Corporation
Room 270
59 Executive Park South
Atlanta, GA 30329

With Copy to:

Vice President and General Counsel

BellSouth Advertising & Publishing Corporation

Room 430

59 Executive Park South Atlanta, GA 30329

If to CARRIER:

Vice President - Network Services and Administration

141 National Business Parkway

Suite 120

Annapolis Junction, MA 20701

13. MISCELLANEOUS. This Agreement represents the entire Agreement between the parties with respect to the subject matter hereof and supersedes any previous oral or written communications, representations, understandings, or agreements with respect thereto. It may be executed in counterparts, each of which shall be deemed an original. All prior and contemporaneous written or oral agreements, representations, warranties, statements, negotiations, and /or understandings by and between the parties, whether express or implied, are superseded, and there are no representations or warranties, either oral or written, express or implied, not herein contained. This Agreement shall be governed by the laws of the state of Georgia.

IN WITNESS WHEREOF, the parties have executed this Agreement by their duly authorized representatives in one or more counterparts, each of which shall constitute an original, on the dates set forth below.

BELLSOUTH ADVERTISING & PUBLISHING CORPORATION

By: By

Title: Disoby - LEC Saterfec

Date: July W, 1996

AMERICAN COMMUNICATIONS

SERVICES, INC.:

Bu Salve

Title: Uce thousaled - Suc A

Date: 81196

APPROVED AS TO FORM:

RANDALY I CADENHEAD

EXHIBIT A

ACCOUNT INFORMATION SECTION (Items in this section are mandatory)

- I. Main Telephone Number: Main line of telephone service that all other numbers are associated to. (Area Code/NXX/Line Numbers)
- 2. Published Telephone Number: Telephone number to appear in the directory.
- 3. Old Telephone Number: If the number is changing, enter the OLD Telephone Number.
- 4. Type of Directory Service: Bus (Business) or Res (Residence)
- 5. Order Type: N New connect order; D Disconnect service order; C Change of listings; R Directory delivery only.
- 6. <u>Due Date</u>: Date that service is requested.
- 7. Carrier Name: The name of the local exchange carrier and operating company code.
- 8. Carrier Number: Operating Company Number

PRIMARY LISTING INFORMATION SECTION (Items in this section are mandatory)

- Listed Name: The way the listing is to appear in the directory. (maximum 1,000 characters including spaces)
 Caption arrangements should be formatted per guidelines. Non-Pub or Non-List situations should be indicated.
- 10. <u>Listed Address</u>: Current address may include street number street name, city, state, and zip code. (Note: P.O. Box or Route not acceptable). Omitted address shown as (OAD). (maximum 250 characters)
- 11. Service Address: Physical location of the telephone.
- 12. Community Name: The name of the community where the listing appears. (i.e.: the Atlanta Directory may have a Community name of Buckhead).
- 13. Zip code: 5 or 9 character code.
- 14. Yellow Pages Heading: The Yellow Page heading where customer wants his listing to appear. (Valid for Business Primary Listings only).
- 15. <u>Directory Name</u>: Name of the directory where Customer desires listing to appear (including town section if applicable). If consistent with existing central office and directory configuration, listing will be included. If different, a Foreign Listing will be charged. Directory appearance entitled free is based on the central office prefix. Entitlement for appearance in other directories will be at the rate of a Foreign Listing (FL).

BILLING INFORMATION SECTION (Items in this section are requested but optional)

- 16. Billing: Name to appear on bill.
- 17. Billing Address: Street number, street name, city, state, zip.
- 18. Contact Telephone Number: Telephone number to contact regarding billing.
- 19. Responsible Person: Owner's name or partners' names or 2 corporate officers.
- 20. Type of Ownership: Sole owner; Partnership or Corporation-
- 21. Tax ID Number or Social Security Number: If sole owner, must have social security number.

DIRECTORY DELIVERY INFORMATION SECTION (Items in this section are mandatory)

- 22. Name: Personal or business name.
- 23. <u>Delivery Address</u>: Street number, street name, city, state, zip code of where directories are to be delivered.
- 24. Directory (Book ID): Bolt code of the directory.
- 25. Number of books now: for immediate delivery/replacement.
- 26. Number of books annually: 0 3 residence, 0 5 business, then negotiated.

REMARKS SECTION (As required)

27. Remarks: Free flow field used by Carrier for any additional information

PRIVATE/PROPRIETARY

Contains private and/or proprietary information. May not be used or disclosed outside the BellSouth companies except pursuant to a written agreement.

Attachment 12

Bona Fide Request and New Business Requests Process

BONA FIDE REQUEST AND NEW BUSINESS REQUESTS PROCESS

- 1.0 The Parties agree that e.spire is entitled to order any Network Element, Interconnection option, service option or Resale Service required to be made available by the Communications Act of 1934, as modified by the Telecommunications Act of 1996 (the "Act"), FCC requirements or State Commission requirements. e.spire also shall be permitted to request the development of new or revised facilities or service options which are not required by the Act. Procedures applicable to requesting the addition of such facilities or service options are specified in this Attachment 12.
- Bona Fide Requests ("BFR") are to be used when e.spire makes a request of BellSouth to provide a new or modified network element, interconnection option, or other service option pursuant to the Act that was not previously included in the Agreement. New Business Requests ("NBRs") are to be used when e.spire makes a request of BellSouth to provide a new or custom capability or function to meet e.spire's business needs that was not previously included in the Agreement. The BFR/NBR process is intended to facilitate the two-way exchange of information between e.spire and BellSouth, necessary for accurate processing of requests in a consistent and timely fashion.
- A BFR shall be submitted in writing by e.spire and shall specifically identify the required service date, technical requirements, space requirements and/or such specifications that clearly define the request such that BellSouth has sufficient information to analyze and prepare a response. Such a request also shall include a e.spire's designation of the request as being (i) pursuant to the Telecommunications Act of 1996 (i.e. a "BFR") or (ii) pursuant to the needs of the business (i.e. a "NBR"). The request shall be sent to e.spire's Account Executive.
- e.spire may cancel a BFR or NBR at any time. If e.spire cancels the request more than three (3) business days after submitting it, e.spire shall pay BellSouth's reasonable and demonstrable costs of processing and/or implementing the BFR or NBR up to the date of cancellation. If e.spire does not cancel a BFR or NBR, e.spire shall pay BellSouth's reasonable and demonstrable costs of processing and implementing the request.
- Within fifteen (15) business days of its receipt of a BFR or NBR from e.spire, BellSouth shall respond to e.spire by providing a preliminary analysis of such Interconnection, Network Element, or other facility or service option that is the subject of the BFR or NBR. The preliminary analysis shall confirm that BellSouth will either offer access to the

Interconnection, Network Element, or other facility or service option, or provide an explanation of why it is not technically feasible and/or why the request does not qualify as an Interconnection, Network Element, or is not otherwise required to be provided under the Act.

- 6.0 If BellSouth determines that the Interconnection, Network Element, or other facility or service option that is the subject of the BFR is technically feasible, BellSouth shall propose a firm price and a detailed implementation plan within forty (40) business days after receipt of the BFR. BellSouth may, but shall not be required, to provide a firm time and cost proposal for a NBR.
- 7.0 Within thirty (30) business days after its receipt of (i) a refusal of BellSouth to provide a BFR or NBR price quote, or (ii) the BFR or NBR price quote and implementation plan from BellSouth, e.spire must either confirm or cancel its order for such facility or service option. If it believes such quote is not consistent with the requirements of the Act, e.spire may at that time seek FCC or state Commission arbitration of its request, as appropriate. Any such arbitration applicable to Network Elements and/or Interconnection shall be conducted in accordance with standards prescribed in Section 252 of the Act.
- Unless e.spire agrees otherwise, all prices shall be consistent with the pricing principles of the Act, FCC and/or the State Commission.
- 9.0 If either Party to a BFR or NBR believes that the other Party is not requesting, negotiating, or processing the Bona Fide Request in good faith, or disputes a determination, or price or cost quote, such Party may seek FCC or state Commission resolution of the dispute, as appropriate.
- 10.0 Upon agreement to the terms of a BFR or NBR, an amendment to the Agreement may be required.

Amendment to Agreement between BellSouth Telecommunications, Inc. and e.spire Communications, Inc. dated July 25, 2000

Pursuant to this Amendment (the "Amendment"), BellSouth Telecommunications, Inc. ("BellSouth") and e.spire Communications, Inc. ("e.spire"), hereinafter referred to collectively as the "Parties", hereby agree to amend that certain Interconnection Agreement (the "Agreement") between BellSouth and e.spire dated July 25, 2000.

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and e.spire hereby covenant and agree as follows:

- 1. Part C, Schedule of E.Spire Operating Subsidiaries, of the General Terms and Conditions is hereby amended to add "ACSI Local Switched Services, Inc.".
- 2. All of the other provisions of the Agreement, dated July 25, 2000, shall remain in full force and effect.
- 3. The Parties agree that either or both of the Parties is authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

BellSouth Telecommunications, Inc.	e.spire Communications, Inc.
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

AMENDMENT TO THE

AGREEMENT BETWEEN e.spire COMMUNICATIONS, INC. AND BELLSOUTH TELECOMMUNICATIONS, INC. DATED JANUARY 1, 2000

Pursuant to this Agreement, (the "Amendment"), e.spire Communications, Inc., ("e.spire"), a Delaware corporation on behalf of itself and its operating subsidiaries (See Agreement), and BellSouth Telecommunications, Inc. ("BellSouth"), a Georgia corporation, having an office at 675 W. Peachtree Street, Atlanta, Georgia, 30375, on behalf of itself, hereinafter referred to collectively as the "Parties," hereby agree to amend that certain Physical Collocation Master Agreement between the Parties dated January 1, 2000 ("Agreement").

WHEREAS, BellSouth and e.spire entered into the Agreement on January 1, 2000, and;

WHEREAS, the Parties desire to amend the Agreement to include Virtual Collocation language, and to add Virtual Collocation Co-Carrier Cross Connect Rates for the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee.

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. The Parties agree that the Agreement between e.spire and BellSouth is hereby amended to add the following to Section 3:
 - 3.4 <u>Virtual Collocation</u>. BellSouth will provide virtual collocation in accordance with the Terms and Conditions as contained in BellSouth's FCC No 1 Tariff. The rates are set forth in Exhibit A to this Agreement.
- 2. The Parties agree that the Agreement between e.spire and BellSouth is hereby amended to add to Exhibit A the following rates in Exhibit 1 attached hereto for the States of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee.
- 3. All of the other provisions of the Agreement, dated January 1, 2000, shall remain in full force and effect.
- 4. In the event that the rates contained in Exhibit 1 are modified by a final order of the Commission, the Parties mutually agree that e.spire may, on thirty (30) day's written notice, require that the parties adopt the Commission final order rates. In the event that such new rates are not adopted within ninety (90) days after such notice, the Dispute shall be referred to the Dispute Resolution procedures set forth in Section 26 of the Agreement.

	TNESS WHEREOF, the Parties live duly authorized representative			Amendment to be executed by their below.	
BellSouth Telecommunications, Inc. e.spire Communications, Inc.					
By:	Signature On File		By:	Signature On File	
Name:	G.R. Follensbee		Name:	James C. Falvey	
Title:	Senior Director		Title:	Senior Vice President- Regulatory Affairs	
Date:	12/3/01		Date:	December 3, 2001	

5. Either or both of the Parties is authorized to submit this Amendment to the respective state

regulatory authorities for approval subject to Section 252(e) of the Federal

Telecommunications Act of 1996.

BELLSOUTH/e.spire COMMUNICATIONS, INC. RATES – ALABAMA VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$2,848.30
	Cable Fees			
ESPCX	Cable Installation Charge	Per Cable	NA	\$2,750.00
ESPSX	Cable Support Charge	Per Cable	\$13.35	NA
	Cross-Connect Fees			
UEAC2	2-Wire Cross-Connect	Per 2-Wire Loop	\$0.28	\$30.76 (First)/ \$29.40 (Add'l) Disconnect \$12.75 (First)/ \$11.38 (Add'l) Manual Svc Order \$19.99
UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.56	\$66.71 (First)/ \$50.43 (Add'I) Disconnect \$12.82 (First)/ \$11.39 (Add'I) Manual Svc Order \$19.99
CNC2F	2-Fiber Cross-Connect	Per Connection	\$12.10	\$55.46 (First)/ \$39.18 (Add'I) Disconnect \$16.83 (First)/ \$13.27 (Add'I) Manual Svc Order \$19.99
CNC4F	4-Fiber Cross-Connect	Per Connection	\$21.75	\$66.71(First)/ \$50.43 (Add'I) Disconnect \$21.86 (First)/ \$18.31 (Add'I) Manual Svc Order\$19.99
CNC1X	Cross-Connect (BellSouth SPA)	Per DS1-Special	\$7.50	\$155.00 (First)/ \$14.00 (Add'l)
CNDS1	Cross-Connect (BellSouth SWA)	Per DS1	\$7.50	\$155.00 (First)/ \$14.00 (Add'l)
CND3X	Cross-Connect (BellSouth SPA)	Per DS3-Special	\$56.25	\$151.90 (First)/ \$11.83 (Add'I)
CNDS3	Cross-Connect (BellSouth SWA)	Per DS3	\$56.25	\$151.90 (First)/ \$11.83 (Add'l)

	T	T	1	2 01 15
	Co-Carrier Cross-Connect Fees			
PE1DS	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Linear Foot	\$0.0038	
	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Cable		\$535.37
PE1ES	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Linear Foot	\$0.0026	
	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Cable		\$535.37
	Floor Space Fees			
ESPVX	Floor Space	Per Square Foot	\$3.20	NA
ESPAX	Floor Space	Per Ampere	\$3.48	NA
	2 2 2 302 2	, , , , , , , , , , , , , , , , , , ,		
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Maintenance in CO Labor Rate	First ½ Hour and Each ½ 0r Fraction Thereof		
	Basic Time			\$30.64
	additional 1/2			
	Overtime			\$35.77
	Premium Time			\$40.90
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
	Security Escort Expenses		First ½ Hour or Fraction Thereof	Each Additional ½ or Fraction Thereof
SPTBX	Basic Time, Normally Scheduled Work Hours		\$41.00	\$25.00
SPTOX	Overtime. Outside of Normally Scheduled Working Hours on a Scheduled Work Day		\$48.00	\$30.00
SPTPX	Premium Time, Outside of Scheduled Work day		\$55.00	\$35.00

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – FLORIDA VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$4122.00
	Cable Fees			

r				3 of 15
ESPCX	Cable Installation Charge	Per Cable	NA	\$965.00
ESPSX	Cable Support Charge	Per Cable	\$13.35	NA
	Cross-Connect Fees			
UEAC2	2-Wire Cross-Connect	Per 100 Ckts	\$5.02	\$1157.00
UEAC4	4-Wire Cross-Connect	Per 100 Ckts	\$5.02	\$1157.00
CNC2F	2-Fiber Cross-Connect	Per Connection	\$6.71	\$2431.00
CNC4F	4-Fiber Cross-Connect	Per Connection	\$6.71	\$2431.00
CNC1X	Cross-Connect (BellSouth	Per DS1-Special	\$7.50	\$155.00 (First)/
	SPA)			\$14.00 (Add'l)
CNDS1	Cross-Connect (BellSouth	Per DS1	\$7.50	\$155.00 (First)/
	SWA)			\$14.00 (Add'l)
CND3X	Cross-Connect (BellSouth	Per DS3-Special	\$56.25	\$151.90 (First)/
	SPA)			\$11.83 (Add'l)
CNDS3	Cross-Connect (BellSouth	Per DS3	\$56.25	\$151.90 (First)/
	SWA)			\$11.83 (Add'l)
	Co-Carrier Cross-Connect			
	Fees			
PE1DS	Co-Carrier Cross-Connect-	Per Linear Foot	\$0.0041	
	Copper or Coaxial Cable			
	Support Structure			.
	Co-Carrier Cross-Connect-	Per Cable		\$535.54
	Copper or Coaxial Cable			
DE4E0	Support Structure	D 1: E (Ф0.0000	
PE1ES	Co-Carrier Cross-Connect-	Per Linear Foot	\$0.0028	
	Fiber Cable Support Structure	Dor Coble		ΦΕΩΕ Ε <i>Δ</i>
	Co-Carrier Cross-Connect-	Per Cable		\$535.54
	Fiber Cable Support Structure			
	Floor Chang Food			
ESPVX	Floor Space Fees Floor Space	Per Square Foot	\$3.20	NA
ESPAX	Floor Space	Per Square Foot Per Ampere	\$3.48	NA
ESPAX	Floor Space	Pel Ampele	φ3. 4 0	INA
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Maintenance in CO Labor Rate	First ½ Hour and		ψ130.01
CINEX	Waintenance in CO Labor Nate	Each ½ 0r		
		Fraction Thereof		
	Basic Time	Tradion mercor		\$30.64
	Dadio Timo			ψ00.0 -
	additional 1/2			
	Overtime			\$35.77
	Premium Time			\$40.90
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
		- 'F		,
	Security Escort Expenses		First ½ Hour or	Each Additional
			Fraction	½ or Fraction
			Thereof	Thereof

SPTBX	Basic Time, Normally	\$41.00	\$25.00
	Scheduled Work Hours		
SPTOX	Overtime. Outside of Normally	\$48.00	\$30.00
	Scheduled Working Hours on a		
	Scheduled Work Day		
SPTPX	Premium Time, Outside of	\$55.00	\$35.00
	Scheduled Work day		

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – GEORGIA VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$2,848.30
	Cable Fees			
ESPCX	Cable Installation Charge	Per Cable	NA	\$2,750.00
ESPSX	Cable Support Charge	Per Cable	\$13.35	NA
	Cross-Connect Fees			
UEAC2	2-Wire Cross-Connect	Per 2-Wire Loop	\$0.283	\$24.56 (First)/ \$23.56 (Add'I) Disconnect \$9.20 (First)/ \$8.30 (Add'I) Manual Svc Order \$19.99
UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.566	\$24.75 (First)/ \$23.70 (Add'l) Disconnect \$9.03 (First)/ \$8.10 (Add'l) Manual Svc Order \$19.99
CNC2F	2-Fiber Cross-Connect	Per Connection	\$2.88	\$41.72 (First)/ \$30.36 (Add'I) Disconnect \$10.43 (First)/ \$8.36 (Add'I) Manual Svc Order \$2.20
CNC4F	4-Fiber Cross-Connect	Per Connection	\$5.76	\$51.03 (First)/ \$39.67 (Add'l) Disconnect \$13.71 (First)/ \$11.65 (Add'l) Manual Svc Order \$2.20
CNC1X	Cross-Connect (BellSouth SPA)	Per DS1-Special	\$7.50	\$155.00 (First)/ \$14.00 (Add'I)

-			1	5 of 15
CNDS1	Cross-Connect (BellSouth SWA)	Per DS1	\$7.50	\$155.00 (First)/ \$14.00 (Add'l)
CND3X	Cross-Connect (BellSouth SPA)	Per DS3-Special	\$56.25	\$151.90 (First)/ \$11.83 (Add'I)
CNDS3	Cross-Connect (BellSouth SWA)	Per DS3	\$56.25	\$151.90 (First)/ \$11.83 (Add'l)
	Co-Carrier Cross-Connect Fees			
PE1DS	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Linear Foot	\$0.0034	
	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Cable		\$553.43
PE1ES	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Linear Foot	\$0.0023	
	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Cable		\$553.43
	Floor Space Fees			
ESPVX	Floor Space	Per Square Foot	\$3.20	NA
ESPAX	Floor Space	Per Ampere	\$3.48	NA
	Training Expenses Per Trainee	1	·	
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Maintenance in CO Labor Rate	First ½ Hour and Each ½ 0r Fraction Thereof		
	Basic Time			\$30.64
	additional 1/2			
	Overtime			\$35.77
	Premium Time			\$40.90
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
	Security Escort Expenses		First ½ Hour or Fraction Thereof	Each Additional ½ or Fraction Thereof
SPTBX	Basic Time, Normally Scheduled Work Hours		\$41.00	\$25.00
SPTOX	Overtime. Outside of Normally Scheduled Working Hours on a Scheduled Work Day		\$48.00	\$30.00
SPTPX	Premium Time, Outside of Scheduled Work day		\$55.00	\$35.00

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – KENTUCKY VIRTUAL COLLOCATION

1	1		T	6 of 15
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$2,848.30
	Cable Fees			
ESPCX	Cable Installation Charge	Per Cable	NA	\$2,750.00
ESPSX	Cable Support Charge	Per Cable	\$13.35	NA
	Cross-Connect Fees			
UEAC2	2-Wire Cross-Connect	Per 2-Wire Loop	\$0.31	\$54.21 (First)/ \$51.07 (Add'l) Manual Svc Order \$19.99
UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.62	\$54.23 (First)/ \$50.96 (Add'l) Manual Svc Order \$19.99
CNC2F	2-Fiber Cross-Connect	Per Connection	\$15.64	\$41.56 (First)/ \$29.82 (Add'l) Manual Svc Order \$2.20
CNC4F	4-Fiber Cross-Connect	Per Connection	\$28.11	\$50.53 (First)/ \$38.78 (Add'I) Manual Svc Order \$2.20
CNC1X	DS1 Cross-Connect	Per DS1	\$1.50	\$44.07 (First)/ \$31.86 (Add'I) Disconnect \$12.76 (First)/\$11.53 (Add'I)
CND3X	DS3 Cross-Connect	Per DS3	\$56.25	\$151.90 (First)/ \$11.83 (Add'l)
	Co-Carrier Cross-Connect Fees			
PE1DS	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Linear Foot	\$0.0045	
	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Cable		\$535.55
PE1ES	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Linear Foot	\$0.003	
	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Cable		\$535.55
	Floor Space Fees			
ESPVX	Floor Space	Per Square Foot	\$3.20	NA
		•	· ·	
ESPAX	Floor Space	Per Ampere	\$3.48	NA

				7 01 13
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Maintenance in CO Labor Rate	First ½ Hour and		
		Each ½ 0r		
		Fraction Thereof		
	Basic Time			\$30.64
	additional 1/2			
	Overtime			\$35.77
	Premium Time			\$40.90
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
	Security Escort Expenses		First ½ Hour or	Each Additional
			Fraction	½ or Fraction
			Thereof	Thereof
SPTBX	Basic Time, Normally		\$41.00	\$25.00
	Scheduled Work Hours			
SPTOX	Overtime. Outside of Normally		\$48.00	\$30.00
	Scheduled Working Hours on a			
	Scheduled Work Day			
SPTPX	Premium Time, Outside of		\$55.00	\$35.00
	Scheduled Work day			

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – LOUISIANA VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$1770.40
	Cable Fees			
ESPCX	Cable Installation Charge	Per Cable	NA	\$841.54
ESPSX	Cable Support Charge	Per Cable	\$16.02	NA
	Cross-Connect Fees			
UEAC2	2-Wire Cross-Connect	Per 2-Wire Loop	\$0.0296	\$11.94 (First)/
				\$11.46 (Add'l)
UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.0591	\$12.04 (First)/
				\$11.53 (Add'l)
CNC2F	2-Fiber Cross-Connect	Per Connection	\$2.65	\$20.29 (First)/
				\$14.76 (Add'I)
CNC4F	4-Fiber Cross-Connect	Per Connection	\$5.31	\$24.81 (First)/
				\$19.29 (Add'l)
CNC1X	DS1 Cross-Connect	Per DS1	\$1.04	\$21.39 (First)/
				\$15.47 (Add'I)
CND3X	DS3 Cross-Connect	Per DS3	\$13.21	\$20.28 (First)/
				\$14.76 (Add'l)

	1			8 01 13
	Co-Carrier Cross-Connect			
	Fees			
PE1DS	Co-Carrier Cross-Connect-	Per Linear Foot	\$0.0036	
	Copper or Coaxial Cable			
	Support Structure			
	Co-Carrier Cross-Connect-	Per Cable		\$534.79
	Copper or Coaxial Cable			
	Support Structure			
PE1ES	Co-Carrier Cross-Connect-	Per Linear Foot	\$0.0024	
	Fiber Cable Support Structure			
	Co-Carrier Cross-Connect-	Per Cable		\$534.79
	Fiber Cable Support Structure			
	Floor Space Fees			
ESPVX	Floor Space	Per Square Foot	\$3.20	NA
ESPAX	Floor Space	Per Ampere	\$8.32	NA
		·		
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Maintenance in CO Labor Rate	First ½ Hour and		
		Each ½ 0r		
		Fraction Thereof		
	Basic Time		\$27.12	\$10.42
			,	
	additional 1/2			
	Overtime		\$35.42	\$13.45
	Premium Time		\$43.72	\$16.49
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
	·	·		
	Security Escort Expenses		First ½ Hour or	Each Additional
			Fraction	½ or Fraction
			Thereof	Thereof
SPTBX	Basic Time, Normally		\$16.44	\$10.42
	Scheduled Work Hours			
SPTOX	Overtime. Outside of Normally		\$21.41	\$13.45
	Scheduled Working Hours on a			
	Scheduled Work Day			
SPTPX	Premium Time, Outside of		\$26.38	\$16.49
	Scheduled Work day			,
L		l .	L.	l

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – MISSISSIPPI VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$1212.25 Disconnect \$0.51

Discor S22.6				_	9 of 15
ESPCX Cable Installation Charge Per Cable NA \$926.: Discorous \$22.6					
Discor \$22.6 ESPSX Cable Support Charge Per Cable \$15.24 NA Cross-Connect Fees		Cable Fees			
Cross-Connect Fees	ESPCX	Cable Installation Charge	Per Cable	NA	\$926.27 Disconnect \$22.62
Cross-Connect Fees	ESPSX	Cable Support Charge	Per Cable	\$15.24	NA
UEAC2 2-Wire Cross-Connect Per 2-Wire Loop \$0.0268 \$12.37 \$11.8 Discon \$6.04 \$5.45 Manu Order \$12.47 \$11.9 Discon \$6.59 \$5.91 Manu Order \$1.24 \$11.9 Discon \$6.59 \$5.91 Manu Order \$1.24 \$12.37 \$		- control of the grant g		- +	
UEAC2 2-Wire Cross-Connect Per 2-Wire Loop \$0.0268 \$12.37 \$11.8 Discon \$6.04 \$5.45 Manu Order \$12.47 \$11.9 Discon \$6.59 \$5.91 Manu Order \$1.24 \$11.9 Discon \$6.59 \$5.91 Manu Order \$1.24 \$12.37 \$		Cross-Connect Fees			
S11.9d Discort S6.59 S5.91 Manu Order	UEAC2		Per 2-Wire Loop	\$0.0268	\$12.37 (First)/ \$11.87 (Add'I) Disconnect \$6.04 (First)/ \$5.45 (Add'I) Manual Svc Order \$19.99
S15.28 Discort S7.61 \$6.10 Manu Order	UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.0536	\$12.47 (First)/ \$11.94 (Add'I) Disconnect \$6.59 (First)/ \$5.91 (Add'I) Manual Svc Order \$19.99
\$19.97 Discord \$10.00 \$8.50 Manu Order	CNC2F	2-Fiber Cross-Connect	Per Connection	\$2.91	\$21.01 (First)/ \$15.29 (Add'I) Disconnect \$7.61 (First)/ \$6.10 (Add'I) Manual Svc Order \$19.99
\$16.02 Discort \$6.60 \$5.97 CND3X DS3 Cross-Connect Per DS3-Special \$14.49 \$21.07 \$15.29 Discort Discort \$7.61 \$7.61	CNC4F	4-Fiber Cross-Connect	Per Connection	\$5.82	\$25.70 (First)/ \$19.97 (Add'I) Disconnect \$10.01 (First)/ \$8.50 (Add'I) Manual Svc Order \$19.90
\$15.29 Discor \$7.61	CNC1X	DS1 Cross-Connect	Per DS1-Special	\$1.14	\$22.16 (First)/ \$16.02 (Add'I) Disconnect \$6.60 (First)/ \$5.97 (Add'I)
	CND3X	DS3 Cross-Connect	Per DS3-Special	\$14.49	\$21.01 (First)/ \$15.29 (Add'I) Disconnect \$7.61 (First)/ \$6.10 (Add'I)
Co-Carrier Cross-Connect Fees					
PE1DS Co-Carrier Cross-Connect- Per Linear Foot \$0.0037	PF1DS		Per Linear Foot	\$0.0037	

F		T .	1	10 01 13
	Copper or Coaxial Cable			
	Support Structure			
	Co-Carrier Cross-Connect-	Per Cable		\$534.65
	Copper or Coaxial Cable			
	Support Structure			
PE1ES	Co-Carrier Cross-Connect-	Per Linear Foot	\$0.0025	
	Fiber Cable Support Structure			
	Co-Carrier Cross-Connect-	Per Cable		\$534.65
	Fiber Cable Support Structure			
	Floor Space Fees			
ESPVX	Floor Space	Per Square Foot	\$5.74	NA
ESPAX	Floor Space	Per Ampere	\$7.33	NA
LOIAX	1 loor opace	1 of Ampere	ψ1.55	IVA
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Maintenance in CO Labor Rate	First ½ Hour and		
		Each ½ 0r		
		Fraction Thereof		
	Basic Time		\$28.09	\$10.79
	additional 1/2			
	Overtime		\$36.69	\$13.94
	Premium Time		\$45.28	\$17.08
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
	Security Escort Expenses		First ½ Hour or	Each Additional
			Fraction	½ or Fraction
			Thereof	Thereof
SPTBX	Basic Time, Normally		\$17.02	\$10.79
	Scheduled Work Hours			
SPTOX	Overtime. Outside of Normally		\$22.17	\$13.94
	Scheduled Working Hours on a			
	Scheduled Work Day			
SPTPX	Premium Time, Outside of		\$27.32	\$17.08
	Scheduled Work day			

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – NORTH CAROLINA VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$2,848.30
	Cable Fees			
ESPCX	Cable Installation Charge	Per Cable	NA	\$2,750.00
ESPSX	Cable Support Charge	Per Cable	\$13.35	NA

Version 1Q00 6/5/00

	 	Т	1	11 of 15
	Cross-Connect Fees			
UEAC2	2-Wire Cross-Connect	Per 2-Wire Loop	\$0.09	\$41.78 (First)/ \$39.23 (Add'l) Disconnect \$4.75(First)/ \$4.75 (Add'l) Manual Svc Order \$19.99
UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.18	\$41.91 (First)/ \$39.25(Add'I) Disconnect \$4.73 (First)/ \$4.73 (Add'I) Manual Svc Order \$19.99
CNC2F	2-Fiber Cross-Connect	Per Connection	\$15.99	\$67.34 (First)/ \$48.55 (Add'l) Manual Svc Order \$19.99
CNC4F	4-Fiber Cross-Connect	Per Connection	\$28.74	\$82.35(First)/ \$63.56 (Add'I) Manual Svc Order \$19.90
CNC1X	DS1 Cross-Connect	Per DS1	\$0.97	\$71.02(First)/ \$51.08 (Add'I)
CND3X	DS3 Cross-Connect	Per DS3	\$56.25	\$151.90 (First)/ \$11.83 (Add'l)
	Co-Carrier Cross-Connect Fees			
PE1DS	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Linear Foot	\$0.0041	
	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Cable		\$532.72
PE1ES	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Linear Foot	\$0.0028	
	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Cable		\$532.72
	Floor Space Fees		1	
ESPVX	Floor Space	Per Square Foot	\$3.20	NA
ESPAX	Floor Space	Per Ampere	\$3.48	NA NA
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Labor Rate	First ½ Hour and Each ½ 0r Fraction Thereof		V 100101

	Basic Time			\$30.64
	additional 1/2			
	Overtime			\$35.77
	Premium Time			\$40.90
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
	Security Escort Expenses		First ½ Hour or	Each Additional
			Fraction	½ or Fraction
			Thereof	Thereof
SPTBX	Basic Time, Normally		\$41.00	\$25.00
	Scheduled Work Hours			
SPTOX	Overtime. Outside of Normally		\$48.00	\$30.00
	Scheduled Working Hours on a			
	Scheduled Work Day			
SPTPX	Premium Time, Outside of		\$55.00	\$35.00
	Scheduled Work day			

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – SOUTH CAROLINA VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
EAF	Application Fee	Per Location	NA	\$2,848.30
	Cable Fees			
ESPCX	Cable Installation Charge	Per Cable	NA	\$2,750.00
ESPSX	Cable Support Charge	Per Cable	\$13.35	NA
	Cross-Connect Fees			
UEAC2	2-Wire Cross-Connect	Per 2-Wire Loop	\$0.3648	\$41.50 (First)/
ULACZ	2-Wile Closs-Collifect	rei 2-vviie Loop	ψ0.3040	\$38.94 (Add'l)
				Manual Svc
				Order \$19.99
UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.7297	\$41.56 (First)/
02,101	Tring Grade Comment		ψο Ξο.	\$38.90 (Add'l)
				Manual Svc
				Order \$19.99
CNC2F	2-Fiber Cross-Connect	Per Connection	\$15.06	\$69.28 (First)/
				\$48.89 (Add'l)
				Manual Svc
				Order \$2.20
CNC4F	4-Fiber Cross-Connect	Per Connection	\$27.08	\$84.07 (First)/
				\$63.68 (Add'I)
				Manual Svc
				Order \$2.20

Version 1Q00 6/5/00

			-	13 of 15
CNC1X	Cross-Connect (BellSouth SPA)	Per DS1-Special	\$7.50	\$155.00 (First)/ \$14.00 (Add'l)
CNDS1	Cross-Connect (BellSouth SWA)	Per DS1	\$7.50	\$155.00 (First)/ \$14.00 (Add'l)
CND3X	Cross-Connect (BellSouth SPA)	Per DS3-Special	\$56.25	\$151.90 (First)/ \$11.83 (Add'l)
CNDS3	Cross-Connect (BellSouth SWA)	Per DS3	\$56.25	\$151.90 (First)/ \$11.83 (Add'l)
	Co-Carrier Cross-Connect			
	Fees			
PE1DS	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Linear Foot	\$0.0033	
	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Cable		\$536.56
PE1ES	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Linear Foot	\$0.0022	
	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Cable		\$536.56
	Floor Space Fees			
ESPVX	Floor Space	Per Square Foot	\$3.20	NA
ESPAX	Floor Space	Per Ampere	\$3.48	NA
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Labor Rate	First ½ Hour and Each ½ 0r Fraction Thereof		
	Basic Time			\$30.64
	additional 1/2			
	Overtime			\$35.77
	Premium Time			\$40.90
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
	Security Escort Expenses		First ½ Hour or Fraction Thereof	Each Additional ½ or Fraction Thereof
SPTBX	Basic Time, Normally Scheduled Work Hours		\$41.00	\$25.00
SPTOX	Overtime. Outside of Normally Scheduled Working Hours on a Scheduled Work Day		\$48.00	\$30.00
SPTPX	Premium Time, Outside of Scheduled Work day		\$55.00	\$35.00

BELLSOUTH//e.spire COMMUNICATIONS, INC. RATES – TENNESSEE VIRTUAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)				
EAF	Application Fee	Per Location	NA	\$2,848.30				
	Cable Fees							
ESPCX	Cable Installation Charge	Per Cable	NA	\$2,750.00				
ESPSX	Cable Support Charge	Per Cable	\$13.35	NA				
	Cross-Connect Fees							
UEAC2	2-Wire Cross-Connect	Per 2-Wire Loop	\$0.57	\$11.62 (First)/ \$9.90 (Add'I) Disconnect \$10.38 (First)/ \$8.66 (Add'I) Manual Svc Order \$19.99				
UEAC4	4-Wire Cross-Connect	Per 4-Wire Loop	\$0.57	\$11.81 (First)/ \$10.04 (Add'I) Disconnect \$10.44 (First)/ \$8.67 (Add'I) Manual Svc Order \$19.99				
CNC2F	2-Fiber Cross-Connect	Per Connection	\$15.64	\$41.56 (First)/ \$29.82 (Add'l)				
CNC4F	4-Fiber Cross-Connect	Per Connection	\$28.11	\$50.53 (First)/ \$38.78 (Add'l)				
CNC1X	DS1 Cross-Connect	Per DS1-Special	\$1.319	\$32.22(First)/ \$17.76 (Add'I) Disconnect \$10.46 (First)/ \$8.75 (Add'I)				
CND3X	DS3 Cross-Connect	Per DS3-Special	\$56.25	\$151.90 (First)/ \$11.83 (Add'l)				
	Co-Carrier Cross-Connect Fees							
PE1DS	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Linear Foot	\$0.0045					
	Co-Carrier Cross-Connect- Copper or Coaxial Cable Support Structure	Per Cable		\$555.03				
PE1ES	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Linear Foot	\$0.0031					
	Co-Carrier Cross-Connect- Fiber Cable Support Structure	Per Cable		\$555.03				

Version 1Q00 6/5/00

				15 01 15
	Floor Space Fees			
ESPVX	Floor Space	Per Square Foot	\$3.20	NA
ESPAX	Floor Space	Per Ampere	\$3.48	NA
	Training Expenses Per Trainee			
CTRLD	Living Expenses	Per Day		\$136.67
CTRLX	Labor Rate	First ½ Hour and		
		Each ½ 0r		
ļ		Fraction Thereof		
	Basic Time			\$30.64
	additional 1/2			
	Overtime			\$35.77
	Premium Time			\$40.90
CTRTA	Air Fare/Travel Expense	Per Trip		\$555.00
ı	Security Escort Expenses		First ½ Hour or	Each Additional
			Fraction	½ or Fraction
			Thereof	Thereof
SPTBX	Basic Time, Normally		\$41.00	\$25.00
	Scheduled Work Hours			
SPTOX	Overtime. Outside of Normally		\$48.00	\$30.00
	Scheduled Working Hours on a			
	Scheduled Work Day		1	
SPTPX	Premium Time, Outside of		\$55.00	\$35.00
	Scheduled Work day			

AMENDMENT TO THE

AGREEMENT BETWEEN E.SPIRE COMMUNICATIONS, INC. AND BELLSOUTH TELECOMMUNICATIONS, INC. DATED JANUARY 1, 2000

Pursuant to this Agreement (the "Amendment"), e.spire Communications, Inc. ("e.spire") and BellSouth Telecommunications, Inc. ("BellSouth"), hereinafter referred to collectively as the "Parties", hereby agree to amend that certain Interconnection Agreement between the Parties dated January 1, 2000 ("Agreement").

WHEREAS, BellSouth and e.spire entered into an Interconnection Agreement on January 1, 2000, and;

NOW THEREFORE, in consideration of the mutual promises and covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

1. The General Terms and Conditions – Part A, Section 53.1, is hereby amended to change the e-spire contact information as follows:

e.spire Communications, Inc.

Mr. James C. Falvey Senior Vice President-Regulatory Affairs e.spire Communications, Inc. 7125 Columbia Gateway Drive, Suite 200 Columbia, MD 21046

- 2. All of the other provisions of the Agreement, dated January 1, 2000, shall remain in full force and effect.
- 3. Either or both of the Parties are authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

e.spsire Communications, Inc.	Bensouth Telecommunications, Inc.
By: Original on File	By: Original on File
Name: <u>James C. Falvey</u>	Name: C. W. Boltz
Title: SVP – Regulatory Affairs	Title: Managing Director
Date: <u>04/22/2002</u>	Date: 05/01/2002

AMENDMENT TO THE

AGREEMENT BETWEEN XSPEDIUS MANAGEMENT CO., LLC AND BELLSOUTH TELECOMMUNICATIONS, INC. DATED JANUARY 1, 2000

Pursuant to this Amendment, (the "Amendment"), Xspedius Management Co., LLC, on behalf of its operating affiliates ("XMC") as set forth in Attachment A, and BellSouth Telecommunications, Inc. ("BellSouth"), hereinafter referred to collectively as the "Parties", hereby agree to amend that certain Interconnection Agreement between the Parties dated January 1, 2000 ("Agreement").

WHEREAS, BellSouth and e.spire Communications, Inc. ("e.spire") entered into the Agreement on January 1, 2000;

WHEREAS, on March 22, 2001 e.spire commended cases under chapter 11 of title 11 of the United States Code; and

WHEREAS, pursuant to bankruptcy court order, e.spire assumed the Agreement and then assigned the Agreement to XMC;

WHEREAS, since January 1, 2000, BellSouth has engaged in cost dockets before the state commissions in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina, and Tennessee ("Cost Dockets");

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. The Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee rates contained in Exhibit A of Attachment 1 of the Agreement are hereby deleted in entirety and replaced with the rates in Exhibit A of this Amendment, attached hereto and incorporated herein by this reference.
- 2. The Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee rates contained in Exhibit H of Attachment 1 of the Agreement are hereby deleted in entirety and replaced with the rates in Exhibit B of this Amendment, attached hereto and incorporated herein by this reference.
- 3. The Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee rates contained in Exhibit C of Attachment 2 of the Agreement are hereby deleted in entirety and replaced with the rates in Exhibit C of this Amendment, attached hereto and incorporated herein by this reference

- 4. The Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee rates contained in Exhibit A of Attachment 3 of the Agreement are hereby deleted in entirety and replaced with the rates in Exhibit D of this Amendment, attached hereto and incorporated herein by this reference.
- 5. The Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee rates contained in Exhibit A of Attachment 5 of the Agreement are hereby deleted in entirety and replaced with the rates in Exhibit E of this Amendment, attached hereto and incorporated herein by this reference.
- 6. The Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee rates contained in Exhibit A of Attachment 7 of the Agreement are hereby deleted in entirety and replaced with the rates in Exhibit F of this Amendment, attached hereto and incorporated herein by this reference.
- 7. Attachment 2, Section 17.7, Port/Loop Combinations, is hereby deleted in its entirety and replaced with new Section 17.7, as set forth in Exhibit G attached hereto and incorporated herein by this reference.
- 8. The Parties agree to add to the Agreement a new Section 17.8 of Attachment 2, Ordinarily Combined UNE Combinations, as set forth in Exhibit G attached hereto and incorporated herein by this reference.

All of the other provisions of the Agreement, dated January 1, 2000, shall remain in full force and effect.

Either or both of the Parties are authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.

This Amendment shall be deemed effective 5 calendar days following the date of the last signature of both Parties and shall apply to the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, South Carolina and Tennessee.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives.

Xspedius Management Co., LLC, on behalf of its operating affiliates BellSouth Telecommunications, Inc. By: __Original On File By: __Original On File Name: __James C. Falvey Name: Elizabeth R.A. Shiroishi Title: Senior VP, Regulatory Affairs Title: __Assistant Director

Date: __10-25-02_____

Date: __10-24-02_____

ATTACHMENT A

Xspedius Management Co., LLC Operating Affiliates

Xspedius Management Co. Switched Services, LLC (AL, FL, GA, KY, LA, MS, SC and TN) Xspedius Management Co. of Charleston, LLC (SC) Xspedius Management Co. of Columbia, LLC (SC) Xspedius Management Co. of Greenville, LLC (SC) Xspedius Management Co. of Spartanburg, LLC (SC) Xspedius Management Co. of Chattanooga, LLC (TN) Xspedius Management Co. of Atlanta, LLC (GA) Xspedius Management Co. of Louisville, LLC (KY) Xspedius Management Co. of Baton Rouge, LLC (LA) Xspedius Management Co. of Louisiana, LLC (LA) Xspedius Management Co. of Shreveport, LLC (LA) Xspedius Management Co. of Jackson, LLC (MS) Xspedius Management Co. of Birmingham, LLC (AL) Xspedius Management Co. of Mobile, LLC (AL) Xspedius Management Co. of Montgomery, LLC (AL) Xspedius Management Co. of Jacksonville, LLC (FL)

RESALE DISCOUNTS AND RATES

		ALABAMA	FLORIDA	GEORGIA	KENTUCKY	LOUISIANA	MISSISSIPPI	SOUTH CAROLINA	TENNESSEE
APPLICABL	E DISCOU	NTS							
RESIDENCE	3	16.3%	21.83%	20.3%	16.79%	20.72%	15.75%	14.8%	16%
BUSINESS		16.3%	16.81%	17.3%	15.54%	20.72%	15.75%	14.8%	16%
CSAs*						9.05%		8.98%	
* Unless noted in	n this row, the	discount for Busi	ness will be the applica	able discount rate f	for CSAs.				
OPERATIO	NAL SUPPO	ORT SYSTEM	MS (OSS) RATES						
<u>ELEMENT</u>	<u>USOC</u>								
Electronic LSR	SOMEC	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
Manual LSR	SOMAN	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99

ODLIE/ADLIE	F/CMDS - Alabama												Attack manuf.	4	Exhibit: B	
ODUF/ADUF	-/CWD3 - Alabama				1	1							Attachment:			
													Incremental		Incremental	
							Submitted	Submitted		Charge -	Charge -	Charge -				
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													131	Addi	Diac 1at	Disc Add I
						Rec	Nonre	curring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C	MDS/EODUF															
ENHAN	NCED OPTIONAL DAILY USAGE FILE (EODUF)															
	EODUF: Message Processing, per message					0.22206700										
ACCES	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.00185100										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00011300										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.00001100										
	ODUF: Message Processing, per message				N/A	0.00249900										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.76000000										
	• • • • • • • • • • • • • • • • • • • •															
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00009400										
CENTE	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.00400000										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00100000					İ					1
Notes:	If no rate is identified in the contract, the rate for the specific	service	or fun	ction will be as set	forth in appli	icable BellSout	h tariff.		1							

10/04/02 Page 1 of 8

ODUF/ADUF	F/CMDS - Florida												Attachment:	1	Exhibit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted		Charge -	Charge -	Charge -
		lust a ut														Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RA	TES(\$)			per LSR		Order vs.	Order vs.	Order vs.	Order vs.
		m									po. 20.1	po. zo.	Electronic-	Electronic-		Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															Disc 1st	Disc Add I
						Rec	Nonre	curring	Nonrecurring	Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C																
ENHA	NCED OPTIONAL DAILY USAGE FILE (EODUF)															
	EODUF: Message Processing, per message					0.080698										
ACCES	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.001656										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012450										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000071										
	ODUF: Message Processing, per message				N/A	0.002146										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.91										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010375										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
											İ					
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 2 of 8

ODUF/ADUF	F/CMDS - Georgia												Attachment:	1	Exhibit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted		Charge -	Charge -	Charge -
											Elec			Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	por Lore		Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															DISC 1St	DISC Add I
						Rec		curring	Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C	MDS/EQDUE															
			-													
ENHA	NCED OPTIONAL DAILY USAGE FILE (EODUF)		-			0.0034555										
4005	EODUF: Message Processing, per message		-			0.0034555										
ACCE	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.0079506										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0000434										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000090										
	ODUF: Message Processing, per message				N/A	0.0046462										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	28.85										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0000434										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004	•									
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001				·				·		
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fur	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 3 of 8

ODUF/ADUF	F/CMDS - Kentucky												Attachment:	1	Exhibit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted		Charge -	Charge -	Charge -
											Elec		Manual Svc	Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	por Lore	Electronic-	Electronic-		Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													151	Add I	DISCISE	DISC Add I
						Rec		curring	Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C	MDS/EQDUE															
	NCED OPTIONAL DAILY USAGE FILE (EODUF)		<u> </u>													
ENHAI	EODUF: Message Processing, per message		<u> </u>			0.235889										
4000			<u> </u>			0.233669										
ACCES	SS DAILY USAGE FILE (ADUF)				N1/A	0.004057										
	ADUF: Message Processing, per message				N/A	0.001857										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0001245										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000136										
	ODUF: Message Processing, per message				N/A	0.002506										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.90										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010372										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)						•					,				
	CMDS: Message Processing, per message				N/A	0.004	•									
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific	service	or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 4 of 8

ODUF/ADUF	F/CMDS - Louisiana												Attachment:	1	Exhibit: B	
		I										Submitted		Charge -	Incremental Charge - Manual Svc	Incremental Charge - Manual Syc
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'I	Order vs.	Order vs. Electronic- Disc Add'l
							Nonre	curring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C																
ENHAI	NCED OPTIONAL DAILY USAGE FILE (EODUF)					0.229779										
ACCE	EODUF: Message Processing, per message					0.229779										
ACCE	SS DAILY USAGE FILE (ADUF)		-		N/A	0.001825										
-	ADUF: Message Processing, per message		-		IN/A	0.001825										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012147										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000117										
	ODUF: Message Processing, per message				N/A	0.002446										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.54										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010122										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004	•					,				
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001	•									
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 5 of 8

ODUF/ADUF	F/CMDS - Mississippi												Attachment:	1	Exhibit: B	
													Incremental			Incremental
											Elec	Submitted		Charge -	Charge - Manual Svc	Charge -
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
	1	m						(+)			per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													ist	Add I	DISC 1St	DISC Add I
						Rec	Nonre	curring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODLIE (A DIJE (C	MADO/FODUF															
ODUF/ADUF/C																
ENHAI	NCED OPTIONAL DAILY USAGE FILE (EODUF)					0.004045										
	EODUF: Message Processing, per message					0.234915										
ACCE	SS DAILY USAGE FILE (ADUF)					0.001001										
	ADUF: Message Processing, per message				N/A	0.001861										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012278										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000063										
	ODUF: Message Processing, per message				N/A	0.002509										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.97										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010232										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)						•			•		,				
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific	service	or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 6 of 8

ODUF/ADUF	F/CMDS - South Carolina												Attachment:	1	Exhibit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted		Charge -	Charge -	Charge -
											Elec			Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	por Lore		Electronic-		Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															DISC 1St	DISC Add I
						Rec		curring	Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C	MDS/EQDUE															
	NCED OPTIONAL DAILY USAGE FILE (EODUF)															
LINITAL	EODUF: Message Processing, per message					0.241298										
ACCE	SS DAILY USAGE FILE (ADUF)					0.241290										
ACCES					N/A	0.001856										
	ADUF: Message Processing, per message				N/A	0.001836										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012515										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000216										
	ODUF: Message Processing, per message				N/A	0.002508										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.84										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010429										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001	•									
Notes:	If no rate is identified in the contract, the rate for the specific	service	or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 7 of 8

ODUF/ADUF	F/CMDS - Tennessee												Attachment:	1	Exhibit: B	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	TES(\$)				Submitted		Charge -	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						5	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		•
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C																
ENHA	NCED OPTIONAL DAILY USAGE FILE (EODUF)															
	EODUF: Message Processing, per message					0.22977900										
ACCES	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.00182500										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012147										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.00000440										
	ODUF: Message Processing, per message				N/A	0.00244600										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.54000000										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00003390										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific	service	or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 8 of 8

LINBUN	IDI ED	NETWORK ELEMENTS - Alabama												Attach	ment: 2	Evk:	bit: C
ONDU	IDLED	NET WORK ELEMENTS - AIGUATIA	1		ı		1					Svo Ordor	Suc Order	Incremental			
													Submitted	Charge -			
															Charge -	Charge -	Charge -
CATEGO)RY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Elec		Manual Svc	Manual Svc	Manual Svc	
OA! LOC		NATE ELEMENTO	m	20.10	500	0000			ππι Ευ(ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							_ 1	Nonrec	curring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	he "Zon	e" shown in the sections for stand-alone loops or loops as part of	of a com	binatio	n refers to Geographi	ically Deavera	ged UNE Zones	. To view Geog	raphically Deav	eraged UNE Zo	ne Designation	ns by Centra	l Office, refe	r to Internet W	/ebsite:		-
l l	nttp://ww	w.interconnection.bellsouth.com/become_a_clec/html/interconne	ection.h	tm													
OPERAT	IONAL S	SUPPORT SYSTEMS															
1	NOTE: (1) Electronic Service Order: CLEC should contact its contract	t negot	iator if	it prefers the state :	specific elect	tronic service o	rdering charge	es as ordered b	y the State Co	mmissions. T	he electron	ic service o	rdering charg	e currently co	ntained in th	is rate
6	xhibit is	s the BellSouth regional electronic service ordering charge.	CLEC	may ele	ect either the state s	pecific Comr	mission ordered	d rates for the	electronic serv	ice ordering ch	arges, or CLE	C may elect	the region	al electronic s	service orderii	ng charge.	
1	NOTE: (2	2) Any element that can be ordered electronically will be bill	ed acco	rding	to the SOMEC rate li	isted in this	category. Pleas	se refer to Bell	South's Busine	ss Rules for L	ocal Ordering	(BBR-LO) to	determine	if a product of	can be ordere	d electronical	ly. For
l t	hose ele	ements that cannot be ordered electronically at present per t	he BBR	LO, th	ne listed SOMEC rate	e in this cate	gory reflects th	e charge that v	vould be billed	to a CLEC on	ce electronic o	rdering cap	abilities co	me on-line fo	r that element	. Otherwise,	the manual
	ordering	charge, SOMAN, will be applied to a CLECs bill when it sub	mits ar	LSR t	o BellSouth.												
	Е	Electronic OSS Charge, per LSR, submitted via BST's OSS															
		nteractive interfaces (Regional)				SOMEC		3.50									
		Manual Service Order Charge, per LSR, Disconnect Only (AL)				SOMAN				1.97							
		ATE ADVANCEMENT CHARGE															
		he Expedite charge will be maintained commensurate with I	BellSou	th's FO	CC No.1 Tariff, Section	on 5 as appli	cable.		-								
1 1		JNE Expedite Charge per Circuit or Line Assignable USOC, per	1			I]	_		
		Day			ALL UNE	SDASP		200.00									<u> </u>
		CHANGE ACCESS LOOP															
		ANALOG VOICE GRADE LOOP															
		-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	12.58	37.81	17.56	23.49	5.30		15.66				
		-Wire Analog Voice Grade Loop - Service Level 1- Zone 2			UEANL	UEAL2	21.05	37.81	17.56	23.49	5.30		15.66				
		-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	34.34	37.81	17.56	23.49	5.30		15.66				
		oop Testing - Basic 1st Half Hour			UEANL	URET1		34.16					15.66				
		oop Testing - Basic Additional Half Hour			UEANL	URETA		19.85					15.66				
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
		UVL-SL1)			UEANL	UREWO		15.78	8.94				15.66				
		ngineering Information Document (EI)			UEANL	UEANM		13.44									ļ
		Manual Order Coordination for UVL-SL1s (per loop)			UEANL	UEAMC		8.15									
		Order Coordination for Specified Conversion Time for UVL-SL1															
- 		per LSR)			UEANL	OCOSL		18.09									
		Jnbundled COPPER LOOP			LIEO	LIEONY	44.00	04.44	45.40	04.05	1.15		45.00				
		-Wire Unbundled Copper Loop - Non-Designed Zone 1	!		UEQ UEQ	UEQ2X UEQ2X	11.20 13.27	34.14 34.14	15.10	21.25 21.25	4.15 4.15		15.66 15.66				
-		Wire Unbundled Copper Loop - Non-Designed - Zone 2	<u> </u>		UEQ	UEQ2X	15.27	34.14	15.10 15.10	21.25	4.15		15.66				
		Wire Unbundled Copper Loop - Non-Designed - Zone 3 Order Coordination 2 Wire Unbundled Copper Loop - Non-	'	3	UEQ	UEQZX	15.07	34.14	15.10	21.25	4.15		15.00				
		Designed (per loop)			UEQ	USBMC		8.15									
		Ingineering Information Document			UEQ	USDIVIC		13.44					15.66				
		oop Testing - Basic 1st Half Hour			UEQ	URET1		34.16					15.66				-
		oop Testing - Basic 1st Hall Hour			UEQ	URETA		19.85					15.66				1
		CLEC to CLEC Conversion Charge Without Outside Dispatch			OLQ	OKLIA		13.00					13.00				
		UCL-ND)			UEQ	UREWO		14.27	7.43				15.66				
UNBUNI		CHANGE ACCESS LOOP			OLQ	ORLIVO		14.27	7.40				10.00				1
		ANALOG VOICE GRADE LOOP															
		-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or															
		Ground Start Signaling - Zone 1		1	UEA	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66				
		-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or			027	O L / LLL		00.00	00.00				10.00				
		Ground Start Signaling - Zone 2		2	UEA	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66				
		P-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or															
		Ground Start Signaling - Zone 3	1	3	UEA	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66	1	I		
		Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.09									
		-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
		Battery Signaling - Zone 1	l	1	UEA	UEAR2	14.38	88.00	55.00	47.24	7.44		15.66		1		
	2	-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
L_		Battery Signaling - Zone 2	<u> </u>	2	UEA	UEAR2	22.85	88.00	55.00	47.24	7.44	<u> </u>	15.66	<u> </u>	<u> </u>		
	2	-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	В	Battery Signaling - Zone 3	<u> </u>	3	UEA	UEAR2	36.14	88.00	55.00	47.24	7.44		15.66	<u> </u>	<u></u>		
		Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.09									
		CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36				15.66				
		ANALOG VOICE GRADE LOOP															
		-Wire Analog Voice Grade Loop - Zone 1			UEA	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66				
	4	-Wire Analog Voice Grade Loop - Zone 2		2	UEA	UEAL4	38.58	131.97	94.51	59.14	14.50		15.66				

10/04/02 Page 1 of 362

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhi	bit: C
											Svc Order	Svc Order		Incremental		Incremental
											1	Submitted		Charge -	Charge -	Charge -
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Elec	Manually		Manual Svc	Manual Svc	
CATEGORI	RATE ELEMENTS	m	Zone	ВСЗ	0300			KATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonreci		Nonrecurring					Rates(\$)		
	A Mire Apples Vision Conda Long 7000 2		_	UEA	UEAL4	60.02	First	Add'I 94.51	First	Add'I 14.50	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Analog Voice Grade Loop - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UEA	OCOSL	60.02	131.97 18.09	94.51	59.14	14.50		15.66				
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36			1	15.66				+
2-WIR	E ISDN DIGITAL GRADE LOOP			0271	0.1.2.1.0		02	00.00				10.00				
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	21.88	117.24	79.77	52.88	10.54		15.66				
	2-Wire ISDN Digital Grade Loop - Zone 2			UDN	U1L2X	32.85	117.24	79.77	52.88	10.54		15.66				
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	48.55	117.24	79.77	52.88	10.54		15.66				
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		18.09					15.00				
0.1405	CLEC to CLEC Conversion Charge without outside dispatch	A TIDLE		UDN	UREWO		91.63	44.16				15.66				_
2-WIR	E ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP 2 Wire Unbundled ADSL Loop including manual service inquiry	AIIBLE	LOOP	<u> </u>												
	& facility reservation - Zone 1		1	UAL	UAL2X	11.01	110.00	68.00	47.24	7.44		15.66				
+	2 Wire Unbundled ADSL Loop including manual service inquiry		<u> </u>	O/ IL	ONLEN	11.01	110.00	00.00	77.27	,,,,,		10.00				1
	& facility reservation - Zone 2		2	UAL	UAL2X	12.73	110.00	68.00	47.24	7.44		15.66				
	2 Wire Unbundled ADSL Loop including manual service inquiry						j									
	& facility reservation - Zone 3			UAL	UAL2X	14.30	110.00	68.00	47.24	7.44	ļ	15.66				
L	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.09									
	2 Wire Unbundled ADSL Loop without manual service inquiry &		١.,		1141 0147	44.04	00.00	F7.00	47.04	7.44		45.00				
—	facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry &		1	UAL	UAL2W	11.01	90.00	57.00	47.24	7.44	1	15.66				
	facility reservaton - Zone 2		2	UAL	UAL2W	12.73	90.00	57.00	47.24	7.44		15.66				
	2 Wire Unbundled ADSL Loop without manual service inquiry &			UAL	UALZVV	12.73	90.00	37.00	47.24	7.44		13.00				
	facility reservaton - Zone 3		3	UAL	UAL2W	14.30	90.00	57.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.09									
	CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		86.20	40.40				15.66				
2-WIR	E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE L	_OOP													
	2 Wire Unbundled HDSL Loop including manual service inquiry					0.74	440.00	00.00	47.04	7.44		45.00				
—	& facility reservation - Zone 1 2 Wire Unbundled HDSL Loop including manual service inquiry		1	UHL	UHL2X	8.74	110.00	68.00	47.24	7.44	1	15.66				
	& facility reservation - Zone 2		2	UHL	UHL2X	10.17	110.00	68.00	47.24	7.44		15.66				
 	2 Wire Unbundled HDSL Loop including manual service inquiry			OFF	OFFICEX	10.17	110.00	00.00	71.27	7.44		13.00				1
	& facility reservation - Zone 3		3	UHL	UHL2X	11.44	110.00	68.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.09									
	2 Wire Unbundled HDSL Loop without manual service inquiry															
L	and facility reservation - Zone 1		1	UHL	UHL2W	8.74	90.00	57.00	47.24	7.44		15.66				
	2 Wire Unbundled HDSL Loop without manual service inquiry		_			40.47	00.00	F7.00	47.04	7.44		45.00				
h	and facility reservation - Zone 2 2 Wire Unbundled HDSL Loop without manual service inquiry		2	UHL	UHL2W	10.17	90.00	57.00	47.24	7.44		15.66				
	and facility reservation - Zone 3	1	3	UHL	UHL2W	11.44	90.00	57.00	47.24	7.44		15.66	1			
	Order Coordination for Specified Conversion Time (per LSR)	1		UHL	OCOSL	11.44	18.09	07.00	77.27	717	1	10.00	1	1		1
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO	1	86.14	40.40				15.66				†
4-WIR	E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE L	_00P													
	4 Wire Unbundled HDSL Loop including manual service inquiry							<u> </u>								
	and facility reservation - Zone 1	ļ	1	UHL	UHL4X	13.95	148.36	68.00	51.70	9.73	ļ	15.66	ļ			1
	4-Wire Unbundled HDSL Loop including manual service inquiry		_			45.50	440.00	00.00	54.70	0 =0		45.00				
\vdash	and facility reservation - Zone 2 4-Wire Unbundled HDSL Loop including manual service inquiry	 	2	UHL	UHL4X	15.56	148.36	68.00	51.70	9.73		15.66				-
	and facility reservation - Zone 3	1	3	UHL	UHL4X	15.25	148.36	68.00	51.70	9.73		15.66	1			
	Order Coordination for Specified Conversion Time (per LSR)	1		UHL	OCOSL	10.20	18.09	00.00	31.70	3.13	 	13.00				
	4-Wire Unbundled HDSL Loop without manual service inquiry					1							İ			1
	and facility reservation - Zone 1		1	UHL	UHL4W	13.95	94.00	57.00	51.70	9.73	<u></u>	15.66				
	4-Wire Unbundled HDSL Loop without manual service inquiry							<u> </u>								
	and facility reservation - Zone 2		2	UHL	UHL4W	15.56	94.00	57.00	51.70	9.73	ļ	15.66	ļ			
	4-Wire Unbundled HDSL Loop without manual service inquiry				L c											
 	and facility reservation - Zone 3	1		UHL UHL	UHL4W	15.25	94.00	57.00	51.70	9.73		15.66	-			1
\vdash	Order Coordination for Specified Conversion Time (per LSR) CLEC to CLEC Conversion Charge without outside dispatch	 		UHL	OCOSL UREWO	-	18.09 86.14	40.40	+		 	15.66	1	1	-	
4-WIR	E DS1 DIGITAL LOOP		-	OI IL	OKEWO	 	00.14	40.40	1		 	13.00				
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	4-Wire DS1 Digital Loop - Zone 1	1	1	USL	USLXX	82.55	252.47	157.54	44.70	11.71	1	15.66	1	1		†
LL	T THIS DOT DIGITAL LOOP - ZOITE 1	l		1001	COLAN	02.00	202.41	107.04	77.70	11.71	1	15.00	l	I	I	Ь

UNBUN	DLED	NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhib	oit: C
												Svc Order	Svc Order	Incremental	Incremental		Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGO	RY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
						\bot											
							Rec	Nonrec		Nonrecurring					Rates(\$)		
		A William DOA District Laws - 7 case 0		_	1101	1101.307	454.40	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-		4-Wire DS1 Digital Loop - Zone 2			USL	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				
		4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX OCOSL	314.52	252.47	157.54	44.70	11.71		15.66				
-		Order Coordination for Specified Conversion Time (per LSR)			USL	UREWO		18.09 101.09	43.05				15.66				
H 4		CLEC to CLEC Conversion Charge without outside dispatch 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP		-	USL	UKEWU		101.09	43.03				15.66				
	WINE	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	26.09	126.27	88.80	59.14	14.50		15.66				
		4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	35.95	126.27	88.80	59.14	14.50		15.66				
 		4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	37.88	126.27	88.80	59.14	14.50		15.66				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	37.88	126.27	88.80	59.14	14.50		15.66				
		Order Coordination for Specified Conversion Time (per LSR)		Ť	UDL	OCOSL	555	18.09	22.30		30				İ		
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 2		2	UDL	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL	UDL64	37.88	126.27	88.80	59.14	14.50		15.66				
		Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		18.09									
		CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		102.13	49.75				15.66				
2-	-WIRE	Unbundled COPPER LOOP															
		2-Wire Unbundled Copper Loop/Short including manual service															
		inquiry & facility reservation - Zone 1		1	UCL	UCLPB	11.01	112.46	65.30	47.24	7.44		15.66				
		2-Wire Unbundled Copper Loop/Short including manual service															
		inquiry & facility reservation - Zone 2		2	UCL	UCLPB	12.73	112.46	65.30	47.24	7.44		15.66				
		2 Wire Unbundled Copper Loop/Short including manual service															
		inquiry & facility reservation - Zone 3		3	UCL	UCLPB	14.30	112.46	65.30	47.24	7.44		15.66				
		Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.15	8.15								
		2-Wire Unbundled Copper Loop/Short without manual service															
		inquiry and facility reservation - Zone 1	ı	1	UCL	UCLPW	11.01	91.46	54.30	47.24	7.44		15.66				
		2-Wire Unbundled Copper Loop/Short without manual service		_			40.70		=	4= 0.4							
		inquiry and facility reservation - Zone 2	- 1	2	UCL	UCLPW	12.73	91.46	54.30	47.24	7.44		15.66				
		2-Wire Unbundled Copper Loop/Short without manual service		3		LIOL DIA	44.00	04.40	5400	47.04	7.44		45.00				
-		inquiry and facility reservation - Zone 3	ı	3	UCL	UCLPW UCLMC	14.30	91.46	54.30	47.24	7.44		15.66				
-		Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLIVIC		8.15	8.15								
		2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone 1		1	UCL	UCL2L	31.42	112.46	65.30	47.24	7.44		15.66				
		2-Wire Unbundled Copper Loop/Long - includes manual svc.		-	UCL	UCLZL	31.42	112.40	05.50	47.24	7.44		13.00				
		inquiry and facility reservation - Zone 2		2	UCL	UCL2L	55.01	112.46	65.30	47.24	7.44		15.66				
		2-Wire Unbundled Copper Loop/Long - includes manual svc.			OOL	OOLZL	33.01	112.40	00.00	77.27	7.77		13.00				
		inquiry and facility reservation - Zone 3		3	UCL	UCL2L	80.00	112.46	65.30	47.24	7.44		15.66				
		Order Coordination for Unbundled Copper Loops (per loop)		Ť	UCL	UCLMC	33.55	8.15	8.15				.0.00				
		2-Wire Unbundled Copper Loop/Long - without manual service			1	1		25	20								
		inquiry and facility reservation - Zone 1	- 1	1	UCL	UCL2W	31.42	91.46	54.30	47.24	7.44		15.66				
		2-Wire Unbundled Copper Loop/Long - without manual service															
		inquiry and facility reservation - Zone 2	- 1	2	UCL	UCL2W	55.01	91.46	54.30	47.24	7.44		15.66				
		2-Wire Unbundled Copper Loop/Long - without manual service								1							
		inquiry and facility reservation - Zone 3	- 1	3	UCL	UCL2W	80.00	91.46	54.30	47.24	7.44		15.66				
		Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.15	8.15	<u> </u>							
		CLEC to CLEC Conversion Charge without outside dispatch															
		(UCL-Des)			UCL	UREWO		97.23	42.48				15.66				
4-	WIRE	COPPER LOOP															
		4-Wire Copper Loop/Short - including manual service inquiry				1											
LL		and facility reservation - Zone 1		1	UCL	UCL4S	17.36	135.21	88.05	51.70	9.73		15.66				
		4-Wire Copper Loop/Short - including manual service inquiry		_				,									
		and facility reservation - Zone 2		2	UCL	UCL4S	20.76	135.21	88.05	51.70	9.73		15.66				
		4-Wire Copper Loop/Short - including manual service inquiry		_		1,101.10		40= 0:					4-0-				
-		and facility reservation - Zone 3		3	UCL	UCL4S	28.21	135.21	88.05	51.70	9.73		15.66				
\vdash		Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.15	8.15	 					1		
		4-Wire Copper Loop/Short - without manual service inquiry and	Ι.	1	UCL	UCL4W	17.36	114.21	67.05	51.70	9.73		15.66				
		facility reservation - Zone 1		1	JUUL	UCL4VV	17.30	114.21	67.05	51.70	9.73	l	10.00		I.		

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	oit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
-	4-Wire Copper Loop/Short - without manual service inquiry and						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	facility reservation - Zone 2	١.,	2	UCL	UCL4W	20.76	114.21	67.05	51.70	9.73		15.66				ı
 	4-Wire Copper Loop/Short - without manual service inquiry and	<u> </u>		UCL	UCL4VV	20.70	114.21	07.03	31.70	9.13		13.00				
	facility reservation - Zone 3	l i	3	UCL	UCL4W	28.21	114.21	67.05	51.70	9.73		15.66				ı
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.15	8.15								
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 1		1	UCL	UCL4L	49.35	135.21	88.05	51.70	9.73		15.66				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															ı
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	92.45	135.21	88.05	51.70	9.73		15.66				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3		3	UCL	UCL4L	127.39	135.21	88.05	51.70	9.73		15.66				, ,
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	127.59	8.15	8.15	31.70	3.13		10.00				
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 1	- 1	1	UCL	UCL4O	49.35	114.21	67.05	51.70	9.73		15.66				
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 2		2	UCL	UCL4O	92.45	114.21	67.05	51.70	9.73		15.66				
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 3	١.,	3	UCL	UCL4O	127.39	114.21	67.05	51.70	9.73		15.66				ı
-	Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLMC	127.39	8.15	8.15	51.70	9.73		15.66				
+	CLEC to CLEC conversion Charge without outside dispatch			UCL	UREWO		97.23	42.48				15.66				
LOOP MODIFI							9110									
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft	ı		UAL, UHL, UCL, UEQ, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULM2L		0.00	0.00				15.66				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft	l ,		UCL, ULS, UEQ	ULM2G		170.51	170.51				15.66				
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18K ft			UHL, UCL	ULM4L		0.00	0.00				15.66				
 	Unbundled Loop Modification Removal of Load Coils - 4 Wire	<u> </u>		OFIL, OCL	OLIVIAL		0.00	0.00				13.00				
	pair greater than 18k ft	1		UCL	ULM4G		170.51	170.51				15.66				ı
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop	I		UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULMBT		32.41	32.41				15.66				
SUB-LOOPS	Black at an															
Sub-L	oop Distribution Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-				 											
	Up	ı		UEANL	USBSA		244.42					15.66				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	- 1		UEANL	USBSB		22.64					15.66				
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up	I		UEANL	USBSC		177.45					15.66				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up	1		UEANL	USBSD		55.15					15.66				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN2	11.21	65.80	30.96	45.25	6.70		15.66				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN2	11.94	65.80	30.96	45.25	6.70		15.66				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN2	16.86	65.80	30.96	45.25	6.70		15.66				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		Ĭ	UEANL	USBMC	.0.00	8.15	8.15	.0.20	30						
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN4	8.46	79.03	44.19	49.71	9.07		15.66				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -		2	UEANL	USBN4				49.71	9.07						
<u> </u>	Zone 2	1	2	UEANL	USBN4	16.67	79.03	44.19	49.71	9.07	<u> </u>	15.66		I	I	1

UNBUN	IDLE	NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	oit: C
CATEGO		RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
											B'						
-							Rec	Nonrec First	urring Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
-		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -						FIISL	Add I	FIISL	Auu i	SOWIEC	SOWAN	SOWAN	SOWAN	SOWAN	SOWAN
		Zone 3		3	UEANL	USBN4	32.57	79.03	44.19	49.71	9.07		15.66				, ,
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.15	8.15								
		Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	1		UEANL	USBR2	2.27	53.01	18.17	45.25	6.70		15.66				
		0.10				1100140		0.45	0.45								, ,
-		Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	-		UEANL UEANL	USBMC USBR4	5.16	8.15 59.25	8.15 24.41	49.71	9.07		15.66				
-		Sub-Loop 4-vviile intrabuliding Network Cable (INC)			UEAINL	USBK4	5.16	59.25	24.41	49.71	9.07		13.00				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.15	8.15								,
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1	UEF	UCS2X	6.22	65.80	30.96	45.25	6.70		15.66				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		2	UEF	UCS2X	8.76	65.80	30.96	45.25	6.70		15.66				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		3	UEF	UCS2X	11.27	65.80	30.96	45.25	6.70		15.66				
		Onder Consideration for Habrardie 10.1. Language 2.1.		1	UEF	USBMC		0.45	0.4-								ļ
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1	UEF	UCS4X	6.11	8.15 79.03	8.15 44.19	49.71	9.07		15.66				
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2			UEF	UCS4X	12.61	79.03	44.19	49.71	9.07		15.66				
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3			UEF	UCS4X	15.36	79.03	44.19	49.71	9.07		15.66				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.15	8.15								
U	Inbund	lled Sub-Loop Modification															
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load							=				4= 00				,
		Coil/Equip Removal per 2-W PR			UEF	ULM2X		175.78	5.10				15.66				
		Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR			UEF	ULM4X		175.78	5.10				15.66				,
-		Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged			ULI	OLIVIAX		173.76	3.10				13.00				
		Tap Removal, per PR unloaded			UEF	ULM4T		278.20	6.11				15.66				,
U	Inbund	lled Network Terminating Wire (UNTW)															
		Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.40	30.01					15.66				
N		k Interface Device (NID)															
		Network Interface Device (NID) - 1-2 lines			UENTW UENTW	UND12 UND16		43.23	28.38				15.66				
		Network Interface Device (NID) - 1-6 lines Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		63.97 5.87	49.11 5.87				15.66 15.66				
-		Network Interface Device Cross Connect - 4W			UENTW	UNDC4		5.87	5.87				15.66				
SUB-LOC	OPS							-									
S	ub-Lo	op Feeder															1
		USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
		Distribution Facility set-up			UDN,UCL,UDL,UDC	USBFW		244.42					15.66				-
		USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up			UEA, UDN.UCL.UDL.UDC	USBFX		22.64	22.64				15.66				1
		USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		519.95	11.32				15.66				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice				30Di Z		313.33	11.02				10.00				
		Grade - Zone 1		1	UEA	USBFA	8.03	93.00	56.48	54.51	13.67		15.66				1
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice															
		Grade - Zone 2		2	UEA	USBFA	12.00	93.00	56.48	54.51	13.67		15.66				
		Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,				l											1
 		Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR		3	UEA UEA	USBFA OCOSL	20.39	93.00 18.09	56.48	54.51	13.67		15.66				
-		Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice	-	-	OLA	UUUSL		18.09				-	-				
		Grade - Zone 1		1	UEA	USBFB	8.03	93.00	56.48	54.51	13.67		15.66				ļ
		Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice		<u> </u>			5.55	33.55	55.16	001	.0.07		.0.00				
		Grade - Zone 2	L_	2	UEA	USBFB	12.00	93.00	56.48	54.51	13.67		15.66				<u> </u>
		Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice														_	
		Grade - Zone 3	1	3	UEA	USBFB	20.39	93.00	56.48	54.51	13.67		15.66				,
-		Order Coordination for Specified Time Conversion, per LSR	1	-	UEA	OCOSL		18.09							-		
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1		1	UEA	USBFC	8.03	93.00	56.48	54.51	13.67		15.66				
 		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,	-	- '-	0=/1	2021 0	0.03	33.00	JU. + 0	34.31	10.07	1	10.00				
		Voice Grade - Zone 2		2	UEA	USBFC	12.00	93.00	56.48	54.51	13.67		15.66				, !
	-		•	-	•										•		-

UNBUNDLI	ED NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	bit: C
											Svc Order	Svc Order				Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi	_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect		1	oss	Rates(\$)		'
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse		_		LIODEO	00.00	00.00	50.40	54.54	40.07		45.00				j '
	Battery, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR		3	UEA UEA	USBFC OCOSL	20.39	93.00 18.09	56.48	54.51	13.67		15.66				
-	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice			OLA	OCOSE		10.09									
	Grade - Zone 1		1	UEA	USBFD	19.21	107.56	70.09	62.05	17.40		15.66				İ
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 2		2	UEA	USBFD	23.47	107.56	70.09	62.05	17.40		15.66				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice		_													İ
	Grade - Zone 3		3	UEA	USBFD	39.63	107.56	70.09	62.05	17.40		15.66				-
-	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			UEA	OCOSL		18.09									-
	Grade - Zone 1		1	UEA	USBFE	19.21	107.56	70.09	62.05	17.40		15.66				İ
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			02,1	002. 2	10.21	101.00	70.00	02.00			10.00				
	Grade - Zone 2		2	UEA	USBFE	23.47	107.56	70.09	62.05	17.40		15.66				İ
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice							-								
	Grade - Zone 3		3	UEA	USBFE	39.63	107.56	70.09	62.05	17.40		15.66				1
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL	44.07	18.09	00.00	55.04	40.00		45.00				
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1 Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN UDN	USBFF USBFF	14.87 21.69	106.16 106.16	68.69 68.69	55.64 55.64	13.29 13.29		15.66 15.66				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3			UDN	USBFF	32.51	106.16	68.69	55.64	13.29		15.66				
	Order Coordination For Specified Conversion Time, Per LSR		3	UDN	OCOSL	32.31	18.09	00.03	33.04	13.23		15.00				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDC	USBFS	14.87	106.16	68.69	55.64	13.29		15.66				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	21.69	106.16	68.69	55.64	13.29		15.66				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	32.51	106.16	68.69	55.64	13.29		15.66				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1			USL	USBFG	55.09	101.85	64.38	62.05	17.40		15.66				
-	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2			USL	USBFG USBFG	124.69 294.62	101.85 101.85	64.38 64.38	62.05	17.40 17.40		15.66 15.66				
-	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3 Order Coordination For Specified Conversion Time, Per LSR		3	USL USL	OCOSL	294.62	18.09	64.38	62.05	17.40		15.00				-
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	5.75	83.78	46.32	53.02	10.67		15.66				<u> </u>
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone															
	2		2	UCL	USBFH	4.93	83.78	46.32	53.02	10.67		15.66				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone															
	3			UCL	USBFH	3.96	83.78	46.32	53.02	10.67		15.66				
	Order Coordination For Specified Conversion Time, per LSR Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1			UCL UCL	OCOSL USBFJ	12.71	18.09 100.99	63.53	57.90	13.26		15.66				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1 Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2			UCL	USBFJ	9.69	100.99	63.53	57.90 57.90	13.26		15.66				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	14.37	100.99	63.53	57.90	13.26		15.66				
	Order Coordination For Specified Conversion Time, per LSR		_	UCL	OCOSL		18.09		000							
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	19.20	101.85	64.38	62.05	17.40		15.66				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	21.64	101.85	64.38	62.05	17.40		15.66				
\vdash	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	23.75	101.85	64.38	62.05	17.40	<u> </u>	15.66				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 1		1	UDL	USBFO	19.20	101.85	64.38	62.05	17.40		15.66				
 	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -			ODL	JJDI-U	19.20	101.05	04.38	02.05	17.40	 	15.00				
	Zone 2		2	UDL	USBFO	21.64	101.85	64.38	62.05	17.40		15.66				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -				1	204		000	52.50	+0						
	Zone 3	<u> </u>	3	UDL	USBFO	23.75	101.85	64.38	62.05	17.40	<u></u>	15.66				
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		18.09	· · · · ·								
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		l .	LIDI	HODES					.=						1
	Zone 1		1	UDL	USBFP	19.20	101.85	64.38	62.05	17.40	 	15.66				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2		2	UDL	USBFP	21.64	101.85	64.38	62.05	17.40		15.66				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -				30011	21.04	101.03	04.38	02.03	17.40	1	10.00				
	Zone 3		3	UDL	USBFP	23.75	101.85	64.38	62.05	17.40		15.66				
	Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		18.09									
SUB-LOOPS																
Sub-L	Loop Feeder	<u> </u>		LIEO	41.50	10.55					<u> </u>	1				
	Sub Loop Feeder - DS3 - Per Mile Per Month Sub Loop Feeder - DS3 - Facility Termination Per Month	 		UE3 UE3	1L5SL USBF1	13.55 332.40	3,400,58	407.00	160.47	90.97	 	15.66		-		
	Cas 200p 1 00001 DOO 1 domity formination 1 of Month	' '		020	3051	332.70	5,-100.50	-407.00	100.47	30.31	1	10.00	1	1		

UNBUN	NDLE	D NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Exhi	bit: C
CATEGO		RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
		O. I. I. and Francis OTO 4. Darking Darking			LIDLOY	41.501		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-		Sub Loop Feeder – STS-1 – Per Mile Per Month Sub Loop Feeder - STS-1 - Facility Termination Per Month	H		UDLSX UDLSX	1L5SL USBF7	13.55 357.36	3,400.58	407.00	160.47	90.97		15.66				
-		Sub Loop Feeder - OC-3 - Per Mile Per Month	i i		UDLO3	1L5SL	10.28	3,400.30	407.00	100.47	30.31		13.00				
		Sub Loop Feeder - OC-3 - Facility Termination Protection Per															
		Month	I		UDLO3	USBF5	54.89										<u> </u>
		Sub Loop Feeder - OC-3 - Facility Termination Per Month Sub Loop Feeder - OC-12 - Per Mile Per Month			UDLO3 UDL12	USBF2 1L5SL	538.69 12.66	3,400.58	407.00	160.47	90.97		15.66				
		Sub Loop Feeder - OC-12 - Per Mile Per Month Sub Loop Feeder - OC-12 - Facility Termination Protection Per	-		UDL12	ILSSL	12.00										
		Month	- 1		UDL12	USBF6	620.18										
		Sub Loop Feeder - OC-12 - Facility Termination Per Month	ı		UDL12	USBF3	1,729.00	3,400.58	407.00	160.47	90.97		15.66				
		Sub Loop Feeder - OC-48 - Per Mile Per Month	ı		UDL48	1L5SL	41.51										_
		Sub Loop Feeder - OC-48 - Facility Termination Protection Per Month			UDL48	USBF9	310.30										
 		Sub Loop Feeder - OC-48 - Facility Termination Per Month		-	UDL48	USBF4	1,495.00	3,586.58	407.00	160.47	90.97	1	15.66				
		Sub Loop Feeder - OC-12 Interface On OC-48	Ė		UDL48	USBF8	350.09	804.67	407.00	160.47	90.97		15.66				
UNBUN	DLED L	OOP CONCENTRATION															
		Unbundled Loop Concentration - System A (TR008) Unbundled Loop Concentration - System B (TR008)			ULC	UCT8A UCT8B	364.17 43.70	325.41 135.59	325.41 135.59				15.66 15.66				_
		Unbundled Loop Concentration - System B (TR008) Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	395.12	325.41	325.41				15.00				
		Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	73.64	135.59	135.59				15.66				
		Unbundled Loop Concentration - DS1 Loop Interface Card			ULC	UCTCO	4.16	63.29	46.07	16.79	4.70		15.66				
		Unbundled Loop Concentration - ISDN Loop Interface (Brite															
-		Card) Unbundled Loop Concentration - UDC Loop Interface (Brite			UDN	ULCC1	6.60	10.54	10.48	5.39	5.36		15.66				.
		Card)			UDC	ULCCU	6.60	10.54	10.48	5.39	5.36		15.66				
		Unbundled Loop Concentration2 Wire Voice-Loop Start or			CDC	02000	0.00	10.04	10.40	0.00	0.00		10.00				
		Ground Start Loop Interface (POTS Card)			UEA	ULCC2	1.65	10.54	10.48	5.39	5.36		15.66				
		Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery											4= 00				
		Loop Interface (SPOTS Card) Unbundled Loop Concentration - 4 Wire Voice Loop Interface			UEA	ULCCR	9.81	10.54	10.48	5.39	5.36		15.66				
		(Specials Card)			UEA	ULCC4	5.85	10.54	10.48	5.39	5.36		15.66				
		Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	28.60	10.54	10.48	5.39	5.36		15.66				1
		Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface			UDL	ULCC7	8.67	10.54	10.48	5.39	5.36		15.66				
		Unbundled Loop Concentration - Digital 56 Kbps Data Loop								= 00			4= 00				
\vdash		Interface Unbundled Loop Concentration - Digital 64 Kbps Data Loop			UDL	ULCC5	8.67	10.54	10.48	5.39	5.36		15.66				
		Interface			UDL	ULCC6	8.67	10.54	10.48	5.39	5.36		15.66				
UNE OT	HER, F	ROVISIONING ONLY - NO RATE			-												1
		NID - Dispatch and Service Order for NID installation			UENTW	UNDBX	0.00	0.00									
		UNTW Circuit Id Establishment, Provisioning Only - No Rate			UENTW UEANL,UEF,UEQ,U	UENCE	0.00	0.00									
		Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN	0.00	0.00									
UNE OT	HER, F	ROVISIONING ONLY - NO RATE				0.12011	0.00	0.00									
					UAL,UCL,UDC,UDL,												
		Unbundled Contact Name, Provisioning Only - no rate Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no			UDN,UEA,UHL,ULC	UNECN	0.00	0.00									+
		rate			UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
		Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no															1
		rate		ļ	UEA,USL,UCL,UDL	USBFR	0.00	0.00									<u> </u>
\vdash		Unbundled DS1 Loop - Superframe Format Option - no rate Unbundled DS1 Loop - Expanded Superframe Format option -		<u> </u>	USL	CCOSF	0.00	0.00				ļ					
		no rate			USL	CCOEF	0.00	0.00									
HIGH CA	APACIT	TY UNBUNDLED LOCAL LOOP					0.00	0.00									
		High Capacity Unbundled Local Loop - DS3 - Per Mile per month			UE3	1L5ND	8.38										
		High Capacity Unbundled Local Loop - DS3 - Facility Termination per month			UE3	UE3PX	308.98	451.52	263.94	119.49	83.58		15.66				

UNBUND	DLED	NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhi	bit: C
CATEGOR		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Submitted Manually	Incremental Charge -	Incremental Charge - Manual Svc Order vs.		Incremental Charge -
			""											Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic- Disc Add'l
							Rec	Nonred		Nonrecurring					Rates(\$)		
-		High Capacity Unbundled Local Loop - STS-1 - Per Mile per						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		month			UDLSX	1L5ND	8.38										
		High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month			UDLSX	UDLS1	319.83	451.52	263.94	119.49	83.58		15.66				
LOOP MAR					ODLOX	UDLST	319.03	451.52	263.94	119.49	63.36		13.66				
		Loop Makeup - Preordering Without Reservation, per working or															1
		spare facility queried (Manual).			UMK	UMKLW		20.00	20.00								
		Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).			UMK	UMKLP		21.00	21.00								
		Loop MakeupWith or Without Reservation, per working or															
LINDUND		spare facility queried (Mechanized) EDICATED TRANSPORT			UMK	PSUMK		0.59	0.59								
		EDICATED TRANSPORT INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	a perio	d - below DS3=one	month, DS3/	STS-1=four mo	nths									+
		FFICE CHANNEL - DEDICATED TRANSPORT		g pone		1											
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -				41 =>07											
		Per Mile per month Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			U1TVX	1L5XX	0.008838										
		Facility Termination			U1TVX	U1TV2	21.13	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade				41 =>07											
		Rev Bat Per Mile per month Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat			U1TVX	1L5XX	0.008838										+
		Facility Termination			U1TVX	U1TR2	21.13	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -			LIATE OV	41.500/	0.000000										
		Per Mile per month Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade			U1TVX	1L5XX	0.008838										<u> </u>
		- Facility Termination			U1TVX	U1TV4	18.73	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile				41 =>07											
		per month Interoffice Channel - Dedicated Transport - 56 kbps - Facility			U1TDX	1L5XX	0.008838										
		Termination			U1TDX	U1TD5	15.12	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile				41 =>07											
		per month Interoffice Channel - Dedicated Transport - 64 kbps - Facility			U1TDX	1L5XX	0.008838										-
		Termination			U1TDX	U1TD6	15.12	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per				41 =>07											
		month Interoffice Channel - Dedicated Tranport - DS1 - Facility			U1TD1	1L5XX	0.18										+
		Termination			U1TD1	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per															
		month Interoffice Channel - Dedicated Transport - DS3 - Facility			U1TD3	1L5XX	4.09										+
	ľ	Termination per month			U1TD3	U1TF3	703.52	278.75	162.76	60.20	58.46		15.66				
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month			U1TS1	1L5XX	4.09										
		Interoffice Channel - Dedicated Transport - STS-1 - Facility			01151	ILSAX	4.09										
	ŀ	Termination			U1TS1	U1TFS	701.37	278.75	162.76	60.20	58.46		15.66				
		CHANNEL - DEDICATED TRANSPORT	<u> </u>	<u> </u>		200/070											
NO		OCAL CHANNEL DEDICATED TRANSPORT - minimum billin Local Channel - Dedicated - 2-Wire Voice Grade	g perio	a - peid	ULDVX	ULDV2	13.97	193.10	33.17	36.64	3.20		15.66				
		Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat			ULDVX	ULDR2	13.97	193.10	33.17	36.64	3.20		15.66				
		Local Channel - Dedicated - 4-Wire Voice Grade			UNDVX	ULDV4	14.93	193.53	33.60	27.11	3.67		15.66	İ			<u> </u>
		Local Channel - Dedicated - DS1 - Zone 1		1	ULDD1	ULDF1	35.76	177.47	153.72	22.19	15.26		15.66				
		Local Channel - Dedicated - DS1 - Zone 2		2	ULDD1	ULDF1	49.98	177.47	153.72	22.19	15.26		15.66				
		Local Channel - Dedicated - DS1 - Zone 3		3	ULDD1	ULDF1	107.63	177.47	153.72	22.19	15.26		15.66	ļ			ļ
		Local Channel - Dedicated - DS3 - Per Mile per month		<u> </u>	ULDD3	1L5NC	6.92	151 50	100.01	110.10	20.50	<u> </u>	45.00				
		Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1- Per Mile per month	<u> </u>	 	ULDD3 ULDS1	ULDF3 1L5NC	416.54 6.92	451.52	463.94	119.49	83.58	 	15.66				
		Local Channel - Dedicated - STS-1- Per Mile per Month Local Channel - Dedicated - STS-1 - Facility Termination		 	ULDS1	ULDFS	408.49	451.52	463.94	119.49	83.58	-	15.66				
DARK FIBI	ER										23.00						†
	-							l l									

LINIDIIN	IDI EI	NETWORK ELEMENTS - Alabama												Attack		Fkit	-it. C
UNDUI	NDLEL	D NET WORK ELEMENTS - Alabama	1			1						Cua Oudan	Cur Ouden	Incremental	ment: 2	Incremental	oit: C
												Submitted	Submitted		Charge -	Charge -	Charge -
047506		DATE EL EMENTO	Interi	-	D00	11000			D.4.T.F.O.(A)			Elec	,	Manual Svc	Manual Svc		Manual Svc
CATEGO	DRY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
		Thereof per month - Local Channel			UDF	1L5DC	60.32										
		NRC Dark Fiber - Local Channel			UDF	UDFC4		639.09	137.87	317.06	197.66		15.66				
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
		Thereof per month - Interoffice Channel			UDF	1L5DF	22.34										
		NRC Dark Fiber - Interoffice Channel			UDF	UDF14		639.09	137.87	317.06	197.66		15.66				
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
		Thereof per month - Local Loop			UDF	1L5DL	60.32										
		NRC Dark Fiber - Local Loop			UDF	UDFL4		639.09	137.87	317.06	197.66		15.66				
8XX ACC		EN DIGIT SCREENING															
1		8XX Access Ten Digit Screening, Per Call			OHD		0.00056										
		8XX Access Ten Digit Screening, Reservation Charge Per 8XX			0.12		0.00000										
		Number Reserved			OHD	N8R1X		2.58	0.44				15.66				
 		8XX Access Ten Digit Screening, Per 8XX No. Established W/O	 	 	55		 	2.00	0.44			1	10.00				
		POTS Translations	1	1	OHD			5.94	0.81	4.57	0.54		15.66		Ì		
 		8XX Access Ten Digit Screening, Per 8XX No. Established With	 	 	מויס	1	+	5.54	0.01	4.57	0.34	1	13.00	1	1	1	
		POTS Translations			OHD	N8FTX		5.94	0.81	4.57	0.54		15.66				
		8XX Access Ten Digit Screening, Customized Area of Service	1	<u> </u>	OUD	INOFIA		5.94	0.01	4.57	0.34		13.00				
		Per 8XX Number			OHD	N8FCX		2.58	4.00				45.00				
-			-	-	OHD	N8FCX		2.58	1.29				15.66				
		8XX Access Ten Digit Screening, Multiple InterLATA CXR			OLID	NOTAN		0.00	4.70				45.00				
		Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		3.02	1.73				15.66				
		8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		3.02	0.44				15.66				
		8XX Access Ten Digit Screening, Call Handling and Destination															
		Features			OHD	N8FDX		2.58					15.66				
		8XX Access Ten Digit Screening, w/ 8FL No. Delivery			OHD		0.000565										
		8XX Access Ten Digit Screening, w/ POTS No. Delivery			OHD		0.000565										
LINE IN	ORMA	TION DATA BASE ACCESS (LIDB)															
		LIDB Common Transport Per Query			OQT		0.00002										
		LIDB Validation Per Query			OQU		0.012002										
		LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		34.32	42.08				15.66				
SIGNAL																	
		CCS7 Signaling Connection, Per 56Kbps Facility					15.46	35.53	35.53	16.44	16.44		15.66				
		CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	130.83										
		CCS7 Signaling Usage, Per Call Setup Message					0.0000142										
		CCS7 Signaling Usage, Per TCAP Message			UDB		0.0000569										
		CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	15.46	35.53	35.53	16.44	16.44		15.66				
		CCS7 Signaling Connection, Per link (B link) (also known as D					İ			İ							
		link)			UDB	TPP++	15.46	35.53	35.53	16.44	16.44		15.66				
		CCS7 Signaling Usage, Per ISUP Message			UDB	1	0.0000142										
		CCS7 Signaling Usage Surrogate, per link per LATA		1	UDB	STU56	650.33										
		CCS7 Signaling Point Code, per Originating Point Code								i i							
		Establishment or Change, per STP affected	1	1	UDB	CCAPO		29.01	29.01	35.57	35.57		15.66		Ì		
E911 SE	RVICE	and a second of the second of	1	1	-					22.01	22.01	1	12.30		1		
T		Local Channel - Dedicated - 2-wr Voice Grade		 		1	13.97	193.10	33.17	36.64	3.20	1	15.66		†		
+		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile				1	0.008838	.55.10	55.17	33.04	0.20	†	.0.50		 		
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility		 		1	3.300000					1	1		†		
		Termination	1	1			21.13	40.54	27.41	16.74	6.90		15.66		Ì		
 		Local Channel - Dedicated - DS1 - Zone 1	 	 		+	35.76	177.47	153.72	22.19	15.26	1	15.66				
 		Local Channel - Dedicated - DS1 - Zone 2	 	 		1	49.98	177.47	153.72	22.19	15.26	1	15.66	1	1	1	
 		Local Channel - Dedicated - DS1 - Zone 3		 		1	107.63	177.47	153.72	22.19	15.26	1	15.66		1		
 		Interoffice Transport - Dedicated - DS1 Per Mile	 	1		+	0.18	1//.4/	100.72	22.19	15.26	 	13.00	1	 	-	
\vdash		interonice transport - Dedicated - DST Per Mile	-	-		1	0.18								-	-	
		Intereffice Transport Dedicated DC4 Des Facility Transferre	1	1			00.40	00.07	04.04	40.05	44.44		45.00		Ì		
CALLE	O NIARE	Interoffice Transport - Dedicated - DS1 Per Facility Termination	 	1		1	60.16	89.27	81.81	16.35	14.44	-	15.66	-	1	-	
CALLING	J NAM	E (CNAM) SERVICE			001/			20.65		04.11		1					
\vdash		CNAM For DB Owners - Service Establishment	<u> </u>	<u> </u>	OQV	1		22.95		21.11				1		1	
\vdash		CNAM For Non DB Owners - Service Establishment	ļ		OQV	-		22.95		21.11		ļ					
1 1		CNAM For DB Owners - Service Provisioning With Point Code	1	1	001				=00 - :				1		Ì		
		Establishment	1	1	OQV	1		990.88	732.84	268.93	197.74	Î.	ı	l	1	i	

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachi	nent: 2	Exhi	bit: C
CHECHEL	The state of the s										Svc Order	Svc Order	Incremental	Incremental		Incremental
												Submitted		Charge -	Charge -	Charge -
											Elec	Manually		Manual Svc	Manual Svc	
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc			RATES(\$)								
CATEGORI	NATE ELEMENTS	m	Zone	B03	0300			KAILS(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
			-				Managa		Name and a second to a	D:			000	D-4(A)		
			1			Rec	Nonrec		Nonrecurring					Rates(\$)		
L							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CNAM For Non DB Owners - Service Provisioning With Point															
	Code Establishment			OQV			342.33	245.14	275.25	197.74						
	CNAM for DB Owners, Per Query			OQV		0.000902										
	CNAM for Non DB Owners, Per Query			OQV		0.000902										
LNP Query S																
	LNP Charge Per query					0.000757										
	LNP Service Establishment Manual						12.52		11.51			15.66				
	LNP Service Provisioning with Point Code Establishment						593.49	303.20		197.74		15.66				
OPERATOR	CALL PROCESSING															
	Oper. Call Processing - Oper. Provided, Per Min Using BST		1													
	LIDB			1		1.20						l				
			+	-	+	1.20			 		}	 			 	-
	Oper. Call Processing - Oper. Provided, Per Min Using		1	İ		4.04									Ì	1
	Foreign LIDB		1		-	1.24					1					
	Oper. Call Processing - Fully Automated, per Call - Using BST		1	İ]]			1	
	LIDB					0.20										
	Oper. Call Processing - Fully Automated, per Call - Using											1			<u> </u>	_
	Foreign LIDB					0.20										
INWARD OP	ERATOR SERVICES															
	Inward Operator Services - Verification, Per Minute					1.15										
	Inward Operator Services - Verification and Emergency Interrupt															
	- Per Minute					1.15										
BRANDING -	OPERATOR CALL PROCESSING		1			1.10										
	ity based CLEC		1													
raciii	Recording of Custom Branded OA Announcement		1		CBAOS		7,000.00	7,000.00			1	15.66				
			1		CBAUS		7,000.00	7,000.00				15.00				
	Loading of Custom Branded OA Announcement per shelf/NAV															
	per OCN				CBAOL		500.00	500.00				15.66				
UNEF	CLEC															
	Recording of Custom Branded OA Announcement						7,000.00	7,000.00				15.66				
	Loading of Custom Branded OA Announcement per shelf/NAV															
	per OCN						500.00	500.00				15.66				
DIRECTORY	ASSISTANCE SERVICES															
DIRE	CTORY ASSISTANCE ACCESS SERVICE															
	Directory Assistance Access Service Calls, Charge Per Call					0.275										
DIRE	CTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (I	DACC)														
	Directory Assistance Call Completion Access Service (DACC),	1	1													
	Per Call Attempt					0.10										
NILIM	BER SERVICES INTERCEPT ACCESS SERVICE		+	-		0.10					1					
	ASSISTANCE SERVICES		1	-	+	 			 		1	 			-	-
	CTORY ASSISTANCE DATA BASE SERVICE (DADS)		1	 	+				 		 				ļ	-
DIKE			-	1	1						1					-
L	Directory Assistance Data Base Service Charge Per Listing		1		2222	0.04					1					.
	Directory Assistance Data Base Service, per month		1		DBSOF	150.00										
	DIRECTORY ASSISTANCE		1	ļ							Į					
Facil	ty Based CLEC															
	Recording and Provisioning of DA Custom Branded		1										-			
	Announcement		1	AMT	CBADA]	6,000.00	6,000.00				15.66			1	
	Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00				15.66				
UNEF	CLEC															
	Recording of DA Custom Branded Announcement			İ			3,000.00	3,000.00	i i		İ	15.66			İ	1
	Loading of DA Custom Branded Announcement per Switch per			1			.,	.,	1		l				1	1
	OCN			1			1.170.00	1,170.00				15.66				
SELECTIVE I			+	 	+	 	1,170.00	1,170.00	 		 	15.00			 	1
SELECTIVE	Selective Routing Per Unique Line Class Code Per Request Per		1	 	+	 					}				 	
			1	İ	USRCR		04.70	04.70	4444	4444		45.00			Ì	1
VIDTUAL CO	Switch		-	1	USKCK		84.70	84.70	14.11	14.11	1	15.66				-
VIRTUAL CO			<u> </u>	l	4							ļ				
	Virtual Collocation - Application Cost			AMTFS	EAF		1,205.26	1,205.26	0.51	0.51		15.66				
	Virtual Collocation - Cable Installation Cost, per cable		<u> </u>	AMTFS	ESPCX		859.71	859.71	22.49	22.49	ļ	15.66				ļ
		1	1	AMTFS	ESPVX	3.22			1		1	Ī	ı	ı	Ì	1
	Virtual Collocation - Floor Space, per sq. ft. Virtual Collocation - Power, per fused amp			AMTFS	ESPAX	7.83										

UNBUND	DLED NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Exhi	bit: C
CATEGOR		Interi m	Zone	BCS	usoc			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
\vdash	Virtual Collocation - Cable Support Structure, per entrance						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	cable			AMTFS	ESPSX	14.97										
				UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, AMTFS, UDL, UNCVX, UNCDX,												
	Virtual Collocation - 2-wire Cross Connects (loop)			UNCNX	UEAC2	0.03	12.30	11.80	6.03	5.44		15.66				
	Virtual Collocation - 4-wire Cross Connects (loop)			UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX AMTFS,UDL12,	UEAC4	0.05	12.39	11.87	6.39	5.73		15.66				
	Virtual Collocation - 2-Fiber Cross Connects			UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC2F	2.84	20.89	15.20	7.38	5.92		15.66				
				AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12,												
	Virtual Collocation - 4-Fiber Cross Connects			ULD48, UDF	CNC4F	5.69	25.55	19.86	9.71	8.25		15.66				
	Virtual collocation - Special Access & UNE, cross-connect per DS1			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL, UNLD1	CNC1X	1.11	22.03	15.93	6.40	5.79		15.66				
	Virtual collocation - Special Access & UNE, cross-connect per DS3			USL, ULC, AMTFS, U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	14.16	20.89	15.20	7.38	5.92		15.66				
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable															
igsquare	Support Structure, per linear foot			AMTFS	VE1CB	0.0026										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable			AMTFS	VE1CD	0.0038										
	Support Structure,per cable Virtual Collocation - Co-Carrier Cross Connects - Public Cable Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AMTFS	VE1CC		535.37					15.66				
	Cable Support Structure, per cable			AMTFS	VE1CE		535.37					15.66				
	Virtual Collocation Cable Records - per request			AMTFS	VE1BA		1,518.57	1,518.57	265.99	265.99		15.66				
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable record			AMTFS	VE1BB		653.83	653.83	378.24	378.24		15.66				
	Virtual Collocaiton Cable Records - VG/DS0 Cable, per each 100 pair			AMTFS	VE1BC		9.62	9.62	11.79	11.79		15.66				<u> </u>
	Virtual Collocation Cable Records - DS1, per T1TIE			AMTFS	VE1BD		4.50	4.50	5.52	5.52		15.66				
$\vdash \vdash$	Virtual Collocation Cable Records - DS3, per T3TIE			AMTFS	VE1BE		15.75	15.75	19.32	19.32	ļ	15.66				
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber records			AMTFS	VE1BF		168.97	168.97	154.25	154.25		15.66				
$\vdash \vdash$	Virtual collocation - Security Escort - Basic, per half hour	-		AMTES	SPTBX		16.93	10.73	ļ		<u> </u>	15.66				
$\vdash \vdash$	Virtual collocation - Security Escort - Overtime, per half hour Virtual collocation - Security Escort - Premium, per half hour	-		AMTFS AMTFS	SPTOX SPTPX		22.05 27.17	13.86 16.98	<u> </u>		<u> </u>	15.66 15.66				
$\vdash \vdash$	Virtual collocation - Security Escort - Premium, per nair nour Virtual collocation - Maintenance in CO - Basic, per half hour	+	-	AMTES	CTRLX		27.17	10.73	 		 	15.66				1
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		36.47	13.86				15.66				
	Virtual collocation - Maintenance in CO - Overtime, per hair noun			AMTFS	SPTPM		45.02	16.98				15.66				
VADTUAL C	COLLOCATION	1		, , 0	J. 11 IVI	1	-10.0Z	10.00				10.00				1

UNBUNDI	LED NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	bit: C
CATEGORY		Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge -		Incremental Charge -	Incremental Charge -
		1				_ 1	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		-
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-															
-	Wire Analog - Res Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-	1	ļ	UEPSR	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus			UEPSB	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN			UEPSX	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			OLFSX	VLINZ	0.03	12.30	11.00	0.03	3.44		13.00				
	ISDN			UEPTX	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1	1		UEPEX	VE1R4	0.05	12.39	11.87	6.39	5.44		15.66				
VIRTUAL CO	OLLOCATION	-		UEPEX	VE IK4	0.05	12.39	11.07	0.39	5.44		13.00				\vdash
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line															
	Splitting			UEPSR, UEPSB	VE1LS	0.03	12.30	11.80	6.03	5.44		15.66				
PHYSICAL (COLLOCATION IPhysical Collocation 2 Wire Cross Connects (Loop) for Line	-														\vdash
	Physical Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	PE1LS	0.03	12.30	11.80	6.03	5.44		15.66				
AIN SELECT	TIVE CARRIER ROUTING			OEI OIX, OEI OB	1 2 120	0.00	12.00	11.00	0.00	0.44		10.00				
	Regional Service Establishment			SRC	SRCEC		101,098.91		8,590.70			15.66				
	End Office Establishment			SRC	SRCEO		169.88	169.88	1.70	1.70		15.66				
	Query NRC, per query			SRC		0.002749										
AIN - BELLS	SOUTH AIN SMS ACCESS SERVICE															
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			A1N	CAMSE		39.44	39.44	40.69	40.69		15.66				
	AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		7.83	7.83	9.09	9.09		15.66				
	AIN SMS Access Service - Port Connection - ISDN Access	1		A1N	CAM1P		7.83	7.83	9.09	9.09		15.66				
	AIN SMS Access Service - User Identification Codes - Per User				0,		7.00	7.00	0.00	0.00		10.00				
	ID Code			A1N	CAMAU		35.00	35.00	27.06	27.06		15.66				
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			A1N	CAMRC		41.88	41.88	11.71	11.71		15.66				
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.002188										
—	AIN SMS Access Service - Session, Per Minute					0.59										
	AIN SMS Access Service - Company Performed Session, Per Minute					0.73										
AIN - BELLS	SOUTH AIN TOOLKIT SERVICE	-				0.73										
1	AIN Toolkit Service - Service Establishment Charge, Per State,	1	1		t				1							
	Initial Setup			CAM	BAPSC		39.44	39.44	40.69	40.69		15.66				
	AIN Toolkit Service - Training Session, Per Customer				BAPVX		4,202.17	4,202.17				15.66				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt				BAPTT		7.83	7.83	9.09	9.09		15.66				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		7.83	7.83	9.09	9.09		15.66				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		7.83	7.83	9.09	9.09		15.66				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				ВАРТО		34.47	34.47	14.36	14.36		15.66				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP				BAPTC		34.47	34.47	14.36	14.36		15.66				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code				BAPTF		34.47	34.47	14.36	14.36		15.66				
	AIN Toolkit Service - Query Charge, Per Query	1	 		DAPIF	0.05	34.47	34.47	14.30	14.30		10.00				\vdash
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.00582										
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes					0.00382										
	p	1	1	l	1	0.00			1		1	1	1	l		

UNRU	NDI F	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhib	nit: C
0.120			1									Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			l									Elec			Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						- ()			per Lor	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							_	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
		Subscription			CAM	BAPMS	10.17	7.83	7.83	5.50	5.50		15.66				
		AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
		Subscription			CAM	BAPLS	2.87	8.66	8.66				15.66				
		AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service															
		Subscription			CAM	BAPDS	7.39	7.83	7.83	5.50	5.50		15.66				
		AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit									0.00						
		Service Subscription			CAM	BAPES	0.10	8.66	8.66				15.66				
ENHAN	CED EX	(TENDED LINK (EELs)															
		New Density Zone 1 EELs are available in the following MSA	s: Orlan	do. FL	Miami, FL: Ft, Laud	derdale. FL: A	Atlanta, Ga: Ne	v Orleans, LA.									
		Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-						, =, . ,		1					İ		
		In all states, EEL network elements shown below also apply t					erted to UNE ra	tes. A Switch	As Is Charge a	pplies to curre	ntly combined	facilities co	onverted to	UNEs.(Non-re	curring rates	do not apply	.)
		In All States the EEL network elements apply to ordinarily co															'
	2-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)				.,		,						
		First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport			(+											
		Combination - Zone 1	l	1	UNCVX	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed	1	<u> </u>	J. 1.5 V/) _ / \L_L	14.50	00.00	33.00	77.27	,.44		10.00				
		Transport Combination - Zone 2	l	2	UNCVX	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		-	ONOVA	OL7 ILE	22.00	00.00	00.00	77.27	7		10.00				
		Transport Combination - Zone 3		3	UNCVX	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66				
		Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	ONOVA	OLALZ	30.14	00.00	33.00	77.27	7.44		13.00				
		per month			UNC1X	1L5XX	0.18										
		Interoffice Transport - Dedicated - DS1 combination - Facility			ONOTA	TEO/O	0.10										
		Termination per month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				
		DS1 Channelization System Per Month			UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66				
		Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.56	6.58	4.72	10.34	5.15		15.66				
		Each Additional 2-Wire VG Loop(SL 2) in the same DS1			UNCVA	IDIVG	0.30	0.30	4.72	†			13.00				
		Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66				
-		Each Additional 2-Wire VG Loop(SL2) in the same DS1			UNCVX	ULALZ	14.30	88.00	33.00	47.24	7.44	-	13.00				
		Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66				
-		Each Additional 2-Wire VG Loop(SL2) in the same DS1			UNCVA	UEALZ	22.00	00.00	55.00	41.24	7.44	-	13.00				
		Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66				
		Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVA	UEALZ	30.14	00.00	55.00	41.24	7.44		15.66				
		per month			UNCVX	1D1VG	0.56	6.58	4.72				15.66				
					UNCVA	IDIVG	0.36	0.30	4.72				13.66				
		Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	l	1	UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66		Ì		
—	1-WIDE	IIS CHARGE EVOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROE	ICE TO		UNCCC		5.59	5.59	0.98	0.98	1	13.00				
\vdash	+-vviKE	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	LKUFF	ICE IK	ANOFURI (EEL)	 				 							
		Transport Combination - Zone 1	l	4	UNCVX	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66				
—		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	1	- '	OI NO V A	ULAL4	25.54	131.97	₹.51	35.14	14.50	1	15.00				
		Transport Combination - Zone 2	l	2	UNCVX	UEAL4	38.58	131.97	94.51	59.14	14.50		15.66		Ì		
—		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	1		OIVOVA	ULAL4	30.38	131.9/	94.31	J9.14	14.30	 	13.00	1	1		
		Transport Combination - Zone 3	l	3	UNCVX	UEAL4	60.02	131.97	94.51	59.14	14.50		15.66		Ì		
—		Interoffice Transport - Dedicated - DS1 combination - Per Mile	1	- 3	0110 1/	OLAL4	00.02	151.51	34.31	55.14	17.50		10.00		 		
		Per Month	l	1	UNC1X	1L5XX	0.18			I			15.66		Ì		
\vdash		Interoffice Transport - Dedicated - DS1 - Facility Termination Per	1	 	011017	LUAA	0.10			 			10.00		 		
		Month	l	1	UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66		Ì		
\vdash		Channelization - Channel System DS1 to DS0 combination Per	1	 	5JA	31111	00.10	03.21	01.01	10.33	17.44		10.00		 		
		Month	l	1	UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66		Ì		
\vdash		Voice Grade COCI - DS1 to DS0 Channel System combination -	1	 	5JA	.7104 1	107.19	31.04	02.37	10.34	3.75		10.00		 		
		per month	l		UNCVX	1D1VG	0.56	6.58	4.72	1			15.66				
—		Additional 4-Wire Analog Voice Grade Loop in same DS1	1	1	J	1.5140	0.50	0.00	7.12	 			10.00				
		Interoffice Transport Combination - Zone 1	l	1	UNCVX	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66				
—		Additional 4-Wire Analog Voice Grade Loop in same DS1	1	- '-	011017	JLAL4	20.04	131.31	₹.31	39.14	14.30	 	13.00	1	1		
		Interoffice Transport Combination - Zone 2	l	2	UNCVX	UEAL4	38.58	131.97	94.51	59.14	14.50		15.66		Ì		
		Additional 4-Wire Analog Voice Grade Loop in same DS1	1		OIVOVA	ULAL4	30.38	131.9/	94.31	J9.14	14.30	 	13.00	1	1		
		Interoffice Transport Combination - Zone 3	l	3	UNCVX	UEAL4	60.02	131.97	94.51	59.14	14.50		15.66				
		Voice Grade COCI - DS1 to DS0 Channel System combination -	1	-	011017	JLAL4	00.02	131.31	₹.31	39.14	14.30	 	13.00	1	1		
		per month	l		UNCVX	1D1VG	0.56	6.58	4.72	1			15.66				
ullet		per monu	L	1	OINOVA	טיוטו	0.36	0.00	4.72	1		1	15.00	l			

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental		Incremental Charge -	
						Rec	Nonrec	urring	Nonrecurring	Disconnect		ı	oss	Rates(\$)	l.	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Currently Combined Network Elements Switch -As-	1		LINGAV	LINICOC		5 50	5.50	0.00	0.00		45.00				
4-WID	Is Charge E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	EEICE	UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				-
7-7711	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	, i ioc	TRANSFORT (LLL)												
	Transport Combination - Zone 1		1	UNCDX	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 2		2	UNCDX	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	37.88	126.27	88.80	59.14	14.50		15.66				
+	Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCDX	UDL36	37.00	120.27	00.00	59.14	14.50		15.00				
	Per Month			UNC1X	1L5XX	0.18										
	Interoffice Transport - Dedicated - DS1 - combination Facility															
	Termination Per Month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				
	Channelization - Channel System DS1 to DS0 combination Per			LINICAV	MQ1	407.40	04.04	CO 57	40.54	0.70		45.00				
+	OCU-DP COCI (data) - DS1 to DS0 Channel System - per			UNC1X	IVIQ1	107.19	91.04	62.57	10.54	9.79		15.66				
	month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.58	4.72				15.66				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1								=0.44							
	Interoffice Transport Combination - Zone 2 Additional 4-Wire 56Kbps Digital Grade Loopin same DS1	1	2	UNCDX	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	37.88	126.27	88.80	59.14	14.50		15.66				
	OCU-DP COCI (data) - DS1 to DS0 Channel System -		Ŭ	ONODA	ODLOG	07.00	120.27	00.00	00.14	14.00		10.00				
	combination per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.58	4.72				15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-	-														
	Is Charge			UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WIR	E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	OFFICE	TRANSPORT (EEL)												1
	Transport Combination - Zone 1		1	UNCDX	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		·	0.1027	00201	20.00	120.21	00.00	00.11	1 1100		10.00				
	Transport Combination - Zone 2		2	UNCDX	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		_													
	Transport Combination - Zone 3	ļ	3	UNCDX	UDL64	37.88	126.27	88.80	59.14	14.50		15.66				ļ
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.18										
	Interoffice Transport - Dedicated - DS1 combination - Facility			ONOTA	TESTON	0.10										
	Termination Per Month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				
	Channelization - Channel System DS1 to DS0 combination Per															
	Month	ļ		UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66				ļ
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.58	4.72				15.66				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1			OHODA	10100	1.19	0.36	7.12				10.00				
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1						-									
 	Interoffice Transport Combination - Zone 2	<u> </u>	2	UNCDX	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	37.88	126.27	88.80	59.14	14.50		15.66				1
	OCU-DP COCI (data) - DS1 to DS0 Channel System		3	5ODA	35204	57.00	120.21	00.00	33.14	17.30		10.00				
I	combination - per month (2.4-64kbs)	L		UNCDX	1D1DD	1.19	6.58	4.72			<u></u>	15.66		<u> </u>	<u> </u>	<u> </u>
	Nonrecurring Currently Combined Network Elements Switch -As-															
4 14/15	Is Charge		OF TO 4	UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTI 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	EKUFFI	CE IKA	MINOPURI (EEL)	-				-							<u> </u>
	Transport - Zone 1		1	UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice					02.00	202	101.04				.0.00				
	Transport - Zone 2		2	UNC1X	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice		_	LINGAV	LICL VV	04450	050.47	457.51	44 70			45.00				1
	Transport - Zone 3	<u> </u>	3	UNC1X	USLXX	314.52	252.47	157.54	44.70	11.71	<u> </u>	15.66		<u> </u>	<u> </u>	<u> </u>

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile					1111	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Per Month			UNC1X	1L5XX	0.18										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTI	EROFFI	CE TRA		UNCCC		5.59	5.59	0.90	0.90		15.00				+
	First DS1Loop in DS3 Interoffice Transport Combination - Zone			•												
	1		1	UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone					ĺ										
	Intereffice Transport Dedicated DC2		3	UNC1X	USLXX	314.52	252.47	157.54	44.70	11.71		15.66				ļ
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month			UNC3X	1L5XX	4.09										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per															
	month			UNC3X	U1TF3	703.52	278.75	162.76	60.20	58.46		15.66				
	DS3 to DS1 Channel System combination per month DS3 Interface Unit (DS1 COCI) combination per month			UNC3X UNC1X	MQ3 UC1D1	176.20 13.47	178.14 6.58	93.97 4.72	33.26	31.83		15.66 15.66				<u> </u>
	Additional DS1Loop in DS3 Interoffice Transport Combination -			UNCIA	UCIDI	13.47	6.56	4.72				15.00				
	Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination -		1	UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				
	Zone 2		2	UNC1X	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				
	Additional DS1Loop in DS3 Interoffice Transport Combination -		3	UNC1X	USLXX	314.52	050.47	457.54	44.70	11.71		45.00				
	Zone 3 DS3 Interface Unit (DS1 COCI) combination per month		3	UNC1X UNC1X	UC1D1	13.47	252.47 6.58	157.54 4.72	44.70	11.71		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC3X	UNCCC		5.59	5.59	6.98	6.98		15.66				
2-WIR	E VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE IN	EROFF	ICE TR	ANSPORT (EEL)												
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per		3	UNCVA	UEALZ	30.14	00.00	55.00	47.24	7.44		15.00				
	Mile Per Month			UNCVX	1L5XX	0.008838										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV2	21.13	40.54	27.41	16.74	6.90		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	•		UNCVX	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WIR	E VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE IN	EROFF	ICE TR		014000		3.33	5.59	0.90	0.90		13.00				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	38.58	131.97	94.51	59.14	14.50		15.66				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	60.02	131.97	94.51	59.14	14.50		15.66				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Month			UNCVX	1L5XX	0.008838										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV4	18.73	40.54	27.41	16.74	6.90		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCVX	UNCCC		5.59	5.59	6.98	6.98		15.66				
DS3 D	IGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRAI	NSPOR	T (EEL)												
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month			UNC3X	1L5ND	8.89										
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month			UNC3X	UE3PX	327.71	451.52	263.94	119.49	83.58		15.66				

CATEGORY RATE ELEMENTS Interi m Zone BCS USOC RATES(\$) Submit Elect per L: Rec Nonrecurring Nonrecurring Disconnect Rec First Add'l First Add'l SOME Interoffice Transport - Dedicated - DS3 - Per Mile per month UNC3X 1L5XX 4.09	SR per LSR	d Charge - Manual Svo Order vs. Electronic- 1st	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs.
Rec First Add'l SOME Interoffice Transport - Dedicated - DS3 - Per Mile per month UNC3X 1L5XX 4.09	EC SOMAN	OSS		Disc 1st	Electronic- Disc Add'l
Interoffice Transport - Dedicated - DS3 - Per Mile per month UNC3X 1L5XX 4.09	EC SOMAN		S Rates(\$)	001141	
		SOMAN	SOMAN	SOMAN	SOMAN
Interoffice Transport - Dedicated - DS3 combination - Facility		+	+		
Termination per per month	15.66	;			
Nonrecurring Currently Combined Network Elements Switch -As-					
Is Charge	15.66	i			_
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL) High Capacity Unbundled Local Loop - STS1 combination - Per		-	-		
Mile per month					
High Capacity Unbundled Local Loop - STS1 combination -					1
Facility Termination per month UNCSX UDLS1 339.21 451.52 263.94 119.49 83.58	15.66	5			
Interoffice Transport - Dedicated - STS1 combination - Per Mile					
per month UNCSX 1L5XX 4.09 Interoffice Transport - Dedicated - STS1 combination - Facility		-	-		
	15.66	;			
Nonrecurring Currently Combined Network Elements Switch -As-	10.00				1
Is Charge	15.66	i			
2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)					<u> </u>
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 1 UNCNX U1L2X 21.88 117.24 79.77 52.88 10.54	15.66				
First 2-Wire ISDN Loop in a DS1 Interoffice Combination	13.00	'			+
Transport - Zone 2 2 UNCNX U1L2X 32.85 117.24 79.77 52.88 10.54	15.66	;			
First 2-Wire ISDN Loop in a DS1 Interoffice Combination					
Transport - Zone 3 3 UNCNX U1L2X 48.55 117.24 79.77 52.88 10.54	15.66	;			<u> </u>
Interoffice Transport - Dedicated - DS1 combination - Per Mile UNC1X 1L5XX 0.18 Interoffice Transport - Dedicated - DS1 combination - Facility					1
	15.66	,			
Channelization - Channel System DS1 to DS0 combination -	10.00				1
per month	15.66	;			
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System					
combination - per month	15.66	j	-		
Administrate 24 wire 15Dt Loop in same D3 interiorine transport	15.66	;			
Additional 2-wire ISDN Loop in same DS1Interoffice Transport	10.00				1
Combination - Zone 2 2 UNCNX U1L2X 32.85 117.24 79.77 52.88 10.54	15.66	;			
Additional 2-wire ISDN Loop in same DS1Interoffice Transport					
Combination - Zone 3	15.66	j	-		
2-wire ison-coordigatile) - UST to DSt Challing System UNCNX UC1CA 2.56 6.58 4.72					
Nonrecurring Currently Combined Network Elements Switch -As-					1
Is Charge	15.66	;			ļ
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL)			ļ		.
First DS1 Loop in STS1 Interoffice Transport Combination -	15.66				
Zottle 1 ONCIA 62.39 232.47 137.34 44.70 11.71 First DS1 Loop in STS1 Interoffice Transport Combination -	13.00	+	+		+
Zone 2 2 UNC1X USLXX 154.18 252.47 157.54 44.70 11.71	15.66	;			
First DS1 Loop in STS1 Interoffice Transport Combination -					
Zone 3 JUNC1X USLXX 314.52 252.47 157.54 44.70 11.71	15.66	5			<u> </u>
Interoffice Transport - Dedicated - STS1 combination - Per Mile					
Interoffice Transport - Dedicated - STS1 combination - Facility		+	+		†
Termination UNCSX U1TFS 701.37 278.75 162.76 60.20 58.46	15.66				
STS1 to DS1 Channel System conbination per month UNCSX MQ3 176.20 178.14 93.97 33.26 31.83	15.66	5			
DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 13.47 6.58 4.72					<u> </u>
Additional DS1Loop in STS1 Interoffice Transport Combination -	15.66				
Additional DS1Loop in STS1 Interoffice Transport Combination -	13.00		1		†
Zone 2 2 UNC1X USLXX 154.18 252.47 157.54 44.70 11.71	15.66	<u>; </u>		<u> </u>	
Additional DS1Loop in STS1 Interoffice Transport Combination -					
Zone 3 3 UNC1X USLXX 314.52 252.47 157.54 44.70 11.71	15.66	· I	I .		

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachr	ment: 2	Exhil	oit: C
0.1.201.22											Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									po. 2011	po. 2011	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													151	Add I	DISC ISL	DISC Add I
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		-
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.47	6.58	4.72								
	Nonrecurring Currently Combined Network Elements Switch -As-	-														
	Is Charge			UNCSX	UNCCC		5.59	5.59	6.98	6.98		15.66				ł
4-WI	RE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE T	RANS	PORT (EEL)												
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															ł
	Combination - Zone 1		1	UNCDX	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				1
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															1
	Combination - Zone 2		2	UNCDX	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				1
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															ł
	Combination - Zone 3		3	UNCDX	UDL56	37.88	126.27	88.80	59.14	14.50		15.66				ļ
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -		1						_							i
	Per Mile			UNCDX	1L5XX	0.008838										ļ
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -				I				1							i
	Facility Termination		<u> </u>	UNCDX	U1TD5	15.12	40.54	27.41	16.74	6.90		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-	1	1		1				I							i
	Is Charge			UNCDX	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WI	RE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERO	FFICE T	RANS	PORT (EEL)												
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport															1
	Combination - Zone 1		1	UNCDX	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				ļ
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		_													1
	Combination - Zone 2		2	UNCDX	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		_													ł
	Combination - Zone 3		3	UNCDX	UDL64	37.88	126.27	88.80	59.14	14.50		15.66				
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															1
	Per Mile			UNCDX	1L5XX	0.008838										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -			LINICDY	LIATEC	45.40	40.54	07.44	40.74	0.00		45.00				1
	Facility Termination Nonrecurring Currently Combined Network Elements Switch -As-	-	-	UNCDX	U1TD6	15.12	40.54	27.41	16.74	6.90		15.66				
	Is Charge	1		UNCDX	UNCCC		5.59	5.59	6.98	6.98		15.66				1
ADDITIONAL	. NETWORK ELEMENTS			UNCDA	UNCCC		5.59	5.59	0.90	0.90		15.66				-
	n used as a part of a currently combined facility, the non-recuri	ma cha	race do	not apply but a S	witch As Is a	hargo doos ann	N.				1					-
	n used as a part of a currently combined facility, the hori-recurr								†							
	ecurring Currently Combined Network Elements "Switch As Is"					A3 I3 Charge C	does not.		†							
1	Nonrecurring Currently Combined Network Elements Switch -As-		(00	pp.ioo to odoi: ooiii	1						1					1
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		5.59	5.59	6.98	6.98		15.66				ł
	Nonrecurring Currently Combined Network Elements Switch -As-		1		3550		0.03	0.00	5.50	0.90		10.00		1	1	
	Is Charge - 56/64 kbps	1	1	UNCDX	UNCCC		5.59	5.59	6.98	6.98		15.66				i
	Nonrecurring Currently Combined Network Elements Switch -As-	 	†				2.00	2.00	2.00	2.00						
	Is Charge - DS1	1	1	UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				i
	Nonrecurring Currently Combined Network Elements Switch -As-		1		1									İ	İ	ſ
	Is Charge - DS3	1	1	UNC3X	UNCCC		5.59	5.59	6.98	6.98		15.66				i
	Nonrecurring Currently Combined Network Elements Switch -As-															ĺ
	Is Charge - STS1	1	1	UNCSX	UNCCC		5.59	5.59	6.98	6.98		15.66				i
NOT	E: Local Channel - Dedicated Transport - minimum billing perior	d - Belo	w DS3:	one month, DS3 ar	nd above=fou	r months										ĺ
	Local Channel - Dedicated - 2-Wire Voice Grade			UNCXV	ULDV2	13.97	193.10	33.17	36.64	3.20		15.66				
	Local Channel - Dedicated - 4-Wire Voice Grade			UNCXV	ULDV4	14.93	193.53	33.60	37.11	3.67		15.66				
	Local Channel - Dedicated - DS1 per month Zone 1		1	UNC1X	ULDF1	35.76	177.47	153.72	22.19	15.26		15.66				
	Local Channel - Dedicated -DS1 Per Month Zone 2		2	UNC1X	ULDF1	49.98	177.47	153.72	22.19	15.26		15.66				<u> </u>
	Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	107.63	177.47	153.72	22.19	15.26		15.66				
	Local Channel - Dedicated - DS3 - Per Mile per month		<u> </u>	UNC3X	1L5NC	6.92			1							
	Local Channel - Dedicated - DS3 - Facility Termination	ļ		UNC3X	ULDF3	416.54	451.52	263.94	119.49	83.58		15.66				
	Local Channel - Dedicated - STS-1- Per Mile per month	<u> </u>	<u> </u>	UNCSX	1L5NC	6.92			L							
	Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	408.49	451.52	263.94	119.49	83.58		15.66				
	onal Features & Functions:	ļ			ļ											
MUL	TIPLEXERS	<u> </u>	<u> </u>	LIVEDA	1404	404.00	04.01	00 ==	40 = 1	0 =0		45.00				
	Channelization - DS1 to DS0 Channel System	 	-	UXTD1	MQ1	101.06	91.04	62.57	10.54	9.79	-	15.66			-	
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per			UDL	1D1DD	1.12	6.58	4.72	1			15.66				i
<u> </u>	month (2.4-64kbs)	<u> </u>	<u> </u>	UDL	טטוטון	1.12	86.0	4.72	<u> </u>		l	10.00		l	I	L

UNBU	NDLE	D NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Fxhil	oit: C
CATEG		RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
							Rec	Nonrec		Nonrecurring		001150	001441		Rates(\$)	0011411	001441
		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		month			UDN	UC1CA	2.41	6.58	4.72				15.66				
		Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.53	6.58	4.72				15.66				
		DS3 to DS1 Channel System per month			UXTD3	MQ3	166.13	178.14	93.97	33.26	31.83		15.66				
		STS1 to DS1 Channel System per month			UXTS1	MQ3	166.13	178.14	93.97	33.26	31.83		15.66				
-		DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	12.70	6.58	4.72				15.66				
		DS3 Interface Unit (DS1 COCI) used with Local Channel per month			ULDD1	UC1D1	12.70	6.58	4.72				15.66				
		DS3 Interface Unit (DS1 COCI) used with Interoffice Channel			CLDD1	00101	12.70	0.00	7.72				10.00				
		per month			U1TD1	UC1D1	12.70	6.58	4.72				15.66				
		OCAL EXCHANGE SWITCHING(PORTS)															
		ge Ports				l	ا										
		Although the Port Rate includes all available features in GA, I	KY, LA	& IN, t	ne desired features	will need to b	be ordered usin	ig retail USOCs	i								
	Z-VVIRE	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.38	2.38	2.27	1.42	1.33		15.66				
		Excitating Forts - 2-vviile Arialog Line Fort- Nes.			OLI OK	OLITE	1.50	2.50	2.21	1.72	1.55		13.00				
		Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire VG unbundled AL extended local			HEDOD	UEPAR	1.38	2.20	2.27	1.42	1.33		45.00				
		dialing parity Port with Caller ID - Res. Exchange Ports - 2-Wire VG unbundled res, low usage line port			UEPSR	UEPAR	1.38	2.38	2.21	1.42	1.33		15.66				$\overline{}$
		with Caller ID (LUM)			UEPSR	UEPAP	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire VG Alabama Residence Dialing Plan															
		without Caller Id			UEPSR	UEPWA	1.38	2.38	2.27	1.42	1.33		15.66				<u> </u>
		2-Wire voice unbundled Low Usage Line Port without Caller ID															
		Capability			UEPSR	UEPRT	1.38	2.38	2.27	1.42	1.33		15.66				ļ
-	FEATU	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00				15.66				
	FLATO	All Available Vertical Features			UEPSR	UEPVF	1.98	0.00	0.00				15.66				
	2-WIRE	VOICE GRADE LINE PORT RATES (BUS)			OLI OIK	OLI VI	1.50	0.00	0.00				10.00				
		Exchange Ports - 2-Wire Analog Line Port without Caller ID -															
		Bus			UEPSB	UEPBL	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire VG unbundled Line Port with															, l
-		unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.38	2.38	2.27	1.42	1.33		15.66				, !
		Exchange Ports - 2-Wire VG unbundled AL extended local			OLI OD	OLI DO	1.00	2.00	2.21	1.42	1.00		10.00				
		dialing parity Port with Caller ID - Bus.		<u>L</u>	UEPSB	UEPAW	1.38	2.38	2.27	1.42	1.33	<u> </u>	15.66				
		Exhange Ports - 2-Wire VG unbundled incoming only port with							· · · · · · · · · · · · · · · · · · ·								
$\vdash \vdash$		Caller ID - Bus			UEPSB	UEPB1	1.38	2.38	2.27	1.42	1.33	ļ	15.66				
		Exchange Ports - 2-Wire Voice Alabama Business Dialing Plan without Caller ID			UEPSB	UEPWB	1.38	2.38	2.27	1.42	1.33		15.66				1
\vdash		2-Wire voice unbundled Incoming Only Port without Caller ID			OLI OD	OLFWD	1.30	2.30	2.21	1.42	1.33		13.00				$\overline{}$
		Capability			UEPSB	UEPBE	1.38	2.38	2.27	1.42	1.33		15.66				l
		Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00		50		15.66				
	FEATU								•		•						
<u> </u>	-vo::-	All Available Vertical Features		ļ	UEPSB	UEPVF	1.98	0.00	0.00			<u> </u>	15.66				
\vdash	EXCHA	NGE PORT RATES (DID & PBX) 2-Wire VG Unbundled 2-Way PBX Trunk - Res		-	UEPSE	UEPRD	1.38	31.27	14.85	13.94	0.90	 	15.66				
\vdash		2-Wire VG Unbundled 2-Way PBX Trunk - Res 2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSE UEPSP	UEPRD	1.38	31.27	14.85	13.94	0.90		15.66				
		2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.38	31.27	14.85	13.94	0.90		15.66				
		2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.38	31.27	14.85	13.94	0.90		15.66				
		2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.38	31.27	14.85	13.94	0.90		15.66				
		2-Wire Voice Unbundled 2-Way PBX Alabama Calling Port			UEPSP	UEPA2	1.38	31.27	14.85	13.94	0.90		15.66				
\vdash		2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.38	31.27	14.85	13.94	0.90		15.66				
\vdash		2-Wire Vice Unbundled 2-Way PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP UEPSP	UEPXA UEPXB	1.38 1.38	31.27 31.27	14.85 14.85	13.94 13.94	0.90 0.90	<u> </u>	15.66 15.66				
\vdash		2-Wire Voice Unbundled PBX I'dli Terminal Hoter Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.38	31.27	14.85	13.94	0.90	 	15.66				
					-												

UNRUNDI	ED NETWORK ELEMENTS - Alabama												Attach	ment: 2	Fyhil	bit: C
ONBONDI	LED NEI WORK ELEMENTO - Alabama										Submitted	Submitted	Incremental Charge -	Incremental Charge -	Incremental Charge -	Incremental Charge -
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Elec per LSR	Manually per LSR	Manual Svc Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Manual Svc Order vs. Electronic- Disc 1st	Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred			Disconnect				Rates(\$)		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	<u> </u>		UEPSP	UEPXD	1.38	First 31.27	Add'I 14.85	First 13.94	Add'l 0.90	SOMEC	SOMAN 15.66	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPSP	UEPAD	1.30	31.21	14.00	13.94	0.90		13.00				
	Capable Port			UEPSP	UEPXE	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy						-									
	Administrative Calling Port			UEPSP	UEPXL	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port	ļ		UEPSP	UEPXM	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPSP	UEPXO	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.38	31.27	14.85	13.94	0.90		15.66				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00	10.01	0.00		15.66				
FEA	TURES															
	All Available Vertical Features			UEPSP UEPSE	UEPVF	1.98	0.00	0.00		-		15.66			·	
EXC	HANGE PORT RATES (COIN)					1.00						4= 00				
NOT	Exchange Ports - Coin Port E: Transmission/usage charges associated with POTS circuit so	witched		will also apply to a	iravit awitah	1.38	2.38	2.27	1.42	1.33		15.66	20110			
	E: Access to B Channel or D Channel Packet capabilities will be													s Request Pro	CASS	
	D LOCAL EXCHANGE SWITCHING(PORTS)	I	010	J Iniough Britiness	Dusiness ite	quest i rocess.	reaces for the	packet capabi	lities will be de	terrimica via t	T Dona i ic	ic requesti	Dusines.	Requestivo	0000.	
	HANGE PORT RATES															
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.05	119.31	18.74	59.90	3.76		15.66				
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID															
	capability			UEPDD	UEPDD	60.09	202.02	95.69	72.59	2.46		15.66				
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered			UEPTX UEPSX UEPTX UEPSX	U1PMA UEPVF	9.79 1.98	72.77 0.00	52.99 0.00	47.79	10.74		15.66				
NOT	E: Transmission/usage charges associated with POTS circuit so	witched	licado						ission by R-Ch	annels associ	isted with 2	wire ISDN r	orte			
i inot	E: Access to B Channel or D Channel Packet capabilities will be	e availal	ole onl	v through BFR/New	Business Re	eauest Process.	Rates for the	packet capabi	lities will be de	etermined via t	he Bona Fid	le Reauest/I	New Busines:	s Request Pro	cess.	
NOT	E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles	e availal	ole onl	UEPTX UEPSX	U1UMA	0.00	0.00	packet capabi 0.00	lities will be de	termined via t	he Bona Fid	le Request/l	New Busines	s Request Pro	cess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port		ole onl						79.18	20.06	he Bona Fid	le Request/I 15.66	New Busines	s Request Pro	cess.	
UNB	Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY	·	ole onl	UEPTX UEPSX	U1UMA	0.00	0.00	0.00			he Bona Fid		New Busines	s Request Pro	cess.	
UNB	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE	·	ole onl	UEPTX UEPSX UEPEX	U1UMA UEPEX	0.00 84.32	0.00 203.81	0.00 101.56	79.18	20.06	he Bona Fid	15.66	New Busines	s Request Pro	cess.	
UNB	Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY	·	ole onl	UEPTX UEPSX	U1UMA	0.00	0.00	0.00			he Bona Fid		New Busines	s Request Pro	cess.	
UNB	Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res	·	ole onl	UEPTX UEPSX UEPEX UEPVR	U1UMA UEPEX UERAC	0.00 84.32 1.38	0.00 203.81 2.38	0.00 101.56 2.27	79.18	20.06	he Bona Fid	15.66	New Busines	s Request Pro	cess.	
UNB	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE	·	ole onl	UEPTX UEPSX UEPEX	U1UMA UEPEX	0.00 84.32	0.00 203.81	0.00 101.56	79.18	20.06	he Bona Fid	15.66	New Busines	s Request Pro	cess.	
UNB	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR	U1UMA UEPEX UERAC UERLC	0.00 84.32 1.38	0.00 203.81 2.38 2.38	0.00 101.56 2.27 2.27	79.18 1.42 1.42	20.06 1.33 1.33		15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res -Recurring	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	U1UMA UEPEX UERAC UERLC UERLC UERTE	0.00 84.32 1.38 1.38	203.81 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27	79.18 1.42 1.42 1.42	20.06 1.33 1.33 1.33		15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion -	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERLC UERTE UERTR	0.00 84.32 1.38 1.38	2.38 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42	20.06 1.33 1.33 1.33		15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	U1UMA UEPEX UERAC UERLC UERLC UERTE	0.00 84.32 1.38 1.38	203.81 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27	79.18 1.42 1.42 1.42	20.06 1.33 1.33 1.33		15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERLC UERTE UERTR USAC2	0.00 84.32 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38	2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42	20.06 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNB	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERLC UERTE UERTR	0.00 84.32 1.38 1.38	2.38 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42	20.06 1.33 1.33 1.33		15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)	·		UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERLC UERTE UERTR USAC2	0.00 84.32 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38	2.27 2.27 2.27 2.27 2.27 0.10	79.18 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERLC UERTE UERTR USAC2	0.00 84.32 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38	2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42	20.06 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERAC UERTE UERTE UERTE UERTE UERTE UERTE UEACC USACC	0.00 84.32 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38 0.10 0.10	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10	79.18 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERTE UERTE UERTE UERTE UERTE UERTE UEACC USACC UERAC UERAC	0.00 84.32 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38 0.10 0.10	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10	79.18 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res- Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERTE UERTE UERTE UERTE UERTE UERTE UERTE UERTE UERTE USAC2 UERAC UERAC	0.00 84.32 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38 0.10 0.10	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27	79.18 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus	·	ole on!	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERTE UERTE UERTE UERTE UERTE UERTE UEACC USACC UERAC UERAC	0.00 84.32 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38 0.10 0.10	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10	79.18 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus	·	ole on!	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UERAC UERTE UERTE UERTE UERTE UERTE UERTE UERTE UERTE UERTE USAC2 UERAC UERAC	0.00 84.32 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38 0.10 0.10	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27	79.18 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
Non UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling	·	ole on!	UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB	UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC UERAC UERAC UERAC UERAC	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
Non UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling - Recurring Unbundled Remote Call Forwarding Service - Conversion -	·	ole on!	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB UEPVB UEPVB	UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC UERAC UERAC UERAC UERTE UERTE UERTE UERTE UERTE	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	Cess.	
Non UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port IUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY IUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res - Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) Unbundled Remote Call Forwarding Service - Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling - Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	·	ole on!	UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB	UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC UERAC UERAC UERAC UERAC	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
Non UNE	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	·	ole on!	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB UEPVB UEPVB	USAC2 UERAC USACC UERAC USACC UERAC UERTE USACC UERAC	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
Non Non Non	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	·	ole on!	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB UEPVB UEPVB	UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC UERAC UERAC UERAC UERTE UERTE UERTE UERTE UERTE	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
Non Non Non Non Non Non Non Non Non Non	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB UEPVB UEPVB	USAC2 UERAC USACC UERAC USACC UERAC UERTE USACC UERAC	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	Cess.	
Non Non Non Non Non Non Non Non Non Non	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) DLOCAL SWITCHING, PORT USAGE Office Switching Function, Per MOU	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB UEPVB UEPVB	USAC2 UERAC USACC UERAC USACC UERAC UERTE USACC UERAC	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Busines	s Request Pro	cess.	
Non Non Non Non Non Non Non Non	Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res- Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling -Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) DLOCAL SWITCHING, PORT USAGE Office Switching (Port Usage)	·	ole onl	UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB UEPVB UEPVB	USAC2 UERAC USACC UERAC USACC UERAC UERTE USACC UERAC	0.00 84.32 1.38 1.38 1.38 1.38 1.38 1.38 1.38	0.00 203.81 2.38 2.38 2.38 0.10 0.10 2.38 2.38 2.38 2.38	0.00 101.56 2.27 2.27 2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27 2.27	79.18 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	20.06 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66 15.66	New Business	s Request Pro	cess.	

UNBU	NDLE	D NETWORK ELEMENTS - Alabama												Attach	ment: 2	Exhi	bit: C
												Svc Order	Svc Order	Incremental			Incrementa
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
												Elec		Manual Svc			Manual Svo
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									po. 20.1	po. 2011	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																2.00 .01	2.007.441
							Rec	Nonre			Disconnect				Rates(\$)		
		T 1 0 11 1 5 11 5 10 10 11						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Tandem Switching Function Per MOU					0.000095										
	<u> </u>	Tandem Trunk Port - Shared, Per MOU					0.0002015										
	Commo	on Transport Common Transport - Per Mile, Per MOU				+	0.0000023										
		Common Transport - Fer Mile, Fer MOU Common Transport - Facilities Termination Per MOU				-	0.0000023										
LINDLIN	DI ED B	PORT/LOOP COMBINATIONS - COST BASED RATES				1	0.0003224					1					
		ased Rates are applied where BellSouth is required by FCC ar	nd/or St	ate Cor	mmission rule to nro	ovide Unbun	l dled Local Swi	tching or Swite	ch Ports								
		es shall apply to the Unbundled Port/Loop Combination - Cos								ed Port section	of this Rate F	yhihit					
		fice and Tandem Switching Usage and Common Transport Us											n Port/Loor	Combinatio	ns.		
		st and additional Port nonrecurring charges apply to Not Curr															
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	, 5 .						J goo ona							İ	
		ort/Loop Combination Rates				1	İ				l						
		2-Wire VG Loop/Port Combo - Zone 1		1		1	12.70				l						
		2-Wire VG Loop/Port Combo - Zone 2		2			21.19										
		2-Wire VG Loop/Port Combo - Zone 3		3			34.80										
	UNE Lo	oop Rates															
		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	11.55										
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	20.04										
		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	33.65										
	2-Wire	Voice Grade Line Port Rates (Res)															
		2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire voice Grade unbundled Alabama extended local dialing							40.00				4= 00				
		parity port with Caller ID - res			UEPRX	UEPAR	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire voice unbundles res, low usage line port with Caller ID			UEPRX	UEPAP	4.45	40.40	19.83	24.91	6.63		45.00				
		(LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan			UEPKX	UEPAP	1.15	40.19	19.83	24.91	0.03		15.66				
		without Caller ID			UEPRX	UEPWA											
		2-Wire voice unbundled Low Usage Line Port without Caller ID			OLFKA	OLFWA						1					
		Capability			UEPRX	UEPRT	1.15	40.19	19.83	24.91	6.63		15.66				
	FEATU				OLITON	OLI IXI	1.10	40.10	10.00	24.01	0.00		10.00				
		All Features Offered			UEPRX	UEPVF	1.98	0.00	0.00				15.66				
		NUMBER PORTABILITY						0.00	0.00								
		Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
		Switch-as-is			UEPRX	USAC2		0.10	0.10				15.66				
	ADDITI	ONAL NRCs															
		2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
		Activity			UEPRX	USAS2	0.00	0.00	0.00				15.66				
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
		ort/Loop Combination Rates															
		2-Wire VG Loop/Port Combo - Zone 1	ļ	1		ļ	12.70								ļ		
		2-Wire VG Loop/Port Combo - Zone 2	!	2		1	21.19									 	
-	IINIE I :	2-Wire VG Loop/Port Combo - Zone 3	 	3		1	34.80									-	-
-	UNE LO	oop Rates	 	1	UEPBX	UEPLX	11.55									-	1
		2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2	 		UEPBX	UEPLX	20.04						 		1	-	
-		2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	1		UEPBX	UEPLX	33.65						-			1	
	2-Wire	Voice Grade Line Port (Bus)	 	J	OLFDA	JLFLA	33.05								1	1	
 		2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.15	40.19	19.83	24.91	6.63		15.66		1	 	
		2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.15	40.19	19.83	24.91	6.63		15.66		1	 	
1		2-Wire voice unbundled port with Gallet + 12404 ib - bus	1		UEPBX	UEPBO	1.15	40.19	19.83	24.91	6.63	<u> </u>	15.66		1	 	
		2-Wire voice Grade unbundled Alabama extended local dialing	1			1 20	0			251	5.50		.0.00			1	
		parity port with Caller ID - bus	1		UEPBX	UEPAW	1.15	40.19	19.83	24.91	6.63		15.66			1	
		2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.15	40.19	19.83	24.91	6.63		15.66			İ	İ
		2-Wire Voice Unbundled Alabama Business Dialing Plan without				1			. ,,-				- · · ·			İ	
										24.91	6.63						

CATEGORY RATE ELEMENTS Interi m Zone BCS USOC BCS USOC RATES(\$) Svc Order	UNBU	NDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhib	oit: C
ATTECHN MATERIAN INTO A BOOK OF A BO													Svc Order	Svc Order				
AFEONY RAFE ELEMENTS IN BOOK STORM S													Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
AND Color				Interi									Elec	Manually			Manual Svc	Manual Svc
Section Sect	CATEG	ORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
Part													-	-	Electronic-	Electronic-	Electronic-	Electronic-
Part																		
COLA MURRER FORT AGENTY COLOR CO								1										
Particle Particle				1				Rec					COMEC	COMAN			COMAN	COMAN
Cock Walker Port Package February	-		2-Wire voice unbundled Incoming Only Port without Caller ID						FIISL	Auu i	FIISL	Add I	SOWIEC	SOWAN	SUMAN	SOWAN	SOWAN	SOWAN
LOCAL NUMBER FORTABLITY LOPEN LO						UEPBX	UEPBE	1.15	40.19	19.83	24.91	6.63		15.66				
		LOCAL				02. 5%	02. 02	0	10.10		2	0.00		10.00				
SECURES						UEPBX	LNPCX	0.35										
NONECURRING CHARGES (INCE) - CURRENTY COMMINDO		FEATU																
SVIVE Vision Clark Loop Life Plot Controllation - Conversacio - John Controllation - Conversacio - John Controllation - Cont						UEPBX	UEPVF	1.98	0.00	0.00				15.66				
South-work South-work South So		NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
ADDITIONAL RICE 2-Win Voice Grafue Loop Virth P-Write Like PORT (RES - PRX) 2-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 3-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 4-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 4-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 4-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 5-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 5-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 5-Win Voice Grafue Loop Virth 2-Wine Like PORT (RES - PRX) 5-Win Voice Grafue Like Port Resides (RES - PRX) 5-Win Voice Grafue Like Port Resides (RES - PRX) 5-Win Voice Grafue Like Port Resides (RES - PRX) 5-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 6-Win Voice Grafue Like Port Resides (RES - PRX) 7-Win Voice Grafue Like Port Resides (RES - PRX) 7-Win Voice Grafue Like Port Resides (RES - PRX) 7-Win Voice Grafue Like Port Resides (RES - PRX) 7-Win Voice Grafue Like Port Resides (RES - PRX) 7-Win Voice Grafue Like Port Resides (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES - PRX) 7-Win Voice Grafue Like (RES -																		
CPPS						UEPBX	USAC2		0.10	0.10				15.66				
Activity Colored Col		ADDITI																
2-Wink Voice GRADE LOP WITH ZWINE LINE PORT (RES - PBX)																		
Well Compose						UEPBX	USAS2		0.00	0.00				15.66				
2-Wire Vot LoopProt Corrob. Zeno 2 1 1 1 1 1 1 1 1 1																		
2-Wire VS LoopPer Corbo: Zene 2 2 2 2 1.19		UNE P			<u> </u>			10 =0										
WHE Loop Plates																		
UPPR UPPR							-				-							
2-Vive Voice Grade Loop (St. 1) - Zone 1		LIME L			3		-	34.80			-							
2-Wire Vote Grade Log (St. 1) - Zone 3		ONE LO			1	HEDDC	HEDLY	11.55										
2-Wire Votes Grade Line Protecting (St. 1) - Zone 3 3 UEPRG UEPK 33.65																		
E-Wire Vote Grade Line Port Rates (RES - PRX)	-										†							
Section Sect		2-Wire			3	OLITIO	OLI LX	33.03										
Res		_ ******																
LOCAL NUMBER PORTABILITY LUPRG LINPCP 3.15 0.00 0.00 15.66						UEPRG	UEPRD	1.15	69.08	32.41	37.43	6.20		15.66				
Coal Number Portability (1 per port)		LOCAL										0.20						
FATURES						UEPRG	LNPCP	3.15	0.00	0.00				15.66				
NONRÉCURRING CHARGES (NRCs) - CURRENTLY COMBINED																		
Conversion - Switch-As-Is			All Features Offered			UEPRG	UEPVF	1.98	0.00	0.00				15.66				
Conversion - Switch-As-is UEPRG USAC2 7.91 1.90 15.66		NONRE																
ADDITIONAL NRCS ADDITIONAL NRCS																		
2-Wire Voice Grade Loop (Ine Port Combination (PBX) - UEPRG USAS2 0.00 0.00 0.00 0.00 15.66						UEPRG	USAC2		7.91	1.90				15.66				
Subsequent Activity		ADDITI																
PRX Subsequent Activity - Change/Rearrange Multiline Hunt Group 15.66 15																		
Croup						UEPRG	USAS2	0.00	0.00	0.00				15.66				
2-Wire Voice Grade Loop (St. 1) - Zone 1				1							I		1	4		1		
UNE Port/Loop Combination Rates	<u> </u>	0 14/15-		!	<u> </u>		1		7.32	7.32	!			15.66		 		
2-Wire VG Loop/Port Combo - Zone 1				!	<u> </u>		1				!					 		
2-Wire VG Loop/Port Combo - Zone 2 2 21.19	\vdash	UNE P		1	4		+	10.70			 					 		
2-Wire Voice Grade Loop (St. 1) - Zone 1				1			1				 		1	-		1		
UNE Loop Rates	\vdash			1			+				 					 		
2-Wire Voice Grade Loop (SL 1) - Zone 1		UNFI		 			1	54.60			-							
2-Wire Voice Grade Loop (SL 1) - Zone 2 2 UEPPX UEPLX 20.04		U.112 EV		 	1	UEPPX	UEPLX	11.55			-							
2-Wire Voice Grade Loop (SL 1) - Zone 3 3 UEPPX UEPLX 33.65				1							I		 	 		 		
2-Wire Voice Grade Line Port Rates (BUS - PBX)				†	_						t					1		
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus UEPPX UEPPO 1.15 69.08 32.41 37.43 6.20 15.66 Line Side Unbundled Contward PBX Trunk Port - Bus UEPPX UEPPO 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled 1-Coming PBX Trunk Port - Bus UEPPX UEPP1 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled 2-Way Combination PBX Alabama Calling Port UEPPX UEPPX UEPPA 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Ports UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPX UE		2-Wire		1	<u> </u>		1	22.00			İ					İ		
Line Side Unbundled Outward PBX Trunk Port - Bus UEPPX UEPPO 1.15 69.08 32.41 37.43 6.20 15.66								i i										
Line Side Unbundled Outward PBX Trunk Port - Bus UEPPX UEPPO 1.15 69.08 32.41 37.43 6.20 15.66			Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	1		UEPPX	UEPPC	1.15	69.08	32.41	37.43	6.20	1	15.66		1		
Calling Port				Ì		UEPPX	UEPPO	1.15	69.08	32.41	37.43	6.20		15.66				
Calling Port						UEPPX	UEPP1	1.15	69.08	32.41	37.43	6.20		15.66				
2-Wire Voice Unbundled PBX LD Terminal Ports UEPPX UEPLD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port UEPXX UEPXX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPXX UEPXX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPXX UEPXX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPXX UEPXX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPXX UEPXX UEPXX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPXX U			2-Wire Voice Unbundled 2-Way Combination PBX Alabama															
2-Wire Voice Unbundled 2-Way Combination PBX Usage Port UEPX UEPXA 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPX UEPXB 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPX UEPX UEPX UEPX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPX UEPX UEPX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPX UEPX UEPX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD				<u></u>												<u> </u>		
2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPX U															_		_	
2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPX UEPX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPX UEPX UEPX 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32.41 37.43 6.20 15.66 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 1.15 69.08 32																		
2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPPX UEPXD 1.15 69.08 32.41 37.43 6.20 15.66 5.20 5.40 5.40 5.40 5.40 5.40 5.40 5.40 5.4				<u> </u>														
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD				ļ														
				ļ		UEPPX	UEPXD	1.15	69.08	32.41	37.43	6.20		15.66				
				1	1								1			Ì		
			Capable Port	1	1	UEPPX	UEPXE	1.15	69.08	32.41	37.43	6.20		15.66				

Admi 2-Wir Roon 2-Win Discc 2-Win LOCAL NUM LOCAL FEATURES AII F. NONRECURI 2-Wir Conv ADDITIONAL 2-Wire UNE POT/Lo 12-Wir 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 3-Wir	Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED Jire Voice Grade Loop/ Line Port Combination (PBX) - nversion - Switch-As-Is AL NRCs Jire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt sup	Interi	Zone	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXL UEPXM UEPXO UEPXS	Rec 1.15 1.15 1.15	Nonrec First 69.08	Add'I 32.41	Nonrecurring First 37.43	Disconnect Add'I		Submitted Manually per LSR	Manual Svc Order vs. Electronic- 1st	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
Admi 2-Wir Roon 2-Win Discc 2-Win LOCAL NUM LOCAL FEATURES AII F. NONRECURI 2-Wir Conv ADDITIONAL 2-Wire UNE POT/Lo 12-Wir 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 3-Wir	ministrative Calling Port //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy m Calling Port //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital count Room Calling Port //ire Voice Unbundled 1-Way Outgoing PBX Measured Port MBER PORTABILITY tal Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED //ire Voice Grade Loop/ Line Port Combination (PBX) -			UEPPX UEPPX UEPPX UEPPX	UEPXM UEPXO UEPXS	1.15 1.15 1.15	First 69.08	Add'I 32.41	First	Add'l	SOMEC				SOMAN	SOMAN
Admi 2-Wir Roon 2-Win Discc 2-Win LOCAL NUM LOCAL FEATURES AII F. NONRECURI 2-Wir Conv ADDITIONAL 2-Wire UNE POT/Lo 12-Wir 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 3-Wir	ministrative Calling Port //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy m Calling Port //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital count Room Calling Port //ire Voice Unbundled 1-Way Outgoing PBX Measured Port MBER PORTABILITY tal Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED //ire Voice Grade Loop/ Line Port Combination (PBX) -			UEPPX UEPPX UEPPX UEPPX	UEPXM UEPXO UEPXS	1.15 1.15 1.15	69.08	32.41			SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
Admi 2-Wir Roon 2-Win Discc 2-Win LOCAL NUM LOCAL FEATURES AII F. NONRECURI 2-Wir Conv ADDITIONAL 2-Wire UNE POT/Lo 12-Wir 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 3-Wir	ministrative Calling Port //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy m Calling Port //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital count Room Calling Port //ire Voice Unbundled 1-Way Outgoing PBX Measured Port MBER PORTABILITY tal Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED //ire Voice Grade Loop/ Line Port Combination (PBX) -			UEPPX UEPPX UEPPX UEPPX	UEPXM UEPXO UEPXS	1.15 1.15			37.43	6.20		45.00	'	Ì	ļ j	1
2-Wir Roon 2-Wir Roon 2-Wir Roon 1-Wir Roon 1-Wir Roon 1-Wir Roon 1-Wir Roon 2-Wir Roon 3-Wir Roon 3-Wir Roon 4-Wir	Vire Voice Unbundled 2-Way PBX Hotel/Hospital Economy om Calling Port Vire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital count Room Calling Port Vire Voice Unbundled 1-Way Outgoing PBX Measured Port MBER PORTABILITY al Number Portability (1 per port) Seatures Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop/ Line Port Combination (PBX) - nversion - Switch-As-Is AL NRCs Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Line Port Combination (PBX) - Sequent Activity Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vire Voice Combination Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vire Voice Combination Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vire Voice Combination Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vire Voice Combination Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vire Voice Combination Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vire Voice Grade Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vire Voice Grade Vire Voice Grade Vire Voice Grade Vire Voice Grade Vire Voice Grade Vire Voice Vire Voi			UEPPX UEPPX UEPPX UEPPX	UEPXM UEPXO UEPXS	1.15 1.15			37.43	6.20						
Roon 2-Win Discc 2-Win LOCAL NUM LOCAL NUM FEATURES All Fe NONRECURI 2-Win Conv ADDITIONAL 2-Win Subs PBX Grou UNE Port/Lo 2-Wine 2-Win	om Calling Port Vire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital count Room Calling Port Vire Voice Unbundled 1-Way Outgoing PBX Measured Port MBER PORTABILITY all Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop/ Line Port Combination (PBX) - nversion - Switch-As-Is AL NRCS Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt sup			UEPPX UEPPX UEPPX	UEPXO UEPXS	1.15	69.08			_		15.66			├	
Disac 2-Wire LOCAL NUM LOCAL NUM FEATURES All Fi NONRECURI 2-Wir Conv ADDITIONAL PBX Grou 2-WIRE VOIC UNE POrt/Lo 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 3-Wir	count Room Calling Port //ire Voice Unbundled 1-Way Outgoing PBX Measured Port MBER PORTABILITY ral Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED //ire Voice Grade Loop/ Line Port Combination (PBX) - nversion - Switch-As-Is AL NRCs //ire Voice Grade Loop/ Line Port Combination (PBX) - ssequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt sup			UEPPX UEPPX	UEPXS			32.41	37.43	6.20		15.66				I
2-Wire Voice 2-Wi	Vire Voice Unbundled 1-Way Outgoing PBX Measured Port MBER PORTABILITY all Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop/ Line Port Combination (PBX) - nversion - Switch-As-Is AL NRCS Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity Vire Voice Grade Loop/ Change/Rearrange Multiline Hunt Vip			UEPPX UEPPX	UEPXS		00.00	00.44	07.40	0.00		45.00				
LOCAL NUM Local FEATURES All Fe NONRECURI 2-Wir Conv ADDITIONAL 2-Wir Subs PBX Grou UNE Port/Lo 12-Wir 2-Wir	MBER PORTABILITY all Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop/ Line Port Combination (PBX) - neversion - Switch-As-Is AL NRCs Vire Voice Grade Loop/ Line Port Combination (PBX) - ssequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt sup			UEPPX		1.15	69.08 69.08	32.41 32.41	37.43 37.43	6.20 6.20		15.66 15.66		 		
Local	ral Number Portability (1 per port) S Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop/ Line Port Combination (PBX)					1.10	03.00	32.41	37.43	0.20		13.00	$\overline{}$	 	 	
All Fe NONRECURI 2-Wir Conv ADDITIONAL 2-Wir Subs PBX Grou UNE Port/Lo 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 2-Wire 3	Features Offered RRING CHARGES (NRCs) - CURRENTLY COMBINED Jire Voice Grade Loop/ Line Port Combination (PBX) - nversion - Switch-As-Is AL NRCs Jire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt sup			LIEPPX	LNPCP	3.15	0.00	0.00				15.66	i			
NONRECURI 2-Win Conv ADDITIONAL 2-Win Subs PBX Grou 2-WIRE VOIC UNE POrt/Lo 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 3-	RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop/ Line Port Combination (PBX) - neversion - Switch-As-Is AL NRCs Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt Jup			LIEPPX												1
2-Wir Conv ADDITIONAL 2-Wis Subs PBX Grou 2-WIRE VOIC UNE POrIL-D 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii 2-Wii	Vire Voice Grade Loop/ Line Port Combination (PBX) - nversion - Switch-As-Is AL NRCs Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt sup			OLI I A	UEPVF	1.98	0.00	0.00				15.66	ļ!			
Conv ADDITIONAL 2-Wire Subs PBX Grou 2-WIRE VOIC UNE Port/Lo 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir Bloch	nversion - Switch-As-Is AL NRCs Vire Voice Grade Loop/ Line Port Combination (PBX) - osequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt pup												<u>'</u>			
ADDITIONAL 2-Win Subs PBX Grou 2-WIRE VOIC UNE POrt/Lo 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 3-Win	AL NRCs Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt sup			UEPPX	USAC2		7.04	4.00				45.00	i '	j ,	i !	1
2-Wir Subs PBX Grou 2-WIRE VOIC UNE POrt/Lo 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 2-Wir 3-Wir 3-Wir 3-Wir 3-Wir 3-Wir 3-Wir 3-Wir 3-Wir 3-Wir	Vire Voice Grade Loop/ Line Port Combination (PBX) - sequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt sup		-	UEPPX	USAC2		7.91	1.90				15.66			 	
Subs PBX Grou 2-WIRE VOIC UNE POrt/Lo 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 3-Wi	osequent Activity X Subsequent Activity - Change/Rearrange Multiline Hunt oup				+									 		
2-WIRE VOIC UNE POT/Lo 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win Block	pup			UEPPX	USAS2	0.00	0.00	0.00				15.66	i '		i !	l
2-WIRE VOIC UNE POrt/Lo. 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 3-Win 3-Win 3-Win 3-Win 3-Win 3-Win 3-Win 3-Win 3-Win 3-Win 3-Win 3-Win							7.00	7.00				45.00				
UNE Port/Lo 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win 2-Win Block	ICE CDADE LOOD WITH 2 WIDE ANALOC LINE COIN DO	DT			-		7.32	7.32				15.66		├ ──	 	
2-Win 2-Wi	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PO	T	1											\vdash		
2-Wii 2-Wii UNE Loop R 2-Wii 2-Wii 2-Wii 2-Wii 2-Wire Voice 2-Wii Block	/ire VG Coin Port/Loop Combo – Zone 1	1	1		+	12.70										
2-Wir UNE Loop R 2-Wir 2-Wir 2-Wir 2-Wire Voice 2-Wire Block	Vire VG Coin Port/Loop Combo – Zone 2		2			21.19										
2-Wir 2-Wir 2-Wir 2-Wire Voice 2-Wir Block	/ire VG Coin Port/Loop Combo – Zone 3		3			34.80										
2-Wir 2-Wir 2-Wire Voice 2-Wir Block	Rates															
2-Wire Voice 2-Wire Block	Vire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	11.55										
2-Wire Voice 2-Wir Block	Vire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	20.04										
2-Wii Block	/ire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	33.65							ļ!		 	
Block	ce Grade Line Ports (COIN)	1	1		-									├ ──	 	
	/ire Coin 2-Way without Operator Screening and without cking (AL, KY, LA, MS)			UEPCO	UEPRF	1.15	40.19	19.83	24.91	6.63		15.66	i '	j ,	j .	1
2-001	Vire Coin 2-Way with Operator Screening (AL, KY)			UEPCO	UEPRE	1.15	40.19	19.83	24.91	6.63		15.66				
2-Wii	Vire Coin 2-Way with Operator Screening and Blocking: 011, V976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRA	1.15	40.19	19.83	24.91	6.63		15.66				
	Vire Coin 2-Way with Operator Screening and 011 Blocking			UEPCO	UEPRB	1.15	40.19	19.83	24.91	6.63		15.66				·
	, LA, MS) /ire Coin 2-Way with Operator Screening & Blocking:	1	-	UEPCO	UEPKB	1.15	40.19	19.83	24.91	6.63		15.66		\vdash	 	
	0/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.15	40.19	19.83	24.91	6.63		15.66				I
	Vire Coin Outward with Operator Screening and 011 Blocking	ı		LIEBOO	HEDDIA	4.45	10.10	40.00	04.04	0.00		45.00				1
	, FL) /ire Coin Outward with Operator Screening and Blocking:			UEPCO	UEPRK	1.15	40.19	19.83	24.91	6.63		15.66		\vdash		
011,	, 900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRH	1.15	40.19	19.83	24.91	6.63		15.66				
	Vire Coin Outward Operator Screening & Blocking: 900/976, DDD, 011+, and Local (AL, KY, LA, MS)			UEPCO	UEPCN	1.15	40.19	19.83	24.91	6.63		15.66			i !	İ
	/ire 2-Way Smartline with 900/976 (all states except LA)	1		UEPCO	UEPCK	1.15	40.19	19.83	24.91	6.63		15.66	$\overline{}$			
	Vire Coin Outward Smartline with 900/976 (all states except	1				5			2	0.00		.0.00				
LA)				UEPCO	UEPCR	1.15	40.19	19.83	24.91	6.63		15.66	ļ			
	AL UNE COIN PORT/LOOP (RC)			LIEBOO	UDEOU	4.50	40.40	40.00	04.61	0.00		45.00	ļ!	↓ _		
	E Coin Port/Loop Combo Usage (Flat Rate) MBER PORTABILITY	1	1	UEPCO	URECU	1.56	40.19	19.83	24.91	6.63		15.66		├ ───┤	├──┤	
	al Number Portability (1 per port)	+	1	UEPCO	LNPCX	0.35								\vdash	 	i
	RRING CHARGES - CURRENTLY COMBINED	1	1	021 00	LIVI OA	0.33			+					 	 	·
2-Wii		-			1											
	/ire Voice Grade Loop / Line Port Combination - Conversion	1		UEPCO	USAC2		0.10	0.10				15.66	ļ!	ļ	Ļ	1
ADDITIONAL	itch-as-is	-	1		1								<u>'</u>	├	├	
2-Wii Activi	itch-as-is AL NRCs	1		UEPCO	USAS2		0.00	0.00				45.60	, '	1 !	i !	İ
	itch-as-is AL NRCs Vire Voice Grade Loop/Line Port Combination - Subsequent		PORT (+		2.20					15.66	•	1		
UNE Port/Lo	itch-as-is AL NRCs Vire Voice Grade Loop/Line Port Combination - Subsequent	E LINE I		RES)	1	1						15.66	 			

	NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Exhil	ibit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge - Manual Svo Order vs.
						Rec	Nonrec	urring	Nonrecurring	Disconnect		ı	oss	Rates(\$)		-
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			15.76										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			24.23 37.52										
	op Rates		3			37.32										
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	14.38										+
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	22.85										+
	2-Wire Voice Grade Loop (SL2) - Zone 3			UEPFR	UECF2	36.14										
	Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.38	90.38	57.27	48.66	8.77		15.66				
1	2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res			UEPFR	UEPAR	1.38	90.38	57.27	48.66	8.77		15.66				
(2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPFR	UEPAP	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Unbundled Alabama Residence Dialing Plan			LIEDED	UEPWA	1.38	00.00	57.27	40.00	8.77		45.00				
	without Caller ID PFFICE TRANSPORT			UEPFR	UEPWA	1.38	90.38	57.27	48.66	8.77		15.66				+
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															+
-	Termination Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFR	U1TV2	21.13	40.54	27.41	16.74	6.90						<u> </u>
	or Fraction Mile			UEPFR	1L5XX	0.008838										
	All Features Offered		<u> </u>	UEPFR	UEPVF	1.98	0.00	0.00				15.66				+
	NUMBER PORTABILITY			OLITIK	OLI VI	1.50	0.00	0.00				13.00				+
	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										+
NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED					0.00										
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFR	USAC2		8.48	1.87				15.66				
1	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-With-Change			UEPFR	USACC		8.48	1.87				15.66				
	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	ORT (
UNE Por	rt/Loop Combination Rates		<u> </u>	,												1
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			15.76										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			24.23										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			37.52							ļ			
	op Rates		1	UEPFB	UECF2	14.38			1		1					+
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	14.38 22.85							-			+
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3			UEPFB	UECF2	36.14										+
	Voice Grade Line Port (Bus)				220.2	33.14			İ							†
	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.38	90.38	57.27	48.66	8.77		15.66				1
1	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - bus			UEPFB	UEPAW	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.38	90.38	57.27	48.66	8.77	ļ	15.66				
	2-Wire Voice Unbundled Alabama Business Dialing Plan without Caller ID			UEPFB	UEPWB	1.38	90.38	57.27	48.66	8.77		15.66				
	NUMBER PORTABILITY															
	Local Number Portability (1 per port)		ļ	UEPFB	LNPCX	0.35							ļ			
	DIFFICE TRANSPORT Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility			LIEDED	1147.60	04.10	40.51	07.11	40 = 1	0.00						1
	Termination Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFB	U1TV2	21.13	40.54	27.41	16.74	6.90						
	or Fraction Mile		1	UEPFB	1L5XX	0.008838					ļ		ļ			4
FEATUR				UEPFB	UEPVF	1.98	0.00	0.00	-			15.66				+

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	bit: C
													Incremental	Incremental	Incremental	Incremental
												Submitted	•	Charge -	Charge -	Charge -
04750000	DATE EL EMENTO	Interi	-	D00				D.4.T.F.Q(A)			Elec		Manual Svc	Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect		ı	oss	Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFB	LICACO		8.48	4.07				45.00				
-	Combination - Conversion - Switch-as-is 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFB	USAC2		8.48	1.87				15.66				-
	Combination - Conversion - Switch with change			UEPFB	USACC		8.48	1.87				15.66				
2-WIRI	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)						-									
UNE P	ort/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			15.76										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			24.23										
LINE	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			37.52										-
UNE L	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	14.38										\vdash
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	22.85										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	36.14										
2-Wire	Voice Grade Line Port Rates (BUS - PBX)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	1.38	119.27	69.85	61.18	8.34		15.66				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.38	119.27	69.85	61.18	8.34		15.66				
	Line Side Unbundled Incoming PBX Trunk Port - Bus 2-Wire Voice Unbundled 2-Way Combination PBX Alabama			UEPFP	UEPP1	1.38	119.27	69.85	61.18	8.34		15.66				-
	Calling Port			UEPFP	UEPA2	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPFP	UEPXE	1.38	119.27	69.85	61.18	8.34		15.66				
-	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPFP	UEPXE	1.38	119.27	69.85	61.18	8.34		15.00				-
	Administrative Calling Port			UEPFP	UEPXL	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy								• • • • • • • • • • • • • • • • • • • •							
	Room Calling Port			UEPFP	UEPXM	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPFP	UEPXO	1.38	119.27	69.85	61.18	8.34		15.66				
1.004	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port - NUMBER PORTABILITY			UEPFP	UEPXS	1.38	119.27	69.85	61.18	8.34		15.66				├
LUCAL	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.66				\vdash
INTER	OFFICE TRANSPORT			OLITI	LIVI OI	3.13	0.00	0.00				13.00				
1	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility					1										
	Termination	<u> </u>		UEPFP	U1TV2	21.13	40.54	27.41	16.74	6.90						<u>1</u> l
T	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile							· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·						
	or Fraction Mile			UEPFP	1L5XX	0.008838										↓
FEATU	All Features Offered			UEPFP	UEPVF	1.98	0.00	0.00				15.66				\vdash
NONRI	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLFIF	OLF VF	1.96	0.00	0.00				15.00				\vdash
NONKI	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port				1		-					 				\vdash
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		8.48	1.87				15.66				1
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch with change			UEPFP	USACC		8.48	1.87				15.66				igsquare
	PORT/LOOP COMBINATIONS - COST BASED RATES	D055			1											└──
	E VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK ort/Loop Combination Rates	PORT			1											\vdash
UNE P	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1		1	22.40										\vdash
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2		1	30.88	-					 				\vdash
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3		1	44.17										
UNE L	oop Rates															
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	14.38										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	22.85										<u> </u>
LIME D	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3	1	3	UEPPX	UECD1	36.14										\vdash
UNEP	VII Nate	l	1	l .	1	l l			I		l	l		l		

UNBUNDLE	D NETWORK ELEMENTS - Alabama													Attachi	ment: 2	Exhi	bit: C
														Incremental			Incremental
													Submitted		Charge -	Charge -	Charge -
CATEGORY	RATE ELEMENTS	Interi	Zone		cs	USOC			RATES(\$)			Elec					Manual Svc
CATEGORT	RATE ELEMENTS	m	Zone	-	.03	0300			KATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec			g Disconnect		•		Rates(\$)	•	
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - 2-Wire DID Port			UEPPX		UEPD1	8.02	207.31	73.74	107.14	11.20		15.66				
NONR	ECURRING CHARGES - CURRENTLY COMBINED	-															├
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-is			UEPPX		USAC1		7.31	1.87								İ
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion			OL: . X		00,101		7.01									
	with BellSouth Allowable Changes			UEPPX		USA1C		7.31	1.87								İ
ADDIT	IONAL NRCs																
LL	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX		USAS1		26.78	26.78								└
Teleph	none Number/Trunk Group Establisment Charges			UEPPX		NDT	0.00	0.00	0.00								-
	DID Trunk Termination (One Per Port) Additional DID Numbers for each Group of 20 DID Numbers	-		UEPPX		NDT ND4	0.00	0.00	0.00								-
	DID Numbers, Non- consecutive DID Numbers, Per Number			UEPPX		ND5	0.00	0.00	0.00								<u> </u>
	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00								
LOCAI	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00								
	E ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	INE SIDI	E PORT														-
UNE P	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	-				+											-
	UNE Zone 1		1	UEPPB	UEPPR	,	27.28										İ
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -			OLITE	OLITI	ì	27.20										
	UNE Zone 2		2	UEPPB	UEPPR		37.86										İ
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 3		3	UEPPB	UEPPR		53.84										
UNE L	oop Rates			LIEDDD	UEPPR	1101.01/	40.00										└
-	2-Wire ISDN Digital Grade Loop - UNE Zone 1	-	1	UEPPB	UEPPR	USL2X	19.03										
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	29.62										İ
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR		45.60										
UNE P	ort Rate																
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	8.24	190.01	132.76	100.67	21.28		15.66				
NONR	ECURRING CHARGES - CURRENTLY COMBINED																
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion			UEPPB	UEPPR	USACB	0.00	38.51	27.02				15.66				ĺ
ADDIT	TONAL NRCs	1		OLFFB	ULFFR	USACB	0.00	30.31	21.02				13.00		1		
	L NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-CHA	NNEL USER PROFILE ACCESS:																
	CVS/CSD (DMS/5ESS)	1	<u> </u>	UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)	1	<u> </u>	UEPPB UEPPB	UEPPR UEPPR	U1UCB U1UCC	0.00	0.00	0.00						1		
R-CHA	ICSD INNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C MS 8	: TNI	UEPPB	UEPPK	01000	0.00	0.00	0.00						+		
Б-СПА	CVS/CSD (DMS/5ESS)	1	T,	UEPPB	UEPPR	U1UCD	0.00	0.00	0.00						 		
	CVS (EWSD)	1		UEPPB	UEPPR	U1UCE	0.00	0.00	0.00	İ					1		
	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								
USER	TERMINAL PROFILE																
	User Terminal Profile (EWSD only)	1	<u> </u>	UEPPB	UEPPR	U1UMA	0.00	0.00	0.00			ļ					├
VERTI	CAL FEATURES All Vertical Features - One per Channel B User Profile	+	!	UEPPB	UEPPR	UEPVF	1.98	0.00	0.00			 			-		
INTER	OFFICE CHANNEL MILEAGE	+	 	JLFFD	ULPPR	JLF VF	1.88	0.00	0.00	1	1	+					
IIII EK	Interoffice Channel mileage each, including first mile and	1	<u> </u>														
	facilities termination		1		UEPPR	M1GNC	21.14	40.54	27.41	16.74	6.90						1
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.008838	0.00	0.00				0.00				
	E DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNI	K PORT															
UNE P	ort/Loop Combination Rates	 	<u> </u>				ļ			-	-						
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		4	UEPPP			166.87										1
 	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	+		JLFFF			100.07			1		+					
	Zone 2		2	UEPPP			238.50										1
	•			•				l		•		•			•		

UNBUNDLE	ED NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			1	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
						Rec	Nonred		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		3	UEPPP		398.85										
LINE I	Loop Rates		3	UEPPP	+	398.85										
ONE	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP	USL4P	82.55										
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP	USL4P	154.18										
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	314.52										
UNE F	Port Rate															
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	84.32	456.28	259.10	123.88	31.77		15.66				
NONE	RECURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion -Switch-as-is			UEPPP	USACP	0.00	119.07	78.56				15.66				
ADDI	Combination - Conversion -Switch-as-is TIONAL NRCs	1	-	ULFFF	USACP	0.00	119.07	78.36	+		1	10.00	1		1	
ADDI	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-	1	 		+				+		<u> </u>	1				
	Inward/two way Tel Nos. (except NC)			UEPPP	PR7TF		0.49									
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -								†							
	Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		11.51									
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -															
	Subsequent Inward Tel Numbers			UEPPP	PR7ZT		23.02									
LOCA	AL NUMBER PORTABILITY			LIEDDD	LNDON	4 75										
INTER	Local Number Portability (1 per port) RFACE (Provsioning Only)			UEPPP	LNPCN	1.75										
INTER	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00			1					1
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New o	or Additional "B" Channel															
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.53									
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.53									
0411	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.53									
CALL	TYPES Inward			UEPPP	PR7C1	0.00	0.00	0.00			1	1				<u> </u>
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								1
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								1
Intero	office Channel Mileage				111100											
	Fixed Each Including First Mile			UEPPP	1LN1A	60.34	89.27	81.81	16.35	14.44		15.66				
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.18										
	RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE F	Port/Loop Combination Rates			LIEBBO												
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	<u> </u>	2	UEPDC UEPDC	+	142.64 214.26			 							-
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3	1		UEPDC	+	374.61			+		<u> </u>	1				
UNE I	Loop Rates	<u> </u>	Ť	02. 00	1	37 7.01			†							
	4-Wire DS1 Digital Loop - UNE Zone 1	Ì	1	UEPDC	USLDC	82.55										
	4-Wire DS1 Digital Loop - UNE Zone 2			UEPDC	USLDC	154.18										
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	314.52										
UNE F	Port Rate			LIEDDO	UDDAT	00.00	454.40	050.00	117.00	44	<u> </u>	45.00				
NONE	4-Wire DDITS Digital Trunk Port RECURRING CHARGES - CURRENTLY COMBINED	<u> </u>		UEPDC	UDD1T	60.09	454.49	253.23	117.29	14.17	 	15.66				
NONK	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	1	-		+				+							
	- Switch-as-is	1		UEPDC	USAC4		129.49	67.02				15.66				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with DS1 Changes			UEPDC	USAWA		129.49	67.02				15.66				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk				USAWB							15.66				
ADDI	- Conversion with Change - Trunk TIONAL NRCs	 	-	UEPDC	USAVVB		129.49	67.02	+		 	15.00	1		-	
וטטא	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -	1	1		+				 							
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		14.48	14.48				15.66				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent	l					-									
	Channel Activation/Chan - 1-Way Outward Trunk	1	1	UEPDC	UDTTB		14.48	14.48				15.66				

UNBI	NDLF	D NETWORK ELEMENTS - Alabama												Attach	ment: 2	Fyhil	bit: C
300												Svc Order	Svc Order	Incremental		Incremental	
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec	Manually		Manual Svc		Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						-(1)			per LSK	per LOK	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							_	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsgnt Channel															
		Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		14.48	14.48				15.66				
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
		Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		14.48	14.48				15.66				
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
		Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		14.48	14.48				15.66				
	BIPOL	AR 8 ZERO SUBSTITUTION															
		B8ZS -Superframe Format			UEPDC	CCOSF		0.00	600.00								
		B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	600.00								
	Alterna	ate Mark Inversion															
		AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
		AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
	Teleph	one Number/Trunk Group Establisment Charges															
		Telephone Number for 2-Way Trunk Group		1	UEPDC	UDTGX	0.00										
		Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00										
		Telephone Number for 1-Way Inward Trunk Group Without DID		1	UEPDC	UDTGZ	0.00										
		DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00	0.00									
		DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00										
		Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00								
		Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00								
	Dedica	ted DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS	Digita	Loop	with 4-Wire DDITS	Trunk Port											
		Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
		Termination)			UEPDC	1LNO1	60.16	89.27	81.81	16.35	14.44		15.66				
		,															
		Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.18	0.00	0.00								
		Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities															
		Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
		Interoffice Channel Mileage - Additional rate per mile - 9-25															
		miles			UEPDC	1LNOB	0.18	0.00	0.00								
		Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
		Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							
		,															
		Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.18	0.00	0.00								
		Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00							
		Central Office Termininating Point			UEPDC	CTG	0.00										
	4-WIRE	DS1 LOOP WITH CHANNELIZATION WITH PORT															
		is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	ivations	5													
		system can have up to 24 combinations of rates depending on			ber of ports used												
		\$1 Loop															
		4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	82.55	0.00	0.00								
		4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	154.18	0.00	0.00								
		4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	314.52	0.00	0.00								
		SO Channelization Capacities (D4 Channel Bank Configuration	ns)														
		24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	101.40	0.00	0.00								
		48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	202.80	0.00	0.00								
		96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	405.60	0.00	0.00								
		144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	608.40	0.00	0.00								
		192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	811.20	0.00	0.00								
		240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,014.00	0.00	0.00								
		288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,216.80	0.00	0.00								
		384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,622.40	0.00	0.00								
		480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,028.00	0.00	0.00								
		576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,433.60	0.00	0.00								
		672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,839.20	0.00	0.00								
	Non-Re	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chanı	neliztio	n with Port - Conve	rsion Charge	Based on a Sy	stem									
		mum System configuration is One (1) DS1, One (1) D4 Channe															
	Multipl	es of this configuration functioning as one are considered Ad	ld'I afte	r the m	inimum system cor	nfiguration is	counted.										
		NRC - Conversion (Currently Combined) with or without															1
l	L	BellSouth Allowed Changes	L	Ш_	UEPMG	USAC4	0.00	150.48	8.36	<u> </u>	<u> </u>	<u> </u>	15.66	<u> </u>	<u> </u>	<u> </u>	1

UNBU	NDLE	NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Exhi	bit: C
CATEG		RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			1	Submitted Manually	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'l		Incremental Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
	Custom	Additions at End Hosel continue Where 4 Wine DC4 Loop with	h Char		ion with Bout Comb	in ation Com		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Additions at End User Locations Where 4-Wire DS1 Loop wit ot Currently Combined) in all states, except in Density Zone 1				ination Curr	ently Exists and										
	IAGM (IA	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port	ОГТОР	O IVI SP													
		and Assoc Fea Activation			UEPMG	VUMD4	0.00	716.11	468.04	148.75	17.65		15.66				
	Bipolar	8 Zero Substitution															
		Clear Channel Capability Format, superframe - Subsequent			UEPMG	CCOSF	0.00	0.00	600.00								
		Activity Only Clear Channel Capability Format - Extended Superframe -			UEPIVIG	CCOSF	0.00	0.00	600.00								
		Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	600.00								ĺ
	Alterna	te Mark Inversion (AMI)															
		Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
		Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
		ge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
	Exchar	ge Ports				 						<u> </u>	1				
		Line Cide Combination Channelland DDV Tauri Dest. Business			LIEDDY	LIEDOV	4.45	0.00	0.00	0.00	0.00		45.00				ĺ
		Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business			UEPPX UEPPX	UEPCX	1.15 1.15	0.00	0.00	0.00	0.00	 	15.66 15.66	1			-
		Line Side Odtward Charmenzed FBX Trunk Fort - Business			OLFFX	OLFOX	1.13	0.00	0.00	0.00	0.00		13.00				
		Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.15	0.00	0.00	0.00	0.00		15.66				İ
		2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	8.05	0.00	0.00	0.00	0.00		15.66				
		2-Wire Channelized PBX Area Calling Service Combination Port															
		(AL Only)			UEPPX	UEPA4	1.15	0.00	0.00				15.66				
		2 Wire Channelized PBX Area Calling Service Outgoing Only															
		Port (AL Only)			UEPPX	UEPA3	1.15	0.00	0.00				15.66				
	Feature	Activations - Unbundled Loop Concentration															
		Feature (Service) Activation for each Line Side Port Terminated in D4 Bank			UEPPX	1PQWM	0.56	54.55					15.66				
		Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			UEPPX	1PQWU	0.56	77.03					15.66				
	Teleph	one Number/ Group Establishment Charges for DID Service															
		DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
		DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
		Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00								
		Reserve Non-Consecutive DID Numbers Reserve DID Numbers			UEPPX	ND6 NDV	0.00	0.00	0.00	-		1					—
	l ocal N	Reserve DID Numbers lumber Portability			UEPPX	INDV	0.00	0.00	0.00			1	1	1			
	Local I	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	FEATU	RES - Vertical and Optional				1	50	0.00	2.30								
		witching Features Offered with Line Side Ports Only															
		All Features Available			UEPPX	UEPVF	1.98	0.00	0.00								
		op Rates	<u> </u>			1	ļ					ļ					
ONBON		ENTREX PORT/LOOP COMBINATIONS - COST BASED RATES		Ctot- 1	Commission mula (provide Hel	undled !! C	vitabin = == 0	itah Derte			<u> </u>	1	-			├
		Based Rates are applied where BellSouth is required by FCC ires shall apply to the Unbundled Port/Loop Combination - C								dled Port socti	on of this Date	Evhibit	1				
		Office and Tandem Switching Usage and Common Transport											Coin Port/Lo	on Combinat	ions.		
		irst and additional Port nonrecurring charges apply to Not Cu														Additional NE	Cs mav
	apply a	Iso and are categorized accordingly.	•			•		•	9 0900					,			
		set Rates for Unbundled Centrex Port/Loop Combination will		otiated	on an Individual Ca	ase Basis, un	til further notice	э.									
		CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo				+	1			-		 	1				
		ort/Loop Combination Rates (Non-Design)				1	1										
	JINE P	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+	1					<u> </u>	1				
		Non-Design		1	UEP91		12.70										1
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP91		21.19										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OEFSI	1	21.19					1	1	1			
		Non-Design		3	UEP91		34.80										1
	UNE P	ort/Loop Combination Rates (Design)		Ť	01	1	300										
					l .	1	1			1		1	1	·			

RATE ELEMENTS Interi m Zone BCS USOC RATES(\$) BCS USOC RATES(\$) RATES(\$)	UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhi	bit: C
ATE ELEMENTS mark most m												Svc Order	Svc Order				
ATTECHNICATION AND REAL PROPERTY OF THE PROPER												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
A			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
Best-roots Bes	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
Page Page														Electronic-	Electronic-	Electronic-	Electronic-
No. No.														1st	Add'l	Disc 1st	Disc Add'l
No. No.							I	Nonrec	rurring	Nonrecurring	Disconnect			OSS	Rates(\$)	l	L
A Wine Visit Corporative Vacas Grade Part (Contrarge) For Common 1 DEPsi 15.55							Rec					SOMEC	SOMAN			SOMAN	SOMAN
2-Vive VL cappe Vive Varie Grade Part (CorresopPort Control)		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
Design D				1	UEP91		15.53										
SWING VS LOUGH VALUE CORRESPOND				2	LIED01		24.00										
Design Section Secti					UEF91	-	24.00										
URF Ports URF Ports URF Port URF Por				3	UEP91		37.29										
2-Wine Value Close Long (St.1) - Zono 2	UNE L																
2-Wire Votes Grafe Loop (St. 1- Zene 3 3 UPP91 UECS2 336.5				1	UEP91	UECS1	11.55										
SWIN Vision Grante Long (St. 2) - Zerum 1																	
2-Vitre Votoe Grade Dept (2-) Zene 2 2 UEP91 UECS2 25.6																	
2-Wire Vend Grante Long (St. 2) - Zono 3 3 UPP91 UFP74 UFP74 1.15 40.19 19.80 24.91 6.60 15.60 1.5	 											ļ					
All States (Except North Carolina and Sout Carolina)	$\vdash \vdash \vdash$											<u> </u>		ļ	ļ		
All States (Except North Carolina and Sout Carolina)	LIME		1	3	UEP91	UECS2	36.14					<u> </u>	ļ	 	 		
2-Wire Votor Grafe Port (Centres A) Esses Local Area																	-
2-Wire Voso Grade Port Centrex with Center Report UEP91 UEP92 1.15 40.19 19.83 24.91 6.63 15.66	All Sta				I IEDQ1	ΠΕΡΥΔ	1 15	40.19	10.83	2/ 01	6.63	1	15.66				
Ava Ava	 			1	OLI 31	JLI IA	1.13	40.19	13.03	24.31	0.03	 	13.00				
2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area UEP91 UEP7H 1.15 40.19 19.83 2.491 6.63 15.66					UEP91	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
Area																	
Center/2 Basic Local Area UEP91 UEPVX 1.15 90.38 57.27 48.66 8.77 15.66					UEP91	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
2-Wire Voice Grade Port, DIF Serving Wire Center - 800 Service UEP91 UEPY2 1.15 90.36 57.27 48.66 8.77 15.66		2-Wire Voice Grade Port (Centrex from diff Serving Wire															
Term - Basic Local Area					UEP91	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
2-Wire Voice Grade Port Terminated in on Megalink or equivalent UEP91 UEP92 1.15 40.19 19.83 24.91 6.63 15.66																	
Basic Local Area					UEP91	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				
2 2 2 2 2 2 2 2 2 2					LIEBOA	LIEDVO	4.45	40.40	10.00	04.04	0.00		45.00				
Basic Local Area	-				UEP91	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				-
AL, KY, LA, MS, & TN Only					LIEP91	HEPY2	1 15	40 19	19.83	24 91	6.63		15.66				
2-Wire Voice Grade Port (Centrex 800 termination)	AL K				OLI 31	OLI 12	1.10	40.13	13.03	24.51	0.03		13.00				
2-Wire Voice Grade Port (Centrex 800 termination)	-,				UEP91	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66				
2-Wire Voice Grade Port (Centex from diff Serving Wire Center - 800 Service UEP91 UEPQX 1.15 90.38 57.27 48.66 8.77 15.66																	
Center 2		2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				
2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service UEP91 UEPQ2 1.15 90.38 57.27 48.66 8.77 15.66		2-Wire Voice Grade Port (Centrex from diff Serving Wire															
Term					UEP91	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
2-Wire Voice Grade Port terminated in on Megalink or equivalent UEP91																	
2-Wire Voice Grade Port Terminated on 800 Service Term	 	lerm		<u> </u>	UEP91	UEPQZ	1.15	90.38	57.27	48.66	8.77	<u> </u>	15.66	 	 	-	
2-Wire Voice Grade Port Terminated on 800 Service Term		2 Wire Voice Grade Port terminated in an Magaliak or againment		1	LIEDO1	LIEBOO	1 15	40.40	10.00	24.04	6.63		15.66	1	1		
Local Switching	\vdash			!								 		-	1	-	
Centrex Intercom Funtionality, per port UEP91 URECS 0.5488	Local		1	 	OLI 31	JLI QZ	1.13	40.19	13.03	24.31	0.03	<u> </u>	13.00	 	 		
Local Number Portability Local Number Portability (1 per port) UEP91 LNPCC 0.35	Local			<u> </u>	UEP91	URECS	0.5488										
Local Number Portability (1 per port)	Local				- "	1	3.0.00							İ	Ì		<u> </u>
All Standard Features Offered, per port				1	UEP91	LNPCC	0.35										
All Select Features Offered, per port UEP91 UEPVS 0.00 405.52	Featur																
All Centrex Control Features Offered, per port UEP91 UEPVC 1.98																	
NARS Unbundled Network Access Register - Combination UEP91 UARCX 0.00 0.00 0.00 0.00 Unbundled Network Access Register - Indial UEP91 UARIX 0.00 0.00 0.00 Unbundled Network Access Register - Outdial UEP91 UARIX 0.00 0.00 0.00 Unbundled Network Access Register - Outdial UEP91 UARIX 0.00 0.00 0.00 UEP91 UARIX 0.00 0.00 0.00 UEP91 UARIX 0.00 0.00 UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UARIX UEP91 UEP91 UARIX UEP91 UEP9								405.52						ļ	ļ		<u> </u>
Unbundled Network Access Register - Combination				<u> </u>	UEP91	UEPVC	1.98					ļ					
Unbundled Network Access Register - Indial UEP91	NARS		1	}	I IED01	LIABOV	0.00	0.00	0.00			<u> </u>	ļ	 	 		
Unbundled Network Access Register - Outdial	\vdash			 									-			-	
Miscellaneous Terminations	 		1	 								1	1	1	1	1	1
2-Wire Trunk Side	Misce			1	OL: 31	CAROA	0.00	0.00	0.00								
Trunk Side Terminations, each UEP91 CENA6 8.05 119.31 18.74 59.90 3.76 15.66 Interoffice Channel Mileage - 2-Wire			1	†		1	1					1		1	1		
Interoffice Channel Mileage - 2-Wire Interoffice Channel Facilities Termination - Voice Grade UEP91 M1GBC 21.13 40.54 27.41 16.74 6.90 15.66 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Feature Activations (DS0) Centrex Loops on Channelized DS1 Service					UEP91	CENA6	8.05	119.31	18.74	59.90	3.76		15.66	Ì	Ì		†
Interoffice Channel Facilities Termination - Voice Grade UEP91 M1GBC 21.13 40.54 27.41 16.74 6.90 15.66 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Interoffice Channel mileage, per mile or fraction of mileage, per mileage, p	Intero	ffice Channel Mileage - 2-Wire												İ	<u> </u>		
Interoffice Channel mileage, per mile or fraction of mile UEP91 M1GBM 0.008838 Feature Activations (DS0) Centrex Loops on Channelized DS1 Service		Interoffice Channel Facilities Termination - Voice Grade						40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.008838										
D4 Channel Bank Feature Activations			e					, The state of the					<u> </u>				<u> </u>
	D4 Ch	annel Bank Feature Activations		<u> </u>								l		l	l		

CATEGORY NATE ELEMENTS Inter BCS USOC NATE Section Colored	UNBUNDI	ED NETWORK ELEMENTS - Alabama												Attach	ment: 2	Fyhil	oit: C
RATE ELEMENTS RATE E	SHOUNDL	LD III. WORK ELLINERTO - Alabama										Svc Order	Svc Order				Incremental
ACTIONY RATE LEMENTS PROTECTION RATE PLANES 1905 PROTECTION RATE PLANES 1905																	Charge -
APPEN Part																	Manual Svc
Biochander Bio	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)								
Section Part			m						-(1)			per LSK	per LOK				
Pattern Actions on D. A. Course Service Course Se																	
Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation Cont														1st	Add'I	Disc 1st	DISC Add'I
Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation on D-I Chinnel Basic Politics Content Loop State Figure Activation Cont							_	Nonred	urrina	Nonrecurring	Disconnect			oss	Rates(\$)		
Festina Authorison of the Charmel Search Control							Rec					SOMEC	SOMAN			SOMAN	SOMAN
Petition Anticotro on the Charmon Binds Prison Stote Loop Stote Petition Anticotro on the Charmon Binds Prison Stote Loop Stote Petition Anticotro on the Charmon Binds Prison Stote Loop Stote Petition Anticotro on the Charmon Binds Prison Stote Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Anticotro on the Charmon Binds Prison Line Loop Stote Petition Line Loop Stote Lo		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.56										
Feature Activation on D-4 Charmed Board ATT TUNK Side Load Septiment Sep																	
Feature Activation on D-4 Criminal Basic Factors (1997) UEP91 FOWP 0.56 UEP91		Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.56										
Sec																	
Peature Activation on D.4 Chinnel Block Phinds Line Loop Seld UEP91 PCWP 0.56					UEP91	1PQW7	0.56										
Different Wise Context		Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
Feature Achiestics on D-4 Charmel Series (Fig. 14 page 55) UEPP 170 V/O 0.56					UEP91	1PQWP	0.56										
Feature Activation on D-4 Charmel Stark Tips Limit Transit Code																	
Feature Activation on D-4 Charmel Stark Tips Limit Transit Code		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.56										
Set																	
Feature Activation on D.4 Channel Book WATS Loop Site UEP91 DPWW 0.56		Slot			UEP91	1PQWQ	0.56									1	
Non-Recurring Charges (RRC) Associated with UNEP Centres Convention - Currently Depth of Selection - Currently Depth of		Feature Activation on D-4 Channel Bank WATS Loop Slot															
Comunitation Currently Committed Seath Asia with allowed otherwise per per of currently Curren	Non-		1	İ	İ	1				1				İ	İ	İ	
Convenience of Eventina Centerina Centerina Centerina (Centerina Centerina				İ		1								İ	1	İ	
Convertion of Entiring Centines Common Block UEP91 MIACS 0.00 667.21 1.566			1		UEP91	USAC2		0.10	0.10				15,66	l		I	
New Centres Standard Common Block			†											1	1	t	
New Centrex Customized Common Block UEP91 MINCC 0.00 697.21 1.5.66				İ			0.00							İ	1	İ	
Secondary Block, per Block UEP91 MCCC1 0.00 78.02 15.66																	
NAME Establishment Charge, Per Occasion UEP91 URECA 0.00 72.73 15.66																	
UNEP CENTREX SESS (Valid in All States)																	
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Combo	UNF				02. 0.	ONLON	0.00	72.70					10.00				
Number N																	
2.Wirk VSL Loop/2-Wire Voice Grade Port (Centrex)Port Combo-Non-Design 2.Wire VSL Loop/2-Wire Voice Grade Port (Centrex)Port Combo-Non-Design 2.Wire VSL Loop/2-Wire Voice Grade Port (Centrex)Port Combo-Non-Design 3. UEP96 34.80															1		
Non-Design	0														1		
2. Wife VS Loop/2-Wife Voice Grade Port (Centrex/Port Combo-Non-Design 2. Wife VS Loop/2-Wife Voice Grade Port (Centrex/Port Combo-Non-Design 3. UEP95 34.80 34.80				1	LIEP95		12 70										
Non-Design 2 UEP95 21.19				<u> </u>	0L1 00		12.70								1		
Non-Design Non				2	LIEP95		21 19										
Non-Design 3 UEP95 34.80					02. 00		20								1		
UNE Port/Loop Combination Rates (Design)				3	LIEP95		34.80										
2-Wire VIce Carde Port (Centrex) Port Combo- Design 1 UEP95 15.53	LINE			-	OLI SO	+	04.00								1		
Design	ONL					+									1		
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design				1	LIEP95		15 53										
Design 2 UFP95 24.00					OLI 93	+	10.00					1			1		
2-Wire Voice Grade Loop (St. 1) - Zone 1				2	LIEP95		24 00										
Design 3 UP95 37.29				-	OLI SO	+	24.00								1		
UNE Loop Rate				3	LIEDOS		37 20										
2-Wire Voice Grade Loop (St. 1) - Zone 1	IINE		 	- 3	OL1 30	+ -	31.29			 		1		 	 	 	
2-Wire Voice Grade Loop (SL 1) - Zone 2	OINE		 	1	HEP95	LIECS1	11 55					1	1	1	1	 	
2-Wire Voice Grade Loop (SL 1) - Zone 3	 		1	2						 				 	 	 	
2-Wire Voice Grade Loop (SL 2) - Zone 1	 		 							 		1		 	 	 	
2-Wire Voice Grade Loop (St. 2) - Zone 2 2 UEP95 UECS2 22.85	 		1	_						 				 	 	 	
2-Wire Voice Grade Loop (SL 2) - Zone 3 3 UEP95 UECS2 36.14 UNE Port Rate	 		 							 		1		 	 	 	
UNE Port Rate All States			 									1	1	1	1	 	
All States 2-Wire Voice Grade Port (Centrex) Basic Local Area UEP95 UEPYA 1.15 40.19 19.83 24.91 6.63 15.66	IINE		 	-	OL1 30	02002	30.14					1	1	1	1	 	
2-Wire Voice Grade Port (Centrex) Basic Local Area UEP95 UEPY8 1.15 40.19 19.83 24.91 6.63 15.66			 	1		+						1	1	1	1	 	
2-Wire Voice Grade Port (Centrex 800 termination)	All S		1		LIEP95	ΠΕΡΥΔ	1 15	<i>1</i> 0.10	10.92	2/ 01	6 63		15.66	 	 	 	
2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local UEP95 UEPYH 1.15 40.19 19.83 24.91 6.63 15.66	 		 	1								1		 	 	 	
Area	 				OL: 30	OLI ID	1.13	70.13	10.00	27.31	0.03	†	15.00	1	1	1	
2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area UEP95 UEPYM 1.15 90.38 57.27 48.66 8.77 15.66			1		LIEP95	UEPYH	1 15	<u>4</u> 0 10	10.83	24 01	6 63		15.66	l		I	
Center/2 Basic Local Area	 		 	1	02.100	JE: 111	1.13	70.13	19.00	27.31	0.03	1	10.00	 	 	 	
2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service UEP95 UEPYZ 1.15 90.38 57.27 48.66 8.77 15.66					LIEP95	HEPYM	1 15	90 38	57 27	48 66	8 77		15.66		1		
Term - Basic Local Area	 		 	1	OL1 30	OLI IIVI	1.13	30.00	31.21	40.00	0.77	1	13.00	 	 	 	
2-Wire Voice Grade Port terminated in on Megalink or equivalent UEP95			1		I IEDOS	I IEDV7	1 15	00.30	57 27	10 66	0 77		15.60	l		I	
Basic Local Area	 		-	 	OLF 33	OLF 12	1.15	90.38	51.21	40.00	0.77	-	10.00	-	 		
2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area UEP95 UEPY2 1.15 40.19 19.83 24.91 6.63 15.66					LIEDOE	LIEDVO	1 45	40.40	10.00	24.04	6.00		15.00		1		
Basic Local Area	 		-	 	ULFSO	UEF 19	1.15	40.19	19.83	∠4.91	6.63	-	10.00	-	 		
AL, KY, LA, MS, SC, & TN Only			1		LIEDOS	LIEDVO	4 45	40.40	10.00	24.04	6.00		15.60	l		I	
	AI L		 	1	OLF 90	UEFIZ	1.15	40.19	19.83	24.91	0.03	-	13.00		 	-	
	AL, r	2-Wire Voice Grade Port (Centrex)	 	1	UEP95	UEPQA	1.15	40.19	19.83	24.91	6.63	-	15.66		 	-	

UNBU	INDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhib	bit: C
0.120												Svc Order	Svc Order	Incremental		Incremental	
												Submitted	1		Charge -	Charge -	Charge -
			Interi									Elec			Manual Svc		Manual Svc
CATEG	ORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									po. zo.t	po. 20.1	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																D130 131	DISC Add I
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire Voice Grade Port (Centrex from diff Serving Wire															
		Center)2			UEP95	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
		Term			UEP95	UEPQZ	1.15	90.38	57.27	48.66	8.77		15.66				
		L.,, .,						40.40					4= 00				
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	1.15	40.19	19.83	24.91	6.63		15.66				
<u> </u>	Local S	Switching	<u> </u>	1	LIEBOE	LIDEOO	0.5400							1		1	├
-	1	Centrex Intercom Funtionality, per port	ļ	1	UEP95	URECS	0.5488					1			-		
<u> </u>	∟ocai I	Number Portability	!	1	LIEDOE	LNDCC	0.05			1		}		1	 	1	
	Feature	Local Number Portability (1 per port)	ļ	1	UEP95	LNPCC	0.35					1			-		
-	reature		 	+	UEP95	UEPVF	1.98			 		1	-	-	 		
		All Standard Features Offered, per port All Select Features Offered, per port			UEP95 UEP95	UEPVS	0.00	405.52									
		All Centrex Control Features Offered, per port		1	UEP95 UEP95	UEPVS	1.98	405.52									-
	NARS	All Certifex Control Features Offered, per port			UEF95	UEFVC	1.90										
	NAKS	Unbundled Network Access Register - Combination		1	UEP95	UARCX	0.00	0.00	0.00								-
		Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial		1	UEP95	UAR1X	0.00	0.00	0.00								-
		Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial	1	1	UEP95	UAROX	0.00	0.00	0.00			1					-
	Miscol	Ianeous Terminations			ULF 93	UARUX	0.00	0.00	0.00								
		Trunk Side	1	1								1					-
	Z-WIIG	Trunk Side Terminations, each		+	UEP95	CEND6	8.05	119.31	18.74	59.90	3.76	1	15.66				
	4-Wire	Digital (1.544 Megabits)			OL1 00	CENTRO	0.00	110.01	10.74	00.00	0.70		10.00				
		DS1 Circuit Terminations, each		1	UEP95	M1HD1	60.09	202.02	95.69	72.59	2.46		15.66				
		DS0 Channels Activated, each			UEP95	M1HDO	0.00	14.46	00.00	12.00	2.10		15.66				
	Interof	fice Channel Mileage - 2-Wire															
		Interoffice Channel Facilities Termination			UEP95	MIGBC	21.13	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.008838										
	Feature	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
	D4 Cha	annel Bank Feature Activations															
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.56										
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.56										
		Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
		Slot			UEP95	1PQW7	0.56										
		Feature Activation on D-4 Channel Bank Centrex Loop Slot -		1	l	1											1
ļ		Different Wire Center	ļ	 	UEP95	1PQWP	0.56								ļ		I
1				1													1
<u></u>		Feature Activation on D-4 Channel Bank Private Line Loop Slot	<u> </u>	 	UEP95	1PQWV	0.56					<u> </u>		ļ	 	ļ	
1		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop	1	1	LIEDOE	400140	0 =0								Ì		1
		Slot	<u> </u>	 	UEP95	1PQWQ	0.56					1					
-	Non D	Feature Activation on D-4 Channel Bank WATS Loop Slot ecurring Charges (NRC) Associated with UNE-P Centrex	 	+	UEP95	1PQWA	0.56			 		1	-	-	 		
	NON-R		 	+		+				-				-	-	-	
1		NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port	1	1	UEP95	USAC2]	0.10	0.10				15.66		1		1
		Conversion of Existing Centrex Common Block, each	 	+	UEP95 UEP95	USACZ	 	37.75	16.58	-		-	15.66		 	-	
-		New Centrex Standard Common Block	1	1	UEP95	M1ACS	0.00	667.21	10.56			1	15.66				
—		New Centrex Standard Common Block	1	+	UEP95	M1ACC	0.00	667.21		 			15.66		 		
—		NAR Establishment Charge, Per Occasion	1	+	UEP95	URECA	0.00	72.73		 			15.66		 		
—	IINF-P	CENTREX - DMS100 (Valid in All States)	1	+	OE1 30	UNLUA	0.00	12.13		 		1	13.00		 		
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo	1	+		+	 			 					 		
I		ort/Loop Combination Rates (Non-Design)	1	1		+						1			 		—
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1		1						1		1	1	1	t
		Non-Design	1	1	UEP9D		12.70								1		1
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			-												
		Non-Design		2	UEP9D		21.19										1
	•		•	•				l.				•			•		

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			1	Submitted	_	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Non-Design		3	UEP9D		34.80										Ĭ
UNE	Port/Loop Combination Rates (Design)			02.02		000										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-														
-	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	UEP9D		15.53										
	Design		2	UEP9D		24.00										Ĭ
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP9D		37.29										
UNE	.oop Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	11.55										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	20.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	33.65										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	14.38										
-	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D UEP9D	UECS2 UECS2	22.85 36.14										
UNE	Port Rate			OLI OD	CLOCL	00.14										
ALL S	TATES															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area 2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local			UEP9D	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
	Area			UEP9D	UEPYC	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area			UEP9D	UEPYV	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 Basic Local Area			UEP9D	UEPYO	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	1.15	90.38	57.27	48.66	8.77		15.66				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area 2 Wire Voice Crade Port (Centrex/differ SWC /EBS M5009)3, 3			UEP9D	UEPYS	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.15	90.38	57.27	48.66	8.77		15.66				<u> </u>

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachr	nent: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			OLI OD	OLI 10	1.10	30.00	07.27	40.00	0.77		10.00				
	Basic Local Area			UEP9D	UEPY6	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEF9D	UEPTI	1.15	90.36	51.21	40.00	0.11		15.00				
	Term			UEP9D	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			LIEDOD	LIEDVO	4.45	40.40	40.00	24.04	0.00		45.00				
	Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term Basic			UEP9D	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				
	Local Area	L		UEP9D	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66				
AL, K	Y, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.15	40.19	19.83 19.83	24.91 24.91	6.63		15.66				_
 	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D UEP9D	UEPQB UEPQC	1.15 1.15	40.19 40.19	19.83	24.91	6.63 6.63		15.66 15.66				+
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	1.15	40.19	19.83	24.91	6.63		15.66				
 	2-Wire Voice Grade Port (Centrex / EBS-M5312)3 2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D UEP9D	UEPQG UEPQT	1.15 1.15	40.19 40.19	19.83 19.83	24.91 24.91	6.63 6.63		15.66 15.66				<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5006)3			UEP9D	UEPQU	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				_
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication)3			UEP9D	UEPQW	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	1.15	90.38	57.27	48.66	8.77		15.66				
	0 M/ 1/2 0 1- B (0 1 / 1// 0 M/ 0 / EB0 ME440)0 0			LIEDOD	UEPQR	1.15	90.38	57.07	40.00	0.77		45.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.15	90.38	57.27	48.66	8.77		15.66				
	·															
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			021 00	JE1 W1	1.13	30.30	51.21	40.00	0.77		10.00				
	Term			UEP9D	UEPQZ	1.15	90.38	57.27	48.66	8.77		15.66				
	2 Wire Vales Crade Port terminated in an Magaliali as assistant			LIEDOD	UEPQ9	4 45	40.19	10.00	24.91	6.00		15.00				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D UEP9D	UEPQ9 UEPQ2	1.15 1.15	40.19	19.83 19.83	24.91	6.63 6.63		15.66 15.66				
Loca	Switching								201	3.00		.0.00				
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.5488										
Loca	Number Portability			HEDOD	LNDCC	0.0-										
Featu	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										-
1.5000	All Standard Features Offered, per port			UEP9D	UEPVF	1.98										
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	405.52									
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	1.98						<u> </u>				

	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	i Exhil	bit: C
											Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		""									-		Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
		<u> </u>				1	Nonrec	urring	Nonrecurring	Disconnect			220	Rates(\$)		<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NARS								7144		7144				00		
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00								
	aneous Terminations															
	Trunk Side															
	Trunk Side Terminations, each			UEP9D	CEND6	8.05	119.31	18.74	59.90	3.76		15.66				
	Digital (1.544 Megabits)			UEP9D	M1HD1	00.00	202.02	05.00	70.50	0.40		45.00				.
	DS1 Circuit Terminations, each DS0 Channels Activiated per Channel			UEP9D	M1HD0	60.09 0.00	14.46	95.69	72.59	2.46		15.66				
	ice Channel Mileage - 2-Wire			UEP9D	MITHDO	0.00	14.46					15.66				
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	21.13	40.54	27.41	16.74	6.90		15.66				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.008838	40.04	21.41	10.74	0.30		13.00				
Feature	Activations (DS0) Centrex Loops on Channelized DS1 Service	e		OLI OD	IVIIODIVI	0.000000										
	nnel Bank Feature Activations	Ī														
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.56										
. !	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.56										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP9D	1PQW7	0.56										
. !	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP9D	1PQWP	0.56										<u> </u>
. !					450147											
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.56										
. !	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.56										
	Feature Activation on D-4 Channel Bank WATS Loop Slot	1		UEP9D	1PQWQ	0.56										
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex			OLF9D	IFQWA	0.50										-
	NRC Conversion Currently Combined Switch-As-Is with allowed															
. !	changes, per port			UEP9D	USAC2		0.10	0.10				15.66				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		37.75	16.58				15.66				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	667.21					15.66				
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	667.21					15.66				
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.73					15.66				
	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															ļ
UNE Po	ort/Loop Combination Rates (Non-Design)	ļ			1											<u> </u>
. ['	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	,	LIEDOE		40.70						1				
	Non-Design	!	1	UEP9E	1	12.70			ļ					1	1	├
. ['	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	1	2	UEP9E		21.19						1				
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	 		OLF JL	1	21.19			 					1	1	
. [!	Non-Design		3	UEP9E		34.80										
	ort/Loop Combination Rates (Design)	1			1	04.00						 				
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1			İ											
. [!	Design	1	1	UEP9E		15.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design	<u></u>	2	UEP9E		24.00								<u></u>	<u></u>	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design	<u> </u>	3	UEP9E		37.29										
	pop Rate	ļ	لبا	115005	LIEGE:											
	2-Wire Voice Grade Loop (SL 1) - Zone 1	ļ		UEP9E	UECS1	11.55										
	2-Wire Voice Grade Loop (SL 1) - Zone 2	ļ	2	UEP9E	UECS1	20.04										├
	2-Wire Voice Grade Loop (SL 1) - Zone 3	 	3	UEP9E UEP9E	UECS1	33.65										
	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2	 	2	UEP9E UEP9E	UECS2 UECS2	14.38 22.85						 				
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3	 	3	UEP9E	UECS2	36.14			 					1	1	
		1	J	OLI OL	32032	30.14			 					1	1	
UNE Po	ort Kate											ī		ī		

UNBUNI	DLED	NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental		Incremental
													Submitted		Charge -	Charge -	Charge -
CATEGOR		RATE ELEMENTS	Interi	7	BCS	USOC			DATEC(®)			Elec	Manually		Manual Svc	Manual Svc	
CATEGOR	KY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec		Nonrecurring					Rates(\$)		
		2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.15	First 40.19	Add'I 19.83	First 24.91	Add'l 6.63	SOMEC	SOMAN 15.66	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			OLF 9L	OLFIA	1.13	40.19	19.03	24.51	0.03		13.00				+
		Area			UEP9E	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
	- 1	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
		Area			UEP9E	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
1		2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP9E	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OLF9L	OLFTW	1.13	90.30	31.21	40.00	0.77		13.00				+
		Term - Basic Local Area			UEP9E	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				
	-	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
		Basic Local Area			UEP9E	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP9E	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66				
AL		LA, MS, & TN Only			OLF9L	OLF 12	1.13	40.19	19.03	24.51	0.03		13.00				-
		2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66				-
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQB	1.15	40.19	19.83	24.91	6.63		15.66				
	- 2	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEF9E	UEPQIVI	1.15	90.36	57.27	40.00	0.77		13.66	1			+
		Term			UEP9E	UEPQZ	1.15	90.38	57.27	48.66	8.77		15.66				
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66				
		2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.15	40.19	19.83	24.91	6.63		15.66				
Lo		witching Centrex Intercom Funtionality, per port			UEP9E	URECS	0.5488										+
Lc		umber Portability			OLF 9L	UKLCS	0.3400										+
		Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Fe	atures																
		All Standard Features Offered, per port			UEP9E	UEPVF	1.98	107.50									
		All Select Features Offered, per port All Centrex Control Features Offered, per port			UEP9E UEP9E	UEPVS UEPVC	0.00 1.98	405.52						-			-
N/	ARS	All Centrex Control Features Offered, per port		1	UEP9E	UEPVC	1.98										+
		Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00								1
		Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00								
		Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00								
		neous Terminations runk Side															4
Z-V		Trunk Side Trunk Side Terminations, each			UEP9E	CEND6	8.05	119.31	18.74	59.90	3.76	1	15.66				+
4-1		Digital (1.544 Megabits)		†		3200	0.00	110.01	10.74	33.30	5.70		10.00	<u> </u>			†
		DS1 Circuit Terminations, each			UEP9E	M1HD1	60.09	202.02	95.69	72.59	2.46		15.66				
		DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	14.46					15.66				
Int		ce Channel Mileage - 2-Wire Interoffice Channel Facilities Termination		<u> </u>	UEP9E	MIGBC	21.13	40.54	27.41	16.74	6.90	1	15.66				
		Interoffice Channel Facilities Termination Interoffice Channel mileage, per mile or fraction of mile		!	UEP9E UEP9E	MIGBC	0.008838	40.54	21.47	10.74	0.90	1	15.00	+			
Fe		Activations (DS0) Centrex Loops on Channelized DS1 Service	e	†	021 02	.vii	0.000008							<u> </u>			†
	Char	nnel Bank Feature Activations															
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.56										<u> </u>
	I.	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.56										
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot		 	OLFAE	IFQVVO	0.06					<u> </u>		 			
		Slot			UEP9E	1PQW7	0.56										
		Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
$oxed{oxed}$	ļ	Different Wire Center		<u> </u>	UEP9E	1PQWP	0.56					1					<u> </u>
	I.	Footure Activation on D.4 Channel Book Britists Line Land Clat			UEP9E	1PQWV	0.56										
\vdash		Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop		 	OLFAE	IFQVVV	0.06					<u> </u>		 			
		Slot			UEP9E	1PQWQ	0.56										
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.56										<u> </u>

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachi	ment: 2	Exhil	bit: C
											Svc Order	Svc Order				Incremental
												Submitted		Charge -	Charge -	Charge -
		Interi									Elec	Manually	_	Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		""											Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
					1	1	Nonroo	rina	Monroourring	Dissennest			220	Potos(\$)		
-					-	Rec	Nonrec First	Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN		Rates(\$) SOMAN	SOMAN	SOMAN
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex				+		FIISL	Auu i	FIISL	Auu i	SOWIEC	JOWAN	JOWAN	JOWAN	JOWAN	JOWAN
1101111	NRC Conversion Currently Combined Switch-As-Is with allowed				+											
	changes, per port			UEP9E	USAC2		0.10	0.10				15.66				
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		37.75	16.58				15.66				
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	667.21					15.66				
	New Centrex Customized Common Block			UEP9E	M1ACC	0.00	667.21					15.66				
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	72.73					15.66				
	CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)															
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE F	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	_	LIEDOS	1	40.70							1	1		
 	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	 	1	UEP93	+	12.70					 	-	 	 	-	
	Non-Design	1	2	UEP93	1	21.19							1	1		
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLF 93	+	21.19						1				
	Non-Design	1	3	UEP93	1	34.80							1	1		
UNF F	Port/Loop Combination Rates (Design)			02. 00	1	54.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP93		15.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP93		24.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP93		37.29										
UNE L	oop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1			UEP93	UECS1	11.55										_
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP93 UEP93	UECS1 UECS1	20.04 33.65										
	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	14.38										-
	2-Wire Voice Grade Loop (SL 2) - Zone 1			UEP93	UECS2	22.85										
	2-Wire Voice Grade Loop (SL 2) - Zone 3			UEP93	UECS2	36.14										+
UNE F	Port Rate		Ŭ	02.00	02002	00.11										
	Y, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP93	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP93	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire	1		LIEDOS	LIEDA		22.22		40.00			45.00	1	1		
	Center)2 Basic Local Area	 	-	UEP93	UEPYM	1.15	90.38	57.27	48.66	8.77	 	15.66	 	 	1	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area	1		UEP93	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66	1	1		
 	2-Wire Voice Grade Port terminated in on Megalink or equivalent	 		OLF 30	OLF IZ	1.15	90.38	31.21	40.00	0.77	 	10.00	1	1	1	1
	- Basic Local Area	l		UEP93	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				
 	2-Wire Voice Grade Port Terminated on 800 Service Term -	1		00	52. 10	1.15	70.13	10.00	27.01	0.00	1	10.00	1	1	1	
	Basic Local Area	1		UEP93	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66	1	1		
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66	İ	İ	<u> </u>	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire	l		l												
 	Center)2	<u> </u>		UEP93	UEPQM	1.15	90.38	57.27	48.66	8.77	<u> </u>	15.66	ļ	ļ	ļ	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1		LIEDOS	LIEDOZ		20.00		40.00			45.00	1	1		
 	Term	!		UEP93	UEPQZ	1.15	90.38	57.27	48.66	8.77	<u> </u>	15.66	-	 		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1		UEP93	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66	1	1		
 	2-Wire Voice Grade Port terminated in on Megalink of equivalent	 		UEP93	UEPQ2	1.15	40.19	19.83	24.91	6.63	1	15.66			-	+
Local	Switching	1		OL1 30	JLI VZ	1.15	40.13	19.03	24.31	0.03	 	13.00				
20001	Centrex Intercom Funtionality, per port	1		UEP93	URECS	0.5488							1	1		
Local	Number Portability				1	2.2.00										
	Local Number Portability (1 per port)			UEP93	LNPCC	0.35										
											•		•	•		-

RUNDLE	D NETWORK ELEMENTS - Alabama			•	-	1					Ι			ment: 2		ibit: C
EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually		Charge -	Incremental Charge - Manual Svc Order vs.	Charge
		""											Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electron Disc Add
						n	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Feature	es															
	All Standard Features Offered, per port			UEP93	UEPVF	1.98										
	All Centrex Control Features Offered, per port			UEP93	UEPVC	1.98										Ī
NARS																
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00								
	laneous Terminations															
2-Wire	Trunk Side															
	Trunk Side Terminations, each			UEP93	CEND6	8.05	119.31	18.74	59.90	3.76		15.66				
4-Wire	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP93	M1HD1	60.09	202.02	95.69	72.59	2.46		15.66				
	DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	14.46					15.66				
Interof	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP93	MIGBC	21.13	40.54	27.41	16.74	6.90		15.66				
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.008838										
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 Cha	nnel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.56										
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.56										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															Ī
	Slot			UEP93	1PQW7	0.56										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															Ī
	Different Wire Center			UEP93	1PQWP	0.56										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.56										
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop															
	Slot	<u> </u>		UEP93	1PQWQ	0.56					<u> </u>					
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.56										
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port		1	UEP93	USAC2		0.10	0.10			ļ	15.66				
	Conversion of Existing Centrex Common Block, each		1	UEP93	USACN		37.75	16.58			ļ	15.66				
	New Centrex Standard Common Block	<u> </u>	 	UEP93	M1ACS	0.00	667.21					15.66			-	
_	New Centrex Customized Common Block	 	1	UEP93	M1ACC	0.00	667.21				1	15.66		-	1	+
	NAR Establishment Charge, Per Occasion	 	1	UEP93	URECA	0.00	72.73				1	15.66		-	1	₩
	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD	 	1	 	+	1					1			-	1	+
	2 - Requires Interoffice Channel Mileage		1	-	+	 									 	+
	- Requires Specific Customer Premises Equipment	ioot t-	roto t	 	Conord Fee	no and Candidia									 	+
	Rates displaying an "R" in Interim column are interim and sub							Abia Fubility -	nd to sumb the		 	l mliaabla		 	 	+
INOTE:	Where the state Commission has adopted rates for the rate ele	ements	contail	nea nerein, it is the	intent of the	Parties to reflec	T SUCH PATES IN	inis Exhibit a	ng to anniv the		rent with an	DUCADIE		I	1	1

IUNBU	NDLE	D NETWORK ELEMENTS - Florida												Attach	ment: 2	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental		Incrementa
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec			Manual Svc		Manual Svo
CATEG	OPV	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)				-				
CAILG	OKI	RATE ELEMENTS	m	Zone	603	0300			KATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec		Nonrecurring					Rates(\$)	_	
								First	Add'l	First	Add'l			SOMAN		SOMAN	SOMAN
'	The "Zo	one" shown in the sections for stand-alone loops or loops as part of	of a com	binatio	n refers to Geograph	ically Deavera	aged UNE Zones	. To view Geog	graphically Dea	veraged UNE Zo	ne Designation	ns by Centra	I Office, refe	r to Internet W	ebsite:		
,	http://w	ww.interconnection.bellsouth.com/become_a_clec/html/interconne	ection.h	tm													
OPERA	TIONAL	SUPPORT SYSTEMS															
		(1) Electronic Service Order: CLEC should contact its contract	t negot	iator if	it prefers the state	specific elec	tronic service o	rdering charge	es as ordered l	ov the State Cor	mmissions. T	he electron	c service o	rdering charg	e currently co	ntained in th	is rate
,		is the BellSouth regional electronic service ordering charge.															
		(2) Any element that can be ordered electronically will be bill															li. Fan
		elements that cannot be ordered electronically at present per t				e in this cate	gory reflects the	e charge that v	vould be billed	to a CLEC onc	ce electronic o	ordering cap	abilities co	me on-line to	r that elemen	t. Otherwise,	the manual
	orderin	g charge, SOMAN, will be applied to a CLECs bill when it sub	mits ar	LSR t	o BellSouth.												
		Manual Service Order Charge, per LSR, Disconnect Only (FL)				SOMAN				1.83							
		Electronic OSS Charge, per LSR, submitted via BST's OSS	l											l			
'		interactive interfaces (Regional)	l	1		SOMEC		3.50				l	1		Ì	Ì	l
UNE SI	RVICE	DATE ADVANCEMENT CHARGE											i				
		The Expedite charge will be maintained commensurate with	BellSou	th's FC	C No.1 Tariff. Secti	ion 5 as annli	cable.			1		1	1		1	1	i
-		UNE Expedite Charge per Circuit or Line Assignable USOC, per	I		o ito: i raini, ocoti	on o as appn	oubic.										
,		Day	l		ALL UNE	SDASP		200.00		1			l				1
LINDIA	DIEDE	XCHANGE ACCESS LOOP	 	-	ALL UINL	SUASE	1	200.00		 		-	 	-	 	 	
			 	-		1	1			 		1	 		1	1	1
└	∠-WIRE	ANALOG VOICE GRADE LOOP	 	<u> </u>		4				ļ							
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	12.79	49.57	22.83	25.62	6.57		11.90				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	17.27	49.57	22.83	25.62	6.57		11.90				
'		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	33.36	49.57	22.83	25.62	6.57		11.90				
		Unbundled Copper Loop, Non-Designed 2-Wire Voice Grade															
'		Copper Flagging - Zone 1		1	UEANL	UEQ2F	12.79	49.57	22.83	25.62	6.57		11.90				
		Unbundled Copper Loop, Non-Designed 2-Wire Voice Grade															
'		Copper Flagging - Zone 2		2	UEANL	UEQ2F	17.27	49.57	22.83	25.62	6.57		11.90				
		Unbundled Copper Loop, Non-Designed 2-Wire Voice Grade			OLTUVE	OLGEI	11.21	10.01	22.00	20.02	0.01		11.50				
'				3	UEANL	UEQ2F	33.36	49.57	22.83	25.62	6.57		11.90				
		Copper Flagging - Zone 3		3			33.30			25.62	6.57		11.90				
		Engineering Information Document (EI)			UEANL	UEANM		12.28	12.28								
'		Unbundled Miscellaneous Rate Element, CLEC Conversion,															
		UVL-SL1 Loop on Copper to UVL-SL1 Flagging			UEANL	UREGG		4.1845	0.6042								
'		Loop Testing - Basic 1st Half Hour			UEANL	URET1		77.09					11.90				
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		33.12					11.90				
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
,		(UVL-SL1)			UEANL	UREWO		15.78	8.94				11.90				
—		Engineering Information Document (EI)			UEANL	UEANM		12.28	12.28								
$\vdash \vdash \vdash$		Manual Order Coordination for UVL-SL1s (per loop)	1	1	UEANL	UEAMC		9.00	9.00	 		 	1		†	†	
\vdash		Order Coordination for Specified Conversion Time for UVL-SL1	 	 	O = / 11 11	OL, WIO	1	3.00	3.00	1		 	l	1	1	1	1
,		(per LSR)	l		UEANL	OCOSL		23.02	23.02	1			l				1
$igwdapsilon^{\prime\prime}$	0 14/15-		-	-	ULANL	UCUSL		23.02	23.02	 		1			ļ	ļ	-
$\vdash \vdash \vdash$	Z-WIRE	Unbundled COPPER LOOP	<u> </u>	.	LIFO	LIEOSY				10.0-		1		-			
└─ ─		2-Wire Unbundled Copper Loop - Non-Designed Zone 1	<u> </u>		UEQ	UEQ2X	13.83	41.64	19.02	19.65	5.09	ļ	11.90				ļ
<u> </u>		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2			UEQ	UEQ2X	15.29	41.64	19.02	19.65	5.09		11.90				
L!		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3		3	UEQ	UEQ2X	20.29	41.64	19.02	19.65	5.09		11.90				
l ,		Order Coordination 2 Wire Unbundled Copper Loop - Non-	l	1									1				
1		Designed (per loop)	l	1	UEQ	USBMC		9.00	9.00]		I]]	1	1	1
		Engineering Information Document			UEQ			12.28	12.28				11.90				
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		77.09		1			11.90	ĺ			
		Loop Testing - Basic Additional Half Hour			UEQ	URETA		33.12		1		1	11.90		1	1	1
\vdash		CLEC to CLEC Conversion Charge Without Outside Dispatch	 	 		5.1.2.77	1	55.1Z		 		 	11.50	 	 	 	
'		(UCL-ND)	l		LIEO	LIDEWO		4407	7.40	1			11.00				1
LINISTS	DI E5 -		 	!	UEQ	UREWO	1	14.27	7.43	 		 	11.90		 	 	
		XCHANGE ACCESS LOOP	.			+	1					1	ļ	1			
igsquare	2-WIRE	ANALOG VOICE GRADE LOOP															
'		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	l	l .	l	1						I	l]	1	1	1
Ш_		Ground Start Signaling - Zone 1		1	UEA	UEAL2	14.50	135.75	82.47	63.53	12.01	ļ	11.90				ļ
]		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	1	1						1		1]]	1	1	1
└ '		Ground Start Signaling - Zone 2	<u>L_</u>	2	UEA	UEAL2	19.57	135.75	82.47	63.53	12.01	<u> </u>	11.90	<u> </u>	<u></u>	<u></u>	<u> </u>
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or															
1 '		Ground Start Signaling - Zone 3	l	3	UEA	UEAL2	37.82	135.75	82.47	63.53	12.01	l	11.90		Ì	Ì	İ
		Order Coordination for Specified Conversion Time (per LSR)	1	Ť	UEA	OCOSL		23.02		12.20		i	1	1	1	1	i e
1 *			 	 			1	20.02		 		 			 	 	
\vdash		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															

Version 3Q02: 09/06/02

UNRUN	DI FI	NETWORK ELEMENTS - Florida												Attach	ment: 2	Fyhi	ibit: C
CIADOIA	DEEL	O NETWORK ELLINENTS - Florida	1		1		1					Svc Order	Svc Order	Incremental	Incremental		
													Submitted		Charge -	Charge -	Charge -
												Elec		Manual Svc	Manual Svc		
CATEGO	RY	RATE ELEMENTS	Interi	Zone	BCS	usoc			RATES(\$)								
CAILGO		KATE ELEMENTO	m	20116	B00	0000			KATEO(ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
								Nonred	urring	Nonrecurring	a Disconnect			088	Rates(\$)		
-				1			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse						THOL	Auu i	11130	Addi	JONEC	JOINAN	JOINAIN	JONAN	JOHAN	JONAN
		Battery Signaling - Zone 2		2	UEA	UEAR2	19.57	135.75	82.47	63.53	12.01		11.90				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse			OLA	OLAKZ	13.37	133.73	02.47	00.00	12.01		11.50				+
		Battery Signaling - Zone 3		3	UEA	UEAR2	37.82	135.75	82.47	63.53	12.01		11.90				
		Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL	07.02	23.02	02.11	00.00	12.01		11.00				+
		CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.71	36.35				11.90				+
4		ANALOG VOICE GRADE LOOP			0271	0.12.110		0	00.00				11.00				1
H		4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	23.02	167.86	115.15	67.08	15.56		11.90				+
		4-Wire Analog Voice Grade Loop - Zone 2			UEA	UEAL4	31.07	167.86	115.15	67.08	15.56		11.90				+
\vdash		4-Wire Analog Voice Grade Loop - Zone 3	1		UEA	UEAL4	60.02	167.86	115.15	67.08	15.56		11.90	1	1		
\vdash		Order Coordination for Specified Conversion Time (per LSR)	†	Ť	UEA	OCOSL	55.62	23.02		51.00	.5.00				1		
	-	CLEC to CLEC Conversion Charge without outside dispatch	†		UEA	UREWO	1	87.71	36.35		1		11.90		İ		1
2		ISDN DIGITAL GRADE LOOP	1		t			÷	22.30		1				1	Ì	
		2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	21.76	147.69	94.41	62.23	10.71		11.90				1
		2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	29.38	147.69	94.41	62.23	10.71		11.90				
		2-Wire ISDN Digital Grade Loop - Zone 3			UDN	U1L2X	56.76	147.69	94.41	62.23	10.71		11.90				1
		Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		23.02									1
		CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.61	44.15				11.90				
2	-WIRE	ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIBLE	LOOP	,												1
		2 Wire Unbundled ADSL Loop including manual service inquiry															1
		& facility reservation - Zone 1		1	UAL	UAL2X	12.65	149.53	103.85	75.05	15.63		11.90				
		2 Wire Unbundled ADSL Loop including manual service inquiry															
		& facility reservation - Zone 2		2	UAL	UAL2X	17.08	149.53	103.85	75.05	15.63		11.90				
		2 Wire Unbundled ADSL Loop including manual service inquiry															1
		& facility reservation - Zone 3		3	UAL	UAL2X	33.00	149.53	103.85	75.05	15.63		11.90				
		Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		23.02									
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
	ŀ	facility reservaton - Zone 1		1	UAL	UAL2W	12.65	124.83	71.12	60.64	9.12		11.90				
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
	ŀ	facility reservaton - Zone 2		2	UAL	UAL2W	17.08	124.83	71.12	60.64	9.12		11.90				
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
		facility reservaton - Zone 3		3	UAL	UAL2W	33.00	124.83	71.12	60.64	9.12		11.90				
		Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		23.02									
		CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		86.19	40.39				11.90				
2		HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
		2 Wire Unbundled HDSL Loop including manual service inquiry															
		& facility reservation - Zone 1		1	UHL	UHL2X	9.97	159.09	113.41	75.05	15.63		11.90				
	ŀ	2 Wire Unbundled HDSL Loop including manual service inquiry															
		& facility reservation - Zone 2		2	UHL	UHL2X	13.46	159.09	113.41	75.05	15.63		11.90				
		2 Wire Unbundled HDSL Loop including manual service inquiry	1		L												1
		& facility reservation - Zone 3		3	UHL	UHL2X	26.00	159.09	113.41	75.05	15.63		11.90				
$\vdash \vdash$		Order Coordination for Specified Conversion Time (per LSR)	ļ		UHL	OCOSL		23.02								ļ	
		2 Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 1		1	UHL	UHL2W	9.97	134.40	80.69	60.64	9.12		11.90				
		2 Wire Unbundled HDSL Loop without manual service inquiry			l												
$\vdash \vdash$		and facility reservation - Zone 2	1	2	UHL	UHL2W	13.46	134.40	80.69	60.64	9.12		11.90	-	1	1	+
		2 Wire Unbundled HDSL Loop without manual service inquiry		_		1 11 11 2047	20.00	404.40	00.00	CO C4	0.40		44.00				
$\vdash \vdash$		and facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)	1	3	UHL	UHL2W	26.00	134.40	80.69	60.64	9.12		11.90	-	1	1	+
\vdash		CLEC to CLEC Conversion Charge without outside dispatch	 	1	UHL	OCOSL UREWO		23.02 86.12	40.39				11.90				+
 		HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIDIE	LOOP	UHL	UKEWU	1	გი.12	40.39	-	-		11.90	-	-	 	+
4		4 Wire Unbundled HDSL Loop including manual service inquiry	IBLE	LUUP	-		1			-	-			-	-	 	+
		and facility reservation - Zone 1	1	1	UHL	UHL4X	15.69	193.31	138.98	77.15	12.61		11.90				1
$\vdash \!$		4-Wire Unbundled HDSL Loop including manual service inquiry	1		OI IL	UNL4A	15.09	183.31	130.98	11.15	12.01		11.90	1	1		+
		and facility reservation - Zone 2	1	2	UHL	UHL4X	21.17	193.31	138.98	77.15	12.61	1	11.90		Ì		1
1 1		4-Wire Unbundled HDSL Loop including manual service inquiry	 		OI IL	OI IL4A	21.17	133.31	130.30	11.13	12.01		11.50		 	1	+
			1	1	i	ı	1			I	I	1	l	I	Ì	1	1
		and facility reservation - Zone 3		3	UHL	UHL4X	40.90	193.31	138.98	77.15	12.61		11.90				

ONROND	LED	NETWORK ELEMENTS - Florida										Ι -	_		ment: 2		bit: C
CATEGOR	Y	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		4-Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 1		1	UHL	UHL4W	15.69	168.62	115.47	62.74	11.22		11.90				<u> </u>
		4-Wire Unbundled HDSL Loop without manual service inquiry					04.47	100.00	445.47	00.74	44.00		44.00				
		and facility reservation - Zone 2		2	UHL	UHL4W	21.17	168.62	115.47	62.74	11.22		11.90				ļ
		4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		3	UHL	UHL4W	40.90	168.62	115.47	62.74	11.22		11.90				
		Order Coordination for Specified Conversion Time (per LSR)		3	UHL	OCOSL OCOSL	40.90	23.02	110.47	62.74	11.22		11.90				
		CLEC to CLEC Conversion Charge without outside dispatch		1	UHL	UREWO		86.12	40.39	1			11.90				
4-V		DS1 DIGITAL LOOP			OTIL	OKEWO		00.12	40.00				11.00				
		4-Wire DS1 Digital Loop - Zone 1		1	USL	USLXX	73.44	313.75	181.48	61,22	13.53		11.90				
		4-Wire DS1 Digital Loop - Zone 2			USL	USLXX	99.13	313.75	181.48	61.22	13.53		11.90				
	4	4-Wire DS1 Digital Loop - Zone 3			USL	USLXX	191.51	313.75	181.48	61.22	13.53		11.90		1		
		Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		23.02									
		CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		101.07	43.04				11.90				
4-V		19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															
		4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	26.39	161.56	108.85	67.08	15.56		11.90				
		4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	35.62	161.56	108.85	67.08	15.56		11.90				ļ
		Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	68.82	161.56	108.85		15.56		11.90				ļ
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 1			UDL	UDL56	26.39	161.56	108.85	67.08	15.56		11.90				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56	35.62	161.56	108.85	67.08	15.56		11.90				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	68.82	161.56	108.85	67.08	15.56		11.90				
		Order Coordination for Specified Conversion Time (per LSR)		4	UDL	OCOSL UDL64	20, 20	23.02 161.56	400.05	67.00	45.50		44.00				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 1 4 Wire Unbundled Digital Loop 64 Kbps - Zone 2			UDL UDL	UDL64	26.39 35.62	161.56	108.85 108.85	67.08 67.08	15.56 15.56		11.90 11.90				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 3			UDL	UDL64	68.82	161.56	108.85	67.08	15.56		11.90				
		Order Coordination for Specified Conversion Time (per LSR)		3	UDL	OCOSL	00.02	23.02	100.03	07.00	13.30		11.90				
		CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		102.11	49.74				11.90				+
2-V		Unbundled COPPER LOOP			002	O.KETTO		.02					11.00				
		2-Wire Unbundled Copper Loop/Short including manual service				1											
		nquiry & facility reservation - Zone 1		1	UCL	UCLPB	12.65	148.50	102.82	75.05	15.63		11.90				
		2-Wire Unbundled Copper Loop/Short including manual service															
		nquiry & facility reservation - Zone 2		2	UCL	UCLPB	17.08	148.50	102.82	75.05	15.63		11.90				
	2	Wire Unbundled Copper Loop/Short including manual service															
	i	nquiry & facility reservation - Zone 3		3	UCL	UCLPB	33.00	148.50	102.82	75.05	15.63		11.90				<u> </u>
		Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
		2-Wire Unbundled Copper Loop/Short without manual service															
		nquiry and facility reservation - Zone 1		1	UCL	UCLPW	12.65	123.81	70.09	60.64	9.12		11.90				
		2-Wire Unbundled Copper Loop/Short without manual service		_			4= 00		=								
		nquiry and facility reservation - Zone 2		2	UCL	UCLPW	17.08	123.81	70.09	60.64	9.12		11.90				ļ
		2-Wire Unbundled Copper Loop/Short without manual service		_	UCL	LICI DW	22.00	400.04	70.00	CO C4	0.40		11.90				
		nquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLPW	33.00	123.81 9.00	70.09 9.00	60.64	9.12		11.90				
		2-Wire Unbundled Copper Loop/Long - includes manual srvc.			UCL	UCLIVIC		9.00	9.00								
		nquiry and facility reservation - Zone 1		1	UCL	UCL2L	37.07	148.50	102.82	75.05	15.63		11.90				
		2-Wire Unbundled Copper Loop/Long - includes manual svc.		i i	002	COLLE	01.01	140.00	102.02	70.00	10.00		11.00				+
		nquiry and facility reservation - Zone 2		2	UCL	UCL2L	50.04	148.50	102.82	75.05	15.63		11.90				
		2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	i	nquiry and facility reservation - Zone 3		3	UCL	UCL2L	96.67	148.50	102.82	75.05	15.63		11.90				
	(Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
		2-Wire Unbundled Copper Loop/Long - without manual service								ĺ							
		nquiry and facility reservation - Zone 1		1	UCL	UCL2W	37.07	123.81	70.09	60.64	9.12		11.90				
		2-Wire Unbundled Copper Loop/Long - without manual service	1							1]		
		nquiry and facility reservation - Zone 2		2	UCL	UCL2W	50.04	123.81	70.09	60.64	9.12	ļ	11.90				
		2-Wire Unbundled Copper Loop/Long - without manual service	1		l <u>.</u> .	l <u>.</u>					_				1		
		nquiry and facility reservation - Zone 3	ļ	3	UCL	UCL2W	96.67	123.81	70.09	60.64	9.12		11.90				↓
		Order Coordination for Unbundled Copper Loops (per loop)	<u> </u>		UCL	UCLMC		9.00	9.00	 				ļ	 	ļ	↓
	- 19	CLEC to CLEC Conversion Charge without outside dispatch (UCL -Des)	1		UCL	UREWO		07.04	42.47				44.00		1		
	(COPPER LOOP		!	UCL	UKEWU		97.21	42.47			ļ	11.90				↓

UNBUND	DLED NETWORK ELEMENTS - Florida												Attachr	ment: 2	Exhi	bit: C
CATEGOR		Interi m	Zone	BCS	usoc			RATES(\$)			II.	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1 AM:		1	UCL	UCL4S	18.03	177.87	132.76	77.15	17.73		11.90				
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 2		2	UCL	UCL4S	24.34	177.87	132.76	77.15	17.73		11.90				
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 3		3	UCL	UCL4S	47.02	177.87	132.76	77.15	17.73		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)		5	UCL	UCLMC	47.02	9.00	9.00	77.13	17.73		11.30				
	4-Wire Copper Loop/Short - without manual service inquiry and			002	0020		0.00	0.00								
	facility reservation - Zone 1		1	UCL	UCL4W	18.03	153.18	100.03	62.74	11.22		11.90				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 2		2	UCL	UCL4W	24.34	153.18	100.03	62.74	11.22		11.90				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 3		3	UCL	UCL4W	47.02	153.18	100.03	62.74	11.22		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	47.02	9.00	9.00	02.14	11.22		11.50				†
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1		1	UCL	UCL4L	64.52	177.87	132.76	77.15	17.73		11.90				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2		2	UCL	UCL4L	87.09	177.87	132.76	77.15	17.73		11.90				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	168.25	177.87	132.76	77.15	17.73		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 1		1	UCL	UCL4O	64.52	153.18	100.03	62.74	11.22		11.90				
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 2		2	UCL	UCL4O	87.09	153.18	100.03	62.74	11.22		11.90				
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 3		3	UCL	UCL4O	168.25	153.18	100.03	62.74	11.22		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	100.20	9.00	9.00	02.14	11.22		11.00				
	CLEC to CLEC Conversion Charge without outside dispatch			UCL	UREWO		97.21	42.47				11.90				
LOOP MOI	DIFICATION															
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft			UAL, UHL, UCL, UEQ, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULM2L		0.00	0.00				11.90				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire															
	greater than 18k ft Unbundled Loop Modification Removal of Load Coils - 4 Wire			UCL, ULS, UEQ	ULM2G		343.12	343.12				11.90				
	less than or equal to 18K ft Unbundled Loop Modification Removal of Load Coils - 4 Wire			UHL, UCL	ULM4L		0.00	0.00				11.90				
	pair greater than 18k ft			UCL	ULM4G		343.12	343.12				11.90				
	Unbundled Loop Modification Removal of Bridged Tap Removal,			UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL,												
	per unbundled loop			USL	ULMBT		10.52	10.52				11.90				
SUB-LOOF											1					1
Su	Sub-Loop Distribution Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-															
	Up	- 1		UEANL	USBSA		487.23	487.23				11.90				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	ı		UEANL	USBSB		6.25	6.25				11.90				
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up	- 1		UEANL	USBSC		169.25	169.25				11.90				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up	1		UEANL	USBSD		38.65	38.65				11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN2	7.61	60.19	21.78	47.50	5.26		11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2			UEANL	USBN2	10.27	60.19	21.78	47.50	5.26		11.90				

CATEGORY RATE ELEMENTS Interest 200																	
ATECHNIST RATE ELEMENTS RATE CLIME	UNBUNDLE	D NETWORK ELEMENTS - Florida												Attachi	ment: 2	Exhi	oit: C
### PACT PLEMENTS Head Zone BCS USOC PACT Special Mineral River												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
## APTER_EMEMONS												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
March Marc			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
Sub-Lists Description Desc	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
15 15 15 15 15 15 15 15			""													Electronic-	Electronic-
Section Part																	Disc Add'l
Sec. Log Destitution Per 2 Was Analog Vasca Grade Loop 3 UPENAL USSSYD 19.65 60.19 21.76 47.50 5.50 11.90 11																D130 13t	DISC Add I
Sub-Logo Distriction for 2 Week Analog Vance Grands Loop							Boo	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
Const Constitution for Lithounded Shi-Loops, per sub-loop pay Size-Not 1985 6019 5,70 5,30 11,00							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
District Confidence for Unbounded Sub-Locus, per sub-roco pier		Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -															
Statute Stat		Zone 3		3	UEANL	USBN2	19.85	60.19	21.78	47.50	5.26		11.90				
Sub-Loop Distribution Per 4-Vivin Analog Visios Grade Loop - 2																	
September 1		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00								
Sub-Loop Datablator Per 4-Wer Analog Vices Grade Loop - 2		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
Description Control		Zone 1		1	UEANL	USBN4	8.12	68.83	30.42	49.71	6.60		11.90				
Sub-Loco Distribution for Unbundled Sub-Loco Depart		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
2016 Contribution for Unbounded Sub-Logo, per sub-loop pair UEANL USBM2 0.00		Zone 2		2	UEANL	USBN4	10.96	68.83	30.42	49.71	6.60		11.90				
Order Coordination for Unbundled Sub-Loops, per Bub-loop part UEANL USBINC S.00 0.00		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
Sub-Loop 2-Wine Introducting Network Cabler (NC)		Zone 3		3	UEANL	USBN4	21.18	68.83	30.42	49.71	6.60		11.90				
Sub-Loop 2-Wine Introducting Network Cabler (NC)							İ										
Sub-Loop 2-Wine Introducting Network Cabler (NC)		Order Coordination for Unbundled Sub-Loops, per sub-loop pair	1	1	UEANL	USBMC		9.00	9.00			1			I	Ì	
Chiefer Coordination for Unboundled Sub-Loop, per sub-loop part UEANL USBNC 0.00					UEANL		3.50			47.50	5.26		11.90				
Sale-Loop 4-Vive Intributinities Shi-Loop Date UEAN, USBR4 6.66 5.511 77.51 49.71 6.60 11.00		•															
Sub-Luce 4-Vive International Sub-Luce (RIC) UERN, USBR4 6.68 5.591 17.51 49.71 6.60 11.00		Order Coordination for Unbundled Sub-Loops, per sub-loop pair	1	1	UEANL	USBMC		9.00	9.00			1			I	Ì	
2 Vivic Copper Unbounded Sub-Loop Distribution - Zone 2			1		UEANL	USBR4	6.68	55.91	17.51	49.71	6.60		11.90				
2 Vivic Copper Unbounded Sub-Loop Distribution - Zone 2																	
2 Wire Copper Unbursdied Sub-Loop Britshution - Zone 2 1 2 UEF UCS2X		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00								
2 Wire Copper Unbursdied Sub-Loop Britshution - Zone 2 1 2 UEF UCS2X		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1	1	UEF	UCS2X	6.25	60.19	21.78	47.50	5.26		11.90				
2 Wire Cooper Undurided Sub-Loop, per sub-loop part UEF USBMC 9.00 9.			1	2	UEF		8.44	60.19	21.78	47.50	5.26		11.90				
Order Coordination for Unbundled Sub-Loops, per sub-loop pair UEF USBMC 9.00			1		UEF												
4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1 1 1 UEF UCS4X 5.20 68.83 30.42 49.71 6.60 11.90																	
4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		9.00	9.00								
A Wire Copper Unbundled Sub-Loop Detribution - Zone 2 2 UEF			-	1	UEF	UCS4X	5.20	68.83		49.71	6.60		11.90				
A Wire Copper Unbundled Sub-Loop Dearhounded Sub-Loops, per sub-loop pair UEF USBMC 9.00 9			1				7.02	68.83	30.42	49.71	6.60		11.90				
Order Coordination for Unbundled Sub-Loops, per sub-loop pair UEF USBMC 9.00			1	3	UEF					49.71			11.90				
Unbundled Sub-Loop Modification William UEF ULMZX										-							
Unbundled Sub-Loop Modification - 2-W Copper Dist Load UEF		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		9.00	9.00								
Unbundled Sub-Loop Modification - 2-W Copper Dist Load UEF ULMX 10.11 10.11 11.90 11.90 11.90	Unbur																
Col/Equip Removal per 2-W PR																	
Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W P R					UEF	ULM2X		10.11	10.11				11.90				
ColiEquip Removal per 4-W PR																	
Unbundled Sub-loop Modification - 2-wif-w Copper Dist Bridged Tap Removal, per PR unloaded UEF ULMAT 15.58 15.58 11.90 1					UEF	ULM4X		10.11	10.11				11.90				
Tap Removal, per PR unloaded																	
Unbundled Network Terminating Wire (UNTW) per Pair UENTW UENPP 0.286 18.02 11.90		Tap Removal, per PR unloaded			UEF	ULM4T		15.58	15.58				11.90				
Unbundled Network Terminating Wire (UNTW) per Pair	Unbur																
Network Interface Device (NID) Network Interface Device (NID) - 1-2 lines UENTW UND12 68.08 42.80 11.90					UENTW	UENPP	0.2286	18.02	18.02				11.90				
Network Interface Device (NID) - 1-2 lines	Netwo		1	1		İ				İ					İ	İ	
Network Interface Device (NID) - 1-6 lines			1	1	UENTW	UND12		68.08	42.80	İ			11.90		İ	İ	
Network Interface Device Cross Connect - 2 W					UENTW	UND16			85.20				11.90				
Network Interface Device Cross Connect - 4W																	
Sub-Loop Sub-Loop Feeder													11.90				
Sub-Loop Feeder USL-Feeder, DSO Set-up per Cross Box location - CLEC UEA, UDN, UCL, UDL, UDC USBFW 487.23 11.90 11.90 USL Feeder - DSO Set-up per Cross Box location - per 25 pair UEA, UDN, UCL, UDL, UDC USBFX 6.25 6.25 11.90 USL Feeder - DSI Set-up at DSX location, per DSI termination USL Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 UEA USBFA 8.05 92.75 51.24 58.45 13.07 11.90 USL Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 2 UEA USBFA USBFA 21.00 92.75 51.24 58.45 13.07 11.90 USL Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 3 3 UEA USBFA 21.00 92.75 51.24 58.45 13.07 11.90 USBFA 11.90 USBFA 21.00 92.75 51.24 58.45 13.07 11.90 USBFA 21.00 92.75 51.24 58.45 13.07 11.90 USBFA 21.00 92.75 51.24 58.45 13.07 11.90 USBFA 21.00 92.75 51.24 58.45 13.07 11.90 USBFA 21.00 92.75 51.24 58.45 13.07 11.90 USBFA 21.00 92.75 51.24 58.45 23.07 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 UEA USBFA 23.02 US	SUB-LOOPS		1	1		İ				İ					İ	İ	
USL-Feeder, DS0 Set-up per Cross Box location - CLEC UEA, UDN,UCL,UDL,UDC USBFW 487.23 11.90 11.90		oop Feeder	1	1		İ				İ					İ	İ	
Distribution Facility set-up					UEA,	1											
USL Feeder - DS0 Set-up per Cross Box location - per 25 pair Set-up S			1	1		USBFW		487.23				1	11.90		I	Ì	
Set-up																	
USL Feeder DS1 Set-up at DSX location, per DS1 termination			1			USBFX		6.25	6.25			İ	11.90		1		
Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1																	
Grade - Zone 1																	
Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice 2 UEA USBFA 10.87 92.75 51.24 58.45 13.07 11.90			1	1	UEA	USBFA	8.05	92.75	51.24	58.45	13.07		11.90		1		
Grade - Zone 2						1											
Voice Grade - Zone 3			1	2	UEA	USBFA	10.87	92.75	51.24	58.45	13.07	1	11.90		I	Ì	
Voice Grade - Zone 3		Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,						_									
Order Coordination for Specified Conversion Time, per LSR UEA OCOSL 23.02 Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice			1	3	UEA	USBFA	21.00	92.75	51.24	58.45	13.07	İ	11.90		1		
Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice																	
		Grade - Zone 1	1	1	UEA	USBFB	8.05	92.75	51.24	58.45	13.07	İ	11.90		1		

CATEGORY Unbu Grad Unbu Grad Unbu Voice Unbu Voice Unbu Grad	RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS RATE ELEMENTS REPUBLICANO PROBLEM PROB	Interi	2 3 1 2 3	BCS UEA UEA UEA UEA UEA	USBFB USBFB OCOSL USBFC	Rec 10.87 21.00	Nonrec First 92.75 92.75 23.02	RATES(\$) curring Add'I 51.24	Nonrecurring First 58.45	Disconnect Add'I 13.07	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR SOMAN	Incremental Charge - Manual Svc Order vs. Electronic- 1st	nent: 2 Incremental Charge - Manual Svc Order vs. Electronic- Add'I Rates(\$) SOMAN		it: C Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
Unbu Grad Unbu Grad Orde Unbu Voice Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ibundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR ibundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 ibundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 ibundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse itery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR ibundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 ibundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		2 3	UEA UEA UEA UEA	USBFB USBFB OCOSL USBFC	10.87	92.75 92.75	curring Add'l 51.24	First	Add'l	Submitted Elec per LSR	Submitted Manually per LSR SOMAN	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I Rates(\$)	Charge - Manual Svc Order vs. Electronic- Disc 1st	Manual Svc Order vs. Electronic- Disc Add'l
Unbu Grad Unbu Grad Orde Unbu Voice Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ibundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR ibundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 ibundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 ibundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse itery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR ibundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 ibundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		2 3	UEA UEA UEA UEA	USBFB USBFB OCOSL USBFC	10.87	92.75 92.75	curring Add'l 51.24	First	Add'l	Elec per LSR	Manually per LSR	Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs. Electronic- Add'I Rates(\$)	Manual Svc Order vs. Electronic- Disc 1st	Manual Svc Order vs. Electronic- Disc Add'l
Unbu Grad Unbu Grad Orde Unbu Voice Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ibundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR ibundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 ibundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 ibundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse itery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR ibundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 ibundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 ibundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		2 3	UEA UEA UEA UEA	USBFB USBFB OCOSL USBFC	10.87	92.75 92.75	curring Add'l 51.24	First	Add'l	per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'I Rates(\$)	Order vs. Electronic- Disc 1st	Order vs. Electronic- Disc Add'l
Grad Unbu Grad Unbu Grad Unbu Voice Unbu Voice Unbu Grad	ade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		3 1 2	UEA UEA UEA	USBFB OCOSL USBFC	10.87	92.75 92.75	Add'I 51.24	First	Add'l	•	SOMAN	Electronic- 1st OSS	Electronic- Add'l Rates(\$)	Electronic- Disc 1st	Electronic- Disc Add'l
Grad Unbu Grad Unbu Grad Unbu Voice Unbu Voice Unbu Grad	ade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		3 1 2	UEA UEA UEA	USBFB OCOSL USBFC	10.87	92.75 92.75	Add'I 51.24	First	Add'l	SOMEC		1st OSS	Add'I Rates(\$)	Disc 1st	Disc Add'l
Grad Unbu Grad Unbu Grad Unbu Voice Unbu Voice Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		3 1 2	UEA UEA UEA	USBFB OCOSL USBFC	10.87	92.75 92.75	Add'I 51.24	First	Add'l	SOMEC		OSS	Rates(\$)		
Grad Unbu Grad Unbu Grad Unbu Voice Unbu Voice Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		3 1 2	UEA UEA UEA	USBFB OCOSL USBFC	10.87	92.75 92.75	Add'I 51.24	First	Add'l	SOMEC				SOMAN	SOMAN
Grad Unbu Grad Unbu Grad Unbu Voice Unbu Voice Unbu Grad	ade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		3 1 2	UEA UEA UEA	USBFB OCOSL USBFC	10.87	92.75 92.75	51.24			SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
Grad Unbu Grad Unbu Grad Unbu Voice Unbu Voice Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		3 1 2	UEA UEA UEA	USBFB OCOSL USBFC	21.00	92.75		58.45	13.07		11.90				· ·
Unbu Grad Orde Unbu Voice Unbu Voice Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Orde Unbu Grad Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	bundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice ade - Zone 3 der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		3 1 2	UEA UEA UEA	USBFB OCOSL USBFC	21.00	92.75		30.43	13.07		11.90				
Grad Orde Unbu Voice Unbu Voice Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Orde Unbu Grad Orde Unbu Grad Orde Unbu Grad Unbu Grad Orde Unbu Grad	ade - Zone 3 der Coordination for Specified Time Conversion, per LSR blundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 blundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 blundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR blundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 blundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 blundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 blundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3		1 2	UEA UEA UEA	OCOSL USBFC			54.04								
Orde Unbu Voice Unbu Voice Unbu Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Orde Unbu Grad Orde Unbu Grad Orde Unbu Grad Orde Unbu Grad Unbu Grad	der Coordination for Specified Time Conversion, per LSR bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3		1 2	UEA UEA UEA	OCOSL USBFC				58.45	13.07		11.90				í
Unbu Voice Unbu Batte Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu	bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3		2	UEA UEA	USBFC			31.24	30.43	13.07		11.90				
Voice Unbu Voice Unbu Hatte Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Orde Unbu Grad Orde Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ice Grade - Zone 1 bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		2	UEA			20.02									f
Unbu Voice Unbu Batte Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Orde Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad	bundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, ice Grade - Zone 2 blundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 blundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 blundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2		Ī	UEA		8.05	92.75	51.24	58.45	13.07		11.90				í
Voice Unbu Batte Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu	ice Grade - Zone 2 bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3		Ī		LISBEC				00.10							
Unbu Batte Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	bundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3		3		OOD! O	10.87	92.75	51.24	58.45	13.07		11.90				í
Batte Orde Unbb Grad Unbb Grad Unbb Grad Unbb Grad Orde Unbb Grad Unbb Grad Unbb Grad Unbb Grad Unbb Grad Unbb Grad	ttery, Voice Grade - Zone 3 der Coordination For Specified Conversion Time, per LSR blundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 blundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 blundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 2 ade - Zone 3		3	ΙΙΕΔ												í
Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu	bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3				USBFC	21.00	92.75	51.24	58.45	13.07		11.90				1
Grad Unbu Grad Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	ade - Zone 1 bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3			UEA	OCOSL		23.02									
Unbu Grad Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad	bundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3		1													1
Grad Unbu Grad Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Unbu Unbu Unbu Unbu Unbu	ade - Zone 2 bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3		1	UEA	USBFD	17.26	106.92	64.46	63.54	14.83		11.90				
Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad	bundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice ade - Zone 3	1		-												
Grad Orde Unbb Grad Unbb Grad Unbb Grad Unbb Grad Unbb Grad Unbb	ade - Zone 3		2	UEA	USBFD	23.29	106.92	64.46	63.54	14.83		11.90				
Orde Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu Grad Unbu				l	I											í
Unbu Grad Unbu Grad Unbu Grad Orde	der Coordination For Specified Conversion Time, Per LSR		3	UEA	USBFD	45.00	106.92	64.46	63.54	14.83		11.90				
Grad Unbu Grad Unbu Grad Unbu Grad Unbu			ļ	UEA	OCOSL		23.02									
Unbu Grad Unbu Grad Orde Unbu	bundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		1		HODEE	47.00	400.00	04.40	00.54	44.00		44.00				ł
Grad Unbu Grad Orde Unbu	ade - Zone 1		1	UEA	USBFE	17.26	106.92	64.46	63.54	14.83		11.90				
Unbu Grad Orde Unbu	bundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice ade - Zone 2		2	UEA	USBFE	23.29	106.92	64.46	63.54	14.83		11.90				í
Grad Orde Unbu	bundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			UEA	USBFE	23.29	106.92	04.40	63.54	14.03		11.90				
Orde Unbu	ade - Zone 3		3	UEA	USBFE	45.00	106.92	64.46	63.54	14.83		11.90				í
Unbu	der Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL	45.00	23.02	04.40	03.54	14.00		11.30				
	bundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	17.04	109.71	66.68	60.21	12.49		11.90				
I Unbi	bundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN	USBFF	23.00	109.71	66.68	60.21	12.49		11.90				
	bundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	44.43	109.71	66.68	60.21	12.49		11.90				í
Orde	der Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		23.02									
Unbı	bundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDC	USBFS	17.04	109.71	66.68	60.21	12.49		11.90				I
Unbi	bundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		2	UDC	USBFS	23.00	109.71	66.68	60.21	12.49		11.90				ĺ .
	bundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		3	UDC	USBFS	44.43	109.71	66.68	60.21	12.49		11.90				<u> </u>
	bundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	USL	USBFG	46.27	133.77	78.02	85.16	21.21		11.90				Ļ
	bundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2	USL	USBFG	62.45	133.77	78.02	85.16	21.21		11.90				
	bundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	USL	USBFG	120.65	133.77	78.02	85.16	21.21		11.90		ļ		
	der Coordination For Specified Conversion Time, Per LSR			USL	OCOSL	7.0-	23.02	10.01	50.51	10.00		44.00		ļ		
	bundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	7.25	85.27	42.24	58.54	10.82		11.90				
Unbl	bundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		2	UCL	USBFH	9.79	85.27	42.24	58.54	10.82		11.90				ł
L Inhi	bundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone			UCL	USDFF	9.79	03.27	42.24	30.34	10.82		11.90				
2	ibariatea Sub-Loop i eeder Loop, 2-vville Copper Loop - 2011e		3	UCL	USBFH	18.92	85.27	42.24	58.54	10.82		11.90				1
Orde	der Coordination For Specified Conversion Time, per LSR	1	- 3	UCL	OCOSL	10.32	23.02	42.24	30.34	10.02		11.50				
	b-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	14.22	99.66	57.20	60.98	12.28		11.90				
	b-Loop Feeder - Per 4-Wire Copper Loop - Zone 2			UCL	USBFJ	19.20	99.66	57.20	60.98	12.28		11.90				í
	b-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	37.09	99.66	57.20	60.98	12.28		11.90				i
	der Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		23.02		1			,,,				1
	b-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	18.68	100.62	58.16	63.54	14.83		11.90				i
Sub-	b-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2	UDL	USBFN	25.21	100.62	58.16	63.54	14.83		11.90				i
	b-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	48.71	100.62	58.16	63.54	14.83		11.90				
	b-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -			-												
Zone			1	UDL	USBFO	18.68	100.62	58.16	63.54	14.83		11.90				
	b-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		1 .	l	1 7				I							1
	ne 2		2	UDL	USBFO	25.21	100.62	58.16	63.54	14.83		11.90				
	b-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -					40 = :	400	=0 :-								ł
	ne 3		3	UDL	USBFO	48.71	100.62	58.16	63.54	14.83		11.90		ļ		
				UDL	OCOSL		23.02							ļ		
Sub- Zone	der Coordination For Specified Time Conversion, per LSR b-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -	1	1	UDL	USBFP	18.68			1					ī		1

UNBUNDLE	D NETWORK ELEMENTS - Florida										Ι -			nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		_													
	Zone 2		2	UDL	USBFP	25.21	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -			UDI	HODED	40.74	400.00	50.40	00.54	44.00		44.00				
	Zone 3		3	UDL	USBFP OCOSL	48.71	100.62	58.16	63.54	14.83		11.90				
SUB-LOOPS	Order Coordination For Specified Conversion Time, per LSR			UDL	UCUSL		23.02									+
	Dop Feeder															
Sub-LC	Sub Loop Feeder - DS3 - Per Mile Per Month			UE3	1L5SL	15.69										
	Sub Loop Feeder - DS3 - Fer Mile Fer Month Sub Loop Feeder - DS3 - Facility Termination Per Month	-		UE3	USBF1	347.59	3,402.59	407.15	166.83	94.58		11.90				
	Sub Loop Feeder – STS-1 – Per Mile Per Month	-i-		UDLSX	1L5SL	15.69	0,402.00	407.10	100.00	04.00		11.00				+
+	Sub Loop Feeder - STS-1 - Facility Termination Per Month	i	1	UDLSX	USBF7	402.09	3,402.59	407.15	166.83	94.58		11.90				†
	Sub Loop Feeder – OC-3 – Per Mile Per Month	Ė			1L5SL	11.90	2, 102.00	.00		230						1
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per	<u> </u>							1							
	Month	- 1	1	UDLO3	USBF5	62.98										
	Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3	USBF2	547.22	3,402.59	407.15	166.83	94.58		11.90				
	Sub Loop Feeder - OC-12 - Per Mile Per Month			UDL12	1L5SL	14.65										
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per			_				-								
	Month	- 1		UDL12	USBF6	502.47										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	- 1			USBF3	1,577.00	3,402.59	407.15	166.83	94.58		11.90				
	Sub Loop Feeder - OC-48 - Per Mile Per Month	I		UDL48	1L5SL	48.06										ļ
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per															
	Month	I		UDL48	USBF9	251.80										ļ
	Sub Loop Feeder - OC-48 - Facility Termination Per Month	- 1		UDL48	USBF4	1,589.00	3,588.59	407.15	168.35	95.43		11.90				ļ
	Sub Loop Feeder - OC-12 Interface On OC-48			UDL48	USBF8	331.15	804.98	407.15	168.35	95.43		11.90				
UNBUNDLED I	LOOP CONCENTRATION				LIOTOA	110.10	050.40	050.40				44.00				
	Unbundled Loop Concentration - System A (TR008)			ULC ULC	UCT8A UCT8B	449.49 53.44	359.42 149.76	359.42 149.76				11.90 11.90				├
	Unbundled Loop Concentration - System B (TR008) Unbundled Loop Concentration - System A (TR303)				UCT3A	487.33	359.42	359.42				11.90				
	Unbundled Loop Concentration - System A (TR303)				UCT3B	90.05	149.76	149.76				11.90				
	Unbundled Loop Concentration - DS1 Loop Interface Card				UCTCO	5.04	71.70	51.52	18.49	4.82		11.90				
	Unbundled Loop Concentration - ISDN Loop Interface (Brite			OLC	00100	3.04	71.70	31.32	10.45	4.02		11.90				
	Card)			UDN	ULCC1	8.00	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - UDC Loop Interface (Brite			ODIT	OLOO1	0.00	10.00	10.00	0.77	0.70		11.50				
	Card)			UDC	ULCCU	8.00	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration2 Wire Voice-Loop Start or			-		2.23				2.70						
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	2.00	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery															
	Loop Interface (SPOTS Card)			UEA	ULCCR	11.90	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface															
	(Specials Card)			UEA	ULCC4	7.10	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	34.68	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop									·						
	Interface			UDL	ULCC7	10.51	16.59	16.50	6.77	6.73	ļ	11.90				
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop		1	l	l I] _]	_						
ļ	Interface			UDL	ULCC5	10.51	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop			LIBI												
LINE OTHER S	Interface			UDL	ULCC6	10.51	16.59	16.50	6.77	6.73		11.90			ļ	
UNE UTHER, F	PROVISIONING ONLY - NO RATE NID - Dispatch and Service Order for NID installation		-	UENTW	UNDBX	0.00	0.00		 							
			 						 							
 	UNTW Circuit Id Establishment, Provisioning Only - No Rate		 	UENTW	UENCE	0.00	0.00		 		 	 				
	Unbundled Contract Name, Provisioning Only - No Rate			UEANL,UEF,UEQ,U ENTW	UNECN	0.00	0.00					1				
LINE OTHER	PROVISIONING ONLY - NO RATE			LINI VV	CINECIN	0.00	0.00		+		}	-			1	
J.I.E. OTTIER, F	NOTICIONINO ONET - NO NATE				1				 		 					
				UAL,UCL,UDC,UDL,								1				
	Unbundled Contact Name, Provisioning Only - no rate			UDN,UEA,UHL,ULC	UNECN	0.00	0.00									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no			,,		0.00	3.55		1							1

UNBUNI	DLE	NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEGOR	RY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
							Rec		curring		Disconnect				Rates(\$)		
		Habita diad Cub Lasa Fandar A Wire Cases Batti historia						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			UEA.USL.UCL.UDL	USBFR	0.00	0.00									
		Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
		Unbundled DS1 Loop - Expanded Superframe Format option -															
		no rate			USL	CCOEF	0.00	0.00									
HIGH CAP	ACII	Y UNBUNDLED LOCAL LOOP High Capacity Unbundled Local Loop - DS3 - Per Mile per															
		month			UE3	1L5ND	10.92										
		High Capacity Unbundled Local Loop - DS3 - Facility															
		Termination per month			UE3	UE3PX	386.88	556.37	343.01	139.13	96.84		11.90				
		High Capacity Unbundled Local Loop - STS-1 - Per Mile per month			UDLSX	1L5ND	10.92										
		High Capacity Unbundled Local Loop - STS-1 - Facility		<u> </u>						1							
		Termination per month			UDLSX	UDLS1	426.60	556.37	343.01	139.13	96.84		11.90			1.83	
LOOP MA																	
		Loop Makeup - Preordering Without Reservation, per working or spare facility gueried (Manual).			UMK	UMKLW		52.17	52.17								
		Loop Makeup - Preordering With Reservation, per spare facility			O.V.	O.V.II. K.E.V.		02.11	02.11								
		queried (Manual).			UMK	UMKLP		55.07	55.07								
		Loop MakeupWith or Without Reservation, per working or			UMK	DOLIMIZ		0.0704	0.0704								
UNRUNDI	FD D	spare facility queried (Mechanized) PEDICATED TRANSPORT			UMK	PSUMK		0.6784	0.6784								
		INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	od - below DS3=one	month, DS3/	STS-1=four mo	nths									
IN		OFFICE CHANNEL - DEDICATED TRANSPORT		Ĭ													
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
		Per Mile per month Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			U1TVX	1L5XX	0.0091										
		Facility Termination			U1TVX	U1TV2	25.32	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade			011111	011112	20.02	17.00	00	10.01	7.00		11.00				
		Rev Bat Per Mile per month			U1TVX	1L5XX	0.0091										
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination	-		U1TVX	U1TR2	25.32	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -			UTIVA	UTIKZ	25.52	47.33	31.76	10.31	7.03		11.90				
		Per Mile per month			U1TVX	1L5XX	0.0091										
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade															
		- Facility Termination Interoffice Channel - Dedicated Transport - 56 kbps - per mile			U1TVX	U1TV4	22.58	47.35	31.78	18.31	7.03		11.90				
		per month			U1TDX	1L5XX	0.0091										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
		Termination		<u> </u>	U1TDX	U1TD5	18.44	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			U1TDX	1L5XX	0.0091			1							
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility		1	STIDA	ILUAA	0.0031										
		Termination			U1TDX	U1TD6	18.44	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			LIATDA	41.577	0.1053			1							
		month Interoffice Channel - Dedicated Tranport - DS1 - Facility		1	U1TD1	1L5XX	0.1856			 							
		Termination			U1TD1	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per															
		month		ļ	U1TD3	1L5XX	3.87			ļ							<u> </u>
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			U1TD3	U1TF3	1,071.00	335.46	219.28	72.03	70.56		11.90				
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per		<u> </u>			.,57 1.00	300.70	210.20	72.55	70.00		11.50				
		month		<u> </u>	U1TS1	1L5XX	3.87										
		Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination			U1TS1	U1TFS	1.056.00	335.46	219.28	72.03	70.56		44.00				
1.0		CHANNEL - DEDICATED TRANSPORT		 	01101	UIIFO	00.060,1	333.46	219.28	12.03	70.56		11.90				
		LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing	g perio	d - belo	w DS3=one month.	DS3/STS-1=1	four months			1							
		Local Channel - Dedicated - 2-Wire Voice Grade - Zone 1		1	ULDVX	ULDV2	21.94	265.84	46.97	37.63	4.00		11.90				

CATEGORY	RATE ELEMENTS RATE ELEMENTS Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat	Interi m	Zone	BCS	USOC						Svc Order Submitted	Svc Order Submitted	Incremental	ment: 2 Incremental Charge -	Incremental Charge -	Incremental Charge -
CATEGORY	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat		Zone	BCS	usoc										Charge -	Channa
CATEGORY	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat		Zone	BCS	usoc											. Charge -
CATEGORY	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat		Zone	BCS	USOC						Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat	m						RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat										per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat															
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat												1st	Add'l	Disc 1st	Disc Add'l
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat					_	Nonrec	urrina	Nonrecurring	Disconnect			oss	Rates(\$)		
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat	_				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat		2	ULDVX	ULDV2	29.62	265.84	46.97	37.63	4.00		11.90				
			3	UNDVX	ULDV2	57.22	265.84	46.97	37.63	4.00		11.90				
	Zone 1		1	ULDVX	ULDR2	21.94	265.84	46.97	37.63	4.00		11.90				i
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat															
	Zone 2		2	ULDVX	ULDR2	29.62	265.84	46.97	37.63	4.00		11.90				1
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat															
1 1	Zone 3		3	ULDVX	ULDR2	57.22	265.84	46.97	37.63	4.00		11.90				1
	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 1		1	UNDVX	ULDV4	22.81	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 2			UNDVX	ULDV4	30.79	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 3		3	UNDVX	ULDV4	59.48	266.54	47.67	44.22	5.33		11.90				
l i	Local Channel - Dedicated - DS1 - Zone 1		1	ULDD1	ULDF1	35.28	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated - DS1 - Zone 2		2	ULDD1	ULDF1	47.63	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated - DS1 - Zone 3		3	ULDD1	ULDF1	92.01	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated - DS3 - Per Mile per month			ULDD3	1L5NC	8.50										
	Local Channel - Dedicated - DS3 - Facility Termination			ULDD3	ULDF3	531.91	556.37	343.01	139.13	96.84		11.90				
	Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	8.50										
	Local Channel - Dedicated - STS-1 - Facility Termination			ULDS1	ULDFS	540.69	556.37	343.01	139.13	96.84		11.90				
DARK FIBER	•															
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															1
	Thereof per month - Local Channel			UDF	1L5DC	55.04										i
	NRC Dark Fiber - Local Channel			UDF	UDFC4		751.34	193.88	356.21	230.11		11.90				i
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															1
	Thereof per month - Interoffice Channel			UDF	1L5DF	26.85										i
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		751.34	193.88	356.21	230.11		11.90				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Loop			UDF	1L5DL	55.04										1
	NRC Dark Fiber - Local Loop			UDF	UDFL4		751.34	193.88	356.21	230.11		11.90				1
8XX ACCESS T	EN DIGIT SCREENING															1
	8XX Access Ten Digit Screening, Per Call			OHD		0.0006252										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX															i
	Number Reserved			OHD	N8R1X		4.15	0.70				11.90				<u></u>
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O															i
	POTS Translations			OHD			8.78	1.18	5.77	0.70		11.90				
	8XX Access Ten Digit Screening, Per 8XX No. Established With															i
\vdash	POTS Translations	ļ		OHD	N8FTX		8.78	1.18	5.77	0.70		11.90		ļ		
1 1	8XX Access Ten Digit Screening, Customized Area of Service			0.15												1
\vdash	Per 8XX Number			OHD	N8FCX		4.15	2.07				11.90				
1 1	8XX Access Ten Digit Screening, Multiple InterLATA CXR			OLID.	NOTATI											1
\vdash	Routing Per CXR Requested Per 8XX No.	<u> </u>		OHD	N8FMX		4.85	2.78				11.90				
\vdash	8XX Access Ten Digit Screening, Change Charge Per Request	<u> </u>		OHD	N8FAX		4.85	0.70				11.90				
	8XX Access Ten Digit Screening, Call Handling and Destination			OLID	NOEDY							44.00				1
\vdash	Features	<u> </u>		OHD	N8FDX		4.15	4.15				11.90		1		
1 1	OVA A Tr. Birit O (OFI N. Bull			OLID	I	0.00000=0					1	1		Ì		i
\vdash	8XX Access Ten Digit Screening, w/ 8FL No. Delivery, per query	}		OHD	+	0.0006252								-		
	8XX Access Ten Digit Screening, w/ POTS No. Delivery, per			OLID	I	0.0000050					1	1		Ì		i
LINE INFORM	query	<u> </u>		OHD	+	0.0006252								1		
LINE INFORMA	ITION DATA BASE ACCESS (LIDB) LIDB Common Transport Per Query	 	l .	OQT	+	0.0000203					ļ	ļ		1		
\vdash					+	0.0000203										
	LIDB Validation Per Query	<u> </u>	.	OQU OQT, OQU	NRPBX	0.0136959	55.13	55.13	55.13	55.13		11.90				
SIGNALING (C	LIDB Originating Point Code Establishment or Change	-		UQ1, UQU	INCLDY		55.13	55.13	55.13	55.13		11.90		 		
JIGNALING (C	CCS7 Signaling Termination, Per STP Port	 		UDB	PT8SX	135.05					-	 		-		
\vdash	CCS7 Signaling Termination, Per STP Port CCS7 Signaling Usage, Per TCAP Message	1		UDB	F 100A	0.0000607								-		
\vdash	CCS7 Signaling Osage, Per TCAP Message CCS7 Signaling Connection, Per link (A link)	-		UDB	TPP++	17.93	43.57	43.57	18.31	18.31		11.90		 		
\vdash	CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D	}	 	סטס	177++	17.93	43.57	43.57	18.31	18.31	1	11.90		1		i
	link)			UDB	TPP++	17.93	43.57	43.57	18.31	18.31	1	11.90		1		1
\vdash	CCS7 Signaling Usage, Per ISUP Message	1		UDB	IFFTT	0.0000152	43.57	45.57	10.31	10.31		11.90				
	CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	694.32								1		

UNBUNDLE	D NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CCS7 Signaling Point Code, per Originating Point Code															İ
	Establishment or Change, per STP affected			UDB	CCAPO		46.03	46.03	46.03	46.03		11.90				
E911 SERVICE						21.21	205.01									
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 1					21.94	265.84	46.97	37.63	4.00		11.90				-
-	Local Channel - Dedicated - 2-wr Voice Grade - Zone 2					29.62	265.84	46.97	37.63	4.00	1	11.90			-	
-	Local Channel - Dedicated - 2-wr Voice Grade - Zone 3 Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile				-	57.22 0.0091	265.84	46.97	37.63	4.00		11.90			-	
-	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Nille Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility				-	0.0091									-	
	Termination					25.32	47.35	31.78	18.31	7.03		11.90				
	Local Channel - Dedicated - DS1 - Zone 1		1		1	35.28	216.65	183.54	21.47	19.05	1	11.90				
	Local Channel - Dedicated - DS1 - Zone 2					47.63	216.65	183.54		19.05		11.90				
	Local Channel - Dedicated - DS1 - Zone 3				1	92.01	216.65	183.54		19.05		11.90	1	1	1	t
	Interoffice Transport - Dedicated - DS1 Per Mile				1	0.1856	2.0.00	.00.07		.5.50			1	1	1	t
			†		1	21.1300			1				1	1	1	
	Interoffice Transport - Dedicated - DS1 Per Facility Termination					88.44	105.54	98.47	21.47	19.05		11.90	1	1	I	1
CALLING NAM	IE (CNAM) SERVICE				İ				1						1	
	CNAM For DB Owners - Service Establishment			OQV			25.35	25.35	19.01	19.01		11.90				
	CNAM For Non DB Owners - Service Establishment			OQV			25.35	25.35	19.01	19.01		11.90				
	CNAM For DB Owners - Service Provisioning With Point Code															
	Establishment			OQV			1,592.00	1,177.00	352.36	259.09		11.90				
	CNAM For Non DB Owners - Service Provisioning With Point															
	Code Establishment			OQV			546.51	393.82	358.06	259.09		11.90				
	CNAM for DB Owners, Per Query			OQV		0.001024										
	CNAM for Non DB Owners, Per Query			OQV		0.001024										
LNP Query Se																
	LNP Charge Per query			OQV		0.000852										
	LNP Service Establishment Manual						13.83	13.83	12.71	12.71		11.90				
	LNP Service Provisioning with Point Code Establishment						655.50	334.88	297.03	218.40		11.90				
OPERATOR CA	ALL PROCESSING		<u> </u>													
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					4.00										
					-	1.20										
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1 24										
	Oper. Call Processing - Fully Automated, per Call - Using BST		<u> </u>			1.24										
	LIDB					0.20										
	Oper. Call Processing - Fully Automated, per Call - Using					0.20										
	Foreign LIDB					0.20										
INWARD OPER	RATOR SERVICES					0.20										
111111111111111111111111111111111111111	Inward Operator Services - Verification, Per Call					1.00										
	Inward Operator Services - Verification and Emergency Interrupt				İ				1				1	İ	1	
	- Per Call					1.95										
BRANDING - C	PERATOR CALL PROCESSING															
	/ based CLEC															
	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				11.90				
	Loading of Custom Branded OA Announcement per shelf/NAV															
	per OCN				CBAOL		500.00	500.00				11.90				
UNEP																
	Recording of Custom Branded OA Announcement						7,000.00	7,000.00				11.90				
_	Loading of Custom Branded OA Announcement per shelf/NAV]	1	_	1
	per OCN						500.00	500.00			<u> </u>	11.90				1
	SSISTANCE SERVICES				 				ļ				ļ	ļ	ļ	1
DIREC	TORY ASSISTANCE ACCESS SERVICE				 				ļ				ļ	ļ	ļ	1
	Directory Assistance Access Service Calls, Charge Per Call		 		ļ	0.275			ļ						.	1
DIREC	TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (D	ACC)			ļ				ļ		ļ					
	Directory Assistance Call Completion Access Service (DACC),					0.10									1	1
DIDECTORY	Per Call Attempt		<u> </u>		1	0.10			ļ	-	}		 	 	!	
	SSISTANCE SERVICES				1				ļ		<u> </u>		 	ļ	-	
DIREC	TORY ASSISTANCE DATA BASE SERVICE (DADS)		<u> </u>		1	201			1						-	├
	Directory Assistance Data Base Service Charge Per Listing		1			0.04			L	l	1			l	1	

UNBL	JNDLE	D NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEC	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
DDANI	DING D	Directory Assistance Data Base Service, per month IRECTORY ASSISTANCE				DBSOF	150.00										
BRANI		Based CLEC															
	raciiity	Recording and Provisioning of DA Custom Branded															+
		Announcement			AMT	CBADA		6,000.00	6,000.00				11.90				
		Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00				11.90				
	UNEP (
		Recording of DA Custom Branded Announcement						3,000.00	3,000.00				11.90				
		Loading of DA Custom Branded Announcement per Switch per						4 4=0 00	==								
CELEC	TIVE RO	OCN						1,170.00	1,170.00				11.90				-
SELEC	JIIVE K	Selective Routing Per Unique Line Class Code Per Request Per															+
		Switch				USRCR		93.55	93.55	12.71	12.71		11.90				
VIRTU	AL COL	OCATION															†
		Virtual Collocation - Application Cost			AMTFS	EAF		4,122.00	1,249.00				11.90				
		Virtual Collocation - Cable Installation Cost, per cable			AMTFS	ESPCX	12.45	965.00					11.90				
		Virtual Collocation - Floor Space, per sq. ft.			AMTFS	ESPVX	4.25										
		Virtual Collocation - Power, per fused amp			AMTFS	ESPAX	6.95										
		Virtual Collocation - Cable Support Structure, per entrance cable			AMTFS	ESPSX	13.35										
		Virtual Collocation - 2-wire Cross Connects (loop)			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, AMTFS, UDL, UNCVX, UNCDX, UNCNX	UEAC2	0.0502	11.57	11.57				11.90				
		Virtual Collocation - 4-wire Cross Connects (loop)			UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX	UEAC4	0.0502	11.57	11.57				11.90				
		Virtual Collocation - 2-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC2F	6.71	2,431.00					11.90				
					AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12,												
	 	Virtual Collocation - 4-Fiber Cross Connects			ULD48, UDF	CNC4F	6.71	2,431.00					11.90				₩
		Virtual collocation - Special Access & UNE, cross-connect per DS1			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL, UNLD1	CNC1X	7.50	155.00	14.00				11.90				
		Virtual collocation - Special Access & UNE, cross-connect per DS3			USL,ULC,AMTFS,U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	56.25	151.90	11.83				11.90				
		Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable															
	1	Support Structure, per linear foot			AMTFS,CLO	VE1CB	0.0028										<u> </u>
		Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft			AMTFS, CLO	VE1CD	0.0041										
		Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable			AMTFS	VE1CC		535.54					11.90				
		Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			ANTEO	VE40E											
	 	Cable Support Structure, per cable Virtual Collocation Cable Records - per request			AMTFS AMTFS	VE1CE VE1BA		535.54 1,525.00	1,525.00	267.08	267.08		11.90				+

ONBONDER	ED NETWORK ELEMENTS - Florida												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable															
	record			AMTFS	VE1BB		656.50	656.50	379.78	379.78						
	Virtual Collocation Cable Records - VG/DS0 Cable, per each															
	100 pair			AMTFS	VE1BC		9.66	9.66	11.84	11.84						
	Virtual Collocation Cable Records - DS1, per T1TIE			AMTFS	VE1BD		4.52	4.52	5.54	5.54						
	Virtual Collocation Cable Records - DS3, per T3TIE			AMTFS	VE1BE		15.82	15.82	19.40	19.40						
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber			AMTEC	VE1BF		169.67	169.67	454.00	154.89						
	records Virtual collocation - Security Escort - Basic, per quarter hour			AMTFS AMTFS	SPTBQ	-	169.67	169.67	154.89	154.89		11.90			-	
	virtual collocation - Security Escort - Basic, per quarter flour			AIVITO	SPIBQ		10.09		1			11.90				
	Virtual collocation - Security Escort - Overtime, per quarter hour	1		AMTFS	SPTOQ		13.64				1	11.90		1	I	
	Virtual Collocation - Occurry Escort - Overtime, per qualter flour			, uviii O	51 100	 	13.04		 		 	11.50		 	t	t
	Virtual collocation - Security Escort - Premium, per quarter hour	1		AMTFS	SPTPQ		16.40				1	11.90		1	I	
		1		T					†					1	1	1
	Virtual Collocation - DS-1/DCS Cross Connects, PER 28 CKTS	l		AMTFS	VE11S	226.39	1,950.00					11.90			1	
	2															
	Virtual Collocation - DS-1.DSX Cross Connects, PER 28 CKTS			AMTFS	VE11X	11.51	1,950.00					11.90				
	Virtual Collocation - DS-3/DCS Cross Connects, PER CKT			AMTFS	VE13S	56.97	528.00					11.90				
	Virtual Collocation - DS-3/DSC Cross Connects, PER CKT			AMTFS	VE13X	10.06	528.00					11.90				
	Virtual collocation - Maintenance in CO - Basic, per quarter hour			AMTFS	SPTRE		10.89					11.90				
	Virtual collocation - Maintenance in CO - Overtime, per quarter															
	hour			AMTFS	SPTOE		13.64					11.90				
	Virtual collocation - Maintenance in CO - Premium per quarter															
VIRTUAL COL	hour			AMTFS	SPTPE		16.40					11.90				
VIRTUAL COL	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-															
	Wire Analog - Res			UEPSR	VE1R2	0.524	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-	-		ULFSK	VLTINZ	0.324	11.57	11.57				11.90				
	Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.524	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			OLI GI	VETICE	0.024	11.07	11.07				11.00				
	Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.524	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire						-									
	Analog Bus			UEPSB	VE1R2	0.524	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire															
	ISDN			UEPSX	VE1R2	0.524	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire															
	ISDN			UEPTX	VE1R2	0.524	11.57	11.57				11.90		ļ		
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire	1		LIEBEY	VE45:						1					
VIDTUAL CC	ISDN DS1	 		UEPEX	VE1R4	0.524	11.57	11.57				11.90			1	1
VIRTUAL COL	Virtual Collocation-2 Wire Cross Connects (Loop) for Line	 		1	+	 									 	
	Splitting	l		UEPSR, UEPSB	VE1LS	0.0297	33.86	31.95				11.90			1	
PHYSICAL CO		1		OLFON, UEFOB	VL ILO	0.0297	33.00	31.95	+		-	11.90		1	 	
oioal oc	Physical Collocation-2 Wire Cross Connects (Loop) for Line			<u> </u>	+	 								 	 	1
	Splitting			UEPSR. UEPSB	PE1LS	0.0276	8.22	7.22	5.74	4.58		11.90				
AIN SELECTI	VE CARRIER ROUTING			02. 0. t, 02. 02		0.02.0	0.22		0			11.00				
	Regional Service Establishment			SRC	SRCEC		193,444.00		7,737.00			11.90				
	End Office Establishment			SRC	SRCEO		187.36	187.36	0.69	0.69		11.90				
	Query NRC, per query			SRC		0.0031868										
AIN - BELLSC	OUTH AIN SMS ACCESS SERVICE							•		•						
	AIN SMS Access Service - Service Establishment, Per State,	l]							1					
	Initial Setup			A1N	CAMSE		43.56	43.56	44.93	44.93		11.90				
		1		1			_				1			1	I	
	AIN SMS Access Service - Port Connection - Dial/Shared Access	ļ		A1N	CAMDP	ļ	8.64	8.64	10.03	10.03		11.90			-	-
	AIN SMS Access Service - Port Connection - ISDN Access	 		A1N	CAM1P	1	8.64	8.64	10.03	10.03		11.90		1	!	!
	AIN SMS Access Service - User Identification Codes - Per User ID Code	l	1	A1N	CAMAU		38.66	38.66	29.88	29.88		11.90		Ì	1	1

UNBUNDLE	D NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			A1N	CAMRC		75.10	75.10	12.93	12.93		11.90				l
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)			AIN	CAIVIRC	0.0028	75.10	75.10	12.93	12.93		11.90				
	AIN SMS Access Service - Storage, Fer Unit (100 Kilobytes) AIN SMS Access Service - Session, Per Minute					0.7809										
	AIN SMS Access Service - Company Performed Session, Per					0.7005										
	Minute					0.4609										l
AIN - BELLSOU	JTH AIN TOOLKIT SERVICE															
	AIN Toolkit Service - Service Establishment Charge, Per State,															
	Initial Setup			CAM	BAPSC		43.56	43.56	44.93	44.93		11.90				
	AIN Toolkit Service - Training Session, Per Customer				BAPVX		8,439.00	8,439.00				11.90				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, Term. Attempt				BAPTT		8.64	8.64	10.03	10.03		11.90	ļ		ļ	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DARTS											1
	DN, Off-Hook Delay				BAPTD		8.64	8.64	10.03	10.03	1	11.90	 	-	 	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN. Off-Hook Immediate				BAPTM		8.64	8.64	10.03	10.03		11.90				1
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPTM		8.64	8.64	10.03	10.03	-	11.90	-	-	-	
	DN, 10-Digit PODP				ВАРТО		38.06	38.06	15.86	15.86		11.90				l
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAFTO		36.00	36.00	13.80	13.80		11.90				
	DN. CDP				BAPTC		38.06	38.06	15.86	15.86		11.90				ĺ
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DAI 10		30.00	30.00	15.00	13.00		11.50				
	DN, Feature Code				BAPTF		38.06	38.06	15.86	15.86		11.90				
	AIN Toolkit Service - Query Charge, Per Query					0.0535927	00.00	00.00	10.00	10.00		11.00				
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit															
	Subscription, Per Node, Per Query					0.0063698										İ
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access															
	Account, Per 100 Kilobytes					0.06										
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
	Subscription			CAM	BAPMS	8.34	8.64	8.64	6.08	6.08		11.90				
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service															İ
	Subscription			CAM	BAPLS	3.73	9.56	9.56				11.90				
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service			0444	DARRO	4.70	0.04	0.04	0.00	0.00		44.00				İ
	Subscription AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit			CAM	BAPDS	4.73	8.64	8.64	6.08	6.08		11.90				—
	Service Subscription			CAM	BAPES	0.12	9.56	9.56				11.90				
ENHANCED EX	(TENDED LINK (EELs)			CAIVI	DAPES	0.12	9.56	9.56				11.90				
	New Density Zone 1 EELs are available in the following MSA:	s: Orlan	do. Fl	· Miami. Fl · Ft. I au	derdale. FI : A	Atlanta, Ga: Nev	w Orleans, I.A.									
	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-					manta, Ga, mo										
	In all states, EEL network elements shown below also apply to					erted to UNE ra	tes. A Switch	As Is Charge a	pplies to curre	ntly combined	facilities co	onverted to	UNEs.(Non-re	curring rates	do not apply	.)
NOTE:	In All States the EEL network elements apply to ordinarily cor	mbined	netwo	rk elements.(No Swi	itch As Is Cha	arge.) When or	dering ordinar	ily combined	network elemer	nts, Non-recur	ring rates de	o apply.				
2-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												
, <u> </u>	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport															
	Combination - Zone 1		1	UNCVX	UEAL2	14.50	135.75	82.47	63.53	12.01	ļ	11.90				1
.	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		_	l <u>.</u>	l							l	1		1	1
	Transport Combination - Zone 2		2	UNCVX	UEAL2	19.57	135.75	82.47	63.53	12.01		11.90				
1	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		2	LINCVA	LIEALO	07.00	405.75	00.47	00.50	40.04		44.00	1		1	1
	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCVX	UEAL2	37.82	135.75	82.47	63.53	12.01	ļ	11.90	 		 	
	per month			UNC1X	1L5XX	0.1856										1
	Interoffice Transport - Dedicated - DS1 combination - Facility	-		014017	ILUAA	0.1000			1		}		1	1	1	
	Termination per month			UNC1X	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				1
	DS1 Channelization System Per Month			UNC1X	MQ1	146.77	101.42	71.62		10.49		11.90	1		1	t
1	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	1.38	10.07	7.08		121.10		11.90	1		1	
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1						_									
	Interoffice Transport Combination - Zone 1	L	1	UNCVX	UEAL2	14.50	135.75	82.47	63.53	12.01	<u></u>	11.90		<u> </u>		1
	Each Additional 2-Wire VG Loop(SL2) in the same DS1]]	1
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	19.57	135.75	82.47	63.53	12.01		11.90				<u> </u>
	Each Additional 2-Wire VG Loop(SL2) in the same DS1			l	1											1
	Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	37.82	135.75	82.47	63.53	12.01		11.90		<u> </u>]	<u>i</u>

UNBUNDLE	D NETWORK ELEMENTS - Florida												Attachi	nent: 2	Exhib	oit: C
											Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		Interi	_								Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												1	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
			-			_ 1	Nonrec	urring	Nonrecurring	Disconnect		l	oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Voice Grade COCI - DS1 to DS0 Channel System combination -															
	per month			UNCVX	1D1VG	1.38	10.07	7.08				11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-															
4 14/15	Is Charge E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	FDOFF	IOF TO	UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIR	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	EROFF	ICE IR	ANSPORT (EEL)	-											
	Transport Combination - Zone 1		1	UNCVX	UEAL4	23.02	167.86	115.15	67.08	15.56		11.90				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		† ·	0.1017	02/121	20.02	.07.00	110.10	07.00	10.00		11100				
	Transport Combination - Zone 2		2	UNCVX	UEAL4	31.07	167.86	115.15	67.08	15.56		11.90				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
\vdash	Transport Combination - Zone 3		3	UNCVX	UEAL4	60.02	167.86	115.15	67.08	15.56		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1856										
\vdash	Interoffice Transport - Dedicated - DS1 - Facility Termination Per		-	OINC IA	ILOAA	U. 1806			+			-				
	Month			UNC1X	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				
	Channelization - Channel System DS1 to DS0 combination Per			0110171		00.11	100.01	00.11		10.00		11.00				
	Month			UNC1X	MQ1	146.77	101.42	71.62	11.09	10.49		11.90				
	Voice Grade COCI - DS1 to DS0 Channel System combination -															
	per month			UNCVX	1D1VG	1.38	10.07	7.08				11.90				
	Additional 4-Wire Analog Voice Grade Loop in same DS1			110000		00.00	407.00	445.45	07.00	45.50		44.00				
-	Interoffice Transport Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1		1	UNCVX	UEAL4	23.02	167.86	115.15	67.08	15.56		11.90				
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	31.07	167.86	115.15	67.08	15.56		11.90				
	Additional 4-Wire Analog Voice Grade Loop in same DS1			0.1017	02/121	01.07	.07.00	110.10	07.00	10.00		11100				
	Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	60.02	167.86	115.15	67.08	15.56		11.90				
	Voice Grade COCI - DS1 to DS0 Channel System combination -															
	per month			UNCVX	1D1VG	1.38	10.07	7.08				11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.00				
4-WID	Is Charge E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTER	EEICE				8.98	8.98	8.98	8.98		11.90				
7-7711	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	I	TRANSFORT (EEE)												
	Transport Combination - Zone 1		1	UNCDX	UDL56	26.39	161.56	108.85	67.08	15.56		11.90				
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 2		2	UNCDX	UDL56	35.62	161.56	108.85	67.08	15.56		11.90				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice			LINODY	1101 50	00.00	101 50	400.05	07.00	45.50		44.00				
\vdash	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCDX	UDL56	68.82	161.56	108.85	67.08	15.56		11.90		-		
	Per Month			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 - combination Facility				. 20, 01	3.1000										
	Termination Per Month		<u>L</u>	UNC1X	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				
	Channelization - Channel System DS1 to DS0 combination Per							· · · · · · · · · · · · · · · · · · ·		· · · · · ·						
\vdash	Month		<u> </u>	UNC1X	MQ1	146.77	101.42	71.62	11.09	10.49		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	2.10	10.07	7.08	1			11.90				
 	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		-	OINCDA	טטוטו	2.10	10.07	7.08	1			11.90				
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	26.39	167.86	115.15	67.08	15.56		11.90				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			-					1							
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	35.62	167.86	115.15	67.08	15.56		11.90				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1				L											
\vdash	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	68.82	167.86	115.15	67.08	15.56		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)			UNCDX	1D1DD	2.10	10.07	7.08	1			11.90				
 	Nonrecurring Currently Combined Network Elements Switch -As-			OINODA	טטוטו	2.10	10.07	7.08	 			11.90				
	Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIR	E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL)												
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1		1	UNCDX	UDL64	26.39	161.56	108.85	67.08	15.56		11.90				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		2	LINCDY	UDL64	25.00	404.50	400.05	07.00	45.50		44.00				
	Transport Combination - Zone 2	l	2	UNCDX	UDL04	35.62	161.56	108.85	67.08	15.56	<u> </u>	11.90		L		

ONRONDLE	D NETWORK ELEMENTS - Florida					1						,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec First	curring Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$)	SOMAN	SOMAN
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice						First	Auu i	Filst	Addi	SOMEC	JOWAN	JOWAN	JOWAN	JOWAN	JOWAN
	Transport Combination - Zone 3		3	UNCDX	UDL64	68.82	161.56	108.85	67.08	15.56		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	146.77	101.42	71.62	11.09	10.49		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System								11.00	10.10						
	combination - per month (2.4-64kbs) Additional 4-Wire 64Kbps Digital Grade Loopin same DS1			UNCDX	1D1DD	2.10	10.07	7.08				11.90				
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	26.39	167.86	115.15	67.08	15.56		11.90				<u> </u>
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	35.62	167.86	115.15	67.08	15.56		11.90				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	68.82	167.86	115.15	67.08	15.56		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	2.10	10.07	7.08				11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-					2.10			0.00	0.00						
4-WIRI	Is Charge E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE	ROFFI	CF TRA	UNC1X ANSPORT (FFL)	UNCCC		8.98	8.98	8.98	8.98		11.90				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1		1	UNC1X	USLXX	73.44	313.75	181.48	61.22	13.53		11.90				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice		i i													
	Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice		2	UNC1X	USLXX	99.13	313.75	181.48	61.22	13.53		11.90				
	Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNC1X	USLXX	191.51	313.75	181.48	61.22	13.53		11.90				
	Per Month Interoffice Transport - Dedicated - DS1 combination - Facility			UNC1X	1L5XX	0.1856										
	Termination Per Month			UNC1X	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIRI	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE	ROFFI	CE TRA	ANSPORT (EEL)												
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	73.44	313.75	181.48	61.22	13.53		11.90				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	99.13	313.75	181.48	61.22	13.53		11.90				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	191.51	313.75	181.48	61.22	13.53		11.90				
	Interoffice Transport - Dedicated - DS3 combination - Per Mile		3				313.73	101.40	01.22	13.33		11.90				
	Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per			UNC3X	1L5XX	3.87										
	month			UNC3X	U1TF3	1,071.00	335.46	219.28	72.03	70.56		11.90				
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3	211.19	199.28	118.64	40.34	39.07		11.90				.
	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination -			UNC1X	UC1D1	13.76	10.07	7.08				11.90				
	Zone 1		1	UNC1X	USLXX	73.44	313.75	181.48	61.22	13.53		11.90				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	99.13	313.75	181.48	61.22	13.53		11.90				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	191.51	313.75	181.48	61.22	13.53		11.90				
	DS3 Interface Unit (DS1 COCI) combination per month		Ľ	UNC1X	UC1D1	13.76	10.07	7.08				11.90			<u></u>	
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge			UNC3X	UNCCC		8.98	8.98	8.98	8.98		11.90				
2-WIRI	E VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	EROFF	ICE TF				5.50	3.30	3.30	5.50		50				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1			UNCVX	UEAL2	14.50	135.75	82.47	63.53	12.01		11.90				

CATEGORY RATE ELEMENTS Initial Zoole BCS USOC Week RATEGORY Sections Deciminate Deciminate Deciminate Deciminate Deciminate	UNBUNDLI	ED NETWORK ELEMENTS - Florida												Attachi	nent: 2	Exhil	bit: C
2				Zone	BCS	USOC			.,,			Submitted Elec	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
2. Verwin Long under with Zene VS Interolifice Transport 2 DACVX UEAL2 18.57 136.79 62.47 63.50 12.91 11.90							Rec					COMEC	COMAN			COMAN	SOMAN
Combination - Zane 2		2-WireVG Loop used with 2-wire VG Interoffice Transport	1					FIRST	Add I	FIRST	Addi	SOMEC	SOWAN	SUMAN	SOWAN	SOWAN	SUMAN
Combination		Combination - Zone 2		2	UNCVX	UEAL2	19.57	135.75	82.47	63.53	12.01		11.90				
Miles Per Manufil Therefore Transport - Dedicated - 2-Wire Vices Grade Miles Per Manufil Therefore Transport - Dedicated - 2-Wire Vices Grade Internet Stretch - New York Internet Stretch - N		Combination - Zone 3		3	UNCVX	UEAL2	37.82	135.75	82.47	63.53	12.01		11.90				
Intereffice Transport - Dedicated - 2-Weive Vaco Grade UNCVX					UNCVX	1L5XX	0.0091										
Nonecouring Currently Combined Newton: Elements Settlich As-Pain (NICC)						LIATVO		47.25	24.70	10.21	7.02		11.00				
CAMINE VOICE GRADE EXTENDED LOOP 4 WHISE VOICE GRADE INTEROFFICE TRANSPORT (EEL)					UNCVX	UTIVZ	25.32	47.35	31.78	18.31	7.03		11.90				
Advirency Coop used with -4-wer VG Interoffice Transport 1 UNCVX						UNCCC		8.98	8.98	8.98	8.98		11.90				
Combration - Zone 1	4-WIR		TEROFF	ICE TR	ANSPORT (EEL)												
Combination - Zone 2 2 UNCVX UEAL4 31.07 167.86 115.15 67.06 15.56 11.90		Combination - Zone 1		1	UNCVX	UEAL4	23.02	167.86	115.15	67.08	15.56		11.90				
4-Wire/GLopu used with 4-wee VG interdifice Transport 3 UNCVX UEAL4 60.02 167.86 115.15 67.08 15.56 11.90				2	UNCVX	UEAL4	31.07	167.86	115.15	67.08	15.56		11.90				
InterOffice Transport - Dedicated -4-Wire VG combination - Per UNCVX		4-WireVG Loop used with 4-wire VG Interoffice Transport		3													
Interofice Transport - Declarated - Series month		Interoffice Transport - Dedicated - 4-wire VG combination - Per															
Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNCVX UNCCC 8.98 8.98 8.98 8.98 11.90		Interoffice Transport - Dedicated - 4- Wire Voice Grade						47.25	24.70	10.21	7.02		11.00				
DS3 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)		Nonrecurring Currently Combined Network Elements Switch -As-	-				22.38										
High Capacity Unbundled Local Loop - DS3 combination - Per UNC3X	Dear		CE TO A	NEDOD		UNCCC		8.98	8.98	8.98	8.98		11.90				\vdash
Mile per month	D33 L		I	NOFUR	I (CCL)												$\vdash \vdash \vdash$
Facility Termination per month		Mile per month			UNC3X	1L5ND	10.92										
Interoffice Transport - Dedicated - DSS combination - Facility UNC3X		Facility Termination per month						556.37	343.01	139.13	96.84		11.90				
Termination per per month					UNC3X	1L5XX	3.87										
Is Charge					UNC3X	U1TF3	1,071.00	335.46	219.28	72.03	70.56		11.90				
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)					LINC3X	LINCCC		8 98	8 98	8 98	8 98		11 90				
High Capacity Unbundled Local Loop - STS1 combination - Per UNCSX	STS1		FICE TE	RANSPO		011000		0.30	0.30	0.30	0.30		11.30				—
Facility Termination per month					UNCSX	1L5ND	10.92										
Interoffice Transport - Dedicated - STS1 combination - Per Mile per month		High Capacity Unbundled Local Loop - STS1 combination -						556.37	343.01	139.13	96.84		11.90				
Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month UNCSX U1TFS 1,056.00 335.46 219.28 72.03 70.56 11.90							3 07	-			-						
Nonrecurring Currently Combined Network Elements Switch -As- UNCSX UNCCC 8.98 8.98 8.98 11.90		Interoffice Transport - Dedicated - STS1 combination - Facility							212				11.5				
Scharge UNCSX UNCCC 8.98 8.98 8.98 8.98 11.90			-				1,056.00										
First 2-Wire ISDN Loop in a DS1 Interoffice Combination 1 UNCNX U1L2X 21.76 147.69 94.41 62.23 10.71 11.90 11.90		Is Charge			UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				
Transport - Zone 1	2-WIR		RT (EEL)								1					
Transport - Zone 2 2 UNCNX U1L2X 29.38 147.69 94.41 62.23 10.71 11.90		Transport - Zone 1		1	UNCNX	U1L2X	21.76	147.69	94.41	62.23	10.71		11.90				<u> </u>
First 2-Wire ISDN Loop in a DS1 Interoffice Combination 3 UNCNX U1L2X 56.76 147.69 94.41 62.23 10.71 11.90				2	UNCNX	U1L2X	29.38	147.69	94.41	62.23	10.71		11.90				1
Interoffice Transport - Dedicated - DS1 combination - Per Mile UNC1X 1L5XX 0.1856 Interoffice Transport - Dedicated - DS1 combinition - Facility Termination per month UNC1X U1TF1 88.44 105.54 98.47 21.47 19.05 11.90				3	UNCNX	U1L2X	56.76	147.69	94.41	62.23			11.90				
Termination per month UNC1X U1TF1 88.44 105.54 98.47 21.47 19.05 11.90		Interoffice Transport - Dedicated - DS1 combination - Per Mile															
		Termination per month			UNC1X	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				
		Channelization - Channel System DS1 to DS0 combination - per month			UNC1X	MQ1	146.77	101.42	71.62	11.09	10.49					_	
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System						UC1CA											

UNBUND	LED NETWORK ELEMENTS - Florida													nent: 2		bit: C
CATEGORY	7 RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	14.1% 10.1.10D111 1						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1		1	UNCNX	U1L2X	21.76	147.69	94.41	62.23	10.71		11.90				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport		2	UNCNX	U1L2X	29.38	147.69	94.41	62.23	10.71		11.90				<u> </u>
	Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System		3	UNCNX	U1L2X	56.76	147.69	94.41	62.23	10.71		11.90				<u> </u>
	combintation- per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCNX	UC1CA	3.66	10.07	7.08				11.90				<u> </u>
	Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-W	IRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROF	FICE T	RANSPORT (EEL)												
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	73.44	313.75	181.48	61.22	13.53		11.90				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	99.13	313.75	181.48	61.22	13.53		11.90				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	191.51	313.75	181.48	61.22	13.53		11.90				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month			UNCSX	1L5XX	3.87										
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination			UNCSX	U1TFS	1,056.00	335.46	219.28	72.03	70.56		11.90				
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	211.19	199.28	118.64	40.34	39.07		44.00				
	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination -			UNC1X	UC1D1	13.76	10.07	7.08				11.90				
	Zone 1 Additional DS1Loop in STS1 Interoffice Transport Combination -		1	UNC1X	USLXX	73.44	313.75	181.48	61.22	13.53		11.90				
	Zone 2 Additional DS1Loop in STS1 Interoffice Transport Combination -		2	UNC1X	USLXX	99.13	313.75	181.48	61.22	13.53		11.90				<u> </u>
	Zone 3		3	UNC1X	USLXX	191.51	313.75	181.48	61.22	13.53		11.90				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.76	10.07	7.08				11.90				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-W	IRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROI	FFICE T	RANS	PORT (EEL)												
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	26.39	161.56	108.85	67.08	15.56		11.90				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	35.62	161.56	108.85	67.08	15.56		11.90				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	68.82	161.56	108.85	67.08	15.56		11.90				
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.0091										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	U1TD5	18.44	47.35	31.78	18.31	7.03		11.90				
/ NA / NA /	Is Charge IRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROL	EEICE T	DANC	UNCDX	UNCCC		8.98	8.98	8.98	8.98		11.90				<u> </u>
4-77	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport	Trice I	KANO	FURT (EEL)												
	Combination - Zone 1 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		1	UNCDX	UDL64	26.39	161.56	108.85	67.08	15.56		11.90				<u> </u>
	Combination - Zone 2 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		2	UNCDX	UDL64	35.62	161.56	108.85	67.08	15.56		11.90				
	Combination - Zone 3 Interoffice Transport - Dedicated - 4-wire 64 kbps combination -		3	UNCDX	UDL64	68.82	161.56	108.85	67.08	15.56		11.90				
	Per Mile Interoffice Transport - Dedicated - 4-wire 64 kbps combination -			UNCDX	1L5XX	0.0091										<u> </u>
	Facility Termination Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	U1TD6	18.44	47.35	31.78	18.31	7.03		11.90				<u> </u>
	Is Charge LI NETWORK ELEMENTS			UNCDX	UNCCC		8.98	8.98	8.98	8.98		11.90				

IINBIIND	LED NETWORK ELEMENTS - Florida												Attach	ment: 2	Exhib	nit: C
SHOOKD	LLD IAL I WORK LLLINLIA I 3 - FIUIIUA					1					Svc Order	Svc Order	Incremental		Incremental	Incremental
1		1									Submitted	Submitted		Charge -	Charge -	Charge -
											Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc]		RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
5,1125511		m		200				= = (+)			perLSK	per LSK				
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonrec	curring	Nonrecurring	Disconnect		1	oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Wh	en used as a part of a currently combined facility, the non-recurr	rna cha	nes do	not apply but a Si	witch As Is c	harge does and		Auu	11100	Auui	COMILO	COMPAR	COMPAN	COMPAR	COMPAR	- COMPART
	en used as ordinarily combined network elements in All States, t															
	recurring Currently Combined Network Elements "Switch As Is"					1										
1	Nonrecurring Currently Combined Network Elements Switch -As-		(0110	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,												
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		8.98	8.98	8.98	8.98		11.90				1
	Nonrecurring Currently Combined Network Elements Switch -As-			0.10171	0.1000		0.00	0.00	0.00	0.00		11.00				
	Is Charge - 56/64 kbps			UNCDX	UNCCC		8.98	8.98	8.98	8.98		11.90				i
	Nonrecurring Currently Combined Network Elements Switch -As-						0.00									
	Is Charge - DS1	1		UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				i
	Nonrecurring Currently Combined Network Elements Switch -As-			0.10.1%	0.1000		0.00	0.00	0.00	0.00		11.00				
	Is Charge - DS3			UNC3X	UNCCC		8.98	8.98	8.98	8.98		11.90				ı
	Nonrecurring Currently Combined Network Elements Switch -As-				3		0.00	0.00	5.50	0.50		11.50				
1 1	Is Charge - STS1	1		UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				i
NO.	FE: Local Channel - Dedicated Transport - minimum billing perior	d - Belo	w DS3-			r months	0.30	0.90	0.90	0.90		11.30				
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 1			UNCVX	ULDV2	21.94	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 2		2	UNCVX	ULDV2	29.62	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 3			UNCXV	ULDV2	57.22	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade Zone 1			UNCVX	ULDV4	22.81	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade Zone 2			UNCVX	ULDV4	30.79	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade Zone3			UNCXV	ULDV4	59.48	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade 20163 Local Channel - Dedicated - DS1 per month Zone 1		1	UNC1X	ULDF1	35.28	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated - DS1 Per Month Zone 2		2	UNC1X	ULDF1	47.63	216.65	183.54	24.30	16.95		11.90				
-	Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	92.01	216.65	183.54	24.30	16.95		11.90				
-	Local Channel - Dedicated - DS3 - Per Mile per month		3	UNC3X	1L5NC	8.50	210.03	103.34	24.30	10.55		11.90				
-	Local Channel - Dedicated - DS3 - Fel Wille per Month Local Channel - Dedicated - DS3 - Facility Termination			UNC3X	ULDF3	531.91	556.37	343.01	139.13	96.84		11.90				
-	Local Channel - Dedicated - DSS - Facility Termination Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC	8.50	336.37	343.01	139.13	90.04		11.90				
	Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	540.69	556.37	343.01	139.13	96.84		11.90				
Ont	ional Features & Functions:			UNCOX	OLDI 3	340.09	330.37	343.01	139.13	30.04		11.90				
	LTIPLEXERS				1											
10.0	Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	146.77	101.42	71.62	11.09	10.49		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per			OXIDI	IVIQI	140.77	101.42	71.02	11.09	10.49		11.90				
	month (2.4-64kbs)			UDL	1D1DD	2.10	10.07	7.08				11.90				1
-	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per			ODL	טטוטו	2.10	10.07	7.00				11.90				
	month			UDN	UC1CA	3.66	10.07	7.08				11.90				1
				UEA	1D1VG	1.38	10.07					11.90				
 	Voice Grade COCI - DS1 to DS0 Channel System - per month DS3 to DS1 Channel System per month	 		UXTD3	MQ3	211.19	199.28	7.08 118.64	40.34	39.07		11.90				
\vdash	STS1 to DS1 Channel System per month	 		UXTS1	MQ3	211.19	199.28	118.64	40.34	39.07		11.90				
 	DS3 Interface Unit (DS1 COCI) used with Loop per month	1		USL	UC1D1	13.76	10.07	7.08	40.34	39.07		11.90				
 	DS3 Interface Unit (DS1 COCI) used with Local Channel per	 		00L	30101	13.70	10.07	1.00	 			11.50				
1 1	month	1		ULDD1	UC1D1	13.76	10.07	7.08	I			11.90				i
 	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel	1		ו ממשט	OCIDI	13.76	10.07	1.00	 			11.90				
1 1	per month	1		U1TD1	UC1D1	13.76	10.07	7.08	I			11.90				i
LINBLINDI F	D LOCAL EXCHANGE SWITCHING(PORTS)	 		וטווט	30101	13.70	10.07	1.00	 			11.50				
	hange Ports	 			1				 			-				
	FE: Although the Port Rate includes all available features in GA,	KYIA	R TNI 41	ne desired features	will need to b	ne ordered usin	ng retail USOC		 							
	IRE VOICE GRADE LINE PORT RATES (RES)	, LA	ا ۱۱۹٫۱۱	io desired realures	need to t	o ordered usil	ig retail 0300		 							
2-41	Exchange Ports - 2-Wire Analog Line Port- Res.	 		UEPSR	UEPRL	1.40	3.74	3.63	1.88	1.80		11.90				
 	Energy Forto 2 Wild Filliandy Line Fort INCo.	 	1	521 OK	JEINE	1.70	5.74	5.05	1.00	1.00		11.00				
1 1	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.	1		UEPSR	UEPRC	1.40	3.74	3.63	1.88	1.80		11.90				i
 	Exercises 1 310 2 Wild Finding Enter of with Odiler ID - 1065.	1		J. J.	02110	1.40	5.74	5.05	1.00	1.00		11.30				
1 1	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.	1		UEPSR	UEPRO	1.40	3.74	3.63	1.88	1.80		11.90				i
 	Exchange Ports - 2-Wire VG unbundled Florida area calling with	1		OLI OIL	OLI NO	1.40	5.74	5.05	1.00	1.00		11.30				
1 1	Caller ID - Res.	1		UEPSR	UEPAF	1.40	3.74	3.63	1.88	1.80		11.90				i
 	Exchange Ports - 2-Wire VG unbundled Florida Residence Area	1		0L1 0K	OLI AI	1.40	5.74	5.05	1.00	1.00		11.30				
1 1	Calling Plan, without Caller ID capability	1		UEPSR	UEPA9	1.40	3.74	3.63	1.88	1.80		11.90				i
 	Exchange Ports - 2-Wire VG unbundled Florida extended	 		OLFOR	OLFAS	1.40	3.14	3.03	1.08	1.80		11.90				
	dialing port for use with CREX7 and Caller ID	1		UEPSR	UEPA1	1.40	3.74	3.63	1.88	1.80		11.90				i
 	Exchange Ports - 2-Wire VG unbundled Florida extended	 		OLI OIX	JLI AI	1.40	3.14	3.03	1.00	1.00		11.50				
	dialing port for use with CREX7, without Caller ID capability			UEPSR	UEPA8	1.40	3.74	3.63	1.88	1.80		11.90				ı
	Takening port for use with ONLAT, without Galler ID capability	1		OLFON	JLFA0	1.40	3.74	3.03	1.68	1.60		11.90		l		

<u>JNBUNDL</u>	ED NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LUM)			UEPSR	UEPAP	1.40	3.74	3.63	1.88	1.80		11.90				
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPSR	UEPRT	1.40	3.74	3.63	1.88	1.80		11.90				
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00				11.90				
FEAT	TURES			LIEBOD								44.00				
	All Available Vertical Features			UEPSR	UEPVF	2.26	0.00	0.00				11.90				
2-WII	RE VOICE GRADE LINE PORT RATES (BUS)		<u> </u>													
	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus			UEPSB	UEPBL	1.40	3.74	3.63	1.88	1.80		11.90				
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.40	3.74	3.63	1.88	1.80		11.90				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.40	3.74	3.63	1.88	1.80		11.90				
	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus			UEPSB	UEPB1	1.40	3.74	3.63	1.88	1.80		11.90				
	2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPSB	UEPBE	1.40	3.74	3.63	1.88	1.80		11.90				
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00				11.90				
FEAT	TURES															
	All Available Vertical Features			UEPSB	UEPVF	2.26	0.00	0.00				11.90				
EXC	HANGE PORT RATES (DID & PBX)															
	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus		<u> </u>	UEPSP	UEPPO	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1 UEPLD	1.40 1.40	39.06	18.18	12.35 12.35	0.7187	1	11.90				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus 2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP UEPSP	UEPLD	1.40	39.06 39.06	18.18 18.18	12.35	0.7187 0.7187		11.90 11.90				
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPSP	UEPXE	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPSP	UEPXL	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPSP	UEPXM	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPSP	UEPXO	1.40	39.06	18.18	12.35	0.7187		11.90				ļ
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port		ļ	UEPSP	UEPXS	1.40	39.06	18.18	12.35	0.7187	<u> </u>	11.90		ļ		<u> </u>
	Subsequent Activity TURES		-	UEPSP	USASC	0.00	0.00	0.00				11.90	1			ļ
FEA	All Available Vertical Features	-	 	UEPSP UEPSE	UEPVF	2.26	0.00	0.00	 		1	11.90		-	-	1
EYC	HANGE PORT RATES (COIN)			OLI OF OLFOE	OLF VI	2.20	0.00	0.00	 		1	11.90	 		-	-
EXC	Exchange Ports - Coin Port				+	1.40	3.74	3.63	1.88	1.80	 	11.90	 			†
NOT	E: Transmission/usage charges associated with POTS circuit so	witched	usage	will also apply to c	ircuit switche								ports.			İ
NOT	E: Access to B Channel or D Channel Packet capabilities will be	availa	ole only	y through BFR/New	Business Re	quest Process.	Rates for the	packet capab	ilities will be de	termined via t	he Bona Fid	de Request/	New Busines:	s Request Pro	cess.	İ
UNBUNDLED	D LOCAL EXCHANGE SWITCHING(PORTS)															
EXC	HANGE PORT RATES							•		•						
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.73	78.41	15.82	41.94	4.26		11.90			1.83	
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability			UEPDD	UEPDD	54.95	151.11	77.75	48.81	3.10		11.90			1.83	
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	8.83	46.83	50.68		11.93		11.90			1.83	
	All Features Offered			UEPTX UEPSX	UEPVF	2.26	0.00	0.00				11.90			1.83	
NOT	E: Transmission/usage charges associated with POTS circuit sy													Ì	<u> </u>	ļ
	E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles	availa	ole only	UEPTX UEPSX	U1UMA	quest Process. 0.00	Rates for the 0.00	packet capab 0.00		termined via t	the Bona Fig	de Request/	New Business	s Request Pro	cess.	

UNBUNDLE	D NETWORK ELEMENTS - Florida												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge -	Incrementa Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	NDLED PORT with REMOTE CALL FORWARDING CAPABILITY															
UNBU	NDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res			UEPVR	UERAC	1.40	3.74	3.63	1.88	1.80	1	11.90				<u> </u>
	Oriburidied Remote Call Forwarding Service, Area Calling, Res			UEPVK	UERAC	1.40	3.74	3.03	1.00	1.00		11.90				
	Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.40	3.74	3.63	1.88	1.80		11.90				
	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.40	3.74	3.63	1.88	1.80		11.90				
	Unbundled Remote Call Forwarding Service, IntraLATA - Res			UEPVR	UERTR	1.40	3.74	3.63	1.88	1.80		11.90				
Non-Re	ecurring				-											
	Unbundled Remote Call Forwarding Service - Conversion -															
	Switch-as-is			UEPVR	USAC2		0.102	0.102				11.90				
	Unbundled Remote Call Forwarding Service - Conversion with															1
	allowed change (PIC and LPIC)			UEPVR	USACC		0.102	0.102								
UNBU	NDLED REMOTE CALL FORWARDING - Bus															
					l									1		
	Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVB	UERAC	1.40	3.74	3.63	1.88	1.80	ļ	11.90				<u> </u>
												44.00				
	Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERLC	1.40	3.74	3.63	1.88	1.80		11.90				
	Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVB	UERTE UERTR	1.40 1.40	3.74 3.74	3.63	1.88	1.80 1.80		11.90 11.90				
	Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service Expanded and			UEPVB	UERIR	1.40	3.74	3.63	1.88	1.80		11.90				
	Exception Local Calling			UEPVB	UERVJ	1.40	3.74	3.63	1.88	1.80		11.90				
Non-Pa	ecurring			UEPVB	UERVJ	1.40	3.74	3.03	1.88	1.80		11.90				
NOII-R	Unbundled Remote Call Forwarding Service - Conversion -															
	Switch-as-is			UEPVB	USAC2		0.102	0.102				11.90				
	Unbundled Remote Call Forwarding Service - Conversion with			02. 10	00/102		0.102	002								1
	allowed change (PIC and LPIC)			UEPVB	USACC		0.102	0.102								
UNBUNDLED I	LOCAL SWITCHING, PORT USAGE															
End Of	ffice Switching (Port Usage)															
	End Office Switching Function, Per MOU					0.0007662										
	End Office Trunk Port - Shared, Per MOU					0.000164										
Tander	m Switching (Port Usage) (Local or Access Tandem)															
	Tandem Switching Function Per MOU					0.0001319										
	Tandem Trunk Port - Shared, Per MOU					0.000235										
Comm	on Transport															
	Common Transport - Per Mile, Per MOU					0.0000035										
LINDUNDI ED I	Common Transport - Facilities Termination Per MOU				+	0.0004372										
	PORT/LOOP COMBINATIONS - COST BASED RATES assed Rates are applied where BellSouth is required by FCC ar	d/or St	ate Co	nmission rule to no	ovide Unbur	dlad Local Swit	tching or Swite	h Porte			 			-	-	
	es shall apply to the Unbundled Port/Loop Combination - Cos								ed Port section	of this Rate F	xhihit					1
	ffice and Tandem Switching Usage and Common Transport Us											n Port/Loon	Combination	18		
	st and additional Port nonrecurring charges apply to Not Curr															
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	1						J g								
	ort/Loop Combination Rates				İ											1
	2-Wire VG Loop/Port Combo - Zone 1		1			14.11										
	2-Wire VG Loop/Port Combo - Zone 2		2			18.23										
	2-Wire VG Loop/Port Combo - Zone 3		3	_		33.04										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1			UEPRX	UEPLX	12.94										<u> </u>
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	17.06								ļ	ļ	
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	31.87					<u> </u>			 	ļ	
2-Wire	Voice Grade Line Port Rates (Res)		-	LIEDDY	LIEDDI	4.47	00.00	00.00			}	44.00		 	1	
	2-Wire voice unbundled port - residence		-	UEPRX UEPRX	UEPRL	1.17 1.17	90.00 90.00	90.00			}	11.90 11.90		 	1	
	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.17	90.00	90.00			-	11.90				
-+	2-14116 voice unbunuled port outgoing only - 165		1	OLFINA	OLFRU	1.17	90.00	90.00			}	11.90		1	1	
	2-Wire voice unbundled Florida Area Calling with Caller ID - res			UEPRX	UEPAF	1.17	90.00	90.00				11.90		1		
	2-Wire voice unbundles res, low usage line port with Caller ID			JE1 100	JEI /3I	1.17	30.00	30.00			1	11.50		 	 	†
	= parado roo, ron doago mio port with odilor ib			UEPRX	UEPAP	1.17	90.00	90.00	ı	1	1	11.90	1	1	1	1

ONDUNDE	ED NETWORK ELEMENTS - Florida		1	1	1						Com Cont	Comp Control	Attachr			bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled Florida extended dialing port for use with CREX7 and Caller ID			UEPRX	UEPA1	1.17	90.00	90.00				11.90				
	2-Wire voice unbundled Florida extended dialing port for use with CREX7, without Caller ID capability			UEPRX	UEPA8	1.17	90.00	90.00				11.90				
	2-Wire voice unbundled Florida Area Calling Port without Caller ID Capability			UEPRX	UEPA9	1.17	90.00	90.00				11.90				
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX	UEPRT	1.17	90.00	90.00				11.90				
FEAT																
	All Features Offered			UEPRX	UEPVF	2.26	0.00	0.00				11.90				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPRX	USAC2		0.102	0.102				11.90				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPRX	USACC		0.102	0.102				11.90				
ADDIT	TONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPRX	USAS2	0.00	0.00	0.00				11.90				
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.11										
	2-Wire VG Loop/Port Combo - Zone 2		2			18.23										
	2-Wire VG Loop/Port Combo - Zone 3		3			33.04										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	12.94										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	17.06										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	31.87										
2-Wire	e Voice Grade Line Port (Bus)			UEDDV								44.00				
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.17	90.00	90.00				11.90				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC UEPBO	1.17	90.00	90.00				11.90				
	2-Wire voice unbundled port outgoing only - bus			UEPBX UEPBX	UPEB1	1.17 1.17	90.00 90.00	90.00				11.90 11.90			-	
	2-Wire voice unbundled incoming only port with Caller ID - Bus 2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPBX	UEPBE	1.17	90.00	90.00				11.90				
LOCA	L NUMBER PORTABILITY			OLFBA	OLFBL	1.17	90.00	90.00	+			11.50				
LOUA	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35			+							
FEAT				02. 5%	Litti OX	0.00										
	All Features Offered			UEPBX	UEPVF	2.26	0.00	0.00				11.90				
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPBX	USAC2		0.102	0.102				11.90				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPBX	USACC		0.102	0.102				11.90				
ADDIT	TONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2		0.00	0.00				11.90				
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)		1				-		1							
UNE F	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.11		•		•				_		
	2-Wire VG Loop/Port Combo - Zone 2		2			18.23							·			
	2-Wire VG Loop/Port Combo - Zone 3		3			33.04										
UNE L	oop Rates		ļ						ļ <u> </u>						1	
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	12.94										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	17.06									-	
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	31.87										
2-Wire	e Voice Grade Line Port Rates (RES - PBX)		1	l							<u> </u>					<u> </u>

CATEORY RATE ELEMENTS Minds Zone BCS USOC RATE(EL) Sharing Charge	UNBUNDI F	D NETWORK ELEMENTS - Florida												Attach	ment: 2	Fyhil	oit: C
APPLICATION Bodd Application Applica	SHOUNDEE	- TIONIC ELEMENTO - FIORICA										Svc Order	Svc Order				Incremental
CATEGORY RATE ELEMENTS Mode M																	Charge -
CATEGORY RATE ELEMENTS m 2006 RATE December																	Manual Svc
No. No.	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)								Order vs.
Second Content of Co			m						***			per Lor	per Lor				Electronic-
None None																	
Mode														ist	Addi	DISC 1St	Disc Add'l
2-Win Vi U Putunded Composition 2-Way PPS Traft Prof - UEPRG UEP							В	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
Res							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN			SOMAN	SOMAN
Res		2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -															
Card Number Proteinty 1 per part 0,5PRG 1,0PCP 3.15 0.000 0.000 11:00		Res			UEPRG	UEPRD	1.17	90.00	90.00				11.90				
FATURES A In Faultus Offered A In Faultus Offe	LOCAL	NUMBER PORTABILITY															
FATURES					UEPRG	LNPCP	3.15	0.00	0.00				11.90				
NONSECURING CHARGES (NECS) - CURRENTLY COMBINED	FEATU																
2-Wile Voice Grade Long Line Part Commission (PBQ) - Commission (PBQ		All Features Offered			UEPRG	UEPVF	2.26	0.00	0.00				11.90				
2-Wile Voice Grade Long Line Part Commission (PBQ) - Commission (PBQ	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
Comerators - South-Au-Sec UEPRG USAC2 8.45 191 11.90																	
Conversion - Solido No Charge USACC B.45 1.91 11.50					UEPRG	USAC2		8.45	1.91				11.90				
ADOTTONAL NECS Developed Service Control of PEX																	
ADDITIONAL MRCs 20th Viole Graft Loop (Line Port Combination (PBIX) 11.50 11.5					UEPRG	USACC		8.45	1.91				11.90				
Service Voice Grade Loop Line Port Combination (PRS) - Subsequent Activity Vision Visio	ADDIT																
Subsequent Activity - Change Rearrange Multime Hund																	
Croup Croup Cross Cros					UEPRG	USAS2	0.00	0.00	0.00				11.90				
Croup Croup Cross Cros																	
WIRRE VOICE GRADE LOOP WITH ZWIRRE LINE PORT (BUS - PBX)								7.09	7.09				11.90				
New Forticop Combination Rates	2-WIRE																
2-Wire Vols Lope/Port Combo - Zone 2																	
2.Wife VGL LopeProt Combo - Zone 2				1			14.11										
2, Wife Voto (Dept Combo - 2 one 3 3 3, 3, 4 3, 3, 4 3, 3, 4 4 5 5 5																	
NUME Loop Rates		2-Wire VG Loop/Port Combo - Zone 3		3			33.04										
2-Wire Voice Grade Loop (St. 1) - Zone 1	UNE Lo																
2-Wire Voice Grade Loop (St.1) - Zone 2 2 UEPPX UEPLX 17.06				1	UEPPX	UEPLX	12.94										
2-Wire Voice Grade Loop (St. 1) - Zone 3 3 UEPPX UEPLX 31.87				2													
2-Wire Voice Oracle Line Port Rates (BUS - PDX)				3													
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	2-Wire																
Line Side Unburdled Dutward PBX Trunk Port - Bus																	
Line Side Unburdled Duward PBX Trunk Port - Bus UEPPX UEPP 1.17 90.00 90.00 11.90		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.17	90.00	90.00				11.90				
Line Side Unbundled PEX Tunk Port - Bus																	
2-Wire Voice Unbundled PBX LD Terminal Ports					UEPPX		1.17	90.00	90.00								
2-Wire Voice Unbundled 2-Way Combination PBX Usage Port UEPPX UEPX 1.17 90.00 90.00 11.9					UEPPX	UEPLD	1.17						11.90				
2-Wire Voice Unbundled PBX 10I Terminal Hotel Ports UEPPX UEPXB 1.17 90.00 90.00 11.90					UEPPX	UEPXA							11.90				
2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPPX UEPX					UEPPX	UEPXB											
2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPPX UEPX																	
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD																	
Capable Port				İ		1			22.30		İ			İ	İ	İ	İ
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPX UEXEX UEPX UEPX UEPX UEPX UEPX UEPX UEPX UEPX UEXEX UEPX UEPX UEPX UEPX UEPX UEPX UEPX UEPX UEX					UEPPX	UEPXE	1.17	90.00	90.00				11.90		Ì		
Administrative Calling Port						1									1		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPX UEPX UEPX UEPX					UEPPX	UEPXL	1.17	90.00	90.00				11.90				
Room Calling Port				İ		1					İ			İ	İ	İ	İ
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPPX UEX					UEPPX	UEPXM	1.17	90.00	90.00				11.90		Ì		
Discount Room Calling Port																	
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UEPPX UEPXS 1.17 90.00 90.00 11.90					UEPPX	UEPXO	1.17	90.00	90.00				11.90		Ì		
LOCAL NUMBER PORTABILITY LOCAL Number Portability (1 per port) UEPPX LNPCP 3.15 0.00 0.00 11.9				İ											İ		
Local Number Portability (1 per port)	LOCAL			İ							İ			İ	İ	İ	İ
FEATURES				İ	UEPPX	LNPCP	3.15	0.00	0.00		İ		11.90	İ	İ	İ	İ
All Features Offered																	
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED				İ	UEPPX	UEPVF	2.26	0.00	0.00		İ		11.90	İ	İ	İ	İ
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is	NONRE			İ											İ		
Conversion - Switch-As-Is				İ		1					İ			İ	İ	İ	İ
2-Wire Voice Grade Loop/ Line Port Combination (PBX) -					UEPPX	USAC2		8.45	1.91				11.90				
Conversion - Switch with Change				İ		1		20			İ			İ	İ	İ	İ
ADDITIONAL NRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group 7.86 11.90					UEPPX	USACC		8.45	1.91				11.90		Ì		
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group 7.86 7.86 11.90	ADDIT					1		20							1		
Subsequent Activity				i –		1						İ			İ		
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group 7.86 7.86 11.90					UEPPX	USAS2	0.00	0.00	0.00				11.90		Ì		
Group 7.86 7.86 11.90						1	3.50	3.50	3.30						1		
								7.86	7.86				11.90		Ì		
1 12-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT	2-WIRE	EVOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	RT	1		+				1		1			 		

UNBI	JNDLF	D NETWORK ELEMENTS - Florida												Attach	ment: 2	Fyhil	oit: C
												Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Intor	1								Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			l m											Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
L				<u> </u>				N.		T. N	D']			,	
							Rec	Nonred		Nonrecurring					Rates(\$)		
	 			<u> </u>		_		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	UNE PO	ort/Loop Combination Rates					44.44										
-		2-Wire VG Coin Port/Loop Combo – Zone 1		1			14.11										
-		2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3		3			18.23 33.04										
	LINEL	pop Rates		3		+	33.04										
		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	12.94					1					
		2-Wire Voice Grade Loop (SL1) - Zone 1		2	UEPCO	UEPLX	17.06			1							
-	1	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	31.87					1					
	2-Wire	Voice Grade Line Ports (COIN)		3	OLI OO	OLILX	31.07			1							
	Z-VVIIC	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,		1		+				1							
1		900/976. 1+DDD (FL)			UEPCO	UEP2F	1.17	90.00	90.00				11.90	1	I		
 	1	2-Wire Coin 2-Way with Operator Screening and 011 Blocking	1		021 00	JL1 21	1.17	30.00	30.00			<u> </u>	11.30	 	I		
		(FL)			UEPCO	UEPFA	1.17	90.00	90.00				11.90	1	I		
	<u> </u>	2-Wire Coin 2-Way with Operator Screening and Blocking:		l	02.00	SEITA	1.17	55.00	30.00	 			11.30		-		
		900/976, 1+DDD, 011+, and Local (FL)		1	UEPCO	UEPCG	1.17	90.00	90.00				11.90		1		
	<u> </u>	2-Wire Coin Outward with Operator Screening and 011 Blocking		l	02.1 00	OL1 00	1.17	30.00	30.00	 			11.90		-		
1		(AL, FL)			UEPCO	UEPRK	1.17	90.00	90.00				11.90	1	I		
	1	2-Wire Coin Outward with Operator Screening and Blocking:	-		021 00	JEI IIII	1.17	30.00	30.00	 			11.30	 	t		
		900/976, 1+DDD, 011+ (FL)		1	UEPCO	UEPOF	1.17	90.00	90.00				11.90		1		
—	<u> </u>	2-Wire Coin Outward with Operator Screening and Blocking:		1		52. 51	1.17	55.56	55.50	 			11.50		 		1
		900/976, 1+DDD, 011+, and Local (FL, GA)		1	UEPCO	UEPCQ	1.17	90.00	90.00				11.90	Ì	I		
		2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.17	90.00	90.00				11.90	1	1		
		2-Wire Coin Outward Smartline with 900/976 (all states except		1			,	55.00	22.00				50	1	<u> </u>		
1		LA)		1	UEPCO	UEPCR	1.17	90.00	90.00				11.90	Ì	I		
	ADDITI	ONAL UNE COIN PORT/LOOP (RC)								† †				İ	İ		
	1	UNE Coin Port/Loop Combo Usage (Flat Rate)		1	UEPCO	URECU	1.86	90.00	90.00	1			11.90	İ	İ	İ	İ
	LOCAL	NUMBER PORTABILITY				1 1			22.30						1	İ	
		Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
		CURRING CHARGES - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
		Switch-as-is			UEPCO	USAC2		0.102	0.102				11.90				
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
		Switch with change			UEPCO	USACC		0.102	0.102				11.90				
	ADDITI	ONAL NRCs															
		2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
		Activity			UEPCO	USAS2		0.00	0.00	<u> </u>			11.90	<u> </u>	<u> </u>	<u></u>	
		VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	PORT (RES)												
	UNE Po	ort/Loop Combination Rates															
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			20.10	<u> </u>									
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			24.05										
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			29.49										
		pop Rates															,
		2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	18.48										
		2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	22.43	`									
		2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	27.87										
	2-Wire	Voice Grade Line Port Rates (Res)															
		2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.62	250.00	250.00				11.90		1		
		2-Wire voice unbundled port with Caller ID - res		<u> </u>	UEPFR	UEPRC	1.62	250.00	250.00				11.90		ļ		
		2-Wire voice unbundled port outgoing only - res		 	UEPFR	UEPRO	1.62	250.00	250.00				11.90	ļ	.		
l				1	l	1								Ì	I		
	1	2-Wire voice unbundled Florida Area Calling with Caller ID - res		 	UEPFR	UEPAF	1.62	250.00	250.00				11.90	ļ	.		ļ
		2-Wire voice unbundles res, low usage line port with Caller ID		1		==,-									1		
	L	(LUM)		 	UEPFR	UEPAP	1.62	250.00	250.00				11.90	ļ	.		ļ
	INTER	DFFICE TRANSPORT		 		<u> </u>								ļ	.		
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility			l	1 1								1	I		
		Termination		 	UEPFR	U1TV2	26.52							ļ	.		<u> </u>
1		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile		1	l	1								Ì	I		
		or Fraction Mile			UEPFR	1L5XX	0.0091						ļ]
	FEATU	RES															L

HINRHINDI	.ED NETWORK ELEMENTS - Florida												Attach	ment: 2	Evhil	oit: C
C. TDO INDE	LE RETRORN LEEMENTO TIONGO										Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
											Elec	Manually		Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						_	Nonrec	urrina	Nonrecurring Disco	onnect			oss	Rates(\$)		
						Rec	First	Add'l		dd'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	All Features Offered			UEPFR	UEPVF	2.26	0.00	0.00				11.90				
LOC	AL NUMBER PORTABILITY					_										
	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		16.97	3.73				11.90				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		16.97	3.73				11.90				
2-WI	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	E LINE I	ORT (BUS)												
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			20.10										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			24.05										
l l	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			29.49										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	18.48										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	22.43										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	27.87										
2-Wi	re Voice Grade Line Port (Bus)															
	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.62	250.00	250.00				11.90				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.62	250.00	250.00				11.90				
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.62	250.00	250.00				11.90				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.62	250.00	250.00				11.90				
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFB	LNPCX	0.35										
INTE	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFB	U1TV2	26.52										
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFB	1L5XX	0.0091										
FEA	TURES															
	All Features Offered			UEPFB	UEPVF	2.26	0.00	0.00				11.90				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFB	USAC2		16.97	3.73				11.90				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch with change			UEPFB	USACC		16.97	3.73				11.90				
2-WI	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			20.10										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			24.05										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			29.49										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	18.48										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	22.43										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	27.87										
2-Wi	re Voice Grade Line Port Rates (BUS - PBX)															
								<u> </u>								
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	1.62	250.00	250.00				11.90				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.62	250.00	250.00				11.90				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.62	250.00	250.00				11.90				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.62	250.00	250.00				11.90				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.62	250.00	250.00				11.90				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.62	250.00	250.00				11.90				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.62	250.00	250.00				11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.62	250.00	250.00				11.90				
1 🗍	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD							-								
<u> </u>	Capable Port	<u>L_</u>		UEPFP	UEPXE	1.62	250.00	250.00	<u> </u>		<u></u>	11.90	<u></u>	<u> </u>	<u> </u>	
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
1 1	Administrative Calling Port	<u> </u>	<u></u>	UEPFP	UEPXL	1.62	250.00	250.00	<u> </u>			11.90	<u> </u>	<u> </u>	<u> </u>	

UNBU	NDLE	D NETWORK ELEMENTS - Florida													Attach	ment: 2	Exhi	bit: C
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	вся	6	usoc			RATES(\$)			1	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
								Rec	Nonrec	urring	Nonrecurrin	g Disconnect		1	oss	Rates(\$)	1	1
								Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy																
		Room Calling Port			UEPFP		UEPXM	1.62	250.00	250.00				11.90				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital																
		Discount Room Calling Port			UEPFP		UEPXO	1.62	250.00	250.00				11.90				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP		UEPXS	1.62	250.00	250.00				11.90				
	LOCAL	NUMBER PORTABILITY																
		Local Number Portability (1 per port)			UEPFP		LNPCP	3.15	0.00	0.00				11.90				
	INTER	OFFICE TRANSPORT																
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility																
		Termination			UEPFP		U1TV2	26.52										
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile	l													1		
		or Fraction Mile			UEPFP		1L5XX	0.0091					ļ]
	FEATU]
		All Features Offered			UEPFP		UEPVF	2.26	0.00	0.00				11.90		1		
	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED	ļ				ļ .						ļ			ļ		ļ
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port	l		l		1			_				1		I		
		Combination - Conversion - Switch-as-is			UEPFP		USAC2		16.97	3.73				11.90				
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port																
		Combination - Conversion - Switch with change			UEPFP		USACC		16.97	3.73				11.90				
		PORT/LOOP COMBINATIONS - COST BASED RATES																
		VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT															
	UNE P	ort/Loop Combination Rates																
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1				23.21										
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2				28.28										
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3				46.53										
	UNE L	pop Rates																
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX		UECD1	14.50						11.90			1.83	
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX		UECD1	19.57						11.90			1.83	
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX		UECD1	37.82						11.90			1.83	
	UNE P	ort Rate			LIEBBY			0.74	0.00					11.00				
		Exchange Ports - 2-Wire DID Port			UEPPX		UEPD1	8.71	850.00	75.00				11.90			1.83	
	NONRE	CURRING CHARGES - CURRENTLY COMBINED																
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -			LIEDDY		110404		7.05	4.07				44.00				
		Switch-as-is			UEPPX		USAC1		7.85	1.87				11.90				
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion																
	ADDIT	with BellSouth Allowable Changes			UEPPX		USA1C		7.85	1.87				11.90				
<u> </u>	ADDITI	ONAL NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX		USAS1		32.26	32.26				11.90				
	Talank	one Number/Trunk Group Establisment Charges			UEPPX		USAST		32.20	32.26				11.90				
	reiepn				UEPPX		NDT	0.00	0.00	0.00				44.00			4.00	
		DID Trunk Termination (One Per Port)		-	UEPPX		NDI	0.00	0.00	0.00				11.90			1.83	
		DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers			UEPPX		ND7	0.00	0.00	0.00				11.90			1.83	
		Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		NDZ ND4	0.00	0.00	0.00				11.90			1.83	
		DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX		ND5	0.00	0.00	0.00				11.90			1.83	
		Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00				11.90			1.83	
		Reserve Non-Consecutive DID numbers Reserve DID Numbers					NDV	0.00	0.00					11.90			1.83	
 	LOCAL	Reserve DID Numbers NUMBER PORTABILITY	-		UEPPX		IADA	0.00	0.00	0.00		1	 	11.90	-		1.83	1
├	LUCAL	Local Number Portability (1 per port)	 		UEPPX		LNPCP	3.15	0.00	0.00		-	<u> </u>			 	-	
 	2-WIDE	EISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LIF	NE SIDI	DODI			LINECE	3.15	0.00	0.00								
		ort/Loop Combination Rates	1 310	I			1					1	1	1	1	 	1	1
-	JINE PO	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	1		ł		1					1	1	1	1	 	1	1
		UNE Zone 1	l	1	UEPPB	UEPPR		32.09								1		
 		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	-		UEFFB	UEPPR	+ +	32.09				 	<u> </u>		-	-	 	
		UNE Zone 2	l	2	UEPPB	UEPPR		38.15								I		
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	 		UEFFB	UEPPR	 	38.15				-	<u> </u>			 	-	
			l	3	HEDDD	UEPPR		59.94								I		
1		UNE Zone 3	-	3	UEPPB	UEPPR	+ +	59.94				 	<u> </u>		-	-	 	
				1							1		1		1	1	•	1

ONROND	DLED NETWORK ELEMENTS - Florida		1									_	_		ment: 2		bit: C
CATEGOR	RY RATE ELEMENTS	Interi m	Zone	E	3CS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	30.77						11.90			1.83	
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	52.56						11.90			1.83	
UN	NE Port Rate			UEPPB	UEPPR	UEPPB	7.38	525.00	400.00				44.00			1.83	
NO	Exchange Port - 2-Wire ISDN Line Side Port ONRECURRING CHARGES - CURRENTLY COMBINED			UEPPB	UEPPR	UEPPB	7.38	525.00	400.00			-	11.09			1.83	
NO	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port	_	1	1													
	Combination - Conversion			UEPPB	UEPPR	USACB	0.00	25.22	17.00				11.90			1.83	
AD	DDITIONAL NRCs			02	OL: III	00/102	0.00	20.22					11.00			1.00	
	OCAL NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-C	CHANNEL USER PROFILE ACCESS:																
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
	CSD		<u> </u>	UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
	CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS : SER TERMINAL PROFILE	SC,MS, 8	TN)	 						ļ						ļ	<u> </u>
US			<u> </u>	LIEDDD	HEDDD	11411540	0.00	0.00	0.00								
VE	User Terminal Profile (EWSD only) ERTICAL FEATURES	_	1	UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								1
VE	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	2.26	0.00	0.00				11.90				
INT	TEROFFICE CHANNEL MILEAGE			UEPPB	UEPPK	UEFVF	2.20	0.00	0.00				11.90				
1141	Interoffice Channel mileage each, including first mile and			1		+											
	facilities termination			LIFPPR	UEPPR	M1GNC	18.4491	47.35	31.78	18.31	7.03		11.90			1.83	
	Interoffice Channel mileage each, additional mile				UEPPR	M1GNM	0.0091	0.00	0.00				11.90			1.83	
4-W	WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUN	IK PORT															
UN	NE Port/Loop Combination Rates																
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																
	Zone 1		1	UEPPP			156.18										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																
	Zone 2		2	UEPPP			181.87										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			UEPPP			074.05										
	Zone 3		3	UEPPP			274.25										
UN	NE Loop Rates 4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	73.44						11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	99.13						11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	191.51						11.90			1.83	
UN	NE Port Rate			OLITI		OOLHI	101.01						11.50			1.00	
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	82.74	1,150.00	1,150.00				11.90			1.83	
NO	ONRECURRING CHARGES - CURRENTLY COMBINED														1		
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port																
	Combination - Conversion -Switch-as-is			UEPPP		USACP	0.00	84.17	61.38				11.90			1.83	
AD	DDITIONAL NRCs																
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-																
	Inward/two way Tel Nos. (except NC)	-	ļ	UEPPP		PR7TF		0.5412		ļ			11.90			1.83	<u> </u>
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -		1	LIEDDE		PR7TO		40.74	40.74				44.00			4.00	
	Outward Tel Numbers (All States except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -	+	<u> </u>	UEPPP		PK/IU		12.71	12.71	-			11.90		-	1.83	-
	Subsequent Inward Tel Numbers		1	UEPPP		PR7ZT		25.42	25.42				11.90			1.83	
10	OCAL NUMBER PORTABILITY	+		OLITE		1 1/1/21		20.42	20.42				11.30			1.03	
	Local Number Portability (1 per port)			UEPPP		LNPCN	1.75										
INT	TERFACE (Provsioning Only)						0								İ		
	Voice/Data			UEPPP		PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP		PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP		PR71E	0.00	0.00	0.00								
Nev	ew or Additional "B" Channel																
	New or Additional - Voice/Data B Channel			UEPPP		PR7BV	0.00	15.48					11.90		ļ	1.83	ļ
	New or Additional - Digital Data B Channel New or Additional Inward Data B Channel		<u> </u>	UEPPP		PR7BF	0.00	15.48					11.90			1.83	ļ
			1	UEPPP		PR7BD	0.00	15.48		1	ı	1	11.90	ı	1	1.83	İ

UNBUNDL	ED NETWORK ELEMENTS - Florida											,		ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual So Order vs Electronic Disc Add
						D	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Inward			UEPPP	PR7C1	0.00	0.00	0.00								
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Interd	office Channel Mileage				1											
	Fixed Each Including First Mile		1	UEPPP UEPPP	1LN1A	88.6256 0.1856	105.54	98.47	21.47	19.05		11.90			1.93	
4 10/15	Each Airline-Fractional Additional Mile RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT			UEPPP	1LN1B	0.1856										-
	Port/Loop Combination Rates				+						-				-	
ONE	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC	+	128.39						11.90			1.83	
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		154.08						11.90			1.83	
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3			UEPDC		246.46						11.90		1	1.83	
UNE	Loop Rates		Ť	- ::										İ	50	
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	73.44						11.90		Ì	1.83	
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	99.13						11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 3			UEPDC	USLDC	191.51			<u> </u>			11.90			1.83	
UNE	Port Rate															
	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	54.95						11.90			1.83	
NONE	RECURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Switch-as-is			UEPDC	USAC4		95.31	46.71				11.90			1.83	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with DS1 Changes			UEPDC	USAWA		95.31	46.71				11.90			1.83	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination											44.00				
ADDI	- Conversion with Change - Trunk			UEPDC	USAWB		95.31	46.71				11.90			1.83	
ADDI	TIONAL NRCs 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -		<u> </u>		_											
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent		1	ULFDC	ODITA		13.09	13.09			1	11.90			1.00	
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel			OLI DO	OBTIB		10.00	10.00				11.50			1.00	
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		15.69	15.69				11.90			1.83	
BIPO	LAR 8 ZERO SUBSTITUTION															
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	655.00				11.90			1.83	
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	655.00				11.90			1.83	
Alteri	nate Mark Inversion															
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
I elep	phone Number/Trunk Group Establisment Charges	<u> </u>	<u> </u>	LIEBBO	LIDTOY	0.00						44.00			4.00	
	Telephone Number for 2-Way Trunk Group			UEPDC UEPDC	UDTGX	0.00						11.90			1.83 1.83	
	Telephone Number for 1-Way Outward Trunk Group		<u> </u>	UEPDC	UDTGY	0.00						11.90 11.90			1.83	
	Telephone Number for 1-Way Inward Trunk Group Without DID DID Numbers, Establish Trunk Group and Provide First Group	}	1	OLPDO	UDIGZ	0.00			1			11.90		1	1.83	-
	of 20 DID Numbers		1	UEPDC	NDZ	0.00	0.00	0.00]			11.90		1	1.83	
	DID Numbers for each Group of 20 DID Numbers	 		UEPDC	ND4	0.00	0.00	0.00				11.90		 	1.83	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00						11.90			1.83	
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				11.90		İ	1.83	
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00				11.90			1.83	
Dedic	cated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS	1 Digital	Loop						i i							
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities								i i							
	Termination)	<u></u>		UEPDC	1LNO1	88.44	105.54	98.47	21.47	19.05		11.90			1.83	
				_												
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities		1]					<u> </u>	_	
	Termination)	1	1	UEPDC	1LNO2	0.00	0.00	0.00	<u> </u>		<u> </u>			1	1	1

ARUNDLE	D NETWORK ELEMENTS - Florida										1			ment: 2		bit: C
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		2	RATES(\$)	,			Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel Mileage - Additional rate per mile - 9-25												1 '			
	miles			UEPDC	1LNOB	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities			LIEBBO	41 1100	0.00	0.00	0.00	0.00				1 '			
_	Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00					igwdots		
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.1856	0.00	0.00					1 '			
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00				 	$\vdash \vdash \vdash$		
	Central Office Termininating Point		1	UEPDC	CTG	0.00	0.00	0.00	0.00							<u> </u>
4-WIRE	E DS1 LOOP WITH CHANNELIZATION WITH PORT			OLI DO	010	0.00							\vdash			1
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	ivations														
	system can have up to 24 combinations of rates depending on			ber of ports used												
	\$1 Loop						İ									
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	73.44	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2			UEPMG	USLDC	99.13	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	191.51	0.00	0.00		_						
UNE D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)											L			
	24 DSO Channel Capacity - 1 per DS1	<u> </u>		UEPMG	VUM24	118.06	0.00	0.00			ļ	11.90	'	Ļ'	1.83	<u> </u>
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	236.12	0.00	0.00				11.90	 '		1.83	
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	472.24	0.00	0.00				11.90			1.83	
	144 DS0 Channel Capacity - 1 per 6 DS1s	<u> </u>	<u> </u>	UEPMG	VUM14	708.36	0.00	0.00				11.90			1.83	
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	944.48	0.00	0.00				11.90	 '		1.83	
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG UEPMG	VUM20 VUM28	1,180.60 1,416.72	0.00	0.00				11.90 11.90	 '		1.83 1.83	
	288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 16 DS1s		<u> </u>	UEPMG	VUM38	1,416.72	0.00	0.00				11.90	 		1.83	-
-	480 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM40	2,361.20	0.00	0.00				11.90	——		1.83	
	576 DS0 Channel Capacity - 1 per 24 DS1s		1	UEPMG	VUM57	2,833,44	0.00	0.00				11.90			1.83	
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3,305.68	0.00	0.00				11.90	\vdash		1.83	
Non-Re	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chani	neliztio					0.00				11.00			1.00	
	mum System configuration is One (1) DS1, One (1) D4 Channe															
Multipl	es of this configuration functioning as one are considered Ad	dd'l afte	r the m	ninimum system con	figuration is	counted.										
	NRC - Conversion (Currently Combined) with or without			, in the second												
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	96.77	4.24				11.90	l '			
	n Additions at End User Locations Where 4-Wire DS1 Loop wi				ination Curre	ently Exists and										
New (N	lot Currently Combined) in all states, except in Density Zone 1	of Top	8 MSA	\'s												
	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port			l			_						1 '	1 '		
	and Assoc Fea Activation			UEPMG	VUMD4	0.00	726.11	468.21	145.32	17.24		11.90	 '	 '		
Bipola	r 8 Zero Substitution				ļ								 '	 '		
	Clear Channel Capability Format, superframe - Subsequent	1		LIEDMC	CCCC	0.00	0.00	055.00				44.00	1 '	1 '		
-	Activity Only	 	-	UEPMG	CCOSF	0.00	0.00	655.00	 		1	11.90	 	\vdash		1
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	655.00				11.90	1 '	1 '		
Alterna	ite Mark Inversion (AMI)	 	 	OLI IVIG	CCOLI	0.00	0.00	055.00	 			11.90		$\vdash \vdash \vdash$		
Aiteilla	Superframe Format	 	 	UEPMG	MCOSF	0.00	0.00	0.00			1		 	\vdash		1
	Extended Superframe Format	 	<u> </u>	UEPMG	MCOPO	0.00	0.00	0.00			1			\vdash		1
Exchar	nge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port		1	2.00	2.00	2.00								
	nge Ports		T		1								[
		1														
	Line Side Combination Channelized PBX Trunk Port - Business	<u></u>	L	UEPPX	UEPCX	1.38	0.00	0.00	0.00	0.00		11.90	<u> </u>	<u> </u>	1.83	
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.38	0.00	0.00	0.00	0.00		11.90			1.83	
													1	1		
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.38	0.00	0.00	0.00	0.00		11.90	 '	<u> </u>	1.83	
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	8.71	0.00	0.00	0.00	0.00		11.90	<u> </u>		1.83	
	e Activations - Unbundled Loop Concentration	<u> </u>									<u> </u>			igspace		<u> </u>
Feature		•	1	1	1	1			_	_			1 '	1 '		1
Feature	Feature (Service) Activation for each Line Side Port Terminated															1
Feature	in D4 Bank			UEPPX	1PQWM	0.66	25.40	13.41	3.96	3.93		11.90	ļ.,,	\vdash	1.83	ļ
Feature	in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated															
	in D4 Bank			UEPPX UEPPX	1PQWM 1PQWU	0.66 0.66	25.40 78.16	13.41	3.96 56.03	10.95		11.90			1.83	

Version 3Q02: 09/06/02

CATEGORY RATE ELEMENTS	UNRUI	NDI FI	NETWORK ELEMENTS - Florida												Attachi	ment: 2	Exhib	nit: C
CATEGORY RATE ELEMENTS Interior 2 miles 2 mile													Svc Order	Svc Order				Incremental
## DES USC RATES(S) PER STATE LEMENTS Based Company																		Charge -
CATEGORY RATE ELEMENTS				١														Manual Svc
Best 176 Girg and Provide 12 (2) Distance Committe	CATEG	ORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)				,				Order vs.
Section Process Proc				m						,			per Loix	per Lor				Electronic-
Pauls Trx Gip and Provide 1at 20 IDI Not. PET. GA. NC.A. SC)																		
Page Page															151	Add I	DISC 1St	Disc Add'l
Section Sect								_	Nonre	curring	Nonrecurring	g Disconnect			oss	Rates(\$)	L L	
DID Numbers - propused 420 - Visited at States DEPPK NP4 0.00 0.03 0.00 11150								Rec			First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Number N			Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00				11.90				
Reserve Pron-Consecution CD Numbers DUEPPX NOE 0.00 0.0			DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				11.90				
Reserve DD Number Portability Load Number Portability Load Number Portability Load Number Portability Load Sunday Lo			Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00				11.90				
Local Renther Profitability Commission			Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00				11.90				
Cock Number Printering - 1 per port OUPPY UPPY 3-15 O.00 O.0			Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				11.90				
PEATURES Vertical and Optional		Local N	lumber Portability															
Local Svitching Fastures Officed with Line Side Ports Only UEPPX UEPVF 2.26 0.00 0.00 11.00 1.16 Market Raine Shall apply entere BellSouth is not required to provide unbundled local svitching or switch ports per FCG and/or State Commission nules. Unbundled portfloop combinations that are Currently Combined or Not Currently Combined or Not Currently Combined in Zone 1 of the Top 8 MSAS in BellSouth's region for one disease with 4 or more D80 equivalent lines. Unbundled portfloop combinations that are Currently Combined or Not Currently Combined in Zone 1 of the Top 8 MSAS in BellSouth's region from end users with 4 or more D80 equivalent lines. Unbundled portfloop combinations that are Currently Combined or Not Currently Combined in Zone 1 of the Top 8 MSAS in BellSouth's region from end users with 4 or more D80 equivalent lines. In the Top 8 MSAS in BellSouth Stall bill the rates in the Cost-Based section proceding in lites of the Market Rates and reserves the right to true-up the billing difference. In the Top 8 MSAS in BellSouth Stall bill the rates in the Cost-Based section proceding in lites of the Market Rates and reserves the right to true-up the billing difference. In the Market Rate or unbundled prior includes all substales features in all states. End Office and Tandens Switching Usage and Common Transport Usage rates in the Fort section of this site exhibit shall apply to all combinations of looplyon review of elements accept for UMC Combined Section And MCC currently Combined Section (MSC). Market Rate or unbundled prior includes all substales features in all states. End Office and Tandens Switching Usage are listed in the First and Additional NRC currently Switching And Additional NRC currently Combined sections, the Nonrecurring charges are listed in the NRC - Currently Combined Section (MSC). Market Rate or unbundled prior switching charges are listed in the First and Additional NRC currently Combined Section (MSC). Market Rate (MSC). Market Rate (MSC). Market Rate (MSC). Market Rate			Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
MR Features Available		FEATU	RES - Vertical and Optional															
UNBINITIOLE PORT LOOP COMBINATIONS - MARKET RATES Washer Tases and lapply where efficient is not registed unbundled local switching or with ports per FCC and/or State Commission rules. Washer Tases and lapply where efficient is not received unbundled local switching or with ports per FCC and/or State Commission rules. The Top 8 MS/face combinations that are currently Combined or Not Currently Combined in Zone 1 of the Top 8 MS/face is a commission rules. The Top 8 MS/face combinations that are currently Combined or Not Currently Combined in Zone 1 of the Top 8 MS/face is a commission rules. The Top 8 MS/face combinations that are currently Combined or Not Currently Combined in Zone 1 of the Top 8 MS/face is a commission rules. The Top 8 MS/face combinations that are currently Combined in Zone 1 of the Top 8 MS/face is a commission rules. The Top 8 MS/face combinations with the Top 8 MS/face is a commission rules. The Top 8 MS/face combination on the Combination of Top 1 of the Top 8 MS/face is a commission rules. The Top 8 MS/face combination on the Currently Combined section preceding in the Ord Post Post Post Post Post Post Post Post	1	Local S	witching Features Offered with Line Side Ports Only															
Market Rises shall apply where BellSouth is not required to provide unbunded local switching or awitch ports per FCC and/or Size Commission on Inc. Unbundled port/loop combinations that are Currently Combined or Not Currently Combined in Jose 1 or the 750 states (Commission or Size Commission or			All Features Available			UEPPX	UEPVF	2.26	0.00	0.00				11.90			1.83	
This includes: Unbundled portloop combinations that are Currently Combined or Not Currently Combined in Zone 1 of the Top & MSAS in BellSouth's region for end users with 4 or more DS0 equivalent lines. The Top & MSAS in BellSouth's region for red users with 4 or more DS0 equivalent lines. The Market Rate for unbundled ports includes all available features in all states. End Office and Transdom Windows and accordingly. The Market Rate for unbundled ports includes all available features in all states. End Office and Transdom Windows are assignation of the Top of the Common Transport Usage are listed in the First and Additional NRC columns for each Port USOC. For Currently Combined scenarios, the Nonrecurring charges are listed in the First and Additional NRC columns for each Port USOC. For Currently Combined scenarios, the Nonrecurring charges are listed in the First and Additional NRC columns for each Port USOC. For Currently Combined scenarios, the Nonrecurring charges are listed in the First and Additional NRC columns for each Port USOC. For Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the																		
This includes: Unbundled portloop combinations that are Currently Combined or Not Currently Combined in Zone 1 of the Top & MSAS in BellSouth's region for red users with 4 or more DS0 equivalent lines. The Top & MSAS in BellSouth's region are Ft, (Ortando, Ft, Landerdale, Mismit) CA, (Atlantal), LA (New Observal), MSC (Greensberg-Winston Salem-Highphonit/Charleste-Bastonia-Rock Hill), Till (Relabvillo). The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports includes all available features in all states. The Market Rate for unbundled ports include a value available features in all states. The Market Rate for unbundled port include for the market Rates and reserves the right to the whole Annual Rates and reserves the right to the whole Annual Rates and reserves the right to the whole Annual Rates and reserves the right to the whole Annual Rates and reserves the right to the whole Annual Rates and reserves the right to the whole Annual Rates and reserves the right to the whole Annual Rates and reserves the right to the whole Annual Rates an				unbun	dled lo	al switching or swit	tch ports per	FCC and/or St	ate Commission	n rules.								
The Top 8 MSAs in BellSouth's region are: TL (Orlando, Pt. Luuderdale, Mismi); CA (Altanus); LA (New Orlanna); MC (Greensboro-Winston Salem-Highpoint/Charlotte Castonia-Rock HBI); TN (Nashvillo). BellSouth shall bill the rates in the Cost-Based section preceding in lieu of the Market Rates in this cost-Based section preceding in lieu of the Market Rates in this cost-Based section preceding in lieu of the Market Rates in the Section						-	I .											
The Top 8 MSAs in BellSouth's region are FL (Orlando, Pt. Laudedale, Mlami); GA (Atlanta); LA (New Orlands); KC (Greensboro-Winston Salem-Highpoint/Charlette-Gastonia-Rock HBI; TN (Nashvillo). BellSouth shall bill the I rate in the Cost-Based section preceding in lisu of the Market Rates in this cost-Based section preceding in lisu of the Market Rates in this cost-Based section preceding in lisu of the Market Rates in the Salem Rates in the Sale		Unbun	dled port/loop combinations that are Currently Combined or I	Not Cur	rently (combined in Zone 1	of the Top 8	MSAS in BellS	outh's region	for end users	with 4 or more	DS0 equivaler	t lines.					
BellSouth currently is developing the billing capability to mechanically bill the recurring and non-recurring Market Rates in the section except for noncourring charges for not currently combined in FL and NC. In the interim where BellSouth canner Rates, BellSouth shall bill the rate bill the Society that the rate bill the Society of the Market Rates and reverse the right to the billing difference. The Market Rate for unburdled ports includes all available features in all states. I I I I I I I I I I I I I I I I I I I														e).				
Rates, BellSouth shall bill the rates in the Cost-Based section proceding in lieu of the Market Rates and reserves the right to true-up the billing difference. The Market Rate for unbunding dopris includes all validable features in all states. Ent Office and Tandem Svitching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of looppion network elements except for UNE Coin Port/Loop Combinations which have a flat rate u. (ISSC: USE															. In the interi	m where Bell	South cannot	bill Market
The Marker Rate for unbundled ports includes all available features in all states. End Office and Trandem Switching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements except for UNE Coin Port/Loop Combinations which have a flar rate (USOC: URECU). For Not Currently Combined scenarios, the Nonrecurring charges are listed in the First and Additional NRC routers of the NRC with a special part of the NRC was apply also and are categorized accordingly. 2 WINE VOICE GRADE LOOP WITE 2-WINE LAND PORT (RES) WILL PLAND WITE 2-WINE LAND PORT (RES) WILL PLAND WITE 2-WINE LAND PORT (RES) WILL PLAND WITE 2-WINE LAND PORT (RES) WILL PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE LAND PORT (RES) WINE PLAND WITE 2-WINE PLAND WITE 2-WINE PLAND WITE 2-WINE 2												,						
End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements except for UNE Coin Port/Loop Combinations which have a flat rate (USOC: UNECO). For Not Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios, the Nonrecurring charges are listed in the NRC - Currently Combined scenarios. Valving Voice Condet Loop WiTT 2 WiTT Except Valving Currently Currentl								1										
USOC: URECU).						e Port section of the	is rate exhib	it shall apply to	all combination	ons of loop/po	rt network ele	ments except	for UNE Coi	n Port/Loor	Combination	ns which have	a flat rate us	age charge
For Not Currently Combined scenarios the Nonrecurring charges are listed in the First and Additional NRC any apply also and are categorized accordingly.																		-99-
Additional NRCs may apply also and are categorized accordingly. 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LIRP PORT (RES) VINE Port/Loop Combination Rates 2-Wire VIOL ExppProt Combio - Zone 2 1 1 26,94				lictod	in tha E	iret and Additional	NPC column	e for each Dor	LISOC For C	urrently Comb	inad coonariae	the Menrecui	ring charge	e aro lietod	in the NPC - (Currently Con	hinad caction	•
2-Wire Voice Grade LoOP WITH 2-WIRE LINE PORT (RES)				iisteu	iii tiie r	ii st aiiu Auuitioiiai	NAC COIGINI	S IOI Eacii FOII	USOC. FOI C	urrently Comb	illeu scellarios	s, the Noniecui	illig charge	s are iisteu	iii tile NKC - V	Currently Con	ibilieu sectioi	
UNE Port/Logo Combination Rates							1	1			1	1						
2-Wire Vol Loop/Port Combo - Zone 2 2 31.06					<u> </u>													
2-Wire VSL Loop/Fort Combo - Zone 2 2 31.06		UNE PO			-			20.04										
2-Wire Voice Grade Long (SL1) - Zone 1																		
UNE Loop Rates																		
2-Wire Voice Grade Loop (SL1) - Zone 1	-				3			45.87										
2-Wire Voice Grade Loop (SL1) - Zone 2 2 UEPRX UEPLX 17.06		UNE LO			_	HEDDY	LIEDLY	40.04										
2-Wire Voice Grade Loop (SL1) - Zone 3 3 UEPRX UEPLX 31.87																		
2-Wire voice unbundled port - residence																		
2-Wire voice unbundled port -residence		0 14/:			3	UEPKX	UEPLX	31.87										
2-Wire voice unbundled port with Caller ID - res		2-wire			-	HEDDY	LIEDDI	44.00	00.00	00.00				44.00				
2-Wire voice unbundled port outgoing only - res					-													
2-Wire voice unbundled Florida Area Calling with Caller ID - res UEPRX UEPAF 14.00 90.00 90.00 90.00 11.90																		
2-Wire voice unbundles res, low usage line port with Caller ID (LUM) UEPRX UEPAP 14.00 90.00 90.00 11.90 11.90			2-vvire voice unbunatea port outgoing only - res		<u> </u>	UEPKX	UEPRO	14.00	90.00	90.00				11.90		-		
2-Wire voice unbundles res, low usage line port with Caller ID (LUM) UEPRX UEPAP 14.00 90.00 90.00 11.90 11.90			O Miles veins veins and od Florido Asso Colling with College D			LIEDDY	LIEBAE	44.00	00.00	00.00				44.00				
CLUM UEPRX UEPAP 14.00 90.00 90.00 11.90 11.90 11.90 2-Wire voice unbundled Low Usage Line Port without Caller ID UEPRX UEPRT 14.00 90.00 90.00 90.00 11.90 11.90					1	UEPKA	UEPAF	14.00	90.00	90.00	1	1	1	11.90		 		
2-Wire voice unbundled Low Usage Line Port without Caller ID UEPRX					1	LIEDDY	LIEDAS	44.00	20.00	20.00				44.00		I		
Capability					-	UEPKX	UEPAP	14.00	90.00	90.00				11.90		 		
2-Wire voice unbundled Florida extended dialing port for use with CREX7 and Caller ID 2-Wire voice unbundled Florida extended dialing port for use with CREX7, without Caller ID capability UEPRX UEPA8 14.00 90.00 90.00 11.90 2-Wire voice unbundled Florida extended dialing port for use with CREX7, without Caller ID capability UEPRX UEPA8 14.00 90.00 90.00 11.90 11.90 LOCAL NUMBER PORTABILITY Local Number Portability (1 per port) UEPRX UEPRX UEPA9 14.00 90.00 90.00 11.90 11.90 11.90 2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is UEPRX UEPRX USAC2 41.50 41.50 41.50 11.90 ADDITIONAL NRCS NRC - 2-Wire Voice Grade Loop/Line Port Combination -					1	LIEDDY	LIEDET	44.00	00.00	00.00				44.00		I		
With CREX7 and Caller ID					<u> </u>	UEPKA	UEPKI	14.00	90.00	90.00	1	1	-	11.90		-		
2-Wire voice unbundled Florida extended dialing port for use with CREX7, without Caller ID capability UEPRX UEPA8 14.00 90.00 90.00 11.90					1	HEDDY	LIEBAA	44.00	00.00	00.00				44.00		I		
with CREX7, without Caller ID capability					<u> </u>	UEPKX	UEPA1	14.00	90.00	90.00				11.90		-		
2-Wire voice unbundled Florida Årea Calling Port without Caller					1		l==				Ì							
D Capability					<u> </u>	UEPKX	UEPA8	14.00	90.00	90.00		ļ		11.90				
LOCAL NUMBER PORTABILITY Local Number Portability (1 per port) UEPRX LNPCX 0.35 LNPCX 0.						HEDDY	LIEDAS											
Local Number Portability (1 per port)					<u> </u>	UEPKX	UEPA9	14.00	90.00	90.00		ļ		11.90				
FEATURES					 		L											
All Features Offered						UEPRX	LNPCX	0.35										
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Switch with change UEPRX USAC2 41.50 41.50 11.90 LEPRX USACC 41.50 41.50 11.90 NRC - 2-Wire Voice Grade Loop/Line Port Combination -	ļ.	FEATU			<u> </u>		l	ļ			ļ		1	ļ		1		
2-Wire Voice Grade Loop / Line Port Combination - Switch with change UEPRX USACC 41.50 41.50 11.90 11.			All Features Offered		 	UEPRX	UEPVF	0.00	0.00	0.00				11.90		.		
2-Wire Voice Grade Loop / Line Port Combination - Switch with change UEPRX USACC 41.50 41.50 11.90 11.					1		1]						1		I		
change					<u> </u>	UEPRX	USAC2		41.50	41.50				11.90				
ADDITIONAL NRCs NRC - 2-Wire Voice Grade Loop/Line Port Combination -																		
NRC - 2-Wire Voice Grade Loop/Line Port Combination -						UEPRX	USACC		41.50	41.50				11.90				
		ADDITI				-												
	T		•		1				<u> </u>									-
						UEPRX	USAS2		0.00	0.00				11.90				
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)		2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)			-												

Version 3Q02: 09/06/02

NRANDI	ED NETWORK ELEMENTS - Florida	,		,									Attachr			oit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incrementa Charge - Manual Sv Order vs.
													Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic Disc Add'
						_	Nonrec	urrina	Nonrecurring	Disconnect			oss	Rates(\$)	1	1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			26.94										
	2-Wire VG Loop/Port Combo - Zone 2		2			31.06										
	2-Wire VG Loop/Port Combo - Zone 3		3			45.87										
UNF	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	12.94										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	17.06										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	31.87			+							
2-W	ire Voice Grade Line Port (Bus)		Ť	02. 5/	02.2.	01.01			+							
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	14.00	90.00	90.00	+			11.90				
	2-Wire voice unbundled port with Caller + E484 ID - bus	1	1	UEPBX	UEPBC	14.00	90.00	90.00	 		-	11.90			 	
	2-Wire voice unbundled port outgoing only - bus	1	1	UEPBX	UEPBO	14.00	90.00	90.00				11.90				
	2-Wire voice unbundled incoming Only Port without Caller ID	 	 	021 0/	02.00	17.00	30.00	30.00	 			11.00			 	
	Capability			UEPBX	UEPBE	14.00	90.00	90.00				11.90				
1.00	CAL NUMBER PORTABILITY	1	1	OLI DA	OLI DL	14.00	90.00	50.00			1	11.50			1	
LOC	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35			+							
NON	IRECURRING CHARGES - CURRENTLY COMBINED			UEPBA	LINPUX	0.35			+							
NON	IRECORRING CHARGES - CORRENTLY COMBINED		<u> </u>						-							
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is	:		UEPBX	USAC2		41.50	41.50				11.90				
	2-Wire Voice Grade Loop / Line Port Combination - Switch with															
	change			UEPBX	USACC		41.50	41.50				11.90				
ADD	OITIONAL NRCs															
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -															
	Subsequent			UEPBX	USAS2		0.00	0.00				11.90				
	IRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			26.94										
	2-Wire VG Loop/Port Combo - Zone 2		2			31.06										
	2-Wire VG Loop/Port Combo - Zone 3		3			45.87										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRG	UEPLX	12.94										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRG	UEPLX	17.06										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRG	UEPLX	31.87										
2-W	ire Voice Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	14.00	90.00	90.00				11.90				
1.00	CAL NUMBER PORTABILITY			UEPRG	UEPRD	14.00	90.00	90.00	+			11.90				-
LOC			<u> </u>	UEPRG	LNPCP	3.15	0.00	0.00	-							
FFA	Local Number Portability (1 per port) TURES			UEPRG	LINECE	3.13	0.00	0.00	-							
FEA			<u> </u>	UEPRG	UEPVF	0.00	0.00	0.00	-			44.00				
NON	All Features Offered IRECURRING CHARGES - CURRENTLY COMBINED			UEPRG	UEPVF	0.00	0.00	0.00				11.90				
NON	IRECURRING CHARGES - CURRENTLY COMBINED								<u> </u>							<u> </u>
	OME Visit On the Land Visit Dark On the Control of the Anta-			LIEDDO	110400		44.50	44.50				44.00				
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is		<u> </u>	UEPRG	USAC2		41.50	41.50				11.90				
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with															
	Change			UEPRG	USACC		41.50	41.50				11.90				
ADL	DITIONAL NRCs															
	2 Wire Loop/Line Side Port Combination - Non feature -															
_	Subsequent Activity- Nonrecurring	!	<u> </u>				0.00	0.00				11.90			ļ	
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt		1								I				Ì	
	Group		<u> </u>				7.09	7.09				11.90				
	IRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
UNE	Port/Loop Combination Rates	1													ļ	
	2-Wire VG Loop/Port Combo - Zone 1		1			26.94										
	2-Wire VG Loop/Port Combo - Zone 2		2			31.06										
	2-Wire VG Loop/Port Combo - Zone 3		3			45.87										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	12.94		-								
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	17.06										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	31.87										

JNBUNDL	ED NETWORK ELEMENTS - Florida												ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)		Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring Disconne				Rates(\$)		
						Nec	First	Add'l	First Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wir	e Voice Grade Line Port Rates (BUS - PBX)														
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00			11.90				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00			11.90				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD														
	Capable Port			UEPPX	UEPXE	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy														
	Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy														
	Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital														
	Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00			11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00			11.90				
LOCA	AL NUMBER PORTABILITY														
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00							
FEAT	TURES														
	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00			11.90				
NONE	RECURRING CHARGES - CURRENTLY COMBINED														
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPPX	USAC2		41.50	41.50			11.90				
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with														
	Change			UEPPX	USACC		41.50	41.50			11.90				
ADDI	TIONAL NRCs														
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2	0.00	0.00	0.00			11.90				
	2 Wire Loop/Line Side Port Combination - Non feature -														ĺ
	Subsequent Activity- Nonrecurring						0.00	0.00			11.90				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt														
	Group						7.09	7.09			11.90				
2-WIF	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POP	RT													
UNE	Port/Loop Combination Rates														ĺ
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			26.94									
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			31.06									1
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			45.87									1
UNE	Loop Rates														1
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	12.94									
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	17.06									1
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	31.87			i						
2-Wir	e Voice Grade Line Port Rates (Coin)								1						
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,								1						
	900/976, 1+DDD (FL)			UEPCO	UEP2F	14.00	90.00	90.00			11.90				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking		1		1				† †		1	İ	İ	İ	1
	(FL)			UEPCO	UEPFA	14.00	90.00	90.00			11.90				
	2-Wire Coin 2-Way with Operator Screening and Blocking:		1		1				† †		1	İ	İ	İ	1
	900/976, 1+DDD, 011+, and Local (FL)			UEPCO	UEPCG	14.00	90.00	90.00			11.90				
	2-Wire Coin Outward with Operator Screening and 011 Blocking		1		1			22.30	† †		1	İ	İ	İ	1
	(AL, FL)		1	UEPCO	UEPRK	14.00	90.00	90.00			11.90	I	Ì		
	2-Wire Coin Outward with Operator Screening and Blocking:		t					22.30			1	t	1		
	900/976, 1+DDD, 011+ (FL)			UEPCO	UEPOF	14.00	90.00	90.00			11.90				
	2-Wire Coin Outward with Operator Screening and Blocking:		t				55.50	55.50			150	t	1		
	900/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	14.00	90.00	90.00			11.90				
1.00/	AL NUMBER PORTABILITY		<u> </u>								1	1	1		
ILUCA				UEPCO	LNPCX	0.35									

UNRI	JNDI F	D NETWORK ELEMENTS - Florida													Attach	ment: 2	Fyhil	oit: C
5.40	,110LL	I TET TOTAL ELEMENTO - I IONG											Svc Order	Svc Order	Incremental		Incremental	
													Submitted	Submitted		Charge -	Charge -	Charge -
													Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATE	GORY	RATE ELEMENTS	Interi	Zone	ВС	s	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m										per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
															1st	Add'l	Disc 1st	Disc Add'l
								_ 1	Nonred	urring	Nonrecurring	g Disconnect			oss	Rates(\$)	ı	ı
								Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	NONRE	CURRING CHARGES - CURRENTLY COMBINED																
		2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPCO		USAC2		41.50	41.50				11.90				
		2-Wire Voice Grade Loop/ Line Port Combination - Switch with																
		Change			UEPCO		USACC		41.50	41.50								
	ADDITI	ONAL NRCs																
		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPCO		USAS2		0.00	0.00				11.90				
UNBU	NDLED F	PORT/LOOP COMBINATIONS - MARKET BASED RATES																
		VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT															
		ort/Loop Combination Rates	1					1			İ	1				İ	İ	İ
	1	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1	1	1				69.50			İ	1				İ	İ	İ
	1	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2	1	2				74.57			t	t				t		
	1	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3	1	3				92.82			1	1	İ	1				
	UNF I	pop Rates	1	Ť				02.02			t	t	1			t	1	1
	J. 12 E	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1	1	1	UEPPX		UECD1	14.50			t	t	1	11.90		t	1.83	1
	1	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2	1	2	UEPPX		UECD1	19.57			t	t	1	11.90		t	1.83	1
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3			UEPPX		UECD1	37.82						11.90			1.83	
		ort Rate			OLITA		OLOD1	01.02						11.00			1.00	
	O.V.E.T.	Exchange Ports - 2-Wire DID Port			UEPPX		UEPD1	55.00	850.00	75.00				11.90			1.83	
	NONRE	ECURRING CHARGES - CURRENTLY COMBINED			OLITA		OLI DI	00.00	000.00	70.00				11.50			1.00	
	HOHILE	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -												1				
		Switch-As-Is Top 8 MSAs only			UEPPX		USAC1		850.00	75.00				11.90				
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion			OLITA		OOAOT		030.00	75.00				11.30				
		with BellSouth Allowable Changes Top 8 MSAs only			UEPPX		USA1C		850.00	75.00				11.90				
	ADDIT	ONAL NRCs			OLFFX		USAIC		030.00	75.00				11.90				
		2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX		USAS1		32.26	32.26				11.90				
		one Number/Trunk Group Establisment Charges			OLITA		OOAOT		32.20	32.20				11.30				
	relepii	DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00				11.90			1.83	
		DID Numbers, Establish Trunk Group and Provide First Group			OLITA		NDI	0.00	0.00	0.00				11.30			1.00	
		of 20 DID Numbers			UEPPX		NDZ	0.00	0.00	0.00				11.90			1.83	
		Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		ND4	0.00	0.00	0.00				11.90			1.83	
-		DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX		ND5	0.00	0.00	0.00	-	-		11.90		-	1.83	
-		Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00	-	-		11.90		-	1.83	
-		Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00	-	-		11.90		-	1.83	
—		. NUMBER PORTABILITY	1		OLFFA		NDV	0.00	0.00	0.00	 	 	1	11.90		 	1.03	1
-		Local Number Portability (1 per port)	ł	1	UEPPX		LNPCP	3.15	0.00	0.00			 	 		-	-	-
—		EISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	DOD*			LINE OF	3.13	0.00	0.00			-				-	-
—		ort/Loop Combination Rates	IAE SIDE	FURI	1		-	 					-				-	-
—	DIVE P	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	 				-						-				-	-
1	1		1	1	UEPPB	UEPPR		04.74			1	1						
_	+	UNE Zone 1	1	1	UEPPB	UEPPR	-	94.71			-	-	-	 		-		
1	1	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2	1	2	UEPPB	UEPPR		100.77			1	1						
\vdash	1	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	 		UEFFB	UEPPR	-	100.77			 	 	1			 		
	1		1	3	LIEDDD	HEDDO		400.50			1	1						
<u> </u>	LINE :	UNE Zone 3	l	3	UEPPB	UEPPR		122.56			 	1	1	1		1		-
-	UNE LO	Dop Rates	1	_	LIEDES	HEDDO	LICL OV	0171			 	 	1	44.00		 	1.00	-
	1	2-Wire ISDN Digital Grade Loop - UNE Zone 1	!	7	UEPPB	UEPPR	USL2X	24.71			1	1	1	11.90		1	1.83	
1		2 Wire ISDN Digital Crade Loop LINE 7 0	1	2	UEPPB	UEPPR	USL2X	30.77			I	I		44.00		I	1.83	
<u> </u>	1	2-Wire ISDN Digital Grade Loop - UNE Zone 2	!								1	1	1	11.90		1		
<u> </u>	1.157= -	2-Wire ISDN Digital Grade Loop - UNE Zone 3	!	3	UEPPB	UEPPR	USL2X	52.56			.	.		11.90		-	1.83	1
	UNE P	ort Rate	!		LIEDES	LIEDES	LIEBES	20.0	F0= 0-	100.5	.	.				-		1
		Exchange Port - 2-Wire ISDN Line Side Port	ļ		UEPPB	UEPPR	UEPPB	70.00	525.00	400.00				11.09			1.83	
	NONRE	CURRING CHARGES - CURRENTLY COMBINED	ļ								.	.				.		
		2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port	1		l		l				I	I				I		
		Combination - Conversion - Top 8 MSAs only	<u> </u>		UEPPB	UEPPR	USACB	0.00	215.00	215.00	1	1		11.90			1.83	
		ONAL NRCs	<u> </u>								ļ	ļ	<u> </u>			ļ		
		NUMBER PORTABILITY	ļ															l
		Local Number Portability (1 per port)	ļ		UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
	B-CHA	NNEL USER PROFILE ACCESS:				_												

INBLINDI	.ED NETWORK ELEMENTS - Florida													Attach	ment: 2	Evhil	oit: C
,,,DONDE	LE TEL TORRE LEMENTO - I IONG											Svc Order	Svc Order	Incremental		Incremental	
												Submitted	Submitted		Charge -	Charge -	Charge -
		1										Elec	Manually		Manual Svc		Manual Svo
CATEGORY	RATE ELEMENTS	Interi	Zone	F	BCS	usoc			RATES(\$)						Order vs.	Order vs.	Order vs.
		m]		1			==(4)			per LSR	per LSR	Order vs.	Electronic-		
			1											Electronic-		Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
$\overline{}$	+	1		1		1	_ 1	Nonrec	urrina	Nonrecurring	Disconnect			088	Rates(\$)		
-+-		1					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-+-	CVS/CSD (DMS/5ESS)	1		UEPPB	UEPPR	U1UCA	0.00	0.00	0.00	11100	Addi	COME	COMPAN	COMPAR	COMPAR	COMPAR	COMPAR
	CVS (EWSD)	1		UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
-+-	CSD	1		UEPPB		U1UCC	0.00	0.00	0.00								
B-CH	HANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C.MS. 8	: TN)	02	OL: : : t	0.000	0.00	0.00	0.00								
	R TERMINAL PROFILE	1	1														
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VER	TICAL FEATURES			02	02	0.10.11.11	0.00	0.00	0.00								
	All Vertical Features - One per Channel B User Profile	1		UEPPB	UEPPR	UEPVF	2.26	0.00	0.00				11.90				
INTE	ROFFICE CHANNEL MILEAGE	1															
	Interoffice Channel mileage each, including first mile and	1															
1	facilities termination			UEPPB	UEPPR	M1GNC	18.4491	47.35	31.78	18.31	7.03	1	11.90	Ì	I	1.83	
-	Interoffice Channel mileage each, additional mile	†			UEPPR	M1GNM	0.0091	0.00	0.00	.5.51	50		11.90	1	t	1.83	
4-WII	RE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	K PORT	1	2=: 10	UZ. 1 IX		5.0001	0.00	0.00	 			11.50		 	1.00	
	Port/Loop Combination Rates	1	1	1		+				 					 		
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1	1	1		1				t	1			1	t		
1	Zone 1		1	UEPPP			973.44			1					1		
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1	<u> </u>	52111		1	0,0.44			†				1	<u> </u>		
1	Zone 2		2	UEPPP			999.13			I		1		Ì	I		
-+-	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1		OL: II			555.10										
	Zone 3		3	UEPPP			1,091.51										
LINE	Loop Rates	1		OLITI			1,031.31										
ONE	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	73.44						11.90			1.83	
-+-	4-Wire DS1 Digital Loop - UNE Zone 2	1	2	UEPPP		USL4P	99.13						11.90			1.83	
-+-	4-Wire DS1 Digital Loop - UNE Zone 3	1	3	UEPPP		USL4P	191.51						11.90			1.83	
LINE	Port Rate	1		OL: II		OOL-II	101.01						11.50			1.00	
ONE	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	900.00	1,150.00	1,150.00				11.90			1.83	
NON	RECURRING CHARGES - CURRENTLY COMBINED	1		OLITI		OLITI	500.00	1,100.00	1,100.00				11.50			1.00	
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port	1															
	Combination - Conversion -Switch-As-Is Top 8 MSAs only			UEPPP		USACP	0.00	925.00	925.00				11.90			1.83	
ADDI	ITIONAL NRCs	1		OL: II		00/101	0.00	020.00	020.00				11.50			1.00	
ADDI	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-	1															
	Inward/two way Telephone Numbers (except NC)			UEPPP		PR7TF		0.5412					11.90			1.83	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -	1		OLITI		1 187 11		0.5412		†			11.50			1.00	
	Outward Tel Numbers (All States except NC)			UEPPP		PR7TO		12.71	12.71				11.90			1.83	
-+-	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -	1		OL: II		110710		12.71	12.71				11.50			1.00	
	Subsequent Inward Telephone Numbers			UEPPP		PR7ZT		25.42	25.42	I		1	11.90	Ì	I	1.83	
1.00	AL NUMBER PORTABILITY	+	1	OLI FF		11/41		20.42	20.42	 			11.50	 	 	1.03	
	Local Number Portability (1 per port)	+	1	UEPPP		LNPCN	1.75			 	1	1	1	1	 		
INTE	RFACE (Provsioning Only)	+	1	JE: 11		_141 014	1.75			 				 	 		
	Voice/Data	1	1	UEPPP		PR71V	0.00	0.00	0.00	-					-		
-+-	Digital Data	+	1	UEPPP		PR71D	0.00	0.00	0.00	 				 	 		
-+-	Inward Data	+	1	UEPPP		PR71E	0.00	0.00	0.00	 				 	 		
New	or Additional "B" Channel	+	1	JEI'FF		1 1X/ 1L	0.00	0.00	0.00	 	1	1	1	1	 		
IACM (New or Additional - Voice/Data B Channel	+	1	UEPPP		PR7BV	0.00	20.00		 	1	1	11.90	1	 	1.83	
-+-	New or Additional - Voice/Data B Channel	+	1	UEPPP		PR7BF	0.00	20.00		 	1	1	11.90	1	 	1.83	
-+-	New or Additional Inward Data B Channel	+		UEPPP		PR7BD	0.00	20.00		 			11.90	 	 	1.83	
CALI	L TYPES	1	1	J_111		. 13700	0.00	20.00		 			11.30		 	1.03	
- JOALL	Inward	+	1	UEPPP		PR7C1	0.00	0.00	0.00	 				 	 		
-+-	Outward	+	1	UEPPP		PR7C0	0.00	0.00	0.00	 				 	 		
	Two-way	+	1	UEPPP		PR7CC	0.00	0.00	0.00	 				 	 		
		+	1	J_111		. 11750	0.00	0.00	0.00	 					 		
Interd	office Channel Mileage		i .	LIEDDD		1LN1A	88.6256	105.54	98.47	21.47	19.05		11.90		 	1.93	
Interd	office Channel Mileage Eixed Each Including First Mile	1		UFPPP			00.0200	100.04	30.47	∠1.+1	19.00		11.50	1	1	1.00	
Interd	Fixed Each Including First Mile			UEPPP			0.1856										
	Fixed Each Including First Mile Each Airline-Fractional Additional Mile			UEPPP		1LN1B	0.1856										
4-WIF	Fixed Each Including First Mile Each Airline-Fractional Additional Mile RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT						0.1856										
4-WIF	Fixed Each Including First Mile Each Airline-Fractional Additional Mile RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT Port/Loop Combination Rates		1	UEPPP									11 00			1 82	
4-WIF	Fixed Each Including First Mile Each Airline-Fractional Additional Mile RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT Port/Loop Combination Rates 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1 2	UEPPP			128.39						11.90			1.83	
4-WIF	Fixed Each Including First Mile Each Airline-Fractional Additional Mile RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT Port/Loop Combination Rates		1 2 3	UEPPP									11.90 11.90 11.90			1.83 1.83 1.83	

UNBU	NDLE	NETWORK ELEMENTS - Florida												Attach	ment: 2	Exhil	oit: C
												Svc Order	Svc Order	Incremental		Incremental	
			1	1								Submitted	Submitted		Charge -	Charge -	Charge -
												Elec	Manually		Manual Svc		Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)								
CAILG	OKI	RATE ELEMENTS	m	Zone	603	0300			KATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
				<u> </u>			Rec	Nonred		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	73.44						11.90			1.83	
		4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	99.13						11.90			1.83	
		4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	191.51						11.90			1.83	
	UNE Po																
		4-Wire DDITS Digital Trunk Port	1	1	UEPDC	UDD1T	750.00	1.019.56	479.87	204.92	20.10		11.90			1.83	
		CURRING CHARGES - CURRENTLY COMBINED		1	OLI DO	ODDII	700.00	1,010.00	410.01	204.02	20.10		11.00			1.00	
	HOHIL	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination		1													
					LIEDDO	USAC4		05.04	40.74				44.00			4.00	
		- Switch-As-Is Top 8 MSAs only		<u> </u>	UEPDC	USAC4		95.31	46.71				11.90			1.83	
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
		- Conversion with DS1 Changes Top 8 MSAs only			UEPDC	USAWA		95.31	46.71				11.90			1.83	
		<u> </u>															
]		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	1	1		1				I	1		I	1	1	1	1
		- Conversion with Change - Trunk Top 8 MSAs only	1	1	UEPDC	USAWB		95.31	46.71	I	1		11.90	1	1	1.83	1
	ידוחח	ONAL NRCs	 	 		55		55.51	70.71	t		1	11.50	 	t	1.55	
	וווטטא	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -	 	1		1				 		1	 	1	1	1	
			1	1	LIEDDC	UDTTA		45.00	45.00	I		1	44.00	Ì	I	4.00	I
		Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDITA		15.69	15.69				11.90			1.83	
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
		Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		15.69	15.69				11.90			1.83	
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
		Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		15.69	15.69				11.90			1.83	
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
		Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		15.69	15.69				11.90			1.83	
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan		1	02. 50	00110		10.00	10.00				11.00			1.00	
		Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		15.69	15.69				11.90			1.83	
	DIDOL (R 8 ZERO SUBSTITUTION	1	1	UEPDC	ODITE		13.69	13.69				11.90			1.03	
						00005										4.00	
		B8ZS -Superframe Format			UEPDC	CCOSF		0.00	655.00				11.90			1.83	
		B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	655.00				11.90			1.83	
	Alterna	te Mark Inversion															
		AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
		AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
	Teleph	one Number/Trunk Group Establisment Charges															
		Telephone Number for 2-Way Trunk Group	1	1	UEPDC	UDTGX	0.00						11.90			1.83	
		Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00						11.90			1.83	
			1		UEPDC	UDTGZ	0.00					ļ	11.90			1.83	
		Telephone Number for 1-Way Inward Trunk Group Without DID	-	-	ULFDC	UDIGE	0.00			-		 	11.90	-	-	1.83	-
		DID Numbers, Establish Trunk Group and Provide First Group	1	1		1				I	1			1	1		1
		of 20 DID Numbers	<u> </u>	<u> </u>	UEPDC	NDZ	0.00	0.00	0.00	ļ		1	11.90		ļ	1.83	ļ
		DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00						11.90			1.83	<u> </u>
		DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00						11.90			1.83	
		Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				11.90			1.83	
		Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00				11.90			1.83	
	Dedicat	ed DS1 (Interoffice Channel Mileage) -															
		for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port								1							
\vdash	00	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities	 	 		+ -	1			 		1	 	 	1	1	
		Termination)			LIEDDC	1LNO1	88.44	105.54	98.47	21.47	10.05		11.90			1 00	
\vdash		reminauo(I)		 	UEPDC	ILINUT	88.44	105.54	98.47	21.47	19.05	1	11.90	 	-	1.83	-
		Literative Observative Administrative and a second	1	1	LIEDDO	41.110.	6 10=-			I	1		I	1	1	1]
		Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.1856	0.00	0.00	ļ		ļ					ļ
		Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities				1				1							
		Termination)	<u></u>	Щ_	UEPDC	1LNO2	0.00	0.00	0.00		<u> </u>	<u> </u>		L			<u> </u>
		Interoffice Channel Mileage - Additional rate per mile - 9-25															
		miles			UEPDC	1LNOB	0.1856	0.00	0.00	1							
		Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
		Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							
\vdash		· ommanorij	 	 	021 00	121100	0.00	0.00	0.00	0.00	1	1	 	1	1	1	1
		Intereffice Channel Milesen Additional anternation 25 C5 C7	1	1	LIEDDO	41 NOC	0.4050	0.00	0.00	I	1		I	1	1	1	1
\vdash		Interoffice Channel Mileage - Additional rate per mile - 25+ miles	<u> </u>	 	UEPDC	1LNOC	0.1856	0.00	0.00		ļ	!	.	ļ			
		Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00							
		Central Office Termininating Point		<u> </u>	UEPDC	CTG	0.00					<u> </u>					<u> </u>
		DS1 LOOP WITH CHANNELIZATION WITH PORT			-				-								
	System	is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	ivations	5													

UNBUNDI	ED NETWORK ELEMENTS - Florida												Attach	ment: 2	Fyhil	oit: C
CABOADE	LE TETTORICE LEMENTO TIONGA										Svc Order	Svc Order	Incremental		Incremental	Incremental
						1					Submitted	Submitted		Charge -	Charge -	Charge -
		l									Elec	Manually		Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						_ 1	Nonrec	urrina	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
A sv	stem can have various rate combinations based on type and nu	mber of	ports	used												
	DS1 Loop															
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	73.44	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	99.13	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	191.51	0.00	0.00								
UNE	DSO Channelization Capacities (D4 Channel Bank Configuratio	ns)														
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	118.06	0.00	0.00				11.90			1.83	
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	236.12	0.00	0.00				11.90			1.83	
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	472.24	0.00	0.00				11.90			1.83	
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	708.36	0.00	0.00				11.90			1.83	
	192 DS0 Channel Capacity -1 per 8 DS1s	1	1	UEPMG	VUM19	944.48	0.00	0.00	1	İ		11.90	İ	İ	1.83	
	240 DS0 Channel Capacity - 1 per 10 DS1s	1	1	UEPMG	VUM20	1,180.60	0.00	0.00	1	İ		11.90	İ	İ	1.83	
	288 DS0 Channel Capacity - 1 per 12 DS1s	1		UEPMG	VUM28	1,416.72	0.00	0.00				11.90	İ	İ	1.83	
	384 DS0 Channel Capacity - 1 per 16 DS1s	1		UEPMG	VUM38	1,888.96	0.00	0.00				11.90	İ	İ	1.83	
	480 DS0 Channel Capacity - 1 per 20 DS1s		t	UEPMG	VUM40	2,361.20	0.00	0.00				11.90	1	t	1.83	
	576 DS0 Channel Capacity -1 per 24 DS1s		t	UEPMG	VUM57	2.833.44	0.00	0.00				11.90	1	t	1.83	
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3,305.68	0.00	0.00				11.90			1.83	
Non-	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop wit	h Chanı	neliztio													
	nimum System configuration is One (1) DS1, One (1) D4 Channel															
	iples of this configuration functioning as one are considered A															
	NRC - Conversion (Currently Combined) with or without	1	1													
	BellSouth Allowed Changes - Top 8 MSAs Only			UEPMG	USAC4	0.00	450.00	50.00				11.90				
Svst	em Additions Where Currently Combined and New (Not Current	ly Comb	pined)													
	ensity Zone 1 Top 8 MSAs	ĺ	, , ,													
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc															
	Fea Activation -			UEPMG	VUMD4	0.00	950.00	600.00	200.00	30.00		11.90				
Bipo	lar 8 Zero Substitution											11.90				
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	655.00				11.90				
	Clear Channel Capability Format - Extended Superframe -															
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	655.00				11.90				
Alter	nate Mark Inversion (AMI)															
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
Exch	ange Ports Associated with 4-Wire DS1 Loop with Channelizati	on with	Port													
	lange Ports		1		1	†			1	İ			İ	İ	İ	
		1	1		1	† 1			1	İ			İ	İ	İ	
	Line Side Combination Channelized PBX Trunk Port - Business		1	UEPPX	UEPCX	14.00	0.00	0.00	0.00	0.00		11.90	Ì	I	1.83	
	Line Side Outward Channelized PBX Trunk Port - Business	1		UEPPX	UEPOX	14.00	0.00	0.00	0.00	0.00		11.90	İ	İ	1.83	
		1	1		1					1			İ	İ		
	Line Side Inward Only Channelized PBX Trunk Port without DID		1	UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00		11.90	Ì	I	1.83	
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port	1	1	UEPPX	UEPDM	55.00	0.00	0.00	0.00	0.00		11.90	İ	İ	1.83	
Feat	ure Activations - Unbundled Loop Concentration	1	1		1					1			İ	İ		
	Feature (Service) Activation for each Line Side Port Terminated	1			1	† 1							İ	İ	İ	
	in D4 Bank		1	UEPPX	1PQWM	0.66	40.00	20.00	6.00	5.00		11.90	Ì	I	1.83	
	Feature (Service) Activation for each Trunk Side Port Terminated	ı			1							1				
	in D4 Bank		1	UEPPX	1PQWU	0.66	110.00	30.00	65.00	20.00		11.90	Ì	I	1.83	
Tele	phone Number/ Group Establishment Charges for DID Service	1	1		1					1			İ	İ		
	DID Trunk Termination (1 per Port)	1	1	UEPPX	NDT	0.00	0.00	0.00	1	İ		11.90	İ	İ	İ	
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)	1	1	UEPPX	NDZ	0.00	0.00	0.00	1	İ		11.90	İ	İ	İ	
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				11.90				
	Non-Consecutive DID Numbers - per number	1	1	UEPPX	ND5	0.00	0.00	0.00	1	İ		11.90	İ	İ	İ	
	Reserve Non-Consecutive DID Numbers		t	UEPPX	ND6	0.00	0.00	0.00				11.90	1	t	t	
	Reserve DID Numbers		1	UEPPX	NDV	0.00	0.00	0.00				11.90	1	1		
Loca	Number Portability		t		İ			2.30					1	t	t	
	Local Number Portability - 1 per port		t	UEPPX	LNPCP	3.15	0.00	0.00					1	t	t	
FFA	TURES - Vertical and Optional		t		1 3.	50	3.50	0.00					1	t	t	
	Switching Features Offered with Line Side Ports Only	1	1		1	 					<u> </u>	 	 	—		
	All Features Available	1	1	UEPPX	UEPVF	2.26	0.00	0.00			-	11.90	 	—	1.83	
	/ III / Gataroo / Wallabio	1		02. I A	OL: V:	2.20	0.00	0.00	l	l	1	11.30	1	1	1.00	L

UNBUN	IDLEI	NETWORK ELEMENTS - Florida												Attachi	ment: 2	Exhil	bit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGO	RY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									P	F	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec		curring	Nonrecurring					Rates(\$)		
LINIBLINIB		ENTREX PORT/LOOP COMBINATIONS - COST BASED RATES						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Based Rates are applied where BellSouth is required by FCC		State ('ammissian rula ta t	nrovido Unb	undlad Lasal C	witching or Cu	vitab Barta								1
		ures shall apply to the Unbundled Port/Loop Combination - C								dlad Bart sacti	on of this Date	Evhibit					1
		Office and Tandem Switching Usage and Common Transport											oin Port/Lo	on Combinat	ions		-
		first and additional Port nonrecurring charges apply to Not Cu														Additional NR	Ce may
		Iso and are categorized accordingly.	arrentiy	COILID	ned combos. Tor	ouriently co	mbmed comb	os, the nomec	unning charges	Silali be tilose	identified in t	ne nomecu	ining - Guire	antiy Combine	eu sections.	Additional Ni	.os may
		ket Rates for Unbundled Centrex Port/Loop Combination will	he nead	otiated	on an Individual Ca	se Basis, un	til further notic	Α.	1	1		1	I		1	l	
		CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)		liaica	on an marviadar oa	Dusis, un		-									
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo	<u> </u>														
		ort/Loop Combination Rates (Non-Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
		Non-Design		1	UEP91		14.11										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -]		
		Non-Design		2	UEP91		18.23										<u> </u>
	_	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -]				1]		
		Non-Design		3	UEP91		33.04		ļ	ļ					ļ		<u> </u>
U	JNE Po	ort/Loop Combination Rates (Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -					40.50										
		Design		1	UEP91		16.53										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP91		21.60										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEF91		21.00										-
		Design		3	UEP91		37.85										
10	INF L	pop Rate		3	OLI 31		37.03										
		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	12.94										
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	17.06										
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1	31.87										
		2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	15.36										
		2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	20.43										
		2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	36.68										
	INE Po																
А	III Stat	es (Except North Carolina and Sout Carolina)			LIEDOA	LIEDVA	4.47	50.04	00.40	07.50	0.07		44.00				
		2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	1.17	53.31	26.46	27.50	8.37		11.90				
		2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP91	UEPYB	1 17	53.31	26.46	27.50	8.37		11.00				
		2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			OLF31	UEPID	1.17	55.51	26.46	21.50	0.37		11.90			-	
		Area			UEP91	UEPYH	1.17	53.31	26.46	27.50	8.37		11.90		1		
-+		2-Wire Voice Grade Port (Centrex from diff Serving Wire			0.	J=:	1.17	55.51	20.40	27.50	0.07	<u> </u>	11.50		 		
		Center)2 Basic Local Area			UEP91	UEPYM	1.17	139.49	86.10	65.41	13.81		11.90		1		
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service							220	22711			50				
		Term - Basic Local Area			UEP91	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90		1		
		2-Wire Voice Grade Port terminated in on Megalink or equivalent															
		- Basic Local Area			UEP91	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				
		2-Wire Voice Grade Port Terminated on 800 Service Term -															
		Basic Local Area			UEP91	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90		ļ		1
G	eorgi	a and Florida Only			LIEDO.								44.5-		ļ		<u> </u>
		2-Wire Voice Grade Port (Centrex)			UEP91	UEPHA	1.17	53.31	26.46	27.50	8.37		11.90				_
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPHB UEPHH	1.17	53.31	26.46	27.50	8.37	1	11.90 11.90		 		1
		2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	1.17	53.31	26.46	27.50	8.37	-	11.90		-		
		2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90		1		
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OL1 31	JEI I IIVI	1.17	135.49	00.10	00.41	10.01		11.50		1	1	1
		Term			UEP91	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90		1		
-+		· - · · · ·				JE	,	100.40	55.70	55.71	10.01	1	11.50		1	1	
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	1.17	53.31	26.46	27.50	8.37		11.90		1		
		2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	1.17	53.31	26.46	27.50	8.37		11.90			İ	
L		witching									·						
		Centrex Intercom Funtionality, per port			UEP91	URECS	0.7384										
	ocal N	lumber Portability															

Version 3Q02: 09/06/02

<u> </u>													Attach	ment: 2	Fyhil	oit: C
	ED NETWORK ELEMENTS - Florida	1									Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
											Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						- (1)			per LSK	per LOK	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						_	Nonrec	urrina	Nonrecurring	Disconnect			oss	Rates(\$)		
-						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-	Local Number Portability (1 per port)			UEP91	LNPCC	0.35			1							
Featur																
	All Standard Features Offered, per port			UEP91	UEPVF	2.26						11.90				
-	All Select Features Offered, per port			UEP91	UEPVS	0.00	370.70					11.90				
-	All Centrex Control Features Offered, per port			UEP91	UEPVC	2.26						11.90				
NARS																
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00				11.90				
-	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00				11.90				
-	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				11.90				
Misce	Ilaneous Terminations															
	e Trunk Side	1			1				† †				İ	İ		İ
	Trunk Side Terminations, each			UEP91	CENA6	8.81										
Intero	office Channel Mileage - 2-Wire	1		-	1				† †				İ	İ		
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	25.32			†				1	1		l
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0091			† †			İ	İ	İ		
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Service	ce			1				† †			İ	İ	İ		
	nannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.66										
-	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP91	1PQW7	0.66										
-+	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			02. 0.		0.00										
	Different Wire Center			UEP91	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			02. 0.		0.00										
	Slot			UEP91	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.66										
Non-F	Recurring Charges (NRC) Associated with UNE-P Centrex															
	Conversion - Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP91	USAC2		21.50	8.42				11.90				
-+	Conversion of Existing Centrex Common Block			UEP91	USACN		5.17	8.32				11.90				
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	618.82					11.90				
-+	New Centrex Customized Common Block			UEP91	M1ACC	0.00	618.82					11.90				
	Secondary Block, per Block	1	1	UEP91	M2CC1	0.00	71.31					11.90	1	1		1
$\overline{}$	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	66.48		1			11.90	1	1		
UNF-F	P CENTREX - 5ESS (Valid in All States)	1	1	- ** - *	1	0.00	550						1	1		1
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo	1	1		+ +								1	1		1
	Port/Loop Combination Rates (Non-Design)				1								1	1		
- 1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1			1								1	1		1
	Non-Design		1	UEP95	1 1	14.11						1				
-+	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		<u> </u>	02. 00	+											
	Non-Design		2	UEP95		18.23										
-+	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	-		+ +	.0.20						 	 	 		1
	Non-Design		3	UEP95		33.04										
UNE F	Port/Loop Combination Rates (Design)			OLI 00	+	00.04										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1			+ +				 		l					
	Design		1	UEP95		16.53										
-+	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	†		+ +	. 5.56						 	 	 		1
	Design		2	UEP95	1	21.60]		1		Ì	Ì		
-+	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1		02. 00	+ +	21.50						 	 	 		1
	Design		3	UEP95	1	37.85]		1		Ì	Ì		
LIME	Loop Rate	1	-	OLI 33	+ +	37.03			+ +				 	 		-
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP95	UECS1	12.94			 			1	1	1		
ONEL		+			UECS1	17.06			+ +		-	 	 	 		1
ONE	2-Wire Voice Grade Loop (SL 1) Zono 2															
ONE I	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP95												
ONE I	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		3	UEP95 UEP95 UEP95	UECS1 UECS2	31.87 15.36										

UNBUNDLE	D NETWORK ELEMENTS - Florida	,		,										nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	36.68										
	Port Rate															
All Sta				LIEBAE	LUEDVA.		== = = = = = = = = = = = = = = = = = = =									
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95	UEPYH	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area		<u> </u>	UEP95	UEPYM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP95	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			LIEDOE	LIEDVO	4.47	50.04	00.40	07.50	0.07		44.00				
	- Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term -	1	-	UEP95	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				
	Basic Local Area			UEP95	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90				
	Y, LA, MS, SC, & TN Only															
FL & (GA Only															
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPHA	1.17	53.31	26.46		8.37		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)		<u> </u>	UEP95	UEPHB	1.17	53.31	26.46		8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPHH	1.17	53.31	26.46	27.50	8.37		11.90				<u> </u>
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90				
	O.Wire Veice Conde Boot terraineted in an Manalista or annivelent						53.31									
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term		1	UEP95 UEP95	UEPH9 UEPH2	1.17 1.17	53.31	26.46 26.46	27.50 27.50	8.37 8.37		11.90 11.90				-
Local	Switching			OLF 95	OLFTIZ	1.17	33.31	20.40	21.30	0.57		11.90				1
Looui	Centrex Intercom Funtionality, per port			UEP95	URECS	0.7384										
Local	Number Portability															
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Featu	res															
	All Standard Features Offered, per port			UEP95	UEPVF	2.26										
	All Select Features Offered, per port			UEP95	UEPVS	0.00	370.70					11.90				
	All Centrex Control Features Offered, per port			UEP95	UEPVC	2.26										
NARS			<u> </u>	LIEDOE	LIADOY	0.00	0.00	0.00				44.00				
	Unbundled Network Access Register - Combination	1		UEP95 UEP95	UARCX UAR1X	0.00	0.00	0.00				11.90 11.90				ļ
	Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial	1	1	UEP95 UEP95	UAROX	0.00	0.00	0.00				11.90				
Misce	Illaneous Terminations			021 00	J, 11.07.	0.00	0.00	0.00	<u> </u>			11.00				
	e Trunk Side															
-	Trunk Side Terminations, each			UEP95	CEND6	8.81										†
4-Wire	e Digital (1.544 Megabits)								1						İ	
	DS1 Circuit Terminations, each			UEP95	M1HD1	54.95										
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	15.69	•		•		11.90				
Intero	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination	ļ		UEP95	MIGBC	25.32			ļ							<u> </u>
Faster	Interoffice Channel mileage, per mile or fraction of mile	<u> </u>	<u> </u>	UEP95	MIGBM	0.0091			1						-	
	re Activations (DS0) Centrex Loops on Channelized DS1 Service nannel Bank Feature Activations	e	-		+ -				-						-	
D4 Ch	Feature Activation on D-4 Channel Bank Centrex Loop Slot		 	UEP95	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop	!	-	UEP95	1PQW6	0.66			 							
	Slot			UEP95	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP95	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.66										

	ED NETWORK ELEMENTS - Florida	1	1		1						Core Constru	Corn Condi		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			LIEDOE	400000	0.00										
	Slot			UEP95 UEP95	1PQWQ 1PQWA	0.66										
Non-E	Feature Activation on D-4 Channel Bank WATS Loop Slot Recurring Charges (NRC) Associated with UNE-P Centrex		-	UEF95	IPQVA	0.66			1							+
NOII-N	NRC Conversion Currently Combined Switch-As-Is with allowed															+
	changes, per port			UEP95	USAC2	0.00	21.50	8.42				11.90				
	Conversion of Existing Centrex Common Block, each			UEP95	USACN	0.00	5.17	8.32				11.90			1	1
-	New Centrex Standard Common Block			UEP95	M1ACS	0.00	618.82					11.90				
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	618.82					11.90				1
	NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	66.48					11.90				
	P CENTREX - DMS100 (Valid in All States)															
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo									·						
UNE P	Port/Loop Combination Rates (Non-Design)														1	
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1		LIEDOD	1 1											
	Non-Design	 	1	UEP9D	1 1	14.11			 						!	+
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	l	2	UEP9D	1	18.23									I	1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEP9D		18.23										
	Non-Design		3	UEP9D		33.04										
LINE E	Port/Loop Combination Rates (Design)		- 3	OLI 3D	+	33.04										+
OIVE !	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+											+
	Design		1	UEP9D		16.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		<u> </u>	02.05		10.00										†
	Design		2	UEP9D		21.60										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															1
	Design		3	UEP9D		37.85										<u> </u>
UNE L	oop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	12.94										
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP9D	UECS1	17.06										<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	31.87										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D UEP9D	UECS2	15.36 20.43									-	+
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2 UECS2	36.68										+
LINE	Port Rate		3	OLF3D	ULCGZ	30.00					1					\vdash
	TATES															+
ALL	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.17						11.90				†
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
1	Area	l		UEP9D	UEPYB	1.17	53.31	26.46	27.50	8.37		11.90			I	1
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
	Area			UEP9D	UEPYC	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local															
	Area			UEP9D	UEPYD	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local	l		LIEDOD	LIEDVE		50.01	00.10	07.50	0.00		44.60			1	1
+-	Area		1	UEP9D	UEPYE	1.17	53.31	26.46	27.50	8.37		11.90			1	₩
1	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local			UEP9D	UEPYF	1.17	53.31	26.46	27.50	8.37		11.90				
-+-	Area 2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local		1	OEFSD	UEFTF	1.17	53.31	∠0.46	21.50	8.37	1	11.90			 	+
1	Area			UEP9D	UEPYG	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local			02	020	1.17	55.51	20.40	27.00	0.07		71.00			†	†
1	Area			UEP9D	UEPYT	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local				1											1
[Area	L		UEP9D	UEPYU	1.17	53.31	26.46	27.50	8.37	<u></u>	11.90			<u> </u>	
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local				ĺ											
	Area			UEP9D	UEPYV	1.17	53.31	26.46	27.50	8.37		11.90			1	
1	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local				1							,				
	Area 2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local	ļ		UEP9D	UEPY3	1.17	53.31	26.46	27.50	8.37		11.90			-	
			1	1	ſ				1		1	i l			l	1

CHECINDLE	D NETWORK ELEMENTS - Florida			I	1						Cup Onder	Cup Cude		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonred		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYM	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 Basic Local Area			UEP9D	UEPYO	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.17	139.49	86.10	65.41	13.81		11.90				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	1.17	139.49	86.10	65.41	13.81		11.90				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY5	1.17	139.49	86.10	65.41	13.81		11.90				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	1.17				13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service						139.49	86.10	65.41							
	Term 2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90				
	Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term Basic			UEP9D	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				
El 9 C	Local Area GA Only			UEP9D	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90			-	-
r L & C	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPHA	1.17	53.31	26.46	27.50	8.37		11.90				+
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPHB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPHC	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHD	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPHE	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPHF	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHT	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPHU	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPHV	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPH3	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)		1	UEP9D	UEPHH	1.17	53.31	26.46	27.50	8.37		11.90				1
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication)3			UEP9D	UEPHW	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPHJ	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2. 3	-	 	UEP9D	UEPHO	1.17	139.49	86.10	65.41	13.81	}	11.90			-	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D UEP9D	UEPHP UEPHQ	1.17 1.17	139.49 139.49	86.10 86.10	65.41 65.41	13.81 13.81		11.90 11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	1.17	139.49	86.10	65.41	13.81		11.90				

UNBUNDLI	ED NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9D	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPH9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPH2	1.17	53.31	26.46	27.50	8.37		11.90		ļ	ļ	
Local	Switching								1					1	1	
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.7384			ļ					ļ	ļ	
Local	Number Portability															
_	Local Number Portability (1 per port)		<u> </u>	UEP9D	LNPCC	0.35					1					
Featu			 	LIEDOD	LIED) "E	0.00			!	-	}			!	!	
-	All Standard Features Offered, per port		 	UEP9D UEP9D	UEPVF UEPVS	2.26	070.70		!	-	}	44.00		!	!	
_	All Select Features Offered, per port					0.00	370.70					11.90				
NADO	All Centrex Control Features Offered, per port			UEP9D	UEPVC	2.26										
NARS				LIEDOD	LIADOV	0.00	0.00	0.00				44.00				
-	Unbundled Network Access Register - Combination		-	UEP9D UEP9D	UARCX UAR1X	0.00	0.00	0.00	 		1	11.90 11.90		 	 	
	Unbundled Network Access Register - Inward															
Minor	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00				11.90				
									-							
2-Wir	e Trunk Side Trunk Side Terminations, each			UEP9D	CEND6	8.81			-							
4-\Mir	e Digital (1.544 Megabits)			UEP9D	CENDO	0.01					1					
4-4411	DS1 Circuit Terminations, each			UEP9D	M1HD1	54.95										
	DS0 Channels Activiated per Channel			UEP9D	M1HD0	0.00	15.69					11.90				
Intoro	office Channel Mileage - 2-Wire			OLF 9D	WITTE	0.00	13.09				1	11.90				
interd	Interoffice Channel Facilities Termination			UEP9D	MIGBC	25.32					1					
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0091										
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Service	Δ	1	OLI 3D	IVIIODIVI	0.0031			1		1					
	nannel Bank Feature Activations		1		+				1		1					
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.66			<u> </u>					<u> </u>	<u> </u>	
-	. satars / totration on b + chariter bank centrex Loop Glot		<u> </u>	021 00	11 00 110	0.00			I		1			I	I	1
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.66			1							
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center		1	UEP9D	1PQWP	0.66								1	1	
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			UEP9D	1PQWV	0.66								ļ		
	Slot			UEP9D	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.66										
Non-F	Recurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		21.50	8.42				11.90				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		5.17	8.32	1			11.90		1	1	
_	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	618.82		ļ			11.90		ļ	ļ	
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	618.82		1			11.90		1	1	
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	66.48		.			11.90		1	.	
	P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)								.					1	.	
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo								.					1	.	
UNE	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	1	1						I					I	I	
1	Non-Design		1	UEP9E		14.11										

UNBUNDL	ED NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonred	urring	Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP9E		18.23										ļ
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_	LIEDOE		00.04										
LINE	Non-Design		3	UEP9E	+	33.04										<u> </u>
UNE	Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+											
	Design		1	UEP9E		16.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLI SL	+	10.55										
	Design		2	UEP9E		21.60										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -				1											
	Design		3	UEP9E		37.85										
UNE	Loop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	12.94										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	17.06										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	31.87										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	15.36										↓
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	20.43										ļ
LINE	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	36.68										
	Port Rate FL, KY, LA, MS, & TN only				-											
AL, F	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Fort (Centrex 800 termination)Basic Local			OLI SL	OLITA	1.17	33.31	20.40	27.50	0.57		11.50				+
	Area			UEP9E	UEPYB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			02. 02	02		00.01	20.10	21.00	0.01		11100				
	Area			UEP9E	UEPYH	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP9E	UEPYM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP9E	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP9E	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				_
	2-Wire Voice Grade Port Terminated on 800 Service Term -			LIEDOE	LIEDVO	4 47	50.04	00.40	27.50	0.07		44.00				
Flori	Basic Local Area da Only			UEP9E	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90				
FIOTI	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPHA	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Fort (Centrex 800 termination)			UEP9E	UEPHB	1.17	53.31	26.46		8.37		11.90				1
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPHH	1.17	53.31	26.46		8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2		1	UEP9E	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90		1		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service							-								
	Term			UEP9E	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90				
				l												
	2-Wire Voice Grade Port terminated in on Megalink or equivalent		<u> </u>	UEP9E	UEPH9	1.17	53.31	26.46	27.50	8.37		11.90				
<u> </u>	2-Wire Voice Grade Port Terminated on 800 Service Term		!	UEP9E	UEPH2	1.17	53.31	26.46	27.50	8.37		11.90	ļ	ļ		
Loca	I Switching		<u> </u>	LIEDOE	LIDECC	0.7004			1				1	 	1	
1 000	Centrex Intercom Funtionality, per port I Number Portability		 	UEP9E	URECS	0.7384			 		-		1	-	-	+
Loca	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35			 				-		-	
Featu			!	02.1 02	2111 00	0.55										†
	All Standard Features Offered, per port		†	UEP9E	UEPVF	2.26								1		1
	All Select Features Offered, per port		<u> </u>	UEP9E	UEPVS	0.00	370.70					11.90				1
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	2.26						-				
NAR																
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00		•		11.90				
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00			<u> </u>	11.90				
	Unbundled Network Access Register - Outdial		!	UEP9E	UAROX	0.00	0.00	0.00	ļ			11.90	ļ	ļ		
	ellaneous Terminations re Trunk Side		<u> </u>		1											├

UNBUNDLE	ED NETWORK ELEMENTS - Florida												Attachi	ment: 2	Exhil	bit: C
											Svc Order	Svc Order				Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec	urring	Nonrecurrin	g Disconnect			oss	Rates(\$)	I	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
4-Wire	e Digital (1.544 Megabits)															
-	DS1 Circuit Terminations, each DS0 Channel Activated Per Channel			UEP9E UEP9E	M1HD1 M1HDO	54.95 0.00	15.69					11.90				
Intero	ffice Channel Mileage - 2-Wire			UEP9E	MILLIPO	0.00	15.69					11.90				
Intere	Interoffice Channel Facilities Termination			UEP9E	MIGBC	25.32										
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0091										
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 Ch	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.66										
	Factors Astination on D. A. Channel Deals EV line Cide I and Clat			LIEDOE	1PQW6	0.00										
 	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop	-	-	UEP9E	IPQVVO	0.66			+	+						1
1 1	Slot			UEP9E	1PQW7	0.66			1							
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -	1			1	5.50			1	1			1	1		
	Different Wire Center	<u> </u>		UEP9E	1PQWP	0.66			<u> </u>	1	L	<u> </u>	<u> </u>			
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			LIEDOE	4001410	0.00										
	Slot Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E UEP9E	1PQWQ 1PQWA	0.66 0.66										
Non-R	Recurring Charges (NRC) Associated with UNE-P Centrex			UEP9E	IPQWA	0.00										
NOII-IV	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9E	USAC2		21.50	8.42				11.90				
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		5.17	8.32				11.90				
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	618.82					11.90				
	New Centrex Customized Common Block			UEP9E	M1ACC	0.00	618.82					11.90				
Nerte	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	66.48					11.90				
	I - Required Port for Centrex Control in 1AESS, 5ESS & EWSD 2 - Requires Interoffice Channel Mileage				-	-										
	3 - Requires Specific Customer Premises Equipment				+											
	CENTREX PORT/LOOP COMBINATIONS - MARKET RATES															
	ket Rates are applied where BellSouth is not required by FCC					indled Local Sw	itching or Swi	tch Ports.								
	curring Charges for all Standard Centrex and Centrex Conrol Fe															
	d Office and Tandem Switching Usage and Common Transport															
	first and additional Port nonrecurring charges apply to Not Co	urrently	Comb	ned Combos. For	Currently Co	mbined Combo	s, the nonrecu	rring charges	shall be those	e identified in t	he Nonrecu	rring - Curre	ently Combine	ed sections.	Additional NR	≀Cs may
	also and are categorized accordingly.									1						1
	CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)														ļ
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design)															
ONL	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															1
	Non-Design		1	UEP91		26.94										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -						İ									
	Non-Design		2	UEP91		31.06										<u> </u>
1 1 -	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo		_			I T	\exists		_							
LINE -	Non-Design		3	UEP91		45.87				1	<u> </u>		ļ	ļ	ļ	
UNE	Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1			<u> </u>	 			 	+	 	-		-		
1 1	Design		1	UEP91		29.36	1		1				1	1		
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1				20.00			1	1			1	1		
	Design		2	UEP91		34.43			1							
İ	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -					į i	İ									
\vdash	Design		3	UEP91		50.68			ļ	1	ļ					
UNE L	oop Rate		.	LIEDOA	LIEOC :	10.01				1						<u> </u>
\vdash	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP91 UEP91	UECS1 UECS1	12.94 17.06	-		 	+	<u> </u>	1	 	 		1
\vdash	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91 UEP91	UECS1	17.06 31.87			 	+	 					
 	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	15.36	+		 	+	 					
	2-Wire Voice Grade Loop (SL 2) - Zone 1		2	UEP91	UECS2	20.43			1					1		
	2-Wire Voice Grade Loop (SL 2) - Zone 3			UEP91	UECS2	36.68									1	

UNRII	NDI FI	NETWORK ELEMENTS - Florida												Attach	ment: 2	Exhib	oit: C
5.400	.1066	THE THORN ELLINER TO THORN										Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Intori									Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																Disc 1st	DISC Add I
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	UNE Po																
	All Stat	es (Except North Carolina and Sout Carolina)															
		2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	14.00	70.00	35.00	35.00	10.00		11.90				
		2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
		Area			UEP91	UEPYB	14.00	70.00	35.00	35.00	10.00		11.90				
		2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local					44.00	=									
		Area			UEP91	UEPYH	14.00	70.00	35.00	35.00	10.00		11.90				
		2-Wire Voice Grade Port (Centrex from diff Serving Wire			LIEDO4	LIEDVAA	44.00	400.00	440.00	05.00	20.00		44.00				
		Center)2 Basic Local Area			UEP91	UEPYM	14.00	180.00	110.00	85.00	20.00		11.90				
	1	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1		UEP91	UEPYZ	14.00	180.00	110.00	85.00	20.00		11.90		1		
-	<u> </u>	Term - Basic Local Area 2-Wire Voice Grade Port terminated in on Megalink or equivalent	 		UEF91	UEPTZ	14.00	180.00	110.00	გე.00	∠0.00		11.90	-	-		
	l	Note Stade Port terminated in on Megalink or equivalent Basic Local Area			UEP91	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90				
-	1	2-Wire Voice Grade Port Terminated on 800 Service Term -	1		טבו פו	OLFIS	14.00	70.00	33.00	33.00	10.00	1	11.90				
	l	Basic Local Area			UEP91	UEPY2	14.00	70.00	35.00	35.00	10.00		11.90				
	Georgi	and Florida Only	1		OLI 91	OLI 12	14.00	70.00	35.00	55.00	10.00		11.30				
	Jeorgia	2-Wire Voice Grade Port (Centrex)	†		UEP91	UEPHA	14.00	70.00	35.00	35.00	10.00	<u> </u>	11.90		 		
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPHB	14.00	70.00	35.00	35.00	10.00		11.90				
		2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	14.00	70.00	35.00	35.00	10.00		11.90				
		2-Wire Voice Grade Port (Centrex from diff Serving Wire															
		Center)2			UEP91	UEPHM	14.00	180.00	110.00	85.00	20.00		11.90				
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
		Term			UEP91	UEPHZ	14.00	180.00	110.00	85.00	20.00		11.90				
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	14.00	70.00	35.00	35.00	10.00		11.90				
		2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90				
	Local S	witching															
		Centrex Intercom Funtionality, per port			UEP91	URECS	0.7384										
	Local	umber Portability			LIEDO4	LNPCC	0.25										
	Feature	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
		All Standard Features Offered, per port			UEP91	UEPVF	0.00					1	11.90				
-		All Select Features Offered, per port			UEP91	UEPVS	0.00	370.70					11.90				
		All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00	370.70					11.90				
	NARS	The Control Co			02. 0.	02. 10	0.00						11.00				
		Unbundled Network Access Register - Combination	†		UEP91	UARCX	0.00	0.00	0.00				11.90		1		
		Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00				11.90		İ		
		Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00	İ			11.90	İ			
		aneous Terminations															
		Trunk Side															
		Trunk Side Terminations, each			UEP91	CENA6	8.81										
	Interoff	ice Channel Mileage - 2-Wire						, and the second									
		Interoffice Channel Facilities Termination - Voice Grade	<u> </u>		UEP91	M1GBC	25.32										
	<u> </u>	Interoffice Channel mileage, per mile or fraction of mile	<u> </u>		UEP91	M1GBM	0.0091			ļ					ļ		
		Activations (DS0) Centrex Loops on Channelized DS1 Service	e			+								ļ	 		
<u> </u>	ט4 Cha	nnel Bank Feature Activations	 	ļ	LIEDO4	400000	0.00			1				1	 		
\vdash	 	Feature Activation on D-4 Channel Bank Centrex Loop Slot	-		UEP91	1PQWS	0.66								-		
	1	Feature Activation on D-4 Channel Bank FX line Side Loop Slot	1		UEP91	1PQW6	0.66						1		1		
\vdash	 	Feature Activation on D-4 Channel Bank FX Trunk Side Loop	 		021 31	11 4770	0.00			1				1	1		
	1	Slot	1		UEP91	1PQW7	0.66						1		1		
	l -	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			J. J.	3,117	0.00										
	l	Different Wire Center			UEP91	1PQWP	0.66										
	 				-		2.20			Ì					1		
	1	Feature Activation on D-4 Channel Bank Private Line Loop Slot	1		UEP91	1PQWV	0.66						1		1		
		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
		Slot			UEP91	1PQWQ	0.66							<u> </u>	<u> </u>		
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.66										
	Non-Re	curring Charges (NRC) Associated with UNE-P Centrex							-								-

UNBUNDI	ED NETWORK ELEMENTS - Florida												Attach	ment: 2	Exhil	bit: C
											Svc Order	Svc Order	Incremental		Incremental	
1											Submitted	Submitted		Charge -	Charge -	Charge -
1		Intori									Elec	Manually		Manual Svc		Manual Svo
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
i		m						.,,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
i													1st	Add'l	Disc 1st	Disc Add'l
i													151	Add I	DISC ISL	DISC Add I
						_	Nonrec	urring	Nonrecurring	g Disconnect			oss	Rates(\$)	·	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Conversion - Currently Combined Switch-As-Is with allowed															
1	changes, per port			UEP91	USAC2		21.50	8.42				11.90				
	Conversion of Existing Centrex Common Block			UEP91	USACN		5.17	8.32				11.90				
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	618.82					11.90				
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	618.82					11.90				
	Secondary Block, per Block			UEP91	M2CC1	0.00	71.31					11.90				
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	66.48					11.90				
UNE	P CENTREX - 5ESS (Valid in All States)															
	re VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
i I	Non-Design		1	UEP95		26.94						1		1		1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -											1		1		i
i l	Non-Design		2	UEP95		31.06								I	I	l
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	 		† †	000				1			1	t	†	l
i I	Non-Design		3	UEP95		45.87						1		1		1
UNE	Port/Loop Combination Rates (Design)			0L1 00		40.07					1					
- OILE	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+											
1	Design		1	UEP95		29.36										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		<u> </u>	OLI 33		23.30										
1	Design		2	UEP95		34.43										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEF93	+	34.43					-			-	-	
1	Design		3	UEP95		50.68										
LINE	Loop Rate		3	UEF93	+	30.06										
UNE	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	12.94										
\vdash			2	UEP95 UEP95	UECS1	17.06										
\vdash	2-Wire Voice Grade Loop (SL 1) - Zone 2		3		UECS1											
	2-Wire Voice Grade Loop (SL 1) - Zone 3	ļ		UEP95		31.87										
	2-Wire Voice Grade Loop (SL 2) - Zone 1	ļ	1	UEP95	UECS2	15.36										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	20.43										
 	2-Wire Voice Grade Loop (SL 2) - Zone 3	ļ	3	UEP95	UECS2	36.68										
	Port Rate	ļ	-		_											
All S				LIEDOS	LIEDVA	44.00	70.00	05.00	05.00	40.00		44.00				
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	14.00	70.00	35.00	35.00	10.00		11.90				
i l	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			LIEBOE	LIEDY"		=							I	I	l
	Area	1	1	UEP95	UEPYH	14.00	70.00	35.00	35.00	10.00		11.90				ļ
i l	2-Wire Voice Grade Port (Centrex from diff Serving Wire													I	I	l
\longmapsto	Center)2 Basic Local Area	1	ļ	UEP95	UEPYM	14.00	180.00	110.00	85.00	20.00		11.90		.	.	ļ
i l	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			l	I I									I	I	l
\longmapsto	Term - Basic Local Area	1	ļ	UEP95	UEPYZ	14.00	180.00	110.00	85.00	20.00		11.90		.	.	ļ
i l	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1		l	1									I	I	l
igwdow	- Basic Local Area		<u> </u>	UEP95	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90		ļ	ļ	
1 1	2-Wire Voice Grade Port Terminated on 800 Service Term -		1							Ì	1	İ		I		l
igsquare	Basic Local Area		1	UEP95	UEPY2	14.00	70.00	35.00	35.00	10.00		11.90				
	(Y, LA, MS, SC, & TN Only															
FL 8	GA Only															
\Box	2-Wire Voice Grade Port (Centrex)			UEP95	UEPHA	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPHB	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPHH	14.00	70.00	35.00	35.00	10.00		11.90				
T -	2-Wire Voice Grade Port (Centrex from diff Serving Wire												I			
	Center)2			UEP95	UEPHM	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
i l	Term			UEP95	UEPHZ	14.00	180.00	110.00	85.00	20.00		11.90		I	I	l
1 l	2-Wire Voice Grade Port terminated in on Megalink or equivalent	:		UEP95	UEPH9	14.00	70.00	35.00	35.00	10.00		11.90		I	I	l
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90				
Loca	I Switching											1				İ
LUCA						0 7001										
Loca	Centrex Intercom Funtionality, per port			UEP95	URECS	0.7384										

CATEGORY	D NETWORK ELEMENTS - Florida				, 											bit: C
CATEGORY											Svc Order	Svc Order	Incremental	ment: 2 Incremental		
CATEGORY											Submitted	Submitted		Charge -	Charge -	Charge -
CATEGORY											Elec	Manually		Manual Svc		Manual Svo
	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						.,,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
i													1st	Add'l	Disc 1st	Disc Add'l
						_	Nonrec	urring	Nonrecurring	g Disconnect			oss	Rates(\$)	·	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Feature	es															
	All Standard Features Offered, per port			UEP95	UEPVF	0.00										
	All Select Features Offered, per port			UEP95	UEPVS	0.00	370.70					11.90				
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00										
NARS																
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				11.90				
Miscel	laneous Terminations															
2-Wire	Trunk Side															
	Trunk Side Terminations, each			UEP95	CEND6	8.81										
	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP95	M1HD1	54.95										
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	15.69					11.90				
Interof	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP95	MIGBC	25.32										
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0091										
Featur	e Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Cha	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP95	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP95	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP95	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.66										
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP95	USAC2	0.00	21.50	8.42				11.90				
	Conversion of Existing Centrex Common Block, each	ļ		UEP95	USACN		5.17	8.32	ļ			11.90		.	.	↓
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	618.82		ļ			11.90		.	.	└
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	618.82					11.90				├
	NAR Establishment Charge, Per Occasion	_		UEP95	URECA	0.00	66.48					11.90		-	-	├
	CENTREX - DMS100 (Valid in All States)	<u> </u>			1				ļ				1	-	-	├
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo	<u> </u>			1				ļ				1	-	-	├
UNE P	ort/Loop Combination Rates (Non-Design)	_			1									-	-	├
. 1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	,	LIEDOD		20.01					1			I	I	1
	Non-Design	<u> </u>	1	UEP9D	1	26.94			ļ				1	-	-	├
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_	LIEDOD		04.00								1	1	1
	Non-Design	 	2	UEP9D	1	31.06			ļ	1	ļ		-	 	 	
.	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			LIEDOD		45.07					1			I	I	1
LINES	Non-Design ort/Loop Combination Rates (Design)	 	3	UEP9D	1	45.87			ļ	1	ļ		-	 	 	
UNE PO		<u> </u>	-		+ -				 					 	 	
.	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	1	4	UEP9D		29.36					1			I	I	1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	-	ロトタロ	+	29.30			+			1		1	1	
.	Design		2	UEP9D		34.43					1			I	I	1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	<u> </u>		OFLAD	+	34.43			 					 	 	
.	Design 2-wire voice Grade Port (Centrex)Port Combo -		3	UEP9D		50.68					1			I	I	1
I INIE I	posign	 	3	UEF9D	+	50.08			 	-			-	-	-	
UNE LO	2-Wire Voice Grade Loop (SL 1) - Zone 1	-	1	UEP9D	UECS1	12.94			1	-			-			+
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	 		UEP9D UEP9D	UECS1	17.06			 					-	-	
				UEP9D	UECS1	31.87			1		l	ĺ	i	l .	İ	1

UNBUNDLI	ED NETWORK ELEMENTS - Florida												Attachi	ment: 2	Exhil	oit: C
											Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
CATEGORY	DATE EL EMENTO	Interi	7	DO0	11000			DATEO(A)			Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Dee	Nonrec	curring	Nonrecurring	g Disconnect		1	oss	Rates(\$)	I	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	15.36										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	20.43										
LINE	2-Wire Voice Grade Loop (SL 2) - Zone 3 Port Rate		3	UEP9D	UECS2	36.68										
	TATES															
ALL	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	14.00						11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP9D	UEPYB	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
 	Area			UEP9D	UEPYC	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	14.00	70.00	35.00	35.00	10.00		11.90				
 	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local			OLI 3D	OLI ID	14.00	70.00	33.00	33.00	10.00		11.50				
	Area			UEP9D	UEPYE	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local	Ì														
	Area			UEP9D	UEPYF	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local									40						
	Area			UEP9D	UEPYG	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	14.00	70.00	35.00	35.00	10.00		11.90				
—	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local	1		UEP9D	UEPYI	14.00	70.00	35.00	35.00	10.00		11.90				
	Area			UEP9D	UEPYU	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local															
	Area			UEP9D	UEPYV	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local															
-	Area	ļ		UEP9D	UEPY3	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	14.00	70.00	35.00	35.00	10.00		11.90				
—	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp	1		UEF9D	UEFTH	14.00	70.00	35.00	35.00	10.00		11.90				
	Indication))3 Basic Local Area			UEP9D	UEPYW	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3															
	Basic Local Area			UEP9D	UEPYJ	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2 Basic Local Area			UEP9D	UEPYM	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			LIEDOD	UEPYO	14.00	70.00	35.00	35.00	10.00		11.90				
 	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEFIU	14.00	70.00	35.00	35.00	10.00		11.90				
	Basic Local Area			UEP9D	UEPYP	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			-		50		22.30	22.30	12.30						
	Basic Local Area			UEP9D	UEPYQ	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			l												
	Basic Local Area	ļ		UEP9D	UEPYR	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	14.00	180.00	110.00	85.00	20.00		11.90				
 	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3	1		OLFSD	UEF15	14.00	180.00	110.00	85.00	∠0.00		11.90				
	Basic Local Area			UEP9D	UEPY4	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3				1		.00.00		55.30	20.00						
	Basic Local Area			UEP9D	UEPY5	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3															
	Basic Local Area	<u> </u>		UEP9D	UEPY6	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			LIEDOD	LIEDVZ	44.00	400.00	440.00	05.00	20.00		44.00				
\vdash	Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	-		UEP9D	UEPY7	14.00	180.00	110.00	85.00	20.00		11.90				
	Term			UEP9D	UEPYZ	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1				00	.00.00		55.00	20.00		50				
	Basic Local Area	1		UEP9D	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90	1	1		
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic															
	Local Area	<u> </u>		UEP9D	UEPY2	14.00	70.00	35.00	35.00	10.00		11.90	ļ	ļ		
FL &	GA Only	l														

<u> UNBUNDLE</u>	D NETWORK ELEMENTS - Florida													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			1	Svc Order Submitted Manually per LSR	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Boo	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPHA	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPHB	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPHC	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHD	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPHE	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPHF	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHT	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPHU	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPHV	14.00	70.00	35.00		10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPH3	14.00	70.00	35.00		10.00		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)		1	UEP9D	UEPHH	14.00	70.00	35.00		10.00		11.90	İ	İ	İ	
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp	1	t		1			22.30					t		1	
	Indication)3	l		UEP9D	UEPHW	14.00	70.00	35.00	35.00	10.00		11.90				1
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3	1	1	UEP9D	UEPHJ	14.00	70.00	35.00		10.00	l -	11.90	t	1	1	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)	1	1		520	14.00	70.00	35.50	30.00	10.00	l -	11.50	t	1	1	
	2			UEP9D	UEPHM	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	14.00	180.00	110.00		20.00		11.90				
	2 Wile voice crade For (centres affer 6W6/EB6 F6E1/2, 6			OLI OD	OLITIO	14.00	100.00	110.00	00.00	20.00		11.50				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPHQ	14.00	180.00	110.00		20.00		11.90				
	2-Wile Voice Grade Fort (Certifex differ SWC /LB3-3209)2, 3			OLF 9D	ULFIIQ	14.00	100.00	110.00	65.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wile Voice Grade Fort (Certifex diller SWC /LBS-WST12)2, 3	-		OLFBD	OLFTIK	14.00	100.00	110.00	65.00	20.00		11.90	-			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	14.00	180.00	110.00	85.00	20.00		11.90				
	2-wire voice Grade Port (Centrex/diller SWC /EBS-Nb312)2, 3			UEP9D	UEPH5	14.00	180.00	110.00	85.00	20.00		11.90				
	2 Mins Vains Conda Book (Control/differ CMC /EBC ME200)2 2			LIEDOD	UEPH4	44.00	400.00	440.00	05.00	20.00		44.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	14.00	180.00	110.00	85.00	20.00		11.90				
	O MESSAGE OF THE POST OF THE P			LIEDOD	LIEDLIE	44.00	400.00	440.00	05.00	00.00		44.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	14.00	180.00	110.00	85.00	20.00		11.90				
				l												
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	14.00	180.00	110.00	85.00	20.00		11.90				
				l												
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9D	UEPHZ	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPH9	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90				
Local	Switching															<u> </u>
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.7384										<u> </u>
Local	Number Portability		 		1				<u> </u>				.		ļ	
	Local Number Portability (1 per port)	ļ	<u> </u>	UEP9D	LNPCC	0.35							ļ			
Featur			<u> </u>		1								1			
	All Standard Features Offered, per port			UEP9D	UEPVF	0.00				·						
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	370.70					11.90				
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00					ļ					
NARS			1								ļ					
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00			ļ	11.90				
	Unbundled Network Access Register - Outdial	ļ	<u> </u>	UEP9D	UAROX	0.00	0.00	0.00				11.90	ļ			
	laneous Terminations	ļ	<u> </u>										ļ			
2-Wire	Trunk Side										ļ					
	Trunk Side Terminations, each			UEP9D	CEND6	8.81										
4-Wire	Digital (1.544 Megabits)										ļ					
	DS1 Circuit Terminations, each			UEP9D	M1HD1	54.95										
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	15.69					11.90				
Intero	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	25.32										
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0091										1

IUNBUNDI F	ED NETWORK ELEMENTS - Florida												Attach	ment: 2	Fyhil	oit: C
3.12311222											Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
											Elec		Manual Svc	Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						,			per Loix	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															Disc 1st	DISC Add I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	re Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Ch	nannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.66										
 	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.66			-							
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEF9D	IPQV/	0.00										
	Different Wire Center			UEP9D	1PQWP	0.66										
	Difficient Wife Contor			OLI OD	11 00111	0.00										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.66			I			1		I	I	
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			-					1					1	1	İ
	Slot			UEP9D	1PQWQ	0.66								1	1	
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.66										
Non-F	Recurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		21.50	8.42				11.90				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		5.17	8.32	ļ			11.90				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	618.82		ļ			11.90		ļ	ļ	
\vdash	New Centrex Customized Common Block		<u> </u>	UEP9D	M1ACC	0.00	618.82		-			11.90		-	-	
	NAR Establishment Charge, Per Occasion		<u> </u>	UEP9D	URECA	0.00	66.48		!			11.90		!	!	1
	P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)		<u> </u>		1				!					!	!	1
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design)	-	 		1				 			 				-
ONE P	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				1 -				t					t	t	1
	Non-Design		1	UEP9E		26.94								1	1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	- '-	J_1 J_	1 1	20.04			†					†	†	
	Non-Design		2	UEP9E		31.06			I			1		I	I	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		Ť		1				1					1	1	
	Non-Design		3	UEP9E		45.87			I			1		I	I	
UNE F	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9E	<u> </u>	29.36									<u> </u>	<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9E	<u> </u>	34.43			ļ					1	1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_											1	1	
	Design		3	UEP9E	ļ	50.68										
UNE L	Loop Rate		1	LIEDOE	LIECC4	40.04			.	-				1	1	
\vdash	2-Wire Voice Grade Loop (SL 1) - Zone 1		2	UEP9E UEP9E	UECS1	12.94 17.06			 					 	 	-
\vdash	2-Wire Voice Grade Loop (SL 1) - Zone 2	-	3	UEP9E UEP9E	UECS1 UECS1	17.06 31.87			 			 				-
 	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E UEP9E	UECS1	15.36			 					 	 	-
	2-Wire Voice Grade Loop (SL 2) - Zone 1	1	2	UEP9E	UECS2	20.43			 					t	t	
 	2-Wire Voice Grade Loop (SL 2) - Zone 2		3	UEP9E	UECS2	36.68			+					t	 	
UNF F	Port Rate	1			32332	55.55			†					†	†	
	L, KY, LA, MS, & TN only				1				1					1	1	
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP9E	UEPYB	14.00	70.00	35.00	35.00	10.00		11.90			<u> </u>	<u> </u>
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local						_	-								
\vdash	Area			UEP9E	UEPYH	14.00	70.00	35.00	35.00	10.00		11.90			1	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			l	[]				I			1		I	I	
	Center)2 Basic Local Area			UEP9E	UEPYM	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			LIEDOE	LIED) (Z		,							1	1	
\vdash	Term - Basic Local Area		-	UEP9E	UEPYZ	14.00	180.00	110.00	85.00	20.00		11.90		1	1	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			LIEDOE	LIEDVO	44.00	70.00	25.00	25.00	40.00		44.00		1	1	
	- Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term -		-	UEP9E	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90		 	 	-
					1				1							

INBUNDLED NET	WORK ELEMENTS - Florida													ment: 2	1	bit: C
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremen
				ĺ							Submitted	Submitted	Charge -	Charge -	Charge -	Charge
		Interi		ĺ							Elec	Manually		Manual Svc		Manual S
ATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs
		m									po. 20.1	po. 2011	Electronic-	Electronic-	Electronic-	Electroni
													1st	Add'l	Disc 1st	Disc Add
															Disc 1st	Disc Add
						Rec	Nonred	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Florida Only																
2-Wire	Voice Grade Port (Centrex)			UEP9E	UEPHA	14.00	70.00	35.00	35.00	10.00		11.90				
2-Wire	Voice Grade Port (Centrex 800 termination)			UEP9E	UEPHB	14.00	70.00	35.00	35.00	10.00		11.90				
2-Wire	Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPHH	14.00	70.00	35.00	35.00	10.00		11.90				
2-Wire	Voice Grade Port (Centrex from diff Serving Wire															
Center)	2			UEP9E	UEPHM	14.00	180.00	110.00	85.00	20.00		11.90				
2-Wire	Voice Grade Port, Diff Serving Wire Center - 800 Service															
Term				UEP9E	UEPHZ	14.00	180.00	110.00	85.00	20.00		11.90				
2-Wire	Voice Grade Port terminated in on Megalink or equivalent		1	UEP9E	UEPH9	14.00	70.00	35.00	35.00	10.00		11.90				
	Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90				
Local Switchin																
	Intercom Funtionality, per port	l		UEP9E	URECS	0.7384								İ		
Local Number																
	umber Portability (1 per port)			UEP9E	LNPCC	0.35										
Features																
	dard Features Offered, per port			UEP9E	UEPVF	0.00										
	ct Features Offered, per port		1	UEP9E	UEPVS	0.00	370.70					11.90				
	trex Control Features Offered, per port			UEP9E	UEPVC	0.00	0.00				1	11.00				
NARS	and definition in database different, per perit			02. 02	02. 10	0.00					1					
	lled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00			1	11.90				
	Iled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00				11.90				
	dled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00				11.90				
Miscellaneous				OLI OL	O/II/O//	0.00	0.00	0.00				11.00				
2-Wire Trunk S																
	side Terminations, each			UEP9E	CEND6	8.81										
	1.544 Megabits)			OLI 3L	CLINDO	0.01										
	rcuit Terminations, each			UEP9E	M1HD1	54.95										
	annel Activated Per Channel			UEP9E	M1HDO	0.00	15.69					11.90				
	nnel Mileage - 2-Wire	<u> </u>	 	OLI OL	WITTE	0.00	10.00					11.50				
	ce Channel Facilities Termination			UEP9E	MIGBC	25.32										
	ce Channel mileage, per mile or fraction of mile	<u> </u>	 	UEP9E	MIGBM	0.0091										
	tions (DS0) Centrex Loops on Channelized DS1 Service			OLI 3L	IVIIODIVI	0.0031					1					
	nk Feature Activations	,e			+						-					-
	Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.66					-					-
reature	Activation on D-4 Channel Bank Centrex Loop Stot			UEF9E	IFQWS	0.00					-					-
Footure	Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.66										
	Activation on D-4 Channel Bank FX Trunk Side Loop	<u> </u>		UEF9E	IFQW6	0.00										
Slot	Activation on D-4 Channel Bank FX Trunk Side Loop			UEP9E	1PQW7	0.66										
	Anti-estina an D. 4 Channal Bank Contract Land Clat	<u> </u>		UEF9E	IFQW/	0.00										
	Activation on D-4 Channel Bank Centrex Loop Slot - t Wire Center	1	1	UEP9E	1PQWP	0.66								l		
Dillerer	IL VVIII CETILET	-	 	OLFSE	IFQWF	0.06								-	1	
Foot	Activation on D.4 Channel Bank Brigate Line Land Clat	1	1	LIEDOE	1PQWV	0.60								l		
	Activation on D-4 Channel Bank Private Line Loop Slot	 	<u> </u>	UEP9E	IPQWV	0.66					-			1	1	
	Activation on D-4 Channel Bank Tjie Line/Trunk Loop	1	1	LIEDOE	4001410	0.00								l		
Slot	1			UEP9E	1PQWQ	0.66					ļ					
	Activation on D-4 Channel Bank WATS Loop Slot	 	<u> </u>	UEP9E	1PQWA	0.66					-			1	1	!
	Charges (NRC) Associated with UNE-P Centrex	<u> </u>	ļ	-	+											<u> </u>
	onversion Currently Combined Switch-As-Is with allowed		1	LIEDOE	LICACO		04.50	0.40				44.00		l		
	s, per port	<u> </u>	ļ	UEP9E	USAC2		21.50	8.42				11.90				<u> </u>
	sion of Existing Centrex Common Block, each	<u> </u>	ļ	UEP9E	USACN		5.17	8.32				11.90				<u> </u>
	entrex Standard Common Block	<u> </u>	ļ	UEP9E	M1ACS	0.00	618.82					11.90				<u> </u>
	entrex Customized Common Block	ļ	<u> </u>	UEP9E	M1ACC	0.00	618.82					11.90		ļ		
	stablishment Charge, Per Occasion	ļ	<u> </u>	UEP9E	URECA	0.00	66.48					11.90		ļ		
	red Port for Centrex Control in 1AESS, 5ESS & EWSD	ļ	<u> </u>											ļ		<u> </u>
	es Interoffice Channel Mileage		<u> </u>		_											<u> </u>
	es Specific Customer Premises Equipment	<u> </u>	<u> </u>													
Note: Rates di	splaying an "R" in Interim column are interim and sub	ject to	rate tru	ie-up as set forth ir	n General Tern	ns and Condition	ons.					l		ĺ	I	I

UNBL	JNDLEI	D NETWORK ELEMENTS - Georgia												Attach	ment: 2	Exhil	bit: C
												Svc Order Submitted	Svc Order Submitted	Incremental Charge -	Incremental Charge -	Incremental Charge -	Incrementa Charge -
			Interi									Elec		Manual Svc	Manual Svc	Manual Svc	Manual Sv
CATE	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			'''									-		Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
	1					-		Nonrec	vein a	Nonrecurring	Disconnect			000	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN		SOMAN	SOMAN
	The "70	Lone" shown in the sections for stand-alone loops or loops as part o	of a com	hinatio	n refers to Geographi	ically Deavers	aged LINE Zones									SUMAN	SOWAN
		ww.interconnection.bellsouth.com/become_a_clec/html/interconne			Trefers to Geographi	ically Deavers	iged ONL Zones	. To view Geog	napriicany Dea	veraged ONL 20	one Designation	is by Cerilia	i Office, refe	i to internet vi	repsite.		
OBED		SUPPORT SYSTEMS	I	1		1				1		1	ı			1	1
OFER		(1) Electronic Service Order: CLEC should contact its contract	t negot	iator if	it prefers the state	specific elec	tronic service o	rdering charge	s as ordered b	v the State Co	mmissions T	he electron	ic service o	dering charg	e currently co	ntained in th	is rate
		is the BellSouth regional electronic service ordering charge.															io rate
		(2) Any element that can be ordered electronically will be bill															lly. For
		elements that cannot be ordered electronically at present per t															
		g charge. SOMAN, will be applied to a CLECs bill when it sub				c iii tiiio oato	gory remedia in	c onarge mar i	rould be billed	. 10 4 0220 011	oc cicon onio c	rucing out	abilities ee	inc on mic io	i tilat cicilicii	Other wise,	tile manaai
	O GC III	Electronic OSS Charge, per LSR, submitted via BST's OSS		LOIL	o Benooutin.												
		interactive interfaces (Regional)				SOMEC		3.50									
UNE S	ERVICE	DATE ADVANCEMENT CHARGE	1					5.50							1	1	
		The Expedite charge will be maintained commensurate with	BellSou	th's FC	C No.1 Tariff, Section	on 5 as appli	cable.								1		
		UNE Expedite Charge per Circuit or Line Assignable USOC, per	1		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,												
l	1	Day	1		ALL UNE	SDASP		200.00					1		I	1	
UNBU	NDLED E	XCHANGE ACCESS LOOP															
		ANALOG VOICE GRADE LOOP				1											
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	14.21	42.54	31.33					18.94	8.42		
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	16.41	42.54	31.33					18.94	8.42		
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	26.08	42.54	31.33					18.94	8.42		
		Loop Testing - Basic 1st Half Hour			UEANL	URET1		78.92	78.92					18.94	8.42		
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		23.33	23.33					18.94	8.42		
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
		(UVL-SL1)			UEANL	UREWO		15.75	8.92								
		Engineering Information Document (EI)			UEANL	UEANM		28.72	28.72								
		Manual Order Coordiantion for UVL-SL1s (per loop)			UEANL	UEAMC		16.11	16.11								
		Order Coordination for Specified Conversion Time for UVL-SL1															
		(per LSR)		<u> </u>	UEANL	OCOSL		35.74	35.74					10.01	0.40		
		2 Wire Unbundled Copper Loop Non-Designed- Zone 1		1	UEQ	UEQ2X		11.02	44.69	25.65	7.06			18.94	8.42		
		2 Wire Unbundled Copper Loop Non-Designed- Zone 2 2 Wire Unbundled Copper Loop Non-Designed-Zone 3		3	UEQ UEQ	UEQ2X UEQ2X		12.72 20.22	44.69 44.69	25.65 25.65	7.06 7.06			18.94 18.94	8.42 8.42		
		Order Coordination 2 Wire Unbundled Copper Loop - Non-		3	UEQ	UEQZX		20.22	44.69	25.05	7.06			18.94	8.42		
		Designed (per loop)			UEQ	USBMC		16.11	16.11					18.94	8.42		
	1	Engineering Information Document (EI)			UEQ	USDIVIC		28.72	28.72					18.94	8.42		
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		78.92	78.92					18.94	8.42		
	1	Loop Testing - Basic Additional Half Hour			UEQ	URETA		23.33	23.33					18.94	8.42		
		CLEC to CLEC Conversion Charge Without Outside Dispatch			024	O.K.E.I.X		20.00	20.00					10.01	02		
		(UCL-ND)			UEQ	UREWO		14.25	7.42					18.94	8.42		
UNBU	NDLED E	EXCHANGE ACCESS LOOP				1											
		ANALOG VOICE GRADE LOOP				1									1		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or															
<u>L</u>	<u> </u>	Ground Start Signaling - Zone 1	<u> </u>	_1	UEA	UEAL2	16.84	104.17	78.10	<u> </u>		<u> </u>	<u></u>	18.94	8.42	<u> </u>	
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or															
		Ground Start Signaling - Zone 2		2	UEA	UEAL2	19.45	104.17	78.10					18.94	8.42		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or]	
		Ground Start Signaling - Zone 3		3	UEA	UEAL2	30.92	104.17	78.10					18.94	8.42		
		Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		35.74									
	1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1										1		_]	
	ļ	Battery Signaling - Zone 1	ļ	1	UEA	UEAR2	16.84	104.17	78.10					18.94	8.42	ļ	
l	1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1						=0 :-				1			Ì	
l	<u> </u>	Battery Signaling - Zone 2	ļ	2	UEA	UEAR2	19.45	104.17	78.10					18.94	8.42		
l	1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1	_	1154	LIEADO	00.00	404 :-	70.10				1	40.01		Ì	
	 	Battery Signaling - Zone 3	 	3	UEA	UEAR2	30.92	104.17	78.10					18.94	8.42	 	
 	 	Order Coordination for Specified Conversion Time (per LSR)	 		UEA UEA	OCOSL UREWO		35.74 87.72	36.36					18.94	8.42	-	1
	4-M1D	CLEC to CLEC Conversion Charge without outside dispatch ANALOG VOICE GRADE LOOP	<u> </u>		UEA	UKEWU	-	87.72	36.36					18.94	8.42		
	4-WIKE	4-Wire Analog Voice Grade Loop - Zone 1	 	4	UEA	UEAL4	22.26	206.95	170.57					18.94	8.42	 	<u> </u>
 		14-vviie Arialog voice Glade Loop - Zolle I	ı	1 1			22.20										<u> </u>
		A-Wire Analog Voice Grade Loop Zono 2		2	IEΔ	I I E A I 4	25.70	206.05	170 F7					10 04	0 // 0		
		4-Wire Analog Voice Grade Loop - Zone 2 4-Wire Analog Voice Grade Loop - Zone 3		2	UEA UEA	UEAL4 UEAL4	25.70 40.86	206.95 206.95	170.57 170.57					18.94 18.94	8.42 8.42		

Version 3Q02: 09/06/02 Page 88 of 362

CATEGORY RATE ELIMENTS Maria Deep BCS USOC RATE Deep D	UNBUNDI F	D NETWORK ELEMENTS - Georgia												Attachi	ment: 2	Fyhil	bit: C
ATE FLEMENTS Married M	ONDONDEL											Svc Order	Svc Order				
RATE REMEMTS Horse Long																	
## ATERIAL PRINTS Mark Dotter 120 Dott																	Manual Svc
Biother Biot	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)								
			m						- (1)			per LSK	per LOK				
Processing Pro																	
STATE CONTROLLED CON														151	Add I	DISC ISL	DISC Add I
CRE'D CLAST COMPANIES CO							Dee	Nonrec	urring	Nonrecurring	g Disconnect			oss	Rates(\$)		
2000 100							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
SWING BOTH Digital Grade Logs - Zeros 1		CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36					18.94	8.42		
Syme BKN Digit Close Large Zone 2	2-WIRI																ĺ
2-years (SEX) Option Contention from tools (200) 3 DN 1911 2 DN 1912 44.07 23.30 180.35 180.3				1													1
Order Constitution for Specialist Convenient First peril (189) USE																	<u> </u>
CLEC to CLEC Conversion Charge eliminal cutation department UNN UNE NOT 120.08 33.04 18.94 8.42				3			40.17		180.35					18.94	8.42		1
SWIRE ASYMETRICAL DIGITAL SUBSCRIBER LINE (ADS.) COMPATIBLE LOOP 1 MAI																	
2 Niver Uncommoded ADSL Loop including manual service requiry 1 1 UAL			<u> </u>			UREWO		120.98	33.04					18.94	8.42		
Statilly reservation. 7 June 1 1 1 1 1 1 1 1 1 1	2-WIRI		ATIBLE	LOOI	•												
2 Wile Inhumbed AVSL Loop Including manual service requiry 1 2 UAL UNLZX 12.97 44.66 31.55 25.65 7.06 18.04 8.42			١.		l												i
A facility reservation - Zone 2 1 2 UAL UALZX 1257 44.69 31.55 25.65 7.06 18.94 8.42	 			1	UAL	UAL2X	11.23	44.69	31.55	25.65	7.06			18.94	8.42	 	
2. Were Unbounded APSEL to SQL productions around service requiry 6. 1 3 UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 CODSS. 1. 1. 1. UAL UNIXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62 44.69 31.55 25.65 7.06 19.94 8.42 10.44 UNIXXX 20.62			l .	_	LIAI	LIALOV	40.07	44.00	24.55	05.05	7.00	1		40.04	0.40	1	1
A facility reservation - Zone 3 1 3 UAL UAL UAL 2002 34-60 31-55 26-65 7-06 18-94 8-42 UAL				- 2	UAL	UAL2X	12.97	44.69	31.55	25.65	7.06			18.94	8.42	-	
Onter Countminister for Specified Commission from (per LSR) UAL OODSL 35.74]			2	LIAL	LIALOV	20.60	44.60	24 55	25.65	7.00	1		10.04	0.40	1	1
2 Wire Unbundled ADSL Loop without manual service inquiry & 1	\vdash		 	3			20.62		31.55	∠5.05	7.06	-		18.94	8.42	-	
Solity reservation - Zone 1 1 1 UAL UALZW 11.23 44.69 31.55 25.66 7.06 18.94 8.42			 	<u> </u>	UAL	UCUSL		35.74						-	-	-	
2 2 2 2 2 2 2 2 2 2				1	LIAI	1101 200	11 22	44.60	21.55	25.65	7.06			19.04	0 12		ı
Social tynesenation - Zone 2				-	UAL	UALZVV	11.23	44.03	31.33	25.05	7.00			10.54	0.42		
2 Wife Informed ADSL Loop without manual service inquiry & 1 3 UAL UAL2W 20.62 44.69 31.55 25.65 7.06 18.94 8.42			1	2	ΠΔΙ	HAL 2W	12 97	44 69	31.55	25.65	7.06			18 94	8 42		1
Procing reservation - Zone 3			-	-	O/ IL	OTILLEVY	12.07	44.00	01.00	20.00	7.00			10.04	0.42		—
Order Coordination for Specified Conversion Time (per LSR)			1	3	ΠΔΙ	11AL 2W/	20.62	44 69	31 55	25.65	7.06			18 94	8 42		i
CLEC to CLEC Convention Charge without outside disparch UAL UREWO 44.69 29.29 18.94 8.42			<u> </u>	Ŭ			20.02		01.00	20.00	7.00			10.04	0.42		
A WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HOSL) COMPATIBLE LOOP			1						29.29					18.94	8.42		
2 Wire Unbounded HDSL Loop including manual service inquiry 1 1 UHL UHL2X 7.88 44.69 31.55 25.65 7.06 18.94 8.42	2-WIR		TIBLE	LOOP													
Stacility reservation - Zone 1																	
8 facility reservation - Zone 2 2 Wire Inhundled HDSL Loop including manual service inquiry 8 facility reservation - Zone 3 1 3 UHL UHLZX 14.46 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1 1 1 UHL UHLZW 7.88 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2 2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2 3 UHL UHLZW 9.09 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2 3 UHL UHLZW 9.09 44.69 31.55 25.65 7.06 18.94 8.42 4 Wire Unbundled HDSL Loop mittout manual service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop industry and service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIRE Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIR Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIR Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIR Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIR Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIR Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2 4 UWIR Unbundled HDSL Loop			- 1	1	UHL	UHL2X	7.88	44.69	31.55	25.65	7.06			18.94	8.42		ĺ
2 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 3		2 Wire Unbundled HDSL Loop including manual service inquiry															
Stability reservation - Zone 3		& facility reservation - Zone 2	- 1	2	UHL	UHL2X	9.09	44.69	31.55	25.65	7.06			18.94	8.42		1
Order Coordination for Specified Conversion Time (per LSR) UHL UHL2W 7.88 44.69 31.55 25.65 7.06 18.94 8.42																	ĺ
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1			- 1	3			14.46		31.55	25.65	7.06			18.94	8.42		L
and facility reservation - Zone 1					UHL	OCOSL		35.74									
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2 1 2 UHL					l												1
and facility reservation - Zone 2			<u> </u>	1	UHL	UHL2W	7.88	44.69	31.55	25.65	7.06			18.94	8.42		
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3			١.	_	l	l											1
and facility reservation - Zone 3				2	UHL	UHL2W	9.09	44.69	31.55	25.65	7.06			18.94	8.42		+
Order Coordination for Specified Conversion Time (per LSR)			١.	_			11.10	44.00	24.55	25.05	7.00			40.04	0.40		i .
CLEC to CLEC Conversion Charge without outside dispatch UHL UREWO 44.69 31.55 18.94 8.42			-	3			14.46		31.33	25.65	7.06			10.94	0.42		
4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	-		<u> </u>						31.55					18 94	8 42		
A Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 1	4-WIRI		TIBI F	LOOP		0		44.00	01.00			 		10.54	0.72	 	<u> </u>
and facility reservation - Zone 1	1. 34.114		<u> </u>	T										1	1	1	
4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2			1	1	UHL	UHL4X	10.39	44.69	31.55	25.65	7.06			18.94	8.42		1
and facility reservation - Zone 2			1							,,,,,	122					İ	
and facility reservation - Zone 3]		1	2	UHL	UHL4X	12.00	44.69	31.55	25.65	7.06	1		18.94	8.42	1	1
and facility reservation - Zone 3		4-Wire Unbundled HDSL Loop including manual service inquiry															
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		and facility reservation - Zone 3	1	3	UHL	UHL4X	19.07	44.69	31.55	25.65	7.06			18.94	8.42		1
and facility reservation - Zone 1					UHL	OCOSL		35.74									ĺ
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2																	1
and facility reservation - Zone 2			I	1	UHL	UHL4W	10.39	44.69	31.55	25.65	7.06			18.94	8.42		L
A-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3			1 .	_								1				1	1
and facility reservation - Zone 3	\vdash			2	UHL	UHL4W	12.00	44.69	31.55	25.65	7.06			18.94	8.42	 	+
Order Coordination for Specified Conversion Time (per LSR)			Ι.	_	l		10.0-	44.00	04.55	05.05	7.00	1		10.01	0.40	1	1
CLEC to CLEC Conversion Charge without outside dispatch UHL UREWO 44.69 31.55 18.94 8.42	 			3			19.07		31.55	25.65	7.06			18.94	8.42	 	
4-WIRE DS1 DIGITAL LOOP USL USLXX 55.53 429.98 268.18 18.94 8.42 4-Wire DS1 Digital Loop - Zone 2 2 USL USLXX 64.13 429.98 268.18 18.94 8.42	 		 	<u> </u>					24.55					40.04	0.40	 	
4-Wire DS1 Digital Loop - Zone 1 1 USL USLXX 55.53 429.98 268.18 18.94 8.42 4-Wire DS1 Digital Loop - Zone 2 2 USL USLXX 64.13 429.98 268.18 18.94 8.42	4.14/101			1	UUL	UKEWU		44.69	31.55			-		18.94	8.42	-	
4-Wire DS1 Digital Loop - Zone 2 2 USL USLXX 64.13 429.98 268.18 18.94 8.42	4-WIRI		1	1	LIGI	LISLYY	55.52	120 00	268 10	1		1		18 04	Ω 42	1	
	H + + -		1							1		1				1	
	 	4-Wire DS1 Digital Loop - Zone 3	1			USLXX	101.93	429.98	268.18	 				18.94	8.42	 	

UNBUNDL	ED NETWORK ELEMENTS - Georgia			1							T -	1 -		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		35.74									
4 140	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		100.91	42.97					18.94	8.42		ļ
4-WI	RE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP		-	UDL	UDL19	25.75	348.55	241.20					18.94	8.42		_
	4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19 UDL19	25.75 29.74	348.55	241.20					18.94	8.42	-	
	4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital 19.2 Kbps		3	UDL	UDL19	29.74 47.27	348.55	241.20					18.94	8.42		
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	25.75	348.55	241.20					18.94	8.42		
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56	29.74	348.55	241.20					18.94	8.42		
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	47.27	348.55	241.20					18.94	8.42		+
	Order Coordination for Specified Conversion Time (per LSR)		Ŭ	UDL	OCOSL		35.74	220					.0.0.	02		†
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	25.75	348.55	241.20					18.94	8.42		
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2	1	2	UDL	UDL64	29.74	348.55	241.20	i i				18.94	8.42		
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3	1	3	UDL	UDL64	47.27	348.55	241.20	i i				18.94	8.42		
	Order Coordination for Specified Conversion Time (per LSR)	1		UDL	OCOSL		35.74		i i							
	CLEC to CLEC Conversion Charge without outside dispatc h			UDL	UREWO		101.95	49.66					18.94	8.42		
2-WI	RE Unbundled COPPER LOOP															
	2-Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 1	I	1	UCL	UCLPB	12.02	44.69	31.55	25.65	7.06			18.94	8.42		
	2-Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 2	- 1	2	UCL	UCLPB	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	22.07	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
	2-Wire Unbundled Copper Loop/Short without manual service		1		LIOL BIA	40.00	44.00	04.55	05.05	7.00			10.01	0.40		
	inquiry and facility reservation - Zone 1		1	UCL	UCLPW	12.02	44.69	31.55	25.65	7.06			18.94	8.42	-	
	2-Wire Unbundled Copper Loop/Short without manual service		2	UCL	UCLPW	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
	inquiry and facility reservation - Zone 2 2-Wire Unbundled Copper Loop/Short without manual service			UCL	UCLPVV	13.00	44.69	31.33	25.05	7.00			10.94	0.42		1
	inquiry and facility reservation - Zone 3		3	UCL	UCLPW	22.07	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLMC	22.07	16.11	16.11	25.05	7.00			10.54	0.42		
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.			OOL	COLIVIO		10.11	10.11								†
	inquiry and facility reservation - Zone 1	1	1	UCL	UCL2L	35.56	44.69	31.55	25.65	7.06			18.94	8.42		
	2-Wire Unbundled Copper Loop/Long - includes manual svc.					-										
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL2L	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3	- 1	3	UCL	UCL2L	65.28	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 1	- 1	1	UCL	UCL2W	35.56	44.69	31.55	25.65	7.06			18.94	8.42		
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 2	I	2	UCL	UCL2W	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 3		3	UCL	UCL2W	65.28	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
	CLEC to CLEC Conversion Charge without outside dispatch						44.00									
4 18/1	(UCL-Des) RE COPPER LOOP	1		UCL	UREWO		44.69	31.55					18.94	8.42	-	
4-771	4-Wire Copper Loop/Short - including manual service inquiry															-
	and facility reservation - Zone 1		1	UCL	UCL4S	12.02	44.69	31.55	25.65	7.06			18.94	8.42		
	4-Wire Copper Loop/Short - including manual service inquiry	+-'-	+-	JUL	30140	12.02	77.03	31.33	25.05	7.00			10.54	0.42	 	
	and facility reservation - Zone 2	1 1	2	UCL	UCL4S	13.88	44.69	31.55	25.65	7.06	1		18.94	8.42	I	
	4-Wire Copper Loop/Short - including manual service inquiry	 '			55240	10.00	77.00	01.00	20.00	7.50			10.04	0.42	1	1
	and facility reservation - Zone 3	Li	3	UCL	UCL4S	22.07	44.69	31.55	25.65	7.06			18.94	8.42	1	
	Order Coordination for Unbundled Copper Loops (per loop)	T ·	Ť	UCL	UCLMC		16.11	16.11							1	1
	4-Wire Copper Loop/Short - without manual service inquiry and	1							i i						1	
	facility reservation - Zone 1	1	1	UCL	UCL4W	12.02	44.69	31.55	25.65	7.06	1		18.94	8.42	I	
	4-Wire Copper Loop/Short - without manual service inquiry and								ĺ							
	facility reservation - Zone 2	1	2	UCL	UCL4W	13.88	44.69	31.55	25.65	7.06	1		18.94	8.42	I	

UNBUNDLE	D NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 3	ı	3	UCL	UCL4W	22.07	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
	4-Wire Unbundled Copper Loop/Long - includes manual svc.	١.	1	UCL	UCL4L	05.50	44.00	04.55	05.05	7.06			40.04	0.40		l
	inquiry and facility reservation - Zone 1 4-Wire Unbundled Copper Loop/Long - includes manual svc.	-	1	UCL	UCL4L	35.56	44.69	31.55	25.65	7.06	-		18.94	8.42		
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
	4-Wire Unbundled Copper Loop/Long - includes manual svc.			OOL	OOL4L	41.07	44.03	31.33	23.03	7.00			10.54	0.72		
	inquiry and facility reservation - Zone 3	1	3	UCL	UCL4L	65.28	44.69	31.55	25.65	7.06			18.94	8.42		ĺ
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
	4-Wire Unbundled Copper Loop/Long - without manual svc.												1		1	
	inquiry and facility reservation - Zone 1		1	UCL	UCL4O	35.56	44.69	31.55	25.65	7.06			18.94	8.42	<u> </u>	
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 2	I	2	UCL	UCL4O	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
	4-Wire Unbundled Copper Loop/Long - without manual svc.			l <u>.</u> .												1
	inquiry and facility reservation - Zone 3		3	UCL	UCL4O	65.28	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Unbundled Copper Loops (per loop)	_		UCL UCL	UCLMC		16.11 44.69	16.11 31.55					18.94	8.42		-
LOOP MODIFIC	CLEC to CLEC conversion Charge without outside dispatch	-		UCL	UREWO		44.69	31.55					18.94	8.42		
LOOP WIODIFIC	ATION			UAL. UHL. UCL.							-					
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft	I		UEQ, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULM2L		0.00	0.00					18.94	8.42		
	Unbundled Loop Modification, Removal of Load Coils - 2 wire	١.			ULM2G		0.00	0.00					40.04	8.42		İ
	greater than 18k ft Unbundled Loop Modification Removal of Load Coils - 4 Wire		<u> </u>	UCL, ULS, UEQ	ULM2G		0.00	0.00					18.94	8.42		
	less than or equal to 18K ft			UHL, UCL	ULM4L		0.00	0.00					18.94	8.42		İ
	Unbundled Loop Modification Removal of Load Coils - 4 Wire			OTIE, OOL	OLIVIAL		0.00	0.00					10.54	0.42		
	pair greater than 18k ft	- 1		UCL	ULM4G		0.00	0.00					18.94	8.42		ĺ
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop	I		UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULMBT		0.00	0.00					18.94	8.42		
SUB-LOOPS																
Sub-Lo	op Distribution		<u> </u>						ļ							1
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-			LIFANII	LICDO A		101.00	404.00					40.01	2.12	1	1
	Up	-	-	UEANL	USBSA		421.08	421.08					18.94	8.42	-	
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up			UEANL	USBSB		67.10	67.10					18.94	8.42	1	1
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Op Sub-Loop - Per Building Equipment Room - CLEC Feeder		1	ULANL	USDSD		01.10	67.10					10.94	0.42	1	
	Facility Set-Up	1		UEANL	USBSC		394.74	394.74					18.94	8.42	1	1
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel	•			- 3500		55 4	55 4					.5.54	5.42		
	Set-Up	- 1		UEANL	USBSD		154.57	154.57					18.94	8.42	1	1
	Unbundled Sub-Loops, Riser Cable, 2-Wire per Loop, Working							·					1		1	
	and Spare Loop Activation			UEANL	USBRC	1.37	2.48	2.48	1.74	1.74			18.94	8.42		<u> </u>
	Unbundled Sub-Loops, Riser Cable, 4-Wire per Loop, Working															1
	and Spare Loop Activation			UEANL	USBRD	2.74	4.96	4.96	1.74	1.74			18.94	8.42		↓
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -															1
	Statewide		SW	UEANL	USBN2	9.12	207.01	171.32	1	-			18.94	8.42	 	
	Order Coordination for Unbundled Sub Leans, per sub-least and			UEANL	USBMC		34.22	34.22								1
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -		<u> </u>	UEANL	USBIVIC		34.22	34.22	-					-		
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Statewide		sw	UEANL	USBN4	8.32	219.35	72.99	123.72	28.77			18.94	8.42		
1	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.22	34.22					1		1	1
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	ı		UEANL	USBR2	1.37	2.48	41.59	115.85	19.17			18.94	8.42	1	
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC) -															
1	Intermediary Access Terminal (IAT)			UEANL	USBRC	1.37	2.48	2.48	1.74	1.74			18.94	8.42		1

NRONDLE	ED NETWORK ELEMENTS - Georgia												Attachi	nent: 2	Exhi	bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.22	34.22								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC) -			UEAINL	USBIVIC		34.22	34.22								
	Intermediary Access Terminal (IAT)			UEANL	USBRD	2.74	4.96	4.96	1.74	1.74			18.94	8.42		
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)			UEANL	USBR4	2.96	176.46	55.11	122.17	19.57			18.94	8.42		
	Sas 2005 1 11110 Intrasarianing Hothert Sasio (Into)	·		0271112	CCBITT	2.00	170.10	00		10.01			10.01	0.12		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.22	34.22								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	ı		UEF	UCS2X	5.54	175.16	55.50	108.86	24.53			18.84	8.42		
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	1	2	UEF	UCS2X	5.54	175.16	55.50	108.86	24.53			18.94	8.42		
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	-	3	UEF	UCS2X	5.54	175.16	55.50	108.86	24.53			18.94	8.42		
								·		·						
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.22	34.22								
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	!	1	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77			18.94	8.42		
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	- !	2	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77			18.94	8.42		
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		3	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77			18.94	8.42		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.22	34.22								
Unbu	ndled Network Terminating Wire (UNTW)			UEF	USBIVIC		34.22	34.22								
Olibu	Unbundled Network Terminating Wire (UNTW) per Pair		1	UENTW	UENPP	1.37	2.48	2.48	1.74	1.74			18.94	8.42		
Notwo	ork Interface Device (NID)		1	OLIVIW	OLIVIT	1.57	2.40	2.40	1.74	1.74			10.34	0.42		
1101111	Network Interface Device (NID) - 1-2 lines			UENTW	UND12		86.37	56.69					18.94	8.42		
	Network Interface Device (NID) - 1-6 lines	i		UENTW	UND16		127.93	98.21					18.94	8.42		
	Network Interface Device Cross Connect - 2 W	1		UENTW	UNDC2		6.15	6.15					18.94	8.42		
	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		6.15	6.15								
B-LOOPS																
Sub-L	Loop Feeder															
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
	Distribution Facility set-up			UDN,UCL,UDL,UDC	USBFW		421.08						18.94	8.42		
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair			UEA,												
	set-up		1	UDN,UCL,UDL,UDC USL	USBFX USBFZ		67.10 521.57	67.10 11.30					18.94 18.94	8.42 8.42		
_	USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice			USL	USBFZ		521.57	11.30					18.94	8.42		
	Grade- Statewide		sw	UEA	USBFA	8.58	206.44	170.05					18.94	8.42		
	Order Coordination for Specified Conversion Time, per LSR		SW	UEA	OCOSL	0.50	35.74	170.03					10.54	0.72		
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice															
	Grade - Statewide		sw	UEA	USBFB	8.58	206.44	170.05					18.94	8.42		
	Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		35.74									
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
	Voice Grade Loop - Statewide		SW	UEA	USBFC	8.58	206.44	170.05					18.94	8.42		
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		35.74									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Statewide		SW	UEA	USBFD	19.91	243.41	81.32	134.77	33.93			18.94	8.42		ļ
_	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		<u> </u>	UEA	OCOSL		35.74							-	-	
	Grade - Statewide		sw	UEA	USBFE	19.91	243.41	81.32	134.77	33.93			18.94	8.42		
	Order Coordination For Specified Conversion Time, Per LSR	-	SW	UEA	OCOSL	19.91	35.74	81.32	134.77	33.93			18.94	8.42	1	1
-	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI -		 	ULA	JUUJL		33.74									<u> </u>
	Statewide		sw	UDN	USBFF	17.73	208.50	62.31	119.68	29.58			18.94	8.42		
	Order Coordination For Specified Conversion Time, Per LSR		<u> </u>	UDN	OCOSL		35.74	12.01								
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		SW	UDC	USBFS	17.73	208.50	62.31	119.68	29.58			19.99	19.99	19.99	19.9
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide			USL	USBFG	79.30	203.69	128.76	124.09	34.80			19.99	19.99	19.99	19.9
	Order Coordination For Specified Conversion Time, Per LSR			USL	OCOSL		35.74									
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop -													_	_	
	Statewide		SW	UCL	USBFH	7.22	195.38	63.15	119.68	29.58			18.94	8.42		
_	Order Coordination For Specified Conversion Time, per LSR		<u> </u>	UCL	OCOSL	10.55	35.74		101 ==							
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewide Order Coordination For Specified Conversion Time, per LSR		SW	UCL	USBFJ OCOSL	13.72	243.41 35.74	81.32	134.77	33.93			18.94	8.42	-	
																1

UNBUNDI F	D NETWORK ELEMENTS - Georgia												Δttach	ment: 2	Fyhil	oit: C
SHOUNDEE											Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		lustani									Elec	Manually		Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	0.1.1						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewide		sw	UDL	USBFO	24.50	243.41	81.32	134.77	33.93			19.99	19.99	19.99	19.99
	Order Coordination For Specified Time Conversion, per LSR		SW	UDL	OCOSL	24.50	35.74	81.32	134.77	33.93	ļ		19.99	19.99	19.99	19.99
-	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -			ODL	OCOSL		33.74									
	Statewide		sw	UDL	USBFP	24.50	243.41	81.32	134.77	33.93			19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, per LSR		311	UDL	OCOSL	24.00	35.74	01.02	104.77	00.00			10.00	10.00	10.00	10.00
SUB-LOOPS																
Sub-Lo	oop Feeder															
	Sub Loop Feeder - DS3 - Per Mile Per Month	1		UE3	1L5SL	12.80										
	Sub Loop Feeder - DS3 - Facility Termination Per Month	I		UE3	USBF1	329.94	3,396.56	406.50	163.61	92.75			18.94	8.42		
	Sub Loop Feeder – STS-1 – Per Mile Per Month			UDLSX	1L5SL	12.80										
	Sub Loop Feeder - STS-1 - Facility Termination Per Month	l I		UDLSX	USBF7	372.78	3,396.56	406.50	163.61	92.75			18.94	8.42		
	Sub Loop Feeder – OC-3 – Per Mile Per Month	ı		UDLO3	1L5SL	9.71										
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per	l .		LIBLOS	LIODE-											
 	Month Sub-Lean Fooder, OC 3, Facility Termination Per Month	++	1	UDLO3 UDLO3	USBF5 USBF2	57.79	3.396.56	406.50	163.61	92.75	1		18.94	8.42	 	
\vdash	Sub Loop Feeder - OC-3 - Facility Termination Per Month Sub Loop Feeder - OC-12 - Per Mile Per Month	l I	+	UDLO3 UDL12	1L5SL	524.13 11.95	3,396.56	406.50	163.61	92.75	 		18.94	8.42	-	
-	Sub Loop Feeder - OC-12 - Per Mile Per Month Sub Loop Feeder - OC-12 - Facility Termination Protection Per			UDL12	ILSSL	11.95										
	Month	1		UDL12	USBF6	519.09										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	l i		UDL12	USBF3	1.570.00	3,396.56	406.50	163.61	92.75			18.94	8.42		
	Sub Loop Feeder - OC-48 - Per Mile Per Month	i		UDL48	1L5SL	39.20	0,000.00	100.00	100.01	02.70			10.01	0.12		
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per															
	Month	- 1		UDL48	USBF9	259.99										
	Sub Loop Feeder - OC-48 - Facility Termination Per Month	ı		UDL48	USBF4	1,505.00	3,582.56	406.50	163.61	92.75			18.94	8.42		
	Sub Loop Feeder - OC-12 Interface On OC-48	I		UDL48	USBF8	323.43	803.69	406.50	163.61	92.75			18.94	8.42		
UNBUNDLED I	LOOP CONCENTRATION															
	Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	441.42	650.81	650.81					19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	52.97	271.17	271.17					19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	478.93	650.81	650.81					19.99	19.99	19.99	19.99
-	Unbundled Loop Concentration - System B (TR303) Unbundled Loop Concentration - DS1 Loop Interface Card		-	ULC ULC	UCT3B UCTCO	89.26 5.04	271.17 126.57	271.17 92.14	33.57	9.40			19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Unbundled Loop Concentration - ISDN Loop Interface (Brite			OLC	00100	5.04	120.57	92.14	33.37	9.40			19.99	19.99	19.99	19.99
	Card)			UDN	ULCC1	8.00	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - UDC Loop Interface (Brite	1		ODIT	OLOG1	0.00	21.07	20.00	10.70	10.71			10.00	10.00	10.00	10.00
	Card)			UDC	ULCCU	8.00	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration2 Wire Voice-Loop Start or															
	Ground Start Loop Interface (POTS Card)	<u>L</u>	<u> </u>	UEA	ULCC2	2.00	21.07	20.96	10.78	10.71	<u> </u>	<u> </u>	19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery															
	Loop Interface (SPOTS Card)		1	UEA	ULCCR	11.89	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface	1														
	(Specials Card)	ļ	1	UEA	ULCC4	7.09	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
 	Unbundled Loop Concentration - TEST CIRCUIT Card	1	1	ULC	UCTTC	34.67	21.07	20.96	10.78	10.71	1		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface	1		UDL	ULCC7	10.51	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
 	Unbundled Loop Concentration - Digital 56 Kbps Data Loop	l	1	ODL	OLCO1	10.51	21.07	20.90	10.76	10.71			15.99	15.99	15.99	15.99
	Interface	1		UDL	ULCC5	10.51	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop	1				10.01	21.07	20.30	10.70	10.71			10.59	10.00	10.00	10.00
	Interface			UDL	ULCC6	10.51	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
UNE OTHER, F	PROVISIONING ONLY - NO RATE															
	NID - Dispatch and Service Order for NID installation			UENTW	UNDBX	0.00	0.00									
	UNTW Circuit Id Establishment, Provisioning Only - No Rate			UENTW	UENCE	0.00	0.00	•								
		1		UEANL,UEF,UEQ,U									1	1]	
<u> </u>	Unbundled Contract Name, Provisioning Only - No Rate	ļ		ENTW	UNECN	0.00	0.00		ļ				ļ	ļ	ļ	
UNE OTHER, F	PROVISIONING ONLY - NO RATE	ļ	1													
		1		TIME FIGURES CIE:]				1	1	1	
	Unbundled Contact Name, Provisioning Only - no rate	1		UAL,UCL,UDC,UDL, UDN,UEA,UHL,ULC	UNECN	0.00	0.00						1	1	1	
 	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no	1	+	UDIN,UEA,UHL,ULC	UNEUN	0.00	0.00		-		 					
	rate	1		UEA,UDN,UCL,UDC	LISBEO	0.00	0.00						1	1	1	
	rato	1	1	OLA,ODIN,OOL,ODO	טטטו ע	0.00	0.00		1		1	1	1	1	1	

CATEGORY															Incremental	Incrementa
	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonre		Nonrecurring Di					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			UEA.USL.UCL.UDL	USBFR	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
	Unbundled DS1 Loop - Expanded Superframe Format option -			OOL	00001	0.00	0.00									
	no rate			USL	CCOEF	0.00	0.00									
HIGH CAPACIT	Y UNBUNDLED LOCAL LOOP															
	High Capacity Unbundled Local Loop - DS3 - Per Mile per															
	month			UE3	1L5ND	8.90										
	High Capacity Unbundled Local Loop - DS3 - Facility			LIEO	LIEODY	000.04	000 50	400.40					07.55	07.55	40.00	40.00
	Termination per month High Capacity Unbundled Local Loop - STS-1 - Per Mile per			UE3	UE3PX	390.34	639.50	426.40					37.55	37.55	18.03	18.03
	month		1	UDLSX	1L5ND	8.90]					1		
	High Capacity Unbundled Local Loop - STS-1 - Facility		l	SELOX	. 20112	0.30			 							1
	Termination per month			UDLSX	UDLS1	421.59	639.50	426.40					37.55	37.55	18.03	18.03
LOOP MAKE-U	P															
	Loop Makeup - Preordering Without Reservation, per working or															
	spare facility queried (Manual).			UMK	UMKLW		35.00	35.00								
	Loop Makeup - Preordering With Reservation, per spare facility						4= 00	4= 00								
	queried (Manual).			UMK	UMKLP		45.00	45.00								
	Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)			UMK	PSUMK		0.075	0.075								
INBUNDLED D	EDICATED TRANSPORT			UIVIK	POUNK		0.075	0.075								
	INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimur	m billin	a perio	d - below DS3=one	month, DS3/5	STS-1=four mo	nths									
	OFFICE CHANNEL - DEDICATED TRANSPORT		J		,											
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			U1TVX	1L5XX	0.0222										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -															
	Facility Termination			U1TVX	U1TV2	17.07	79.61	36.08					18.94	18.94		
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade			11477.07	41.5307	0.0000										
	Rev Bat Per Mile per month Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat			U1TVX	1L5XX	0.0222										
	Facility Termination			U1TVX	U1TR2	17.07	79.61	36.08					18.94	18.94		
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile			OTTVX	OTTIVE	17.07	73.01	30.00					10.54	10.54		
	per month			U1TDX	1L5XX	0.0222										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination			U1TDX	U1TD5	16.45	79.61	36.08					18.94	18.94		
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
	per month			U1TDX	1L5XX	0.0222										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility			LIATOV	U1TD6	40.45	70.04	20.00					40.04	40.04		
	Termination Interoffice Channel - Dedicated Channel - DS1 - Per Mile per		-	U1TDX	סעווט	16.45	79.61	36.08	 				18.94	18.94		
	month		1	U1TD1	1L5XX	0.4523]					1		
	Interoffice Channel - Dedicated Tranport - DS1 - Facility				. 20, 01	3.4020			 							
	Termination		1	U1TD1	U1TF1	78.47	147.07	111.75]				18.94	18.94		
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per															
	month			U1TD3	1L5XX	2.72										
	Interoffice Channel - Dedicated Transport - DS3 - Facility		1]							
	Termination per month		 	U1TD3	U1TF3	788.00	511.10	330.77	\vdash				37.55	37.55	18.03	18.03
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month			U1TS1	1L5XX	2.72										
	month Interoffice Channel - Dedicated Transport - STS-1 - Facility		 	01101	ILOAA	2.12			 				1	1	1	
	Termination		1	U1TS1	U1TFS	783.63	511.10	449.91]				61.19	61.19	3.17	3.17
	CHANNEL - DEDICATED TRANSPORT				55	, 00.00	311.10	440.01	 				51.19	51.13	5.17	0.17
	LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing	g perio	d - belo	w DS3=one month,	DS3/STS-1=f	our months							l		l	
	Local Channel - Dedicated - 2-Wire Voice Grade			ULDVX	ULDV2	13.91	382.95	62.40					18.94	8.42		
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat			ULDVX	ULDR2	13.91	382.95	62.40					18.94	18.94		
	Local Channel - Dedicated - 4-Wire Voice Grade Local Channel - Dedicated - DS1			UNDVX ULDD1	ULDV4 ULDF1	14.99 38.36	368.44 356.15	64.05 312.89					18.94 44.22	8.42 44.22	18.03	18.03

UNBUNDL	ED NETWORK ELEMENTS - Georgia				1									ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Nonrec	urrina	Nonrecurring Dis	sconnect			oss	Rates(\$)	l.	
						Rec	First	Add'l		Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Channel - Dedicated - DS3 - Facility Termination			ULDD3	ULDF3	515.91	639.50	426.31					37.55	37.55		18.03
	Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	6.92										
	Local Channel - Dedicated - STS-1 - Facility Termination			ULDS1	ULDFS	517.56	639.50	426.31					18.94	18.94		
DARK FIBER																
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Channel			UDF	1L5DC	44.22										
	NRC Dark Fiber - Local Channel			UDF	UDFC4		1,355.29	273.69					18.94	18.94		
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Interoffice Channel			UDF	1L5DF	44.22										
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		1,355.29	273.69					18.94	18.94		
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction				l											
	Thereof per month - Local Loop			UDF	1L5DL	44.22							10.01	10.01		
0VV ACCESS	NRC Dark Fiber - Local Loop TEN DIGIT SCREENING	 	<u> </u>	UDF	UDFL4		1,355.29	273.69					18.94	18.94	!	
BAA ACCESS		1		OLID	_	0.0004000										
\vdash	8XX Access Ten Digit Screening, Per Call	+	!	OHD	+	0.0004868			 					-		-
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserved			OHD	N8R1X		6.57	0.76					18.94	18.94	1	
\vdash	8XX Access Ten Digit Screening, Per 8XX No. Established W/O	 	 	טווט	INOINIA		0.07	0.76	 				10.94	10.94	 	-
	POTS Translations			OHD			12.81	1.45					18.94	18.94	1	
-	8XX Access Ten Digit Screening, Per 8XX No. Established With	+		OTID			12.01	1.40					10.34	10.54		
	POTS Translations			OHD	N8FTX		12.81	1.45					18.94	18.94		
	8XX Access Ten Digit Screening, Customized Area of Service		1	OHD	INOL IX		12.01	1.40					10.54	10.54		
	Per 8XX Number			OHD	N8FCX		4.46	2.23					18.94	18.94		
	8XX Access Ten Digit Screening, Multiple InterLATA CXR	1		01.15	1101 071		0	2.20					.0.01	10.01		
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		5.22	2.99					18.94	18.94		
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		7.33	0.76					18.94	18.94	1	
	8XX Access Ten Digit Screening, Call Handling and Destination															
	Features			OHD	N8FDX		4.72	4.46					18.94	18.94		
LINE INFORM	IATION DATA BASE ACCESS (LIDB)															
	LIDB Common Transport Per Query			OQT		0.0000338										
	LIDB Validation Per Query			OQU		0.0105974										
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		50.30						18.94	18.94		
SIGNALING (
	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	133.99										
	CCS7 Signaling Usage, Per TCAP Message			UDB		0.000087										
$oxed{oxed}$	CCS7 Signaling Connection, Per link (A link)	ļ	<u> </u>	UDB	TPP++	17.05	131.96	131.96					18.94	18.94	ļ	
	CCS7 Signaling Connection, Per link (B link) (also known as D						,								I	
	link)			UDB	TPP++	17.05	131.96	131.96					18.94	18.94		
	CCS7 Signaling Usage, Per ISUP Message			UDB	OTUEO	0.0000354										
	CCS7 Signaling Usage Surrogate, per link per LATA	1		UDB	STU56	340.67										
	CCS7 Signaling Point Code, per Originating Point Code			LIDD	CCARO		40.00	40.00					40.04	18.94		
	Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code	1		UDB	CCAPO		40.00	40.00					18.94	18.94		
	Establishment or Change, Per Stp Affected			UDB	CCAPD		8.00	8.00					18.94	18.94		
CALLING NA	ME (CNAM) SERVICE	1		ODB	CCAFD		8.00	8.00					10.54	10.94		
CALLING IVA	CNAM for DB Owners, Per Query	+		OQV		0.01										
	CNAM for Non DB Owners, Per Query		1	OQV		0.01										
	CNAM (Non-Databs Owner), NRC, applies when using the	1			+	0.01			 					 	I	<u> </u>
	Character Based User Interface (CHUI)			OQV	CDDCH		595.00	595.00					18.94	18.94	I	
OPERATOR (CALL PROCESSING	1	1				,,,,,,	222.30						15.51	1	
	Oper. Call Processing - Oper. Provided, Per Min Using BST															
	LIDB					1.20								1	I	
	Oper. Call Processing - Oper. Provided, Per Min Using															
	Foreign LIDB					1.24								1	I	
	Oper. Call Processing - Fully Automated, per Call - Using BST															
	LIDB	<u> </u>	<u>L</u>			0.20								<u> </u>	<u></u>	
	Oper. Call Processing - Fully Automated, per Call - Using						_	-								
	Foreign LIDB	<u> </u>				0.20						<u> </u>				
INWARD OP	RATOR SERVICES															

UNBUNDLE	D NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Inward Operator Svcs - Verification, Per Minute					1.15										
	Inward Operator Services - Verification and Emergency Interrupt - Per Minute					1.15										
BRANDING - O	PERATOR CALL PROCESSING					1.15					1			-	-	
	based CLEC															
	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00					19.99	19.99	19.99	19.99
	Loading of Custom Branded OA Announcement per shelf/NAV				02/100		7,000.00	1,000.00					10.00	10.00	10.00	10.00
	per OCN				CBAOL		500.00	500.00					19.99	19.99		
UNEP C	CLEC															
	Recording of Custom Branded OA Announcement		1			1	7,000.00	7,000.00					19.99	19.99	19.99	19.99
	Loading of Custom Branded OA Announcement per shelf/NAV					İ										
	per OCN		<u>L</u>				500.00	500.00					19.99	19.99		
	SSISTANCE SERVICES							-								
	TORY ASSISTANCE ACCESS SERVICE															
	Directory Assistance Access Service Calls, Charge Per Call	l				0.275	,									
	TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (E	DACC)	<u> </u>								ļ					ļ
	Directory Assistance Call Completion Access Service (DACC),															
	Per Call Attempt					0.10										
	SSISTANCE SERVICES															
	TORY ASSISTANCE DATA BASE SERVICE (DADS)					0.04			-						-	
	Directory Assistance Data Base Service Charge Per Listing Directory Assistance Data Base Service, per month				DBSOF	150.00										
	IRECTORY ASSISTANCE				DBSOF	150.00										
	Based CLEC															
	Recording and Provisioning of DA Custom Branded															
	Announcement			AMT	CBADA		6,000.00	6,000.00					18.94	8.42		
	Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00					18.94	8.42		
UNEP C							.,	.,								
	Recording of DA Custom Branded Announcement						3,000.00	3,000.00					18.94	8.42		
	Loading of DA Custom Branded Announcement per Switch per							·								
	OCN						1,170.00	1,170.00					18.94	8.42		
SELECTIVE RC																
	Selective Routing Per Unique Line Class Code Per Request Per												_	_		
	Switch		<u> </u>		USRCR		180.62	180.62					33.67	7.88		
VIRTUAL COLL							,									
	Virtual Collocation - Application Cost		<u> </u>	AMTFS	EAF		2,848.30	2,848.30	ļ				19.99	19.99	.	
	Virtual Collocation - Cable Installation Cost, per cable	ļ	<u> </u>	AMTES	ESPCX	0.00	2,750.00	2,750.00			ļ		19.99	19.99		ļ
	Virtual Collocation - Floor Space, per sq. ft.	<u> </u>	<u> </u>	AMTES	ESPVX	3.20			ļ					-	-	
	Virtual Collocation - Power, per fused amp	l	1	AMTFS	ESPAX	3.48			 		1			 	 	1
	Virtual Collocation - Cable Support Structure, per entrance cable	l		AMTFS	ESPSX	13.35								1	1	
	capie			UEANL,UEA,UDN,U	ESPSX	13.35					1			-	-	
				DC,UAL,UHL,UCL,U												
				EQ, AMTFS, UDL,												
		l		UNCVX, UNCDX,]									1	I	
	Virtual Collocation - 2-wire Cross Connects (loop)			UNCNX	UEAC2	0.0283	24.56	23.56	9.20	8.30			19.99	19.99	19.99	19.99
	(50		1.20	2.00				13.30	1	12.30
		l		UEA,UHL,UCL,UDL,										1	I	
		l		AMTFS, UAL, UDN,]									1	I	
	Virtual Collocation - 4-wire Cross Connects (loop)	<u> </u>	<u>L</u>	UNCVX, UNCDX	UEAC4	0.0566	24.75	23.70	9.03	8.10	<u> </u>		19.99	19.99	19.99	19.99
				AMTFS,UDL12,				· · · · · · · · · · · · · · · · · · ·								
		l		UDLO3, U1T48,										1	1	
		l		U1T12, U1T03,]									1	I	
		l		ULDO3, ULD12,										1	1	
1 1 1	Virtual Collocation - 2-Fiber Cross Connects			ULD48, UDF	CNC2F	2.88	41.72	30.36	10.43	8.36			2.20	2.20		1

UNBUNDL	ED NETWORK ELEMENTS - Georgia					1						,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Nonrec	rurring	Nonrecurring	Disconnect			oss	Rates(\$)	l	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12,								50				
	Virtual Collocation - 4-Fiber Cross Connects			ULD48, UDF USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1,	CNC4F	5.76	51.03	39.67	13.71	11.65			2.20	2.20		
	Virtual collocation - Special Access & UNE, cross-connect per DS1			U1TD1, USLEL, UNLD1	CNC1X	7.50	155.00	14.00					19.99	19.99		
	Virtual collocation - Special Access & UNE, cross-connect per DS3			USL, ULC, AMTFS, U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	56.25	151.90	11.83					19.99	19.99		
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			AMTFS	VE1CB	0.0023										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0034										
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable			AMTFS	VE1CC	0.000 1	553.43						19.99			
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable			AMTFS	VE1CE		553.43						19.99			
	Virtual Collocation Cable Records - per request Virtual Collocation Cable Records - VG/DS0 Cable, per cable			AMTFS	VE1BA		1,706.00	1,706.00								
	record			AMTFS	VE1BB		922.38	922.38								
	Virtual Collocation Cable Records - VG/DS0 Cable, per each 100 pair			AMTFS	VE1BC		18.00	18.00								
	Virtual Collocation Cable Records - DS1, per T1TIE			AMTFS	VE1BD		8.43	8.43								
	Virtual Collocation Cable Records - DS3, per T3TIE Virtual Collocation Cable Records - Fiber Cable, per 99 fiber			AMTFS	VE1BE		29.49	29.49								
	records			AMTFS	VE1BF		278.61	278.61								
	Virtual collocation - Security Escort - Basic, per half hour			AMTFS	SPTBX		41.00	25.00					19.99	19.99		
	Virtual collocation - Security Escort - Overtime, per half hour			AMTFS	SPTOX		48.00	30.00					19.99	19.99		
	Virtual collocation - Security Escort - Premium, per half hour			AMTFS	SPTPX		55.00	35.00					19.99	19.99		
	Virtual collocation - Maintenance in CO - Basic, per half hour			AMTFS	CTRLX		30.64	30.64					19.99	19.99		
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		35.77	35.77					19.99	19.99		
VIRTUAL CO	Virtual collocation - Maintenance in CO - Premium per half hour DLLOCATION			AMTFS	SPTPM		40.90	40.90					19.99	19.99		
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2- Wire Analog - Res			UEPSR	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus			UEPSB	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN			UEPSX	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			UEPTX	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			UEPEX	VE1R4	0.50	12.60	12.60					18.94	8.42		
VIRTUAL CO	DLLOCATION															
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.03	24.56	23.56	9.20	8.30			19.99	19.99		

UNBUNDLE	D NETWORK ELEMENTS - Georgia				•						•	,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
													1st	Add'l	Disc 1st	DISC Add I
						Rec	Nonred		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
PHYSICAL CO																
	Physical Collocation-2 Wire Cross Connects (Loop) for Line			LIEBOD LIEBOD	DE41.0	0.0040	44.04	44.40					40.00	40.00		
AIN CELECTIV	Splitting E CARRIER ROUTING			UEPSR, UEPSB	PE1LS	0.0318	11.94	11.46					19.99	19.99		
AIN SELECTIV				000	00000		004 700 00						19.99	19.99	19.99	19.99
	Regional Service Establishment End Office Establishment			SRC SRC	SRCEC SRCEO		391,788.00 320.53	320.53					19.99	19.99	19.99	19.99
	Line/Port NRC, per end user			SRC	SRCLP		2.06	2.06					19.99	19.99	19.99	19.99
	Query NRC, per query			SRC	OROLI	0.000448	2.00	2.00					13.33	13.33	15.55	15.55
AIN - BELL SOL	UTH AIN SMS ACCESS SERVICE			Orto		0.000440										
7	AIN SMS Access Service - Service Establishment, Per State,															
	Initial Setup			A1N	CAMSE		90.25	90.25					18.94	18.94		
	'				1		22.20		1						1	1
	AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		29.66	29.66					18.94	18.94	1	1
	AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		29.66	29.66					18.94	18.94		
	AIN SMS Access Service - User Identification Codes - Per User															
	ID Code			A1N	CAMAU		84.43	84.43					18.94	18.94		
	AIN SMS Access Service - Security Card, Per User ID Code,															
	Initial or Replacement			A1N	CAMRC		35.44	35.44					18.94	18.94		
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0023										
	AIN SMS Access Service - Session, Per Minute					0.0795604										
	AIN SMS Access Service - Company Performed Session, Per															
	Minute					2.08										
AIN - BELLSO	UTH AIN TOOLKIT SERVICE															
	AIN Toolkit Service - Service Establishment Charge, Per State,			0444	D 4 D 0 0		00.74	00.74					40.04	40.04		
	Initial Setup AIN Toolkit Service - Training Session, Per Customer			CAM	BAPSC BAPVX		86.74	86.74 8.348.00					18.94	18.94		
	AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPVX		8,348.00	8,348.00	-				18.94	18.94		
	DN, Term. Attempt				BAPTT		19.13	19.13					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DAFII		19.13	19.13					10.54	10.94		
	DN, Off-Hook Delay				BAPTD		114.80	114.80					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DAI 1D		114.00	114.00					10.34	10.54		
	DN, Off-Hook Immediate				BAPTM		19.13	19.13					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				27.11.11.1		10.10	10.10					.0.01	10.01		
	DN, 10-Digit PODP				BAPTO		70.06	70.06					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, CDP				BAPTC		70.06	70.06					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, Feature Code				BAPTF		70.06	70.06					18.94	18.94		
	AIN Toolkit Service - Query Charge, Per Query					0.0209223										
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit															
	Subscription, Per Node, Per Query					0.0053137										
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access															
	Account, Per 100 Kilobytes					1.46										
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service			l											1	1
	Subscription			CAM	BAPMS	15.96	22.64	22.64					18.94	18.94	-	-
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service			CAM	DADI C	0.0004400	20.04	20.04					40.04	40.04	1	1
	Subscription		<u> </u>	CAM	BAPLS	0.0861109	22.64	22.64	 				18.94	18.94	!	!
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			CAM	BAPDS	15.87	22.64	22.64					18.94	18.94	1	1
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit		-	CAIVI	DAPUS	15.87	22.04	22.04	 				10.94	10.94	-	-
1	Service Subscription		l	CAM	BAPES	0.0028704	22.64	22.64]				18.94	18.94	I	I
ENHANCED EX	TSEIVICE SUBSCRIPTION KTENDED LINK (EELs)	-		C, UVI	טרו בט	0.0020104	22.04	22.04	 				10.34	10.94	t	t
	New Density Zone 1 EELs are available in the following MSA	s: Orlan	do. FI	: Miami, FL: Ft. I au	derdale. Fl ·	Atlanta, Ga: Ne	w Orleans. L A							1	I	I
	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-					,,,	, - ,		†						1	1
	In all states, EEL network elements shown below also apply t					erted to UNE ra	ites. A Switch	As Is Charge a	pplies to curre	ntly combined	facilities co	onverted to	UNEs.(Non-re	curring rates	do not apply	.)
	In All States the EEL network elements apply to ordinarily co														1	ĺ
	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT															
İ	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport								İ							
1	Combination - Zone 1	1	1	UNCVX	UEAL2	16.84	104.14	78.10				ı	18.94	8.42	1	1

OMBUNDLE	D NETWORK ELEMENTS - Georgia			1	1	l					Cua Ord	Cua Orden	Attachr	nent: 2 Incremental		bit: C Incremental
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Submitted Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring Dis					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2		2	UNCVX	UEAL2	19.45	104.14	78.10					18.94	8.42		
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3		3	UNCVX	UEAL2	30.92	104.14	78.10					18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per month			UNC1X	1L5XX	0.4523										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	DS1 Channelization System Per Month			UNC1X	MQ1	126.22	.000						00.00	271.10	10.00	11.00
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	1.17	12.02	8.66					18.94	8.42		
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.84	104.14	78.10					18.94	8.42		
	Each Additional 2-Wire VG Loop(SL2) in the same DS1		-	ONOVA	OLALZ	10.04	104.14	70.10					10.34	0.42		
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	19.45	104.14	78.10					18.94	8.42		
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	30.92	104.14	78.10					18.94	8.42		
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	1.17	12.02	8.66					18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	22,26	206.95	170.57					18.94	8.42		
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	25.70	206.95	170.57					18.94	8.42		
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	40.86	206.95	170.57					18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		3	UNC1X	1L5XX	0.4523	200.93	170.57					10.54	0.42		
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per			UNCIA	ILOAA	0.4525										
	Month Channelization - Channel System DS1 to DS0 combination Per			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Month			UNC1X	MQ1	126.22										
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	1.17	12.02	8.66								
	Additional 4-Wire Analog Voice Grade Loop in same DS1															İ
	Interoffice Transport Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1		1	UNCVX	UEAL4	22.26	206.95	170.57					18.94	8.42		
	Interoffice Transport Combination - Zone 2 Additional 4-Wire Analog Voice Grade Loop in same DS1		2	UNCVX	UEAL4	25.70	206.95	170.57					18.94	8.42		
	Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVX	UEAL4	40.86	206.95	170.57					18.94	8.42		
	per month			UNCVX	1D1VG	1.17	12.02	8.66					18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge			UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIR	E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	DFFICE	TRANSPORT (EEL)												
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	25.75	384.56	241.20					18.94	8.42		1
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	29.74	384.56	241.20					18.94	8.42		
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	47.27	384.56	241.20					18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.4523	304.30	271.20					10.54	0.72		
	Interoffice Transport - Dedicated - DS1 - combination Facility						404.0-						20.0-	a= 1-	40.0-	
	Termination Per Month Channelization - Channel System DS1 to DS0 combination Per			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Month	<u> </u>	<u> </u>	UNC1X	MQ1	126.22										<u> </u>

UNBUNDLI	ED NETWORK ELEMENTS - Georgia			T	1	1					1 -		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)		Submitte Elec per LSF	d Submitted Manually	Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring Disconr				Rates(\$)		
							First	Add'l	First Add	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	1.86	12.02	8.66				18.94	8.42		
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	25.75	384.56	241.20				18.94	8.42		
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	29.74	384.56	241.20				18.94	8.42		
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	47.27	384.56	241.20				18.94	8.42		
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)			UNCDX	1D1DD	1.86	12.02	8.66				18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As-					1.00		11.27							
4-WIE	Is Charge	INTERC	FFICE	UNC1X TRANSPORT (FFL)	UNCCC		12.97	11.27		+	-	18.94	8.42		
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		,, , ICE	TARIO ON (CEL)	1										
	Transport Combination - Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		1	UNCDX	UDL64	25.75	348.55	241.20			1	18.94	8.42		
	Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		2	UNCDX	UDL64	29.74	348.55	241.20				18.94	8.42		
	Transport Combination - Zone 3		3	UNCDX	UDL64	47.27	348.55	241.20				18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.4523									
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	78.47	194.63	141.51				33.63	27.49	19.88	11.85
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	126.22									
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.86	12.02	8.66				18.94	8.42		
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	25.75	348.55	241.20				18.94	8.42		
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2			UNCDX	UDL64	29.74	348.55	241.20				18.94	8.42		
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1														
	Interoffice Transport Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System		3	UNCDX	UDL64	47.27	348.55	241.20				18.94	8.42		
	combination - per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	1D1DD	1.86	12.02	8.66				18.94	8.42		
	Is Charge			UNC1X	UNCCC		12.97	11.27				45.46	15.72		
4-WIF	RE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE	ROFFI	CE TRA	ANSPORT (EEL)											
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1		1	UNC1X	USLXX	55.53	443.20	138.69				18.94	8.42		
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	64.13	443.20	138.69				18.94	8.42		
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3		3	UNC1X	USLXX	101.93	443.20	138.69				18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.4523									
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	78.47	194.63	141.51				33.63	27.49	19.88	11.85
	Nonrecurring Currently Combined Network Elements Switch -As-					70.47					1			13.00	11.05
4-18/15	Is Charge RE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE	POEE	CE TD	UNC1X	UNCCC		12.97	11.27				45.46	15.72		
4-141	First DS1Loop in DS3 Interoffice Transport Combination - Zone	NOFFE		UNC1X	USLXX	55.53	443,20	138.69			1	18.94	8.42		
	First DS1Loop in DS3 Interoffice Transport Combination - Zone												-		
	2 First DS1Loop in DS3 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	64.13	443.20	138.69			1	18.94	8.42		
	3 Interoffice Transport - Dedicated - DS3 combination - Per Mile		3	UNC1X	USLXX	101.93	443.20	138.69			1	18.94	8.42		-
	Per Month			UNC3X	1L5XX	2.72									

UNBUNDLE	D NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec First	urring Add'l	Nonrecurring First	g Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - DS3 - Facility Termination per						11130	Auu i	11130	Addi	JOINEC	JONIAN	JOWAN	JOINAN	JOHAN	JOHIAN
	month			UNC3X	U1TF3	788.00	198.45	153.15					37.55	37.55	18.03	18.03
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3	137.73	196.66	204.61					18.94	8.42		
-	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination -			UNC1X	UC1D1	11.02	12.02	8.66					18.94	8.42		
	Zone 1		1	UNC1X	USLXX	55.53	443.20	138.69					18.94	8.42		
	Additional DS1Loop in DS3 Interoffice Transport Combination -		_													
	Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination -		2	UNC1X	USLXX	64.13	443.20	138.69					18.94	8.42		-
	Zone 3		3	UNC1X	USLXX	101.93	443.20	138.69					18.94	8.42		İ
	DS3 Interface Unit (DS1 COCI) combination per month		Ĭ	UNC1X	UC1D1	11.02	12.02	8.66					18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As-															
	ls Charge	<u></u>		UNC3X	UNCCC		12.97	11.27					45.46	15.72		
2-WIRI	VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT 2-WireVG Loop used with 2-wire VG Interoffice Transport	EROFF	ICE TE	RANSPORT (EEL)												!
	Combination - Zone 1		1	UNCVX	UEAL2	16.84	104.14	78.10					18.94	8.42		ĺ
	2-WireVG Loop used with 2-wire VG Interoffice Transport			OI TO VX	OL/ ILL	10.04	104.14	70.10					10.04	0.42		
	Combination - Zone 2		2	UNCVX	UEAL2	19.45	104.14	78.10					18.94	8.42		
	2-WireVG Loop used with 2-wire VG Interoffice Transport															
	Combination - Zone 3 Interoffice Transport - Dedicated - 2-wire VG combination - Per		3	UNCVX	UEAL2	30.92	104.14	78.10					18.94	8.42		
	Mile Per Month			UNCVX	1L5XX	0.0222										ĺ
	Interoffice Transport - Dedicated - 2- Wire Voice Grade				1-21-1											
	combination - Facility Termination per month			UNCVX	U1TV2	17.07	79.61	36.08					18.94	18.94		
	Nonrecurring Currently Combined Network Elements Switch -As-												4= 40			ĺ
4-WID	Is Charge VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EROFE	ICE TE	UNCVX	UNCCC		12.97	11.27					45.46	15.72		
7-1/11(1	4-WireVG Loop used with 4-wire VG Interoffice Transport	LICOLI	IOL II	LANGI OKT (LLL)												—
	Combination - Zone 1		1	UNCVX	UEAL4	22.26	206.95	170.57					18.94	8.42		ĺ
	4-WireVG Loop used with 4-wire VG Interoffice Transport															
	Combination - Zone 2		2	UNCVX	UEAL4	25.70	206.95	170.57					18.94	8.42		!
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	40.86	206.95	170.57					18.94	8.42		
	Interoffice Transport - Dedicated - 4-wire VG combination - Per															
	Mile Per Month			UNCVX	1L5XX	0.0222										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV4	17.07	79.61	36.08					18.94	18.94		ĺ
	Nonrecurring Currently Combined Network Elements Switch -As-			UNCVX	01174	17.07	79.61	36.08					18.94	18.94		
	Is Charge			UNCVX	UNCCC		12.97	11.27					45.46	15.72		ĺ
DS3 D	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRA	NSPOR	RT (EEL)												
	High Capacity Unbundled Local Loop - DS3 combination - Per		1	LINGSV	1L5ND	8.90			1							
	Mile per month High Capacity Unbundled Local Loop - DS3 combination -			UNC3X	1L5ND	8.90										
	Facility Termination per month			UNC3X	UE3PX	390.34	639.50	426.40					37.55	37.55	18.03	18.03
	Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	2.72										
	Interoffice Transport - Dedicated - DS3 combination - Facility			l												
	Termination per per month Nonrecurring Currently Combined Network Elements Switch -As-		<u> </u>	UNC3X	U1TF3	788.00	198.45	153.15	.				37.55	37.55	18.03	18.03
	Inchrecurring Currently Combined Network Elements Switch -As- Is Charge		1	UNC3X	UNCCC		12.97	11.27	1				45.46	15.72		
STS1 I	DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROF	FICE TE	RANSP				.2.07		1				10.40	.0.72		
	High Capacity Unbundled Local Loop - STS1 combination - Per															
	Mile per month			UNCSX	1L5ND	8.90										
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month			UNCSX	UDLS1	421.59	639.50	426.40					37.55	37.55	18.03	18.03
	Interoffice Transport - Dedicated - STS1 combination - Per Mile		1	UNUUN	ODLO I	421.39	039.50	420.40					31.33	31.33	10.03	10.03
	per month .	L		UNCSX	1L5XX	2.72			<u> </u>					<u></u>	<u></u>	<u> </u>
	Interoffice Transport - Dedicated - STS1 combination - Facility															
1	Termination per month			UNCSX	U1TFS	783.63	198.45	449.91	1	l	<u> </u>		37.55	37.55	18.03	18.03

UNBUNDLI	D NETWORK ELEMENTS - Georgia												Attach	ment: 2	Exhil	oit: C
											Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		Interi	l_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonrec	urring	Nonrecurring	Disconnect		1	oss	Rates(\$)	I	ı
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNCSX	UNCCC		12.97	11.27					45.46	15.72		
2-WIF	E ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL)													
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination			LINGNIV	1141.00/	04.00	200 00	400.00					40.04	0.40		
	Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination		1	UNCNX	U1L2X	21.89	233.38	180.38					18.94	8.42		
	Transport - Zone 2		2	UNCNX	U1L2X	25.27	233.38	180.38					18.94	8.42		
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination			ONONA	UTLZX	25.21	255.50	100.50					10.54	0.42		
	Transport - Zone 3		3	UNCNX	U1L2X	40.17	233.38	180.38					18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.4523										
	Interoffice Transport - Dedicated - DS1 combintion - Facility						_									
	Termination per month			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Channelization - Channel System DS1 to DS0 combination -			LINICAY	MO1	400.00										
\vdash	per month 2 wire ISDN COCI (RRITE) DS1 to DS0 Channel System	<u> </u>		UNC1X	MQ1	126.22			-							
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month	1		UNCNX	UC1CA	3.37	12.02	8.66			1		33.63	27.49	19.88	11.85
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport	<u> </u>		0.40147	JUIUA	3.31	12.02	0.00	+		1		33.03	21.49	13.00	11.03
	Combination - Zone 1		1	UNCNX	U1L2X	21.89	233.38	180.38					18.94	8.42		
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 2		2	UNCNX	U1L2X	25.27	233.38	180.38					18.94	8.42		
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 3		3	UNCNX	U1L2X	40.17	233.38	180.38					18.94	8.42		
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System			LINGNIV	110404	0.07	40.00	0.00					00.00	07.40	40.00	44.05
	combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCNX	UC1CA	3.37	12.02	8.66					33.63	27.49	19.88	11.85
	Is Charge			UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIF	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROF	FICE TI		011000		12.01	11.27					40.40	10.72		
	First DS1 Loop in STS1 Interoffice Transport Combination -			, ,												
	Zone 1		1	UNC1X	USLXX	55.53	443.20	138.69					18.94	8.42		
	First DS1 Loop in STS1 Interoffice Transport Combination -															
	Zone 2		2	UNC1X	USLXX	64.13	443.20	138.69					18.94	8.42		
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	HELVY	101.93	442.20	120.60					18.94	8.42		
	Interoffice Transport - Dedicated - STS1 combination - Per Mile		3	UNCIX	USLXX	101.93	443.20	138.69					18.94	8.42		
	Per Month			UNCSX	1L5XX	2.72										
	Interoffice Transport - Dedicated - STS1 combination - Facility			200/.	. 20, 01	2.72			1	1						
	Termination	1		UNCSX	U1TFS	783.63	198.45	449.91			1		37.55	37.55	18.08	18.03
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	182.04	196.66	204.61					37.55	37.55	18.08	18.03
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.02	12.02	8.66					37.55	37.55	18.08	18.03
	Additional DS1Loop in STS1 Interoffice Transport Combination -			LINICAY	LICLYS	55.50	440.00	100.00					1000			
	Zone 1 Additional DS1Loop in STS1 Interoffice Transport Combination -	 	1	UNC1X	USLXX	55.53	443.20	138.69	 	-			18.94	8.42		
	Zone 2	1	2	UNC1X	USLXX	64.13	443.20	138.69			1		18.94	8.42		
	Additional DS1Loop in STS1 Interoffice Transport Combination -			001/	552,00	04.10	++0.20	100.00					10.54	0.42		
	Zone 3		3	UNC1X	USLXX	101.93	443.20	138.69					18.94	8.42		
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.02	12.02	8.66					18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As-	1									1		1	1		
4	Is Charge		DANC	UNCSX	UNCCC		12.97	11.27	ļ		ļ		45.46	15.72		
4-WIF	E 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE	KANSI	OKI (EEL)	1				1	 			 	 		
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1	1	1	UNCDX	UDL56	25.75	384.56	241.20			1		18.94	8.42		
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport	<u> </u>		0.4007	JDLJU	23.13	304.30	241.20	+		1		10.34	0.42		
	Combination - Zone 2		2	UNCDX	UDL56	29.74	384.56	241.20					18.94	8.42		
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport													1	1	
	Combination - Zone 3		3	UNCDX	UDL56	47.27	384.56	241.20					18.94	8.42		
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -				I											
	Per Mile	ļ		UNCDX	1L5XX	0.0222			ļ							
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -	1		UNCDX	U1TD5	16.45	147.07	111.75			1		33.63	27.49	19.88	11.85
	Facility Termination	1	<u> </u>	UNCDY	לעווט	16.45	147.07	111./5	L	l	l	<u> </u>	33.63	27.49	19.88	11.85

UNB	UNDLE	D NETWORK ELEMENTS - Georgia			1		1								ment: 2		bit: C
CATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Nonrecurring Currently Combined Network Elements Switch -As-															
	4 1400	Is Charge		<u> </u>	UNCDX	UNCCC		12.97	11.27					45.46	15.72		<u> </u>
	4-WIRE	64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROI	FFICE I	RANS	PORT (EEL)												.
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		١,	LINCDY	LIDL C4	25.75	348.55	241.20					40.04	8.42		
		Combination - Zone 1 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport			UNCDX	UDL64	25.75	340.33	241.20					18.94	0.42		
		Combination - Zone 2		2	UNCDX	UDL64	29.74	348.55	241.20					18.94	8.42		
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport			ONODA	ODLOT	20.14	040.00	241.20	1				10.04	0.42		
		Combination - Zone 3		3	UNCDX	UDL64	47.27	348.55	241.20					18.94	8.42		
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
		Per Mile			UNCDX	1L5XX	0.0222										
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	1	Facility Termination			UNCDX	U1TD6	16.45	147.07	111.75					33.63	27.49	19.88	11.85
		Nonrecurring Currently Combined Network Elements Switch -As-															
		Is Charge			UNCDX	UNCCC		12.97	11.27					45.46	15.72		
ADDI		NETWORK ELEMENTS		<u>. </u>													
-	wnen	used as a part of a currently combined facility, the non-recurr	ng cna	rges ac	not apply, but a s	Witch As is c	narge does app	oly.									
		used as ordinarily combined network elements in All States, th curring Currently Combined Network Elements "Switch As Is"					As is Charge	does not.									
	Nonrec	Nonrecurring Currently Combined Network Elements Switch As-	Charge	(One a	applies to each con	ibination)										-	
		Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		12.97	11.27					18.94	18.94		
		Nonrecurring Currently Combined Network Elements Switch -As-			ONOVA	ONCCC		12.57	11.21	1				10.54	10.54		
		Is Charge - 56/64 kbps			UNCDX	UNCCC		12.97	11.27					18.94	18.94		
		Nonrecurring Currently Combined Network Elements Switch -As-						_									
		Is Charge - DS1			UNC1X	UNCCC		12.97	11.27					18.94	18.94		
		Nonrecurring Currently Combined Network Elements Switch -As-															
		ls Charge - DS3			UNC3X	UNCCC		12.97	11.27					18.94	18.94		
		Nonrecurring Currently Combined Network Elements Switch -As-															
		Is Charge - STS1		<u> </u>	UNCSX	UNCCC		12.97	11.27					18.94	18.94		ļ
	NOTE:	Local Channel - Dedicated Transport - minimum billing period	d - Belo	w DS3				070.07	00.40					40.04	40.04		
		Local Channel - Dedicated - 2-Wire Voice Grade Local Channel - Dedicated - 4-Wire Voice Grade			UNCXV	ULDV2 ULDV4	13.91 14.99	272.07	60.43 60.43					18.94 18.94	18.94 18.94		
		Local Channel - Dedicated - 4-Wire voice Grade Local Channel - Dedicated - DS1			UNC1X	ULDF1	38.36	272.07 356.15	312.89	-				10.94	10.94		<u> </u>
		Local Channel - Dedicated - DS1 - Per Mile per month			UNC3X	1L5NC	6.92	330.13	312.09								1
		Local Channel - Dedicated - DS3 - Facility Termination			UNC3X	ULDF3	515.91	639.50	426.31	1				18.94	18.94		
		Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC	6.92	000.00	120.01					10.01	10.01		
		Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	517.56	639.50	426.31					18.94	18.94		
	Option	al Features & Functions:															
	MULTI	PLEXERS															
		Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	126.22	198.22	123.59					14.75	6.55	10.70	
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per		1	l					1						_	
	1	month (2.4-64kbs)	ļ		UDL	1D1DD	1.86	12.02	8.66	<u> </u>				14.75	6.55	10.70	
l		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per	l		LIDNI	110404	0.0-	10.00	0.00						0.55	10	
<u> </u>	-	month		 	UDN	UC1CA	3.37	12.02	8.66					14.75	6.55	10.70	
	+	Voice Grade COCI - DS1 to DS0 Channel System - per month DS3 to DS1 Channel System per month			UEA UXTD3	1D1VG MQ3	1.17 182.04	12.02 265.91	8.66 188.78	 				14.75 14.75	6.55 6.55		
	+	STS1 to DS1 Channel System per month	 		UXTS1	MQ3	182.04	265.91	188.78	1				14.75	6.55	10.70	
-	+	DS3 Interface Unit (DS1 COCI) used with Loop per month	-		USL	UC1D1	11.02	12.02	8.66	1				14.75	6.55	10.70	
		DS3 Interface Unit (DS1 COCI) used with Local Channel per				-0.5.		.2.02	2.00	1				70	3.00	.5.70	1
1	1	month		1	ULDD1	UC1D1	11.02	12.02	8.66					14.75	6.55	10.70	
	1	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel								1							1
L	1_	per month	<u></u>	L	U1TD1	UC1D1	11.02	12.02	8.66	<u> </u>				14.75	6.55	10.70	<u></u>
UNBU		OCAL EXCHANGE SWITCHING(PORTS)												_			
		nge Ports															
		Although the Port Rate includes all available features in GA, I	KY, LA	& TN, t	he desired features	will need to b	e ordered usir	ng retail USOC	· ·								1
	2-WIRE	VOICE GRADE LINE PORT RATES (RES)	ļ	ļ	LIEDOD	HEDE:				1							
	1	Exchange Ports - 2-Wire Analog Line Port- Res.	 	 	UEPSR	UEPRL	1.85	17.16	17.16	1				18.94	8.42	!	├
		1	l	1	UEPSR	UEPRC	1.85	17.16	17.16	1		Ì		18.94	8.42		1

ONRONDL	ED NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.85	17.16	17.16					18.94	8.42		
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LUM)			UEPSR	UEPAP	1.85		17.16					18.94	8.42		
	Exchange Ports - 2-Wire Voice Georgia basic dialing port without Caller ID			UEPSR	UEPWC	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port for use with Caller ID - res			UEPSR	UEPWQ	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - outgoing only			UEPSR	UEPWR	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPSR	UEPRT	1.85	17.16	17.16					18.94	8.42		
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00					18.94	8.42		
FEAT	URES				1	2.20	2.20	2.30	1							
	All Available Vertical Features			UEPSR	UEPVF	0.00	0.00	0.00					18.94	8.42		
2-WIR	RE VOICE GRADE LINE PORT RATES (BUS)															
	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus			UEPSB	UEPBL	1.85	17.16	17.16					18.94	8.42		
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.85	17.16	17.16					18.94	8.42		
	Exchange Ports - 2-Wire Voice Georgia Business Basic Dialing Port, with Caller ID capability			UEPSB	UEPWP	1.85	17.16	17.16					18.94	8.42		
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.85	17.16	17.16					18.94	8.42		
	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus			UEPSB	UEPB1	1.85	17.16	17.16					18.94	8.42		
	Exchange Ports - 2-Wire Voice Georgia Business Dialing Plan without Caller ID			UEPSB	UEPWD	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPSB	UEPBE	1.85	17.16	17.16					18.94	8.42		
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00					18.94	8.42		
FEAT	URES			LIEDOD	LIEDVE	0.00	0.00	0.00					40.04	0.40		
EVCL	All Available Vertical Features IANGE PORT RATES (DID & PBX)			UEPSB	UEPVF	0.00	0.00	0.00					18.94	8.42		
EXCH	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia extended dialing port, PBX 1-Way Outdial Trunk			UEPSE	UEPPO	1.85	17.16	17.16					18.94	8.42		
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.85	17.16	17.16					18.94	8.42		
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.85	17.16	17.16					18.94	8.42		
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.85	17.16	17.16					18.94	8.42		
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.85	17.16	17.16					18.94	8.42		
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPSP	UEPXD	1.85	17.16	17.16					18.94	8.42		
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPSP	UEPXE	1.85	17.16	17.16					18.94	8.42		
	Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPSP	UEPXL	1.85	17.16	17.16					18.94	8.42		<u> </u>
	Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			UEPSP	UEPXM	1.85	17.16	17.16					18.94	8.42		
	Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port		-	UEPSP UEPSP	UEPXO UEPXS	1.85 1.85	17.16 17.16	17.16 17.16	 				18.94 18.94	8.42 8.42	-	
	2-Wire voice unbundled Georgia basic dialing port - 1-Way			UEPSP	UEPWS	1.85	17.16	17.16					18.94	8.42		
	Oudial Trunk 2-Wire voice unbundled Georgia basic dialing port - 2-Way Trunk			UEPSP	UEPWS	1.85	17.16	17.16					18.94	8.42		

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachi	ment: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		Nonred	RATES(\$)	I Monarcourie	g Disconnect		Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I Rates(\$)	Charge -	Charge
						Rec	First	Add'l	First	Add'l	COMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled Georgia basic dialing port - 2-way PBX						FIISL	Auu i	FIISL	Add I	SOWIEC	SOWAN	SOWAN	SUMAN	SOWAN	SOWAN
	Trunk			UEPSP	UEPPQ	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX LD			02. 0.	02Q	1.00							10.01	02		
	Terminal Ports			UEPSP	UEPPS	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX Toll															
	Terminal Ports			UEPSP	UEPPT	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															
	DDD Terminal Port			UEPSP	UEPPU	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX LD			LIEDOD	UEPPV	4.05	47.40	47.40					40.04	0.40		
-	Terminal Switchboard Port 2-Wire voice unbundled Georgia basic dialing port - PBX LD			UEPSP	UEPPV	1.85	17.16	17.16		<u> </u>			18.94	8.42		
	Terminal Switchboard DDD Capable Port		1	UEPSP	UEPPW	1.85	17.16	17.16	I			1	18.94	8.42		
	Subsequent Activity		1	UEPSP	USASC	0.00	0.00	0.00	†	†			18.94	8.42		
FEATU				-		2.20	2.30	2.30	1	Ì				1		
	All Available Vertical Features			UEPSP UEPSE	UEPVF	0.00	0.00	0.00					18.94	8.42		
EXCHA	NGE PORT RATES (COIN)															
	Exchange Ports - Coin Port					2.05	17.16	17.16					18.94	8.42		
	Transmission/usage charges associated with POTS circuit sy													L		
	Access to B Channel or D Channel Packet capabilities will be	availa	ble only	through BFR/New	Business Re	quest Process.	Rates for the	packet capabi	lities will be d	etermined via i	the Bona Fig	le Request/	New Business	s Request Pro	cess.	
	LOCAL EXCHANGE SWITCHING(PORTS) UNGE PORT RATES															
LACITA	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	11.35	61.91	61.91					19.99	19.99	19.99	19.9
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID			OLI LX	OLI I Z	11.00	01.01	01.01					10.00	10.00	10.00	10.0
	capability			UEPDD	UEPDD	120.80	108.38	60.88					19.99	19.99	19.99	19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	13.47	47.37	47.37					39.98	39.98		
	All Features Offered			UEPTX UEPSX	UEPVF	0.00	0.00	0.00								
	Transmission/usage charges associated with POTS circuit sy															
NOTE:	Access to B Channel or D Channel Packet capabilities will be	availa	ble only							etermined via t	the Bona Fig	le Request/	New Business	s Request Pro	cess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port			UEPTX UEPSX UEPEX	U1UMA UEPEX	0.00 163.16	0.00 186.80	0.00 186.80					37.88	37.88		
LINBUR	NDLED PORT with REMOTE CALL FORWARDING CAPABILITY	,		UEPEX	UEPEX	163.16	100.00	100.00					37.00	37.00		
	IDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE															-
	Unbundled Remote Call Forwarding Service, Area Calling, Res			UEPVR	UERAC	1.85	17.16	17.16					18.94	8.42		
	g,			-												
	Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.85	17.16	17.16					18.94	8.42		
	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.85	17.16	17.16					18.94	8.42		
	Unbundled Remote Call Forwarding Service, IntraLATA - Res			UEPVR	UERTR	1.85	17.16	17.16					18.94	8.42		
Non-Re	ecurring		!		<u> </u>					ļ	<u> </u>			ļ	ļ	
	Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is			UEPVR	USAC2		2.01	0.31	1				33.67	7.88	11.17	3.9
-	Unbundled Remote Call Forwarding Service - Conversion with		 	OLF VK	USAUZ		2.01	0.31	 	<u> </u>			33.67	1.88	11.17	3.8
	allowed change (PIC and LPIC)			UEPVR	USACC		2.01	0.31								
UNBUN	NDLED REMOTE CALL FORWARDING - Bus			02. ***	00/100		2.01	0.01								
					1					1					1	
	Unbundled Remote Call Forwarding Service, Area Calling - Bus		<u> </u>	UEPVB	UERAC	1.85	17.16	17.16					18.94	8.42		
	Unbundled Remote Call Forwarding Service, Local Calling - Bus		<u> </u>	UEPVB	UERLC	1.85	17.16	17.16	ļ	ļ	ļ		18.94	8.42		
	Unbundled Remote Call Forwarding Service, InterLATA - Bus		!	UEPVB	UERTE	1.85	17.16	17.16		ļ	<u> </u>		18.94	8.42	ļ	
	Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service Expanded and		!	UEPVB	UERTR	1.85	17.16	17.16	 	1	 		18.94	8.42		
	Exception Local Calling		1	UEPVB	UERVJ	1.85	17.16	17.16	1			1	18.94	8.42		
Non-Re	ecurring		1	OLI VD	OLIVO	1.03	17.10	17.10	-	†			10.54	0.42		
	Unbundled Remote Call Forwarding Service - Conversion -		<u> </u>						1							
	Switch-as-is			UEPVB	USAC2		2.01	0.31	1				33.67	7.88	11.17	3.9
	Unbundled Remote Call Forwarding Service - Conversion with															
	allowed change (PIC and LPIC)			UEPVB	USACC		2.01	0.31								
	OCAL SWITCHING, PORT USAGE															
End Of	fice Switching (Port Usage)		<u> </u>			0.001000					ļ					
	End Office Switching Function, Per MOU				1	0.0016333									L	<u> </u>

LIMBLE	IDI EI	O NETWORK ELEMENTS Coordia												A441-		F. 1. 1	0
UNBU	NDLEI	D NETWORK ELEMENTS - Georgia		1		1	1					Con Onder	Cur Onden		ment: 2		bit: C
		1												Incremental		Incremental	
		1										Submitted			Charge -	Charge -	Charge -
CATEC	SDV.	DATE ELEMENTO	Interi	7	BCS	USOC			DATEC(#)			Elec		Manual Svc	Manual Svc		
CATEGO	JKT	RATE ELEMENTS	m	Zone	BCS	USUC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		·												Electronic-	Electronic-	Electronic-	Electronic-
		·												1st	Add'l	Disc 1st	Disc Add'l
—				-			1	Name	curring	Nonrecurring	- Di			000	Rates(\$)		
				-			Rec					001150	001111			001441	SOMAN
		End Office Trunk Port - Shared, Per MOU		-			0.0001564	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOWAN	SOMAN	SOWAN
	Tandon	n Switching (Port Usage) (Local or Access Tandem)		1			0.0001364										
	lanuen	Tandem Switching Function Per MOU		-			0.0006757										
-		Tandem Trunk Port - Shared, Per MOU		1			0.0000737										
-	Commo	on Transport					0.0002120										
		Common Transport - Per Mile, Per MOU					0.000008										
-		Common Transport - Facilities Termination Per MOU					0.0004152										
UNRUN	DI ED P	PORT/LOOP COMBINATIONS - COST BASED RATES					0.000+102										
		ased Rates are applied where BellSouth is required by FCC an	nd/or St	tate Co	mmission rule to nr	ovide Unbun	dled Local Swi	tching or Swit	ch Ports								
		es shall apply to the Unbundled Port/Loop Combination - Cos								ed Port section	of this Rate F	yhihit					-
		fice and Tandem Switching Usage and Common Transport Us											n Port/Loor	Combinatio	18		-
		st and additional Port nonrecurring charges apply to Not Curre															t
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	J, U						J J 900 0110	20000 idei			,		I		
		ort/Loop Combination Rates	1	1		1	1				1				1		t
 		2-Wire VG Loop/Port Combo - Zone 1	1	1			12.59				1				1		1
		2-Wire VG Loop/Port Combo - Zone 2	†	2		1	14.26				1				1		1
		2-Wire VG Loop/Port Combo - Zone 3		3			21.62										
—		pop Rates	†	Ť													
		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	10.80										
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	12.47										
		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	19.83										
	2-Wire	Voice Grade Line Port Rates (Res)															
		2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.79	22.14	15.25	8.45	3.91			37.06	7.88	11.17	3.91
		2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundles res, low usage line port with Caller ID															
		(LUM)			UEPRX	UEPAP	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled Georgia basic dialing port without Caller															
		ID capability - res			UEPRX	UEPWC	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled Georgia basic dialing port for use with															
		Caller ID - res			UEPRX	UEPWQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled Georgia basic dialing port - outgoing															
		only			UEPRX	UEPWR	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled Low Usage Line Port without Caller ID															
		Capability			UEPRX	UEPRT	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	FEATU																
		All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
\coprod		NUMBER PORTABILITY															
igsquare		Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
└	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED	<u> </u>	<u> </u>		ļ											1
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -			l	l											1
\sqcup		Switch-as-is	<u> </u>	<u> </u>	UEPRX	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -	1									1	1				I
\vdash		Switch with change		1	UEPRX	USACC		2.01	0.3108					33.67	7.88		
\longrightarrow	ADDITI	ONAL NRCs	<u> </u>	1						ļ							
		2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
\sqcup	0.140'-	Activity	<u> </u>	1	UEPRX	USAS2	0.00	0.00	0.00	ļ				33.67	7.88	11.17	3.91
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
		ort/Loop Combination Rates	<u> </u>	 		 	40 =-										-
		O Wine MC Lean/Bast Comba 77 14	1	1		 	12.59 14.26			1	1		ļ		 		1
		2-Wire VG Loop/Port Combo - Zone 1)						1			i	Ì	1	ı	1
	UNE Po	2-Wire VG Loop/Port Combo - Zone 2		2		1											
	UNE Po	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3		3			21.62										
	UNE Po	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates		3	HEDDY	LIEDLY	21.62										
	UNE Po	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		3	UEPBX	UEPLX	21.62 10.80										
	UNE Po	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		1 2	UEPBX	UEPLX	21.62 10.80 12.47										
	UNE Po	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 op Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		3			21.62 10.80										
	UNE Po	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		1 2	UEPBX	UEPLX	21.62 10.80 12.47	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91

UNBUNDL	ED NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	O.M.			LIEDDY	LIEDDO		First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled port outgoing only - bus	+	1	UEPBX UEPBX	UEPBO UPEB1	1.79 1.79	22.14	15.25	8.45 8.45	3.91			33.67 33.67	7.88 7.88	11.17 11.17	3.91 3.91
	2-Wire voice unbundled incoming only port with Caller ID - Bus	+	1	UEPBX	UPEB1	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port, without Caller ID capability - bus			UEPBX	UEPWD	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port for use with Caller ID - bus			UEPBX	UEPWP	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPBX	UEPBE	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
LOC	AL NUMBER PORTABILITY					_										
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEA	TURES															
	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED									-						
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPBX	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop / Line Port Combination - Conversion Switch with change	-		UEPBX	USACC		2.01	0.3108								
ADD	ITIONAL NRCs			02. 27.	00/100		2.01	0.0100								
1.22	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2		0.00	0.00					33.67	7.88	11.17	3.91
2-WI	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			OLFBA	USASZ		0.00	0.00					33.07	7.00	11.17	3.91
	Port/Loop Combination Rates															+
OILE	2-Wire VG Loop/Port Combo - Zone 1	1	1		-	12.59										
	2-Wire VG Loop/Port Combo - Zone 2		2			14.26			İ							
	2-Wire VG Loop/Port Combo - Zone 3		3			21.62										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	10.80										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	12.47										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	19.83										
2-Wi	re Voice Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia extended dialing port, PBX 1- Way Outdial Trunk			UEPRG	UEPPO	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00					33.67	7.88	11.17	3.91
FEA	TURES															
	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPRG	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPRG	USACC		2.01	0.3108					33.67	7.88	11.17	3.91
ADD	ITIONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.91
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64					19.99	19.99	19.99	19.99
2-WI	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	1	1		1		17.04	17.07			<u> </u>	 	10.00	10.00	10.00	10.00
	Port/Loop Combination Rates	1			1				†							
	2-Wire VG Loop/Port Combo - Zone 1	1	1			12.59										
	2-Wire VG Loop/Port Combo - Zone 2		2			14.26			<u> </u>							
	2-Wire VG Loop/Port Combo - Zone 3		3			21.62										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	10.80			ļl							ļ
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	12.47			ļ							
	2-Wire Voice Grade Loop (SL 1) - Zone 3	1	3	UEPPX	UEPLX	19.83			ļ							_
2-Wi	re Voice Grade Line Port Rates (BUS - PBX)	1	1						l			l				<u> </u>

ONBONDLE	D NETWORK ELEMENTS - Georgia													ment: 2	1	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Dee	Nonrec	urring	Nonrecurring	Disconnect		•	oss	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
-	Line Side Unbundled Incoming PBX Trunk Port - Bus		-	UEPPX	UEPP1	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.79	22.14	15.25	8.45	3.91			33.67	7.88		3.9
	2-Wire Voice Unburidled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.79	22.14	15.25	8.45	3.91			37.06	7.88		
				UEPPX	UEPXA					3.91						
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports					1.79	22.14	15.25	8.45				33.67	7.88	11.17	3.9
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
	Capable Port			UEPPX	UEPXE	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Administrative Calling Port			UEPPX	UEPXL	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPPX	UEPXM	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			OLITA	OLI AW	1.70	22.17	10.20	0.40	0.01			00.01	7.00	11.17	0.01
				UEPPX	UEPXO	1.79	22.14	45.05	8.45	3.91			33.67	7.88	11.17	3.91
	Discount Room Calling Port							15.25								
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 1-Way															
	Oudial Trunk			UEPPX	UEPWS	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 2-Way															
	Trunk			UEPPX	UEPWT	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 2-way PBX															
	Trunk			UEPPX	UEPPQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX LD			OLI I X	0 <u></u>	0		10.20	0.10	0.01			00.01	7.00		0.01
				UEPPX	UEPPS	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	Terminal Ports			UEPPX	UEPPS	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX Toll					. =0										
	Terminal Ports			UEPPX	UEPPT	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															
	DDD Terminal Port			UEPPX	UEPPU	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															
	Terminal Switchboard Port			UEPPX	UEPPV	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															1
	Terminal Switchboard DDD Capable Port			UEPPX	UEPPW	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
															11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX 2-Way															0.0.
	Trunk			UEPPX	UEPPC	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
1.004	L NUMBER PORTABILITY		-	ULFFX	ULFFC	1.79	22.14	13.23	0.43	3.91			33.07	7.00	11.17	3.91
LUCA				LIEDDY	LNDCD	2.45	0.00	0.00					33.67	7.00	44.47	2.04
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00					33.67	7.88	11.17	3.91
FEAT																
	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch-As-Is			UEPPX	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															1
	Conversion - Switch with Change			UEPPX	USACC		2.01	0.3108					33.67	7.88	11.17	3.91
ADDIT	IONAL NRCs															
7,0011	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -		-		+ +				 					 	 	+
	Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.91
			1	OLI I A	JUAUL	0.00	0.00	0.00	 				55.07	7.00	11.17	3.91
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt						44.04	44.04					40.00	40.00	40.00	40.00
0 1477-	Group	<u> </u>	-		+		14.64	14.64	 				19.99	19.99	19.99	19.99
	E VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	(I			\bot											
UNE F	Port/Loop Combination Rates		<u> </u>	ļ					ļ						ļ	4
	2-Wire VG Coin Port/Loop Combo – Zone 1		1		1	12.69										1
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			14.36										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			21.72										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	10.80									1	1
	2-Wire Voice Grade Loop (SL1) - Zone 2		2		UEPLX	12.47								1	+	+

UNBUNDL	ED NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonred			g Disconnect				Rates(\$)		
				LIEBOO	LUEBLY.		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2 14/:	2-Wire Voice Grade Loop (SL1) - Zone 3 re Voice Grade Line Ports (COIN)		3	UEPCO	UEPLX	19.83								-		
2-0011	2-Wire Coin 2-Way with Operator Screening (GA)			UEPCO	UEPGC	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,			OLI GO	OLI GO	1.03	22.14	10.20	0.43	3.91			33.07	7.00	11.17	3.9
	900/976, 1+DDD (GA)			UEPCO	UEP2G	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking															
	(GA)			UEPCO	UEPGA	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Coin 2-Way with Operator Screening and 900/976															
	Blocking (GA)			UEPCO	UEPGB	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Coin 2-Way with Operator Screening and Blocking:			LIEBCO	UEPCH	1.89	22.44	45.05	8.45	3.91			33.67	7.88	11.17	2.0
	900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and 011 Blocking			UEPCO	UEPCH	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	(GA, KY, MS)			UEPCO	UEPRJ	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Coin Outward with Operator Screening and Blocking:			52. 00	CLITTO	1.00	22.14	10.20	0.40	5.91			55.67	7.00		5.5
	900/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Coin Outward Smartline with 900/976 (all states except															
	LA)			UEPCO	UEPCR	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
ADD	TIONAL UNE COIN PORT/LOOP (RC)			LIEBOO	LIBEOU	0.50										
1.00	UNE Coin Port/Loop Combo Usage (Flat Rate) AL NUMBER PORTABILITY			UEPCO	URECU	3.59	0.00	0.00					33.67	7.88	11.17	3.9
LUC	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35								-		
NON	RECURRING CHARGES - CURRENTLY COMBINED			ULFCO	LINEUX	0.33										
NON	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPCO	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch with change			UEPCO	USACC		2.01	0.31					33.67	7.88	11.17	3.91
ADD	TIONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
0.14/1	Activity	- 1 1615 1	DODT (UEPCO	USAS2		0.00	0.00					33.67	7.88	11.17	3.9
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE Port/Loop Combination Rates	LINE	PORT (KES)	+									-		
ONE	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		+	18.69								1		1
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			21.30										1
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			32.77										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	16.84										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	19.45								ļ		ļ
0.140	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	30.92										
2-WII	re Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence		1	UEPFR	UEPRL	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res		1	UEPFR	UEPRC	1.85	121.33	95.26	8.45	3.91			37.06	7.88	11.17	3.9
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundles res, low usage line port with Caller ID				1	50	56	22.20	50	2.31				1.30	· · · · · ·	1
	(LUM)		<u>L</u>	UEPFR	UEPAP	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port, without														1	
	Caller ID capability - res		<u> </u>	UEPFR	UEPWC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port for use with Caller ID - res		1	UEPFR	UEPWQ	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - outgoing		 	OLFFR	UEFWQ	1.85	121.33	95.26	8.45	3.91			33.07	7.88	11.17	3.9
	only		1	UEPFR	UEPWR	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
INTE	ROFFICE TRANSPORT						.250	00.20	5.10	3.31			55.57	50	1	0.0
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility												1			
	Termination		<u>L</u>	UEPFR	U1TV2	17.07	79.61	36.08		<u></u>			<u> </u>	<u></u>	<u> </u>	<u> </u>
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile														1	
1	or Fraction Mile		<u> </u>	UEPFR	1L5XX	0.0222										
	HIDES	1	1	1	1					l	l	Ì	l	l .	1	
FEAT	All Features Offered		1	UEPFR	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91

ONRON	ULEL	NETWORK ELEMENTS - Georgia			T							I	·		ment: 2		bit: C
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
N		CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFR	USAC2		93.83	93.83					33.67	7.88	11.17	3.9
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
		Combination - Conversion - Switch-With-Change			UEPFR	USACC		93.83	93.83					33.67	7.88		
		VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	E LINE I	PORT (BUS)												
U		rt/Loop Combination Rates															
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.69										
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			21.30										
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			32.77										
U		op Rates									-						
		2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	16.84										
		2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	19.45										
		2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	30.92										
2-	-Wire \	/oice Grade Line Port (Bus)															
		2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
		2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
		2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
		2-Wire voice unbundled Georgia basic dialing port, without															1
		Caller ID capability - bus			UEPFB	UEPWD	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled Georgia basic dialing port for use with															
		Caller ID - bus			UEPFB	UEPWP	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
L	OCAL	NUMBER PORTABILITY															
		Local Number Portability (1 per port)			UEPFB	LNPCX	0.35										
IN		FFICE TRANSPORT															
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
		Termination			UEPFB	U1TV2	17.07	79.61	36.08								
	1	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
		or Fraction Mile			UEPFB	1L5XX	0.0222										
F	EATUR																
		All Features Offered			UEPFB	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.9
N		CURRING CHARGES (NRCs) - CURRENTLY COMBINED												-			
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
		Combination - Conversion - Switch-as-is			UEPFB	USAC2		93.83	93.83					33.67	7.88	11.17	3.9
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			02.1.5	00/102		00.00	00.00					00.01	7.00		0.0
		Combination - Conversion - Switch with change			UEPFB	USACC		93.83	93.83								
2-		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
		rt/Loop Combination Rates	†	t	1		İ							1	1		
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1	†	1	1	1	18.69							t	1		
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	†	2	1		21.30							1	1		
-+		2-Wire VG Loop/IO Tranport/Port Combo - Zone 3	 	3		1 1	32.77			1		1	l	 			
11		op Rates	 	۲		+ +	02.77					1	l	 			
		2-Wire Voice Grade Loop (SL2) - Zone 1	1	1	UEPFP	UECF2	16.84							-			†
		2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2	 	2	UEPFP	UECF2	19.45					1	l	 			
		2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3	 	3	UEPFP	UECF2	30.92			1		1	l	 			
2.		/oice Grade Line Port Rates (BUS - PBX)	 	۲		02012	00.02					1	l	 			
			 	 		+ +						1	l	 			
		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
		Line Side Unbundled Outward PBX Trunk Port - Bus	1		UEPFP	UEPPO	1.85	121.33	95.26	8.45	3.91	İ		33.67	7.88	11.17	
		Line Side Unbundled Incoming PBX Trunk Port - Bus	†	t	UEPFP	UEPP1	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
		2-Wire Voice Unbundled PBX LD Terminal Ports		1	UEPFP	UEPLD	1.85	121.33	95.26	8.45	3.91	1		33.67	7.88	11.17	3.9
		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port		t	UEPFP	UEPXA	1.85	121.33	95.26	8.45	3.91			37.06	7.88	11.17	
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports		t	UEPFP	UEPXB	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	
-+		2-Wire Voice Unbundled PBX LD DDD Terminal Ploter Fors		t	UEPFP	UEPXC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
-+		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port		t	UEPFP	UEPXD	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.9
- t		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD		1		132.7.2	50	.200	00.20	0.40	3.01	1		33.07		,	3.5
		Capable Port	1		UEPFP	UEPXE	1.85	121.33	95.26	8.45	3.91		1	33.67	7.88	11.17	3.91

RATE ELEMENTS	UNBUNDLI	ED NETWORK ELEMENTS - Georgia			1									T -		ment: 2		bit: C
2 2000 200	CATEGORY	RATE ELEMENTS		Zone	BCS		USOC						Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
Description Print								Rec										
Commission Colling Part Commission Colling Part Colling Part								1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Room Calling Part Section Sect		Administrative Calling Port			UEPFP	UE	EPXL	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
Decount Room Calling Part USPPP USPPS 180 172:33 65.08 8.45 3.91 33.07 7.88 11.17		Room Calling Port			UEPFP	UE	EPXM	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
2-Wine Views Linchmode (1-Wing Culgaring PMX Materianse Port UIPPP UIPPS 1.86 121.33 66.26 8.46 3.01 33.67 7.86 11.17																		l
2-Vive vace unbundled Georgia basis dialing post - V-Vay UEPPP UEPVT 1.65 121.33 66.26 6.45 3.91 33.67 7.86 11.17	-																	3.91
Oudst Trusk ULPPP ULPWS 1.85 121.33 56.28 6.45 391 3367 7.86 11.17					UEPFP	UE	EPXS	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
Trusk		Oudial Trunk			UEPFP	UE	EPWS	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
Local Number Potalbility (1 per port)		Trunk			UEPFP	UE	EPWT	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
Network Tempor Delicated - 2 Wire Voce Grade - Facility UEPFP UTV_2 17.07 78.61 36.08 UEPFP UTV_2 17.07 78.61 36.08 UEPFP UTV_2 17.07 78.61 36.08 UEPFP UTV_2 17.07 78.61 36.08 UEPFP UTV_2 17.07 78.61 36.08 UEPFP UTV_2 17.07 78.61 UEPFP UTV_2 17.07 78.61 UEPFP UTV_2 17.07 78.61 UEPFP UTV_2 UTV_	LOCA				LIEDED		IDOD	0.45	0.00	0.00					00.07	7.00	44.47	0.04
Interedifical Transport - Decicated - 2 Wire Votoc Grade - Facility Termination UEPPP UTIV2 17.07 79.61 36.08	INITE		 	-	UEPFP	LN	NPCP	3.15	0.00	0.00	1		1		33.67	7.88	11.17	3.91
Intendifical Transport - Dedicated - 2 Wire Voice Grade - Per Mile UEPPP	INTER	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility			LIEDED		4T) /O	47.07	70.04	20.00								
Features		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile							79.61	36.08								
All Features Offended UEPFP UEPF UEPF 0.00 0.0	FEAT				UEPFP	1L	_5XX	0.0222										
NONRECURRING CHARGES (NINCs) = CURRENTLY COMBINED	FEAT			<u> </u>	LIEDED	1.15	ED\/E	0.00	0.00	0.00					22.67	7.00	11 17	3.91
2-Wire Loop / Dedicated 10 Transport / 2 Wire Line Port UEPPP USAC2 93.83 93.83 33.67 7.88 11.17	NONE				UEFFF	UE	EFVF	0.00	0.00	0.00					33.67	7.00	11.17	3.91
Combination - Conversion - Switch-as-8 UEPPP USACC 93.83 93.83 93.83 33.67 7.88 11.17 11.1	I I I I I I I I I I I I I I I I I I I																	
2 2 2 2 2 2 3 3 3 3					UEPFP	us	SAC2		93.83	93.83					33.67	7.88	11.17	3.91
UNBUNDLED FORTIL.ODP COMBINATIONS - COST BASED RATES		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port																3.91
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT	UNBUNDI ED				OLFIF	00	SACC		93.03	33.03					33.07	7.00	11.17	3.91
NE POPULOGO Combination Rates			PORT		İ													
2-Wire VSL Loop/2-Wire DID Trunk Port Combo - UNE Zone 2																		
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3 3 42.27		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1				28.19										
WIK Loop Rates																		
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1				3				42.27										
2-Wire Analog Voice Grade Loop - (St2) - UNE Zone 2 2 UEPPX UED1 19.45 104.17 78.10	UNE I			<u></u>														
2-Wire Analog Voice Grade Loop - (\$L2) - UNE Zone 3 3 UEPPX UECD1 30.92 104.17 104.10																		
UNE Port Rate	-																	—
Exchange Ports - 2-Wire DID Port UEPPX UEPD1 11.35 61.91 61.91 33.67 7.88	LINE			3	UEPPX	UE	ECDI	30.92	104.17	104.10			-					
NONRECURRING CHARGES - CURRENTLY COMBINED	UNL				LIEPPX	LIE	FPD1	11 35	61 91	61 91					33.67	7.88		-
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - UEPPX	NONE				02				01.01	01.01					00.07	7.00		
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion UEPPX		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -			LIEPPX	LIS	SAC1		93.38	93.38					33.67	7.88		
ADDITIONAL NRCs		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion																
Telephone Number/Trunk Group Establisment Charges	ADDI			 	OLI I A	03	0,710		30.00	30.00	1		-		33.07	7.00		-
DID Trunk Termination (One Per Port)			1		I						1		<u> </u>	 		1	1	—
DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers UEPPX NDZ 0.00					UEPPX	NE	DT	0.00	0.00	0.00								
Additional DID Numbers for each Group of 20 DID Numbers UEPPX ND4 0.00 0.00 0.00 0.00		DID Numbers, Establish Trunk Group and Provide First Group				NE	DZ	0.00	0.00	0.00								
DID Numbers, Non- consecutive DID Numbers Per Number UEPPX ND5 0.00 0.00 0.00 0.00																		
Reserve DID Numbers		DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX													
LOCAL NUMBER PORTABILITY UEPPX LNPCP 3.15 0.00 0.00 UEPX LNPCP 3.15 0.00 UEPX LNPCP 3.15						NE	D6	0.00	0.00									
Local Number Portability (1 per port)					UEPPX	NE	DV	0.00	0.00	0.00								L
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT UNE Port/Loop Combination Rates WISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1	LOCA				L													
UNE Port/Loop Combination Rates ZW ISDN Digital Grade Loop/ZW ISDN Digital Line Side Port - UNE Zone 1	<u> </u>					LN	NPCP	3.15	0.00	0.00								1
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - 1 UEPPB UEPPR 35.36			NE SIDE	PORT	1						1				-	-	-	
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	UNE	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -			l													
UNE Zone 2		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		1														

UNBUND	DLED NETWORK ELEMENTS - Georgia														ment: 2		oit: C
CATEGORY	RY RATE ELEMENTS	Interi m	Zone	В	cs	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 3		3	UEPPB	UEPPR		53.64										
UNI	IE Loop Rates																
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	21.89	252.32	188.77					19.99	19.99		
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	25.27	252.32	188.77					19.99	19.99		
	2-Wire ISDN Digital Grade Loop - UNE Zone 2 2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X USL2X	40.17	252.32	188.77	+				19.99	19.99		ļ
LINE	NE Port Rate		3	OLFFB	ULFFR	USLZA	40.17	232.32	100.77					15.55	19.99		
ON	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	13.47	47.37	47.37					19.99	19.99		
NO	DNRECURRING CHARGES - CURRENTLY COMBINED			OLITO	OLITIK	OLITB	10.47	47.57	47.57	+				13.33	13.33		
1401	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port	1	1							+							
	Combination - Conversion	1		UEPPB	UEPPR	USACB	0.00	93.38	93.38					19.99	19.99	1	1
ADI	DDITIONAL NRCs	1	<u> </u>				0.00	55.56	33.30	 					.5.55	1	
	2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy																
	Non Feature/Add Trunk			UEPPB	UEPPR	USASB		165.95						19.99	19.99		
LOC	OCAL NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-C	CHANNEL USER PROFILE ACCESS:																
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
	CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C,MS, 8	k TN)														
USE	SER TERMINAL PROFILE																
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VEF	RTICAL FEATURES																
	All Vertical Features - One per Channel B User Profile	1		UEPPB	UEPPR	UEPVF	0.00	0.00	0.00					19.99	19.99		
INI	TEROFFICE CHANNEL MILEAGE																
	Interoffice Channel mileage each, including first mile and facilities termination			UEPPB	UEPPR	M1GNC	16.47	79.61	36.08					19.99	19.99		
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.0222	0.00	0.00	+			0.00	19.99	19.99		
4-10	WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNI	K DODT		UEPPB	UEPPK	IVITGINIVI	0.0222	0.00	0.00				0.00				
	NE Port/Loop Combination Rates	KFOKI								+							
0.41	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE									+							
	Zone 1		1	UEPPP			218.69										
+	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			02			210.00										
	Zone 2		2	UEPPP			227.29										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		_	02			227.20										
	Zone 3		3	UEPPP			265.09										
UNE	NE Loop Rates	1				1											
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	55.53	448.92	276.60					19.99	19.99		
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	64.13	448.92	276.60					19.99	19.99		
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	101.93	448.92	276.60					19.99	19.99		
UNI	IE Port Rate																
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	163.16	186.80	186.80					19.99	19.99		
NOI	DNRECURRING CHARGES - CURRENTLY COMBINED	ļ															
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port	1				110465											
	Combination - Conversion -Switch-as-is	 	<u> </u>	UEPPP		USACP	0.00	269.96	269.96	 				19.99	19.99	 	
ADI	DDITIONAL NRCs	1	<u> </u>	1						 					1	 	
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-	1		UEPPP		DDZTE		0.0600							1	1	1
	Inward/two way Tel Nos. (except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -	1	!	UEPPP		PR7TF		0.9686		 					-	-	-
	Outward Tel Numbers (All States except NC)	1		UEPPP		PR7TO		22.75	22.75						1	1	1
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -	+	<u> </u>	UEPPP		FR/IU		22.75	22.75	+					-	-	
	Subsequent Inward Tel Numbers	1		UEPPP		PR7ZT		45.49	45.49								
LOC	CAL NUMBER PORTABILITY	1	 	J-111		. 11/21		70.73	45.45	+					 	 	
	Local Number Portability (1 per port)	1		UEPPP		LNPCN	1.75			 					 	 	
INT	TERFACE (Provsioning Only)	1	<u> </u>	J 1 1			1.75								1	1	
	Voice/Data	1	t	UEPPP		PR71V	0.00	0.00	0.00	† †					1	1	
	Digital Data	1		UEPPP		PR71D	0.00	0.00	0.00	 		-					

UNBUNDL	ED NETWORK ELEMENTS - Georgia											,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
					-	1	Nonrec	urring	Nonrecurring	Disconnect			088	Rates(\$)		
			1			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00		Addi	JOINEO	JOHAN	JONAN	JONAN	JOHIAN	JONAN
New	or Additional "B" Channel															
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	28.71						19.99	19.99		
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	28.71						19.99	19.99		
	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	28.71						19.99	19.99		
CALI	_ TYPES															
	Inward			UEPPP	PR7C1	0.00	0.00	0.00								
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way	ļ	1	UEPPP	PR7CC	0.00	0.00	0.00								
Inter	office Channel Mileage			LIEDDD	41.514.6	70 0000	4.47.07	111.75	0.00				10.00	40.00		
	Fixed Each Including First Mile Each Airline-Fractional Additional Mile	1	1	UEPPP UEPPP	1LN1A 1LN1B	78.9223	147.07	111.75	0.00		 		19.99	19.99	 	-
A_\A/11	Each Airline-Fractional Additional Mile RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT	1	1	UEPPP	ILINIB	0.4523			 		 				 	
	Port/Loop Combination Rates	1	1	+	+ -				+		 			1	 	-
ONL	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		176.33										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	1	2	UEPDC	_	184.93			1					 	I	<u> </u>
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3	1	3	UEPDC	1	222.73			1					1	1	
UNE	Loop Rates		Ť						†						1	
	4-Wire DS1 Digital Loop - UNE Zone 1	1	1	UEPDC	USLDC	55.53	448.92	276.00	1				19.99	19.99	1	
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	64.13	448.92	276.60					19.99	19.99		
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	101.93	448.92	276.60					19.99	19.99		
UNE	Port Rate															
	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	120.80	89.44	52.46					19.99	19.99		
NON	RECURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Switch-as-is	ļ	1	UEPDC	USAC4		269.96	269.96					19.99	19.99		
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination			LIEDDO			000.00	200.00					40.00	40.00		
	Conversion with DS1 Changes Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination			UEPDC	USAWA		269.96	269.96					19.99	19.99		
	- Conversion with Change - Trunk			LIEDDO	USAWB		200.00	200.00					19.99	40.00		
ADDI	TIONAL NRCs	-	-	UEPDC	USAWB		269.96	269.96					19.99	19.99		
ADDI	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent		1													
	Service Activity Per Service Order			UEPDC	USAS4		147.47	147.47								
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -			OLI DO	00/104		147.47	147.47								
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		28.71	28.71					19.99	19.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		28.71	28.71					19.99	19.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		28.71	28.71					19.99	19.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		28.71	28.71					19.99	19.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		28.71	28.71					19.99	19.99		
BIPO	LAR 8 ZERO SUBSTITUTION	ļ	1													
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	600.00								
Alton	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	600.00								
Alter	nate Mark Inversion AMI -Superframe Format	1	1	UEPDC	MCOSF		0.00	0.00	 		1				-	
	AMI - Extended SuperFrame Format		1	UEPDC	MCOPO		0.00	0.00	1		1			1	 	
Teler	phone Number/Trunk Group Establisment Charges	1		02.100	7110010		0.00	0.00	 		 			 	t	
1 6161	Telephone Number for 2-Way Trunk Group	1	1	UEPDC	UDTGX	0.00			1					 	I	<u> </u>
	Telephone Number for 1-Way Outward Trunk Group	1		UEPDC	UDTGY	0.00								1	1	
	Telephone Number for 1-Way Inward Trunk Group Without DID	1		UEPDC	UDTGZ	0.00								1	1	
	DID Numbers, Establish Trunk Group and Provide First Group		1		1	2.20			1					İ	1	
	of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00						1	I	
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00										
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00										
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00								

SHOUNDE	ED NETWORK ELEMENTS - Georgia												Attach	ment: 2	Evhil	oit: C
	LD NETWORK ELEMENTO - Georgia										Svc Order	Svc Order	Incremental		Incremental	Incrementa
		1									Submitted	Submitted		Charge -	Charge -	Charge -
		1_									Elec	Manually		Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)				-				
CATEGORI	KATE ELEMENTO	m	20116	Воо	0000			NATEO(ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
	-						Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		l .
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Dedic	cated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	1 Digita	Loop	with 4-Wire DDITS 1	runk Port											
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	78.47	147.07	111.75					19.99	19.99		
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities															
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25															
	miles			UEPDC	1LNOB	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
					1											
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.4523	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15										
	Central Office Termininating Point			UEPDC	CTG	0.00										
	RE DS1 LOOP WITH CHANNELIZATION WITH PORT															
	em is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti															
	System can have up to 24 combinations of rates depending on	type ar	nd num	ber of ports used												
UNE	DS1 Loop		L .													
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG UEPMG	USLDC	55.53	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2	-	2			64.13	0.00	0.00								
LINE	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	101.93	0.00	0.00								
UNE	DSO Channelization Capacities (D4 Channel Bank Configuration 24 DSO Channel Capacity - 1 per DS1	ns)		UEPMG	VUM24	102.64	0.00	0.00					19.99	19.99		
	48 DSO Channel Capacity - 1 per DS1			UEPMG	VUM48	205.28	0.00	0.00					19.99	19.99		
+-	96 DSO Channel Capacity -1 per 4 DS1s			UEPMG	VUM96	410.56	0.00	0.00			1		19.99	19.99		
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	615.84	0.00	0.00					19.99	19.99		
	192 DS0 Channel Capacity - 1 per 8 DS1s			UEPMG	VUM19	821.12	0.00	0.00					19.99	19.99		
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,026.40	0.00	0.00					19.99	19.99		
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,231.68	0.00	0.00					19.99	19.99		
+-	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,642.24	0.00	0.00					19.99	19.99		
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,052.80	0.00	0.00					19.99	19.99		
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,463.36	0.00	0.00					19.99	19.99		
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,873.92	0.00	0.00					19.99	19.99		
Non-I	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chani	neliztio	n with Port - Conve	rsion Charge	Based on a Sy	stem									
	nimum System configuration is One (1) DS1, One (1) D4 Channe															
	ples of this configuration functioning as one are considered Ac															
	NRC - Conversion (Currently Combined) with or without															
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	328.35	16.52					19.99	19.99		
	em Additions at End User Locations Where 4-Wire DS1 Loop wit				ination Curre	ntly Exists and	· ·									
New ((Not Currently Combined) in all states, except in Density Zone 1	of Top	8 MSA	's												
	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port	1														
<u> _</u>	and Assoc Fea Activation			UEPMG	VUMD4	0.00	738.61	462.53	144.05	17.09			19.99	19.99		
Bipol	lar 8 Zero Substitution															
	Clear Channel Capability Format, superframe - Subsequent	1		LIEBNO	00005									1		
	Activity Only	 		UEPMG	CCOSF	0.00	0.00	600.00	ļ	-			1	 		
	Clear Channel Capability Format - Extended Superframe -	1		LIEDMC	CCOFF	0.00	0.00	000.00						1		
A14.0	Subsequent Activity Only nate Mark Inversion (AMI)	 		UEPMG	CCOEF	0.00	0.00	600.00		-	-		-	 		
Aitern	Superframe Format	├		UEPMG	MCOSF	0.00	0.00	0.00						-		
-+-	Extended Superframe Format	 		UEPMG	MCOSF	0.00	0.00	0.00	1		-		1	1		-
Evch	ange Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port	OLFIVIG	IVICOPO	0.00	0.00	0.00	 				1	1		
	ange Ports Associated with 4-wire DST Loop with Chamienzand	OII WILLI	. 0/1		1				 				1	1		
LACITO		 			+									 		-
	Line Side Combination Channelized PBX Trunk Port - Business	1		UEPPX	UEPCX	1.79	0.00	0.00	0.00	0.00			33.67	7.88		1
-+	Line Side Outward Channelized PBX Trunk Port - Business		1	UEPPX	UEPOX	1.79	0.00	0.00	0.00	0.00			33.67	7.88		
		<u> </u>					3.00	3.00	5.00	3.00			55.07			
			1	i	•				1	1	1		1		1	l
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.79	0.00	0.00	0.00	0.00			33.67	7.88		

UNBUNDI	LED NETWORK ELEMENTS - Georgia												Attachr	nent: 2	Exhi	bit: C
CATEGORY		Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Feat	ture Activations - Unbundled Loop Concentration														├	
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank			UEPPX	1PQWM	0.62	25.09	13.25	3.99	3.97			33.67	7.88		
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			HEDDY	45004/11	0.00	77.04	40.00	50.40	44.04			00.07	7.00	ĺ	
Tolo	ephone Number/ Group Establishment Charges for DID Service			UEPPX	1PQWU	0.62	77.21	18.20	56.49	11.04			33.67	7.88	 	1
1010	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00							—	
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00								
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
Loc	cal Number Portability			LIEBBY	Lungs											ļ
	Local Number Portability - 1 per port		ļ	UEPPX	LNPCP	3.15	0.00	0.00	ļ						├	4
	ATURES - Vertical and Optional		<u> </u>		ļ				1							
Loc	cal Switching Features Offered with Line Side Ports Only All Features Available			UEPPX	UEPVF	0.00	0.00	0.00								
IINDIINDI E	ED PORT LOOP COMBINATIONS - MARKET RATES		<u> </u>	UEPPX	UEPVF	0.00	0.00	0.00								
	rket Rates shall apply where BellSouth is not required to provide	unhung	dled lo	al switching or swi	tch norte ner	FCC and/or St	ata Commissio	n rulee								+
	s includes:	unbunc	l lea lo	ar switching or swi	l	l CC and/or of	ate Commissio	ni ruies.							 	+
	bundled port/loop combinations that are Currently Combined or N	Not Cur	rently (Combined in Zone 1	of the Top 8	MSAS in BellS	outh's region	for end users	with 4 or more	DS0 equivaler	t lines.					
	Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderda											e).				
	South currently is developing the billing capability to mechanica												. In the interi	m where Bells	South cannot	bill Market
	es, BellSouth shall bill the rates in the Cost-Based section preced															
	Market Rate for unbundled ports includes all available features in														L	
	d Office and Tandem Switching Usage and Common Transport Us OC: URECU).	age rat	es in th	e Port section of th	is rate exhib	it shall apply to	all combination	ons of loop/po	ort network elen	nents except	for UNE Coi	n Port/Loop	Combination	s which have	a flat rate us	sage charge
For	Not Currently Combined scenarios the Nonrecurring charges are	listed	in the F	irst and Additional	NRC column	s for each Port	USOC. For Co	urrently Comb	ined scenarios,	, the Nonrecur	ring charge	s are listed	in the NRC - C	Currently Com	bined sectio	n.
	ditional NRCs may apply also and are categorized accordingly.															
2-W	/IRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
UNE	E Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			24.80										
	2-Wire VG Loop/Port Combo - Zone 2		2			26.47										
	2-Wire VG Loop/Port Combo - Zone 3		3			33.83									├	
UNE	E Loop Rates			HEDDY	UEPLX	40.00										
	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		1	UEPRX UEPRX	UEPLX	10.80 12.47										<u> </u>
	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	19.83										
2-W	/ire Voice Grade Line Port (Res)	-	-	OLI AX	OLI LX	13.03			1		 					
- "	2-Wire voice unbundled port - residence			UEPRX	UEPRL	14.00	90.00	90.00	t		1	 	33.67	7.88	11.17	3.91
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	14.00	90.00	90.00					33.67	7.88	11.17	
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port without Caller															
-	ID capability - res 2-Wire voice unbundled Georgia basic dialing port for use with			UEPRX	UEPWC	14.00	90.00	90.00			-		33.67	7.88	11.17	3.91
	Caller ID - res 2-Wire voice unbundled Georgia basic dialing port - outgoing		-	UEPRX	UEPWQ	14.00	90.00	90.00			1		33.67	7.88	11.17	3.91
	only			UEPRX	UEPWR	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX	UEPRT	14.00	90.00	90.00					33.67	7.88	11.17	3.91
LOC	CAL NUMBER PORTABILITY				ļ						ļ					1
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35			ļ							ļ
FEA	ATURES		<u> </u>	HEDDY	LIED) (E	0.00	0.00	0.00	 		1		20.0-	7.00	11.7-	1
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00	 		1		33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPRX	USAC2		41.50	41.50					33.67	7.88	11.17	3.91

Version 3Q02: 09/06/02 Page 115 of 362

UNBUNDL'	ED NETWORK ELEMENTS - Georgia												Attach	ment: 2	Exhi	bit: C
											Svc Order	Svc Order	Incremental		Incremental	
											Submitted			Charge -	Charge -	Charge -
		Intori									Elec	Manually	Manual Svc	Manual Svc		
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													151	Add I	Disc 1st	DISC Add I
						Dee	Nonrec	urring	Nonrecurring	g Disconnect		•	oss	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop / Line Port Combination - Switch with															
	change			UEPRX	USACC		41.50	41.50					33.67	7.88	11.17	3.91
ADDI	TIONAL NRCs															
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -															
	Subsequent			UEPRX	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.91
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			24.80										
	2-Wire VG Loop/Port Combo - Zone 2		2			26.47										
\vdash	2-Wire VG Loop/Port Combo - Zone 3	ļ	3			33.83			.					.	.	↓
UNE	oop Rates				lues::											
\vdash	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	10.80										
\vdash	2-Wire Voice Grade Loop (SL1) - Zone 2	<u> </u>	2	UEPBX	UEPLX	12.47			_	 	<u> </u>		ļ	-	-	
	2-Wire Voice Grade Loop (SL1) - Zone 3	<u> </u>	3	UEPBX	UEPLX	19.83			-					-	-	↓
2-Wir	Voice Grade Line Port (Bus)	 	-	LIEDDY	LIEDE	11.00	00.00	20.00	1	-	1	-	20.00	7.00	44.4-	
\vdash	2-Wire voice unbundled port without Caller ID - bus	 		UEPBX UEPBX	UEPBL UEPBC	14.00 14.00	90.00 90.00	90.00	 	 	1	-	33.67 33.67	7.88 7.88	11.17 11.17	
	2-Wire voice unbundled port with Caller + E484 ID - bus	-		UEPBX	UEPBO	14.00	90.00						33.67	7.88	11.17	
\vdash	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBU	14.00	90.00	90.00	-		1		33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port, without			UEPBX	UEPWD	14.00	90.00	90.00					33.67	7.88	11.17	3.91
—	Caller ID capability - bus 2-Wire voice unbundled Incoming Only Port without Caller ID			UEPBX	UEPWD	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	Capability			UEPBX	UEPBE	14.00	90.00	90.00					33.67	7.88	11.17	3.91
———	2-Wire Voice Unbundled Alabama Business Dialing Plan without			UEPBA	UEPBE	14.00	90.00	90.00					33.67	7.00	11.17	3.91
	Caller ID			UEPBX	UEPWB	14.00	90.00	90.00					33.67	7.88	11.17	3.91
-	2-Wire voice unbundled Georgia basic dialing port for use with			OLI DX	OLI WB	14.00	30.00	30.00	1		1		33.07	7.00	11.17	3.31
	Caller ID - bus			UEPBX	UEPWP	14.00	90.00	90.00					33.67	7.88	11.17	3.91
I OC/	L NUMBER PORTABILITY			OLI DX	OLI WI	14.00	50.00	50.00					00.07	7.00		0.01
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEAT																
	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NONE	ECURRING CHARGES - CURRENTLY COMBINED															1
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPBX	USAC2		41.50	41.50					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop / Line Port Combination - Switch with															
	change			UEPBX	USACC		41.50	41.50					33.67	7.88	11.17	3.91
ADDI	TIONAL NRCs															
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -							·							1	
	Subsequent			UEPBX	USAS2		0.00	0.00					33.67	7.88	11.17	3.91
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)								1					1	1	<u> </u>
UNE	Port/Loop Combination Rates	ļ	L .								ļ					↓
\vdash	2-Wire VG Loop/Port Combo - Zone 1	<u> </u>	1		+	24.80			_	 	<u> </u>		ļ	-	-	
\vdash	2-Wire VG Loop/Port Combo - Zone 2	<u> </u>	2		+	26.47			_	 	<u> </u>		ļ	-	-	
1000	2-Wire VG Loop/Port Combo - Zone 3	<u> </u>	3		+	33.83			-					-	-	↓
UNE	Loop Rates	 	1	LIEDDO	LIEDLY	40.00			1	-	1	-		1	1	
+-+-	2-Wire Voice Grade Loop (SL1) - Zone 1			UEPRG	UEPLX	10.80 12.47			 					 	 	
\vdash	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	 	3	UEPRG UEPRG	UEPLX UEPLX	19.83			-	-			-	-	-	
2-///:*	e Voice Grade Line Port Rates (RES - PBX)	 	3	OLFING	OLFLA	18.03			+	1	}	-	1	 	 	+
2-4411	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -	 	1		+				 	1	1	 	1	 	 	
	Res			UEPRG	UEPRD	14.00	90.00	90.00	1				33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia extended dialing port, PBX 1-		1	52. NO	CLIND	14.00	55.00	33.00	<u> </u>				55.07	7.00		5.51
	Way Outdial Trunk	1	1	UEPRG	UEPPO	14.00	90.00	90.00	I				33.67	7.88	11.17	3.91
	2-Wire voice unbundled Low Usage Line Port without Caller ID		1		1		55.50	55.50	t	1	1		55.57		· · · · · · ·	5.51
1 1	Capability	1	1	UEPRX	UEPRT	14.00	90.00	90.00	I				33.67	7.88	11.17	3.91
LOC/	L NUMBER PORTABILITY	l			1				İ	İ				1	1	
1 1	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00	1	İ			İ	1	1	1
FEAT	URES	1									İ					1
	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
MONIT	ECURRING CHARGES - CURRENTLY COMBINED						-									1

2- GI ADDITION 2 SI PI 2-WIRE V UNE Port	RATE ELEMENTS Wire Voice Grade Loop/ Line Port Combination - Switch-As-IsWire Voice Grade Loop/ Line Port Combination - Switch with change NAL NRCs	Interi m	Zone	BCS UEPRG	USOC	Rec	Nonrec	RATES(\$)	Name of the last		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
2- GI ADDITION 2 SI PI 2-WIRE V UNE Port	-Wire Voice Grade Loop/ Line Port Combination - Switch with hange NAL NRCs Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity - Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt			UEPRG		Rec		urring	Na							1
2- GI ADDITION 2 SI PI 2-WIRE V UNE Port	-Wire Voice Grade Loop/ Line Port Combination - Switch with hange NAL NRCs Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity - Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt			UEPRG			First	Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
ADDITION 2 SI PI G 2-WIRE V UNE Ports	Change NAL NRCs Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity - Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group				USAC2		41.50	41.50					33.67	7.88	11.17	3.91
ADDITION 2 SI PI G 2-WIRE V UNE Ports	NAL NRCs Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity - Nonrecurring BX Subsequent Activity - Change/Rearrange Multiline Hunt Froup			UEPRG	USACC		41.50	41.50					33.67	7.88	11.17	3.91
2 Si Pi G 2-WIRE V UNE Port	Wire Loop/Line Side Port Combination - Non feature - Jubsequent Activity - Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt Froup			UEPRG	USACC		41.50	41.50					33.07	1.00	11.17	3.91
SI PI G 2-WIRE V UNE Port	Subsequent Activity- Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group															
2-WIRE V UNE Port	Group						0.00	0.00					33.67	7.88	11.17	3.91
UNE Port	ANCE COARE LOOP WITH 2-WIDE LINE DODT (DUS - DDV)						14.64	14.64					19.99	19.99	19.99	19.99
	OICE GRADE LOOF WITH 2-WIRE LINE FORT (BO3 - FBX)															
	t/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			24.80										
	-Wire VG Loop/Port Combo - Zone 2		2			26.47										
	-Wire VG Loop/Port Combo - Zone 3		3			33.83										
UNE Loo																<u> </u>
	-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	10.80										L
	-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	12.47										<u> </u>
	-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	19.83										L
2-Wire Vo	oice Grade Line Port Rates (BUS - PBX)															
	ine Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	ine Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	ine Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	14.00	90.00	90.00					33.67	7.88	11.17	3.91
C	P-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPPX	UEPXE	14.00	90.00	90.00					33.67	7.88	11.17	3.91
Ad	t-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPPX	UEPXL	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port -Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			UEPPX	UEPXM	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00					33.67	7.88	11.17	3.91
2-	-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	l-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk			UEPPX	UEPWS	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	-Wire voice unbundled Georgia basic dialing port - 2-Way			UEPPX	UEPWT	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	r-Wire voice unbundled Georgia basic dialing port - 2-way PBX			UEPPX	UEPPQ	14.00	90.00	90.00					33.67	7.88	11.17	3.91
2-	-Wire voice unbundled Georgia basic dialing port - PBX LD			UEPPX	UEPPS	14.00	90.00	90.00					33.67	7.88	11.17	3.91
2-	-Wire voice unbundled Georgia basic dialing port - PBX Toll															
	erminal Ports -Wire voice unbundled Georgia basic dialing port - PBX LD			UEPPX	UEPPT	14.00	90.00	90.00					33.67	7.88	11.17	3.91
D	DDD Terminal Port -Wire voice unbundled Georgia basic dialing port - PBX LD			UEPPX	UEPPU	14.00	90.00	90.00					33.67	7.88	11.17	3.91
Te	-Wile voice unbundled Georgia basic dialing port - PBX LD erminal Switchboard Port Wire voice unbundled Georgia basic dialing port - PBX LD			UEPPX	UEPPV	14.00	90.00	90.00					33.67	7.88	11.17	3.91
Te	erminal Switchboard DDD Capable Port			UEPPX	UEPPW	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	NUMBER PORTABILITY Ocal Number Portability (1 per port)	-	1	UEPPX	LNPCP	0.45	0.00	0.00								
FEATURE		├	 	UEPPA	LINPCP	3.15	0.00	0.00	 							₩
	NI Features Offered	 	1	UEPPX	UEPVF	0.00	0.00	0.00	-				33.67	7.88	11.17	3.91
	CURRING CHARGES - CURRENTLY COMBINED	-	-	ULPFA	UEFVF	0.00	0.00	0.00					33.67	1.88	11.17	3.91

UNRI	JNDLF	D NETWORK ELEMENTS - Georgia												Attach	ment: 2	Fyhi	oit: C
5,450												Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATE	GORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
1			""									1	1	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
-	1		1				1	Nonrec	rrina	Nonrecurring	Dissennest			000	Rates(\$)		
-	1		1			+	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
								FIISL	Auu i	Filat	Auu i	SOWIEC	SOWAN	JOWAN	JOWAN	SOWAN	JOWAN
		2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPPX	USAC2		41.50	41.50					33.67	7.88	11.17	3.91
	1	2-Wire Voice Grade Loop/ Line Port Combination - Switch with	1		OLI I X	CONCE		41.00	41.00					00.01	7.00	11.17	0.01
		Change			UEPPX	USACC		41.50	41.50					33.67	7.88	11.17	3.91
	ADDIT	IONAL NRCs															
		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.91
		2 Wire Loop/Line Side Port Combination - Non feature -															
		Subsequent Activity- Nonrecurring						0.00	0.00					33.67	7.88	11.17	3.91
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
	0.1000	Group	<u> </u>			+		14.64	14.64					19.99	19.99	19.99	19.99
<u> </u>		E VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	KI			+								 	1	 	-
—	UNE P	ort/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1	 	1		+	24.80			<u> </u>					 		
-	-	2-Wire VG Coin Port/Loop Combo – Zone 2	1	2		-	26.47								-		
	+	2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3	 	3		+	33.83							 	t	 	
	UNF I	pop Rates	1	5		+	55.65								-		
 		2-Wire Voice Grade Loop (SL1) - Zone 1	1	1	UEPCO	UEPLX	10.80							 	I	 	
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	12.47								1		
		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	19.83										
	2-Wire	Voice Grade Line Port Rates (Coin)															
		2-Wire Coin 2-Way with Operator Screening (GA)			UEPCO	UEPGC	14.00	90.00	90.00					33.67	7.88	11.17	3.91
		2-Wire Coin 2-Way with Operator Screening and Blocking: 011,															
		900/976, 1+DDD (GA)			UEPCO	UEP2G	14.00	90.00	90.00					33.67	7.88	11.17	3.91
		2-Wire Coin 2-Way with Operator Screening and 011 Blocking															
		(GA)			UEPCO	UEPGA	14.00	90.00	90.00					33.67	7.88	11.17	3.91
		2-Wire Coin 2-Way with Operator Screening and 900/976			LIEDOO	UEPGB	44.00	90.00	00.00					33.67	7.88	11.17	0.04
		Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking:			UEPCO	UEPGB	14.00	90.00	90.00					33.67	7.88	11.17	3.91
1		900/976, 1+DDD, 011+,and Local (GA)	1		UEPCO	UEPCH	14.00	90.00	90.00					33.67	7.88	11.17	3.91
-	1	2-Wire Coin Outward with Operator Screening and 011Blocking	†		0L1 00	OLI OII	14.00	90.00	50.00	 				33.07	7.00	11.17	3.31
1		(GA, KY, MS)	1		UEPCO	UEPRJ	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	1	2-Wire Coin Outward with Operator Screening and Blocking:	†				00	22.00	22.00					22.01			2.0.
1		900/976, 1+DDD, 011+, and Local (FL, GA)	1		UEPCO	UEPCQ	14.00	90.00	90.00					33.67	7.88	11.17	3.91
	LOCAL	NUMBER PORTABILITY	<u> </u>														
		Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
	NONRE	CURRING CHARGES - CURRENTLY COMBINED															
			1			I										l —	I
		2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is	ļ		UEPCO	USAC2		41.50	41.50					33.67	7.88	11.17	3.91
		2-Wire Voice Grade Loop/ Line Port Combination - Switch with	1		LIEDOO	USACC		41.50	41.50					33.67	7.88	11.17	2.24
-	ADDIT	Change ONAL NRCs	<u> </u>		UEPCO	USACC		41.50	41.50	-				33.67	7.88	11.17	3.91
-	ADDITI	IONAL NRGS	<u> </u>			+				-					-		
		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent	1		UEPCO	USAS2		0.00	0.00					33.67	7.88	11.17	3.91
UNBII	NDI FD I	PORT/LOOP COMBINATIONS - MARKET BASED RATES	!		021 00	55,152		0.00	0.00	 				55.57	7.00	11.17	5.31
5.150		VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT												1		
		ort/Loop Combination Rates	1							1				İ	1	İ	
	1	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1	1	1			99.84								1		l
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			102.45										
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			113.92										
	UNE L	pop Rates															
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1	<u> </u>	1	UEPPX	UECD1	16.84	104.78	78.10								
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2	ļ	2	UEPPX	UECD1	19.45	104.78	78.10								
<u> </u>		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3	<u> </u>	3	UEPPX	UECD1	30.92	104.78	104.10					 	-	 	ļ
\vdash	UNE P	ort Rate Exchange Ports - 2-Wire DID Port	 		UEPPX	UEPD1	83.00	850.00	75.00	 				33.67	7.88	 	
-	NONDE	ECURRING CHARGES - CURRENTLY COMBINED	1		UEPPA	UEPUI	83.00	გეს.00	75.00					33.67	7.88		
1	NONKE	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -				+				H				1	 	1	1
		Switch-As-Is Top 8 MSAs only	1		UEPPX	USAC1		850.00	75.00					33.67	7.88	1	
		o months of the order of the	1	<u> </u>	0=11A	00/101		550.00	10.00	1		1	1	33.07	1.00	1	l

UNBU	INDLE	D NETWORK ELEMENTS - Georgia												,		ment: 2		bit: C
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	E	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
	1								Name		I Name a commission of	- Di						
	-				ļ			Rec	Nonrec		Nonrecurring		COMEC	SOMAN	SOMAN	Rates(\$)	SOMAN	SOMAN
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion					1		First	Add'l	First	Add'l	SOMEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMAN
		with BellSouth Allowable Changes Top 8 MSAs only			UEPPX		USA1C		850.00	75.00					33.67	7.88		
	ADDITI	ONAL NRCs		1	OLFFX		USAIC		650.00	75.00			1		33.07	7.00		
		one Number/Trunk Group Establisment Charges																
	. с.ор	DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00								
		DID Numbers, Establish Trunk Group and Provide First Group					1	0.00										
		of 20 DID Numbers			UEPPX		NDZ	0.00	0.00	0.00								
		Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		ND4	0.00	0.00	0.00								
		DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX		ND5	0.00	0.00	0.00								
		Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00								
		Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00								
	LOCAL	NUMBER PORTABILITY																
		Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00								
		ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	PORT	<u> </u>													
	UNE P	ort/Loop Combination Rates					ļ										1	
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
		UNE Zone 1		1	UEPPB	UEPPR		81.89										
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		l _														
		UNE Zone 2		2	UEPPB	UEPPR		85.27										
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
		UNE Zone 3		3	UEPPB	UEPPR		100.17										
	UNE LO	pop Rate		-	LIEDDD	HEDDD	LICLOY	24.00	252.22	400.77					40.00	10.00		
		2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USLZX	21.89	252.32	188.77					19.99	19.99		
		2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	25.27	252.32	188.77					19.99	19.99		
		2-Wire ISDN Digital Grade Loop - UNE Zone 2 2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X USL2X	40.17	252.32	188.77					19.99	19.99		
	LINE D	ort Rate		3	OLFFB	ULFFR	USLZX	40.17	232.32	100.77			1		15.55	19.99		
	ONLI	Exchange Port - 2-Wire ISDN Line Side Port		1	UEPPB	UEPPR	UEPPB	60.00	525.00	400.00			1		19.99	19.99		
	NONRE	ECURRING CHARGES - CURRENTLY COMBINED		1	OLITE	OLITIK	OLITB	00.00	323.00	400.00			1		13.33	13.33		
		2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																
		Combination - Conversion - Top 8 MSAs only			UEPPB	UEPPR	USACB	0.00	215.00	215.00					19.99	19.99		
	ADDITI	ONAL NRCs			OL: ID	<u> </u>	00/102	0.00	210.00	2.0.00					10.00	10.00		
		2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy																
		Non Feature/Add Trunk			UEPPB	UEPPR	USASB		165.95						19.99	19.99		
	LOCAL	NUMBER PORTABILITY																
		Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
	B-CHA	NNEL USER PROFILE ACCESS:																
		CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
		CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
		CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
		NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC	C,MS, &	TN)														
	USER	FERMINAL PROFILE																
		User Terminal Profile (EWSD only)		<u> </u>	UEPPB	UEPPR	U1UMA	0.00	0.00	0.00							.	
	VERTIC	CAL FEATURES		<u> </u>	==	=							ļ					
\vdash	INITES	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00					19.99	19.99	-	
$\vdash \vdash$	INTER	OFFICE CHANNEL MILEAGE		 	ļ		1				1	-	}			1	!	
		Interoffice Channel mileage each, including first mile and		1	HEDDE	LIEDDD	MACNIC	40.47	70.01	20.00					40.00	40.00	I	
	-	facilities termination Interoffice Channel mileage each, additional mile		1		UEPPR UEPPR	M1GNC M1GNM	16.47 0.0222	79.61 0.00	36.08 0.00					19.99	19.99	 	
\vdash	4-WIDE	Interoffice Channel mileage each, additional mile DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	POPT	1	UEPPB	UEFFR	IVITGINIVI	0.0222	0.00	0.00							+	
		ort/Loop Combination Rates	PURI	1	 		1						1				1	-
\vdash	JINE PO	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			1		1				1		1			1	t	
		Zone 1		1	UEPPP			955.53									1	
	1	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			OLFFF		1	955.55					1				1	1
		Zone 2		2	UEPPP			964.13								1	I	
	1	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			ULPPP		1	904.13			1		1			1	t	
		Zone 3		3	UEPPP			1,001.93								1	I	
\vdash	UNF I	pop Rates		_	SE111		1	1,001.00									-	
		4-Wire DS1 Digital Loop - UNE Zone 1	-	1	UEPPP		USL4P	55.53	448.92	276.60			 		19.99	19.99	t	1

ARONDER	D NETWORK ELEMENTS - Georgia			1	1						10 0 :			ment: 2		oit: C
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring					Rates(\$)	•	•
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Digital Loop - UNE Zone 2			UEPPP	USL4P	64.13	448.92	276.60					19.99	19.99		
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	101.93	448.92	276.60					19.99	19.99		
UNE P	ort Rate															
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	900.00	1,200.00	1,200.00					19.99	19.99		
NONR	ECURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion -Switch-As-Is Top 8 MSAs only			UEPPP	USACP	0.00	925.00	925.00					19.99	19.99		
ADDIT	TONAL NRCs			UEPPP	USACE	0.00	925.00	925.00					19.99	19.99		
ADDII	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-								1							
	Inward/two way Telephone Numbers (except NC)			UEPPP	PR7TF		0.9686									
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -			J 1 1	. 107 11		0.0000		†							
	Outward Tel Numbers (All States except NC)			UEPPP	PR7TO	l	22.75	22.75								
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -				1	İ			†							
	Subsequent Inward Telephone Numbers			UEPPP	PR7ZT	l	45.49	45.49								
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPP	LNPCN	1.75										
INTER	FACE (Provsioning Only)															
	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New o	r Additional "B" Channel															
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	28.71						19.99	19.99		
	New or Additional - Digital Data B Channel	<u> </u>		UEPPP	PR7BF	0.00	28.71						19.99	19.99		
0411	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	28.71						19.99	19.99		
CALL	TYPES			UEPPP	PR7C1	0.00	0.00	0.00	-							
-	Inward Outward			UEPPP	PR7C0	0.00	0.00	0.00	-		1					
-	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Intero	ffice Channel Mileage			OLITI	11000	0.00	0.00	0.00	+							
	Fixed Each Including First Mile			UEPPP	1LN1A	78.9223	147.07	111.75	0.00				19.99	19.99		
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.4523			****							
4-WIR	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE P	ort/Loop Combination Rates															
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		176.33										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		184.93		-								
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		222.73										
UNE L	oop Rates															
	4-Wire DS1 Digital Loop - UNE Zone 1	<u> </u>	1	UEPDC	USLDC	55.53	448.92	276.00					19.99	19.99		
	4-Wire DS1 Digital Loop - UNE Zone 2	ļ	2	UEPDC	USLDC	64.13	448.92	276.60	ļ		ļ		19.99	19.99		
1111	4-Wire DS1 Digital Loop - UNE Zone 3	 	3	UEPDC	USLDC	101.93	448.92	276.60					19.99	19.99		
UNE P	ort Rate			LIEDDO	UDD1T	750.00	4 044 40	477.07	200.70	20.70			10.00	40.00		
NOND	4-Wire DDITS Digital Trunk Port ECURRING CHARGES - CURRENTLY COMBINED	 		UEPDC	וויטטט	750.00	1,011.43	477.87	206.70	20.70	 		19.99	19.99	-	
NUNK	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	├	-						 		1	-				\vdash
	- Switch-As-Is Top 8 MSAs only			UEPDC	USAC4		269.96	269.96					19.99	19.99		
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with DS1 Changes Top 8 MSAs only			UEPDC	USAWA		269.96	269.96					19.99	19.99		
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk Top 8 MSAs only			UEPDC	USAWB		269.96	269.96					19.99	19.99		
ADDIT	IONAL NRCs 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent				+ -											
	Service Activity Per Service Order 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -			UEPDC	USAS4		147.47	147.47								
	Subsequent Channel Activation/Chan - 2-Way Trunk	ļ		UEPDC	UDTTA		28.71	28.71	 		ļ		19.99	19.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		28.71	28.71					19.99	19.99		

RUNDLE	D NETWORK ELEMENTS - Georgia				<u> </u>									ment: 2		bit: C
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)		II.	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual St Order vs Electronic Disc Add
							Nonrec	rrina	Nonrecurring Di	i a a a n n a a t			000	Rates(\$)		
					-	Rec	First	Add'l	First	Add'l	SOMEC	COMAN	SOMAN	SOMAN	SOMAN	SOMAN
-	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel		1				FIISL	Auu i	FIISL	Add I	SOWIEC	SOWAN	SOWAN	SOWAN	SOWAN	SUMAN
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		28.71	28.71					19.99	19.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			OLFDC	ODITO		20.71	20.71					15.55	19.99		
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		28.71	28.71					19.99	19.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsgnt Chan			02. 20	05.15		20.7 1	20					10.00	10.00		
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		28.71	28.71					19.99	19.99		
BIPOL	AR 8 ZERO SUBSTITUTION															
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	600.00								
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	600.00								
Alterna	ate Mark Inversion															
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Teleph	one Number/Trunk Group Establisment Charges			L												
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00										
	Telephone Number for 1-Way Outward Trunk Group	ļ	<u> </u>	UEPDC	UDTGY	0.00			 							ļ
	Telephone Number for 1-Way Inward Trunk Group Without DID		 	UEPDC	UDTGZ	0.00									1	
	DID Numbers, Establish Trunk Group and Provide First Group			LIEDDO	NDZ	0.00	0.00	0.00								
	of 20 DID Numbers DID Numbers for each Group of 20 DID Numbers			UEPDC UEPDC	NDZ ND4	0.00	0.00	0.00								
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00										
-	Reserve Non-Consecutive DID Nos.		1	UEPDC	ND6	0.00	0.00	0.00								1
	Reserve DID Numbers		1	UEPDC	NDV	0.00	0.00	0.00								1
Dedica	ted DS1 (Interoffice Channel Mileage) -		1	OLI DO	INDV	0.00	0.00	0.00								1
	O for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port															
1,70,00	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	78.47	147.07	111.75					19.99	19.99		
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities															
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25															
	miles			UEPDC	1LNOB	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.4523	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15										
4 14/100	Central Office Termininating Point		<u> </u>	UEPDC	CTG	0.00										
	E DS1 LOOP WITH CHANNELIZATION WITH PORT	4 !													-	
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti em can have various rate combinations based on type and nu			uaad												
	em can nave various rate combinations based on type and nui S1 Loop	nber or	ports	usea											-	1
UNE D	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	55.53	0.00	0.00							-	1
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	64.13	0.00	0.00								1
_	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	101.93	0.00	0.00								
UNF D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)	3	OLI WO	OOLDO	101.95	0.00	0.00								
0.12.2	24 DSO Channel Capacity - 1 per DS1	1		UEPMG	VUM24	102.64	0.00	0.00					19.99	19.99		
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	205.28	0.00	0.00					19.99	19.99	1	
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	410.56	0.00	0.00					19.99	19.99	1	
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	615.84	0.00	0.00		İ			19.99	19.99		
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	821.12	0.00	0.00					19.99	19.99		
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,026.40	0.00	0.00					19.99	19.99		
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,231.68	0.00	0.00					19.99	19.99		
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,642.24	0.00	0.00					19.99	19.99		
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,052.80	0.00	0.00					19.99	19.99		<u> </u>
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,463.36	0.00	0.00					19.99	19.99	1	<u> </u>
_																1
	672 DS0 Channel Capacity - 1 per 28 DS1s ecurring Charges (NRC) Associated with 4-Wire DS1 Loop witl		L	UEPMG	VUM67	2,873.92	0.00	0.00					19.99	19.99		

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring		201150	001441		Rates(\$)	0011411	001111
Multir	les of this configuration functioning as one are considered Ac	dd'I afta	r tho n	ninimum system con	figuration is	counted	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Iwiditip	NRC - Conversion (Currently Combined) with or without	lu i aite	linen	Illillidili systelli coli		Counted.										
1	BellSouth Allowed Changes - Top 8 MSAs Only			UEPMG	USAC4	0.00	450.00	50.00					19.99	19.99		
	m Additions Where Currently Combined and New (Not Currentle	ly Comb	oined)													
In Den	sity Zone 1 Top 8 MSAs															
İ	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc			UEPMG	V/LIMD4	0.00	050.00	000.00	200.00	20.00			40.00	10.00		
Rinol	Fea Activation - ar 8 Zero Substitution			UEPIVIG	VUMD4	0.00	950.00	600.00	200.00	30.00			19.99	19.99		
Бірої	Clear Channel Capability Format, superframe - Subsequent		1		-											
	Activity Only			UEPMG	CCOSF	0.00	0.00	600.00								
	Clear Channel Capability Format - Extended Superframe -															
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	600.00								
Altern	ate Mark Inversion (AMI)															
\vdash	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
Evehr	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port	UEPMG	MCOPO	0.00	0.00	0.00								
	nge Ports Associated with 4-wire DST Loop with Chamienzath	On with	FOIL		1	+									1	
Exona	Inger one															
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	14.00	0.00	0.00	0.00	0.00			33.67	7.88		
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	14.00	0.00	0.00	0.00	0.00			33.67	7.88		
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00			33.67	7.88		
<u> </u>	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	83.00	0.00	0.00	0.00	0.00			33.67	7.88		
Featur	re Activations - Unbundled Loop Concentration				-	-										
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank			UEPPX	1PQWM	0.62	40.00	20.00	6.00	5.00			33.67	7.88		
	Feature (Service) Activation for each Trunk Side Port Terminated			OLITA	II QVVIVI	0.02	40.00	20.00	0.00	5.00			33.07	7.00		
	in D4 Bank			UEPPX	1PQWU	0.62	110.00	30.00	65.00	20.00			33.67	7.88		
Telepi	none Number/ Group Establishment Charges for DID Service															
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
igwdot	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00								
\vdash	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers			UEPPX UEPPX	ND5 ND6	0.00	0.00	0.00							-	
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
Local	Number Portability			OLITA	NDV	0.00	0.00	0.00								
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	JRES - Vertical and Optional															
Local	Switching Features Offered with Line Side Ports Only							•								
<u> </u>	All Features Available	<u> </u>	<u> </u>	UEPPX	UEPVF	0.00	0.00	0.00							ļ	ļ
ONBONDLED	CENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: t Based Rates are applied where BellSouth is required by FCC	5 200/2-	State	Commission sule 4-	provide Ust.	undled Leest C	witching or C:	itch Borto							-	
	tures shall apply to the Unbundled Port/Loop Combination - C								dled Port section	on of this Pate	Evhihit					
	Office and Tandem Switching Usage and Common Transport											oin Port/Lo	op Combinat	ions.		
	first and additional Port nonrecurring charges apply to Not Co												•		Additional NF	RCs mav
	also and are categorized accordingly.	,			,		-,						,			,
5. Ma	rket Rates for Unbundled Centrex Port/Loop Combination will	be neg	otiated	on an Individual Ca	se Basis, un	til further notice	э.									
	CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)														
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo													ļ	1	
UNE P	Port/Loop Combination Rates (Non-Design)	<u> </u>	<u> </u>													ļ
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	1	4	UEP91		12.59										
\vdash	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	 	 	DEPSI	-	12.59									-	
1 1	Non-Design		2	UEP91		14.26										
	international processing in the contract of th	+		0=101	†	17.20								 	†	
\vdash	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP91		21.62										
UNE P	Non-Design Port/Loop Combination Rates (Design)		3	UEP91		21.62									<u> </u>	
UNE P	Non-Design		3	UEP91		21.62										

UNBUNDLE	D NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	O Mira VC Laar / O Mira Vaiga Crada Dart (Cantras) Dart Careha						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP91		21.24										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_	LIEDO4		00.74										
	Design		3	UEP91		32.71										ļ
UNE L	oop Rate		1	LIEDO4	LIECCA	40.00										
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91 UEP91	UECS1 UECS1	10.80 12.47										
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1	12.47										
	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	16.84										.
-	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	19.45					1					
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	30.92					1					
UNE P		1	3	051.91	ULUUZ	30.92			 					 	 	
	ates (Except North Carolina and Sout Carolina)	1	1		+				1		1				1	
All Sta	2-Wire Voice Grade Port (Centrex) Basic Local Area	-	 	UEP91	UEPYA	1.79	22.14	15.25	8.45	3.91	-		33.67	7.88		+
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local	-	 	OLFBI	ULFTA	1.79	22.14	15.25	0.40	3.91	-		33.07	7.68		+
	Area			UEP91	UEPYB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP91	UEPYH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP91	UEPYM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			02. 0.	02			10.20	0.10	0.01			00.01	7.00		
	Term - Basic Local Area			UEP91	UEPYZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP91	UEPY9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP91	UEPY2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
Georg	ia and Florida Only															
	2-Wire Voice Grade Port (Centrex)			UEP91	UEPHA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPHB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPHM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OLI 31	OLITIM	1.75	22.17	10.20	0.43	3.31			33.07	7.00		
	Term			UEP91	UEPHZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.5554										
Local	Number Portability															
	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Featu		 	<u> </u>	LIEDO4	LIEDVE	0.00			1		1					
	All Standard Features Offered, per port	<u> </u>	<u> </u>	UEP91	UEPVF	0.00	454.00		ļ		<u> </u>			ļ	-	<u> </u>
	All Select Features Offered, per port	<u> </u>	<u> </u>	UEP91	UEPVS	0.00	454.69		ļ						-	↓
NASC	All Centrex Control Features Offered, per port	1	_	UEP91	UEPVC	0.00			1		1			1	 	
NARS		1	_	LIEDO4	LIADOV	0.00	0.00	0.00	ļ		1		33.67	7.00	 	
	Unbundled Network Access Register - Combination	1	_	UEP91 UEP91	UARCX	0.00	0.00	0.00	ļ		}			7.88 7.88	 	
	Unbundled Network Access Register - Indial	1	_		UAR1X		0.00		ļ		}		33.67		 	
881	Unbundled Network Access Register - Outdial	 	1	UEP91	UAROX	0.00	0.00	0.00			1		33.67	7.88	 	<u> </u>
	Illaneous Terminations	 	1	 	+						1				 	
2-vvire	Trunk Side Trunk Side Terminations, each	!	 	UEP91	CENA6	11.35	61.91	61.91	 		 		33.67	7.88		
Into		-	1	UEPSI	CENAO	11.35	01.91	01.91					33.67	7.88	 	
intero	ffice Channel Mileage - 2-Wire Interoffice Channel Facilities Termination - Voice Grade	1	_	UEP91	M1GBC	17.07			ļ		}			1	 	
		1	_						ļ		}			1	 	
Factor	Interoffice Channel mileage, per mile or fraction of mile	<u> </u>	_	UEP91	M1GBM	0.0222			ļ		}			1	 	
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e	_	1					ļ		}			1	 	
D4 Ch	annel Bank Feature Activations		 	LIEDO4	400040	0.00			1		1				1	├
	Feature Activation on D-4 Channel Bank Centrex Loop Slot	l	<u> </u>	UEP91	1PQWS	0.62					l					<u> </u>

UNBUNDI	ED NETWORK ELEMENTS - Georgia												Attach	ment: 2	Exhil	bit: C
		1									Svc Order	Svc Order	Incremental		Incremental	
											Submitted	Submitted		Charge -	Charge -	Charge -
											Elec	Manually	Manual Svc	Manual Svc		Manual Svo
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						_	Nonrec	urrina	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.62										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP91	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP91	1PQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.62										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP91	1PQWQ	0.62										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.62										
Non	-Recurring Charges (NRC) Associated with UNE-P Centrex	1	İ						İ	İ			İ	İ	İ	1
1.0	Conversion - Currently Combined Switch-As-Is with allowed	1	1	1		İ				1				t	1	
	changes, per port	1		UEP91	USAC2	l	2.01	0.3108					33.67	7.88	Ì	
	New Centrex Standard Common Block	1	1	UEP91	M1ACS	0.00	659.41	3.5.30		1			33.67	7.88	1	t
	New Centrex Customized Common Block	1	1	UEP91	M1ACC	0.00	659.41			1			33.67	7.88	1	t
	Secondary Block, per Block	1	i e	UEP91	M2CC1	0.00	77.10					1	33.67	7.88	1	
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	71.88				1		33.67	7.88		1
UNF	-P CENTREX - 5ESS (Valid in All States)			02. 0.	OTTE OF T	0.00	7 1.00				1		00.01	7.00		1
	ire VG Loop/2-Wire Voice Grade Port (Centrex) Combo					1					1					1
	Port/Loop Combination Rates (Non-Design)															
0.12	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo															
	Non-Design		1	UEP95		12.59										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo		<u> </u>	02. 00		12.00					1					1
	Non-Design		2	UEP95		14.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo			OL: 50		14.20					1					1
	Non-Design		3	UEP95		21.62										
UNF	Port/Loop Combination Rates (Design)			02. 00		202					1					1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	_				1					1					1
	Design		1	UEP95		18.63										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo		<u> </u>	OL: 50		10.00					1					1
	Design		2	UEP95		21.24										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo			02. 00							1					1
	Design		3	UEP95		32.71										
LINE	Loop Rate			02. 00		02.7 1					1					1
O.V.	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	10.80					1					1
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1		UEP95	UECS1	12.47								 		1
	2-Wire Voice Grade Loop (SL 1) - Zone 2	 	3	UEP95	UECS1	19.83								-		-
	2-Wire Voice Grade Loop (SL 2) - Zone 1	1	1	UEP95	UECS2	16.84								<u> </u>		
	2-Wire Voice Grade Loop (SL 2) - Zone 2	1	2	UEP95	UECS2	19.45				1			1	†	1	t
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3	1	3	UEP95	UECS2	30.92				1			1	†	1	t
UNF	E Port Rate	1	T T		32002	33.32				1				t	1	
	States	1	1	 	+ +	+				1			1	†	1	t
	2-Wire Voice Grade Port (Centrex) Basic Local Area	1		UEP95	UEPYA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex) Basic Essai 7 (ed. 2-Wire Voice Grade Port (Centrex 800 termination)	1	1	UEP95	UEPYB	1.79	22.14	15.25	8.45	3.91			33.67	7.88	1	t
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local	1	i e					.0.20	5.46	3.01		1	55.07		1	
	Area	1		UEP95	UEPYH	1.79	22.14	15.25	8.45	3.91	1	İ	33.67	7.88	Ì	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire	1	1		1	5		.0.20	5.70	3.31			55.57	7.50	1	t
	Center)2 Basic Local Area	1		UEP95	UEPYM	1.79	22.14	15.25	8.45	3.91		1	33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1							50					1.50	1	1
	Term - Basic Local Area	1		UEP95	UEPYZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88	Ì	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	t		1	1	5		.0.20	5.70	3.31			55.57	7.50	1	1
	- Basic Local Area	1		UEP95	UEPY9	1.79	22.14	15.25	8.45	3.91		1	33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term -	1	1	1	02.10	1.73	22.17	10.20	5.45	5.91			55.57	7.50		1
	Basic Local Area	1		UEP95	UEPY2	1.79	22.14	15.25	8.45	3.91		1	33.67	7.88		
FLA	& GA Only	1	1		J 12	1.79	22.17	10.20	5.45	5.91			55.57	7.50	1	t
j. L (2-Wire Voice Grade Port (Centrex)	1	1	UEP95	UEPHA	1.79	22.14	15.25	8.45	3.91	-	 	33.67	7.88	 	
				J_1 JJ	OL: 11/A	1.13	44.14	10.20	0.73	0.01	1		00.07	1.00		1
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPHB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachr			bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring		001150			Rates(\$)		
	OME Vision On the Boat (October Complete October Miles				+ +		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPHM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPHZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent		-	UEP95	UEPH9	1.79	22.14	15.25	8.45	3.91			33.67	7.88 7.88		
Local S	2-Wire Voice Grade Port Terminated on 800 Service Term Switching			UEP95	UEPH2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		<u> </u>
Local S	Centrex Intercom Funtionality, per port		1	UEP95	URECS	0.5554										1
Local I	Number Portability			OL: 00	OILEGO	0.000+										
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Feature					1 1											1
	All Standard Features Offered, per port			UEP95	UEPVF	0.00							33.67	7.88		
	All Select Features Offered, per port			UEP95	UEPVS	0.00	454.69						33.67	7.88		
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00							33.67	7.88		
NARS					<u> </u>											ļ
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00					33.67	7.88		
	Unbundled Network Access Register - Indial			UEP95	UAR1X UAROX	0.00	0.00	0.00					33.67	7.88		ļ
	Unbundled Network Access Register - Outdial laneous Terminations		1	UEP95	UARUX	0.00	0.00	0.00					33.67	7.88		
	Trunk Side				+ +						-					
2-44116	Trunk Side Terminations, each		1	UEP95	CEND6	11.35	61.91	61.91					33.67	7.88		-
4-Wire	Digital (1.544 Megabits)			OL: 00	OLINDO	11.00	01.01	01.01					00.07	7.00		
	DS1 Circuit Terminations, each			UEP95	M1HD1	120.80	89.44	52.46					33.67	7.88		
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	28.71						33.67	7.88		
	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP95	MIGBC	17.07										
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0222										ļ
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e			\bot											<u> </u>
D4 Cha	annel Bank Feature Activations			LIEBOE	400140	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS											
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.62										<u> </u>
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP95	1PQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop		-	UEP95	1PQWV	0.62					-					
.	Slot			UEP95	1PQWQ	0.62										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.62										
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
,	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP95	USAC2		2.01	0.3108					33.67	7.88		
	New Centrex Standard Common Block		1	UEP95 UEP95	M1ACS	0.00	659.41	0.3108					33.67	7.88		
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	659.41						33.67	7.88		1
	NAR Establishment Charge, Per Occasion		†	UEP95	URECA	0.00	71.88						33.67	7.88		
	CENTREX - DMS100 (Valid in All States)				1											1
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE Po	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		_1	UEP9D		12.59										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Non-Design		2	UEP9D		14.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP9D		21.62										

UNBUNDLI	ED NETWORK ELEMENTS - Georgia													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring		001150	001441		Rates(\$)	0011411	001441
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Design		1	UEP9D		18.63										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9D		21.24										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP9D		32.71										
UNE	Loop Rate			LIEDOD	LIEGOA	40.00										
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D UEP9D	UECS1 UECS1	10.80 12.47					1					<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	19.83										-
-	2-Wire Voice Grade Loop (SL 1) - Zone 3		1	UEP9D	UECS2	16.84					1					
- 	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2	1	2	UEP9D	UECS2	19.45			+						<u> </u>	
	2-Wire Voice Grade Loop (SL 2) - Zone 3	1	3	UEP9D	UECS2	30.92			1		1				1	†
UNE I	Port Rate		_													
	STATES															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP9D	UEPYB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
	Area			UEP9D	UEPYC	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local															
	Area			UEP9D	UEPYD	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local			UEP9D	UEPTE	1.79	22.14	15.25	8.45	3.91			33.67	7.88		-
	Area			UEP9D	UEPYF	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local			OLF 9D	OLFII	1.79	22.14	13.23	0.43	3.91			33.07	7.00		
	Area			UEP9D	UEPYG	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															
	Area			UEP9D	UEPYT	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local															
	Area			UEP9D	UEPYU	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local															
	Area			UEP9D	UEPYV	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
1	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local	l														
	Area	 	 	UEP9D	UEPY3	1.79	22.14	15.25	8.45	3.91	}		33.67	7.88	ļ.	
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area	1	1	UEP9D	UEPYH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp	-	 	OEFSD	UEFTH	1.79	22.14	15.25	8.45	3.91	1		33.07	7.88	 	
	Indication))3 Basic Local Area	l		UEP9D	UEPYW	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3			02. 00	JE1 1 1 1 1	1.75	22.14	10.20	0.40	5.91			55.07	7.00		
	Basic Local Area	l		UEP9D	UEPYJ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2 Basic Local Area	<u> </u>		UEP9D	UEPYM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3															
	Basic Local Area			UEP9D	UEPYO	1.79	22.14	15.25	8.45	3.91	<u> </u>		33.67	7.88		1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3	1	1	LIEBOD	LIEDY'S											
	Basic Local Area	 	<u> </u>	UEP9D	UEPYP	1.79	22.14	15.25	8.45	3.91	1		33.67	7.88	1	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area	1	1	UEP9D	UEPYQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3	1	1	UEP9D	UEPTU	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Basic Local Area	1	1	UEP9D	UEPYR	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3	1	1	02100	OLI IIX	1.79	22.14	10.23	0.43	5.91			33.07	7.00	<u> </u>	
	Basic Local Area	1	1	UEP9D	UEPYS	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3						-							1		
	Basic Local Area	<u> </u>	L	UEP9D	UEPY4	1.79	22.14	15.25	8.45	3.91	<u></u>	<u></u>	33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3															
	Basic Local Area	<u> </u>	L	UEP9D	UEPY5	1.79	22.14	15.25	8.45	3.91	<u> </u>	<u> </u>	33.67	7.88	<u> </u>	<u> </u>

UNRI	INDI FI	D NETWORK ELEMENTS - Georgia												Attach	ment: 2	Fyhil	bit: C
CIVE	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,											Svc Order	Svc Order	Incremental		Incremental	
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec	Manually	Manual Svc	Manual Svc		Manual Svo
CATE	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									per Lore	por Lore	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																D130 13t	DISC Add I
							Rec	Nonred			g Disconnect				Rates(\$)		
							1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3															
		Basic Local Area			UEP9D	UEPY6	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3													=		
		Basic Local Area			UEP9D	UEPY7	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			LIEDOD	LIEDVZ	4.70	22.44	45.05	0.45	2.04			22.67	7.00		
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPYZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		Basic Local Area			UEP9D	UEPY9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	1	2-Wire Voice Grade Port Terminated on 800 Service Term Basic			OLF 9D	OLF19	1.79	22.14	13.23	0.40	3.91	1		33.07	7.00		
1		Local Area			UEP9D	UEPY2	1.79	22.14	15.25	8.45	3.91		1	33.67	7.88	1	
-	FL & G	A Only	1		021 00	JL1 12	1.75	22.14	10.20	0.40	5.91	<u> </u>	 	33.07	7.00	 	
		2-Wire Voice Grade Port (Centrex)	l		UEP9D	UEPHA	1.79	22.14	15.25	8.45	3.91			33.67	7.88	1	
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPHB	1.79	22.14	15.25	8.45	3.91			33.67	7.88	1	
	1	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPHC	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHD	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPHE	1.79	22.14	15.25		3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPHF	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHT	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPHU	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPHV	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPH3	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPHH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp													=		
-		Indication)3			UEP9D	UEPHW	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPHJ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-ville voice Grade Fort (Certitex from all Serving wife Certier)			UEP9D	UEPHM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	1.79	22.14	15.25	8.45	3.91	1		33.67	7.88		
		2-Wile Voice Glade For (Centrex differ GWC /EBG-FGE F)2, 3			OLI 3D	OLITIO	1.75	22.14	13.23	0.40	3.31			33.07	7.00		-
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPHQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
							-										
	<u> </u>	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3	<u></u>	L	UEP9D	UEPHS	1.79	22.14	15.25	8.45	3.91	<u> </u>	<u></u>	33.67	7.88		<u> </u>
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
			l		-								1]]	
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	1.79	22.14	15.25	8.45	3.91			33.67	7.88	ļ	
			l							_							
	1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3	ļ	 	UEP9D	UEPH6	1.79	22.14	15.25	8.45	3.91			33.67	7.88	ļ	
		2 Miro Voice Crade Bort (Centre VIIII - CMC /EBC MEC(C)			UEP9D	UEPH7	4 70	00.44	45.05	0.45	3.91		1	33.67	7.88	1	
-	1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3	 	.	UEP9D	UEPH/	1.79	22.14	15.25	8.45	3.91	-		33.67	7.88	 	+
1		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term	l		UEP9D	UEPHZ	1.79	22.14	15.25	8.45	3.91		1	33.67	7.88	1	
-	1	Tomi	1	1	OLFAD	UEPHZ	1.79	22.14	15.25	8.45	3.91	-	 	33.07	7.88	1	
1		2-Wire Voice Grade Port terminated in on Megalink or equivalent	l		UEP9D	UEPH9	1.79	22.14	15.25	8.45	3.91		1	33.67	7.88	1	
-	1	2-Wire Voice Grade Port Terminated in on Weganink of equivalent	-		UEP9D	UEPH2	1.79	22.14	15.25	8.45	3.91		 	33.67	7.88	 	
	Local S	Switching					0		.0.20	3.40	5.51			33.07			t
		Centrex Intercom Funtionality, per port			UEP9D	URECS	0.5554								1		
	Local N	Number Portability			-						İ				1		
		Local Number Portability (1 per port)			UEP9D	LNPCC	0.35				İ				1		
	Feature	es															
		All Standard Features Offered, per port			UEP9D	UEPVF	0.00	_									
		All Select Features Offered, per port			UEP9D	UEPVS	0.00	454.69						33.67	7.88		
		All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00										
	NARS					1									1		
		Unbundled Network Access Register - Combination	1		UEP9D	UARCX	0.00	0.00	0.00					33.67	7.88]	

UNBU	NDLE	D NETWORK ELEMENTS - Georgia												Attachr	ment: 2	Exhil	bit: C
												Svc Order	Svc Order				Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			l '''											Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
						+	l	Nonrec	urring	Nonrecurring Dis	sconnect			OSS	Rates(\$)		
						1	Rec	First	Add'l		Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
		Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00					33.67	7.88		
		Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00					33.67	7.88		
		laneous Terminations															
	2-Wire	Trunk Side				_											
	4 100	Trunk Side Terminations, each			UEP9D	CEND6	11.35										
-	4-Wire	Digital (1.544 Megabits) DS1 Circuit Terminations, each			UEP9D	M1HD1	120.80	89.44	52.46					33.67	7.88		
		DS0 Channels Activiated per Channel			UEP9D	M1HD0	0.00	28.71	52.46					33.67	7.88		
-	Interof	fice Channel Mileage - 2-Wire			UEF9D	MITHDO	0.00	20.71						33.07	1.00		ļ
	III.C. O.	Interoffice Channel Facilities Termination			UEP9D	MIGBC	17.07										+
		Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0222										
	Feature	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e	i –	-	1	****	İ						l		İ	
		annel Bank Feature Activations		1													
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.62										
															1		
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot		<u> </u>	UEP9D	1PQW6	0.62								ļ		ļ
1		Feature Activation on D-4 Channel Bank FX Trunk Side Loop		1						[1		1		
		Slot			UEP9D	1PQW7	0.62										
		Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
-		Different Wire Center			UEP9D	1PQWP	0.62										
		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.62										
-		Feature Activation on D-4 Channel Bank Tije Line/Trunk Loop			UEF9D	IPQVV	0.02										ļ
		Slot			UEP9D	1PQWQ	0.62										
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.62										
	Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex			<u> </u>		7.4										
		NRC Conversion Currently Combined Switch-As-Is with allowed															
		changes, per port			UEP9D	USAC2		2.01	0.3108					33.67	7.88		
		New Centrex Standard Common Block			UEP9D	M1ACS	0.00	659.41						33.67	7.88		
		New Centrex Customized Common Block			UEP9D	M1ACC	0.00	659.41						33.67	7.88		
		NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	71.88						33.67	7.88		
		- Required Port for Centrex Control in 1AESS, 5ESS & EWSD				1											_
		P - Requres Interoffice Channel Mileage - Requires Specific Customer Premises Equipment				-											-
LINBLIN		CENTREX PORT/LOOP COMBINATIONS - MARKET RATES				1											
ONDON		ket Rates are applied where BellSouth is not required by FCC	and/or	State C	ommission rule to r	provide Unbu	Indled Local Sw	itching or Swit	tch Ports.								
		urring Charges for all Standard Centrex and Centrex Conrol Fe					1	toning or our									
		Office and Tandem Switching Usage and Common Transport					nibit shall apply	to all combina	tions of loop/	port network eleme	ents except	t for UNE C	oin Port/Lo	op Combinat	ions.		
	4. The	first and additional Port nonrecurring charges apply to Not Cu	urrently	Comb	ined Combos. For	Currently Co	mbined Combo	s, the nonrecu	rring charges	shall be those ide	ntified in th	ne Nonrecu	rring - Curre	ently Combine	ed sections.	Additional NR	Cs may
		also and are categorized accordingly.															
		CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)		-												
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo													ļ		
<u> </u>	UNE P	ort/Loop Combination Rates (Non-Design)	ļ	<u> </u>		+	ļ			ļ <u></u>					ļ		<u> </u>
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	l .	LIEBO4	1											
<u> </u>		Non-Design	 	1	UEP91	1	24.80			 					 		1
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	ĺ	2	UEP91	1	26.47										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OL1 01	+	20.47			 			 		 		
		Non-Design	ĺ	3	UEP91	1	33.83										
	UNE P	ort/Loop Combination Rates (Design)		Ť			55.55								1		
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1													
<u></u>		Design	<u></u>	1	UEP91	1	30.84			<u>[</u>]			<u> </u>	<u> </u>			
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -]		
		Design		2	UEP91		33.45										<u> </u>
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			l			T					1		1		
<u> </u>		Design		3	UEP91		44.92										
<u></u>	UNE L	pop Rate		<u> </u>	LIEDO4	UE004	10.00							-	 	-	
-		2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	l	2	UEP91 UEP91	UECS1 UECS1	10.80 12.47								 		
		2-VVIIG VOICE Glade LOOP (OL 1) - ZUITE Z	<u> </u>		OFLSI	ULUSI	12.4/						<u> </u>	I	i	l	

NRONDFF	D NETWORK ELEMENTS - Georgia			ı	, .									ment: 2		oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP91	UECS1	19.83										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	16.84										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	19.45										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	30.92										
UNE P																
All Sta	tes (Except North Carolina and Sout Carolina)					44.00		45.00		10.00						
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP91	UEPYB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP91	UEPYH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP91	UEPYM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP91	UEPYZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP91	UEPY9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP91	UEPY2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
Georg	a and Florida Only				1				ļ							
	2-Wire Voice Grade Port (Centrex)			UEP91	UEPHA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPHB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPHM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP91	UEPHZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.5554										
Local	Number Portability															
	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Featur				LIEDO4	LIED) (E	0.00										
-	All Scloot Features Offered, per port		 	UEP91	UEPVE	0.00	454.69		 					-	-	
	All Select Features Offered, per port All Centrex Control Features Offered, per port			UEP91 UEP91	UEPVS UEPVC	0.00	454.69		 				-	-	-	
NARS			 	OFLAI	OLF VC	0.00			1				1	1	1	
III	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00	 				33.67	7.88		
-	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00					33.67	7.88		
	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00					33.67	7.88		
Miscel	laneous Terminations								1					1	İ	
	Trunk Side															
	Trunk Side Terminations, each			UEP91	CENA6	11.35	61.91	61.91					33.67	7.88		
Interof	fice Channel Mileage - 2-Wire						•	•								
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	17.07										
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0222										
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	е			1											
D4 Cha	Annel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.62										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.62				_						
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop								1							
_	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP91	1PQW7	0.62										
_	Different Wire Center			UEP91	1PQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.62										

UNBUN	DLE	D NETWORK ELEMENTS - Georgia												Attach	ment: 2	Exhib	oit: C
32011						T I						Svc Order	Svc Order	Incremental		Incremental	Incremental
1			1									Submitted	Submitted		Charge -	Charge -	Charge -
			Inter.									Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATEGO	RY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						.,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
																Disc 1st	
														1st	Add'l	DISC 1St	Disc Add'l
							B	Nonrec	curring	Nonrecurring	Disconnect			oss	Rates(\$)	U U	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
		Slot			UEP91	1PQWQ	0.62										ł
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.62										
N	on-Re	curring Charges (NRC) Associated with UNE-P Centrex															
		Conversion - Currently Combined Switch-As-Is with allowed															
		changes, per port			UEP91	USAC2		2.01	0.3108					33.67	7.88		f
		New Centrex Standard Common Block			UEP91	M1ACS	0.00	659.41						33.67	7.88		í
		New Centrex Customized Common Block			UEP91	M1ACC	0.00	659.41						33.67	7.88		
		Secondary Block, per Block			UEP91	M2CC1	0.00	77.10						33.67	7.88		í
		NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	71.88						33.67	7.88		1
	NE-P	CENTREX - 5ESS (Valid in All States)															i
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo															1
		ort/Loop Combination Rates (Non-Design)															1
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														1
		Non-Design	1	1	UEP95	1	24.80					1		Ì	Ì		1
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															1
		Non-Design	1	2	UEP95	1	26.47					1		Ì	Ì		1
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															í
		Non-Design		3	UEP95		33.83										ł
U	NE Po	ort/Loop Combination Rates (Design)															í
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															í
		Design		1	UEP95		30.84										ł
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Design		2	UEP95		33.45										ł
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															í
		Design		3	UEP95		44.92										ł
U	NE Lo	oop Rate															í
		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	10.80										í
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	12.47										í
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	19.83										í
		2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	16.84										í
		2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	19.45										í
		2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	30.92										í
U	NE Po	ort Rate															í
Α	II Stat																ĺ
		2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															, <u></u>
		Area			UEP95	UEPYH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex from diff Serving Wire	1			1	l										ł
		Center)2 Basic Local Area	<u> </u>		UEP95	UEPYM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		1
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1	1		1	l					1		Ì	Ì		1
		Term - Basic Local Area	ļ		UEP95	UEPYZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port terminated in on Megalink or equivalent	1			1 7	\exists					1	<u> </u>	<u> </u>]		1
		- Basic Local Area	ļ		UEP95	UEPY9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port Terminated on 800 Service Term -	1	1			l					1		Ì	Ì		1
		Basic Local Area	ļ		UEP95	UEPY2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
F	L & G	A Only	<u> </u>			1				1				ļ	ļ		
		2-Wire Voice Grade Port (Centrex)	ļ		UEP95	UEPHA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
LL		2-Wire Voice Grade Port (Centrex 800 termination)	ļ		UEP95	UEPHB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex with Caller ID)1	ļ		UEP95	UEPHH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex from diff Serving Wire	1	1	l <u>_</u> _	1	l					1		Ì	Ì		1
		Center)2	<u> </u>		UEP95	UEPHM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		1
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1		l	1 1	l										í
		Term	<u> </u>		UEP95	UEPHZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		1
1			1		l	1	l										ł
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPH9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port Terminated on 800 Service Term	<u> </u>		UEP95	UEPH2	14.00	90.00	45.00	20.00	10.00		ļ	33.67	7.88		
L	ocal S	witching				1							<u>l</u>]	l		<u>. </u>

UNBUNDLE	D NETWORK ELEMENTS - Georgia													ment: 2		oit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual St Order vs Electronic Disc Add
						Dan.	Nonrec	urring	Nonrecurring D	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.5554										
Local	Number Portability															
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Featur																
	All Standard Features Offered, per port			UEP95	UEPVF	0.00							33.67	7.88		
	All Select Features Offered, per port			UEP95	UEPVS	0.00	454.69						33.67	7.88		
NABO	All Centrex Control Features Offered, per port	<u> </u>		UEP95	UEPVC	0.00							33.67	7.88		
NARS	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00	<u> </u>				33.67	7.88		
-	Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00					33.67	7.88		
	Unbundled Network Access Register - Indiai			UEP95	UAROX	0.00	0.00	0.00					33.67	7.88		
Miscel	laneous Terminations	†	†	021 00	5/11/5/	0.00	0.00	0.00	 				33.07	7.00	 	
	Trunk Side		 		1											
1	Trunk Side Terminations, each		1	UEP95	CEND6	11.35	61.91	61.91	1				33.67	7.88	1	
4-Wire	Digital (1.544 Megabits)		i –		1											
	DS1 Circuit Terminations, each			UEP95	M1HD1	120.80	89.44	52.46					33.67	7.88	<u> </u>	
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	28.71						33.67	7.88		
Interof	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP95	MIGBC	17.07										
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0222										
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 Ch	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.62										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.62										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP95	1PQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.62										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP95	1PQWQ	0.62										
N 5	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.62										
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex				+				 							
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port		1	UEP95	USAC2		2.01	0.3108					33.67	7.88	1	
	New Centrex Standard Common Block		1	UEP95	M1ACS	0.00	659.41	0.5106	 				33.67	7.88		
	New Centrex Standard Common Block	 	1	UEP95	M1ACC	0.00	659.41		 				33.67	7.88		
	NAR Establishment Charge, Per Occasion		†	UEP95	URECA	0.00	71.88						33.67	7.88	1	
UNE-P	CENTREX - DMS100 (Valid in All States)		<u> </u>			2.00	00						22.0.	1.00		
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo		1													
	ort/Loop Combination Rates (Non-Design)			<u> </u>										<u> </u>	<u> </u>	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Non-Design		1	UEP9D		24.80										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9D		26.47										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9D		33.83										
UNE P	ort/Loop Combination Rates (Design)	1	i													
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Design		1	UEP9D		30.84										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9D		33.45										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9D		44.92										
UNE I	oop Rate		Ť	02.00	1	77.32										
J L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	10.80									1	
	2-Wire Voice Grade Loop (SL 1) - Zone 2	1		UEP9D	UECS1	12.47			t		i		1	1	†	

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachi	ment: 2	Exhib	oit: C
J.150115EE											Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		Intori									Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													151	Auu i	DISC 1St	DISC Add I
						1	Nonrec	curring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	19.83										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.84										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	19.45										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	30.92										
LINE E	Port Rate			OLI OD	OLOGE	00.02										
	TATES															
ALL 3	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
\vdash				UEP9D	UEPTA	14.00	90.00	45.00	20.00	10.00			33.07	7.00		
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			LIEDOD	LIEDVD	44.00	00.00	45.00	00.00	40.00			00.07	7.00		
$\overline{}$	Area			UEP9D	UEPYB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
\vdash	Area	!		UEP9D	UEPYC	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
1 1	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local	1	1								l	1				
\vdash	Area	ļ		UEP9D	UEPYD	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local															
	Area]		UEP9D	UEPYE	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local															
	Area			UEP9D	UEPYF	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local															
	Area			UEP9D	UEPYG	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															
	Area			UEP9D	UEPYT	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			02.05	02	11.00	00.00	10.00	20.00	10.00			00.07	7.00		
	Area			UEP9D	UEPYU	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local			OLI 3D	OLITO	14.00	30.00	43.00	20.00	10.00			33.07	7.00		
	Area			UEP9D	UEPYV	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
\vdash				UEP9D	UEPTV	14.00	90.00	45.00	20.00	10.00			33.07	1.00		
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local			LIEDOD	LIEDVO	44.00	00.00	45.00	20.00	40.00			22.67	7.00		
	Area			UEP9D	UEPY3	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local															
	Area			UEP9D	UEPYH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication))3 Basic Local Area			UEP9D	UEPYW	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3															
	Basic Local Area			UEP9D	UEPYJ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2 Basic Local Area			UEP9D	UEPYM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3															
	Basic Local Area			UEP9D	UEPYO	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3	Ì									İ					
1 1	Basic Local Area	1		UEP9D	UEPYP	14.00	90.00	45.00	20.00	10.00	1		33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3	1			1					1				1		
1 1	Basic Local Area	1	1	UEP9D	UEPYQ	14.00	90.00	45.00	20.00	10.00	l	1	33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3	1				00	22.00				İ		22.01	1.00		
1 1	Basic Local Area	1		UEP9D	UEPYR	14.00	90.00	45.00	20.00	10.00		1	33.67	7.88		
\vdash	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3	 			J=: .1\	14.00	55.50	70.00	20.00	10.00		l	55.57	7.50		
1 1	Basic Local Area	1		UEP9D	UEPYS	14.00	90.00	45.00	20.00	10.00		1	33.67	7.88		
\vdash	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3	 	1	OLI 3D	JLI IJ	14.00	90.00	45.00	20.00	10.00	1	l	33.07	1.00		
1 1	Basic Local Area	1		UEP9D	UEPY4	14.00	90.00	45.00	20.00	10.00	1		33.67	7.88		
\vdash	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3	1	1	טבו שט	OLF 14	14.00	90.00	45.00	20.00	10.00	-	-	33.07	1.00		
1 1		1		UEP9D	UEPY5	14.00	90.00	45.00	20.00	10.00	1		33.67	7.88		
\vdash	Basic Local Area	 	-	UEF9D	UEPTO	14.00	90.00	45.00	20.00	10.00	 	-	33.67	7.88		
1 1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3	1	1	LIEDOD	LIEDVO	44.00	00.00	45.00	00.00	40.00	l	1	00.00	7.00		
\vdash	Basic Local Area	!		UEP9D	UEPY6	14.00	90.00	45.00	20.00	10.00	1	ļ	33.67	7.88		
1 1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3	1	1								l	1				
\vdash	Basic Local Area	<u> </u>	<u> </u>	UEP9D	UEPY7	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
1 1	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1							1		1					
	Term	ļ		UEP9D	UEPYZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1	1						_		i	1				
	Basic Local Area	<u></u>	L	UEP9D	UEPY9	14.00	90.00	45.00	20.00	10.00	<u> </u>	<u> </u>	33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic															
1 1	Local Area	1	1	UEP9D	UEPY2	14.00	90.00	45.00	20.00	10.00	l	1	33.67	7.88		

UNRIII	NDI F	NETWORK ELEMENTS - Georgia												Attach	ment: 2	Fyhil	bit: C
3.4001		ALL HORR LELIELITO Georgia										Svc Order	Svc Order	Incremental		Incremental	
1												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						- ()			per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																DISC 1St	DISC Add I
							Rec	Nonrec	urring	Nonrecurring	g Disconnect			oss	Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	FL & G																
		2-Wire Voice Grade Port (Centrex)			UEP9D	UEPHA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPHB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPHC	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHD	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPHE	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPHF	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHT	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
\vdash		2-Wire Voice Grade Port (Centrex / EBS-M5208)3		\vdash	UEP9D	UEPHU	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
 		2-Wire Voice Grade Port (Centrex / EBS-M5216)3	<u> </u>		UEP9D	UEPHV	14.00	90.00	45.00	20.00	10.00			33.67	7.88	1	├
\vdash		2-Wire Voice Grade Port (Centrex / EBS-M5316)3	<u> </u>		UEP9D	UEPH3 UEPHH	14.00	90.00	45.00	20.00	10.00			33.67	7.88 7.88	1	├
 		2-Wire Voice Grade Port (Centrex with Caller ID)	 	—	UEP9D	UEPHH	14.00	90.00	45.00	20.00	10.00		ļ	33.67	7.88	-	
		2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			LIEDOD	LIEDLIM/	44.00	00.00	45.00	20.00	10.00			22.67	7.00		
\vdash		Indication)3 2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3	├	 	UEP9D UEP9D	UEPHW UEPHJ	14.00 14.00	90.00	45.00 45.00	20.00 20.00	10.00 10.00		 	33.67 33.67	7.88 7.88		
\vdash		2-Wire Voice Grade Port (Centrexivisg Wtg Lamp Indication)3 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)	 		UELAD	UEPHJ	14.00	90.00	45.00	∠0.00	10.00			33.67	7.88	-	
		2-Wile voice Grade Port (Centrex from all Serving Wile Center)			UEP9D	UEPHM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
-		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wile Voice Grade Fort (Certifex diller SWC /LB3-F3L1)2, 3			OLF3D	OLFTIO	14.00	90.00	45.00	20.00	10.00			33.07	7.00		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPHQ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-ville voice diade i dit (dentiex dinei dvvo /EBO-3203/2, 3			OLI 3D	OLITIQ	14.00	30.00	45.00	20.00	10.00			33.07	7.00		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		z mie reies erase i en (eenmen amer erre /zze me rz/z, e			02.02	02	1 1.00	00.00	.0.00	20.00	10.00			00.07	7.00		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		· · · · · · · · · · · · · · · · · · ·															
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
		Term			UEP9D	UEPHZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
						1							1		I		1
ļ		2-Wire Voice Grade Port terminated in on Megalink or equivalent	ļ		UEP9D	UEPH9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		1
		2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPH2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
 		witching	<u> </u>		LIEBOD	LIDEOO	0.555.1								-		├
\vdash		Centrex Intercom Funtionality, per port	 		UEP9D	URECS	0.5554			-	-				1		
\vdash	Local N	umber Portability	 		LIEDOD	LNDCC	0.25			-	-				1		
\vdash	Feature	Local Number Portability (1 per port)	 	-	UEP9D	LNPCC	0.35			ļ	ļ			-	 		
-		All Standard Features Offered, per port	 		UEP9D	UEPVF	0.00			-	-			-	-	-	
-		All Standard Features Offered, per port All Select Features Offered, per port	 	-	UEP9D UEP9D	UEPVF	0.00	454.69		1	1		 	33.67	7.88	1	
 		All Centrex Control Features Offered, per port	 		UEP9D	UEPVS	0.00	404.09		1	1			33.07	1.08	1	
$\vdash \vdash \vdash$	NARS	7 at Control Control i Gatares Chereu, per port	 		JL1 3D	JL1 VC	0.00			1	1			1	t	1	
\vdash		Unbundled Network Access Register - Combination	 		UEP9D	UARCX	0.00	0.00	0.00				 	33.67	7.88		
 		Unbundled Network Access Register - Inward	†		UEP9D	UAR1X	0.00	0.00	0.00				 	33.67	7.88		—
		Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00	1	1			33.67	7.88		t
		aneous Terminations					0.00	0.00	3.30	Ì	1			55.57	7.50		
		Frunk Side								Ì	1				1		
		Trunk Side Terminations, each			UEP9D	CEND6	11.35										
i i		Digital (1.544 Megabits)															
		DS1 Circuit Terminations, each			UEP9D	M1HD1	120.80	89.44	52.46					33.67	7.88		
		DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	28.71						33.67	7.88		1
		ice Channel Mileage - 2-Wire															
		Interoffice Channel Facilities Termination			UEP9D	MIGBC	17.07										

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted Manually	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svo Order vs.
						D	Nonrec	curring	Nonrecurrin	g Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0222										
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 Ch	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.62										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.62										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.62										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.62										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.62										
Non-R	lecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP9D	USAC2		2.01	0.3108					33.67	7.88		
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	659.41		1	1	1	1	33.67	7.88	İ	†
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	659.41					İ	33.67	7.88		1
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	71.88			İ		İ	33.67	7.88		1
Note 1	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD		1							İ		İ				1
	2 - Requres Interoffice Channel Mileage									İ		İ				1
	3 - Requires Specific Customer Premises Equipment		1							İ		İ				1
	Rates displaying an "R" in Interim column are interim and sub	iect to	rate tru	e-up as set forth in	General Tern	ns and Condition	ns.			İ		İ				1

LINDIII	IDI E	D NETWORK ELEMENTS - Kentucky												A44		FLi	-: 0
UNDU	NDLE	D NETWORK ELEMENTS - Kentucky			I	1	1					Cua Oudan	Cur Onden	Incremental	ment: 2	Incremental	oit: C Incremental
												Submitted			Charge -	Charge -	Charge -
CATEG	nev	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CAILO	J. ()	KATE EEEMENTO	m	20116	500	0000			KAT LO(ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							_	Nonrec	curring	Nonrecurring	Disconnect			oss	Rates(\$)		ı
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	The "Zo	one" shown in the sections for stand-alone loops or loops as part	of a com	binatio	n refers to Geographi	ically Deavera	aged UNE Zones	. To view Geog	raphically Dea	veraged UNE Z	one Designation	ns by Centra	l Office, refe	r to Internet W	ebsite:		
	nttp://w	ww.interconnection.bellsouth.com/become_a_clec/html/interconn	ection.h	tm													
		SUPPORT SYSTEMS															
	NOTE:	(1) Electronic Service Order: CLEC should contact its contract	ct nego	tiator if	it prefers the state	specific elec	tronic service o	rdering charge	es as ordered l	y the State Co	mmissions. T	he electron	ic service o	rdering charg	e currently co	ntained in th	is rate
	exhibit	is the BellSouth regional electronic service ordering charge.	CLEC	may ele	ect either the state s	pecific Com	mission ordered	rates for the	electronic serv	ice ordering c	narges, or CLE	C may elec	the region	al electronic	service orderii	ng charge.	
	NOTE:	(2) Any element that can be ordered electronically will be bill	led acco	rding	to the SOMEC rate li	isted in this	category. Pleas	e refer to Bell	South's Busine	ess Rules for L	ocal Ordering	(BBR-LO) t	determine	if a product	an be ordere	d electronical	ly. For
	hose e	elements that cannot be ordered electronically at present per	the BBR	R-LO, th	ne listed SOMEC rate	e in this cate	gory reflects the	e charge that v	vould be billed	to a CLEC on	ce electronic c	ordering cap	abilities co	me on-line fo	r that element	. Otherwise,	the manual
	orderin	g charge, SOMAN, will be applied to a CLECs bill when it sub	bmits ar	LSR t	o BellSouth.												
		Manual Service Order Charge, per LSR, Disconnect Only (KY)				SOMAN				0.99							
		Electronic OSS Charge, per LSR, submitted via BST's OSS															
		interactive interfaces (Regional)				SOMEC		3.50									
		DATE ADVANCEMENT CHARGE															
	NOTE:	The Expedite charge will be maintained commensurate with	BellSou	th's FO	CC No.1 Tariff, Section	on 5 as appli	icable.										
		UNE Expedite Charge per Circuit or Line Assignable USOC, per	1	1]]		<u> </u>
		Day			ALL UNE	SDASP		200.00									
		EXCHANGE ACCESS LOOP															
	2-WIRE	ANALOG VOICE GRADE LOOP															
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	10.56	46.66	22.57	26.65	7.65		7.86				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	15.34	46.66	22.57	26.65	7.65		7.86				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	31.11	46.66	22.57	26.65	7.65		7.86				
		Loop Testing - Basic 1st Half Hour			UEANL	URET1		46.88	46.88				7.86				
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		24.16	24.16				7.86				
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
		(UVL-SL1)			UEANL	UREWO		15.78	8.94				7.86				
		Engineering Information Document (EI)			UEANL	UEANM		13.49	13.49								
		Manual Order Coordination for UVL-SL1s (per loop)			UEANL	UEAMC		9.00	9.00								
		Order Coordination for Specified Conversion Time for UVL-SL1															
		(per LSR)			UEANL	OCOSL		23.01	23.01								
	2-WIRE	Unbundled COPPER LOOP															
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1		1	UEQ	UEQ2X	10.58	44.97	20.89	25.64	6.65		7.86				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2		2	UEQ	UEQ2X	11.51	44.97	20.89	25.64	6.65		7.86				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3		3	UEQ	UEQ2X	13.19	44.97	20.89	25.64	6.65		7.86				
		Order Coordination 2 Wire Unbundled Copper Loop - Non-															
		Designed (per loop)			UEQ	USBMC		9.00	9.00								
—		Engineering Information Document	!		UEQ	LIDET4		13.49	13.49				7.00				
		Loop Testing - Basic 1st Half Hour	!	 	UEQ	URET1		46.88	46.88	-			7.86	 	 		
-		Loop Testing - Basic Additional Half Hour	1	-	UEQ	URETA		24.16	24.16			-	7.86	-	ļ		
		CLEC to CLEC Conversion Charge Without Outside Dispatch (UCL-ND)	1	1	LIFO	LIBEWO		44.07	7.40				7.00	Ì	Ì		
LINDUAL	י בם י	I(UCL-ND) EXCHANGE ACCESS LOOP	 	 	UEQ	UREWO	1	14.27	7.43			-	7.86	-	-		-
		ANALOG VOICE GRADE LOOP	 	 		1	1					-		-	-		-
-	L-VVIRE	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	 	-		+	<u> </u>			-	-			-	-		-
		Ground Start Signaling - Zone 1	1	1	UEA	UEAL2	12.67	134.89	81.87	73.65	14.88		7.86	Ì	Ì		
\vdash		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	1		ULA	UEALZ	12.07	134.89	81.87	73.05	14.88	-	7.86	1	1		1
		Ground Start Signaling - Zone 2	1	2	UEA	UEAL2	17.45	134.89	81.87	73.65	14.88		7.86	Ì	Ì		
\vdash		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	1		ULA	ULALZ	17.45	134.09	01.07	13.05	14.68	-	1.00	1	1		1
		Ground Start Signaling - Zone 3	1	3	UEA	UEAL2	33.22	134.89	81.87	73.65	14.88		7.86				
\vdash		Order Coordination for Specified Conversion Time (per LSR)	 	۲	UEA	OCOSL	30.22	23.01	01.07	75.05	17.00		7.00	 	 		
\vdash		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	 		OLA.	JUUGL		23.01						 	 		
		Battery Signaling - Zone 1	1	1	UEA	UEAR2	12.67	134.89	81.87	73.65	14.88		7.86				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	 			2=,\2	12.01	104.00	01.07	70.00	14.50		7.00				
		Battery Signaling - Zone 2	1	2	UEA	UEAR2	17.45	134.89	81.87	73.65	14.88		7.86	1	1		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1	 			0	.000	301	. 5.00	00			1			
		Battery Signaling - Zone 3	1	3	UEA	UEAR2	33.22	134.89	81.87	73.65	14.88		7.86	Ì	Ì		
		Order Coordination for Specified Conversion Time (per LSR)	1	۲	UEA	OCOSL	55.22	23.01	01.07	7 0.00	14.50		7.50	1	1		1
 		CLEC to CLEC Conversion Charge without outside dispatch	1	<u> </u>	UEA	UREWO		87.72	36.36			t	7.86	 	 		
	4-WIRF	ANALOG VOICE GRADE LOOP	1					32	33.00					1			
		4-Wire Analog Voice Grade Loop - Zone 1	1	1	UEA	UEAL4	29.26	164.11	112.36	78.91	18.66		7.86	1	1		1
		4-Wire Analog Voice Grade Loop - Zone 2	1		UEA	UEAL4	34.25	164.11	112.36	78.91	18.66		7.86	1	1		
							020										

Version 3Q02: 09/06/02 Page 135 of 362

ONRONDI	LED	NETWORK ELEMENTS - Kentucky	,		,										nent: 2		bit: C
CATEGORY	,	RATE ELEMENTS	Interi	Zone	BCS	usoc			RATES(\$)			Submitted Elec	Submitted Manually	Charge - Manual Svc	Charge - Manual Svc		Charge - Manual Sv
CATEGORT		RAIE ELEMENIS	m	Zone	ВСЗ	0300			KATES(\$)			per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Order vs. Electronic- Disc 1st	Order vs. Electronic Disc Add'
							Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)	•	
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		l-Wire Analog Voice Grade Loop - Zone 3		3	UEA	UEAL4	85.06	164.11	112.36	78.91	18.66		7.86				
		Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		23.01									
		CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36				7.86				
2-W		SDN DIGITAL GRADE LOOP															
		2-Wire ISDN Digital Grade Loop - Zone 1			UDN	U1L2X	18.44	146.77	95.02	71.38	13.83		7.86				
		2-Wire ISDN Digital Grade Loop - Zone 2		3	UDN UDN	U1L2X U1L2X	25.08 42.87	146.77	95.02 95.02	71.38	13.83		7.86				
		2-Wire ISDN Digital Grade Loop - Zone 3 Order Coordination For Specified Conversion Time (per LSR)		3	UDN	OCOSL	42.87	146.77 23.01	95.02	71.38	13.83		7.86				
		CLEC to CLEC Conversion Charge without outside dispatch		-	UDN	UREWO		91.63	44.16				7.86				-
2.10		ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIDI E	1.00		UKLVVO		91.03	44.10				7.00				
2-44		Wire Unbundled ADSL Loop including manual service inquiry	A HIDEL	LOGI													-
	&	k facility reservation - Zone 1		1	UAL	UAL2X	10.82	141.98	79.73	69.02	11.47		7.86				
	&	k facility reservation - Zone 2		2	UAL	UAL2X	11.79	141.98	79.73	69.02	11.47		7.86				
	&	A facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UAL UAL	UAL2X OCOSL	12.87	141.98 23.01	79.73	69.02	11.47		7.86				
		Wire Unbundled ADSL Loop without manual service inquiry &			UAL	UCUSL		23.01									
	fa	acility reservaton - Zone 1		1	UAL	UAL2W	10.82	121.18	69.00	69.09	11.54		7.86				
	fa	Wire Unbundled ADSL Loop without manual service inquiry & acility reservaton - Zone 2		2	UAL	UAL2W	11.79	121.18	69.00	69.09	11.54		7.86				
	fa	Wire Unbundled ADSL Loop without manual service inquiry & acility reservaton - Zone 3		3	UAL	UAL2W	12.87	121.18	69.00	69.09	11.54		7.86				
		Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		23.01									
		CLEC to CLEC Conversion Charge without outside dispatch	<u></u>		UAL	UREWO		86.20	40.40				7.86				
2-W	2	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA Wire Unbundled HDSL Loop including manual service inquiry	IIBLE			11111 01/	0.75	454.54	00.00	00.00	44.54		7.00				
	2	A facility reservation - Zone 1 2 Wire Unbundled HDSL Loop including manual service inquiry		1	UHL	UHL2X	8.75	151.54	89.29	69.09	11.54		7.86				
	2	& facility reservation - Zone 2 2 Wire Unbundled HDSL Loop including manual service inquiry		2	UHL	UHL2X	9.56	151.54	89.29	69.09	11.54		7.86				
		k facility reservation - Zone 3		3	UHL	UHL2X	10.61	151.54	89.29	69.09	11.54		7.86				
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		23.01									
	а	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL2W	8.75	130.74	78.56	69.09	11.54		7.86				
	а	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2		2	UHL	UHL2W	9.56	130.74	78.56	69.09	11.54		7.86				
	а	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		3	UHL	UHL2W	10.61	130.74	78.56	69.09	11.54		7.86				
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		23.01									
<u> </u>		CLEC to CLEC Conversion Charge without outside dispatch	L	00-	UHL	UREWO		86.14	40.40	 			7.86			 	
4-W	4	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA Wire Unbundled HDSL Loop including manual service inquiry	IIBLE				40.05		100.50				7.00				
	4	and facility reservation - Zone 1 I-Wire Unbundled HDSL Loop including manual service inquiry		1	UHL	UHL4X	13.95	185.75	123.50	74.95	14.69		7.86				
	4	and facility reservation - Zone 2 I-Wire Unbundled HDSL Loop including manual service inquiry		2	UHL	UHL4X	15.68	185.75	123.50	74.95	14.69		7.86				
		and facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UHL UHL	UHL4X OCOSL	16.98	185.75 23.01	123.50	74.95	14.69		7.86				
	4	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL4W	13.95	164.95	114.04	77.32	15.80		7.86				
	4	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2		2	UHL	UHL4W	15.68	164.95	114.04	77.32	15.80		7.86				
	4	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		3	UHL	UHL4W	16.98	164.95	114.04	77.32	15.80		7.86				
		Order Coordination for Specified Conversion Time (per LSR)	1		UHL	OCOSL	10.00	23.01	117.07	77.02	10.00		7.00			1	
		CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.14	40.40				7.86			Ì	
4-W		DS1 DIGITAL LOOP			İ					† †						İ	
	4	I-Wire DS1 Digital Loop - Zone 1		1	USL	USLXX	86.47	306.69	174.44	65.83	14.55		7.86				

ONBONDE	D NETWORK ELEMENTS - Kentucky	1		ı	1 1							001		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Digital Loop - Zone 2			USL	USLXX	114.10	306.69	174.44	65.83	14.55		7.86				
	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	297.76	306.69	174.44	65.83	14.55		7.86				
	Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		23.01	40.04								
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		101.09	43.04								
4-WIR	E 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP 4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	27.59	157.81	106.06	78.91	18.66		7.86				
	4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	32.48	157.81	106.06	78.91	18.66		7.86				
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	36.37	157.81	106.06	78.91	18.66		7.86				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	27.59	157.81	106.06	78.91	18.66		7.86				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	32.48	157.81	106.06	78.91	18.66		7.86				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	36.37	157.81	106.06	78.91	18.66		7.86				
	Order Coordination for Specified Conversion Time (per LSR)		Ť	UDL	OCOSL	33.57	23.01	.00.00	. 5.51	.0.00					1	
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	27.59	157.81	106.06	78.91	18.66		7.86			1	1
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2		2	UDL	UDL64	32.48	157.81	106.06	78.91	18.66		7.86				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL	UDL64	36.37	157.81	106.06	78.91	18.66		7.86				
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		23.01									
	CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		102.13	49.75				7.86				
2-WIR	E Unbundled COPPER LOOP															
	2-Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 1		1	UCL	UCLPB	10.82	140.95	78.70	69.09	11.54		7.86				
	2-Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 2		2	UCL	UCLPB	11.79	140.95	78.70	69.09	11.54		7.86				
	2 Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	12.87	140.95	78.70	69.09	11.54		7.86				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
	2-Wire Unbundled Copper Loop/Short without manual service				LIOI DW	40.00	100.15	07.07	00.00	44.54		7.00				
	inquiry and facility reservation - Zone 1		1	UCL	UCLPW	10.82	120.15	67.97	69.09	11.54		7.86				
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2		2	UCL	UCLPW	11.79	120.15	67.97	69.09	11.54		7.86				
	2-Wire Unbundled Copper Loop/Short without manual service			UCL	UCLFVV	11.79	120.15	67.97	69.09	11.54		7.00				
	inquiry and facility reservation - Zone 3		3	UCL	UCLPW	12.87	120.15	67.97	69.09	11.54		7.86				
	Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLMC	12.07	9.00	9.00	09.09	11.54		7.00				
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.			OOL	OCLIVIC		3.00	3.00	t							
	inquiry and facility reservation - Zone 1		1	UCL	UCL2L	24.91	140.95	78.70	69.09	11.54		7.86				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.		<u> </u>	002	00222	2	1 10.00	70.70	00.00			7.00				
	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	36.94	140.95	78.70	69.09	11.54		7.86			1	
	2-Wire Unbundled Copper Loop/Long - includes manual svc.		† -		1	33.54			55.55	54					1	1
	inquiry and facility reservation - Zone 3		3	UCL	UCL2L	69.95	140.95	78.70	69.09	11.54		7.86			1	
	Order Coordination for Unbundled Copper Loops (per loop)		1	UCL	UCLMC		9.00	9.00								
	2-Wire Unbundled Copper Loop/Long - without manual service				i i											
	inquiry and facility reservation - Zone 1		1	UCL	UCL2W	24.91	120.15	67.97	69.09	11.54		7.86			<u></u>	<u> </u>
	2-Wire Unbundled Copper Loop/Long - without manual service							· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·						
	inquiry and facility reservation - Zone 2		2	UCL	UCL2W	36.94	120.15	67.97	69.09	11.54		7.86				
	2-Wire Unbundled Copper Loop/Long - without manual service												·		1	
	inquiry and facility reservation - Zone 3		3	UCL	UCL2W	69.95	120.15	67.97	69.09	11.54		7.86			ļ	ļ
	Order Coordination for Unbundled Copper Loops (per loop)		<u> </u>	UCL	UCLMC		9.00	9.00	ļ							
	CLEC to CLEC Conversion Charge without outside dispatch				LIDEILLO]						I	1
4 14/15	(UCL-Des)		<u> </u>	UCL	UREWO		97.23	42.48	 			7.86			-	ļ
4-WIR	E COPPER LOOP		1		+				 						 	
	4-Wire Copper Loop/Short - including manual service inquiry		4	UCL	UCL4S	16.92	170.31	108.06	74.95	14.69		7.86			I	1
	and facility reservation - Zone 1 4-Wire Copper Loop/Short - including manual service inquiry	-	+-	UUL	UUL48	16.92	170.31	108.06	74.90	14.69		7.80			 	1
	and facility reservation - Zone 2		2	UCL	UCL4S	17.36	170.31	108.06	74.95	14.69		7.86			1	
—	4-Wire Copper Loop/Short - including manual service inquiry			JUL	00140	17.30	170.31	100.00	14.53	14.09		1.00			t	
	and facility reservation - Zone 3		3	UCL	UCL4S	28.10	170.31	108.06	74.95	14.69		7.86			I	l
 	Order Coordination for Unbundled Copper Loops (per loop)	1	-	UCL	UCLMC	20.10	9.00	9.00	14.55	14.09		7.00			I	
	4-Wire Copper Loop/Short - without manual service inquiry and		†		002.00		5.50	5.50	†						1	1
1	facility reservation - Zone 1		1	UCL	UCL4W	16.92	149.52	97.33	74.95	14.69	ĺ	7.86				

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attach	nent: 2	Exhil	bit: C
CHECKEL	- NETWORK ELEMENTO ROMANY										Svc Order	Svc Order	Incremental		Incremental	
											Submitted	Submitted		Charge -	Charge -	Charge -
		Intori									Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													151	Add I	DISC ISI	DISC Add I
							Nonrec	urring	Nonrecurring	Disconnect		•	oss	Rates(\$)		•
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 2		2	UCL	UCL4W	17.36	149.52	97.33	74.95	14.69		7.86				
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 3		3	UCL	UCL4W	28.10	149.52	97.33	74.95	14.69		7.86				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 1		1	UCL	UCL4L	46.91	170.31	108.06	74.95	14.69		7.86				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	45.78	170.31	108.06	74.95	14.69		7.86				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
1 1	inquiry and facility reservation - Zone 3	1	3	UCL	UCL4L	171.34	170.31	108.06	74.95	14.69		7.86	Ì			I
	Order Coordination for Unbundled Copper Loops (per loop)	Ì		UCL	UCLMC		9.00	9.00								
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 1		1	UCL	UCL4O	46.91	149.52	97.33	74.95	14.69		7.86				
	4-Wire Unbundled Copper Loop/Long - without manual svc.	1		İ					1	1			İ	İ	İ	İ
1 1	inquiry and facility reservation - Zone 2	1	2	UCL	UCL4O	45.78	149.52	97.33	74.95	14.69		7.86	Ì			I
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL4O	171.34	149.52	97.33	74.95	14.69		7.86				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
	CLEC to CLEC Conversion Charge without outside dispatch															
	(UCL-Des)			UCL	UREWO		97.23	42.48				7.86				
LOOP MODIF																
				UAL, UHL, UCL,												
				UEQ, ULS, UEA,												
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire			UEANL, UDL, UDC,												
	pair less than or equal to 18k ft			UDN, UDL, USL	ULM2L		9.24	9.24				7.86				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire			, , , , , , , , , , , , , , , , , , , ,			_									
	greater than 18k ft			UCL, ULS, UEQ	ULM2G		342.24	342.24				7.86				
	Unbundled Loop Modification Removal of Load Coils - 4 Wire															
	less than or equal to 18K ft			UHL, UCL	ULM4L		9.24	9.24				7.86				
	Unbundled Loop Modification Removal of Load Coils - 4 Wire			,												
	pair greater than 18k ft			UCL	ULM4G		342.24	342.24				7.86				
	Fam greater treat to			UAL, UHL, UCL,												
				UEQ, UEF, ULS,												
				UEA, UEANL, UDL,												
	Unbundled Loop Modification Removal of Bridged Tap Removal,			UDC, UDN, UDL,												
	per unbundled loop			USL	ULMBT		10.47	10.47				7.86				
SUB-LOOPS		1	t	1					t				1			t
	Loop Distribution	1		İ					İ	İ			İ	İ	İ	İ
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-															
1 1	Up	1		UEANL	USBSA		207.91	207.91	I			7.86	Ì			I
	'															
1 1	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	1		UEANL	USBSB		12.50	12.50	I			7.86	Ì			I
	Sub-Loop - Per Building Equipment Room - CLEC Feeder	1							İ				İ			İ
1 1	Facility Set-Up	1		UEANL	USBSC		80.87	80.87	I			7.86	Ì			I
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel	1		İ					İ	İ			İ	İ	İ	İ
1 1	Set-Up	- 1		UEANL	USBSD		45.04	45.04	1			7.86				1
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -	1		İ					1	İ			İ	İ	İ	İ
1 1	Zone 1	1	1	UEANL	USBN2	6.34	85.03	39.05	59.81	7.90		7.86	Ì			I
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -	1		İ						1			İ	İ	İ	İ
	Zone 2	1	2	UEANL	USBN2	9.06	85.03	39.05	59.81	7.90		7.86				1
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -															
	Zone 3	1	3	UEANL	USBN2	14.82	85.03	39.05	59.81	7.90		7.86				
		1				-				1.00			İ			İ
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	1		UEANL	USBMC		9.00	9.00	I				1			I
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -	1					2.00	2.00	İ				İ			İ
1 1	Zone 1	1	1	UEANL	USBN4	8.14	102.31	56.32	65.24	10.88		7.86	Ì			I
\vdash	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -	1		İ					1			1	İ			1
				UEANL	USBN4	8.63	102.31	56.32	65.24	10.88		7.86				1

HINRH	NDI EI	NETWORK ELEMENTS - Kentucky												Attach	ment: 2	Evhil	oit: C
ONBOI	ADEEL	NETWORK ELEMENTS - Remucky	1									Svc Order	Svc Order	Incremental			Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGO	nev	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)				,				
CATEGO	JKI	RATE ELEMENTS	m	Zone	503	0300			KAILS(4)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
—							I	Nonrec	urring	Nonrecurring	Disconnect	-	l	066	Rates(\$)	l	
-							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -						FIISL	Add I	FIISL	Add I	SOMEC	SUMAN	SUMAN	SOWAN	SOWAN	SOWAN
		Zone 3		3	UEANL	USBN4	25.60	102.31	56.32	65.24	10.88		7.86				
\vdash		ZOTIE 3		3	UEAINL	USBIN4	23.60	102.31	30.32	03.24	10.00		7.00				
		Onder Consideration for Habrardtad Cub Large and sub-large asia			UEANL	USBMC		9.00	9.00								
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBR2	2.57	68.35	22.36	50.04	7.00		7.86				
\vdash		Sub-Loop 2-Wire Intrabuilding Network Cable (INC)		-	UEANL	USBRZ	2.57	68.35	22.36	59.81	7.90		7.86				
		Onder Consideration for Habrardtad Cub Large and sub-large asia			UEANL	USBMC		9.00	9.00								
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair	 		UEANL		4.98			CE 04	10.88		7.86				
		Sub-Loop 4-Wire Intrabuilding Network Cable (INC)			UEANL	USBR4	4.98	76.49	30.51	65.24	10.88		7.86				
		Order Coordination for Unbundled Cub Lases are sub-lases as	1		LIEANI	USBMC		0.00	0.00				1		I		
\vdash		Order Coordination for Unbundled Sub-Loops, per sub-loop pair		1	UEANL UEF		5.45	9.00 85.03	9.00 39.05	59.81	7.90	-	7.86		-		
\vdash		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1			UEF	UCS2X		85.03 85.03							 		
\vdash		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	<u> </u>	2		UCS2X	7.06 9.67		39.05	59.81	7.90	1	7.86	-	 	-	
\vdash		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		3	UEF	UCS2X	9.67	85.03	39.05	59.81	7.90	-	7.86		1		
		Onder Consideration for Habitandle 10.1.1	1		luce.	LICDMO		0.00	0.00				1		I		
\vdash		Order Coordination for Unbundled Sub-Loops, per sub-loop pair	⊢. −	1	UEF UEF	USBMC UCS4X	7.09	9.00 102.31	9.00 56.32	65.24	10.88	1	7.86	-	 	-	
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1															
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	l I		UEF	UCS4X	8.66	102.31	56.32	65.24	10.88		7.86				
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	I	3	UEF	UCS4X	19.40	102.31	56.32	65.24	10.88		7.86				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		9.00	9.00								
	Unbunc	dled Sub-Loop Modification															
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load															
		Coil/Equip Removal per 2-W PR			UEF	ULM2X		5.23	5.23				7.86				
		Unbundled Sub-loop Modification - 4-W Copper Dist Load															
		Coil/Equip Removal per 4-W PR			UEF	ULM4X		5.23	5.23				7.86				
		Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged															
		Tap Removal, per PR unloaded			UEF	ULM4T		7.97	7.97				7.86				
!	Unbund	lled Network Terminating Wire (UNTW)															
		Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.53	23.51	23.51				7.86				
$\perp \perp \prime$		k Interface Device (NID)															
		Network Interface Device (NID) - 1-2 lines			UENTW	UND12		73.53	49.47				7.86				
		Network Interface Device (NID) - 1-6 lines			UENTW	UND16		115.96	91.91				7.86				
		Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		8.56	8.56				7.86				
		Network Interface Device Cross Connect - 4W			UENTW	UNDC4		8.56	8.56				7.86				
SUB-LO																	
\coprod	Sub-Lo	op Feeder	ļ														
		USL-Feeder, DS0 Set-up per Cross Box location - CLEC	1		UEA,										1		
igsquare		Distribution Facility set-up	ļ		UDN,UCL,UDL,UDC	USBFW		207.91					7.86				
1 T		USL Feeder - DS0 Set-up per Cross Box location - per 25 pair	1		UEA,								1		_		
igsquare		set-up	ļ		UDN,UCL,UDL,UDC	USBFX		12.50	12.50				7.86				
igsquare		USL Feeder DS1 Set-up at DSX location, per DS1 termination	ļ		USL	USBFZ		527.98	11.32				7.86				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice	1												1		
		Grade - Zone 1	<u> </u>	1	UEA	USBFA	7.67	114.83	64.61	72.34	17.21		7.86				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice	1											1		I	
		Grade - Zone 2]	2	UEA	USBFA	9.70	114.83	64.61	72.34	17.21		7.86				
l T		Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,	1	1									1				
		Voice Grade - Zone 3	<u> </u>	3	UEA	USBFA	19.53	114.83	64.61	72.34	17.21		7.86	<u> </u>	<u> </u>		
		Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		23.01									
		Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice	1											1		I	
		Grade - Zone 1	<u> </u>	1	UEA	USBFB	7.67	114.83	64.61	72.34	17.21		7.86				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice															
		Grade - Zone 2	<u>L</u>	2	UEA	USBFB	9.70	114.83	64.61	72.34	17.21	<u></u>	7.86	<u> </u>	<u> </u>	<u> </u>	
		Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice															
		Grade - Zone 3	1	3	UEA	USBFB	19.53	114.83	64.61	72.34	17.21		7.86				
		Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL	<u> </u>	23.01									
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
		Voice Grade - Zone 1	1	1	UEA	USBFC	7.67	114.83	64.61	72.34	17.21		7.86				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
		Voice Grade - Zone 2	1	2	UEA	USBFC	9.70	114.83	64.61	72.34	17.21	1	7.86				

UNBUNDI F	D NETWORK ELEMENTS - Kentucky												Attach	ment: 2	Fyhil	oit: C
SHESHEL											Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
1		Intor			1						Elec	Manually	Manual Svc	Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
1		m			1								Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							N		T 51	. B'						
					1	Rec	Nonrec		Nonrecurring		201150	001141		Rates(\$)	001111	001441
-	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse		<u> </u>		+		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Battery. Voice Grade - Zone 3		3	UEA	USBFC	19.53	114.83	64.61	72.34	17.21		7.86				
	Order Coordination For Specified Conversion Time, per LSR		3	UEA	OCOSL	19.55	23.01	04.01	72.34	17.21		7.00				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice			OLA	CCCCL		20.01									
	Grade - Zone 1		1	UEA	USBFD	22.82	131.73	79.98	81.82	51.56		7.86				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 2		2	UEA	USBFD	27.24	131.73	79.98	81.82	51.56		7.86				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice															
	Grade - Zone 3		3	UEA	USBFD	61.41	131.73	79.98	81.82	51.56		7.86				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		23.01									
1 1	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice	1		1154	LICDET	20.00	404 70	70.00	04.65	£4.50		7.00	1	1	1	
\vdash	Grade - Zone 1	 	1	UEA	USBFE	22.82	131.73	79.98	81.82	51.56	1	7.86	 	 	 	
1 1	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2		2	UEA	USBFE	27.24	131.73	79.98	81.82	51.56		7.86				
 	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			OLA	OODI L	21.24	131.73	19.90	01.02	31.30		1.00	1	1		-
	Grade - Zone 3		3	UEA	USBFE	61.41	131.73	79.98	81.82	51.56		7.86				
	Order Coordination For Specified Conversion Time, Per LSR		Ť	UEA	OCOSL	01.41	23.01	70.00	01.02	01.00		7.50				
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	13.00	131.79	80.04	74.16	16.60		7.86				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN	USBFF	16.95	131.79	80.04	74.16	16.60		7.86				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	28.95	131.79	80.04	74.16	16.60		7.86				
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		23.01									
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDC	USBFS	13.00	131.79	80.04	74.16	16.60		7.86				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	16.95	131.79	80.04	74.16	16.60		7.86				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	28.95	131.79	80.04	74.16	16.60		7.86				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	USL	USBFG	62.57	125.43	73.68	81.82	21.56		7.86				
-	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2			USL USL	USBFG USBFG	87.71 273.33	125.43 125.43	73.68 73.68	81.82 81.82	21.56 21.56	-	7.86 7.86				
-	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3 Order Coordination For Specified Conversion Time, Per LSR		3	USL	OCOSL	213.33	23.01	73.00	01.02	21.50	-	7.00				
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	6.44	105.31	53.57	71.16	13.61		7.86				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		<u> </u>	002	002	0	.00.01	00.01	7 11.10	10.01		7.00				
	2		2	UCL	USBFH	5.78	105.31	53.57	71.16	13.61		7.86				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone															
	3		3	UCL	USBFH	4.25	105.31	53.57	71.16	13.61		7.86				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		23.01									
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1			UCL	USBFJ	11.33	125.55	73.80	77.12	16.86		7.86				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2	<u> </u>	2		USBFJ	10.18	125.55	73.80	77.12	16.86		7.86				
\vdash	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3	 	3	UCL	USBFJ	10.32	125.55	73.80	77.12	16.86		7.86	 	 	 	-
	Order Coordination For Specified Conversion Time, per LSR Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	 	1	UCL UDL	OCOSL USBFN	20.78	23.01 125.43	73.68	81.82	21.56		7.86	-	-	-	-
 	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	1	2	UDL	USBFN	26.41	125.43	73.68	81.82	21.56	-	7.86	 	 	 	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	†	_	UDL	USBFN	23.10	125.43	73.68	81.82	21.56	1	7.86	1	1	1	1
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -	1	Ĭ		1	200	.20.70	. 0.30	552	250			İ	İ	Ì	
	Zone 1	1	1	UDL	USBFO	20.78	125.43	73.68	81.82	21.56		7.86	1	1	1	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 2		2	UDL	USBFO	26.41	125.43	73.68	81.82	21.56		7.86				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 3	ļ	3	UDL	USBFO	23.10	125.43	73.68	81.82	21.56		7.86	ļ	ļ	ļ	
\vdash	Order Coordination For Specified Time Conversion, per LSR	ļ		UDL	OCOSL		23.01									
1 1	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -	1	_	UDL	LIGHED	00.70	405 40	70.00	04.00	04.50		7.00	1	1	1	
\vdash	Zone 1 Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -	 	1	UUL	USBFP	20.78	125.43	73.68	81.82	21.56	-	7.86	-	-	-	1
1 1	Zone 2		2	UDL	USBFP	26.41	125.43	73.68	81.82	21.56		7.86				
 	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -	1	 _	ODL .	JODITE	20.41	120.43	13.00	01.02	21.30	-	7.00	 	 	 	
1 1	Zone 3	1	3	UDL	USBFP	23.10	125.43	73.68	81.82	21.56		7.86	1	1	1	
	Order Coordination For Specified Conversion Time, per LSR		Ť	UDL	OCOSL		23.01	. 2.00	502	_::00						
SUB-LOOPS	, , , , , , , , , , , , , , , , , , , ,				1											1
Sub-L	pop Feeder															
	Sub Loop Feeder - DS3 - Per Mile Per Month			UE3	1L5SL	15.38										
	Sub Loop Feeder - DS3 - Facility Termination Per Month			UE3	USBF1	346.30	3,402.59	407.14	160.86	91.19		7.86				

UNBUN	NDLE	D NETWORK ELEMENTS - Kentucky											•		ment: 2		bit: C
CATEGO	DRY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect		1	oss	Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Sub Loop Feeder – STS-1 – Per Mile Per Month	I		UDLSX	1L5SL	15.38										
-		Sub Loop Feeder - STS-1 - Facility Termination Per Month			UDLSX	USBF7	372.80 11.67	3,402.59	407.14	160.86	91.19		7.86				
-		Sub Loop Feeder – OC-3 – Per Mile Per Month Sub Loop Feeder - OC-3 - Facility Termination Protection Per			UDLO3	1L5SL	11.07										
		Month	l ,		UDLO3	USBF5	58.27										
		Sub Loop Feeder - OC-3 - Facility Termination Per Month	i		UDLO3	USBF2	564.68	3,402.59	407.14	160.86	91.19		7.86			İ	1
		Sub Loop Feeder - OC-12 - Per Mile Per Month	-		UDL12	1L5SL	14.36										
		Sub Loop Feeder - OC-12 - Facility Termination Protection Per															
		Month	I		UDL12	USBF6	658.35										
-		Sub Loop Feeder - OC-12 - Facility Termination Per Month	<u> </u>		UDL12	USBF3	1,778.00	3,402.59	407.14	160.86	91.19		7.86				
\vdash		Sub Loop Feeder - OC-48 - Per Mile Per Month Sub Loop Feeder - OC-48 - Facility Termination Protection Per		 	UDL48	1L5SL	47.11	-		+				-		-	
		Month	L		UDL48	USBF9	330.39									1	
		Sub Loop Feeder - OC-48 - Facility Termination Per Month	l i	<u> </u>	UDL48	USBF4	1,533.00	3,587.59	407.14	160.86	91.19		7.86				
		Sub Loop Feeder - OC-12 Interface On OC-48	П		UDL48	USBF8	372.76	804.96	407.14	160.86	91.19		7.86				
UNBUNE	DLED L	LOOP CONCENTRATION															
		Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	423.72	359.34	359.34				7.86				
		Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	51.60	149.72	149.72				7.86				
-		Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	460.27	359.34	359.34				7.86				
-		Unbundled Loop Concentration - System B (TR303) Unbundled Loop Concentration - DS1 Loop Interface Card			ULC ULC	UCT3B UCTCO	86.95 4.90	149.72 71.69	149.72 51.51	22.99	6.00		7.86 7.86				
		Unbundled Loop Concentration - ISDN Loop Interface (Brite			ULC	UCTCO	4.90	71.69	51.51	22.99	6.00		7.00			1	
		Card)			UDN	ULCC1	7.78	16.59	16.50	8.42	8.37		7.86				
		Unbundled Loop Concentration - UDC Loop Interface (Brite								****						İ	İ
		Card)			UDC	ULCCU	7.78	16.59	16.50	8.42	8.37		7.86				
		Unbundled Loop Concentration2 Wire Voice-Loop Start or															
		Ground Start Loop Interface (POTS Card)			UEA	ULCC2	1.95	16.59	16.50	8.42	8.37		7.86				
		Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery			1154	ULCCR	11.58	40.50	16.50	8.42	8.37		7.00				
-		Loop Interface (SPOTS Card) Unbundled Loop Concentration - 4 Wire Voice Loop Interface			UEA	ULCCR	11.58	16.59	16.50	8.42	8.37		7.86				
		(Specials Card)			UEA	ULCC4	6.90	16.59	16.50	8.42	8.37		7.86				
		Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	33.74	16.59	16.50	8.42	8.37		7.86			İ	İ
		Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop															
		Interface			UDL	ULCC7	10.23	16.59	16.50	8.42	8.37		7.86				
		Unbundled Loop Concentration - Digital 56 Kbps Data Loop															
		Interface			UDL	ULCC5	10.23	16.59	16.50	8.42	8.37		7.86				
		Unbundled Loop Concentration - Digital 64 Kbps Data Loop			UDL	111.000	40.00	40.50	40.50	0.40	8.37		7.00				
LINE OT	HED E	Interface PROVISIONING ONLY - NO RATE			UDL	ULCC6	10.23	16.59	16.50	8.42	8.37		7.86				
ONE OIL	∟ıx, ı	NID - Dispatch and Service Order for NID installation		1	UENTW	UNDBX	0.00	0.00									
		UNTW Circuit Id Establishment, Provisioning Only - No Rate		1	UENTW	UENCE	0.00	0.00									
		, , , , , , , , , , , , , , , , , , ,			UEANL,UEF,UEQ,U		0.00										
		Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN	0.00	0.00									
UNE OT	HER, F	PROVISIONING ONLY - NO RATE															
					l <u>.</u> <u>.</u>												
		Habitandlad Contact Name Description in Only an arts			UAL,UCL,UDC,UDL,	LINIEGNI	0.00	0.00									
-		Unbundled Contact Name, Provisioning Only - no rate Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no			UDN,UEA,UHL,ULC	UNECN	0.00	0.00									
		rate			UEA,UDN,UCL,UDC	USBEO	0.00	0.00									
		Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no		1	02,403,4002,030	002. Q	0.00	0.00									
		rate	L	L	UEA,USL,UCL,UDL	USBFR	0.00	0.00		<u> </u>		<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>
		Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
		Unbundled DS1 Loop - Expanded Superframe Format option -									· · · · · · · · · · · · · · · · · · ·						
		no rate		<u> </u>	USL	CCOEF	0.00	0.00									
HIGH CA	APACI	TY UNBUNDLED LOCAL LOOP		<u> </u>										-		1	1
		High Capacity Unbundled Local Loop - DS3 - Per Mile per month		1	UE3	1L5ND	9.25										
 		High Capacity Unbundled Local Loop - DS3 - Facility		 	OLO	ILOND	9.25			1							
		Termination per month	l		UE3	UE3PX	308.31	551.38	338.08	173.00	120.42		7.86			I	I

UNBUNDL	ED NETWORK ELEMENTS - Kentucky													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	17.1 0 7.111 11 11 17 0.70 1 B 147						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per			LIDLOV	41 END	0.05										
	month High Capacity Unbundled Local Loop - STS-1 - Facility			UDLSX	1L5ND	9.25										
	Termination per month			UDLSX	UDLS1	320.51	551.38	338.08	173.00	120.42		7.86				
LOOP MAKE				UDLSX	UDLST	320.31	331.36	330.00	173.00	120.42		7.00				
LOOI MAKE	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).			UMK	UMKLW		23.40	23.40								
	Loop Makeup - Preordering With Reservation, per spare facility															
	queried (Manual).			UMK	UMKLP		24.85	24.85								
	Loop MakeupWith or Without Reservation, per working or															
	spare facility queried (Mechanized)			UMK	PSUMK		0.67	0.67								
	DEDICATED TRANSPORT : INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	I-:!!!:		d balanı DÖ2 anı		CTC 4 faces										
	E: INTEROFFICE CHANNEL DEDICATED TRANSPORT - MINIMU ROFFICE CHANNEL - DEDICATED TRANSPORT	m billin	g perio	oa - below D53=one	e month, DS3/	515-1=rour mo	ntns		-							
INTE	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -				+						-					-
	Per Mile per month Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Interoffice Channel - Dedicated Transport - 2- Wire Voice Grade -			U1TVX	1L5XX	0.01										
	Facility Termination			U1TVX	U1TV2	29.11	47.34	31.78	22.77	8.75		7.86				
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per month			U1TVX	1L5XX	0.01	-									
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat					9.0.										
	Facility Termination			U1TVX	U1TR2	29.11	47.34	31.78	22.77	8.75		7.86				
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -															
	Per Mile per month			U1TVX	1L5XX	0.01										
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade															
	- Facility Termination			U1TVX	U1TV4	25.86	47.34	31.78	22.77	8.75		7.86				
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile			U1TDX	1L5XX	0.0115										
	per month Interoffice Channel - Dedicated Transport - 56 kbps - Facility			UTIDX	ILDAX	0.0115										-
	Termination			U1TDX	U1TD5	20.97	47.35	31.78	22.77	8.75		7.86				
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile		1	UTIDA	01103	20.91	47.33	31.76	22.11	0.73		7.00				
	per month			U1TDX	1L5XX	0.0115										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility				1				†							
	Termination			U1TDX	U1TD6	20.97	47.35	31.78	22.77	8.75		7.86				
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			U1TD1	1L5XX	0.23										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility			l	L				ı T			l				
	Termination		<u> </u>	U1TD1	U1TF1	96.04	105.52	98.46	23.09	20.49		7.86			ļ	
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			U1TD3	1L5XX	4.97			j							
	Interoffice Channel - Dedicated Transport - DS3 - Facility		!	U11D3	ILOXX	4.97			 						 	-
	Termination per month			U1TD3	U1TF3	1,175.15	335.40	219.24	89.57	87.75		7.86				
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per			U1TS1	1L5XX	4.97	000.40	210.24	55.57	01.10		7.50				
	month Interoffice Channel - Dedicated Transport - STS-1 - Facility	1	<u> </u>	01191	ILOAX	4.97			 		-				1	
	Termination			U1TS1	U1TFS	1,149.51	335.40	219.24	89.57	87.75		7.86				
LOCA	AL CHANNEL - DEDICATED TRANSPORT	1		0.101	00	1,140.01	333.40	210.24	03.37	01.73	1	7.00			1	1
	E: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	g perio	d - bel	ow DS3=one month	n, DS3/STS-1=f	our months			† †							
	Local Channel - Dedicated - 2-Wire Voice Grade			ULDVX	ULDV2	18.57	265.78	46.96	46.79	4.98		7.86			<u> </u>	
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat			ULDVX	ULDR2	18.57	265.78	46.96	46.79	4.98		7.86				
	Local Channel - Dedicated - 4-Wire Voice Grade			UNDVX	ULDV4	19.86	266.48	47.65	47.54	5.73		7.86				
	Local Channel - Dedicated - DS1 - Zone 1		1	ULDD1	ULDF1	40.46	209.60	176.51	30.21	21.07		7.86				
	Local Channel - Dedicated - DS1 - Zone 2		2	ULDD1	ULDF1	43.39	209.60	176.51	30.21	21.07		7.86			ļ	
 	Local Channel - Dedicated - DS1 - Zone 3	1	3	ULDD1	ULDF1 1L5NC	164.50	209.60	176.51	30.21	21.07	1	7.86				1
	Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination		!	ULDD3 ULDD3	ULDF3	8.74 576.05	551.38	338.08	173.00	120.42		7.86			 	
	Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	8.74	331.38	330.08	173.00	120.42		1.80		1	1	
	Local Channel - Dedicated - STS-1 - Fer Mile per Month Local Channel - Dedicated - STS-1 - Facility Termination		<u> </u>	ULDS1	ULDFS	543.24	551.38	338.08	173.00	120.42		7.86		 	+	
DARK FIBER			1		1022.0	5-10.24	501.00	000.00	170.00	120.42	1	7.00		 	 	1

UNBUNDLE	D NETWORK ELEMENTS - Kentucky										•			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Monroe	rrina	Nonrecurring	Disconnect			220	Rates(\$)		<u></u>
						Rec	Nonrec First	Add'l	First	Add'l	COMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction						FIISL	Auu i	FIISL	Add I	SOWIEC	SUMAN	SOWAN	SOWAN	SUMAN	SOWAN
	Thereof per month - Local Channel			UDF	1L5DC	47.01										
	NRC Dark Fiber - Local Channel			UDF	UDFC4	47.01	732.53	192.67	377.27	241.67		7.86				+
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction			00.	02.0.		7 02.00	102.01	011.21	211101		7.00				
	Thereof per month - Interoffice Channel			UDF	1L5DF	30.74										
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		732.53	192.67	377.27	241.67		7.86				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Loop			UDF	1L5DL	47.01										
	NRC Dark Fiber - Local Loop			UDF	UDFL4		732.53	192.67	377.27	241.67		7.86				
8XX ACCESS	TEN DIGIT SCREENING															
	8XX Access Ten Digit Screening, Per Call			OHD		0.0006478										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserved			OHD	N8R1X		4.14	0.70				7.86				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translations			OHD			8.78	1.18	7.08	0.86		7.86				
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translations			OHD	N8FTX		8.78	1.18	7.08	0.86		7.86				
	8XX Access Ten Digit Screening, Customized Area of Service								7.00	0.80						
	Per 8XX Number 8XX Access Ten Digit Screening, Multiple InterLATA CXR			OHD	N8FCX		4.14	2.07				7.86				
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		4.85	2.78				7.86				
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		4.85	0.70				7.86				
	8XX Access Ten Digit Screening, Call Handling and Destination															
	Features			OHD	N8FDX		4.14	4.14				7.86				
	8XX Access Ten Digit Screening w/ 8FL No. Delivery,			OHD		0.0006478										
LINE NECES	8XX Access Ten Digit Screening, w/ POTS No. Delivery,			OHD		0.0006478										
LINE INFORM	ATION DATA BASE ACCESS (LIDB) LIDB Common Transport Per Query			OQT		0.000023			ļ						-	
	LIDB Validation Per Query			OQU		0.000023			+						-	ļ
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX	0.0137322	55.12		67.59			7.86				-
SIGNALING (041, 040	THE BA		00.12		07.00			7.00				
1	CCS7 Signaling Connection, Per 56 Kbps Facility			UDB	TPP++	20.71	43.56	43.56	22.45	22.45						
	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	151.39										
	CCS7 Signaling Usage, Per TCAP Message			UDB		0.0000656										
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	20.71	43.56	43.56	22.45	22.45		7.86				
	CCS7 Signaling Connection, Per link (B link) (also known as D															
	link)			UDB	TPP++	20.71	43.56	43.56	22.45	22.45		7.86				
	CCS7 Signaling Usage, Per ISUP Message		<u> </u>	UDB	OTUEO	0.0000164										
	CCS7 Signaling Usage Surrogate, per link per LATA CCS7 Signaling Point Code, per Originating Point Code		<u> </u>	UDB	STU56	751.08			1							
	Establishment or Change, per STP affected			UDB	CCAPO		46.02	46.02	56.43	56.43		7.86				
	CCS7 Signaling Point Code, per Destination Point Code		1	LIDD	00455		40.00	40.00	50.40	50.40		7.00				
E911 SERVICE	Establishment or Change, Per Stp Affected		 	UDB	CCAPD		46.02	46.02	56.43	56.43		7.86		1	!	
ESTI SEKVICI	Local Channel - Dedicated - 2-wr Voice Grade		<u> </u>		+	18.57	265.78	46.96	46.79	4.98		7.86		-	-	
 	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile	1	-		+	0.0115	200.78	46.96	46.79	4.98	1	7.86			 	1
 	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility	-			+	0.0115			 						t	
	Termination		1		I	29.11	47.34	31.78	22.77	8.75		7.86			I	
	Local Channel - Dedicated - DS1 - Zone 1	1			1	40.46	209.60	176.51	30.21	21.07		7.86			1	
	Local Channel - Dedicated - DS1 - Zone 2				1	43.39	209.60	176.51	30.21	21.07		7.86		İ	1	
	Local Channel - Dedicated - DS1 - Zone 3					164.50	209.60	176.51	30.21	21.07		7.86				
	Interoffice Transport - Dedicated - DS1 Per Mile					0.23										
	Interoffice Transport - Dedicated - DS1 Per Facility Termination					96.04	105.52	98.46	23.09	20.49		7.86				
CALLING NAM	ME (CNAM) SERVICE				1											
	CNAM For DB Owners - Service Establishment			OQV			25.34	25.34	23.30	23.30		7.86				
	CNAM For Non DB Owners - Service Establishment			OQV			25.34	25.34	23.30	23.30		7.86				
	CNAM For DB Owners - Service Provisioning With Point Code															
	Establishment			OQV	_1		1,591.54	1,177.08	431.95	317.61		7.86			1	<u> </u>

UNBUNDLE	D NETWORK ELEMENTS - Kentucky													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CNAM For Non DB Owners - Service Provisioning With Point															
	Code Establishment			OQV			546.40	393.74	438.93	317.61		7.86				
	CNAM for DB Owners, Per Query			OQV		0.0010348										
	CNAM for Non DB Owners, Per Query			OQV		0.0010348										
	CNAM (Non-Databs Owner), NRC, applies when using the															
	Character Based User Interface (CHUI)			OQV	CDDCH		595.00	595.00				7.86				
LNP Query Ser																
	LNP Charge Per query					0.0008695										
	LNP Service Establishment Manual						13.82	13.82	12.71	12.71		7.86				
	LNP Service Provisioning with Point Code Establishment						953.27	487.00	431.95	317.61		7.86				
OPERATOR CA	ALL PROCESSING		<u> </u>						ļļ					ļ	ļ	ļ
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.20										
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24										
	Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20										
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB					0.20										
INWARD ORES	RATOR SERVICES					0.20										
INVVARD OF EN	Inward Operator Services - Verification, Per Call					1.00										
	Inward Operator Services - Verification, Fer Carr Inward Operator Services - Verification and Emergency Interrupt					1.00										
	- Per Call					1.95										
BRANDING - O	PERATOR CALL PROCESSING															
Facility	based CLEC															
	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				7.86				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN				CBAOL		500.00	500.00				7.86				
UNEP (
	Recording of Custom Branded OA Announcement						7,000.00	7,000.00				7.86				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN						500.00	500.00				7.86				
DIRECTORY AS	SSISTANCE SERVICES						000.00	000.00				7.00				
	TORY ASSISTANCE ACCESS SERVICE															
	Directory Assistance Access Service Calls, Charge Per Call					0.275										
DIRECT	TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (E	DACC)				0.2.0										
	Directory Assistance Call Completion Access Service (DACC), Per Call Attempt					0.10										
DIRECTORY AS	SSISTANCE SERVICES					0.10										
	TORY ASSISTANCE DATA BASE SERVICE (DADS)		1		+											
	Directory Assistance Data Base Service Charge Per Listing					0.04										
	Directory Assistance Data Base Service, per month				DBSOF	150.00			1					1	1	1
BRANDING - D	IRECTORY ASSISTANCE		1		1									İ	1	1
	/ Based CLEC		İ	İ					i				İ	İ	İ	1
	Recording and Provisioning of DA Custom Branded															
	Announcement			AMT	CBADA		6,000.00	6,000.00				7.86				1
	Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00				7.86				
UNEP (
	Recording of DA Custom Branded Announcement						3,000.00	3,000.00				7.86				
	Loading of DA Custom Branded Announcement per Switch per OCN						1,170.00	1,170.00				7.86				
SELECTIVE RO			1		1		.,170.00	.,170.00				7.50	1	1	1	t
	Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		93.53	93.53	15.58	15.58		7.86				
			 		JONOR		93.33	93.33	15.58	10.08		1.00		 	 	
VIDTUAL COLL		1	1	1	1	1			1	1	l	ĺ	i	1	1	1
VIRTUAL COLI				AMTES	FΔF		2 /10 00	2 /10 00	1.01	1 01		7 00				
VIRTUAL COLI	Virtual Collocation - Application Cost			AMTES	EAF		2,419.86	2,419.86	1.01	1.01		7.86				
VIRTUAL COLI				AMTFS AMTFS AMTFS	EAF ESPCX ESPVX	7.99	2,419.86 1,729.11	2,419.86 1,729.11	1.01 45.16	1.01 45.16		7.86 7.86				

UNBUNDLE	D NETWORK ELEMENTS - Kentucky													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring					Rates(\$)		
	Virtual Collegation Coble Support Structure, per entrance					1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation - Cable Support Structure, per entrance cable			AMTFS	ESPSX	17.38										
	Virtual Collocation - 2-wire Cross Connects (loop)			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, AMTFS, UDL, UNCVX, UNCDX, UNCNX	UEAC2	0.0309	24.68	23.68	12.14	10.95		7.86				
	Virtual Collocation - 2-wire Cross Connects (100p)			UNCINA	UEACZ	0.0309	24.00	23.00	12.14	10.95		7.00				
	Virtual Collocation - 4-wire Cross Connects (loop)			UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX AMTFS.UDL12.	UEAC4	0.0619	24.88	23.82	12.77	11.46		7.86				
	Virtual Collocation - 2-Fiber Cross Connects			UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC2F	3.80	41.94	30.51	14.76	11.84		7.86				
	virtual conceanors 24 per cross connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12,	ONOZI	3.00	41.54	30.31	14.70	11.04		7.00				
	Virtual Collocation - 4-Fiber Cross Connects			ULD48, UDF	CNC4F	7.59	51.29	39.87	19.41	16.49		7.86				
	Virtual collocation - Special Access & UNE, cross-connect per DS1			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL, UNLD1	CNC1X	1.48	44.23	31.98	12.81	11.57						
	Virtual collocation - Special Access & UNE, cross-connect per DS3			USL,ULC,AMTFS,U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	18.89	41.93	30.51	14.75	11.83						
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable						41.33	30.31	14.73	11.00						
	Support Structure, per linear foot			AMTFS	VE1CB	0.003										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0045										
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable			AMTFS	VE1CC		535.55									
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable			AMTFS	VE1CE		535.55									
	Virtual Collocation Cable Records - per request			AMTFS	VE1BA		1,524.45	980.01	267.02	267.02						
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable record			AMTFS	VE1BB		656.37	656.37	379.70	379.70						
	Virtual Collocation Cable Records - VG/DS0 Cable, per each 100 pair			AMTFS	VE1BC		9.65	9.65	11.84	11.84						
	Virtual Collocation Cable Records -DS1, per T1TIE			AMTES	VE1BD		4.52	4.52	5.54	5.54						
	Virtual Collocation Cable Records - DS3, per T3TIE Virtual Collocation Cable Records - Fiber Cable, per 99 fiber		 	AMTFS	VE1BE		15.81	15.81	19.39	19.39		1			1	
	records			AMTFS	VE1BF		169.63	169.63	154.85	154.85						
	Virtual collocation - Security Escort - Basic, per half hour		<u> </u>	AMTES	SPTBX		33.98	21.53			-					
	Virtual collocation - Security Escort - Overtime, per half hour Virtual collocation - Security Escort - Premium, per half hour			AMTFS AMTFS	SPTOX SPTPX		44.26 54.54	27.81 34.09	 			-				
	Virtual collocation - Security Escort - Premium, per half hour		 	AMTFS	CTRLX		56.07	21.53				 				
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		73.23	27.81								
	Virtual collocation - Maintenance in CO - Overtime, per han nour			AMTFS	SPTPM		90.39	34.09								
VIRTUAL COL				-			22.20	2 30								Ì

UNBUNDLE	D NETWORK ELEMENTS - Kentucky			I .	T	1							Attachr			bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2- Wire Analog - Res			UEPSR	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus			UEPSB	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86				
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire															
	ISDN Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSX	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86				
	ISDN			UEPTX	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			UEPEX	VE1R4	1.48	44.23	31.98	12.81	11.57		7.86				
VIRTUAL COL																
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.309	24.68	23.68	12.14	10.95		7.86				
PHYSICAL CO	LOCATION Physical Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	PE1LS	0.0333	24.68	23.68	12.14	10.95		7.86				
AIN SELECTIV	E CARRIER ROUTING			, , , , , , , , , , , , , , , , , , , ,												
	Regional Service Establishment			SRC	SRCEC		193,401.00	193,401.00	9,483.34	9,483.34		7.86				
	End Office Establishment			SRC	SRCEO		194.09	194.09	0.85	0.85		7.86				└
	Line/Port NRC, per end user			SRC	SRCLP	0.0007500	2.06	2.06				7.86				└
AIN BELLEO	Query NRC, per query JTH AIN SMS ACCESS SERVICE			SRC		0.0037502										
AIN - BELLSO	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			A1N	CAMSE		43.55	43.55	44.93	44.93		7.86				
	·															
	AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access			A1N A1N	CAMDP CAM1P		8.64 8.64	8.64 8.64	10.03 10.03	10.03 10.03		7.86 7.86				├
	AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code			A1N	CAMAU		38.65	38.65	29.88	29.88		7.86				
	AIN SMS Access Service - Security Card, Per User ID Code,															
	Initial or Replacement AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)			A1N	CAMRC	0.0025	75.08	75.08	12.93	12.93		7.86				
	AIN SMS Access Service - Storage, Fer Onit (100 Kilobytes) AIN SMS Access Service - Session, Per Minute					0.666										-
	AIN SMS Access Service - Company Performed Session, Per Minute					0.4608										
AIN - BELL SO	JTH AIN TOOLKIT SERVICE				+	0.4008										
AII BEEEGO	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			CAM	BAPSC		43.55	43.55	44.93	44.93		7.86				
	AIN Toolkit Service - Training Session, Per Customer			CAIVI	BAPVX		8,436.93	8,436.93	44.55	44.55		7.86				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt				BAPTT		8.64	8.64	10.03	10.03		7.86				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		8.64	8.64	10.03	10.03		7.86				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		8.64	8.64	10.03	10.03		7.86				
	DN, OIT-HOOK Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				ВАРТО		51.01	51.01	18.50	18.50		7.86				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPTC		51.01	51.01	18.50	18.50		7.86				
	DN, Feature Code AIN Toolkit Service - Query Charge, Per Query				BAPTF	0.0549207	51.01	51.01	18.50	18.50	-	7.86				
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.0066492										

UNBU	JNDLE	D NETWORK ELEMENTS - Kentucky												Attachi	ment: 2	Exhil	oit: C
												Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted			Charge -	Charge -	Charge -
			1									Elec		Manual Svc			Manual Svc
CATE	GORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		· · · · · · · · · · · · · · · · · · ·	m						==(+)			per LSK	per LSK		Electronic-	Electronic-	Electronic-
														Electronic-			
														1st	Add'l	Disc 1st	Disc Add'l
							_	Nonred	currina	Nonrecurring	Disconnect			oss	Rates(\$)	ı	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		AIN Toolkit Service - SCP Storage Charge, Per SMS Access															
		Account, Per 100 Kilobytes					0.07										
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
		Subscription			CAM	BAPMS	7.87	8.64	8.64	6.08	6.08		7.86				
		AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
		Subscription			CAM	BAPLS	3.26	9.56	9.56				7.86				
		AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service															
		Subscription			CAM	BAPDS	4.72	8.64	8.64	6.08	6.08		7.86				
		AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit															
		Service Subscription			CAM	BAPES	0.11	9.56	9.56				7.86				
ENHA		(TENDED LINK (EELs)															
		New Density Zone 1 EELs are available in the following MSA:					Atlanta, Ga; Ne	w Orleans, LA,									
		Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-															
		In all states, EEL network elements shown below also apply t												UNEs.(Non-re	ecurring rates	do not apply	.)
		In All States the EEL network elements apply to ordinarily con				tch As Is Cha	arge.) When or	dering ordinar	ily combined	network elemer	nts, Non-recur	ring rates d	o apply.		1		
	2-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)	1									ļ		
		First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport															
		Combination - Zone 1		1	UNCVX	UEAL2	12.67	125.22	60.48	59.69	7.84		7.86				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
		Transport Combination - Zone 2		2	UNCVX	UEAL2	17.45	125.22	60.48	59.69	7.84		7.86				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
		Transport Combination - Zone 3		3	UNCVX	UEAL2	33.22	125.22	60.48	59.69	7.84		7.86				
		Interoffice Transport - Dedicated - DS1 combination - Per Mile															
		per month			UNC1X	1L5XX	0.19										
		Interoffice Transport - Dedicated - DS1 combination - Facility															
		Termination per month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86				
		DS1 Channelization System Per Month			UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86				
		Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.62	6.71	4.84				7.86				
		Each Additional 2-Wire VG Loop(SL 2) in the same DS1															
	<u> </u>	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	12.67	125.22	60.48	59.69	7.84		7.86				
		Each Additional 2-Wire VG Loop(SL2) in the same DS1			111000		47.45	405.00	00.40	50.00	7.04		7.00				
	<u> </u>	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	17.45	125.22	60.48	59.69	7.84		7.86				
		Each Additional 2-Wire VG Loop(SL2) in the same DS1		3	111000	UEAL2	00.00	405.00	60.48	50.00	7.84		7.00				
	-	Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	33.22	125.22	60.48	59.69	7.84		7.86				
		Voice Grade COCI - DS1 to DS0 Channel System combination -			LINOVA	1D1VG	0.62	6.71	4.04				7.00				
	1	per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCVX	IDIVG	0.62	0.71	4.84				7.86				
		Is Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86				
-	4-WIDE	IS Charge VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	FROFE	ICE TO		OINCCC	1	0.98	0.98	11.17	11.17		7.00		 	1	
-	T-VVIKE	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	LNOFF	JE IK	ANDI ONI (EEL)	-	 			 					t	1	
		Transport Combination - Zone 1	1	1	UNCVX	UEAL4	29.26	125.22	60.48	59.69	7.84		7.86		I		
	1	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		-	ONOVA	OL/1L4	20.20	120.22	00.40	00.00	7.04	1	7.00				
		Transport Combination - Zone 2	l	2	UNCVX	UEAL4	34.25	125.22	60.48	59.69	7.84		7.86		1		
	1	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	1	T -			320	.20.22	33.40	55.00	7.04		50		<u> </u>		
		Transport Combination - Zone 3	1	3	UNCVX	UEAL4	85.06	125.22	60.48	59.69	7.84		7.86		I		
	<u> </u>	Interoffice Transport - Dedicated - DS1 combination - Per Mile	1				55.56	.20.22	55.10	55.55	54				t		
		Per Month	1	1	UNC1X	1L5XX	0.19						1		I		
	1	Interoffice Transport - Dedicated - DS1 - Facility Termination Per				1				İ					İ	İ	
		Month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86				
		Channelization - Channel System DS1 to DS0 combination Per															
		Month	l		UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86		1		
		Voice Grade COCI - DS1 to DS0 Channel System combination -											-				
		per month	1	1	UNCVX	1D1VG	0.62	6.71	4.84				7.86		I		
		Additional 4-Wire Analog Voice Grade Loop in same DS1															
		Interoffice Transport Combination - Zone 1	1	1	UNCVX	UEAL4	29.26	125.22	60.48	59.69	7.84		7.86		I		
		Additional 4-Wire Analog Voice Grade Loop in same DS1															
	<u> </u>	Interoffice Transport Combination - Zone 2	<u></u>	2	UNCVX	UEAL4	34.25	125.22	60.48	59.69	7.84	<u></u>	7.86		<u> </u>		
		Additional 4-Wire Analog Voice Grade Loop in same DS1															
		Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	85.06	125.22	60.48	59.69	7.84		7.86				

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attachi	ment: 2	Exhib	oit: C
J.155115E	TOTAL SECURITY TOTAL										Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonrec	urring	Nonrecurring	Disconnect		l	220	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOM AN	SOMAN	SOMAN
	Voice Grade COCI - DS1 to DS0 Channel System combination -							7.00.		71441	0020			00	00	
	per month			UNCVX	1D1VG	0.62	6.71	4.84				7.86				1
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86				ı
4-WII	RE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL)												
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice						40= 00		==							1
	Transport Combination - Zone 1		1	UNCDX	UDL56	27.59	125.22	60.48	59.69	7.84		7.86				
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	32.48	125.22	60.48	59.69	7.84		7.86				1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice			ONODA	ODESO	32.40	120.22	00.40	33.03	7.04		7.00				
	Transport Combination - Zone 3		3	UNCDX	UDL56	36.37	125.22	60.48	59.69	7.84		7.86				ı
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		T	-	1			22.70	22.30					1		·
	Per Month	<u></u>	<u> </u>	UNC1X	1L5XX	0.19					<u> </u>	<u> </u>				1
	Interoffice Transport - Dedicated - DS1 - combination Facility							· · · · · · · · · · · · · · · · · · ·					-			
	Termination Per Month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86				
1 1	Channelization - Channel System DS1 to DS0 combination Per	1	1	LINIOAN	luc.											ı
\vdash	Month OCU-DP COCI (data) - DS1 to DS0 Channel System - per	 	<u> </u>	UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86		-		
	month (2.4-64kbs)			UNCDX	1D1DD	1.32	6.71	4.84				7.86				1
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			ONODA	10100	1.02	0.71	4.04				7.00				
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	27.59	125.22	60.48	59.69	7.84		7.86				1
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1				-				30.00							
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	32.48	125.22	60.48	59.69	7.84		7.86				1
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	36.37	125.22	60.48	59.69	7.84		7.86				
	OCU-DP COCI (data) - DS1 to DS0 Channel System -			LINCDY	4D4DD	4.00	C 74	4.04				7.00				1
	combination per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	1D1DD	1.32	6.71	4.84				7.86				
	Is Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86				1
4-WII	RE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE				0.00	0.00				7.00				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1		1	UNCDX	UDL64	27.59	125.22	60.48	59.69	7.84		7.86				1
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 2		2	UNCDX	UDL64	32.48	125.22	60.48	59.69	7.84		7.86				
1 1	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice	1	3	LINICDY	UDL64	26.27	105.00	60.40	E0.00	7.04		7.00				ı
 	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile	1	3	UNCDX	UDL04	36.37	125.22	60.48	59.69	7.84		7.86				
1 1	Per Month	1	1	UNC1X	1L5XX	0.19						1				ı
 	Interoffice Transport - Dedicated - DS1 combination - Facility	<u> </u>	<u> </u>		. 20,01	5.10										
I	Termination Per Month	<u></u>	L	UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32	<u> </u>	7.86				<u> </u>
	Channelization - Channel System DS1 to DS0 combination Per															
	Month		<u> </u>	UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86				.
	OCU-DP COCI (data) - DS1 to DS0 Channel System	1	1	LINIODY	10105		:									ı
\vdash	combination - per month (2.4-64kbs)	 	<u> </u>	UNCDX	1D1DD	1.32	6.71	4.84				7.86		-		
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		4	UNCDX	UDL64	27.59	125.22	60.48	59.69	7.84		7.86				ı
 	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	1	+	OINODA	UDLU4	21.59	120.22	00.40	35.09	7.04		1.00				 I
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	32.48	125.22	60.48	59.69	7.84		7.86				ı
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1													1		·
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	36.37	125.22	60.48	59.69	7.84		7.86				
	OCU-DP COCI (data) - DS1 to DS0 Channel System	1										1				
	combination - per month (2.4-64kbs)	ļ	<u> </u>	UNCDX	1D1DD	1.32	6.71	4.84				7.86				1
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	1		UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86				İ
4-19/11	Is Charge RE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTI	FROFF	CE TP		UNCCC		0.98	0.98	11.17	11.17	1	7.80				
4-4411	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	LAGERI	LIKA	THO ON TELL								 				
	Transport - Zone 1	1	1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86				ı
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	l														
	Transport - Zone 2		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86				1

UNBUNDI F	D NETWORK ELEMENTS - Kentucky												Attach	ment: 2	Exhib	oit: C
SHOUNDEL	- Nemucky										Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		Intor			1						Elec			Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
1		l m			1								Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						1										
						Rec	Nonrec		Nonrecurring		201150	001441		Rates(\$)	001441	001141
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Transport - Zone 3		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86				
-	Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCIX	USLAA	291.76	210.70	114.60	63.96	17.97		7.00				
	Per Month			UNC1X	1L5XX	0.19										
	Interoffice Transport - Dedicated - DS1 combination - Facility			ONOTA	120701	0.10										
	Termination Per Month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86				
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86				
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTI	ROFFI	CE TRA	NSPORT (EEL)												
	First DS1Loop in DS3 Interoffice Transport Combination - Zone															
	1		1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86		ļ		
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		_											1		
 	[2		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone			LINGAY	1101.307	007.70	040 =0	444.00	00.00	47.00		7.00		1		
 	Intereffice Transport Dedicated DCC	1	3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86		 		
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month			UNC3X	1L5XX	4.09										
\vdash	Interoffice Transport - Dedicated - DS3 - Facility Termination per	-	 	OINCOA	ILOAA	4.09						 		-		
	month			UNC3X	U1TF3	966.89	350.56	141.58	48.00	23.39		7.86				
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3	158.20	115.48	56.53	15.12	5.30		7.86				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.80	6.71	4.84	.0.12	0.00		7.86				
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 1		1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86				
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 2		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86				
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 3		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.80	6.71	4.84				7.86				
	Nonrecurring Currently Combined Network Elements Switch -As-			LINIONY			0.00	0.00	44.47	44.47		7.00				
O MID	Is Charge E VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	FERAFE	ICE TO	UNC3X	UNCCC		8.98	8.98	11.17	11.17		7.86				
Z-WIR	2-WireVG Loop used with 2-wire VG Interoffice Transport	EKUFF	ICE IK	ANSPURI (EEL)	+											
	Combination - Zone 1		1	UNCVX	UEAL2	12.67	125.22	60.48	59.69	7.84		7.86		1		
 	2-WireVG Loop used with 2-wire VG Interoffice Transport		+ '-	J V/A	JL/ 11L	12.07	120.22	00.40	33.09	7.04		7.00				
	Combination - Zone 2		2	UNCVX	UEAL2	17.45	125.22	60.48	59.69	7.84		7.86		1		
	2-WireVG Loop used with 2-wire VG Interoffice Transport		ΤĒ	-	T			22.10	22.30				1			
	Combination - Zone 3		3	UNCVX	UEAL2	33.22	125.22	60.48	59.69	7.84		7.86		1		
	Interoffice Transport - Dedicated - 2-wire VG combination - Per															
	Mile Per Month			UNCVX	1L5XX	0.01										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade				1											
	combination - Facility Termination per month			UNCVX	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86	ļ	ļ		
	Nonrecurring Currently Combined Network Elements Switch -As-			LINOVA	LINICCO		0.00	0.00	44.47	44.47		7.00		1		
A-WID	Is Charge E VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EBOE	ICE TO	UNCVX	UNCCC		8.98	8.98	11.17	11.17		7.86				
4-WIR	4-WireVG Loop used with 4-wire VG Interoffice Transport	ERUFF	ICE IK	ANOFURI (EEL)	+							-	1	1		
	Combination - Zone 1		1	UNCVX	UEAL4	29.26	125.22	60.48	59.69	7.84		7.86		1		
	4-WireVG Loop used with 4-wire VG Interoffice Transport	1	- '-	J	JL/1L7	23.20	120.22	00.40	33.09	7.04		7.00	1	1		
	Combination - Zone 2		2	UNCVX	UEAL4	34.25	125.22	60.48	59.69	7.84		7.86		1		
	4-WireVG Loop used with 4-wire VG Interoffice Transport															
I	Combination - Zone 3	<u></u>	3	UNCVX	UEAL4	85.06	125.22	60.48	59.69	7.84	<u> </u>	7.86	<u> </u>	<u></u>		
	Interoffice Transport - Dedicated - 4-wire VG combination - Per															
	Mile Per Month			UNCVX	1L5XX	0.01										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade				I							1]		
	combination - Facility Termination per month			UNCVX	U1TV4	21.28	98.09	53.67	56.31	22.42		7.86		ļ		
	Nonrecurring Currently Combined Network Elements Switch -As-	1												1		
D00 5	Is Charge	L NE TD 4 :	Henes	UNCVX	UNCCC		8.98	8.98	11.17	11.17		7.86				
DS3 D	IGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC High Capacity Unbundled Local Loop - DS3 combination - Per	,⊏ IKAI	NSPOR	I (EEL)	+									 		
	Mile per month			UNC3X	1L5ND	9.25						1		1		
	Iville per month	l	l	OINOON	ILUND	5.20			1		1	l	l	l .		

UNBUNDL	LED NETWORK ELEMENTS - Kentucky													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	Litely Conserve High Handle and DOO and Conference						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X UNC3X	UE3PX 1L5XX	308.31 4.09	237.36	147.69	83.43	32.67		7.86				
	Interoffice Transport - Dedicated - DS3 - Per Mile per month Interoffice Transport - Dedicated - DS3 combination - Facility			UNCSA	ILSAA	4.09										+
	Termination per per month Nonrecurring Currently Combined Network Elements Switch -As-			UNC3X	U1TF3	966.89	350.56	141.58	48.00	23.39		7.86				1
	Is Charge			UNC3X	UNCCC		8.98	8.98	11.17	11.17		7.86				
STS	1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFF	FICE TF	RANSP	ORT (EEL)												
	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month			UNCSX	1L5ND	9.25										
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile			UNCSX	UDLS1	320.51	237.36	147.69	83.43	32.67		7.86				
	per month Interoffice Transport - Dedicated - \$151 combination - Per Mile per month Interoffice Transport - Dedicated - \$T\$1 combination - Facility			UNCSX	1L5XX	4.09										<u> </u>
	Termination per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCSX	U1TFS	945.79	350.56	141.58	48.00	23.39		7.86				
	Is Charge			UNCSX	UNCCC		8.98	8.98	11.17	11.17		7.86				
2-WI	IRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	T (EEL)													
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1		1	UNCNX	U1L2X	18.44	125.22	60.48	59.69	7.84		7.86				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2	UNCNX	U1L2X	25.08	125.22	60.48	59.69	7.84		7.86				
	Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCNX UNC1X	U1L2X 1L5XX	42.87 0.19	125.22	60.48	59.69	7.84		7.86				
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86				
	Channelization - Channel System DS1 to DS0 combination - per month			UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month			UNCNX	UC1CA	2.84	6.71	4.84				7.86				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1		1	UNCNX	U1L2X	18.44	125.22	60.48	59.69	7.84		7.86				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport		2	UNCNX	U1L2X	25.08	125.22	60.48	59.69	7.84		7.86				ļ
	Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System		3	UNCNX	U1L2X	42.87	125.22	60.48	59.69	7.84		7.86				<u> </u>
	combination- per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCNX	UC1CA	2.84	6.71	4.84				7.86				
. 1	Is Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86				
4-WI	IRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROF	FICE T	RANSPORT (EEL)												
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86				<u> </u>
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86				
	Per Month Interoffice Transport - Dedicated - STS1 combination - Per Mille Per Month Interoffice Transport - Dedicated - STS1 combination - Facility			UNCSX	1L5XX	4.09										
	Termination STS1 to DS1 Channel System conbination per month			UNCSX	U1TFS MQ3	945.79 158.20	350.56 115.48	141.58 56.53	48.00 15.12	23.39 5.30		7.86 7.86				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.80	6.71	4.84	10.12	3.30		7.86				
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86				
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2			UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86				

ONRONDI	LED NETWORK ELEMENTS - Kentucky			1							1 -			ment: 2	1	bit: C
CATEGORY	r RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional DS1Loop in STS1 Interoffice Transport Combination -															
	Zone 3		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.80	6.71	4.84				7.86				
	Nonrecurring Currently Combined Network Elements Switch -As	;-		LINIOOV	UNCCC		0.00	0.00	44.47	44.47		7.00				
4-10/	Is Charge IRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERC	DEELCE :	DANG	UNCSX	UNCCC		8.98	8.98	11.17	11.17		7.86				
4-44	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport	FFICE	KANS	FORT (EEE)												
	Combination - Zone 1		1	UNCDX	UDL56	27.59	125.22	60.48	59.69	7.84		7.86				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		<u> </u>	0.1027	02200	27.00	120.22	00.10	00.00			7.00				
	Combination - Zone 2		2	UNCDX	UDL56	32.48	125.22	60.48	59.69	7.84		7.86				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
	Combination - Zone 3		3	UNCDX	UDL56	36.37	125.22	60.48	59.69	7.84		7.86				
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -							· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·						
	Per Mile		<u> </u>	UNCDX	1L5XX	0.01										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -		1		===											
	Facility Termination		<u> </u>	UNCDX	U1TD5	17.25	98.09	53.67	56.31	22.42		7.86			ļ.	
	Nonrecurring Currently Combined Network Elements Switch -As	i-		UNCDX	UNCCC		0.00	8.98	11.17	11.17		7.06				
4-10/	Is Charge I'IRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERC	DEELCE :	DANG		UNCCC		8.98	8.98	11.17	11.17		7.86				
4-44	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport	TICL	KANS	TOKT (EEE)												
	Combination - Zone 1		1	UNCDX	UDL64	27.59	125.22	60.48	59.69	7.84		7.86				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		<u> </u>	CHODA	ODLOT	27.00	120.22	00.40	00.00	7.04		7.00				
	Combination - Zone 2		2	UNCDX	UDL64	32.48	125.22	60.48	59.69	7.84		7.86				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport						-									
	Combination - Zone 3		3	UNCDX	UDL64	36.37	125.22	60.48	59.69	7.84		7.86				
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	Per Mile			UNCDX	1L5XX	0.01										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -				l											
	Facility Termination			UNCDX	U1TD6	17.25	98.09	53.67	56.31	22.42		7.86				
	Nonrecurring Currently Combined Network Elements Switch -As	i-		LINODY	1111000		0.00	0.00	44.47	44.47		7.00				
ADDITIONA	Is Charge			UNCDX	UNCCC		8.98	8.98	11.17	11.17		7.86				
	en used as a part of a currently combined facility, the non-recu	rna cha	rase de	not apply but a	Switch As Is c	narge does and	Ny									
	en used as a part of a currently combined facility, the hori-recal															
	recurring Currently Combined Network Elements "Switch As Is"					7.0 10 0.1a. go (
	Nonrecurring Currently Combined Network Elements Switch -As		Ī													
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		8.98	8.98	11.17	11.17		7.86				
	Nonrecurring Currently Combined Network Elements Switch -As	;-														
	Is Charge - 56/64 kbps			UNCDX	UNCCC		8.98	8.98	11.17	11.17		7.86				
	Nonrecurring Currently Combined Network Elements Switch -As	i-	1	l												
	Is Charge - DS1		<u> </u>	UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86			ļ.	
	Nonrecurring Currently Combined Network Elements Switch -As Is Charge - DS3	-	1	UNC3X	UNCCC		8.98	8.98	11.17	11.17		7.86				
	Nonrecurring Currently Combined Network Elements Switch -As		 	UNCOA	UNCCC		8.98	8.98	11.17	11.17		7.80			 	
	Is Charge - STS1	1		UNCSX	UNCCC		8.98	8.98	11.17	11.17		7.86				
NOT	TE: Local Channel - Dedicated Transport - minimum billing perio	d - Relo	w DS3			r months	0.30	0.30	11.17	11.17		1.00			<u> </u>	
	Local Channel - Dedicated - 2-Wire Voice Grade	1 20.0	1	UNCXV	ULDV2	18.57	265.78	46.96	46.79	4.98		7.86				
	Local Channel - Dedicated - 4-Wire Voice Grade			UNCXV	ULDV4	19.86	266.48	47.65	47.54	5.73		7.86				
	Local Channel - Dedicated - DS1 per month Zone 1		1	UNC1X	ULDF1	40.46	209.60	176.51	30.21	21.07		7.86				
	Local Channel - Dedicated -DS1 Per Month Zone 2		2	UNC1X	ULDF1	43.39	209.60	176.51	30.21	21.07		7.86				
	Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	164.50	209.60	176.51	30.21	21.07		7.86				
	Local Channel - Dedicated - DS3 - Per Mile per month		<u> </u>	UNC3X	1L5NC	8.74										
	Local Channel - Dedicated - DS3 - Facility Termination		<u> </u>	UNC3X	ULDF3	576.05	551.38	338.08	173.00	120.42		7.86				
	Local Channel - Dedicated - STS-1- Per Mile per month	1	<u> </u>	UNCSX	1L5NC	8.74	FF1.00	200.00	170.00	100.10		7.00			1	
B4111	Local Channel - Dedicated - STS-1 - Facility Termination LTIPLEXERS	1	1	UNCSX	ULDFS	543.24	551.38	338.08	173.00	120.42		7.86		-	1	1
MUL	Channelization - DS1 to DS0 Channel System	1	 	UXTD1	MQ1	113.33	101.40	71.60	13.79	13.04		7.86			 	
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per	+	 	וטואט	IVIQI	113.33	101.40	7 1.00	13.79	13.04		7.00		1		

UNBUNDLE	D NETWORK ELEMENTS - Kentucky						· · · · · ·	· · · · · · · · · · · · · · · · · · ·					Attachr	nent: 2	Exhil	oit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	0 : IODN 0001 (PDITE)						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
.	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month			UDN	UC1CA	2.04	10.07	7.08				7.00				
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	2.84 0.6228	10.07	7.08				7.86 7.86				
	DS3 to DS1 Channel System per month			UXTD3	MQ3	158.20	199.23	118.62	50.16	48.59		7.86				
	STS1 to DS1 Channel System per month			UXTS1	MQ3	158.20	199.23	118.62	50.16	48.59		7.86				
	DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	11.80	10.07	7.08	30.10	40.55		7.86				
	DS3 Interface Unit (DS1 COCI) used with Local Channel per			002	COIDI	11.00	10.07	7.00				7.00				
ı İ	month			ULDD1	UC1D1	11.80	10.07	7.08				7.86				
	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel			-												
ı İ	per month			U1TD1	UC1D1	11.80	10.07	7.08				7.86				
	LOCAL EXCHANGE SWITCHING(PORTS)															
	nge Ports															
	Although the Port Rate includes all available features in GA, I	KY, LA	& TN, t	he desired features	will need to b	e ordered usin	g retail USOCs	•								
2-WIRE	E VOICE GRADE LINE PORT RATES (RES)															
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.49	3.74	3.63	2.23	2.13		7.86				
ı İ																
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.49	3.74	3.63	2.23	2.13		7.86				
ı İ	Firebooks Bosts - 2 Wise Apples Line Dest sutering sells. Des			UEPSR	UEPRO	1.49	3.74	3.63	2.23	2.13		7.86				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res. Exchange Ports - 2-Wire VG unbundled KY extended local			UEPSK	UEPRU	1.49	3.74	3.03	2.23	2.13		7.86				
.	dialing parity Port with Caller ID - Res.			UEPSR	UEPRM	1.49	3.74	3.63	2.23	2.13		7.86				
	Exchange Ports - 2-Wire VG unbundled res, low usage line port			UEFOR	UEPRIVI	1.49	3.74	3.03	2.23	2.13		7.00				
ı İ	with Caller ID (LUM)			UEPSR	UEPAP	1.49	3.74	3.63	2.23	2.13		7.86				
	Exchange Ports - 2-Wire Voice Kentucky Residence Dialing Plan			OLI OIX	OL174	1.40	0.7 4	0.00	2.20	2.10		7.00				
ı İ	without Caller ID			UEPSR	UEPWE	1.49	3.74	3.63	2.23	2.13		7.86				
	2-Wire voice unbundled Low Usage Line Port without Caller ID			02. 0.1	02. 112	11.10	0.7 1	0.00	2.20	2.10		7.00				
ı İ	Capability			UEPSR	UEPRT	1.49	3.74	3.63	2.23	2.13		7.86				
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00	_			7.86				
FEATU	JRES															
	All Available Vertical Features			UEPSR	UEPVF	0.00	0.00	0.00				7.86				
2-WIRE	E VOICE GRADE LINE PORT RATES (BUS)															
ı İ	Exchange Ports - 2-Wire Analog Line Port without Caller ID -															
	Bus			UEPSB	UEPBL	1.49	3.74	3.63	2.23	2.13		7.86				
ı İ	Exchange Ports - 2-Wire VG unbundled Line Port with															
	unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.49	3.74	3.63	2.23	2.13		7.86				
. 1	Fush areas Posts - 0 Wise Applies Line Post sustain 1 - 1 - 1		1	LIEDOD	LIEDDO	4 40	2.74	2.00	0.00	0.40		7.00				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus. Exchange Ports - 2-Wire VG unbundled KY extended local		<u> </u>	UEPSB	UEPBO	1.49	3.74	3.63	2.23	2.13	ļ	7.86				
. 1	dialing parity Port with Caller ID - Bus.		1	UEPSB	UEPBM	1.49	3.74	3.63	2.23	2.13		7.86				
	Exhange Ports - 2-Wire VG unbundled incoming only port with		1	ULFOD	UEPDIVI	1.49	3.74	3.03	2.23	2.13		1.00				
.	Caller ID - Bus			UEPSB	UEPB1	1.49	3.74	3.63	2.23	2.13		7.86				
	Exchange Ports - 2-Wire Voice Kentucky Business Dialing Plan	1	†	021 00	02.01	1.73	5.74	5.05	2.23	2.13		7.00				
. 1	without Caller ID		1	UEPSB	UEPWF	1.49	3.74	3.63	2.23	2.13		7.86				
	2-Wire voice unbundled Incoming Only Port without Caller ID			-				2.30		0						
, 1	Capability		1	UEPSB	UEPBE	1.49	3.74	3.63	2.23	2.13		7.86				
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00				7.86				
FEATU	JRES															
	All Available Vertical Features			UEPSB	UEPVF	0.00	0.00	0.00				7.86				
EXCH#	ANGE PORT RATES (DID & PBX)															
	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.49	39.05	18.17	15.38	0.89		7.86				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus		<u> </u>	UEPSP	UEPPC	1.49	39.05	18.17	15.38	0.89	ļ	7.86				
. 1	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus		<u> </u>	UEPSP	UEPPO	1.49	39.05	18.17	15.38	0.89		7.86				
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus		<u> </u>	UEPSP	UEPP1	1.49	39.05	18.17		0.89		7.86				
		ı	1	UEPSP	UEPLD	1.49	39.05	18.17	15.38 15.38	0.89		7.86 7.86				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			LIEDOD												ı
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.49	39.05	18.17								
	2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.49	39.05	18.17	15.38	0.89		7.86				
	2-Wire Voice Unbundled PBX LD Terminal Ports								15.38 15.38							

	S - Kentucky												Attachi	ment: 2	Exhib	oit: C
ATEGORY RATE	ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge -		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual S Order vs Electronic Disc Add
						 	Nonred	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire Voice Unbundled PBX	LD Terminal Switchboard IDD															
Capable Port				UEPSP	UEPXE	1.49	39.05	18.17	15.38	0.89		7.86				
2-Wire Voice Unbundled 2-Wa	ay PBX Kentucky Room Area															
Calling Port Without LUD	,			UEPSP	UEPXF	1.49	39.05	18.17	15.38	0.89		7.86				
2-Wire Voice Unbundled PBX	Kentucky LUD Area Calling Port			UEPSP	UEPXG	1.49	39.05	18.17	15.38	0.89		7.86				
2-Wire Voice Unbundled PBX	Kentucky Premium Callling Port			UEPSP	UEPXH	1.49	39.05	18.17	15.38	0.89		7.86				
Port Without LUD	ay PBX Kentucky Area Callling			UEPSP	UEPXJ	1.49	39.05	18.17	15.38	0.89		7.86				
2-Wire Voice Unbundled 2-Wa	ay PBX Hotel/Hospital Economy															
Administrative Calling Port				UEPSP	UEPXL	1.49	39.05	18.17	15.38	0.89		7.86				
2-Wire Voice Unbundled 2-Wa Room Calling Port	ay PBX Hotel/Hospital Economy			UEPSP	UEPXM	1.49	39.05	18.17	15.38	0.89		7.86				
	ay Outgoing PBX Hotel/Hospital	-		0=1 01	OLI XIVI	1.73	55.05	10.17	10.00	0.09		7.00		t	 	
Discount Room Calling Port	ay outgoing i bx note, no opital			UEPSP	UEPXO	1.49	39.05	18.17	15.38	0.89		7.86				
	ay Outgoing PBX Measured Port			UEPSP	UEPXS	1.49	39.05	18.17	15.38	0.89		7.86				
Subsequent Activity	ay datgoing i Brimoadarda i oit			UEPSP	USASC	0.00	0.00	0.00	10.00	0.00		7.86				
FEATURES						0.00										
All Available Vertical Features	i			UEPSP UEPSE	UEPVF	0.00	0.00	0.00				7.86				
EXCHANGE PORT RATES (COIN)																
Exchange Ports - Coin Port						1.49	3.74	3.63	2.23	2.13		7.86				
Local Switching Features offered w	ith Port															
NOTE: Transmission/usage charge	s associated with POTS circuit sw	vitched	usage	will also apply to	circuit switche	ed voice and/or	circuit switch	ed data transm	ission by B-Ch	annels associ	ated with 2-	wire ISDN p	orts.			
NOTE: Access to B Channel or D C	hannel Packet capabilities will be	availab	ole onl	y through BFR/Nev	/ Business Re	quest Process.	Rates for the	packet capabi	lities will be de	termined via tl	ne Bona Fid	le Request/l	New Business	s Request Pro	cess.	
Exchange port - 4-wire ISDN t	runk port -all available features															
included					UEPEX	101.60	188.36	95.15	61.92	22.67		7.86				
BUNDLED LOCAL EXCHANGE SWITCHI	NG(PORTS)															
EXCHANGE PORT RATES																
Exchange Ports - 2-Wire DID				UEPEX	UEPP2	10.51	92.18	15.82	52.16	5.30		7.86				
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port																
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability	- 4-Wire DS1 Port with DID			UEPDD	UEPDD	74.77	164.86	77.74	60.69	3.86		7.86				
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN	- 4-Wire DS1 Port with DID			UEPDD UEPTX UEPSX	UEPDD U1PMA	74.77 13.46	164.86 60.60	77.74 50.67								
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered	- 4-Wire DS1 Port with DID N Port (See Notes below.)			UEPDD UEPTX UEPSX UEPTX UEPSX	UEPDD U1PMA UEPVF	74.77 13.46 0.00	164.86 60.60 0.00	77.74 50.67 0.00	60.69 32.83	3.86 14.17		7.86 7.86				
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw			UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to	UEPDD U1PMA UEPVF circuit switche	74.77 13.46 0.00 ed voice and/or	164.86 60.60 0.00 circuit switche	77.74 50.67 0.00 ed data transm	60.69 32.83 ission by B-Ch	3.86 14.17 annels associ		7.86 7.86 wire ISDN p		Poguest Pre		
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit swhannel Packet capabilities will be			UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to our of through BFR/New	UEPDD U1PMA UEPVF circuit switcher Business Re	74.77 13.46 0.00 ed voice and/or quest Process.	164.86 60.60 0.00 circuit switche Rates for the	77.74 50.67 0.00 ed data transm packet capabi	60.69 32.83 ission by B-Ch	3.86 14.17 annels associ		7.86 7.86 wire ISDN p		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles			UEPDD UEPTX UEPSX UEPTX UEPSX ewill also apply to a y through BFR/New UEPTX UEPSX	UEPDD U1PMA UEPVF circuit switcher Business Re	74.77 13.46 0.00 ed voice and/or quest Process. 0.00	164.86 60.60 0.00 circuit switch Rates for the	77.74 50.67 0.00 ed data transm packet capabi 0.00	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port - Channel Profiles N DS1 Port	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to our of through BFR/New	UEPDD U1PMA UEPVF circuit switcher Business Re	74.77 13.46 0.00 ed voice and/or quest Process.	164.86 60.60 0.00 circuit switche Rates for the	77.74 50.67 0.00 ed data transm packet capabi	60.69 32.83 ission by B-Ch	3.86 14.17 annels associ		7.86 7.86 wire ISDN p		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit swhannel Packet capabilities will be N Port - Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY	availab		UEPDD UEPTX UEPSX UEPTX UEPSX ewill also apply to a y through BFR/New UEPTX UEPSX	UEPDD U1PMA UEPVF circuit switcher Business Re	74.77 13.46 0.00 ed voice and/or quest Process. 0.00	164.86 60.60 0.00 circuit switch Rates for the	77.74 50.67 0.00 ed data transm packet capabi 0.00	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to o y through BFR/New UEPTX UEPSX UEPEX	UEPDD U1PMA UEPVF circuit switcher Business Re U1UMA UEPEX	74.77 13.46 0.00 ad voice and/or quest Process. 0.00 101.60	164.86 60.60 0.00 circuit switch Rates for the 0.00 188.36	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p le Request/I		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit swhannel Packet capabilities will be N Port - Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY	availab		UEPDD UEPTX UEPSX UEPTX UEPSX ewill also apply to a y through BFR/New UEPTX UEPSX	UEPDD U1PMA UEPVF circuit switcher Business Re	74.77 13.46 0.00 ed voice and/or quest Process. 0.00	164.86 60.60 0.00 circuit switch Rates for the	77.74 50.67 0.00 ed data transm packet capabi 0.00	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 2-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port - Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY JARDING SERVICE - RESIDENCE arding Service, Area Calling, Res	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to to the thing of the t	UEPDD U1PMA UEPVF irrcuit switcher Business Re U1UMA UEPEX UERAC	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60	164.86 60.60 0.00 circuit switch: Rates for the 0.00 188.36	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p le Request/f		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE arding Service, Area Calling, Researding Service, Local Calling - Res	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to o y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR	UEPDD U1PMA UEPVF circuit switcher / Business Re U1UMA UEPEX UERAC UERAC	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60	164.86 60.60 0.00 circuit switch Rates for the 0.00 188.36 3.74	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15 3.63	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p le Request/f 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port - Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY JARDING SERVICE - RESIDENCE Parding Service, Area Calling, Resurarding Service, Local Calling - Resarding Service, InterLATA - Res	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to to the thing of the t	UEPDD U1PMA UEPVF irrcuit switcher Business Re U1UMA UEPEX UERAC	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60	164.86 60.60 0.00 circuit switch: Rates for the 0.00 188.36	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p le Request/f		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 2-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORM Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE arding Service, Area Calling, Researding Service, Local Calling - Res	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche y Business Re U1UMA UEPEX UEPEX UERAC UERAC UERLC UERTE	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60 1.49	164.86 60.60 0.00 circuit switch. Rates for the 0.00 188.36 3.74 3.74	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15 3.63 3.63	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p le Request/l 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port - Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE rarding Service, Area Calling, Resurding Service, Local Calling - Resarding Service, InterLATA - Resarding Service, IntraLATA - Resarding Service, IntraLATA - Res	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche v Business Re U1UMA UEPEX UEPEX UERAC UERAC UERLC UERTE UERTR	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60 1.49	164.86 60.60 0.00 circuit switche Rates for the 0.00 188.36 3.74 3.74 3.74	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15 3.63 3.63 3.63	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN p le Request/i 7.86 7.86 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Switch-as-is	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port - Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY JARDING SERVICE - RESIDENCE Parding Service, Area Calling, Res rarding Service, Local Calling - Res rarding Service, InterLATA - Res rarding Service, IntraLATA - Res rarding Service - Conversion -	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche y Business Re U1UMA UEPEX UEPEX UERAC UERAC UERLC UERTE	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60 1.49	164.86 60.60 0.00 circuit switch. Rates for the 0.00 188.36 3.74 3.74	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15 3.63 3.63	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 wire ISDN p le Request/l 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORM Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Non-Recurring Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY JARDING SERVICE - RESIDENCE rarding Service, Area Calling, Res rarding Service, Local Calling - Res rarding Service, InterLATA - Res rarding Service, IntraLATA - Res rarding Service Conversion - rarding Service Conversion with	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to or y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF circuit switche r Business Re U1UMA UEPEX UERAC UERAC UERTE UERTE UERTE USAC2	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60 1.49	164.86 60.60 0.00 circuit switch Rates for the 0.00 188.36 3.74 3.74 3.74 0.10	77.74 50.67 0.00 od data transm packet capabi 0.00 95.15 3.63 3.63 3.63 0.10	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN p le Request/i 7.86 7.86 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Non-Recurring Unbundled Remote Call Forw Switch-as-is	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE rarding Service, Area Calling, Reservice, Local Calling - Researding Service, InterLATA - Researding Service, IntraLATA - Researding Service - Conversion - rearding Service - Conversion with C)	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche v Business Re U1UMA UEPEX UEPEX UERAC UERAC UERLC UERTE UERTR	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60 1.49	164.86 60.60 0.00 circuit switche Rates for the 0.00 188.36 3.74 3.74 3.74	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15 3.63 3.63 3.63	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN p le Request/i 7.86 7.86 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw allowed change (PIC and LPIG	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE rarding Service, Area Calling, Reservice, Local Calling - Researding Service, InterLATA - Researding Service, IntraLATA - Researding Service - Conversion - rearding Service - Conversion with C)	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to or y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF circuit switche r Business Re U1UMA UEPEX UERAC UERAC UERTE UERTE UERTE USAC2	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60 1.49	164.86 60.60 0.00 circuit switch Rates for the 0.00 188.36 3.74 3.74 3.74 0.10	77.74 50.67 0.00 od data transm packet capabi 0.00 95.15 3.63 3.63 3.63 0.10	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN p le Request/i 7.86 7.86 7.86 7.86		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 2-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw allowed change (PIC and LPIC UNBUNDLED REMOTE CALL FORW	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE rarding Service, Area Calling, Reservice, Local Calling - Researding Service, InterLATA - Researding Service, IntraLATA - Researding Service - Conversion - rearding Service - Conversion with C)	availab		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to or y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF circuit switche r Business Re U1UMA UEPEX UERAC UERAC UERTE UERTE UERTE USAC2	74.77 13.46 0.00 ed voice and/or quest Process. 0.00 101.60 1.49	164.86 60.60 0.00 circuit switch Rates for the 0.00 188.36 3.74 3.74 3.74 0.10	77.74 50.67 0.00 od data transm packet capabi 0.00 95.15 3.63 3.63 3.63 0.10	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN p le Request/i 7.86 7.86 7.86 7.86		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw allowed change (PIC and LPIc UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Allowed Change (PIC and LPIc UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Allowed Change (PIC and LPIc) UNBUNDLED REMOTE CALL FORW	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE rarding Service, Area Calling, Res rarding Service, Local Calling - Res rarding Service, InterLATA - Res rarding Service, IntraLATA - Res rarding Service - Conversion - rarding Service - Conversion with C) /ARDING - Bus rarding Service, Area Calling - Bus	availab		UEPDD UEPTX UEPSX UEPTX UEPSX Will also apply to y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche / Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC	74.77 13.46 0.00 d voice and/or quest Process. 0.00 101.60 1.49 1.49 1.49 1.49	164.86 60.60 0.00 circuit switche Rates for the 0.00 188.36 3.74 3.74 3.74 0.10 0.10	77.74 50.67 0.00 ed data transm packet capabi 3.63 3.63 3.63 0.10 0.10	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN p le Request/I 7.86 7.86 7.86 7.86 7.86		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Non-Recurring Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw allowed change (PIC and LPIC UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Allowed Change (PIC and LPIC UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port - Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE PARTING SERVICE, Area Calling, Res PARTING SERVICE, Local Calling - Res PARTING SERVICE - CONVERSION - PARTING SERVICE - CONVERSION - PARTING SERVICE - CONVERSION - PARTING SERVICE - CONVERSION - PARTING SERVICE - CONVERSION WITH PARTING SERVICE - CONVERSION - PARTING SERVICE - CONVERSION - PARTING SERVICE - CONVERSION WITH PARTING SERVICE - CONVERSION - PARTING SERVICE, AREA CALLING - BUS PARTING SERVICE, AREA CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE, LOCAL CALLING - BUS PARTING SERVICE - LOCALLING - BUS PARTING SERVICE - LOCAL CALLING - BUS PARTING SERVIC	availab		UEPDD UEPTX UEPSX UEPTX UEPSX UEPTX UEPSX Will also apply to v UEPTX UEPSX UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche / Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC	74.77 13.46 0.00 d voice and/or quest Process. 10.00 101.60 1.49 1.49 1.49 1.49 1.49 1.49	164.86 60.60 0.00 circuit switcher Rates for the 0.00 188.36 3.74 3.74 3.74 0.10 0.10 3.74 3.74	77.74 50.67 0.00 od data transm packet capabi 3.63 3.63 3.63 3.63 3.63 3.63 3.63 3.6	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN ple Request/t 7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw allowed change (PIC and LPIC UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE arding Service, Area Calling, Reserring Service, Local Calling - Researding Service, InterLATA - Researding Service, InterLATA - Researding Service - Conversion - varding Service - Conversion with CD ARDING - Bus arding Service, Area Calling - Bus arding Service, Area Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, InterLATA - Bus	availab		UEPDD UEPTX UEPSX UEPTX UEPSX I UEPTX UEPSX I Will also apply to 0 UEPTX UEPSX UEPTX UEPSX UEPEX UEPVR	UEPDD U1PMA UEPVF ircuit switcher r Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC UERAC UERAC UERTE UERTR	74.77 13.46 0.00 d voice and/or quest Process. 0.00 101.60 1.49 1.49 1.49 1.49 1.49 1.49 1.49 1.49	164.86 60.60 0.00 circuit switcher Rates for the 9.000 188.36 3.74 3.74 3.74 0.10 0.10 3.74 3.74 3.74	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15 3.63 3.63 3.63 3.63 0.10 0.10 3.63 3.63 3.63	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86		s Request Pro	cess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw allowed change (PIC and LPIC UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be N Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE rarding Service, Area Calling, Res rarding Service, Local Calling - Res rarding Service, InterLATA - Res rarding Service - Conversion - rarding Service - Conversion with C) ARDING - Bus rarding Service, Area Calling - Bus rarding Service, Local Calling - Bus rarding Service, Local Calling - Bus rarding Service, Local Calling - Bus rarding Service, Local Calling - Bus rarding Service, Local Calling - Bus rarding Service, Local Calling - Bus rarding Service, InterLATA - Bus rarding Service, InterLATA - Bus	availab		UEPDD UEPTX UEPSX UEPTX UEPSX UEPTX UEPSX Will also apply to v UEPTX UEPSX UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche / Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC	74.77 13.46 0.00 d voice and/or quest Process. 10.00 101.60 1.49 1.49 1.49 1.49 1.49 1.49	164.86 60.60 0.00 circuit switcher Rates for the 0.00 188.36 3.74 3.74 3.74 0.10 0.10 3.74 3.74	77.74 50.67 0.00 od data transm packet capabi 3.63 3.63 3.63 3.63 3.63 3.63 3.63 3.6	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 Wire ISDN ple Request/t 7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86		s Request Pro	ocess.	
Exchange Ports - 2-Wire DID Exchange Ports - DDITS Port capability Exchange Ports - 2-Wire ISDN All Features Offered NOTE: Transmission/usage charge NOTE: Access to B Channel or D C Exchange Ports - 2-Wire ISDN Exchange Ports - 2-Wire ISDN Exchange Ports - 4-Wire ISDN UNBUNDLED PORT with REMOTE C UNBUNDLED PORT with REMOTE C UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Switch-as-is Unbundled Remote Call Forw allowed change (PIC and LPIC UNBUNDLED REMOTE CALL FORW Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw Unbundled Remote Call Forw	- 4-Wire DS1 Port with DID N Port (See Notes below.) s associated with POTS circuit sw hannel Packet capabilities will be Port Channel Profiles N DS1 Port CALL FORWARDING CAPABILITY ARDING SERVICE - RESIDENCE arding Service, Area Calling, Reserring Service, Local Calling - Researding Service, InterLATA - Researding Service, InterLATA - Researding Service - Conversion - varding Service - Conversion with CD ARDING - Bus arding Service, Area Calling - Bus arding Service, Area Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, Local Calling - Bus arding Service, InterLATA - Bus	availab		UEPDD UEPTX UEPSX UEPTX UEPSX I UEPTX UEPSX I Will also apply to 0 UEPTX UEPSX UEPTX UEPSX UEPEX UEPVR	UEPDD U1PMA UEPVF ircuit switcher r Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC UERAC UERAC UERTE UERTR	74.77 13.46 0.00 d voice and/or quest Process. 0.00 101.60 1.49 1.49 1.49 1.49 1.49 1.49 1.49 1.49	164.86 60.60 0.00 circuit switcher Rates for the 9.000 188.36 3.74 3.74 3.74 0.10 0.10 3.74 3.74 3.74	77.74 50.67 0.00 ed data transm packet capabi 0.00 95.15 3.63 3.63 3.63 3.63 0.10 0.10 3.63 3.63 3.63	60.69 32.83 ission by B-Ch lities will be de	3.86 14.17 annels associ termined via the		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86		s Request Pro	ocess.	

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge -	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Names		Name and a second and	. Dianamant					2.00 .01	2.007.007.
						Rec	Nonred First	urring Add'l	Nonrecurring First	Add'l	COMEC	SOMAN	SOMAN	Rates(\$)	SOMAN	SOMAN
	Unbundled Remote Call Forwarding Service - Conversion -						FIISL	Add I	FIISL	Add I	SOMEC	SOWAN	SOWAN	SOWAN	SOWAN	SOWAN
	Switch-as-is			UEPVB	USAC2		0.10	0.10				7.86				
	Unbundled Remote Call Forwarding Service - Conversion with													İ	İ	
	allowed change (PIC and LPIC)			UEPVB	USACC		0.10	0.10								
	LOCAL SWITCHING, PORT USAGE															
End	Office Switching (Port Usage)					0.0011071										ļ
	End Office Switching Function, Per MOU End Office Trunk Port - Shared, Per MOU					0.0011971 0.0002112										
Tand	em Switching (Port Usage) (Local or Access Tandem)					0.0002112										
Tana	Tandem Switching Function Per MOU					0.000194										
	Tandem Trunk Port - Shared, Per MOU					0.0002416								İ	İ	
Com	mon Transport															
	Common Transport - Per Mile, Per MOU					0.000003										
	Common Transport - Facilities Termination Per MOU	ļ				0.0007466								ļ	ļ	
	PORT/LOOP COMBINATIONS - COST BASED RATES	1/2: 2:	-4		and de 10-1-	414414	abla a co Oct	ala Dante						1	1	
	Based Rates are applied where BellSouth is required by FCC ar ares shall apply to the Unbundled Port/Loop Combination - Cos								nd Port coction	of this Data E	vhihit					ļ
	Office and Tandem Switching Usage and Common Transport Us											n Port/Loor	Combination	ns.		
	irst and additional Port nonrecurring charges apply to Not Curr															
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	,						g g				,		T	İ	
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			10.79										
	2-Wire VG Loop/Port Combo - Zone 2		2			15.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			31.74										
UNE	Loop Rates			UEPRX	UEPLX	0.04										<u> </u>
	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2			UEPRX	UEPLX	9.64 14.37										
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPRX	UEPLX	30.59										-
2-Wii	re Voice Grade Line Port Rates (Res)			OLITOX	OLI DX	00.00										1
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire voice Grade unbundled Kentucky extended local dialing															
	parity port with Caller ID - res			UEPRX	UEPRM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled Kentucky Residence Dialing Plan			OLFKA	ULFAF	1.13	21.25	13.45	2.03	2.07		7.00				
ı	without Caller ID	1	1	UEPRX	UEPWE	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire voice unbundled Low Usage Line Port without Caller ID				1	0			50	,,				1	1	†
	Capability			UEPRX	UEPRT	1.15	21.29	15.49	2.85	2.67		7.86	<u> </u>			<u> </u>
FEA1	TURES															
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00				7.86				
LOCA	AL NUMBER PORTABILITY			LIEDDY	LNDCV	0.35										ļ
NON	Local Number Portability (1 per port) RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPRX	LNPCX	0.35										
NON	2-Wire Voice Grade Loop / Line Port Combination - Conversion -													1	1	
	Switch-as-is			UEPRX	USAC2		0.10	0.10				7.86				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -													1		
	Switch with change			UEPRX	USACC		0.10	0.10				7.86				
ADDI	TIONAL NRCs							-								
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent	l														
	Activity	ļ		UEPRX	USAS2	0.00	0.00	0.00				7.86		1	1	<u> </u>
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	<u> </u>	 										ļ			<u> </u>
UNE	Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		1		 	10.79								 	 	
	2-Wire VG Loop/Port Combo - Zone 1	 	2			15.52								 	 	
	2-Wire VG Loop/Port Combo - Zone 3	1	3			31.74							1	†	†	
LINE	Loop Rates	l												1	1	<u> </u>
ONE				UEPBX	UEPLX	9.64										

ONRO	NULE	D NETWORK ELEMENTS - Kentucky			1								1 -		ment: 2		bit: C
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Dee	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Grade Loop (SL1) - Zone 2			UEPBX	UEPLX	14.37										
		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	30.59										ļ
	2-Wire	Voice Grade Line Port (Bus)			LIEBBY .			01.00									
		2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire voice unbundled port with Caller + E484 ID - bus 2-Wire voice unbundled port outgoing only - bus			UEPBX UEPBX	UEPBC UEPBO	1.15 1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67		7.86 7.86				
		2-Wire voice Grade unbundled Kentucky extended local dialing			OLFBA	OLFBO	1.13	21.29	13.43	2.03	2.07		7.00				
		parity port with Caller ID - bus			UEPBX	UEPBM	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.15	21.29	15.49	2.85	2.67		7.86				1
		2-Wire Voice Unbundled Kentucky Business Dialing Plan					-										
		without Caller ID			UEPBX	UEPWF	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire voice unbundled Incoming Only Port without Caller ID															
		Capability			UEPBX	UEPBE	1.15	21.29	15.49	2.85	2.67		7.86				1
	LOCAL	NUMBER PORTABILITY															
		Local Number Portability (1 per port)	ļ	<u> </u>	UEPBX	LNPCX	0.35										
	FEATU				LIEBBY .	1155) (5											
	NOND	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00				7.86				
	NONKI	CURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion -			-	+											
		Switch-as-is			UEPBX	USAC2		0.10	0.10				7.86				
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -			OLFBA	U3AU2		0.10	0.10				7.00				
		Switch with change			UEPBX	USACC		0.10	0.10				7.86				
	ADDIT	IONAL NRCs			02. 27.	00/100		0.10	00				7.00				
		2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
		Activity			UEPBX	USAS2		0.00	0.00				7.86				
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
	UNE P	ort/Loop Combination Rates															
		2-Wire VG Loop/Port Combo - Zone 1		1			10.79										
		2-Wire VG Loop/Port Combo - Zone 2		2			15.52										
	LINE	2-Wire VG Loop/Port Combo - Zone 3		3	-	+	31.74										
	UNE L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	9.64										
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	14.37										
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	30.59										
	2-Wire	Voice Grade Line Port Rates (RES - PBX)		Ŭ	02.1.0	02.2.	00.00										1
		2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -															
		Res			UEPRG	UEPRD	1.15	21.29	15.49	2.85	2.67		7.86				
	LOCAL	NUMBER PORTABILITY															
		Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				7.86				
	FEATU																
		All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00				7.86				
	NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPRG	USAC2		8.45	1.91				7.86				
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			UEPRG	USACZ		0.40	1.91				7.00				
		Conversion - Switch with Change			UEPRG	USACC		8.45	1.91				7.86				
	ADDIT	IONAL NRCs	†		021110	00,100		0.40	1.31			1	7.00			1	
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -		1	1	†											†
	L	Subsequent Activity	<u> </u>	L	UEPRG	USAS2	0.00	0.00	0.00			<u> </u>	7.86		<u> </u>		<u></u>
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt				İ											
		Group						7.86	7.86				7.86				1
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)				\bot	, The state of the										<u> </u>
	UNE P	ort/Loop Combination Rates	ļ	L.	ļ		10 ==										ļ
		2-Wire VG Loop/Port Combo - Zone 1	 	1	 	+ +	10.79			1						1	
-		2-Wire VG Loop/Port Combo - Zone 2	-	2	 	+	15.52			 					-	1	
	LINE I	2-Wire VG Loop/Port Combo - Zone 3	 	3	-	+ +	31.74			 						 	
	SINE L	2-Wire Voice Grade Loop (SL 1) - Zone 1	 	1	UEPPX	UEPLX	9.64			1					1		
		2-Wire Voice Grade Loop (SL 1) - Zone 2	 		UEPPX	UEPLX	14.37			1		1			1	1	+

NRONDLE	D NETWORK ELEMENTS - Kentucky										Ι -			nent: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	30.59										
2-Wire	Voice Grade Line Port Rates (BUS - PBX)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.15	21.29	15.49	2.85	2.67		7.86				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.15	21.29	15.49	2.85	2.67		7.86				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPPX	UEPXE	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area															
	Calling Port without LUD			UEPPX	UEPXF	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port			UEPPX	UEPXG	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled PBX Kentucky Premium Calling Port			UEPPX	UEPXH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD			UEPPX	UEPXJ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPPX	UEPXL	1.15	21.29	15.49	2.85	2.67		7.86				
	Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			UEPPX	UEPXM	1.15	21.29	15.49	2.85	2.67		7.86				
	Discount Room Calling Port			UEPPX	UEPXO	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.15	21.29	15.49	2.85	2.67		7.86				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								
FEAT																
	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00				7.86				
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPPX	USAC2		8.45	1.91				7.86				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPPX	USACC		8.45	1.91				7.86				
ADDIT	TONAL NRCs						00									
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			UEPPX	USAS2	0.00	0.00	0.00				7.00				
	Subsequent Activity PBX Subsequent Activity - Change/Rearrange Multiline Hunt			UEFFX	USASZ	0.00						7.86				
	Group						7.86	7.86				7.86				
	E VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	RT														
UNE F	ort/Loop Combination Rates															
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			10.79										<u> </u>
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			15.52										
	2-Wire VG Coin Port/Loop Combo – Zone 3	ļ	3			31.74										_
UNE L	oop Rates	 	<u> </u>		lues:						1					_
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	9.64										<u> </u>
	2-Wire Voice Grade Loop (SL1) - Zone 2	 	2	UEPCO	UEPLX	14.37					1					_
	2-Wire Voice Grade Loop (SL1) - Zone 3	 	3	UEPCO	UEPLX	30.59					1					_
2-Wire	Voice Grade Line Ports (COIN)										ļ					↓
	2-Wire Coin 2-Way without Operator Screening and without	l		LIEDCO	LIEDDE	4 45	04.00	45.40	0.05	0.07		7.00				
	Blocking (AL, KY, LA, MS)	<u> </u>	<u> </u>	UEPCO	UEPRF	1.15	21.29	15.49	2.85	2.67		7.86			1	
	2-Wire Coin 2-Way with Operator Screening (AL, KY)	l	1	UEPCO	UEPRE	1.15	21.29	15.49	2.85	2.67	1	7.86				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,	l	1	LIEDOO	LIEDDA		04.00	45.00	0.00	0.00		7.00				
	900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking			UEPCO	UEPRA	1.15	21.29	15.49	2.85	2.67		7.86				
	(KY) 2-Wire Coin 2-Way with Operator Screening & Blocking:			UEPCO	UEPKA	1.15	21.29	15.49	2.85	2.67		7.86				-
	900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.15	21.29	15.49	2.85	2.67		7.86				

UNBUNDLE	D NETWORK ELEMENTS - Kentucky				, ,									ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		T
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin Outward without Blocking and without Operator			LIEDOO	LIEDDN	4.45	04.00	45.40	0.05	0.07		7.00				
	Screening (KY, LA, MS)			UEPCO	UEPRN	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS)			UEPCO	UEPRJ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Coin Outward Operator Screening & Blocking: 900/976,															
	1+DDD, 011+, and Local (AL, KY, LA, MS)			UEPCO	UEPCN	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Coin Outward Smartline with 900/976 (all states except			LIEDOO	LIEBOD	4.45	04.00	45.40	0.05	0.07		7.00				
ADDITI	LA)			UEPCO	UEPCR	1.15	21.29	15.49	2.85	2.67	-	7.86		-	 	
	ONAL UNE COIN PORT/LOOP (RC)			UEPCO	URECU	2.57	21.29	15.49	2.85	2.67					 	
	UNE Coin Port/Loop Combo Usage (Flat Rate) NUMBER PORTABILITY	-		ULFCU	UKECU	2.5/	21.29	15.49	∠.85	2.07					-	
				UEPCO	LNPCX	0.35										-
	Local Number Portability (1 per port) CURRING CHARGES - CURRENTLY COMBINED	-		ULFCU	LINEUX	0.35			1	1	-		1	1	1	
NONKE	2-Wire Voice Grade Loop / Line Port Combination - Conversion -				1 -				1	1			1	1	1	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - 2-Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPCO	USAC2		0.10	0.10				7.86				
	Switch with change			UEPCO	USACC		0.10	0.10				7.86				
ADDITI	ONAL NRCs			UEPCO	USACC		0.10	0.10			1	7.00				
ADDITI	2-Wire Voice Grade Loop/Line Port Combination - Subsequent				+											
	Activity			UEPCO	USAS2		0.00	0.00				7.86				
2-WIRE	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	ORT (00,102		0.00	0.00				1.00				
	ort/Loop Combination Rates															
0.12.	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			13.90										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			18.68										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			34.45										
	pop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	12.67										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	17.45										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	33.22										
	Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.23	128.96	64.11	61.92	9.97		7.86				
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.23	128.96	64.11	61.92	9.97		7.86				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.23	128.96	64.11	61.92	9.97		7.86			ļ	ļ
	2-Wire voice Grade unbundled Kentucky extended local dialing parity port with Caller ID - res			UEPFR	UEPRM	1.23	128.96	64.11	61.92	9.97		7.86				
	2-Wire voice unbundles res, low usage line port with Caller ID]]					1	
	(LUM) 2-Wire Voice Unbundled Kentucky Residence Dialing Plan			UEPFR	UEPAP	1.23	128.96	64.11	61.92	9.97	 	7.86				
	without Caller ID	L		UEPFR	UEPWE	1.23	128.96	64.11	61.92	9.97	<u></u>	7.86			<u> </u>	
INTERC	OFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFR	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86				
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFR	1L5XX	0.0095										
FEATU						2.2300			1	1					1	
	All Features Offered			UEPFR	UEPVF	0.00	0.00	0.00		İ		7.86			İ	
	NUMBER PORTABILITY				1											
	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFR	USAC2		9.03	1.87				7.86				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-With-Change			UEPFR	USACC		9.03	1.87				7.86				
2-WIRE	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	ORT (1		0.00		1	1					İ	
	ort/Loop Combination Rates		(• ,	1				1	1					İ	
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		1	13.90										

<u>UNBUND</u> L	ED NETWORK ELEMENTS - Kentucky													ment: 2		bit: C
		Interi									Svc Order Submitted Elec	Svc Order Submitted Manually	Incremental Charge - Manual Svc	Incremental Charge - Manual Svc	Incremental Charge - Manual Svc	Incrementa Charge - Manual Sv
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Order vs. Electronic- Disc 1st	Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	-	2			18.68			1							
LINE	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 Loop Rates	+	3		+	34.45			+ +							
UNE	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	12.67										
	2-Wire Voice Grade Loop (SL2) - Zone 1	+	2	UEPFB	UECF2	17.45					1					
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	33.22			+							
2-Wii	re Voice Grade Line Port (Bus)		1 3	OLFIB	OLGI Z	33.22			+ +							
2-7711	2-Wire voice unbundled port without Caller ID - bus		1	UEPFB	UEPBL	1.23	128.96	64.11	61.92	9.97		7.86				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.23	128.96	64.11	61.92	9.97		7.86				
	2-Wire voice unbundled port with called 12-04-15 Bus			UEPFB	UEPBO	1.23	128.96	64.11	61.92	9.97		7.86				
 	2-Wire voice Grade unbundled Kentucky extended local dialing	1			52. 50	1.20	120.00	04.11	01.02	5.51		7.50	1	 	t	
	parity port with Caller ID - bus			UEPFB	UEPBM	1.23	128.96	64.11	61.92	9.97		7.86	1	1	I	
	2-Wire voice unbundled incoming only port with Caller ID - Bus	1		UEPFB	UEPB1	1.23	128.96	64.11	61.92	9.97		7.86	1	1	1	
	2-Wire Voice Unbundled Kentucky Business Dialing Plan		1	İ	1	0		2		2.3.			İ	İ	1	
	without Caller ID			UEPFB	UEPWF	1.23	128.96	64.11	61.92	9.97		7.86				
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFB	LNPCX	0.35										
INTE	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFB	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86				
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFB	1L5XX	0.0095										
FEAT	TURES															
	All Features Offered			UEPFB	UEPVF	0.00	0.00	0.00				7.86				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFB	USAC2		9.03	1.87				7.86				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch with change			UEPFB	USACC		9.03	1.87				7.86				
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	1														
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			13.90										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			18.68										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			34.45										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	12.67										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	17.45										
	2-Wire Voice Grade Loop (SL2) - Zone 3	1	3	UEPFP	UECF2	33.22										
2-Wii	re Voice Grade Line Port Rates (BUS - PBX)															
	L'a Citattal a la LO al Carlo a Carlo a Dove DDV Tarak Dark Dark			UEPFP	LIEDDO	4.00	404.07	70.05	75.05	0.70		7.00				
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	-	1	UEPFP	UEPPC UEPPO	1.23	164.27	78.65	75.05	8.73		7.86				
	Line Side Unbundled Outward PBX Trunk Port - Bus	+	1	UEPFP	UEPP0	1.23	164.27 164.27	78.65	75.05 75.05	8.73 8.73		7.86 7.86				
	Line Side Unbundled Incoming PBX Trunk Port - Bus 2-Wire Voice Unbundled PBX LD Terminal Ports	+	1	UEPFP	UEPLD	1.23		78.65								
			1	UEPFP	UEPXA	1.23 1.23	164.27 164.27	78.65 78.65	75.05 75.05	8.73 8.73		7.86 7.86				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports	+		UEPFP	UEPXA			78.65	75.05	8.73		7.86				
	2-Wire Voice Unbundled PBX I'oli Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Port	+	 	UEPFP	UEPXB	1.23 1.23	164.27 164.27	78.65 78.65	75.05 75.05	8.73		7.86	-	-		
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	+	 	UEPFP	UEPXC	1.23	164.27	78.65	75.05 75.05	8.73		7.86	-	-		
1	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD	+	1	OLITE	ULFAD	1.23	104.27	70.00	75.05	0.73	1	7.00		1	1	1
	Capable Port			UEPFP	UEPXE	1.23	164.27	78.65	75.05	8.73		7.86	1	1	I	
	2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area	1	1	02111	OLI AL	1.23	107.27	70.00	75.05	0.73		1.00			 	
	Calling Port without LUD			UEPFP	UEPXF	1.23	164.27	78.65	75.05	8.73		7.86	Ì	l	I	
	2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port	+		UEPFP	UEPXG	1.23	164.27	78.65	75.05	8.73		7.86	 	 	t	
	2-Wire Voice Unbundled PBX Kentucky Premium Calling Port	1	+	UEPFP	UEPXH	1.23	164.27	78.65	75.05	8.73		7.86			-	
	2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port	1	+	0=.11	JEI AII	1.20	104.27	70.00	70.00	0.70		7.50			-	
				UEPFP	UEPXJ	1.23	164.27	78.65	75.05	8.73		7.86			1	
	IWITHOUT I UI)										1					1
	without LUD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1				1.20										

UNB	UNDLE	D NETWORK ELEMENTS - Kentucky			•		1					Ι			ment: 2		bit: C
CATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Do.	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
		Room Calling Port			UEPFP	UEPXM	1.23	164.27	78.65	75.05	8.73		7.86				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
		Discount Room Calling Port			UEPFP	UEPXO	1.23	164.27	78.65	75.05	8.73		7.86				.
	LOCAL	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port L NUMBER PORTABILITY			UEPFP	UEPXS	1.23	164.27	78.65	75.05	8.73		7.86				<u> </u>
	LUCAL	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00								
	INTER	OFFICE TRANSPORT			OLFIF	LINE CE	3.13	0.00	0.00								
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
		Termination			UEPFP	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86				
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile								77.7							
		or Fraction Mile			UEPFP	1L5XX	0.0095										
	FEATU	JRES															
		All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00				7.86				
	NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
		Combination - Conversion - Switch-as-is			UEPFP	USAC2		9.03	1.87				7.86				
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFP	LICACO		0.00	1.87				7.00				
LINDI	INDI ED	Combination - Conversion - Switch with change PORT/LOOP COMBINATIONS - COST BASED RATES			UEPFP	USACC		9.03	1.87				7.86				
UND		E VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT														1
		ort/Loop Combination Rates	I														
	OITE !	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1		-	21.30										
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			26.08										
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			41.85										
	UNE L	oop Rates															1
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	12.67						7.86				
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	17.45						7.86				
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	33.22						7.86				
	UNE P	ort Rate			UEDDV			000.11		400.00							
-	NOND	Exchange Ports - 2-Wire DID Port ECURRING CHARGES - CURRENTLY COMBINED			UEPPX	UEPD1	8.63	336.11	27.75	132.37	9.31	1	7.86				
	NONKI	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion				-											
		with BellSouth Allowable Changes			UEPPX	USA1C		7.85	1.87				7.86				
	ADDIT	IONAL NRCs			OZ. I X	00/110		7.00					1.00				
	7.55	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX	USAS1		32.25	32.25				7.86				
	Teleph	none Number/Trunk Group Establisment Charges															
		DID Trunk Termination (One Per Port)			UEPPX	NDT	0.00	0.00	0.00				7.86				
		Additional DID Numbers for each Group of 20 DID Numbers			UEPPX	ND4	0.00	0.00	0.00				7.86				
		DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX	ND5	0.00	0.00	0.00				7.86				<u> </u>
	-	Reserve Non-Consecutive DID numbers	ļ		UEPPX	ND6	0.00	0.00	0.00	1		<u> </u>	7.86			ļ	<u> </u>
	1.0041	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				7.86				
	LOCAL	L NUMBER PORTABILITY			UEPPX	LNPCP	3.15	0.00	0.00			1					
	2-WIDI	Local Number Portability (1 per port) E ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LII	NE SIDI	DOD1		LINECE	3.13	0.00	0.00				-		-		
		ort/Loop Combination Rates	I SIDI	I						1							+
	0.12	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -															
		UNE Zone 1		1	UEPPB UEPPR	1	25.69										
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -															1
		UNE Zone 2		2	UEPPB UEPPR		31.92										
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -									·						
	<u> </u>	UNE Zone 3		3	UEPPB UEPPR		50.21								1		
<u> </u>	UNE L	oop Rates	ļ	<u> </u>	LIEDDD LIEDDS	1101.07	40.10					ļ	7.00			ļ	<u> </u>
	-	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB UEPPR	USL2X	16.10						7.86		1	1	
i		2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB UEPPR	USL2X	22.33						7.86		I		
		2-Wire ISDN Digital Grade Loop - UNE Zone 2 2-Wire ISDN Digital Grade Loop - UNE Zone 3	 	3	UEPPB UEPPR	USL2X USL2X	40.63						7.86		 	1	
	UNF P	ort Rate			OLITO OLFFR	JULZA	40.03			1		 	7.00		 	+	
	U.1L	Exchange Port - 2-Wire ISDN Line Side Port	-		UEPPB UEPPR	UEPPB	9.59	320.53	289.13	92.19	17.56	 	7.86		t	 	

ONBOND	LEC	NETWORK ELEMENTS - Kentucky														ment: 2		bit: C
CATEGOR	Y	RATE ELEMENTS	Interi m	Zone	E	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
								Rec	Nonrec		Nonrecurring					Rates(\$)		
								Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NO	NRE	CURRING CHARGES - CURRENTLY COMBINED																
		2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																
		Combination - Conversion			UEPPB	UEPPR	USACB	0.00	22.77	17.00				7.86				
AD	DITIO	ONAL NRCs																
LO	CAL	NUMBER PORTABILITY																ĺ
		Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-C	CHAN	INEL USER PROFILE ACCESS:																
		CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								1
		CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								1
		CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								1
B-C	CHAN	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C,MS, 8	(TN)														1
		CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								
		CVS (EWSD)		1	UEPPB	UEPPR	U1UCE	0.00	0.00	0.00			1					1
		CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								
US		ERMINAL PROFILE		1									İ					1
		User Terminal Profile (EWSD only)		1	UEPPB	UEPPR	U1UMA	0.00	0.00	0.00			İ					1
VEI		AL FEATURES																
		All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00								
INT	ΓERO	OFFICE CHANNEL MILEAGE																
		Interoffice Channel mileage each, including first mile and																
		facilities termination			LIFPPR	UEPPR	M1GNC	29.12	47.34	31.78	22.77	8.75		7.86				
		Interoffice Channel mileage each, additional mile				UEPPR	M1GNM	0.01	0.00	0.00		0.70		7.86				
4-W		DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	K PORT		02	02		0.01	0.00	0.00				7.00				
		rt/Loop Combination Rates	1															
0.11		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																+
		Zone 1		1	UEPPP			170.06										
		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		-	OLFFF			170.00										
		Zone 2		2	UEPPP			197.70										
-		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			OLFFF			197.70									-	
				3	UEPPP			204.25										
		Zone 3 op Rates		3	UEPPP			381.35										
UN				_	UEPPP		LICL 4D	00.47						7.00				
		4-Wire DS1 Digital Loop - UNE Zone 1		1			USL4P	86.47						7.86				
		4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	114.10						7.86				
		4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	297.76						7.86				
UN		rt Rate					l											
		Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	83.59	736.16	382.74	159.48	48.82		7.86				
NO		CURRING CHARGES - CURRENTLY COMBINED																
		4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port			1													
		Combination - Conversion -Switch-as-is			UEPPP		USACP	0.00	81.70	1.37				7.86				
AD		ONAL NRCs																
		4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-																
		Inward/two way Tel Nos. (except NC)			UEPPP		PR7TF		0.54					7.86				1
		4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -																
		Outward Tel Numbers (All States except NC)			UEPPP		PR7TO		12.71	12.71				7.86				
		4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -			1									[<u> </u>	_	
		Subsequent Inward Tel Numbers			UEPPP		PR7ZT		25.41	25.41				7.86		ļ		ļ
LO		NUMBER PORTABILITY																
		Local Number Portability (1 per port)			UEPPP		LNPCN	1.75										
INT		ACE (Provsioning Only)																
		Voice/Data			UEPPP	-	PR71V	0.00	0.00	0.00								
		Digital Data			UEPPP		PR71D	0.00	0.00	0.00								
		Inward Data			UEPPP		PR71E	0.00	0.00	0.00								
Nev		Additional "B" Channel																
		New or Additional - Voice/Data B Channel			UEPPP		PR7BV	0.00	15.48					7.86				
		New or Additional - Digital Data B Channel			UEPPP		PR7BF	0.00	15.48					7.86				
		New or Additional Inward Data B Channel			UEPPP		PR7BD	0.00	15.48					7.86				
CA	LL T	YPES																
		Inward		1	UEPPP		PR7C1	0.00	0.00	0.00			1					1
— 		Outward			UEPPP		PR7C0	0.00	0.00	0.00		İ	İ				1	

NDUNDLE	ED NETWORK ELEMENTS - Kentucky			1	1						0	001		ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		•
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Intero	ffice Channel Mileage															
	Fixed Each Including First Mile			UEPPP	1LN1A	96.27	105.52	98.46	23.09	20.49		7.86				
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.23										
	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE F	Port/Loop Combination Rates		<u> </u>			4.47.00										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		147.99										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		175.62										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		359.28										
UNE L	.oop Rates 4-Wire DS1 Digital Loop - UNE Zone 1	 	1	UEPDC	USLDC	86.47						7.86		 	1	
	4-Wire DS1 Digital Loop - UNE Zone 1 4-Wire DS1 Digital Loop - UNE Zone 2	-		UEPDC	USLDC	114.10			1			7.86		-	 	
	4-Wire DS1 Digital Loop - UNE Zone 2 4-Wire DS1 Digital Loop - UNE Zone 3	 	3	UEPDC	USLDC	297.76						7.86		-	-	
LINE	Port Rate	-	J	OLFDC	USLDC	291.76			1			1.00		-	 	
ONE P	4-Wire DDITS Digital Trunk Port	1		UEPDC	UDD1T	61.52	780.61	375.52	176.19	16.98		7.86		1		
NOND	ECURRING CHARGES - CURRENTLY COMBINED			UEPDC	UDDTT	61.52	700.01	3/3.32	170.19	10.90		7.00				+
NONK	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	1			+ +		i							 	1	\vdash
	- Switch-as-is			UEPDC	USAC4		92.84	46.70				7.86				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination			OLI DO	00/104		32.04	40.70				7.00				+
	- Conversion with DS1 Changes			UEPDC	USAWA		92.84	46.70				7.86				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination			OLI DO	OOAWA		32.04	40.70	1			7.00				+
	- Conversion with Change - Trunk			UEPDC	USAWB		92.84	46.70				7.86				
ADDIT	TONAL NRCs			OLI DO	CONTRA		32.04	40.70	1			7.00				+
7.55	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -				+											t
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		15.09	15.09				7.86				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent			OLI DO	OBTIA		10.00	10.00				7.00				
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		15.09	15.09				7.86				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		15.09	15.09				7.86				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsgnt Chan															
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		15.09	15.09				7.86				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															1
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		15.09	15.09				7.86				
BIPOL	AR 8 ZERO SUBSTITUTION															
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	730.00				7.86				Ī
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	730.00				7.86				
Altern	ate Mark Inversion															
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Telepl	hone Number/Trunk Group Establisment Charges															
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00	0.00	0.00				7.86				
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00		0.00				7.86				
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00	0.00	0.00				7.86				
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00	0.00	0.00				7.86				ļ
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00	0.00	0.00				7.86				
_	Reserve Non-Consecutive DID Nos.	!		UEPDC	ND6	0.00	0.00	0.00				7.86		ļ	ļ	
- In	Reserve DID Numbers	D::-		UEPDC	NDV	0.00	0.00	0.00	1			7.86			ļ	
Dedic	ated DS1 (Interoffice Channel Mileage) - FX/FC0 for 4-Wire DS1	Digital	Loop	with 4-Wire DDITS	Trunk Port											
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)			UEPDC	1LNO1	96.04	105.52	98.46	23.09	20.49		7.86				
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.23	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities	l			I											
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.45	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)			UEPDC	1LNO3	0.00	0.00	0.00								

UNBL	JNULEI	D NETWORK ELEMENTS - Kentucky											,		ment: 2		bit: C
CATEG	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	Charge - Manual Sv Order vs. Electronic
														1st	Add'l	Disc 1st	Disc Add'
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		l.
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.45	0.00	0.00								
		Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								
		Central Office Termininating Point			UEPDC	CTG	0.00										
		EDS1 LOOP WITH CHANNELIZATION WITH PORT n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	votiono			_											
		system can have up to 24 combinations of rates depending on			her of norte used												
		S1 Loop	type an	la man	liber or ports useu												
		4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	86.47	0.00	0.00								
		4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	114.10	0.00	0.00								
		4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	297.76	0.00	0.00								
		SO Channelization Capacities (D4 Channel Bank Configuration	ıs)														
		24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	111.16	0.00	0.00	<u> </u>			7.86				
		48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	222.32	0.00	0.00		•		7.86				
		96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	444.64	0.00	0.00				7.86				
		144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	666.96	0.00	0.00				7.86				
		192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	889.28	0.00	0.00				7.86				
		240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,111.60	0.00	0.00				7.86				
		288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG UEPMG	VUM28 VUM38	1,333.92 1,778.56	0.00	0.00				7.86 7.86				
		480 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,223.20	0.00	0.00				7.86				<u> </u>
		576 DS0 Channel Capacity - 1 per 24 DS1s			UEPMG	VUM57	2,667.84	0.00	0.00				7.86				1
		672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3.112.48	0.00	0.00				7.86				
		ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	Chann	eliztio					0.00				7.00				
		mum System configuration is One (1) DS1, One (1) D4 Channe															
		es of this configuration functioning as one are considered Ac															
		NRC - Conversion (Currently Combined) with or without															
		BellSouth Allowed Changes			UEPMG	USAC4	0.00	94.30	4.24				7.86				
		n Additions at End User Locations Where 4-Wire DS1 Loop wit				oination Curre	ently Exists and										
		ot Currently Combined) in all states, except in Density Zone 1	of Top	8 MSA	A's												
		1 DS1/D4 Channel Bank - Additionally Add NRC for each Port															
		and Assoc Fea Activation			UEPMG	VUMD4	0.00	718.89	469.86	149.83	17.77		7.86				
		r 8 Zero Substitution															
		Clear Channel Capability Format, superframe - Subsequent Activity Only			UEPMG	CCOSF	0.00	0.00	730.00				7.86				
		Clear Channel Capability Format - Extended Superframe -			UEPING	CCOSF	0.00	0.00	730.00				7.86				-
		Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	730.00				7.86				
		ate Mark Inversion (AMI)			OLI WO	CCOLI	0.00	0.00	730.00				7.00				
		Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
		Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	Exchan	nge Ports Associated with 4-Wire DS1 Loop with Channelization	n with	Port													
	Exchan	nge Ports															
		Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.15	0.00	0.00	0.00	0.00		7.86				ļ
		Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.15	0.00	0.00	0.00	0.00		7.86				
				l	LIEBBY	LIED () (- 0.5				
	1	Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX UEPPX	UEP1X UEPDM	1.15 8.65	0.00	0.00	0.00	0.00		7.86 7.86			1	
	Foaturo	e Activations - Unbundled Loop Concentration		-	ULFFA	OEPDIVI	8.05	0.00	0.00	0.00	0.00	 	7.80		-	1	
	- eature	Feature (Service) Activation for each Line Side Port Terminated			1	+									-	1	
		in D4 Bank		l	UEPPX	1PQWM	0.62	25.40	13.41	4.17	4.15		7.86				
		Feature (Service) Activation for each Trunk Side Port Terminated				~	0.02	20.40	10.41	7.17	4.13		7.00				
		in D4 Bank		l	UEPPX	1PQWU	0.62	78.15	19.68	59.05	11.54		7.86				
		one Number/ Group Establishment Charges for DID Service															
		DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00	<u> </u>			7.86			<u> </u>	
		DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				7.86				
. —		Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00				7.86				
		Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00				7.86		l		
		Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				7.86				

CATEGORY RATE ELEMENTS Mind Zone BCS USOC RATES(S) Submitted Submitted Charges College Submitted Charges College Submitted Charges College Submitted Charges College Submitted Charges College Submitted Submitted Charges College Submitted														ı			
CATEGORY RATE ELEMENTS Intel Zone BCS USOC RATES(\$)	<u>JNDLED</u>	NETWORK ELEMENTS - Kentucky															bit: C
CATEGORY RATE ELEMENTS Intel Zone BCS USOC RATE(SE) ERCENOIS ERCONOIS ERCENOIS						Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental						
CATEORY RATE ELEMENTS												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
NATE SUPPLY NATIONAL PROPERTY NATIONAL P			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
Bectonia Branch	GORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
Test			m									,	P		Electronic-	Electronic-	Electronic-
Cold Number Portability I per pint															Add'l	Disc 1st	Disc Add'l
Local Number Portability Local Number Portab														131	Auu	Diac 1at	Disc Auu i
Cost Number Permissibility							Dan	Nonre	curring	Nonrecurring	g Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
PRATURES Vertical and Optional	Local Nu	umber Portability														·	
URBONICE O'ENTREX PORTICOPY COMBINATIONS. COST BASES PURES UEPPY 0.00 0	L	ocal Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00							·	
MERINATER FORTADOP COMBINATIONS - COST BASED RATE	FEATUR	ES - Vertical and Optional															
UNBLONGED CENTREX PORTILOP COMBINATIONS - COST BASED RATES 1. Cost Based Rates are applied where Belliston it required by PCC and/or State Commission rule to provide Unbundied Local Switching or 8witch Ports. 2. Features shall apply to the Unbundied PortLoop Combination and the state section in the same manner as they are applied to the Stand-Adores Unbundied Port section of this Rate State	Local Sv	witching Features Offered with Line Side Ports Only														·	
1. Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission rule to provide Unbundled Local Switching or Switch Ports. 2. Features shall apply to the Unbundled Port section of this Rate Exhibit. 3. End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements accept for UNE Coin PortLoop Combination. 4. The Irst and additional Province Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements accept for UNE Coin PortLoop Combinations. 4. The Irst and additional Province Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements accept for UNE Coin PortLoop Combinations. 5. Market Rates for Unbundled Centres PortLoop Combined see Seals, until further notice. WINE-PORTLOOP Combination Rates (Review) Port Common PortLoop Combinations will be negotiated on an Individual Case Basis, until further notice. WINE-PORTLOOP Combination Rates (Review) Port Common PortLoop Combination Rates (Review) Port Common PortLoop Combination Rates (Review) Port Common PortLoop Combination Rates (Review) Port Common Port (Supplement Volume Common Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates (Review) Port Common Portloop Combination Rates	P	All Features Available			UEPPX	UEPVF	0.00	0.00	0.00							·	
2. Features shall apply to the Unburndled Port/Loop Combination - Cost Based Rate section in this same manner as they are applied to the Stand-Alone Unburndled Port section of this page and the Port section of this rate exhibit shall and combinations of Component Port of New Coin Port/Loop Combinations (A. The first and additional Port noncuring charges apply to Net Currently Combined Combos. For Currently Combined Combos. For Currently Combined Combos. For Currently Combined Combos. For Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combos Combos Combos Combos. The Currently Combos Com	NDLED CE	ENTREX PORT/LOOP COMBINATIONS - COST BASED RATES	S													·	
2. Features shall apply to the Unburndled Port/Loop Combination - Cost Based Rate section in this same manner as they are applied to the Stand-Alone Unburndled Port section of this page and the Port section of this rate exhibit shall and combinations of Component Port of New Coin Port/Loop Combinations (A. The first and additional Port noncuring charges apply to Net Currently Combined Combos. For Currently Combined Combos. For Currently Combined Combos. For Currently Combined Combos. For Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combined Combos. The Currently Combos Combos Combos Combos. The Currently Combos Com	1. Cost E	Based Rates are applied where BellSouth is required by FCC	and/or	State C	ommission rule to	orovide Unbu	undled Local S	witching or Sv	vitch Ports.							·	
S. End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements except Tor UNE Coin Port/Loop Combinations apply also and are categorized accordingly. 8. Marker Rates for full-bridged demirer Perfor/Loop Combination will be negotiated on an Individual Case Basis, until further notice.										dled Port secti	ion of this Rate	Exhibit.					
A. The first and additional Port nonrecurring charges apply to Not Currently Combined Combos. For Currently Combined Combos, the nonrecurring charges shall be those identified in the Nonrecurring charges shall													Coin Port/Lo	op Combinat	ions.		
Spark and are categorized accordingly.																Additional NF	Cs mav
S. Market Rates for Unbundled Centrox PortLoop Combination will be regoritated on an Individual Case Basis, until further notice.			arrenting	0011101	ilea dombos. Tor v	ourrently oo	mbmed comb	oo, and normed	arring orial gos	onan be mose	o identifica ili t	ne nomeou	ining Curry	citily combine	ou scotions.	additional res	.oo may
UNIF. P CENTREX - (Table In ALF, CA, XY, LA, MS, STN only) 2-Wire Vot Loop?-Wire Vote Grade Port (Centrex) Port Combo UNIF. PortLoop Combination Rates (Non-Design) UNIF. PortLoop Combination Rates (Non-Design) UNIF. PortLoop Combination Rates (Non-Design) UNIF. PortLoop Combination Rates (Non-Design) UNIF. PortLoop Combination Rates (PortLoop Combination Rates (PortLoop Rates Combination Rates (PortLoop Rates Combination Rates (PortLoop Rates C			he necc	ntiated :	on an Individual Car	se Rasis un	til further notic	Δ	ı			I	1	ı	1		
2-Wire Vol Loop/2-Wire Voice Grade Port (Centrex) Combo				Juaceu	on an marvidual Ca	l pasis, uiii	in runnier nouc	i.	1	1	1	1	-	1	1	 '	1
Number N				-									-			 	-
2-Wire Vol Loop/2-Wire Volce Grade Port (Centres) Port Combo 1 UEP91 10.79				-									-			 	-
Non-Design 1 UEP91 10.79			-			 	 	-		1	1	-			 	 '	
2. Wire VS Loop/2-Wire Voice Grade Port (Centrex/Port Combo-Non-Design 1. UEP91 15.52 1. VIVE PORT Loop (Centrex) Port Combo-Non-Design 2. Wire VS Loop/2-Wire Voice Grade Port (Centrex) Port Combo-Design 1. UEP91 1. VIVE PORT Loop (Centrex) Port Combo-Design 1. UEP91 1. VIVE VS Loop/2-Wire Voice Grade Port (Centrex) Port Combo-Design 1. UEP91 1. VIVE VS Loop/2-Wire Vs Loop (Centrex) Port Combo-Design 1. UEP91 1. VIVE VS Loop/2-Wire Vs Loop (Centrex) Port Combo-Design 2. Wire VS Loop/2-Wire Vs Loop (Centrex) Port Combo-Design 2. Wire VS Loop/2-Wire Vs Loop (Centrex) Port Combo-Design 2. Wire VS Loop/2-Wire Vs Loop (Centrex) Port Combo-Design 3. UEP91 3. 8.60 2. Wire VS Loop/2-Wire Vs Loop (Centrex) Port Combo-Design 3. UEP91 3. 8.60 3. UEP91 3. 4.37 3. UEP91 3. 4.37 3. UEP91 3. 4.37 3. UEP91 3. 4.37 3. UEP91 3. 4.37 3. UEP91 3. 4.37 3. UEP91 3. 4.37 3. UEP91 3. 4.37 3. UEP91 3. 3. UEP91 3. 3. UEP91 3. 3. UEP91 3. 3. UEP91 3.			1	4	LIEDO4		10.70									1 '	
Non-Design 2 UEP91 15.52				1	UEP91		10.79										-
2-Wire Vol Loop/2-Wire Volce Grade Port (Centrex)Port Combo-Non-Design				_	LIEBO4		45.50									1 '	
Non-Design 3 UEP91 31.74				2	UEP91		15.52						ļ				
UNE Port/Loop Combination Rates (Besign)				_												1 '	
2 Wife VIc Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 1 UEP91 13.8.2				3	UEP91		31.74										
Design																 '	
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2 UEP91 18.60		. , ,	i													1 '	
Design 2 UPP1 18.60				1	UEP91		13.82									 '	
Design Surprise Supplementary Suppleme																1 '	
Design				2	UEP91		18.60									L	
UNE Loop Rate																1	
2-Wire Voice Grade Loop (St. 1) - Zone 1				3	UEP91		34.37										
2-Wire Voice Grade Loop (SL 1) - Zone 2																	
2-Wire Voice Grade Loop (SL 1) - Zone 3 3 UEP91 UECS1 30.59 7.86 2-Wire Voice Grade Loop (SL 2) - Zone 1 1 UEP91 UECS2 12.67 7.86 2-Wire Voice Grade Loop (SL 2) - Zone 2 2 UEP91 UECS2 17.45 7.86 2-Wire Voice Grade Loop (SL 2) - Zone 3 3 UEP91 UECS2 33.22 7.86 2-Wire Voice Grade Loop (SL 2) - Zone 3 3 UEP91 UECS2 33.22 7.86 2-Wire Voice Grade Loop (SL 2) - Zone 3 3 UEP91 UECS2 33.22 7.86 2-Wire Voice Grade Port (Centrex) Basic Local Area UEP91 UEPYA 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex with Caller ID)18asic Local Area UEP91 UEPYH 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area UEP91 UEPYH 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area UEP91 UEPYH 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center - 800 Service Term - Basic Local Area UEP91 UEPY2 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area UEP91 UEPY2 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area UEP91 UEPY2 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port terminated on 800 Service Term - Basic Local Area UEP91 UEPY2 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex) UEP91 UEP92 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex) UEP91 UEPQA 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex) UEP91 UEPQA 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex) UEP91 UEPQA 1.15 21.29 15.49 2.85 2.67 7.86 2-Wire Voice Grade Port (Centrex) UEP91 UEPQA 1.15 21.29 15.49 2.85																	
2-Wire Voice Grade Loop (St. 2) - Zone 1																1	
2-Wire Voice Grade Loop (St. 2) - Zone 2 2 UEP91 UECS2 17.45				3												L	
2-Wire Voice Grade Loop (SL 2) - Zone 3 3 UEP91 UECS2 33.22	2	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	12.67						7.86				
UNE Ports All States (Except North Carolina and Sout Carolina)																L	
All States (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Port (Centrex) Basic Local Area UEP91 UEPYA 1.15 21.29 15.49 2.85 2.67 7.86				3	UEP91	UECS2	33.22						7.86				
2-Wire Voice Grade Port (Centrex) Basic Local Area UEP91 UEPYA 1.15 21.29 15.49 2.85 2.67 7.86																	
2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area UEP91 UEPYB 1.15 21.29 15.49 2.85 2.67 7.86	All State	s (Except North Carolina and Sout Carolina)															
Area	2	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86				
2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area	2	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local														1	
Area			L		UEP91	UEPYB	1.15	21.29	15.49	2.85	2.67	<u> </u>	7.86	<u> </u>	<u> </u>	<u> </u>	
Area	2	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
Center)2 Basic Local Area			1		UEP91	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86		Ì	1 '	
Center)2 Basic Local Area	2	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service UEP91			1		UEP91	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86		Ì	1 '	
Term - Basic Local Area																	
2-Wire Voice Grade Port terminated in on Megalink or equivalent UEP91			1		UEP91	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86		Ì	1 '	
Basic Local Area						İ	1							İ	İ	ſ	
2-Wire Voice Grade Port Terminated on 800 Service Term - UEP91			1		UEP91	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86		Ì	1 '	
Basic Local Area							1					İ			İ		
AL, KY, LA, MS, & TN Only			1		UEP91	UEPY2	1.15	21 29	15 49	2 85	2 67		7.86		Ì	1 '	
2-Wire Voice Grade Port (Centrex) UEP91 UEPQA 1.15 21.29 15.49 2.85 2.67 7.86							0	220	.5.40	2.00	2.07	İ			1		
2-Wire Voice Grade Port (Centrex 800 termination) UEP91 UEPQB 1.15 21.29 15.49 2.85 2.67 7.86			1		UEP91	UEPQA	1 15	21 29	15 49	2.85	2.67	1	7 86			 	
												 			-		
2-Wire Voice Grade Port (Centrex with Caller ID)1 UEP91 UEPQH 1.15 21.29 15.49 2.85 2.67 7.86			 									1			 		
2-Wire Voice Grade Port (Centrex from diff Serving Wire			 		OL1 01	ا الا	1.15	21.29	15.49	2.00	2.07	1	7.00	1	1		
2-vivile votice Grade Fort (Centrex from this Serving vivile UEP91 UEP9M 1.15 21.29 15.49 2.85 2.67 7.86			1		I IFP91	LIEPOM	1 15	21.20	15 /0	2.95	2.67		7 96		Ì	1 '	
Certiferia Certiferia Vire Voice Grade Port, Diff Serving Wire Center - 800 Service			1	1	OL1 31	JLI QIVI	1.15	21.29	15.49	2.00	2.07	1	7.00	1	1	 '	1
2-vvire voice Grade Port, Dill Serving wire Center - 800 Service UEP91 UEPQZ 1.15 21.29 15.49 2.85 2.67 7.86			1		I IEDQ1	LIEDO7	1 15	24.20	15.40	2.05	2.67		7 00		Ì	1 '	

Version 3Q02: 09/06/02 Page 163 of 362

UNBUN	DLE	D NETWORK ELEMENTS - Kentucky					1								ment: 2		bit: C
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Charge - Manual Sv Order vs. Electronic Disc Add'
								Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)	L	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86				
Lo	ocal S	Switching															
		Centrex Intercom Funtionality, per port			UEP91	URECS	0.8873						7.86				
Lo		lumber Portability															
		Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Fe	eature				UEP91	UEPVF	0.00						7.00				
		All Standard Features Offered, per port All Select Features Offered, per port			UEP91	UEPVS	0.00	405.66					7.86 7.86				
		All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00	405.66				1	7.86				
NI.	ARS	All Centres Control i eatures Offered, per port			OLI 91	OLF VO	0.00			 			1.00		 	 	
		Unbundled Network Access Register - Combination	1		UEP91	UARCX	0.00	0.00	0.00				7.86			-	
		Unbundled Network Access Register - Indial	1		UEP91	UAR1X	0.00	0.00	0.00				7.86		1	1	
		Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				7.86				
М	liscell	aneous Terminations															
2-		Trunk Side															
		Trunk Side Terminations, each			UEP91	CENA6	10.51	92.18	15.82	52.16	5.30		7.86				
In		ice Channel Mileage - 2-Wire															
		Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	29.11						7.86				
		Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.01						7.86				
		Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4		nnel Bank Feature Activations															
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.62						7.86				
		Factors Astination on D.4 Channel Book EV line Cide Land Clat			LIEDOA	400000	0.62						7.00				
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop		-	UEP91	1PQW6	0.62						7.86				
		Slot			UEP91	1PQW7	0.62						7.86				
		Feature Activation on D-4 Channel Bank Centrex Loop Slot -			OLF91	IFQW/	0.02						7.00				
		Different Wire Center			UEP91	1PQWP	0.62						7.86				
		Directions while content			02. 0.		0.02						7.00				
		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.62						7.86				
		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
		Slot			UEP91	1PQWQ	0.62						7.86				
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.62						7.86				
No	lon-Re	curring Charges (NRC) Associated with UNE-P Centrex															
		Conversion - Currently Combined Switch-As-Is with allowed															
		changes, per port			UEP91	USAC2		0.102	0.102				7.86				
		Conversion of Existing Centrex Common Block	ļ		UEP91	USACN		18.95	8.32	.					ļ	ļ	
		New Centrex Standard Common Block	ļ		UEP91	M1ACS	0.00	669.80	78.32	111.05	13.27		7.86			-	
		New Centrex Customized Common Block	1		UEP91	M1ACC	0.00	669.80	78.32	111.05	13.27	1	7.86		 	1	1
		Secondary Block, per Block	1		UEP91 UEP91	M2CC1 URECA	0.00	78.32 72.75	78.32	13.27	13.27	1	7.86 7.86		 	1	1
<u> </u>	NE-D	NAR Establishment Charge, Per Occasion CENTREX - 5ESS (Valid in All States)	 		UEPSI	UKECA	0.00	12.15		 		-	7.86			 	-
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo	1		 	+				H				1	1	 	
		ort/Loop Combination Rates (Non-Design)			 	+				 					 	 	
- 10	(2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1			+								1	1	†	1
		Non-Design	1	1	UEP95		10.79								1	I	
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design	<u> </u>	2	UEP95		15.52			<u> </u>		<u> </u>	<u> </u>	<u> </u>		<u> </u>	<u></u>
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design		3	UEP95		31.74							<u></u>			
UI	NE Po	ort/Loop Combination Rates (Design)						_	•		•						
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	l]												
		Design	ļ	1	UEP95		13.82								ļ	ļ	
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	l	_	LIEBOE											1	
		Design	<u> </u>	2	UEP95		18.60			ļ				ļ	ļ	-	
1		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Design	l	3	UEP95		34.37									1	

DUNDLE	D NETWORK ELEMENTS - Kentucky	1	1	1							0	001	Attachr			bit: C
EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring	Disconnect				Rates(\$)		•
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE L	pop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1			UEP95	UECS1	9.64						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	14.37						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	30.59						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	12.67						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	17.45						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	33.22						7.86				
	ort Rate															
All Sta																
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86				
_	2-Wire Voice Grade Port (Centrex 800 termination)	ļ		UEP95	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86			ļ	
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local	l	1	l											1	
	Area			UEP95	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP95	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP95	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP95	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP95	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86				
AL, KY	, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86				
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.8873						7.86				
Local	lumber Portability															
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Featur																
	All Standard Features Offered, per port			UEP95	UEPVF	0.00	105.00					7.86				
	All Select Features Offered, per port		<u> </u>	UEP95	UEPVS	0.00	405.66					7.86				
NADO	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00						7.86				
NARS	Haland Halanda Anna Barista Canting			LIEDOS	LIADOV	0.00	0.00	0.00				7.00				
	Unbundled Network Access Register - Combination		<u> </u>	UEP95	UARCX	0.00	0.00	0.00				7.86				
	Unbundled Network Access Register - Indial		<u> </u>	UEP95	UAR1X	0.00	0.00	0.00				7.86				
	Unbundled Network Access Register - Outdial		<u> </u>	UEP95	UAROX	0.00	0.00	0.00				7.86				
	aneous Terminations															
2-wire	Trunk Side			LIEDOS	OFNIDO	40.54	00.40	45.00	50.40	5.00		7.00				
4 10/:	Trunk Side Terminations, each Digital (1.544 Megabits)	1	<u> </u>	UEP95	CEND6	10.51	92.18	15.82	52.16	5.30		7.86			1	!
4-wire	DS1 Circuit Terminations, each	 		UEP95	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86				
_	DS0 Channels Activated, each	-	-	UEP95	M1HD0	0.00	15.09	11.14	60.69	3.80		7.86			-	-
Intorof	Fice Channel Mileage - 2-Wire	-	 	ULFSO	INITIDO	0.00	15.09		+			7.80			 	-
interof	Interoffice Channel Facilities Termination	-	 	UEP95	MIGBC	29.11			+			7.86			 	-
-	Interoffice Channel Facilities Termination Interoffice Channel mileage, per mile or fraction of mile	 		UEP95	MIGBC	0.01						7.86				
1	e Activations (DS0) Centrex Loops on Channelized DS1 Service	<u> </u>	 	OLF 30	IVIIODIVI	0.01			+			1.00			 	-
Foative		· ·	├	 	+							7.86				-
	nnol Pank Foature Activations															
	Innel Bank Feature Activations			HED05	100//9	0.62			1							
	nnel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.62						7.86				

NBUNDLI	ED NETWORK ELEMENTS - Kentucky	1			1						Core Condition	Core Cord	Attachr			oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		l .
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.62						7.86				
_	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP95	TPQW7	0.62						7.80				
	Different Wire Center			UEP95	1PQWP	0.62						7.86				
					i i											
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.62						7.86				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP95	1PQWQ	0.62						7.86				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.62	-					7.86				
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex				1 2											
	NRC Conversion Currently Combined Switch-As-Is with allowed				i i											
	changes, per port			UEP95	USAC2		0.102	0.102				7.86				
	Conversion of Existing Centrex Common Block, each			UEP95	USACN		18.95	8.32	111.05			7.86				
	New Centrex Standard Common Block			UEP95	M1ACS M1ACC	0.00	669.80	78.32 78.32	111.05 111.05	13.27 13.27		7.86 7.86				
	New Centrex Customized Common Block NAR Establishment Charge, Per Occasion			UEP95 UEP95	URECA	0.00	669.80 72.75	78.32	111.05	13.27		7.86				
UNF-	P CENTREX - DMS100 (Valid in All States)			OLF 93	UKLCA	0.00	12.13					7.00				
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo				1											
	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP9D		10.79										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo		_			4==0										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP9D		15.52										
	Non-Design		3	UEP9D		31.74										
UNE	Port/Loop Combination Rates (Design)		3	OLI 3D	+	31.74										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9D		13.82										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9D		18.60										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		3	UEP9D		34.37										
LINE	Design Loop Rate		3	UEP9D	+	34.37	-									
OIAL I	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	9.64						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	14.37						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	30.59						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	12.67						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	17.45						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	33.22						7.86				
	Port Rate STATES				+		-									
ALL	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Fort (Centrex 800 termination)Basic Local			-2.02	J 1/1	1.13	21.23	10.73	2.00	2.01		7.00				
	Area			UEP9D	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local			l												
_	Area			UEP9D	UEPYC	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area	l		UEP9D	UEPYD	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local			OLFBD	ULFID	1.13	21.29	15.49	2.00	2.07		7.00				
	Area			UEP9D	UEPYE	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local															
	Area			UEP9D	UEPYF	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local			LIEDOD	LIEDY CO		2.25		2 25							
	Area	ļ		UEP9D	UEPYG	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	1.15	21.29	15.49	2.85	2.67		7.86				
-+	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local	1		OL1 3D	JEITT	1.13	21.29	13.43	2.00	2.07		7.00				
1	Area	l		UEP9D	UEPYU	1.15	21.29	15.49	2.85	2.67	1	7.86			1	l

UNDUNDLE	D NETWORK ELEMENTS - Kentucky			1							1 -			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonre		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local															
	Area		<u> </u>	UEP9D	UEPYV	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local			UEP9D	UEPTS	1.15	21.29	15.49	2.00	2.07		7.00				-
	Area			UEP9D	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			OLI 3D	OLI III	1.10	21.23	10.43	2.00	2.07		7.00				+
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3					_	-									
	Basic Local Area			UEP9D	UEPYJ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2 Basic Local Area			UEP9D	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3															
	Basic Local Area			UEP9D	UEPYO	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPYP	1.15	21.29	15.49	2.85	2.67		7.86				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPTP	1.15	21.29	15.49	2.00	2.07		7.00				
	Basic Local Area			UEP9D	UEPYQ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			02. 05	02 Q		220	101.10	2.00	2.0.		7.00				1
	Basic Local Area			UEP9D	UEPYR	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3															1
	Basic Local Area			UEP9D	UEPYS	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3															
	Basic Local Area			UEP9D	UEPY4	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3															
	Basic Local Area			UEP9D	UEPY5	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			OLFBD	OLFTO	1.13	21.29	13.45	2.03	2.07		7.00				
	Basic Local Area			UEP9D	UEPY7	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			02. 02	02		21120	10.10	2.00	2.07		7.00			1	İ
	Term			UEP9D	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	Basic Local Area			UEP9D	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic															
	Local Area			UEP9D	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86				
AL, K	Y, LA, MS, SC, & TN Only			LIEDOD	LIEBOA	1.15	04.00	45.40	0.05	0.07		7.86				
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)		<u> </u>	UEP9D UEP9D	UEPQA UEPQB	1.15 1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67		7.86 7.86				
	2-Wire Voice Grade Port (Centrex 600 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	1.15	21.29	15.49	2.85	2.67		7.86			1	
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.15	21.29	15.49	2.85	2.67		7.86			İ	
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	1.15	21.29	15.49	2.85	2.67		7.86				1
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3		<u> </u>	UEP9D	UEPQV	1.15	21.29	15.49	2.85	2.67		7.86			ļ	ļ
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3		<u> </u>	UEP9D	UEPQ3	1.15	21.29	15.49	2.85	2.67	1	7.86		1	1	
	2-Wire Voice Grade Port (Centrex with Caller ID) 2-Wire Voice Grade Port (Centrex/Caller ID/Msq Wtg Lamp		<u> </u>	UEP9D	UEPQH	1.15	21.29	15.49	2.85	2.67	-	7.86			-	
	Indication)3		1	UEP9D	UEPQW	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.15	21.29	15.49	2.85	2.67		7.86			t	
	2-Wire Voice Grade Fort (Centrex/Msg Wtg Lamp Indication)3				J X0	1.10	21.23	10.40	2.00	2.07	1	7.00		1	†	
	2			UEP9D	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86			1	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.15	21.29	15.49	2.85	2.67		7.86		<u> </u>		
				1												
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.15	21.29	15.49	2.85	2.67		7.86			<u> </u>	
1 -	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3	l	<u> </u>	UEP9D	UEPQQ	1.15	21.29	15.49	2.85	2.67	L	7.86		<u> </u>		

UNBl	UNDLE	D NETWORK ELEMENTS - Kentucky													ment: 2		bit: C
CATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP9D	UEPQ7	1.15	21.29	15.49	2.85	2.67		7.86				
		Term			UEP9D	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86				
		2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86				
	Local S	Switching															
		Centrex Intercom Funtionality, per port			UEP9D	URECS	0.8873						7.86				
	Local I	Number Portability															
		Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										
	Feature																
		All Standard Features Offered, per port			UEP9D	UEPVF	0.00						7.86				
		All Select Features Offered, per port			UEP9D	UEPVS	0.00	405.66					7.86				
		All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00						7.86				
	NARS																
		Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				7.86				
		Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				7.86				
		Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00				7.86				
		laneous Terminations															
	2-Wire	Trunk Side															
		Trunk Side Terminations, each			UEP9D	CEND6	10.51	92.18	15.82	52.16	5.30		7.86				
	4-Wire	Digital (1.544 Megabits)															
		DS1 Circuit Terminations, each			UEP9D	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86				
		DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	15.09					7.86				
	Interof	fice Channel Mileage - 2-Wire															
		Interoffice Channel Facilities Termination			UEP9D	MIGBC	29.11						7.86				
		Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.01						7.86				
		e Activations (DS0) Centrex Loops on Channelized DS1 Service	е														
	D4 Cha	nnel Bank Feature Activations															
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.62						7.86				
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.62						7.86				
		Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.62						7.86				
		Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.62						7.86				
	1	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.62						7.86				
	1	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
		Slot			UEP9D	1PQWQ	0.62			 			7.86			 	
	Na = =	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.62			ļ .			7.86	1		-	
	Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex								ļ							↓
		NRC Conversion Currently Combined Switch-As-Is with allowed	1		LIEBOD	110466]					l	I	
	-	changes, per port			UEP9D	USAC2		0.102	0.102	ļ .			7.86	1		-	
	-	Conversion of existing Centrex Common Block, each			UEP9D	USACN		18.95	8.32				7.86			-	!
	_	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	669.80	78.32	111.05	13.27		7.86				
L		New Centrex Customized Common Block			UEP9D	M1ACC	0.00	669.80	78.32	111.05	13.27	ļ	7.86				
	1	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.75					7.86				l

UNBUNDI	ED NETWORK ELEMENTS - Kentucky												Attach	ment: 2	Fyhil	bit: C
ONDONDE.	TOTAL ELEMENTO ROMANN										Svc Order	Svc Order	Incremental		Incremental	
											Submitted	Submitted		Charge -	Charge -	Charge -
		1									Elec	Manually				Manual Svo
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per LSK	per LSK	Electronic-		Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						D	Nonred	urring	Nonrecurring	g Disconnect		•	oss	Rates(\$)	•	•
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE-	P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
	Non-Design		1	UEP9E		10.79										
i I	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo															
\vdash	Non-Design		2	UEP9E		15.52										
i I	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		3	LIEDOE		04.74										
 	Non-Design		3	UEP9E		31.74										
UNE	Port/Loop Combination Rates (Design)	1	+						 				-	1	 	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP9E		13.82			1						1	
 	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	+ '-	OL1 3L	+	13.02			t				1	1	t	+
(I	Design		2	UEP9E		18.60			I				1		I	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1			1	10.00			I		<u> </u>		 	1	I	†
1 1	Design		3	UEP9E		34.37			1						1	
UNE	Loop Rate					99										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	9.64						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	14.37						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	30.59						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	12.67						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	17.45						7.86				
igsquare	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	33.22						7.86				
	Port Rate															
AL, F	L, KY, LA, MS, & TN only						21.22					=				
	2-Wire Voice Grade Port (Centrex) Basic Local Area		1	UEP9E	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9E	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86				
\vdash	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local	-	-	UEF9E	UEFIB	1.15	21.29	15.49	2.00	2.07		7.00				
	Area			UEP9E	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			OLI OL	OLI III	1.13	21.23	13.43	2.00	2.01		7.00				
i I	Center)2 Basic Local Area			UEP9E	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP9E	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	:														
<u> </u>	- Basic Local Area		<u> </u>	UEP9E	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86	<u> </u>	<u> </u>	<u> </u>	<u></u>
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area		1	UEP9E	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86				
AL, K	Y, LA, MS, & TN Only	1							ļ							
$\vdash \vdash \vdash$	2-Wire Voice Grade Port (Centrex)	1	1	UEP9E	UEPQA	1.15	21.29	15.49	2.85	2.67		7.86		ļ		
	2-Wire Voice Grade Port (Centrex 800 termination)	1	1	UEP9E	UEPQB	1.15	21.29	15.49	2.85	2.67		7.86	1	ļ	!	
+	2-Wire Voice Grade Port (Centrex with Caller ID)1	1	1	UEP9E	UEPQH	1.15	21.29	15.49	2.85	2.67	1	7.86	 	 	1	1
1	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86	1		I	
 	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1	1	OL1 3L	OLI QIVI	1.13	21.29	15.49	2.00	2.07		7.00	1	1	t	+
(I	Term			UEP9E	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86				
		1			J X_	1.13	21.23	10.40	2.00	2.07		7.50	1	1	1	
(I	2-Wire Voice Grade Port terminated in on Megalink or equivalent	:		UEP9E	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86	1		I	
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86				
Loca	Switching															
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.8873						7.86				
Local	Number Portability							-								
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35						7.86				
Featu			ļ	L	<u> </u>				ļ				ļ	ļ	ļ	
1 1	All Standard Features Offered, per port	1	1	UEP9E	UEPVF	0.00	10=					7.86		ļ		
			1	UEP9E	UEPVS	0.00	405.66		1	ı	1	7.86	l	1	1	1
	All Select Features Offered, per port	-	+									7.00				
NARS	All Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00						7.86				

ONRONDE	ED NETWORK ELEMENTS - Kentucky	ı		T	1						10	06		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00								
	ellaneous Terminations															
2-1011	e Trunk Side Trunk Side Terminations, each			UEP9E	CEND6	10.51	92.18	15.82	52.16	5.30		7.86				
4-10/17	e Digital (1.544 Megabits)			UEP9E	CENDO	10.51	92.10	15.62	32.10	5.30	1	7.00				1
4-7711	DS1 Circuit Terminations, each			UEP9E	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86				
	DS0 Channel Activated Per Channel			UEP9E	M1HD0	0.00	15.09	77.74	00.03	3.00	1	7.86				
Interd	office Channel Mileage - 2-Wire			OLI OL	WITIEG	0.00	10.00					7.00				1
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	29.11						7.86				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.01						7.86				
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e														1
	hannel Bank Feature Activations		1								Ì					
<u> </u>	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.62						7.86				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot		<u> </u>	UEP9E	1PQW6	0.62						7.86				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP9E	1PQW7	0.62						7.86				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP9E	1PQWP	0.62						7.86				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.62						7.86				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			LIEDOE	400140	0.00						7.00				
	Slot			UEP9E	1PQWQ	0.62 0.62						7.86				
Non	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.62						7.86				
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex NRC Conversion Currently Combined Switch-As-Is with allowed														-	
	changes, per port			UEP9E	USAC2		0.102	0.102				7.86				
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		18.95	8.32			1	7.00				
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	669.80	78.32	111.05	13.27		7.86				
	New Centrex Customized Common Block			UEP9E	M1ACC	0.00	669.80	78.32	111.05	13.27		7.86				
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	72.75					7.86				
UNE-	P CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)															
2-Wir	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
	Non-Design		1	UEP93		10.79										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	l	l		1									_	
	Non-Design	ļ	2	UEP93		15.52					ļ					<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_	LIEDOS		04 7.									1	
	Non-Design	 	3	UEP93	+	31.74					1				 	
UNE	Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	 	 		+						1				 	
	Design	1	1	UEP93		13.82									I	
-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	 	+-	OLF 33	+	13.02					}				+	1
	Design		2	UEP93		18.60									1	
<u> </u>	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	<u> </u>			10.00									1	1
	Design	1	3	UEP93		34.37									I	
UNE	Loop Rate															
1	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP93	UECS1	9.64					Ì					1
<u> </u>	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	14.37										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	30.59										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	12.67										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	17.45										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	33.22										<u> </u>
	Port Rate	ļ	<u> </u>												ļ	
AL, K	Y, LA, MS, & TN only	<u> </u>	<u> </u>	L							<u> </u>				ļ	<u> </u>
ı	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86				<u> </u>

NRONDE	ED NETWORK ELEMENTS - Kentucky			1										nent: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec	urring	Nonrecurring					Rates(\$)		•
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP93	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP93	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			UEP93	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86				
	Center)2 Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP93	UEPYIVI	1.15	21.29	15.49	2.85	2.07		7.80				
	Term - Basic Local Area			UEP93	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			OLI SO	OLI IZ	1.10	21.20	10.40	2.00	2.07		7.00				
	- Basic Local Area			UEP93	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP93	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86			I	
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP93	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			LIEBOO	LIEDO7	4.45	04.00	45.40	0.05	2.67		7.00				
_	Term			UEP93	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86			-	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port Terminated in on Wegalink of equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86				
Local	Switching			OLI 93	OLI QZ	1.10	21.23	10.40	2.00	2.07		7.00				
Looui	Centrex Intercom Funtionality, per port			UEP93	URECS	0.8873						7.86				
Local	Number Portability				01.200										1	
	Local Number Portability (1 per port)			UEP93	LNCCC	0.35										
Featu	res															
	All Standard Features Offered, per port			UEP93	UEPVF	0.00						7.86				
	All Centrex Control Features Offered, per port			UEP93	UEPVC	0.00						7.86				
NARS																
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00								
_	Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial			UEP93 UEP93	UAR1X UAROX	0.00	0.00	0.00							-	
Misco	Illaneous Terminations			UEP93	UARUA	0.00	0.00	0.00	-						-	
	e Trunk Side				+											
2-7711	Trunk Side Terminations, each			UEP93	CEND6	10.51	92.18	15.82	52.16	5.30		7.86				
4-Wire	e Digital (1.544 Megabits)				5220		32.10	.5.02	52.10	3.00					1	
	DS1 Circuit Terminations, each			UEP93	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86			1	1
	DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	15.09					7.86				
Intero	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP93	MIGBC	29.11						7.86				
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.01						7.86				
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	е														
D4 Ch	nannel Bank Feature Activations			LIEDOS	100000	0.62						7.00			1	1
-	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.62			+			7.86			+	-
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.62						7.86			I	
-	Feature Activation on D-4 Channel Bank FX Trunk Side Loop			02. 00		0.02						7.50			-	<u> </u>
	Slot			UEP93	1PQW7	0.62						7.86				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -					-										
	Different Wire Center			UEP93	1PQWP	0.62			l			7.86			<u></u>	
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.62						7.86				
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop															
	Slot			UEP93	1PQWQ	0.62						7.86			ļ	
B1	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.62						7.86			1	
Non-F	Recurring Charges (NRC) Associated with UNE-P Centrex NRC Conversion Currently Combined Switch-As-Is with allowed				+ +										1	
	INFO CONVENSION CURRENTLY COMBINED SWITCH-AS-IS WITH AllOWED I		1	i	USAC2			0.102	1					l	1	ì

UNBU	NDLE	NETWORK ELEMENTS - Kentucky												Attachi	ment: 2	Exhil	oit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			•••									-		Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	I	I
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Conversion of Existing Centrex Common Block, each			UEP93	USACN		18.95	8.32				7.86				
		New Centrex Standard Common Block			UEP93	M1ACS	0.00	669.80	78.32	111.05	13.27		7.86				
		New Centrex Customized Common Block			UEP93	M1ACC	0.00	669.80	78.32	111.05	13.27		7.86				
		NAR Establishment Charge, Per Occasion			UEP93	URECA	0.00	72.75					7.86				
	Note 1	Required Port for Centrex Control in 1AESS, 5ESS & EWSD															
	Note 2	- Requres Interoffice Channel Mileage															
		Requires Specific Customer Premises Equipment															
	Note: F	Rates displaying an "R" in Interim column are interim and sub	ject to	rate tru	e-up as set forth in	General Tern	ns and Condition	ons.			•						

UNBUNDLE	D NETWORK ELEMENTS - Louisiana													ment: 2		oit: C
													Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonrec		Managarinia	g Disconnect		ļ.	000	Rates(\$)		
-					-	Rec	First	Add'l	First		COMEC	COMAN	SOMAN		SOMAN	SOMAN
The "7	I one" shown in the sections for stand-alone loops or loops as part of	of a com	hinatio	n refers to Geographi	ically Deavers	aged LINE Zones									SOWAN	SOWAN
	www.interconnection.bellsouth.com/become a clec/html/interconne			r reiers to Geograpin	ically Deavers	aged OIVE Zones	. TO VIEW GEOG	napriicany Dea	veraged ONE Z	one besignation	is by Certifa	i Onice, reie	i to internet vv	ebsite.		
	L SUPPORT SYSTEMS	I COLOTI.II	1		1	1			1			ı		ı		
	: (1) Electronic Service Order: CLEC should contact its contract	t negot	iator if	it prefers the state	specific elec	tronic service o	rdering charge	s as ordered b	ov the State Co	ommissions. T	he electron	ic service o	dering charg	e currently co	ntained in th	is rate
	t is the BellSouth regional electronic service ordering charge.															
	(2) Any element that can be ordered electronically will be bill															lv. For
	elements that cannot be ordered electronically at present per t															
	ng charge, SOMAN, will be applied to a CLECs bill when it sub				oo oa.o	.go.,	o o go	20 2			· aoig oap			0.0		oaaa.
O G G G G	Electronic OSS Charge, per LSR, submitted via BST's OSS	l l	LOICE	o Benooutin.												
	interactive interfaces (Regional)				SOMEC		3.50								, '	ł
UNE SERVICE	DATE ADVANCEMENT CHARGE															
NOTE:	The Expedite charge will be maintained commensurate with	BellSou	th's FC	C No.1 Tariff, Section	on 5 as appl	icable.										ĺ
	UNE Expedite Charge per Circuit or Line Assignable USOC, per			,											i	ĺ
LI	Day	<u></u>		ALL UNE	SDASP	<u> </u>	200.00		<u> </u>			<u></u>		<u> </u>	<u>. </u>	<u> </u>
UNBUNDLED	EXCHANGE ACCESS LOOP														1	ĺ
2-WIR	E ANALOG VOICE GRADE LOOP															<u> </u>
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	12.90	36.54	16.87				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	23.33	36.54	16.87				15.20				L
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	48.43	36.54	16.87				15.20				L
	Loop Testing - Basic 1st Half Hour			UEANL	URET1		33.17	33.17				15.20			<u> </u>	
	Loop Testing - Basic Additional Half Hour			UEANL	URETA		19.28	19.28				15.20				
	CLEC to CLEC Conversion Charge Without Outside Dispatch														, '	1
	(UVL-SL1)			UEANL	UREWO		15.75	8.93				15.20				
	Engineering Information Document (EI)			UEANL	UEANM		13.04	13.04								
	Manual Order Coordination for UVL-SL1s (per loop) Order Coordination for Specified Conversion Time for UVL-SL1			UEANL	UEAMC		7.92	7.92								
	(per LSR)			UEANL	OCOSL		17.56	17.56							, '	1
2-WID	E Unbundled COPPER LOOP			OLANL	OCOSL		17.50	17.50								
Z-VVIK	2-Wire Unbundled Copper Loop - Non-Designed Zone 1		1	UEQ	UEQ2X	12.40	35.27	15.60				15.20				
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	l i		UEQ	UEQ2X	14.32	35.27	15.60				15.20			$\overline{}$	
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	l i		UEQ	UEQ2X	16.87	35.27	15.60				15.20				
	Order Coordination 2 Wire Unbundled Copper Loop - Non-															
	Designed (per loop)			UEQ	USBMC		7.92	7.92							, '	1
	Engineering Information Document			UEQ			13.04	13.04								
	Loop Testing - Basic 1st Half Hour			UEQ	URET1		33.17	33.17				15.20			i	
	Loop Testing - Basic Additional Half Hour			UEQ	URETA		19.28	19.28				15.20				
	CLEC to CLEC Conversion Charge Without Outside Dispatch															1
	(UCL-ND)			UEQ	UREWO		14.25	7.42				15.20				ļ
	EXCHANGE ACCESS LOOP															
2-WIR	E ANALOG VOICE GRADE LOOP														<u>'</u>	
1 1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	l			LIEALS										, '	l
\vdash	Ground Start Signaling - Zone 1	ļ	1	UEA	UEAL2	14.93	102.10	65.72							<u>'</u>	
1 1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	l	2	UEA	UEAL2	25.05	102.10	65.72				45.00			, '	i
\vdash	Ground Start Signaling - Zone 2 2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	-	2	UEA	UEAL2	25.35	102.10	65.72				15.20				
			3	UEA	UEAL2	50.46	102.10	65.72				15.20			, '	ł
 	Ground Start Signaling - Zone 3 Order Coordination for Specified Conversion Time (per LSR)	1	3	UEA	OCOSL	50.46	102.10	05.72				15.20				
 	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	 		OLA	JUUGL	1	17.30		1	1				1		
	Battery Signaling - Zone 1	1	1	UEA	UEAR2	14.93	102.10	65.72				15.20			, '	ĺ
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1		<i></i> (J L / 11 \Z	14.33	102.10	03.72		1		10.20				
1	Battery Signaling - Zone 2	1	2	UEA	UEAR2	25.35	102.10	65.72				15.20			, '	ĺ
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1	Ť			20.00	.02.10	33.72				.0.20			$\overline{}$	
1 1	Battery Signaling - Zone 3	l	3	UEA	UEAR2	50.46	102.10	65.72				15.20			, '	l
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		17.56		İ					l		ſ
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.59	36.30				15.20			1	ĺ
4-WIR	E ANALOG VOICE GRADE LOOP								<u> </u>							ĺ
	4-Wire Analog Voice Grade Loop - Zone 1			UEA	UEAL4	30.81	127.40	91.02				15.20			i	<u> </u>
	4-Wire Analog Voice Grade Loop - Zone 2		2	UEA	UEAL4	38.32	127.40	91.02				15.20				
	4-Wire Analog Voice Grade Loop - Zone 3		3	UEA	UEAL4	60.39	127.40	91.02				15.20			, ,	l

Version 3Q02: 09/06/02 Page 173 of 362

UNBUND	LED	NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Fyhil	oit: C
ONDOND		THE TWO RIVER ELEMENTO LOGISTATIO										Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec			Manual Svc	Manual Svc	Manual Svc
CATEGORY	Y	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						***			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
														151	Add I	DISC ISI	DISC Add I
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		17.56									
		CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.59	36.30				15.20				
2-W		ISDN DIGITAL GRADE LOOP															
		2-Wire ISDN Digital Grade Loop - Zone 1			UDN	U1L2X	22.09	113.34	76.96				15.20				
		2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	35.28	113.34	76.96				15.20				
		2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	65.18	113.34	76.96				15.20				
		Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		17.56									
		CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.49	44.09				15.20				
2-W		ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	AHBLE	LOOP	' T												
	E	2 Wire Unbundled ADSL Loop including manual service inquiry				LIALOV	40.00	447.00	00.00				45.00		1		
\vdash		& facility reservation - Zone 1	 	1	UAL	UAL2X	12.29	117.08	68.36	 			15.20	-	 	-	-
	I.	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 2		2	UAL	UAL2X	14.09	117.08	68.36]		1	15.20		I		
\vdash	!	& facility reservation - Zone 2 2 Wire Unbundled ADSL Loop including manual service inquiry	 		UAL	UALZA	14.09	117.08	08.36	 		-	15.20		 		
		& facility reservation - Zone 3		3	UAL	UAL2X	15.75	117.08	68.36]		1	15.20		I		
 		Order Coordination for Specified Conversion Time (per LSR)		3	UAL	OCOSL	13.73	17.56	00.30	1			13.20				
-		2 Wire Unbundled ADSL Loop without manual service inquiry &			OAL	00000		17.50									
		facility reservation - Zone 1		1	UAL	UAL2W	12.29	92.83	56.02				15.20				
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
		facility reservaton - Zone 2		2	UAL	UAL2W	14.09	92.83	56.02				15.20				
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
	l	facility reservaton - Zone 3		3	UAL	UAL2W	15.75	92.83	56.02				15.20				
	-	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		17.56									
		CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		86.07	40.34				15.20				
2-W		HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
	1	2 Wire Unbundled HDSL Loop including manual service inquiry															
		& facility reservation - Zone 1		1	UHL	UHL2X	9.79	125.50	76.77				15.20				
	- 13	2 Wire Unbundled HDSL Loop including manual service inquiry		2	UHL	UHL2X	11.52	405.50	70.77				45.00				
		& facility reservation - Zone 2		2	UHL	UHL2X	11.52	125.50	76.77	 			15.20				
	ľ	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 3		3	UHL	UHL2X	12.74	125.50	76.77				15.20				
+		Order Coordination for Specified Conversion Time (per LSR)		3	UHL	OCOSL	12.74	17.56	10.11	1			13.20				
+		2 Wire Unbundled HDSL Loop without manual service inquiry			OTIL	00000		17.50		1							
		and facility reservation - Zone 1		1	UHL	UHL2W	9.79	101.24	64.43				15.20				
		2 Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 2		2	UHL	UHL2W	11.52	101.24	64.43				15.20		1		
		2 Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 3		3	UHL	UHL2W	12.74	101.24	64.43				15.20		1		
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		17.56									
		CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.00	40.34				15.20				
4-W		HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP						ļ					ļ		
		4 Wire Unbundled HDSL Loop including manual service inquiry						,]		1					
\vdash		and facility reservation - Zone 1	ļ	1	UHL	UHL4X	16.24	153.26	104.54	ļ			15.20				
		4-Wire Unbundled HDSL Loop including manual service inquiry		_	l	111111 457	40.0-	450.00	10151]		1	45.00		I		
\vdash		and facility reservation - Zone 2	 	2	UHL	UHL4X	16.65	153.26	104.54	 			15.20	-	 		-
		4-Wire Unbundled HDSL Loop including manual service inquiry		3	UHL	UHL4X	17.34	150.00	104.54]		1	15.20		I		
\vdash		and facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)	├	3	UHL	OCOSL	17.34	153.26 17.56	104.54	 		-	15.∠0				
 		4-Wire Unbundled HDSL Loop without manual service inquiry			OI IL	UCUSL		17.50		 				-	 	-	-
1 1		and facility reservation - Zone 1		1	UHL	UHL4W	16.24	129.00	92.20				15.20		1		
		4-Wire Unbundled HDSL Loop without manual service inquiry		<u> </u>		J	10.24	.20.00	32.20				.0.20		1		
		and facility reservation - Zone 2		2	UHL	UHL4W	16.65	129.00	92.20]		1	15.20				
		4-Wire Unbundled HDSL Loop without manual service inquiry								†							
		and facility reservation - Zone 3		3	UHL	UHL4W	17.34	129.00	92.20			1	15.20		I		
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		17.56									
		CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.00	40.34		•		15.20				
4-W		DS1 DIGITAL LOOP															
\vdash		4-Wire DS1 Digital Loop - Zone 1	ļ	1	USL	USLXX	85.70	245.16	152.98	ļ			15.20		ļ		
		4-Wire DS1 Digital Loop - Zone 2		2	USL	USLXX	194.96	245.16	152.98	l		l	15.20		1	l]

UNBUND	LEDI	NETWORK ELEMENTS - Louisiana													Δttach	ment: 2	Eyhil	oit: C
ONDOND	LLD	VETWORK ELEMENTO Edulation											Svc Order	Svc Order	Incremental		Incremental	Incremental
													Submitted	Submitted		Charge -	Charge -	Charge -
													Elec		Manual Svc	Manual Svc		Manual Svc
CATEGORY	Y	RATE ELEMENTS	Interi	Zone	В	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m							.,,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
															1st	Add'l	Disc 1st	Disc Add'l
															151	Add I	DISC ISI	DISC Add I
								Rec	Nonrec	urring	Nonrecurring	g Disconnect			oss	Rates(\$)		
									First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
\perp		Nire DS1 Digital Loop - Zone 3		3	USL		USLXX	491.94	245.16	152.98				15.20				
\perp		der Coordination for Specified Conversion Time (per LSR)			USL		OCOSL		17.56									
		EC to CLEC Conversion Charge without outside dispatch			USL		UREWO		100.93	42.98				15.20				
4-W		.2, 56 OR 64 KBPS DIGITAL GRADE LOOP			LIDI		LIDI 40	00.00	101.00	05.40				45.00				
\vdash		Wire Unbundled Digital 19.2 Kbps Wire Unbundled Digital 19.2 Kbps			UDL UDL		UDL19 UDL19	30.99 36.78	121.86 121.86	85.48 85.48	-			15.20 15.20				
\vdash		Wire Unbundled Digital 19.2 Kbps Wire Unbundled Digital 19.2 Kbps			UDL		UDL19	38.92	121.86	85.48	-			15.20		-		
\vdash		Wire Unbundled Digital Loop 56 Kbps - Zone 1			UDL		UDL56	30.99	121.86	85.48				15.20				
		Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL		UDL56	36.78	121.86	85.48				15.20				
		Vire Unbundled Digital Loop 56 Kbps - Zone 3			UDL		UDL56	38.92	121.86	85.48				15.20				
		der Coordination for Specified Conversion Time (per LSR)			UDL		OCOSL	55.02	17.56	55.10								
		Vire Unbundled Digital Loop 64 Kbps - Zone 1	İ	1	UDL		UDL64	30.99	121.86	85.48				15.20			1	
		Vire Unbundled Digital Loop 64 Kbps - Zone 2		2	UDL		UDL64	36.78	121.86	85.48				15.20				
		Vire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL		UDL64	38.92	121.86	85.48				15.20				
		der Coordination for Specified Conversion Time (per LSR)			UDL		OCOSL		17.56									
\perp		EC to CLEC Conversion Charge without outside dispatch			UDL		UREWO		101.97	49.67				15.20				
2-W		bundled COPPER LOOP																
		Wire Unbundled Copper Loop/Short including manual service						40.00						4= 00				
\vdash		uiry & facility reservation - Zone 1		1	UCL		UCLPB	12.29	116.18	67.46				15.20				
		Wire Unbundled Copper Loop/Short including manual service		2	UCL		UCLPB	14.09	440.40	67.46				45.00				
		uiry & facility reservation - Zone 2 Wire Unbundled Copper Loop/Short including manual service		2	UCL		UCLPB	14.09	116.18	67.46				15.20				
		quiry & facility reservation - Zone 3		3	UCL		UCLPB	15.75	116.18	67.46				15.20				
		der Coordination for Unbundled Copper Loops (per loop)		3	UCL		UCLMC	15.75	7.92	7.92	1			13.20				
\vdash		Wire Unbundled Copper Loop/Short without manual service			002		COLIVIO		7.02	1.02								
		quiry and facility reservation - Zone 1		1	UCL		UCLPW	12.29	91.92	55.12				15.20				
		Wire Unbundled Copper Loop/Short without manual service							0.1.0									
		uiry and facility reservation - Zone 2		2	UCL		UCLPW	14.09	91.92	55.12				15.20				
	2-V	Vire Unbundled Copper Loop/Short without manual service																
		uiry and facility reservation - Zone 3		3	UCL		UCLPW	15.75	91.92	55.12				15.20				
\sqcup		der Coordination for Unbundled Copper Loops (per loop)			UCL		UCLMC		7.92	7.92								
		Nire Unbundled Copper Loop/Long - includes manual srvc.																
\vdash		uiry and facility reservation - Zone 1		1	UCL		UCL2L	17.21	116.18	67.46				15.20				
		Wire Unbundled Copper Loop/Long - includes manual svc.		2			1101 01	04.00	440.40	07.40				45.00				
\vdash		uiry and facility reservation - Zone 2		2	UCL		UCL2L	24.98	116.18	67.46	-			15.20				
		Wire Unbundled Copper Loop/Long - includes manual svc. purity and facility reservation - Zone 3		3	UCL		UCL2L	39.57	116.18	67.46	I			15.20		I	1	
\vdash		der Coordination for Unbundled Copper Loops (per loop)	}	3	UCL		UCLZL	38.37	7.92	7.92	 	1		15.20		 	1	
+-+		Wire Unbundled Copper Loop/Long - without manual service	 		OOL		COLIVIO		1.52	1.52	 					 	 	
		quiry and facility reservation - Zone 1		1	UCL		UCL2W	17.21	91.92	55.12	I			15.20		I	1	
		Wire Unbundled Copper Loop/Long - without manual service							202									
		uiry and facility reservation - Zone 2		2	UCL		UCL2W	24.98	91.92	55.12				15.20		1		
		Wire Unbundled Copper Loop/Long - without manual service	İ														1	
		uiry and facility reservation - Zone 3		3	UCL		UCL2W	39.57	91.92	55.12				15.20		<u> </u>	<u> </u>	
		der Coordination for Unbundled Copper Loops (per loop)			UCL		UCLMC		7.92	7.92		_						
		EC to CLEC Conversion Charge without outside dispatch			1]	
		CL-Des)	<u> </u>		UCL		UREWO		91.92	42.47				15.20				
4-W		OPPER LOOP	ļ		ļ		ļ											
		Nire Copper Loop/Short - including manual service inquiry					1101.40	00.67	400.00	00.00	I			45.00		I	1	
\vdash		d facility reservation - Zone 1	-	1	UCL		UCL4S	22.27	139.69	90.96	 			15.20		 	-	
		Wire Copper Loop/Short - including manual service inquiry		2	UCL		UCL4S	18.95	139.69	90.96				15.20		1		
		d facility reservation - Zone 2 Wire Copper Loop/Short - including manual service inquiry	 		UCL		UCL45	18.95	139.69	90.96	-			15.20		-		
		d facility reservation - Zone 3		3	UCL		UCL4S	10.99	139.69	90.96				15.20		1		
\vdash		der Coordination for Unbundled Copper Loops (per loop)		J	UCL		UCLMC	10.99	7.92	7.92	 			13.20		 	1	
+-+		Wire Copper Loop/Short - without manual service inquiry and	 		UUL		OCLIVIC		1.92	1.92	 					 	 	
		cility reservation - Zone 1		1	UCL		UCL4W	22.27	115.43	78.63				15.20		1		
		Wire Copper Loop/Short - without manual service inquiry and		Ė						. 0.00	t	1		.0.20		1	1	
1 1		cility reservation - Zone 2	1	2	UCL		UCL4W	18.95	115.43	78.63	I			15.20		I	Ì	

UNBUNDI	ED NETWORK ELEMENTS - Louisiana												Attach	ment: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Copper Loop/Short - without manual service inquiry and				1101 414	40.00	445.40	70.00				45.00				'
-	facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL UCL	UCL4W UCLMC	10.99	115.43 7.92	78.63 7.92				15.20				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.			UCL	UCLIVIC		7.92	7.92								
	inquiry and facility reservation - Zone 1		1	UCL	UCL4L	26.17	139.69	90.96				15.20				'
	4-Wire Unbundled Copper Loop/Long - includes manual svc.		-	002	002.2	20.11	100.00	00.00				10.20				
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	28.47	139.69	90.96				15.20				'
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	62.93	139.69	90.96				15.20				<u> </u>
	Order Coordination for Unbundled Copper Loops (per loop) 4-Wire Unbundled Copper Loop/Long - without manual svc.	1		UCL	UCLMC		7.92	7.92								<u> </u>
	inquiry and facility reservation - Zone 1		1	UCL	UCL4O	26.17	115.43	78.63				15.20				1 '
	4-Wire Unbundled Copper Loop/Long - without manual svc.		<u> </u>			20.17	110.40	70.00				10.20				
	inquiry and facility reservation - Zone 2		2	UCL	UCL4O	28.47	115.43	78.63				15.20				1 '
	4-Wire Unbundled Copper Loop/Long - without manual svc.															1
	inquiry and facility reservation - Zone 3	<u> </u>	3	UCL	UCL4O	62.93	115.43	78.63				15.20				 '
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		7.92	7.92								 '
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)			UCL	UREWO		91.92	42.47				15.20				'
LOOP MODII				UCL	UKEWU		91.92	42.47				13.20				—
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft			UAL, UHL, UCL, UEQ, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULM2L		0.00	0.00				15.20				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire															
	greater than 18k ft			UCL, ULS, UEQ	ULM2G		0.00	0.00				15.20				<u> </u>
	Unbundled Loop Modification Removal of Load Coils - 4 Wire				ULM4L		0.00	0.00				15.20				'
	less than or equal to 18K ft Unbundled Loop Modification Removal of Load Coils - 4 Wire			UHL, UCL	ULIVI4L		0.00	0.00				15.20				
	pair greater than 18k ft			UCL	ULM4G		0.00	0.00				15.20				'
SUB-LOOPS	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop			UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULMBT		12.15	12.15				15.20				
	oop Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set- Up	ı		UEANL	USBSA		144.09	144.09				15.20				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	ı		UEANL	USBSB		10.99	10.99				15.20				
	Sub-Loop - Per Building Equipment Room - CLEC Feeder	Ì														
	Facility Set-Up	I		UEANL	USBSC		86.16	86.16				15.20				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up	1		UEANL	USBSD		27.13	27.13				15.20				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1	I	1	UEANL	USBN2	7.57	63.89	30.06				15.20				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2	ı	2	UEANL	USBN2	12.75	63.89	30.06				15.20				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	ı	3	UEANL	USBN2	21.45	63.89	30.06				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
	Zone 1		1	UEANL	USBN4	11.76	76.75	42.92				15.20				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN4	16.84	76.75	42.92				15.20				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN4	19.27	76.75	42.92				15.20				

UNBU	INDLE	D NETWORK ELEMENTS - Louisiana													ment: 2		bit: C
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						-	1	Nonrec	urring	Nonrecurring D	Disconnect			088	Rates(\$)	L	
				-			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
								FIISt	Add I	FIRST	Addi	SOMEC	SUMAN	SUMAN	SOWAN	SUMAN	SUMAN
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
							0.04						45.00				
		Sub-Loop 2-Wire Intrabuilding Network Cable (INC)		1	UEANL	USBR2	2.91	51.48	17.65				15.20				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
		Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	ı		UEANL	USBR4	6.58	57.54	23.71				15.20				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1	UEF	UCS2X	6.26	63.89	30.06				15.20				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		2	UEF	UCS2X	10.07	63.89	30.06				15.20				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		3	UEF	UCS2X	12.70	63.89	30.06				15.20				
						1											
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair	l	1	UEF	USBMC		7.92	7.92							I	1
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1	UEF	UCS4X	8.03	76.75	42.92	 			15.20			†	I
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	H	2	UEF	UCS4X	10.71	76.75	42.92	 			15.20			t	1
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		3	UEF	UCS4X	6.08	76.75	42.92				15.20		1	t	1
		T TYTHE COPPER CHIDUNGIEG COD-LOOP DISTIBUTION - 2018 3	- ' -	3	OLI	JUJ4A	0.08	10.15	42.92				15.20		1	 	
		Order Coordination for Habundle 1 Out Lane and the said	l	1	UEF	USBMC		7.92	7.92							I	1
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBINC		7.92	7.92								
		dled Sub-Loop Modification															
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load															
		Coil/Equip Removal per 2-W PR			UEF	ULM2X		0.00	0.00				15.20				
		Unbundled Sub-loop Modification - 4-W Copper Dist Load															
		Coil/Equip Removal per 4-W PR			UEF	ULM4X		0.00	0.00				15.20				
		Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged															
		Tap Removal, per PR unloaded			UEF	ULM4T		224.55	4.29				15.20				
		dled Network Terminating Wire (UNTW)															
		Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.3454	14.72	14.72				15.20				
		k Interface Device (NID)															
		Network Interface Device (NID) - 1-2 lines		+	UENTW	UND12		42.26	27.83				15.20				
		Network Interface Device (NID) - 1-6 lines		+	UENTW	UND16		62.86	48.43	<u> </u>			15.20				
		Network Interface Device (NID) - 1-0 lines Network Interface Device Cross Connect - 2 W		+	UENTW	UNDC2		5.73	5.73	+			15.20				-
		Network Interface Device Cross Connect - 2 W		-	UENTW	UNDC4		5.73	5.73				15.20				
CLID I	2000	Network Interface Device Cross Connect - 4vv		1	UENTW	UNDC4		5.73	5.73				15.20				
SUB-LO		Facility															
		op Feeder		1													
		USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
		Distribution Facility set-up	 	1	UDN,UCL,UDL,UDC	USBFW		144.09					15.20				
		USL Feeder - DS0 Set-up per Cross Box location - per 25 pair	l	1	UEA,											I	1
		set-up			UDN,UCL,UDL,UDC			10.99	10.99				15.20				
		USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		568.98	11.30				15.20				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice												-			
		Grade - Zone 1	l	1	UEA	USBFA	8.71	89.81	54.35				15.20				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice															
		Grade - Zone 2	l	2	UEA	USBFA	13.64	89.81	54.35				15.20				
		Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,		1		1				i i							
		Voice Grade - Zone 3	l	3	UEA	USBFA	30.21	89.81	54.35				15.20				
		Order Coordination for Specified Conversion Time, per LSR	1		UEA	OCOSL	00.21	17.56	04.00				10.20				
		Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice	-	+	52/1	3000L		17.50		 						t	1
		Grade - Zone 1	l	1	UEA	USBFB	8.71	89.81	54.35				15.20			I	1
	-	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice	 	+ '	OLA	OODI D	0.71	09.01	34.35	 			15.20			 	
			l	_	1154	HODES	40.01	20.01	-10-				45.00			I	1
		Grade - Zone 2		2	UEA	USBFB	13.64	89.81	54.35				15.20		1	1	-
		Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice	l	_									,			I	1
		Grade - Zone 3		3	UEA	USBFB	30.21	89.81	54.35				15.20				
		Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		17.56									
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,														1	
		Voice Grade - Zone 1		1	UEA	USBFC	8.71	89.81	54.35				15.20				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
ı		Voice Grade - Zone 2	l	2	UEA	USBFC	13.64	89.81	54.35				15.20			1	
		Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse				1									ĺ		
		Battery, Voice Grade - Zone 3	I	3	UEA	USBFC	30.21	89.81	54.35	i l		l	15.20		1	1	1

UNBUNDU	D NETWORK ELEMENTS - Louisiana												Δttach	ment: 2	Fyhil	oit: C
SINDONDE	LO INC. INCINCLE LINE IN 10 - Louisiana										Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		١									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						_ 1	Nonrec	urring	Nonrecurring Di	isconnect			oss	Rates(\$)	ı	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		17.56									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 1		1	UEA	USBFD	21.44	103.69	67.31				15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 2		2	UEA	USBFD	24.66	103.69	67.31				15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice															
	Grade - Zone 3		3	UEA	USBFD	42.84	103.69	67.31				15.20				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		17.56									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
	Grade - Zone 1		1	UEA	USBFE	21.44	103.69	67.31				15.20		1		
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		1]			1		I		
	Grade - Zone 2		2	UEA	USBFE	24.66	103.69	67.31				15.20		1		
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice													1		
	Grade - Zone 3		3	UEA	USBFE	42.84	103.69	67.31				15.20				
\vdash	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		17.56							ļ		
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	15.44	102.58	66.20				15.20				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN	USBFF	23.32	102.58	66.20				15.20				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	44.57	102.58	66.20				15.20				
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		17.56									
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	15.44	102.58	66.20				15.20				
-	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		2	UDC	USBFS	23.32	102.58	66.20				15.20				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	44.57	102.58	66.20				15.20				
-	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	USL	USBFG	55.38	98.15	61.77				15.20				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2	USL	USBFG	167.83	98.15	61.77				15.20				
-	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	USL USL	USBFG	469.87	98.15	61.77	-			15.20				
-	Order Coordination For Specified Conversion Time, Per LSR		1	UCL	OCOSL USBFH	6.96	17.56 81.36	44.98				15.20				
-	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1 Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		-	UCL	USBER	0.90	01.30	44.90				15.20				
	onbundied Sub-Loop Feeder Loop, 2-wife Copper Loop - Zone		2	UCL	USBFH	4.97	81.36	44.98				15.20				
-	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone			UCL	USBITT	4.37	01.30	44.30	+			13.20				
	2		3	UCL	USBFH	3.99	81.36	44.98				15.20				
	Order Coordination For Specified Conversion Time, per LSR		3	UCL	OCOSL	3.33	17.56	44.30				13.20				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	15.68	98.07	61.69				15.20				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2	UCL	USBFJ	9.68	98.07	61.69				15.20				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	6.39	98.07	61.69				15.20				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL	0.00	17.56									
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	22.61	98.15	61.77				15.20		1	İ	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	22.87	98.15	61.77				15.20		1	İ	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	24.25	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 1		1	UDL	USBFO	22.61	98.15	61.77				15.20		<u></u>	<u></u>	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -						_									
	Zone 2		2	UDL	USBFO	22.87	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 3		3	UDL	USBFO	24.25	98.15	61.77				15.20		1		
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		17.56							ļ		
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -													1		
\vdash	Zone 1		1	UDL	USBFP	22.61	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		_	LIDI	LIODES							4-0-		1		
\vdash	Zone 2		2	UDL	USBFP	22.87	98.15	61.77	 			15.20		-	1	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		_	LIDI	HODES	04.0-	00.1-	04 ==]			45.00		I		
\vdash	Zone 3		3	UDL	USBFP	24.25	98.15	61.77	 			15.20		!	1	
SUB-LOOPS	Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		17.56		 					 		
	oon Fooder	-			+				 			 				
Sub-L	oop Feeder Sub Loop Feeder - DS3 - Per Mile Per Month			UE3	1L5SL	17.00			+			-		+	1	
 	Sub Loop Feeder - DS3 - Fer Mile Fer Month Sub Loop Feeder - DS3 - Facility Termination Per Month	i	 	UE3	USBF1	368.44	3,397.56	406.56	 			15.20		 		
 	Sub Loop Feeder – STS-1 – Per Mile Per Month	i		UDLSX	1L5SL	17.00	5,551.50	400.30	 			13.20		t	1	
 	Sub Loop Feeder - STS-1 - Facility Termination Per Month	i		UDLSX	USBF7	395.92	3,397.56	406.56	 			15.20		t		
	1000 200p 1 00001 OTO 1 1 dollary formination 1 of MOHILI	<u>'</u>	<u> </u>	COLON	00017	J3J.3Z	0,001.00	700.00	<u> </u>			10.20		1	l	

3.123.12EI	ED NETWORK ELEMENTS - Louisiana												Attach	ment: 2	Fyhil	oit: C
1											Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
											Elec	Manually		Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						.,,			per Loix	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															DISC 1St	DISC Add I
						Rec	Nonred		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub Loop Feeder – OC-3 – Per Mile Per Month	- 1		UDLO3	1L5SL	12.90										
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per															
	Month			UDLO3	USBF5	60.45	0.007.50	400.50				45.00				
	Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3	USBF2	594.77	3,397.56	406.56				15.20				
	Sub Loop Feeder - OC-12 - Per Mile Per Month Sub Loop Feeder - OC-12 - Facility Termination Protection Per	ı		UDL12	1L5SL	15.87										
	Month			UDL12	USBF6	683.03										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	Li		UDL12	USBF3	1,922.00	3,397.56	406.56				15.20				
	Sub Loop Feeder - OC-48 - Per Mile Per Month	l i		UDL48	1L5SL	52.07	0,007.00	400.00				10.20				
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per															
	Month	1		UDL48	USBF9	341.64							1	I	1	
	Sub Loop Feeder - OC-48 - Facility Termination Per Month	i		UDL48	USBF4	1,663.00	3,582.56	406.56				15.20				
	Sub Loop Feeder - OC-12 Interface On OC-48	I		UDL48	USBF8	385.45	803.80	406.56				15.20				
UNBUNDLED	LOOP CONCENTRATION															
	Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	374.26	316.00	316.00				15.20				
	Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	53.40	131.67	131.67				15.20				
	Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	412.08	316.00	316.00				15.20				
	Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	89.98	131.67	131.67				15.20				
	Unbundled Loop Concentration - DS1 Loop Interface Card	ļ		ULC	UCTCO	5.12	61.46	44.74				15.20				
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Card)			UDN	ULCC1	8.12	10.23	10.18				15.20				
	Unbundled Loop Concentration - UDC Loop Interface (Brite Card)			UDC	ULCCU	8.12	10.23	10.18				15.20				
	Unbundled Loop Concentration2 Wire Voice-Loop Start or															
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	2.03	10.23	10.18				15.20				
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery															
	Loop Interface (SPOTS Card)			UEA	ULCCR	12.07	10.23	10.18				15.20				
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface					= 00						4= 00				
	(Specials Card)			UEA	ULCC4	7.20	10.23	10.18				15.20				
 	Unbundled Loop Concentration - TEST CIRCUIT Card Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop			ULC	UCTTC	35.19	10.23	10.18				15.20				
	Interface			UDL	ULCC7	10.67	10.23	10.18				15.20				
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop			ODL	OLOGI	10.07	10.25	10.10				13.20				
	Interface			UDL	ULCC5	10.67	10.23	10.18				15.20				
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop	1											1	1	1	
	Interface			UDL	ULCC6	10.67	10.23	10.18				15.20		1		
UNE OTHER,	PROVISIONING ONLY - NO RATE	1										1	1		1	
	NID - Dispatch and Service Order for NID installation			UENTW	UNDBX	0.00	0.00									
	UNTW Circuit Id Establishment, Provisioning Only - No Rate			UENTW	UENCE	0.00	0.00									
				UEANL,UEF,UEQ,U									1	_]	
<u> </u>	Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN	0.00	0.00									
UNE OTHER,	PROVISIONING ONLY - NO RATE	1	<u> </u>										ļ		ļ	
				UAL,UCL,UDC,UDL,									1	I	1	
	Unbundled Centest Name Provisioning Only, so and				LINEON	0.00	0.00						1	I	1	
\vdash	Unbundled Contact Name, Provisioning Only - no rate Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no	1	 	UDN,UEA,UHL,ULC	UNECN	0.00	0.00						-		-	
	onbunuled Sub-Loop Feeder-Z Wire Cross Box Jumper - No			UEA,UDN,UCL,UDC	USBFQ	0.00	0.00						1	I	1	
 	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no			OLA, ODIN, OOL, ODC	UUDI Q	0.00	0.00						1	 		
	rate			UEA,USL,UCL,UDL	USBFR	0.00	0.00						1		1	
	Unbundled DS1 Loop - Superframe Format Option - no rate	1		USL	CCOSF	0.00	0.00						1	1	1	
	Unbundled DS1 Loop - Expanded Superframe Format option -	1				2.00	2.00									
	no rate			USL	CCOEF	0.00	0.00						1		1	
HIGH CAPAC	ITY UNBUNDLED LOCAL LOOP															
	High Capacity Unbundled Local Loop - DS3 - Per Mile per															
\vdash	month	1		UE3	1L5ND	10.04							ļ	1	ļ	
	High Capacity Unbundled Local Loop - DS3 - Facility			LIEO	LIEODY	000 0 1	400.40	050.00				45.00				
1 1	Termination per month	1	1	UE3	UE3PX	362.34	438.46	256.30				15.20				
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per					ı										

UNBUN	IDLE	D NETWORK ELEMENTS - Louisiana													ment: 2		bit: C
CATEGO	PRY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)	•	•
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		High Capacity Unbundled Local Loop - STS-1 - Facility															
		Termination per month			UDLSX	UDLS1	374.56	438.46	256.30				15.20				
LOOP MA																	
		Loop Makeup - Preordering Without Reservation, per working or															
		spare facility queried (Manual).		1	UMK	UMKLW		23.29	23.29							-	
		Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).			UMK	UMKLP		24.70	24.70								
-		Loop MakeupWith or Without Reservation, per working or		1	OWIN	OWINE		24.70	24.70								
		spare facility queried (Mechanized)			UMK	PSUMK		0.19	0.19								
UNBUND		DEDICATED TRANSPORT						00	00								
		INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimum	m billin	g perio	d - below DS3=one	month, DS3/	STS-1=four mo	nths									
		OFFICE CHANNEL - DEDICATED TRANSPORT				,				İ					1		
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
		Per Mile per month			U1TVX	1L5XX	0.013										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -		1	l	L				1		1				_	
		Facility Termination			U1TVX	U1TV2	22.60	39.36	26.62				15.20				
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade															
		Rev Bat Per Mile per month			U1TVX	1L5XX	0.013										
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat			11477.07	LIATEDO	00.00	00.00	00.00				45.00				
-		Facility Termination		1	U1TVX	U1TR2	22.60	39.36	26.62				15.20			-	
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month			U1TVX	1L5XX	0.013										
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade		<u> </u>	UTIVX	ILDAX	0.013			-							
		- Facility Termination			U1TVX	U1TV4	19.81	39.36	26.62				15.20				
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile			OTTVX	01114	13.01	39.30	20.02				13.20				
		per month			U1TDX	1L5XX	0.013										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility			01127	120701	0.0.0									1	
		Termination			U1TDX	U1TD5	15.61	39.37	26.62				15.20				
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
		per month			U1TDX	1L5XX	0.013										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
		Termination			U1TDX	U1TD6	15.61	39.37	26.62				15.20				
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
		month			U1TD1	1L5XX	0.2652										
		Interoffice Channel - Dedicated Tranport - DS1 - Facility				=							4= 00				
-		Termination		1	U1TD1	U1TF1	70.47	86.69	79.44				15.20			-	
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			U1TD3	1L5XX	6.04										
\vdash		Interoffice Channel - Dedicated Transport - DS3 - Facility			0.100	LUAA	0.04			1					1	t	
		Termination per month			U1TD3	U1TF3	850.45	270.69	158.05				15.20			1	
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per				30	300.40	2,0.00	100.00	1			10.20		1	1	
		month		1	U1TS1	1L5XX	6.04					1					
		Interoffice Channel - Dedicated Transport - STS-1 - Facility				1				1					İ	1	
		Termination		L	U1TS1	U1TFS	830.19	270.69	158.05			<u> </u>	15.20			<u> </u>	<u></u>
		CHANNEL - DEDICATED TRANSPORT															
N	IOTE:	LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	g perio	d - belo													
		Local Channel - Dedicated - 2-Wire Voice Grade			ULDVX	ULDV2	18.32	187.51	32.21				15.20				
$\vdash \vdash$		Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat		<u> </u>	ULDVX	ULDR2	18.32	187.51	32.21	 			15.20				
		Local Channel - Dedicated - 4-Wire Voice Grade		<u> </u>	UNDVX	ULDV4 ULDF1	19.41	187.94	32.63				15.20			1	
\vdash		Local Channel - Dedicated - DS1 - Zone 1		1	ULDD1		39.18	172.34	149.27	1			15.20		 	 	
\vdash		Local Channel - Dedicated - DS1 - Zone 2 Local Channel - Dedicated - DS1 - Zone 3		3	ULDD1 ULDD1	ULDF1 ULDF1	121.58 70.02	172.34 172.34	149.27 149.27	1			15.20 15.20			-	-
\vdash		Local Channel - Dedicated - DS1 - Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month		3	ULDD1	1L5NC	70.02	172.34	149.27	1			15.20			-	-
\vdash		Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination		1	ULDD3	ULDF3	469.44	438.46	256.30	1			15.20			+	
\vdash		Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1- Per Mile per month		 	ULDS1	1L5NC	7.82	+30.40	230.30	 			13.20		1	t	
\vdash		Local Channel - Dedicated - STS-1 - Fer Wille per Month Local Channel - Dedicated - STS-1 - Facility Termination			ULDS1	ULDFS	457.22	438.46	256.30	† †		 	15.20		 	I	
DARK FIE	BER					1322.0	707.22	100.40	200.00	1			10.20		1	1	
T		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction				1				1					1	1	
		Thereof per month - Local Channel		1	UDF	1L5DC	52.23									1	

ONRONDLE	ED NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			1	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic Disc Add'l
							Nonrec	urring	Nonrecurrin	a Disconnect			oss	Rates(\$)		1
+						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
+	NRC Dark Fiber - Local Channel			UDF	UDFC4		620.60	133.88	11130	Addi	COMILO	15.20	COMPAR	COMPAR	COMPAR	COMPAR
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction			05.	02.0.		020.00	100.00				10.20				
	Thereof per month - Interoffice Channel			UDF	1L5DF	25.28										
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		620.60	133.88				15.20				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Loop			UDF	1L5DL	52.23										
	NRC Dark Fiber - Local Loop			UDF	UDFL4		620.60	133.88				15.20				
XX ACCESS	TEN DIGIT SCREENING															
	8XX Access Ten Digit Screening, Per Call			OHD		0.0006387										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX															
	Number Reserved			OHD	N8R1X		2.51	0.43				15.20				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O															
	POTS Translations			OHD			5.77	0.78		1		15.20				
1	8XX Access Ten Digit Screening, Per 8XX No. Established With		1											1	1	
	POTS Translations		<u> </u>	OHD	N8FTX		5.77	0.78		ļ	ļ	15.20				
1	8XX Access Ten Digit Screening, Customized Area of Service		1	OUD	NOTO							4-0-		1	1	
	Per 8XX Number			OHD	N8FCX		2.51	1.26				15.20				
	8XX Access Ten Digit Screening, Multiple InterLATA CXR			OUD	NOTAN		0.00	4.00				45.00				
-	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		2.93	1.68				15.20				
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		2.93	0.43				15.20				
	8XX Access Ten Digit Screening, Call Handling and Destination			OHD	N8FDX		0.54					15.20				
	Features			ОНО	NOFDX		2.51					15.20				
	8XX Access Ten Digit Screening, w/ 8XX No. Delivery, per query			OHD		0.0006387										
+	8XX Access Ten Digit Screening, w/ box No. Delivery, per query		-	OLID		0.0000367										
	query			OHD		0.0006387										
INF INFORM	ATION DATA BASE ACCESS (LIDB)			OTID		0.0000307										
	LIDB Common Transport Per Query			OQT		0.0000221										
	LIDB Validation Per Query			OQU		0.0135077										
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		33.33					15.20				
SIGNALING (,												
ì	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	147.60										
	CCS7 Signaling Usage, Per TCAP Message			UDB		0.000064										
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	15.77	34.50	34.50				15.20				
	CCS7 Signaling Connection, Per link (B link) (also known as D															
	link)			UDB	TPP++	15.77	34.50	34.50				15.20				
	CCS7 Signaling Usage, Per ISUP Message			UDB		0.000016										
	CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	732.10										
1	CCS7 Signaling Point Code, per Originating Point Code		l		1											
	Establishment or Change, per STP affected			UDB	CCAPO		28.17	28.17	ļ	ļ	ļ	15.20		ļ	ļ	1
	CCS7 Signaling Point Code, per Destination Point Code		1											1	1	
1	Establishment or Change, Per Stp Affected		<u> </u>	UDB	CCAPD		28.17	28.17		ļ	ļ	15.20				
911 SERVICI			<u> </u>			10.0-	405 5 :			ļ	ļ					
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 1				+	18.32	187.51	32.21		1		15.20				1
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 2		<u> </u>		+	18.32	187.51	32.21	-	1	ļ	15.20		-	-	1
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 3		 		+	18.32 0.013	187.51	32.21	 	 	 	15.20		 	 	1
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility		 		+	0.013			 	 	 			 	 	1
1	Termination		1		1	22.60	39.36	26.62				15.20		1	1	
-	Local Channel - Dedicated - DS1 - Zone 1		-		+	39.18	172.34	149.27		1		15.20				
	Local Channel - Dedicated - DS1 - Zone 1 Local Channel - Dedicated - DS1 - Zone 2				+	121.58	172.34	149.27		1	1	15.20				1
	Local Channel - Dedicated - DS1 - Zone 2 Local Channel - Dedicated - DS1 - Zone 3				+	70.02	172.34	149.27		1	1	15.20		1	1	1
-	Interoffice Transport - Dedicated - DS1 - Zone 3				+	0.2652	172.34	143.27		1	1	13.20				1
	interented transport - Dedicated - DOT Fel IVIIIe				+	0.2002			1	1	 			1	1	
	Interoffice Transport - Dedicated - DS1 Per Facility Termination		1		1	70.47	86.69	79.44				15.20		1	1	
CALLING NAM	ME (CNAM) SERVICE				1	70.47	55.03	70.44				10.20				
	CNAM For DB Owners - Service Establishment			OQV	1		22.29			1		15.20		1	1	
	CNAM For Non DB Owners - Service Establishment		1	OQV	+		22.29			†	1	15.20				1

UNBUNDLE	D NETWORK ELEMENTS - Louisiana												ment: 2		oit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)		Submitted Elec per LSR	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec	urring	Nonrecurring Disconne				Rates(\$)		
						Nec	First	Add'l	First Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CNAM For DB Owners - Service Provisioning With Point Code														
	Establishment			OQV			962.22	711.64			15.20				
	CNAM For Non DB Owners - Service Provisioning With Point			001/			000 40	200.05			45.00				
	Code Establishment CNAM for DB Owners, Per Query			OQV OQV	1	0.0010217	332.43	238.05			15.20				
-	CNAM for Non DB Owners, Per Query CNAM for Non DB Owners, Per Query			OQV	+	0.0010217									
LNP Query Ser				OQV	+	0.0010217									
	LNP Charge Per query			OQV	+	0.0008559									
	LNP Service Establishment Manual			٠,		0.000000	12.16				15.20				
	LNP Service Provisioning with Point Code Establishment						576.33	294.43			15.20				
	ALL PROCESSING				1							<u> </u>		<u> </u>	
	Oper. Call Processing - Oper. Provided, Per Min Using BST														
	LIDB					1.20									
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24									
	Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20									
	Oper. Call Processing - Fully Automated, per Call - Using														
	Foreign LIDB					0.20									
INWARD OPER	ATOR SERVICES														
	Inward Operator Services - Verification, Per Minute					1.15									
	Inward Operator Services - Verification and Emergency Interrupt														
	- Per Minute					1.15									
	PERATOR CALL PROCESSING														
Facility	based CLEC				00100		=				1= 00				
-	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00			15.20				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN				CBAOL		500.00	500.00			15.20				
UNEP (CBAOL		300.00	300.00			13.20				
ONE! (Recording of Custom Branded OA Announcement				+		7,000.00	7,000.00			15.20				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN						500.00	500.00			15.20				
DIRECTORY AS	SSISTANCE SERVICES				+		000.00	000.00			10.20				
	TORY ASSISTANCE ACCESS SERVICE														
	Directory Assistance Access Service Calls, Charge Per Call					0.275									
DIRECT	TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (D	ACC)													
	Directory Assistance Call Completion Access Service (DACC),														
	Per Call Attempt					0.10									
	SSISTANCE SERVICES				1									ļ	
	TORY ASSISTANCE DATA BASE SERVICE (DADS)		ļ		+										
	Directory Assistance Data Base Service Charge Per Listing				DRECE	0.04				-	1			 	
	Directory Assistance Data Base Service, per month IRECTORY ASSISTANCE		-		DBSOF	150.00					1	-	-	 	-
	Based CLEC		-		+					+	1	1	1	1	
lacility	Recording and Provisioning of DA Custom Branded				+					+	+			 	
	Announcement			AMT	CBADA		6,000.00	6,000.00			15.20				
	Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00			15.20			Ì	
UNEP (CLEC							•							
	Recording of DA Custom Branded Announcement						3,000.00	3,000.00			15.20				
	Loading of DA Custom Branded Announcement per Switch per OCN			_			1,170.00	1,170.00			15.20				
SELECTIVE RO							.,	.,							
	Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		82.25	82.25			15.20				
VIRTUAL COLI		 			301.01.		02.23	02.23		+	10.20			 	
OAL GOLL	Virtual Collocation - Application Cost			AMTFS	EAF		1,770.40			-	15.20		1	 	1
	Virtual Collocation - Cable Installation Cost, per cable			AMTFS	ESPCX		841.54				15.20			1	
	Virtual Collocation - Floor Space, per sq. ft.			AMTFS	ESPVX	3.20									
	Virtual Collocation - Power, per fused amp			AMTFS	ESPAX	8.32					1				

ONBONDER	D NETWORK ELEMENTS - Louisiana												Attach	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge -	Increment Charge - Manual Sv Order vs. Electronic Disc Add
		ļ				Rec	Nonred			g Disconnect	COMEC	COMAN		Rates(\$)	COMAN	COMAN
	Virtual Collocation - Cable Support Structure, per entrance						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	cable			AMTFS	ESPSX	16.02										
	cable			UEANL,UEA,UDN,U	LOI OX	10.02					-					
				DC,UAL,UHL,UCL,U												
				EQ, AMTFS, UDL,												
				UNCVX, UNCDX,												
	Virtual Collocation - 2-wire Cross Connects (loop)			UNCNX	UEAC2	0.0296	11.94	11.46				15.20				
1				UEA,UHL,UCL,UDL,												
				AMTFS, UAL, UDN,												
-	Virtual Collocation - 4-wire Cross Connects (loop)			UNCVX, UNCDX	UEAC4	0.0591	12.04	11.53			-	15.20				
				AMTFS,UDL12, UDLO3, U1T48,												
				U1T12, U1T03,												
				ULDO3, ULD12,												
	Virtual Collocation - 2-Fiber Cross Connects			ULD48, UDF	CNC2F	2.65	20.29	14.76				15.20				
	Virtual Controlation 2 1 ibon Cross Comments			AMTFS,UDL12,	0.102.	2.00	20.20	0				10.20				
				UDLO3, U1T48,												
				U1T12, U1T03,												
				ULDO3, ULD12,												
	Virtual Collocation - 4-Fiber Cross Connects			ULD48, UDF	CNC4F	5.31	24.81	19.29				15.20				
				USL,ULC,AMTFS,												
				ULR, UXTD1,												
				UNC1X, ULDD1,												
	Virtual collocation - Special Access & UNE, cross-connect per			U1TD1, USLEL,												
	DS1			UNLD1	CNC1X	1.04	21.39	15.47				15.20				
				USL,ULC,AMTFS,U E3, U1TD3, UXTS1,												
				UXTD3, UNC3X,												
				UNCSX, ULDD3,												
	Virtual collocation - Special Access & UNE, cross-connect per			U1TS1, ULDS1,												
	DS3			UDLSX, UNLD3	CND3X	13.21	20.28	14.76				15.20				
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable															
	Support Structure, per linear foot			AMTFS	VE1CB	0.0024										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax															
	Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0036										
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable															
	Support Structure,per cable			AMTFS	VE1CC		534.79					15.20				
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax															
-	Cable Support Structure, per cable			AMTES	VE1CE	40.07	534.79				-	15.20				
	Virtual Collocation Cable Records - per request Virtual Collocation Cable Records - VG/DS0 Cable, per cable		<u> </u>	AMTFS	VE1BA	10.97				-	+					
	record			AMTFS	VE1BB	5.29										
	Virtual Collocation Cable Records - VG/DS0 Cable, per each			AWITTO	VETOD	5.25					-					
	100 pair			AMTFS	VE1BC	0.08										
	Virtual Collocation Cable Records - DS1, per T1TIE			AMTFS	VE1BD	0.04				1			İ			
	Virtual Collocation Cable Records - DS3, per T3TIE	l		AMTFS	VE1BE	0.13										
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber															
	records			AMTFS	VE1BF	1.37										
	Virtual collocation - Security Escort - Basic, per half hour			AMTFS	SPTBX		16.44	10.42				15.20				
	Virtual collocation - Security Escort - Overtime, per half hour	ļ		AMTFS	SPTOX		21.41	13.45		1		15.20		ļ	ļ	
<u> </u>	Virtual collocation - Security Escort - Premium, per half hour	ļ		AMTFS	SPTPX		26.38	16.49		 		15.20		1	1	
	Virtual collocation - Maintenance in CO - Basic, per half hour	ļ		AMTFS	CTRLX		27.12	10.42	ļ	ļ	1	15.20				
	Marcal college Marcal control of the		1	ALTEO	ортом		05.40	40.1-		I		45.00		1	1	
 	Virtual collocation - Maintenance in CO - Overtime, per half hour	-	-	AMTFS	SPTOM		35.42	13.45		+	1	15.20		 	 	
	Virtual collocation - Maintenance in CO - Premium per half hour		1	AMTFS	SPTPM		43.72	16.49		I		15.20		I	I	
	LOCATION		-	AIVIIFO	OF I FIVI	1	43.72	10.49	1	1	 	15.20		-	-	1

CATEGORY RATE ELEMENTS Intert	UNBUNDLE	D NETWORK ELEMENTS - Louisiana		•	1	,					Ι	Ι -		ment: 2		bit: C
Main Collocation - 2-wine Cross Connect, Exchange Port 2- UPPSR VETR2 0.0296 11.04 11.46 11.55 15.55	CATEGORY	RATE ELEMENTS	Zone	BCS	usoc			RATES(\$)			Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
Wintural Collocation - Zwine Cross Connect, Exchange Port 2- Wintural Collocation - Zwine Cross Connects (Loop) for Line Spiting - Zwine						Rec								Rates(\$)		
Wire Analog - Res						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Virtual Collocation 2-Virter Cross Connect, Exchange Port 2-Wire UEPSP VETR2 0.0296 11.94 11.46 15.5				HEDOD	\/E4D0	0.0000	44.04	44.40				45.00				
Wire Line State PEX Trunk - Bus Verifical Collection 2-Wire Cross Connect, Exchange Port 2-Wire Verifical Content PAX Trunk - Res.				UEPSR	VE1R2	0.0296	11.94	11.46				15.20				
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Vision Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Vision Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 4-Wire Cross Connect, Exchange Port 2-Wire Sp. No. Collocation 4-Wire Cross Connect, Exchange Port 4-Wire Sp. No. Collocation 4-Wire Cross Connect, Exchange Port 4-Wire Sp. No. Collocation 4-Wire Cross Connect, Exchange Port 4-Wire Sp. No. Collocation 4-Wire Cross Connects (Loop) for Line Sp. No. C				LIEPSP	VF1R2	0.0296	11 94	11 46				15.20				
Wirtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire UEPSB VE1R2 0.0296 11.94 11.46 15.5					1										İ	
Analog Bus VERZ				UEPSE	VE1R2	0.0296	11.94	11.46				15.20				
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire UEPSX VETR2 0.0296 11.94 11.46 11.55 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 15.55 11.94 11.46 11.46 15.55 11.94 11.46 11																
SISN UEFSX VETR2 0.0296 11.94 11.46 15.5				UEPSB	VE1R2	0.0296	11.94	11.46				15.20				
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire UEPTX VETR2 0.0256 11.94 11.46 11.55 15.55 Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire UEPSX VETR4 0.0661 12.04 11.53 15.55 VIRTUAL COLLOCATION UEPSX VETR4 0.0661 12.04 11.53 15.55 VIRTUAL COLLOCATION UEPSX VETR4 0.0661 12.04 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX UEPSX VETLS 0.0226 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 11.94 11.46 0.00 0.00 15.55 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 VIRTUAL COLLOCATION UEPSX, UEPSX VETLS 0.0236 VIRTUAL COLLOCATION UEPSX VETLS V				LIEBOY	\/E4D0	0.0000	44.04	44.40				45.00				
SISON UPFTX VETR2 0.0296 11.94 11.46 15.5				UEPSX	VE1R2	0.0296	11.94	11.46				15.20				ļ
Virtual Colocation 4-Wire Cross Connects, Exchange Port 4-Wire UEPEX VE1R4 0.691 12.04 11.53 15.			1	UEPTX	VE1R2	0.0296	11.94	11.46				15.20				
ISDN DS1			<u> </u>			3.3200		40							1	
Virtual Collocation-2 Wire Cross Connects (Loop) for Line Spitting UEPSR, UEPSB VE1LS 0.0296 11.94 11.46 0.00 0.00 15.		ISDN DS1		UEPEX	VE1R4	0.0591	12.04	11.53				15.20		<u> </u>		
Spitting	VIRTUAL COL							-								
PHYSICAL COLLOCATION						0.00						4=			1	
Physical Collocation-2 Wire Cross Connects (Loop) for Line UEPSR, UEPSB PE1LS 0.0318 11.94 11.46 15.5	DUVEICAL CO			UEPSR, UEPSB	VE1LS	0.0296	11.94	11.46	0.00	0.00		15.20				
Spitting	PHYSICAL CC															
AIN SELECTIVE CARRIER ROUTING				HEDGD HEDGR	DE1LS	0.0318	11 04	11 /6				15.20				
Regional Service Establishment	AIN SELECTI			OLI OK, OLI OD	I L ILO	0.0510	11.54	11.40				13.20				
Query NRC, per query				UEBIB	SRCEC		100,209.33					15.20				
AIN - BELLSOUTH AIN SMS ACCESS SERVICE AIN SMS ACCESS Service - Service Establishment, Per State, Initial Setup		End Office Establishment			SRCEO		164.29	164.29				15.20				
AIN SMS Access Service - Service Establishment, Per State, nitial Setup				UEBIB		0.0030293										
Initial Setup	AIN - BELLSO															
ANI SMS Access Service - Port Connection - ISDN Access A1N CAMIP 7.60 7.60 15.5				A1N	CAMSE		38.30	38.30				15.20				
ANI SMS Access Service - Port Connection - ISDN Access A1N		AIN SMS Access Service - Port Connection - Dial/Shared Access		Δ1N	CAMDE		7 60	7.60				15.20				
AIN SMS Access Service - User Identification Codes - Per User ID Code												15.20				
AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement																
Initial or Replacement				A1N	CAMAU		33.99	33.99				15.20				
AIN SMS Access Service - Storage, Per Unit (100 Kilobytes) 0.0022 0.5795 0.5																
AIN SMS Access Service - Session, Per Minute AIN SMS Access Service - Company Performed Session, Per Minute AIN SMS Access Service - Company Performed Session, Per Minute AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Triager Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger, Per All Toolkit Service - Trigger, Per All Toolkit Service - Trigger, Per All Toolkit Service - Trigger, Per All Toolkit Service - Type 1 Node Charge, Per All Toolkit	<u> </u>			A1N	CAMRC		41.39	41.39				15.20				
AIN SMS Access Service - Company Performed Session, Per Minute 0.8104 AIN - BELLSOUTH AIN TOOLKIT SERVICE AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup AIN Toolkit Service - Trigining Session, Per Customer BAPVX 4,175.10 4,175.10 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay BAPTD 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTM 7,60 7,60 155. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP 1,00-Digit PODP BAPTM 7,00-Digit PODP 1,00-Digit PODP																ļ
Minute						0.5795									-	-
AIN - BELLSOÜTH AIN TOOLKIT SERVICE AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup AIN Toolkit Service - Triaining Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit						0.8104										
Initial Setup	AIN - BELLSC			İ	1	3.3.3.										<u> </u>
AIN Toolkit Service - Training Session, Per Customer BAPVX 4,175.10 4,175.10 15.3 AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt BAPTT 7.60 7.60 15.3 AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay BAPTD 7.60 7.60 7.60 15.3 AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate BAPTM 7.60 7.60 7.60 15.3 AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTO 33.47 33.47 15.3 AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP BAPTC 33.47 33.47 33.47 15.3 AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code BAPTF 33.47 33.47 33.47 15.3 AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code BAPTF 33.47 33.47 33.47 15.3 AIN Toolkit Service - Query Charge, Per Query AIN Toolkit Service - Trype 1 Node Charge, Per AIN Toolkit Service - Trype 1 Node Ch																
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per BAPTT 7.60 7.60 15			ļ	CAM		ļ						15.20				
DN, Term. Attempt	\vdash		<u> </u>	ļ	BAPVX	ļ	4,175.10	4,175.10	ļ		1	15.20				
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate BAPTM 7.60 7.60 15. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTO 33.47 33.47 15. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP BAPTC 33.47 33.47 15. AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Query AIN Toolkit Service - Type 1 Node Charge, Per AlN Toolkit			1	1	DADTT	[7.60	7.60				15.20				
DN, Off-Hook Delay	\vdash		1	1	DAFII	 	7.00	00.1			}	15.20			-	
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP BAPTO 33.47 33.47 15.: AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Query AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit AIN Toolkit Service - Trigger Access Charge, Per AIN Toolkit			1	1	BAPTD	[7.60	7.60				15.20				
DN, Off-Hook Immediate				İ	1	1		50								<u> </u>
DN, 10-Digit PODP		DN, Off-Hook Immediate	<u> </u>		BAPTM		7.60	7.60		<u></u>		15.20				
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP BAPTC 33.47 33.47 15.: AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Query AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit]												
DN, CDP	igwdown		ļ		BAPTO		33.47	33.47				15.20				
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Query AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit			1	1	BADTC	[22 47	22 47				15.20				
DN, Feature Code	\vdash		1	1	DAFIC	 	33.47	33.47			}	15.20			-	
AIN Toolkit Service - Query Charge, Per Query 0.0536446 0.0536446 AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit			1	1	BAPTF		33,47	33.47				15.20				
AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit				İ		0.0536446						12.20		İ	1	
Subscription, Per Node, Per Query		AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit														
0.00000		Subscription, Per Node, Per Query				0.006569										<u> </u>
AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes 0.06			1	1										1	I	

UNBL	JNDLE	D NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhil	oit: C
												Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted			Charge -	Charge -	Charge -
												Elec		Manual Svc	Manual Svc		Manual Svc
CATE	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						• • •			per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
														-		Disc 1st	Disc Add I
							Rec	Nonred	curring	Nonrecurring	g Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
		Subscription			CAM	BAPMS	10.90	7.60	7.60				15.20				ı
		AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
		Subscription			CAM	BAPLS	2.80	8.41	8.41				15.20				ı
		AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service															i
		Subscription			CAM	BAPDS	8.20	7.60	7.60				15.20				1
		AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit															i
		Service Subscription			CAM	BAPES	0.09	8.41	8.41				15.20				1
ENHA		(TENDED LINK (EELs)															
<u> </u>		New Density Zone 1 EELs are available in the following MSA					Atlanta, Ga; Ne	w Orleans, LA,							ļ		
		Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem								L <u></u>	<u> </u>				L	L	
L		In all states, EEL network elements shown below also apply t												UNEs.(Non-re	curring rates	do not apply.)
		In All States the EEL network elements apply to ordinarily co				tch As Is Ch	arge.) When or	dering ordinar	ily combined i	network eleme	nts, Non-recur	ring rates de	o apply.				
	2-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												
		First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport	1	l .	1110101								4		1		1
	-	Combination - Zone 1	<u> </u>	1	UNCVX	UEAL2	14.93	94.21	45.09				15.20				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		_					4= 00								i
	ļ	Transport Combination - Zone 2		2	UNCVX	UEAL2	25.35	94.21	45.09				15.20				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		_			== 40		4= 00								i
		Transport Combination - Zone 3		3	UNCVX	UEAL2	50.46	94.21	45.09				15.20				
		Interoffice Transport - Dedicated - DS1 combination - Per Mile															1
		per month			UNC1X	1L5XX	0.2652										
		Interoffice Transport - Dedicated - DS1 combination - Facility			LINIOAN	U1TF1	70.47	4 40 50	400.00				45.00				1
		Termination per month			UNC1X	_	70.47	143.58	103.88				15.20				
		DS1 Channelization System Per Month			UNC1X	MQ1	105.09	59.97	12.96				15.20				
		Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.6497	5.91	4.26								
		Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1			UNCVX	UEAL2	14.93	94.21	45.09				15.20				1
		Each Additional 2-Wire VG Loop(SL2) in the same DS1		1	UNCVX	UEAL2	14.93	94.21	45.09				15.20				
		Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	25.35	94.21	45.09				15.20				i
-		Each Additional 2-Wire VG Loop(SL2) in the same DS1			UNCVX	UEALZ	25.35	94.21	45.09				15.20				
		Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	50.46	94.21	45.09				15.20				1
-		Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVA	ULALZ	30.40	34.21	43.09				13.20				
		per month			UNCVX	1D1VG	0.6497	5.91	4.26								i
-		Nonrecurring Currently Combined Network Elements Switch -As-			ONOVA	IDIVO	0.0431	3.31	4.20								
		Is Charge			UNC1X	UNCCC		5.43	5.43				15.20				i
—	4-WIRE	INSTRUCTION OF THE PROPERTY OF	FROFF	ICF TR		311000		5.45	5.45				10.20				
1		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice			()	1							 		 		
1		Transport Combination - Zone 1	1	1	UNCVX	UEAL4	30.81	94.21	45.09				15.20		1		1
	1	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		İ		1									İ		
		Transport Combination - Zone 2		2	UNCVX	UEAL4	38.32	94.21	45.09				15.20				1
		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
1		Transport Combination - Zone 3	1	3	UNCVX	UEAL4	60.39	94.21	45.09				15.20		1		1
		Interoffice Transport - Dedicated - DS1 combination - Per Mile															
		Per Month	<u> </u>	<u>L</u>	UNC1X	1L5XX	0.2652			<u></u>	<u> </u>				<u> </u>		ı
		Interoffice Transport - Dedicated - DS1 - Facility Termination Per]		
		Month		<u></u>	UNC1X	U1TF1	70.47	143.58	103.88				15.20				
1		Channelization - Channel System DS1 to DS0 combination Per								<u> </u>						-	
	1	Month	<u> </u>	<u> </u>	UNC1X	MQ1	105.09	59.97	12.96								1
		Voice Grade COCI - DS1 to DS0 Channel System combination -	1										1		Ì		i
	1	per month	ļ	<u> </u>	UNCVX	1D1VG	0.6497	5.91	4.26								
		Additional 4-Wire Analog Voice Grade Loop in same DS1	1	l .	l <u></u> .	1]						l		Ì		i
		Interoffice Transport Combination - Zone 1	ļ	1	UNCVX	UEAL4	30.81	94.21	45.09				15.20		ļ		
		Additional 4-Wire Analog Voice Grade Loop in same DS1	1	_											1		1
		Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	38.32	94.21	45.09				15.20				
		Additional 4-Wire Analog Voice Grade Loop in same DS1		_	1,10,0,0								4-0-				1
	1	Interoffice Transport Combination - Zone 3	ļ	3	UNCVX	UEAL4	60.39	94.21	45.09				15.20				
		Voice Grade COCI - DS1 to DS0 Channel System combination -	1	1	LINGVO	4041/0	0.040=		4.00				1		1		1
		per month	<u> </u>		UNCVX	1D1VG	0.6497	5.91	4.26								

	D NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Fxhil	bit: C
											Svc Order	Svc Order	Incremental		Incremental	
											Submitted	Submitted		Charge -	Charge -	Charge -
		Intori									Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
													151	Add I	DISC ISI	DISC Add I
						_	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	U	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNC1X	UNCCC		5.43	5.43				15.20				1
4-WIRI	E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE				0.10	0.10				10.20				—
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice															—
	Transport Combination - Zone 1		1	UNCDX	UDL56	30.99	94.21	45.09				15.20				i
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice		-	CHODA	ODLOG	00.00	0 T.Z.1	40.00				10.20				
	Transport Combination - Zone 2		2	UNCDX	UDL56	36.78	94.21	45.09				15.20				i
 	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice			ONODA	ODESO	30.70	34.21	40.00				13.20				
	Transport Combination - Zone 3		3	UNCDX	UDL56	38.92	94.21	45.09				15.20				1
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			ONODA	ODESO	30.32	34.21	40.00				13.20				
	Per Month			UNC1X	1L5XX	0.2652										i
\vdash	Interoffice Transport - Dedicated - DS1 - combination Facility	1		UNUIA	ILOAA	0.2032				-	-	-		-		
]]	Termination Per Month	1		UNC1X	U1TF1	70.47	143.58	103.88				15.20		Ì		1
	Channelization - Channel System DS1 to DS0 combination Per	 	-	UNC IA	UIIFI	70.47	143.58	103.88	1		 	15.∠0		 		
				LINIOAY		405.00	50.07	40.00								i
	Month	1		UNC1X	MQ1	105.09	59.97	12.96	1	-		ļ		1		
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per			LINODY	40400	4.00	5.04	4.00								i
	month (2.4-64kbs)			UNCDX	1D1DD	1.38	5.91	4.26								
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															1
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	30.99	94.21	45.09				15.20				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															i
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	36.78	94.21	45.09				15.20				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															i
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	38.92	94.21	45.09				15.20				<u> </u>
	OCU-DP COCI (data) - DS1 to DS0 Channel System -															ĺ
	combination per month (2.4-64kbs)			UNCDX	1D1DD	1.38	5.91	4.26								<u> </u>
	Nonrecurring Currently Combined Network Elements Switch -As-															ĺ
	Is Charge			UNC1X	UNCCC		5.43	5.43				15.20				L
4-WIRE	E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL)												L
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															ĺ
	Transport Combination - Zone 1		1	UNCDX	UDL64	30.99	94.21	45.09				15.20				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															1
	Transport Combination - Zone 2		2	UNCDX	UDL64	36.78	94.21	45.09				15.20				L
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															ĺ
	Transport Combination - Zone 3		3	UNCDX	UDL64	38.92	94.21	45.09				15.20				<u> </u>
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															1
	Per Month	<u> </u>		UNC1X	1L5XX	0.2652				<u></u>						<u> </u>
	Interoffice Transport - Dedicated - DS1 combination - Facility															1
1 1	Termination Per Month	l		UNC1X	U1TF1	70.47	143.58	103.88			l	15.20		Ì		1
	Channelization - Channel System DS1 to DS0 combination Per															
1 1	Month	l		UNC1X	MQ1	105.09	59.97	12.96			l	1		Ì		1
	OCU-DP COCI (data) - DS1 to DS0 Channel System															
1 1	combination - per month (2.4-64kbs)	l		UNCDX	1D1DD	1.38	5.91	4.26			l	1		Ì		1
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1										İ					
1 1	Interoffice Transport Combination - Zone 1	1	1	UNCDX	UDL64	30.99	94.21	45.09				15.20		Ì		1
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1										İ					
	Interoffice Transport Combination - Zone 2	l	2	UNCDX	UDL64	36.78	94.21	45.09			1	15.20				1
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1			-						İ				İ		ſ
1 1	Interoffice Transport Combination - Zone 3	l	3	UNCDX	UDL64	38.92	94.21	45.09			1	15.20				1
 	OCU-DP COCI (data) - DS1 to DS0 Channel System	1		-			¥ ·· ·							1		
1 1	combination - per month (2.4-64kbs)	l		UNCDX	1D1DD	1.38	5.91	4.26			l	1		Ì		1
 	Nonrecurring Currently Combined Network Elements Switch -As-	1					0.01	20						1		
1 1	Is Charge	l		UNC1X	UNCCC		5.43	5.43			1	15.20				1
4-WIRI	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE	ROFFI	CE TRA				50	3.10		1				1		
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	1			1					1				1		
	Transport - Zone 1	1	1	UNC1X	USLXX	85.70	169.22	100.89				15.20		Ì		1
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	1			30230	55.75	100.22	100.00		1		10.20		1		
	Transport - Zone 2	1	2	UNC1X	USLXX	194.96	169.22	100.89				15.20		Ì		1
-	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	l	_			.000	.00.22		<u> </u>		 	.5.20		 		
1 1	Transport - Zone 3	l	3	UNC1X	USLXX	491.94	169.22	100.89			l	15.20		Ì		1

UNBUND	DLF	O NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Fxhil	bit: C
ONDONE		THE THORK ELEMENTO Education										Svc Order	Svc Order	Incremental		Incremental	
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGOR	Y	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			""											Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
										T N1	B'				D-1(A)		
							Rec	Nonrec		Nonrecurring		001150	SOMAN		Rates(\$)	001441	001141
		Interoffice Transport - Dedicated - DS1 combination - Per Mile						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Per Month			UNC1X	1L5XX	0.2652										
-		Interoffice Transport - Dedicated - DS1 combination - Facility			UNCIX	ILJAA	0.2032			+							
		Termination Per Month			UNC1X	U1TF1	70.47	143.58	103.88				15.20				
-		Nonrecurring Currently Combined Network Elements Switch -As-			011017	01111	70.47	140.00	100.00				10.20				
		Is Charge			UNC1X	UNCCC		5.43	5.43				15.20				
4-V	VIRE	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTI	ROFFI	CE TRA													
		First DS1Loop in DS3 Interoffice Transport Combination - Zone			` '												
		1		1	UNC1X	USLXX	85.70	169.22	100.89				15.20				İ
		First DS1Loop in DS3 Interoffice Transport Combination - Zone															
		2		2	UNC1X	USLXX	194.96	169.22	100.89				15.20				
	Ţ	First DS1Loop in DS3 Interoffice Transport Combination - Zone			l <u>-</u>	L				1 7		1]		_		1
		3		3	UNC1X	USLXX	491.94	169.22	100.89	.			15.20		ļ		1
		Interoffice Transport - Dedicated - DS3 combination - Per Mile								1		1	1		I		1
		Per Month			UNC3X	1L5XX	6.04			1					1		
		Interoffice Transport - Dedicated - DS3 - Facility Termination per			UNC3X	LIATEO	050.45	200.00	404.40			1	45.00				1
-		month DS3 to DS1 Channel System combination per month			UNC3X UNC3X	U1TF3 MQ3	850.45 201.48	296.68 107.05	121.16 48.07	+			15.20				
		DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.78	5.91	4.26								-
		Additional DS1Loop in DS3 Interoffice Transport Combination -			UNCIX	OCIDI	11.76	5.91	4.20	+ +							
		Zone 1		1	UNC1X	USLXX	85.70	169.22	100.89				15.20				
		Additional DS1Loop in DS3 Interoffice Transport Combination -		<u> </u>	ONOTA	COLOC	00.70	100.22	100.00				10.20				
		Zone 2		2	UNC1X	USLXX	194.96	169.22	100.89				15.20				
		Additional DS1Loop in DS3 Interoffice Transport Combination -															
		Zone 3		3	UNC1X	USLXX	491.94	169.22	100.89				15.20				İ
		DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.78	5.91	4.26								
		Nonrecurring Currently Combined Network Elements Switch -As-															
		Is Charge			UNC3X	UNCCC		5.43	5.43				15.20				
2-V		VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE IN	EROFF	ICE TR	ANSPORT (EEL)												
		2-WireVG Loop used with 2-wire VG Interoffice Transport											4=00				İ
		Combination - Zone 1		1	UNCVX	UEAL2	14.93	94.21	45.09				15.20				
		2-WireVG Loop used with 2-wire VG Interoffice Transport		2	LINIOVA	UEAL2	25.25	04.04	45.00				45.00				
-		Combination - Zone 2 2-WireVG Loop used with 2-wire VG Interoffice Transport			UNCVX	UEAL2	25.35	94.21	45.09	-			15.20				
		Combination - Zone 3		3	UNCVX	UEAL2	50.46	94.21	45.09				15.20				İ
		Interoffice Transport - Dedicated - 2-wire VG combination - Per		3	ONOVA	OLALZ	30.40	34.21	45.05				13.20				
		Mile Per Month			UNCVX	1L5XX	0.013			1		1	1		I		1
		Interoffice Transport - Dedicated - 2- Wire Voice Grade					5.5.10								1		
		combination - Facility Termination per month			UNCVX	U1TV2	22.60	72.60	41.75			1	15.20				1
		Nonrecurring Currently Combined Network Elements Switch -As-								İ							
		Is Charge			UNCVX	UNCCC		5.43	5.43				15.20				
4-V		VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EROFF	ICE TR	ANSPORT (EEL)												
		4-WireVG Loop used with 4-wire VG Interoffice Transport	1	l . –		L ¬				1 T		1	l		_		1
		Combination - Zone 1		1	UNCVX	UEAL4	30.81	94.21	45.09	.			15.20		ļ		1
		4-WireVG Loop used with 4-wire VG Interoffice Transport		2	LINIONA								4		1		1
		Combination - Zone 2	1	2	UNCVX	UEAL4	38.32	94.21	45.09	+ +			15.20		1		
		4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	60.39	94.21	45.09	1		1	15.20		I		1
\vdash	-	Interoffice Transport - Dedicated - 4-wire VG combination - Per		3	OINCVA	UEAL4	60.39	94.21	45.09	+			15.20		+		
		Mile Per Month			UNCVX	1L5XX	0.013								1		1
		Interoffice Transport - Dedicated - 4- Wire Voice Grade					0.010			 					1		t
		combination - Facility Termination per month			UNCVX	U1TV4	19.81	72.60	41.75				15.20		1		1
		Nonrecurring Currently Combined Network Elements Switch -As-															
		Is Charge			UNCVX	UNCCC		5.43	5.43	1		1	15.20		I		1
DS	3 DIC	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRA	NSPOR	T (EEL)												
		High Capacity Unbundled Local Loop - DS3 combination - Per							· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·						
		Mile per month			UNC3X	1L5ND	10.04										
		High Capacity Unbundled Local Loop - DS3 combination -								1		1	1		I		1
		Facility Termination per month		l	UNC3X	UE3PX	362.34	188.45	125.51								1

UNBUND	LED	NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhil	oit: C
CITECITE	Ť	TELITORIC ELEMENTO Education										Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	′	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
	-						_	Nonrec	urrina	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	lr	nteroffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	6.04										
		nteroffice Transport - Dedicated - DS3 combination - Facility															I
\vdash		ermination per per month			UNC3X	U1TF3	850.45	296.68	121.16				15.20				
		lonrecurring Currently Combined Network Elements Switch -As- s Charge			UNC3X	UNCCC		5.43	5.43				15.20				I
STS		GITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROF	FICE TE	ANSP		UNCCC		5.43	5.43				15.20				
0.0		ligh Capacity Unbundled Local Loop - STS1 combination - Per	1	1	I (LLL)												
		file per month			UNCSX	1L5ND	10.04										I
		ligh Capacity Unbundled Local Loop - STS1 combination -															i
		acility Termination per month			UNCSX	UDLS1	374.56	188.45	125.51								ļ
		nteroffice Transport - Dedicated - STS1 combination - Per Mile			LINGOV	41.5307	0.04										I
		er month hteroffice Transport - Dedicated - STS1 combination - Facility			UNCSX	1L5XX	6.04										
		remination per month			UNCSX	U1TFS	830.19	296.68	121.16				15.20				l
		Ionrecurring Currently Combined Network Elements Switch -As-				7	3000		.20	†		l –	.0.20				
	Is	s Charge			UNCSX	UNCCC		5.43	5.43				15.20				<u> </u>
2-W		SDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL	.)													
		rirst 2-Wire ISDN Loop in a DS1 Interoffice Combination															I
		ransport - Zone 1		1	UNCNX	U1L2X	22.09	94.21	45.09				15.20				1
		First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2	UNCNX	U1L2X	35.28	94.21	45.09				15.20				I
		irst 2-Wire ISDN Loop in a DS1 Interoffice Combination			UNCINA	UTLZA	35.26	34.21	45.09				13.20				
		ransport - Zone 3		3	UNCNX	U1L2X	65.18	94.21	45.09				15.20				I
		nteroffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.2652										
		nteroffice Transport - Dedicated - DS1 combintion - Facility															
		ermination per month			UNC1X	U1TF1	70.47	143.58	103.88				15.20				
		Channelization - Channel System DS1 to DS0 combination - er month			UNC1X	MQ1	105.09	59.97	12.96								I
 		-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System			UNCIA	IVIQT	105.09	59.97	12.90								
		ombination - per month			UNCNX	UC1CA	2.96	5.91	4.26								I
		dditional 2-wire ISDN Loop in same DS1Interoffice Transport															
		Combination - Zone 1		1	UNCNX	U1L2X	22.09	94.21	45.09				15.20				l
		dditional 2-wire ISDN Loop in same DS1Interoffice Transport		_					4= 00								I
		Combination - Zone 2		2	UNCNX	U1L2X	35.28	94.21	45.09				15.20				1
		additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3		3	UNCNX	U1L2X	65.18	94.21	45.09				15.20				İ
\vdash		-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System	1	3	014014/	UILZA	05.10	3 4 .21	45.05				13.20				
		ombintaion- per month			UNCNX	UC1CA	2.96	5.91	4.26								l
		Ionrecurring Currently Combined Network Elements Switch -As-							-		-						
<u> </u>		s Charge	<u></u>		UNC1X	UNCCC		5.43	5.43				15.20				
4-W		OS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN First DS1 Loop in STS1 Interoffice Transport Combination -	II EROF	FICE T	KANSPORT (EEL)					 		-					
		one 1		1	UNC1X	USLXX	85.70	169.22	100.89				15.20				l
		irst DS1 Loop in STS1 Interoffice Transport Combination -	1	† ·			33.70		.00.00				.0.20		1		
		one 2		2	UNC1X	USLXX	194.96	169.22	100.89				15.20				<u> </u>
		irst DS1 Loop in STS1 Interoffice Transport Combination -															
$\sqcup \! \! \perp$		one 3		3	UNC1X	USLXX	491.94	169.22	100.89				15.20				
		nteroffice Transport - Dedicated - STS1 combination - Per Mile Per Month			UNCSX	1L5XX	6.04										I
$\vdash \vdash \vdash$		rer Month hteroffice Transport - Dedicated - STS1 combination - Facility	1	-	OINCOV	ILDAX	6.04					-	-				
		ermination			UNCSX	U1TFS	830.19	296.68	121.16				15.20				l
		TS1 to DS1 Channel System conbination per month	İ		UNCSX	MQ3	201.48	107.05	48.07								
		S3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.78	5.91	4.26								
		dditional DS1Loop in STS1 Interoffice Transport Combination -															
$\vdash \vdash$	Z	one 1	<u> </u>	1	UNC1X	USLXX	85.70	169.22	100.89				15.20				
		additional DS1Loop in STS1 Interoffice Transport Combination - done 2		2	UNC1X	USLXX	194.96	169.22	100.89				15.20				İ
\vdash		Idditional DS1Loop in STS1 Interoffice Transport Combination -	 		014017	JOLAA	134.30	103.22	100.09	 		 	13.20				
1 1		Cone 3		3	UNC1X	USLXX	491.94	169.22	100.89				15.20		1		I

<u> </u>	IDLE	D NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhi	bit: C
CATEGO	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec			g Disconnect	001150	001441		Rates(\$)	001441	001411
		DS3 Interface Unit (DS1 COCI) combination per month		<u> </u>	UNC1X	UC1D1	11.78	First 5.91	Add'I 4.26	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Nonrecurring Currently Combined Network Elements Switch -As-			UNCIA	ОСТОТ	11.70	5.91	4.20			1					+
		Is Charge			UNCSX	UNCCC		5.43	5.43				15.20				
4	-WIRE	56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE 1	RANS								1					†
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															1
		Combination - Zone 1		1	UNCDX	UDL56	30.99	94.21	45.09				15.20				
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		_													
		Combination - Zone 2		2	UNCDX	UDL56	36.78	94.21	45.09				15.20				
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		3	UNCDX	UDL56	38.92	04.04	45.09				45.00				
		Combination - Zone 3 Interoffice Transport - Dedicated - 4-wire 56 kbps combination -		3	UNCDX	UDL56	38.92	94.21	45.09			+	15.20			-	+
		Per Mile			UNCDX	1L5XX	0.013										
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination -			CHODA	120/01	0.010										+
		Facility Termination			UNCDX	U1TD5	15.61	72.60	41.75				15.20				
		Nonrecurring Currently Combined Network Elements Switch -As-															
		Is Charge			UNCDX	UNCCC		5.43	5.43				15.20				
4	-WIRE	64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERO	FFICE 1	RANS	PORT (EEL)												
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport															
		Combination - Zone 1		1	UNCDX	UDL64	30.99	94.21	45.09				15.20				
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	36.78	94.21	45.09				15.20				
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	38.92	94.21	45.09				15.20				
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
		Per Mile			UNCDX	1L5XX	0.013										
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination			UNCDX	U1TD6	15.61	72.60	41.75				15.20				
		Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCDX	UNCCC		5.43	5.43				15.20				
		IETWORK ELEMENTS															
		used as a part of a currently combined facility, the non-recurr															
		used as ordinarily combined network elements in All States, th					As Is Charge	does not.									
IN.	vonrec	curring Currently Combined Network Elements "Switch As Is" Nonrecurring Currently Combined Network Elements Switch -As-		(One a	applies to each cor	nbination)											
		Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		5.43	5.43				15.20				
		Nonrecurring Currently Combined Network Elements Switch -As-			ONOVA	011000		0.40	0.40				10.20				†
		Is Charge - 56/64 kbps			UNCDX	UNCCC		5.43	5.43				15.20				
		Nonrecurring Currently Combined Network Elements Switch -As-															
		ls Charge - DS1			UNC1X	UNCCC		5.43	5.43				15.20				
		Nonrecurring Currently Combined Network Elements Switch -As-															
		ls Charge - DS3			UNC3X	UNCCC		5.43	5.43				15.20				
		Nonrecurring Currently Combined Network Elements Switch -As-			LINICOV	LINICCC		5 40	F 40				45.00				
	IOTE:	Is Charge - STS1 Local Channel - Dedicated Transport - minimum billing period	d - Bolo	w Des	UNCSX	UNCCC	r months	5.43	5.43				15.20				
- 10	IOIE.	Local Channel - Dedicated Transport - Infilming benefit Local Channel - Dedicated - 2-Wire Voice Grade	u - Beio	W D33	UNCXV	ULDV2	18.32	187.51	32.21			1					+
		Local Channel - Dedicated - 4-Wire Voice Grade			UNCXV	ULDV4	19.41	187.94	32.63								1
		Local Channel - Dedicated - DS1 per month Zone 1		1	UNC1X	ULDF1	39.18	172.34	149.27				15.20				
		Local Channel - Dedicated -DS1 Per Month Zone 2		2	UNC1X	ULDF1	121.58	172.34	149.27				15.20				
		Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	70.02	172.34	149.27				15.20				
		Local Channel - Dedicated - DS3 - Per Mile per month			UNC3X	1L5NC	7.82		·								
		Local Channel - Dedicated - DS3 - Facility Termination			UNC3X	ULDF3	469.44	438.46	256.30				15.20			1	<u> </u>
		Local Channel - Dedicated - STS-1- Per Mile per month		ļ	UNCSX	1L5NC	7.82	100.10	050.00			1	15.20				
-	Intion	Local Channel - Dedicated - STS-1 - Facility Termination al Features & Functions:			UNCSX	ULDFS	457.22	438.46	256.30			1				 	
		PLEXERS	-	-	-							+				+	+
		Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	105.09	88.41	60.76			-	15.20			t	+
 		OCU-DP COCI (data) - DS1 to DS0 Channel System - per	1				100.09	55.71	33.70			1	10.20		1	1	
		month (2.4-64kbs)		1	UDL	1D1DD	1.38	6.39	4.58				15.20		l	I	1

UNBUNDLE	D NETWORK ELEMENTS - Louisiana			,									ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				d Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring Discon				Rates(\$)		
						Nec	First	Add'l	First Add	i'i SOM	EC SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per														
	month			UDN	UC1CA	2.96	6.39	4.58			15.20				
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.6497	6.39	4.58			15.20				
	DS3 to DS1 Channel System per month			UXTD3	MQ3	201.48	172.99	91.25			15.20				
	STS1 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) used with Loop per month			UXTS1 USL	MQ3 UC1D1	201.48 11.78	172.99 6.39	91.25 4.58		-	15.20 15.20		+		
 	DS3 Interface Unit (DS1 COCI) used with Local Channel per			USL	OCIDI	11.70	0.39	4.50		+	13.20	'	1		
	month			ULDD1	UC1D1	11.78	6.39	4.58							
	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel			OLDD 1	00101	11.70	0.00	4.00							
	per month			U1TD1	UC1D1	11.78	6.39	4.58							
Acces	s to DCS - Customer Reconfiguration (FlexServ)						0.00								
	LOCAL EXCHANGE SWITCHING(PORTS)		i									1	1		
Excha	nge Ports											<u> </u>			
	Although the Port Rate includes all available features in GA, I	KY, LA	& TN, t	he desired features	will need to b	e ordered usir	g retail USOCs	5							
2-WIR	VOICE GRADE LINE PORT RATES (RES)														
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.52	2.31	2.21			15.20)			<u> </u>
			1							1		. [1	1	
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.52	2.31	2.21			15.20)			
	L														
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.52	2.31	2.21			15.20)			.
	Exchange Ports - 2-Wire VG unbundled LA extended local			LIEDOD	LIEDAG	4.50	0.04	0.04			45.00				
+	dialing parity Port with Caller ID - Res.			UEPSR	UEPAS	1.52	2.31	2.21			15.20)			ļ
	Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RUL)			UEPSR	UEPAG	1.52	2.31	2.21			15.20				
	Exchange Ports - 2-Wire VG unbundled res, low usage line port			UEFOR	UEPAG	1.52	2.31	2.21			15.20	'	-		
	with Caller ID (LUM)			UEPSR	UEPAP	1.52	2.31	2.21			15.20				
 	Exchange Ports - 2-Wire VG Louisiana Residence Dialing Plan			ULFOR	ULFAF	1.32	2.31	2.21		+	13.20	'	1		
	without Caller ID			UEPSR	UEPWG	1.52	2.31	2.21			15.20	,			
	Exchange Ports - 2-Wire VG Louisiana Residence Area Plus			OLI OIX	OLI WO	1.02	2.01	2.21			10.20	`-	1		-
	without Caller ID			UEPSR	UEPRQ	1.52	2.31	2.21			15.20	,			
	2-Wire voice unbundled Low Usage Line Port without Caller ID										.,,,,,				
	Capability			UEPSR	UEPRT	1.52	2.31	2.21			15.20)			
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00			15.20)			
FEAT	IRES														
	All Available Vertical Features			UEPSR	UEPVF	0.00	0.00	0.00			15.20)			
2-WIR	E VOICE GRADE LINE PORT RATES (BUS)														
	Exchange Ports - 2-Wire Analog Line Port without Caller ID -		1							1 -]	1
	Bus		<u> </u>	UEPSB	UEPBL	1.52	2.31	2.21			15.20	1	ļ		
	Exchange Ports - 2-Wire VG unbundled Line Port with					!				1					
\vdash	unbundled port with Caller+E484 ID - Bus.		<u> </u>	UEPSB	UEPBC	1.52	2.31	2.21			15.20	<u>' </u>	 	 	
	Evaluation Ports 2 Wire Applied Line Port systeming and D		1	LIEDOD	LIEDRO	4.50	0.01	0.01		1	45.00	, I	1	1	
 	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus. Exchange Ports - 2-Wire VG unbundled LA extended local	1	}	UEPSB	UEPBO	1.52	2.31	2.21	 		15.20	<u>'</u>	 	 	
	dialing parity Port with Caller ID - Bus.		1	UEPSB	UEPAX	1.52	2.31	2.21		1	15.20	, I	1	1	
 	Exhange Ports - 2-Wire VG unbundled incoming only port with		 	ULFOD	UEPAX	1.52	2.31	2.21	 		15.20	' 	+	-	
	Caller ID - Bus		1	UEPSB	UEPB1	1.52	2.31	2.21		1	15.20	ı I		1	1
	Exchange Ports - 2-Wire VG unbundled Louisiana Bus Area		1	OL1 0D	CLIDI	1.32	2.31	2.21			13.20	<u> </u>	1		
	Calling Port with Caller ID - Bus (BUC)		1	UEPSB	UEPAA	1.52	2.31	2.21		1	15.20	ı I		1	1
	Exchange Ports - 2-Wire Voice Louisiana Business Dialing Plan		†	- "		52	2.01		 	1	.0.20	1	1	1	
	without Caller ID		1	UEPSB	UEPWH	1.52	2.31	2.21		1	15.20	ı [1	1
	Exchange Ports - 2-Wire Voice Louisiana Business Area Calling		1												
	Port without Caller ID			UEPSB	UEPBA	1.52	2.31	2.21		1	15.20)			
	2-Wire voice unbundled Incoming Only Port without Caller ID														
	Capability			UEPSB	UEPBE	1.52	2.31	2.21			15.20				
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00			15.20				
FEATU															
	All Available Vertical Features			UEPSB	UEPVF	0.00	0.00	0.00			15.20) [
EXCH	ANGE PORT RATES (DID & PBX)														
	2-Wire VG Unbundled 2-Way PBX Trunk - Res		<u> </u>	UEPSE	UEPRD	1.52	30.37	14.42			15.20)	<u> </u>		<u></u>

UNBUNDL F	D NETWORK ELEMENTS - Louisiana											Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			d Submitted Manually	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		
						1	Nonrec	urring	Nonrecurring Discon	noct		088	Rates(\$)		
 						Rec	First	Add'l	First Add		SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.52	30.37	14.42	Tilot Auc	1 JOINEC	15.20	JOMAN	JONIAN	JOHAN	JOHAN
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.52	30.37	14.42			15.20				
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.52	30.37	14.42			15.20				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Port			UEPSP	UEPL2	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.52	30.37	14.42			15.20				
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD				I								1		
ļļ	Capable Port			UEPSP	UEPXE	1.52	30.37	14.42			15.20				ļ
	2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Callling Port			UEPSP	UEPXK	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPSP	UEPXL	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPSP	UEPXM	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPSP	UEPXO	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local														
	Discount Calling Port			UEPSP	UEPXP	1.52	30.37	14.42			15.20				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.52	30.37	14.42			15.20				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00			15.20				
FEATU															
	All Available Vertical Features			UEPSP UEPSE	UEPVF	0.00	0.00	0.00			15.20				
EXCHA	ANGE PORT RATES (COIN)														
	Exchange Ports - Coin Port	L				1.52	2.31	2.21			15.20				
	Transmission/usage charges associated with POTS circuit sv												L		
	Access to B Channel or D Channel Packet capabilities will be LOCAL EXCHANGE SWITCHING(PORTS)	avallai	oie oniy	tnrough BFR/New	Business Re	quest Process.	Rates for the	раскет сарарі	lities will be determined	Via the Bona i	ide Request/	New Business	s Request Pro	cess.	
	ANGE PORT RATES										+				
EXCITA	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.29	115.85	18.20			15.20				
 	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID			OLI LX	OLITZ	0.23	110.00	10.20			13.20				
	capability			UEPDD	UEPDD	68.47	196.18	92.92			15.20				
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	10.07	70.76	51.46			15.20				
<u> </u>	All Features Offered			UEPTX UEPSX	UEPVF	0.00	0.00	0.00			10.20		1		Ì
NOTE:	Transmission/usage charges associated with POTS circuit sv	vitched	usage						nission by B-Channels a	ssociated with	2-wire ISDN	ports.			
	Access to B Channel or D Channel Packet capabilities will be			through BFR/New	Business Re								s Request Pro	cess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0.00	0.00	0.00							
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPEX	UEPEX	94.82	197.92	98.62			15.20				
	NDLED PORT with REMOTE CALL FORWARDING CAPABILITY														
UNBUN	NDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE				L										
	Unbundled Remote Call Forwarding Service, Area Calling, Res			UEPVR	UERAC	1.52	2.31	2.21			15.20		ļ		
													1		
 	Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.52	2.31	2.21			15.20	1	 	1	1
 	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.52	2.31	2.21			15.20		 	ļ	
Non D	Unbundled Remote Call Forwarding Service, IntraLATA - Res			UEPVR	UERTR	1.52	2.31	2.21			15.20	1	 	-	1
Non-Re	ecurring Unbundled Remote Call Forwarding Service - Conversion -				 						+	1	 	-	1
	Switch-as-is			UEPVR	USAC2		0.10	0.10			15.20				
	Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)			UEPVR	USACC		0.10	0.10							
UNBUN	NDLED REMOTE CALL FORWARDING - Bus										\bot		ļ		ļ
	Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVB	UERAC	1.52	2.31	2.21			15.20				
	i	1	1		1	1						1	I	1	
	l.,, .,,,, .,,,,,,,,,,,,,,,,,,,,,,,,,,,					!									
	Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVB UEPVB	UERLC UERTE	1.52 1.52	2.31 2.31	2.21 2.21			15.20 15.20				

Version 3Q02: 09/06/02

	ETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			II.	Submitted	Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec			g Disconnect				Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	oundled Remote Call Forwarding Service, IntraLATA - Bus			UEPVB	UERTR	1.52	2.31	2.21				15.20				
	bundled Remote Call Forwarding Service Expanded and					. =0						4= 00				
	ception Local Calling			UEPVB	UERVJ	1.52	2.31	2.21				15.20				
Non-Recurr																
	oundled Remote Call Forwarding Service - Conversion -			UEPVB	USAC2		0.10	0.10				15.20				
	oundled Remote Call Forwarding Service - Conversion with			UEPVB	USAC2		0.10	0.10				15.20				
	wed change (PIC and LPIC)			UEPVB	USACC		0.10	0.10								
	AL SWITCHING, PORT USAGE			OLF VB	USACC		0.10	0.10								
	Switching (Port Usage)															
	d Office Switching Function, Per MOU					0.001868										
End	d Office Trunk Port - Shared, Per MOU					0.00018										
	witching (Port Usage) (Local or Access Tandem)	†		1	+	2.00010				-	1			I	1	
	ndem Switching Function Per MOU					0.0001067										
	ndem Trunk Port - Shared, Per MOU		1	İ		0.000222				1				1		
Common T			1	İ		5.300222				1				1		
	mmon Transport - Per Mile, Per MOU					0.0000032										
	mmon Transport - Facilities Termination Per MOU					0.0003748										
	T/LOOP COMBINATIONS - COST BASED RATES															
Cost Based	Rates are applied where BellSouth is required by FCC ar	nd/or St	ate Co	mmission rule to p	rovide Unbun	dled Local Swi	tching or Swite	h Ports.								
	hall apply to the Unbundled Port/Loop Combination - Cos								d Port section	of this Rate E	xhibit.					
	and Tandem Switching Usage and Common Transport Us											n Port/Loop	Combination	ns.		
	nd additional Port nonrecurring charges apply to Not Curr															
	ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)				1			<u> </u>			1					
	oop Combination Rates															
	Vire VG Loop/Port Combo - Zone 1															
			1 1			13.13										
			1 2			13.13 23.75										
	/ire VG Loop/Port Combo - Zone 2		2			23.75										
2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3															
UNE Loop	/ire VG Loop/Port Combo - Zone 2 /ire VG Loop/Port Combo - Zone 3 Rates		2	UEPRX	UEPLX	23.75 49.62										
UNE Loop I	/ire VG Loop/Port Combo - Zone 2 /ire VG Loop/Port Combo - Zone 3 Rates /ire Voice Grade Loop (SL1) - Zone 1		3	UEPRX UEPRX	UEPLX UEPLX	23.75 49.62 11.77										
2-W UNE Loop I 2-W 2-W	/ire VG Loop/Port Combo - Zone 2 /ire VG Loop/Port Combo - Zone 3 Rates //ire Voice Grade Loop (SL1) - Zone 1 //ire Voice Grade Loop (SL1) - Zone 2		3	UEPRX	UEPLX	23.75 49.62 11.77 22.39										
2-W UNE Loop I 2-W 2-W 2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3		2 3 1 2			23.75 49.62 11.77										
2-W UNE Loop I 2-W 2-W 2-W 2-Wire Voic	/ire VG Loop/Port Combo - Zone 2 /ire VG Loop/Port Combo - Zone 3 Rates //ire Voice Grade Loop (SL1) - Zone 1 //ire Voice Grade Loop (SL1) - Zone 2		2 3 1 2	UEPRX	UEPLX	23.75 49.62 11.77 22.39 48.26	38.85	19.08				15.20				
2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Dec Grade Line Port Rates (Res) Vire voice unbundled port - residence		2 3 1 2	UEPRX UEPRX	UEPLX UEPLX	23.75 49.62 11.77 22.39 48.26	38.85 38.85	19.08				15.20 15.20				
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Dece Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port with Caller ID - res		2 3 1 2	UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL	23.75 49.62 11.77 22.39 48.26	38.85 38.85 38.85					15.20 15.20 15.20				
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Dec Grade Line Port Rates (Res) Vire voice unbundled port - residence		2 3 1 2	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	23.75 49.62 11.77 22.39 48.26 1.36	38.85	19.08				15.20				
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 COMBO CO		2 3 1 2	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	23.75 49.62 11.77 22.39 48.26 1.36	38.85	19.08				15.20				
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port with Caller ID - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing		2 3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO	23.75 49.62 11.77 22.39 48.26 1.36 1.36	38.85 38.85	19.08 19.08				15.20 15.20				
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Ce Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing ity port with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res		2 3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO	23.75 49.62 11.77 22.39 48.26 1.36 1.36	38.85 38.85	19.08 19.08				15.20 15.20 15.20				
2-W UNE Loop 2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port with Caller ID - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res		2 3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAS	23.75 49.62 11.77 22.39 48.26 1.36 1.36	38.85 38.85 38.85	19.08 19.08				15.20 15.20				
2-W UNE Loop 2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Vire Voice Grade Loop (SL1) - Zone 3 Vire Voice Grade Loop (SL1) - Zone 3 Vire voice unbundled port - residence Vire voice unbundled port with Caller ID - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res VIL) Vire voice unbundled sers, low usage line port with Caller ID		2 3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAS	23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36	38.85 38.85 38.85 38.85	19.08 19.08 19.08				15.20 15.20 15.20				
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Ce Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Viry port with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled sers, low usage line port with Caller ID M)		2 3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAS UEPAG	23.75 49.62 11.77 22.39 48.26 1.36 1.36	38.85 38.85 38.85	19.08 19.08				15.20 15.20 15.20				
2-W UNE Loop 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W (RU 2-W (LU 2-W 2-W (LU 2-W 2-W (LU 2-W 2-W 2-W 2-W (LU 2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port vith Caller ID - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundles res, low usage line port with Caller ID M) Vire Voice Unbundled Louisiana Residence Dialing Plan		2 3 1 2	UEPRX				15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Ce Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Viry port with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled sers, low usage line port with Caller ID M)		2 3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAS UEPAG	23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36	38.85 38.85 38.85 38.85	19.08 19.08 19.08				15.20 15.20 15.20				
2-W UNE Loop I 2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 22 Vire Voice Grade Loop (SL1) - Zone 3 22 Vire Voice Grade Loop (SL1) - Zone 3 23 24 Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Residence Dialing Plan vire Voice Unbundled Louisiana Residence Dialing Plan vout Caller ID Vire voice unbundled Louisiana Area Plus Port without Caller		2 3 1 2	UEPRX				15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop I 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 Compared Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID M) Vire Voice Unbundled Louisiana Residence Dialing Plan Nout Caller ID Vire voice unbundled Louisiana Area Plus Port without Caller Capability		2 3 1 2	UEPRX				15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 22 Vire Voice Grade Loop (SL1) - Zone 3 22 Vire Voice Grade Loop (SL1) - Zone 3 23 24 Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Residence Dialing Plan vire Voice Unbundled Louisiana Residence Dialing Plan vout Caller ID Vire voice unbundled Louisiana Area Plus Port without Caller		2 3 1 2	UEPRX				15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3		2 3 1 2	UEPRX br>19.08				15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port ortigoring only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Residence Dialing Plan Lout Caller ID Vire Voice Unbundled Louisiana Area Plus Port without Caller Loapability Vire voice unbundled Louisiana Area Plus Port without Caller ID Dability Sepatures Offered		2 3 1 2	UEPRX br>19.08				15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus Port with Caller ID Vire Voice Unbundled Louisiana Residence Dialing Plan Nout Caller ID Vire voice unbundled Louisiana Area Plus Port without Caller Capability Vire voice unbundled Low Usage Line Port without Caller ID Daibility		2 3 1 2	UEPRX br>19.08 19.08				15.20 15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port ortigoring only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Residence Dialing Plan Lout Caller ID Vire Voice Unbundled Louisiana Area Plus Port without Caller Loapability Vire voice unbundled Louisiana Area Plus Port without Caller ID Dability Sepatures Offered		2 3 1 2	UEPRX br>19.08 19.08				15.20 15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Residence Dialing Plan Nout Caller ID Vire voice unbundled Louisiana Area Plus Port without Caller Capability Vire voice unbundled Louisiana Area Plus Port without Caller ID Dability Seatures Offered MBER PORTABILITY		2 3 1 2	UEPRX br>19.08 19.08				15.20 15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID M) Vire Voice Unbundled Louisiana Residence Dialing Plan Nout Caller ID Vire voice unbundled Louisiana Area Plus Port without Caller Capability Vire voice unbundled Low Usage Line Port without Caller ID ability BER PORTABILITY al Number Portability (1 per port)		2 3 1 2	UEPRX br>19.08 19.08				15.20 15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice unbundled port outgoing only - res Vire voice unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus Port with Caller ID Min Vire Voice Unbundled Louisiana Residence Dialing Plan Nout Caller ID Vire voice unbundled Louisiana Area Plus Port without Caller Capability Vire voice unbundled Louisiana Area Plus Port without Caller ID Dability S Features Offered MBER PORTABILITY Lai Number Portability (1 per port) RRING CHARGES (NRCs) - CURRENTLY COMBINED		2 3 1 2	UEPRX br>19.08 19.08				15.20 15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 De Grade Line Port Rates (Res) Vire voice unbundled port - residence Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus With Caller ID Will Vire voice unbundled Louisiana Residence Dialing Plan Lout Caller ID Vire Voice Unbundled Louisiana Area Plus Port without Caller Capability Vire voice unbundled Louisiana Area Plus Port without Caller ID Dability Seatures Offered MBER PORTABILITY La Number Portability (1 per port) RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop / Line Port Combination - Conversion -		2 3 1 2	UEPRX br>19.08 19.08 19.08				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20								
2-W UNE Loop	Vire VG Loop/Port Combo - Zone 2 Vire VG Loop/Port Combo - Zone 3 Rates Vire Voice Grade Loop (SL1) - Zone 1 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 2 Vire Voice Grade Loop (SL1) - Zone 3 CC Grade Line Port Rates (Res) Vire voice Grade Loop (SL1) - Zone 3 CC Grade Line Port Rates (Res) Vire voice unbundled port vith Caller ID - res Vire voice unbundled port outgoing only - res Vire voice Grade unbundled Louisiana extended local dialing vity port with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Area Plus with Caller ID - res Vire voice unbundled Louisiana Residence Dialing Plan Nout Caller ID Vire Voice Unbundled Louisiana Area Plus Port without Caller Capability Vire voice unbundled Low Usage Line Port without Caller ID Dability SE Features Offered MBER PORTABILITY Lat Number Portability (1 per port) RRING CHARGES (NRCs) - CURRENTLY COMBINED Vire Voice Grade Loop / Line Port Combination - Conversion - tich-as-is		2 3 1 2	UEPRX br>19.08 19.08 19.08				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20								

ONBONDL	ED NETWORK ELEMENTS - Louisiana			1										nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						_	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
	Activity			UEPRX	USAS2	0.00	0.00	0.00				15.20				
2-WIF	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			13.13										
	2-Wire VG Loop/Port Combo - Zone 2		2			23.75									20.00	
	2-Wire VG Loop/Port Combo - Zone 3		3			49.62										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL1) - Zone 3	1	3	UEPBX	UEPLX	48.26			ļ							
2-Wir	e Voice Grade Line Port (Bus)	1		L					ļ			L				
	2-Wire voice unbundled port without Caller ID - bus	1		UEPBX	UEPBL	1.36	38.85	19.08				15.20				
	2-Wire voice unbundled port with Caller + E484 ID - bus		<u> </u>	UEPBX	UEPBC	1.36	38.85	19.08	ļ			15.20				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.36	38.85	19.08				15.20				<u> </u>
	2-Wire voice Grade unbundled Louisiana extended local dialing		1	l	[Ì	1
	parity port with Caller ID - bus			UEPBX	UEPAX	1.36	38.85	19.08				15.20				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.36	38.85	19.08				15.20				
	2-Wire voice unbundled Louisiana Bus Area Calling Port with Caller ID (BUC)			UEPBX	UEPAA	1.36	38.85	19.08				15.20				
	2-Wire Voice Unbundled Louisiana Business Dialing Plan without Caller ID			UEPBX	UEPWH	1.36	38.85	19.08				15.20				
	2-Wire voice unbundled Louisiana Business Area Calling Port without Caller ID Capability			UEPBX	UEPBA	1.36	38.85	19.08				15.20				
	2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPBX	UEPBE	1.36	38.85	19.08				15.20				
LOCA	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEAT	URES															
	All Features Offered		<u> </u>	UEPBX	UEPVF	0.00	0.00	0.00				15.20				
NONE	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED		<u> </u>													.
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPBX	USAC2		0.10	0.10				15.20				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	1		HEDDY	110400		0.40	0.40				45.00				
ADD	Switch with change	1	1	UEPBX	USACC		0.10	0.10	 	ļ	-	15.20			 	-
ADDI	TIONAL NRCs	1	1	 	+				 	ļ	-				 	-
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity		1	UEPBX	USAS2		0.00	0.00				15.20			1	1
2-10/10	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)	<u> </u>	-	OLPDA	USASZ		0.00	0.00		-		15.20			-	
	Port/Loop Combination Rates	1		 	1 1				1	1					1	
UNE	2-Wire VG Loop/Port Combo - Zone 1	1	1	 	1 1	13.13			1	1					1	
	2-Wire VG Loop/Port Combo - Zone 2	1	2	 	+	23.75			 	 					 	
	2-Wire VG Loop/Port Combo - Zone 3	1	3	 	+	49.62			 	 					 	
UNF	Loop Rates	1		 	+ +	43.02			 						 	
0.12	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEPRG	UEPLX	11.77					<u> </u>				 	<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	2	UEPRG	UEPLX	22.39				1					1	
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	48.26				1					1	
2-Wir	e Voice Grade Line Port Rates (RES - PBX)		Ť		1				1	1					1	
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.36	66.91	31.29				15.20				
LOCA	AL NUMBER PORTABILITY			1	1		00.01	020		1		.0.20			1	
	Local Number Portability (1 per port)	1		UEPRG	LNPCP	3.15	0.00	0.00	1	İ		15.20			İ	
FEAT	URES				1	20	2.00	2.00		1					1	
1 1	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00		1		15.20			1	
NONE	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED	1		1	1				1	İ					İ	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPRG	USAC2		7.68	1.85				15.20				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPRG	USACC		7.68	1.85				15.20				

IINRI	IND! F	D NETWORK ELEMENTS - Louisiana												Attach	ment: 2	Exhib	nit: C
CIADO	,.1VLC	NETWORK ELLINERTO - Louisiana										Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Indan:									Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATE	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																D130 131	Disc Add I
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	ADDITI	ONAL NRCs															
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -											4= 00				
		Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				15.20				
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
	O MUDE	Group VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)						7.11	7.11				15.20				
		crt/Loop Combination Rates		<u> </u>		+						-					
	UNE PO	2-Wire VG Loop/Port Combo - Zone 1		1			13.13										
		2-Wire VG Loop/Port Combo - Zone 1		2		+	23.75										
-	1	2-Wire VG Loop/Port Combo - Zone 3		3			49.62					1					
		pop Rates		3		+	45.02			1							
	J.12 E	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	11.77					<u> </u>		 	 		
-	 	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	22.39			 				 	 		
 	1	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	48.26					<u> </u>		 	 		
—	2-Wire	Voice Grade Line Port Rates (BUS - PBX)		Ť			.0.20							1			
		1000 1000 1000 1000 1000				1								1	1		
	1	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus		1	UEPPX	UEPPC	1.36	66.91	31.29				15.20				
	†	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.36	66.91	31.29	1			15.20	İ	1		
	1	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled 2-Way Combination PBX Louisiana															
		Calling Port			UEPPX	UEPL2	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
		Capable Port			UEPPX	UEPXE	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional															
		Calling Port			UEPPX	UEPXK	1.36	66.91	31.29				15.20				
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPPX	UEPXL	4.00	00.04	24.20				45.00				
		Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPPX	UEPAL	1.36	66.91	31.29				15.20				
		Room Calling Port			UEPPX	UEPXM	1.36	66.91	31.29				15.20				
-	1	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			OLFFX	OLFAIN	1.30	00.91	31.29			1	13.20				
1	1	Discount Room Calling Port			UEPPX	UEPXO	1.36	66.91	31.29				15.20	1	1		
\vdash	 	2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local			OLI I A	JLI AU	1.50	00.31	31.29	 			10.20	 	 		
1	1	Discount Calling Port			UEPPX	UEPXP	1.36	66.91	31.29				15.20	1	1		
	†	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port		†	UEPPX	UEPXS	1.36	66.91	31.29				15.20	1	1		
	LOCAL	NUMBER PORTABILITY											1				
		Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00				15.20				
	FEATU																
		All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00				15.20				
	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -						_]		
L	ļ	Conversion - Switch-As-Is			UEPPX	USAC2		7.68	1.85				15.20				
1	1	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -				1				[1]		
<u> </u>	L	Conversion - Switch with Change			UEPPX	USACC		7.68	1.85				15.20	ļ	ļ		
	ADDITI	ONAL NRCs		ļ													
	1	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -		1	HEDDY	110400	0.00	0.00	0.00				45.00				
-	1	Subsequent Activity		1	UEPPX	USAS2	0.00	0.00	0.00			-	15.20	-	-		
	1	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group		1				7.11	7.11				15.20				
<u> </u>	2-WIDE	Group EVOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	T T	<u> </u>		+		7.11	7.11	 			15.20				
-		e voice grade Loop with z-wire analog line coin Por ort/Loop Combination Rates	1	1		+				+ +							
-	JINE PO	2-Wire VG Coin Port/Loop Combo – Zone 1		1		+	13.13			 							
—	 	2-Wire VG Coin Port/Loop Combo – Zone 2		2		+	23.75			 				 	 		
	 	2-Wire VG Coin Port/Loop Combo – Zone 3		3		+	49.62			 				1	1		
	UNET	pop Rates		Ť		+	70.02			 							
	J.IL L	70p00	L							1		1	1	1	1		

NBUNDLE	NETWORK ELEMENTS - Louisiana			П	, r								Attachr			oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring Disc					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL1) - Zone 1			UEPCO	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPCO	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	48.26										
	Voice Grade Line Ports (COIN)				+											
	2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)			UEPCO	UEPRF	1.36	38.85	19.08				15.20				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,			UEPCO	UEPKF	1.30	30.00	19.06				15.20				
	900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRA	1.36	38.85	19.08				15.20				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking			OLI OO	OLITICA	1.50	30.03	13.00				13.20				
	(AL. LA. MS)			UEPCO	UEPRB	1.36	38.85	19.08				15.20				
	2-Wire Coin 2-Way with Operator Screening & Blocking:			02. 00	OZ. KB	1.00	00.00	10.00				10.20				
	900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.36	38.85	19.08				15.20				
	2-Wire Coin Outward without Blocking and without Operator															
	Screening (KY, LA, MS)			UEPCO	UEPRN	1.36	38.85	19.08				15.20				
	2-Wire Coin Outward with Operator Screening and 011 Blocking															
	(LA)			UEPCO	UEPLA	1.36	38.85	19.08				15.20				
	2-Wire Coin Outward with Operator Screening and Blocking:															
	011, 900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRH	1.36	38.85	19.08				15.20				
	2-Wire Coin Outward Operator Screening & Blocking: 900/976,															
	1+DDD, 011+, and Local (AL, KY, LA, MS)			UEPCO	UEPCN	1.36	38.85	19.08				15.20				
	2-Wire Coin 2-Way Smartline with 900/976 (Louisiana only)			UEPCO	UEPNA	1.36	38.85	19.08				15.20				
	2-Wire Coin Outward Smartline with 900/976 (Louisiana only)			UEPCO	UEPCB	1.36	38.85	19.08				15.20				
	ONAL UNE COIN PORT/LOOP (RC)															
	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	1.81	0.00	0.00				15.20				
	NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
	CURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPCO	USAC2		0.10	0.10				15.20				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -											4= 00				
	Switch with change			UEPCO	USACC		0.10	0.10				15.20				
	ONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0.00	0.00				15.20				
	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	ODT /		U3A32		0.00	0.00				15.20				
	ort/Loop Combination Rates	LINE	JORT (KES)	-											
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		+ +	16.45										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1 2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2		+	26.87									1	
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3		+ +	51.98										
	op Rates		Ť		1	355									1	
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	14.93									1	
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	25.35									1	
	2-Wire Voice Grade Loop (SL2) - Zone 3			UEPFR	UECF2	50.46									İ	
	Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.52	104.41	67.93				15.20				
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.52	104.41	67.93				15.20				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.52	104.41	67.93				15.20				
	2-Wire voice Grade unbundled Louisiana extended local dialing]	
	parity port with Caller ID - res			UEPFR	UEPAS	1.52	104.41	67.93				15.20				
	2-Wire voice unbundled Louisiana Area Plus with Caller ID - res			l	[]										1	1
	(RUL)			UEPFR	UEPAG	1.52	104.41	67.93				15.20				
	2-Wire voice unbundles res, low usage line port with Caller ID						,		1							
	(LUM)			UEPFR	UEPAP	1.52	104.41	67.93				15.20				
	2-Wire Voice Unbundled Louisiana Residence Dialing Plan			LIEDED	LIEDC		,					4= 00			1	1
	without Caller ID			UEPFR	UEPWG	1.52	104.41	67.93				15.20				
	OFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															

or F FEATURES All LOCAL NU Loc NONRECUI 2-\(\text{V} \) Cor 2-WIRE VO UNE Port/L 2-\(\text{V} \)	teroffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile Fraction Mile SS I Features Offered UMBER PORTABILITY Deal Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-with-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE //Loop Combination Rates Wire VG Loop/IO Transport/Port Combo - Zone 1	Interi m	Zone	BCS UEPFR UEPFR	usoc 1L5XX	Rec	Nonrec First		Nonrecurring Disc	Sul p	ubmitted S Elec	Submitted	Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic Disc Add'l
or F FEATURES All LOCAL NU LOCAL NU COT	Fraction Mile IS I Features Offered UMBER PORTABILITY ocal Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP / 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates				1L5XX					connect				Dotoo(¢)		1
or F FEATURES All LOCAL NU LOCAL NU COT COT 2-\hat{V} COT	Fraction Mile IS I Features Offered UMBER PORTABILITY ocal Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP / 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates				1L5XX		First	A -1 -111		Joinicot .			oss	rates(a)		
or F FEATURES All LOCAL NU LOCAL NU LOCAL NU LOCAL NU COT 2-\hat{V} COT 2-\hat{V} COT 2-\hat{V} COT	Fraction Mile IS I Features Offered UMBER PORTABILITY ocal Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP / 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates				1L5XX			Add'l	First A	Add'I S	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
FEATURES All	I Features Offered UMBER PORTABILITY coal Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE //Loop Combination Rates				1L5XX									1		
All	I Features Offered UMBER PORTABILITY DOZAI Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE //Loop Combination Rates			UEPFR		0.013										
LOCAL NU Loc NONRECUI 2-W Cor 2-WIRE VO UNE POrt/L 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	UMBER PORTABILITY coal Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port combination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port combination - Conversion - Switch-With-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE //Loop Combination Rates			UEPFR												
Loc NONRECUI 2-W Cor 2-W Rev Cor 2-W Rev 2-W 2-W Rev 2-W	ocal Number Portability (1 per port) URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP / 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates			l	UEPVF	0.00	0.00	0.00				15.20				
NONRECUI	URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates			1												
2-W Cor 2-WIRE VO UNE POrIVL 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE //Loop Combination Rates			UEPFR	LNPCX	0.35								1		
Cor 2-WiRE VO UNE Port/L 2-W 2-W UNE Loop 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	ombination - Conversion - Switch-as-is Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE //Loop Combination Rates															
2-W COV UNE Port/L 2-W 2-W 2-W UNE Loop 2-W 2-W 2-W 2-W 2-Wire Voic 2-W 2-W 2-W	Wire Loop / Dedicated IO Transport / 2 Wire Line Port ombination - Conversion - Switch-With-Change OICE LOOP 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates															
Cor 2-WIRE VO UNE Port/L 2-W	ombination - Conversion - Switch-With-Change OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates			UEPFR	USAC2		8.24	1.81				15.20				
2-WIRE VO UNE Port/L 2-W 2-W UNE Loop 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE /Loop Combination Rates															
UNE Port/L 2-W 2-W UNE Loop 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Loop Combination Rates			UEPFR	USACC		8.24	1.81				15.20				
2-W 2-W UNE Loop 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W		LINE	PORT (BUS)												
2-W 2-W UNE Loop 2-W 2-W 2-Wire Voic 2-Wire 2-W 2-W	Wire VG Loop/IO Tranport/Port Combo - Zone 1															
2-W UNE Loop 2-W 2-W 2-Wire Voic 2-W-2-W			1			16.45								·		
UNE Loop	Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			26.87										
2-Wire Voic 2-Wire 2-Wire 2-Wire Voic 2-Wire Voic	Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			51.98								·		
2-W 2-Wire Void 2-Wire 2-W 2-W																
2-Wire Void 2-Wire 2-W 2-W	Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	14.93										
2-Wire Void 2-W 2-W	Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	25.35										
2-W 2-W	Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	50.46										
2-W	ice Grade Line Port (Bus)															
2-W	Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.52	104.41	67.93				15.20				
	Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.52	104.41	67.93				15.20				
. I 12-V	Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.52	104.41	67.93				15.20				
	Wire voice Grade unbundled Alabama extended local dialing															
	arity port with Caller ID - bus			UEPFB	UEPAW								,			
	Wire voice Grade unbundled Louisiana extended local dialing															
	arity port with Caller ID - bus			UEPFB	UEPAX	1.52	104.41	67.93				15.20	,			
	Wire voice unbundled incoming only port with Caller ID - Bus		†	UEPFB	UEPB1	1.52	104.41	67.93				15.20				
	Wire voice unbundled Louisiana Bus Area Calling Port with			02.75	02. 5.	02		01.00				.0.20			 	1
	aller ID (BUC)			UEPFB	UEPAA	1.52	104.41	67.93				15.20	,			
	Wire Voice Unbundled Louisiana Business Dialing Plan			OLITE	OL1701	1.02	104.41	07.00			+	10.20			 	
	ithout Caller ID			UEPFB	UEPWH	1.52	104.41	67.93				15.20	,			
	UMBER PORTABILITY			OLITB	OLI VVII	1.52	104.41	07.33				13.20				
	ocal Number Portability (1 per port)			UEPFB	LNPCX	0.35										
	FICE TRANSPORT		1	UEPFB	LINPUX	0.33					+				-	1
	teroffice Transport - Dedicated - 2 Wire Voice Grade - Facility				-						+	\longrightarrow				
	ermination			UEPFB	U1TV2	22.60	20.26	26.62				15 20	,			
	teroffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile	<u> </u>	-	OLPED	UTIVZ	22.00	39.36	20.02			\longrightarrow	15.20			 	
	Fraction Mile	l	1	UEPFB	1L5XX	0.013				1			,	1	1	1
FEATURES		-	-	UEPFB	ILDAA	0.013			 		\longrightarrow	\longrightarrow			 '	
		 	1	LIEDED	UEPVF	0.00	0.00	0.00			\longrightarrow	15.00			 	
	Features Offered			UEPFB	UEPVF	0.00	0.00	0.00				15.20				ļ
	URRING CHARGES (NRCs) - CURRENTLY COMBINED Wire Loop / Dedicated IO Transport / 2 Wire Line Port	1	<u> </u>		_						\longrightarrow				 	
				LIEDED	110400		0.04	4.04				45.00				
	ombination - Conversion - Switch-as-is	l	-	UEPFB	USAC2		8.24	1.81		-	\longrightarrow	15.20			<u> </u>	
	Wire Loop / Dedicated IO Transport / 2 Wire Line Port	l	1	LIEDED	110400		201	4.61		1		45.00	,	1	1	1
	ombination - Conversion - Switch with change	<u> </u>	<u> </u>	UEPFB	USACC		8.24	1.81				15.20			 	
	OICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	<u> </u>	<u> </u>	-	1						\longrightarrow				├ ──	
	/Loop Combination Rates		<u> </u>			10.15										ļ
	Wire VG Loop/IO Tranport/Port Combo - Zone 1	<u> </u>	1	-	1	16.45					\longrightarrow				├ ──	
	Wire VG Loop/IO Tranport/Port Combo - Zone 2	<u> </u>	2	-	1	26.87					\longrightarrow				├ ──	
	Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			51.98								·	<u> </u>	<u> </u>
UNE Loop			<u> </u>	LIEDED	1,505									·	<u> </u>	<u> </u>
	Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	14.93								·	<u> </u>	<u> </u>
	Wire Voice Grade Loop (SL2) - Zone 2			UEPFP	UECF2	25.35					\longrightarrow			·	ļ	
	Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	50.46								·	ļ	<u> </u>
2-Wire Void	ice Grade Line Port Rates (BUS - PBX)								1							
Line			1	1									l	l		

ONRONDI	LED NETWORK ELEMENTS - Louisiana											,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Name		Name and a second and	- Dianamant					2.00 .01	2.007.444.
		-	<u> </u>			Rec	Nonrec First	Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
	Line Side Unbundled Outward PBX Trunk Port - Bus		1	UEPFP	UEPPO	1.52	132.47	82.14	FIISt	Auu i	SOMEC	15.20	JOWAN	JOWAN	SOWAN	JOWAN
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.52	132.47	82.14				15.20				+
	2-Wire Voice Unbundled 2-Way Combination PBX Louisiana		1	OLITI	OLI I	1.02	132.47	02.14			1	13.20				+
	Calling Port			UEPFP	UEPL2	1.52	132.47	82.14				15.20				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.52	132.47	82.14				15.20				+
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.52	132.47	82.14				15.20				+
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.52	132.47	82.14				15.20				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.52	132.47	82.14				15.20				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.52	132.47	82.14				15.20				1
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD					_	-									1
	Capable Port	1		UEPFP	UEPXE	1.52	132.47	82.14				15.20				1
	2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional															1
	Calling Port	1		UEPFP	UEPXK	1.52	132.47	82.14				15.20				1
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															1
	Administrative Calling Port	1	1	UEPFP	UEPXL	1.52	132.47	82.14				15.20				I
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPFP	UEPXM	1.52	132.47	82.14				15.20				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															1
	Discount Room Calling Port			UEPFP	UEPXO	1.52	132.47	82.14				15.20				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local															
	Discount Calling Port			UEPFP	UEPXP	1.52	132.47	82.14				15.20				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	1.52	132.47	82.14				15.20				
LOC	CAL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.20				
INTE	EROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFP	U1TV2	22.60	39.36	26.62				15.20				
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFP	1L5XX	0.013										
FEA	TURES															
	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00				15.20				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		8.24	1.81				15.20				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch with change			UEPFP	USACC		8.24	1.81				15.20				
	D PORT/LOOP COMBINATIONS - COST BASED RATES															
	IRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	KPORT	<u> </u>													
UNE	Port/Loop Combination Rates		_			00.00										-
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			23.20										-
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			33.62										-
LINIE	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			58.73										-
UNE	Loop Rates 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	14.93						15.20				-
			2													+
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX UEPPX	UECD1 UECD1	25.35 50.46						15.20 15.20				+
LIME	E Port Rate		3	UEPPX	UECDI	50.46						15.20				+
ONE	Exchange Ports - 2-Wire DID Port	+	 	UEPPX	UEPD1	8.27	217.95	83.92			 	15.20		-	1	+
NON	RECURRING CHARGES - CURRENTLY COMBINED	1	1	OLI I A	021 01	0.27	211.30	05.92			1	13.20		1	1	+
1401	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -	. 	 	 	+ +				 		1				1	+
	Switch-as-is	1	1	UEPPX	USAC1		7.10	1.81				15.20				1
 	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion	1		0=117	00,101		7.10	1.01				10.20		1	1	
	with BellSouth Allowable Changes	1		UEPPX	USA1C		7.10	1.81				15.20				1
ADD	DITIONAL NRCs	1	1		55,110		7.10	1.01			 	10.20		1	<u> </u>	+
7,00	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk	1		UEPPX	USAS1		26.01	26.01				15.20		1	1	
Tele	phone Number/Trunk Group Establisment Charges	1		52. TX	30,101		20.01	20.01			1	10.20				
1 316	DID Trunk Termination (One Per Port)	1		UEPPX	NDT	0.00	0.00	0.00			l -	15.20		1		
	Additional DID Numbers for each Group of 20 DID Numbers	1	1	UEPPX	ND4	0.00	0.00	0.00			1	15.20		 	†	+
	DID Numbers, Non- consecutive DID Numbers , Per Number	+	 	UEPPX	ND5	0.00	0.00	0.00			†	15.20			†	+

ONRONDI	LED	NETWORK ELEMENTS - Louisiana												_		nent: 2		bit: C
CATEGORY	,	RATE ELEMENTS	Interi m	Zone	В	cs	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
								D	Nonrec	urring	Nonrecurring I	Disconnect			oss	Rates(\$)		
								Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	R	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00				15.20				
	R	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00				15.20				
LOC	CAL N	IUMBER PORTABILITY																
	Lo	ocal Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00								
2-W	IRE IS	SDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LII	NE SIDE	PORT														
UNE		t/Loop Combination Rates																
		W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																i .
		INE Zone 1		1	UEPPB	UEPPR		27.48										l
		W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																1
		INE Zone 2		2	UEPPB	UEPPR		40.34										1
		W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -			1			l					1				I	1
		INE Zone 3		3	UEPPB	UEPPR]	70.99										
UNE		p Rates			L		L				ļ						ļ	
	2-	-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	19.09						15.20				
					1		l	l					1				I	1
		-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	31.95			ļ			15.20				
		-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	62.60						15.20				
UNE		t Rate			ļ													
		xchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	8.39	184.10	128.42				15.20				1
NON		URRING CHARGES - CURRENTLY COMBINED																
		-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																i
		Combination - Conversion			UEPPB	UEPPR	USACB	0.00	37.40	26.23				15.20				1
		NAL NRCs																
LOC		IUMBER PORTABILITY																
		ocal Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-C		NEL USER PROFILE ACCESS:																
		CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
		CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
		CSD		TA1\	UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								+
B-C		NEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SO	C,MS, &	IN)	UEPPB	UEPPR	LIALICD	0.00	0.00	0.00	-							
-		CVS/CSD (DMS/5ESS) CVS (EWSD)			UEPPB	UEPPR	U1UCD			0.00	-							
-					UEPPB	UEPPR	U1UCE U1UCF	0.00	0.00	0.00	-							
Her		CSD ERMINAL PROFILE		-	UEPPB	UEFFR	UTUCF	0.00	0.00	0.00								
031		Jser Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00							-	
VED		AL FEATURES			UEPPB	UEPPR	UTUIVIA	0.00	0.00	0.00								-
VER		Il Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00	+			15.20			 	
INT		FICE CHANNEL MILEAGE			JEITD	OLCER	OLI VI	0.00	0.00	0.00	+			13.20			 	
11411		nteroffice Channel mileage each, including first mile and			 		 	i			 						 	
		acilities termination			LIFPPR	UEPPR	M1GNC	22.613	39.36	26.62	1		1	15.20			I	1
		nteroffice Channel mileage each, additional mile					M1GNM	0.013	0.00	0.00	 		 	15.20			I	
4-W		DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	PORT		OLITE	OLITIK	WITCHNI	0.010	0.00	0.00				10.20				
		t/Loop Combination Rates			1		1	<u> </u>			 						†	
J.,,_		W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			1												1	
		Cone 1		1	UEPPP			180.52					1				I	1
	4\	W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE one 2		2	UEPPP			289.78										
-+		W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		<u> </u>	1		1				 						†	
		Cone 3		3	UEPPP			586.76					1				I	1
UNE		p Rates		Ť	 						1						t	1
		-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	85.70			i			15.20			1	
		-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	194.96						15.20				
		-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	491.94			i t			15.20			İ	\Box
UNE		t Rate						1			† †						İ	
		xchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	94.82	443.08	251.60	i t			15.20			İ	ſ
NON		URRING CHARGES - CURRENTLY COMBINED									i						1	
		-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port			Ì													
1		Combination - Conversion -Switch-as-is			UEPPP		USACP	0.00	115.63	76.29				15.20				1
		NAL NRCs			i –						i i							

UNBUNDLED NETWORK ELEMENTS - Louisiana												ment: 2		bit: C
CATEGORY RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)		Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs Electronic Disc Add
					Rec	Nonrec	urring	Nonrecurring Disconnect			oss	Rates(\$)		
					Rec	First	Add'l	First Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-														
Inward/two way Tel Nos. (except NC)			UEPPP	PR7TF		0.48				15.20				
4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -														
Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		11.18	11.18			15.20				
4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -														
Subsequent Inward Tel Numbers			UEPPP	PR7ZT		22.35	22.35			15.20				
LOCAL NUMBER PORTABILITY														
Local Number Portability (1 per port)			UEPPP	LNPCN	1.75									
INTERFACE (Provsioning Only)														
Voice/Data			UEPPP	PR71V	0.00	0.00	0.00							
Digital Data			UEPPP	PR71D	0.00	0.00	0.00							
Inward Data			UEPPP	PR71E	0.00	0.00	0.00							
New or Additional "B" Channel														
New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.11				15.20				
New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.11				15.20				
New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.11				15.20				
CALL TYPES														
Inward			UEPPP	PR7C1	0.00	0.00	0.00							
Outward			UEPPP	PR7C0	0.00	0.00	0.00							
Two-way			UEPPP	PR7CC	0.00	0.00	0.00							
Interoffice Channel Mileage														
Fixed Each Including First Mile			UEPPP	1LN1A	70.7352	86.69	79.44			15.20				
Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.2652									
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT														
UNE Port/Loop Combination Rates														
4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		154.17					15.20				
4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		263.43					15.20				
4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		560.41					15.20				
UNE Loop Rates														
4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	85.70					15.20				
4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	194.96					15.20				
4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	491.94					15.20				
UNE Port Rate														
4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	68.47	441.34	245.90			15.20				
NONRECURRING CHARGES - CURRENTLY COMBINED														
4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	1													
- Switch-as-is			UEPDC	USAC4		125.75	65.08			15.20				
4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination														1
- Conversion with DS1 Changes	<u> </u>	1	UEPDC	USAWA		125.75	65.08			15.20				
4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination		1										I		1
- Conversion with Change - Trunk			UEPDC	USAWB		125.75	65.08			15.20				
ADDITIONAL NRCs														
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -														
Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		14.06	14.06			15.20				
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent														
Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		14.06	14.06			15.20				
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel														
Activation/Chan Inward Trunk w/out DID		1	UEPDC	UDTTC		14.06	14.06			15.20	ļ		ļ	
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan	1	1]]	1
Activation Per Chan - Inward Trunk with DID		<u> </u>	UEPDC	UDTTD		14.06	14.06			15.20				
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan	1	1	l								Ì		Ì	1
Activation / Chan - 2-Way DID w User Trans	<u> </u>	1	UEPDC	UDTTE		14.06	14.06			15.20				
BIPOLAR 8 ZERO SUBSTITUTION	<u> </u>	1	L	1		_				L				
B8ZS -Superframe Format	ļ	1	UEPDC	CCOSF		0.00	605.00		_	15.20				
B8ZS - Extended Superframe Format	ļ	 	UEPDC	CCOEF		0.00	605.00		_	15.20	ļ		ļ	
Alternate Mark Inversion	<u> </u>	1												
AMI -Superframe Format		<u> </u>	UEPDC	MCOSF		0.00	0.00							
AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00			ļ				
Telephone Number/Trunk Group Establisment Charges		<u> </u>						<u> </u>		<u> </u>		L		

UNB	UNDLE	D NETWORK ELEMENTS - Louisiana	,		•							,			ment: 2		bit: C
ATE	EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Charge -	Charge - Manual Sv Order vs. Electronic
														1st	Add'l	Disc 1st	Disc Add'
							Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)		
								First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	_	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00						15.20				
		Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00						15.20				
		Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00						15.20				
		DID Numbers for each Group of 20 DID Numbers			UEPDC UEPDC	ND4	0.00						15.20 15.20				
	-	DID Numbers, Non- consecutive DID Numbers, Per Number Reserve Non-Consecutive DID Nos.			UEPDC	ND5 ND6	0.00	0.00	0.00				15.20				
		Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00				15.20				
	Dedica	ted DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	Digital	Loop			0.00	0.00	0.00				10.20				
	200.00	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities				1											
		Termination)			UEPDC	1LNO1	70.47	86.69	79.44				15.20				
		,				1						Ì			1		
		Interoffice Channel Mileage - Additional rate per mile - 0-8 miles	1		UEPDC	1LNOA	0.2652	0.00	0.00								
		Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities															
		Termination)			UEPDC	1LNO2	0.00	0.00	0.00						<u> </u>		
		Interoffice Channel Mileage - Additional rate per mile - 9-25							-								
		miles			UEPDC	1LNOB	0.2652	0.00	0.00								
		Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities							·								
		Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							
	_	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.2652	0.00	0.00								
		Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00							
	4 14/15/	Central Office Termininating Point			UEPDC	CTG	0.00										
		E DS1 LOOP WITH CHANNELIZATION WITH PORT															
		n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti				+											
		system can have up to 24 combinations of rates depending on S1 Loop	type ar	na nun	nber of ports used												
	UNE D	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	85.70	0.00	0.00			1	15.20				1
		4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	194.96	0.00	0.00	1		1	15.20				
		4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	491.94	0.00	0.00				15.20				
	UNF D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)		020	00220	.0	0.00	0.00				10.20				
	0.12.2	24 DSO Channel Capacity - 1 per DS1	,		UEPMG	VUM24	97.35	0.00	0.00				15.20				
		48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	194.70	0.00	0.00				15.20				
		96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	389.40	0.00	0.00				15.20				
		144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	584.10	0.00	0.00				15.20				
		192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	778.80	0.00	0.00				15.20				
		240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	973.50	0.00	0.00				15.20				
		288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,168.20	0.00	0.00				15.20				
		384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,557.60	0.00	0.00				15.20				
		480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	1,947.00	0.00	0.00				15.20				
		576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,336.40	0.00	0.00				15.20				
	1	672 DS0 Channel Capacity - 1 per 28 DS1s	L	L	UEPMG	VUM67	2,725.80	0.00	0.00				15.20			ļ	
		ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with						stem								ļ	
		mum System configuration is One (1) DS1, One (1) D4 Channe															
	wultip	les of this configuration functioning as one are considered Ac	ia'i afte	r tne m	ımımum system cor	inguration is	counted.					}				ļ.	
		NRC - Conversion (Currently Combined) with or without	1		LIEDMC	LISAC4	0.00	440.40	0.40				45.00				
	Cunter	BellSouth Allowed Changes Additions at End User Locations Where 4-Wire DS1 Loop with	h Char	nol:-c	UEPMG	USAC4	0.00	146.13	8.12	 		 	15.20		-	1	
		lot Currently Combined) in all states, except in Density Zone 1					ining Exists and	'							-		
	IASM (I	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port	J. TOP	J 141 J	1	1				 		1			1	1	
		and Assoc Fea Activation	1		UEPMG	VUMD4	0.00	715.54	467.54				15.20				
	Binola	r 8 Zero Substitution	1		021 1010	V 51V154	0.00	7 10.04	407.34			1	10.20			1	<u> </u>
	poia	Clear Channel Capability Format, superframe - Subsequent	1		1	1											
		Activity Only	l		UEPMG	CCOSF	0.00	0.00	605.00				15.20				
	1	Clear Channel Capability Format - Extended Superframe -				1	2.20	2.20							İ		
		Subsequent Activity Only	1		UEPMG	CCOEF	0.00	0.00	605.00				15.20				
	Alterna	ate Mark Inversion (AMI)			-												
		Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
		Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
		nge Ports Associated with 4-Wire DS1 Loop with Channelization			1	1							-		i	1	

UNBU	NDLE	D NETWORK ELEMENTS - Louisiana												Attachr	nent: 2	Exhil	oit: C
CATEG		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchar	nge Ports															
																	ı l
		Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.52	0.00	0.00	0.00	0.00		15.20				i l
		Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.52	0.00	0.00	0.00	0.00		15.20				1
																	1
		Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.52	0.00	0.00	0.00	0.00		15.20				<u> </u>
		2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	8.29	0.00	0.00	0.00	0.00		15.20				1
	Feature	Activations - Unbundled Loop Concentration															
		Feature (Service) Activation for each Line Side Port Terminated															1
		in D4 Bank			UEPPX	1PQWM	0.6497	25.36	13.40				15.20				i l
		Feature (Service) Activation for each Trunk Side Port Terminated															1
		in D4 Bank			UEPPX	1PQWU	0.6497	78.05	18.40				15.20				ı l
	Teleph	one Number/ Group Establishment Charges for DID Service															
		DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00				15.20				
		DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				15.20				
		Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00				15.20				
		Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00				15.20				
		Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				15.20				
	Local N	lumber Portability															
		Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								$\overline{}$
	FFATU	RES - Vertical and Optional			OLI I X	2.1. 0.	0.10	0.00	0.00								$\overline{}$
		Switching Features Offered with Line Side Ports Only															$\overline{}$
		All Features Available			UEPPX	UEPVF	0.00	0.00	0.00				15.20				$\overline{}$
UNRUN		PORT LOOP COMBINATIONS - MARKET RATES			CLITA	OLI VI	0.00	0.00	0.00				10.20				
O.T.DO.		Rates shall apply where BellSouth is not required to provide	unhung	dled loc	al switching or swit	tch norte ne	r ECC and/or St	ata Commissio	n rules								
	This in		unbun	1	an switching or swi	l	1 00 ana/or oa	ate commission	iii ruico.								
		dled port/loop combinations that are Currently Combined or N	Not Cur	rently (Combined in Zone 1	of the Ton 8	MSAS in BellS	outh's region t	for end users	with 4 or more	DS0 equivalen	t lines					
		p 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderda											e)				
		uth currently is developing the billing capability to mechanica												In the interi	m where Bells	South cannot	bill Market
		BellSouth shall bill the rates in the Cost-Based section preced								.g ogoo .c.					20	, out ouor	Din martor
		arket Rate for unbundled ports includes all available features i			ine market reaces an	1	le right to true t	up the billing t	annerenoe.								
		fice and Tandem Switching Usage and Common Transport Us			e Port section of th	is rate evhib	it chall annly to	all combination	one of loon/no	rt network eler	nente evcent	or LINE Coi	n Port/Loon	Combination	se which have	a flat rate us	ane charne
		: URECU).	age rai	63 III U	ie i oit section of th	is rate exilib	it siiaii appiy to	an combination	ons or loop/po	it lietwork elei	nenta except	OI OILE COI	0.0 LOOP	Combination	is willell llave	a nat rate us	age charge
-		t Currently Combined scenarios the Nonrecurring charges are	liotod	n tha E	irot and Additional	NDC solumn	o for sook Dort	HEAC For C	urrantly Camb	nad acanarica	the Negroous	rina aharaa	o ara liatad i	in the NDC (Commonthy Com	hinad acatio	
			iisteu	ın me r	irst and Additional	NKC Column	is for each Fort	USUC. FOI CI	arrently Comb	nieu scenanos	, the Nonrecur	ring charge	s are iisteu	in the NKC - C	Juriently Com	billed Section	١.
		onal NRCs may apply also and are categorized accordingly.				1											
-		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)		-													
	UNE PO	ort/Loop Combination Rates		-			05.77										
—	1	2-Wire VG Loop/Port Combo - Zone 1	1	1		 	25.77										
—	1	2-Wire VG Loop/Port Combo - Zone 2	1	3		 	36.39										
—		2-Wire VG Loop/Port Combo - Zone 3	 	3		 	62.26										
—	UNE LO	pop Rates	 	1	LIEDDV	LIEDLY	44 77										
—	 	2-Wire Voice Grade Loop (SL1) - Zone 1	 	2	UEPRX	UEPLX UEPLX	11.77 22.39										
—	1	2-Wire Voice Grade Loop (SL1) - Zone 2	1		UEPRX	UEPLX	22.39 48.26										
—	2 M:	2-Wire Voice Grade Loop (SL1) - Zone 3	1	3	UEPRX	UEPLX	48.26										
	∠-vvire	Voice Grade Line Port (Res)	1	1	HEDDY	LIEDE:	44.00	00.00	20.00				45.00				
-	.	2-Wire voice unbundled port - residence		<u> </u>	UEPRX	UEPRL	14.00	90.00	90.00			1	15.20				
-	.	2-Wire voice unbundled port with Caller ID - res		<u> </u>	UEPRX	UEPRC	14.00	90.00	90.00			1	15.20				
	<u> </u>	2-Wire voice unbundled port outgoing only - res	<u> </u>	<u> </u>	UEPRX	UEPRO	14.00	90.00	90.00			ļ	15.20				
	1	2-Wire voice Grade unbundled Louisiana extended local dialing	l	1	LIEDDY	LIEDAG	44.00	00.00	20.00				45.00				, I
	<u> </u>	parity port with Caller ID - res	<u> </u>	<u> </u>	UEPRX	UEPAS	14.00	90.00	90.00			ļ	15.20				
	l	2-Wire voice unbundled Louisiana Area Plus with Caller ID - res	l														, I
<u> </u>	 	(RUL)	 	 	UEPRX	UEPAG	14.00	90.00	90.00			ļ	15.20				
1	1	2-Wire voice unbundled Louisiana Area Plus with Caller ID - res	l	1	l	L											i
	ļ	(AC7)		<u> </u>	UEPRX	UEPAH	14.00	90.00	90.00				15.20				
1	1	2-Wire voice unbundles res, low usage line port with Caller ID	l	1													i
		(LUM)			UEPRX	UEPAP	14.00	90.00	90.00				15.20				
	l	2-Wire voice unbundled Low Usage Line Port without Caller ID	l									1					ı I
		Capability			UEPRX	UEPRT	14.00	90.00	90.00				15.20				1

Version 3Q02: 09/06/02

UNBUNDLE	D NETWORK ELEMENTS - Louisiana			,								,		nent: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
					+ +		Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled Louisiana Area Plus Port without Caller							7.00		71441						
	ID Capability			UEPRX	UEPRQ	14.00	90.00	90.00				15.20				
LOCAL	NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
FEATU	RES															
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00				15.20				
NONRE	CURRING CHARGES - CURRENTLY COMBINED															ĺ
																Ì
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPRX	USAC2		41.50	41.50				15.20				
	2-Wire Voice Grade Loop / Line Port Combination - Switch with															
	change			UEPRX	USACC		41.50	41.50				15.20				
ADDITI	ONAL NRCs															
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -		1													
	Subsequent			UEPRX	USAS2		0.00	0.00				15.20				
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)							-								
UNE Po	ort/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			25.77										
	2-Wire VG Loop/Port Combo - Zone 2		2			36.39										1
	2-Wire VG Loop/Port Combo - Zone 3		3			62.26										ĺ
UNE Lo	oop Rates															ĺ
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	11.77										ĺ
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	22.39										ĺ
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	48.26										
2-Wire	Voice Grade Line Port (Bus)															
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	14.00	90.00	90.00				15.20				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	14.00	90.00	90.00				15.20				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	14.00	90.00	90.00				15.20				ĺ
	2-Wire voice Grade unbundled Louisiana extended local dialing															
	parity port with Caller ID - bus			UEPBX	UEPAX	14.00	90.00	90.00				15.20				
	2-Wire voice unbundled Louisiana Bus Area Calling Port with															
	Caller ID (BUC)			UEPBX	UEPAA	14.00	90.00	90.00				15.20				
	2-Wire voice unbundled Incoming Only Port without Caller ID															
	Capability			UEPBX	UEPBE	14.00	90.00	90.00				15.20				
	2-Wire Voice Unbundled Louisiana Business Dialing Plan															1
	without Caller ID			UEPBX	UEPWH	14.00	90.00	90.00				15.20				
	2-Wire voice unbundled Louisiana Business Area Calling Port															
	without Caller ID Capability			UEPBX	UEPBA	14.00	90.00	90.00				15.20				
LOCAL	NUMBER PORTABILITY															1
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
NONRE	CURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPBX	USAC2		41.50	41.50				15.20				
	2-Wire Voice Grade Loop / Line Port Combination - Switch with															
	change			UEPBX	USACC		41.50	41.50				15.20				
ADDITI	ONAL NRCs															
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -															
	Subsequent			UEPBX	USAS2		0.00	0.00				15.20				
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
UNE Po	ort/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			25.77										
	2-Wire VG Loop/Port Combo - Zone 2		2			36.39										
	2-Wire VG Loop/Port Combo - Zone 3		3			62.26										
UNE Lo	pop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRG	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRG	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRG	UEPLX	48.26										
2-Wire	Voice Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -															1
	Res			UEPRG	UEPRD	14.00	90.00	90.00				15.20				

ONRC	UNDLE	D NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhi	bit: C
CATEC	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		N	RATES(\$)	Non	a Diagrama		Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
	-						Rec	Nonrec First	Add'l	First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	LOCAL	I NUMBER PORTABILITY						FIISL	Add I	FIISL	Add I	SOWIEC	SOWAN	SOWAN	SOWAN	SOWAN	SOWAN
	LOCAL	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15										
	NONRE	CURRING CHARGES - CURRENTLY COMBINED			OLI IKO	LIVI OI	0.10										
												1					1
		2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPRG	USAC2		41.50	41.50				15.20				
		2-Wire Voice Grade Loop/ Line Port Combination - Switch with															
		Change			UEPRG	USACC		41.50	41.50				15.20				
		ONAL NRCs															
		2 Wire Loop/Line Side Port Combination - Non feature -															
		Subsequent Activity- Nonrecurring						0.00	0.00				15.20				
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
		Group						14.64	14.64				15.20				
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
	UNE Po	ort/Loop Combination Rates															
		2-Wire VG Loop/Port Combo - Zone 1		1			25.77										
		2-Wire VG Loop/Port Combo - Zone 2		2			36.39										
		2-Wire VG Loop/Port Combo - Zone 3		3			62.26										
	UNE Lo	pop Rates															
		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	11.77										
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	22.39										
		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	48.26										
	2-Wire	Voice Grade Line Port Rates (BUS - PBX)															
		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00				15.20				
		Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00				15.20				
		Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled 2-Way Combination PBX Louisiana															
		Calling Port			UEPPX	UEPL2	14.00						15.20				
		2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
		Capable Port			UEPPX	UEPXE	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional															
		Calling Port			UEPPX	UEPXK	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
		Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
		Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
		Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00				15.20				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local															
		Discount Calling Port			UEPPX	UEPXP	14.00	90.00	90.00				15.20				
	1.00	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	<u> </u>		UEPPX	UEPXS	14.00	90.00	90.00		1		15.20			-	
	LUCAL	NUMBER PORTABILITY	 		LIEDDY	LNDCS		0.00									<u> </u>
	FF 4 T.	Local Number Portability (1 per port)	<u> </u>		UEPPX	LNPCP	3.15	0.00	0.00		1					-	
	FEATU		<u> </u>		LIEDDY	LIED: /=					1					-	
	Never	All Features Offered	<u> </u>		UEPPX	UEPVF	0.00	0.00	0.00		1		15.20			-	
	NONRE	CURRING CHARGES - CURRENTLY COMBINED	<u> </u>		-						1					-	
	1	O.Wise Veice Condo Lace / Line Bort Combination Co. 2011 Ac. In	1		LIEDDY	LICACO		44.50	44.50		1		45.00		l	I	
	+	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is	1		UEPPX	USAC2		41.50	41.50	-	1	1	15.20		1	 	
	1	2-Wire Voice Grade Loop/ Line Port Combination - Switch with	1		LIEDDY	LICACO		44.50	44.50		1		45.00		l	I	
	ADDIT	Change	1		UEPPX	USACC		41.50	41.50	-	1	1	15.20		1	 	
	AUUIII	ONAL NRCs	 		_						 	1			-	 	
	1	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent	l	1	UEPPX	USAS2		0.00	0.00	1	1	1	15.20		1	1	

ONROND	LED NETWORK ELEMENTS - Louisiana			1										ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec			g Disconnect				Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2 Wire Loop/Line Side Port Combination - Non feature -											4= 00				
	Subsequent Activity- Nonrecurring		1				0.00	0.00				15.20				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64				15.20				
2 14/1	IRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN P	ODT	-				14.04	14.04				15.20				
	E Port/Loop Combination Rates	JKI	1		-											
ONL	2-Wire VG Coin Port/Loop Combo – Zone 1		1			25.77										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			36.39										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			62.26										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL1) - Zone 3			UEPCO	UEPLX	48.26			İ	İ					1	
2-Wi	ire Voice Grade Line Port Rates (Coin)															
	2-Wire Coin 2-Way without Operator Screening and without															
	Blocking (AL, KY, LA, MS)			UEPCO	UEPRF	14.00	90.00	90.00				15.20				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 01	١,														
	900/976, 1+DDD (AL, KY, LA, MS, SC)			UEPCO	UEPRA	14.00	90.00	90.00				15.20				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking															
	(AL, LA, MS)			UEPCO	UEPRB	14.00	90.00	90.00				15.20				
	2-Wire Coin 2-Way with Operator Screening & Blocking:															
	900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	14.00	90.00	90.00				15.20				
	2-Wire Coin Outward without Blocking and without Operator															
	Screening (KY, LA, MS)			UEPCO	UEPRN	14.00	90.00	90.00				15.20				
	2-Wire Coin Outward with Operator Screening and 011 Blocking	g														
	(LA)		1	UEPCO	UEPLA	14.00	90.00	90.00				15.20				
	2-Wire Coin Outward with Operator Screening and Blocking:			LIEBOO	LIEDDII	44.00	00.00	00.00				45.00				
	011, 900/976, 1+DDD (AL, KY, LA, MS)		<u> </u>	UEPCO	UEPRH	14.00	90.00	90.00				15.20				
	2-Wire Coin Outward Operator Screening & Blocking: 900/976 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCN	14.00	90.00	90.00				15.20				
1.00	CAL NUMBER PORTABILITY	_	-	UEPCO	UEPCN	14.00	90.00	90.00				15.20				
LOC	Local Number Portability (1 per port)	-	1	UEPCO	LNPCX	0.35								-	-	
NON	NRECURRING CHARGES - CURRENTLY COMBINED		1	OLFCO	LINFOX	0.33										
NON	WESSIKING SHAKSES - COKKENTET GOMBINED	_	1		-											
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPCO	USAC2		41.50	41.50				15.20				
+	2-Wire Voice Grade Loop/ Line Port Combination - Switch with	<u> </u>	1	021 00	00/102		41.00	41.00				10.20				
	Change			UEPCO	USACC		41.50	41.50				15.20				
ADD	DITIONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequen	:		UEPCO	USAS2		0.00	0.00				15.20				
UNBUNDLE	D PORT/LOOP COMBINATIONS - MARKET BASED RATES		1													
2-WI	IRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUI	IK PORT														
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			50.93										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			61.35										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3	ļ		86.46	, and the second									<u> </u>
UNE	Loop Rates		<u> </u>	L										1	1	
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	14.93					ļ	15.20				
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	25.35				 	<u> </u>	15.20		-	-	
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	50.46			1	 	ļ	15.20		!	!	
UNE	E Port Rate	+	+	LIEDDY	HEDD4	20.00	600.00	45.00	ļ	 	 	45.00		 	 	1
NO	Exchange Ports - 2-Wire DID Port NRECURRING CHARGES - CURRENTLY COMBINED	+	+	UEPPX	UEPD1	36.00	600.00	45.00	ļ	 	 	15.20		 	 	1
NON	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination	-	1								1			 	 	
	Switch-As-Is Top 8 MSAs only	1		UEPPX	USAC1		100.00	42.50				15.20		1	1	
 	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion		+	UEFFA	USACT		100.00	4∠.50	1	1	 	15.∠0		 	 	}
	with BellSouth Allowable Changes Top 8 MSAs only			UEPPX	USA1C		100.00	42.50				15.20		1	1	
ADD	DITIONAL NRCs	+	1	OLI-FA	JOATO		100.00	42.50			1	13.20		1	1	1
עטאן ו	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX	USAS1		45.00	45.00			.	15.20				ļ

UNI	RUNDLE	D NETWORK ELEMENTS - Louisiana											,			ment: 2		bit: C
CAT	EGORY	RATE ELEMENTS	Interi m	Zone	E	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
								 	Nonrec	urring	Nonrecurring	Disconnect			066	Rates(\$)		
	_							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Tolonh	lone Number/Trunk Group Establisment Charges		1	1				FIISL	Auu i	LIISI	Add I	SOWIEC	SUMAN	SOWAN	SOWAN	SOWAN	SUMAN
	relepii	DID Trunk Termination (One Per Port)		1	UEPPX		NDT	0.00	0.00	0.00			1	15.20				
		Additional DID Numbers for each Group of 20 DID Numbers		1	UEPPX		ND4	0.00	0.00	0.00			1	15.20				-
		DID Numbers, Non- consecutive DID Numbers , Per Number		1	UEPPX		ND5	0.00	0.00	0.00			1	15.20				
		Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00				15.20				-
		Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00				15.20				
	LOCAL	NUMBER PORTABILITY			OL: IX			0.00	0.00	0.00				10.20				
		Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00								
	2-WIRE	ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	POR														
		ort/Loop Combination Rates																
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB	UEPPR		84.09										
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		_														
		UNE Zone 2 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	-	2	UEPPB	UEPPR		96.95										
		UNE Zone 3		3	UEPPB	UEPPR		127.60										
	UNE L	oop Rates																
		2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	19.09						15.20				
		2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	31.95						15.20				
		2-Wire ISDN Digital Grade Loop - ONE Zone 2 2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	62.60						15.20				—
	LINE P	ort Rate		3	OLITE	OLITIK	OOLZX	02.00						13.20				-
	O.V.E.	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	65.00	525.00	400.00				15.20				
	NONR	ECURRING CHARGES - CURRENTLY COMBINED																
		2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																
		Combination - Conversion - Top 8 MSAs only			UEPPB	UEPPR	USACB	0.00	230.00	230.00				15.20				
	ADDIT	IONAL NRCs																
	LOCAL	NUMBER PORTABILITY																
		Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
	B-CHA	NNEL USER PROFILE ACCESS:																
		CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
		CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
		CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
	В-СНА	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C,MS, &	i IN)														.
		CVS/CSD (DMS/5ESS) CVS (EWSD)		-	UEPPB	UEPPR	U1UCD U1UCE	0.00	0.00 0.00	0.00								
		CSD		1	UEPPB UEPPB	UEPPR UEPPR	U1UCF	0.00	0.00	0.00			1					1
	LISER	TERMINAL PROFILE		1	OLFFB	ULFFR	01001	0.00	0.00	0.00			1					1
	OOLIK	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
	VERTI	CAL FEATURES																
		All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00				15.20				
	INTER	OFFICE CHANNEL MILEAGE																
		Interoffice Channel mileage each, including first mile and																
		facilities termination			UEPPB	UEPPR	M1GNC	22.613	39.36	26.62				15.20				
		Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.013	0.00	0.00				15.20				
		E DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	(PORT															
	UNE P	ort/Loop Combination Rates																
		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP			935.70										İ
		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1	Ė														
		Zone 2 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		2	UEPPP			1,044.96										
		Zone 3		3	UEPPP			1,341.94										1
	UNE L	oop Rates																
		4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	85.70						15.20				
		4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	194.96						15.20				
		4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	491.94						15.20				
	UNE P	ort Rate																
		Exchange Ports - 4-Wire ISDN DS1 Port		1	UEPPP		UEPPP	850.00	1,150.00	1,150.00			<u> </u>	15.20				<u></u>

ONRONDLED N	IETWORK ELEMENTS - Louisiana	,		,										ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Б	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NONRECU	RRING CHARGES - CURRENTLY COMBINED															
	Vire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															
	mbination - Conversion -Switch-As-Is Top 8 MSAs only			UEPPP	USACP	0.00	950.00	950.00				15.20				
ADDITION			1													
	Vire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-		1													
	vard/two way Telephone Numbers (except NC)			UEPPP	PR7TF		0.48					15.20				
	Vire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -		1	02			0.10					10.20				
	itward Tel Numbers (All States except NC)			UEPPP	PR7TO		11.18	11.18				15.20				
	Vire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -			OLITI	11010		11.10	11.10				10.20				†
	bsequent Inward Telephone Numbers			UEPPP	PR7ZT		22.35	22.35				15.20				
	JMBER PORTABILITY		 	OLITI	11(12)		22.00	22.00			<u> </u>	13.20				+
	cal Number Portability (1 per port)	1	 	UEPPP	LNPCN	1.75					1			 	 	
	E (Provsioning Only)	1	1	OLITE	LINFOIN	1.75					1	-	-	-		+
	ice/Data	 	 	UEPPP	PR71V	0.00	0.00	0.00	1		1	-			 	
	gital Data	 	 	UEPPP	PR71D	0.00	0.00	0.00	1		1	-			 	
	gitai Data vard Data	<u> </u>	 	UEPPP	PR71D PR71E	0.00	0.00	0.00	-		 				-	
		 	1	UEPPP	PK/1E	0.00	0.00	0.00			1			-	 	
	ditional "B" Channel		<u> </u>	LIEDDD	DDZD) (0.00	4444					45.00				4
	w or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.11					15.20				
	w or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.11					15.20				
	w or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.11					15.20				
CALL TYPI																
	vard			UEPPP	PR7C1	0.00	0.00	0.00								1
	itward			UEPPP	PR7C0	0.00	0.00	0.00								1
	o-way			UEPPP	PR7CC	0.00	0.00	0.00								
	Channel Mileage															
	ed Each Including First Mile			UEPPP	1LN1A	70.7532	86.69	79.44				15.20				
	ch Airline-Fractional Additional Mile			UEPPP	1LN1B	0.2652										
	31 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
	Loop Combination Rates															
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		154.17						15.20				
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		263.43						15.20				
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		560.41						15.20				
UNE Loop																ĺ
	Vire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	85.70						15.20				ĺ
4-V	Vire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	194.96						15.20				ĺ
4-V	Vire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	491.94						15.20				
UNE Port F	Rate															
4-V	Vire DDITS Digital Trunk Port			UEPDC	UDD1T	750.00	1,006.28	479.28	0.00	0.00		15.20				1
NONRECU	RRING CHARGES - CURRENTLY COMBINED						,									
4-V	Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	witch-As-Is Top 8 MSAs only			UEPDC	USAC4		125.75	65.08				15.20				
	,															
4-V	Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	Conversion with DS1 Changes Top 8 MSAs only			UEPDC	USAWA		125.75	65.08				15.20				
	onteresent man per entangee rep e mere entry			02. 50	00/11//		120.70	00.00				10.20				+
4-W	Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	Conversion with Change - Trunk Top 8 MSAs only			UEPDC	USAWB		125.75	65.08				15.20				
ADDITION			 	OLI DO	OOMAND		120.70	00.00			<u> </u>	10.20				+
	Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -	 	 	<u> </u>	+				1		1	1	1	1	t	
	bsequent Channel Activation/Chan - 2-Way Trunk	l	1	UEPDC	UDTTA		14.06	14.06				15.20			1	
	Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent	1	1	OLFDO	ODITA		14.00	14.06	1		1	15.20	1	1	 	
	annel Activation/Chan - 1-Way Outward Trunk	l	1	UEPDC	UDTTB		14.06	14.06				15.20		l	I	
		1	<u> </u>	UEPUC	חחוות		14.06	14.06	1		1	15.20	-	1	 	
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel	l	1	LIEDDO	LIDTTO		44.00	44.00			I	45.00		Ì	I	
	tivation/Chan Inward Trunk w/out DID		!	UEPDC	UDTTC		14.06	14.06			1	15.20			1	
	Nire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan	l	1	LIEBBO	LIDTER							4-0-		l	I	
	tivation Per Chan - Inward Trunk with DID	<u> </u>	ļ	UEPDC	UDTTD		14.06	14.06				15.20	1		-	
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan	l	1												1	
	tivation / Chan - 2-Way DID w User Trans	<u> </u>	<u> </u>	UEPDC	UDTTE		14.06	14.06			ļ	15.20			ļ	ļ
IRIPOLAR S	8 ZERO SUBSTITUTION	l	1	1							<u> </u>					1

ONRONDE	ED NETWORK ELEMENTS - Louisiana			,										nent: 2		bit: C
											Svc Order Submitted Elec	Svc Order Submitted Manually	Incremental Charge - Manual Svc	Incremental Charge - Manual Svc	Incremental Charge - Manual Svc	Incrementa Charge - Manual Sv
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Order vs. Electronic- Disc 1st	Order vs. Electronic Disc Add
							Nonrec	urring	Nonrecurring Dis	sconnect			oss	Rates(\$)	1	1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	605.00				15.20				
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	605.00				15.20				1
Alteri	nate Mark Inversion															1
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Telep	hone Number/Trunk Group Establisment Charges															
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00						15.20				
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00						15.20				
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00						15.20				
	DID Numbers, Establish Trunk Group and Provide First Group															
	of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00	ļ <u> </u>			15.20			ļ	ļ
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00						15.20			1	<u> </u>
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00						15.20			.	
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				15.20			.	
	Reserve DID Numbers		<u> </u>	UEPDC	NDV	0.00	0.00	0.00	 			15.20			-	
	cated DS1 (Interoffice Channel Mileage) -															
FX/F0	CO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port															
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)			UEPDC	1LNO1	70.47	86.69	79.44				15.20				
	lateration Channel Mileson Additional rate and mile O Carillan			LIEDDO	41.000	0.0050	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities			UEPDC	1LNOA	0.2652	0.00	0.00								
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25			UEPDC	ILNO2	0.00	0.00	0.00								
	miles			UEPDC	1LNOB	0.2652	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities			UEPDC	ILNOB	0.2652	0.00	0.00								
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.2652	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								
	Central Office Termininating Point			UEPDC	CTG	0.00	0.00	0.00	+	•						†
4-WIF	RE DS1 LOOP WITH CHANNELIZATION WITH PORT			OLI DO	010	0.00										1
	em is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti	vations														†
	stem can have various rate combinations based on type and nur			used												
	DS1 Loop															
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	85.70	0.00	0.00				15.20				
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	194.96	0.00	0.00				15.20				
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	491.94	0.00	0.00				15.20				
UNE	DSO Channelization Capacities (D4 Channel Bank Configuration	1s)														
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	97.35	0.00	0.00				15.20				
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	194.70	0.00	0.00				15.20				
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	389.40	0.00	0.00				15.20				
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	584.10	0.00	0.00				15.20				
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	778.80	0.00	0.00				15.20				
	240 DS0 Channel Capacity - 1 per 10 DS1s		<u> </u>	UEPMG	VUM20	973.50	0.00	0.00				15.20				ļ
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,168.20	0.00	0.00				15.20				ļ
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,557.60	0.00	0.00				15.20				
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	1,947.00	0.00	0.00				15.20			-	<u> </u>
	576 DS0 Channel Capacity -1 per 24 DS1s		<u> </u>	UEPMG	VUM57	2,336.40	0.00	0.00	 			15.20			!	
	672 DS0 Channel Capacity - 1 per 28 DS1s	Ob	L	UEPMG	VUM67	2,725.80	0.00	0.00				15.20			1	
	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with						stern								 	
	nimum System configuration is One (1) DS1, One (1) D4 Channe														 	
Multi	ples of this configuration functioning as one are considered Ac	u i arte	tne m	ıınımum system co T	miguration is	counted.			 						 	
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only		1	UEPMG	USAC4	0.00	450.00	50.00				15.20			I	
Cueta	Bell South Allowed Changes - 10p 8 MSAS Only em Additions Where Currently Combined and New (Not Currently	v Comb	inca'	UEPIVIG	USAC4	0.00	450.00	50.00	 			15.20			-	-
	em Additions where currently combined and New (Not Currentle ensity Zone 1 Top 8 MSAs	y comb	inea)		+				 							
in De	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc		-		+				 							
																•

UNBUNDL	ED NETWORK ELEMENTS - Louisiana												Attachr	nent: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
						1	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		<u> </u>
—						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
Bipo	lar 8 Zero Substitution															
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	605.00				15.20				
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	605.00				15.20				
Alter	nate Mark Inversion (AMI)			OLI WO	CCOLI	0.00	0.00	003.00				13.20				
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	ange Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
Exch	ange Ports								-							
1 1	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	14.00	0.00	0.00				15.20				
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	14.00	0.00	0.00	1			15.20				
						İ	İ									
	Line Side Inward Only Channelized PBX Trunk Port without DID		<u> </u>	UEPPX	UEP1X	14.00	0.00	0.00	ļl			15.20				
Fast	2-Wire Trunk Side Unbundled Channelized DID Trunk Port Ire Activations - Unbundled Loop Concentration			UEPPX	UEPDM	36.00	0.00	0.00				15.20				
reatt	Feature (Service) Activation for each Line Side Port Terminated															-
	in D4 Bank			UEPPX	1PQWM	0.6497	40.00	20.00				15.20				
	Feature (Service) Activation for each Trunk Side Port Terminated															
	in D4 Bank			UEPPX	1PQWU	0.6497	110.00	30.00				15.20				
Telep	phone Number/ Group Establishment Charges for DID Service			LIEBBY .	LIDT							45.00				
-	DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States			UEPPX UEPPX	NDT ND4	0.00	0.00	0.00				15.20 15.20				
-	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00	1			15.20				1
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00	1			15.20				
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				15.20				
Loca	Number Portability															
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	FURES - Vertical and Optional I Switching Features Offered with Line Side Ports Only								-							
Loca	All Features Available			UEPPX	UEPVF	0.00	0.00	0.00				15.20				
UNBUNDLED	CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES	S			_											
	st Based Rates are applied where BellSouth is required by FCC															
	atures shall apply to the Unbundled Port/Loop Combination - C															
	d Office and Tandem Switching Usage and Common Transport														A 1 120 1 NE	1
	e first and additional Port nonrecurring charges apply to Not Co , also and are categorized accordingly.	urrentiy	Comp	inea Combos. For	Currently Co	mbinea Combo	s, the nonrect	irring charges	snall be those	identified in t	ne Nonrecu	rring - Curre	ently Combine	ed sections.	Additional NF	kCs may
	arket Rates for Unbundled Centrex Port/Loop Combination will	be neg	otiated	on an Individual Ca	se Basis, unt	til further notice	a. I		1						I	
UNE-	P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only		L													
2-Wii	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE	Port/Loop Combination Rates (Non-Design)		ļ													
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	1	1	UEP91		13.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	OLFSI		13.13			 		 					
1 1	Non-Design		2	UEP91		23.75										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			-												
	Non-Design		3	UEP91		49.62										
UNE	Port/Loop Combination Rates (Design)								 							
1 1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP91		16.29			1							
 	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		-	OLI 31		10.29			 							
	Design		2	UEP91		26.71										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP91		48.26			ļl							
UNE	Loop Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	11.77					-					ļ
 	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP91 UEP91	UECS1	22.39			 		1					
	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP91	UECS1	48.26										
	,	•	<u> </u>	<u> </u>		,			·				·	·		

NBUNDLE	D NETWORK ELEMENTS - Louisiana			1	-						0	06	Attachr			oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 2) - Zone 1			UEP91	UECS2	14.93										
	2-Wire Voice Grade Loop (SL 2) - Zone 2			UEP91	UECS2	25.35										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	50.46										
UNE Po																
	es (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			UEP91	UEPYA	1.36	38.85	19.08	-			15.20				
	Area			UEP91	UEPYB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP91	UEPYH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP91	UEPYM	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP91	UEPYZ	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP91	UEPY9	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP91	UEPY2	1.36	38.85	19.08				15.20				
	, LA, MS, & TN Only			02. 0.	022	1.00	00.00	10.00				10.20				
,,	2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPQB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPQM	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP91	UEPQZ	1.36	104.41	67.93				15.20				
				UEP91	UEPQ9	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ9	1.36	38.85	19.08	-			15.20				
	Switching	-	-	OLF91	ULFQZ	1.30	30.03	19.00				13.20				
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.8577										
	lumber Portability			OLI OI	OILEGO	0.0077										
	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Feature																
	All Standard Features Offered, per port			UEP91	UEPVF	0.00										
	All Select Features Offered, per port			UEP91	UEPVS	0.00	412.25					15.20				
	All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00										
NARS																
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00				15.20				
	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00				15.20				
	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				15.20				
	aneous Terminations															
	Trunk Side			LIEDO4	OFNIAO	0.00	115.05	10.00				45.00				
	Trunk Side Terminations, each			UEP91	CENA6	8.29	115.85	18.20				15.20				
	ice Channel Mileage - 2-Wire Interoffice Channel Facilities Termination - Voice Grade	 	-	UEP91	M1GBC	22.60	39.36	26.62	 			15.20				-
	Interoffice Channel mileage, per mile or fraction of mile	1		UEP91	M1GBC M1GBM	0.013	38.30	20.02			1	15.20				
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e	 	021 31	IVITODIVI	0.013			+							
	nnel Bank Feature Activations	Ī	l		+				 							
- 3	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP91	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP91	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.6497						15.20				

UNBII	NDLF	D NETWORK ELEMENTS - Louisiana												Attach	ment: 2	Fyhil	oit: C
	·											Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec	Manually		Manual Svc		Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							B	Nonrec	urring	Nonrecurring I	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
		Slot			UEP91	1PQWQ	0.6497						15.20				
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.6497						15.20				
	Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
		Conversion - Currently Combined Switch-As-Is with allowed															
		changes, per port			UEP91	USAC2		0.10	0.10				15.20				
		Conversion of Existing Centrex Common Block			UEP91	USACN	0.00	36.66	16.10								
		New Centrex Standard Common Block			UEP91	M1ACS	0.00	680.40					15.20				
		New Centrex Customized Common Block			UEP91	M1ACC	0.00	680.40					15.20				
		Secondary Block, per Block			UEP91	M2CC1	0.00	79.31					15.20				
		NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	73.93					15.20				
		CENTREX - 5ESS (Valid in All States)															
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo		<u></u>		<u> </u>											
	UNE P	ort/Loop Combination Rates (Non-Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-[
L !		Non-Design	<u>L</u>	1	UEP95	[13.13			<u>1 </u>		<u></u>	<u> </u>	<u></u>	<u> </u>	<u> </u>	<u> </u>
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design		2	UEP95		23.75										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design		3	UEP95		49.62										
	UNE P	ort/Loop Combination Rates (Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
		Design		1	UEP95		16.29										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Design		2	UEP95		26.71										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Design		3	UEP95		51.82										
	UNE L	oop Rate															
		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	11.77										
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	22.39										
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	48.26										
		2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	14.93										
		2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	25.35										
		2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	50.46										
		ort Rate															
	All Stat							, and the second			·						
		2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.36	38.85	19.08				15.20				
		2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.36	38.85	19.08				15.20				
		2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local									·				1	1	
		Area			UEP95	UEPYH	1.36	38.85	19.08				15.20				l
		2-Wire Voice Grade Port (Centrex from diff Serving Wire		1						1 T		1	<u> </u>	<u> </u>	_	_	1
		Center)2 Basic Local Area			UEP95	UEPYM	1.36	104.41	67.93				15.20				
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service		1								1					1
		Term - Basic Local Area			UEP95	UEPYZ	1.36	104.41	67.93				15.20				
		2-Wire Voice Grade Port terminated in on Megalink or equivalent		1								1					1
		- Basic Local Area			UEP95	UEPY9	1.36	38.85	19.08				15.20				
		2-Wire Voice Grade Port Terminated on 800 Service Term -		1						1 T		1	<u> </u>	<u> </u>	_	_	1
		Basic Local Area			UEP95	UEPY2	1.36	38.85	19.08				15.20	ļ			l
	AL, KY	, LA, MS, SC, & TN Only															
		2-Wire Voice Grade Port (Centrex)		<u> </u>	UEP95	UEPQA	1.36	38.85	19.08	1			15.20		1	1	
		2-Wire Voice Grade Port (Centrex 800 termination)	1	<u> </u>	UEP95	UEPQB	1.36	38.85	19.08	1			15.20		ļ	ļ	
		2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.36	38.85	19.08				15.20	ļ			l
		2-Wire Voice Grade Port (Centrex from diff Serving Wire			l					1					1	1	
		Center)2	1	<u> </u>	UEP95	UEPQM	1.36	104.41	67.93	1			15.20		ļ	ļ	
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			l	[]						1		1	I	I	
		Term	1	<u> </u>	UEP95	UEPQZ	1.36	104.41	67.93	ļ			15.20		ļ	ļ	
					l	[]						1		1	I	I	
		2-Wire Voice Grade Port terminated in on Megalink or equivalent	1	<u> </u>	UEP95	UEPQ9	1.36	38.85	19.08	1			15.20		ļ	ļ	
1		2-Wire Voice Grade Port Terminated on 800 Service Term	1		UEP95	UEPQ2	1.36	38.85	19.08				15.20				

NRUNDI	LED NETWORK ELEMENTS - Louisiana													nent: 2		oit: C
ATEGORY	Y RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
		+			+		Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
		_			+	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOM AN	SOMAN	SOMAN
Loca	al Switching						11130	Addi	11130	Auu	COMILO	COMPAR	COMPAR	COMPAR	COMPAN	COMPAR
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.8577						15.20				
Loca	al Number Portability			02. 00	0.1200	0.0077						10.20				
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Feat	tures															
	All Standard Features Offered, per port			UEP95	UEPVF	0.00						15.20				
	All Select Features Offered, per port			UEP95	UEPVS	0.00	412.25					15.20				
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00						15.20				
NAR	RS															
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				15.20				
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				15.20				
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				15.20				
	cellaneous Terminations															
2-W	fire Trunk Side															
	Trunk Side Terminations, each			UEP95	CEND6	8.29	115.85	18.20				15.20				
4-Wi	fire Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP95	M1HD1	68.47	196.18	92.92				15.20				
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	14.06					15.20				
Inte	roffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP95	MIGBC	22.60	39.36	26.62				15.20				
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.013										
Feat	ture Activations (DS0) Centrex Loops on Channelized DS1 Serv	ice														
D4 C	Channel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP95	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP95	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP95	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.6497						15.20				
Non	n-Recurring Charges (NRC) Associated with UNE-P Centrex		<u> </u>	ļ	1											
1	NRC Conversion Currently Combined Switch-As-Is with allowed														Ì	
	changes, per port		<u> </u>	UEP95	USAC2		0.10	0.10				15.20				
	Conversion of Existing Centrex Common Block, each		<u> </u>	UEP95	USACN		36.66	16.10				15.20			ļ	
	New Centrex Standard Common Block		<u> </u>	UEP95	M1ACS	0.00	680.40					15.20				
	New Centrex Customized Common Block		<u> </u>	UEP95	M1ACC	0.00	680.40					15.20			ļ	
	NAR Establishment Charge, Per Occasion		<u> </u>	UEP95	URECA	0.00	73.93					15.20				ļ
	E-P CENTREX - DMS100 (Valid in All States)		<u> </u>												ļ	
	/ire VG Loop/2-Wire Voice Grade Port (Centrex) Combo		<u> </u>													
UNE	E Port/Loop Combination Rates (Non-Design)		<u> </u>												ļ	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	'-													Ì	
_	Non-Design		1	UEP9D		13.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	-	2	LIEDOD		00.7-									Ì	
_	Non-Design	+	- 2	UEP9D	+	23.75			 						1	
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design	-	3	UEP9D		49.62									Ì	
LINE		+	3	UEP9D	+	49.62										
UNE	E Port/Loop Combination Rates (Design)	+	!	 	+				 						1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	' 1	4	UEP9D		16.00									1	1
+	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	-		OFLAD	+	16.29			 						-	-
	Design 2-wire voice Grade Port (Centrex)Port Combo	-	2	UEP9D		26.74										
-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	+		OFLAD	+	26.71			 						 	
	Design 2-wire voice Grade Port (Centrex)Port Combo	1	3	UEP9D		51.82									1	1
	IDESIGN	1	1 3	ULLAD	1	51.62			1		1				1	i

UNBUNDI	.ED NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhib	oit: C
CIADOIADE	LE TELLITORIA ELEMENTO LOGISTATIO										Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
											Elec		Manual Svc			Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						.,,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
														Add'l		
													1st		Disc 1st	Disc Add'l
						Rec	Nonred	urring	Nonrecurring	Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	22.39										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	48.26										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	14.93										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	25.35										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	50.46										
	Port Rate															
ALL	STATES															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.36	38.85	19.08				15.20				
1 1	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local		1									1		I		
\vdash	Area	<u> </u>	<u> </u>	UEP9D	UEPYB	1.36	38.85	19.08				15.20		ļ		
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local		1	l	1							l		I		
	Area			UEP9D	UEPYC	1.36	38.85	19.08				15.20				
1 1	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local		1									1		I		
	Area			UEP9D	UEPYD	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local		1]			1		I		
	Area		1	UEP9D	UEPYE	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local															
	Area			UEP9D	UEPYF	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local															
	Area			UEP9D	UEPYG	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															
	Area			UEP9D	UEPYT	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local															
	Area			UEP9D	UEPYU	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local															
	Area			UEP9D	UEPYV	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local															
	Area			UEP9D	UEPY3	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local															
	Area			UEP9D	UEPYH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3															
	Basic Local Area			UEP9D	UEPYJ	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)		1	l	1							l		I		
	2 Basic Local Area	ļ	 	UEP9D	UEPYM	1.36	104.41	67.93	ļ			15.20		.		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3		1											1		
\vdash	Basic Local Area	ļ	 	UEP9D	UEPYO	1.36	104.41	67.93	ļ			15.20		.		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3]					I		
\vdash	Basic Local Area	1	<u> </u>	UEP9D	UEPYP	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3											4= 00				
\vdash	Basic Local Area	1	<u> </u>	UEP9D	UEPYQ	1.36	104.41	67.93	 			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3		1	LIEDOD	LIEDY'S							4-0-		1		
\vdash	Basic Local Area	 	<u> </u>	UEP9D	UEPYR	1.36	104.41	67.93				15.20		-		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3		1	LIEDOD	LIEDVO	4.00	404 **	07.00				45.00		I		
\vdash	Basic Local Area	 	<u> </u>	UEP9D	UEPYS	1.36	104.41	67.93				15.20		-		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3		1	LIEDOD	LIEDY4	4.00	404.41	07.00]			45.00		I		
	Basic Local Area	 		UEP9D	UEPY4	1.36	104.41	67.93				15.20				
1 1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3		1	LIEDOD	LIEDY'S	4.00	404.41	07.00				45.00		1		
	Basic Local Area	 	1	UEP9D	UEPY5	1.36	104.41	67.93				15.20		 		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3		1	LIEDOD	LIEDVO	4.00	404.41	07.00				45.00		1		
\vdash	Basic Local Area	1	<u> </u>	UEP9D	UEPY6	1.36	104.41	67.93	 			15.20				
1 1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3		1	LIEDOD	LIEDY (=							4-0-		1		
\vdash	Basic Local Area	 	<u> </u>	UEP9D	UEPY7	1.36	104.41	67.93				15.20		-		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			LIEDOD	LIEDYZ	4.00	404.41	07.00]			45.00		I		
\vdash	Term		<u> </u>	UEP9D	UEPYZ	1.36	104.41	67.93				15.20		-		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1	1	UEP9D	UEPY9	1.36	38.85	19.08			l	15.20				
	Basic Local Area	<u> </u>	<u> </u>	UEF9D	UEPTS	1.36	38.85	19.08			1	15.20		l		

UNBUNDL	ED NETWORK ELEMENTS - Louisiana			,										ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increments Charge - Manual Sv Order vs. Electronic Disc Add
							N		T 81	B'						
						Rec	Nonrec		Nonrecurring		001150			Rates(\$)		
			<u> </u>				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic															
	Local Area			UEP9D	UEPY2	1.36	38.85	19.08				15.20				
AL, I	(Y, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3	1	t	UEP9D	UEPQU	1.36	38.85	19.08	1			15.20		1	Ì	1
 	2-Wire Voice Grade Fort (Centrex / EBS-M5236)3	-	!	UEP9D	UEPQV	1.36	38.85	19.08	 			15.20		 	 	1
 	2-Wire Voice Grade Port (Centrex / EBS-M5316)3	 	 	UEP9D	UEPQ3	1.36	38.85	19.08				15.20	1	1	1	1
	2-Wire Voice Grade Port (Centrex vith Caller ID)		1	UEP9D	UEPQH	1.36	38.85	19.08				15.20				
		-	!	OFLAD	UEPUN	1.30	30.83	19.08	-		 	13.20	-	-	1	
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp	l	1	UEP9D	UEPQW	4.00	20.05	40.00			l	45.00				
	Indication)3					1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPQM	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.36	104.41	67.93				15.20				
	·															
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	1.36	104.41	67.93				15.20				
			1													
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.36	104.41	67.93				15.20				
	2-Wile Voice Grade Fort (Certifex diller SWC /LB3-W5112)2, 3			OLFBD	ULFQK	1.30	104.41	07.53				13.20				1
	2 Mins Vains Canda Bart (Cantana) differ CMIC (EBC ME212)2 2			UEP9D	UEPQS	1.36	404.44	67.93				45.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.30	104.41	67.93				15.20				ļ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.36	104.41	67.93				15.20				
	, ,															
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3	l	1	UEP9D	UEPQ7	1.36	104.41	67.93			l	15.20				
 	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1	1		J Q,	1.00	104.41	07.00				10.20		†	1	I
	Term	l	1	UEP9D	UEPQZ	1.36	104.41	67.93]			15.20		Ì		
 	Tomi	1	1	OLI 3D	טבו עב	1.30	104.41	01.33	1			13.20	1	 	1	
	2 Wire Voice Crade Port terminated in an Manalinia as a state	l	1	UEP9D	LIEDOO	4.00	20.05	40.00]			45.00		Ì		
—	2-Wire Voice Grade Port terminated in on Megalink or equivalent	 	1		UEPQ9	1.36	38.85	19.08	 			15.20		 	1	1
⊢ .	2-Wire Voice Grade Port Terminated on 800 Service Term		1	UEP9D	UEPQ2	1.36	38.85	19.08				15.20	1		1	1
Loca	l Switching															
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.8577										
Loca	l Number Portability															
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										
Feat	ures															
	All Standard Features Offered, per port			UEP9D	UEPVF	0.00						15.20				
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	412.25		i i			15.20				
<u> </u>	All Centrex Control Features Offered, per port		1	UEP9D	UEPVC	0.00			i i			15.20	İ	İ	Ì	1
NAR			1		7=	2.00			1		1			1	1	1
THE STATE OF THE S	Unbundled Network Access Register - Combination	l	1	UEP9D	UARCX	0.00	0.00	0.00			1	15.20	1			1
 	Unbundled Network Access Register - Combination Unbundled Network Access Register - Inward	 	 	UEP9D	UAR1X	0.00	0.00	0.00	 		 	15.20	 	 	1	1
	Unbundled Network Access Register - Inward Unbundled Network Access Register - Outdial	 	 	UEP9D	UAROX	0.00	0.00	0.00				15.20			 	
			1	05790	UARUX	0.00	0.00	0.00				15.20	-		1	1
	ellaneous Terminations	<u> </u>	<u> </u>									ļ			ļ	
2-Wi	re Trunk Side		<u> </u>													
	Trunk Side Terminations, each			UEP9D	CEND6	8.29	115.85	18.20				15.20				
4-Wi	re Digital (1.544 Megabits)		<u> </u>								<u> </u>					
	DS1 Circuit Terminations, each			UEP9D	M1HD1	68.47	196.18	98.62		-		15.20				
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	14.06					15.20				

INBUNDLE	D NETWORK ELEMENTS - Louisiana													ment: 2	1	oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Nonrec	urring	Nonrecurring	g Disconnect				Rates(\$)	2.00	2.007.007
					1	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Interof	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	22.60	39.36	26.62				15.20				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.013										
	e Activations (DS0) Centrex Loops on Channelized DS1 Servic	е														
D4 Cha	annel Bank Feature Activations				100110	0.010=						1= 00				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP9D	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP9D	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tijle Line/Trunk Loop		 	OLI 3D	11 02 44 4	0.0437			1	1	 	13.20			1	
	Slot		l	UEP9D	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.6497			1	1		15.20				
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		0.10	0.10				15.20				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		36.66	16.10				15.20				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	680.40					15.20				
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	680.40					15.20				
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	73.93					15.20				
	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE P	ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo-															
	Non-Design		1	UEP9E		13.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLFBL	+	13.13										
	Non-Design		2	UEP9E		23.75										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_	02. 02		20.10										
	Non-Design		3	UEP9E		49.62										
UNE P	ort/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9E		16.29										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			l	1 7	\exists]						1
	Design		2	UEP9E		26.71					ļ					
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo		_	LIEDOE	1 1	54.00										
LINE !	Design oop Rate		3	UEP9E	+	51.82					 				-	
ONE L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	11.77			1	1	1				1	
-	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	22.39			1	1	 			1	1	
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	48.26					 				1	
	2-Wire Voice Grade Loop (SL 2) - Zone 1			UEP9E	UECS2	14.93										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	25.35			İ	İ				İ		
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	50.46										
	ort Rate						_	•	_							
AL, FL	, KY, LA, MS, & TN only			L					ļ	ļ						
	2-Wire Voice Grade Port (Centrex) Basic Local Area		 	UEP9E	UEPYA	1.36	38.85	19.08		ļ	<u> </u>	15.20			ļ	
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9E	UEPYB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local		1	J_1 J_	02.10	1.50	30.03	13.00			 	10.20			1	
	Area		1	UEP9E	UEPYH	1.36	38.85	19.08				15.20				1
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP9E	UEPYM	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
1	Term - Basic Local Area	l	l	UEP9E	UEPYZ	1.36	104.41	67.93				15.20]]

NRONDF	ED NETWORK ELEMENTS - Louisiana			ı	•						T -			nent: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						_	Nonrec	urring	Nonrecurring	Disconnect		1	oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP9E	UEPY9	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP9E	UEPY2	1.36	38.85	19.08				15.20				
AL. K	(Y, LA, MS, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP9E	UEPQM	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9E	UEPQZ	1.36	104.41	67.93				15.20			I	I
				İ	1				İ						İ	1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	1.36	38.85	19.08				15.20			I	I
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.36	38.85	19.08			İ	15.20			1	
Loca	Switching				1 1	50	22.00				İ				1	
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.8577										
Loca	I Number Portability															
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Featu																
· out	All Standard Features Offered, per port			UEP9E	UEPVF	0.00						15.20				
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	412.25					15.20				
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00						15.20				-
NARS				02. 02	02. 70	0.00						10.20				-
10.111	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00								-
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00								-
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00								-
Misc	ellaneous Terminations			02. 02	07.11.071	0.00	0.00	0.00								-
	re Trunk Side															
	Trunk Side Terminations, each			UEP9E	CEND6	8.29	115.85	18.20				15.20				-
4-Wii	re Digital (1.544 Megabits)			OLI OL	OLINDO	0.20	110.00	10.20				10.20				-
7 ****	DS1 Circuit Terminations, each			UEP9E	M1HD1	68.47	196.18	92.92				15.20				
-	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	14.06	02.02				15.20				1
Inter	office Channel Mileage - 2-Wire			OLI OL	WITTIBO	0.00	14.00					10.20				
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	22.60	39.36	26.62				15.20				1
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.013	00.00	20.02				10.20				-
Feati	ure Activations (DS0) Centrex Loops on Channelized DS1 Service	e		OLI OL	WIIGEWI	0.010										
	hannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.6497						15.20			t	
						2.2.07									1	1
J	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.6497						15.20			I	I
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop					3.5.57						.0.20			1	1
J	Slot			UEP9E	1PQW7	0.6497						15.20			I	I
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -				1	3.5.57						.0.20			<u> </u>	<u> </u>
1	Different Wire Center			UEP9E	1PQWP	0.6497						15.20			I	
					~,,,	3.0-137						10.20			t	t
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.6497						15.20			I	
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop				1										t	1
	Slot			UEP9E	1PQWQ	0.6497						15.20			1	
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.6497						15.20			t	
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex					3.0.07						.0.20			t	
- 1.3	NRC Conversion Currently Combined Switch-As-Is with allowed				<u> </u>						İ				1	
	changes, per port			UEP9E	USAC2		0.10	0.10				15.20			I	
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		36.66	16.10			i	15.20			1	1
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	680.40					15.20			t	
	New Centrex Standard Common Block			UEP9E	M1ACC	0.00	680.40					15.20			t	
-	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	73.93				-	15.20			—	
IINE	P CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)			J_1 JL	UNLUA	0.00	10.00				1	15.20			t	
OITE-	re VG Loop/2-Wire Voice Grade Port (Centrex) Combo														1	

DNROND	LED NE	TWORK ELEMENTS - Louisiana			•										nent: 2		bit: C
ATEGORY	′	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increments Charge - Manual Sv Order vs. Electronic Disc Add
	-						1	Nonros	rrina	Monroourring	n Diagonnoot			000	Potoc/\$\		
						+	Rec	Nonrec		Nonrecurring		COMEC	COMAN		Rates(\$)	COMAN	COMAN
LINIT	- Daw/I a	op Combination Rates (Non-Design)				+		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE		re VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				_											
		Design		1	UEP93		13.13										
		re VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		<u> </u>	OLF 93		13.13										
		Design		2	UEP93		23.75										
		re VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLI 33		25.75										
		Design		3	UEP93		49.62										
UNE		op Combination Rates (Design)		Ŭ	02.00		10.02										
		re VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Desid			1	UEP93		16.29										
	2-Wir	re VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Desig			2	UEP93		26.71									I	
	2-Wir	re VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Desig		<u></u>	3	UEP93	<u> </u>	51.82			<u> </u>	<u></u>					<u> </u>	
UNE	Loop R																
		re Voice Grade Loop (SL 1) - Zone 1		1	UEP93	UECS1	11.77										
		re Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	22.36										
		re Voice Grade Loop (SL 1) - Zone 3			UEP93	UECS1	48.26										
		re Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	14.93										
		re Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	25.35										
		re Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	50.46										
	Port Ra																
AL,		MS, & TN only															
		re Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.36	38.85	19.08				15.20				
		re Voice Grade Port (Centrex 800 termination)Basic Local											4= 00				
	Area				UEP93	UEPYB	1.36	38.85	19.08				15.20				
		re Voice Grade Port (Centrex with Caller ID)1Basic Local			UEP93	UEPYH	1.36	38.85	19.08				45.00				
	Area				UEP93	UEPTH	1.30	38.85	19.08				15.20				
		re Voice Grade Port (Centrex from diff Serving Wire er)2 Basic Local Area			UEP93	UEPYM	1.36	104.41	67.93				15.20				
		re Voice Grade Port, Diff Serving Wire Center - 800 Service			UEF93	UEPTIVI	1.30	104.41	67.93				15.20				
		1 - Basic Local Area			UEP93	UEPYZ	1.36	104.41	67.93				15.20				
-		re Voice Grade Port terminated in on Megalink or equivalent			OLF 93	OLFIZ	1.30	104.41	07.93				13.20				
		sic Local Area			UEP93	UEPY9	1.36	38.85	19.08				15.20				
		re Voice Grade Port Terminated on 800 Service Term -			02.00	020		00.00	.0.00				10.20				
		c Local Area			UEP93	UEPY2	1.36	38.85	19.08				15.20				
		re Voice Grade Port (Centrex)			UEP93	UEPQA	1.36	38.85	19.08				15.20				
		re Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.36	38.85	19.08				15.20				
		re Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.36	38.85	19.08				15.20				
	2-Wir	re Voice Grade Port (Centrex from diff Serving Wire												_			
	Cente	er)2		<u> </u>	UEP93	UEPQM	1.36	104.41	67.93	L	<u></u>		15.20			<u></u>	
		re Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term	1			UEP93	UEPQZ	1.36	104.41	67.93				15.20				
		re Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.36	38.85	19.08	ļ			15.20			.	ļ
		re Voice Grade Port Terminated on 800 Service Term		<u> </u>	UEP93	UEPQ2	1.36	38.85	19.08				15.20				
Loc	al Switch			<u> </u>	LIEDOS	LIDECO	0.0575			—						-	
1		rex Intercom Funtionality, per port er Portability		<u> </u>	UEP93	URECS	0.8577			1						!	1
Loc		er Portability I Number Portability (1 per port)		 	UEP93	LNCCC	0.35			 						 	1
East	Local tures	i Number Foliability (1 per port)	-	 	OFLAS	LINCOL	0.35			<u> </u>						 	}
геа		tandard Features Offered, per port			UEP93	UEPVF	0.00			1			15.20			t	1
		entrex Control Features Offered, per port	-	 	UEP93	UEPVF	0.00			<u> </u>			15.20			 	}
NAF		enties Control / Edities Offered, per port			OL1 33	OLF VO	0.00						13.20			1	1
IVAL		undled Network Access Register - Combination	-		UEP93	UARCX	0.00	0.00	0.00	 			15.20			t	
-+		undled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00				15.20			I	
		undled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00				15.20			1	
Mis		us Terminations		-	- "		0.00	3.50	5.50		1		.0.20			t	1
	ire Trunk					1					l					1	†

ONRONDLED	NETWORK ELEMENTS - Louisiana			ı		1					1.	Ι -		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		Nave	RATES(\$)	Name	a Diagonal		Submitted Manually	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec			g Disconnect				Rates(\$)		
	T 10:1 T 1:1				051100		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Trunk Side Terminations, each			UEP93	CEND6	8.27	115.85	18.20				15.20				
	Digital (1.544 Megabits)			LIEDOO	MALIDA	00.47	100.10	00.00				45.00				
	DS1 Circuit Terminations, each DS0 Channels Activated, Per Channel			UEP93 UEP93	M1HD1 M1HDO	68.47 0.00	196.18 14.06	92.92				15.20 15.20				
	ice Channel Mileage - 2-Wire			UEF93	IVITIDO	0.00	14.00				-	15.20				
	Interoffice Channel Facilities Termination			UEP93	MIGBC	22.60	39.36	26.62		1	1	15.20				
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.013	39.30	20.02		+	1	13.20				
	Activations (DS0) Centrex Loops on Channelized DS1 Service	-		OLI 33	IVIIODIVI	0.013				+	1					
	nnel Bank Feature Activations	ĭ									-					
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.6497					1	15.20				
	r datare / terrament en 2 - terraminer barric dentrex 2005 diet			02.00		0.0101						10.20				
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop					0.0.0										
	Slot			UEP93	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP93	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop															
	Slot			UEP93	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.6497						15.20				
	curring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP93	USAC2		0.10	0.10				15.20				
	Conversion of Existing Centrex Common Block, each			UEP93	USACN		36.66	16.10				15.20				
	New Centrex Standard Common Block			UEP93	M1ACS	0.00	680.40					15.20				
	New Centrex Customized Common Block			UEP93	M1ACC	0.00	680.40					15.20				
	NAR Establishment Charge, Per Occasion			UEP93	URECA	0.00	73.93					15.20				
	Required Port for Centrex Control in 1AESS, 5ESS & EWSD															
	- Requires Interoffice Channel Mileage				-					+	-					
	Requires Specific Customer Premises Equipment ENTREX PORT/LOOP COMBINATIONS - MARKET RATES				-						-					
	et Rates are applied where BellSouth is not required by FCC	and/or	State C	ommission rulo to I	provide Unbu	ndlad Lacal Su	vitching or Swi	tch Dorte			-					
	rring Charges for all Standard Centrex and Centrex Conrol Fe					liuleu Local Sv	viccining or Swi	icii Fuits.		1	1					
	Office and Tandem Switching Usage and Common Transport					ihit shall annly	to all combina	tions of loon/	nort network e	alaments avcer	t for LINE (Coin Port/Lo	on Combinat	ione		
	irst and additional Port nonrecurring charges apply to Not Cu														Additional NE	Cc may
	is and additional Fort homecuring charges apply to Not Co	urrentiy	COIIID	ineu Combos. For	Currently Co	ilibilied Collibe	os, the homecu	iring charges	Silali De lilose	e identined iii i	ille Nolliecu	iring - Curr	entry Combin	eu sections.	Additional N	NOS Illay
	CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)	1			1	1			l			1	ı		ı	1
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo										-					
	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1		İ					1				İ		
	Non-Design		1	UEP91		25.77				I				1		
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	i –		İ				l	1	1		İ		İ	
	Non-Design		2	UEP91		36.39				I				1		
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		i		1											
	Non-Design	<u></u>	3	UEP91	<u> </u>	62.26	<u> </u>		<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u></u>	<u> </u>	<u> </u>
UNE Po	rt/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -							-								
	Design		1	UEP91		28.93				1						
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	1							_				1		
	Design		2	UEP91		39.35				1	1	ļ				
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			l						1						
	Design		3	UEP91	ļ	64.46				1						
	op Rate		<u> </u>		<u> </u>					_				ļ		ļ
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	11.77				_						
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	22.39				1	1			ļ		
	2 Mura Voica Grada Loop (SL 1) Zona 2	i	3	UEP91	UECS1	48.26			ı	1	1	1	I		l	1
	2-Wire Voice Grade Loop (SL 1) - Zone 3		·	LIEDO4							1					
	2-Wire Voice Grade Loop (St. 1) - Zone 1 2-Wire Voice Grade Loop (St. 2) - Zone 1		1	UEP91 UEP91	UECS2 UECS2	14.93 25.35										

NRONDLE	D NETWORK ELEMENTS - Louisiana			1										ment: 2		bit: C
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremen Charge Manual S Order vs Electroni Disc Add
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	50.46										
UNE P																
All Sta	tes (Except North Carolina and Sout Carolina)															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP91	UEPYB	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP91	UEPYH	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP91	UEPYM	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP91	UEPYZ	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP91	UEPY9	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP91	UEPY2	14.00	50.00	25.00				15.20				
AL, K	Y, LA, MS, & TN Only				1	00	22.00									
	2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPQB	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPQM	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP91	UEPQZ	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ9	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ2	14.00	50.00	25.00				15.20				
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.8577										
Local	Number Portability Local Number Portability (1 per port)			UEP91	LNPCC	0.35									-	
Featu				UEP91	LNPCC	0.35					-				-	
reatui	All Standard Features Offered, per port			UEP91	UEPVF	0.00										
	All Select Features Offered, per port			UEP91	UEPVS	0.00	412.25					15.20				
	All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00										
NARS					1		_								1	
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00				15.20				
	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00				15.20		-	1	1
Micco	Unbundled Network Access Register - Outdial Ianeous Terminations			UEP91	UAROX	0.00	0.00	0.00	 		-	15.20			 	
	Trunk Side				+ +				 					1	 	
	Trunk Side Terminations, each			UEP91	CENA6	8.29	115.85	18.20				15.20		1	†	1
Intero	ffice Channel Mileage - 2-Wire				1	5.20			1						1	
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	22.60	39.36	26.62				15.20				
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.013										
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	е					•	•		•						
D4 Ch	annel Bank Feature Activations			L	1		, and the second									<u> </u>
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.6497						15.20				
+-	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP91	1PQW6	0.6497					1	15.20			1	
+	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP91	1PQW7	0.6497						15.20				
_	Different Wire Center			UEP91	1PQWP	0.6497						15.20				
_	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			UEP91	1PQWV	0.6497						15.20			-	
	Slot			UEP91	1PQWQ	0.6497						15.20			1	
1	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.6497			1		1	15.20		1	1	1

UNBU	NDLF	D NETWORK ELEMENTS - Louisiana												Attachi	ment: 2	Exhil	oit: C
												Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted			Charge -	Charge -	Charge -
			Intori									Elec	Manually	Manual Svc	Manual Svc		Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
														151	Add I	DISC ISL	DISC Add I
							Dee	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	•	•
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Non-Re	curring Charges (NRC) Associated with UNE-P Centrex															
		Conversion - Currently Combined Switch-As-Is with allowed															
		changes, per port			UEP91	USAC2		0.10	0.10				15.20				
		Conversion of Existing Centrex Common Block			UEP91	USACN	0.00	36.66	16.10								
		New Centrex Standard Common Block			UEP91	M1ACS	0.00	680.40					15.20				
		New Centrex Customized Common Block			UEP91	M1ACC	0.00	680.40					15.20				
		Secondary Block, per Block			UEP91	M2CC1	0.00	79.31					15.20				
		NAR Establishment Charge, Per Occasion	ļ		UEP91	URECA	0.00	73.93					15.20				
		CENTREX - 5ESS (Valid in All States)	ļ														
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo	1			1				1		}			!	!	
\longmapsto	UNE PO	ort/Loop Combination Rates (Non-Design)	1			+				 		1	-		 	 	
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	1	4	UEP95		25.77								I	I	
\vdash		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1		OEF90	+	25.77			<u> </u>		 				 	
		Non-Design		2	UEP95		36.39								I	I	
\vdash		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1		OE1 30	+	30.39			 		1	-		 	 	
1		Non-Design		3	UEP95		62.26								I	I	
	UNF Po	ort/Loop Combination Rates (Design)			OLI 50		02.20										
	···- · ·	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-														
		Design		1	UEP95		28.93										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Design		2	UEP95		39.35										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Design		3	UEP95		64.46										
	UNE Lo	oop Rate															
		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	11.77										
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	22.39										
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	48.26										
		2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	14.93										
		2-Wire Voice Grade Loop (SL 2) - Zone 2	ļ	2	UEP95	UECS2	25.35										
		2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	50.46										
		ort Rate															
	All Stat				LIEDOE	LIEDVA	44.00	50.00	25.00				45.00				
\vdash		2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA UEPYB	14.00	50.00 50.00	25.00			1	15.20				
-		2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			UEP95	UEPTB	14.00	50.00	25.00				15.20				
		Area			UEP95	UEPYH	14.00	50.00	25.00				15.20		1	1	
\vdash		2-Wire Voice Grade Port (Centrex from diff Serving Wire	1		OLI 33	OLI III	14.00	50.00	25.00	 			13.20		 	 	
1		Center)2 Basic Local Area			UEP95	UEPYM	14.00	135.00	90.00				15.20		I	I	
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service					20		22.30	1					1	1	
		Term - Basic Local Area			UEP95	UEPYZ	14.00	135.00	90.00				15.20		1	1	
		2-Wire Voice Grade Port terminated in on Megalink or equivalent															
		- Basic Local Area	<u> </u>		UEP95	UEPY9	14.00	50.00	25.00	<u> </u>		<u></u>	15.20		<u> </u>	<u> </u>	
		2-Wire Voice Grade Port Terminated on 800 Service Term -															
		Basic Local Area			UEP95	UEPY2	14.00	50.00	25.00				15.20				
	AL, KY	LA, MS, SC, & TN Only													1	1	
\square		2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	14.00	50.00	25.00				15.20		ļ	ļ	
$\vdash \!$		2-Wire Voice Grade Port (Centrex 800 termination)	1		UEP95	UEPQB	14.00	50.00	25.00			<u> </u>	15.20		-	-	
\vdash		2-Wire Voice Grade Port (Centrex with Caller ID)1	1		UEP95	UEPQH	14.00	50.00	25.00			1	15.20		1	1	
		2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	14.00	135.00	90.00				15.20		I	I	
\vdash		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1		OLF 30	ULF QIVI	14.00	135.00	90.00	1		1	15.20		t	t	
		Term			UEP95	UEPQZ	14.00	135.00	90.00				15.20		I	I	
\vdash			1		02. 00	CLI WZ	14.00	100.00	30.00				10.20		-	-	
		2-Wire Voice Grade Port terminated in on Megalink or equivalent	:1		UEP95	UEPQ9	14.00	50.00	25.00				15.20		I	I	
		2-Wire Voice Grade Port Terminated in 61 Megalinik of equivalent			UEP95	UEPQ2	14.00	50.00	25.00	1			15.20		1	1	
		witching	1														
		Centrex Intercom Funtionality, per port	1		UEP95	URECS	0.8577						15.20				
-		lumber Portability															

NRAND	LEC	NETWORK ELEMENTS - Louisiana			•										ment: 2		bit: C
ATEGOR	Y	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	Charge - Manual Sv Order vs. Electronic
														1st	Add'l	Disc 1st	Disc Add'
							Rec	Nonrec		Nonrecurring					Rates(\$)		
					LIEDAE			First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Fac	ature	Local Number Portability (1 per port)		<u> </u>	UEP95	LNPCC	0.35										-
rea		All Standard Features Offered, per port		<u> </u>	UEP95	UEPVF	0.00						15.20				
		All Select Features Offered, per port			UEP95	UEPVS	0.00	412.25					15.20			1	
		All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00						15.20				
NA	RS																
		Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				15.20				
		Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				15.20				
		Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				15.20				
		aneous Terminations															
2-V		Trunk Side	1	-	LIEDOE	CENDS	9.00	115.05	10.00	ļ —		1	15.00		 	1	1
4-V		Trunk Side Terminations, each Digital (1.544 Megabits)	 		UEP95	CEND6	8.29	115.85	18.20	-		-	15.20			 	-
4-1		DS1 Circuit Terminations, each	1	 	UEP95	M1HD1	68.47	196.18	92.92	1		1	15.20		1	 	1
		DS0 Channels Activated, each	 	1	UEP95	M1HDO	0.00	14.06	32.32	 			15.20			 	
Inte		ce Channel Mileage - 2-Wire			OL: 00	MITIEG	0.00	14.00					10.20				
		Interoffice Channel Facilities Termination			UEP95	MIGBC	22.60	39.36	26.62				15.20				
		Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.013										
Fea	ature	Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4		nnel Bank Feature Activations															
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.6497						15.20				
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.6497						15.20				
		Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.6497						15.20				
		Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP95	1PQWP	0.6497						15.20				
		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.6497						15.20				
		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP95	1PQWQ	0.6497						15.20				
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.6497						15.20			1	
No		curring Charges (NRC) Associated with UNE-P Centrex														1	
		NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP95	USAC2		0.10	0.10				15.20				
_		Conversion of Existing Centrex Common Block, each			UEP95	USACN		36.66	16.10				15.20				
		New Centrex Standard Common Block	1		UEP95	M1ACS	0.00	680.40	10.10				15.20				
		New Centrex Customized Common Block			UEP95	M1ACC	0.00	680.40					15.20				
		NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	73.93					15.20				
UN	E-P (CENTREX - DMS100 (Valid in All States)															
		/G Loop/2-Wire Voice Grade Port (Centrex) Combo															
UN		rt/Loop Combination Rates (Non-Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP9D		25.77										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9D		36.39										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9D		62.26										
UN		rt/Loop Combination Rates (Design)			OLI OD		02.20										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP9D		28.93										
		Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9D		39.35										
-+		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -										1					
LIN		Design op Rate		3	UEP9D		64.46					-				-	1
ON		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	11.77			 		 				†	
		2-Wire Voice Grade Loop (SL 1) - Zone 2	<u> </u>	2	UEP9D	UECS1	22.39									1	
-+		2-Wire Voice Grade Loop (SL 1) - Zone 3	1		UEP9D	UECS1	48.26					1			1	1	1

ONBONDE	D NETWORK ELEMENTS - Louisiana			1									ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)		Submitted Elec per LSR	Submitted Manually	Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonred	curring	Nonrecurring Disconnect			oss	Rates(\$)		
							First	Add'l	First Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	14.93									
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	25.35									
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	50.46									
	Port Rate														
ALLS	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	14.00	50.00	25.00	 	-	15.20	-			
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			OLF3D	OLFIA	14.00	30.00	23.00	 		13.20				
	Area 2-Wire Voice Grade Port (Centrex doc termination) Basic Local			UEP9D	UEPYB	14.00	50.00	25.00			15.20				
	Area			UEP9D	UEPYC	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area				UEPYV		50.00								
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local			UEP9D		14.00		25.00			15.20				
	Area 2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local			UEP9D	UEPY3	14.00	50.00	25.00			15.20				
	Area 2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			UEP9D	UEPYH	14.00	50.00	25.00			15.20				
	Indication))3 Basic Local Area 2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3			UEP9D	UEPYW	14.00	50.00	25.00			15.20				
	Basic Local Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPYJ	14.00	50.00	25.00			15.20				
	2 Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPYM	14.00	135.00	90.00			15.20				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPYO	14.00	135.00	90.00			15.20				
	Basic Local Area			UEP9D	UEPYP	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area			UEP9D	UEPY9	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic														
	Local Area Y, LA, MS, SC, & TN Only			UEP9D	UEPY2	14.00	50.00	25.00	 	+	15.20	1			

UNBUNDLEI	D NETWORK ELEMENTS - Louisiana							· <u> </u>				Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Manually per LSR	Incremental	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
					+	1	Nonrec	urring	Nonrecurring Disconnec			088	Rates(\$)		L
					1	Rec	First	Add'l	First Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	14.00	50.00	25.00	Tilat Auu I	JOHILO	15.20	JONAN	JOHAN	JOHAN	JONAN
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	14.00	50.00	25.00		+	15.20				
	2-Wire Voice Grade Port (Centrex 666 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	14.00	50.00	25.00	†		15.20				-
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	14.00	50.00	25.00	†		15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5309)3			UEP9D	UEPQE	14.00	50.00	25.00		-	15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	14.00	50.00	25.00		+	15.20				——
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	14.00	50.00	25.00		+	15.20				——
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	14.00	50.00	25.00		-	15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	14.00	50.00	25.00	†		15.20				
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	14.00	50.00	25.00	†		15.20				
†	2-Wire Voice Grade Port (Centrex / Ebb-Wiss 10)3			UEP9D	UEPQH	14.00	50.00	25.00		+	15.20				—
	2-Wire Voice Grade Port (Centrex With Caller ID/Msg Wtg Lamp			021 00	0L1 Q11	14.00	50.00	25.00		+	10.20		 		
	Indication)3			UEP9D	UEPQW	14.00	50.00	25.00			15.20				
- 	2-Wire Voice Grade Port (Centrex/Msq Wtq Lamp Indication)3		1	UEP9D	UEPQJ	14.00	50.00	25.00		+	15.20		 		
	2-Wire Voice Grade Port (Centrex/ring Vitg Early Indication) 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			OLI 3D	OLI QU	14.00	30.00	23.00	†		13.20				
	2			UEP9D	UEPQM	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	14.00	135.00	90.00			15.20				
	2-ville voice Grade i Gri (Geritiex diller GWG/EBG-i GE1/2, G			OLI 3D	OLI QO	14.00	155.00	30.00			13.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port (Centrex differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	14.00	135.00	90.00			15.20				
	2-Wile Voice Grade Fort (Certifex diller SWC /LB3-3209)2, 3			OLF 9D	ULFQQ	14.00	133.00	90.00			13.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	14.00	135.00	90.00			15.20				
	2-Wile Voice Grade Fort (Certite/Valler SWC/LBS-WST12)2, 3			OLF 9D	ULFUN	14.00	133.00	90.00	-		13.20				├
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	14.00	135.00	90.00			15.20				
	2-vviie voice Grade Fort (Ceritiex/diller SVVC /EBS-ivi5312)2, 3			UEP9D	UEPQS	14.00	133.00	90.00	+	-	15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	14.00	135.00	90.00			15.20				
	2-Wile Voice Grade Fort (Certite/Valler SWC/LBS-W5000)2, 3			OLF 9D	ULF Q4	14.00	133.00	90.00	-		13.20				├
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	14.00	135.00	90.00			15.20				
	2-vviie voice Grade Fort (Certitex differ SvvC /LB3-ivi5200)2, 3			OLF3D	ULFQ3	14.00	133.00	90.00	+	-	13.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	14.00	135.00	90.00			15.20				
	2-vviie voice Grade Fort (Ceritiex/diller SVVC/EBS-ivi5216)2, 3			UEP9D	UEPQ6	14.00	133.00	90.00	+	-	15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	14.00	135.00	90.00			15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OLF3D	ULFQ1	14.00	133.00	90.00	+	-	13.20				
	Term			UEP9D	UEPQZ	14.00	135.00	90.00			15.20				
	Telli			OLF 9D	ULFQZ	14.00	133.00	90.00			13.20				-
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	14.00	50.00	25.00			15.20				
	2-Wire Voice Grade Port Terminated in 611 Weganitk of equivalent			UEP9D	UEPQ2	14.00	50.00	25.00			15.20				
l ocal S	Switching			021 00	JL1 42	14.00	50.00	25.00		+	10.20		 		
Local C	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.8577			†						
l ocal N	lumber Portability			021 00	511255	3.0311				-	1				——
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35				+					
Feature				SE. 0D	,,	0.00				-					——
	All Standard Features Offered, per port			UEP9D	UEPVF	0.00				-	15.20				—
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	412.25			-	15.20				
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00	712.20		 	+	15.20		 	 	—
NARS	and desired a desired desired por port			00	50	0.00				+	10.20	1	1	1	
IVAILO	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00		-	15.20				——
-	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00		-	15.20				——
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00	 	+	15.20		 	 	—
Miscell	aneous Terminations			00	5, 11, 5, 1	0.00	0.00	0.00		+	10.20	1	1	1	
	Trunk Side				1 1					+		1	1	1	
2-11116	Trunk Side Terminations, each			UEP9D	CEND6	8.29	115.85	18.20		+	15.20		 		
4-Wire	Digital (1.544 Megabits)			0-	5250	0.20	110.00	10.20		-	10.20				—
	DS1 Circuit Terminations, each			UEP9D	M1HD1	68.47	196.18	98.62		-	15.20				—
	DS0 Channels Activiated per Channel			UEP9D	M1HD0	0.00	14.06	30.02		+	15.20		 		
	ice Channel Mileage - 2-Wire		1	OLI 3D	WITTIDO	0.00	14.00			+	13.20		 		
meron	Interoffice Channel Facilities Termination			UEP9D	MIGBC	22.60	39.36	26.62	 	+	15.20		1		
	Interoffice Channel mileage, per mile or fraction of mile		_	UEP9D	MIGBM	0.013	55.50	20.02	.	-	10.20	l		 	

UNBUNDI F	D NETWORK ELEMENTS - Louisiana												Attach	ment: 2	Evhil	bit: C
SHOONDLE	D NETWORK ELLINERTO - Louisiana										Svc Order	Svc Order	Incremental		Incremental	Incremental
Ī											Submitted	Submitted		Charge -	Charge -	Charge -
Ī		1									Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc			RATES(\$)								
CATEGORI	NATE ELEMENTO	m	20116	500	0000			NATEO(ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
					1	_ 1	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
 						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Featur	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e					11100	Addi	11130	Auu	COME	COMPAR	COMPAR	COMPAN	COMPAN	COMPAR
	annel Bank Feature Activations	Ī														
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP9D	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP9D	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot	l	1	UEP9D	1PQWV	0.6497			Ì			15.20		Ì		Ì
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
<u> </u>	Slot		Ш.	UEP9D	1PQWQ	0.6497			<u> </u>	<u> </u>	<u> </u>	15.20		<u> </u>	<u> </u>	<u> </u>
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.6497						15.20				
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		0.10	0.10	L			15.20			<u> </u>	
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		36.66	16.10				15.20				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	680.40					15.20				
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	680.40					15.20				
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	73.93					15.20				
UNE-P	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE P	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP9E		25.77										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP9E		36.39										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP9E		62.26										
UNE P	ort/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo															
	Design		1	UEP9E		28.93										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_	l												
\vdash	Design		2	UEP9E	1	39.35			 					 	1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo		_	LIEDOE		04.40										
I IAIT !	Design Parts		3	UEP9E		64.46			-					-		-
UNE L	oop Rate		1	UEP9E	IJECS4	11.77			ļ					 		
\vdash	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	-	2	UEP9E UEP9E	UECS1 UECS1	22.39						 		-		-
\vdash		-	3	UEP9E UEP9E	UECS1	48.26			1			 		1	1	1
\vdash	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1	-	1	UEP9E UEP9E	UECS1 UECS2	48.26 14.93			1			 		1	1	1
\vdash	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2	-	2	UEP9E UEP9E	UECS2	25.35			1			 		1	1	1
\vdash	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	50.46					1					
LINE D	Port Rate	-	- 3	OLI OL	02002	30.40			 					 		
	-, KY, LA, MS, & TN only		1		1							l				
75,15	2-Wire Voice Grade Port (Centrex) Basic Local Area		1	UEP9E	UEPYA	14.00	50.00	25.00	1	1		15.20		1	1	1
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local		1		J=: ./\	14.00	55.55	20.00				10.20				
	Area			UEP9E	UEPYB	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local		†			00	22.00							1		1
	Area			UEP9E	UEPYH	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire		1		1				İ	İ				İ	İ	İ
	Center)2 Basic Local Area			UEP9E	UEPYM	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service		†					22.30	1					1		1
	Term - Basic Local Area	1	1	UEP9E	UEPYZ	14.00	135.00	90.00				15.20		Ì		Ì
	2-Wire Voice Grade Port terminated in on Megalink or equivalent				T 1			22.30	İ						İ	
	- Basic Local Area			UEP9E	UEPY9	14.00	50.00	25.00				15.20		1		1
				1	1				1					1		Ì
	2-Wire Voice Grade Port Terminated on 800 Service Term -				1											

<u>JNBUNDL</u>	ED NETWORK ELEMENTS - Louisiana													nent: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	Charge - Manual Sv Order vs. Electronic
							Names		Namananimi	- Di			1st	Add'l	Disc 1st	Disc Add'
					+	Rec	Nonrec First	urring Add'l	First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
AL. P	(Y, LA, MS, & TN Only							,,,,,,		71441	0020					00
	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQB	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPQM	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9E	UEPQZ	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port Terminated in on Wegalink or equivalent	1	 	UEP9E	UEPQ2	14.00	50.00	25.00			 	15.20			 	
Loca	Switching	1	 	OLI SL	ULFUZ	14.00	50.00	25.00		 	 	13.20			 	
LUCA	Centrex Intercom Funtionality, per port	1	 	UEP9E	URECS	0.8577			1	1	1				 	
Loca	Number Portability			OLF 9L	UKLCS	0.0377					+				-	ļ
Loca	Local Number Portability (1 per port)	 	 	UEP9E	LNPCC	0.35				-	-					
F4				UEP9E	LNPCC	0.35										
Featu	All Standard Features Offered, per port	1	 	UEP9E	UEPVF	0.00					 	15.20			-	-
-	All Select Features Offered, per port	 		UEP9E UEP9E	UEPVF	0.00	412.25				1	15.20			 	
_							412.25									
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00						15.20				
NAR																
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00								
	ellaneous Terminations															
2-Wii	e Trunk Side															
	Trunk Side Terminations, each			UEP9E	CEND6	8.29	115.85	18.20				15.20				
4-Wii	e Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9E	M1HD1	68.47	196.18	92.92				15.20				
	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	14.06					15.20				
Inter	office Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	22.60	39.36	26.62				15.20				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.013										
Featu	ire Activations (DS0) Centrex Loops on Channelized DS1 Service	ce														
D4 C	hannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.6497						15.20				
	·															
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP9E	1PQW6	0.6497						15.20				
	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP9E	1PQW7	0.6497						15.20				
	Different Wire Center			UEP9E	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9E	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot	<u> </u>		UEP9E	1PQWA	0.6497					-	15.20				
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex	ļ	ļ								1					
	NRC Conversion Currently Combined Switch-As-Is with allowed	1	1			l									I	
	changes, per port	<u> </u>	<u> </u>	UEP9E	USAC2		0.10	0.10			1	15.20			1	
	Conversion of Existing Centrex Common Block, each	<u> </u>	<u> </u>	UEP9E	USACN		36.66	16.10			1	15.20			1	
	New Centrex Standard Common Block	ļ	<u> </u>	UEP9E	M1ACS	0.00	680.40					15.20			.	
	New Centrex Customized Common Block	ļ		UEP9E	M1ACC	0.00	680.40]		15.20				
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	73.93					15.20				
	P CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)							-								
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-														
1	Non-Design		1	UEP93		25.77					1					

UNBLINDI	ED NETWORK ELEMENTS - Louisiana												Attach	ment: 2	Fyhil	oit: C
CHOCKEL	LD INC. INC. IN LELINICIA I O - LOUISIANA										Svc Order	Svc Order	Incremental		Incremental	Incremental
1											Submitted	Submitted		Charge -	Charge -	Charge -
											Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)								
		m		200	5555			= = (+)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
\vdash						1	Nonrec	urring	Nonrecurring	n Disconnect		I	OSS	Rates(\$)	I.	I.
\vdash						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
\vdash	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -						11131	Auu i	11130	Auu	JOINEC	JONAN	JONAN	JOHAN	JOHAN	JONAN
	Non-Design		2	UEP93		36.36										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			ULF 93		30.30										
	Non-Design		3	UEP93		62.26										
LINE		<u> </u>	3	UEF93	-	02.20										
UNE	Port/Loop Combination Rates (Design)	<u> </u>			-											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1		LIEDOS		20.02										
\vdash	Design		1	UEP93		28.93										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_													
$\vdash \vdash \vdash$	Design Court	<u> </u>	2	UEP93	1	39.35						ļ				
1 1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		l -													
	Design	ļ	3	UEP93		64.46										
UNE	Loop Rate		<u> </u>		1	ļ										
igsquare	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP93	UECS1	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	22.36										
igsquare	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	48.26										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	14.93										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	25.35										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	50.46										
UNE	Port Rate															
AL, K	Y, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP93	UEPYB	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP93	UEPYH	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP93	UEPYM	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP93	UEPYZ	14.00	135.00	90.00				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP93	UEPY9	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port Terminated on 800 Service Term -			02. 00	020		00.00	20.00				10.20				
1	Basic Local Area			UEP93	UEPY2	14.00	50.00	25.00				15.20				
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	14.00	50.00	25.00				15.20				
\vdash	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	14.00	50.00	25.00				15.20				
+-+-	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1	 	 	UEP93	UEPQH	14.00	50.00	25.00				15.20				
\vdash	2-Wire Voice Grade Port (Centrex with Carler ID)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire	 	 	OE1 30	טבו עוו	14.00	50.00	20.00	 			13.20		 		
1	Center)2	1	1	UEP93	UEPQM	14.00	135.00	90.00				15.20		I		
\vdash		 	├	OFL 20	OLF QIVI	14.00	135.00	90.00	-		 	15.20		-		
1	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1	1	UEP93	UEPQZ	14.00	135.00	90.00				15.20		I		
\vdash	Term	 	├	OLF93	UEFUL	14.00	135.00	90.00	-		 	15.∠0		-		
1 1	2 Wire Voice Crade Bort terminated in an Manadial and a late			LIEDOS	LIEDOO	44.00	50.00	05.00				45.00				
\vdash	2-Wire Voice Grade Port terminated in on Megalink or equivalent	 	1	UEP93	UEPQ9	14.00	50.00	25.00	1	-	1	15.20		 	-	-
	2-Wire Voice Grade Port Terminated on 800 Service Term	-	1	UEP93	UEPQ2	14.00	50.00	25.00			-	15.20		-		1
Local	Switching	-	<u> </u>	LIEDOO	LIDECC						1			1		
\vdash	Centrex Intercom Funtionality, per port	<u> </u>	<u> </u>	UEP93	URECS	0.8577			-	ļ				-	1	1
Local	Number Portability	<u> </u>	<u> </u>			0.5-						ļ				
	Local Number Portability (1 per port)	ļ	ļ	UEP93	LNCCC	0.35										
Featu		ļ	 	LIEBAA										.		
igsquare	All Standard Features Offered, per port	<u> </u>	<u> </u>	UEP93	UEPVF	0.00					1	15.20		1		
igsquare	All Centrex Control Features Offered, per port			UEP93	UEPVC	0.00						15.20				
NARS																
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00				15.20				
	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00				15.20				
	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00				15.20				
	ellaneous Terminations							-								
2-Wir	e Trunk Side															
	Trunk Side Terminations, each			UEP93	CEND6	8.27	115.85	18.20				15.20				
4-Wir	e Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP93	M1HD1	68.47	196.18	92.92				15.20				

BUNDLE	D NETWORK ELEMENTS - Louisiana												Attachr	nent: 2	Exhi	bit: C
FEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
							Nonrecu	ırrina	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	DS0 Channels Activated. Per Channel			UEP93	M1HDO	0.00	14.06					15.20				
Interoff	ice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP93	MIGBC	22.60	39.36	26.62				15.20				
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.013										
	Activations (DS0) Centrex Loops on Channelized DS1 Service	е														
	nnel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP93	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP93	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop Slot			UEP93	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.6497						15.20				
	curring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP93	USAC2		0.10	0.10				15.20				
	Conversion of Existing Centrex Common Block, each			UEP93	USACN		36.66	16.10				15.20				
	New Centrex Standard Common Block			UEP93	M1ACS	0.00	680.40					15.20				
	New Centrex Customized Common Block			UEP93	M1ACC	0.00	680.40					15.20				
	NAR Establishment Charge, Per Occasion			UEP93	URECA	0.00	73.93					15.20				
	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD															
	- Requres Interoffice Channel Mileage															
Note 3	- Requires Specific Customer Premises Equipment															

UNBUNDI F	D NETWORK ELEMENTS - Mississippi												Δttach	ment: 2	Exhil	bit: C
CHECHEL		1				1					Svc Order	Svc Order	Incremental			
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
											Elec		Manual Svc	Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)				,				
CATEGORI	NATE ELEMENTO	m	20116	500	0000			KAT LO(ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonred	urring	Nonrecurring	Disconnect		I	OSS	Rates(\$)	1	L
					-	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN		SOMAN	SOMAN
The "7	one" shown in the sections for stand-alone loops or loops as part (of a com	hinatio	n refers to Geographi	ically Deaver	aged LINE Zones									JOHAN	JOHIAN
	www.interconnection.bellsouth.com/become_a_clec/html/interconn			in refers to Geographi	ically Deavers	aged OIVE Zones	. TO VIEW GEOG	graprincally Dea	veraged ONL 20	nie Designation	is by Certifa	i Onice, reie	i to internet vi	repaile.		
	L SUPPORT SYSTEMS	ection.n	1111				1								1	
	(1) Electronic Service Order: CLEC should contact its contract	ct nogot	istori	it profess the state	enocific aloc	tronic corvice o	rdorina charac	e as ordered l	w the State Co	mmissions T	ho electron	io convico o	doring charg	o currently or	ntained in th	is rate
																.s rate
	t is the BellSouth regional electronic service ordering charge.															
	(2) Any element that can be ordered electronically will be bill															
	elements that cannot be ordered electronically at present per t				e in this cate	gory reflects th	e charge that v	vould be billed	to a CLEC on	ce electronic o	ordering cap	abilities co	me on-line to	r that elemen	t. Otherwise,	the manual
orderin	ng charge, SOMAN, will be applied to a CLECs bill when it sub	omits ar	LSR	o BellSouth.									•			
	Manual Service Order Charge, per LSR, Disconnect Only (MS)				SOMAN				1.97							
	Electronic OSS Charge, per LSR, submitted via BST's OSS															
	interactive interfaces (Regional)				SOMEC		3.50									
	DATE ADVANCEMENT CHARGE															
NOTE:	The Expedite charge will be maintained commensurate with	BellSou	th's F	CC No.1 Tariff, Section	on 5 as appl	icable.										
	UNE Expedite Charge per Circuit or Line Assignable USOC, per															
	Day	<u> </u>		ALL UNE	SDASP		200.00									<u> </u>
	EXCHANGE ACCESS LOOP									-						
2-WIRE	E ANALOG VOICE GRADE LOOP									-						
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	12.03	37.92	17.55	23.48	5.25		15.75				
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	16.87	37.92	17.55	23.48	5.25		15.75				
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	25.68	37.92	17.55	23.48	5.25		15.75				
	2-Wire Analog Voice Grade Loop - Service Level 1-Zone 4		4	UEANL	UEAL2	43.85	37.92	17.55	23.48	5.25		15.75				
	Loop Testing - Basic 1st Half Hour			UEANL	URET1		34.36					15.75				
	Loop Testing - Basic Additional Half Hour			UEANL	URETA		19.97					15.75				
	CLEC to CLEC Conversion Charge Without Outside Dispatch			UEANL	UREWO		15.75	8.92				15.75				
	Engineering Information Document (EI)			UEANL	UEANM		13.51	13.51								
	Manual Order Coordination for UVL-SL1s (per loop)			UEANL	UEAMC		8.20	8.20								
	Order Coordination for Specified Conversion Time for UVL-SL1															
	(per LSR)			UEANL	OCOSL		18.19	18.19								
2-WIRE	E Unbundled COPPER LOOP															
	2-Wire Unbundled Copper Loop - Non-Designed Zone 1	1	1	UEQ	UEQ2X	11.01	36.53	16.16	22.66	4.42		15.75				
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	1	2	UEQ	UEQ2X	11.51	36.53	16.16	22.66	4.42		15.75				
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	1	3	UEQ	UEQ2X	11.57	36.53	16.16	22.66	4.42		15.75				
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 4	1	4	UEQ	UEQ2X	13.10	36.53	16.16	22.66	4.42		15.75				
	Order Coordination 2 Wire Unbundled Copper Loop - Non-															
	Designed (per loop)	1		UEQ	USBMC		8.20	8.20				1		I	Ì	1
	Engineering Information Document	1		UEQ	1	İ	13.51	13.51						İ	İ	
	Loop Testing - Basic 1st Half Hour	1		UEQ	URET1	1	34.36					15.75		İ	İ	
	Loop Testing - Basic Additional Half Hour	1		UEQ	URETA		19.97					15.75		1	İ	
	CLEC to CLEC Conversion Charge Without Outside Dispatch	1		UEQ	UREWO		14.24	7.42				15.75		1	İ	
UNBUNDLED F	EXCHANGE ACCESS LOOP	1		1	1	İ								İ	İ	
	E ANALOG VOICE GRADE LOOP	1		İ	1									1	İ	
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	1		İ	1									1	İ	
	Ground Start Signaling - Zone 1	1	1	UEA	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75		1		1
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	†	<u> </u>	1	1	.0.55	.00.00	55.20	32.32	.0.07				t	1	
	Ground Start Signaling - Zone 2	1	2	UEA	UEAL2	18.75	105.96	68.28	52.82	10.37		15.75		I	Ì	1
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or			02/1	O E / KEE	10.10	100.00	00.20	02.02	10.01	1	10.70				
	Ground Start Signaling - Zone 3		3	UEA	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	 	_		1	27.00	100.00	00.20	02.02	10.07		10.70		 		—
	Ground Start Signaling - Zone 4		4	UEA	UEAL2	45.72	105.96	68.28	52.82	10.37		15.75				
 	Order Coordination for Specified Conversion Time (per LSR)	 	-	UEA	OCOSL	40.12	18.19	00.20	32.02	10.37		13.73		 	 	
 	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	 		0_/\	JUUDE	 	10.19							 	 	
	Battery Signaling - Zone 1	1	1	UEA	UEAR2	13.89	105.96	68.28	52.82	10.37		15.75		1		1
 	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	 	_	027	OLANZ	15.05	103.30	00.20	32.02	10.37	1	13.73		 	1	
	Battery Signaling - Zone 2	1	2	UEA	UEAR2	18.75	105.96	68.28	52.82	10.37		15.75		1		1
 	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1		ULA	ULARZ	10.75	105.96	00.28	5∠.0∠	10.37	 	15.75		 	1	
	Battery Signaling - Zone 3	1	3	UEA	UEAR2	27.55	105.96	68.28	52.82	10.37		15.75		I	Ì	1
 	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	 	3	OLA	ULARZ	21.35	105.96	00.28	32.02	10.37		15.75			 	
	Battery Signaling - Zone 4	1	4	UEA	UEAR2	45.72	105.96	68.28	52.82	10.37		15.75		1		1
\vdash	Order Coordination for Specified Conversion Time (per LSR)	 	4	UEA		45.72	105.96	55.28	5∠.82	10.37	 	15.75		 		
1 1	Torder Coordination for Specified Conversion Time (per LSR)	1		UEA	OCOSL		18.19				L	l				

Version 3Q02: 09/06/02

ONBON	DLE	D NETWORK ELEMENTS - Mississippi												Attach	ment: 2	Exhi	bit: C
CATEGO		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.56	36.29				15.75				
4-		ANALOG VOICE GRADE LOOP						100.00									
		4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	27.47	132.27	94.59	60.68	14.64		15.75				
		4-Wire Analog Voice Grade Loop - Zone 2		2	UEA	UEAL4 UEAL4	38.26	132.27	94.59	60.68	14.64		15.75				
		4-Wire Analog Voice Grade Loop - Zone 3 4-Wire Analog Voice Grade Loop - Zone 4		3	UEA UEA	UEAL4 UEAL4	50.03 50.03	132.27 132.27	94.59 94.59	60.68 60.68	14.64 14.64		15.75 15.75				
		Order Coordination for Specified Conversion Time (per LSR)		4	UEA	OCOSL	50.03	18.19	94.59	00.00	14.04		15.75				+
		CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.56	36.29				15.75				+
2-		ISDN DIGITAL GRADE LOOP			OLA	UKLWO		07.50	30.29				13.73				+
		2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	21.01	117.61	79.92	52.82	10.37		15.75				+
		2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	27.59	117.61	79.92	52.82	10.37	 	15.75	1	1	 	
		2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	37.34	117.61	79.92	52.82	10.37		15.75			1	†
h t		2-Wire ISDN Digital Grade Loop - Zone 4		4	UDN	U1L2X	59.18	117.61	79.92	52.82	10.37		15.75			1	†
		Order Coordination For Specified Conversion Time (per LSR)		<u> </u>	UDN	OCOSL	220	18.19								İ	†
		CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.46	44.07				15.75				
2-		ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIBLE	LOOF	,												1
		2 Wire Unbundled ADSL Loop including manual service inquiry															1
		& facility reservation - Zone 1		1	UAL	UAL2X	11.11	121.27	70.81	50.38	7.93		15.75				
		2 Wire Unbundled ADSL Loop including manual service inquiry															
		& facility reservation - Zone 2		2	UAL	UAL2X	11.47	121.27	70.81	50.38	7.93		15.75				
		2 Wire Unbundled ADSL Loop including manual service inquiry															1
		& facility reservation - Zone 3		3	UAL	UAL2X	11.74	121.27	70.81	50.38	7.93		15.75				
		2 Wire Unbundled ADSL Loop including manual service inquiry															
		& facility reservation - Zone 4		4	UAL	UAL2X	12.69	121.27	70.81	50.38	7.93		15.75				
		Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.19									
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
		facility reservaton - Zone 1		1	UAL	UAL2W	11.11	96.15	58.03	50.38	7.93		15.75				
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
		facility reservaton - Zone 2		2	UAL	UAL2W	11.47	96.15	58.03	50.38	7.93		15.75				
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
		facility reservaton - Zone 3		3	UAL	UAL2W	11.74	96.15	58.03	50.38	7.93		15.75				
		2 Wire Unbundled ADSL Loop without manual service inquiry &															
<u> </u>		facility reservaton - Zone 4		4	UAL	UAL2W	12.69	96.15	58.03	50.38	7.93		15.75				
		Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.19	40.00				45.75				
		CLEC to CLEC Conversion Charge without outside dispatch	TIDI E I	000	UAL	UREWO		86.04	40.33				15.75				
2-		HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA 2 Wire Unbundled HDSL Loop including manual service inquiry	IIBLE	LOOP													
		& facility reservation - Zone 1		1	UHL	UHL2X	8.75	129.98	79.52	50.38	7.93		15.75				
-		2 Wire Unbundled HDSL Loop including manual service inquiry		-	UNL	UHLZA	0.75	129.90	79.52	30.36	7.93		15.75				+
		& facility reservation - Zone 2		2	UHL	UHL2X	9.22	129.98	79.52	50.38	7.93		15.75				
		2 Wire Unbundled HDSL Loop including manual service inquiry			UNL	UHLZX	9.22	129.90	79.52	50.36	7.93		15.75				+
		& facility reservation - Zone 3		3	UHL	UHL2X	9.87	129.98	79.52	50.38	7.93		15.75				
		2 Wire Unbundled HDSL Loop including manual service inquiry		3	OFIL	UTILZX	5.07	129.90	19.52	30.36	7.55		13.73				
		& facility reservation - Zone 4		4	UHL	UHL2X	10.46	129.98	79.52	50.38	7.93		15.75				
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL	10.10	18.19	7 0.02	00.00	7.00		10.70				
		2 Wire Unbundled HDSL Loop without manual service inquiry			OTIL	00002		10.10									†
		and facility reservation - Zone 1		1	UHL	UHL2W	8.75	104.86	66.74	50.38	7.93		15.75				
		2 Wire Unbundled HDSL Loop without manual service inquiry			-		5.70			22.00							†
		and facility reservation - Zone 2		2	UHL	UHL2W	9.22	104.86	66.74	50.38	7.93		15.75				
		2 Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 3		3	UHL	UHL2W	9.87	104.86	66.74	50.38	7.93	1	15.75			1	
		2 Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 4	<u></u>	4	UHL	UHL2W	10.46	104.86	66.74	50.38	7.93	<u></u>	15.75	<u> </u>		<u> </u>	<u> </u>
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.19									
		CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		85.98	40.33				15.75				
4-		HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
		4 Wire Unbundled HDSL Loop including manual service inquiry												I			
		and facility reservation - Zone 1		1	UHL	UHL4X	13.78	158.74	108.28	56.72	10.68	l	15.75				1

HINDH	NDI E	D NETWORK ELEMENTS - Mississippi												Attach	ment: 2	Evhil	oit: C
OIABO	NULE	P MET MORK ELEMIEM 19 - MISSISSIPPI										Svc Order	Svc Order	Incremental			
												Submitted			Charge -	Charge -	Charge -
												Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'I	Disc 1st	Disc Add'l
																D130 131	DISC Add I
							Rec	Nonred		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		4-Wire Unbundled HDSL Loop including manual service inquiry		_				.== = .									
		and facility reservation - Zone 2		2	UHL	UHL4X	13.43	158.74	108.28	56.72	10.68		15.75				
		4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 3		3	UHL	UHL4X	15.59	158.74	108.28	56.72	10.68		15.75				
		4-Wire Unbundled HDSL Loop including manual service inquiry		3	UNL	UHL4A	15.59	156.74	100.20	30.72	10.00		15.75				
		and facility reservation - Zone 4		4	UHL	UHL4X	14.46	158.74	108.28	56.72	10.68		15.75				
-		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL	14.40	18.19	100.20	30.72	10.00		13.73				
		4-Wire Unbundled HDSL Loop without manual service inquiry			0.12	00002		10.10									
		and facility reservation - Zone 1		1	UHL	UHL4W	13.78	133.62	95.50	56.72	10.68		15.75				
		4-Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 2		2	UHL	UHL4W	13.43	133.62	95.50	56.72	10.68		15.75	<u></u>			
		4-Wire Unbundled HDSL Loop without manual service inquiry															
		and facility reservation - Zone 3		3	UHL	UHL4W	15.59	133.62	95.50	56.72	10.68		15.75				
1 7		4-Wire Unbundled HDSL Loop without manual service inquiry				I											
		and facility reservation - Zone 4		4	UHL	UHL4W	14.46	133.62	95.50	56.72	10.68		15.75				
-		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.19									
	4 WIDE	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		85.98	40.33				15.75				
	4-WIKE	DS1 DIGITAL LOOP 4-Wire DS1 Digital Loop - Zone 1		1	USL	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				
-		4-Wire DS1 Digital Loop - Zone 1			USL	USLXX	129.38	253.93	158.45	46.10	12.07		15.75				
-		4-Wire DS1 Digital Loop - Zone 2			USL	USLXX	206.74	253.93	158.45	46.10	12.07	1	15.75				
		4-Wire DS1 Digital Loop - Zone 3			USL	USLXX	458.46	253.93	158.45	46.10	12.07		15.75				
-		Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL	400.40	18.19	100.40	40.10	12.01		10.70				
		CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		100.90	42.96				15.75				
	4-WIRE	19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															
		4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	27.44	126.53	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	34.55	126.53	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	40.76	126.53	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	32.25	126.53	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 1			UDL	UDL56	27.44	126.53	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	34.55	126.53	88.85	60.68	14.64		15.75				
-		4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL UDL	UDL56	40.76 32.25	126.53	88.85	60.68	14.64 14.64		15.75				
-		4 Wire Unbundled Digital Loop 56 Kbps - Zone 4 Order Coordination for Specified Conversion Time (per LSR)		4	UDL	UDL56 OCOSL	32.25	126.53 18.19	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	27.44	126.53	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 1			UDL	UDL64	34.55	126.53	88.85	60.68	14.64	 	15.75				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3		UDL64	40.76	126.53	88.85	60.68	14.64		15.75				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 4		4		UDL64	32.25	126.53	88.85	60.68	14.64		15.75				
		Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL	50	18.19	22.30	22.30	31						
		CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		101.94	49.66				15.75				
	2-WIRE	Unbundled COPPER LOOP															
		2-Wire Unbundled Copper Loop/Short including manual service															
		inquiry & facility reservation - Zone 1		1	UCL	UCLPB	11.11	120.34	69.87	50.38	7.93		15.75				
		2-Wire Unbundled Copper Loop/Short including manual service		_													
		inquiry & facility reservation - Zone 2		2	UCL	UCLPB	11.47	120.34	69.87	50.38	7.93	}	15.75	1	1		
		2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3		3	UCL	UCLPB	11.74	120.34	69.87	50.38	7.93		15.75				
		2 Wire Unbundled Copper Loop/Short including manual service		3	UCL	UCLPB	11.74	120.34	09.87	50.38	7.93	1	15.75	1	1		
		inquiry & facility reservation - Zone 4		4	UCL	UCLPB	12.69	120.34	69.87	50.38	7.93		15.75				
		Order Coordination for Unbundled Copper Loops (per loop)		7	UCL	UCLMC	12.03	8.20	8.20	55.56	7.95		10.70				
		2-Wire Unbundled Copper Loop/Short without manual service						3.20	0.20								
		inquiry and facility reservation - Zone 1		1	UCL	UCLPW	11.11	95.21	57.09	50.38	7.93		15.75				
		2-Wire Unbundled Copper Loop/Short without manual service					_							1	1		
		inquiry and facility reservation - Zone 2		2	UCL	UCLPW	11.47	95.21	57.09	50.38	7.93		15.75	<u> </u>	<u> </u>		
		2-Wire Unbundled Copper Loop/Short without manual service															
		inquiry and facility reservation - Zone 3		3	UCL	UCLPW	11.74	95.21	57.09	50.38	7.93		15.75				
		2-Wire Unbundled Copper Loop/Short without manual service				1											
		inquiry and facility reservation - Zone 4		4	UCL	UCLPW	12.69	95.21	57.09	50.38	7.93		15.75				
		Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20			<u> </u>		l	l		

UNBUNDLE	D NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonre			Disconnect				Rates(\$)		
	DWG-11-1						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone 1		1	UCL	UCL2L	29.29	120.34	69.87	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.		-	UCL	UCLZL	29.29	120.34	09.07	50.56	7.93		15.75				1
	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	43.46	120.34	69.87	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL2L	64.44	120.34	69.87	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 4		4	UCL	UCL2L	87.60	120.34	69.87	50.38	7.93		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20								ļ
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone 1		1	UCL	UCL2W	29.29	95.21	57.09	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - without manual service		+ -	UCL	UCLZVV	29.29	93.21	37.09	30.36	7.93		13.73				
	inquiry and facility reservation - Zone 2		2	UCL	UCL2W	43.46	95.21	57.09	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 3		3	UCL	UCL2W	64.44	95.21	57.09	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 4		4	UCL	UCL2W	87.60	95.21	57.09	50.38	7.93		15.75				
	Order Coordination for Unbundled Copper Loops (per loop) CLEC to CLEC Conversion Charge without outside dispatch			UCL	UCLMC		8.20	8.20								ļ
	(UCL-Des)			UCL	UREWO		95.21	42.40				15.75				
4-WIR	E COPPER LOOP		1	OOL	OIKEWO		95.21	42.40				13.73				
	4-Wire Copper Loop/Short - including manual service inquiry				1										İ	
	and facility reservation - Zone 1		1	UCL	UCL4S	17.30	144.68	94.22	56.72	10.68		15.75				
	4-Wire Copper Loop/Short - including manual service inquiry															
	and facility reservation - Zone 2		2	UCL	UCL4S	18.84	144.68	94.22	56.72	10.68		15.75				ļ
	4-Wire Copper Loop/Short - including manual service inquiry		3		1101.40	04.00	444.00	04.00	50.70	40.00		45.75				
	and facility reservation - Zone 3		3	UCL	UCL4S	21.33	144.68	94.22	56.72	10.68		15.75				ļ
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 4		4	UCL	UCL4S	21.33	144.68	94.22	56.72	10.68		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)		_	UCL	UCLMC	21.00	8.20	8.20	00.72	10.00		10.70				
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 1		1	UCL	UCL4W	17.30	119.56	81.44	56.72	10.68		15.75				
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 2		2	UCL	UCL4W	18.84	119.56	81.44	56.72	10.68		15.75				
	4-Wire Copper Loop/Short - without manual service inquiry and		3	UCL	UCL4W	21.33	110.50	81.44	50.70	10.68		45.75				
	facility reservation - Zone 3 4-Wire Copper Loop/Short - without manual service inquiry and		3	UCL	UCL4VV	21.33	119.56	81.44	56.72	10.68		15.75				
	facility reservation - Zone 4		4	UCL	UCL4W	21.33	119.56	81.44	56.72	10.68		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)		Ė	UCL	UCLMC	21.00	8.20	8.20	00.72	10.00		10.70				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															1
	inquiry and facility reservation - Zone 1		1	UCL	UCL4L	54.72	144.68	94.22	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	97.47	144.68	94.22	56.72	10.68		15.75				ļ
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3		3	UCL	UCL4L	106.06	144.68	94.22	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.		3	UCL	UCL4L	106.06	144.68	94.22	56.72	10.68		15.75				1
	inquiry and facility reservation - Zone 4		4	UCL	UCL4L	106.06	144.68	94.22	56.72	10.68		15.75				
1	Order Coordination for Unbundled Copper Loops (per loop)		Ė	UCL	UCLMC		8.20	8.20								1
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 1		1	UCL	UCL4O	54.72	119.56	81.44	56.72	10.68		15.75			1	
	4-Wire Unbundled Copper Loop/Long - without manual svc.		_		1101.40	07.4-	440.50		50.70	10.00		45.75				
	inquiry and facility reservation - Zone 2	-	2	UCL	UCL4O	97.47	119.56	81.44	56.72	10.68		15.75		-	1	
1	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 3		3	UCL	UCL4O	106.06	119.56	81.44	56.72	10.68		15.75				
 	4-Wire Unbundled Copper Loop/Long - without manual service		-		55240	100.00	113.30	01.44	30.72	10.00	 	10.10			†	
	inquiry and facility reservation - Zone 4		4	UCL	UCL4O	106.06	119.56	81.44	56.72	10.68		15.75			1	
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20								
	CLEC to CLEC Conversion Charge without outside dispatch															
	(UCL-Des)			UCL	UREWO		95.21	42.40				15.75				

CATEGORY RATE ELEMENTS Name 2006 BCS USOC PATES(S)	LINBLINDI	ED NETWORK ELEMENTS - Mississippi												Attach	ment: 2	Evhi	bit: C
Mac First Age(1) SOME				Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	
Description Comparison Co							Pos	Nonrec	urring	Nonrecurring	Disconnect						
Debundled Loop Modification, Removal of Lood Citils - 2 We part feets than or equal to 18.8 Loop Loop Loop Loop Loop Loop Loop Lo							Kec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Patient feet Local Addressers, Personal of Load Cole - 2 Wire Dec. LOC. U.S. U.S. U.S. U.S. U.S. U.S. U.S. U.	LOOP MODIF	ICATION															
Desire than 16th 15th 17th 17th 15		pair less than or equal to 18k ft			UEQ, ULS, UEA, UEANL, UDL, UDC,	ULM2L		32.57	32.57				15.75				
Unburidies Loop Modification Removal of Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire Use A Look Colls - 4 Wire					LICE LIES LIES	LILMOC		171 40	171 40				15.75				
Most not not registed trapp Medicification Removal of Load Cols - 4 Wire part greater than 16 N. U.S.					UCL, ULS, UEQ	ULIVIZG		171.49	171.49				15.75				
Description Section					UHL, UCL	ULM4L		32.57	32.57				15.75				
URL URL URL URL URL URL URL URL URL URL																	
SUB-LOOP For Cross Box Location - CLEC Feeder Facility Set- UEANIL USBSB UEANIL USBSD UE		pair greater than 18k ft				ULM4G		171.49	171.49				15.75				
Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set 1					UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL,	ULMBT		32.59	32.59				15.75				
Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Lip UEANL USBSA 259.888 115.75																	
Upan	Sub-L																
Sub-Loop - Per Bulding Equipment Room - CLEC Feeder 1			ı		UEANL	USBSA		259.69					15.75				ļ
Sub-Loop - Per Bulding Equipment Room - CLEC Feeder 1		Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	1		UEANL	USBSB		22.77					15.75				
Set-Up 1		Sub-Loop - Per Building Equipment Room - CLEC Feeder	ı														
Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - 1					LIEANI	HEBED		56.20					15.75				
Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - 1 2 UEANL USBN2 9.51 66.18 31.14 45.36 6.71 15.75		Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -		1			7.15		31.14	45.36	6.71						
Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - 1 3 UEANL USBN2 12.45 66.18 31.14 45.36 6.71 15.75		Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -		2													
Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - 4 UEANIL USBNZ 18.26 66.18 31.14 45.36 6.71 15.75		Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -	<u> </u>														
Order Coordination for Unbundled Sub-Loops, per sub-loop pair UEANL USBMC 8.20		Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -	'														
Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - 1 UEANL USBN4 7.30 79.49 44.45 51.27 9.35 15.75		Zone 4		4	UEANL	USBN2	18.26	66.18	31.14	45.36	6.71		15.75				
Zone 1					UEANL	USBMC		8.20	8.20								<u> </u>
Zone 2		Zone 1		1	UEANL	USBN4	7.30	79.49	44.45	51.27	9.35		15.75				
Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - 3 UEANL USBN4 16.73 79.49 44.45 51.27 9.35 15.75				2	UEANL	USBN4	13.92	79.49	44.45	51.27	9.35		15.75				
Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - 4 UEANL USBN4 16.73 79.49 44.45 51.27 9.35 15.75		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -		3													
Order Coordination for Unbundled Sub-Loops, per sub-loop pair UEANL USBMC Sub-Loop 2-Wire Intrabuilding Network Cable (INC) I UEANL USBR2 2.29 53.32 18.28 45.36 6.71 15.75		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
Sub-Loop 2-Wire Intrabuilding Network Cable (INC) I UEANL USBR2 2.29 53.32 18.28 45.36 6.71 15.75		LUIG		4	OLAINL	000144	10.73	79.49	44.40	51.27	9.35	 	15.75				
Sub-Loop 2-Wire Intrabuilding Network Cable (INC) I UEANL USBR2 2.29 53.32 18.28 45.36 6.71 15.75				L			<u> </u>			<u> </u>		<u> </u>		<u> </u>	<u> </u>		<u> </u>
Sub-Loop 4-Wire Intrabuilding Network Cable (INC) I UEANL USBR4 4.40 59.60 24.55 51.27 9.35 15.75			I		UEANL	USBR2	2.29	53.32		45.36	6.71		15.75				
Sub-Loop 4-Wire Intrabuilding Network Cable (INC) I UEANL USBR4 4.40 59.60 24.55 51.27 9.35 15.75		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.20	8.20	1							
2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1			I				4.40			51.27	9.35		15.75				
2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		Order Coordination for Habrard and Cab Lana are as the same			LIFANII	LICDMC		0.00	0.00	1							
2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2			-	1			6.06			45.36	6 71	-	15 75				
2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3			i i									t					
2 Wire Copper Unbundled Sub-Loop Distribution - Zone 4			İ	_										İ			
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			LIEE	LISBMC		g 20	g 20								
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1	1	UEF	UCS4X	5.10	79.49	44.45	51.27	9.35	1	15.75	1	1		1

IINRI	INDI E	D NETWORK ELEMENTS - Mississippi												Attach	ment: 2	Exhib	it. C
ONDO	NULL											Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec			Manual Svc	Manual Svc	Manual Svc
CATE	SORV	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)				Manually				
CAIL	JONI	RATE ELEMENTS	m	Zone	BC3	0300			KAILS(4)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
-	1						-	Nonrec	urring	Monrocurring	g Disconnect		l .	000	Rates(\$)		
-	-			-			Rec	First	Add'l	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	<u> </u>	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		2	UEF	UCS4X	9.11	79.49	44.45	51.27	9.35	SOWIEC	15.75	SUMAN	SOWAN	SUMAN	SUMAN
	<u> </u>																
<u> </u>	<u> </u>	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	ı		UEF	UCS4X	14.00	79.49	44.45	51.27	9.35		15.75				
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4		4	UEF	UCS4X	14.00	79.49	44.45	51.27	9.35		15.75				
																	i
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.20	8.20								,
	Unbun	dled Sub-Loop Modification															
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load															1
		Coil/Equip Removal per 2-W PR			UEF	ULM2X		176.80	5.13				15.75				1
		Unbundled Sub-loop Modification - 4-W Copper Dist Load															1
		Coil/Equip Removal per 4-W PR			UEF	ULM4X		176.80	5.13				15.75				1
1	1	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged		1]							l				
	1	Tap Removal, per PR unloaded		1	UEF	ULM4T		279.81	6.15	I	Ì		15.75	Ì	Ì		ı
	Unbun	dled Network Terminating Wire (UNTW)															
		Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.3366	30.55					15.75				
		k Interface Device (NID)															
		Network Interface Device (NID) - 1-2 lines			UENTW	UND12		43.84	28.90				15.75				
	1	Network Interface Device (NID) - 1-6 lines			UENTW	UND16		65.30	50.36	1	İ		15.75	İ	İ		
		Network Interface Device Cross Connect - 2 W		1	UENTW	UNDC2		5.94	5.94				15.75				
		Network Interface Device Cross Connect - 4W			UENTW	UNDC4		5.94	5.94				15.75				
SUB-I	OOPS	Network interface Device Gross Cormect 444			OLIVIV	ONDO		0.04	0.04				10.70				
OOD-L		pop Feeder		 													
	Sub-LC	USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
		Distribution Facility set-up			UDN,UCL,UDL,UDC	LICDEW		259.69					15.75				i
	<u> </u>	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair			UEA,	USBFW		259.09					15.75				
						LIODEY		00.77	22.77				45.75				i
	1	set-up			UDN,UCL,UDL,UDC	USBFX		22.77					15.75				
-	-	USL Feeder DS1 Set-up at DSX location, per DS1 termination		-	USL	USBFZ		534.46	11.30				15.75				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice		١.													i
		Grade - Zone 1		1	UEA	USBFA	7.98	93.23	56.50	54.45	13.51		15.75				,
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice		_													1
		Grade - Zone 2		2	UEA	USBFA	10.39	93.23	56.50	54.45	13.51		15.75				
		Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,															1
		Voice Grade - Zone 3		3	UEA	USBFA	16.11	93.23	56.50	54.45	13.51		15.75				1
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start Loop,															i
		Voice Grade - Zone 4		4	UEA	USBFA	28.37	93.23	56.50	54.45	13.51		15.75				1
		Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		18.19									1
1		Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice]	I								1			
L	<u></u>	Grade - Zone 1	<u></u>	1	UEA	USBFB	7.98	93.23	56.50	54.45	13.51	L	15.75	<u> </u>	<u> </u>		1
		Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice															
L	<u> </u>	Grade - Zone 2	<u></u>	2	UEA	USBFB	10.39	93.23	56.50	54.45	13.51	<u></u>	15.75	<u></u>	<u> </u>		<u> </u>
		Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice															
		Grade - Zone 3		3	UEA	USBFB	16.11	93.23	56.50	54.45	13.51		15.75				ı
		Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice															
	1	Grade - Zone 4		4	UEA	USBFB	28.37	93.23	56.50	54.45	13.51		15.75	Ì	Ì		ı
		Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		18.19									
	1	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,			İ					1	İ			İ	İ		
	1	Voice Grade - Zone 1		1	UEA	USBFC	7.98	93.23	56.50	54.45	13.51		15.75	Ì	Ì		ı
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
		Voice Grade - Zone 2		2	UEA	USBFC	10.39	93.23	56.50	54.45	13.51		15.75				ı
	1	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,		- -		·	.0.00	55.25	55.56	010	.5.51			†	 		
	1	Voice Grade - Zone 3		3	UEA	USBFC	16.11	93.23	56.50	54.45	13.51		15.75	Ì	Ì		ı
—	+	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,		۲	0=/1	505.0	10.11	30.23	30.30	54.45	13.31		15.75	 	 		i
	1	Voice Grade - Zone 4		4	UEA	USBFC	28.37	93.23	56.50	54.45	13.51		15.75	Ì	Ì		ı
	+	Order Coordination For Specified Conversion Time, per LSR		4	UEA	OCOSL	20.37	18.19	00.00	54.45	13.51		15.75				
-	1				UĽA	UUUSL		18.19		 							ı
1	1	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice		١,	Liea	LICDED	04.00	407.74	70.00	00.00	47.04		45.75	Ì	Ì		ı
├	1	Grade - Zone 1		1	UEA	USBFD	21.69	107.71	70.03	63.68	17.64		15.75				
		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice		_		LICDES	00.00	407.71	70.00	20.00	47.01		45.75				ı
	<u> </u>	Grade - Zone 2		2	UEA	USBFD	26.06	107.71	70.03	63.68	17.64		15.75				
1	1	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice		l _	l= .			,						Ì	Ì		ı
		Grade - Zone 3		3	UEA	USBFD	34.77	107.71	70.03	63.68	17.64	l	15.75	l	l		

UNDUNDLE	D NETWORK ELEMENTS - Mississippi			1	1						C C1	Cura Curti		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonre		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 4		4	UEA	USBFD	34.77	107.71	70.03	63.68	17.64		15.75				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		18.19									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice				HODEE	04.00	407.74	70.00	00.00	47.04		45.75				
	Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		1	UEA	USBFE	21.69	107.71	70.03	63.68	17.64		15.75				
	Grade - Zone 2		2	UEA	USBFE	26.06	107.71	70.03	63.68	17.64		15.75				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			ULA	USBI L	20.00	107.71	70.03	03.00	17.04		13.73				
	Grade - Zone 3		3	UEA	USBFE	34.77	107.71	70.03	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire Analog Voice Grade Loop-Start		Ŭ	02/1	005. 2	0	101111	. 0.00	00.00			10.10				1
	Loop - Zone 4		4	UEA	USBFE	34.77	107.71	70.03	63.68	17.64		15.75				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		18.19									
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	14.60	106.46	68.78	55.58	13.13		15.75				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2		2	UDN	USBFF	18.78	106.46	68.78	55.58	13.13		15.75				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3			UDN	USBFF	25.47	106.46	68.78	55.58	13.13		15.75				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 4		4	UDN	USBFF	41.41	106.46	68.78	55.58	13.13		15.75				ļ
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		18.19									
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDC	USBFS	14.60	106.46	68.78	55.58	13.13		15.75				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	18.78	106.46	68.78	55.58	13.13		15.75				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		3	UDC	USBFS	25.47	106.46	68.78	55.58	13.13		15.75				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		4	UDC	USBFS	41.41	106.46	68.78	55.58	13.13		15.75				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		2	USL	USBFG USBFG	55.19 100.03	101.97 101.97	64.29 64.29	63.68	17.64 17.64		15.75				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	USL USL	USBFG	183.66	101.97	64.29	63.68 63.68	17.64		15.75 15.75				
+	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 4		4	USL	USBFG	430.04	101.97	64.29	63.68	17.64		15.75				1
-	Order Coordination For Specified Conversion Time, Per LSR		-4	USL	OCOSL	430.04	18.19	04.23	05.00	17.04		13.73				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone			OOL	COOCE		10.10									1
	1		1	UCL	USBFH	5.88	84.27	46.59	53.14	10.70		15.75				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone															
	2		2	UCL	USBFH	5.21	84.27	46.59	53.14	10.70		15.75				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone															
	3		3	UCL	USBFH	4.40	84.27	46.59	53.14	10.70		15.75				
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 4		4	UCL	USBFH	3.63	84.27	46.59	53.14	10.70		15.75				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		18.19									
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	13.49	101.58	63.90	59.71	13.67		15.75				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2			UCL	USBFJ	10.96	101.58	63.90	59.71	13.67		15.75				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	8.59	101.58	63.90	59.71	13.67		15.75				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 4 Order Coordination For Specified Conversion Time, per LSR		4	UCL	USBFJ OCOSL	8.59	101.58 18.19	63.90	59.71	13.67		15.75			1	
+	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	22.89	101.97	64.29	63.68	17.64	1	15.75			1	
- 	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	-	2	UDL	USBFN	25.11	101.97	64.29	63.68	17.64		15.75			+	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	30.84	101.97	64.29	63.68	17.64	1	15.75			1	
1	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	41.05	101.97	64.29	63.68	17.64		15.75				†
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -				1			520	22,00							İ
	Zone 1		1	UDL	USBFO	22.89	101.97	64.29	63.68	17.64		15.75				
İ	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															1
	Zone 2		2	UDL	USBFO	25.11	101.97	64.29	63.68	17.64		15.75				<u> </u>
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -					_										
	Zone 3		3	UDL	USBFO	30.84	101.97	64.29	63.68	17.64		15.75				<u> </u>
1	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 4		4	UDL	USBFO	41.05	101.97	64.29	63.68	17.64		15.75			ļ	.
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		18.19								ļ	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		4	UDL	USBFP	22.89	101.97	64.29	63.68	17.64		15.75				
	Zone 1	-	1	UDL	USBFF	22.89	101.97	64.29	80.68	17.04		15.75		-	1	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2		2	UDL	USBFP	25.11	101.97	64.29	63.68	17.64		15.75				
-+	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -	 		JDL	OODER	20.11	101.97	04.29	03.00	17.04		13.73			1	
	Zone 3		3	UDL	USBFP	30.84	101.97	64.29	63.68	17.64		15.75				

UNBUND	DLED	NETWORK ELEMENTS - Mississippi												Attachi	ment: 2	Exhi	bit: C
CATEGOR		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -															
		Zone 4		4	UDL	USBFP	41.05	101.97	64.29	63.68	17.64		15.75				
		Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		18.19									
SUB-LOOI																	
Su		pp Feeder															
		Sub Loop Feeder - DS3 - Per Mile Per Month	ı		UE3	1L5SL	18.88										
		Sub Loop Feeder - DS3 - Facility Termination Per Month	- 1		UE3	USBF1	349.41	3,396.56	406.45	157.96	89.54		15.75				
		Sub Loop Feeder – STS-1 – Per Mile Per Month	ı		UDLSX	1L5SL	18.88										
		Sub Loop Feeder - STS-1 - Facility Termination Per Month	ı		UDLSX	USBF7	376.07	3,396.56	406.45	157.96	89.54		15.75				
		Sub Loop Feeder – OC-3 – Per Mile Per Month	ı		UDLO3	1L5SL	14.33										
		Sub Loop Feeder - OC-3 - Facility Termination Protection Per								1					Ì		
		Month	- 1		UDLO3	USBF5	58.63										
		Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3	USBF2	569.22	3,396.56	406.45	157.96	89.54	<u> </u>	15.75				
		Sub Loop Feeder - OC-12 - Per Mile Per Month	ı		UDL12	1L5SL	17.63										
		Sub Loop Feeder - OC-12 - Facility Termination Protection Per				l				1							
		Month	-		UDL12	USBF6	662.39										
		Sub Loop Feeder - OC-12 - Facility Termination Per Month	I		UDL12	USBF3	1,795.00	3,396.56	406.45	157.96	89.54		15.75				
		Sub Loop Feeder - OC-48 - Per Mile Per Month	ı		UDL48	1L5SL	57.83										
		Sub Loop Feeder - OC-48 - Facility Termination Protection Per															
		Month			UDL48	USBF9	331.52										
		Sub Loop Feeder - OC-48 - Facility Termination Per Month			UDL48	USBF4	1,545.00	3,581.56	406.45	157.96	89.54		15.75				
		Sub Loop Feeder - OC-12 Interface On OC-48	I		UDL48	USBF8	374.04	803.60	406.45	157.96	89.54		15.75				
UNBUNDL		OOP CONCENTRATION															
		Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	36367	327.30	327.30				15.75				
		Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	47.56	136.37	136.37				15.75				
		Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	397.35	327.30	327.30				15.75				
		Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	80.15	136.37	136.37				15.75				
		Unbundled Loop Concentration - DS1 Loop Interface Card			ULC	UCTCO	4.52	63.65	46.34	17.31	4.85		15.75				
		Unbundled Loop Concentration - ISDN Loop Interface (Brite				l											
		Card)			UDN	ULCC1	7.17	10.60	10.54	5.56	5.53		15.75				
		Unbundled Loop Concentration - UDC Loop Interface (Brite				l											
		Card)			UDC	ULCCU	7.17	10.60	10.54	5.56	5.53		15.75				
		Unbundled Loop Concentration2 Wire Voice-Loop Start or					4.00	40.00									
		Ground Start Loop Interface (POTS Card)			UEA	ULCC2	1.80	10.60	10.54	5.56	5.53		15.75				
		Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery	1					40					4===		Ì		
 -		Loop Interface (SPOTS Card)			UEA	ULCCR	10.66	10.60	10.54	5.56	5.53		15.75				
		Unbundled Loop Concentration - 4 Wire Voice Loop Interface	1												Ì		
 -		(Specials Card)			UEA	ULCC4	6.36	10.60	10.54	5.56	5.53		15.75				
		Unbundled Loop Concentration - TEST CIRCUIT Card		1	ULC	UCTTC	31.07	10.60	10.54	5.56	5.53	1	15.75		-		1
		Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop	1		UDL	ULCC7	0.40	40.00	40.54	5.50	5.53		45.75		Ì		
		Interface		1	UDL	ULUU/	9.42	10.60	10.54	5.56	5.53		15.75				
		Unbundled Loop Concentration - Digital 56 Kbps Data Loop			LIDI	111.005	0.40	40.00	40.54	5.50	F F2		45.75				
 		Interface Unbundled Loop Concentration - Digital 64 Kbps Data Loop			UDL	ULCC5	9.42	10.60	10.54	5.56	5.53		15.75				
					UDL	ULCC6	9.42	40.00	40.54	5.50	5.53		45.75				
LINE OTH		Interface ROVISIONING ONLY - NO RATE			UDL	ULCC6	9.42	10.60	10.54	5.56	5.53		15.75				
UNE OTHE					UENTW	UNDBX	0.00	0.00									
		NID - Dispatch and Service Order for NID installation UNTW Circuit Id Establishment, Provisioning Only - No Rate		1	UENTW	UENCE	0.00	0.00		+		1					1
		ONTIVE CITCUIT IN ESTABLISHMENT, PROVISIONING ONLY - NO Rate		1	UEANL,UEF,UEQ,U	OLINGE	0.00	0.00		+ +				-	-	-	1
	l,	Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN	0.00	0.00		1							
LINE OTHE		ROVISIONING ONLY - NO RATE		1	LIA1 AA	OINLOIN	0.00	0.00		+ +		1		1	1	1	1
ONE OTHE	_rx, r/r	CONDIGNATO ONE I - NO NATE	-	1			1			 		1			 		1
	ļ				UAL.UCL.UDC.UDL.					1							
	l,	Unbundled Contact Name, Provisioning Only - no rate	1		UDN,UEA,UHL,ULC	LINECN	0.00	0.00							Ì		
		Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no	-	1	SSIT, SEI I, OI IE, OEO	STALOIN	0.00	0.00		 		1			 		1
ı I		rate		1	UEA,UDN,UCL,UDC	USBEO	0.00	0.00							1		
		Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no		1	OL7, ODIN, OOL, ODC	טטטו ע	0.00	0.00		 							
	,	·		1	UEA,USL,UCL,UDL	USBFR	0.00	0.00				1			ĺ		
	l,	rate															

UNBUNDL	ED NETWORK ELEMENTS - Mississippi			1										ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled DS1 Loop - Expanded Superframe Format option -															
HICH CABAC	no rate CITY UNBUNDLED LOCAL LOOP			USL	CCOEF	0.00	0.00									
HIGH CAPAC	High Capacity Unbundled Local Loop - DS3 - Per Mile per				-											+
	month			UE3	1L5ND	11.20										
	High Capacity Unbundled Local Loop - DS3 - Facility			023	TESIND	11.20									1	+
	Termination per month			UE3	UE3PX	326.15	454.13	265.47	123.23	86.19		15.75				
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per															1
	month			UDLSX	1L5ND	11.20										
	High Capacity Unbundled Local Loop - STS-1 - Facility															
	Termination per month			UDLSX	UDLS1	338.55	454.13	265.47	123.23	86.19		15.75				
LOOP MAKE																
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).			UMK	UMKLW		24.12	24.12								
	Loop Makeup - Preordering With Reservation, per spare facility			UIVIK	UIVIKLVV		24.12	24.12								+
	queried (Manual).			UMK	UMKLP		25.58	25.58								
	Loop MakeupWith or Without Reservation, per working or			0	O I I I I I		20.00	20.00	İ							†
	spare facility queried (Mechanized)			UMK	PSUMK		0.6652	0.6652								
	DEDICATED TRANSPORT															
	E: INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	od - below DS3=one	month, DS3/	STS-1=four mo	nths									
INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			U1TVX	1L5XX	0.0098										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination			U1TVX	U1TV2	22.52	40.77	27.57	47.00	7.44		45.75				
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade			UTTVX	UTIVZ	22.52	40.77	27.57	17.26	7.11		15.75				+
	Rev Bat Per Mile per month			U1TVX	1L5XX	0.0098										
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat.			OTTVX	TEO/OX	0.0000									1	+
	Facility Termination			U1TVX	U1TR2	22.52	40.77	27.57	17.26	7.11		15.75				
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade															
	Per Mile per month			U1TVX	1L5XX	0.0098										
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade															
	- Facility Termination			U1TVX	U1TV4	19.79	40.77	27.57	17.26	7.11		15.75				
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile			LIATOV	1L5XX	0.0000										
	per month Interoffice Channel - Dedicated Transport - 56 kbps - Facility			U1TDX	ILSXX	0.0098										+
	Termination			U1TDX	U1TD5	15.68	40.78	27.57	17.26	7.11		15.75				
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile			OTTEX	01100	10.00	40.70	21.01	17.20	7.11		10.70			1	+
	per month			U1TDX	1L5XX	0.0098										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination			U1TDX	U1TD6	15.68	40.78	27.57	17.26	7.11		15.75				1
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per				41 = 3.07											
	month Interoffice Channel - Dedicated Tranport - DS1 - Facility			U1TD1	1L5XX	0.201										-
	Termination			U1TD1	U1TF1	57.33	89.79	82.28	16.86	14.90		15.75				
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			UTIDI	01111	37.33	05.75	02.20	10.00	14.50		13.73				+
	month			U1TD3	1L5XX	4.76										
	Interoffice Channel - Dedicated Transport - DS3 - Facility								†							†
	Termination per month			U1TD3	U1TF3	641.90	280.37	163.70	62.08	60.29		15.75				
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per															
ļļ	month	ļ	<u> </u>	U1TS1	1L5XX	4.76			ļ						ļ	
	Interoffice Channel - Dedicated Transport - STS-1 - Facility	1	1		LIATES						1	,				
1.00	Termination TRANSPORT	<u> </u>	ļ	U1TS1	U1TFS	644.21	280.37	163.70	62.08	60.29		15.75			ļ	
	AL CHANNEL - DEDICATED TRANSPORT E: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	a nerie	d - bal	DE2-one ment	Dealers 4 4	four months			 						}	
NOTE	Local Channel - Dedicated - 2-Wire Voice Grade	y perio	u - pel	ULDVX	ULDV2	14.91	194.22	33.36	37.79	3.30		15.75		-	 	+
 	Local Channel - Dedicated - 2-Wire Voice Grade Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat	 		ULDVX	ULDR2	14.91	194.22	33.36	37.79	3.30		15.75			1	
											l				1	+
	Local Channel - Dedicated - 4-Wire Voice Grade			UNDVX	ULDV4	15.99	194.66	33.80	38.27	3.78		15.75				

UNBUNDL	ED NETWORK ELEMENTS - Mississippi			1								,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	Local Channel - Dedicated - DS1 - Zone 2		2	ULDD1	ULDF1	35.99	178.50	154.61	22.89	15.74		15.75				
	Local Channel - Dedicated - DS1 - Zone 3		3	ULDD1	ULDF1	221.63	178.50	154.61	22.89	15.74		15.75				
	Local Channel - Dedicated - DS1 - Zone 4		4	ULDD1	ULDF1	221.63	178.50	154.61	22.89	15.74						
	Local Channel - Dedicated - DS3 - Per Mile per month			ULDD3 ULDD3	1L5NC ULDF3	9.66 413.87	454.40	205 47	123,23	00.40		45.75			-	
	Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1- Per Mile per month			ULDD3 ULDS1	1L5NC	413.87 9.66	454.13	265.47	123.23	86.19		15.75				
	Local Channel - Dedicated - STS-1 - Per Mile per month Local Channel - Dedicated - STS-1 - Facility Termination			ULDS1	ULDFS	408.02	454.13	265.47	123.23	86.19		15.75			-	-
DARK FIBER		-		ULDST	ULDFS	406.02	404.10	205.47	123.23	00.19		15.75				
DAKK FIBER	Dark Fiber. Four Fiber Strands. Per Route Mile or Fraction								1							
	Thereof per month - Local Channel			UDF	1L5DC	59.95										
	NRC Dark Fiber - Local Channel			UDF	UDFC4	00.00	642.79	138.67	326.97	203.85		15.75				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction		1		32.0.		0.2.70	.00.07	323.07	200.00		.5.70			1	1
	Thereof per month - Interoffice Channel			UDF	1L5DF	28.27									1	
	NRC Dark Fiber - Interoffice Channel		1	UDF	UDF14		642.79	138.67	326.97	203.85		15.75				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Loop		<u>L</u>	UDF	1L5DL	59.95			<u> </u>					<u> </u>	<u></u>	<u> </u>
	NRC Dark Fiber - Local Loop			UDF	UDFL4		642.79	138.67	326.97	203.85		15.75				
8XX ACCESS	S TEN DIGIT SCREENING															
	8XX Access Ten Digit Screening, Per Call			OHD		0.0006216										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX															
	Number Reserved			OHD	N8R1X		2.60	0.44				15.75				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O															
	POTS Translations			OHD			5.97	0.81	4.60	0.54		15.75				
	8XX Access Ten Digit Screening, Per 8XX No. Established With			0.15												
	POTS Translations			OHD	N8FTX		5.97	0.81	4.60	0.54		15.75				
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number			OHD	N8FCX		2.60	1.30				15.75				
	8XX Access Ten Digit Screening, Multiple InterLATA CXR			OHD	NOFCX		2.60	1.30	1			15.75				
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		3.04	1.74				15.75				
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		3.04	0.44	1			15.75				
	8XX Access Ten Digit Screening, Change Charge Fer Request			OLID	INOI AX		3.04	0.44				13.73				
	Features			OHD	N8FDX		2.60					15.75				
	1 Galares			OTID	THOI DX		2.00					10.70				
	8XX Access Ten Digit Screening, w/ 8FL No. Delivery, per query			OHD		0.0006216										
	8XX Access Ten Digit Screening, w/ POTS No. Delivery, per															
	query			OHD		0.0006216										
LINE INFORM	MATION DATA BASE ACCESS (LIDB)															
	LIDB Common Transport Per Query			OQT		0.0000197										
	LIDB Validation Per Query			OQU		0.0137053										
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		34.52	34.52	42.33	42.33		15.75				
SIGNALING	(CCS7)						, i		ļ					ļ	ļ	
<u> </u>	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	132.21			ļļ						1	
\vdash	CCS7 Signaling Usage, Per TCAP Message		<u> </u>	UDB	TDD	0.0000597			10.5-	10.5-		2				
\vdash	CCS7 Signaling Connection, Per link (A link)	1	<u> </u>	UDB	TPP++	16.55	35.74	35.74	16.53	16.53		15.75				
	CCS7 Signaling Connection, Per link (B link) (also known as D			LIDD	TDD	40	05 = 1	05 = 1	40 =0	40 =0		45		1	I	I
\vdash	link)	1	 	UDB UDB	TPP++	16.55 0.0000149	35.74	35.74	16.53	16.53		15.75		 	 	
\vdash	CCS7 Signaling Usage, Per ISUP Message CCS7 Signaling Usage Surrogate, per link per LATA	1	 	UDB	STU56	0.0000149 683.55			 					 	 	
\vdash	CCS7 Signaling Osage Surrogate, per link per LATA CCS7 Signaling Point Code, per Originating Point Code	1	 	סטט	31000	083.05			 			 		-		
	Establishment or Change, per STP affected			UDB	CCAPO		29.18	29.18	35.78	35.78		15.75		1	I	I
E911 SERVIC		1	!	טטט	COAFO	1	29.18	29.18	33.18	33.78		15.75		1	t	t
LOTT OLIVIC	Local Channel - Dedicated - 2-wr Voice Grade	1	1		+	14.91	194.22	33.36	37.79	3.30		15.75			-	-
 	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile	1	†		+	0.0098	134.22	33.30	31.19	5.50		10.70		 	t	t
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility	1	!			5.0000			† †			 		 	I	I
	Termination					22.52	40.77	27.57	17.26	7.11		15.75		1	I	
	Local Channel - Dedicated - DS1 - Zone 1		1			36.83	178.50	154.61	22.89	15.74		15.75			1	1
	Local Channel - Dedicated - DS1 - Zone 2	1				35.99	178.50	154.61	22.89	15.74		15.75				
	Local Channel - Dedicated - DS1 - Zone 3	1	i –			221.63	178.50	154.61	22.89	15.74		15.75			1	
 	Local Channel - Dedicated - DS1 - Zone 4	1				221.63	178.50	154.61	22.89	15.74		15.75			1	1

UNBUNDLE	D NETWORK ELEMENTS - Mississippi												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	Interoffice Transport - Dedicated - DS1 Per Mile						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-	Interoffice Transport - Dedicated - DST Per Mile					0.2010										<u> </u>
	Interoffice Transport - Dedicated - DS1 Per Facility Termination					57.33	89.79	82.28	16.86	14.90		15.75				
CALLING NAM	I ME (CNAM) SERVICE											15.75		-	-	
CALLING NAM	CNAM For DB Owners - Service Establishment			OQV			23.09	23.09	21.23	21.23		15.75				
	CNAM For Non DB Owners - Service Establishment			OQV			23.09	23.09	21.23	21.23		15.75				
	CNAM For DB Owners - Service Provisioning With Point Code															
	Establishment			OQV			996.62	737.08	270.49	198.89		15.75				
	CNAM For Non DB Owners - Service Provisioning With Point			001/	1				670.05			,				
 	Code Establishment	1		OQV OQV	+	0.0040004	344.32	246.56	276.85	198.89		15.75		 	1	
\vdash	CNAM for DB Owners, Per Query CNAM for Non DB Owners, Per Query	 		OQV	+	0.0010231 0.0010231								 	 	}
LNP Query Se		 		OQ V	+	0.0010231								 	 	
	LNP Charge Per query	1		OQV	+	0.0008477			 					—	-	<u> </u>
	LNP Service Establishment Manual					0.0000	12.59	12.59	11.58	11.58		15.75				
	LNP Service Provisioning with Point Code Establishment						596.94	304.96	270.49	198.89		15.75				
OPERATOR C.	ALL PROCESSING															
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.20										
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24										
	Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20										
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB					0.20										
INWARD OPE	RATOR SERVICES															
	Inward Operator Services - Verification, Per Minute					1.15										
	Inward Operator Services - Verification and Emergency Interrupt - Per Minute					1.15										
	OPERATOR CALL PROCESSING															
Facility	y based CLEC															
	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				15.75				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN				CBAOL		500.00	500.00				15.75				
UNEP																
	Recording of Custom Branded OA Announcement Loading of Custom Branded OA Announcement per shelf/NAV	ļ			+		7,000.00	7,000.00				15.75		1	1	1
	per OCN						500.00	500.00				15.75				
DIRECTORY A	SSISTANCE SERVICES						000.00	000.00				10.70				
	TORY ASSISTANCE ACCESS SERVICE															
	Directory Assistance Access Service Calls, Charge Per Call					0.275										
DIREC	TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (I Directory Assistance Call Completion Access Service (DACC),	DACC)														
	Per Call Attempt	L				0.10					<u></u>			<u> </u>	<u> </u>	
	SSISTANCE SERVICES									•						
DIREC	TORY ASSISTANCE DATA BASE SERVICE (DADS)															
\vdash	Directory Assistance Data Base Service Charge Per Listing	ļ			DDCCE	0.04										ļ
DRANDING 5	Directory Assistance Data Base Service, per month DIRECTORY ASSISTANCE	1		1	DBSOF	150.00								 	1	
	y Based CLEC			 	+									-	-	}
Facility	Recording and Provisioning of DA Custom Branded Announcement			AMT	CBADA		6,000.00	6,000.00				15.75				
 	Loading of Custom Branded Announcement per Switch	-		AMT	CBADA		1,170.00	1,170.00	 			15.75		 		
UNEP		 		7 WVI I	ODADO		1,170.00	1,170.00	+			13.73		 	t	
O.V.E.F	Recording of DA Custom Branded Announcement			<u> </u>	1		3,000.00	3,000.00				15.75				
	Loading of DA Custom Branded Announcement per Switch per						2,222.00	2,222.00	i i							İ
1 1	OCN	l	l				1,170.00	1,170.00	1			15.75				

UNBUNDLE	D NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonred		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Selective Routing Per Unique Line Class Code Per Request Per															
	Switch				USRCR		85.19	85.19	14.19	14.19		15.75				
VIRTUAL COL									0.51							
	Virtual Collocation - Application Cost			AMTFS	EAF		1,212.25		0.51			15.75				
	Virtual Collocation - Cable Installation Cost, per cable			AMTES	ESPCX	5.74	926.27		22.62			15.75				
	Virtual Collocation - Floor Space, per sq. ft.			AMTES	ESPVX	5.74										
	Virtual Collocation - Power, per fused amp			AMTFS	ESPAX	7.33										
	Virtual Collocation - Cable Support Structure, per entrance			AMTEC	FOROY	45.04										
	cable			AMTFS	ESPSX	15.24										
	Virtual Collocation - 2-wire Cross Connects (loop)			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, AMTFS, UDL, UNCVX, UNCDX, UNCNX	UEAC2	0.0268	12.37	11.87	6.04	5.45		15.75				
				AMTFS, UAL, UDN,												
	Virtual Collocation - 4-wire Cross Connects (loop)			UNCVX, UNCDX	UEAC4	0.0536	12.47	11.94	6.59	5.91		15.75				
	Virtual Collocation - 2-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF AMTFS,UDL12,	CNC2F	2.91	21.01	15.29	7.61	6.10		15.75				
	Virtual Collocation - 4-Fiber Cross Connects			UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC4F	5.82	25.70	19.97	10.01	8.50		15.75				
	Virtual Collocation - Special Access & UNE, cross-connect per DS1			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL, UNLD1	CNC1X	1.14	22.16	16.02	6.60	5.97		15.75				
	Virtual collocation - Special Access & UNE, cross-connect per DS3			USL,ULC,AMTFS,U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	14.49	21.01	15.29	7.61	6.10		15.75				
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			AMTFS	VE1CB	0.0025										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax															
	Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable		 	AMTFS	VE1CD	0.0037										1
	Support Structure,per cable			AMTFS	VE1CC		534.65					15.75				
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable			AMTFS	VE1CE		534.65					15.75				
\vdash	Virtual Collocation Cable Records - per request	 	1	AMTFS	VE1CE VE1BA		763.69	763.69	133.77	133.77	1	15.75		1	 	}
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable	 	 	, uviii O	VEIDA		103.09	703.09	133.11	133.77	1			1	 	1
	record			AMTFS	VE1BB		328.81	328.81	190.22	190.22						
	Virtual Collocation Cable Records - VG/DS0 Cable, per each															
	100 pair			AMTFS	VE1BC		4.84	4.84	5.93	5.93						
	Virtual Collocation Cable Records - DS1, per T1TIE			AMTFS	VE1BD		2.27	2.27	2.78	2.78						
	Virtual Collocation Cable Records - DS3, per T3TIE			AMTFS	VE1BE		7.92	7.92	9.72	9.72						
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber records			AMTFS	VE1BF		84.98	84.98	77.58	77.58						
 	Virtual collocation - Security Escort - Basic, per half hour	1	1	AMTFS	SPTBX		17.02	10.79	11.58	11.58	1	15.75			1	1
	Virtual collocation - Security Escort - Basic, per half hour	-	!	AMTFS	SPTOX	-	22.17	13.94	 			15.75		-		1
	Ivilual conocador - Security Escort - Overtime, per fidil flour	1	1	AWITS	SPTPX		22.17	17.08	1		1	15.75	l		l .	

UNBUNDLI	ED NETWORK ELEMENTS - Mississippi						·							ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	Virtual collocation - Maintenance in CO - Basic, per half hour			AMTFS	CTRLX		First 28.09	Add'I 10.79	First	Add'l	SOMEC	SOMAN 15.75	SOMAN	SOMAN	SOMAN	SOMAN
+	Virtual collocation - Maintenance in CO - Basic, per hair nour			AMIFS	CIRLX		28.09	10.79				15.75				-
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		36.69	13.94				15.75				
	Virtual collocation - Maintenance in CO - Premium per half hour			AMTFS	SPTPM		45.28	17.08				15.75				
VIRTUAL CO																
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-															
	Wire Analog - Res Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-			UEPSR	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			OLI OI	VETICE	0.0200	12.07	11.07	0.04	0.40		10.70				
	Voice Grade PBX Trunk - Res	<u> </u>		UEPSE	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire															
\vdash	Analog Bus Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire			UEPSB	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 2-wire Cross Connect, Exchage Port 2-wire			UEPSX	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			OLI OX	VETIVE	0.0200	12.57	11.07	0.04	3.43		10.73				
	ISDN			UEPTX	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire															
	ISDN DS1			UEPEX	VE1R4	0.0536	12.47	11.94	6.59	5.91		15.75				
VIRTUAL CO	Virtual Collocation-2 Wire Cross Connects (Loop) for Line															
	Splitting			UEPSR, UEPSB	VE1LS	0.0268	12.37	11.87	6.04	5.45		15.75				
PHYSICAL CO	DLLOCATION			02. 0.t., 02. 0B	12.20	0.0200	12.01	11.01	0.01	0.10		10.10				
	Physical Collocation-2 Wire Cross Connects (Loop) for Line															
	Splitting			UEPSR, UEPSB	PE1LS	0.0288	12.37	11.87	6.04	5.45		15.75				
AIN SELECTI	VE CARRIER ROUTING Regional Service Establishment			SRC	SRCEC		404 005 40		8,640.51			15.75				ļ
-	End Office Establishment			SRC	SRCEO		101,685.12 167.49	167.49	1.71	1.71		15.75				
	Query NRC, per query			SRC	OROLO	0.0030502	101.40	107.40	1.71			10.70				
AIN - BELLSO	OUTH AIN SMS ACCESS SERVICE															
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			A1N	CAMSE		39.67	39.67	40.92	40.92		15.75				
	AIN ONO A company of the Design of the Division Birlion				OAMBB		7.07	7.07	0.44	0.44		45.75				
-	AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access			A1N A1N	CAMDP CAM1P		7.87 7.87	7.87 7.87	9.14 9.14	9.14 9.14		15.75 15.75				ļ
	AIN SMS Access Service - User Identification Codes - Per User			Ally	OAWIII		7.07	7.07	3.14	3.14		13.73				
	ID Code			A1N	CAMAU		35.21	35.21	27.21	27.21		15.75				
	AIN SMS Access Service - Security Card, Per User ID Code,															
	Initial or Replacement			A1N	CAMRC	0.0004	42.13	42.13	11.78	11.78		15.75				_
-	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes) AIN SMS Access Service - Session, Per Minute					0.0021 0.5649										ļ
	AIN SMS Access Service - Company Performed Session, Per					0.3043										
	Minute					0.8393										
AIN - BELLSO	OUTH AIN TOOLKIT SERVICE															
	AIN Toolkit Service - Service Establishment Charge, Per State,															
	Initial Setup AIN Toolkit Service - Training Session, Per Customer			CAM	BAPSC BAPVX		39.67 4,226.54	39.67 4,226.54	40.92	40.92		15.75 15.75				
	AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DAPVA		4,226.54	4,220.54				15.75				
	DN, Term. Attempt				BAPTT		7.87	7.87	9.14	9.14		15.75				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
\vdash	DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPTD		7.87	7.87	9.14	9.14		15.75				<u> </u>
	DN, Off-Hook Immediate				BAPTM		7.87	7.87	9.14	9.14		15.75				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per			1				7.01	5	J.14						
	DN, 10-Digit PODP				BAPTO		34.67	34.67	14.44	14.44		15.75				<u> </u>
1 1	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per	l			DARTO		04.07	04.07				45				
	DN, CDP	1		<u> </u>	BAPTC		34.67	34.67	14.44	14.44		15.75		<u> </u>		1

UNBUND																
	LED NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
1													Incremental			
											Submitted	Submitted		Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															2.00 .00	2.007.001
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, Feature Code				BAPTF		34.67	34.67	14.44	14.44		15.75				
	AIN Toolkit Service - Query Charge, Per Query					0.0535577										
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit															
	Subscription, Per Node, Per Query					0.0063509										
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access															
	Account, Per 100 Kilobytes					0.06										
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
	Subscription			CAM	BAPMS	11.11	7.87	7.87	5.54	5.54		15.75				
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
$oxed{oxed}$	Subscription			CAM	BAPLS	2.71	8.71	8.71				15.75				
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service										<u> </u>	[
	Subscription			CAM	BAPDS	8.48	7.87	7.87	5.54	5.54		15.75				
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit									<u> </u>				1		
	Service Subscription			CAM	BAPES	0.09	8.71	8.71				15.75				
	EXTENDED LINK (EELs)															
	E: New Density Zone 1 EELs are available in the following MSA					Atlanta, Ga; Nev	v Orleans, LA,									
	E: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salen															
	E: In all states, EEL network elements shown below also apply												UNEs.(Non-re	curring rates	do not apply	r.)
	E: In All States the EEL network elements apply to ordinarily co				itch As Is Ch	arge.) When or	dering ordinar	ily combined r	network elemer	nts, Non-recuri	ring rates de	o apply.				
2-W	IRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 IN	TEROFF	ICE TR	ANSPORT (EEL)												
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport															
	Combination - Zone 1		1	UNCVX	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
	Transport Combination - Zone 2		2	UNCVX	UEAL2	18.75	105.96	68.28	52.82	10.37		15.75				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
	Transport Combination - Zone 3		3	UNCVX	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75				
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport															
	Combination - Zone 4		4	UNCVX	UEAL2	45.72	105.96	68.28	52.82	10.37		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															
	per month			UNC1X	1L5XX	0.1813										
	Interoffice Transport - Dedicated - DS1 combination - Facility															
	Termination per month			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
	DS1 Channelization System Per Month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.5737	6.62	4.74								
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1															
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75				
	Each Additional 2-Wire VG Loop(SL2) in the same DS1															
l	Interoffice Transport Combination - Zone 2	<u></u>	2	UNCVX	UEAL2	18.75	105.96	68.28	52.82	10.37	<u> </u>	15.75		<u> </u>		
	Each Additional 2-Wire VG Loop(SL2) in the same DS1															
	Interoffice Transport Combination - Zone 3	<u> </u>	3	UNCVX	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75	<u> </u>			
	Each Additional 2-Wire VG Loop(SL2) in the same DS1														_	
l	Interoffice Transport Combination - Zone 4		4	UNCVX	UEAL2	45.72	105.96	68.28	52.82	10.37	<u> </u>	15.75		<u> </u>		
	Voice Grade COCI - DS1 to DS0 Channel System combination	-[
	per month			UNCVX	1D1VG	0.5737	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As	-														
l	Is Charge	<u></u>		UNC1X	UNCCC	<u> </u>	5.63	5.63	7.20	7.20	<u> </u>	15.75		<u> </u>		
4-W	IRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 IN	TEROFF	ICE TR	ANSPORT (EEL)												
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1	<u> </u>	1	UNCVX	UEAL4	27.47	132.27	94.59	60.68	14.64	<u> </u>	15.75	<u> </u>			
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice														_	
	Transport Combination - Zone 2		2	UNCVX	UEAL4	38.26	132.27	94.59	60.68	14.64		15.75				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
1 1	Transport Combination - Zone 3		3	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
1 1	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
 	i iist 4-viile Analog voice Grade Loop iii a Do i interonice															
	Transport Combination - Zone 4		4	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
			4	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				

CHECINDLE	ED NETWORK ELEMENTS - Mississippi	1	1	1							Svo Order	Suo Orda-	Attachr Incremental		Incremental	bit: C Incremental
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Charge -	Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
	Channelization - Channel System DS1 to DS0 combination Per			UNCIA	OTIFI	31.72	09.79	02.20	10.00	14.50		13.73				
	Month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	Voice Grade COCI - DS1 to DS0 Channel System combination -															
	per month Additional 4-Wire Analog Voice Grade Loop in same DS1			UNCVX	1D1VG	0.5737	6.62	4.74				15.75			-	
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	27.47	132.27	94.59	60.68	14.64		15.75				
	Additional 4-Wire Analog Voice Grade Loop in same DS1		<u> </u>	0.1017	02/121	2	102.21	0 1.00	00.00			10.70				
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	38.26	132.27	94.59	60.68	14.64		15.75				
	Additional 4-Wire Analog Voice Grade Loop in same DS1															
	Interoffice Transport Combination - Zone 3 Additional 4-Wire Analog Voice Grade Loop in same DS1		3	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	Interoffice Transport Combination - Zone 4		4	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	Voice Grade COCI - DS1 to DS0 Channel System combination -															
	per month			UNCVX	1D1VG	0.5737	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-			LINGAY	1111000		5.00	5.00	7.00	7.00		45.75				
4-WID	Is Charge E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INITED	CEICE	UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75			1	
4-WIK	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	JEFICE	TRANSFORT (EEL)	'											
	Transport Combination - Zone 1		1	UNCDX	UDL56	27.44	126.53	88.85	60.68	14.64		15.75				
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 2		2	UNCDX	UDL56	34.55	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	40.76	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice		3	UNCDA	UDL36	40.76	120.55	00.00	60.66	14.04		13.73				
	Transport Combination - Zone 4		4	UNCDX	UDL56	32.25	126.53	88.85	60.68	14.64		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															
	Per Month			UNC1X	1L5XX	0.1813						15.75				
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Month			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
	Channelization - Channel System DS1 to DS0 combination Per			UNCIA	01111	31.72	09.79	02.20	10.00	14.90		13.73				
	Month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per															
	month (2.4-64kbs)			UNCDX	1D1DD	1.22	6.62	4.74				15.75				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		4	UNCDX	UDL56	27.44	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			UNCDX	ODESO	21.44	120.55	88.83	00.00	14.04		13.73				
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	34.55	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	40.76	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 4		4	UNCDX	UDL56	32.25	126.53	88.85	60.68	14.64		15.75				
	OCU-DP COCI (data) - DS1 to DS0 Channel System -		7	ONODA	ODESO	32.23	120.55	00.03	00.00	14.04		10.70				
	combination per month (2.4-64kbs)			UNCDX	1D1DD	1.22	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-															
4 14/10	Is Charge E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	SECIOE	UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75				
4-WIR	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	JFFICE	TRANSPORT (EEL)	'										-	-
	Transport Combination - Zone 1		1	UNCDX	UDL64	27.44	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 2		2	UNCDX	UDL64	34.55	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	40.76	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		3	UNCDA	UDL04	40.76	120.53	00.85	80.08	14.04		15.75				
	Transport Combination - Zone 4		4	UNCDX	UND64	32.25	126.53	88.85	60.68	14.64		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															
	Per Month	l	1	UNC1X	1L5XX	0.1813			1		1				1	1

UNBUND	DLFD	NETWORK ELEMENTS - Mississippi												Attachi	ment: 2	Fyhil	oit: C
J J. 10	Ī											Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGOR	ŀΥ	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			""											Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
				<u> </u>			Rec	Nonred		Nonrecurring					Rates(\$)		
		standing Transport Dedicated DC4 combination Facility						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		nteroffice Transport - Dedicated - DS1 combination - Facility [Fermination Per Month]			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
\vdash		Channelization - Channel System DS1 to DS0 combination Per			UNCIX	UTIFT	51.72	89.79	82.28	10.80	14.90		15.75				
		Month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
\vdash		DCU-DP COCI (data) - DS1 to DS0 Channel System			UNCIX	IVIQT	102.00	91.37	02.94	10.67	10.10		13.73				
		combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.22	6.62	4.74				15.75				
		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1			0110271	10.00		0.02					10.70				
		nteroffice Transport Combination - Zone 1		1	UNCDX	UDL64	27.44	126.53	88.85	60.68	14.64		15.75				
		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1															
	ı	nteroffice Transport Combination - Zone 2		2	UNCDX	UDL64	34.55	126.53	88.85	60.68	14.64		15.75				
		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1															
		nteroffice Transport Combination - Zone 3		3	UNCDX	UDL64	40.76	126.53	88.85	60.68	14.64		15.75				
		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1				1									1		
		nteroffice Transport Combination - Zone 4	ļ	4	UNCDX	UDL64	32.25	126.53	88.85	60.68	14.64		15.75		1		
		OCU-DP COCI (data) - DS1 to DS0 Channel System															
		combination - per month (2.4-64kbs)	<u> </u>		UNCDX	1D1DD	1.22	6.62	4.74	-			15.75	 	-	 	ļ
		Nonrecurring Currently Combined Network Elements Switch -As-	1		LINICAV	LINICCO		5.00	F 00	7.00	7.00		45.75				
H 4 V		s Charge DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTI	EBOEEL	CE TD/	UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75		-		
4-1		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	EKUFFI	CE IRA	ANSPORT (EEL)												
		Fransport - Zone 1		1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				
		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice		<u>'</u>	ONOTA	OOLXX	73.00	200.90	130.43	40.10	12.07		13.73				
		Fransport - Zone 2		2	UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75				
		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice															
		Fransport - Zone 3		3	UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75				
		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice															
		Fransport - Zone 4		4	UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75				
		nteroffice Transport - Dedicated - DS1 combination - Per Mile															
		Per Month			UNC1X	1L5XX	0.1813										
		nteroffice Transport - Dedicated - DS1 combination - Facility															
		Termination Per Month			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
		Nonrecurring Currently Combined Network Elements Switch -As-	-		LINGAY	1111000		5.00	5.00	7.00	7.00		45.75				
4 V		s Charge DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTI	EBOEEL	CE TD/	UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75		-		
4-1			EKUFFI	CE IRA	ANSPORT (EEL)												
		First DS1Loop in DS3 Interoffice Transport Combination - Zone	1	1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75	1		1	
 -		First DS1Loop in DS3 Interoffice Transport Combination - Zone	 	+-	OI4O IV	JJLAA	79.08	200.93	100.45	40.10	12.07		15.75	 	t	 	
	Į.)	1	2	UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75	1	I	1	
	- li	First DS1Loop in DS3 Interoffice Transport Combination - Zone	<u> </u>	Ť			.20.00	200.00	.00.40	.0.10	.2.07				1		
		3		3	UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75		1		
	- l i	First DS1Loop in DS3 Interoffice Transport Combination - Zone	1	Ė	-					1							
	4	1	1	4	UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75	1	I	1	
		nteroffice Transport - Dedicated - DS3 combination - Per Mile															
		Per Month	<u> </u>		UNC3X	1L5XX	4.29								<u> </u>		<u> </u>
		nteroffice Transport - Dedicated - DS3 - Facility Termination per					_]]	
		month			UNC3X	U1TF3	641.90	280.37	163.70	62.08	60.29		15.75				
<u> </u>		DS3 to DS1 Channel System combination per month	ļ		UNC3X	MQ3	107.85	179.17	94.52	34.30	32.82		15.75	ļ	1	ļ	
		DS3 Interface Unit (DS1 COCI) combination per month	ļ		UNC1X	UC1D1	12.96	6.62	4.74				15.75				
		Additional DS1Loop in DS3 Interoffice Transport Combination -	1		LINICAY	LICL VV	70.00	050.00	450.45	40.40	10.0=		45 7-	1	I	1	
\vdash		Zone 1	1	1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75	 	 	 	-
		Additional DS1Loop in DS3 Interoffice Transport Combination -	1	2	UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75	1	I	1	
$\vdash \vdash$		Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination -	 		OINO IV	USLAA	129.38	253.93	158.45	46.10	12.07		15.75	-		-	
		Additional DS (Loop in DS3 interoffice Transport Combination -	1	3	UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75	1	I	1	
		Additional DS1Loop in DS3 Interoffice Transport Combination -	 	3	014017	JULAA	200.14	200.30	150.45	40.10	12.07		13.73	1	t	1	
		Zone 4	1	4	UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75	1	I	1	
\vdash		DS3 Interface Unit (DS1 COCI) combination per month	1		UNC1X	UC1D1	12.96	6.62	4.74	40.10	12.07	1	15.75	 	I	 	
\vdash		Nonrecurring Currently Combined Network Elements Switch -As-	1			30.51	12.00	0.02	7.17	I		1	10.70	 	I	 	
		s Charge	1		UNC3X	UNCCC		5.63	5.63	7.20	7.20		15.75		1		

ONDONDL	LED NETWORK ELEMENTS - Mississippi	1	1	1							Cup Cade	Cup Code	Attachr			bit: C
CATEGORY	7 RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			1	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svo Order vs.
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-WI	IRE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	TEROFF	ICE TE	RANSPORT (EEL)												
	2-WireVG Loop used with 2-wire VG Interoffice Transport		1	LINCVO	LIEVIO	40.00	105.00	CO 00	50.00	40.07		45.75				
	Combination - Zone 1 2-WireVG Loop used with 2-wire VG Interoffice Transport		1	UNCVX	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75			-	
	Combination - Zone 2		2	UNCVX	UEAL2	18.75	105.96	68.28	52.82	10.37		15.75				
	2-WireVG Loop used with 2-wire VG Interoffice Transport			ONOVA	OL/ LL	10.70	100.00	00.20	02.02	10.07		10.70				1
	Combination - Zone 3		3	UNCVX	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75				
	A.1.2 2-WireVG Loop used with 2-wire VG Interoffice Transport															1
	Combination - Zone 4		4	UNCVX	UEAL2	45.72	105.96	68.28	52.82	10.37		15.75				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per															
	Mile Per Month Interoffice Transport - Dedicated - 2- Wire Voice Grade			UNCVX	1L5XX	0.00088										
	combination - Facility Termination per month			UNCVX	U1TV2	20.32	40.77	27.57	17.26	7.11		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-			ONOVA	OTIVE	20.02	40.77	21.01	17.20	7.11		10.70				+
	Is Charge			UNCVX	UNCCC		5.63	5.63	7.20	7.20		15.75				
4-WI	IRE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE IN	TEROFF	ICE TE													
	4-WireVG Loop used with 4-wire VG Interoffice Transport															
	Combination - Zone 1		1	UNCVX	UEAL4	27.47	132.27	94.59	60.68	14.64		15.75				
	4-WireVG Loop used with 4-wire VG Interoffice Transport		_		l											
	Combination - Zone 2		2	UNCVX	UEAL4	38.26	132.27	94.59	60.68	14.64		15.75				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	4-WireVG Loop used with 4-wire VG Interoffice Transport		3	UNCVA	ULAL4	30.03	132.21	54.55	00.08	14.04		13.73			1	+
	Combination - Zone 4		4	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per															
	Mile Per Month			UNCVX	1L5XX	0.00088										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade															
	combination - Facility Termination per month			UNCVX	U1TV4	17.86	40.77	27.57	17.26	7.11		15.75				<u> </u>
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	1		UNCVX	UNCCC		5.63	5.63	7.20	7.20		15.75				
DS3	IS Charge B DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	F TPA	NSDOE		UNCCC		5.63	5.03	7.20	7.20		15.75				
500	High Capacity Unbundled Local Loop - DS3 combination - Per	I IIIA	101 01		1											+
	Mile per month			UNC3X	1L5ND	11.20										
	High Capacity Unbundled Local Loop - DS3 combination -															
	Facility Termination per month			UNC3X	UE3PX	252.17	454.13	265.47	123.23	86.19		15.75				
	Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	4.29										
	Interoffice Transport - Dedicated - DS3 combination - Facility			LINICOV	U1TF3	641.90	200.27	100.70	CO 00	00.00		45.75				
-	Termination per per month Nonrecurring Currently Combined Network Elements Switch -As-			UNC3X	UIIF3	641.90	280.37	163.70	62.08	60.29		15.75				+
	Is Charge			UNC3X	UNCCC		5.63	5.63	7.20	7.20		15.75				
STS	51 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROF	FICE TF	RANSP		0.1000		0.00	0.00	1.20	7120		10.70			İ	†
	High Capacity Unbundled Local Loop - STS1 combination - Per															1
	Mile per month			UNCSX	1L5ND	11.20										<u> </u>
	High Capacity Unbundled Local Loop - STS1 combination -						,									
	Facility Termination per month			UNCSX	UDLS1	264.35	454.13	265.47	123.23	86.19		15.75				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile per month		1	UNCSX	1L5XX	4.29										
	Interoffice Transport - Dedicated - STS1 combination - Facility	1	 	51400A	ILUAA	4.29					-				t	+
	Termination per month			UNCSX	U1TFS	644.21	280.37	163.70	62.08	60.29		15.75			1	
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNCSX	UNCCC		5.63	5.63	7.20	7.20		15.75				
2-WI	IRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL)	ļ												
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination	1		LINICALY	LIALOV	04.04	447.04	70.00	50.00	40.07		45.75				
	Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination	 	1	UNCNX	U1L2X	21.01	117.61	79.92	52.82	10.37		15.75			 	
	Transport - Zone 2		2	UNCNX	U1L2X	27.59	117.61	79.92	52.82	10.37		15.75				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination	<u> </u>		5.1511/1	S ILLA	27.55	117.01	10.02	52.02	10.07		10.70			1	
		1	3	UNCNX	U1L2X	37.34	117.61	79.92	52.82	10.37	1	15.75			l .	1

CIADOIADE	ED NETWORK ELEMENTS - Mississippi	ı ———		T	1						Sup Carle	Cup Cade		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination		١.			== 40		=	=====							
	Transport - Zone 4		4	UNCNX	U1L2X	59.18	117.61	79.92	52.82	10.37		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.1813										
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per month			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
	Channelization - Channel System DS1 to DS0 combination -			UNCIA	UTIFT	31.72	09.79	02.20	10.00	14.90		15.75				
	per month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System			ONOTA	IVIQI	102.00	31.37	02.34	10.07	10.10		13.73				
	combination - per month			UNCNX	UC1CA	2.62	6.62	4.74				15.75				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 1		1	UNCNX	U1L2X	21.01	117.61	79.92	52.82	10.37		15.75				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 2		2	UNCNX	U1L2X	27.59	117.61	79.92	52.82	10.37		15.75				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 3		3	UNCNX	U1L2X	37.34	117.61	79.92	52.82	10.37		15.75				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 4		4	UNCNX	U1L2X	59.18	117.61	79.92	52.82	10.37		15.75				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System															
	combintaion- per month		<u> </u>	UNCNX	UC1CA	2.62	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNC1X	LINICCO		F CO	F 00	7.00	7.00		45.75				
4 WID	Is Charge E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROE	L FICE T		UNCCC		5.63	5.63	7.20	7.20		15.75				
4-1111	First DS1 Loop in STS1 Interoffice Transport Combination -	IEROF	FICE I	KANSPORT (EEL)	1											-
	Zone 1		1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				
 	First DS1 Loop in STS1 Interoffice Transport Combination -		<u> </u>	0.1.0.1.1	00200	7 0.00	200.00	100.10	.0.10	.2.01		10.10				
	Zone 2		2	UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75				
	First DS1 Loop in STS1 Interoffice Transport Combination -															
	Zone 3		3	UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75				
	First DS1 Loop in STS1 Interoffice Transport Combination -															
	Zone 4		4	UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile															
	Per Month			UNCSX	1L5XX	4.29										
	Interoffice Transport - Dedicated - STS1 combination - Facility				==0											
	Termination		<u> </u>	UNCSX	U1TFS	644.21	280.37	163.70	62.08	60.29		15.75				
	STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month		1	UNCSX UNC1X	MQ3 UC1D1	107.63 12.96	179.17 6.62	94.52 4.74	34.30	32.82		15.75 15.75				
	Additional DS1Loop in STS1 Interoffice Transport Combination -			UNCIA	OCIDI	12.90	0.02	4.74				15.75				
	Zone 1	l	1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				1
	Additional DS1Loop in STS1 Interoffice Transport Combination -		- '-	0.101/	30200	13.00	200.00	130.43	40.10	12.07		10.73			<u> </u>	
	Zone 2		2	UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75				
	Additional DS1Loop in STS1 Interoffice Transport Combination -		T -					.55.10	.5.10							
	Zone 3		3	UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75				
	Additional DS1Loop in STS1 Interoffice Transport Combination -															
	Zone 4		4	UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	12.96	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-	l	1	l	L											1
4 10=	Is Charge	<u> </u>	D 4 N 2	UNCSX	UNCCC		5.63	5.63	7.20	7.20		15.75			ļ	
4-WIR	E 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROI	FFICE 1	KANS	PUKI (EEL)	1									1	ļ.	!
1	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport	l	1	UNCDX	UDL56	27.44	126.53	88.85	60.68	14.04		15.75				1
-+	Combination - Zone 1 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport	-	1	UNCDX	UDLOO	21.44	120.53	88.85	80.08	14.64		15./5				+
1	Combination - Zone 2	l	2	UNCDX	UDL56	34.55	126.53	88.85	60.68	14.64		15.75				I
- 	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport			O140DA	CDLOO	54.55	120.33	00.00	00.00	17.04		10.70			+	1
	Combination - Zone 3	l	3	UNCDX	UDL56	40.76	126.53	88.85	60.68	14.64		15.75				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport						00	22.00	22.00							1
1	Combination - Zone 4	l	4	UNCDX	UDL56	32.25	126.53	88.85	60.68	14.64		15.75				I
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -															
	Per Mile	I	1	UNCDX	1L5XX	0.00088					1	1		1	1	1

UNBUNDLE	ED NETWORK ELEMENTS - Mississippi													ment: 2	1	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		Nonrec	RATES(\$)	Nonrecurring	Diagona		Svc Order Submitted Manually per LSR	Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	First	arring Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -						Filat	Auu i	FIISL	Auu i	SOWIEC	JOWAN	JOWAN	SOWAN	JOWAN	SOWAN
	Facility Termination			UNCDX	U1TD5	14.14	40.78	27.57	17.26	7.11		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-							_								
	Is Charge			UNCDX	UNCCC		5.63	5.63	7.20	7.20		15.75				
4-WIR	E 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERO	FFICE 1	TRANS	PORT (EEL)												
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	27.44	126 52	00.05	60.69	14.64		15 75				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport			UNCDX	UDL64	21.44	126.53	88.85	60.68	14.64		15.75				
	Combination - Zone 2		2	UNCDX	UDL64	34.55	126.53	88.85	60.68	14.64		15.75				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		_	0110271	05201	000	120.00	00.00	00.00			10.70				
	Combination - Zone 3		3	UNCDX	UDL64	40.76	126.53	88.85	60.68	14.64		15.75				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport															
	Combination - Zone 4		4	UNCDX	UDL64	32.25	126.53	88.85	60.68	14.64		15.75				
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	Per Mile			UNCDX	1L5XX	0.00088										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination			UNCDX	U1TD6	14.14	40.78	27.57	17.26	7.11		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	UTID6	14.14	40.78	21.51	17.26	7.11		15.75				
	Is Charge			UNCDX	UNCCC		5.63	5.63	7.20	7.20		15.75				
ADDITIONAL	NETWORK ELEMENTS			0110271	0.1000		0.00	0.00	7.20	7.20		10.70				
	used as a part of a currently combined facility, the non-recurr	ng cha	rges de	o not apply, but a	Switch As Is c	harge does app	oly.									
	used as ordinarily combined network elements in All States, the															
Nonre	curring Currently Combined Network Elements "Switch As Is"	Charge	(One a	applies to each con	nbination)											
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		5.63	5.63	7.20	7.20		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-			LINCDY	LINICCO		5.00	F 00	7.00	7.20		45.75				
	Is Charge - 56/64 kbps Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	UNCCC		5.63	5.63	7.20	7.20		15.75				
	Is Charge - DS1			UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-			0110111	0.1000		0.00	0.00	7.20	7.20		10.70				
	Is Charge - DS3			UNC3X	UNCCC		5.63	5.63	7.20	7.20		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-															
	ls Charge - STS1			UNCSX	UNCCC		5.63	5.63	7.20	7.20		15.75				
NOTE	: Local Channel - Dedicated Transport - minimum billing period	d - Belo	w DS3													
	Local Channel - Dedicated - 2-Wire Voice Grade			UNCXV	ULDV2	14.91	194.22	33.36	37.79	3.30		15.75				
	Local Channel - Dedicated - 4-Wire Voice Grade Local Channel - Dedicated - DS1 per month Zone 1		1	UNCXV UNC1X	ULDV4 ULDF1	15.99 36.83	194.66 178.50	33.80 154.61	38.27 22.89	3.78 15.74		15.75 15.75				
	Local Channel - Dedicated - DS1 Per Month Zone 2		2	UNC1X	ULDF1	35.99	178.50	154.61	22.89	15.74		15.75				
	Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	221.63	178.50	154.61	22.89	15.74		15.75				
	Local Channel - Dedicated - DS1- Per Month Zone 4		4	UNC1X	ULDF1	221.63	178.50	154.61	22.89	15.74		15.75				
	Local Channel - Dedicated - DS3 - Per Mile per month			UNC3X	1L5NC	9.66										
	Local Channel - Dedicated - DS3 - Facility Termination			UNC3X	ULDF3	413.87	454.13	265.47	123.23	86.19		15.75				
	Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC	9.66										
	Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	408.02	454.13	265.47	123.23	86.19		15.75				
	nal Features & Functions:															
MULI	PLEXERS Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per		1	OXIDI	IVIQI	102.03	91.37	02.54	10.07	10.10		13.73				
	month (2.4-64kbs)			UDL	1D1DD	1.22	6.62	4.74				15.75				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per			1	1		3.32	4				.00				t e
	month			UDN	UC1CA	2.62	6.62	4.74			<u> </u>	15.75		<u> </u>		<u> </u>
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.5737	6.62	4.74				15.75				
	DS3 to DS1 Channel System per month			UXTD3	MQ3	170.63	179.17	94.52	34.30	32.82		15.75				
	STS1 to DS1 Channel System per month	<u> </u>		UXTS1	MQ3	170.63	179.17	94.52	34.30	32.82		15.75				
	DS3 Interface Unit (DS1 COCI) used with Loop per month		1	USL	UC1D1	12.96	6.62	4.74				15.75		1	1	
i l	DS3 Interface Unit (DS1 COCI) used with Local Channel per month			ULDD1	UC1D1	12.96	6.62	4.74				15.75				
UNBUNDI ED	LOCAL EXCHANGE SWITCHING(PORTS)		1	ו טבטט ו	ומוסט	12.90	0.02	4.74			1	15.75		1	1	
	inge Ports				+									 	1	t
1=20110	· • · · · ·	l				1					L			1	1	

UNBUNDLE	ED NETWORK ELEMENTS - Mississippi			T		1						,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	: Although the Port Rate includes all available features in GA, I	Y, LA 8	& TN, t	he desired features	will need to b	e ordered usir	ng retail USOCs	3								
2-WIR	E VOICE GRADE LINE PORT RATES (RES)			LUEBOD	LIEBBI		2.22									
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.41	2.39	2.29	1.42	1.33		15.75				
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.41	2.39	2.29	1.42	1.33		15.75				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.41	2.39	2.29	1.42	1.33		15.75				
	Exchange Ports - 2-Wire VG unbundled MS extended local dialing parity Port with Caller ID - Res.			UEPSR	UEPAT	1.41	2.39	2.29	1.42	1.33		15.75				
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LUM)			UEPSR	UEPAP	1.41	2.39	2.29	1.42	1.33		15.75				
	Exchange Ports - 2-Wire Voice Mississippi Residence Dialing Plan without Caller ID			UEPSR	UEPWJ	1.41	2.39	2.29	1.42	1.33		15.75				
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPSR	UEPRT	1.41	2.39	2.29	1.42	1.33		15.75				
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00				15.75				1
FEAT	URES			LUEBOD		0.50	2.22									
0.14/15	All Available Vertical Features			UEPSR	UEPVF	2.56	0.00	0.00				15.75				
2-WIR	E VOICE GRADE LINE PORT RATES (BUS) Exchange Ports - 2-Wire Analog Line Port without Caller ID -			LIEDOD	UEPBL	4.44	2.20	2.20	1.42	4.22		45.75				
	Exchange Ports - 2-Wire VG unbundled Line Port with			UEPSB		1.41	2.39	2.29		1.33		15.75				
	unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.41	2.39	2.29	1.42	1.33		15.75				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus. Exchange Ports - 2-Wire VG unbundled MS extended local			UEPSB	UEPBO	1.41	2.39	2.29	1.42	1.33		15.75				
	dialing parity Port with Caller ID - Bus. Exhange Ports - 2-Wire VG unbundled incoming only port with			UEPSB	UEPAY	1.41	2.39	2.29	1.42	1.33		15.75				
	Caller ID - Bus Exchange Ports - 2-Wire Voice Mississippi Business Dialing Plan			UEPSB	UEPB1	1.41	2.39	2.29	1.42	1.33		15.75				
	without Caller ID 2-Wire voice unbundled Incoming Only Port without Caller ID			UEPSB	UEPWK	1.41	2.39	2.29	1.42	1.33		15.75				
	Capability			UEPSB	UEPBE	1.41	2.39	2.29	1.42	1.33		15.75				
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00				15.75				
FEAT	URES All Available Vertical Features			UEPSB	UEPVF	2.56	0.00	0.00				15.75				
EXCH	ANGE PORT RATES (DID & PBX)			UEFSB	UEFVF	2.56	0.00	0.00				15.75				
LAGII	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPSP	UEPXE	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPSP	UEPXL	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPSP	UEPXM	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPSP	UEPXO	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Economy Calling Port			UEPSP	UEPXQ	1.41	31.45	14.93	14.38	0.92		15.75				
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Optional Calling Port			UEPSP	UEPXR	1.41	31.45	14.93	14.38	0.92		15.75				

ONBONDL	ED NETWORK ELEMENTS - Mississippi												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment: Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		т
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled PBX Port, Mississippi only			UEPSP	UEPA5	1.41	31.45	14.93	14.38	0.92		15.75				ļ
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.41	31.45	14.93	14.38	0.92		15.75				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00				15.75				
FEAT	TURES															ļ
	All Available Vertical Features			UEPSP UEPSE	UEPVF	2.56	0.00	0.00				15.75				ļ
EXCH	HANGE PORT RATES (COIN)															ļ
	Exchange Ports - Coin Port	L				1.41	2.39	2.29	1.42	1.33		15.75				
	E: Transmission/usage charges associated with POTS circuit s															ļ
	E: Access to B Channel or D Channel Packet capabilities will be	availal	ole onl	y through BFR/New	Business Re	quest Process.	Rates for the	packet capabi	lities will be de	termined via t	he Bona Fic	le Request/l	New Business	s Request Pro	ocess.	ļ
	LOCAL EXCHANGE SWITCHING(PORTS)	ļ			1									-	ļ	<u> </u>
EXCH	HANGE PORT RATES	<u> </u>		UEPEX	LIEDDO	0.05	400.00	10.05	04 ==	0.00		45		-	1	
	Exchange Ports - 2-Wire DID Port	ļ		UEPEX	UEPP2	8.25	120.00	18.85	61.77	3.88		15.75		-	1.97	<u> </u>
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID	1		LIEDDD	LIEDDD	50.44	202.42	00.05	74.00	0.54		45.75		I	4.07	
——	capability	1		UEPDD	UEPDD	58.41	203.19	96.25	74.86	2.54 10.76		15.75		1	1.97 1.97	├
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	13.69	73.19	53.30	47.90	10.76		15.75				
NOTE	All Features Offered			UEPTX UEPSX	UEPVF	2.56	0.00	0.00				15.75			1.97	
	E: Transmission/usage charges associated with POTS circuit s														<u> </u>	
NOTE	E: Access to B Channel or D Channel Packet capabilities will be	availal	ole onl						lities will be de	termined via t	he Bona Fic	le Request/l	New Busines	s Request Pro	ocess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0.00	0.00	0.00	04.05							
LINIBI	Exchange Ports - 4-Wire ISDN DS1 Port	ļ		UEPEX	UEPEX	84.63	205.00	102.14	81.65	20.69		15.75			1.97	
	UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY															ļ
UNBU	UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE			LIEDVD	LIEDAO	4.44	0.00	0.00	4.40	4.00		45.75				4
	Unbundled Remote Call Forwarding Service, Area Calling, Res			UEPVR	UERAC	1.41	2.39	2.29	1.42	1.33		15.75				
	Halandia I Barrett Oall France I'm Oar i'm Land Oalling Bar			LIEDVO	LIEBLO		0.00	0.00	4.40	4.00		45.75				
	Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.41	2.39	2.29	1.42	1.33		15.75				4
	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.41	2.39	2.29	1.42	1.33		15.75				
Na.	Unbundled Remote Call Forwarding Service, IntraLATA - Res			UEPVR	UERTR	1.41	2.39	2.29	1.42	1.33		15.75				
Non-I	Recurring															
	Unbundled Remote Call Forwarding Service - Conversion -			LIEDVO			0.0000	0.0000				45.75				
	Switch-as-is	-		UEPVR	USAC2		0.0988	0.0988				15.75				
	Unbundled Remote Call Forwarding Service - Conversion with			LIEDVD	110400		0.0000	0.0000								
LINDI	allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus	-		UEPVR	USACC		0.0988	0.0988								
UNBU	UNDLED REMOTE CALL FORWARDING - BUS	-														ļ
	Haland Hala Daniel Call France Para Control Association Branch			LIEDVD	LIEDAG		0.00	0.00	4.40	4.00		45.75				
	Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVB	UERAC	1.41	2.39	2.29	1.42	1.33		15.75				
	Habita diad Danata Call Familian Canida Lacal Calling Dia			LIEDVD	LIEDLO	4 44	2.20	0.00	4.40	4.00		45.75				
	Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus	-		UEPVB UEPVB	UERLC UERTE	1.41 1.41	2.39 2.39	2.29 2.29	1.42 1.42	1.33 1.33		15.75 15.75				
	Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, IntraLATA - Bus			UEPVB	UERTR	1.41	2.39	2.29	1.42	1.33		15.75				
			-	UEFVB	UEKIK	1.41	2.39	2.29	1.42	1.33		15.75				
	Unbundled Remote Call Forwarding Service Expanded and Exception Local Calling			UEPVB	UERVJ	1.41	2.39	2.29	1.42	1.33		15.75				
Non I				UEPVB	UERVJ	1.41	2.39	2.29	1.42	1.33		15.75				
Non-i	Recurring	-														
	Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is			UEPVB	USAC2		0.0988	0.0988				15.75				
	Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with	 		ULFVD	USAUZ		0.0988	0.0988				15.75		 	 	
	allowed change (PIC and LPIC)			UEPVB	USACC		0.0988	0.0988								
LINDUNDI ED	D LOCAL SWITCHING, PORT USAGE	-		UEPVB	USACC		0.0900	0.0900						-		-
	Office Switching (Port Usage)	 			1									 	 	
Ena	End Office Switching Function, Per MOU	-			+	0.0010269								-	 	
	End Office Switching Function, Per MOU End Office Trunk Port - Shared, Per MOU	-			1	0.0010269			 						 	
Tond	lem Switching (Port Usage) (Local or Access Tandem)	-			1	0.000101			-						 	
rand	Tandem Switching Function Per MOU	1			1	0.0001723					1			1	1	
	Tandem Trunk Port - Shared, Per MOU	-			1	0.0001723			 						 	
Com	mon Transport	-			1	0.0001028			 						 	
Comi	Common Transport - Per Mile, Per MOU	-			1	0.0000026			 						 	
	Common Transport - Per Mille, Per MOU Common Transport - Facilities Termination Per MOU	-			1	0.0000026									 	\vdash
LINDING CO	D PORT/LOOP COMBINATIONS - COST BASED RATES	-			1	0.0004541									 	\vdash
CONFUNDIDED	FOR I/LOUP COMBINATIONS - COST BASED RATES			l												↓
0.120.1222	Based Rates are applied where BellSouth is required by FCC ar	2/0-0														

<u>UNBUN</u> D	LED	NETWORK ELEMENTS - Mississippi												Attach	ment: 2	Exhi	bit: C
ATEGORY		RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -		Incremental Charge -	Increment Charge Manual S Order vs Electronic Disc Add
							Rec	Nonre		Nonrecurring			•		Rates(\$)	•	
								First	Add'l	First	Add'l		SOMAN		SOMAN	SOMAN	SOMAN
		ce and Tandem Switching Usage and Common Transport Us															
		and additional Port nonrecurring charges apply to Not Curr	ently C	ombin	ed Combos. For Cur	rently Combi	ned Combos th	ne nonrecurrin	g charges sha	II be those iden	ntified in the N	onrecurring	j - Currently	Combined s	ections.		
		OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
UNE		t/Loop Combination Rates															
		-Wire VG Loop/Port Combo - Zone 1		1			12.22										
		2-Wire VG Loop/Port Combo - Zone 2		2			17.13										
		2-Wire VG Loop/Port Combo - Zone 3		3			26.26										
		-Wire VG Loop/Port Combo - Zone 4		4			44.91										
UNE		p Rates															
		-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	10.98										
	2-	P-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	15.91										
		-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	25.04										
		-Wire Voice Grade Loop (SL1) - Zone 4		4	UEPRX	UEPLX	43.68										
2-W		oice Grade Line Port Rates (Res)															
	2-	!-Wire voice unbundled port - residence			UEPRX	UEPRL	1.23	40.31	19.84	24.90	6.58		15.75				
		!-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.23	40.31	19.84	24.90	6.58		15.75				
	2-	-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.23	40.31	19.84	24.90	6.58		15.75				
		-Wire voice Grade unbundled Mississippi extended local															
	di	lialing parity port with Caller ID - res			UEPRX	UEPAT	1.23	40.31	19.84	24.90	6.58		15.75				
		-Wire voice unbundles res, low usage line port with Caller ID LUM)			UEPRX	UEPAP	1.23	40.31	19.84	24.90	6.58		15.75				
	2-	t-Wire Voice Unbundled Mississippi Residence Dialing Plan without Caller ID			UEPRX	UEPWJ	1.23	40.31	19.84	24.90	6.58		15.75				
		-Wire voice unbundled Low Usage Line Port without Caller ID		†													
		Capability			UEPRX	UEPRT	1.23	40.31	19.84	24.90	6.58		15.75				
FF4	ATUR				OLITOR	OLITA	1.20	40.01	10.04	24.00	0.00		10.70		-		
		NI Features Offered		†	UEPRX	UEPVF	2.56	0.00	0.00				15.75				1
1.00	CALN	NUMBER PORTABILITY			02.101	02	2.00	0.00	0.00				10.70		-		
		ocal Number Portability (1 per port)			UEPRX	LNPCX	0.35								-		
NO		CURRING CHARGES (NRCs) - CURRENTLY COMBINED		†													1
		-Wire Voice Grade Loop / Line Port Combination - Conversion -		†													1
	S	Switch-as-is -Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPRX	USAC2		0.0988	0.0988				15.75				
	S	Switch with change -Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPRX	USACC		0.0988	0.0988				15.75				<u> </u>
ADI	S	NAL NRCs						0.00	0.00				15.75				
ADI		NAL NRCS -Wire Voice Grade Loop/Line Port Combination - Subsequent		-	-	+								-	-	-	
		Activity			UEPRX	USAS2	0.00	0.00	0.00				15.75		1	1	
2 14		/OICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)		-	OLI NA	UUAUZ	0.00	0.00	0.00				13.75	-			
		t/Loop Combination Rates				-									-	-	
UNI		!-Wire VG Loop/Port Combo - Zone 1		1	 	1	12.22			1		-	1	1	 	 	1
		-Wire VG Loop/Port Combo - Zone 1		2		-	17.13								-	-	
		!-Wire VG Loop/Port Combo - Zone 2		3	+	1	26.26			 				-			1
LINI		p Rates		3		-	20.20								-	-	
UNI		-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	10.98						-	-			1
		!-Wire Voice Grade Loop (SL1) - Zone 1		2	UEPBX	UEPLX	15.91			1		-	1	1	 	 	1
-		!-Wire Voice Grade Loop (SL1) - Zone 2		3	UEPBX	UEPLX	25.04						-	-			1
		!-Wire Voice Grade Loop (SL1) - Zone 3		4	UEPBX	UEPLX	43.68			1		-	1	1	 	 	1
2_14		oice Grade Line Port (Bus)		+	OLI DA	JLI LA	45.00			1		-	1	1	 	 	1
Z-VV		!-Wire voice unbundled port without Caller ID - bus		1	UEPBX	UEPBL	1.23	40.31	19.84	24.90	6.58	-	15.75	1	 	 	1
		!-Wire voice unbundled port with Caller + E484 ID - bus		-	UEPBX	UEPBC	1.23	40.31	19.84	24.90	6.58		15.75	-			1
		-wire voice unbundled port with Caller + E484 ID - bus		1	UEPBX	UEPBO	1.23	40.31	19.84	24.90	6.58		15.75		 	 	-
		-Wire voice unbundled port outgoing only - bus -Wire voice Grade unbundled Mississippi extended local		 	OLFDA	OLFBU	1.23	40.31	19.64	24.90	0.38		15.75	-			1
		i-vvire voice Grade unbundled Mississippi extended local			UEPBX	UEPAY	1.23	40.31	19.84	24.90	6.58		15.75				
		!-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.23	40.31	19.84	24.90	6.58		15.75		†	†	1
		-Wire Voice Unbundled Mississippi Business Dialing Plan				1 -2 -	20	.0.01		250	0.00				1	t	†
		vithout Caller ID			UEPBX	UEPWK	1.23	40.31	19.84	24.90	6.58	l	15.75				1

UNBUNDLE	ED NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring		SOMEC	COMAN		Rates(\$)	COMAN	COMAN
	2-Wire voice unbundled Incoming Only Port without Caller ID						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Capability			UEPBX	UEPBE	1.23	40.31	19.84	24.90	6.58		15.75				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEAT				LIEDDY	LIEDVE	0.50	2.22	0.00				45.75				.
NOND	All Features Offered ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPBX	UEPVF	2.56	0.00	0.00				15.75				
NONK	2-Wire Voice Grade Loop / Line Port Combination - Conversion -		1													+
	Switch-as-is			UEPBX	USAC2		0.0988	0.0988				15.75				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -			-												
	Switch with change			UEPBX	USACC		0.0988	0.0988				15.75				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Subsequent Database Update						0.00	0.00				15.75				
ADDIT	TONAL NRCs	-	1		-									-	1	
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2		0.00	0.00	[15.75				
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			OLI DX	OOAOZ		0.00	0.00				13.73				
	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			12.22										
	2-Wire VG Loop/Port Combo - Zone 2		2			17.13										
	2-Wire VG Loop/Port Combo - Zone 3		3			26.26										
	2-Wire VG Loop/Port Combo - Zone 4		4			44.91										<u> </u>
UNE L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	10.98										
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	15.91										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	25.04										+
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEPRG	UEPLX	43.68										
2-Wire	Voice Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.23	69.37	32.48	37.86	6.17		15.75				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				15.75				<u> </u>
FEAT	All Features Offered		1	UEPRG	UEPVF	2.56	0.00	0.00				15.75				
NONE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED		1	UEFRG	UEPVF	2.50	0.00	0.00				15.75				1
NONK	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															+
	Conversion - Switch-As-Is			UEPRG	USAC2		7.96	1.91				15.75				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch with Change			UEPRG	USACC		7.96	1.91	ļ			15.75		ļ	1	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -						0.00	0.00				45.75			1	
Ι	Subsequent Database Update TONAL NRCs		1		+		0.00	0.00			1	15.75				
ADDII	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -		1		1										 	†
	Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00]			15.75				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt															1
	Group						7.36	7.36				15.75				
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
UNE F	Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		1		1	12.22			 		1				-	
	2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		2		1	17.13			 		1			1	 	+
 	2-Wire VG Loop/Port Combo - Zone 2		3		1	26.26									—	†
	2-Wire VG Loop/Port Combo - Zone 4		4		1	44.91			1							1
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	10.98		•								
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	15.91					1					1
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	25.04			<u> </u>		1			1	1	
2 14/:	2-Wire Voice Grade Loop (SL 1) - Zone 4 2 Voice Grade Line Port Rates (BUS - PBX)	l	4	UEPPX	UEPLX	43.68			 					 	 	
Z-WIFE	VOICE Graue Litte FUIT Rates (DUS - FDA)	l	1		1				1		<u> </u>	l		l	1	ь

ONROND	LED NETWORK ELEMENTS - Mississippi			1										nent: 2		bit: C
CATEGOR	r RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
					1	_	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.23	69.37	32.48	37.86	6.17		15.75				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.23	69.37	32.48	37.86	6.17		15.75				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
	Capable Port			UEPPX	UEPXE	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Administrative Calling Port			UEPPX	UEPXL	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy									• • • • • • • • • • • • • • • • • • • •						
	Room Calling Port			UEPPX	UEPXM	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			CLITA	OLI AWI	1.20	00.07	02.40	07.00	0.17	1	10.70				
	Discount Room Calling Port			UEPPX	UEPXO	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Economy			OLITA	OLI AO	1.25	03.37	32.40	37.00	0.17		13.73				
	Calling Port			UEPPX	UEPXQ	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Optional			OLITA	OLI AQ	1.25	03.37	32.40	37.00	0.17		13.73				
	Calling Port			UEPPX	UEPXR	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.23	69.37	32.48	37.86	6.17	-	15.75			-	
				UEPPX	UEPA5	1.23	69.37	32.48		6.17						
1.00	Mississippi PBX 2-Way Combo Local Opt 2 Calling Port CAL NUMBER PORTABILITY			UEFFA	UEPAS	1.23	09.37	32.40	37.86	0.17		15.75				
LO				UEPPX	LNPCP	3.15	0.00	0.00				15.75				
EE.	Local Number Portability (1 per port) ATURES			UEPPA	LINECE	3.13	0.00	0.00	-		-	15.75			-	
FE/	All Features Offered			UEPPX	UEPVF	2.56	0.00	0.00	-		-	15.75			-	
NO	NRECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEFFA	UEFVF	2.30	0.00	0.00				15.75				
NO	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -				+				-		-				-	
	Conversion - Switch-As-Is			UEPPX	USAC2		7.96	1.91				15.75				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			UEPPA	USACZ		7.90	1.91	-		-	15.75			-	
				UEPPX	USACC		7.96	1.91				15.75				
	Conversion - Switch with Change			UEFFA	USACC		7.90	1.91				15.75				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update						0.00	0.00				15.75				
ADI							0.00	0.00				15.75				
ADI	DITIONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			LIEDDY	110 4 00	0.00	0.00	0.00				45.75				
	Subsequent Activity		-	UEPPX	USAS2	0.00	0.00	0.00				15.75				-
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.36	7.26				15.75				
0.14							7.30	7.36				15.75				
	IRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PO	K I														
UNI	E Port/Loop Combination Rates		1			40.00										
	2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2		2		+	12.22 17.13										-
			3													
	2-Wire VG Coin Port/Loop Combo – Zone 3				+	26.26										-
	2-Wire VG Coin Port/Loop Combo – Zone 4		4		+	44.91										-
UNI	E Loop Rates			LIEBOO	LIEDLY	10.00										
	2-Wire Voice Grade Loop (SL1) - Zone 1	!	1	UEPCO	UEPLX	10.98									-	
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	15.91										
	2-Wire Voice Grade Loop (SL1) - Zone 3	1	3	UEPCO	UEPLX	25.04					-				 	
0.14	2-Wire Voice Grade Loop (SL1) - Zone 4	l	4	UEPCO	UEPLX	43.68			 		1				 	
2-W	/ire Voice Grade Line Ports (COIN)	1	<u> </u>	 							-				 	
	2-Wire Coin 2-Way without Operator Screening and without	1	1	LIEDOO	HEDDE	4.00	40.01	10.01	04.00	0.50		45.75			I	
	Blocking (AL, KY, LA, MS)	!	<u> </u>	UEPCO	UEPRF	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-Way without Operator Screening and without	1	1	LIEBOO	LIED: 10							,			I	
	Blocking; with Dialing Parity (Note 3) (MS)	<u> </u>		UEPCO	UEPMC	1.23	40.31	19.84	24.90	6.58	ļ	15.75				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-W with Operator Screening and Blocking: 011, 900/976, 1+DDD; with Dialing Parity (MS)			UEPCO	UEPMA	1.23	40.31	19.84	24.90	6.58		15.75				

UNDUNDL	ED NETWORK ELEMENTS - Mississippi			1										ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking															
	(AL, LA, MS)			UEPCO	UEPRB	1.23	40.31	19.84	24.90	6.58		15.75				<u> </u>
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking;			LIEBOO	LIEDMD	4.00	40.04	40.04	04.00	0.50		45.75				
	with Dialing Parity (MS) 2-Wire Coin 2-Way with Operator Screening & Blocking:			UEPCO	UEPMB	1.23	40.31	19.84	24.90	6.58		15.75			-	+
	900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-W Operator Screening: 900 Block: 900/976,			OLI OO	OLI OD	1.20	40.01	10.04	24.00	0.00		10.70				†
	1+DDD, 011+, Local; with Dialing Parity (MS)			UEPCO	UEPCJ	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Outward without Blocking and without Operator															
	Screening (KY, LA, MS)			UEPCO	UEPRN	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Outward without Blocking and without Operator															
	Screening; With Dailing Parity (MS)			UEPCO	UEPME	1.23	40.31	19.84	24.90	6.58		15.75				4
	2-Wire Coin Outward with Operator Screening and 011 Blocking			UEPCO	UEPRJ	1.23	40.04	40.04	24.00	6.58		45.75				
	(GA, KY, MS) 2-Wire Coin Outward with Operator Screening and 011			UEPCO	UEPRJ	1.23	40.31	19.84	24.90	6.58		15.75				+
	Blocking; with Dialing Parity (MS)			UEPCO	UEPMD	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Outward with Operator Screening and Blocking:			OLI OO	OLI WID	1.25	40.51	13.04	24.30	0.50		10.70				+
	011, 900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRH	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Outward Operator Screening & Blocking: 900/976,															
	1+DDD, 011+, and Local (AL, KY, LA, MS)			UEPCO	UEPCN	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD,															
	011+, and Local; with Dialing Parity (MS)			UEPCO	UEPCS	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.23	40.31	19.84	24.90	6.58		15.75				<u> </u>
	2-Wire Coin Outward Smartline with 900/976 (all states except I.A)			LIEBOO	LIEDOD	4.00	40.04	40.04	04.00	0.50		45.75				
ADDI	TIONAL UNE COIN PORT/LOOP (RC)			UEPCO	UEPCR	1.23	40.31	19.84	24.90	6.58		15.75			-	+
ADDI	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	4.62	0.00	0.00								+
LOCA	AL NUMBER PORTABILITY			OLI CO	OILLOO	4.02	0.00	0.00								+
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35									1	†
NON	RECURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPCO	USAC2		0.0988	0.0988				15.75				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch with change			UEPCO	USACC		0.0988	0.0988				15.75				
ADDI	TIONAL NRCs														-	-
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0.00	0.00				15.75				
2-WIF	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	PORT (U3A32		0.00	0.00				13.73				+
	Port/Loop Combination Rates	<u> </u>	1	1												1
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		1	15.16									İ	†
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			20.02										1
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			28.82										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 4		4			46.99										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	13.89										
	2-Wire Voice Grade Loop (SL2) - Zone 2		3	UEPFR UEPFR	UECF2 UECF2	18.75									 	
	2-Wire Voice Grade Loop (SL2) - Zone 3 2-Wire Voice Grade Loop (SL2) - Zone 4		4	UEPFR	UECF2	27.55 45.72			 		-			-		+
2-Wir	re Voice Grade Line Port Rates (Res)	1	+	OLI I IX	OLOI Z	40.12								1	t	+
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.27	108.35	70.57	54.24	11.70		15.75				†
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.27	108.35	70.57	54.24	11.70		15.75				1
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.27	108.35	70.57	54.24	11.70		15.75				
	2-Wire voice Grade unbundled Mississippi extended local															
	dialing parity port with Caller ID - res		<u> </u>	UEPFR	UEPAT	1.27	108.35	70.57	54.24	11.70		15.75			1	
	2-Wire voice unbundles res, low usage line port with Caller ID		1								1					
	(LUM)		<u> </u>	UEPFR	UEPAP	1.27	108.35	70.57	54.24	11.70		15.75			1	
	2-Wire Voice Unbundled Mississippi Residence Dialing Plan without Caller ID		1	UEPFR	UEPWJ	1.27	108.35	70.57	54.24	11.70	1	15.75				
	ROFFICE TRANSPORT		1	ULFFR	UEFVVJ	1.27	108.35	/0.5/	54.24	11.70		15.75		 	 	+

UNBUNDL	ED NETWORK ELEMENTS - Mississippi										,	,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFR	U1TV2	20.32	40.77	27.57	17.26	7.11						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFR	1L5XX	0.0088										
FEAT	All Features Offered			UEPFR	UEPVF	2.56	0.00	0.00			1	15.75			-	
1.00	AL NUMBER PORTABILITY			UEPFR	UEPVF	2.30	0.00	0.00				15.75			-	
100	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35									-	
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLFIK	LINFOX	0.33										
I I I I I I I I I I I I I I I I I I I	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															+
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		16.94	3.72				15.75				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port		1	1	15552		.0.04	J Z						İ	1	
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		16.94	3.72				15.75		1	I	
2-WII	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	PORT (BUS)			-									
UNE	Port/Loop Combination Rates															1
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			15.16										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			20.02										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			28.82										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 4		4			46.99										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	13.89										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	18.75										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	27.55										
0.187	2-Wire Voice Grade Loop (SL2) - Zone 4		4	UEPFB	UECF2	45.72										
2-Wii	re Voice Grade Line Port (Bus)			LIEDED	LIEDDI	4.07	100.05	70.57	54.04	44.70		45.75				
	2-Wire voice unbundled port without Caller ID - bus 2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB UEPFB	UEPBL UEPBC	1.27 1.27	108.35 108.35	70.57 70.57	54.24 54.24	11.70 11.70		15.75 15.75				
-	2-Wire voice unburidled port with Caller + £484 ID - bus			UEPFB	UEPBO	1.27	108.35	70.57	54.24	11.70		15.75				
+	2-Wire voice Grade unbundled Mississippi extended local			OLFIB	OLFBO	1.21	100.33	70.57	34.24	11.70	1	13.73				
	dialing parity port with Caller ID - bus			UEPFB	UEPAY	1.27	108.35	70.57	54.24	11.70		15.75				
-	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.27	108.35	70.57	54.24	11.70		15.75				1
	2-Wire Voice Unbundled Mississippi Business Dialing Plan			02.1.5	02. 2.		100.00	7 0.01	02.			10.70				
	without Caller ID			UEPFB	UEPWK	1.27	108.35	70.57	54.24	11.70		15.75				
LOCA	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFB	LNPCX	0.35										1
INTE	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															ĺ
	Termination			UEPFB	U1TV2	20.32	40.77	27.57	17.26	7.11						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFB	1L5XX	0.0088										
FEAT	TURES															
	All Features Offered			UEPFB	UEPVF	2.56	0.00	0.00				15.75				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			LIEDED	LICACO		40.04	2.70				45.75				
 	Combination - Conversion - Switch-as-is 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port	-	 	UEPFB	USAC2		16.94	3.72	1		 	15.75		-		
1	Combination - Conversion - Switch with change			UEPFB	USACC		16.94	3.72				15.75		1	I	
2-W/II	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)		!	OLI I D	UUAUU		10.34	3.12	1		1	13.73		1	t	
	Port/Loop Combination Rates														-	†
J.W.	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			15.16									1	
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2	İ		20.02			İ					İ	1	
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			28.82					Ì					
l	2-Wire VG Loop/IO Tranport/Port Combo - Zone 4		4			46.99			1						1	1
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	13.89										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	18.75										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	27.55		-								
	2-Wire Voice Grade Loop (SL2) - Zone 4		4	UEPFP	UECF2	45.72										
2-Wii	e Voice Grade Line Port Rates (BUS - PBX)															

ONRONDL	ED NETWORK ELEMENTS - Mississippi			1							_			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
1					+ +		Nonre	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
					+		11130	Auu	11100	Addi	COME	COMPAR	COMPAN	COMPAR	COMPAR	COMPAN
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	1.27	137.41	80.14	67.20	11.29		15.75				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.27	137.41	80.14	67.20	11.29		15.75				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.27	137.41	80.14	67.20	11.29		15.75			-	-
			-	UEPFP	UEPLD					11.29						
	2-Wire Voice Unbundled PBX LD Terminal Ports		-			1.27	137.41	80.14	67.20			15.75				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port	-		UEPFP	UEPXA	1.27	137.41	80.14	67.20	11.29		15.75				ļ
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.27	137.41	80.14	67.20	11.29		15.75				ļ
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.27	137.41	80.14	67.20	11.29		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.27	137.41	80.14	67.20	11.29		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
	Capable Port			UEPFP	UEPXE	1.27	137.41	80.14	67.20	11.29		15.75				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	l			1											1
	Administrative Calling Port	1	1	UEPFP	UEPXL	1.27	137.41	80.14	67.20	11.29	1	15.75		Ì		
1	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPFP	UEPXM	1.27	137.41	80.14	67.20	11.29		15.75				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPFP	UEPXO	1.27	137.41	80.14	67.20	11.29		15.75				
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Economy		-	OLITT	OLI AO	1.27	107.41	00.14	07.20	11.20		10.70				+
	Calling Port			UEPFP	UEPXQ	1.27	137.41	80.14	67.20	11.29		15.75				
				UEFFF	UEFAQ	1.27	137.41	60.14	07.20	11.29		15.75				
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Optional															
	Calling Port			UEPFP	UEPXR	1.27	137.41	80.14	67.20	11.29		15.75				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	1.27	137.41	80.14	67.20	11.29		15.75				
	Mississippi PBX 2-Way Combo Local Opt 2 Calling Port			UEPFP	UEPA5	1.27	137.41	80.14	67.20	11.29		15.75				
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.75				
INTE	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFP	U1TV2	20.32	40.77	27.57	17.26	7.11						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFP	1L5XX	0.0088										
FFA	TURES															
	All Features Offered			UEPFP	UEPVF	2.56	0.00	0.00				15.75				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			CLITT	OLI VI	2.00	0.00	0.00				10.70				+
NON	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
				UEPFP	USAC2		16.94	3.72				15.75				
	Combination - Conversion - Switch-as-is 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port		-	UEFFF	USACZ		10.94	3.12				15.75				
				UEPFP	USACC		40.04	0.70				45.75				
INDUNE E	Combination - Conversion - Switch with change			UEPFP	USACC		16.94	3.72				15.75				
	O PORT/LOOP COMBINATIONS - COST BASED RATES	L DOST		-	\rightarrow										-	
	RE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PURI			\rightarrow				1							↓
UNE	Port/Loop Combination Rates	 	<u> </u>	L	\longrightarrow							ļ				
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			21.32										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			26.16										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			34.98										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 4		4			53.15										
UNE	Loop Rates															
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	13.89										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	18.75										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	27.55						i		İ	1	İ
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 4	1	4	UEPPX	UECD1	45.72			1			i		1	1	
IINE	Port Rate	1	<u> </u>	 	·	2			t			1		†	t	†
- ONE	Exchange Ports - 2-Wire DID Port	l	t	UEPPX	UEPD1	7.43	225.96	87.13	114.59	14.25		15.75		 	1.97	†
NON	RECURRING CHARGES - CURRENTLY COMBINED		 	OLI I A	JEI DI	7.40	220.90	07.13	117.33	17.23		15.75		-	1.57	
NON	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -	-	-	-	+				 		 			-	-	
		l		UEPPX	LIGAC1		7.05	1.00			l	15 75			1.97	1
	Switch-as-is	!	-	UEPPA	USAC1		7.35	1.88				15.75		 	1.97	
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion	l		LIEBBY							l					1
	with BellSouth Allowable Changes	 		UEPPX	USA1C		7.35	1.88			ļ	15.75			1.97	
ADD	ITIONAL NRCs															1
	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk	L	L	UEPPX	USAS1		26.94	26.94	L		L	15.75		<u> </u>	1.97	<u> </u>
Teler	phone Number/Trunk Group Establisment Charges	l									l					1

UNBUNDL	ED NETWORK ELEMENTS - Mississippi			1			1								ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	Ē	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00				15.75			1.97	
	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		ND4	0.00	0.00	0.00				15.75			1.97	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX		ND5	0.00	0.00	0.00				15.75			1.97	
	Reserve Non-Consecutive DID numbers Reserve DID Numbers	-		UEPPX		ND6 NDV	0.00	0.00	0.00				15.75			1.97 1.97	
1.004	AL NUMBER PORTABILITY	-		UEPPX		NDV	0.00	0.00	0.00				15.75			1.97	
LUCA	Local Number Portability (1 per port)	-		UEPPX		LNPCP	3.15	0.00	0.00								
2-11/15	RE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	BOD.			LINECE	3.15	0.00	0.00								
	Port/Loop Combination Rates	INC SIDE	I	1													1
ONL	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	1	1														1
	UNE Zone 1		1	UEPPB	UEPPR	,	28.59										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		-	OLITE	OLITIN	`	20.53										
	UNE Zone 2		2	UEPPB	UEPPR		35.00										
 	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		 -	02	02		00.00										1
	UNE Zone 3		3	UEPPB	UEPPR		45.18										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		Ť	02.72	02		10.10										
	UNE Zone 4		4				67.61										
UNE	Loop Rates																
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	18.26						15.75			1.97	
	·																
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	24.67						15.75			1.97	
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	34.85						15.75			1.97	
	2-Wire ISDN Digital Grade Loop - UNE Zone 4		4	UEPPB	UEPPR	USL2X	57.28						15.75			1.97	
UNE	Port Rate																
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	10.33	190.80	133.22	100.72	21.13		15.75			1.97	
NONE	RECURRING CHARGES - CURRENTLY COMBINED																
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																
	Combination - Conversion			UEPPB	UEPPR	USACB	0.00	38.73	27.17				15.75			1.97	
	TIONAL NRCs																
LOCA	AL NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-CH	ANNEL USER PROFILE ACCESS:																
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
B-CH	ANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C,MS, 8	(TN)														
	CVS/CSD (DMS/5ESS)	ļ		UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								
USER	R TERMINAL PROFILE			LIEDDD	UEPPR	11411840	0.00	0.00	0.00								
VEDT	User Terminal Profile (EWSD only) TICAL FEATURES			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VERI	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	2.56	0.00	0.00				15.75			1.97	
INITE	ROFFICE CHANNEL MILEAGE	-		UEPPB	UEPPK	UEFVF	2.56	0.00	0.00				15.75			1.97	
INTE	Interoffice Channel mileage each, including first mile and	-															
	facilities termination			LIEDDD	UEPPR	M1GNC	22.5298	40.77	27.57	17.26	7.11		15.75		1	1.97	
 	Interoffice Channel mileage each, additional mile	1	<u> </u>	UEPPB	UEPPR	M1GNM	0.0098	0.00	0.00	11.20	7.11	1	13.73		 	1.97	
4-WIF	RE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNI	K PORT	<u> </u>	JEIID	JLIIN	IVITOINIVI	0.0090	0.00	0.00			1			 	 	
	Port/Loop Combination Rates	1	1	<u> </u>		+									-	-	
0.12	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																
	Zone 1		1	UEPPP			155.43						1		1	I	
1	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		† –	1			.550								1	1	
	Zone 2		2	UEPPP			205.74						1		I	I	
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1		1		1				İ				İ	İ	İ	
	Zone 3		3	UEPPP			283.10						1		1	I	
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																
	Zone 4		4	UEPPP			534.81						1		1	I	
UNE	Loop Rates	1		1		1				İ				İ	İ	İ	
	4-Wire DS1 Digital Loop - UNE Zone 1	1	1	UEPPP		USL4P	79.08					İ	15.75			1.97	

ONROND	LΕD	NETWORK ELEMENTS - Mississippi			1										ment: 2		bit: C
CATEGORY	r	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs Electronic Disc Add
							Rec	Nonrec	urring	Nonrecurring	Disconnect		•	oss	Rates(\$)	•	•
								First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
		4-Wire DS1 Digital Loop - UNE Zone 2			UEPPP	USL4P	129.38						15.75			1.97	
		1-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	206.74						15.75			1.97	
		4-Wire DS1 Digital Loop - UNE Zone 4		4	UEPPP	USL4P	458.46						15.75			1.97	
UNE		rt Rate			LIEDDD	LIEDDD	70.05	450.00	200 50	407.75	00.70		45.75			4.07	
NO		Exchange Ports - 4-Wire ISDN DS1 Port CURRING CHARGES - CURRENTLY COMBINED			UEPPP	UEPPP	76.35	458.93	260.59	127.75	32.76		15.75			1.97	-
NOI		4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port				+											
		Combination - Conversion -Switch-as-is			UEPPP	USACP	0.00	119.76	79.01				15.75			1.97	
ADD		NAL NRCs															
	4	1-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-															
		nward/two way Tel Nos. (except NC)			UEPPP	PR7TF		0.49					15.75			1.97	
	4	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -							· · · · · · · · · · · · · · · · · · ·						1		
		Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		11.58	11.58				15.75			1.97	
		4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -															
		Subsequent Inward Tel Numbers			UEPPP	PR7ZT		23.15	23.15				15.75			1.97	
LOC		NUMBER PORTABILITY			UEPPP	LNPCN	1.75										
INIT		Local Number Portability (1 per port) ACE (Provsioning Only)			UEPPP	LINPCIN	1.75										
		/oice/Data			UEPPP	PR71V	0.00	0.00	0.00								
		Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
		nward Data			UEPPP	PR71E	0.00	0.00	0.00								
New		Additional "B" Channel				1	2.77										
		New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.61					15.75			1.97	
		New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.61					15.75			1.97	
		New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.61					15.75			1.97	
CAL		/PES															
		nward			UEPPP	PR7C1	0.00	0.00	0.00								
		Outward			UEPPP	PR7C0	0.00	0.00	0.00								
Into		Гwo-way ce Channel Mileage			UEPPP	PR7CC	0.00	0.00	0.00								
inte		Fixed Each Including First Mile			UEPPP	1LN1A	57.53	89.79	82.28	16.66	14.90		15.75			1.97	
-		Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.20	05.75	02.20	10.00	14.50		13.73			1.57	
4-W		DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT			CLITT	ILIVID	0.20										
		rt/Loop Combination Rates				i i											
	4	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		131.78						15.75			1.97	
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		182.07						15.75			1.97	
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		259.44						15.75			1.97	
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4		4	UEPDC		511.15						15.75			1.97	
UNE		op Rates	 		LIEDDO	1101.00	70.00						45.75			1.0-	
		4-Wire DS1 Digital Loop - UNE Zone 1	<u> </u>	1 2	UEPDC UEPDC	USLDC USLDC	79.08 129.38			 			15.75 15.75			1.97 1.97	-
		4-Wire DS1 Digital Loop - UNE Zone 2 4-Wire DS1 Digital Loop - UNE Zone 3	1	3	UEPDC	USLDC	129.38 206.74			1		1	15.75			1.97	
		4-Wire DS1 Digital Loop - UNE Zone 4		4	UEPDC	USLDC	458.46						15.75			1.97	
UNE		rt Rate		_	02. 00	30250	-100.40			†			10.70			1.57	
		4-Wire DDITS Digital Trunk Port	1		UEPDC	UDD1T	52.70	457.12	254.70	120.96	14.61		15.75		Ì	1.97	
NON		CURRING CHARGES - CURRENTLY COMBINED															
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination							· · · · · · · · · · · · · · · · · · ·						1		
		Switch-as-is	<u> </u>		UEPDC	USAC4		130.24	67.41				15.75			1.97	
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination				l		400 - 1									
		Conversion with DS1 Changes	 	-	UEPDC	USAWA		130.24	67.41				15.75		1	1.97	
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination			UEPDC	USAWB		130.24	67.41				15.75		1	1.97	
ADE		Conversion with Change - Trunk	 	-	UEPDC	USAWB		130.24	07.41	 			15.75			1.97	-
ADL		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -	1		 	+				1		1	1		1	 	1
		Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		14.56	14.56				15.75			1.97	
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent			02. 00	35117		14.50	14.50				10.70			1.57	
		Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		14.56	14.56				15.75			1.97	
	4	1-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel				1											
1 1		Activation/Chan Inward Trunk w/out DID	1		UEPDC	UDTTC		14.56	14.56			l	15.75		Ì	1.97	

INDUNUL	ED NETWORK ELEMENTS - Mississippi			T		ı								nent: 2		oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual S Order vs Electroni Disc Add
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		14.56	14.56				15.75			1.97	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		14.56	14.56				15.75			1.97	
BIPOL	AR 8 ZERO SUBSTITUTION															
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	600.00				15.75			1.97	
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	600.00				15.75			1.97	
Altern	ate Mark Inversion															
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Telep	hone Number/Trunk Group Establisment Charges															
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00						15.75			1.97	
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00						15.75			1.97	
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00			1			15.75			1.97	
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00						15.75			1.97	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00			i i			15.75			1.97	
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				15.75			1.97	
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00				15.75			1.97	
Dedic	ated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	Digital	Loop													
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
_	Termination)			UEPDC	1LNO1	57.33	89.79	82.28	16.86	14.90		15.75			1.97	
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities			UEPDC	1LNOA	0.20	0.00	0.00								
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.20	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.20	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00							
	Central Office Termininating Point			UEPDC	CTG	0.00	0.00	0.00	0.00							
4-WIR	E DS1 LOOP WITH CHANNELIZATION WITH PORT					0.00										
	m is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	vations														
	System can have up to 24 combinations of rates depending on			ber of ports used												
	OS1 Loop	type a.	1	loci oi porto doca												
0.12	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	79.08	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	129.38	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	206.74	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 4		4	UEPMG	USLDC	458.46	0.00	0.00				15.75			1.97	
UNF I	OSO Channelization Capacities (D4 Channel Bank Configuration	15)	t i	02.1.10	00220	100.10	0.00	0.00				10.70			1.07	
0.12	24 DSO Channel Capacity - 1 per DS1	,		UEPMG	VUM24	95.06	0.00	0.00				15.75			1.97	
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	190.12	0.00	0.00			1	15.75			1.97	
1	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	380.24	0.00	0.00				15.75			1.97	
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	570.36	0.00	0.00				15.75			1.97	
_	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	760.48	0.00	0.00	1			15.75			1.97	
_	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	950.60	0.00	0.00	 		†	15.75			1.97	
1	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1.140.72	0.00	0.00				15.75			1.97	
1	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,520.96	0.00	0.00				15.75			1.97	
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	1,901.20	0.00	0.00			1	15.75			1.97	
_	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,281.44	0.00	0.00	1			15.75			1.97	
_	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,661.68	0.00	0.00	 		†	15.75			1.97	l
Non-F	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with	Chanr	neliztio					3.00				.5.76				l
	imum System configuration is One (1) DS1, One (1) D4 Channe														1	
	oles of this configuration functioning as one are considered Ac														1	
	NRC - Conversion (Currently Combined) with or without				34.4										1	
	BellSouth Allowed Changes	l		UEPMG	USAC4	0.00	151.35	8.41				15.75			1.97	
	m Additions at End User Locations Where 4-Wire DS1 Loop with							0.41	—		 	10.70			1.01	├ ──

JINDUNULI	ED NETWORK ELEMENTS - Mississippi	1		I	1	l					Cup Cade	Cup Carle		ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				SVC Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
N /	No. O control O colling IV in all states are still Bookin 7 and	<u> </u>					First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
New (Not Currently Combined) in all states, except in Density Zone 1 1 DS1/D4 Channel Bank - Additionally Add NRC for each Port	от гор	8 W S	l's												
	and Assoc Fea Activation			UEPMG	VUMD4	0.00	715.15	327.39	148.05	17.56		15.75			1.97	
Bipola	ar 8 Zero Substitution			-	_		-									
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	600.00				15.75			1.97	
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	600.00				15.75			1.97	
Altern	nate Mark Inversion (AMI)			OLFING	CCOLI	0.00	0.00	000.00				13.73			1.97	
7.0.0011	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	ange Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
Excha	ange Ports															
	Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business			UEPPX UEPPX	UEPCX	1.23 1.23	0.00	0.00	0.00	0.00		15.75 15.75			1.97 1.97	-
	Line Side Oddward Chairneilzed PBX Trunk Port - Business			UEPPX	UEPUX	1.23	0.00	0.00	0.00	0.00		15.75			1.97	
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.23	0.00	0.00	0.00	0.00		15.75			1.97	
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	7.40	0.00	0.00	0.00	0.00		15.75			1.97	
Featu	re Activations - Unbundled Loop Concentration															
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank			UEPPX	1PQWM	0.61	25.36	13.39	4.29	4.26		15.75			1.97	
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			UEPPX	1PQWU	0.61	78.03	18.39	60.66	11.85		15.75			1.97	
Telep	hone Number/ Group Establishment Charges for DID Service			LIEBBY .		2.22										
	DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States			UEPPX UEPPX	NDT ND4	0.00	0.00	0.00				15.75 15.75			1.97 1.97	
	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00				15.75			1.97	-
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00				15.75			1.97	
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				15.75			1.97	
Local	Number Portability															
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	URES - Vertical and Optional															
Local	Switching Features Offered with Line Side Ports Only			LIEDDY	LIED) (E	0.50	0.00	0.00				45.75			4.07	
	All Features Available Mississippi PBX 2-Way Combo Local Opt 2 Calling Port			UEPPX UEPPX	UEPVF UEPA5	2.56 14.00	0.00 90.00	90.00				15.75 15.75			1.97	
IBUNDI ED	CENTREX PORT/LOOP COMBINATIONS - COST BASED RATE:	S		UEPPX	UEFAS	14.00	90.00	90.00				15.75				
	st Based Rates are applied where BellSouth is required by FCC		State (Commission rule to	provide Unbu	undled Local S	witching or Sw	itch Ports.								
	atures shall apply to the Unbundled Port/Loop Combination - C								dled Port section	on of this Rate	Exhibit.					
	d Office and Tandem Switching Usage and Common Transport															
	e first and additional Port nonrecurring charges apply to Not C	urrently	Comb	ined Combos. For	Currently Co	mbined Combo	s, the nonrecu	rring charges	shall be those	identified in the	he Nonrecu	rring - Curre	ently Combine	ed sections.	Additional NF	RCs may
	also and are categorized accordingly.								1		1					
	arket Rates for Unbundled Centrex Port/Loop Combination will		tiated	on an Individual Ca	ase Basis, un	til further notice	э.									
	P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo)														
	Port/Loop Combination Rates (Non-Design)															
ONE	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1 -			1											
	Non-Design		1	UEP91	1	12.22										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP91		17.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		l . ¯	l	1										1	
	Non-Design	-	3	UEP91	-	26.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	1	4	UEP91		44.91										
IINE	Non-Design Port/Loop Combination Rates (Design)	1	4	OEFSI	+	44.91								-	1	
UNE	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+									1	1	
	Design		1	UEP91	1	15.12										
		t = 1			1											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -					1										

Version 3Q02: 09/06/02 Page 257 of 362

UNBUNDLE	D NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -														1	
'	Design		3	UEP91		28.78									'	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
. '	Design		4	UEP91		46.95									'	
UNE Lo	pop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	10.98										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	15.91										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1	25.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP91	UECS1	43.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	13.89										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	18.75										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	27.55			† †		İ	1	1	1	\vdash	
	2-Wire Voice Grade Loop (SL 2) - Zone 4	-	4	UEP91	UECS2	45.72			 		 	 	 	 	\vdash	+
UNE Po		-	+-	J 01	32002	70.72			 		 	 	-	-		
	tes (Except North Carolina and Sout Carolina)	-	 		+ +				 		 	 	-	-		
	2-Wire Voice Grade Port (Centrex) Basic Local Area	1	+	UEP91	UEPYA	1.23	40.31	19.84	24.90	6.58	1	15.75	 	 		
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local	 	+	OL1 31	OLI IA	1.23	40.31	15.04	24.50	0.36	ł	13.75	1	1		
	Area			UEP91	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75			'	
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local		1	UEF91	UEFIB	1.23	40.31	19.04	24.90	0.30		15.75			\vdash	
. '				LIEDO4	LIEDVILI	4.00	40.04	40.04	04.00	0.50		45.75			'	
	Area		 	UEP91	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75				
. '	2-Wire Voice Grade Port (Centrex from diff Serving Wire						400.0=		=						'	
	Center)2 Basic Local Area			UEP91	UEPYM	1.23	108.35	70.57	54.24	11.70		15.75				<u> </u>
. '	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			l											'	
	Term - Basic Local Area			UEP91	UEPYZ	1.23	108.35	70.57	54.24	11.70		15.75				<u> </u>
. '	2-Wire Voice Grade Port terminated in on Megalink or equivalent														'	
	- Basic Local Area			UEP91	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75				<u> </u>
. '	2-Wire Voice Grade Port Terminated on 800 Service Term -														'	
	Basic Local Area			UEP91	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75				
AL, KY	, LA, MS, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75			<u> </u>	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire														1	
	Center)2			UEP91	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75			'	
,	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
. '	Term	l	1	UEP91	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75			1 '	
. '	2-Wire Voice Grade Port terminated in on Megalink or equivalent	l	1	UEP91	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75	Ì	Ì	1 '	
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ2	1.23	40.31	19.84	24.90	6.58		15.75				
	Switching		1		1 1						İ					
	Centrex Intercom Funtionality, per port		1	UEP91	URECS	0.7947			† 1				İ	İ	 	
	Number Portability		1		1				† 1				İ	İ	 	
	Local Number Portability (1 per port)		1	UEP91	LNPCC	0.35			†		l	i	1	1		1
Feature			1		1	2.00			† †		İ	1	1	1	\vdash	
	All Standard Features Offered, per port	1	1	UEP91	UEPVF	2.56			† †			15.75	1	1	\vdash	†
	All Select Features Offered, per port	1	1 -	UEP91	UEPVS	0.00	404.98		t		1	15.75	 	 	\vdash	†
	All Centrex Control Features Offered, per port	l -	1	UEP91	UEPVC	2.56	404.30				1	15.75	 	 	\vdash	†
NARS		1	+	02101	OL. 70	2.50			 		1	10.70	 	 		
INAINO	Unbundled Network Access Register - Combination	 	+	UEP91	UARCX	0.00	0.00	0.00	 		ł	1	1	1		
 '	Unbundled Network Access Register - Indial	l -	+	UEP91	UAR1X	0.00	0.00	0.00	 		1	1	1	1		
	Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial	1	1	UEP91	UAROX	0.00	0.00	0.00	 		1	1	1	1	 	
Miccell	Jundungled Network Access Register - Outdial laneous Terminations	-	+	OCEAL	UARUA	0.00	0.00	0.00	 				 	 	\vdash	+
	Trunk Side	 	1		+ +				 		 	 	 	 	─ ──	
∠-vvire		-	1	UEP91	CENA6	8.25	120.00	40.05	C4 77	3.88	 	45.75	 	 		
	Trunk Side Terminations, each	 	+	OCEAL	CEIVAD	გ.∠ე	120.00	18.85	61.77	3.88	1	15.75			\vdash	
1									1		1	ĺ	1	1	1 '	<u> </u>
Interoff	fice Channel Mileage - 2-Wire		-	LIEDO4	MACEC	00.50	40.7-	27.55	47.00	771		45.75				
Interoff	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	22.52	40.77	27.57	17.26	7.11		15.75				
				UEP91 UEP91	M1GBC M1GBM	22.52 0.0098	40.77	27.57	17.26	7.11		15.75				

CITECIT	NDLE	NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
CATEGO	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							1	Nonroa	rrina	Nonroquerino	Disconnect				Rates(\$)		
					-		Rec	Nonrec First	Add'l	Nonrecurring First	Add'l	COMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-+		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.57	FIISL	Auu i	FIISt	Addi	SOMEC	SOWAN	JOWAN	SOWAN	SOWAN	JOWAN
		T Catalo Notivation on D 4 Chamor Bank Control 2005 Glot			OLI 01	11 0110	0.07										
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.57										
		Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
		Slot			UEP91	1PQW7	0.57										
		Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
		Different Wire Center			UEP91	1PQWP	0.57										
1		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.57										
-+		Feature Activation on D-4 Channel Bank Frivate Line Loop Slot			OLF91	IFQVV	0.57										
1		Slot			UEP91	1PQWQ	0.57										
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.57										
1		curring Charges (NRC) Associated with UNE-P Centrex															
i T		Conversion - Currently Combined Switch-As-Is with allowed															
igsquare		changes, per port			UEP91	USAC2		0.10	0.10				15.75				
ldash		Conversion of Existing Centrex Common Block			UEP91	USACN		37.97	16.68				15.75				
\longmapsto		New Centrex Standard Common Block			UEP91	M1ACS	0.00	666.32					15.75				
\longrightarrow		New Centrex Customized Common Block			UEP91	M1ACC	0.00	666.32					15.75				
\vdash		Secondary Block, per Block			UEP91	M2CC1	0.00	77.91					15.75				
	INE D	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	72.63				1	15.75				
		CENTREX - 5ESS (Valid in All States) VG Loop/2-Wire Voice Grade Port (Centrex) Combo															-
		ort/Loop Combination Rates (Non-Design)															-
- $+$ $$		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
1		Non-Design		1	UEP95		12.22										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		i i	02.00		12.22										
1		Non-Design		2	UEP95		17.13										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design		3	UEP95		26.26										
i l		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
\longmapsto		Non-Design		4	UEP95		44.91										
<u></u> _u		ort/Loop Combination Rates (Design)															
1		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -			LIEBOE		45.40										
\vdash		Design		1	UEP95		15.12										
1		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP95		19.98										
+		Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEF95		19.90										
1		Design		3	UEP95		28.78										
-+		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		Ť	00		20.70										
ı l		Design		4	UEP95		46.95										
l	UNE Lo	pop Rate															
		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	10.98										
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	15.91										
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	25.04										
\longrightarrow		2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP95	UECS1	43.68										
\vdash		2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	13.89					<u> </u>					
		2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95 UEP95	UECS2 UECS2	18.75 27.55			1							
		2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 4		4	UEP95	UECS2	45.72			-		1				 	
		ort Rate		+ -	OLI 33	01002	45.72			1		1					-
	All Stat																
– ť		2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.23	40.31	19.84	24.90	6.58		15.75				
-		2-Wire Voice Grade Port (Centrex 900 termination)			UEP95	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75				
		2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
الـــــا		Area			UEP95	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75				
		2-Wire Voice Grade Port (Centrex from diff Serving Wire												_			
. 1		Center)2 Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP95	UEPYM	1.23	108.35	70.57	54.24	11.70		15.75				
														_			

ONROND	LED NETWORK ELEMENTS - Mississippi			1							_			ment: 2		bit: C
CATEGORY	rate elements	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port terminated in on Megalink or equivalen	t														
	- Basic Local Area			UEP95	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP95	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75				
AL,	KY, LA, MS, SC, & TN Only 2-Wire Voice Grade Port (Centrex)		1	UEP95	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)	+	-	UEP95	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex vith Caller ID)1	-	1	UEP95	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75		-		
	2-Wire Voice Grade Port (Centrex with Carler 15)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire		1	OLF 93	ULFQII	1.23	40.31	15.04	24.90	0.30		13.73				1
	Center)2			UEP95	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OL1 30	OLI QIVI	1.20	100.00	70.01	04.24	11.70		10.70				
	Term			UEP95	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75				
		1			: ~ -	20		. 0.01	J 2 -					1		
	2-Wire Voice Grade Port terminated in on Megalink or equivalen	t		UEP95	UEPQ9	1.23	40.31	19.84	24.90	6.58	1	15.75		I		
	2-Wire Voice Grade Port Terminated on 800 Service Term	1		UEP95	UEPQ2	1.23	40.31	19.84	24.90	6.58		15.75		İ		
FL &	& GA Only	1														
	al Switching															1
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.7947										1
Loc	al Number Portability															1
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										1
Fea	tures															
	All Standard Features Offered, per port			UEP95	UEPVF	2.56						15.75				Ì
	All Select Features Offered, per port			UEP95	UEPVS	0.00	404.98					15.75				
	All Centrex Control Features Offered, per port			UEP95	UEPVC	2.56						15.75				
NAF																
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				15.75				
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				15.75				
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				15.75				
	cellaneous Terminations															
2-W	fire Trunk Side															
	Trunk Side Terminations, each	1	1	UEP95	CEND6	8.25	120.00	18.85	61.77	3.88		15.75				
4-W	fire Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP95	M1HD1	58.41	203.19	96.25	74.86	2.54		15.75				
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	14.56									
Inte	eroffice Channel Mileage - 2-Wire			115505					47.00							
	Interoffice Channel Facilities Termination			UEP95	MIGBC	22.52	40.77	27.57	17.26	7.11		15.75				
	Interoffice Channel mileage, per mile or fraction of mile	<u> </u>		UEP95	MIGBM	0.0098										
	ture Activations (DS0) Centrex Loops on Channelized DS1 Servi Channel Bank Feature Activations	ce	1													
D4 (Feature Activation on D-4 Channel Bank Centrex Loop Slot		1	UEP95	1PQWS	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot	+	-	UEP95	IPQWS	0.57										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.57										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop	+	1	OLF 93	IFQWO	0.37										-
	Slot			UEP95	1PQW7	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -	1	1	OL1 33	11 Q 11 /	0.37								 	1	
	Different Wire Center			UEP95	1PQWP	0.57										
	Directorit Wile Genter			OL1 30	11 QVVI	0.07										1
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.57								1		
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP95	1PQWQ	0.57					1			I		
	Feature Activation on D-4 Channel Bank WATS Loop Slot	1		UEP95	1PQWA	0.57										1
Non	n-Recurring Charges (NRC) Associated with UNE-P Centrex															1
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP95	USAC2		0.10	0.10			1	15.75		I		
	Conversion of Existing Centrex Common Block, each	1		UEP95	USACN		37.97	16.68				15.75				
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	666.32					15.75				
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	666.32					15.75				
	NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	72.63					15.75				1
	E-P CENTREX - DMS100 (Valid in All States)	1	1	1	1						İ				İ	1

UNBUNDL	ED NETWORK ELEMENTS - Mississippi												Attachi	ment: 2	Exhil	oit: C
											Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		l									-		Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
		1					Nonred	rurring	Nonrecurring	Disconnect		J.	oss	Rates(\$)	l	
					+	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wi	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo						11130	Addi	11130	Addi	COMILO	COMPAR	COMPAR	COMPAR	COMPAR	COMPAR
	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP9D		12.22										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP9D		17.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		3	LIEDOD		20, 20										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	<u> </u>	3	UEP9D	+	26.26					-					
	Non-Design	1	4	UEP9D		44.91										
UNE	Port/Loop Combination Rates (Design)		_	OLI OD		44.01										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1			1											
	Design	<u> </u>	1	UEP9D	<u> </u>	15.12					<u> </u>	<u> </u>		<u> </u>	<u> </u>	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9D		19.98										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1		LIEDOD		00 =0						1				
\vdash	Design		3	UEP9D	+	28.78										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		4	UEP9D		46.95										
LINE	Loop Rate	1	4	OLF3D	1	40.53										
ONE	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP9D	UECS1	10.98										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	15.91								İ	t	
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	25.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP9D	UECS1	43.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	13.89										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	18.75										
+-	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	27.55										
LINE	2-Wire Voice Grade Loop (SL 2) - Zone 4 Port Rate		4	UEP9D	UECS2	45.72			1							
	STATES				+											
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.23	40.31	19.84	24.90	6.58		15.75		1		
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP9D	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
	Area			UEP9D	UEPYC	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local			UEP9D	UEPYD	1.23	40.24	10.04	24.00	6.58		15 75		1	1	
\vdash	Area 2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local	 		OFLAD	UEFID	1.23	40.31	19.84	24.90	86.0		15.75				
	Area	1		UEP9D	UEPYE	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local	1			1	20				2.00				1	1	
	Area	<u></u>		UEP9D	UEPYF	1.23	40.31	19.84	24.90	6.58		15.75		<u></u>	<u></u>	
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local	1										1				
	Area	ļ		UEP9D	UEPYG	1.23	40.31	19.84	24.90	6.58		15.75		1	1	
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local	1		LIEDOD	LIEDVE	4.00	40.04	40.04	04.00	0.50		45.75				
\vdash	Area 2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local	 		UEP9D	UEPYT	1.23	40.31	19.84	24.90	6.58		15.75		-	-	
	Area	1		UEP9D	UEPYU	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local	 		S_1 0D	JE: 70	1.23	70.01	13.04	24.30	0.36		10.70		t	 	
	Area			UEP9D	UEPYV	1.23	40.31	19.84	24.90	6.58		15.75		1	1	
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local				1 1											
	Area	1		UEP9D	UEPY3	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local			l										1	1	
	Area	ļ		UEP9D	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75		ļ		
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp	1		LIEDOD	LIEDVA	4.00	40.04	40.04	04.00	0.50		45.75				
\vdash	Indication))3 Basic Local Area 2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3	 		UEP9D	UEPYW	1.23	40.31	19.84	24.90	6.58		15.75		-	-	
	Basic Local Area			UEP9D	UEPYJ	1.23	40.31	19.84	24.90	6.58		15.75		1	1	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)	<u> </u>			1	20		.0.04	200	0.00		.0.70		1	1	
	2 Basic Local Area	1		UEP9D	UEPYM	1.23	108.35	70.57	54.24	11.70		15.75		1	1	

<u> </u>	ED NETWORK ELEMENTS - Mississippi												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring		001150	001111		Rates(\$)	001141	001141
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3				_		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Basic Local Area			UEP9D	UEPYO	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			OLI SD	OLI 10	1.20	100.00	70.07	04.24	11.70		10.70				1
	Basic Local Area			UEP9D	UEPYP	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3															
	Basic Local Area			UEP9D	UEPYQ	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3															
	Basic Local Area			UEP9D	UEPYR	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			OLFBD	OLF13	1.23	100.55	70.37	34.24	11.70		13.73				
	Basic Local Area			UEP9D	UEPY4	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3					_			-							
	Basic Local Area			UEP9D	UEPY5	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3															
	Basic Local Area			UEP9D	UEPY6	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			LIEDOD	LIEDV7	4.00	100.05	70.57	54.04	44.70		45.75				
	Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP9D	UEPY7	1.23	108.35	70.57	54.24	11.70		15.75				<u> </u>
	Term			UEP9D	UEPYZ	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			OLF 9D	OLFIZ	1.23	100.33	70.57	34.24	11.70		13.73				
	Basic Local Area			UEP9D	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic															
	Local Area			UEP9D	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75				
AL, K	Y, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D UEP9D	UEPQC UEPQD	1.23	40.31 40.31	19.84 19.84	24.90	6.58 6.58		15.75 15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3 2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.23 1.23	40.31	19.84	24.90 24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3 2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.23	40.31	19.84	24.90	6.58		15.75				1
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			UEP9D	UEPQW	1.23	40.31	19.84	24.90	6.58		15.75				
	Indication)3 2-Wire Voice Grade Port (Centrex/Msq Wtq Lamp Indication)3			UEP9D	UEPQV	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			OLFBD	ULFQJ	1.23	40.31	15.04	24.50	0.36		13.73				
	2			UEP9D	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.23	108.35	70.57	54.24	11.70		15.75				
	·															1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	1.23	108.35	70.57	54.24	11.70		15.75				
	0 M/ Value O I- Boot (O I / I'' O M/O /EBO ME140)0 0			LIEDOD	LIEDOD	4.00	400.05	70.57	54.04	44.70		45.75				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.23	108.35	70.57	54.24	11.70		15.75				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.23	108.35	70.57	54.24	11.70		15.75				
	2 1 5 1.500 Grado i Gra (Gerta avalliei GVV / EBO-190512)2, 5			02.100	OL. 40	1.23	100.33	10.51	57.24	11.70		10.73			1	†
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.23	108.35	70.57	54.24	11.70	1	15.75				
ĺ	, .															
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.23	108.35	70.57	54.24	11.70		15.75				
	O. M. C. Marian Const. Part (O. Mar. 1977) Children Const.			LIEDOD	LIEDO A			====				,				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.23	108.35	70.57	54.24	11.70		15.75			 	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.23	108.35	70.57	54.24	11.70	1	15.75			l	

UNBUNDL	ED NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						I	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
					-	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service						11130	Auu	11130	даат	COME	COMPAR	COMPAR	COMPAR	COMPAR	COMPAR
	Term			UEP9D	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D UEP9D	UEPQ9 UEPQ2	1.23 1.23	40.31 40.31	19.84 19.84	24.90 24.90	6.58 6.58		15.75				
Loca	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPQZ	1.23	40.31	19.84	24.90	6.38	-	15.75			-	
Loca	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.7947										1
Loca	I Number Portability		1	OLI 3D	OKLOO	0.7347			1							
2000	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35			<u> </u>							1
Feat				02. 02	2.1. 00	0.00										
	All Standard Features Offered, per port			UEP9D	UEPVF	2.56						15.75				
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	404.98		1			15.75		İ	1	
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	2.56			1			15.75		İ	1	
NAR									1						1	1
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				15.75				
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				15.75				
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00				15.75				
Misc	ellaneous Terminations															
2-Wi	re Trunk Side															
	Trunk Side Terminations, each			UEP9D	CEND6	8.25	120.00	18.85	61.77	3.88		15.75				
4-Wi	re Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	58.41	203.19	96.25	74.86	2.54		15.75				
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	14.56									
Inter	office Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	22.52	40.77	27.57	17.26	7.11		15.75				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0098										
	ure Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 C	hannel Bank Feature Activations			UEP9D	1PQWS	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot		1	UEP9D	1PQWS	0.57									-	
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.57										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.57										
			1	l											_	
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.57									1	ļ
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop	l													1	
ļ	Slot	<u> </u>	<u> </u>	UEP9D	1PQWQ	0.57								ļ	-	
Ala	Feature Activation on D-4 Channel Bank WATS Loop Slot	 	-	UEP9D	1PQWA	0.57			 		-			 	 	
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex NRC Conversion Currently Combined Switch-As-Is with allowed	!	 						 					-		
	changes, per port	l	1	UEP9D	USAC2		0.10	0.10				15.75		1	I	
 	Conversion of existing Centrex Common Block, each	1	1	UEP9D	USACN		37.97	16.68	1			15.75		1	 	
 	New Centrex Standard Common Block	 		UEP9D	M1ACS	0.00	666.32	10.00	1			15.75		1	t	
-	New Centrex Standard Common Block	-		UEP9D	M1ACC	0.00	666.32		 			15.75		 	t	
	NAR Establishment Charge, Per Occasion	1		UEP9D	URECA	0.00	72.63		† †		<u> </u>	15.75		 	I	
UNF	P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)	1			0.1.20/1	0.00	72.00		† †		<u> </u>	10.70		 	I	†
	re VG Loop/2-Wire Voice Grade Port (Centrex) Combo	1							† †		<u> </u>			 	I	†
	Port/Loop Combination Rates (Non-Design)								1					İ	1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP9E		12.22										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9E		17.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		3	UEP9E		26.26										
UNF	Non-Design Port/Loop Combination Rates (Design)		4	UEP9E		44.91					-				-	-

ONRONDI	ED NETWORK ELEMENTS - Mississippi			,										ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						_	Nonrec	curring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		†					7144		7.44		00				
	Design		1	UEP9E		15.12										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9E		19.98										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP9E		28.78										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		4	UEP9E		46.95										
UNE	Loop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	10.98										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	15.91										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	25.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP9E	UECS1	43.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	13.89										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	18.75										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	27.55										
	2-Wire Voice Grade Loop (SL 2) - Zone 4		4	UEP9E	UECS2	45.72										
	Port Rate															
AL,	FL, KY, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP9E	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP9E	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP9E	UEPYM	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP9E	UEPYZ	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP9E	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area		<u> </u>	UEP9E	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75				
AL,	KY, LA, MS, & TN Only			LIEDOE	LIEDOA	4.00	40.04	40.04	24.00	0.50		45.75				
	2-Wire Voice Grade Port (Centrex)		ļ	UEP9E	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75				-
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex with Caller ID)1	-	-	UEP9E	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75		-	-	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2	l		UEP9E	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75			1	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1	1	OLI: 9L	OLF QIVI	1.23	100.33	10.51	34.24	11.70	1	13.73			1	+
	Term	l		UEP9E	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75			1	
	1000		†	OLI JL	ULI QL	1.23	100.33	10.31	J4.24	11.70	 	13.73		 	t	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1		UEP9E	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75		1	I	
	2-Wire Voice Grade Port Terminated in on Negarink of equivalent		†	UEP9E	UEPQ2	1.23	40.31	19.84	24.90	6.58	 	15.75		 	t	
Loca	al Switching	1	1	J J_	JE1 42	1.20	70.01	10.04	24.30	0.00		10.75			-	†
	Centrex Intercom Funtionality, per port	1		UEP9E	URECS	0.7947			1		1			 	I	1
Loca	al Number Portability	l	1		5200	0047								1	1	1
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35								İ	1	1
Feat	ures				1	2.20								İ	1	1
	All Standard Features Offered, per port			UEP9E	UEPVF	2.56						15.75			1	1
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	404.98				Ì	15.75				
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	2.56						15.75				1
NAR																1
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00				15.75				1
	Unbundled Network Access Register - Indial	<u></u>		UEP9E	UAR1X	0.00	0.00	0.00				15.75				
	Unbundled Network Access Register - Outdial		L	UEP9E	UAROX	0.00	0.00	0.00				15.75				
	cellaneous Terminations															
2-W	re Trunk Side															
	Trunk Side Terminations, each			UEP9E	CEND6	8.25	120.00	18.85	61.77	3.88		15.75				
4-W	re Digital (1.544 Megabits)															

UNBUNDLE	D NETWORK ELEMENTS - Mississippi													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	DS1 Circuit Terminations, each			UEP9E	M1HD1	58.41	First	Add'I 96.25	First 74.86	Add'l 2.54	SOMEC	SOMAN 15.75	SOMAN	SOMAN	SOMAN	SOMAN
	DS0 Channel Activated Per Channel			UEP9E	M1HD0	0.00	203.19 14.56	96.25	74.86	2.54	-	15.75				
Interes	ffice Channel Mileage - 2-Wire			UEF9E	MINDO	0.00	14.56					15.75				
intero	Interoffice Channel Facilities Termination			UEP9E	MIGBC	22.52	40.77	27.57	17.26	7.11	1	15.75				1
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0098	40.77	21.01	17.20	7.11		13.73				
Featur	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e		OLI OL	IVIIODIVI	0.0000										1
	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.57						15.75				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.57						15.75				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9E	1PQW7	0.57						15.75				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9E	1PQWP	0.57						15.75				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.57						15.75				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9E	1PQWQ	0.57						15.75				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.57						15.75				
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed			UEP9E	USAC2		0.10	0.10				15.75				
	changes, per port Conversion of Existing Centrex Common Block, each			UEP9E	USACN		37.97	0.10 16.68				15.75 15.75				
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	666.32	10.00			1	15.75				1
	New Centrex Customized Common Block			UEP9E	M1ACC	0.00	666.32					15.75				
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	72.63					15.75				1
UNE-P	CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)			02. 02	0112071	0.00	72.00					10.70				
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE P	ort/Loop Combination Rates (Non-Design)															1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														ĺ
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	UEP93		12.22										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP93		17.13										
_	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		3	UEP93		26.26										
	Non-Design		4	UEP93		44.91										
UNE P	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP93		15.12										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP93		19.98										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		3	UEP93		28.78										
UNE I	Design Oop Rate		4	UEP93		46.95										-
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP93	UECS1	10.98								1		
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	15.91										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	25.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP93	UECS1	43.68		-		-			_			
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	13.89										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	18.75										
	2-Wire Voice Grade Loop (SL 2) - Zone 3	ļ	3	UEP93	UECS2	27.55									ļ	<u> </u>
	2-Wire Voice Grade Loop (SL 2) - Zone 4	 	4	UEP93	UECS2	45.72									1	
	ort Rate	-	1		+				 					 	1	
IAL KY	Y, LA, MS, & TN only	1	1	1					1		1	ĺ	1	1	1	1

NRONDLE	D NETWORK ELEMENTS - Mississippi			1							1 -			nent: 2		oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec	urring	Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP93	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP93	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			UEP93	UEPYM	1.23	108.35	70.57	54.24	11.70		15.75				
	Center)2 Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEF93	UEPTIVI	1.23	106.33	70.57	54.24	11.70		15.75				
	Term - Basic Local Area			UEP93	UEPYZ	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			OLI SO	OLI IZ	1.20	100.00	70.07	04.24	11.70		10.70				
	- Basic Local Area			UEP93	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area		1	UEP93	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75			1	1
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP93	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			LIEDOO	115007	4.00	400.05	70.57	54.04	44.70		45.75				
	Term			UEP93	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75				
-	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75				
Local	Switching			UEF93	UEFQZ	1.23	40.31	19.04	24.90	0.30		15.75				
Local	Centrex Intercom Funtionality, per port			UEP93	URECS	0.7947										
Local	Number Portability			02. 00	0.1200	00										
	Local Number Portability (1 per port)			UEP93	LNCCC	0.35										
Featu	res															
	All Standard Features Offered, per port			UEP93	UEPVF	2.56						15.75				
	All Centrex Control Features Offered, per port			UEP93	UEPVC	2.56						15.75				
NARS																
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00				15.75				
	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00				15.75				
Micco	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00				15.75				
	Trunk Side				+											
2-99116	Trunk Side Terminations, each			UEP93	CEND6	8.25	120.00	18.85	61.77	3.88		15.75				
4-Wire	Digital (1.544 Megabits)			OL1 30	OLINDO	0.20	120.00	10.00	01.77	0.00		10.70				
	DS1 Circuit Terminations, each			UEP93	M1HD1	58.41	203.19	96.25	74.86	2.54		15.75				
	DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	14.56					15.75				
Intero	ffice Channel Mileage - 2-Wire								<u> </u>							
	Interoffice Channel Facilities Termination			UEP93	MIGBC	22.52	40.77	27.57	17.26	7.11		15.75				
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.0098										
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	е		ļ	\bot	,									ļ	
D4 Ch	annel Bank Feature Activations		ļ	LIEBOO	4001110										ļ	
_	Feature Activation on D-4 Channel Bank Centrex Loop Slot		 	UEP93	1PQWS	0.57									 	
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot		1	UEP93	1PQW6	0.57									1	1
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop	-	-	OLF 33	IF WVV0	0.57			1						1	
	Slot			UEP93	1PQW7	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			02. 00		0.01										
	Different Wire Center		1	UEP93	1PQWP	0.57]						1	1
L	Feature Activation on D-4 Channel Bank Private Line Loop Slot	<u></u>	L	UEP93	1PQWV	0.57			<u> </u>		<u></u>				<u> </u>	<u> </u>
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop												_			
	Slot			UEP93	1PQWQ	0.57										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.57									ļ	ļ
	ecurring Charges (NRC) Associated with UNE-P Centrex	1	1	1	1											L
Non-R	NRC Conversion Currently Combined Switch-As-Is with allowed															

UNBL	JNDLE	NETWORK ELEMENTS - Mississippi												Attachr	nent: 2	Exhib	oit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
													Submitted		Charge -		Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATE	GORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	l.	ı
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Conversion of Existing Centrex Common Block, each			UEP93	USACN		37.97	16.68								
		New Centrex Standard Common Block			UEP93	M1ACS	0.00	666.32					15.75				
		New Centrex Customized Common Block			UEP93	M1ACC	0.00	666.32					15.75				
		NAR Establishment Charge, Per Occasion			UEP93	URECA	0.00	72.63					15.75				
	Note 1	Required Port for Centrex Control in 1AESS, 5ESS & EWSD															
	Note 2	- Requres Interoffice Channel Mileage															
		Requires Specific Customer Premises Equipment															
	Note: F	Rates displaying an "R" in Interim column are interim and sub	ject to	rate tru	e-up as set forth in	General Term	ns and Condition	ons.	•								

UNB	UNDLE	D NETWORK ELEMENTS - South Carolina												Attach	ment: 2	Fxhil	bit: C
OIVE	UNDEL		l	1			1					Svc Order	Svc Order	Incremental			Incremental
												Submitted			Charge -	Charge -	Charge -
												Elec			Manual Svc		Manual Svo
CATE	GORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)				-				
CAIL	.001(1	KATE ELEMENTO	m	20116	B00	0000			KATEO(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
								Nonred	curring	Nonrecurring	Disconnect		l .	oss	Rates(\$)	l	l .
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN		SOMAN	SOMAN
	The "7	one" shown in the sections for stand-alone loops or loops as part of	of a com	hinatio	n refers to Geograph	ically Deavera	aged UNF Zones									COMPAN	COMPAN
		ww.interconnection.bellsouth.com/become a clec/html/interconne			roioio to occgiapi	Doaron	.gou 0.12 201100		jiapinoanj boa	roragoa orte zo	o Doolgilatio		. 011100, 1010		obolio.		
OPE		L SUPPORT SYSTEMS	1	1						1		1		1		1	1
01 11		(1) Electronic Service Order: CLEC should contact its contract	ct negot	iator if	it prefers the state	specific elec	tronic service o	rdering charge	es as ordered	by the State Cor	mmissions. T	he electron	ic service o	rdering charg	e currently co	ntained in th	is rate
		t is the BellSouth regional electronic service ordering charge.															
		(2) Any element that can be ordered electronically will be bill															ly For
		elements that cannot be ordered electronically at present per t															
		ng charge, SOMAN, will be applied to a CLECs bill when it sub				le III IIIIS Cale	gory reflects th	e charge mar v	would be billed	u to a CLLC onc	e electronic c	nuering cap	Jabilities CO	ille Oli-ille io	i tilat elelileli	i. Otherwise,	uie ilialiuai
	ordeni		Jillits ar	LOK	o belisoutii.	SOMAN				1.97		1	1	1		1	1
	-	Manual Service Order Charge, per LSR, Disconnect Only (SC) Electronic OSS Charge, per LSR, submitted via BST's OSS	1		1	SUIVIAIN				1.97		1	1		1		1
	1	interactive interfaces (Regional)	1		İ	SOMEC		3.50		1				Ì	I	Ì	l
LINE	SERVICE	DATE ADVANCEMENT CHARGE	1		 	JOIVILO	1	3.30		+ +		1	1	1	 	1	1
OINE .		The Expedite charge will be maintained commensurate with	PallCar	th's E	C No 1 Tariff Coat	ion E oo onni	iooblo			-					-		
	NOTE:	UNE Expedite Charge per Circuit or Line Assignable USOC, per	Delisou	ui s re	I No.1 Tariii, Secti	ion 5 as appi	icable.			+		1					
	1	Day	1		ALL UNE	SDASP		200.00		1				Ì	I	Ì	l
LIMPI	INDI ED I	EXCHANGE ACCESS LOOP	 		ALL UIVL	SUMSE	1	200.00		+ +		-		-		 	
CIVE		E ANALOG VOICE GRADE LOOP	1		 	1	1			+ +		1	1	1	 	1	1
-	Z-VVIIN	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	14.94	37.92	17.62	23.56	5.32		15.69		-		
	-	2-Wire Analog Voice Grade Loop - Service Level 1-Zone 1		2	UEANL	UEAL2	21.39	37.92	17.62	23.56	5.32		15.69		-		
	-	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	26.72	37.92	17.62	23.56	5.32		15.69		-		
		Loop Testing - Basic 1st Half Hour		3	UEANL	URET1	20.72	34.23	34.23	23.36	5.32		15.69				
		Loop Testing - Basic 1st Hall Hour Loop Testing - Basic Additional Half Hour			UEANL	URETA		19.90	19.90				15.69				
-	-	CLEC to CLEC Conversion Charge Without Outside Dispatch			ULANL	UKLIA		19.90	19.90	-			13.09		-		
		(UVL-SL1)			UEANL	UREWO		15.81	8.96				15.69				
	-	Engineering Information Document (EI)			UEANL	UEANM		13.47	13.47	+		1	15.69				
-	-	Manual Order Coordination for UVL-SL1s (per loop)			UEANL	UEAMC		8.17	8.17	-					-		
	-	Order Coordination for Specified Conversion Time for UVL-SL1			ULANL	ULAIVIC		0.17	0.17	-					-		
		(per LSR)			UEANL	OCOSL		18.13	18.13								
	2-WIDE	E Unbundled COPPER LOOP			ULANL	OCOSL		10.13	10.13	1							
	Z-VVIKE	2-Wire Unbundled Copper Loop - Non-Designed Zone 1		1	UEQ	UEQ2X	12.94	36.40	16.10	22.66	4.42	1	15.69				
		2 Wire Unbundled Copper Loop - Non-Designed Zone 2	l i		UEQ	UEQ2X	14.51	36.40	16.10	22.66	4.42		15.69				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	i		UEQ	UEQ2X	15.02	36.40	16.10	22.66	4.42	1	15.69				
		Order Coordination 2 Wire Unbundled Copper Loop - Non-	<u>'</u>	3	OLQ	OLQZX	10.02	30.40	10.10	22.00	7.72	1	15.05				
		Designed (per loop)			UEQ	USBMC		8.17	8.17								
—	-	Engineering Information Document	 		UEQ	CODIVIC	†	13.47	13.47	 		1	15.69	 	 	 	
	_	Loop Testing - Basic 1st Half Hour			UEQ	URET1		34.23	34.23	1		1	15.69				
—	-	Loop Testing - Basic 1st Half Hour	 		UEQ	URETA	†	19.90	19.90	 		1	15.69	 	 	 	
 	-	CLEC to CLEC Conversion Charge Without Outside Dispatch	 			0.12171		10.00	10.30	+ +		1	10.00		 		
	1	(UCL-ND)	1		UEQ	UREWO		14.30	7.45	1			15.69	Ì	I	Ì	l
UNBI	JNDLED I	EXCHANGE ACCESS LOOP	†			320		00		† †			.0.00	1	1	1	
		ANALOG VOICE GRADE LOOP	1		 	1	1			† †		1		1	t	1	
<u> </u>	T	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	†		1					† †				1	1	1	
	1	Ground Start Signaling - Zone 1	1	1	UEA	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69	Ì	I	Ì	l
	1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	†	Ė	1	1		.00.00	55.10	33.55			.0.00	1	t	1	i
	1	Ground Start Signaling - Zone 2	1	2	UEA	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69		1		
	1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or	†		1	1 -			22.10	22.23			12.20	1	t	1	i
	1	Ground Start Signaling - Zone 3	1	3	UEA	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69		1		
		Order Coordination for Specified Conversion Time (per LSR)	1		UEA	OCOSL		18.13		1				İ	İ	İ	İ
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	1	Battery Signaling - Zone 1	1	1	UEA	UEAR2	16.68	105.98	68.43	53.05	10.61		15.69	Ì	I	Ì	l
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	1	Battery Signaling - Zone 2	1	2	UEA	UEAR2	23.13	105.98	68.43	53.05	10.61		15.69		1		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse						_									
	1	Battery Signaling - Zone 3	1	3	UEA	UEAR2	28.46	105.98	68.43	53.05	10.61		15.69	Ì	I	Ì	l
		Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.13				İ					
		CLEC to CLEC Conversion Charge without outside dispatch	1		UEA	UREWO		87.90	36.44	1 1			15.69	İ	İ	İ	İ
	4-WIRE	ANALOG VOICE GRADE LOOP															
		4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
		4-Wire Analog Voice Grade Loop - Zone 2	İ	2	UEA	UEAL4	43.89	132.38	94.83		14.61	İ	15.69				

Version 3Q02: 09/06/02 Page 268 of 362

ONBONDL	LED NETWORK ELEMENTS - South Carolina			,										ment: 2		bit: C
															Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi	l_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
											-	_	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
1		-			-		Nonrec	rrina	Nonrecurring	Disconnect			000	Rates(\$)		
			1			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Analog Voice Grade Loop - Zone 3	+	3	UEA	UEAL4	43.38	132.38	94.83	59.35	14.61	SOWIEC	15.69	JOWAN	JOWAN	JOWAN	JOWAN
	Order Coordination for Specified Conversion Time (per LSR)		Ŭ	UEA	OCOSL	40.00	18.13	04.00	00.00	14.01		10.00				†
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.90	36.44				15.69				†
2-WI	IRE ISDN DIGITAL GRADE LOOP															1
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				1
	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				1
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				1
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		18.13									1
	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.82	44.25				15.69				1
2-WI	IRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COM	PATIBLE	LOOF	j												
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 1		1	UAL	UAL2X	12.19	120.84	70.56	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 2		2	UAL	UAL2X	13.71	120.84	70.56	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 3		3	UAL	UAL2X	14.14	120.84	70.56	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.13									
	2 Wire Unbundled ADSL Loop without manual service inquiry &															
	facility reservaton - Zone 1		1	UAL	UAL2W	12.19	95.81	57.82	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop without manual service inquiry &															
	facility reservaton - Zone 2		2	UAL	UAL2W	13.71	95.81	57.82	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop without manual service inquiry &															
	facility reservaton - Zone 3		3	UAL	UAL2W	14.14	95.81	57.82	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.13									
	CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		86.38	40.48				15.69				
2-WI	IRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMP		LOOP													
	2 Wire Unbundled HDSL Loop including manual service inquiry															
	& facility reservation - Zone 1		1	UHL	UHL2X	9.58	129.52	79.24	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop including manual service inquiry															
	& facility reservation - Zone 2		2	UHL	UHL2X	10.92	129.52	79.24	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop including manual service inquiry															
	& facility reservation - Zone 3		3	UHL	UHL2X	11.40	129.52	79.24	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.13									
	2 Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 1		1	UHL	UHL2W	9.58	104.49	66.50	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 2		2	UHL	UHL2W	10.92	104.49	66.50	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 3		3	UHL	UHL2W	11.40	104.49	66.50	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)		<u> </u>	UHL	OCOSL		18.13					4= 00				
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.32	40.48				15.69				
4-WI	IRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMP	ATIBLE	LOOP													
	4 Wire Unbundled HDSL Loop including manual service inquiry			l	1 11 11 437	40.00	450.40	407.00	55.40	40.00		45.00				
	and facility reservation - Zone 1		1	UHL	UHL4X	16.02	158.18	107.89	55.12	10.38		15.69				
	4-Wire Unbundled HDSL Loop including manual service inquiry		١.,	l			450.40					4= 00				
	and facility reservation - Zone 2	-	2	UHL	UHL4X	14.33	158.18	107.89	55.12	10.38		15.69				
	4-Wire Unbundled HDSL Loop including manual service inquiry		_		11111 437	40.04	450.40	407.00	55.40	40.00		45.00				
	and facility reservation - Zone 3	-	3	UHL	UHL4X	16.84	158.18	107.89	55.12	10.38		15.69				
	Order Coordination for Specified Conversion Time (per LSR) 4-Wire Unbundled HDSL Loop without manual service inquiry	+	1	UHL	OCOSL		18.13		 						-	+
	and facility reservation - Zone 1		4	UHL	UHL4W	16.02	133.14	95.16	55.12	10.38		15.69				
 	4-Wire Unbundled HDSL Loop without manual service inquiry	+		OI IL	UNL4VV	10.02	133.14	90.16	55.12	10.38	1	15.69		-	1	+
	and facility reservation - Zone 2		2	UHL	UHL4W	14.33	133.14	95.16	55.12	10.38		15.69				1
<u> </u>		+		UITL	UNL4VV	14.33	133.14	95.16	55.12	10.38		10.09			-	+
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		2	UHL	UHL4W	16.84	133.14	95.16	55.12	10.38		15.69				1
	Order Coordination for Specified Conversion Time (per LSR)	+	3	UHL	OCOSL	10.84	133.14	95.16	55.12	10.38		15.09		-	 	+
	CLEC to CLEC Conversion Charge without outside dispatch	+	 	UHL			18.13 86.32	40.48	 			15.69		-	1	+
4 1471	IRE DS1 DIGITAL LOOP	+	1	UITL	UREWO		80.32	40.48	 			10.09			-	+
4-771	4-Wire DS1 Digital Loop - Zone 1	-	1	USL	USLXX	79.51	253.03	157.89	44.80	11.73		15.69			1	+

ONBONDE	ED NETWORK ELEMENTS - South Carolina												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	l	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Digital Loop - Zone 2		2	USL	USLXX	136.00	253.03	157.89	44.80	11.73		15.69				
	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	229.15	253.03	157.89	44.80	11.73		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		18.13									
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		101.30	43.13				15.69				
4-WIF	RE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	29.93	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	33.99	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	34.74	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1			UDL	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
 	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3	<u> </u>	3	UDL	UDL56	34.74	126.66	89.12	59.35	14.61		15.69		ļ	-	
	Order Coordination for Specified Conversion Time (per LSR)	<u> </u>		UDL	OCOSL	00.00	18.13	00.10	50.05	1101		45.00		ļ	-	
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1			UDL	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
 	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2 4 Wire Unbundled Digital Loop 64 Kbps - Zone 3	 	2	UDL UDL	UDL64 UDL64	33.99 34.74	126.66 126.66	89.12 89.12	59.35 59.35	14.61 14.61		15.69 15.69		 	 	
	Order Coordination for Specified Conversion Time (per LSR)		3	UDL	OCOSL	34.74	126.66	89.12	59.35	14.61		15.69				
-	CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		102.34	49.85	1			15.69				
2 WIE	RE Unbundled COPPER LOOP			UDL	UKEWU		102.34	49.00				15.69				
2-4411	2-Wire Unbundled Copper Loop/Short including manual service															-
	inquiry & facility reservation - Zone 1		1	UCL	UCLPB	12.19	119.91	69.62	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Short including manual service		- ' -	UCL	OCLFB	12.19	119.91	09.02	30.37	7.55		13.09				1
	inquiry & facility reservation - Zone 2		2	UCL	UCLPB	13.71	119.91	69.62	50.37	7.93		15.69				
	2 Wire Unbundled Copper Loop/Short including manual service		-	002	COLIB	10.71	110.01	00.02	00.01	7.00		10.00				+
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	14.14	119.91	69.62	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)		_ <u> </u>	UCL	UCLMC		8.17	8.17	00.07	7.00		10.00				1
	2-Wire Unbundled Copper Loop/Short without manual service			002	CCLING		0	0								
	inquiry and facility reservation - Zone 1		1	UCL	UCLPW	12.19	94.87	56.89	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Short without manual service					-										
	inquiry and facility reservation - Zone 2		2	UCL	UCLPW	13.71	94.87	56.89	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Short without manual service															1
	inquiry and facility reservation - Zone 3		3	UCL	UCLPW	14.14	94.87	56.89	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.															
	inquiry and facility reservation - Zone 1		1	UCL	UCL2L	38.22	119.91	69.62	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	55.33	119.91	69.62	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL2L	67.95	119.91	69.62	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 1		1	UCL	UCL2W	38.22	94.87	56.89	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - without manual service		_					=		=		4= 00				
	inquiry and facility reservation - Zone 2		2	UCL	UCL2W	55.33	94.87	56.89	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - without manual service		_					=				4= 00				
	inquiry and facility reservation - Zone 3		3	UCL	UCL2W UCLMC	67.95	94.87	56.89	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLINC		8.17	8.17								
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)			UCL	UREWO		94.87	42.57				15.60				
4 10/15	RE COPPER LOOP			UCL	UREWU		94.87	42.57				15.69				
4-1/1	4-Wire Copper Loop/Short - including manual service inquiry	1							+		1			1	 	
	and facility reservation - Zone 1	l	1	UCL	UCL4S	19.64	144.17	93.88	55.12	10.38		15.69			1	
 	4-Wire Copper Loop/Short - including manual service inquiry	1		UUL	UUL43	19.04	144.17	93.08	55.12	10.38	-	13.09		1	 	
	and facility reservation - Zone 2	l	2	UCL	UCL4S	20.90	144.17	93.88	55.12	10.38		15.69			1	
 	4-Wire Copper Loop/Short - including manual service inquiry	1	 	00L	00140	20.90	177.17	33.00	55.12	10.30		10.03		 	 	
	and facility reservation - Zone 3	1	3	UCL	UCL4S	19.34	144.17	93.88	55.12	10.38	1	15.69		1	I	
 	Order Coordination for Unbundled Copper Loops (per loop)	1	3	UCL	UCLMC	10.04	8.17	8.17	55.12	10.30	 	10.08		 	I	†
 	4-Wire Copper Loop/Short - without manual service inquiry and	1	1		CCLIVIC		0.17	0.17	† †					 	t	—
	facility reservation - Zone 1	l	1	UCL	UCL4W	19.64	119.13	81.15	55.12	10.38		15.69		Ì	I	

UNBUNDL	ED NETWORK ELEMENTS - South Carolina												Attach	ment: 2	Fxhil	oit: C
CITECITE	D NETWORK ELEMENTO GOURN GALORING										Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		Intori									Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									P	p-0.	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Copper Loop/Short - without manual service inquiry and								== 40							
	facility reservation - Zone 2		2	UCL	UCL4W	20.90	119.13	81.15	55.12	10.38		15.69				
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 3		3	UCL	UCL4W	19.34	119.13	81.15	55.12	10.38		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	4-Wire Unbundled Copper Loop/Long - includes manual svc.		1	UCL	UCL4L	77.29	144.17	93.88	55.40	10.38		15.69				
—	inquiry and facility reservation - Zone 1 4-Wire Unbundled Copper Loop/Long - includes manual svc.		1	UCL	UCL4L	77.29	144.17	93.88	55.12	10.38		15.69				
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	118.78	144.17	93.88	55.12	10.38		15.69				
-	4-Wire Unbundled Copper Loop/Long - includes manual svc.			UCL	UCL4L	110.76	144.17	93.00	33.12	10.36	1	13.09				
	inquiry and facility reservation - Zone 3	1	3	UCL	UCL4L	144.10	144.17	93.88	55.12	10.38		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)	1	3	UCL	UCLMC	144.10	8.17	8.17	55.12	10.30	<u> </u>	10.09	 	 		
	4-Wire Unbundled Copper Loop/Long - without manual svc.	 	1		COLIVIO		0.17	0.17	-							
	inquiry and facility reservation - Zone 1	1	1	UCL	UCL4O	77.29	119.44	81.45	55.12	10.38		15.69	1	1		
	4-Wire Unbundled Copper Loop/Long - without manual svc.	1	† ·			20		30	55.12				1			
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL4O	118.78	119.44	81.45	55.12	10.38		15.69	1	1		
	4-Wire Unbundled Copper Loop/Long - without manual svc.	†	T -	1			7.04	010	002				İ	1		
	inquiry and facility reservation - Zone 3		3	UCL	UCL4O	144.10	119.44	81.45	55.12	10.38		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	CLEC to CLEC Conversion Charge without outside dispatch															
	(UCL-Des)			UCL	UREWO		94.87	42.57				15.69				
LOOP MODI	FICATION															
				UAL, UHL, UCL,												
				UEQ, ULS, UEA,												
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire			UEANL, UDL, UDC,												
	pair less than or equal to 18k ft			UDN, UDL, USL	ULM2L		32.46	32.46				15.69				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire															
	greater than 18k ft			UCL, ULS, UEQ	ULM2G		170.89	170.89				15.69				
	Unbundled Loop Modification Removal of Load Coils - 4 Wire															
-	less than or equal to 18K ft			UHL, UCL	ULM4L		32.46	32.46				15.69				
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18k ft			UCL	ULM4G		170.89	170.89				15.69				
	pair greater triair Tok It			UAL, UHL, UCL,	ULIVI4G		170.69	170.09				15.69				
				UEQ, UEF, ULS,												
				UEA, UEANL, UDL,												
	Unbundled Loop Modification Removal of Bridged Tap Removal,			UDC, UDN, UDL,												
	per unbundled loop			USL	ULMBT		32.48	32.48				15.69				
SUB-LOOPS		†					22.10	50	1							
	Loop Distribution	1			1										1	1
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-															
	Up	L		UEANL	USBSA		241.42	241.42	<u> </u>			15.69	<u> </u>	<u> </u>	<u></u>	<u></u>
		1	1]]]		
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	I		UEANL	USBSB		22.69	22.69				15.69				
	Sub-Loop - Per Building Equipment Room - CLEC Feeder	1	1	l	I				_				1]		
\vdash	Facility Set-Up			UEANL	USBSC		177.84	177.84	.			15.69	ļ			
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel	Ι.		l					1							
\vdash	Set-Up	<u> </u>		UEANL	USBSD		55.58	55.58	-			15.69	 	 	ļ	ļ
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1	1 .	1	UEANL	USBN2	8.87	65.94	31.03	45.35	6.71		15.69	1	1		
\vdash			1	UEAINL	USBN2	8.87	65.94	31.03	45.35	6.71	-	15.69	 	 		-
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2	1 .	2	UEANL	USBN2	12.58	65.94	31.03	45.35	6.71		15.69	1	1		
 	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -	- '-		ULAINL	OODINZ	12.30	05.94	31.03	40.33	0.71		13.09	 	 		
	Zone 3	1 .	3	UEANL	USBN2	14.79	65.94	31.03	45.35	6.71		15.69	1	1		
	2010 0	 		J	335112	14.73	55.54	01.00	70.00	0.71		10.00				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -	†			2 320		3.17	5.17	1							
	Zone 1	1	1	UEANL	USBN4	14.11	79.21	44.29	49.82	9.09		15.69	1	1		
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
1 1	Zone 2	1	2	UEANL	USBN4	19.40	79.21	44.29	49.82	9.09		15.69	Ì	Ì		

UNRU	INDI FI	D NETWORK ELEMENTS - South Carolina												Attach	ment: 2	Fyhil	bit: C
3.400		NET TORK ELLINER TO - OUGH Carolina										Svc Order	Svc Order	Incremental	Incremental		
												Submitted			Charge -	Charge -	Charge -
			Intori									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									po. zo.t	po. 20.1	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																2.00 .00	2.007.001
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -					40.00	=									
		Zone 3		3	UEANL	USBN4	18.90	79.21	44.29	49.82	9.09		15.69				.
		Onder Consideration for Holorodical Cub Loops, and sub-loop aris			UEANL	USBMC		8.17	8.17								
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INC)			UEANL	USBR2	2.41	53.13	18.21	45.35	6.71		15.69				
		Sub-Loop 2-wife intrabuliding Network Cable (INC)	-		UEAINL	USBRZ	2.41	33.13	10.21	45.55	6.71	1	15.69				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
		Sub-Loop 4-Wire Intrabuilding Network Cable (INC)			UEANL	USBR4	5.36	59.38	24.47	49.82	9.09		15.69				+
		Cas 200p 1 trino initiasanang Hothoric Casto (into)	<u> </u>		027.11.12	005.	0.00	00.00		10.02	0.00		10.00				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1	UEF	UCS2X	7.11	65.94	31.03	45.35	6.71		15.69				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	ı	2	UEF	UCS2X	9.83	65.94	31.03	45.35	6.71		15.69				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	ı		UEF	UCS2X	10.48	65.94	31.03	45.35	6.71		15.69				
		·															
	<u></u>	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.17	8.17					<u> </u>			
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	Ī	1	UEF	UCS4X	7.85	79.21	44.29	49.82	9.09		15.69				
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2			UEF	UCS4X	14.17	79.21	44.29	49.82	9.09		15.69				
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	ı	3	UEF	UCS4X	12.64	79.21	44.29	49.82	9.09		15.69				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.17	8.17								
	Unbun	dled Sub-Loop Modification															
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load															
		Coil/Equip Removal per 2-W PR		<u> </u>	UEF	ULM2X		176.17	5.11				15.69				ļ
		Unbundled Sub-loop Modification - 4-W Copper Dist Load			UEF	ULM4X		470 47	5.44				45.00				
		Coil/Equip Removal per 4-W PR		1	UEF	ULIM4X		176.17	5.11				15.69				
		Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR unloaded			UEF	ULM4T		278.82	6.13				15.69				
	Unbun	dled Network Terminating Wire (UNTW)			OLI	OLIVI41		270.02	0.13				13.09				-
	Onban	Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.3303	30.20	30.20				15.69				+
	Networ	k Interface Device (NID)			02	02.1.	0.0000	00.20	00.20				10.00				
		Network Interface Device (NID) - 1-2 lines			UENTW	UND12		43.68	28.79	İ			15.69				
		Network Interface Device (NID) - 1-6 lines			UENTW	UND16		64.42	49.53				15.69				
		Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		5.92	5.92				15.69				
		Network Interface Device Cross Connect - 4W			UENTW	UNDC4		5.92	5.92				15.69				
SUB-L																	
	Sub-Lo	oop Feeder															
		USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
	<u> </u>	Distribution Facility set-up			UDN,UCL,UDL,UDC	USBFW		241.42				<u> </u>	15.69				1
		USL Feeder - DS0 Set-up per Cross Box location - per 25 pair	l		UEA,												
	<u> </u>	set-up	ļ	ļ	UDN,UCL,UDL,UDC			22.69	22.69	-			15.69				↓
	1	USL Feeder DS1 Set-up at DSX location, per DS1 termination	l	-	USL	USBFZ		523.87	11.34	 			15.69	 	ļ		
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice	l	1	UEA	USBFA	8.93	93.28	56.69	54.68	13.74		15.69				
<u> </u>	 	Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice	<u> </u>	 	UEA	USBFA	8.93	93.28	90.09	54.68	13.74	 	15.69				
1		Grade - Zone 2	l	2	UEA	USBFA	11.74	93.28	56.69	54.68	13.74		15.69	1	1		
-	1	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,	 		ULA	JJDI A	11.74	93.28	56.69	34.08	13.74	1	15.69	1	1		1
		Voice Grade - Zone 3	l	3	UEA	USBFA	14.74	93.28	56.69	54.68	13.74		15.69	1	1		
	 	Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL	17.74	18.13	30.09	54.00	15.74	 	10.09	 	 		
 	†	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice	1	<u> </u>		2000		10.10		I		1		 	 		†
		Grade - Zone 1	l	1	UEA	USBFB	8.93	93.28	56.69	54.68	13.74		15.69	1	1		
	1	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice		Ė			2.20	22.20	22.30	230							
		Grade - Zone 2	l	2	UEA	USBFB	11.74	93.28	56.69	54.68	13.74		15.69	1	1		
		Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice						_									
<u></u>	<u></u>	Grade - Zone 3	<u></u>	3	UEA	USBFB	14.74	93.28	56.69	54.68	13.74	<u> </u>	15.69	<u> </u>			
		Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		18.13	•								
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,	l	1]]		
	<u> </u>	Voice Grade - Zone 1		1	UEA	USBFC	8.93	93.28	56.69	54.68	13.74		15.69				<u> </u>
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,	1	1	[. <u></u> .					_				1]		
		Voice Grade - Zone 2		2	UEA	USBFC	11.74	93.28	56.69	54.68	13.74		15.69				1

ONRONDER	D NETWORK ELEMENTS - South Carolina			•										ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge -
-						,									D130 131	DISC Add I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	Habitadiad Cub Lasa Fasdari and O'Mira Applea Deversa		1	-	-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zone 3		3	UEA	USBFC	14.74	93.28	56.69	54.68	13.74		15.69				
	Order Coordination For Specified Conversion Time, per LSR		3	UEA	OCOSL	14.74	18.13	36.69	34.00	13.74		15.69		-		+
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice			OLA	OCOGL		10.13									+
	Grade - Zone 1		1	UEA	USBFD	21.63	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice		<u> </u>	OLA	OODI D	21.00	107.51	70.50	02.20	17.52		13.03				+
	Grade - Zone 2		2	UEA	USBFD	27.57	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice		<u> </u>	02/1	002. 2	27.07	107.01	7 0.00	02.20			10.00				†
	Grade - Zone 3		3	UEA	USBFD	26.04	107.91	70.36	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		18.13			-						1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
	Grade - Zone 1		1	UEA	USBFE	21.63	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
	Grade - Zone 2	L	2	UEA	USBFE	27.57	107.91	70.36	62.26	17.52		15.69		<u> </u>		<u>1</u>
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice												_	_		
	Grade - Zone 3		3	UEA	USBFE	26.04	107.91	70.36	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		18.13									1
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1			UDN	USBFF	17.05	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2		_	UDN	USBFF	20.92	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	23.49	106.47	68.92	55.81	13.37		15.69				<u> </u>
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		18.13									
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDC	USBFS	17.05	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	20.92	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		3	UDC	USBFS	23.49	106.47	68.92	55.81	13.37 17.52	1	15.69 15.69				+
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		2	USL	USBFG	55.85 109.16	102.19 102.19	64.64 64.64	62.26 62.26	17.52		15.69				+
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3			USL	USBFG	203.35	102.19	64.64	62.26	17.52		15.69		-		+
	Order Coordination For Specified Conversion Time, Per LSR		3	USL	OCOSL	203.33	18.13	04.04	02.20	17.52		15.69		-		+
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	5.98	83.97	46.42	53.14	10.69		15.69				+
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		- '-	OOL	OODITI	5.30	00.91	70.72	33.14	10.03		15.05				+
	2		2	UCL	USBFH	4.80	83.97	46.42	53.14	10.69		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone			COL	CODITI	4.00	00.01	40.42	00.14	10.00		10.00				†
	3		3	UCL	USBFH	4.59	83.97	46.42	53.14	10.69		15.69				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		18.13									1
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	13.21	101.22	63.67	58.03	13.29		15.69				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2	UCL	USBFJ	8.28	101.22	63.67	58.03	13.29		15.69				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	8.42	101.22	63.67	58.03	13.29		15.69				1
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		18.13									1
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	21.02	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2	UDL	USBFN	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	20.17	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 1		1	UDL	USBFO	21.02	102.19	64.64	62.26	17.52		15.69				1
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		l _													
	Zone 2		2	UDL	USBFO	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -			LIBI	LIODEO	00.47	100.10	04.04	00.00	47.50		45.00				
	Zone 3		3	UDL UDL	USBFO	20.17	102.19	64.64	62.26	17.52		15.69				-
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		18.13									+
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1	l	4	UDL	USBFP	21.02	102.19	64.64	62.26	17.52		15.69		1		1
 	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -	 		ODL	USBFF	21.02	102.19	04.04	02.20	17.32	1	15.69		t	1	\leftarrow
	Zone 2	l	2	UDL	USBFP	21.30	102.19	64.64	62.26	17.52		15.69		I		
 	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		É	552	00011	21.00	102.19	04.04	02.20	17.52	 	10.09		t	1	+
	Zone 3	l	3	UDL	USBFP	20.17	102.19	64.64	62.26	17.52		15.69		I		
	Order Coordination For Specified Conversion Time, per LSR	1	Ť	UDL	OCOSL	20	18.13	004	02.20	52				1		—
SUB-LOOPS	, , , , , , , , , , , , , , , , , , , ,				1				1							1
	oop Feeder										Ì					1
	Sub Loop Feeder - DS3 - Per Mile Per Month	I		UE3	1L5SL	20.44										1
	Sub Loop Feeder - DS3 - Facility Termination Per Month	1		UE3	USBF1	348.12	3,408.62	407.90	160.83	91.17		15.69				

UNBUNE	DLED NETWORK ELEMENTS - South Carolina		,									,		ment: 2	1	bit: C
CATEGOR	Y RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
			-			ı	Nonrec	urrina	Nonrecurring	Disconnect			088	Rates(\$)		
					1	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub Loop Feeder – STS-1 – Per Mile Per Month		1	UDLSX	1L5SL	20.44	THOL	Auu	11130	Addi	JOHILO	JOINAIN	JONAN	JONAN	JOHIAN	JOHAN
	Sub Loop Feeder - STS-1 - Facility Termination Per Month	1		UDLSX	USBF7	369.07	3,408.62	407.90	160.83	91.17		15.69				
	Sub Loop Feeder – OC-3 – Per Mile Per Month	I		UDLO3	1L5SL	15.51										
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per															
	Month	++	<u> </u>	UDLO3	USBF5	56.04	0.400.00	107.00	400.00	01.17		45.00				
	Sub Loop Feeder - OC-3 - Facility Termination Per Month Sub Loop Feeder - OC-12 - Per Mile Per Month		1	UDLO3 UDL12	USBF2 1L5SL	565.50 19.08	3,408.62	407.90	160.83	91.17		15.69				
	Sub Loop Feeder - OC-12 - Fer Mile Fer Month Sub Loop Feeder - OC-12 - Facility Termination Protection Per	- '	1	UDL12	ILSSL	19.06										
	Month	1		UDL12	USBF6	669.82										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	I		UDL12	USBF3	1,840.00	3,408.62	407.90	160.83	91.17		15.69				
	Sub Loop Feeder - OC-48 - Per Mile Per Month	-		UDL48	1L5SL	62.60										
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per			l	l											
	Month	+ !	<u> </u>	UDL48	USBF9	326.16	0.504.00	407.00	100.00	04.1=		45.00				
	Sub Loop Feeder - OC-48 - Facility Termination Per Month	++		UDL48	USBF4	1,560.00	3,594.62	407.90	160.83	91.17		15.69				
HINBLIND	Sub Loop Feeder - OC-12 Interface On OC-48 ED LOOP CONCENTRATION	+ -	-	UDL48	USBF8	366.86	806.47	407.90	160.83	91.17		15.69			-	
UNBUNDL	Unbundled Loop Concentration - System A (TR008)		1	ULC	UCT8A	318.73	326.13	326.13				15.69				
	Unbundled Loop Concentration - System B (TR008)		1	ULC	UCT8B	46.69	135.89	135.89				15.69				
	Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	351.78	326.13	326.13				15.69				
	Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	78.67	135.89	135.89				15.69				
	Unbundled Loop Concentration - DS1 Loop Interface Card			ULC	UCTCO	4.42	63.43	46.18	16.83	4.71		15.69				
	Unbundled Loop Concentration - ISDN Loop Interface (Brite				l											
	Card)		1	UDN	ULCC1	7.02	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - UDC Loop Interface (Brite Card)			UDC	ULCCU	7.02	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration2 Wire Voice-Loop Start or		1	UDC	ULCCU	7.02	10.56	10.50	5.41	5.57		15.69				
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	1.75	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Batte	'n														
	Loop Interface (SPOTS Card)			UEA	ULCCR	10.42	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface															
	(Specials Card)		<u> </u>	UEA	ULCC4	6.22	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - TEST CIRCUIT Card Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop		-	ULC	UCTTC	30.38	10.56	10.50	5.41	5.37		15.69				
	Interface			UDL	ULCC7	9.21	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop		1	ODL	OLOGI	9.21	10.50	10.50	3.41	3.31		10.00				
	Interface			UDL	ULCC5	9.21	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop															
	Interface			UDL	ULCC6	9.21	10.56	10.50	5.41	5.37		15.69				
UNE OTHE	ER, PROVISIONING ONLY - NO RATE															
	NID - Dispatch and Service Order for NID installation		<u> </u>	UENTW	UNDBX	0.00	0.00		1						1	
	UNTW Circuit Id Establishment, Provisioning Only - No Rate	+	+	UENTW UEANL,UEF,UEQ,U	UENCE	0.00	0.00		 							
	Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN	0.00	0.00									
UNE OTHE	ER, PROVISIONING ONLY - NO RATE		+		5.12511	0.00	0.00		†							
				İ	1				1							
				UAL,UCL,UDC,UDL,												
	Unbundled Contact Name, Provisioning Only - no rate			UDN,UEA,UHL,ULC	UNECN	0.00	0.00									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no			LIEA LIBNI LIGI LIBO	LIODEO	0.00	0.00									
	rate		-	UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			UEA,USL,UCL,UDL	USBFR	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate	1	1	USL	CCOSF	0.00	0.00		 						1	
	Unbundled DS1 Loop - Expanded Superframe Format option -	1	1	İ	1	2.20	2.20									
	no rate			USL	CCOEF	0.00	0.00		<u> </u>							
HIGH CAP	ACITY UNBUNDLED LOCAL LOOP							· · · · ·		· · · · ·						
	High Capacity Unbundled Local Loop - DS3 - Per Mile per				41.515											
	month		<u> </u>	UE3	1L5ND	12.26									1	
i 1	High Capacity Unbundled Local Loop - DS3 - Facility Termination per month	1	1	UE3	UE3PX	306.36	452.52	264.53	119.75	83.77	1	15.69		l	1	1

month High Capacity Termination p LOOP MAKE-UP Loop Makeup spare facility of Loop Makeup queried (Mann Loop Makeup queried (Mann Loop Makeup spare facility of INTEROFFICE CHAN INTEROFFICE CHAN INTEROFFICE CHAN INTEROFFICE CHAN INTEROFFICE CHAN INTEROFFICE CHAN INTEROFFICE CHAN Interoffice Cha Per Mile per m Interoffice Cha Rev Bat Per Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha - Facility Termin Interoffi	up - Preordering Without Reservation, per working or y queried (Manual). up - Preordering With Reservation, per spare facility anual). upWith or Without Reservation, per working or y queried (Mechanized) TRANSPORT	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually	Charge -	Charge -	Charge -	Incremental Charge -
month High Capacity Termination p Loop Makeup spare facility c Loop Makeup queried (Mann Loop Makeup queried (Mann NoTE: INTEROFFICE CHAN Interoffice Cha Per Mile per m Interoffice Cha Rev Bat Per Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Per Mile per m Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Termination	ity Unbundled Local Loop - STS-1 - Facility I per month III - Preordering Without Reservation, per working or III y queried (Manual). III - Preordering With Reservation, per spare facility III - With or Without Reservation, per working or III y queried (Mechanized) TRANSPORT	r									per LSR	per LSR	Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs. Electronic- Add'l	Manual Svc Order vs. Electronic- Disc 1st	Manual Svo Order vs. Electronic- Disc Add'l
month High Capacity Termination p LOOP MAKE-UP Loop Makeup spare facility of Loop Makeup queried (Mann Loop Makeup queried (Mann Loop Makeup spare facility of INTEROFFICE CHAN INTEROFFICE CHAN INTEROFFICE CHAN Interoffice Cha Per Mile per m Interoffice Cha Rev Bat Per Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Per Mile per m Interoffice Cha Per Mile per m Interoffice Cha Facility Termin Interoffice Cha Per mile per m Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Termination	ity Unbundled Local Loop - STS-1 - Facility I per month III - Preordering Without Reservation, per working or III y queried (Manual). III - Preordering With Reservation, per spare facility III - With or Without Reservation, per working or III y queried (Mechanized) TRANSPORT	1				Rec	Nonrec		Nonrecurring					Rates(\$)		
month High Capacity Termination p LOOP MAKE-UP Loop Makeup spare facility of Loop Makeup queried (Mann Loop Makeup queried (Mann Loop Makeup spare facility of INTEROFFICE CHAN INTEROFFICE CHAN INTEROFFICE CHAN Interoffice Cha Per Mile per m Interoffice Cha Rev Bat Per Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Per Mile per m Interoffice Cha Per Mile per m Interoffice Cha Facility Termin Interoffice Cha Per mile per m Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Termination	ity Unbundled Local Loop - STS-1 - Facility I per month III - Preordering Without Reservation, per working or III y queried (Manual). III - Preordering With Reservation, per spare facility III - With or Without Reservation, per working or III y queried (Mechanized) TRANSPORT	r	-		+		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
High Capacity Termination p LOOP MAKE-UP Loop Makeup spare facility c Loop Makeup queried (Manu Loop Makeup spare facility c spare facility c INTEROFFICE INTEROFFICE INTEROFFICE CHAN Interoffice Che Per Mile per m Interoffice Che Facility Termin Interoffice Che Rev Bat Per Interoffice Che Per Mile per m Interoffice Che Rev Bat Per Interoffice Che Per Mile per m Interoffice Che Per Mile per m Interoffice Che Per Mile per m Interoffice Che Per Mile per m Interoffice Che Per Mile per m Interoffice Che Per Mile per m Interoffice Che Per Mile per m Interoffice Che Per month Interoffice Che Termination Interoffice Che Termination Interoffice Che month	per month up - Preordering Without Reservation, per working or y queried (Manual). up - Preordering With Reservation, per spare facility anual). upWith or Without Reservation, per working or y queried (Mechanized) TRANSPORT	r	1	UDLSX	1L5ND	12.26						15.69		ĺ		İ
Termination p LOOP MAKE-UP Loop Makeup spare facility of Loop Makeup queried (Manu Loop Makeup spare facility of gueried (Manu Loop Makeup spare facility of INOTE: INTEROFFICE INTEROFFICE CHAN Interoffice CHAN Interoffice Cha Per Mile per r Interoffice Cha Rev Bat Per Interoffice Cha Facility Termin Interoffice Cha Per Mile per r Interoffice Cha Per Mile per n Interoffice Cha Per Mile per n Interoffice Cha Per Mile per n Interoffice Cha per month Interoffice Cha per month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination Interoffice Cha month Interoffice Cha	per month up - Preordering Without Reservation, per working or y queried (Manual). up - Preordering With Reservation, per spare facility anual). upWith or Without Reservation, per working or y queried (Mechanized) TRANSPORT	r		ODLOX	ILSIND	12.20					\vdash	13.09		 		
Loop Makeup spare facility C Loop Makeup queried (Manu Loop Makeup spare facility c Makeup queried (Manu Loop Makeup spare facility C Makeup spare facility C INTEROFFICE CHAN INTEROFFICE CHAN Interoffice Cha Per Mile per n Interoffice Cha Facility Termir Interoffice Cha Rev Bat Per Mile per n Interoffice Cha Facility Termir Interoffice Cha Per Mile per n Interoffice Cha Per Mile per n Interoffice Cha Per Mile per n Interoffice Cha Per month Interoffice Cha Per month Interoffice Cha Per month Interoffice Cha Per mination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Month Interoffice Cha Month Interoffice Cha Month Interoffice Charmonth I	y queried (Manual). up - Preordering With Reservation, per spare facility anual). upWith or Without Reservation, per working or y queried (Mechanized) TRANSPORT	r		UDLSX	UDLS1	313.49	452.52	264.53	119.75	83.77		15.69		İ		İ
spare facility of Loop Makeup- queried (Manu. Loop Makeup- spare facility of UNBUNDLED DEDICATED TI NOTE: INTEROFFICE INTEROFFICE CHAN Interoffice Cha Per Mile per n Interoffice Cha Rev Bat Per Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Facility Termin Interoffice Cha Per Mile per n Interoffice Cha Per Mile per n Interoffice Cha per month Interoffice Cha per month Interoffice Cha per month Interoffice Cha per month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination D	y queried (Manual). up - Preordering With Reservation, per spare facility anual). upWith or Without Reservation, per working or y queried (Mechanized) TRANSPORT	r								-						
queried (Manu- Loop Makeup- spare facility company facility company facility company facility company facility company facility f	anual). upWith or Without Reservation, per working or y queried (Mechanized) TRANSPORT			UMK	UMKLW		24.04	24.04								
spare facility C UNBUNDLED DEDICATED TI NOTE: INTEROFFICE INTEROFFICE CHAN Interoffice Cha Per Mile per rr Interoffice Cha Facility Termir Interoffice Cha Rev Bat Per Interoffice Cha Facility Termir Interoffice Cha Facility Termir Interoffice Cha Per Mile per rr Interoffice Cha - Facility Termir Interoffice Cha per month Interoffice Cha per month Interoffice Cha per month Interoffice Cha per month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha	y queried (Mechanized) TRANSPORT			UMK	UMKLP		25.49	25.49								
UNBUNDLED DEDICATED TI NOTE: INTEROFFICE CHAN Interoffice Cha Per Mile per m Interoffice Che Rev Bat Per Interoffice Che Facility Termir Interoffice Che Facility Termir Interoffice Che Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termir Interoffice Che - Facility Termination Interoffice Che - Facility Termination Interoffice Che - Facility Termination Interoffice Che - Facility Termination Interoffice Che - Facility Termination Interoffice Che - Termination Interoffice Che	TRANSPORT															
NOTE: INTEROFFICI INTEROFFICE CHAN Interoffice Cha- Per Mile per re Interoffice Cha- Facility Termir Interoffice Cha- Rev Bat Per Interoffice Cha- Facility Termir Interoffice Cha- Per Mile per re Interoffice Cha- Per Mile per re Interoffice Cha- Per Mile per re Interoffice Cha- Per month Interoffice Cha- Per month Interoffice Cha- Per month Interoffice Cha- Per month Interoffice Cha- Per month Interoffice Cha- Per month Interoffice Cha- Termination Interoffice Cha- Termina				UMK	PSUMK		0.34	0.34								
INTEROFFICE CHAN Interoffice Chan Per Mile per nr Interoffice Che Facility Termin Interoffice Che Facility Termin Interoffice Che Facility Termin Interoffice Che Facility Termin Interoffice Che Per Mile per nr Interoffice Che - Facility Term Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Local Chean Interoffice Che Termination Local Chean Interoffice Che Termination Local Chean Interoffice Che Termination Local Chean Interoffice Che Termination Local Chean Interoffice Che Termination Local Chean Interoffice Che Termination Local Chean Interoffice Che Termination Local Chean Interoffice Chean Interoff	OF OUADINE DEDICATED TRANSPORT				DOO!	070.4 (
Interoffice Cha Per Mile per m Interoffice Cha Facility Termir Interoffice Cha Rev Bat Per Interoffice Cha Facility Termir Interoffice Cha Facility Termir Interoffice Cha - Facility Termir Interoffice Cha - Facility Termir Interoffice Cha - Facility Termir Interoffice Cha per month Interoffice Cha per month Interoffice Cha per month Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha month Interoffice Cha month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha	ICE CHANNEL DEDICATED TRANSPORT - minim	um billir	ng peri	od - below DS3=one	e month, DS3/	STS-1=four mo	nths				 			├		—
Per Mile per n Interoffice Che Facility Termir Interoffice Che Rev Bat Per Interoffice Che Facility Termir Interoffice Che Facility Termir Interoffice Che Per Mile per n Interoffice Che - Facility Term Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che Termination Interoffice Che month	Channel - Dedicated Transport - 2-Wire Voice Grade		+		+											
Facility Termir Interoffice Che Rev Bat Per Interoffice Che Facility Termir Interoffice Che Facility Termir Interoffice Che Per Mile per n Interoffice Che - Facility Termi Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che Termination				U1TVX	1L5XX	0.0167								<u> </u>		ļ
Rev Bat Per Interoffice Cha Facility Termir Interoffice Cha Per Mile per m Interoffice Cha Per Mile per m Interoffice Cha Per Mile per m Interoffice Cha Per month Interoffice Cha Per month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Month Interoffice Cha Month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Month Interoffice Cha Month Interoffice Cha Month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha Month Interoffice Cha Termination Interoffice Ch	mination			U1TVX	U1TV2	24.30	40.63	27.47	16.77	6.91		15.69				
Facility Termin Interoffice Che Per Mile per n Interoffice Che - Facility Term Interoffice Che - Facility Term Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che Termination Interoffice Che month Interoffice Che Termination Interoffice Che Termination Interoffice Che month Interoffice Che month Interoffice Che month Interoffice Che month Interoffice Che Termination Interoffice Che Termination Interoffice Che month Interoffice Che month Interoffice Che month Interoffice Che month Interoffice Che Termination LOCAL CHANNEL - I	Channel - Dedicated Transpor t- 2-Wire Voice Grade Per Mile per month	<u> </u>		U1TVX	1L5XX	0.0167										
Per Mile per n Interoffice Che - Facility Term Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che per month Interoffice Che Termination Interoffice Che month Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che month Interoffice Che month Interoffice Che Termination p Interoffice Che Termination p Interoffice Che Termination LOCAL CHANNEL - I				U1TVX	U1TR2	24.30	40.63	27.47	16.77	6.91		15.69				
- Facility Term Interoffice Che per month Interoffice Che Termination Interoffice Che per month Interoffice Che per month Interoffice Che Termination Interoffice Che month Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che month Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination				U1TVX	1L5XX	0.0167										
per month Interoffice Che Termination Interoffice Che per month Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che month Interoffice Che Termination p Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination		Э		U1TVX	U1TV4	21.29	40.63	27.47	16.77	6.91		15.69				
Termination Interoffice Che per month Interoffice Che Termination Interoffice Che month Interoffice Che Termination Interoffice Che Termination Interoffice Che month Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination Interoffice Che Termination	Channel - Dedicated Transport - 56 kbps - per mile			U1TDX	1L5XX	0.0167										
per month Interoffice Che Termination Interoffice Cha month Interoffice Cha Termination Interoffice Cha Termination Interoffice Cha month Interoffice Cha Termination p Interoffice Cha Termination p Interoffice Cha month Interoffice Cha month Interoffice Cha Termination p Interoffice Cha Termination cha LOCAL CHANNEL - I				U1TDX	U1TD5	16.76	40.63	27.47	16.77	6.91		15.69				
Termination Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth Interoffice Chemonth	Channel - Dedicated Transport - 64 kbps - per mile			U1TDX	1L5XX	0.0167										
month Interoffice Che Termination Interoffice Che month Interoffice Che Termination p Interoffice Che month Interoffice Che month Interoffice Che Termination LOCAL CHANNEL - I				U1TDX	U1TD6	16.76	40.63	27.47	16.77	6.91		15.69				
Termination Interoffice Che month Interoffice Cha Termination p Interoffice Cha month Interoffice Cha month Interoffice Cha Termination cha LOCAL CHANNEL - I	Channel - Dedicated Channel - DS1 - Per Mile per			U1TD1	1L5XX	0.3415										
month Interoffice Che Termination p Interoffice Che month Interoffice Che Termination LOCAL CHANNEL - I				U1TD1	U1TF1	77.14	89.47	81.99	16.39	14.48		15.69				
Termination p Interoffice Cha month Interoffice Cha Termination LOCAL CHANNEL - I	Channel - Dedicated Transport - DS3 - Per Mile per			U1TD3	1L5XX	8.02										
month Interoffice Cha Termination LOCAL CHANNEL - I				U1TD3	U1TF3	880.65	279.37	163.12	60.33	58.59		15.69				
Termination LOCAL CHANNEL - D	Channel - Dedicated Transport - STS-1 - Per Mile per			U1TS1	1L5XX	8.02										
				U1TS1	U1TFS	880.55	279.37	163.12	60.33	58.59		15.69				
NOTE: LOCAL CHAN	- DEDICATED TRANSPORT		L													
	ANNEL DEDICATED TRANSPORT - minimum billi	ng perio	d - bel				_	-					·			
	nel - Dedicated - 2-Wire Voice Grade			ULDVX	ULDV2	15.33	193.53	33.24	36.72	3.21	ldot	15.69		<u> </u>		
	nel - Dedicated - 2-Wire Voice Grade Rev Bat	-	1	ULDVX	ULDR2	15.33	193.53	33.24	36.72	3.21		15.69		├		
	nel - Dedicated - 4-Wire Voice Grade		1	UNDVX ULDD1	ULDV4 ULDF1	16.54 42.62	193.97 177.87	33.68 154.06	37.19 22.24	3.68 15.30	\vdash	15.69 15.69		├ ──		
	not Dodicated DS1 Zone 1	-	2	ULDD1	ULDF1 ULDF1	70.32	177.87	154.06	22.24	15.30	\vdash	15.69				
	nel - Dedicated - DS1 - Zone 1 nel - Dedicated - DS1 - Zone 2	+	3	ULDD1	ULDF1	190.68	177.87	154.06	22.24	15.30		15.69		 		-
	nel - Dedicated - DS1 - Zone 2	1		ULDD3	1L5NC	11.93	111.01	104.00	22.24	10.00		10.00		 	1	
	nel - Dedicated - DS1 - Zone 2 nel - Dedicated - DS1 - Zone 3	1	1	ULDD3	ULDF3	446.00	452.52	264.53	119.75	83.77		15.69			1	
Local Channe	nel - Dedicated - DS1 - Zone 2 nel - Dedicated - DS1 - Zone 3 nel - Dedicated - DS3 - Per Mile per month			ULDS1	1L5NC	11.93										
Local Channe DARK FIBER	nel - Dedicated - DS1 - Zone 2 nel - Dedicated - DS1 - Zone 3 nel - Dedicated - DS3 - Per Mile per month nel - Dedicated - DS3 - Facility Termination nel - Dedicated - STS-1- Per Mile per month			ULDS1	ULDFS	435.10	452.52	264.53	119.75	83.77		15.69				

UNBUNDL	ED NETWORK ELEMENTS - South Carolina			T							1			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svo Order vs. Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						B	Nonrec	urring	Nonrecurring	Disconnect		1	oss	Rates(\$)		<u></u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Channel			UDF	1L5DC	97.65	0.10.51	100.1=	0.17.70	100.11		45.00				
	NRC Dark Fiber - Local Channel			UDF	UDFC4		640.51	138.17	317.76	198.11		15.69				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Interoffice Channel			UDF	1L5DF	36.41										
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14	30.41	640.51	138.17	317.76	198.11		15.69				-
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction			ODI	0D1 14		040.01	100.17	017.70	100.11		10.00				
	Thereof per month - Local Loop			UDF	1L5DL	97.65										
	NRC Dark Fiber - Local Loop			UDF	UDFL4		640.51	138.17	317.76	198.11		15.69				
8XX ACCESS	TEN DIGIT SCREENING															
	8XX Access Ten Digit Screening, Per Call			OHD		0.0006673										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserved			OHD	N8R1X		2.59	0.44				15.69				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translations			OHD			5.95	0.81	4.58	0.54		15.69				
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translations			OHD	N8FTX		5.95	0.81	4.58	0.54		15.69				
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number			OHD	N8FCX		2.59	1.30				15.69				
	8XX Access Ten Digit Screening, Multiple InterLATA CXR															
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		3.03	1.74				15.69				
-	8XX Access Ten Digit Screening, Change Charge Per Request 8XX Access Ten Digit Screening, Call Handling and Destination			OHD	N8FAX		3.03	0.44				15.69				
	Features			OHD	N8FDX		2.59	2.59				15.69				
	8XX Access Ten Digit Screening, w/ 8XX No. Delivery			OHD		0.0006673										
	8XX Access Ten Digit Screening, w/ POTS No. Delivery			OHD		0.0006673										
LINE INFORM	IATION DATA BASE ACCESS (LIDB)			007		0.0000040										
	LIDB Common Transport Per Query LIDB Validation Per Query			OQT OQU		0.0000246 0.0138158									-	<u> </u>
	LIDB Originating Point Code Establishment or Change			OQU OQT, OQU	NRPBX	0.0136136	34.40		42.18			15.69				1
SIGNALING (OQ1, OQ0	INICI DX		34.40		42.10			10.00				
Ι	CCS7 Signaling Connection, Per 56 Kbps Facility			UDB	TPP++	16.93	35.61	35.61	16.48	16.48						
	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	163.49										
	CCS7 Signaling Usage, Per TCAP Message			UDB		0.0000692										
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	16.93	35.61	35.61	16.48	16.48		15.69				
	CCS7 Signaling Connection, Per link (B link) (also known as D															
	link) CCS7 Signaling Usage, Per ISUP Message			UDB	TPP++	16.93	35.61	35.61	16.48	16.48		15.69				1
	CCS7 Signaling Usage, Per ISUP Message CCS7 Signaling Usage Surrogate, per link per LATA	 	-	UDB UDB	STU56	0.0000173 791.37			+		-			-		+
 	CCS7 Signaling Osage Surrogate, per link per LATA CCS7 Signaling Point Code, per Originating Point Code	-		555	0.000	131.31			 		1				†	
	Establishment or Change, per STP affected			UDB	CCAPO		29.08	29.08	35.65	35.65		15.69				
	CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected			UDB	CCAPD		29.08	29.08	35.65	35.65		15.69				
E911 SERVIC	E													<u> </u>		<u> </u>
	Local Channel - Dedicated - 2-wr Voice Grade					15.33	193.53	33.24	36.72	3.21		15.69				
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile	1				0.0167			ļl							
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Termination					24.30	40.63	27.47	16.77	6.91		15.69				
	Local Channel - Dedicated - DS1 - Zone 1					42.62	177.87	154.06	22.24	15.30		15.69				
	Local Channel - Dedicated - DS1 - Zone 2					70.32	177.87	154.06	22.24	15.30		15.69			1	ļ
	Local Channel - Dedicated - DS1 - Zone 3	<u> </u>	ļ			190.68	177.87	154.06	22.24	15.30	<u> </u>	15.69				4
	Interoffice Transport - Dedicated - DS1 Per Mile					0.3415									 	
CALLING	Interoffice Transport - Dedicated - DS1 Per Facility Termination		<u> </u>			77.14	89.47	81.99	16.39	14.48		15.69			ļ	
CALLING NA	ME (CNAM) SERVICE CNAM For DB Owners - Service Establishment	<u> </u>		OQV	+		23.00	23.00	21.15	21.15	1	15.69			 	
	CNAM For Non DB Owners - Service Establishment CNAM For Non DB Owners - Service Establishment			OQV	+		23.00	23.00	21.15	21.15	1	15.69				
	CNAM For DB Owners - Service Provisioning With Point Code	 			+		20.00	20.00	21.10	21.10		10.00			-	†
	Establishment	<u></u>	<u>L</u>	OQV			993.09	734.47	269.53	198.18		15.69		<u> </u>	<u> </u>	

ONRONDE	ED NETWORK ELEMENTS - South Carolina			1		1					T -			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Name		l Names accoming	Dianamant						
						Rec	Nonrec		Nonrecurring		001150	001111		Rates(\$)	001111	001111
	CNAM For Non DB Owners - Service Provisioning With Point		<u> </u>		_		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Code Establishment			ogv			343.09	245.69	275.87	198.18		15.69				
	CNAM for DB Owners, Per Query			OQV		0.0010433	343.09	245.09	2/3.0/	190.10		15.69				+
-	CNAM for Non DB Owners, Per Query			OQV	-	0.0010433						-			-	+
LNP Query S				OQV	-	0.0010433						-			-	+
LINE Query 3	LNP Charge Per query				-	0.0008837						-			-	+
-	LNP Service Establishment Manual				-	0.0000037	25.09	25.09	23.07	23.07		15.69			-	+
	LNP Service Establishment wantual LNP Service Provisioning with Point Code Establishment	1	1				594.82	303.88	269.53	198.18		15.69				
ODEDATOR	CALL PROCESSING	1	1				394.62	303.00	209.55	190.10		15.69				
OPERATOR		-	-													
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.20										
	Oper. Call Processing - Oper. Provided, Per Min Using					1.20										+
]	Foreign LIDB	1	1		I	1.24								1	I	1
	Oper. Call Processing - Fully Automated, per Call - Using BST	†							1					1	1	<u> </u>
	LIDB					0.20									1	
	Oper. Call Processing - Fully Automated, per Call - Using					0.20										
	Foreign LIDB					0.20										
INWARD OP	ERATOR SERVICES				-	0.20						1				+
1	Inward Operator Services - Verification, Per Minute				-	1.15						1				+
	Inward Operator Services - Verification and Emergency Interrupt					1.10										+
	- Per Minute					1.15										
BRANDING -	OPERATOR CALL PROCESSING					1.10										+
	ity based CLEC															
i acii	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				15.69				+
	Loading of Custom Branded OA Announcement per shelf/NAV				CDACC		7,000.00	7,000.00				13.03				+
	per OCN				CBAOL		500.00	500.00				15.69				
LINE	PCLEC				CBAOL		300.00	300.00				13.08				+
ONLI	Recording of Custom Branded OA Announcement				-		7,000.00	7,000.00	-			15.69			-	+
-	Loading of Custom Branded OA Announcement per shelf/NAV				-		7,000.00	7,000.00	-			13.08			-	+
	per OCN						500.00	500.00				15.69				
DIRECTORY	ASSISTANCE SERVICES				-		500.00	300.00	-			15.69			-	+
	CTORY ASSISTANCE ACCESS SERVICE				-				-			-			-	+
DIKE	Directory Assistance Access Service Calls, Charge Per Call	<u> </u>	<u> </u>			0.275										
DIDE	CTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (I) A C C \	<u> </u>			0.275										
DIKE		JACC)							-							+
	Directory Assistance Call Completion Access Service (DACC),					0.10										
DIDECTORY	Per Call Attempt ASSISTANCE SERVICES	<u> </u>	<u> </u>			0.10										
	CTORY ASSISTANCE DATA BASE SERVICE (DADS)	-	-													
DIKE						0.04										+
\vdash	Directory Assistance Data Base Service Charge Per Listing		1		DROOF	0.04									 	
DDANDING	Directory Assistance Data Base Service, per month DIRECTORY ASSISTANCE	 	1		DBSOF	150.00			1		1			-	 	+
			1												 	
Facil	ity Based CLEC															+
	Recording and Provisioning of DA Custom Branded			ANAT	CBADA		6 000 00	6 000 00				45.00			1	
 	Announcement	<u> </u>	ļ	AMT	CBADA		6,000.00	6,000.00	1			15.69			-	+
	Loading of Custom Branded Announcement per Switch	 	<u> </u>	AMT	CBADC		1,170.00	1,170.00			1	15.69		-	1	+
UNE	CLEC	 	<u> </u>		-		2.000.00	2 222 22			1	45.00		1	 	
\vdash	Recording of DA Custom Branded Announcement	<u> </u>	ļ		-		3,000.00	3,000.00	1			15.69			-	├
]	Loading of DA Custom Branded Announcement per Switch per	1	1				4 470 00	4 470 00				45.00		l	I	1
OF LEGIL :	OCN		!				1,170.00	1,170.00			1	15.69			1	+
SELECTIVE		 	<u> </u>		-						1	-		1	 	+
]	Selective Routing Per Unique Line Class Code Per Request Per	1	1		LICECE		04.00	04.00				45.00		l	I	
WIDTHALCO	Switch	<u> </u>	ļ		USRCR		84.89	84.89	14.14	14.14		15.69			-	├
VIRTUAL CO			<u> </u>	AMTEO	EAE		1.007.05	4 007 05	0 = 1		ļ	45.00				
\vdash	Virtual Collocation - Application Cost	ļ	<u> </u>	AMTFS	EAF		1,207.95	1,207.95	0.51	0.51	1	15.69				_
\vdash	Virtual Collocation - Cable Installation Cost, per cable	ļ	<u> </u>	AMTFS	ESPCX		794.22	794.22	22.54	22.54	1	15.69				_
$oxed{oxed}$	Virtual Collocation - Floor Space, per sq. ft.	ļ	ļ	AMTFS	ESPVX	3.95										
ļļ	Virtual Collocation - Power, per fused amp	<u> </u>		AMTFS	ESPAX	9.19					<u> </u>				ļ	
]	Virtual Collocation - Cable Support Structure, per entrance	1	1								I			Ì	I	1
	cable	<u>L</u>	<u>L_</u>	AMTFS	ESPSX	18.66			<u> </u>		<u> </u>	<u> </u>		<u> </u>	<u> </u>	1

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incrementa Charge -
							Nonrec	curring	Nonrecurring	Disconnect		l.	oss	Rates(\$)	<u> </u>	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation - 2-wire Cross Connects (Ioop)			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, AMTFS, UDL, UNCVX, UNCDX, UNCNX	UEAC2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation - 4-wire Cross Connects (loop)			UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX	UEAC4	0.0634	12.42	11.90	6.40	5.74		15.69				
	Virtual Collocation - 2-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC2F	2.86	20.94	15.23	7.40	5.93		15.69				
				AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12,	01045	5.74	25.24	40.00	0.70	0.00		45.00				
	Virtual Collocation - 4-Fiber Cross Connects Virtual collocation - Special Access & UNE, cross-connect per			ULD48, UDF USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL,	CNC4F	5.71	25.61	19.90	9.73	8.26		15.69				
	Virtual collocation - Special Access & UNE, cross-connect per DS3			UNLD1 USL,ULC,AMTFS,U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1,	CNC1X CND3X	1.12	22.08	15.96	7.39	5.80		15.69				
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			AMTFS	VE1CB	0.0022	20.94	13.23	7.39	3.93		13.09				
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0033										
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AMTFS	VE1CC		536.56									
	Cable Support Structure, per cable Virtual Collocation Cable Records - per request			AMTFS AMTFS	VE1CE VE1BA		536.56 760.98	489.20	133.29	133.29						
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable record			AMTFS	VE1BB		327.65	327.65	189.54	189.54						
	Virtual Collocation Cable Records - VG/DS0 Cable, per each 100 pair Virtual Collocation Cable Records - DS1, per T1TIE			AMTFS AMTFS	VE1BC VE1BD		4.82 2.26	4.82 2.26	5.91 2.77	5.91 2.77						
	Virtual Collocation Cable Records - DS1, per TTTE Virtual Collocation Cable Records - DS3, per T3TIE			AMTFS	VE1BE		7.90	7.90	9.68	9.68	1	1	 	 		
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber records			AMTFS	VE1BF		84.68	84.68	77.30	77.30						
 	Virtual collocation - Security Escort - Basic, per half hour			AMTES	SPTBX		16.96	10.75			ļ	15.69				
	Virtual collocation - Security Escort - Overtime, per half hour Virtual collocation - Security Escort - Premium, per half hour			AMTFS AMTFS	SPTOX SPTPX		22.10 27.23	13.89 17.02				15.69 15.69				-
	Virtual collocation - Maintenance in CO - Basic, per half hour			AMTFS	CTRLX		27.99	10.75				15.69				
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		36.56	13.89				15.69				
VIRTUAL COL	Virtual collocation - Maintenance in CO - Premium per half hour LOCATION			AMTFS	SPTPM		45.12	17.02				15.69				
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2- Wire Analog - Res			UEPSR	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				

UNBUN	DLE	NETWORK ELEMENTS - South Carolina					•					T -	1 -		ment: 2		bit: C
CATEGOR	RY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
							Rec	Nonred		Nonrecurring					Rates(\$)		
							1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.0317	40.00	44.00	0.04	5.45		45.00				İ
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSP	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
		Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				ĺ
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire					0.00										
		Analog Bus			UEPSB	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				<u> </u>
		Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire															
		ISDN			UEPSX	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			UEPTX	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				ĺ
		Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire			UEPIX	VE IRZ	0.0317	12.32	11.83	6.04	5.45		15.69				
		ISDN DS1			UEPEX	VE1R4	1.12	22.08	15.96	6.42	5.80		15.69				ĺ
VIRTUAL	COLL	OCATION			İ						2.30						
		Virtual Collocation-2 Wire Cross Connects (Loop) for Line															
		Splitting			UEPSR, UEPSB	VE1LS	0.0317	12.32	11.83	6.04	5.45		15.69				
PHYSICAL		LOCATION		<u> </u>	1	1											
		Physical Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	PE1LS	0.0341	12.32	11.83	6.04	5.45		15.69				İ
AIN SELE		E CARRIER ROUTING			OLFSK, OLFSB	FLILS	0.0341	12.32	11.03	0.04	3.43		13.03				
AIN OLLL		Regional Service Establishment			SRC	SRCEC		101,324.34	101,324.34	8.609.85	8.609.85		15.69				
		End Office Establishment			SRC	SRCEO		175.66	175.66	1.70	1.70		15.69				
		Query NRC, per query			SRC		0.0035036										
AIN - BEL		TH AIN SMS ACCESS SERVICE															
		AIN SMS Access Service - Service Establishment, Per State,				044405		00.50	00.50	40.70	40.70		45.00				ĺ
-		Initial Setup			A1N	CAMSE		39.53	39.53	40.78	40.78		15.69				
		AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		7.85	7.85	9.11	9.11		15.69				ĺ
		AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		7.85	7.85	9.11	9.11		15.69				
		AIN SMS Access Service - User Identification Codes - Per User															
		ID Code			A1N	CAMAU		35.08	35.08	27.12	27.12		15.69				
		AIN SMS Access Service - Security Card, Per User ID Code,											4= 00				ĺ
		Initial or Replacement AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)			A1N	CAMRC	0.0027	41.98	41.98	11.74	11.74		15.69				
		AIN SMS Access Service - Storage, Fer Onit (100 Kilobytes) AIN SMS Access Service - Session, Per Minute					0.7121										
		AIN SMS Access Service - Company Performed Session, Per					0.7 121										
		Minute					0.8364										İ
AIN - BEL		ITH AIN TOOLKIT SERVICE															
		AIN Toolkit Service - Service Establishment Charge, Per State,															İ
		Initial Setup			CAM	BAPSC BAPVX		39.53 4,211.54	39.53 4,211.54	40.78 0.00	40.78 0.00		15.69				
 		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per			 	DAFVA		4,211.54	4,∠11.54	0.00	0.00		15.69				
		DN, Term. Attempt				BAPTT		7.85	7.85	9.11	9.11		15.69				1
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
		DN, Off-Hook Delay				BAPTD		7.85	7.85	9.11	9.11		15.69				
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															ĺ
		DN, Off-Hook Immediate				BAPTM		7.85	7.85	9.11	9.11		15.69				
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP		1	1	ВАРТО		34.54	34.54	14.39	14.39		15.69				
 		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per			 	2,1110		34.34	54.54	14.55	14.55		10.08				
L l		DN, CDP	L	L	<u> </u>	BAPTC		34.54	34.54	14.39	14.39	<u> </u>	15.69		<u> </u>	<u> </u>	<u> </u>
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per									-						
		DN, Feature Code			ļ	BAPTF		34.54	34.54	14.39	14.39		15.69				
\vdash		AIN Toolkit Service - Query Charge, Per Query		<u> </u>	1	1	0.0558238										
		AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.0069214										1
\vdash		AIN Toolkit Service - SCP Storage Charge, Per SMS Access			 	+	0.0009214			+ -							
		Account, Per 100 Kilobytes		1	1		0.07										
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
1 1		Subscription	<u></u>	L	CAM	BAPMS	11.87	7.85	7.85	5.52	5.52	<u> </u>	15.69		<u> </u>	<u> </u>	1

UNBL	NDLE	D NETWORK ELEMENTS - South Carolina												Attachi	ment: 2	Exhil	oit: C
												Svc Order	Svc Order	Incremental		Incremental	Incrementa
												Submitted			Charge -	Charge -	Charge -
			Intori									Elec		Manual Svc	Manual Svc		Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																D130 13t	DISC Add I
							Rec	Nonre	curring	Nonrecurring	g Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
		Subscription			CAM	BAPLS	3.51	8.68	8.68				15.69				
		AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service															
		Subscription			CAM	BAPDS	8.48	7.85	7.85	5.52	5.52		15.69				
		AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit															
		Service Subscription			CAM	BAPES	0.12	8.68	8.68				15.69				
ENHAN		(TENDED LINK (EELs)															
		New Density Zone 1 EELs are available in the following MSA:					Atlanta, Ga; Ne	w Orleans, LA,									
		Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-															
		In all states, EEL network elements shown below also apply t												UNEs.(Non-re	ecurring rates	do not apply	.)
		In All States the EEL network elements apply to ordinarily con				tch As Is Ch	arge.) When o	dering ordinar	ily combined	network eleme	nts, Non-recur	ring rates d	o apply.				
	2-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)	1										ļ	
	1	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport	1			L							1		_]	
	ļ	Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69		1		
	1	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed	l	1		1							1		I	Ì	
	ļ	Transport Combination - Zone 2	ļ	2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69		1	ļ	
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
		Transport Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
		Interoffice Transport - Dedicated - DS1 combination - Per Mile															
		per month			UNC1X	1L5XX	0.27										
		Interoffice Transport - Dedicated - DS1 combination - Facility															
		Termination per month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
		DS1 Channelization System Per Month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
		Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.56	6.59	4.73				15.69				
		Each Additional 2-Wire VG Loop(SL 2) in the same DS1															
		Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
		Each Additional 2-Wire VG Loop(SL2) in the same DS1								=			4= 00				
		Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
		Each Additional 2-Wire VG Loop(SL2) in the same DS1		_													
		Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
		Voice Grade COCI - DS1 to DS0 Channel System combination -			1110101	454)(0	0.50	0.50	4.70				45.00				
		per month			UNCVX	1D1VG	0.56	6.59	4.73				15.69				
		Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		5.04	5.04	7.00	7.00		45.00				
	4 14/105	IS CHARGE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT		IOF TO		UNCCC		5.61	5.61	7.00	7.00		15.69				
	4-WIRE	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	ERUFF	ICE IN	ANSPORT (EEL)												
		Transport Combination - Zone 1		4	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
	1	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	1		OIVOVA	ULAL4	32.39	132.38	94.63	39.35	14.01		15.09		 	1	1
	l	Transport Combination - Zone 2	l	2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69		1		
	1	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	1	 -	0110 VX	JEALT	45.09	102.00	37.03	55.55	14.01		10.09		 	 	
	1	Transport Combination - Zone 3	l	3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69		I	1	
		Interoffice Transport - Dedicated - DS1 combination - Per Mile		Ŭ	ONOVA	OL/1L-I	40.00	102.00	04.00	00.00	14.01	1	10.00				
	l	Per Month	l		UNC1X	1L5XX	0.27								1		
-	1	Interoffice Transport - Dedicated - DS1 - Facility Termination Per	1	1			Ų. <u>2</u> 1						l		 		
1	1	Month	l	1	UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69		I	Ì	
	1	Channelization - Channel System DS1 to DS0 combination Per	1			1	J 1	55.77	050	.5.55	10		.0.00		t	1	
		Month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	1	Voice Grade COCI - DS1 to DS0 Channel System combination -	1			1					2.31				t	1	
	1	per month	l	1	UNCVX	1D1VG	0.56	6.59	4.73				15.69		I	Ì	
		Additional 4-Wire Analog Voice Grade Loop in same DS1				1			_		İ				İ	İ	İ
	1	Interoffice Transport Combination - Zone 1	l	1	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69		I	Ì	
		Additional 4-Wire Analog Voice Grade Loop in same DS1															
	1	Interoffice Transport Combination - Zone 2	l	2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69		I	1	
		Additional 4-Wire Analog Voice Grade Loop in same DS1				1		, ,,,,,			1,1,1				İ	İ	
	1	Interoffice Transport Combination - Zone 3	l	3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69		I	Ì	
		Voice Grade COCI - DS1 to DS0 Channel System combination -		Ť	-	1		.52.50	230	33.50					1		İ
	l	per month	l		UNCVX	1D1VG	0.56	6.59	4.73				15.69		1		
		Nonrecurring Currently Combined Network Elements Switch -As-				1			_		İ				İ	İ	İ
1	1	Is Charge	l	1	UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69		I	Ì	
				•								•			•		

UNBUNDI	LED	NETWORK ELEMENTS - South Carolina												Attachi	ment: 2	Exhib	oit: C
CITECITE	1	NETWORK ELEMENTO COULT CATOMIC										Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Interi									Elec			Manual Svc	Manual Svc	Manual Svc
CATEGORY	1	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			""											Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
	_									T 51	. B'				D-1(A)		
						+	Rec	Nonrec	urring Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
4 10/	IDE I	56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTER	EEICE	TDANSDORT (EEL)			First	Add I	FIRST	Addi	SOMEC	SUMAN	SUMAN	SOWAN	SUMAN	SUMAN
4-44		First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	FFICE	TRANSFORT (ELL)												
		Fransport Combination - Zone 1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
		First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice															
		Fransport Combination - Zone 2		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
		First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice															
		Fransport Combination - Zone 3		3	UNCDX	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				
	li li	nteroffice Transport - Dedicated - DS1 combination - Per Mile				1											
		Per Month			UNC1X	1L5XX	0.27										
		nteroffice Transport - Dedicated - DS1 - combination Facility [ermination Per Month]			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
		Channelization - Channel System DS1 to DS0 combination Per	1		OINC IA	UIIFI	01./1	89.47	81.99	10.39	14.48	1	15.09	1	1		
		Month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per	<u> </u>					024	021		0.01		.0.00		1		
		nonth (2.4-64kbs)	1		UNCDX	1D1DD	1.19	6.59	4.73	1			15.69		1		
		Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															
		nteroffice Transport Combination - Zone 1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
		Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		_													
		nteroffice Transport Combination - Zone 2		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
		Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		3	LINCDY	UDL56	34.74	400.00	00.40	50.05	44.64		45.00				
		nteroffice Transport Combination - Zone 3 DCU-DP COCI (data) - DS1 to DS0 Channel System -		3	UNCDX	UDLOO	34.74	126.66	89.12	59.35	14.61	-	15.69				
		combination per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.59	4.73				15.69				
		Nonrecurring Currently Combined Network Elements Switch -As-			CNODA	10100	1.10	0.00	4.70				10.00				
		s Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-W	IRE (64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL)												
		First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
		Fransport Combination - Zone 1		1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
		First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		_													
		Fransport Combination - Zone 2		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
		First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		3	UNCDX	UDL64	34.74	126.66	89.12	59.35	14.61		15.69				
		nteroffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCDX	ODL04	34.74	120.00	09.12	39.33	14.01		13.09				
		Per Month			UNC1X	1L5XX	0.27										
		nteroffice Transport - Dedicated - DS1 combination - Facility					_										
	1	Termination Per Month	<u> </u>		UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69	<u> </u>	<u> </u>		
		Channelization - Channel System DS1 to DS0 combination Per													1		
		Month	ļ		UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
		DCU-DP COCI (data) - DS1 to DS0 Channel System			LINCDY	40400		0.50	4 =				45.00				
		combination - per month (2.4-64kbs) Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	<u> </u>		UNCDX	1D1DD	1.19	6.59	4.73	_		-	15.69		-		
		nteroffice Transport Combination - Zone 1	1	1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69		1		
		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	 	_	5.10DA	JULUT	23.33	120.00	03.12	33.33	17.01		10.09				
		nteroffice Transport Combination - Zone 2		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
	P	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	l												1		
	l	nteroffice Transport Combination - Zone 3		3	UNCDX	UDL64	34.74	126.66	89.12	59.35	14.61		15.69				
		DCU-DP COCI (data) - DS1 to DS0 Channel System	1														
		combination - per month (2.4-64kbs)	ļ		UNCDX	1D1DD	1.19	6.59	4.73				15.69				
		Nonrecurring Currently Combined Network Elements Switch -As-	1		LINICAY	LINICOO		5.01	.	7.00	7.00		45.00				
A_1A1	IDE I	s Charge DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTI	EROFFI	CE TD	UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69	-	-		
4-77		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	LAUFFI	UL IKA	THOFUNI (EEL)					 		-			 		
		Fransport - Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	<u> </u>	Ė			55.57		.000	50	0		.0.00		1		
	7	Fransport - Zone 2	<u>L</u>	2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73	<u> </u>	15.69	<u> </u>	<u> </u>		
		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice															
		Fransport - Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
		nteroffice Transport - Dedicated - DS1 combination - Per Mile	1		LINGAY	41.5307				1			1		1		
	JF	Per Month	<u> </u>	l	UNC1X	1L5XX	0.27			1		1	l	l	l		L

														ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonred		Nonrecurring					Rates(\$)		_
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - DS1 combination - Facility								40.00							
	Termination Per Month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIDE	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE	DOEEL	CE TD		UNCCC		3.01	3.01	7.00	7.00		15.09				
4-4411	First DS1Loop in DS3 Interoffice Transport Combination - Zone	KOFFI	L IK	I	+											
	1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
_	First DS1Loop in DS3 Interoffice Transport Combination - Zone			0.10.17	00201	00.07	200.00	101.00	11.00			10.00				t
	2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone															
	3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
	Interoffice Transport - Dedicated - DS3 combination - Per Mile															
	Per Month			UNC3X	1L5XX	6.42										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per															Ī
	month			UNC3X	U1TF3	704.52	279.37	163.12	60.33	58.59		15.69				
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	8.64	6.59	4.73				15.69				
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
	Additional DS1Loop in DS3 Interoffice Transport Combination -		_													
	Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	Additional DS1Loop in DS3 Interoffice Transport Combination -		_					.==								
	Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	8.64	6.59	4.73				15.69				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				
2-WIDE	IN CHARGE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	EDOEE	ICE TE		UNCCC		5.01	3.01	7.00	7.00		15.69				
Z-VVIKE	2-WireVG Loop used with 2-wire VG Interoffice Transport	LKOFF	ICE II	I	+											
	Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	2-WireVG Loop used with 2-wire VG Interoffice Transport			0.1017	027122	10.00	.00.00	00.10	00.00			10.00				
	Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
	2-WireVG Loop used with 2-wire VG Interoffice Transport											.0.00				
	Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per															1
	Mile Per Month			UNCVX	1L5XX	0.0134										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade															Ī
	combination - Facility Termination per month			UNCVX	U1TV2	19.44	40.63	27.47	16.77	6.91		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNCVX	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIRE	VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EROFF	ICE TF	RANSPORT (EEL)												
	4-WireVG Loop used with 4-wire VG Interoffice Transport		١.									4= 00				
	Combination - Zone 1		1	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
	4-WireVG Loop used with 4-wire VG Interoffice Transport		_	UNCVX	UEAL4	43.89	132.38	94.83	59.35	44.04		45.00				
-+-	Combination - Zone 2 4-WireVG Loop used with 4-wire VG Interoffice Transport		2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	Combination - Zone 3		3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
+	Interoffice Transport - Dedicated - 4-wire VG combination - Per		3	OINOVA	ULAL4	45.50	132.30	34.03	39.33	14.01		13.09		 	 	
	Mile Per Month		1	UNCVX	1L5XX	0.0134								1	I	
	Interoffice Transport - Dedicated - 4- Wire Voice Grade				.20,01	0.0.04									<u> </u>	<u> </u>
	combination - Facility Termination per month		1	UNCVX	U1TV4	17.03	40.63	27.47	16.77	6.91		15.69		1	I	
	Nonrecurring Currently Combined Network Elements Switch -As-					55	0							İ	1	1
	Is Charge		l	UNCVX	UNCCC		5.61	5.61	7.00	7.00		15.69			1	
	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRAI	SPOR	T (EEL)												
	High Capacity Unbundled Local Loop - DS3 combination - Per															
	Mile per month			UNC3X	1L5ND	12.26										
	High Capacity Unbundled Local Loop - DS3 combination -				1			-								
	Facility Termination per month		ı	UNC3X	UE3PX	306.36	452.52	264.53	119.75	83.77	1	15.69		1	1	1

UNBUNDI I	D NETWORK ELEMENTS - South Carolina												Attach	ment: 2	Fyhil	bit: C
											Svc Order	Svc Order	Incremental		Incremental	Incremental
1											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
ł		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
ł		"											Electronic-	Electronic-	Electronic-	Electronic-
1													1st	Add'l	Disc 1st	Disc Add'l
		1					Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
		1				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
- 	Interoffice Transport - Dedicated - DS3 combination - Facility							7.44		7.44	0020					
1	Termination per per month			UNC3X	U1TF3	704.52	279.37	163.12	60.33	58.59		15.69				1
	Nonrecurring Currently Combined Network Elements Switch -As-	•														
	Is Charge			UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				<u></u>
STS1	DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROF	FICE TF	ANSP	ORT (EEL)												
	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month			UNCSX	1L5ND	12.26										1
	High Capacity Unbundled Local Loop - STS1 combination -			UNCOX	ILSIND	12.20								-		
1	Facility Termination per month			UNCSX	UDLS1	313.49	452.52	264.53	119.75	83.77		15.69				1
	Interoffice Transport - Dedicated - STS1 combination - Per Mile															
	per month			UNCSX	1L5XX	6.42										
T T	Interoffice Transport - Dedicated - STS1 combination - Facility	1														1
	Termination per month	ļ		UNCSX	U1TFS	704.44	279.37	163.12	60.33	58.59		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As-	1		LINICEY	LINICCO		F 04	F 04	7.00	7.00		45.00				1
2-WIE	Is Charge E ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (FFI	\ \	UNCSX	UNCCC		5.61	5.61	7.00	7.00	1	15.69		+		
Z-VVIR	First 2-Wire ISDN Loop in a DS1 Interoffice Combination	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			1				+					†		<u> </u>
	Transport - Zone 1		1	UNCNX	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				ĺ
í İ	First 2-Wire ISDN Loop in a DS1 Interoffice Combination															
	Transport - Zone 2		2	UNCNX	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination															ĺ
\vdash	Transport - Zone 3		3	UNCNX	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				
\vdash	Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility	1		UNC1X	1L5XX	0.27								-		
1	Termination per month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				ĺ
	Channelization - Channel System DS1 to DS0 combination -			014017	01111	01.71	00.41	01.00	10.00	14.40		10.00				
	per month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				ĺ
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System															
	combination - per month			UNCNX	UC1CA	2.56	6.59	4.73				15.69				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport								== ==							ĺ
	Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport	1	1	UNCNX	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				
	Combination - Zone 2		2	UNCNX	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				ĺ
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport		_	ONOTAL	OTLEX	02.70	117.00	00.00	00.00	10.01		10.00				—
	Combination - Zone 3		3	UNCNX	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				ĺ
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System							-								
	combintaion- per month	ļ		UNCNX	UC1CA	2.56	6.59	4.73				15.69				
1	Nonrecurring Currently Combined Network Elements Switch -As-	1		LINCAV	LINICCO		F 04	F 04	7.00	7.00		45.00				1
4-18/16	IS Charge E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	ITEPOE	FICE 7	UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69		 		
4-7/18	First DS1 Loop in STS1 Interoffice Transport Combination -	LEKUF	ICE II	MINOPURI (EEL)	+											
1	Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				i
	First DS1 Loop in STS1 Interoffice Transport Combination -									1						
	Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	First DS1 Loop in STS1 Interoffice Transport Combination -															1
	Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
1	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month			UNCSX	1L5XX	6.42										1
- -	Interoffice Transport - Dedicated - STS1 combination - Facility			014007	ILOAA	0.42			1					 		—
	Termination			UNCSX	U1TFS	704.44	279.37	163.12	60.33	58.59		15.69				1
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	8.64	6.59	4.73				15.69				1
	Additional DS1Loop in STS1 Interoffice Transport Combination -			LINGAY	1101.704	22.2						4= 0-				1
\longrightarrow	Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69		1		ĺ
-+-	Additional DS1Loop in STS1 Interoffice Transport Combination -	 		DINOIA	USLAA	100.40	200.00	137.09	44.00	11.73		13.09		 		
i l	Zone 3	1	3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69	1		1	i
	DS3 Interface Unit (DS1 COCI) combination per month	1		UNC1X	UC1D1	8.64	6.59	4.73	100			15.69				

ONRONDL	ED NETWORK ELEMENTS - South Carolina			1	1	1						T -		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						B	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	l.	L
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Currently Combined Network Elements Switch -As	-														
	Is Charge			UNCSX	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WI	RE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERC	FFICE 1	TRANS	PORT (EEL)												
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport			LINODY	LIDI 50	00.00	400.00	00.40	50.05	44.04		45.00				
	Combination - Zone 1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport			ONODA	ODLOG	00.00	120.00	00.12	00.00	14.01		10.00				
	Combination - Zone 3		3	UNCDX	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -															
	Per Mile			UNCDX	1L5XX	0.0134										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -															
	Facility Termination		<u> </u>	UNCDX	U1TD5	13.41	40.63	27.47	16.77	6.91		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As	1	1	LINCDY	LINICOC		F 04	F 04	7.00	7.00		45.00				
A. 1A/I	Is Charge RE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERC	EEICE 3	DANC	UNCDX	UNCCC		5.61	5.61	7.00	7.00	-	15.69			1	
4-771	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport	Frice	CHANS	FORT (EEL)	+									1		
	Combination - Zone 1		1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		† ·	0.105%	02201	20.00	120.00	00.12	00.00			10.00				
	Combination - Zone 2		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport															
	Combination - Zone 3		3	UNCDX	UDL64	34.74	126.66	89.12	59.35	14.61		15.69				
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	Per Mile			UNCDX	1L5XX	0.0134										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -						40.00									
	Facility Termination			UNCDX	U1TD6	13.41	40.63	27.47	16.77	6.91		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As Is Charge			UNCDX	UNCCC		5.61	5.61	7.00	7.00		15.69				
ADDITIONAL	L NETWORK ELEMENTS			ONODA	ONCCC		3.01	3.01	7.00	7.00		13.03				
	n used as a part of a currently combined facility, the non-recur	rng cha	raes de	not apply, but a S	witch As Is c	harge does apr	olv.									
	n used as ordinarily combined network elements in All States,															
	recurring Currently Combined Network Elements "Switch As Is"															
	Nonrecurring Currently Combined Network Elements Switch -As	-														
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		5.61	5.61	7.00	7.00		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As	1	1	LINORY	Liniono		F 0.1		7.00	7		45.00				
	Is Charge - 56/64 kbps Nonrecurring Currently Combined Network Elements Switch -As	1	!	UNCDX	UNCCC		5.61	5.61	7.00	7.00	-	15.69			1	1
.	Is Charge - DS1	1	1	UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As	 	-	014017	514000		3.01	5.01	7.00	7.00	-	13.09		1	1	1
	Is Charge - DS3		1	UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As	-		-												
	Is Charge - STS1		<u>L</u>	UNCSX	UNCCC		5.61	5.61	7.00	7.00		15.69	<u> </u>	<u> </u>		<u> </u>
NOT	E: Local Channel - Dedicated Transport - minimum billing perio	d - Belo	w DS3													
	Local Channel - Dedicated - 2-Wire Voice Grade			UNCXV	ULDV2	15.33	193.53	33.24	36.72	3.21		15.69		ļ		
	Local Channel - Dedicated - 4-Wire Voice Grade	1	<u> </u>	UNCXV	ULDV4	16.54	193.97	33.68	37.19	3.68		15.69	-	1	ļ.	
	Local Channel - Dedicated - DS1 per month Zone 1	1	2	UNC1X	ULDF1	42.62	177.87	154.06	22.24	15.30		15.69			-	
	Local Channel - Dedicated -DS1 Per Month Zone 2 Local Channel - Dedicated - DS1- Per Month Zone 3	1	3	UNC1X UNC1X	ULDF1 ULDF1	70.32 190.68	177.87 177.87	154.06 154.06	22.24 22.24	15.30 15.30		15.69 15.69	1	-	†	
	Local Channel - Dedicated - DS3 - Per Mile per month	1	- 3	UNC3X	1L5NC	11.93	177.07	134.00	22.24	15.50	-	13.09		1	1	1
	Local Channel - Dedicated - DS3 - Facility Termination	1	<u> </u>	UNC3X	ULDF3	446.00	452.52	264.53	119.75	83.77		15.69				
	Local Channel - Dedicated - STS-1- Per Mile per month	1		UNCSX	1L5NC	11.93							1			
	Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	435.10	452.52	264.53	119.75	83.77		15.69				
	onal Features & Functions:															
MUL	TIPLEXERS				1									ļ		
	Channelization - DS1 to DS0 Channel System	1	<u> </u>	UXTD1	MQ1	107.57	91.24	62.71	10.56	9.81		15.69	ļ		ļ	<u> </u>
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)		1	UDL	1D1DD	1.19	6.59	4.73				15.69				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per	1	 	UDL	טטוטו	1.19	86.0	4.73	1		1	15.69	1	1	1	
	month			UDN	UC1CA	2.56	6.59	4.73				15.69		l		

ONRONDE	ED NETWORK ELEMENTS - South Carolina			,		,								ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Names		l Name and a series	Dianamant			000	Detec(f)		
					+	Rec	Nonrec		Nonrecurring First		SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.56	First 6.59	Add'I 4.73	FIISt	Add'l	SOWIEC	15.69	SOWAN	SOWAN	SUMAN	SUMAN
	DS3 to DS1 Channel System per month			UXTD3	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				+
	STS1 to DS1 Channel System per month			UXTS1	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				+
	DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	8.64	6.59	4.73	33.33	31.90		15.69				+
	DS3 Interface Unit (DS1 COCI) used with Local Channel per			OOL	OCIDI	0.04	0.55	4.73				13.03				+
	month			ULDD1	UC1D1	8.64	6.59	4.73				15.69				
	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel			OLDD I	COIDI	0.04	0.00	4.70				10.00				†
	per month			U1TD1	UC1D1	8.64	6.59	4.73				15.69				
UNBUNDLE	LOCAL EXCHANGE SWITCHING(PORTS)						0.00									
	ange Ports															
	E: Although the Port Rate includes all available features in GA,	KY. LA	& TN. t	he desired features	will need to I	oe ordered usir	ng retail USOCs									
	RE VOICE GRADE LINE PORT RATES (RES)						•									1
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.65	2.38	2.28	1.42	1.33		15.69				1
																1
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.65	2.38	2.28	1.42	1.33		15.69				
	, and the second															
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled SC extended local															
	dialing parity Port with Caller ID - Res.			UEPSR	UEPAU	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled South Carolina Area															
	Calling port with Caller ID - Res (LW8)			UEPSR	UEPAJ	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled res, low usage line port															
	with Caller ID (LUM)			UEPSR	UEPAP	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG South Carolina Residence Dialing															
	Plan without Caller ID			UEPSR	UEPWL	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG South Carolina Residence Area															
	Calling Plan without Caller ID capability			UEPSR	UEPRS	1.65	2.38	2.28	1.42	1.33		15.69				
	2-Wire voice unbundled Low Usage Line Port without Caller ID															
	Capability			UEPSR	UEPRT	1.65	2.38	2.28	1.42	1.33		15.69				
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00				15.69				
FEA	TURES															
	All Available Vertical Features			UEPSR	UEPVF	3.04	0.00	0.00				15.69				
2-WI	RE VOICE GRADE LINE PORT RATES (BUS)															
	Exchange Ports - 2-Wire Analog Line Port without Caller ID -															
	Bus			UEPSB	UEPBL	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled Line Port with															
	unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled SC extended local															
	dialing parity Port with Caller ID - Bus.			UEPSB	UEPAZ	1.65	2.38	2.28	1.42	1.33		15.69				
	Exhange Ports - 2-Wire VG unbundled incoming only port with															
	Caller ID - Bus			UEPSB	UEPB1	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled South Carolina Bus															
	Area Calling Port with Caller ID - Bus (LMB)			UEPSB	UEPAB	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire Voice South Carolina Business Dialing															
	Plan without Caller ID			UEPSB	UEPWM	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire Voice South Carolina Business Area			LIEDOD	LIEDDD	4.05	0.00	0.00	4 40	4.00		45.00				
	Calling Port without Caller ID			UEPSB	UEPBB	1.65	2.38	2.28	1.42	1.33		15.69				
	2-Wire voice unbundled Incoming Only Port without Caller ID			LIEDOD	LIEDDE	4.05	2.00	0.00	4.40	4.00	1	45.00				1
	Capability		!	UEPSB	UEPBE	1.65	2.38	2.28	1.42	1.33	ļ	15.69		-	1	+
	Subsequent Activity		<u> </u>	UEPSB	USASC	0.00	0.00	0.00				15.69		1		+
FEA	All Available Vertical Features		!	UEPSB	UEPVF	3.04	0.00	0.00			ļ	15.00		-	1	+
			<u> </u>	UEFOB	UEPVF	3.04	0.00					15.69				+
EVO	All Available Vertical Features		!	 	UEPVF	3.04	0.00	0.00			ļ	15.69		-	1	+
	HANGE PORT RATES (DID & PBX)		<u> </u>	UEPSE	UEPRD	1.65	31.34	14.88	13.97	0.00		15.69				
LAC			1	IUEPOE	IUEPKD	1.65	31.34	14.88	13.97	0.90	i	15.69	l	1	1	1
EXC	2-Wire VG Unbundled 2-Way PBX Trunk - Res 2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus		1	UEPSP	UEPPC	1.65	31.34	14.88	13.97	0.90		15.69				

	ED NETWORK ELEMENTS - South Carolina													ment: 2	Exhil	oit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incrementa Charge - Manual Svo Order vs. Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	•	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP UEPSP	UEPXC	1.65	31.34	14.88 14.88	13.97	0.90 0.90		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPSP	UEPXD	1.65	31.34	14.88	13.97	0.90		15.69				
	Capable Port			UEPSP	UEPXE	1.65	31.34	14.88	13.97	0.90		15.69				
-	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			OLFSF	OLFAL	1.05	31.34	14.00	13.97	0.90		13.09				
	Administrative Calling Port			UEPSP	UEPXL	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1	!	021 01	JEI AL	1.00	31.54	14.00	13.31	0.30		10.03		1	 	
1	Room Calling Port	l		UEPSP	UEPXM	1.65	31.34	14.88	13.97	0.90		15.69			1	
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital	1	†		1		004	50		3.30		.0.00			1	
	Discount Room Calling Port	l		UEPSP	UEPXO	1.65	31.34	14.88	13.97	0.90		15.69			1	
1	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port		i –	UEPSP	UEPXS	1.65	31.34	14.88	13.97	0.90		15.69		İ	İ	
	2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus		1													
	Calling Port			UEPSP	UEPXT	1.65	31.34	14.88	13.97	0.90		15.69				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00				15.69				
FEAT	TURES															
	All Available Vertical Features			UEPSP UEPSE	UEPVF	3.04	0.00	0.00				15.69				
EXC	HANGE PORT RATES (COIN)															
	Exchange Ports - Coin Port					1.65	2.38	2.28	1.42	1.33		15.69				
	l Switching Features offered with Port															
	E: Transmission/usage charges associated with POTS circuit sv													<u> </u>		
	E: Access to B Channel or D Channel Packet capabilities will be	availal	ble onl	y through BFR/New	Business Re	quest Process.	Rates for the	packet capabi	lities will be de	termined via t	he Bona Fic	le Request/l	New Business	Request Pro	cess.	
	D LOCAL EXCHANGE SWITCHING(PORTS) HANGE PORT RATES															
EXC																
	Evolungo Porto 2 Wiro DID Port		1	HEDEV	LIEDDO	0.00	110 F7	10.70	60.03	2 77		15.60				
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.86	119.57	18.78	60.03	3.77		15.69				
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID															
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability			UEPDD	UEPDD	73.62	202.47	95.90	72.75	2.47		15.69				
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPDD UEPTX UEPSX	UEPDD U1PMA	73.62 13.38	202.47 72.93	95.90 53.11								
NOTE	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered	vitched	USage	UEPDD UEPTX UEPSX UEPTX UEPSX	UEPDD U1PMA UEPVF	73.62 13.38 3.04	202.47 72.93 0.00	95.90 53.11 0.00	72.75 47.90	2.47 10.76	ated with 2-	15.69 15.69	oorts.			
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sy			UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c	UEPDD U1PMA UEPVF	73.62 13.38 3.04 ed voice and/or	202.47 72.93 0.00 circuit switche	95.90 53.11 0.00 ed data transm	72.75 47.90 hission by B-Ch	2.47 10.76 annels associ		15.69 15.69 wire ISDN p		s Request Pro	cess.	
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sues. Access to B Channel or D Channel Packet capabilities will be			UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c	UEPDD U1PMA UEPVF	73.62 13.38 3.04 ed voice and/or	202.47 72.93 0.00 circuit switche	95.90 53.11 0.00 ed data transm	72.75 47.90 hission by B-Ch	2.47 10.76 annels associ		15.69 15.69 wire ISDN p		s Request Pro	ocess.	
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles			UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to control of the co	UEPDD U1PMA UEPVF circuit switched Business Re	73.62 13.38 3.04 ed voice and/or quest Process.	202.47 72.93 0.00 circuit switche Rates for the	95.90 53.11 0.00 ed data transm packet capabi	72.75 47.90 hission by B-Ch	2.47 10.76 annels associ		15.69 15.69 wire ISDN p		s Request Pro	ocess.	
NOTE	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sues. Access to B Channel or D Channel Packet capabilities will be	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to coor through BFR/New UEPTX UEPSX	UEPDD U1PMA UEPVF circuit switcher Business Re	73.62 13.38 3.04 ed voice and/or quest Process. 0.00	202.47 72.93 0.00 circuit switche Rates for the	95.90 53.11 0.00 ed data transm packet capabi 0.00	72.75 47.90 hission by B-Ch lities will be de	2.47 10.76 annels associ termined via t		15.69 15.69 wire ISDN p		s Request Pro	cess.	
UNB	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to coor through BFR/New UEPTX UEPSX	UEPDD U1PMA UEPVF circuit switcher Business Re	73.62 13.38 3.04 ed voice and/or quest Process. 0.00	202.47 72.93 0.00 circuit switche Rates for the	95.90 53.11 0.00 ed data transm packet capabi 0.00	72.75 47.90 hission by B-Ch lities will be de	2.47 10.76 annels associ termined via t		15.69 15.69 wire ISDN p		s Request Pro	cess.	
UNB	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to coor through BFR/New UEPTX UEPSX	UEPDD U1PMA UEPVF circuit switcher Business Re	73.62 13.38 3.04 ed voice and/or quest Process. 0.00	202.47 72.93 0.00 circuit switche Rates for the	95.90 53.11 0.00 ed data transm packet capabi 0.00	72.75 47.90 hission by B-Ch lities will be de	2.47 10.76 annels associ termined via t		15.69 15.69 wire ISDN p		s Request Pro	cess.	
UNB	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit so E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPEX	UEPDD U1PMA UEPVF icruit switche Business Re U1UMA UEPEX	73.62 13.38 3.04 ed voice and/or quest Process. 0.00 107.44	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27	95.90 53.11 0.00 ad data transm packet capabi 0.00 101.78	72.75 47.90 hission by B-Ch lities will be de 79.35	2.47 10.76 annels associ termined via t 20.10		15.69 15.69 wire ISDN p le Request/I 15.69		s Request Pro	cess.	
UNB	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPEX UEPVR	UEPDD U1PMA UEPVF circuit switche Business Re U1UMA UEPEX UERAC UERAC	73.62 13.38 3.04 od voice and/or quest Process. 0.00 107.44 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27	95.90 53.11 0.00 od data transm packet capabi 0.00 101.78	72.75 47.90 ission by B-Ch lities will be de 79.35	2.47 10.76 annels associ termined via t 20.10 1.33		15.69 15.69 wire ISDN pele Request/l 15.69 15.69		s Request Pro	ocess.	
UNB	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sve: E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, Local Calling - Res	availal		UEPDD UEPTX UEPSX UEPTX UEPSX I will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF Ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27 2.38 2.38	95.90 53.11 0.00 od data transm packet capabi 101.78 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN p le Request/l 15.69 15.69		s Request Pro	cess.	
UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit ss E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPEX UEPVR	UEPDD U1PMA UEPVF circuit switche Business Re U1UMA UEPEX UERAC UERAC	73.62 13.38 3.04 od voice and/or quest Process. 0.00 107.44 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27	95.90 53.11 0.00 od data transm packet capabi 0.00 101.78	72.75 47.90 ission by B-Ch lities will be de 79.35	2.47 10.76 annels associ termined via t 20.10 1.33		15.69 15.69 wire ISDN pele Request/l 15.69 15.69		s Request Pro	cess.	
UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Recurring	availal		UEPDD UEPTX UEPSX UEPTX UEPSX I will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF Ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27 2.38 2.38	95.90 53.11 0.00 od data transm packet capabi 101.78 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN p le Request/l 15.69 15.69		s Request Pro	cess.	
UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit ss: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion -	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to cy ythrough BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UEPEX UERAC UERAC UERLC UERTE UERTR	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27 2.38 2.38 2.38 2.38	95.90 53.11 0.00 od data transm packet capabi 101.78 2.28 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN ple Request/l 15.69 15.69 15.69 15.69		s Request Pro	ocess.	
UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit ss E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Recurring Unbundled Remote Call Forwarding Service Conversion Switch-as-is	availal		UEPDD UEPTX UEPSX UEPTX UEPSX I will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF Ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27 2.38 2.38	95.90 53.11 0.00 od data transm packet capabi 101.78 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN p le Request/l 15.69 15.69		s Request Pro	cess.	
UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sverices to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED PREMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF circuit switche Business Re U1UMA UEPEX UEPEX UERAC UERLC UERTE UERTE UERTR	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche 0.00 204.27 2.38 2.38 2.38 2.38	95.90 53.11 0.00 old data transm packet capabi 0.00 101.78 2.28 2.28 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN ple Request/l 15.69 15.69 15.69 15.69		s Request Pro	cess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit ss: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to cy ythrough BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UEPEX UERAC UERAC UERLC UERTE UERTR	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27 2.38 2.38 2.38 2.38	95.90 53.11 0.00 od data transm packet capabi 101.78 2.28 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN ple Request/l 15.69 15.69 15.69 15.69		s Request Pro	cess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sverices to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED PREMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF circuit switche Business Re U1UMA UEPEX UEPEX UERAC UERLC UERTE UERTE UERTR	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche 0.00 204.27 2.38 2.38 2.38 2.38	95.90 53.11 0.00 old data transm packet capabi 0.00 101.78 2.28 2.28 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN ple Request/l 15.69 15.69 15.69 15.69		s Request Pro	ocess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sy E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to cy through BFR/New UEPTX UEPSX UEPEX UEPVR	UEPDD U1PMA UEPVF circuit switche Business Re U1UMA UEPEX UERAC UERAC UERTE UERTE UERTR USAC2 USACC	73.62 13.38 3.04 3 voice and/or quest Process. 0.00 107.44 1.65 1.65 1.65	202.47 72.93 0.00 circuit switche 0.00 204.27 2.38 2.38 2.38 2.38 0.10 0.10	95.90 53.11 0.00 dd data transm packet capabi 0.00 101.78 2.28 2.28 2.28 2.28 0.10	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42 1.42	2.47 10.76 annels associ termined via t 20.10 1.33 1.33 1.33		15.69 wire ISDN p le Request/l 15.69 15.69 15.69 15.69 15.69		s Request Pro	ccess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit ss: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF circuit switche Business Re U1UMA UEPEX UEPEX UERAC UERLC UERTE UERTE UERTR	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65	202.47 72.93 0.00 circuit switche 0.00 204.27 2.38 2.38 2.38 2.38	95.90 53.11 0.00 old data transm packet capabi 0.00 101.78 2.28 2.28 2.28 2.28	72.75 47.90 hission by B-Ch lities will be de 79.35 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33		15.69 15.69 wire ISDN ple Request/l 15.69 15.69 15.69 15.69		s Request Pro	cess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit ss: Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED PREMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to cy UEPTX UEPSX UEPTX UEPSX UEPTX UEPSX UEPTX UEPYR UEPVR	UEPDD U1PMA UEPVF Ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 9.00 204.27 2.38 2.38 2.38 2.38 0.10 0.10 2.38	95.90 53.11 0.00 od data transm packet capabi 101.78 2.28 2.28 2.28 2.28 0.10	72.75 47.90 ission by B-Ch lities will be de 79.35 1.42 1.42 1.42 1.42	2.47 10.76 sannels associ termined via t 20.10 1.33 1.33 1.33		15.69 15.69 wire ISDN ple Request/l 15.69 15.69 15.69		s Request Pro	cess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit ss E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC	73.62 13.38 3.04 d voice and/or quest Process. 1.65 1.65 1.65 1.65 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27 2.38 2.38 2.38 2.38 0.10 0.10 0.10 2.38	95.90 53.11 0.00 d data transm packet capabi 101.78 2.28 2.28 2.28 2.28 0.10 0.10 2.28	72.75 47.90 hission by B-Ch itites will be de 79.35 1.42 1.42 1.42 1.42 1.42	2.47 10.76 annels associ termined via t 20.10 1.33 1.33 1.33 1.33		15.69 wire ISDN p le Request/l 15.69 15.69 15.69 15.69 15.69 15.69 15.69		s Request Pro	cess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit so E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Proflies Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus	availal		UEPDD UEPTX UEPSX UEPTX UEPSX viiii also apply to cy y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65 1.65 1.65 1.65 1.65 1.65	202.47 72.93 0.00 circuit switchen 8 0.00 204.27 2.38 2.38 2.38 2.38 2.38 2.38 2.38 2.38	95.90 53.11 0.00 d data transm packet capabi 0.00 101.78 2.28 2.28 2.28 2.28 0.10 0.10 2.28 2.28	72.75 47.90 ission by B-Ch lities will be de 79.35 1.42 1.42 1.42 1.42 1.42 1.42 1.42	2.47 10.76 annels associ termined via t 20.10 1.33 1.33 1.33 1.33 1.33		15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69		s Request Pro	cess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit st. E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus	availal		UEPDD UEPTX UEPSX UEPTX UEPSX will also apply to c y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC	73.62 13.38 3.04 d voice and/or quest Process. 1.65 1.65 1.65 1.65 1.65 1.65	202.47 72.93 0.00 circuit switche Rates for the 0.00 204.27 2.38 2.38 2.38 2.38 0.10 0.10 0.10 2.38	95.90 53.11 0.00 d data transm packet capabi 101.78 2.28 2.28 2.28 2.28 0.10 0.10 2.28	72.75 47.90 hission by B-Ch itites will be de 79.35 1.42 1.42 1.42 1.42 1.42	2.47 10.76 annels associ termined via t 20.10 1.33 1.33 1.33 1.33		15.69 wire ISDN p le Request/l 15.69 15.69 15.69 15.69 15.69 15.69 15.69		s Request Pro	cess.	
UNBI UNBI UNBI	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit so E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Proflies Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY UNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) UNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus	availal		UEPDD UEPTX UEPSX UEPTX UEPSX viiii also apply to cy y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	UEPDD U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERTE UERTR USAC2 USACC UERAC UERAC UERAC	73.62 13.38 3.04 d voice and/or quest Process. 0.00 107.44 1.65 1.65 1.65 1.65 1.65 1.65 1.65	202.47 72.93 0.00 circuit switchen 8 0.00 204.27 2.38 2.38 2.38 2.38 2.38 2.38 2.38 2.38	95.90 53.11 0.00 d data transm packet capabi 0.00 101.78 2.28 2.28 2.28 2.28 0.10 0.10 2.28 2.28	72.75 47.90 ission by B-Ch lities will be de 79.35 1.42 1.42 1.42 1.42 1.42 1.42 1.42	2.47 10.76 annels associ termined via t 20.10 1.33 1.33 1.33 1.33 1.33		15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69		s Request Pro	cess.	

Version 3Q02: 09/06/02 Page 286 of 362

ONBONDL	ED NETWORK ELEMENTS - South Carolina													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonred			g Disconnect				Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Remote Call Forwarding Service - Conversion -											4= 00				
	Switch-as-is	<u> </u>		UEPVB	USAC2		0.10	0.10				15.69				
i l	Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)			UEPVB	USACC		0.10	0.10								
INDINDIE	D LOCAL SWITCHING. PORT USAGE			OLF VB	USACC		0.10	0.10								
	Office Switching (Port Usage)															
Liid	End Office Switching Function, Per MOU					0.0010519										
	End Office Trunk Port - Shared, Per MOU					0.0002136										
Tand	dem Switching (Port Usage) (Local or Access Tandem)															
	Tandem Switching Function Per MOU					0.0001634										
	Tandem Trunk Port - Shared, Per MOU					0.0002863										
Com	mon Transport															
	Common Transport - Per Mile, Per MOU					0.0000045		•	•				_			
	Common Transport - Facilities Termination Per MOU					0.0004095										
	D PORT/LOOP COMBINATIONS - COST BASED RATES	<u> L</u>		L.,		<u> </u>	L			ļ				ļ	ļ	
	Based Rates are applied where BellSouth is required by FCC a									L	l					
	ures shall apply to the Unbundled Port/Loop Combination - Cos															
	Office and Tandem Switching Usage and Common Transport Usage															
	first and additional Port nonrecurring charges apply to Not Curr	rentily Co	ombine	ea Compos. For Cu	rrently Comb	inea Compos ti	ne nonrecurrin	g cnarges sna	i be those ide	ntified in the N	onrecurring	- Currently	Combined se	ections.		
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates									-						
UNE	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										1
UNE	Loop Rates		3			21.11										
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	20.38										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	26.04										
2-Wi	re Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.13	37.93	16.72				15.69				
ı l	2-Wire voice Grade unbundled South Carolina extended local															
	dialing parity port with Caller ID - res			UEPRX	UEPAU	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled South Carolina Area Calling port with Caller ID - res (LW8)			UEPRX	UEPAJ	1.13	37.93	16.72				15.69				
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	1.13	37.93	16.72				15.69				
	2-Wire Voice Unbundled South Carolina Residence Dialing Plan															
	without Caller ID			UEPRX	UEPWL	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled South Carolina Area Calling Port without Caller ID Capability			UEPRX	UEPRS	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX	UEPRT	1.13	37.93	16.72				15.69				
FEA	TURES				_	ļ				.					.	
	All Features Offered	<u> </u>		UEPRX	UEPVF	3.04	0.00	0.00		_		15.69		ļ	-	
LOC	AL NUMBER PORTABILITY	 		UEPRX	LNPCX	0.05				!				 	!	
NO	Local Number Portability (1 per port) RECURRING CHARGES (NRCs) - CURRENTLY COMBINED	 		UEPKA	LINPUX	0.35				 				 	 	-
NON	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPRX	USAC2		0.10	0.10				15.69				
	2-Wirch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPRX	USACC		0.10	0.10				15.69				
ADE	ITIONAL NRCs	 	-	ULPRA	USACC	 	0.10	0.10		-		15.09		-	 	-
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent	 		 	+	 				+				1	+	1
				1		1	1		i e	1	1			•		1
	Activity			UEPRX	USAS2	0.00	0.00	0.00				15.69				
2-WI				UEPRX	USAS2	0.00	0.00	0.00				15.69				

ONRONDLED	NETWORK ELEMENTS - South Carolina				•							1 -		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							N		T. N	D'						
						Rec	Nonred First		Nonrecurring I	Add'l	COMEC	SOMAN	SOMAN	Rates(\$)	SOMAN	SOMAN
	2-Wire VG Loop/Port Combo - Zone 2		2		+	21.52	FIRST	Add'l	First	Addi	SOMEC	SOWAN	SOWAN	SUMAN	SUMAN	SOWAN
	2-Wire VG Loop/Port Combo - Zone 2		3		+	27.17			 							
UNE Loc					+	21.11			 							-
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	13.76			-							
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	20.38										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	26.04										
	/oice Grade Line Port (Bus)															
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.13	37.93	16.72				15.69				
2	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.13	37.93	16.72				15.69				
2	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.13	37.93	16.72				15.69				
	2-Wire voice Grade unbundled South Carolina extended local					_		-								
	dialing parity port with Caller ID - bus			UEPBX	UEPAZ	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.13	37.93	16.72		<u> </u>		15.69				
	2-Wire voice unbundled South Carolina Bus Area Calling Port														1	
	with Caller ID (LMB)			UEPBX	UEPAB	1.13	37.93	16.72				15.69				
	2-Wire Voice Unbundled South Carolina Business Dialing Plan															
	without Caller ID			UEPBX	UEPWM	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled South Carolina Business Area Calling															
	Port without Caller ID Capability			UEPBX	UEPBB	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled Incoming Only Port without Caller ID			HEDDY	LIEDDE	4.40	07.00	40.70				45.00				
	Capability			UEPBX	UEPBE	1.13	37.93	16.72	-			15.69				
	NUMBER PORTABILITY			UEPBX	LNPCX	0.35			<u> </u>						-	
FEATUR	Local Number Portability (1 per port)			UEPBA	LINPUX	0.35			+		-				-	+
	All Features Offered			UEPBX	UEPVF	3.04	0.00	0.00	<u> </u>			15.69				
	CURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLFBX	OLFVI	3.04	0.00	0.00	 			13.09				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -								-							
	Switch-as-is			UEPBX	USAC2		0.10	0.10				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -			02. 5/	00,102		0.10	00				10.00				1
	Switch with change			UEPBX	USACC		0.10	0.10				15.69				
	DNAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															1
	Activity			UEPBX	USAS2		0.00	0.00				15.69				
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
UNE Por	rt/Loop Combination Rates															
2	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										
2	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										
UNE Loc																<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	13.76									1	
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	20.38										ļ
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	26.04			.							<u> </u>
	Voice Grade Line Port Rates (RES - PBX)		-	1	+ +				1					1	1	
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -			LIEDDC	HEDDO	4 40	07.00	40.70				45.00			1	
	Res NUMBER PORTABILITY	-	-	UEPRG	UEPRD	1.13	37.93	16.72	+			15.69		-		
	Local Number Portability (1 per port)	-		UEPRG	LNPCP	3.15	0.00	0.00	+ +			15.69		1	 	
FEATUR		-		OLFING	LINFOR	ა. 15	0.00	0.00	+ +			15.09		1	 	
	All Features Offered	-		UEPRG	UEPVF	3.04	0.00	0.00	+			15.69		 	t	
	CURRING CHARGES (NRCs) - CURRENTLY COMBINED				J 11	0.04	0.00	0.00	+ +		<u> </u>	10.00		 	I	†
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -				1				1					1	1	
	Conversion - Switch-As-Is			UEPRG	USAC2		7.93	1.91	1			15.69			1	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -						-		1					İ	İ	
	Conversion - Switch with Change			UEPRG	USACC		7.93	1.91				15.69		1	I	
ADDITIO	DNAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				15.69				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
	Group						7.34	7.34	1			15.69		Ì	I	1

UNBUNDLE	D NETWORK ELEMENTS - South Carolina													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incrementa Charge - Manual Sv Order vs. Electronic
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonred First	curring Add'l	Nonrecurring First	g Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)				-		THOU	Auu i	11130	Auui	JOHILO	JONAN	JONAN	JONAN	JOHAN	JONAN
	ort/Loop Combination Rates										1				1	
	2-Wire VG Loop/Port Combo - Zone 1		1			14.89					1				1	
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										
UNE Lo	oop Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	20.38										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	26.04										
2-Wire	Voice Grade Line Port Rates (BUS - PBX)															
		1								I					I	
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	ļ		UEPPX	UEPPC	1.13	37.93	16.72	ļ			15.69				ļ
	Line Side Unbundled Outward PBX Trunk Port - Bus	<u> </u>		UEPPX	UEPPO	1.13	37.93	16.72		.	 	15.69			-	
	Line Side Unbundled Incoming PBX Trunk Port - Bus	 		UEPPX	UEPP1	1.13	37.93	16.72	1	!	1	15.69			!	
	2-Wire Voice Unbundled PBX LD Terminal Ports	!		UEPPX	UEPLD	1.13	37.93	16.72	1	 	1	15.69		-	 	-
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports	-		UEPPX UEPPX	UEPXA UEPXB	1.13 1.13	37.93 37.93	16.72 16.72		 	1	15.69 15.69			 	
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXB	1.13	37.93	16.72			1	15.69			-	-
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.13	37.93	16.72		-	-	15.69				_
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPPA	UEPAD	1.13	37.93	10.72			1	15.09			1	
	Capable Port			UEPPX	UEPXE	1.13	37.93	16.72				15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX	UEPXL	1.13	37.93	16.72				15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPPX	UEPXM	1.13	37.93	16.72				15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX	UEPXO	1.13	37.93	16.72				15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.13	37.93	16.72				15.69				
	2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus Calling Port			UEPPX	UEPXT	1.13	37.93	16.72				15.69				
LOCAL	NUMBER PORTABILITY															İ
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00				15.69				1
FEATU																
	All Features Offered			UEPPX	UEPVF	3.04	0.00	0.00				15.69				
NONRE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch-As-Is			UEPPX	USAC2		7.93	1.91				15.69				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			l	1										1	
	Conversion - Switch with Change			UEPPX	USACC		7.93	1.91			1	15.69			ļ	ļ
ADDIT	IONAL NRCs	ļ							ļ							ļ
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00				15.69				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt	1			33,32	0.00	0.00	0.00	1	-	1	10.00			I	†
	Group	1		1			7.34	7.34		1		15.69				
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	RT.			1					1	1			İ	1	1
	ort/Loop Combination Rates															
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			14.89										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			21.52										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			27.17										
UNE Lo	oop Rates									ļ					1	
	2-Wire Voice Grade Loop (SL1) - Zone 1	ļ	1	UEPCO	UEPLX	13.76				ļ	1			ļ	ļ	ļ
	2-Wire Voice Grade Loop (SL1) - Zone 2	ļ	2	UEPCO	UEPLX	20.38				ļ	1			ļ	ļ	ļ
	2-Wire Voice Grade Loop (SL1) - Zone 3	ļ	3	UEPCO	UEPLX	26.04			ļ							ļ
2-Wire	Voice Grade Line Ports (COIN)	ļ								-	<u> </u>				-	
	2-Wire Coin 2-Way without Operator Screening and without Blocking (SC)			UEPCO	UEPSD	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)			UEPCO	UEPSA	1.13	37.93	16.72		1		15.69				

ONBONDL	D NETWORK ELEMENTS - South Carolina			1							T -			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking															
	(SC)			UEPCO	UEPSH	1.13	37.93	16.72				15.69				-
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)			UEPCO	UEPSC	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-Way with Operator Screening and: 900 Blocking:			021 00	021 00	1.10	01.00	10.72				10.00				
	900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCC	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD,															
	011+, Local; Enhanced Call OPT 3YV (SC)			UEPCO	UEPCE	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD,															
	011+, Local; Enhanced Call OPT AP7 (SC) 2-Wire Coin Outward without Blocking and without Operator			UEPCO	UEPCF	1.13	37.93	16.72				15.69				
	Screening (SC)			UEPCO	UEPSG	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward with Operator Screening and 011 Blocking			02. 00	02.00	0	07.00	2				10.00				
	(SC)			UEPCO	UEPSF	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward with Operator Screening and Blocking:															
	011, 900/976, 1+DDD (SC)			UEPCO	UEPSJ	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward with Operator Screening and Blocking:			LIEBOO	LIEDOM	4.40	27.02	40.70				45.00				
	900/976, 1+DDD, 011+, and Local (SC) 2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD,			UEPCO	UEPCM	1.13	37.93	16.72				15.69				
	011+, Local; Enhanced Calling OPT 3YW (SC)			UEPCO	UEPCP	1.13	37.93	16.72				15.69				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.13	37.93	16.72				15.69			İ	
	2-Wire Coin Outward Smartline with 900/976 (all states except															
	LA)			UEPCO	UEPCR	1.13	37.93	16.72				15.69				
ADDI	TIONAL UNE COIN PORT/LOOP (RC)															<u> </u>
1.004	UNE Coin Port/Loop Combo Usage (Flat Rate) L NUMBER PORTABILITY			UEPCO	URECU	4.05	37.93	16.72				15.69				
LOCA	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										1
NONE	ECURRING CHARGES - CURRENTLY COMBINED			02. 00	2.1. 0/1	0.00										
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPCO	USAC2		0.10	0.10				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -			LIEBOO	110400		0.40	0.40				45.00				
ADDI	Switch with change			UEPCO	USACC		0.10	0.10				15.69				1
ADDI	2-Wire Voice Grade Loop/Line Port Combination - Subsequent														1	1
	Activity			UEPCO	USAS2		0.00	0.00				15.69				
2-WIR	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	PORT (RES)												
UNE F	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			22.50										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			30.56										1
IINF I	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 Loop Rates		3		+ +	37.22					1					
O.V.E.	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	20.85										1
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	28.91										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	35.57										
2-Wire	e Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPFR	UEPRL UEPRC	1.65	108.36	70.71 70.71	1.42 1.42	1.33		15.69				
	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res		-	UEPFR UEPFR	UEPRO	1.65 1.65	108.36 108.36	70.71	1.42	1.33 1.33	}	15.69 15.69			-	+
 	2-Wire voice dribdridled port outgoing only 1 res 2-Wire voice Grade unbundled South Carolina extended local			0=1110	02.110	1.00	100.50	70.71	1.72	1.33	1	10.08			†	
	dialing parity port with Caller ID - res	L	L	UEPFR	UEPAU	1.65	108.36	70.71	1.42	1.33	<u></u>	15.69		<u> </u>	<u> </u>	
	2-Wire voice unbundled South Carolina Area Calling port with							-								
	Caller ID - res (LW8)			UEPFR	UEPAJ	1.65	108.36	70.71	1.42	1.33		15.69				ļ
	2-Wire voice unbundles res, low usage line port with Caller ID		1	LIEDED	LIEDAD	4.05	400.00	70.74	4.40	4.00		45.00				
	(LUM) 2-Wire Voice Unbundled South Carolina Residence Dialing Plan		1	UEPFR	UEPAP	1.65	108.36	70.71	1.42	1.33		15.69				
	without Caller ID		1	UEPFR	UEPWL	1.65	108.36	70.71	1.42	1.33		15.69				
INTER	ROFFICE TRANSPORT				J //L	1.00	100.00	70.71	1.72	1.55		10.00				†
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility				1											
	Termination	<u></u>	<u>L</u>	UEPFR	U1TV2	24.30	40.63	27.47	16.77	6.91	<u></u>			<u> </u>	<u> </u>	<u></u>

ONRONE	DLEL	NETWORK ELEMENTS - South Carolina			,										ment: 2		bit: C
CATEGOR	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec	urring	Nonrecurring	Disconnect		•		Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
		or Fraction Mile			UEPFR	1L5XX	0.0167										
FE	EATU																
		All Features Offered			UEPFR	UEPVF	3.04	0.00	0.00				15.69				
LC		NUMBER PORTABILITY															
		Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
NC		CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port						4= 00									
		Combination - Conversion - Switch-as-is			UEPFR	USAC2		17.00	3.74				15.69				
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port						4= 00									
		Combination - Conversion - Switch-With-Change	<u> </u>	<u> </u>	UEPFR	USACC		17.00	3.74				15.69				
		VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	ORI (B02)										-		
UN	NE PO	rt/Loop Combination Rates	<u> </u>				00.50								-		
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1	<u> </u>	1			22.50								-		
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	1	2			30.56					1			1		+
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 3	1	3			37.22					}	-		 	1	
UN		op Rates		1	HEDED	LIEGEO	00.05										
		2-Wire Voice Grade Loop (SL2) - Zone 1			UEPFB	UECF2	20.85										
		2-Wire Voice Grade Loop (SL2) - Zone 2		_	UEPFB	UECF2	28.91										
		2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	35.57										
2-1		/oice Grade Line Port (Bus)			HEDED	LIEDDI	4.05	100.00	70.74	4.40	4.00		45.00				-
		2-Wire voice unbundled port without Caller ID - bus		<u> </u>	UEPFB	UEPBL	1.65	108.36	70.71	1.42	1.33		15.69				
		2-Wire voice unbundled port with Caller + E484 ID - bus		<u> </u>	UEPFB	UEPBC	1.65	108.36	70.71	1.42	1.33		15.69				
		2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.65	108.36	70.71	1.42	1.33		15.69				-
		2-Wire voice Grade unbundled South Carolina extended local			LIEDED	115047	4.05	400.00	70.74	4.40	4.00		45.00				
		dialing parity port with Caller ID - bus		-	UEPFB UEPFB	UEPAZ UEPB1	1.65	108.36 108.36	70.71 70.71	1.42	1.33		15.69				
		2-Wire voice unbundled incoming only port with Caller ID - Bus		-	UEPFB	UEPBI	1.65	108.36	70.71	1.42	1.33		15.69				
		2-Wire voice unbundled South Carolina Bus Area Calling Port with Caller ID (LMB)			UEPFB	UEPAB	1.65	108.36	70.71	1.42	1.33		15.69				
					UEPFB	UEPAB	1.00	108.30	70.71	1.42	1.33		15.69				+
		2-Wire Voice Unbundled South Carolina Business Dialing Plan without Caller ID			UEPFB	UEPWM	1.65	400.00	70.71	4.40	1.33		15.69				
1.0		NUMBER PORTABILITY			UEPFB	UEPWW	1.00	108.36	70.71	1.42	1.33		15.69				+
LC					UEPFB	LNPCX	0.35										+
INI		Local Number Portability (1 per port) PFICE TRANSPORT		<u> </u>	UEPFB	LINECX	0.33										+
IIN																	+
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFB	U1TV2	24.30	40.63	27.47	16.77	6.91						
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile	-		UEPFB	UTIVZ	24.30	40.63	21.41	10.77	0.91		-		-		+
		or Fraction Mile			UEPFB	1L5XX	0.0167										
FF	EATU			1	OLFIB	ILJAA	0.0107					1					+
- '-		All Features Offered			UEPFB	UEPVF	3.04	0.00	0.00				15.69				+
NC		CURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLFIB	OLF VI	3.04	0.00	0.00				13.08				
INC		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port		1		+						1					+
		Combination - Conversion - Switch-as-is			UEPFB	USAC2		17.00	3.74				15.69				
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port	1	 	02.10	30/102		17.00	5.74				15.05		 	1	+
		Combination - Conversion - Switch with change	1	1	UEPFB	USACC		17.00	3.74				15.69		I		
2-1		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)			OLITB	OOACC		17.00	3.74	1			13.03				+
		rt/Loop Combination Rates	1		1	1									t		
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1	1	1		1	22.50			+ +		1			-	<u> </u>	+
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	1	2		+ +	30.56								<u> </u>		
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 3	l	3			37.22								1		†
UN		op Rates	1	Ť		1	J								1	Ì	†
<u> </u>		2-Wire Voice Grade Loop (SL2) - Zone 1	1	1	UEPFP	UECF2	20.85								1	Ì	†
		2-Wire Voice Grade Loop (SL2) - Zone 2	l	2	UEPFP	UECF2	28.91								1		†
		2-Wire Voice Grade Loop (SL2) - Zone 3	1		UEPFP	UECF2	35.57								1	Ì	†
2-\		/oice Grade Line Port Rates (BUS - PBX)			İ	1				1					1		T
	Ī					1											1
		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	l		UEPFP	UEPPC	1.65	137.32	83.31	67.02	11.51		15.69		1		
		Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.65	137.32	83.31	67.02	11.51	İ	15.69		1		
		Line Side Unbundled Incoming PBX Trunk Port - Bus		†	UEPFP	UEPP1	1.65	137.32	83.31	67.02	11.51	1	15.69				1

UNB	SUNDLE	D NETWORK ELEMENTS - South Carolina	,		,		1								ment: 2		bit: C
CATE	EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incrementa Charge - Manual Svo Order vs. Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
								Nonrec	urring	Nonrecurring	Disconnect			088	Rates(\$)		<u> </u>
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.65	137.32	83.31	67.02	11.51	COME	15.69	COMPAR	COMPAR	COMPAR	COMPAR
		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.65	137.32	83.31	67.02	11.51		15.69				
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.65	137.32	83.31	67.02	11.51		15.69			1	
		2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.65	137.32	83.31	67.02	11.51		15.69				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.65	137.32	83.31	67.02	11.51		15.69				1
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPFP	UEPXE	1.65	137.32	83.31	67.02	11.51		15.69				
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPFP	UEPXL	1.65	137.32	83.31	67.02	11.51		15.69				
		Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEFFF	UEPAL	1.00	137.32	03.31	07.02	11.51		15.69				
		Room Calling Port			UEPFP	UEPXM	1.65	137.32	83.31	67.02	11.51		15.69				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital	l	1	LIEDED	LIEDVO	4.05	407.00	00.04	07.00	44.54		45.00		1	I	
	-	Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	 	 	UEPFP UEPFP	UEPXO UEPXS	1.65 1.65	137.32 137.32	83.31 83.31	67.02 67.02	11.51 11.51		15.69 15.69		 	!	
		2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus															
		Calling Port			UEPFP	UEPXT	1.65	137.32	83.31	67.02	11.51		15.69				
	LOCAI	L NUMBER PORTABILITY			LIEDED	LNDOD	0.45	0.00	0.00				45.00				
	INITED	Local Number Portability (1 per port) OFFICE TRANSPORT			UEPFP	LNPCP	3.15	0.00	0.00				15.69				
	INTER	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility					0.1.00	40.00		40.00							
		Termination Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFP	U1TV2	24.30	40.63	27.47	16.77	6.91						
		or Fraction Mile			UEPFP	1L5XX	0.0167										
	FEATU					<u> </u>											
	NOND	All Features Offered			UEPFP	UEPVF	3.04	0.00	0.00				15.69				
	NONK	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port		<u> </u>													
		Combination - Conversion - Switch-as-is			UEPFP	USAC2		17.00	3.74				15.69				
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch with change			UEPFP	USACC		17.00	3.74				15.69				
IINRI	INDI ED	PORT/LOOP COMBINATIONS - COST BASED RATES			UEPFP	USACC		17.00	3.74				15.69				
OND		E VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT														
		ort/Loop Combination Rates	1														1
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			23.75										
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			30.20										
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			35.52										
	UNE L	oop Rates															
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	16.68										
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	23.13										ļ
	LINES	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	28.46									1	
-	UNE P	exchange Ports - 2-Wire DID Port	<u> </u>	<u> </u>	UEPPX	UEPD1	7.06	225.55	87.21	113.08	14.38			15.69		-	
	NONE	ECURRING CHARGES - CURRENTLY COMBINED			UEPPX	UEPDI	7.06	225.55	87.21	113.08	14.38			15.69			-
	NONK	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -															
		Switch-as-is			UEPPX	USAC1		7.32	1.87					15.69			
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes			UEPPX	USA1C		7.32	1.87					15.69			
	ADDIT	IONAL NRCs	ļ	<u> </u>	LIEDDY	110404		00.01						45.00			<u> </u>
<u> </u>	Tolor	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk	 	-	UEPPX	USAS1		26.84		ļ —				15.69	 	1	
-	relepr	none Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port)		-	UEPPX	NDT	0.00	0.00	0.00	+				15.69	-		
		DID Numbers, Establish Trunk Group and Provide First Group															
	_	of 20 DID Numbers	<u> </u>	<u> </u>	UEPPX	NDZ	0.00	0.00	0.00	ļ				15.69	ļ	-	
		Additional DID Numbers for each Group of 20 DID Numbers	l	1	UEPPX	ND4	0.00	0.00	0.00					15.69	 	 	
	-	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX UEPPX	ND5 ND6	0.00	0.00	0.00	 				15.69 15.69		 	
	-	Reserve Non-Consecutive DID numbers Reserve DID Numbers		 	UEPPX	NDV	0.00	0.00	0.00	1		1	1	15.69	1	 	
	LOCAL	L NUMBER PORTABILITY	 		OLI I A	INDV	0.00	0.00	0.00	1				15.09	1	t	
—		Local Number Portability (1 per port)	-	!	UEPPX	LNPCP	3.15	0.00	0.00	 			ł – – – –		 	t	

UNBUND	LED NETWORK ELEMENTS - South Carolina			1			1					1 -			ment: 2		bit: C
CATEGORY	r RATE ELEMENTS	Interi m	Zone	В	cs	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							B	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	l.	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-W	IRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL L	INE SIDE	PORT														
UNE	E Port/Loop Combination Rates																
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB	UEPPR		30.86										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		38.60										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3		3	UEPPB	UEPPR		44.23										
UNI	E Loop Rates		_	LIEDDD	HEDDD	1101.01/	04.00							45.00			
	2-Wire ISDN Digital Grade Loop - UNE Zone 1	1	1	UEPPB	UEPPR	USL2X	21.90			 				15.69	 	 	1
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	29.64							15.69			
1 15 17	2-Wire ISDN Digital Grade Loop - UNE Zone 3	1	3	UEPPB	UEPPR	USL2X	35.27			 				15.69	 	 	
UNI	E Port Rate			LIEDDD	LIEDDD	LIEDDD	0.00	400.54	100.11	400.05	04.07			45.00			
No	Exchange Port - 2-Wire ISDN Line Side Port	+	1	UEPPB	UEPPR	UEPPB	8.96	190.51	133.14	100.95	21.37			15.69	 	 	
NOI	NRECURRING CHARGES - CURRENTLY COMBINED	+	1			1				 					 	 	
ADI	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion			UEPPB	UEPPR	USACB	0.00	38.59	27.08					15.69			
	CAL NUMBER PORTABILITY		<u> </u>														
LOC				UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								1
Б.С	Local Number Portability (1 per port)		<u> </u>	UEPPB	UEPPR	LINPUX	0.35	0.00	0.00								
B-C	HANNEL USER PROFILE ACCESS:			LIEDDD	HEDDD	LIALICA	0.00	0.00	0.00								
	CVS/CSD (DMS/5ESS)		-	UEPPB UEPPB	UEPPR UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00	-							
D 0	CSD	O MC O	TNI	UEPPB	UEPPR	U1UCC	0.00	0.00	0.00	-							
В-С	HANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S CVS/CSD (DMS/5ESS)	L,IVIO, 6	i IN)	UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								1
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								1
	CSD CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								1
Her	ER TERMINAL PROFILE			OLFFB	ULFFR	01001	0.00	0.00	0.00	-					-	-	-
031	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VE	RTICAL FEATURES		1	OLFFB	ULFFR	UTUIVIA	0.00	0.00	0.00								
VLI	All Vertical Features - One per Channel B User Profile	+	1	UEPPB	UEPPR	UEPVF	3.04	0.00	0.00					15.69			
INT	EROFFICE CHANNEL MILEAGE			OLITD	OLITIK	OLI VI	3.04	0.00	0.00					13.03			
	Interoffice Channel mileage each, including first mile and facilities termination			LIEDDR	UEPPR	M1GNC	24.30	40.63	27.47	16.77	6.91			15.69			
	Interoffice Channel mileage each, additional mile					M1GNM	0.0167	0.00	0.00	10.77	0.01			10.00	-		
4-W	IRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUN	K PORT		OLI I D	OLITIK	WITCHWI	0.0107	0.00	0.00								+
	E Port/Loop Combination Rates		1														
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP			176.82										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2		2	UEPPP			241.38										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1															
LIAIT	Zone 3 E Loop Rates	+	3	UEPPP		1	347.84			 					-	-	
UNI	4-Wire DS1 Digital Loop - UNE Zone 1	1	1	UEPPP		USL4P	90.87			 		1		15.69	1	1	+
	4-Wire DS1 Digital Loop - ONE Zone 1	1	2	UEPPP		USL4P	155.43			 		1		15.69	1	1	+
	4-Wire DS1 Digital Loop - ONE Zone 2 4-Wire DS1 Digital Loop - UNE Zone 3	1	3	UEPPP		USL4P	261.89			 		1		15.69	 	 	
LIME	E Port Rate	 	۲	JE: ! !		JULTI	201.09			†				15.05	t	 	
ONL	Exchange Ports - 4-Wire ISDN DS1 Port	1	1	UEPPP		UEPPP	85.95	457.30	259.67	124.15	31.83			15.69	-	 	
NOI	NRECURRING CHARGES - CURRENTLY COMBINED	1		32.11		1	00.00	107.00	200.01	12-7.10	01.00			10.00	t	t	†
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port	1		1		1									I	I	†
	Combination - Conversion -Switch-as-is	1	1	UEPPP		USACP	0.00	119.34	78.73]				15.69	I	I	
ADI	DITIONAL NRCs	1				1				† 1					1	1	
1.01	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-	1				1				† †					1	t	1
	Inward/two way Tel Nos. (except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -			UEPPP		PR7TF		0.49	0.49					15.69			
	Outward Tel Numbers (All States except NC)	1	1	UEPPP		PR7TO		11.54	11.54					15.69			

ONBONDE	D NETWORK ELEMENTS - South Carolina			1										ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
															D130 131	Disc Add I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -				+		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Subsequent Inward Tel Numbers			UEPPP	PR7ZT		23.07	23.07					15.69			
LOCA	L NUMBER PORTABILITY			UEPPP	PRIZI		23.07	23.07					15.69			
LOCA	Local Number Portability (1 per port)			UEPPP	LNPCN	1.75					1					
	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								+
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								1
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New o	r Additional "B" Channel				1										1	
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.56						15.69			
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.56						15.69			1
j	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.56						15.69			
CALL	TYPES															
	Inward			UEPPP	PR7C1	0.00	0.00	0.00								
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Intero	ffice Channel Mileage															
	Fixed Each Including First Mile			UEPPP	1LN1A	77.4815	89.47	81.99	16.39	14.48			15.69	ļ		
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3415										1
	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE P	ort/Loop Combination Rates		<u> </u>													
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		149.77										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		2	UEPDC UEPDC		214.33										
LINE	oop Rates		3	UEPDC	+	320.78										
UNE L	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	90.87							15.69			
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	155.43					1		15.69			-
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	261.89							15.69			+
UNF P	Port Rate		3	OLI DO	OOLDO	201.03							15.05			-
ONE !	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	58.90	455.50	253.79	117.55	14.20			15.69			+
NONR	ECURRING CHARGES - CURRENTLY COMBINED															
1.0	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Switch-as-is			UEPDC	USAC4		129.78	67.17					15.69			
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with DS1 Changes			UEPDC	USAWA		129.78	67.17					15.69			
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with Change - Trunk			UEPDC	USAWB		129.78	67.17					15.69			
ADDIT	TONAL NRCs															
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		14.51	14.51					15.69	ļ		ļ
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel													1	I	
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		14.51	14.51	ļ				15.69	ļ	-	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			LIEBBO	LIDTES		44.51	44.51					45.00	1	I	
	Activation Per Chan - Inward Trunk with DID 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			UEPDC	UDTTD		14.51	14.51	ļ .				15.69	1	!	
				LIEBDO	INDEAC.		44.54	44.54					45.00	1	I	
DIBO	Activation / Chan - 2-Way DID w User Trans AR 8 ZERO SUBSTITUTION	-	-	UEPDC	UDTTE		14.51	14.51	 				15.69	-		
DIFUL	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	605.00	1				15.69	1	t	\vdash
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	605.00	1				15.69	 	 	
Altern	ate Mark Inversion			02. 00	3001		0.00	555.00	 				10.00		-	†
70111	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00							<u> </u>	
1	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00						1	1	
Telepi	none Number/Trunk Group Establisment Charges				1		2.00	2.00						İ	İ	
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00							15.69	İ	İ	
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00			1				15.69		1	1
1	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00			i i				15.69			
	DID Numbers, Establish Trunk Group and Provide First Group				i i											
	of 20 DID Numbers	<u></u>		UEPDC	NDZ	0.00	0.00	0.00	<u> </u>		<u></u>	<u> </u>	15.69	<u> </u>	<u> </u>	<u></u>
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00							15.69			
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00	0.00	0.00	ĺ				15.69			

NDUNDLE	ED NETWORK ELEMENTS - South Carolina												Attachi	ment: 2	Exhi	bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)	N	P		Submitted	Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual S Order v Electron Disc Ad
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00					15.69			
	Reserve DID Numbers	<u></u>		UEPDC	NDV	0.00	0.00	0.00					15.69			ļ
Dedic	ated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	Digital	Loop	with 4-Wire DDITS	Trunk Port											ļ
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	77.14	89.47	81.99	16.39	14.48			15.69			<u> </u>
_	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.3415	0.00	0.00								<u> </u>
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities			LIEDDO	41.1100	0.00	0.00	0.00								
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25	1	1	LIEBBO	AL NICE	0.0445	0.00	0.00								1
+	miles	!	 	UEPDC	1LNOB	0.3415	0.00	0.00					-	-	-	\vdash
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities			LIEDDO	41.1100	0.00	0.00	0.00								
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
_	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC LNPCP	0.3415	0.00	0.00								
_	Local Number Portability, per DS0 Activated			UEPDC		3.15	0.00	0.00								
4 14/15	Central Office Termininating Point			UEPDC	CTG	0.00										
	E DS1 LOOP WITH CHANNELIZATION WITH PORT															
	m is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti															
	System can have up to 24 combinations of rates depending on	type an	d num	ber of ports used												
UNE L	OS1 Loop			LIEDMO	1101.00	00.07	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	90.87	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	155.43	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3	Ц.	3	UEPMG	USLDC	261.89	0.00	0.00								
UNE L	OSO Channelization Capacities (D4 Channel Bank Configuration	ns)											4= 00			
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	82.78	0.00	0.00					15.69			
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	165.56	0.00	0.00					15.69			
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	331.12	0.00	0.00					15.69			
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	496.68	0.00	0.00					15.69			
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	662.24	0.00	0.00					15.69			
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	827.80	0.00	0.00					15.69			
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	993.36	0.00	0.00					15.69			
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,324.48	0.00	0.00					15.69			
_	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	1,655.60	0.00	0.00					15.69			
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	1,986.72	0.00	0.00					15.69			
 _	672 DS0 Channel Capacity - 1 per 28 DS1s		L	UEPMG	VUM67	2,317.84	0.00	0.00					15.69			
	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with						stem									
	imum System configuration is One (1) DS1, One (1) D4 Channe															├
wuitip	oles of this configuration functioning as one are considered Ad	iu i artei	i ine m	iiiiiiium system cor	inguration is	countea.										
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes	1	1	UEPMG	USAC4	0.00	150.81	8.38					15.69			1
Custa	m Additions at End User Locations Where 4-Wire DS1 Loop wit	th Chan						8.38					15.69			
	m Additions at End User Locations where 4-wire DS1 Loop wit Not Currently Combined) in all states, except in Density Zone 1				IIII CUFFE	iliy Exists and										⊢—
ivew (1 DS1/D4 Channel Bank - Additionally Add NRC for each Port	от гор	O IVI 3F	l a	+	 							-	-	-	\vdash
	and Assoc Fea Activation	1	1	UEPMG	VUMD4	0.00	717.71	425.81	149.08	17.69			15.69			1
Pinol	ar 8 Zero Substitution			ULFIVIG	VOIVID4	0.00	717.71	423.01	145.00	17.05			13.09			
Pipole	Clear Channel Capability Format, superframe - Subsequent	 	 		 	+										$\vdash \!$
	Activity Only	1	1	UEPMG	CCOSF	0.00	0.00	605.00								1
-	Clear Channel Capability Format - Extended Superframe -	 	 	021 IVIO	30001	0.00	0.00	303.00								
	Subsequent Activity Only	1	1	UEPMG	CCOEF	0.00	0.00	605.00								1
Altern	ate Mark Inversion (AMI)	1	 	021 IVIO	30021	0.00	0.00	303.00	-							\vdash
Altern	Superframe Format	1	 	UEPMG	MCOSF	0.00	0.00	0.00	-							\vdash
+	Extended Superframe Format	 	 	UEPMG	MCOPO	0.00	0.00	0.00								$\vdash \!$
Excha	inge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port			0.00	0.00	0.00								\vdash
	inge Ports				1											\vdash
LAUITA		1	 		1											\vdash
	Line Side Combination Channelized PBX Trunk Port - Business	1	l	UEPPX	UEPCX	1.13	0.00	0.00	0.00	0.00	I	I	15.69			1
	Line Side Outward Channelized PBX Trunk Port - Business	—	-	UEPPX	UEPOX	1.13	0.00	0.00	0.00	0.00			15.69	 	 	\leftarrow

	ED NETWORK ELEMENTS - South Carolina				1	1					I 0 C .	06	Attachr		Exhib	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						_	Nonred	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.13	0.00	0.00	0.00	0.00			15.69			
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	7.09	0.00	0.00	0.00	0.00			15.69			
Featu	re Activations - Unbundled Loop Concentration															
	Feature (Service) Activation for each Line Side Port Terminated															
	in D4 Bank			UEPPX	1PQWM	0.56	25.45	13.44	4.20	4.17			15.69			
	Feature (Service) Activation for each Trunk Side Port Terminated			UEPPX	1PQWU	0.56	78.31	18.46	59.37	11.60			15.69			
Toloni	in D4 Bank hone Number/ Group Establishment Charges for DID Service			UEPPX	IPQWU	0.56	78.31	18.46	59.37	11.60			15.69			
relepi	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)	 		UEPPX	NDZ	0.00	0.00	0.00							 	
 	DID Numbers - groups of 20 - Valid all States	1		UEPPX	ND4	0.00	0.00	0.00								
 	Non-Consecutive DID Numbers - per number	1		UEPPX	ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Numbers	1		UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
Local	Number Portability															
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	URES - Vertical and Optional															
Local	Switching Features Offered with Line Side Ports Only															
	All Features Available			UEPPX	UEPVF	3.04	0.00	0.00					15.69			
	PORT LOOP COMBINATIONS - MARKET RATES															
	et Rates shall apply where BellSouth is not required to provide	unbund	lled lo	cal switching or swi	itch ports per	FCC and/or St	ate Commission	n rules.								
	ncludes:	<u></u>		<u> </u>	1				<u> </u>							
	ndled port/loop combinations that are Currently Combined or N											,				
I he I	op 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderda outh currently is developing the billing capability to mechanica	ale, Mia	mı); G	A (Atlanta); LA (New	Orleans); NO	Greensboro-	Winston Salem	-Highpoint/Ch	ariotte-Gaston	nat ourrently	N (Nashvill	e).	In the interi	m where Bell	Couth connet	hill Markat
	outh currently is developing the billing capability to mechanically a section preceded.								ig charges for	not currently t	ombined in	FL and NC.	. In the interi	ili wilere beli	South Cannot	DIII Warket
	farket Rate for unbundled ports includes all available features i			life Warket Nates at	T reserves tr	le right to true-	up the billing t	illerence.								
	Office and Tandem Switching Usage and Common Transport Us			he Bert coction of th	in make evolution	t aball anabets					or UNF Coi	- Dawll				
	C: URECU).	ago .u.					all combination	ons of loon/po	rt network eler	nents except			Combination	s which have	e a flat rate us	age charge
				ne Fort Section of th	iis rate exnib	it snaii appiy to	all combination	ons of loop/po	rt network eler	nents except		n Port/Loop	Combination	s which have	e a flat rate us	age charge
		listed	in the l													
	ot Currently Combined scenarios the Nonrecurring charges are	listed i	in the I													
Additi	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly.	listed	in the I													
Additi 2-WIR	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	listed	in the I													
Additi 2-WIR	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly.	listed i	in the I													
Additi 2-WIR	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. EE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates	listed i				s for each Port										
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	listed	1			s for each Port										
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates	listed	1 2 3	First and Additional	NRC column	27.76 34.38 40.04										
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates [2-Wire Voice Grade Loop (SL1) - Zone 1	listed	1 2 3	First and Additional	NRC column	27.76 34.38 40.04										
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2	listed i	1 2 3	First and Additional UEPRX UEPRX	NRC column	27.76 34.38 40.04 13.76 20.38										
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2	listed	1 2 3	First and Additional	NRC column	27.76 34.38 40.04										
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res)	listed	1 2 3	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX	27.76 34.38 40.04 13.76 20.38 26.04	USOC. For Co	urrently Comb				s are listed				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence	listed i	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX	27.76 34.38 40.04 13.76 20.38 26.04	90.00	urrently Comb				s are listed i				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res	listed i	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL	27.76 34.38 40.04 13.76 20.38 26.04 14.00	90.00 90.00	90.00 90.00				15.69 15.69				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res	listed i	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX	27.76 34.38 40.04 13.76 20.38 26.04	90.00	urrently Comb				s are listed i				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled pers, low usage line port with Caller ID	listed i	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL UEPRC UEPRC	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00	90.00 90.00 90.00	90.00 90.00 90.00				15.69 15.69				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID (LUM) 2-Wire voice unbundled Low Usage Line Port without Caller ID	listed	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAP	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00	90.00 90.00 90.00	90.00 90.00 90.00 90.00				15.69 15.69				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire VGico Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice Unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Low Usage Line Port without Caller ID (LUM) 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability	listed i	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL UEPRC UEPRC	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00	90.00 90.00 90.00	90.00 90.00 90.00				15.69 15.69				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Low Usage Line Port without Caller ID (LUM) 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability 2-Wire Voice Unbundled South Carolina Residence Dialing Plan	listed	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRC UEPRO UEPAP	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00				15.69 15.69				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire VGico Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice Unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Low Usage Line Port without Caller ID (LUM) 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability	listed	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAP	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00	90.00 90.00 90.00	90.00 90.00 90.00 90.00				15.69 15.69				
Additi 2-WIR UNE F	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability 2-Wire Voice Unbundled South Carolina Residence Dialing Plan without Caller ID	listed	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRC UEPRO UEPAP	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00				15.69 15.69				
Additi 2-WiR UNE P UNE L	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability 2-Wire Voice Unbundled South Carolina Residence Dialing Plan without Caller ID Capability L NUMBER PORTABILITY	listed	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP UEPAP UEPAP UEPRT UEPWL	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00				15.69 15.69 15.69				
Additi 2-WIR UNE F UNE L 2-Wire	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID res 2-Wire voice unbundled port with Caller ID res 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability 2-Wire Voice Unbundled South Carolina Residence Dialing Plan without Caller ID Capability L NUMBER PORTABILITY L LOCAL Number Portability (1 per port)	listed i	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP UEPAP UEPRT	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00				15.69 15.69 15.69				
Additi 2-WIR UNE F UNE L 2-Wire	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Dort with Caller ID res 2-Wire voice unbundled Low Usage Line Port without Caller ID (LUM) 2-Wire voice unbundled South Carolina Residence Dialing Plan without Caller ID Capability 2-Wire voice unbundled South Carolina Area Calling Port without Caller ID Capability LI NUMBER PORTABILITY Local Number Portability (1 per port) URES	listed	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP UEPRT UEPWL UEPRS	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00 14.00 14.00 14.00 14.00 0.35	90.00 90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00 90.00				15.69 15.69 15.69 15.69 15.69				
Additi 2-WIR UNE F UNE L 2-Wire	ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID res 2-Wire voice unbundled port with Caller ID res 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability 2-Wire Voice Unbundled South Carolina Residence Dialing Plan without Caller ID Capability L NUMBER PORTABILITY L LOCAL Number Portability (1 per port)	listed i	1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP UEPAP UEPAP UEPRT UEPWL	27.76 34.38 40.04 13.76 20.38 26.04 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00				15.69 15.69 15.69				

Version 3Q02: 09/06/02 Page 296 of 362

ONROND	DLED NETWORK ELEMENTS - South Carolina			1							1 -			ment: 2		bit: C
CATEGOR	RY RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	Charge - Manual Sv Order vs. Electronic
													1st	Add'l	Disc 1st	Disc Add'
						Rec	Nonrec		Nonrecurring					Rates(\$)	•	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -															
	Subsequent			UEPRX	USAS2		0.00	0.00				15.69				
	WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UN	NE Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		1 2			27.76 34.38			-						-	
	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3		3			40.04										
LIN	NE Loop Rates	-	3			40.04										
OIV	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	20.38										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	26.04										
2-V	Wire Voice Grade Line Port (Bus)	1	_ ّ		J. 2.	20.04			†					1	1	
	2-Wire voice unbundled port without Caller ID - bus	1		UEPBX	UEPBL	14.00	90.00	90.00	†			15.69		1	1	
	2-Wire voice unbundled port with Caller + E484 ID - bus	l		UEPBX	UEPBC	14.00	90.00	90.00	† †			15.69		1	1	
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	14.00	90.00	90.00	†			15.69				
	2-Wire voice Grade unbundled South Carolina extended local dialing parity port with Caller ID - bus			UEPBX	UEPAZ	14.00	90.00	90.00				15.69				
	2-Wire voice unbundled South Carolina Bus Area Calling Port with Caller ID (LMB)			UEPBX	UEPAB	14.00	90.00	90.00				15.69				
	2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPBX	UEPBE	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled South Carolina Business Dialing Plan without Caller ID			UEPBX	UEPWM	14.00	90.00	90.00				15.69				
	2-Wire voice unbundled South Carolina Business Area Calling															
LO	Port without Caller ID Capability CAL NUMBER PORTABILITY			UEPBX	UEPBB	14.00	90.00	90.00				15.69				
	Local Number Portability (1 per port)	<u> </u>	 	UEPBX	LNPCX	0.35			ļ							<u> </u>
FE	EATURES All Features Offered	 	-	UEPBX	UEPVF	0.00	0.00	0.00	 			15.69		 	 	
AD	DDITIONAL NRCs	 	 	ULPDA	UEFVF	0.00	0.00	0.00	 			15.09		-		1
AD	NRC - 2-Wire Voice Grade Loop/Line Port Combination -	1	 		+				 		1			1	 	1
	Subsequent	l	l	UEPBX	USAS2		0.00	0.00				15.69			1	
2-W	WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)	 	 	OLI DA	00/102		0.00	0.00				13.08		1	t	1
	NE Port/Loop Combination Rates				+											
011	2-Wire VG Loop/Port Combo - Zone 1		1			27.76			†						1	
	2-Wire VG Loop/Port Combo - Zone 2		2			34.38			1					Ì	1	
	2-Wire VG Loop/Port Combo - Zone 3		3			40.04										
UN	NE Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRG	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRG	UEPLX	20.38				_						
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRG	UEPLX	26.04										
2-V	Wire Voice Grade Line Port Rates (RES - PBX)						, and the second									
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	14.00	90.00	90.00				15.69				
LO	OCAL NUMBER PORTABILITY															<u> </u>
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00								
FE	EATURES Of the second s	ļ		LIEBBO	1				ļ			15.5		ļ	ļ	ļ
	All Features Offered	ļ		UEPRG	UEPVF	0.00	0.00	0.00	ļ			15.69			-	
	ONRECURRING CHARGES - CURRENTLY COMBINED	ļ			1										1	
AD	DDITIONAL NRCs	 	-		+				 					 	 	
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring						0.00	0.00				15.69				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64				15.69				
	WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
UN	NE Port/Loop Combination Rates															<u> </u>
1	2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2	ļ	1			27.76			ļ					ļ	1	
			2	I .		34.38									i	

JNBUNDL	ED NETWORK ELEMENTS - South Carolina			,								,		ment: 2		bit: C
		Interi									Svc Order Submitted Elec	Svc Order Submitted Manually	Incremental Charge - Manual Svc	Incremental Charge - Manual Svc	Incremental Charge - Manual Svc	Increment Charge - Manual S
ATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Order vs. Electronic- Disc 1st	Order vs. Electronic Disc Add
						B	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	l.	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	20.38										1
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	26.04										1
2-Wir	e Voice Grade Line Port Rates (BUS - PBX)															
	, ,															1
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00				15.69				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00				15.69				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	14.00	90.00	90.00	-			15.69				†
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port		-	UEPPX	UEPXD	14.00	90.00	90.00				15.69				+
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			OLITA	OLI AD	14.00	30.00	30.00				13.03				
	Capable Port			UEPPX	UEPXE	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			ULFFX	ULFAL	14.00	90.00	90.00				13.09				
	Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy		-	UEFFA	UEFAL	14.00	90.00	90.00	-			15.69				
				HEDDY	LIEDVA	44.00	00.00	00.00				45.00				
	Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00				15.69				
LOCA	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								1
FEAT	URES															
	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00				15.69				
	RECURRING CHARGES - CURRENTLY COMBINED															
ADDI	TIONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2		0.00	0.00				15.69				
	2 Wire Loop/Line Side Port Combination - Non feature -															
	Subsequent Activity- Nonrecurring						0.00	0.00				15.69				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
	Group						7.34	7.34				15.69				
2-WIF	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POP	₹T														
UNE	Port/Loop Combination Rates															
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			27.76										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			34.38										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			40.04										
UNE	Loop Rates															1
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	13.76										1
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	20.38										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	26.04										
2-Wir	e Voice Grade Line Port Rates (Coin)															
	2-Wire Coin 2-Way without Operator Screening and without															
	Blocking (SC)			UEPCO	UEPSD	14.00	90.00	90.00				15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,			02. 00	02.05	11.00	00.00	00.00	t			10.00				+
	900/976, 1+DDD (AL, KY, LA, MS, SC)			UEPCO	UEPRA	14.00	90.00	90.00				15.69				
_	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,				JE. 701	14.00	55.55	55.56	1		ł – – – –	10.00		 	t	
	900/976, 1+DDD (SC)			UEPCO	UEPSA	14.00	90.00	90.00			I	15.69		Ì	I	
-	2-Wire Coin 2-Way with Operator Screening and 011 Blocking			021.00	ULFUA	14.00	90.00	90.00	 		1	15.09		1	 	
	(SC)			UEPCO	UEPSH	44.00	90.00	90.00				15.60			1	
_				UEPCU	UEPSH	14.00	90.00	90.00	<u> </u>		1	15.69		 	 	
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking;			LIEBCO	LIEBOO	44.00	00.00	20.00				45.00		l	I	
	with Dialing Parity (SC)	-		UEPCO	UEPSC	14.00	90.00	90.00				15.69		1	 	
	2-Wire Coin 2-Way with Operator Screening and Blocking:			LIEBCO	LIEBOO	44.00	00.00	20.00				45.00		l	I	
	900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCC	14.00	90.00	90.00	ļ			15.69				₽
	2-Wire Coin 2-W Oper Screen & Blocking: 900/976, 1+DDD,		1	l	1						I	I		1	1	
	011+ & Local; Enhanced Calling OPT 3YV (SC)	1	1	UEPCO	UEPCE	14.00	90.00	90.00	<u> </u>		<u> </u>	15.69	<u></u>	<u> </u>	<u> </u>	1

UNBU	NDLE	D NETWORK ELEMENTS - South Carolina			1	1	,					1 -	T -		ment: 2		bit: C
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	Charge - Manual Sv Order vs. Electronic
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec		Nonrecurring					Rates(\$)		
		0.W. O. O. O. W. O. O. O. O. O. O. O. O. O. O. O. O. O.					1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Coin 2-W Oper Screen & Block: 900/976, 1+DDD, 011+, & Local; Enhanced Calling OPT AP7 (SC)			UEPCO	UEPCF	14.00	90.00	90.00				15.69				
		2-Wire Coin Outward without Blocking and without Operator		1	UEPCO	UEPCF	14.00	90.00	90.00	-			15.69			-	
		Screening (SC)			UEPCO	UEPSG	14.00	90.00	90.00				15.69				
		2-Wire Coin Outward with Operator Screening and 011 Blocking				1										İ	
		(SC)			UEPCO	UEPSF	14.00	90.00	90.00				15.69				
		2-Wire Coin Outward with Operator Screening and Blocking:															
		011, 900/976, 1+DDD (SC)			UEPCO	UEPSJ	14.00	90.00	90.00				15.69				
		2-Wire Coin Outward with Operator Screening and Blocking:			LIEDOO	LIEDOM	44.00	00.00	20.00				45.00				
		900/976, 1+DDD, 011+, and Local (SC) 2-Wire Coin Out Oper Screen & Block: 900/976, 1+DDD, 011+,			UEPCO	UEPCM	14.00	90.00	90.00				15.69				
		& Local; w/ Enhanced Call OPT 3YW (SC)			UEPCO	UEPCP	14.00	90.00	90.00				15.69			1	
	LOCAL	NUMBER PORTABILITY				32. 31	14.00	55.56	30.30	† †			10.00				
		Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
	ADDIT	ONAL NRCs															
		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPCO	USAS2		0.00	0.00				15.69				
UNBUN		ORT/LOOP COMBINATIONS - MARKET BASED RATES VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	DODT														
-	Z-VVIRE	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1	PURI	1		-	73.68			-							-
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2		-	80.13			-							
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3		1	85.46									1	
	UNE L	pop Rates															
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	16.68										
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	23.13										
	LINE D	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	28.46										
		ort Rate Exchange Ports - 2-Wire DID Port			UEPPX	UEPD1	57.00	600.00	75.00				15.69				
		ECURRING CHARGES - CURRENTLY COMBINED			OLFFX	OLFDI	37.00	000.00	75.00				13.03			1	
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -															
		Switch-As-Is Top 8 MSAs only			UEPPX	USAC1		125.00	75.00				15.69				
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion															1
		with BellSouth Allowable Changes Top 8 MSAs only			UEPPX	USA1C		125.00	75.00				15.69				1
	ADDIT	ONAL NRCs			UEDDV			=====					1= 00				
	Tolonh	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk one Number/Trunk Group Establisment Charges			UEPPX	USAS1		53.68					15.69			-	
	reiepn	DID Trunk Termination (One Per Port)			UEPPX	NDT	0.00	0.00	0.00								
		DID Numbers, Establish Trunk Group and Provide First Group			OLITA	INDI	0.00	0.00	0.00	-							
		of 20 DID Numbers			UEPPX	NDZ	0.00	0.00	0.00								
		Additional DID Numbers for each Group of 20 DID Numbers			UEPPX	ND4	0.00	0.00	0.00								
		DID Numbers, Non- consecutive DID Numbers, Per Number			UEPPX	ND5	0.00	0.00	0.00								
		Reserve Non-Consecutive DID numbers			UEPPX	ND6	0.00	0.00	0.00								
	LOCAL	Reserve DID Numbers NUMBER PORTABILITY			UEPPX	NDV	0.00	0.00	0.00								
	LUCAL	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								-
	2-WIRE	EISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LII	NF SIDE	F POR		LINE CE	3.13	0.00	0.00	-							
		ort/Loop Combination Rates					1								Ì	1	
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -													1		
		UNE Zone 1		1	UEPPB UEPPR		76.90										1
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		1 -						Ι Τ							
		UNE Zone 2		2	UEPPB UEPPR	-	84.64			1						1	
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3		3	UEPPB UEPPR		90.27										
	LINE 1	pop Rates		3	UEPPB UEPPR	-	90.27			1				-		-	
	SINE E	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB UEPPR	USL2X	21.90			+ +						 	
				† <u> </u>	5 5 5E/11K		200								İ	1	
		2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB UEPPR	USL2X	29.64										
		2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB UEPPR	USL2X	35.27										
	UNE P	ort Rate															

ONBONDE	LED NETWORK ELEMENTS - South Carolina	_					1								ment: 2		bit: C
CATEGORY	(RATE ELEMENTS	Interi m	Zone	E	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
								Nonrec	urring	Nonrecurring	n Disconnect			220	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Port - 2-Wire ISDN Line Side Port		1	LIEPPR	UEPPR	UEPPB	55.00	525.00	400.00	Tilot	Auu	JOHLC	15.69	JOINAIN	JOHAN	JOHIAN	JOINAIN
NON	NRECURRING CHARGES - CURRENTLY COMBINED			OLITE	OLITIK	OLI I D	00.00	020.00	400.00				10.00				1
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion - Top 8 MSAs only			UEPPB	UEPPR	USACB	0.00	225.00	225.00				15.69				
ADD	DITIONAL NRCs																
LOC	CAL NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								1
B-Cl	HANNEL USER PROFILE ACCESS:																
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								ĺ
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
B-Cl	HANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS	SC,MS, 8	k TN)														
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								
USE	ER TERMINAL PROFILE																
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VER	RTICAL FEATURES																
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	3.04	0.00	0.00								ĺ
INTE	EROFFICE CHANNEL MILEAGE																1
	Interoffice Channel mileage each, including first mile and facilities termination			UEPPB	UEPPR	M1GNC	24.30	60.00	40.00	25.00	10.00		15.69				
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.0167	0.00	0.00								
4-WI	IRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRU	NK PORT															1
UNE	E Port/Loop Combination Rates																1
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP			940.87										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2		2	UEPPP			1,005.43										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		3	UEPPP			1,111.89										
UNE	E Loop Rates		Ŭ	OLITI			1,111.00					1					
ONE	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	90.87						15.69				+
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	155.43						15.69				+
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	261.89						15.69				+
UNE	E Port Rate		Ŭ	OLITI		OOL-11	201.00					1	10.00				+
0.12	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	850.00	1,150.00	1,150.00				15.69				
NON	NRECURRING CHARGES - CURRENTLY COMBINED		1	02		02	000.00	1,100.00	1,100.00				10.00				1
1	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port	1	1	t		1				1					1	t	
	Combination - Conversion -Switch-As-Is Top 8 MSAs only			UEPPP		USACP	0.00	950.00	950.00]			15.69		l	I	
ADD	DITIONAL NRCs			1		1		322.20		İ	İ				İ	İ	1
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-																
	Inward/two way Telephone Numbers (except NC)			UEPPP		PR7TF		0.9822					15.69			1	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -																
	Outward Tel Numbers (All States except NC)			UEPPP		PR7TO		23.02	23.02				15.69			1	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -																
	Subsequent Inward Telephone Numbers			UEPPP		PR7ZT		46.05	46.05				15.69				
LOC	CAL NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPP		LNPCN	1.75										
INTE	ERFACE (Provsioning Only)																
	Voice/Data			UEPPP		PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP		PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP		PR71E	0.00	0.00	0.00								
New	v or Additional "B" Channel																
	New or Additional - Voice/Data B Channel			UEPPP		PR7BV	0.00	40.00									
	New or Additional - Digital Data B Channel			UEPPP		PR7BF	0.00	40.00									
	New or Additional Inward Data B Channel			UEPPP		PR7BD	0.00	40.00									
CAL	LL TYPES																
	Inward			UEPPP		PR7C1	0.00	0.00	0.00								

INRONDLE	D NETWORK ELEMENTS - South Carolina													ment: 2		oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
1						_ 1	Nonrec	urrina	Nonrecurring	Disconnect				Rates(\$)	2.00 .01	2.007.001
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Interof	fice Channel Mileage															
	Fixed Each Including First Mile			UEPPP	1LN1A	77.4815	89.47	81.99	16.39	14.48		15.69				
4 14/101	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3415	-		-							
	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT ort/Loop Combination Rates						-								-	
ONLF	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		840.87										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		905.43										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		1,011.89										
UNE L	oop Rates					, , , , ,										
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	90.87										
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	155.43										
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	261.89									ļ	
UNE P	ort Rate															
NOND	4-Wire DDITS Digital Trunk Port ECURRING CHARGES - CURRENTLY COMBINED			UEPDC	UDD1T	750.00	1,005.07	478.99	213.53	20.94		15.69				
NONK	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Switch-As-Is Top 8 MSAs only			UEPDC	USAC4		259.56	134.33				15.69				
	- GWILCH-A3-13 TOP O WIGAS ONLY			OLI DO	00/104		255.50	104.00				13.03				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with DS1 Changes Top 8 MSAs only			UEPDC	USAWA		259.56	134.33				15.69				
	, and the second second															
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with Change - Trunk Top 8 MSAs only			UEPDC	USAWB		259.56	134.33				15.69				
ADDIT	IONAL NRCs															
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -															
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		29.01	29.01				15.69				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		29.01	29.01				15.69				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel			UEPDC	UDITB		29.01	29.01				15.69				
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		29.01	29.01				15.69				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			OLI DO	OBITO		20.01	20.01				10.00				
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		29.01	29.01				15.69				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		29.01	29.01				15.69				
BIPOL	AR 8 ZERO SUBSTITUTION															
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	605.00								
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	605.00								
Alterna	ate Mark Inversion			UEPDC	MCOSF		0.00	0.00	-							
	AMI - Superframe Format AMI - Extended SuperFrame Format			UEPDC	MCOSF		0.00	0.00								
Telenh	none Number/Trunk Group Establisment Charges			OLFDC	WCOFO		0.00	0.00								
Телері	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00						15.69				
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00	İ					15.69			1	
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00						15.69				
	DID Numbers, Establish Trunk Group and Provide First Group						Ì									
	of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00				15.69				
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00						15.69			ļ	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00	0.00	0.00				15.69				
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				15.69			1	
Dodio	Reserve DID Numbers ated DS1 (Interoffice Channel Mileage) -			UEPDC	NDV	0.00	0.00	0.00	 			15.69		-	 	
	O for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port			 	+	+	+		 							-
FAIR	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities		—		+		ł								t	
	Termination)		l	UEPDC	1LNO1	77.14	89.47	81.99	16.39	14.48		15.69			1	1
	1							230		10				İ	1	
1	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles		l	UEPDC	1LNOA	0.3415	0.00	0.00			I]	1]

	D NETWORK ELEMENTS - South Carolina													ment: 2		bit: C
											Svc Order	Svc Order			Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Intori									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Sv
ATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						.,,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electronic
													1st	Add'l	Disc 1st	Disc Add'
$\overline{}$					+		Nonrec	urring	Nonrecurring	Disconnect		l .	OSS	Rates(\$)		1
			 		-	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-+-	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities						FIISL	Auu i	FIISL	Auu i	SOMEC	JOWAN	JOWAN	SOWAN	SOWAN	JOWAN
				UEPDC	1LNO2	0.00	0.00	0.00								
	Termination)		ļ	UEPDC	TLNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25															
	miles		<u> </u>	UEPDC	1LNOB	0.7598	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.7598	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								
	Central Office Termininating Point			UEPDC	CTG	0.00										
4-WIRI	E DS1 LOOP WITH CHANNELIZATION WITH PORT															
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti	vations									1				1	1
				unnd	-						1					1
	em can have various rate combinations based on type and nur	inder of	ports	uacu	+						 	1		ļ	-	
UNE D	S1 Loop		- -	LIEDIAO	1101.50	22.2-					!	1			1	!
$-\!\!+\!\!-\!\!\!-$	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	90.87	0.00	0.00			<u> </u>	ļ				<u> </u>
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	155.43	0.00	0.00			ļ	1				ļ
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	261.89	0.00	0.00								
UNE D	SO Channelization Capacities (D4 Channel Bank Configuration	าร)														
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	103.47	0.00	0.00				15.69				
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	206.94	0.00	0.00				15.69				
-+	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	413.88	0.00	0.00			1	15.69			1	1
	144 DS0 Channel Capacity - 1 per 6 DS1s		 	UEPMG	VUM14	620.82	0.00	0.00				15.69				
$-\!\!+\!\!-\!\!\!-$																
\longrightarrow	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	827.76	0.00	0.00				15.69				
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,034.70	0.00	0.00				15.69				
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,241.64	0.00	0.00				15.69				
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,655.52	0.00	0.00				15.69				
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,069.40	0.00	0.00				15.69				
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,483.28	0.00	0.00				15.69				
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,897.16	0.00	0.00				15.69				
Non-R	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	Chani	neliztio	n with Port - Conv	ersion Charge	Based on a Sv	stem									
A Mini	mum System configuration is One (1) DS1, One (1) D4 Channel	I Bank	and U	n To 24 DSO Ports	with Feature A	Activations										
	les of this configuration functioning as one are considered Ad															
with	NRC - Conversion (Currently Combined) with or without	id i aite	I the h	T System Co	illiguration is	counted.					1					1
				UEPMG	USAC4	0.00	150.81	8.38				15.69				
	BellSouth Allowed Changes - Top 8 MSAs Only	L		UEPIVIG	USAC4	0.00	150.81	8.38				15.69				
	n Additions Where Currently Combined and New (Not Currently	y Comb	ined)													
In Den	sity Zone 1 Top 8 MSAs					ļ					ļ				ļ	ļ
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc	l	1		1							I		Ì	I	1
	Fea Activation -	<u> </u>	<u>L</u>	UEPMG	VUMD4	0.00	717.71	425.81	149.08	17.69	<u> </u>	15.69		<u> </u>	<u> </u>	<u> </u>
Bipola	r 8 Zero Substitution															
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only	l	1	UEPMG	CCOSF	0.00	0.00	605.00				I		1	1	
	Clear Channel Capability Format - Extended Superframe -				1,222	2.00	2.00	222.00			İ					İ
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	605.00								
Altern	ate Mark Inversion (AMI)	-	 	OLI IVIO	OUOLI	0.00	0.00	303.00			1	 		 	-	1
Aiterna			-	LIEDMO	MOCOF	2.22	0.00	0.00			 	1		ļ	-	
$-\!\!+\!\!-\!\!\!-$	Superframe Format		1	UEPMG	MCOSF	0.00	0.00	0.00			1	1			1	1
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	nge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
Exchar	nge Ports]													
			1		1							1		<u> </u>		
	Line Side Combination Channelized PBX Trunk Port - Business	l	1	UEPPX	UEPCX	14.00	0.00	0.00	0.00	0.00		15.69		1	1	
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	14.00	0.00	0.00	0.00	0.00		15.69				
										. , ,						1
	Line Side Inward Only Channelized PBX Trunk Port without DID	l	1	UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00		15.69		Ì	I	1
-+-	2-Wire Trunk Side Unbundled Channelized DID Trunk Port		 	UEPPX	UEPDM	57.00	0.00	0.00	0.00	0.00	 	15.69		 	1	
Factor		-	 	OLITA	OLI DIVI	31.00	0.00	0.00	0.00	0.00	 	13.09		 	-	1
	e Activations - Unbundled Loop Concentration		1	-		ļ					 	 		 	 	
Featur													ì			i
Featur	Feature (Service) Activation for each Line Side Port Terminated			LIEDDY	400							4-0-				
reatur	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated			UEPPX	1PQWM	0.70	40.00	20.00	6.00	5.00		15.69				

ATEGRAY BATE BLEMENTS INTELLIGIBLES BATE BLEMENTS BATE BLEMENTS	JNDLE	D NETWORK ELEMENTS - South Carolina												Attach	ment: 2	Exhil	bit: C	
ATE REMOND BATE R													Svc Order	Svc Order				
PATE ELEMENTS Interest Part P																		
ATTECHY BATE ELEMENTS M BATE BLANKENS M BAT																		
Map	CATE	GORY	RATE FLEMENTS	Interi	Zone	BCS	USOC			RATES(\$)								
Part Part	OA. L		NATE ELEMENTO	m	20110	200	0000			ππι Ευ(ψ)			per LSR	per LSR				
Recommendation Reco																		
Tolephone Name of Concept 1 1 1 1 1 1 1 1 1															1st	Add'l	Disc 1st	Disc Add'l
Tolephone Name of Concept 1 1 1 1 1 1 1 1 1									Nonrec	urring	Nonrecurring	a Disconnect			oss	Rates(\$)		
Temperature (Companies Allemanter Charges for COD Service)								Rec				·	SOMEC	SOMAN			SOMAN	SOMAN
Did Trans Termination (1 par Pea) USPFX VICT O.03 O.01 O.01 O.02 O.03 O.03 O.03 O.03 O.04 O.05		Telenh	one Number/ Group Establishment Charges for DID Service						11100	Addi	11130	Auui	COMILO	COMPAN	COMPAR	COMPAR	COMPAN	COMPAR
Estab Fix Cop and Fixed Dis Note (ELC) (CA) (CA) (CA) (CA) (CA) (CA) (CA) (C		relepii				LIEDDY	NDT	0.00	0.00	0.00				15.60				
DEN Number's groups of 20 - Votal at Disbases DEPPK DEPPK DEPPK DEPPK DEPK																		-
Non-Corresponded Diffusiones - per nutritiers DEPPK NSS 0.00 0.0					 													-
Reserve Nave-Corresponder Diff Number																		
December DD Numbers																		
Cool Number Proteining 1 per point 1 1 1 1 1 1 1 1 1																		
Dear Number Probability - 1 per port		Local				OLITA	NDV	0.00	0.00	0.00				13.03				
FEATURES - Vertical and Optional		Local				LIEDDY	LNDCD	2 15	0.00	0.00								
		CEATI		-		ULFFX	LINE CE	3.13	0.00	0.00		-	-			-		
Main Features Admissible				1	1		+	-			-		-	-	-	-	-	
		Local		1	1	LIEDDY	HEDVE	2.04	0.00	0.00	-		-	15.60	-	-	-	
1. Coar Based Rates are applied where Belliosum is required by PCC ander State Commission rule to provide Unbounded Local State Commission For State Commission Coards and a supplied to the Standard Boyl to the Unbounded ForUse Combination. Coards and Example State Standard State Commission For State Commission For State Coards and Example State Commission For State Coards and Example State Coa	IINDII	NDI ED 4		<u> </u>	 	ULPFA	UEFVF	3.04	0.00	0.00		 		15.69		 		
2. Features shall apply to the Unburdled Port section of this Rate Exhibit. 3. Find Offices and Traden Switching Busga and Common Transport Usego areas in the Port accident of this rate existion of this rate existion of this rate existion of this rate except for UNE Com Port/Log Combinations. 4. The first and additional Port nonecuring charges apply to Not Currently Combined Combos. For Currently Combined Combos, the nonrecuring charges shall be those identified in the Nonrecuring - Currently Combined Sections. Additional NRCs may apply also and are extengiors accordingly. UNLEY CENTREX - SESS (Valid in All States) 2. Will Composition (Combos of Port (Centros) Port Combo. UNLEY CENTREX - SESS (Valid in All States) 2. Will Composition (Centros) Port (Centros) Port Combo. UNLEY COMPOSITION (Composition Centros) Port Combo. UNLEY COMPOSITION (Composition Centros) Port Combo. UNLEY COMPOSITION (Composition Centros) Port Combo. UNLEY COMPOSITION (Composition Centros) Port Combo. 2. Will Composition Composition (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition Centros (Centros) Port Combo. 2. Will Composition C	UNBU				Ctata (mandala Hab	undlad Lasal C	itabina an C.	ital Danta								+
3. Fix Office and Tanden Switching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all commissions of loopborn network elements except. for UNE Colin PortLoop Combinations. 4. The first and additional Port nonneuring charges apply to Not Correct Combined Combos, the nonneuring charges shall be those identified in the Nonrecuring' -Currenty Combined sections. Additional PIRCs may apply also and are categorized accordingly. 2. When the Complex of the Common Combined Combos. Proceedings of the Pirch Combined Combos and and additional PIRCs and a section of the Pirch Combined Combos. Proceedings of the Pirch Combined Combos. Proceedings of the Pirch Combined Combos. Proceedings of the Pirch Combined Combos. Proceedings of the Pirch Combined Combos. Proceedings of the Pirch Combined Combos. Proceedings of the Pirch Combined Combos. Proceedings of the Pirch Combined Combos. Proceedings of the Pirch Combined Co											 	an aftilia Bata	F. J. 71. 74					
4. The first and additional Port nonrecurring charges apply also and are categorized accordingly.																•		
S. Marker Rates for Unburned Centrex Port/Logo Combination will be negotiated on an Individual Case Basis, until further notice.																		L
S. Market Rates for Unbundled Centres PortLoop Combination will be negotiated on an Individual Case Basis, until further notice.				urrently	Comb	ined Combos. For	Currently Co	mbined Combo	os, the nonrecu	ırring charges	shall be those	identified in t	he Nonrecu	rring - Curre	ently Combin	ed sections.	Additional NR	Cs may
UNEP OR Commission Rates (Name Design)																		
2-Wire Vol Loop/2-Wire Volos Grade Port (Centrex) Combo UNE Port Loop Combination Rates (Non-Design 1 UEPs 14.89 14.				be nego	otiated	on an Individual Ca	ase Basis, un	til further notic	e.									1
UNE PortLoop Combination Rates (Non-Design)																		1
Service Vol Copp2-Write Volor Grade Port (Centrox) Port Combo 1 UEP95 14,89																		l
Non-Design		UNE P	ort/Loop Combination Rates (Non-Design)															[
2-Vifer VGL Loop/2-Wire Votes Grade Port (Centrex)Port Combo 2 UEP95 21.52			2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															[
Non-Design 2 UEP95 21.52					1	UEP95		14.89										1
2 2 2 2 2 2 2 2 2 2			2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															[
Non-Design 3 UEP96 27.17			Non-Design		2	UEP95		21.52										i
UNE PortU.cop Combination Rates (Design)			2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
2-Wire VS Loop/2-Wire Voice Grade Pott (Centrex)Port Combo-Design					3	UEP95		27.17										i
2-Wire VS Loop/2-Wire Voice Grade Pott (Centrex)Port Combo-Design		UNE P	ort/Loop Combination Rates (Design)															
Design																		1
2 Wire Voice Grade Port (Centrex)Port Combo-Design 2 UEP95 24.66					1	UEP95		17.81										i
Design 2 UKP95 24.26																		
2-Wire Voice Grade Port (Centrex)Port Combo- 3 UEP95 29.59					2	UEP95		24.26										i
Design 3 UEP95 29.99								,										
UNE Loop Rate					3	LIEP95		29 59										i
2-Wire Voice Grade Loop (St. 1) - Zone 1		LINE L			-	OLI 00		20.00					1					
2-Wire Voice Grade Loop (St. 1) - Zone 2 2 UEP95 UECS1 20.38		OITE E			1	LIED05	LIECS1	13.76										-
2-Wire Voice Grade Loop (St. 1) - Zone 3 3 UEP95 UECS1 26,04																		-
2-Wire Voice Grade Loop (St. 2) - Zone 1					_													-
2-Wire Voice Grade Loop (SL 2) - Zone 2 2 UEP95 UECS2 23.13 UEP95 UECS2 28.46 UNE Port Rate UNE Port Rate UNE Port Rate UEP95 UEP95 UEP95 UEP96																		
2-Wire Voice Grade Loop (SL 2) - Zone 3 3 UEP95 UECS2 28.46																		
UNE Port Rate																		
All States		LINE D			3	UEF93	UEC32	20.40										
2-Wire Voice Grade Port (Centrex) Basic Local Area UEP95 UEPYA 1.13 40.30 19.90 24.98 6.65 15.69																		
2-Wire Voice Grade Port (Centrex 800 termination)		All Sta				LIEDOS	LIEDYA	4.40	40.00	10.00	04.00	0.05		45.00				
2-Wire Voice Grade Port (Centrex with Caller ID)*Basic Local Area UEP95 UEPYH 1.13 40.30 19.90 24.98 6.65 15.69 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)*2 Basic Local Area UEP95 UEPYM 1.13 108.36 70.71 54.47 11.94 15.69 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area UEP95 UEPYZ 1.13 108.36 70.71 54.47 11.94 15.69 2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area UEP95 UEPYZ 1.13 108.36 70.71 54.47 11.94 15.69 2-Wire Voice Grade Port terminated on 800 Service Term - Basic Local Area UEP95 UEPY9 1.13 40.30 19.90 24.98 6.65 15.69 AL, KY, LA, MS, SC, & TN Only 2-Wire Voice Grade Port (Centrex) UEP95 UEPY2 1.13 40.30 19.90 24.98 6.65 15.69																		+
Area						UEP95	UEPYB	1.13	40.30	19.90	24.98	6.65		15.69				
2-Wire Voice Grade Port (Centrex from diff Serving Wire Center - 800 Service Term - Basic Local Area UEP95 UEPY5 UEPY7 1.13 108.36 70.71 54.47 11.94 15.69 2-Wire Voice Grade Port Diff Serving Wire Center - 800 Service Term - Basic Local Area UEP95 UEPY7 1.13 108.36 70.71 54.47 11.94 15.69 15.69 2-Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area UEP95 UEPY9 1.13 40.30 19.90 24.98 6.65 15.69 AL, KY, LA, MS, SC, & TN Only 2-Wire Voice Grade Port (Centrex) UEP95 UEP95 UEPY2 1.13 40.30 19.90 24.98 6.65 15.69									40.00									i
Center)2 Basic Local Area			, 100			UEP95	UEPYH	1.13	40.30	19.90	24.98	6.65		15.69				1
2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area UEP95 UEPYZ 1.13 108.36 70.71 54.47 11.94 15.69 2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area UEP95 UEPY9 1.13 40.30 19.90 24.98 6.65 15.69 AL, KY, LA, MS, SC, & TN Only 2-Wire Voice Grade Port (Centrex) UEP95 UEPY2 1.13 40.30 19.90 24.98 6.65 15.69				l	1	l	l				l	l	1	l			Ì	1
Term - Basic Local Area		1		 	 	UEP95	UEPYM	1.13	108.36	70.71	54.47	11.94		15.69				
2-Wire Voice Grade Port terminated in on Megalink or equivalent UEP95				l			1					1		1				1
- Basic Local Area UEP95 UEPY9 1.13 40.30 19.90 24.98 6.65 15.69 2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area UEP95 UEPY2 1.13 40.30 19.90 24.98 6.65 15.69 AL, KY, LA, MS, SC, & TN Only UEP95 UEPQA 1.13 40.30 19.90 24.98 6.65 15.69						UEP95	UEPYZ	1.13	108.36	70.71	54.47	11.94		15.69				
2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area UEP95 UEPY2 1.13 40.30 19.90 24.98 6.65 15.69 IEP04 IEP05 UEP04 1.13 40.30 19.90 24.98 6.65 15.69			• .	l	1							1	1	İ			Ì	1
Basic Local Area UEP95 UEPY2 1.13 40.30 19.90 24.98 6.65 15.69						UEP95	UEPY9	1.13	40.30	19.90	24.98	6.65		15.69				
AL, KY, LA, MS, SC, & TN Only 2-Wire Voice Grade Port (Centrex) UEP95 UEPQA 1.13 40.30 19.90 24.98 6.65 15.69			2-Wire Voice Grade Port Terminated on 800 Service Term -	l	1						<u> </u>			l	I			1
2-Wire Voice Grade Port (Centrex) UEP95 UEPQA 1.13 40.30 19.90 24.98 6.65 15.69			Basic Local Area	l	1	UEP95	UEPY2	1.13	40.30	19.90	24.98	6.65	1	15.69			Ì	1
2-Wire Voice Grade Port (Centrex) UEP95 UEPQA 1.13 40.30 19.90 24.98 6.65 15.69		AL, KY	, LA, MS, SC, & TN Only											İ				1
			2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.13	40.30	19.90	24.98	6.65		15.69				1
						UEP95	UEPQB	1,13	40,30	19,90	24.98	6.65		15.69	ĺ			

UNBU	NDLE	D NETWORK ELEMENTS - South Carolina			,		1					,			ment: 2		bit: C
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
								Nonrec	urring	Nonrecurring	Disconnect			088	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.13	40.30	19.90		6.65		15.69				
		2-Wire Voice Grade Port (Centrex from diff Serving Wire															
		Center)2			UEP95	UEPQM	1.13	108.36	70.71	54.47	11.94		15.69				
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
		Term			UEP95	UEPQZ	1.13	108.36	70.71	54.47	11.94		15.69				
		2 Mire Vaios Condo Destatore installing on Manalink on any installed			UEP95	UEPQ9	1.13	40.30	19.90	24.98	6.65		15.69				
		2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ9	1.13	40.30	19.90	24.98	6.65		15.69				
	Local 9	Switching			OLF 93	ULFQZ	1.13	40.30	19.90	24.90	0.03		13.09				
	Looui	Centrex Intercom Funtionality, per port			UEP95	URECS	0.7996										
	Local I	Number Portability			02. 00	0.1200	0.7000									1	
		Local Number Portability (1 per port)		1	UEP95	LNPCC	0.35					Ì					
	Feature	es															
		All Standard Features Offered, per port			UEP95	UEPVF	3.04		•				15.69				
		All Select Features Offered, per port			UEP95	UEPVS	0.00	406.42					15.69				
		All Centrex Control Features Offered, per port			UEP95	UEPVC	3.04						15.69				
	NARS	Historia de la Contractica del Contractica de la															
		Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial			UEP95 UEP95	UARCX UAR1X	0.00	0.00	0.00			1	15.69 15.69			-	
		Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				15.69			-	-
	Miscel	Ianeous Terminations			UEP95	UARUX	0.00	0.00	0.00				15.69				
		Trunk Side															
		Trunk Side Terminations, each			UEP95	CEND6	8.86	119.57	18.78	60.03	3.77		15.69				
		Digital (1.544 Megabits)					0.00									1	
		DS1 Circuit Terminations, each			UEP95	M1HD1	73.62	202.47	95.90	72.75	2.47		15.69				
		DS0 Channels Activated, each			UEP95	M1HDO	0.00	14.51					15.69				
	Interof	fice Channel Mileage - 2-Wire															
		Interoffice Channel Facilities Termination			UEP95	MIGBC	24.30	40.63	27.47	16.77	6.91		15.69				
	_	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0167										
		e Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
	D4 Cha	annel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.56						15.69				
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	IPQW5	0.56						15.69				-
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.56						15.69				
		Feature Activation on D-4 Channel Bank FX Trunk Side Loop			OL: 30	11 00110	0.00						10.00				
		Slot			UEP95	1PQW7	0.56						15.69				
		Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
		Different Wire Center			UEP95	1PQWP	0.56						15.69				
		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.56						15.69			1	
		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop		1	LIEBOE	400000											
		Slot		<u> </u>	UEP95 UEP95	1PQWQ 1PQWA	0.56						15.69			1	
	Non-P	Feature Activation on D-4 Channel Bank WATS Loop Slot ecurring Charges (NRC) Associated with UNE-P Centrex	1	!	UEP95	IPQWA	0.56					1	15.69			 	-
	NOII-R	NRC Conversion Currently Combined Switch-As-Is with allowed	1	 	 	+				1		1	1		1	 	1
		changes, per port			UEP95	USAC2		37.93	16.72				15.69		1	I	
		New Centrex Standard Common Block		1	UEP95	M1ACS	0.00	668.70	.5.72				15.69			1	
		New Centrex Customized Common Block			UEP95	M1ACC	0.00	668.70					15.69				
		NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	72.89					15.69				
		CENTREX - DMS100 (Valid in All States)															
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo			ļ	\perp									ļ	ļ	
	UNE P	ort/Loop Combination Rates (Non-Design)		<u> </u>												ļ	
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1		LIEDOD		44.00									1	
		Non-Design 2 Wire VG Leep/2 Wire Voice Grade Port (Centrey)Port Comba	1	1	UEP9D		14.89					1	-			 	-
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9D		21.52										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLPAD	+	21.52			1					1	 	
		Non-Design		3	UEP9D		27.17									1	
—	UNF P	ort/Loop Combination Rates (Design)		⊢ Ŭ		+	27.17					 	1		 	t	1

UNDUNDLI	ED NETWORK ELEMENTS - South Carolina	1			T I						Cup Onder	Sun Carle		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring			•		Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-	١.													
	Design		1	UEP9D		17.81										<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9D		24.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLI 3D		24.20										
	Design		3	UEP9D		29.59										
UNE I	Loop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	13.76										1
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	20.38										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	26.04										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	23.13										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	28.46										
	Port Rate															
ALL S	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local		1	UEP9D	UEPTA	1.13	40.30	19.90	24.90	6.63		15.69				
	Area			UEP9D	UEPYB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local		1	OLI OD	OLI ID	1.10	40.00	10.00	24.00	0.00		10.00				+
	Area			UEP9D	UEPYC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local				1	11.10										
	Area			UEP9D	UEPYD	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local															1
	Area			UEP9D	UEPYE	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local															
	Area			UEP9D	UEPYF	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local						40.00									
	Area			UEP9D	UEPYG	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local			UEP9D	UEPYT	1.13	40.30	19.90	24.98	6.65		15.69				
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local		+	UEP9D	UEPYI	1.13	40.30	19.90	24.98	6.65		15.69				
	Area			UEP9D	UEPYU	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local	1	1	OLI OD	OLI 10	1.10	40.00	10.00	24.00	0.00		10.00				
	Area			UEP9D	UEPYV	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local															
	Area			UEP9D	UEPY3	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local															ĺ
	Area			UEP9D	UEPYH	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3			UEP9D	UEPYJ	4.40	40.20	40.00	24.98	0.05		15.69				
	Basic Local Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPYJ	1.13	40.30	19.90	24.98	6.65		15.69				
	2 Basic Local Area			UEP9D	UEPYM	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3	1	1	OLI OD	OLI IIVI	1.10	100.00	70.71	04.47	11.04		10.00				
	Basic Local Area			UEP9D	UEPYO	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3															
	Basic Local Area			UEP9D	UEPYP	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3						_	· · · · · · · · · · · · · · · · · · ·								
	Basic Local Area			UEP9D	UEPYQ	1.13	108.36	70.71	54.47	11.94	<u> </u>	15.69				ļ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3		1		1		400									
ļ	Basic Local Area	<u> </u>	1	UEP9D	UEPYR	1.13	108.36	70.71	54.47	11.94	<u> </u>	15.69			ļ	<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3		1	UEP9D	UEPYS	1 10	100 20	70.71	EA A7	11.04		15.69				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3	1	+	OELAD	UEPTS	1.13	108.36	70.71	54.47	11.94	-	10.09		-	-	
	Basic Local Area			UEP9D	UEPY4	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3	1	1	021 00	OL. 17	1.13	100.00	70.71	57.77	11.34	1	10.08			1	†
	Basic Local Area			UEP9D	UEPY5	1.13	108.36	70.71	54.47	11.94		15.69				

NDUND	LED I	NETWORK ELEMENTS - South Carolina			1								_		ment: 2		bit: C
ATEGORY	r	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual S Order vs Electroni Disc Add
							Rec	Nonrec		Nonrecurring					Rates(\$)		
		45 A						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	LIEDVC	4.40	400.00	70.71	54.47	11.94		45.00				
		asic Local Area Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPY6	1.13	108.36	70.71	54.47	11.94		15.69				
		asic Local Area			UEP9D	UEPY7	1.13	108.36	70.71	54.47	11.94		15.69				
	2-\	Wire Voice Grade Port, Diff Serving Wire Center - 800 Service					_		-		-						
		orm			UEP9D	UEPYZ	1.13	108.36	70.71	54.47	11.94		15.69				
		Wire Voice Grade Port terminated in on Megalink or equivalent															
		asic Local Area			UEP9D	UEPY9	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port Terminated on 800 Service Term Basic cal Area			UEP9D	UEPY2	1.13	40.30	19.90	24.98	6.65		15.69				
ΔΙ		A, MS, SC, & TN Only			UEP9D	UEFTZ	1.13	40.30	19.90	24.90	0.00		15.09				
, AL,		Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-\	Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D UEP9D	UEPQT	1.13 1.13	40.30 40.30	19.90 19.90	24.98 24.98	6.65 6.65		15.69 15.69				
		Wire Voice Grade Port (Centrex / EBS-M5208)3 Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQU	1.13	40.30	19.90	24.98	6.65		15.69				-
		Wire Voice Grade Port (Centrex / EBS-M5216)3 Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.13	40.30	19.90	24.98	6.65		15.69				-
		Wire Voice Grade Port (Centrex / EBG-N6516)5 Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.13	40.30	19.90	24.98	6.65		15.69				
		Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			02. 05	02. Q	0	10.00	10.00	200	0.00		10.00				<u> </u>
		dication)3			UEP9D	UEPQW	1.13	40.30	19.90	24.98	6.65		15.69				
	2-\	Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.13	40.30	19.90	24.98	6.65		15.69				
	2-\	Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2				UEP9D	UEPQM	1.13	108.36	70.71	54.47	11.94		15.69				
	2-\	Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.13	108.36	70.71	54.47	11.94		15.69				ļ
	۵,	Mine Veine Crede Best (Control/differ CMC /EBC ME000)2 2			UEP9D	UEPQP	1.13	400.00	70.71	54.47	11.94		15.69				
-		Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQP	1.13	108.36 108.36	70.71	54.47	11.94		15.69				-
_	2-1	wile voice Grade Fort (Gentlewaller GWG/EBG-3209)2, 3			OLI 3D	OLI QQ	1.10	100.50	70.71	34.47	11.54		15.05				
	2-\	Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.13	108.36	70.71	54.47	11.94		15.69				
		,															
	2-\	Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.13	108.36	70.71	54.47	11.94		15.69				
	2-\	Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.13	108.36	70.71	54.47	11.94		15.69				ļ
	2 1	Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3		1	UEP9D	UEPQ5	1.13	108.36	70.71	54.47	11.94		15.69				
	2-1	write voice Grade Port (Centrex/diller SWC /EBS-M5208)2, 3		1	UEPSD	UEPQS	1.13	108.36	70.71	54.47	11.94		15.09				+
	2-1	Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3		1	UEP9D	UEPQ6	1.13	108.36	70.71	54.47	11.94		15.69				
										•							1
	2-\	Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.13	108.36	70.71	54.47	11.94		15.69				
	2-\	Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Te	rm			UEP9D	UEPQZ	1.13	108.36	70.71	54.47	11.94		15.69				
	۵.							40.00					4= 00				
		Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D UEP9D	UEPQ9 UEPQ2	1.13	40.30	19.90 19.90	24.98 24.98	6.65		15.69				
1 00		Wire Voice Grade Port Terminated on 800 Service Term tching		1	OLPAD	UEFQZ	1.13	40.30	19.90	24.98	6.65		15.69				+
LUC		entrex Intercom Funtionality, per port		!	UEP9D	URECS	0.7996						15.69				†
Loc		nber Portability				3.1200	0000			†			.0.00				†
- -50		cal Number Portability (1 per port)			UEP9D	LNPCC	0.35			1							
Fea	tures																
		Standard Features Offered, per port			UEP9D	UEPVF	3.04	•	•		•		15.69				
_		Select Features Offered, per port		<u> </u>	UEP9D	UEPVS	0.00	406.42					15.69				
		Centrex Control Features Offered, per port		!	UEP9D	UEPVC	3.04			ļ			15.69				<u> </u>
NAF		bundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				15.69				├

BUNDLE	D NETWORK ELEMENTS - South Carolina													ment: 2		bit: C
											Svc Order	Svc Order	Incremental	Incremental	Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge
		Interi									Elec	Manually	Manual Svc	Manual Svo	Manual Svc	Manual S
TEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order v
											•		Electronic-	Electronic-	Electronic-	Electron
													1st	Add'l	Disc 1st	Disc Ad
								_								
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				15.69				
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00				15.69				
	aneous Terminations															
	Trunk Side															
	Trunk Side Terminations, each			UEP9D	CEND6	8.86	119.57	18.78	60.03	3.77		15.69				
	Digital (1.544 Megabits)		<u> </u>	LIEBAB		=	222.4					1= 00				
	DS1 Circuit Terminations, each			UEP9D	M1HD1	73.62	202.47	95.90	72.75	2.47		15.69				
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	14.51					15.69				
	ice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	24.30	40.63	27.47	16.77	6.91		15.69				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0167										
	Activations (DS0) Centrex Loops on Channelized DS1 Service	е														
	nnel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.56						15.69				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.56						15.69				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP9D	1PQW7	0.56						15.69				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center		<u> </u>	UEP9D	1PQWP	0.56						15.69				
	Foot on Astinting to B 4 Observal Book Brigate Live Lang Observ			LIEDOD	4500407	0.50						45.00				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.56						15.69				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			LIEDOD	40000	0.56						45.00				
	Slot			UEP9D	1PQWQ							15.69				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.56						15.69				
	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		37.93	16.72				15.69				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	668.70					15.69			ļ	ļ
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	668.70					15.69			ļ	ļ
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.89					15.69			ļ	ļ
	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD		<u> </u>											1		<u> </u>
	- Requres Interoffice Channel Mileage		<u> </u>											1		<u> </u>
INIO40 2	- Requires Specific Customer Premises Equipment		I	1								1	I	1	I	1

IINBIIN	IDI E	D NETWORK ELEMENTS - Tennessee												Attachi	ment: 2	Evhi	bit: C
			Interi										Svc Order Submitted Manually	Incremental Charge -	Incremental Charge - Manual Svc		Incremental Charge -
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Order vs. Electronic- Disc 1st	Order vs. Electronic- Disc Add'l
							Rec	Nonrecurring		Nonrecurring					Rates(\$)		
- 	ho "7	one" shown in the sections for stand-alone loops or loops as part of	of a com	hinatio	a refere to Coograph	colly Decycr		First	Add'I	First	Add'I		SOMAN Office refe		SOMAN	SOMAN	SOMAN
		ww.interconnection.bellsouth.com/become a clec/html/interconne			irrelers to Geographi	cally Deaver	aged ONE Zones	. To view Geog	ларпісану Dea	veraged ONE 20	one Designatio	is by Cerilia	a Office, refe	er to internet vv	ebsite.		
		_ SUPPORT SYSTEMS	COLIOITATI												l		
N	OTE:	(1) Electronic Service Order: CLEC should contact its contract	t nego	iator if	it prefers the state	specific elec	tronic service o	rdering charge	es as ordered l	by the State Co	mmissions. T	he electron	ic service o	rdering charg	e currently co	ntained in th	is rate
		is the BellSouth regional electronic service ordering charge.															
		(2) Any element that can be ordered electronically will be billed															
		elements that cannot be ordered electronically at present per t				in this cate	egory reflects th	e charge that v	vould be billed	to a CLEC on	ce electronic	ordering cap	oabilities co	me on-line fo	r that element	. Otherwise,	the manual
0	raerin	ng charge, SOMAN, will be applied to a CLECs bill when it sub Electronic OSS Charge, per LSR, submitted via BST's OSS	mits ar	LSK	o BellSouth.					1		1			ı		1
		interactive interfaces (Regional)				SOMEC		3.50									
UNE SEF	RVICE	DATE ADVANCEMENT CHARGE															
N	IOTE:	The Expedite charge will be maintained commensurate with	BellSou	th's FC	CC No.1 Tariff, Section	on 5 as appl	icable.										
		UNE Expedite Charge per Circuit or Line Assignable USOC, per															
11015:0:-		Day			ALL UNE	SDASP		200.00				<u> </u>		ļ			
		EXCHANGE ACCESS LOOP					+					 	<u> </u>				
2	-vviKE	ANALOG VOICE GRADE LOOP 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	13.19	31.99	20.02	10.65	1.41	-	-	20.35	10.54	13.32	13.32
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
		Loop Testing - Basic 1st Half Hour			UEANL	URET1		78.92	78.92					20.35	10.54	13.32	13.32
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		23.33	23.33					20.35	10.54	13.32	13.32
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
		(UVL-SL1)			UEANL	UREWO		15.80	8.95					20.35	10.54	13.32	13.32
		Engineering Information Document (EI) Manual Order Coordination for UVL-SL1s (per loop)			UEANL UEANL	UEANM UEAMC		28.80 36.52	28.80 36.52								<u> </u>
		Order Coordination for Specified Conversion Time for UVL-SL1			UEAINL	UEAIVIC		30.32	30.52								
		(per LSR)			UEANL	OCOSL		34.29	34.29								
2	-WIRE	Unbundled COPPER LOOP															
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1	ı		UEQ	UEQ2X	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	- !		UEQ	UEQ2X	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3 Order Coordination 2 Wire Unbundled Copper Loop - Non-	- 1	3	UEQ	UEQ2X	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
		Designed (per loop)			UEQ	USBMC		36.52	36.52								
		Engineering Information Document			UEQ			28.80	28.80					20.35	10.54	13.32	13.32
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		78.92	78.92					20.35	10.54	13.32	13.32
		Loop Testing - Basic Additional Half Hour			UEQ	URETA		23.33	23.33					20.35	10.54	13.32	13.32
		CLEC to CLEC Conversion Charge Without Outside Dispatch			LIEO	LIDEWO		44.00	7.44					00.05	10.51	40.00	40.00
UNRUND	I FD F	(UCL-ND) EXCHANGE ACCESS LOOP			UEQ	UREWO	+	14.29	7.44			1	-	20.35	10.54	13.32	13.32
		ANALOG VOICE GRADE LOOP															
t t		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or										1					<u> </u>
		Ground Start Signaling - Zone 1		1	UEA	UEAL2	16.56	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or			l												
\vdash		Ground Start Signaling - Zone 2		2	UEA	UEAL2	21.63	75.06	48.20	28.70	17.64	 	<u> </u>	20.35	10.54	13.32	13.32
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	28.28	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
\vdash		Order Coordination for Specified Conversion Time (per LSR)		3	UEA	OCOSL	20.20	34.29	40.20	20.70	17.04	 	-	20.35	10.54	13.32	13.32
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse					1	020				1					
		Battery Signaling - Zone 1		1	UEA	UEAR2	16.56	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 2		2	UEA	UEAR2	21.63	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
 		Battery Signaling - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UEA	UEAR2	28.28	75.06 34.29	48.20	28.70	17.64	1	1	20.35	10.54	13.32	13.32
 		CLEC to CLEC Conversion Charge without outside dispatch			UEA UEA	OCOSL UREWO	1	75.06	36.41					20.35	10.54	13.32	13.32
4	-WIRF	E ANALOG VOICE GRADE LOOP			OLA	OKLVVO	+	75.00	30.41			 		20.35	10.54	13.32	13.32
		4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
		4-Wire Analog Voice Grade Loop - Zone 2			UEA	UEAL4	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
		4-Wire Analog Voice Grade Loop - Zone 3			UEA	UEAL4	42.17	122.76	85.57	76.35	39.16		1	20.35	10.54	13.32	13.32

Version 3Q02: 09/06/02 Page 308 of 362

OMBONDE	ED NETWORK ELEMENTS - Tennessee										I	·		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge Manual S Order vs Electroni Disc Add
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		34.29									_
- 1111	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		75.06	36.41					20.35	10.54	13.32	13.3
2-WIR	E ISDN DIGITAL GRADE LOOP		<u> </u>	LIDAL	1141.07/	00.00	440.70	00.00	70.05	00.40			00.05	10.51	40.00	40.0
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	22.22	142.76	88.88	76.35	39.16			20.35	10.54	13.32	13.3
	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X U1L2X	29.02 37.95	142.76 142.76	88.88 88.88	76.35 76.35	39.16 39.16			20.35 20.35	10.54 10.54	13.32 13.32	13.3
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN UDN		37.95		88.88	76.35	39.16			20.35	10.54	13.32	13.3
	Order Coordination For Specified Conversion Time (per LSR) CLEC to CLEC Conversion Charge without outside dispatch		-	UDN	OCOSL UREWO		34.29 91.77	44.22					20.35	10.54	13.32	13.3
2 WID	E ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIDLE	1.00		UREWU		91.77	44.22					20.35	10.54	13.32	13.3
Z-VVIR	2 Wire Unbundled ADSL Loop including manual service inquiry	AIIDLE	LOUR	1												+
	& facility reservation - Zone 1		1	UAL	UAL2X	13.82	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.3
	2 Wire Unbundled ADSL Loop including manual service inquiry		_ '	UAL	UALZA	13.02	270.01	234.03	74.54	39.14			20.33	10.54	13.32	13.3
	& facility reservation - Zone 2		2	UAL	UAL2X	18.05	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.3
	2 Wire Unbundled ADSL Loop including manual service inquiry			UAL	UALZA	10.05	270.01	234.03	74.54	35.14			20.33	10.54	13.32	15.0
	& facility reservation - Zone 3		3	UAL	UAL2X	23.60	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.3
	Order Coordination for Specified Conversion Time (per LSR)		3	UAL	OCOSL	23.00	34.29	234.03	74.54	33.14			20.33	10.54	13.32	13.0
	2 Wire Unbundled ADSL Loop without manual service inquiry &			UAL	OCOGL		34.23									+
	facility reservation - Zone 1		1	UAL	UAL2W	13.82	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	2 Wire Unbundled ADSL Loop without manual service inquiry &	-	<u> </u>	OAL	UALZVV	15.02	31.33	20.02	10.03	1.41			20.55	10.54	10.02	10.0
	facility reservation - Zone 2		2	UAL	UAL2W	18.05	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	2 Wire Unbundled ADSL Loop without manual service inquiry &			OAL	UALZVV	10.00	31.33	20.02	10.03	1.41			20.55	10.54	10.02	10.0
	facility reservation - Zone 3		3	UAL	UAL2W	23.60	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL	25.00	34.29	20.02	10.03	1.41			20.55	10.54	10.02	10.0
	CLEC to CLEC Conversion Charge without outside dispatch	-		UAL	UREWO		31.99	20.02					20.35	10.54	13.32	13.3
2-WIR	E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIRI F	OOP	O/ IL	OILLWO		01.00	20.02					20.00	10.04	10.02	10.0
2 ****	2 Wire Unbundled HDSL Loop including manual service inquiry	I	1													
	& facility reservation - Zone 1		1	UHL	UHL2X	10.83	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.3
	2 Wire Unbundled HDSL Loop including manual service inquiry		<u> </u>	0.1.2	O. I.E.Z.	.0.00	2, 0.0.	201.00	7	00			20.00	10.01	10.02	
	& facility reservation - Zone 2		2	UHL	UHL2X	14.15	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.3
	2 Wire Unbundled HDSL Loop including manual service inquiry									-						
	& facility reservation - Zone 3		3	UHL	UHL2X	18.50	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.3
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		34.29									
	2 Wire Unbundled HDSL Loop without manual service inquiry															1
	and facility reservation - Zone 1	- 1	1	UHL	UHL2W	10.83	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	2 Wire Unbundled HDSL Loop without manual service inquiry															1
	and facility reservation - Zone 2	- 1	2	UHL	UHL2W	14.15	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	2 Wire Unbundled HDSL Loop without manual service inquiry															1
	and facility reservation - Zone 3	I	3	UHL	UHL2W	18.50	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		34.29									
	CLEC to CLEC Conversion Charge without outside dispatch	-		UHL	UREWO		31.99	20.02					20.35	10.54	13.32	13.3
4-WIR	E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
	4 Wire Unbundled HDSL Loop including manual service inquiry															
	and facility reservation - Zone 1		1	UHL	UHL4X	13.93	279.60	244.22	74.54	39.14			20.35	10.54	13.32	13.3
	4-Wire Unbundled HDSL Loop including manual service inquiry															
	and facility reservation - Zone 2		2	UHL	UHL4X	18.20	279.60	244.22	74.54	39.14			20.35	10.54	13.32	13.3
	4-Wire Unbundled HDSL Loop including manual service inquiry															
	and facility reservation - Zone 3		3	UHL	UHL4X	23.80	279.60	244.22	74.54	39.14			20.35	10.54	13.32	13.3
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		34.29								1	
	4-Wire Unbundled HDSL Loop without manual service inquiry	1	1									1			I	
	and facility reservation - Zone 1		1	UHL	UHL4W	13.93	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	4-Wire Unbundled HDSL Loop without manual service inquiry		1	l	I I]			1			I	
	and facility reservation - Zone 2	- 1	2	UHL	UHL4W	18.20	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	4-Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 3		3	UHL	UHL4W	23.80	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
	Order Coordination for Specified Conversion Time (per LSR)	<u> </u>	<u> </u>	UHL	OCOSL		34.29							10		1
	CLEC to CLEC Conversion Charge without outside dispatch		<u> </u>	UHL	UREWO		31.99	20.02					20.35	10.54	13.32	13.3
4-WIR	E DS1 DIGITAL LOOP		L.,	1101	1101.707	F7	040.00	040 =0	00.00	40.15			40.00	0.10	44.00	1
	4-Wire DS1 Digital Loop - Zone 1			USL	USLXX	57.73	313.08	219.72	96.86	40.45		ļ	18.98	8.43		
	4-Wire DS1 Digital Loop - Zone 2		2	USL	USLXX	75.40	313.08	219.72	96.86	40.45	1	l	18.98	8.43	11.95	11

UNBUNDL	ED NETWORK ELEMENTS - Tennessee										•	,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	всѕ	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
		1			+		Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)	l	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	98.59	313.08	219.72	96.86	40.45	COMILO	COMPAR	18.98	8.43	11.95	
	Order Coordination for Specified Conversion Time (per LSR)		Ŭ	USL	OCOSL	50.00	34.59	210.72	30.00	40.40			10.50	0.40	11.00	11.55
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		130.47	40.11					20.35	10.54	13.32	13.32
4-WI	IRE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															10.02
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	31.10	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	40.61	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital 19.2 Kbps		3	UDL	UDL19	53.11	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	31.10	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	40.61	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	53.11	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		34.29									
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1	<u> </u>	1	UDL	UDL64	31.10	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2	ļ	2	UDL	UDL64	40.61	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL	UDL64	53.11	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		34.29									
	CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		102.28	49.82					20.35	10.54	13.32	13.32
2-WI	RE Unbundled COPPER LOOP															+
	2-Wire Unbundled Copper Loop/Short including manual service		1	LICI	LICI DD	12.10	24.00	20.00	40.05	4.44			20.25	40.54	40.00	40.00
	inquiry & facility reservation - Zone 1		1	UCL	UCLPB	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Short including manual service		2		LIOL DD	47.00	04.00	00.00	40.05				00.05	40.54	40.00	40.00
	inquiry & facility reservation - Zone 2	_ ! _	2	UCL	UCLPB	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2 Wire Unbundled Copper Loop/Short including manual service		2	LICI	LICI DD	22.52	24.00	20.00	40.05	4.44			20.25	40.54	40.00	40.00
	inquiry & facility reservation - Zone 3	- '	3	UCL UCL	UCLPB UCLMC	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLIVIC		36.52	36.52								+
	2-Wire Unbundled Copper Loop/Short without manual service		1	UCL	LICI DW	12.10	24.00	20.02	10.65	1 11			20.35	10.54	12.22	12.22
	inquiry and facility reservation - Zone 1		-	UCL	UCLPW	13.19	31.99	20.02	10.65	1.41			20.33	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2	1 1	2	UCL	UCLPW	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Short without manual service	+ '		UCL	OCLF W	17.23	31.99	20.02	10.03	1.41			20.33	10.54	13.32	13.32
	inquiry and facility reservation - Zone 3	1 1	3	UCL	UCLPW	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)	- ' -	3	UCL	UCLMC	22.55	36.52	36.52	10.03	1.71			20.55	10.54	10.02	13.32
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.			OOL	COLIVIO		00.02	00.02								+
	inquiry and facility reservation - Zone 1	1	1	UCL	UCL2L	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL2L	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3	- 1	3	UCL	UCL2L	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 1	1	1	UCL	UCL2W	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 2	- 1	2	UCL	UCL2W	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 3	I	3	UCL	UCL2W	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								
	CLEC to CLEC Conversion Charge without outside dispatch															
	(UCL-Des)			UCL	UREWO		31.99	20.02					20.35	10.54	13.32	13.32
4-WI	RE COPPER LOOP															<u> </u>
	4-Wire Copper Loop/Short - including manual service inquiry	1 .	١.				400 ==		=0					40		
	and facility reservation - Zone 1	 	1	UCL	UCL4S	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Copper Loop/Short - including manual service inquiry	1 .	_	LICI	1101.40	20.05	400.70	05.57	70.05	20.40			20.05	10.54	40.00	40.00
	and facility reservation - Zone 2	 	2	UCL	UCL4S	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Copper Loop/Short - including manual service inquiry	Ι.	3	LICI	1101.40	40.47	400.70	05.57	70.05	20.40			20.25	10.51	40.00	40.00
	and facility reservation - Zone 3	+ -	3	UCL	UCL4S	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)	1	 	UCL	UCLMC		36.52	36.52						 	 	+
	4-Wire Copper Loop/Short - without manual service inquiry and	1 .	4	UCL	LICLAW	24.70	100.70	05 57	76.25	20.40			20.35	10.54	12.20	12.20
	facility reservation - Zone 1 4-Wire Copper Loop/Short - without manual service inquiry and	+ '	1	UCL	UCL4W	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
																1

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Exhi	ibit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		I Increment Charge -
						Rec	Nonrecurring First	Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN		Rates(\$)	SOMAN	SOMAN
	4-Wire Copper Loop/Short - without manual service inquiry and						FIRST	Addi	FIRSt	Addi	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	facility reservation - Zone 3		3	UCL	UCL4W	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.3
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	72.17	36.52	36.52	70.00	00.10			20.00	10.04	10.02	10.0
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															†
	inquiry and facility reservation - Zone 1	- 1	1	UCL	UCL4L	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.3
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 2	ı	2	UCL	UCL4L	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.3
	4-Wire Unbundled Copper Loop/Long - includes manual svc.		3	UCL	UCL4L	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.3
	inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)	ı	3	UCL	UCL4L UCLMC	42.17	36.52	36.52	76.35	39.16			20.35	10.54	13.32	13.3
	4-Wire Unbundled Copper Loop/Long - without manual svc.			UCL	UCLIVIC		30.52	36.32			1					+
	inquiry and facility reservation - Zone 1		1	UCL	UCL4O	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.3
	4-Wire Unbundled Copper Loop/Long - without manual svc.		<u> </u>			0	,0	22.0.	. 2.00	22.10					. 5.02	1
	inquiry and facility reservation - Zone 2	- 1	2	UCL	UCL4O	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.3
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 3	I	3	UCL	UCL4O	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.3
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)			UCL	UREWO		31.99	20.02					20.35	10.54	13.32	13.3
OOP MODIFIC				UCL	UKEWU		31.99	20.02			1	-	20.33	10.54	13.32	13.3
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft Unbundled Loop Modification, Removal of Load Coils - 2 wire	ı		UAL, UHL, UCL, UEQ, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULM2L		65.40	65.40					20.35	10.54	13.32	13.3
	greater than 18k ft	1		UCL, ULS, UEQ	ULM2G		710.71	23.77					20.35	10.54	13.32	13.3
	Unbundled Loop Modification Removal of Load Coils - 4 Wire			,												1
	less than or equal to 18K ft	I		UHL, UCL	ULM4L		65.40	65.40					20.35	10.54	13.32	13.3
	Unbundled Loop Modification Removal of Load Coils - 4 Wire			UCL	ULM4G											
	pair greater than 18k ft Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop	ı		UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULMBT		710.71	23.77 65.44					20.35	10.54	13.32	
SUB-LOOPS	an Dietrikution										1		-			
Sub-Lo	op Distribution Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-										1	-	-			+
	Up	1		UEANL	USBSA		517.25	517.25					20.35	10.54	13.32	13.3
												1				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	I		UEANL	USBSB		42.68	42.68			ļ		20.35	10.54	13.32	13.3
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up	Ι		UEANL	USBSC		313.01	313.01					20.35	10.54	13.32	13.3
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel						400.00									
	Set-Up			UEANL	USBSD		108.06	108.06			1		20.35	10.54	13.32	13.3
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewide		SW	UEANL	USBN2	10.02	148.84	112.34	73.14	36.65			20.35	10.54	13.32	13.3
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.29	34.29								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -										1		1			+
	Zone 1		1	UEANL	USBN4	7.30	147.93	75.11	99.96	16.98			20.35	10.54	13.32	13.3
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN4	9.54	147.93	75.11	99.96	16.98			20.35	10.54	13.32	13.3
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -			-				-			1					
	Zone 3		3	UEANL	USBN4	12.47	147.93	75.11	99.96	16.98		 	20.35	10.54	13.32	13.3
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.29	34.29					1			
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)			UEANL	USBR2	1.35	94.56	29.35			1	1	20.35	10.54	13.32	13.

UNBU	NDLE	NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic Disc Add'l
								Nonrecurring		Nonrecurring	Disconnect			000	Rates(\$)		
				-			Rec					001150	001111			001141	001111
				-				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Onder Consideration for Habrer dead Colo Longo and architecture			LIFANII	LICDMC		24.00	24.00								
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair		<u> </u>	UEANL	USBMC	0.00	34.29	34.29					00.05	10.51	40.00	40.00
		Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	ı		UEANL	USBR4	2.26	116.14	37.10					20.35	10.54	13.32	13.32
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.29	34.29								
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	ı	1	UEF	UCS2X	5.16	110.71	37.89	94.41	13.09			20.35	10.54	13.32	13.32
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	I		UEF	UCS2X	6.74	110.71	37.89	94.41	13.09			20.35	10.54	13.32	13.32
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	ı	3	UEF	UCS2X	8.81	110.71	37.89	94.41	13.09			20.35	10.54	13.32	13.32
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.29	34.29								
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1	UEF	UCS4X	6.52	117.12	44.30	99.96	16.98			20.35	10.54	13.32	13.32
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	i	2	UEF	UCS4X	8.52	117.12	44.30	99.96	16.98	i	1	20.35	10.54	13.32	13.32
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	- i-		UEF	UCS4X	11.14	117.12	44.30	99.96	16.98	ł – – – –		20.35	10.54	13.32	13.32
		Suppor Structured Sub Loop Distribution - 2016 3		-	0-1	300-A	11.14	111.12	44.30	33.30	10.30	1	l	20.33	10.54	10.02	10.02
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.29	34.29								
				-	UEF	USDIVIC		34.29	34.29								
	Unbun	dled Sub-Loop Modification															
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load															
		Coil/Equip Removal per 2-W PR			UEF	ULM2X		335.36	7.82					20.34	10.54	13.32	13.32
		Unbundled Sub-loop Modification - 4-W Copper Dist Load															
		Coil/Equip Removal per 4-W PR			UEF	ULM4X		335.36	7.82					20.35	10.54	13.32	13.32
		Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged															
		Tap Removal, per PR unloaded			UEF	ULM4T		528.48	9.74					20.35	10.54	13.32	13.32
	Unbun	dled Network Terminating Wire (UNTW)															
		Unbundled Network Terminating Wire (UNTW) per Pair		-	UENTW	UENPP	0.4555	2.48	2.48					20.35	10.54	13.32	13.32
		k Interface Device (NID)	·	1	02.11.11	OLI III	0.1000	2.10	2.10					20.00	10.01	10.02	10.02
	INCLINO	Network Interface Device (NID) - 1-2 lines		 	UENTW	UND12		89.69	54.56	0.6391	0.6391			20.35	10.54	13.32	13.32
		Network Interface Device (NID) - 1-2 lines			UENTW	UND16		129.65	94.51	0.6522	0.6522			20.35	10.54	13.32	13.32
				-						0.6322	0.6322						
		Network Interface Device Cross Connect - 2 W		ļ	UENTW	UNDC2		11.11	11.11					20.35	10.54	13.32	13.32
		Network Interface Device Cross Connect - 4W			UENTW	UNDC4		11.11	11.11					20.35	10.54	13.32	13.32
SUB-LO																	
	Sub-Lo	op Feeder															
		USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
		Distribution Facility set-up			UDN,UCL,UDL,UDC	USBFW		517.25						20.35	10.54	13.32	13.32
		USL Feeder - DS0 Set-up per Cross Box location - per 25 pair			UEA,												
		set-up			UDN,UCL,UDL,UDC	USBFX		42.68	42.68					20.35	10.54	13.32	13.32
		USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		531.04	11.34	i i			l	20.35	10.54	13.32	13.32
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice						3004	54	1		i	1	20.00		.0.02	.0.02
		Grade- Statewide	l	sw	UEA	USBFA	12.05	122.24	85.05	76.35	39.16	I	1	20.35	10.54	13.32	13.32
		Order Coordination for Specified Conversion Time, per LSR	-	3**	UEA	OCOSL	12.00	34.29	00.00	7 0.00	55.70	ł – – – –		20.00	10.04	10.02	10.02
		Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice		-	0-/1	JUUGE		57.23		1		1	l	1	t	1	1
		Grade - Statewide	l	sw	UEA	USBFB	12.05	122.24	85.05	76.35	39.16	I	1	20.35	10.54	13.32	13.32
				SW			12.05		85.05	76.35	39.16			20.35	10.54	13.32	13.32
		Order Coordination for Specified Time Conversion, per LSR		 	UEA	OCOSL		34.29		1		1	ļ		-	-	1
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,	l	1	L	LIODES		,		== ==		I	1				
		Voice Grade Loop - Statewide		SW	UEA	USBFC	12.05	122.24	85.05	76.35	39.16		ļ	20.35	10.54	13.32	13.32
		Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		34.29				<u> </u>	<u> </u>				
Ţ		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice		1	<u> </u>							i	1		_		
		Grade - Zone 1	<u></u>	_1	UEA	USBFD	21.52	137.31	61.93	118.04	30.13	<u> </u>	<u> </u>	20.35	10.54	13.32	13.32
		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice							-								
		Grade - Zone 2	l	2	UEA	USBFD	28.11	137.31	61.93	118.04	30.13	I	1	20.35	10.54	13.32	13.32
		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice						1						İ			
		Grade - Zone 3		3	UEA	USBFD	36.76	137.31	61.93	118.04	30.13	1	l	20.35	10.54	13.32	13.32
		Order Coordination For Specified Conversion Time, Per LSR		Ť	UEA	OCOSL	22.7.0	34.29	2.700		22.10						13.02
		Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice	—	 	1	J J J J J L		07.20		1		 	 	 	1	1	1
		Grade - Zone 1	l	1	UEA	USBFE	21.52	137.31	61.93	118.04	30.13	I]	20.35	10.54	13.32	13.32
				- '-	OLA	JUDI L	21.32	101.01	01.33	110.04	30.13	 	 	20.35	10.34	13.32	13.32
		Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice	l	_	Liea	HODEE	20.41	407.01	04.00	440.01	00.40	I	1	20.65	10.51	10.00	10.00
		Grade - Zone 2		2	UEA	USBFE	28.11	137.31	61.93	118.04	30.13	1	ļ	20.35	10.54	13.32	13.32
		Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice	l	_	l							I	1				
		Grade - Zone 3		3	UEA	USBFE	36.76	137.31	61.93	118.04	30.13			20.35	10.54	13.32	13.32
		Order Coordination For Specified Conversion Time, Per LSR	_	i –	UEA	OCOSL		34.29			·	1	1	1	1	1	_

UNBUNDLE	D NETWORK ELEMENTS - Tennessee										1 -	1 -		ment: 2		bit: C
											Submitted	Svc Order Submitted	Charge -	Charge -	Charge -	Charge -
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Elec per LSR	Manually per LSR	Manual Svc Order vs. Electronic-	Manual Svc Order vs. Electronic-	Order vs. Electronic-	Order vs. Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1			UDN	USBFF	16.11	142.83	67.45	104.67	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2		2	UDN	USBFF	21.04	142.83	67.45	104.67	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	27.51	142.83	67.45	104.64	18.53			19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, Per LSR		-	UDN	OCOSL	40.44	34.29	C7.4F	104.67	40.50			19.99	10.00	19.99	19.9
-	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		2	UDC	USBFS USBFS	16.11 21.04	142.83 142.83	67.45 67.45	104.67	18.53 18.53			19.99	19.99 19.99	19.99	
+	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		3	UDC	USBFS	27.51	142.83	67.45	104.64	18.53			19.99	19.99	19.99	19.9
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1			USL	USBFG	39.74	116.00	40.62	106.82	18.91			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2			USL	USBFG	51.90	116.00	40.62	106.82	18.91			19.99	19.99		
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3			USL	USBFG	67.86		40.62	106.82	18.91			19.99	19.99		
	Order Coordination For Specified Conversion Time, Per LSR		Ť	USL	OCOSL	07.00	34.59	10.02	100.02	10.01			10.00	10.00	10.00	10.0.
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	9.52	114.27	38.89	104.64	18.53			19.99	19.99	19.99	19.9
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone															
	2	<u></u>	2	UCL	USBFH	12.43	114.27	38.89	104.64	18.53	<u> </u>	<u> </u>	19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 3		3	UCL	USBFH	16.26	114.27	38.89	104.64	18.53			19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		34.29									
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	14.37	123.41	48.03	110.44	22.53			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2	UCL	USBFJ	18.76	123.41	48.03	110.44	22.53			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3		3	UCL	USBFJ	24.53	123.41	48.03	110.44	22.53			19.99	19.99	19.99	19.9
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		34.29									
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	26.06	116.00	40.62	106.82	18.91			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	34.03	116.00	40.62	106.82	18.91			19.99	19.99		
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	44.50	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 1		1	UDL	USBFO	26.06	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 2		2	UDL	USBFO	34.03	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 3		3	UDL	USBFO	44.50	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		34.29									
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1		1	UDL	USBFP	26.06	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.9
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2		2	UDL	USBFP	34.03	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.9
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3		3	UDL	USBFP	44.50	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.9
	Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		34.29									
SUB-LOOPS																
Sub-L	oop Feeder															
	Sub Loop Feeder - DS3 - Per Mile Per Month	<u> </u>	<u> </u>	UE3	1L5SL	14.11	0 100 0			F0.4.0:				10.5		
	Sub Loop Feeder - DS3 - Facility Termination Per Month	<u> </u>		UE3	USBF1	333.26	3,406.61	407.68	165.17	501.31			20.35	10.54	13.32	
	Sub Loop Feeder - STS-1 - Per Mile Per Month	-	<u> </u>	UDLSX	1L5SL USBF7	14.11	2 400 04	407.68	165.17	F04.04			20.35	10.54	13.32	
	Sub Loop Feeder - STS-1 - Facility Termination Per Month	- !	 		1L5SL	359.02 10.71	3,406.61	407.68	165.17	501.31			20.35	10.54	13.32	+
	Sub Loop Feeder – OC-3 – Per Mile Per Month Sub Loop Feeder - OC-3 - Facility Termination Protection Per			UDLO3												
	Month	<u> </u>	<u> </u>	UDLO3	USBF5	56.64	0.400.01	107.00	405.45	F01.01			00.0=	10.51	10.00	
	Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3	USBF2	546.31	3,406.61	407.68	165.17	501.31			20.35	10.54	13.32	-
	Sub Loop Feeder - OC-12 - Per Mile Per Month	'		UDL12	1L5SL	13.18	-									+
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per Month	١.,		UDL12	USBF6	639.98										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	<u> </u>		UDL12	USBF3	1,697.00	3,406.61	407.68	165.17	501.31			20.35	10.54	13.32	+
+	Sub Loop Feeder - OC-12 - Pacifity Termination Fer Month	H	-	UDL48	1L5SL	43.22	5,400.01	+07.00	105.17	301.31			20.35	10.34	13.32	\leftarrow
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per			00240	72002	70.22			 						-	
	Month	Li		UDL48	USBF9	320.36	1								1	1
- 1	Sub Loop Feeder - OC-48 - Facility Termination Per Month	l i	<u> </u>	UDL48	USBF4	1,457.00	3,592.61	407.68	165.17	501.31			20.35	10.54	13.32	†
	Sub Loop Feeder - OC-12 Interface On OC-48	i		UDL48	USBF8	361.44	806.02	407.68	165.17	501.31			20.35	10.54	13.32	<u> </u>
UNBUNDLED	LOOP CONCENTRATION						1				İ	İ			1	†
	Loop Channelization System		1	ULC	ULCCS	307.07	307.34	74.37	4.18				20.35	10.54	13.32	13.3

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachi	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge -		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -
						Rec	Nonrecurring		Nonrecurring	g Disconnect			oss	Rates(\$)		
							First	Add'l	First	Add'l		SOMAN		SOMAN	SOMAN	SOMAN
	CO Channel Interface - 2-Wire Voice Grade			ULC	ULCC2	1.20	9.57	9.52	8.66	8.60			20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	500.18	613.60	613.60					20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - System B (TR008) Unbundled Loop Concentration - System A (TR303)			ULC ULC	UCT8B UCT3A	54.82 539.00	255.67 613.60	255.67 613.60					20.35 20.35	10.54 10.54	13.32 13.32	13.32 13.32
	Unbundled Loop Concentration - System A (17303)				UCT3B	92.37	255.67	255.67					20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - DS1 Loop Interface Card			ULC	UCTCO	6.23	74.39	53.07	30.23	8.46			20.35	10.54	13.32	
	Unbundled Loop Concentration - ISDN Loop Interface (Brite															
	Card)			UDN	ULCC1	8.46	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - UDC Loop Interface (Brite			1100		0.40	0.00	0.05	0.74	0.05			00.05	40.54	40.00	40.00
-	Card) Unbundled Loop Concentration2 Wire Voice-Loop Start or			UDC	ULCCU	8.46	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	2.32	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery															
	Loop Interface (SPOTS Card)			UEA	ULCCR	12.45	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface														40.00	40.000
	(Specials Card) Unbundled Loop Concentration - TEST CIRCUIT Card			UEA ULC	ULCC4 UCTTC	7.53 35.77	8.69 8.69	8.65 8.65	9.71 9.71	9.65 9.65			20.35 20.35	10.54 10.54	13.32 13.32	13.332 13.32
	Unbundled Loop Concentration - Test Circon Card Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop			OLC	OCTIC	35.77	0.09	0.00	9.71	9.65			20.35	10.54	13.32	13.32
	Interface			UDL	ULCC7	11.03	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface			UDL	ULCC5	11.03	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop			ODL	OLCC3	11.03	0.09	8.03	5.71	9.03			20.33	10.54	13.32	13.32
	Interface			UDL	ULCC6	11.03	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
									9.71							
UNE OTHER, F	ROVISIONING ONLY - NO RATE															
	NID - Dispatch and Service Order for NID installation			UENTW	UNDBX	0.00	0.00									
	UNTW Circuit Id Establishment, Provisioning Only - No Rate			UENTW UEANL,UEF,UEQ,U	UENCE	0.00	0.00									
	Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN	0.00	0.00									
UNE OTHER, E	PROVISIONING ONLY - NO RATE			LIVIVV	ONLON	0.00	0.00									
				UAL,UCL,UDC,UDL,												
	Unbundled Contact Name, Provisioning Only - no rate			UDN,UEA,UHL,ULC	UNECN	0.00	0.00									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no															
 	rate Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no		-	UEA,UDN,UCL,UDC	USBFQ	0.00	0.00		 	 	 	 				
	rate			UEA,USL,UCL,UDL	USBFR	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									1
	Unbundled DS1 Loop - Expanded Superframe Format option -															
	no rate			USL	CCOEF	0.00	0.00									
HIGH CAPACI	TY UNBUNDLED LOCAL LOOP		ļ								<u> </u>	ļ				
	High Capacity Unbundled Local Loop - DS3 - Per Mile per month			UE3	1L5ND	9.19			1	1						
 	High Capacity Unbundled Local Loop - DS3 - Facility			020	ILUIAD	5.19			 	 	 	 				
	Termination per month			UE3	UE3PX	374.24	595.37	304.50	234.83	170.16			36.84	36.84	19.01	19.01
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per															
	month		<u> </u>	UDLSX	1L5ND	9.19										<u> </u>
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month			UDLSX	UDLS1	389.35	505.27	304.50	215 92	151 15			36.84	36.84	19.01	10.01
Note /4): Rates provided in TN for both electronic and manual Loop	Maker	n are in				595.37		215.82	151.15	nents from t	he Tenness			19.01	19.01
LOOP MAKE-U		makeu	are ill	torini ana subject to	- istro-active	ac-up aujusi	ento penulli	y a permanem	Take running on	ese rate elei	101110		negulator)	Authority.		
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).	R		UMK	UMKLW		0.76	0.76								
 	Loop Makeup - Preordering With Reservation, per spare facility	- 11		· · · · · · · · · · · · · · · · · · ·	J.711 (LVV		3.70	5.70								
	queried (Manual).	R		UMK	UMKLP		0.76	0.76	1	1						
İ	Loop MakeupWith or Without Reservation, per working or															
I I I I I I I I I I I I I I I I I I I	spare facility queried (Mechanized)	R	<u> </u>	UMK	PSUMK		0.76	0.76								<u> </u>
	DEDICATED TRANSPORT	:::::		d balani DC2	manth DCC	CTC 4 faux :::				-						↓
NOTE:	INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	a - pelow DS3=one	montn, DS3/	SIS-1=Tour mo	ntns				ĺ					

UNBUNDLE	D NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTER	OFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per month			U1TVX	1L5XX	0.0054										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			UTIVA	ILSAA	0.0054										
	Facility Termination			U1TVX	U1TV2	18.58	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade						55.55									
	Rev Bat Per Mile per month			U1TVX	1L5XX	0.0054										
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat															
	Facility Termination			U1TVX	U1TR2	18.58	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -															
	Per Mile per month Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade			U1TVX	1L5XX	0.0054	-				1					
	- Facility Termination			U1TVX	U1TV4	24.09	37.87	26.02	30.78	13.07			15.08	15.08	8.66	8.66
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile			UTIVA	01174	24.09	37.07	20.02	30.76	13.07			15.00	13.06	8.00	0.00
	per month			U1TDX	1L5XX	0.0174										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination			U1TDX	U1TD5	17.98	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
	per month			U1TDX	1L5XX	0.0174										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			U1TDX	U1TD6	17.98	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			U1TD1	1L5XX	0.3562										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility			OTIDI	ILJAA	0.3302										
	Termination			U1TD1	U1TF1	77.86	112.40	76.27	19.55	14.99			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			0		77.00	112.10		10.00	1 11.00			20.00	21.00	0.00	10.01
	month			U1TD3	1L5XX	2.34										
	Interoffice Channel - Dedicated Transport - DS3 - Facility															
	Termination per month			U1TD3	U1TF3	848.99	395.29	176.56	109.04	105.91			36.84	36.84	19.01	19.01
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per															
	month			U1TS1	1L5XX	2.34										
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination			U1TS1	U1TFS	849.30	395.29	176.56	109.04	105.91			36.84	36.84	19.01	19.01
LOCAL	L CHANNEL - DEDICATED TRANSPORT			01131	01113	049.30	393.29	170.30	105.04	105.91			30.64	30.04	19.01	19.01
	LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing	a perio	d - beld	ow DS3=one month.	DS3/STS-1=f	our months										
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 1	Ĭ		ULDVX	ULDV2	17.18	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2		2	ULDVX	ULDV2	22.44	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3		3	UNDVX	ULDV2	29.34	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat			l												
	Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat		1	ULDVX	ULDR2	17.18	199.33	24.16	54.81	4.80						
	Zone 2		2	ULDVX	ULDR2	22.44	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat			OLDVX	OLDINZ	22.44	199.33	24.10	34.01	4.00						
	Zone 3		3	ULDVX	ULDR2	29.34	199.33	24.16	54.81	4.80						
1	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 1		1	UNDVX	ULDV4	18.18	201.53	24.83	55.52	5.51						
	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 2			UNDVX	ULDV4	23.74	201.53	24.83	55.52	5.51						
	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 3		3	UNDVX	ULDV4	31.05	201.53	24.83	55.52	5.51						
	Local Channel - Dedicated - DS1 - Zone 1		1	ULDD1	ULDF1	36.24	277.35	233.26	33.18	22.30						
	Local Channel - Dedicated - DS1 - Zone 2	ļ	2	ULDD1	ULDF1	47.33	277.35	233.26	33.18	22.30						
	Local Channel - Dedicated - DS1 - Zone 3	1	3	ULDD1	ULDF1	61.89	277.35	233.26	33.18	22.30	1			 	 	
	Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination	-		ULDD3 ULDD3	1L5NC ULDF3	7.15 611.30	595.37	304.50	215.82	151.15	-		36.84	36.84	19.01	19.01
	Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1- Per Mile per month	1		ULDS1	1L5NC	7.15	383.37	304.30	213.02	101.15	1		30.64	30.64	19.01	19.01
	Local Channel - Dedicated - STS-1 - Facility Termination	1		ULDS1	ULDFS	599.59	588.07	297.20	215.82	151.15			20.35	21.09	9.80	10.54
OARK FIBER	The second secon	1			,,,,,,	555.56	555.57	201.20	2.0.02	.510			20.00	255	3.50	
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction								1					1	1	
	Thereof per month - Local Channel			UDF	1L5DC	58.83	<u> </u>						<u> </u>			
	NRC Dark Fiber - Local Channel			UDF	UDFC4		1,121.00	153.19	580.26	357.17			20.35	21.09	9.80	10.54

UNBUNDLE	D NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge -
						Rec	Nonrecurring		Nonrecurring	Disconnect		•		Rates(\$)	•	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Interoffice Channel			UDF	1L5DF	28.74							ļ		1	
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		1,121.00	153.19	580.26	357.17			20.35	21.09	9.80	10.54
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction					=====										
	Thereof per month - Local Loop			UDF	1L5DL	58.83	4 404 00	452.40	500.00	257.47			20.25	24.00	0.00	40.54
	NRC Dark Fiber - Local Loop FEN DIGIT SCREENING		<u> </u>	UDF	UDFL4		1,121.00	153.19	580.26	357.17			20.35	21.09	9.80	10.54
	8XX Access Ten Digit Screening, Per Call			OHD		0.0005192	1									
	8XX Access Ten Digit Screening, Per Call 8XX Access Ten Digit Screening, Reservation Charge Per 8XX			OLID		0.0003192	1									
	Number Reserved			OHD	N8R1X		5.21	0.76					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O	1		0.10	1401.174		J.21	0.70	 			1	20.33	20.33	13.20	15.20
	POTS Translations	l		OHD			11.47	1.46	7.34	0.7602			20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Per 8XX No. Established With						1								12.20	
	POTS Translations	l		OHD	N8FTX		11.47	1.46	7.34	0.7602			20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Customized Area of Service								l i							
	Per 8XX Number			OHD	N8FCX		4.47	2.24					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Multiple InterLATA CXR	l]			
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		5.23	3.00					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		5.97	0.76					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Call Handling and Destination															
I INIE INIEGONA	Features			OHD	N8FDX		4.47						20.35	20.35	13.28	13.28
	ATION DATA BASE ACCESS (LIDB)	 	<u> </u>	OOT	1	0.0000354							 	 	!	1
	LIDB Common Transport Per Query LIDB Validation Per Query	!	 	OQT OQU	+	0.0000354			 				-	-		1
	LIDB Originating Point Code Establishment or Change	 		OQU OQT, OQU	NRPBX	0.0117403	49.03		 				20.35	20.35	13.28	13.28
SIGNALING (C			1	041,040	MINERY		45.03						20.33	20.33	13.20	13.20
S.G.IAZINO (C	CCS7 Signaling Termination, Per STP Port	1		UDB	PT8SX	138.41	† †		 			1	 	 	I	1
	CCS7 Signaling Usage, Per TCAP Message	1		UDB		0.0000916	1						1	1	1	
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	17.84	130.84	130.84	1				20.35	20.35	13.32	13.32
	CCS7 Signaling Connection, Per link (B link) (also known as D															
	link)	<u></u>		UDB	TPP++	17.84	130.84	130.84	<u> </u>		<u></u>	<u> </u>	20.35	20.35	13.32	13.32
	CCS7 Signaling Usage, Per ISUP Message			UDB		0.0000373										
	CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	352.30										
	Signaling Point Code, per Originating Point Code Establishment							·							1	
	or Change, per STP	<u> </u>	<u> </u>	UDB	CCAPO		121.77	121.77					20.35	20.35	13.32	13.32
CALLING NAM	IE (CNAM) SERVICE		-	001/	1	0.0010511									1	1
	CNAM for DB Owners, Per Query CNAM for Non DB Owners, Per Query	 	-	OQV OQV		0.0010541 0.0010541	ļ				1		 	 	1	1
	CNAM for Non DB Owners, Per Query CNAM (Non-Databs Owner), NRC, applies when using the	!	 	OUV	+	0.0010541	 		 			-	-	-		1
	Character Based User Interface (CHUI)			oqv	CDDCH		595.00	595.00					20.35	20.35	13.28	13.28
OPERATOR CA	ALL PROCESSING	1		·	000011		333.00	555.00	 			1	20.33	20.33	13.20	15.20
1	Oper. Call Processing - Oper. Provided, Per Min Using BST														1	
	LIDB					1.08							1			
	Oper. Call Processing - Oper. Provided, Per Min Using				1		i i		i l							
	Foreign LIDB	L	L			1.13	<u> </u>		<u> </u>		<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>	
	Oper. Call Processing - Fully Automated, per Call - Using BST								l i							
	LIDB					0.1010353										
	Oper. Call Processing - Fully Automated, per Call - Using	l]			
	Foreign LIDB				ļ	0.122818	ļļ								1	
	RATOR SERVICES	ļ			-											ļ
	Inward Operator Services - Verification, Per Minute	<u> </u>			+	1.03							 	ļ	-	ļ
	Inward Operator Services - Verification and Emergency Interrupt	l				4.00]				1	1	I	
BRANDING C	- Per Minute PERATOR CALL PROCESSING	 	<u> </u>		1	1.03							 	1	!	1
	bekator Call Processing based CLEC	<u> </u>	<u> </u>		+		 								-	-
racility	Recording of Custom Branded OA Announcement		-		CBAOS		1,555.00	1,553.00	7.03	7.03	1	1	19.99	19.99	19.99	19.99
	Loading of Custom Branded OA Announcement per shelf/NAV	-			ODAGO		1,333.00	1,333.00	1.03	1.03			15.55	19.99	19.99	19.99
			1	1	1				1		1	1	1		1	i
ļ	per OCN				CBAOL		240.71	240.71					19.99	19.99		

UNBUNDLED	NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	Recording of Custom Branded OA Announcement		1			1	First 1.555.00	Add'I 1.555.00	First	Add'l	SOMEC	SOMAN	SOMAN 19.99	SOMAN 19.99	SOMAN 19.99	SOMAN 19.99
L	recording of Custom Branded OA Announcement oldering of Custom Branded OA Announcement per shelf/NAV her OCN						240.71	240.71					19.99	19.99	19.99	19.9
	SISTANCE SERVICES					-	240.71	240.71			-		19.99	19.99		-
	DRY ASSISTANCE ACCESS SERVICE															
	Directory Assistance Access Service Calls, Charge Per Call					0.2286787										
	DRY ASSISTANCE CALL COMPLETION ACCESS SERVICE (I	DACC)				0.2200707										
	Directory Assistance Call Completion Access Service (DACC),	,,,,,,														
	Per Call Attempt					0.0364771										
	R SERVICES INTERCEPT ACCESS SERVICE				1		† †								İ	
	lumber Services Intercept Per Query					0.017793										
	DRY TRANSPORT (DT)						1									
	DT-Local Channel DS1					40.99	277.35	233.26	33.18	22.30			20.35	10.54	13.32	1.4
D	DT-DS1 Level Interoffice per mile					0.3562										
D	DT-DS1 Level Interoffice per facility termination					77.86	112.40	76.27	19.55	14.99			20.35	10.54	13.32	1.4
	SWA Common Transport per Directory Assistance Access Service Per Call					0.000271										
S	SWA Common Transport per Directory Assistance Access Service Per Call Per Mile					0.0000165										
	Access Tandem Switching Per Directory Assistance Access					0.0000100										
S	Service Per Call DT- Directory Assistance Interconnection Per Directory					0.0001875										
A	Assistance Service Call					0.00										
	OT-Installation NRC, Per Trunk or Signaling Connection						204.62	4.43	136.09	4.43			20.35	10.54	13.32	1.4
E	DT Local Channel DS1-Incremental Cost-Manual Svc Order vs Electronic						45.68	1.76	21.75	1.76						
E	DT Interoffice DS1-Incremental Cost-Manual Svc Order vs Electronic						20.35	21.09	9.80	10.54						
	SISTANCE SERVICES															
	DRY ASSISTANCE DATA BASE SERVICE (DADS)															
	Directory Assistance Data Base Service Charge Per Listing					0.0485										
	Directory Assistance Data Base Service, per month				DBSOF	104.13										
	RECTORY ASSISTANCE															
	Based CLEC															
	Recording and Provisioning of DA Custom Branded															
	nnouncement		1	AMT	CBADA		1,555.00	1,553.00	7.03	7.03			20.35	10.54	13.32	1.4
UNEP CL	oading of Custom Branded Announcement per Switch		1	AMT	CBADC		240.71	240.71					20.35	10.54		
							1,555.00	1,553.00	7.00	7.03			20.35	10.54	13.32	
	Recording of DA Custom Branded Announcement Loading of DA Custom Branded Announcement per Switch per					-	1,555.00	1,553.00	7.03	7.03			20.35	10.54	13.32	1.4
	CON						240.71	240.71					20.35	10.54		
SELECTIVE ROU							240.71	240.71					20.33	10.54		
	Selective Routing Per Unique Line Class Code Per Request Per		1								1					1
S	Switch				USRCR		179.60	179.60					20.35	20.35		
VIRTUAL COLLO				ANTEO	E 4 E		0.000.00	0.000.00					0.07	0.04	0.07	4.4
	/irtual Collocation - Application Cost			AMTES	EAF		2,633.00	2,633.00					2.07	2.81	0.67	1.4
	/irtual Collocation - Cable Installation Cost, per cable			AMTFS AMTFS	ESPCX ESPVX	3.91	1,749.00	1,749.00					2.07	2.81	0.67	1.4
	/irtual Collocation - Floor Space, per sq. ft. /irtual Collocation - Power, per fused amp			AMTFS	ESPAX	6.79										
	/irtual Collocation - Power, per rused amp			AIVITS	ESPAN	6.79										
	able			AMTFS	ESPSX	17.87										
				UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, AMTFS, UDL, UNCVX. UNCDX.												
V	/irtual Collocation - 2-wire Cross Connects (loop)			UNCNX	UEAC2	0.57	11.62	9.90	10.38	8.66			2.07	2.81	0.67	1.

LINBLINDI E	D NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Evhi	bit: C
ONDONDEL	D NETWORK ELLINENTS - Termessee										Svc Order	Svc Order	Incremental			
											Submitted			Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
											-		Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
<u> </u>							Nonrecurring		Monrocurring	Disconnect			088	Rates(\$)		
-						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
							THOL	Auu i	11130	Auui	JOHILO	JONAN	JOWAN	JOMAN	JOINAIN	JONIAN
				UEA,UHL,UCL,UDL,												
				AMTFS, UAL, UDN,												
	Virtual Collocation - 4-wire Cross Connects (loop)			UNCVX, UNCDX	UEAC4	0.57	11.81	10.04	10.44	8.67			2.07	2.81	0.67	1.41
				AMTFS,UDL12,												
				UDLO3, U1T48,												
				U1T12, U1T03,												
	Virtual Collocation - 2-Fiber Cross Connects			ULDO3, ULD12, ULD48, UDF	CNC2F	3.03	41.56	29.82	12.96	10.34			2.69	2.69	1.56	1.56
	Virtual Collocation - 2-Fiber Cross Connects			AMTFS,UDL12,	CNCZF	3.03	41.56	29.02	12.90	10.34			2.09	2.09	1.56	1.50
				UDLO3, U1T48,												
				U1T12, U1T03,												
				ULDO3, ULD12,												
	Virtual Collocation - 4-Fiber Cross Connects			ULD48, UDF	CNC4F	6.06	50.53	38.78	16.97	14.35			2.69	2.69	1.56	1.56
				USL,ULC,AMTFS,												
				ULR, UXTD1,												
				UNC1X, ULDD1,												
	Virtual collocation - Special Access & UNE, cross-connect per			U1TD1, USLEL,	0.10414											
	DS1			UNLD1	CNC1X	1.32	32.22	17.76	10.46	8.75			2.07	2.81	0.67	1.41
				USL,ULC,AMTFS,U E3, U1TD3, UXTS1,												
				UXTD3, UNC3X,												
				UNCSX, ULDD3,												
	Virtual collocation - Special Acess & UNE, cross-connect per			U1TS1, ULDS1,												
	DS3			UDLSX, UNLD3	CND3X	12.32	29.97	16.30	12.03	8.99			2.07	2.81	0.67	1.41
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable															
	Support Structure, per linear foot			AMTFS	VE1CB	0.0031										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax															
	Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0045										
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable			AMTFS	VE1CC		555.03						2.07	2.81	0.67	1.41
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AIVITS	VETCC		555.05				1		2.07	2.01	0.67	1.41
	Cable Support Structure, per cable			AMTFS	VE1CE		555.03						2.07	2.81	0.67	1.41
	Virtual Collocation Cable Records - per request			AMTFS	VE1BA		1,711.00									
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable															
	record			AMTFS	VE1BB		925.06									
	Virtual Collocation Cable Records - VG/DS0 Cable, per each	1		l	l											
\vdash	100 pair	<u> </u>	\vdash	AMTES	VE1BC		18.05	18.05	ļ		<u> </u>	<u> </u>	ļ	ļ		ļ
 	Virtual Collocation Cable Records - DS1, per T1TIE Virtual Collocation Cable Records - DS3, per T3TIE		\vdash	AMTFS AMTFS	VE1BD VE1BE		8.45 29.57	8.45 29.57	1		 	-				
 	Virtual Collocation Cable Records - DS3, per 131E Virtual Collocation Cable Records - Fiber Cable, per 99 fiber	1	\vdash	MIVITO	VEIDE		29.57	29.57	+	1	1	1	1	1		1
	records			AMTFS	VE1BF		279.42	279.42								
	Virtual collocation - Security Escort - Basic, per half hour	<u> </u>		AMTFS	SPTBX		33.15	20.44	1				2.07	2.81	0.67	1.41
	Virtual collocation - Security Escort - Overtime, per half hour			AMTFS	SPTOX		41.50	25.61					2.07	2.81	0.67	1.41
	Virtual collocation - Security Escort - Premium, per half hour			AMTFS	SPTPX		49.86	30.79					2.07	2.81	0.67	1.41
	Virtual collocation - Maintenance in CO - Basic, per half hour			AMTFS	CTRLX		30.64	30.64					2.07	2.81	0.67	1.41
	Virtual collocation - Maintenance in CO - Overtime, per half hour	<u> </u>	<u> </u>	AMTFS	SPTOM		35.77	35.77	ļ		<u> </u>	<u> </u>	2.07	2.81	0.67	1.41
	Virtual collocation - Maintenance in CO - Premium per half hour			AMTFS	SPTPM		40.90	40.90					2.07	2.81	0.67	1.41
VIRTUAL COL		1	\vdash	MIVITO	OF I FIVI		40.90	40.90	+	1	1	1	2.07	2.81	0.0/	1.41
TINTOAL COL	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-										1					
	Wire Analog - Res			UEPSR	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire Line Side PBX Trunk - Bus	<u> </u>		UEPSP	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire							· · · · · · · · · · · · · · · · · · ·								
	Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
1 1	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			LIEDOD	VE4D0	0.00	40.00	10.00					00.07	10.51	10.00	
	Analog Bus	1		UEPSB	VE1R2	0.30	19.20	19.20	1	<u> </u>	L	1	20.35	10.54	13.32	1.40

UNBU	NDLEI	NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Exhi	oit: C
3.150		TOTAL CONTRACTOR OF THE PROPERTY OF THE PROPER										Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Intori									Elec	Manually				Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m									P	p	Electronic-		Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																-100 101	
							Rec	Nonrecurring		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire															
		ISDN			UEPSX	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire															
		ISDN		<u> </u>	UEPTX	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
		Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire			HEDEV	\/E4D4	0.50	40.00	40.00					20.05	40.54	40.00	4.40
VIDTU		ISDN DS1 OCATION			UEPEX	VE1R4	0.50	19.20	19.20					20.35	10.54	13.32	1.40
VIKTUA	AL COLL																
		Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.57	11.62	9.90	10.38	8.66			19.99	19.99	19.99	19.99
DUVEIO	M COI	LOCATION			OLFSK, OLFSB	VLILO	0.57	11.02	9.90	10.36	8.00			15.55	19.99	19.99	19.99
1111010	AL 001	Physical Collocation-2 Wire Cross Connects (Loop) for Line				+											
		Splitting			UEPSR, UEPSB	PE1LS	0.0318	11.94	11.46					19.99	19.99	19.99	19.99
AIN SE	LECTIV	E CARRIER ROUTING															
		Regional Service Establishment			SRC	SRCEC		190,638.00						20.35		İ	
		End Office Establishment			SRC	SRCEO		317.55	317.55	3.19	3.19			20.35	20.35	13.28	13.28
		Query NRC, per query			SRC		0.0206047										
AIN - B		JTH AIN SMS ACCESS SERVICE															
		AIN SMS Access Service - Service Establishment, Per State,															
		Initial Setup			A1N	CAMSE		135.56	135.56					20.35	20.35	13.28	13.28
		AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		41.75	41.75					20.35	20.35	13.28	13.28
		AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		41.75	41.75					20.35	20.35	13.28	13.28
		AIN SMS Access Service - User Identification Codes - Per User				_											
		ID Code			A1N	CAMAU		96.63	96.63					20.35	20.35	13.28	13.28
		AIN SMS Access Service - Security Card, Per User ID Code,															
-		Initial or Replacement			A1N	CAMRC	0.0004	113.67	113.67					20.35	20.35	13.28	13.28
-		AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0024 0.0820123							-			
		AIN SMS Access Service - Session, Per Minute AIN SMS Access Service - Company Performed Session, Per		<u> </u>		-	0.0820123										
		Minute					2.27										
ΔIN - R	FLLSOL	JTH AIN TOOLKIT SERVICE				+	2.21										
All C		AIN Toolkit Service - Service Establishment Charge, Per State,															
		Initial Setup			CAM	BAPSC		132.04	132.04					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Training Session, Per Customer				BAPVX		7,915.00	7,915.00					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per						,	, , , , , , , , , , , , , , , , , , , ,								
	<u></u>	DN, Term. Attempt	<u></u>	<u></u>		BAPTT		31.21	31.21		<u></u>	<u> </u>	<u> </u>	20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
		DN, Off-Hook Delay				BAPTD		31.21	31.21					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per]	
		DN, Off-Hook Immediate				BAPTM		31.21	31.21					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DADTO								22.5-			
<u> </u>		DN, 10-Digit PODP				BAPTO		85.24	85.24					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP				BAPTC		05.04	05.04					20.35	20.35	12.00	13.28
—				-		DAF IC		85.24	85.24	-				20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code				BAPTF		85.24	85.24					20.35	20.35	13.28	13.28
—	-	AIN Toolkit Service - Query Charge, Per Query	-			וו ואכ	0.0211882	05.24	05.24	1				20.33	20.35	13.20	13.20
		AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit	1			1	0.0211002			1		1	1	I	1	 	
		Subscription, Per Node, Per Query					0.0054774							1			
		AIN Toolkit Service - SCP Storage Charge, Per SMS Access						i			l			1			İ
	<u></u>	Account, Per 100 Kilobytes	<u></u>	<u></u>		1	1.50	<u> </u>			<u></u>	<u> </u>	<u> </u>	L		<u></u>	<u> </u>
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
		Subscription			CAM	BAPMS	17.43	33.52	33.52		<u></u>			20.35	20.35	13.28	13.28
		AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
		Subscription			CAM	BAPLS	0.1321116	36.23	36.23					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service			L	L		l						I			l
		Subscription			CAM	BAPDS	17.35	33.52	33.52	ļ				20.35	20.35	13.28	13.28
		AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit			CAM	DADEO	0.0544465	20.00	20.00					00.05	20.65	10.00	40.00
		Service Subscription	l	<u> </u>	CAM	BAPES	0.0511435	36.23	36.23		l			20.35	20.35	13.28	13.28

UNR	UNDI FI	NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Fyhil	oit: C
0140	-14DEE	ALL TOTAL LELINEITIO - Tellilessee										Svc Order	Svc Order	Incremental		Incremental	
1			l									Submitted	Submitted		Charge -	Charge -	Charge -
			1									Elec		_	Manual Svc		Manual Svc
CATE	GORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						- ()			per LSK	per LOK	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
														151	Add I	DISC ISL	DISC Add I
							Rec	Nonrecurring		Nonrecurring	Disconnect		•	oss	Rates(\$)	•	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ENHA	NCED EX	TENDED LINK (EELs)															
		New Density Zone 1 EELs are available in the following MSA:					Atlanta, Ga; Ne	w Orleans, LA,									
	NOTE:	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-	High P	oint, N	C; and Nashville, TN												
		In all states, EEL network elements shown below also apply t												UNEs.(Non-re	curring rates	do not apply	.)
		In All States the EEL network elements apply to ordinarily cor				tch As Is Cha	arge.) When o	rdering ordinar	ily combined I	network elemei	nts, Non-recur	ring rates d	o apply.				
	2-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												
		First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport															
		Combination - Zone 1		1	UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
	1	Transport Combination - Zone 2		2	UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
1	1	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed	1	1]		_			<u> </u>	<u> </u>	<u> </u>]	
		Transport Combination - Zone 3		3	UNCVX	UEAL2	28.28	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		Interoffice Transport - Dedicated - DS1 combination - Per Mile															
	1	per month			UNC1X	1L5XX	0.3562										
1	1	Interoffice Transport - Dedicated - DS1 combination - Facility	1	1]		_			<u> </u>	<u> </u>	<u> </u>]	
<u> </u>	<u> </u>	Termination per month		<u> </u>	UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
		DS1 Channelization System Per Month			UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74						
		Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.91	5.70	4.42								
		Each Additional 2-Wire VG Loop(SL 2) in the same DS1															
		Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		Each Additional 2-Wire VG Loop(SL2) in the same DS1															
		Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		Each Additional 2-Wire VG Loop(SL2) in the same DS1															
		Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	28.28	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		Voice Grade COCI - DS1 to DS0 Channel System combination -															
		per month			UNCVX	1D1VG	0.91	5.70	4.42								
		Nonrecurring Currently Combined Network Elements Switch -As-															
		Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
	4-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												
		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
		Transport Combination - Zone 1		1	UNCVX	UEAL4	24.70	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		_													
		Transport Combination - Zone 2		2	UNCVX	UEAL4	32.26	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		_													
<u> </u>		Transport Combination - Zone 3		3	UNCVX	UEAL4	42.18	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
		Interoffice Transport - Dedicated - DS1 combination - Per Mile															
		Per Month		<u> </u>	UNC1X	1L5XX	0.3562										
1	1	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month	1		LINICAV	LIATEA	77.00	474.04	440.40	70.07	20.00			20.05	04.00	0.00	40.54
<u> </u>	1	No. 121	1	1	UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90	1		20.35	21.09	9.80	10.54
1	1	Channelization - Channel System DS1 to DS0 combination Per Month	1	1	UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74			Ì	Ì	Ì	
-	+	Voice Grade COCI - DS1 to DS0 Channel System combination -	-	-	UNCIA	IVIQ I	80.77	105.76	14.48	3.04	2.74			-	-	-	-
1	1	per month	l		UNCVX	1D1VG	0.91	5.70	4.42	1							
-	1	Additional 4-Wire Analog Voice Grade Loop in same DS1	1		OIVOVA	טועט	0.91	5.70	4.42	 		}		1	1	1	1
1	1	Interoffice Transport Combination - Zone 1	1	1	UNCVX	UEAL4	24.70	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
—	+	Additional 4-Wire Analog Voice Grade Loop in same DS1	1	+	0110 1/	JLAL4	24.70	100.70	35.47	12.34	10.00	1		20.35	21.09	9.00	10.54
1	1	Interoffice Transport Combination - Zone 2	l	2	UNCVX	UEAL4	32.26	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	+	Additional 4-Wire Analog Voice Grade Loop in same DS1	-		00 7/	J_/ (L-7	02.20	100.70	00.47	72.54	10.00			20.00	21.03	5.00	10.04
1	1	Interoffice Transport Combination - Zone 3	l	3	UNCVX	UEAL4	42.18	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	1	Voice Grade COCI - DS1 to DS0 Channel System combination -	1	Ť			.2.70		33. H	. 2.54	.0.50	1		20.00	255	5.50	
1	1	per month	1	1	UNCVX	1D1VG	0.91	5.70	4.42	I				Ì	Ì	Ì	
	1	Nonrecurring Currently Combined Network Elements Switch -As-	1	†			3.51	30	72	t				1	1	1	
	1	Is Charge	1		UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
	4-WIRE	56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE			1			1						2.30	
		First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice			` ,												
	1	Transport Combination - Zone 1	1	1	UNCDX	UDL56	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	İ	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice															
1	1	Transport Combination - Zone 2	1	2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
•												•	•	•	•	•	

UNBUNDL	ED NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Exhil	bit: C
											Svc Order	Svc Order	Incremental	Incremental	Incremental	
											Submitted	Submitted		Charge -	Charge -	Charge -
		Interi	_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonrecurring		Nonrecurring	n Disconnect	1	1	088	Rates(\$)		
					+	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice						11130	Auu i	11130	Auu	JOINEC	JOHIAN	JOINAIN	JONAN	JONIAN	JOMAN
	Transport Combination - Zone 3		3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															
	Per Month			UNC1X	1L5XX	0.3562										
	Interoffice Transport - Dedicated - DS1 - combination Facility															
	Termination Per Month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	Channelization - Channel System DS1 to DS0 combination Per															
	Month			UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74						
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per															
	month (2.4-64kbs)			UNCDX	1D1DD	0.91	5.70	4.42								
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1									40.00						
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			UNCDA	UDLS6	40.61	100.76	33.47	72.94	10.00			20.33	21.09	9.60	10.54
1 1	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	OCU-DP COCI (data) - DS1 to DS0 Channel System -		3	ONCDA	ODLSO	55.11	100.70	33.47	72.54	10.00			20.33	21.09	9.60	10.54
	combination per month (2.4-64kbs)			UNCDX	1D1DD	0.91	5.70	4.42								
	Nonrecurring Currently Combined Network Elements Switch -As-			0.10271	.5.55	0.01	0.70	2								
	Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WIF	RE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL)												
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice			, ,												
	Transport Combination - Zone 1		1	UNCDX	UDL64	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 2		2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 3		3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			LINIOAN	1L5XX	0.0500										
	Per Month Interoffice Transport - Dedicated - DS1 combination - Facility			UNC1X	1L5XX	0.3562			1							
	Termination Per Month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	Channelization - Channel System DS1 to DS0 combination Per			UNCIA	01111	77.00	171.24	113.12	70.07	30.90			20.33	21.09	9.00	10.54
	Month			UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74			20.35	21.09	9.80	10.54
	OCU-DP COCI (data) - DS1 to DS0 Channel System			011017		00.11	100.70		0.0 .	2			20.00	21.00	0.00	
	combination - per month (2.4-64kbs)			UNCDX	1D1DD	0.91	5.70	4.42								
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1															
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1							-]	
$\sqcup \sqcup \sqcup$	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1		_													
$\vdash \vdash \vdash$	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	0.91	5.70	4.42	1				1	1	1	
\vdash	Combination - per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-	-	 	OINCDV	טטוטו	0.91	5.70	4.42	 				-	-	-	
]]	Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WIF	IS CHARGE RE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE	ROFFI	CE TR	ANSPORT (FFI)	314000		32.13	24.02	5.12	5.12			20.35	21.09	9.00	10.54
1 1711	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice			()					t	1			1	1	1	
	Transport - Zone 1		1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice															
	Transport - Zone 2		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice							<u> </u>								
	Transport - Zone 3		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS1 combination - Per Mile								I				1	1	1	
igwdow	Per Month		<u> </u>	UNC1X	1L5XX	0.3562			-							
	Interoffice Transport - Dedicated - DS1 combination - Facility			LINICAY	LIATE4	77.00	474.01	440.40	70.00	00.00			20.65	04.65	0.00	10-
L I	Termination Per Month		 	UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
				1	1		1		1		1	•	•		•	1
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54

<u> UNBUND</u> LE	ED NETWORK ELEMENTS - Tennessee													ment: 2	Exhi	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
					ļ	Rec	Nonrecurring		Nonrecurring		001150	001111		Rates(\$)	001141	
	First DS1Loop in DS3 Interoffice Transport Combination - Zone				-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	1		1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.5
	First DS1Loop in DS3 Interoffice Transport Combination - Zone															
	2		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.5
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.5
	Interoffice Transport - Dedicated - DS3 combination - Per Mile		J	ONOTA	OOLXX	90.59	220.40	101.74	19.01	24.00			20.55	21.03	3.00	10.0
	Per Month			UNC3X	1L5XX	2.34										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per															
	month DS3 to DS1 Channel System combination per month			UNC3X UNC3X	U1TF3 MQ3	854.97 222.98	482.01 156.02	153.81 49.41	64.43 17.12	35.43 6.77			20.35	21.09	9.80	10.5
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	17.58	5.70	4.42	17.12	0.77						+
	Additional DS1Loop in DS3 Interoffice Transport Combination -				1		0.70									
	Zone 1		1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.5
	Additional DS1Loop in DS3 Interoffice Transport Combination -		_	LINGAV	LICLYY	75.40	220 40	464.74	70.07	04.00			20.25	24.00	0.00	40.5
	Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination -		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.5
	Zone 3		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.5
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	17.58	5.70	4.42								
	Nonrecurring Currently Combined Network Elements Switch -As-															
2 WID	Is Charge RE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	EDOE	ICE TO	UNC3X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.5
Z-WIR	2-WireVG Loop used with 2-wire VG Interoffice Transport	EROFF	ICE II	RANSPORT (EEL)	-											+
	Combination - Zone 1		1	UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.5
	2-WireVG Loop used with 2-wire VG Interoffice Transport															
	Combination - Zone 2		2	UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.5
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	28.28	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.5
	Interoffice Transport - Dedicated - 2-wire VG combination - Per		3	UNCVX	ULALZ	20.20	100.70	33.47	72.94	10.00			20.33	21.09	9.80	10.0
	Mile Per Month			UNCVX	1L5XX	0.0174										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade															
	combination - Facility Termination per month			UNCVX	U1TV2	21.79	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.5
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCVX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.5
4-WIR	RE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EROFF	ICE T		Citoco		02.70	21.02	0.12	02			20.00	200	0.00	10.0
	4-WireVG Loop used with 4-wire VG Interoffice Transport															1
	Combination - Zone 1		1	UNCVX	UEAL4	24.70	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.5
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	32.26	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.5
	4-WireVG Loop used with 4-wire VG Interoffice Transport			ONOVA	OLAL4	32.20	100.70	33.47	72.54	10.00			20.55	21.03	3.00	10.0
	Combination - Zone 3		3	UNCVX	UEAL4	42.18	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.5
	Interoffice Transport - Dedicated - 4-wire VG combination - Per															
	Mile Per Month Interoffice Transport - Dedicated - 4- Wire Voice Grade			UNCVX	1L5XX	0.0174										
	combination - Facility Termination per month			UNCVX	U1TV4	27.30	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.5
	Nonrecurring Currently Combined Network Elements Switch -As-			OTTO VA	0	27.00	70.00	11.00	00.02	01.00			20.00	21100	0.00	10.0
	Is Charge			UNCVX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.5
DS3 D	DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRA	NSPOF	RT (EEL)												
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month			UNC3X	1L5ND	9.19										
	High Capacity Unbundled Local Loop - DS3 combination -			O1400A	ILUIND	5.19										
	Facility Termination per month			UNC3X	UE3PX	373.47	240.23	180.87	106.78	45.24			20.35	21.09	9.80	10.5
	Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	2.34							_			
	Interoffice Transport - Dedicated - DS3 combination - Facility			UNC3X	U1TF3	854.97	482.01	153.81	64.43	35.43			20.35	21.09	9.80	10.5
	Termination per per month Nonrecurring Currently Combined Network Elements Switch -As-			UNUSA	UTIF3	854.97	482.01	153.81	64.43	35.43			20.35	∠1.09	9.80	10.5
	Is Charge		1	UNC3X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.5
0704	DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROF	FICE TE	ANSP													1

	UNCSX L	USOC	Rec		RATES(\$)			Svc Order Submitted Elec	Submitted	Incremental Charge -	Charge -		oit: C Incremental
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Nonrecurring Currently Combined Network Elements Switch - As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 3 1 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	UNCSX L		Rec		RATES(\$)							Charge -	•
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Nonrecurring Currently Combined Network Elements Switch - As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 3 1 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	UNCSX L		Rec		RATES(\$)								Charge -
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Nonrecurring Currently Combined Network Elements Switch - As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 3 1- First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combinatio	UNCSX L		Rec		RATES(\$)		j			Manual Svc	Manual Svc	Manual Svc	Manual Svc
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Nonrecurring Currently Combined Network Elements Switch - As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 to DS0 combination - per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Jewire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS	UNCSX L		Rec					per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
Mile per month High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As-ls Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Channelization - Channel System DS1 to DS0 combination - Per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport	UNCSX L		Rec					per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
Mile per month High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 1-ETS DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System combination per month Additional DS	UNCSX L		Rec				ļ			1st	Add'l	Disc 1st	Disc Add'l
Mile per month High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch - As-Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 4-WIRE DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 STS1 to	UNCSX L		Rec							151	Add I	DISC ISI	DISC Add I
Mile per month High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 1-ETS DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System combination per month Additional DS	UNCSX L		Rec	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)	•	•
Mile per month High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 1-ETS DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System combination per month Additional DS	UNCSX L		,-	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Mile per month High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 1-ETS DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System combination per month Additional DS	UNCSX L												
Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch - As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL.) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 1 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 2 2 2 2 2 2 2 2 2		1L5ND	9.19		ŀ						1 1		ł
Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch - As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL.) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 1 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 2 2 2 2 2 2 2 2 2													i
per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 1 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 1 3 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interoffice Transport - Dedicated - STS1 combination - Zone 1 1 1	UNCSX 1	UDLS1	394.56	240.23	180.87	106.78	45.24			20.35	21.09	9.80	10.54
Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Channelization - Channel System DS1 to DS0 combination - per month Channelization - Channel System DS1 to DS0 combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch - As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month DS3 Interace Unit (DS1 COCI) combination per month DS3 Interace Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1	UNCSX 1												í
Termination per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TR First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interace Unit (DS1 COCI) combination per month DS3 Interace Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		1L5XX	2.34		ŀ						1 1		ł
Nonrecurring Currently Combined Network Elements Switch -As-													í
Scharge 2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)	UNCSX	U1TFS	849.30	482.01	153.81	64.43	35.43			20.35	21.09	9.80	10.54
2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1													í
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNCSX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone month Nonrecurring Currently Combined Network Elements Switch -As- ls Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1													
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1					ļ		ļ				1 1		f
Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNCNX L	U1L2X	22.22	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1						1					1 7		
Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNCNX L	U1L2X	29.02	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		\neg				1 7					ı – – –		1
Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintation - per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		U1L2X	37.95	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
Termination per month Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-ls Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNC1X 1	1L5XX	0.3562								<u> </u>		<u> </u>
Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1					ŀ						1 1		ł
per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNC1X L	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -Aslis Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - 2					ļ		ļ				1 1		f
combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNC1X N	MQ1	80.77	105.76	14.48	3.04	2.74			20.35	21.09	9.80	10.54
Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1					ŀ						1 1		ł
Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion - per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNCNX L	UC1CA	3.24	5.70	4.42					20.35	21.09	9.80	10.54
Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1					ļ		ļ				1 1		f
Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination-per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1	UNCNX L	U1L2X	22.22	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintation- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1					ļ		ļ				1 1		f
Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per month Nonrecurring Currently Combined Network Elements Switch -As- is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	UNCNX L	U1L2X	29.02	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintation - per month Nonrecurring Currently Combined Network Elements Switch -Asis Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - 2 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - 2 2 3 Additional DS1Loop in STS1 Interoffice Transport Combination - 2 2 3 Intercomplete Transport - Dedicated - STS1 combination - Facility Termination - 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3					ļ		ļ				1 1		f
combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1	UNCNX L	U1L2X	37.95	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
Nonrecurring Currently Combined Network Elements Switch -As- Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1							ļ						f
Is Charge 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TO FIRST DS1 Loop in STS1 Interoffice Transport Combination - Zone 1	UNCNX L	UC1CA	3.24	5.70	4.42					20.35	21.09	9.80	10.54
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TF First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1							2.42						
First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1		UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
Zone 1	RANSPORT (EEL)												
First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	LINICAV	LICL VV		000.40	404.71	70.0-	04.00			00.0-	04.00	0.00	10.51
Zone 2 2	UNC1X L	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	LINICAV		75 40	000.40	404.74	70.07	04.00			20.25	04.00	0.00	10.51
Zone 3 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1	UNC1X L	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	LINICAY	LICL VV	98.59	220,42	404.74	70.07	04.00			20.05	04.00	0.00	40.54
Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1	UNC1X L	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
Interoffice Transport - Dedicated - STS1 combination - Facility Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1	LINCOV	41.577	0.04		ŀ						1 1		ł
Termination STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1	UNCSX 1	1L5XX	2.34								\vdash		
STS1 to DS1 Channel System conbination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1	LINICOV	U1TFS	849.30	492.04	152.04	64.40	25 42			20.25	21.09	0.00	10.54
DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1		MQ3	222.98	482.01 156.02	153.81 49.41	64.43 17.12	35.43 6.77			20.35 20.35	21.09	9.80 9.80	10.54 10.54
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1		UC1D1	17.58	5.70	49.41	11.12	0.77			20.35	21.09	9.80	10.54
Zone 1	OINO IA	COIDI	17.30	5.70	4.42	\vdash				20.33	21.09	9.00	10.54
	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	551A	332701	51.13	220.70	.01.74	13.01	24.00			20.00	21.03	3.00	10.54
Zone 2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
Additional DS1Loop in STS1 Interoffice Transport Combination -	551A	555701	75.40	£20.40	.01.74	13.01	24.00			20.00	21.03	3.00	10.54
Zone 3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
DS3 Interface Unit (DS1 COCI) combination per month		UC1D1	17.58	5.70	4.42	10.07	24.00			20.35	21.09	9.80	10.54
Nonrecurring Currently Combined Network Elements Switch -As-			17.00	5.75	7.72	 				20.00	21.00	0.00	10.54
Is Charge	UNCSX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSF				02.70	27.02	5.12	0.12			20.00	21.00	5.50	10.04
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport				-		 							
Combination - Zone 1			31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54

	IDLED NETWORK ELEMENTS - Tennessee			Т	1	T							Attachn			bit: C
CATEGOR	PRY RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.0174										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination			UNCDX	U1TD5	21.19	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCDX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.5
4-1	-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROI	FFICE T	RANSI	PORT (EEL)												
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile			UNCDX	1L5XX	0.0174										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination			UNCDX	U1TD6	21.19	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.5
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCDX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.5
	NAL NETWORK ELEMENTS															
	When used as a part of a currently combined facility, the non-recurr When used as ordinarily combined network elements in All States, the															
	Ionrecurring Currently Combined Network Elements "Switch As Is"					AS IS Charge	does not.									
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.5
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - 56/64 kbps			UNCDX	UNCCC		52.73	24.62	9.12	9.12						
	Nonrecurring Currently Combined Network Elements Switch -As-												20.35	21.09	9.80	10.5
	Is Charge - DS1			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	
	Is Charge - DS1 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3			UNC1X UNC3X	UNCCC				9.12 9.12	9.12 9.12						10.54
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - STS1			UNC3X UNCSX	UNCCC		52.73	24.62					20.35	21.09	9.80	10.5
NC	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing period	i - Belo		UNC3X UNCSX =one month, DS3 ar	UNCCC UNCCC and above=fou		52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12	9.12 9.12			20.35 20.35 20.35	21.09 21.09 21.09	9.80 9.80 9.80	10.5 10.5
NC	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing perioc	i - Belo	1	UNC3X UNCSX one month, DS3 ar UNCVX	UNCCC UNCCC nd above=fou	17.18	52.73 52.73 52.73 108.76	24.62 24.62 24.62 35.47	9.12 9.12 72.94	9.12 9.12 10.86			20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80	10.5 10.5 10.5
NC	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- ls Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2	i - Belo	1 2	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX	UNCCC UNCCC nd above=fou ULDV2 ULDV2	17.18 22.44	52.73 52.73 52.73 108.76 108.76	24.62 24.62 24.62 35.47 35.47	9.12 9.12 72.94 72.94	9.12 9.12 10.86 10.86			20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80	10.5 10.5 10.5 10.5 10.5
NC	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing period Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3	d - Belo	1 2 3	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX	UNCCC UNCCC nd above=fou ULDV2 ULDV2 ULDV2	17.18 22.44 29.34	52.73 52.73 52.73 108.76 108.76 108.76	24.62 24.62 24.62 35.47 35.47 35.47	9.12 9.12 72.94 72.94 72.94	9.12 9.12 10.86 10.86 10.86			20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80	10.5 10.5 10.5 10.5 10.5 10.5
NO	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing period Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1	i - Belo	1 2 3 1	UNC3X UNCSX one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV2 ULDV2 ULDV4	17.18 22.44 29.34 18.18	52.73 52.73 52.73 108.76 108.76 108.76 108.76	24.62 24.62 24.62 35.47 35.47 35.47 35.47	9.12 9.12 72.94 72.94 72.94 72.94	9.12 9.12 10.86 10.86 10.86 10.86			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5 10.5 10.5 10.5 10.5 10.5 10.5
NG	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing period Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3	i - Belo	1 2 3 1 2	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX	UNCCC UNCCC nd above=fou ULDV2 ULDV2 ULDV2	17.18 22.44 29.34 18.18 23.74	52.73 52.73 52.73 108.76 108.76 108.76	24.62 24.62 24.62 35.47 35.47 35.47	9.12 9.12 72.94 72.94 72.94	9.12 9.12 10.86 10.86 10.86			20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80	10.5 10.5 10.5 10.5 10.5 10.5 10.5 10.5
NO	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2	i - Belo	1 2 3 1 2	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UNCCC UNCCC nd above=fou ULDV2 ULDV2 ULDV2 ULDV2 ULDV4 ULDV4	17.18 22.44 29.34 18.18	52.73 52.73 52.73 108.76 108.76 108.76 108.76	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47	9.12 9.12 72.94 72.94 72.94 72.94 72.94	9.12 9.12 10.86 10.86 10.86 10.86			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5-
NC	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2	i - Belo	1 2 3 1 2 3	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UNCCC UNCCC Id above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDV4	17.18 22.44 29.34 18.18 23.74 31.05	52.73 52.73 52.73 108.76 108.76 108.76 108.76 108.76	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47	9.12 9.12 72.94 72.94 72.94 72.94 72.94 72.94 72.94	9.12 9.12 10.86 10.86 10.86 10.86 10.86 10.86			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5-
NC	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 NOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated -2-Wire Voice Grade Zone 1 Local Channel - Dedicated -2-Wire Voice Grade Zone 2 Local Channel - Dedicated -2-Wire Voice Grade Zone 3 Local Channel - Dedicated -4-Wire Voice Grade Zone 1 Local Channel - Dedicated -4-Wire Voice Grade Zone 2 Local Channel - Dedicated -4-Wire Voice Grade Zone 2 Local Channel - Dedicated -4-Wire Voice Grade Zone 3 Local Channel - Dedicated -5S1 per month Zone 1 Local Channel - Dedicated -DS1 Per Month Zone 2 Local Channel - Dedicated -DS1 - Per Month Zone 2	d - Belo	1 2 3 1 2 3 1	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCXV UNCXV UNCXV UNC1X UNC1X UNC1X UNC1X	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDV4 ULDV4 ULDF1 ULDF1	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89	52.73 52.73 52.73 108.76 108.76 108.76 108.76 108.76 228.40	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74	9.12 9.12 72.94 72.94 72.94 72.94 72.94 72.94 79.87	9.12 9.12 10.86 10.86 10.86 10.86 10.86 24.88			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5-
NC	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 3 Local Channel - Dedicated - DS1 per month Zone 1 Local Channel - Dedicated - DS1 Per Month Zone 2 Local Channel - Dedicated - DS1 Per Month Zone 3 Local Channel - Dedicated - DS1 - Per Month Zone 3 Local Channel - Dedicated - DS1 - Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month	d - Belo	1 2 3 1 2 3 1 2	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCXV UNCXV UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDV4 ULDF1 ULDF1 11.5NC	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15	52.73 52.73 52.73 52.73 108.76 108.76 108.76 108.76 228.40 228.40	24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74	9.12 9.12 72.94 72.94 72.94 72.94 72.94 72.94 79.87 79.87	9.12 9.12 10.86 10.86 10.86 10.86 24.88 24.88 24.88			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5-
NC	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 Norrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing period Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 3 Local Channel - Dedicated - DS1 per month Zone 1 Local Channel - Dedicated - DS1 - Per Month Zone 2 Local Channel - Dedicated - DS1 - Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination	i - Belo	1 2 3 1 2 3 1 2	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCXV UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 ULDF3 ULDF3	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15 611.30	52.73 52.73 52.73 108.76 108.76 108.76 108.76 108.76 108.76 228.40	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74	9.12 9.12 72.94 72.94 72.94 72.94 72.94 79.87 79.87	9.12 9.12 10.86 10.86 10.86 10.86 10.86 24.88			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5- 10.5-
NC	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 NoTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated -2-Wire Voice Grade Zone 1 Local Channel - Dedicated -2-Wire Voice Grade Zone 2 Local Channel - Dedicated -2-Wire Voice Grade Zone 3 Local Channel - Dedicated -4-Wire Voice Grade Zone 1 Local Channel - Dedicated -4-Wire Voice Grade Zone 2 Local Channel - Dedicated -4-Wire Voice Grade Zone 2 Local Channel - Dedicated -4-Wire Voice Grade Zone 3 Local Channel - Dedicated -0-S1 per month Zone 1 Local Channel - Dedicated -DS1 per Month Zone 2 Local Channel - Dedicated -DS1 - Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1- Per Mile per month	i - Belo	1 2 3 1 2 3 1 2 3	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCXV UNCXV UNCXV UNCXV UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNCSX	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 ULDF3 11.5NC	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15 611.30 7.15	52.73 52.73 52.73 108.76 108.76 108.76 108.76 108.76 228.40 228.40 228.40	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74 161.74 161.74	9.12 9.12 72.94 72.94 72.94 72.94 72.94 79.87 79.87 79.87	9.12 9.12 10.86 10.86 10.86 10.86 10.86 24.88 24.88 24.88			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.54 10.54 10.54 10.54 10.55 10.55 10.55 10.55 10.55 10.55
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 Norrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing period Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 3 Local Channel - Dedicated - DS1 per month Zone 1 Local Channel - Dedicated - DS1 - Per Month Zone 2 Local Channel - Dedicated - DS1 - Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination	i - Belo	1 2 3 1 2 3 1 2 3	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCXV UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 ULDF3 ULDF3	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15 611.30	52.73 52.73 52.73 52.73 108.76 108.76 108.76 108.76 228.40 228.40	24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74	9.12 9.12 72.94 72.94 72.94 72.94 72.94 72.94 79.87 79.87	9.12 9.12 10.86 10.86 10.86 10.86 24.88 24.88 24.88			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5 10.5 10.5 10.5 10.5 10.5 10.5 10.5
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - DS1 per month Zone 1 Local Channel - Dedicated - DS1 Per Month Zone 2 Local Channel - Dedicated - DS1- Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - STS-1- Per Mile per month Local Channel - Dedicated - STS-1- Per Mile per month Local Channel - Dedicated - STS-1- Per Mile per month	i - Belo	1 2 3 1 2 3 1 2 3	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCXV UNCXV UNCXV UNCXV UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNCSX	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 ULDF3 11.5NC	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15 611.30 7.15	52.73 52.73 52.73 108.76 108.76 108.76 108.76 108.76 228.40 228.40 228.40	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74 161.74 161.74	9.12 9.12 72.94 72.94 72.94 72.94 72.94 79.87 79.87 79.87	9.12 9.12 10.86 10.86 10.86 10.86 10.86 24.88 24.88 24.88			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80	10.5 10.5 10.5 10.5 10.5 10.5 10.5 10.5
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 NOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - DS1 per month Zone 1 Local Channel - Dedicated - DS1 per Month Zone 2 Local Channel - Dedicated - DS1 - Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1 - Per Mile per month Local Channel - Dedicated - STS-1 - Facility Termination NULTIPLEXERS Channelization - DS1 to DS0 Channel System - per	i - Belo	1 2 3 1 2 3 1 2 3	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNCSX UNCSX	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 ULDF3 ULDF3 ULDF3 ULDFS	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15 611.30 7.15 599.59	52.73 52.73 52.73 108.76 108.76 108.76 108.76 228.40 228.40 228.40 595.37 588.07	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74 161.74 161.74 297.20	9.12 9.12 72.94 72.94 72.94 72.94 72.94 79.87 79.87 79.87 215.82	9.12 9.12 10.86 10.86 10.86 10.86 24.88 24.88 24.88 151.15			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 11.49	10.54 10.54 10.54 10.55 10.55 10.54 10.54 10.54 10.55 10.55 10.54 10.54
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 NoTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - DS1 per month Zone 1 Local Channel - Dedicated - DS1 per Month Zone 2 Local Channel - Dedicated - DS1 - Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination Local Channel - Dedicated - STS-1 - Per Mile per month Local Channel - Dedicated - STS-1 - Facility Termination IULTIPLEXERS Channelization - DS1 to DS0 Channel System	d - Belo	1 2 3 1 2 3 1 2 3	UNC3X UNCSX =One month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCYX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNCSX UNCSX UNCSX	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV4 ULDV4 ULDV4 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 ULDF1 ULDF3 1L5NC ULDFS	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15 611.30 7.15 599.59	52.73 52.73 52.73 108.76 108.76 108.76 108.76 108.76 228.40 228.40 228.40 228.40 24.40 25.37 588.07 141.67	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47 35.47 35.47 35.47 35.47 35.47 161.74 161.74 297.20 77.11	9.12 9.12 72.94 72.94 72.94 72.94 72.94 79.87 79.87 79.87 215.82	9.12 9.12 10.86 10.86 10.86 10.86 24.88 24.88 24.88 151.15			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 11.49	10.54 10.54 10.54 10.54 10.54 10.54 10.55 10.55 10.55 10.55 10.55 10.55 10.55 10.55 10.55 10.55 10.55 10.55
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS3 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1 IOTE: Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade Zone 1 Local Channel - Dedicated - 2-Wire Voice Grade Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 1 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 2 Local Channel - Dedicated - 4-Wire Voice Grade Zone 3 Local Channel - Dedicated - 4-Wire Voice Grade Zone 3 Local Channel - Dedicated - DS1 per month Zone 1 Local Channel - Dedicated - DS1 Per Month Zone 2 Local Channel - Dedicated - DS1- Per Month Zone 3 Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - STS-1- Per Mile per month Local Channel - Dedicated - STS-1- Per Mile per month Local Channel - Dedicated - STS-1- Per Mile per month Local Channel - Dedicated - STS-1- Facility Termination IULTIPLEXERS Channelization - DS1 to DS0 Channel System - per month (2.4-64kbs)	J - Belo	1 2 3 1 2 3 1 2 3	UNC3X UNCSX =one month, DS3 ar UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNCSX UNCSX	UNCCC UNCCC d above=fou ULDV2 ULDV2 ULDV2 ULDV4 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 ULDF3 ULDF3 ULDF3 ULDFS	17.18 22.44 29.34 18.18 23.74 31.05 36.24 47.33 61.89 7.15 611.30 7.15 599.59	52.73 52.73 52.73 108.76 108.76 108.76 108.76 228.40 228.40 228.40 595.37 588.07	24.62 24.62 24.62 35.47 35.47 35.47 35.47 35.47 161.74 161.74 161.74 297.20	9.12 9.12 72.94 72.94 72.94 72.94 72.94 79.87 79.87 79.87 215.82	9.12 9.12 10.86 10.86 10.86 10.86 24.88 24.88 24.88 151.15			20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09 21.09	9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 9.80 11.49	10.5 10.5 10.5 10.5 10.5 10.5 10.5 10.5

UNBUNDL	ED NETWORK ELEMENTS - Tennessee				•							T -		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring	Disconnect		•	oss	Rates(\$)	•	•
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	STS1 to DS1 Channel System per month			UXTS1	MQ3	222.98	308.03	108.47	44.47	42.62			20.35	21.09	9.80	9.80
	DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	17.58	6.07	4.66					20.35	9.80	11.49	1.18
	DS3 Interface Unit (DS1 COCI) used with Local Channel per															
	month	_		ULDD1	UC1D1		6.07	4.66					20.35	9.80	11.49	1.18
	D LOCAL EXCHANGE SWITCHING(PORTS) ange Ports	-			_											-
	ange Forts E: Although the Port Rate includes all available features in GA,	KVIA	2 TNI 4	ho docirod foaturos	s will nood to b	o ordorod usii	na rotail LISOCe				1					
	RE VOICE GRADE LINE PORT RATES (RES)	KI, LA	ο 11 1 , ι	lie desired realures	will fleed to i	l	lg retail 0300s	1								1
	Exchange Ports - 2-Wire Analog Line Port- Res.		1	UEPSR	UEPRL	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.	<u> </u>	<u> </u>	UEPSR	UEPRO	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled TN extended local							-		-						
	dialing parity Port with Caller ID - Res.			UEPSR	UEPAQ	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Plus															
	with Caller ID - Res (AC7)			UEPSR	UEPAH	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling	9		LIEDOD	LIEDAIA	4.00	0.00	0.40	0.00	0.00			00.05	40.54	40.00	4.40
	port with Caller ID - Res (F2R)			UEPSR	UEPAK	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling	9		LIEDOD	LIEDAL	4.00	0.00	0.40	2.00	2.02			20.25	40.54	40.00	4.40
	port with Caller ID - Res (TACER) Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling	_		UEPSR	UEPAL	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	port with Caller ID - Res (TACSR)	9		UEPSR	UEPAM	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
-	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling	+	1	OLFSK	OLFAW	1.09	9.93	5.15	3.00	2.32	1		20.33	10.34	13.32	1.40
	port with Caller ID - Res (1MF2X)	9		UEPSR	UEPAN	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling	1	1	OLI OIL	OLI AIV	1.03	9.93	3.13	3.00	2.32			20.55	10.54	10.02	1.40
	port with Caller ID - Res (2MR)	9		UEPSR	UEPAO	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled res, low usage line por	i l														
	with Caller ID (LUM)			UEPSR	UEPAP	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Port - 2-Wire VG Tennessee Residence Dialing Plan															
	without Caller ID			UEPSR	UEPWN	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Port - 2-Wire VG Tennessee Residence Area Plus															
	without Caller ID			UEPSR	UEPRR	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire voice unbundled Low Usage Line Port without Caller ID				1											
	Capability			UEPSR	UEPRT	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
FEAT	Subsequent Activity	-	1	UEPSR	USASC	0.00	0.00	0.00	-				20.35	10.54	13.32	1.40
FEAT	All Available Vertical Features	-		UEPSR	UEPVF	0.00	0.00	0.00					20.35	10.54	13.32	1.40
2-1//1	RE VOICE GRADE LINE PORT RATES (BUS)	-	1	OLFSK	OLFVI	0.00	0.00	0.00			1		20.33	10.34	13.32	1.40
	Exchange Ports - 2-Wire Analog Line Port without Caller ID -		1													
	Bus			UEPSB	UEPBL	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled Line Port with						0.00		0.00							
	unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	·															
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled TN extended local															
	dialing parity Port with Caller ID - Bus.	1		UEPSB	UEPAV	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
1	Exhange Ports - 2-Wire VG unbundled incoming only port with															
	Caller ID - Bus	+	1	UEPSB	UEPB1	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area			HEDOD	LIEDAG	4.00	0.00	0.40	2.00	0.00			20.05	40.54	40.00	
	Calling Port Economy Option - Bus (TACC1) Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area	+	1	UEPSB	UEPAC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
1	Calling Port Standard Option - Bus (TACC2)			UEPSB	UEPAD	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-W VG unbundled TN Bus 2-Way Collierville	+	1	OLFOD	ULFAD	1.69	9.93	9.19	3.00	2.92			20.35	10.54	13.32	1.40
	& Memphis Local Calling Port - Bus (B2F)			UEPSB	UEPAE	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-W VG unbundled TN Bus 2-Way Collierville	1	1	021 00	OLI AL	1.09	3.33	3.13	5.00	2.32	<u> </u>		20.33	10.34	15.52	1.40
1	& Memphis Local Calling Port			UEPSB	UEPB2	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-W VG unbundled TN, Business Line Inward	,		1	1	i										1
	Collierville & Memphis Local Calling Plan	1	1	UEPSB	UEPB3	1.89	9.93	9.19	3.66	2.92		I	20.35	10.54	13.32	1.40

ONRONDLE	D NETWORK ELEMENTS - Tennessee			1										ment: 2	1	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)	1	1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - 2-Wire Voice Tennessee Business Dialing															
	Plan without Caller ID			UEPSB	UEPWO	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire voice unbundled Incoming Only Port without Caller ID															-
	Capability			UEPSB	UEPBE	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00					20.35	10.54	13.32	1.40
FEAT	URES															
	All Available Vertical Features			UEPSB	UEPVF	0.00	0.00	0.00					20.35	10.54	13.32	1.40
EXCH	ANGE PORT RATES (DID & PBX)															
	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.79		9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
j	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Analog TN 2-Way Calling Plan PBX Trunk - Bus			UEPSP	UEPT2	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire TN Outward Calling Plan PBX Trunk - Bus			UEPSP	UEPTO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 2-Way PBX Tennessee Calling Port			UEPSP	UEPT2	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee															
	Calling Port			UEPSP	UEPTO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.79		9.19	3.66	2.92			20.35	10.54		1.40
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			02. 0.	02.7.0		0.00	0.10	0.00	2.02			20.00	10.01	10.02	
	Capable Port			UEPSP	UEPXE	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			OLI OI	OLI AL	1.73	3.33	3.13	5.00	2.32			20.55	10.54	10.02	1.40
	Administrative Calling Port			UEPSP	UEPXL	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			OLI OI	OLI AL	1.70	0.00	0.10	0.00	2.02			20.00	10.04	10.02	1.40
	Room Calling Port			UEPSP	UEPXM	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-W Voice Unbundled 1-Way Out PBX Hotel/Hospital Economy			OLI OI	OLI AWI	1.70	0.00	0.10	0.00	2.02			20.00	10.04	10.02	1.40
	Administrative Calling Port TN Calling Port			UEPSP	UEPXN	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			OLI OI	OLI XIV	1.75	3.33	3.13	5.00	2.32			20.55	10.54	10.02	1.40
	Discount Room Calling Port			UEPSP	UEPXO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Exchange Ports, PBX Trunk Combination,			OLI OI	OLI AO	1.73	3.33	3.13	3.00	2.32			20.55	10.54	10.02	1.40
	Collierville and Memphis Local Calling Plan			UEPSP	UEPA6	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Exchange Ports, PBX Trunk Combination, first trunk,			OLI OI	OLI AU	1.73	3.33	3.13	5.00	2.32			20.55	10.54	10.02	1.40
	Collierville and Memphis Local Calling Plan			UEPSP	UEPA7	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.79		9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX Collierville and Memphis Calling			OLI OI	OLI AO	1.70	0.00	0.10	0.00	2.02			20.00	10.04	10.02	1.40
	Port			UEPSP	UEPXU	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ			OLI OI	OLI AU	1.75	3.33	3.13	5.00	2.32			20.55	10.54	10.02	1.40
	Calling Port			UEPSP	UEPXV	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00	3.00	2.32			20.35	10.54	13.32	1.40
FEAT				ULFSF	USASC	0.00	0.00	0.00					20.33	10.54	13.32	1.40
1 1	All Available Vertical Features			UEPSP UEPSE	UEPVF	0.00	0.00	0.00					20.35	10.54	13.32	1.40
EVCU	ANGE PORT RATES (COIN)			OLFSF OLFSL	OLFVI	0.00	0.00	0.00					20.33	10.34	13.32	1.40
LXCII	Exchange Ports - Coin Port					2.11	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
NOTE	: Transmission/usage charges associated with POTS circuit so	witched	IISane	will also annly to a	rircuit switche						ated with 2	wire ISDN r		10.54	10.02	1.40
NOTE	: Access to B Channel or D Channel Packet capabilities will be	availal	nle onl	v through BFR/New	Rusiness Re	ulest Process	Rates for the	nacket canahi	ilities will he de	termined via t	he Bona Fir	le Request/	New Rusines	s Request Pro	ncess	
	LOCAL EXCHANGE SWITCHING(PORTS)	. a.anai	0.11			1		- acros oupdo				quo00	40111000			<u> </u>
	ANGE PORT RATES			1			† †							1		
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.97	47.75	47.01	9.21	8.47			20.35	10.54	13.32	1.40
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID					0.01			5.21	0.47			20.00	.5.04	.0.02	11
	capability			UEPDD	UEPDD	35.74	75.93	38.15	8.77	8.04			20.35	10.54	13.32	1.40
- 	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	16.26		29.49		4.10			20.35	10.54		1.40
NOTE	: Transmission/usage charges associated with POTS circuit s	witched	usane								ated with 2-	wire ISDN r		10.04	10.02	1.40
	: Access to B Channel or D Channel Packet capabilities will be													s Request Pro	ocess.	1
	Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0.00	0.00	0.00		The state of the s					1	1
	Exchange Ports - 4-Wire ISDN DS1 Port		-	UEPEX	UEPEX	75.04	148.66	147.18	38.46	36.98			20.35	10.54	13.32	1.40

Version 3Q02: 09/06/02

UNBUNDI	LED NETWORK ELEMENTS - Tennessee													ment: 2		ibit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY															
UNE	BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE															
	Unbundled Remote Call Forwarding Service, Area Calling, Res			UEPVR	UERAC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	
	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	
	Unbundled Remote Call Forwarding Service, IntraLATA - Res			UEPVR	UERTR	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
Non	n-Recurring															
	Unbundled Remote Call Forwarding Service - Conversion -															
	Switch-as-is	ļ	<u> </u>	UEPVR	USAC2	.	1.03	0.29	ļ				20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service - Conversion with	1	1	L	I	I					1	1		l	I	
	allowed change (PIC and LPIC)	<u> </u>		UEPVR	USACC	ļ	1.03	0.29							ļ	↓
UNE	BUNDLED REMOTE CALL FORWARDING - Bus	<u> </u>			1	ļ									ļ	↓
				l	l	1									1	
	Unbundled Remote Call Forwarding Service, Area Calling - Bus	<u> </u>	<u> </u>	UEPVB	UERAC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERLC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	
	Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVB	UERTE	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service, IntraLATA - Bus			UEPVB	UERTR	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service Expanded and															
	Exception Local Calling			UEPVB	UERVJ	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
Non	n-Recurring															
	Unbundled Remote Call Forwarding Service - Conversion -															
	Switch-as-is			UEPVB	USAC2		1.03	0.29					20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service - Conversion with															
	allowed change (PIC and LPIC)			UEPVB	USACC		1.03	0.29								
UNBUNDLE	D LOCAL SWITCHING, PORT USAGE															
End	l Office Switching (Port Usage)															
	End Office Switching Function, Per MOU					0.0008041										
Tan	dem Switching (Port Usage) (Local or Access Tandem)															
	Tandem Switching Function Per MOU					0.0009778										
Con	nmon Transport															
	Common Transport - Per Mile, Per MOU					0.0000064										
	Common Transport - Facilities Termination Per MOU					0.0003871										
	D PORT/LOOP COMBINATIONS - COST BASED RATES															
Cos	t Based Rates are applied where BellSouth is required by FCC a	nd/or St	ate Co	mmission rule to p	ovide Unbun	dled Local Swi	tching or Swite	ch Ports.								
	tures shall apply to the Unbundled Port/Loop Combination - Cos															
	I Office and Tandem Switching Usage and Common Transport Us															
	first and additional Port nonrecurring charges apply to Not Curr	rently C	ombine	ed Combos. For Cu	rrently Comb	ined Combos t	he nonrecurrin	g charges sha	II be those ider	ntified in the N	onrecurring	j - Currently	Combined se	ections.		
	IRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.18										
	2-Wire VG Loop/Port Combo - Zone 2		2			18.01										
	2-Wire VG Loop/Port Combo - Zone 3		3			23.02										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	12.48										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	16.31										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	21.32										
2-W	ire Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice Grade unbundled Tennessee extended local															
	dialing parity port with Caller ID - res		<u> </u>	UEPRX	UEPAQ	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Plus with Caller ID -															
	res (AC7)	<u> </u>	<u></u>	UEPRX	UEPAH	1.70	22.14	15.25	8.45	3.91		15.69			<u></u>	<u> </u>
	2-Wire voice unbundled Tennessee Area Calling port with Caller															
				UEPRX	UEPAK	1.70	22.14	15.25	8.45	3.91		15.69			•	

UNBUNDL	ED NETWORK ELEMENTS - Tennessee			•		1						,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	2-Wire voice unbundled Tennessee Area Calling port with Caller				-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	ID - res (TACER)			UEPRX	UEPAL	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACSR)			UEPRX	UEPAM	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller															
	ID - res (1MF2X) 2-Wire voice unbundled Tennessee Area Calling port with Caller			UEPRX	UEPAN	1.70	22.14	15.25	8.45	3.91		15.69				1
	ID - res (2MR) 2-Wire voice unbundles res, low usage line port with Caller ID			UEPRX	UEPAO	1.70	22.14	15.25	8.45	3.91		15.69				<u> </u>
	(LUM)			UEPRX	UEPAP	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled Tennessee Residence Dialing Plan															
	without Caller ID 2-Wire voice unbundled Tennessee Area Plus Port without		-	UEPRX	UEPWN	1.70	22.14	15.25	8.45	3.91		15.69				-
	Caller ID Capability			UEPRX	UEPRR	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX	UEPRT	1.70	22.14	15.25	8.45	3.91		15.69				
FEAT	FURES			OLI IXX	OLITA	1.70	22.14	13.23	0.43	5.91		15.05				
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00				15.69				
LOC	AL NUMBER PORTABILITY															
NON	Local Number Portability (1 per port) RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPRX	LNPCX	0.35										-
NON	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															†
	Switch-as-is			UEPRX	USAC2		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPRX	USACC		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	-		OLI KX	OOAOO			0.23								
ADD	Subsequent Database Update TIONAL NRCs						0.76					15.69				
ADD	2-Wire Voice Grade Loop/Line Port Combination - Subsequent				+											+
	Activity			UEPRX	USAS2	0.00	0.00	0.00				15.69				
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UNE	Port/Loop Combination Rates		1			14.18										
	2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2	1	2		-	18.01										+
+	2-Wire VG Loop/Port Combo - Zone 3		3			23.02										+
UNE	Loop Rates		Ť													
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	12.48										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	16.31										
2-Wii	2-Wire Voice Grade Loop (SL1) - Zone 3 re Voice Grade Line Port (Bus)		3	UEPBX	UEPLX	21.32										+
2-9911	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.70	22.14	15.25	8.45	3.91		15.69				+
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice Grade unbundled Tennessee extended local															
	dialing parity port with Caller ID - bus		1	UEPBX	UEPAV	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled incoming only port with Caller ID - Bus 2-Wire voice unbundled Tennessee Bus 2-Way Area Calling			UEPBX	UPEB1	1.70	22.14	15.25	8.45	3.91		15.69				
	Port Economy Option (TACC1)			UEPBX	UEPAC	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling Port Standard Option (TACC2)			UEPBX	UEPAD	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Collierville and															
	Memphis Local Calling Port (B2F) 2-Wire Voice Unbundled Tennessee Business Dialing Plan		1	UEPBX	UEPAE	1.70	22.14	15.25	8.45	3.91		15.69				
	without Caller ID			UEPBX	UEPWO	1.70	22.14	15.25	8.45	3.91		15.69				<u> </u>
	Tennessee Inward Collierville and Memphis Local Calling Plan (BUS)			UEPBX	UEPB2	1.70	22.14	15.25	8.45	3.91		15.69				
	Tennessee 2-Way Collierville and Memphis Local Calling Plan (BUS)			UEPBX	UEPB3	1.70	22.14	15.25	8.45	3.91		15.69				

OMBONDE	ED NETWORK ELEMENTS - Tennessee		1	1							0	06		nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPBX	UEPBE	1.70	22.14	15.25	8.45	3.91		15.69				
LOCA	L NUMBER PORTABILITY			OLI DX	OLI DE	1.70	22.14	10.20	0.40	0.01		10.00				
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEAT																
	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00				15.69				
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPBX	USAC2		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															1
	Switch with change			UEPBX	USACC		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	l														
	Subsequent Database Update						0.76					15.69				1
ADDIT	IONAL NRCs															
1 -	2-Wire Voice Grade Loop/Line Port Combination - Subsequent	1							[_	
	Activity	ļ		UEPBX	USAS2	0.00	0.00	0.00	ļ			15.69			ļ	
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
UNE P	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.18										
	2-Wire VG Loop/Port Combo - Zone 2		2			18.01										
	2-Wire VG Loop/Port Combo - Zone 3		3			23.02										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	12.48										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	16.31										ļ
0.180	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	21.32										
2-Wire	e Voice Grade Line Port Rates (RES - PBX)								-						-	
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.70	22.14	15.25	8.45	3.91		15.69				
LOCA	L NUMBER PORTABILITY					-										
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				15.69				
FEAT	URES															
	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00				15.69				
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch-As-Is			UEPRG	USAC2		1.03	0.29				15.69				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch with Change			UEPRG	USACC		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -											4= 00				
ADDIT	Subsequent Database Update TONAL NRCs	 	<u> </u>		+		0.76		 			15.69			!	
ADDII		-	 		+				 						 	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity	1		UEPRG	USAS2	0.00	0.00	0.00]			15.69			I	
-+	PBX Subsequent Activity - Change/Rearrange Multiline Hunt	1		ULPRU	USASZ	0.00	0.00	0.00	 			15.09			+	+
	Group	l					14.64	14.64				15.69			1	
2-W/ID	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	1	1		-		14.04	14.04	+ +		1	13.09			1	+
	Port/Loop Combination Rates				+				 						t	+
OIAE F	2-Wire VG Loop/Port Combo - Zone 1	1	1		+	14.18					<u> </u>				I	†
	2-Wire VG Loop/Port Combo - Zone 2	l	2			18.01			† †						1	
	2-Wire VG Loop/Port Combo - Zone 3		3			23.02			† 1						1	1
UNE L	oop Rates								†						1	İ
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	12.48										1
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	16.31	<u> </u>		<u> </u>							
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	21.32										
2-Wire	Voice Grade Line Port Rates (BUS - PBX)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.70	22.14	15.25	8.45	3.91		15.69				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.70	22.14	15.25	8.45	3.91		15.69				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.70	22.14	15.25	8.45	3.91		15.69				

ONBONDLE	ED NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	всѕ	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	Tour William Bows					1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled 2-Way Combination PBX Tennessee Calling Port			UEPPX	UEPT2	1.70	22.14	15.25	8.45	3.91		15.69				
-	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee			UEFFA	UEP12	1.70	22.14	15.25	0.40	3.91		13.09				
	Calling Port			UEPPX	UEPTO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.70		15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
	Capable Port			UEPPX	UEPXE	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1	1	LIEDDY	LIEDY		00.4	45.65		0.01		45.00				
	Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	 	<u> </u>	UEPPX	UEPXL	1.70	22.14	15.25	8.45	3.91		15.69	1		ļ.	
	Room Calling Port	1	1	UEPPX	UEPXM	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 1W Out PBX Hotel/Hospital Economy			OLIFA	JLFAIVI	1.70	22.14	13.23	0.45	3.91		13.09	-	-	1	
	Administrative Calling Port TN Calling Port	1	1	UEPPX	UEPXN	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			OLITA	OLI XII	1.70	22.17	10.20	0.40	0.01		10.00				
	Discount Room Calling Port			UEPPX	UEPXO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX Collierville and Memphis Calling															
	Port			UEPPX	UEPXU	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ															
	Callling Port			UEPPX	UEPXV	1.70	22.14	15.25	8.45	3.91		15.69				
	Tennessee PBX 2-Way Combo Each Additional Trunk															
	Collierville and Memphis Local Calling Plan			UEPPX	UEPA6	1.70	22.14	15.25	8.45	3.91		15.69				
	Tennessee PBX 2-Way Combo First Trunk Collierville and					. =-										
1.004	Memphis Local Calling Plan IL NUMBER PORTABILITY			UEPPX	UEPA7	1.70	22.14	15.25	8.45	3.91		15.69				
LUCA	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00				15.69				
EEAT	URES			UEFFA	LINPOP	3.13	0.00	0.00			-	15.69				-
FLAT	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00				15.69				
NONR	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLITA	OLI VI	0.00	0.00	0.00				10.00				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch-As-Is			UEPPX	USAC2		1.03	0.29				15.69				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch with Change		<u> </u>	UEPPX	USACC		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	1						-								
	Subsequent Database Update	ļ	<u> </u>				0.76		ļ			15.69				
ADDIT	TIONAL NRCs	ļ	<u> </u>						ļ						ļ	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1	1	UEPPX	LIE A CO	0.00	0.00	0.00	j			45.00				
	Subsequent Activity PBX Subsequent Activity - Change/Rearrange Multiline Hunt	 	 	UEPPA	USAS2	0.00	0.00	0.00	 			15.69	1		 	-
	Group	1	1		1 1		14.64	14.64	j			15.69				
UNF F	Port/Loop Combination Rates	 	 		+		14.04	14.04	 			10.09			+	
0.1.2.1	2-Wire VG Coin Port/Loop Combo – Zone 1	†	1		+ +	14.18							1		1	t
	2-Wire VG Coin Port/Loop Combo – Zone 2	1	2		1	18.01			†							
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			23.02										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	12.48										
	2-Wire Voice Grade Loop (SL1) - Zone 2	ļ	2	UEPCO	UEPLX	16.31			ļ					ļ		
0.1577	2-Wire Voice Grade Loop (SL1) - Zone 3	<u> </u>	3	UEPCO	UEPLX	21.32			ļ .				ļ		ļ	
2-Wire	e Voice Grade Line Ports (COIN)	 	<u> </u>		+										1	
	2-Wire Coin 2-Way without Operator Screening and without Blocking (TN)	1	1	UEPCO	UEPTB	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,	 	 	OLFOO	ULFID	1.70	22.14	15.25	0.40	3.91		15.09	1	1		-
	900/976, 1+DDD (NC, TN)			UEPCO	UEPRP	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking	 		02.1 00	JLI IXI	1.70	22.14	13.23	0.40	5.51		10.09			1	
	(TN)	1		UEPCO	UEPTA	1.70	22.14	15.25	8.45	3.91		15.69		I		

UNBU	NDLF	D NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Exhil	bit: C
0.120												Svc Order	Svc Order	Incremental		Incremental	
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			""											Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrecurring		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Coin 2-Way with Operator Screening: 900 Blocking:															
		900/976, 1+DDD, 011+, and Local (NC, TN)			UEPCO	UEPCA	1.70	22.14	15.25	8.45	3.91		15.69				
		2-Wire Coin Outward with Operator Screening and 011 Blocking															
		(TN)			UEPCO	UEPTC	1.70	22.14	15.25	8.45	3.91		15.69				
		2-Wire Coin Outward with Operator Screening and Blocking:			LIEBOO	LIEDOT	4.70	00.44	45.05	0.45	0.04		45.00				
		900/976, 1+DDD, 011+, and Local (TN)			UEPCO	UEPOT	1.70	22.14	15.25	8.45	3.91		15.69				
		2-Wire 2-Way Smartline with 900/976 (all states except LA) 2-Wire Coin Outward Smartline with 900/976 (all states except			UEPCO	UEPCK	1.88						15.69				
		2-wire Coin Outward Smartline with 900/976 (all states except			UEPCO	UEPCR	1.88						15.69				
	ADDITI	IDNAL UNE COIN PORT/LOOP (RC)			UEPCO	UEPCK	1.00						15.69				-
-	ADDITI	UNE Coin Port/Loop Combo Usage (Flat Rate)	1		UEPCO	URECU	3.45	0.00	0.00	 			15.69	1	 	1	
 		Local Number Portability (1 per port)	1	 	UEPCO	LNPCX	0.35	0.00	0.00	 			15.09	 	 	 	
 		2-Wire Voice Grade Loop / Line Port Combination - Conversion -	 		021 00	LIVIOA	0.35	 		t				1	t	1	
		Switch-as-is	l		UEPCO	USAC2		1.03	0.29	1			15.69		1		1
 	-	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	-		021 00	00,102		1.03	0.29	t			13.09	 	t	 	
1	1	Switch with change	l	1	UEPCO	USACC		1.03	0.29	I			15.69	Ì	I	Ì	1
-		2-Wire Voice Grade Loop/Line Port Combination - Subsequent	1		02. 00	30,100		1.03	0.20	†			10.00	1	<u> </u>		
		Activity			UEPCO	USAS2	0.00	0.00	0.00				15.69				
	2-WIRE	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE	ORT (00,102	0.00	0.00	0.00				10.00				
		ort/Loop Combination Rates		1													
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.45										
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			23.52										
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			30.17										
	UNE Lo	pop Rates															
		2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	16.56										
		2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	21.63										
		2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	28.28										
	2-Wire	Voice Grade Line Port Rates (Res)															
		2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.89	84.99	57.39	32.36	20.56		15.69				
		2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.89	84.99	57.39	32.36	20.56		15.69				
		2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.89	84.99	57.39	32.36	20.56		15.69				
		2-Wire voice Grade unbundled Tennessee extended local															
		dialing parity port with Caller ID - res			UEPFR	UEPAQ	1.89	84.99	57.39	32.36	20.56		15.69				
		2-Wire voice unbundled Tennessee Area Plus with Caller ID -															
		res (AC7)	<u> </u>		UEPFR	UEPAH	1.89	84.99	57.39	32.36	20.56		15.69	 	-	 	
1		2-Wire voice unbundled Tennessee Area Calling port with Caller	l		HEDED	LIEDAY	4.00	04.00	57.00	00.00	00.50		45.00	1	I	1	1
 		ID - res (F2R)	 	<u> </u>	UEPFR	UEPAK	1.89	84.99	57.39	32.36	20.56		15.69	 	!	 	
1		2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER)	l		UEPFR	UEPAL	1.89	84.99	57.39	32.36	20.56		15.69	1	I	1	1
-		2-Wire voice unbundled Tennessee Area Calling port with Caller	-	-	ULFFR	UEPAL	1.89	04.99	51.39	32.36	20.56		15.09	-	-	-	
1		ID - res (TACSR)	l		UEPFR	UEPAM	1.89	84.99	57.39	32.36	20.56		15.69	1	I	1	1
		2-Wire voice unbundled Tennessee Area Calling port with Caller	1	 	OLITA	OLFAIVI	1.09	04.99	31.39	32.30	20.56		15.09	 	 	 	
		ID - res (1MF2X)	l		UEPFR	UEPAN	1.89	84.99	57.39	32.36	20.56		15.69		1		1
-	-	2-Wire voice unbundled Tennessee Area Calling port with Caller	-		O=111K	OLI AIN	1.05	04.33	37.35	52.30	20.30		10.09	 	t	 	
		ID - res (2MR)	l		UEPFR	UEPAO	1.89	84.99	57.39	32.36	20.56		15.69	1	I	1	1
		2-Wire voice unbundles res, low usage line port with Caller ID	1			02.7.0	1.09	04.00	07.00	02.00	20.00		10.00	1	1	1	t
		(LUM)	l		UEPFR	UEPAP	1.89	84.99	57.39	32.36	20.56		15.69	1	I	1	1
		2-Wire Voice Unbundled Tennessee Residence Dialing Plan				1	30	220	230	550			12.20	İ	1	1	
		without Caller ID	l		UEPFR	UEPWN	1.89	84.99	57.39	32.36	20.56		15.69	1	I	1	1
	INTER	OFFICE TRANSPORT															
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
l		Termination	l		UEPFR	U1TV2	18.58	55.39	17.37	27.96	3.51				1		1
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
L	L	or Fraction Mile	<u></u>	<u> </u>	UEPFR	1L5XX	0.0174	<u>] </u>		<u> </u>		<u></u>		<u> </u>	<u> </u>	<u> </u>	1
	FEATU																
		All Features Offered			UEPFR	UEPVF	0.00	0.00	0.00				15.69				
	LOCAL	NUMBER PORTABILITY							-								
		Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED								l			1		l		1

ONBONDE	ED NETWORK ELEMENTS - Tennessee			1							-			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increments Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrecurring		Nonrecurring	Disconnect		•	oss	Rates(\$)	•	•
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															ĺ
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		16.94	3.72				15.69				
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRI	LINE	PORT ((BUS)												
UNE F	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.45										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			23.52										
UNE I	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 Loop Rates		3			30.17										
UNE	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	16.56					1	1				
 	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	21.63	 		1		 			 	1	—
- 	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	28.28			1						1	<u> </u>
2-Wire	e Voice Grade Line Port (Bus)		Ť			20.20								1		
	2-Wire voice unbundled port without Caller ID - bus		1	UEPFB	UEPBL	1.89	84.99	57.39	32.36	20.56		15.69		İ		
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.89	84.99	57.39	32.36	20.56		15.69				1
	2-Wire voice Grade unbundled Tennessee extended local															1
	dialing parity port with Caller ID - bus			UEPFB	UEPAV	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.89	84.99	57.39	32.36	20.56		15.69				1
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling															
	Port Economy Option (TACC1)			UEPFB	UEPAC	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling															
	Port Standard Option (TACC2)			UEPFB	UEPAD	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Collierville and															
	Memphis Local Calling Port (B2F)			UEPFB	UEPAE	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire Voice Unbundled Tennessee Business Dialing Plan			UEPFB	UEPWO	1.89	04.00	F7 20	32.36	20.56		45.00				
	without Caller ID Tennessee Inward Collierville and Memphis Local Calling Plan			UEPFB	UEPWU	1.89	84.99	57.39	32.36	20.56		15.69				
	(BUS)			UEPFB	UEPB2	1.89	84.99	57.39	32.36	20.56		15.69				
	Tennessee 2-Way Collierville and Memphis Local Calling Plan			UEPFB	UEPBZ	1.09	04.99	57.39	32.30	20.36	1	15.69				-
	(BUS)			UEPFB	UEPB3	1.89	84.99	57.39	32.36	20.56		15.69				
LOCA	IL NUMBER PORTABILITY			OLITB	OLI D3	1.03	04.55	37.33	32.30	20.50	1	15.05				+
2007	Local Number Portability (1 per port)		1	UEPFB	LNPCX	0.35										+
INTER	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFB	U1TV2	18.58	55.39	17.37	27.96	3.51						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															1
	or Fraction Mile			UEPFB	1L5XX	0.0174										
FEAT	URES															
	All Features Offered			UEPFB	UEPVF	0.00	0.00	0.00				15.69				
NONE	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFB	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
0.14/15	Combination - Conversion - Switch with change			UEPFB	USACC		16.94	3.72				15.69				
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) Port/Loop Combination Rates		 	-	+				1		1	-		 	1	
UNE	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		+ -	18.45										
- 	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		2		+ -	23.52	 		1		 			 	+	
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3	 		30.17	†		1		1	<u> </u>		 	1	
UNE I	Loop Rates		Ť			55.17								1		
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	16.56								1		
1	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	21.63	i							İ		1
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	28.28										
2-Wire	e Voice Grade Line Port Rates (BUS - PBX)															
													_	_		
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	<u></u>	<u></u>	UEPFP	UEPPC	1.79	106.40	63.08	42.67	18.54	<u> </u>	15.69				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.79	106.40	63.08	42.67	18.54		15.69				

CATEGORY	RATE ELEMENTS										Svc Order	Svc Order	Incremental	Incremental	Incremental	
		Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Charge - Manual Sv Order vs.
													Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic Disc Add'
						_	Nonrecurring		Nonrecurring	Disconnect		l	oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Tennessee Calling Port			UEPFP	UEPT2	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee															
	Calling Port			UEPFP	UEPTO	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports		1	UEPFP	UEPXB	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.79		63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port		1	UEPFP	UEPXD	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPFP	UEPXE	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPFP	UEPXL	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPFP	UEPXM	1.79	106.40	63.08	42.67	18.54		15.69				
<u> </u>	2-Wire Voice Unbundled 1W Out PBX Hotel/Hospital Economy Administrative Calling Port TN Calling Port			UEPFP	UEPXN	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPFP	UEPXO	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	1.79	106.40	63.08	42.67	18.54		15.69				
i	2-Wire Voice Unbundled PBX Collierville and Memphis Calling Port			UEPFP	UEPXU	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Callling Port			UEPFP	UEPXV	1.79	106.40	63.08	42.67	18.54		15.69				
	NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.69				
INTERC	OFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFP	U1TV2	18.58	55.39	17.37	27.96	3.51						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFP	1L5XX	0.0174										
FEATU																<u> </u>
	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00				15.69				<u> </u>
NONRE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															<u> </u>
<u> </u>	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFP	USAC2		16.94	3.72				15.69				
<u> </u>	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch with change			UEPFP	USACC		16.94	3.72				15.69				
	PORT/LOOP COMBINATIONS - COST BASED RATES															<u> </u>
	E VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT														<u> </u>
UNE Po	ort/Loop Combination Rates															<u> </u>
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			18.38										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			19.87										<u> </u>
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			24.78										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	9.60										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	11.09										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	16.00										
	Exchange Ports - 2-Wire DID Port			UEPPX	UEPD1	8.78	45.44	29.94	8.45	3.91			30.89	7.03		
	ECURRING CHARGES - CURRENTLY COMBINED 2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -															
	Switch-as-is 2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion			UEPPX	USAC1		8.76	5.75					30.89	7.03		
	with BellSouth Allowable Changes			UEPPX	USA1C		8.76	5.75					30.89	7.03		
Teleph	none Number/Trunk Group Establisment Charges		-	LIEBBY .	No.										ļ	
	DID Trunk Termination (One Per Port)		-	UEPPX	NDT	0.00	0.00	0.00							ļ	
	Additional DID Numbers for each Group of 20 DID Numbers		-	UEPPX	ND4	0.00	0.00	0.00							ļ	├
\longrightarrow	DID Numbers, Non- consecutive DID Numbers , Per Number Reserve Non-Consecutive DID numbers		1	UEPPX UEPPX	ND5 ND6	0.00	0.00	0.00			ļ					

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												,		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	E	scs	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
								Nonrecurring		Nonrecurring	Disconnect		l .	oss	Rates(\$)	l .	<u> </u>
-							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00	11130	Auu i	JOHLC	JONAN	JOHAN	JONAN	JOHIAN	JOHAN
LOCAL	NUMBER PORTABILITY			OLITA		INDV	0.00	0.00	0.00			1					†
	Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00			1					†
2-WIRE	ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LII	NE SIDE	POR			2.1. 0.	0.10	0.00	0.00			1					†
	ort/Loop Combination Rates																
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB	UEPPR		32.27										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		34.78										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
-+	UNE Zone 3	 	3	UEPPB	UEPPR	LICLOY	44.32	ļ							 	!	├
	2-Wire ISDN Digital Grade Loop - UNE Zone 1	 	1	UEPPB	UEPPR	USL2X	16.20	ļ							 	!	
	2-Wire ISDN Digital Grade Loop - UNE Zone 2	l	2	UEPPB	UEPPR	USL2X	18.71									1	
	2-Wire ISDN Digital Grade Loop - UNE Zone 2 2-Wire ISDN Digital Grade Loop - UNE Zone 3	-	3	UEPPB	UEPPR	USL2X USL2X	28.25	 		1					-	-	-
-	Exchange Port - 2-Wire ISDN Line Side Port		3	UEPPB	UEPPR	UEPPB	16.07	141.75	118.37	49.20	43.26	1		19.99	19.99		-
NOND	ECURRING CHARGES - CURRENTLY COMBINED		-	UEFFB	UEPPR	UEPPB	10.07	141.75	110.37	49.20	43.20	1		19.99	19.99		-
NONKI	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port		-									1					-
	Combination - Conversion			LIFPPR	UEPPR	USACB	0.00	117.23	117.23					19.99	19.99		
ADDIT	IONAL NRCs			02	02	00/102	0.00	20	20					10.00	10.00		†
,,,,,,,,	2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy Non Feature/Add Trunk			UEPPB	UEPPR	USASB		212.88						19.99	19.99		
LOCAL	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								1
B-CHA	NNEL USER PROFILE ACCESS:																
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
B-CHA	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C,MS, &	TN)														
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								ļ
USER	TERMINAL PROFILE				HERRE												
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VERTIO	CAL FEATURES			LIEDDD	HEDDD	LIED) /E	0.00	0.00	0.00								
	All Vertical Features - One per Channel B User Profile	<u> </u>	-	UEPPB	UEPPR	UEPVF	0.00	0.00	0.00	 						-	
	Interoffice Channel mileage each, including first mile and facilities termination	l		UEPPB	UEPPR	M1GNC	17.91	53.99	17.37					19.99	19.99	1	
	Interoffice Channel mileage each, additional mile	 		UEPPB	UEPPR	M1GNM	0.173	0.00	0.00					15.55	19.99	t	\vdash
4-WIRE	E DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	PORT		OLITE	OLITIK	INTOTANT	0.170	0.00	0.00	1							+
	ort/Loop Combination Rates	1															
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP			132.58										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		2	UEPPP			150.25										
	Zone 2 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		3	UEPPP			173.44										
	4-Wire DS1 Digital Loop - UNE Zone 1	 	1	UEPPP		USL4P	57.73	 							1	 	
	4-Wire DS1 Digital Loop - UNE Zone 2	1	2	UEPPP		USL4P	75.40			 		<u> </u>			 	I	†
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	98.59	†							İ	1	
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	74.85	415.53	366.90	89.28	77.43			19.99	19.99		
NONR	CURRING CHARGES - CURRENTLY COMBINED																
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port			LIEDDE		LICACD	0.00	200.50	200 52					40.00	40.00		
ADDIT	Combination - Conversion -Switch-as-is	1		UEPPP		USACP	0.00	328.53	328.53			1		19.99	19.99	1	
AUUII		 	-	 		+		 		1					-		
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way Tel Nos. (except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -			UEPPP		PR7TF		0.94						19.99	19.99		
	Outward Tel Numbers (All States except NC)			UEPPP		PR7TO		22.36	22.36					19.99	19.99		

ONRONDLED NET	WORK ELEMENTS - Tennessee										Ι	1 -		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
1					+		Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)	1	<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
4-Wire	DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -							7.00.		7.00.		00				
	quent Inward Tel Numbers			UEPPP	PR7ZT		44.71	44.70					19.99	19.99		
	ER PORTABILITY															
	Number Portability (1 per port)			UEPPP	LNPCN	1.75										
	Provsioning Only)															
Voice/I	Data			UEPPP	PR71V	0.00	0.00	0.00								
Digital	Data			UEPPP	PR71D	0.00	0.00	0.00								1
Inward	Data			UEPPP	PR71E	0.00	0.00	0.00								1
New or Additi	onal "B" Channel															
New or	r Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	28.39						19.99	19.99		
	r Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	29.11						19.99	19.99		
	r Additional Inward Data B Channel			UEPPP	PR7BD	0.00	29.39						19.99	19.99		
CALL TYPES																
Inward				UEPPP	PR7C1	0.00	0.00	0.00								
Outwa	rd			UEPPP	PR7C0	0.00	0.00	0.00								
Two-w				UEPPP	PR7CC	0.00	0.00	0.00								
Interoffice Cha	annel Mileage															
Fixed F	Each Including First Mile			UEPPP	1LN1A	76.1825	145.98	109.85	19.55				19.99	19.99		1
Each A	Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3525										
4-WIRE DS1 D	IGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE Port/Loo	p Combination Rates															
4W DS	61 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		93.28							19.99	19.99		
4W DS	61 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		110.95							19.99	19.99		
4W DS	61 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		134.14							19.99	19.99		
UNE Loop Rat	tes															
4-Wire	DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	57.53										
4-Wire	DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	75.40										ĺ
4-Wire	DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	98.59										
UNE Port Rate)															
4-Wire	DDITS Digital Trunk Port			UEPDC	UDD1T	35.55	342.80	257.87	61.41	48.49			19.99	19.99		
NONRECURRI	NG CHARGES - CURRENTLY COMBINED															ĺ
4-Wire	DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															i .
- Switc	h-as-is			UEPDC	USAC4		312.91	312.91					19.99	19.99		
	DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	ersion with DS1 Changes			UEPDC	USAWA		312.91	312.91					19.99	19.99		
	DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	ersion with Change - Trunk			UEPDC	USAWB		312.91	312.91					19.99	19.99		
ADDITIONAL I																
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	e Activity Per Service Order			UEPDC	USAS4		94.88	94.88								
	DS1 Loop / 4-Wire DDITS Trunk Port - NRC -	1														
	quent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		108.67	108.67					19.99	19.99		
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	el Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		108.67	108.67					19.99	19.99		
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
	ion/Chan Inward Trunk w/out DID			UEPDC	UDTTC		108.67	108.67					19.99	19.99		
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	ion Per Chan - Inward Trunk with DID			UEPDC	UDTTD		108.67	108.67					19.99	19.99		ļ
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan	1]							l	I	
	ion / Chan - 2-Way DID w User Trans	ļ		UEPDC	UDTTE		108.67	108.67					19.99	19.99	1	ļ
	RO SUBSTITUTION				1		ļ								.	ļ
	-Superframe Format	ļ		UEPDC	CCOSF		0.00	590.00	1		<u> </u>		19.99	19.99	ļ	ļ
	- Extended Superframe Format	ļ		UEPDC	CCOEF		0.00	590.00			<u> </u>		19.99	19.99	ļ	ļ
Alternate Mari					1		ļ								.	ļ
	uperframe Format	ļ		UEPDC	MCOSF		0.00	0.00			<u> </u>				ļ	ļ
	Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00						ļ	.	ļ
	mber/Trunk Group Establisment Charges	ļ		ļ	1						<u> </u>			ļ	ļ	ļ
	one Number for 2-Way Trunk Group		1	UEPDC	UDTGX	0.00							19.99	19.99		
I ITalanh	one Number for 1-Way Outward Trunk Group		l	UEPDC	UDTGY	0.00					1		19.99	19.99		1

<u>NBUNDLE</u>	D NETWORK ELEMENTS - Tennessee												Attachi	ment: 2	Exhi	bit: C
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Charge
													1st	Add'l	Disc 1st	Disc Add
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00							19.99	19.99		
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00							19.99	19.99		
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00							19.99	19.99		
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00								
Dedica	ted DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	Digital	Loop	with 4-Wire DDITS 1	runk Port											
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	75.83	145.98	109.85	19.66	14.99						
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.3525	0.00	0.00							ļ	4
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities	1	1	l		_	_	_							l	
	Termination)	ļ	<u> </u>	UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25	1	1	l											l	
	miles			UEPDC	1LNOB	0.3525	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities	1	1												l	
	Termination)	ļ	<u> </u>	UEPDC	1LNO3	0.00	0.00	0.00								
		1	1												l	
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.3525	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								
	Central Office Termininating Point			UEPDC	CTG	0.00										
	DS1 LOOP WITH CHANNELIZATION WITH PORT															
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act															
	system can have up to 24 combinations of rates depending on	type ar	ıd nun	ber of ports used												
UNE D	S1 Loop															
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	57.73	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	75.40	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	98.59	0.00	0.00								
UNE D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)														
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	131.87	0.00	0.00					19.99	19.99		
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	263.74	0.00	0.00					19.99	19.99		
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	527.48	0.00	0.00					19.99	19.99		
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	791.42	0.00	0.00					19.99	19.99		
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	827.76	0.00	0.00					19.99	19.99		
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,318.70	0.00	0.00					19.99	19.99		
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,582.44	0.00	0.00					19.99	19.99		
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	2,109.92	0.00	0.00					19.99	19.99		
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,637.40	0.00	0.00					19.99	19.99		
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	3,164.88	0.00	0.00					19.99	19.99		
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3,692.36	0.00	0.00					19.99	19.99		
Non-Re	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	n Chanr	eliztio	n with Port - Conve	rsion Charge	Based on a Sy	stem									
A Mini	mum System configuration is One (1) DS1, One (1) D4 Channe	l Bank,	and U	To 24 DSO Ports w	vith Feature A	Activations.										
	es of this configuration functioning as one are considered Ac															
	NRC - Conversion (Currently Combined) with or without															
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	303.61	15.74					19.99	19.99		
System	Additions at End User Locations Where 4-Wire DS1 Loop with	th Chan	nelizat	ion with Port Comb		ntly Exists and	ı									
	lot Currently Combined) in all states, except in Density Zone 1															
,	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port															
	and Assoc Fea Activation	1	1	UEPMG	VUMD4	0.00	704.68	441.48	138.36	16.41			19.99		l	
Bipola	r 8 Zero Substitution							_								
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only	l		UEPMG	CCOSF	0.00	0.00	590.00								
	Clear Channel Capability Format - Extended Superframe -															
	Subsequent Activity Only	1	1	UEPMG	CCOEF	0.00	0.00	590.00							l	
Alterna	ate Mark Inversion (AMI)			İ	1									İ	İ	1
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00						İ	İ	1
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00			İ				İ	
Exchar	nge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port		1	5.50	0.00	0.00			i e			1	1	
	nge Ports			!	+	-								 	-	+

	JNDLE	ED NETWORK ELEMENTS - Tennessee												Attachr	ment: 2	Exhil	oit: C
1												Submitted	Submitted	Incremental Charge -	Incremental Charge -	Incremental Charge -	Incremental Charge -
CATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Elec per LSR	Manually per LSR	Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs. Electronic- Add'l	Order vs. Electronic- Disc 1st	Manual Svc Order vs. Electronic- Disc Add'l
																DISC 1St	DISC Add I
							Rec	Nonrecurring			g Disconnect				Rates(\$)		
				1				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Line Cide Combination Channelined DDV Tauri Dest. Business			UEPPX	UEPCX	1.79	0.00	0.00	0.00	0.00			30.89	7.03		
		Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.79	0.00	0.00	0.00	0.00			30.89	7.03		
		Line Side Odtward Charmenzed FBX Trunk Fort - Business		1	OLITA	OLI OX	1.73	0.00	0.00	0.00	0.00			30.03	7.00		
		Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.79	0.00	0.00	0.00	0.00			30.89	7.03		
		2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	8.97	0.00	0.00	0.00	0.00			30.89	7.03		
	Featur	re Activations - Unbundled Loop Concentration															
		Feature (Service) Activation for each Line Side Port Terminated in D4 Bank			UEPPX	1PQWM	0.66	23.94	12.64	3.82	3.80			30.89	7.03		
		Feature (Service) Activation for each Trunk Side Port Terminated			LIEDDY	1PQWU	0.66	70.07	17.37	54.00	40.57			20.00	7.00		
	Teleni	in D4 Bank hone Number/ Group Establishment Charges for DID Service		1	UEPPX	IPQWU	0.00	73.67	17.37	54.09	10.57			30.89	7.03		
	Tolepi	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00			1	1	-			
		DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00	İ	İ			1			
		Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00								
		Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00								
	Ļ	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
	Local	Number Portability			LIEDDY	LNDOD	0.45	0.00	0.00								
-	FEAT	Local Number Portability - 1 per port		-	UEPPX	LNPCP	3.15	0.00	0.00								
-		URES - Vertical and Optional Switching Features Offered with Line Side Ports Only		1													
	LUCAI	All Features Available			UEPPX	UEPVF	0.00	0.00	0.00								
UNBU	NDLED	PORT LOOP COMBINATIONS - MARKET RATES			OLI I X	02. 1.	0.00	0.00	0.00								
		et Rates shall apply where BellSouth is not required to provide	unbun	dled lo	cal switching or swi	tch ports pe	r FCC and/or S	tate Commissio	n rules.								
		ncludes:															
		ndled port/loop combinations that are Currently Combined or N															
	The To	op 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderda outh currently is developing the billing capability to mechanica	ale, Mia	imi); G	A (Atlanta); LA (New	Orleans); NO	C (Greensboro	Winston Salem	-Highpoint/Ch	arlotte-Gaston	ia-Rock Hill);	TN (Nashvill	e).	In the interi	m where Belli	Cauth cannot	hill Markat
		, BellSouth shall bill the rates in the Cost-Based section preced								ig charges for	not currently	combined in	FL and NO	. In the interi	III WIIEIE DEII	South Cannot	DIII Warket
					tile warket itales all	u reserves ti	ie rigiit to true	up the billing t	annerence.	1							
		larket Rate for unbundled norts includes all available features i	n all st	ates.													
	End O	larket Rate for unbundled ports includes all available features in			l ne Port section of th	is rate exhib	it shall apply t	all combination	ons of loop/po	rt network elei	ments except	for UNE Coi	n Port/Looi	Combination	l ns which have	a flat rate us	age charge
1		larket Rate for unbundled ports includes all available features i Office and Tandem Switching Usage and Common Transport Us C: URECU).			e Port section of th	is rate exhib	it shall apply t	o all combination	ons of loop/po	rt network elei	ments except	for UNE Coi	n Port/Loop	Combination	ns which have	a flat rate us	age charge
	(USOC	Office and Tandem Switching Usage and Common Transport Us	sage rat	tes in tl													
	(USOC For No Additi	Office and Tandem Switching Usage and Common Transport Us C: URECU). or Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly.	sage rat	tes in tl													
	For No Additi 2-WIR	Office and Tandem Switching Usage and Common Transport Us C: URECU). of Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. IE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	sage rat	tes in tl													
	For No Additi 2-WIR	Office and Tandem Switching Usage and Common Transport Us C: URECU). ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. IE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates	sage rat	in the I			ns for each Por										
	For No Additi 2-WIR	Office and Tandem Switching Usage and Common Transport Us C: URECU). ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. BE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) POT/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1	sage rat	in the I			as for each Por										
	For No Additi 2-WIR	Office and Tandem Switching Usage and Common Transport Us C: URECU). C: URECU). C: URECU). C: URECU). C: URECU Combined scenarios the Nonrecurring charges are cional NRCs may apply also and are categorized accordingly. C: VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) CONTURED COMBINITION OF COMBINITION	sage rat	in the I			26.48 30.31										
	(USOC For No Additi 2-WIR UNE P	Office and Tandem Switching Usage and Common Transport Us C: URECU). of Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. IE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I			as for each Por										
	(USOC For No Additi 2-WIR UNE P	Office and Tandem Switching Usage and Common Transport UsC: URECU). of Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. EVOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Loop Rates	sage rat	in the I	First and Additional		26.48 30.31										
	(USOC For No Additi 2-WIR UNE P	Office and Tandem Switching Usage and Common Transport Us C: URECU). of Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. IE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I		NRC column	26.48 30.31 35.32										
	(USOC For No Additi 2-WIR UNE P	Office and Tandem Switching Usage and Common Transport Us C: URECU). C: URECU). C: URECU). C: URECU). C: URECU). C: URECU Service Serv	sage rat	in the I	First and Additional	NRC column	26.48 30.31 35.32										
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Usc: URECU). C: URECU). Or Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. EE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-Wire VG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 8 Voice Grade Line Port (Res)	sage rat	in the I	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX	26.48 30.31 35.32 12.48 16.31 21.32	t USOC. For Ct	urrently Comb					in the NRC - (Currently Com		
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Use: URECU). C: URECU). C: URECU). C: URECU). C: URECU). CE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) POrt/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX	26.48 30.31 35.32 12.48 16.31 21.32	t USOC. For Ct	urrently Comb					in the NRC - C	Currently Com		
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Usc: URECU). C: URECU). Ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) PortLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 -oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 e Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res	sage rat	in the I	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL	26.48 30.31 35.32 12.48 16.31 21.32	90.00 90.00	90.00 90.00					30.89 30.89	7.03 7.03		
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Usc: URECU). C: URECU). Or Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. EE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX	26.48 30.31 35.32 12.48 16.31 21.32	t USOC. For Ct	urrently Comb					in the NRC - C	Currently Com		
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Usc: URECU). C: URECU). Ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL	26.48 30.31 35.32 12.48 16.31 21.32	90.00 90.00	90.00 90.00					30.89 30.89	7.03 7.03		
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Usc: URECU). C: URECU). Or Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. EVOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I	UEPRX												
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Use: URECU). C: URECU). Of Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) PortLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I	UEPRX												
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Use: URECU). C: URECU). Ot Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	sage rat	in the I	UEPRX												
	(USOC For No Additi 2-WIR UNE F	Office and Tandem Switching Usage and Common Transport Usc: URECU). C: URECU). Or Currently Combined scenarios the Nonrecurring charges are ional NRCs may apply also and are categorized accordingly. Its Voice GRADE LOOP WITH 2-WIRE LINE PORT (RES) Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Tennessee extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2R) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER)	sage rat	in the I	UEPRX												

Version 3Q02: 09/06/02

NRONDLE	D NETWORK ELEMENTS - Tennessee			1							1 -			ment: 2		oit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
					1		Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundles res, low usage line port with Caller ID															
	(LUM)			UEPRX	UEPAP	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Low Usage Line Port without Caller ID															
	Capability			UEPRX	UEPRT	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled Tennessee Residence Dialing Plan															
	without Caller ID 2-Wire voice unbundled Tennessee Area Plus Port without			UEPRX	UEPWN	14.00	90.00	90.00					30.89	7.03		
	Caller ID Capability			UEPRX	UEPRR	14.00	90.00	90.00					30.89	7.03		
	NUMBER PORTABILITY			ULFRX	OLFKK	14.00	90.00	90.00					30.09	7.03		
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
FEATU				02.101	2.1. 0/1	0.00										
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00	1				30.89	7.03		
	CURRING CHARGES - CURRENTLY COMBINED															
		-												_		
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPRX	USAC2		41.50	41.50					30.89	7.03		
	2-Wire Voice Grade Loop / Line Port Combination - Switch with															
	change			UEPRX	USACC		41.50	41.50					30.89	7.03		
ADDITI	ONAL NRCs															
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent			UEPRX	USAS2	0.00	0.00	0.00					30.89	7.03		
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)			UEPKA	USAS2	0.00	0.00	0.00					30.69	7.03		
	ort/Loop Combination Rates				+											
	2-Wire VG Loop/Port Combo - Zone 1		1		+	26.48										
	2-Wire VG Loop/Port Combo - Zone 2		2		1	30.31										
	2-Wire VG Loop/Port Combo - Zone 3		3			35.32										
UNE Lo	pop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	12.48										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	16.31										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	21.32										
	Voice Grade Line Port (Bus)			LIEBBY .			22.22									
	2-Wire voice unbundled port without Caller ID - bus 2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX UEPBX	UEPBL UEPBC	14.00 14.00	90.00 90.00	90.00					30.89 30.89	7.03 7.03		
	2-Wire voice unbundled port with Caller + E484 ID - bus 2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice Grade unbundled Tennessee extended local			OLFBX	OLFBO	14.00	90.00	90.00					30.09	7.03		
	dialing parity port with Caller ID - bus			UEPBX	UEPAV	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling			02. B/C	02.7.0		00.00	00.00					00.00	7.00		
	Port Economy Option (TACC1)			UEPBX	UEPAC	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling															
	Port Standard Option (TACC2)			UEPBX	UEPAD	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Bus 2-Way Collierville and															
	Memphis Local Calling Port (B2F)			UEPBX	UEPAE	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Incoming Only Port without Caller ID			LIEDDY	LIEDDE	44.00	00.00	00.00					20.00	7.00		
	Capability 2-Wire Voice Unbundled Tennessee Business Dialing Plan			UEPBX	UEPBE	14.00	90.00	90.00					30.89	7.03	-	
	without Caller ID			UEPBX	UEPWO	14.00	90.00	90.00					30.89	7.03	1	1
	NUMBER PORTABILITY			OLI DA	0L1 VV	14.00	30.00	90.00					30.09	7.03		1
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEATU	RES															
	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00					30.89	7.03		
NONRE	CURRING CHARGES - CURRENTLY COMBINED															
				l	1										1	1
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPBX	USAC2		41.50	41.50					30.89	7.03		
	2-Wire Voice Grade Loop / Line Port Combination - Switch with			LIEDBY	LIEACO		44.50	41.50					30.89	7.00	1	1
	change ONAL NRCs		-	UEPBX	USACC		41.50	41.50	 				30.89	7.03		
AUUIII	NRC - 2-Wire Voice Grade Loop/Line Port Combination -			 	+										 	
	Subsequent			UEPBX	USAS2	0.00	0.00	0.00					30.89	7.03	1	1
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			·	1	3.30	3.50	3.30					33.33		1	
	ort/Loop Combination Rates		t	†	1 1		1		1		1	1		1		

UNBUNDLE	D NETWORK ELEMENTS - Tennessee										Ι		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)		Svc Order Submitted Elec per LSR	Submitted Manually	Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring Disconne				Rates(\$)		
							First	Add'l	First Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/Port Combo - Zone 1		1			26.48									
	2-Wire VG Loop/Port Combo - Zone 2		2			30.31									
	2-Wire VG Loop/Port Combo - Zone 3		3			35.32									
UNE L	oop Rates			LIEBBO	UEDLY	10.10									
	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRG UEPRG	UEPLX	12.48 16.31									
	2-Wire Voice Grade Loop (SL1) - Zone 2		3	UEPRG	UEPLX	21.32									
2-Wire	e Voice Grade Line Port Rates (RES - PBX)			OLI IKO	OLILX	21.02									
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -				+										
	Res			UEPRG	UEPRD	14.00	90.00	90.00				30.89	7.03		
LOCA	L NUMBER PORTABILITY	1			1		22.00	22,00				22.00			
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00							1
FEAT															
	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00				30.89	7.03		
NONR	ECURRING CHARGES - CURRENTLY COMBINED														
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPRG	USAC2		41.50	41.50				30.89	7.03		
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with														
	Change			UEPRG	USACC		41.50	41.50				30.89	7.03		
ADDIT	TIONAL NRCs														
	2 Wire Loop/Line Side Port Combination - Non feature -														
	Subsequent Activity- Nonrecurring						0.00	0.00				30.89	7.03		
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt						44.04	44.64				20.00	7.03		
2 14/15	Group E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)						14.64	14.64				30.89	7.03		
	Port/Loop Combination Rates				+		-								
ONLI	2-Wire VG Loop/Port Combo - Zone 1		1		+	26.48									
	2-Wire VG Loop/Port Combo - Zone 2		2			30.31									
	2-Wire VG Loop/Port Combo - Zone 3		3		+	35.32									
UNE I	oop Rates														
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	12.48									
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	16.31									
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	21.32									
2-Wire	Voice Grade Line Port Rates (BUS - PBX)														
												_	_		
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	1		UEPPX	UEPPC	14.00	90.00	90.00				30.89	7.03		
	Line Side Unbundled Outward PBX Trunk Port - Bus	1		UEPPX	UEPPO	14.00	90.00	90.00				30.89	7.03		
	Line Side Unbundled Incoming PBX Trunk Port - Bus	!	<u> </u>	UEPPX	UEPP1	14.00	90.00	90.00			ļ	30.89	7.03		
	2-Wire Voice Unbundled PBX LD Terminal Ports	 	ļ	UEPPX	UEPLD	14.00	90.00	90.00		_		30.89	7.03		
	2-Wire Voice Unbundled 2-Way Combination PBX Tennessee	1	1	LIEDDY	LIEDTO	44.00	90.00	00.00				30.89	7.03	1	1
	Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee			UEPPX	UEPT2	14.00	90.00	90.00				30.89	7.03		
	Calling Port	1		UEPPX	UEPTO	14.00	90.00	90.00				30.89	7.03		
-+	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port	1	-	UEPPX	UEPXA	14.00	90.00	90.00		-	1	30.89	7.03	1	1
-+-	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports	+		UEPPX	UEPXB	14.00	90.00	90.00				30.89	7.03	<u> </u>	
-+	2-Wire Voice Unbundled PBX LD DDD Terminals Port	1	<u> </u>	UEPPX	UEPXC	14.00	90.00	90.00		_	1	30.89	7.03	 	
- 	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	1		UEPPX	UEPXD	14.00	90.00	90.00				30.89	7.03		
-	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD	1					1								
	Capable Port	1	1	UEPPX	UEPXE	14.00	90.00	90.00				30.89	7.03	1	1
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy														
	Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00				30.89	7.03		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy]	
	Room Calling Port	1		UEPPX	UEPXM	14.00	90.00	90.00				30.89	7.03		
	2-Wire Voice Unbundled 1-W Out PBX Hotel/Hospital Economy	1			[l										
				UEPPX	UEPXN	14.00	90.00	90.00	1		i	30.89	7.03	1	Ì
	Administrative Calling Port TN			UEPPA	ULFAIN	14.00	90.00	50.00	+			30.03	7.03		
_	Administrative Calling Port TN 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00				30.89	7.03		

ONBON	DLE	NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
CATEGOR	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Dan	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Voice Unbundled PBX Collierville and Memphis Calling Port			UEPPX	UEPXU	14.00	90.00	90.00					30.89	7.03		
		2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Callling Port			UEPPX	UEPXV	14.00	90.00	90.00					30.89	7.03		
		Tennessee PBX 2-Way Combo Each Additional Trunk Collierville and Memphis Local Calling Plan			UEPPX	UEPA6	14.00	90.00	90.00					30.89	7.03		
		Tennessee PBX 2-Way Combo First Trunk Collierville and Memphis Local Calling Plan			UEPPX	UEPA7	14.00	90.00	90.00					30.89	7.03		
LC		NUMBER PORTABILITY															
		Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								
FE	EATU						•		•		•						
		All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00					30.89	7.03		
NO	ONRE	CURRING CHARGES - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPPX	USAC2		41.50	41.50					30.89	7.03		
		2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change			UEPPX	USACC		41.50	41.50					30.89	7.03		
ΑI	DDITIO	ONAL NRCs															
		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2	0.00	0.00	0.00					30.89	7.03		
		Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring						0.00	0.00					30.89	7.03		
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64					30.89	7.03		
		VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	RT														
UI		rt/Loop Combination Rates															
		2-Wire VG Coin Port/Loop Combo – Zone 1		1			26.48										
		2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3		2			30.31 35.32										
- 1.11		op Rates		3			35.32										
UI		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	12.48										
		2-Wire Voice Grade Loop (SL1) - Zone 1		2	UEPCO	UEPLX	16.31										
		2-Wire Voice Grade Loop (SL1) - Zone 3			UEPCO	UEPLX	21.32										
2-		Voice Grade Line Port Rates (Coin)								i i							
		2-Wire Coin 2-Way without Operator Screening and without Blocking (TN)			UEPCO	UEPTB	14.00	90.00	90.00					30.89	7.03		
		2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (NC, TN)			UEPCO	UEPRP	14.00	90.00	90.00					30.89	7.03		
		2-Wire Coin 2-Way with Operator Screening and 011 Blocking (TN)			UEPCO	UEPTA	14.00	90.00	90.00					30.89	7.03		
		2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (NC, TN)			UEPCO	UEPCA	14.00	90.00	90.00					30.89	7.03		
		22-Wire Coin Outward with Operator Screening and 011 Blocking (TN)			UEPCO	UEPTC	14.00	90.00	90.00					30.89	7.03		
		2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (TN)			UEPCO	UEPOT	14.00	90.00	90.00					30.89	7.03		
1.0		NUMBER PORTABILITY		1	OLFOO	ULFUI	14.00	90.00	90.00	 				30.09	7.03	 	
		Local Number Portability (1 per port)	1	1	UEPCO	LNPCX	0.35										-
NO		CURRING CHARGES - CURRENTLY COMBINED					3.00										
		2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPCO	USAC2		41.50	41.50					30.89	7.03		
		2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change			UEPCO	USACC		41.50	41.50					30.89	7.03		
AI		DNAL NRCs															
I IN ID		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPCO	USAS2	0.00	0.00	0.00					30.89	7.03		
		ORT/LOOP COMBINATIONS - MARKET BASED RATES VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	DOD=														
		VIIII E LEGALIE I CICIPE BLIS CINI Y - WITH 2-WIPE DID TRIBIN	PURI	1	1							ī	i l		i	1	1

ATE CONY ANY ELEMENTS ANY EL	NRUNDLED	NETWORK ELEMENTS - Tennessee					1	T					1_			ment: 2		bit: C
March Marc	ATEGORY	RATE ELEMENTS		Zone	В	cs	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
Silver No. Substitute Sub								Poc	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		
SWING NO LOUGH POWER DID TINK FOR CHORD - UNE Zone 2 2									First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
EVENT VICTOR VICTOR CONTROL																		
Committee Comm																		
PAYMEN Analog Voron Clonde Lorge - (SLE)** LIME Zone 1				3				56.00										
2-Mins Anatog Voto Grade Logo - 1031 - UNE Zono 2 2 LEPPK UECOT 11.00					HEDDY		LIEOD4	0.00										
2-Wire Analog Victor Grade Logo - (SE) 2- URE Zone 3 UEPPY UEPD 15,000 60,000 8.46 3.91 30,380 7.03																		
Eschange Points 2-Wins DID Port UEPPX UE													1				-	
Non-Recurrence CHARGES - CURRENITY COMBINED				3					600.00	45.00	8.45	3 01			30.80	7.03		
2-Wire Voto Grade Loop / 2-Wire DID Trank Port Combination UEPPX					OLITA		OLIDI	40.00	000.00	45.00	0.40	5.51	1		30.03	7.03		
Satch-Acts Top & MSAc only							+											
2-Wew Voter Grade Logs / 2-Wifer IDD Trush Fort Commension UEPPX					UEPPX		USAC1		100.00	42.50					30.89	7.03		
With BellSouth Allowshie Changes to pt MSAs only UEPPX USAC 100.00 42.50 30.88 7.03							1	1		50	1				22.30	1	1	İ
Totalphone Number/Trunk Group Establishment Charges UEPPX NDT 0.00 0.00 0.00 U.			l		UEPPX		USA1C		100.00	42.50					30.89	7.03	1	
OID Trunk Termination (One Per Port)																		
Additional DID Numbers for each Group of 20 ID Numbers UEPPX NO5 0.00	D	DID Trunk Termination (One Per Port)					NDT	0.00	0.00	0.00								
Reserve Non-Consecutive DID numbers	A·	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		ND4	0.00	0.00	0.00								
Reserve DID Numbers UEPPX NDV 0.00																		
COAL NUMBER PORTABILITY UEPPX UEPPX UEPPR 3.15 0.00 0.00 UEPPX																		
Local Number Portability (1 per port)					UEPPX		NDV	0.00	0.00	0.00								
WIRE ISDN DIGITAL GRADE LOOP WITH ZWIRE ISDN DIGITAL LINE SIDE PORT																		
Wile Port/Log Combination Rates							LNPCP	3.15	0.00	0.00								
2 WisDN Digital Grade Loop/2W ISDN Digital Line Side Port - 1 UEPPB UEPPR 32.27			NE SIDE	PORT														
UNE Zone 1																		
ZW ISDN Digital Grade Loop/ZW ISDN Digital Line Side Port - UNE Zone 2				١.,	LIEDDD	LIEDDO		00.07										
UNE Zone 2				1	UEPPB	UEPPR	1	32.27										
2WISDN Digital Grade Loop - UNE Zone 1 1 UEPPB UEPPR USL2X 16.20				2	LIEDDD	LIEDDD		24 70										
UNE Zone 3					OLFFB	ULFFR		34.76										
2-Wire ISDN Digital Grade Loop - UNE Zone 1				3	LIEPPR	LIEPPR		44 32										
2-Wire ISDN Digital Grade Loop - UNE Zone 2 2 UEPPB UEPPR USL2X 18.71 2-Wire ISDN Digital Grade Loop - UNE Zone 3 3 UEPPB UEPPR USL2X 28.25 Exchange Port 2-Wire ISDN Line Side Port Exchange Port 2-Wire ISDN Line Side Port UEPPB UEPPR UEPPB 80.00 525.00 400.00 75.00 70.00 NONRECURRING CHARGES - CURRENTLY COMBINED 2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion - Top 8 MSAs only UEPPB UEPPR USACB 0.00 225.00 225.00 ADDITIONAL NRCs 2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy Non Feature/Add Trunk UEPPB UEPPR USACB 0.00 225.00 225.00 ADDITIONAL NRCs 2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy Non Feature/Add Trunk UEPPB UEPPR USASB 212.88 10 30.89 7.03 LOCAL NUMBER PORTABILITY UEPPB UEPPR USASB 212.88 10 30.89 7.03 LOCAL NUMBER PORTABILITY UEPPB UEPPR USASB 212.88 10 30.89 7.03 LOCAL NUMBER PORTABILITY UEPPB UEPPR USASB 212.88 10 30.89 7.03 LOCAL NUMBER PORTABILITY UEPPB UEPPR UIVICA 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 DO 0.00 0.00 UEPPB UEPPB UIVICC 0.00 0.00 UEPPB UEPPB UEPPB UIVICC 0.00 0.00 UEPPB UEPPB UEPPB UIVICC 0.00 0.00 UEPPB UEPPB UEPPB UIVICC 0.00 0.00 UEPPB UEPPB UEPPB UIVICC 0.00 0.00 UEPPB UEPPB UEPPB UIVICC 0.00 0.00 UEPPB UEPPB UEPPB UIVICC 0.00 0.00				-			USI 2X											
2-Wire ISDN Digital Grade Loop - UNE Zone 3 3 UEPPB UEPPR UEPPB 80.00 525.00 400.00 75.00 70.00 30.89 7.03	- - -	This is but bigital stade 2005 Site 2016 !			02	<u> </u>	OOLLA	10.20										
2-Wire ISDN Digital Grade Loop - UNE Zone 3 3 UEPPB UEPPR UEPPB 80.00 525.00 400.00 75.00 70.00 30.89 7.03	2.	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	18.71										
NONRECURRING CHARGES - CURRENTLY COMBINED				3	UEPPB	UEPPR	USL2X	28.25										
2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port UEPPB UEPPR USACB 0.00 225.00 225.00 30.89 7.03					UEPPB	UEPPR	UEPPB	80.00	525.00	400.00	75.00	70.00			30.89	7.03		
Combination - Conversion - Top 8 MSAs only	NONREC	CURRING CHARGES - CURRENTLY COMBINED																
ADDITIONAL NRCS 2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy Non Feature / Add Trunk UEPPB UEPPR USASB 212.88 30.89 7.03																		
2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actry UEPPB UEPPR USASB 212.88					UEPPB	UEPPR	USACB	0.00	225.00	225.00					30.89	7.03		
Non Feature/Add Trunk			ļ				 	ļ			ļļ		ļ			ļ	ļ	ļ
LOCAL NUMBER PORTABILITY			İ		LIEBEE	HERRE	110465		0.00-								1	
Local Number Portability (1 per port)			1		UEPPB	UEPPR	USASB	 	212.88		ļ —		 		30.89	7.03	1	
B-CHANNEL USER PROFILE ACCESS:			 	-	LIEDDD	HEDDE	LNDCV	0.25	0.00	0.00	 		1			-		-
CVS/CSD (DMS/5ESS)					UEFFB	UEPPR	LINFUX	0.33	0.00	0.00								
CVS (EWSD)					LIEPPR	LIEPPR	LITLICA	0.00	0.00	0.00								
CSD											 		 			 	t	
B-CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)																	1	
CVS/CSD (DMS/5ESS)			C,MS, &	TN)			1	5.50	5.50	5.50	1					İ	1	İ
CVS (EWSD)			1			UEPPR	U1UCD	0.00	0.00	0.00								
USER TERMINAL PROFILE USER TERMINAL PROFILE USER TERMINAL PROFILE USER TERMINAL PROFILE USER TERMINAL PROFILE USER TERMINAL PROFILE USER TERMINAL PROFILE USER TERMINAL PROFILE USER TERMINAL PROFILE USER USER USER USER USER USER USER USER								0.00										
User Terminal Profile (EWSD only) VERTICAL FEATURES All Vertical Features - One per Channel B User Profile Interoffice Channel mileage each, including first mile and facilities termination UEPPB UEPPR UEPVF 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.	C	CSD				UEPPR	U1UCF	0.00	0.00	0.00								
VERTICAL FEATURES																		
All Vertical Features - One per Channel B User Profile UEPPB UEPPR UEPVF 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.					UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
Interoffice Channel mileage each, including first mile and facilities termination UEPPB UEPPR M1GNC 17.91 53.99 17.37																		
facilities termination UEPPB UEPPR M1GNC 17.91 53.99 17.37			ļ		UEPPB	UEPPR	UEPVF	0.00	0.00	0.00	ļ		ļ			ļ	ļ	ļ
			1				l		====							1	I	1
			!										ļ		ļ	ļ	-	
		nteroffice Channel mileage each, additional mile	L		UEPPB	UEPPR	M1GNM	0.173	0.00	0.00			ļ		ļ	ļ	-	
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT UNE Port/Loop Combination Rates			PURI				+	 			 		 			 	 	

<u>UNBUND</u> LI	ED NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	Charge - Manual Sv Order vs. Electronic
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrecurring First	Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$)	SOMAN	SOMAN
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE						Filst	Auu i	FIISt	Addi	JOINIEC	JOWAN	JOWAN	SOWAN	JOWAN	SOWAN
	Zone 1		1	UEPPP		982.73										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE															
	Zone 2		2	UEPPP		1,000.40										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		_													
	Zone 3 4-Wire DS1 Digital Loop - UNE Zone 1		3 1	UEPPP UEPPP	USL4P	1,023.59 57.73	1									
	4-Wire DS1 Digital Loop - UNE Zone 1 4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP	USL4P USL4P	75.40										
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	98.59	+									
_	Exchange Ports - 4-Wire ISDN DS1 Port		Ŭ	UEPPP	UEPPP	925.00	950.00	950.00	130.00	100.00			30.89	7.03		
NONE	RECURRING CHARGES - CURRENTLY COMBINED			CELLI	CEITI	020.00	300.00	300.00	100.00	100.00			00.00	7.00		
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															
	Combination - Conversion -Switch-As-Is Top 8 MSAs only			UEPPP	USACP	0.00	925.00	925.00			<u> </u>		30.89	7.03		<u></u>
ADDI	TIONAL NRCs															
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-			l			I 7									
	Inward/two way Telephone Numbers (except NC)			UEPPP	PR7TF		0.94								ļ	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -			UEPPP	PR7TO		22.36	22.36								
	Outward Tel Numbers (All States except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -			UEPPP	PR/IU		22.30	22.30								
	Subsequent Inward Telephone Numbers			UEPPP	PR7ZT		44.71	44.70								
LOCA	L NUMBER PORTABILITY			OLITI	110721		44.71	44.70								
	Local Number Portability (1 per port)			UEPPP	LNPCN	1.75	† †									
INTER	RFACE (Provsioning Only)			_												
	Voice/Data			UEPPP	PR71V	0.00		0.00								
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New o	or Additional "B" Channel			UEPPP	DD3D1/	0.00	00.00									
	New or Additional - Voice/Data B Channel New or Additional - Digital Data B Channel			UEPPP	PR7BV PR7BF	0.00	28.39 29.11									
	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00										
CALL	TYPES			ULFFF	FRIBD	0.00	29.39									
UALL	Inward			UEPPP	PR7C1	0.00	0.00	0.00								
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Interd	office Channel Mileage															
	Fixed Each Including First Mile			UEPPP	1LN1A	76.1825	145.98	109.85	19.55							
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3525										
	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT			ļ	-										1	
UNE	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC	+	93.28	+		-		 			-	†	
	4W DS1 Digital Loop/4W DD1TS Trunk Port - UNE Zone 1 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	1	2	UEPDC	+	110.95	 		1		 			1	1	
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		134.14										
UNE I	Loop Rates				İ		† †							İ		
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	57.53								<u> </u>	<u> </u>	
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	75.40										
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	98.59										
UNE	Port Rate			LIEBBO	1100.2	===		480 / -	100.5-							
NONE	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	750.00	982.57	450.10	196.09	19.23	1		30.89	7.03		-
NONE	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination				-		 								-	
	- Switch-As-Is Top 8 MSAs only			UEPDC	USAC4		312.91	312.91					30.89	7.03		
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with DS1 Changes Top 8 MSAs only			UEPDC	USAWA		312.91	312.91					30.89	7.03		
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk Top 8 MSAs only			UEPDC	USAWB		312.91	312.91					30.89	7.03		
ADDI	TIONAL NRCs						5:2:01	2.2.01					22.00			

EGORY	RATE ELEMENTS									ĺ		Svc Order Submitted	Incremental Charge -	Incremental Charge -	Incremental Charge -	Incrementa
,	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Elec per LSR	Manually per LSR	Manual Svc Order vs. Electronic- 1st			Charge - Manual Sv Order vs. Electronic Disc Add
\neg		,					Nonrecurring		Nonrecurring	Disconnect	<u> </u>		oss	Rates(\$)		
\dashv					+	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent				1 1											
	Service Activity Per Service Order	, ,	1	UEPDC	USAS4		94.88	94.88			ļ l				'	
\dashv	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -													·		
	Subsequent Channel Activation/Chan - 2-Way Trunk	, ,	1	UEPDC	UDTTA		108.67	108.67			ļ l		30.89	7.03	'	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent	-						•								
	Channel Activation/Chan - 1-Way Outward Trunk	, ,	l '	UEPDC	UDTTB		108.67	108.67			ŀ		30.89	7.03	1	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
	Activation/Chan Inward Trunk w/out DID	, ,	l '	UEPDC	UDTTC		108.67	108.67			ŀ		30.89	7.03	1	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan														1	
	Activation Per Chan - Inward Trunk with DID	, ,	1	UEPDC	UDTTD		108.67	108.67			ļ l		30.89	7.03	'	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan										ı				ı	
	Activation / Chan - 2-Way DID w User Trans	, !	l '	UEPDC	UDTTE	1	108.67	108.67		!			30.89	7.03	1 '	1
	AR 8 ZERO SUBSTITUTION										ı				ı	
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	590.00								
	B8ZS - Extended Superframe Format	,	·	UEPDC	CCOEF		0.00	590.00			I				,	
Alterna	te Mark Inversion															
	AMI -Superframe Format	,		UEPDC	MCOSF		0.00	0.00			l				1	
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00			ı				ı	
Telepho	one Number/Trunk Group Establisment Charges														1	
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00					ı				ı	
	Telephone Number for 1-Way Outward Trunk Group	,	·	UEPDC	UDTGY	0.00					I				,	
	Telephone Number for 1-Way Inward Trunk Group Without DID	,		UEPDC	UDTGZ	0.00					l				1	
	DID Numbers, Establish Trunk Group and Provide First Group														1	
	of 20 DID Numbers	, ,	l '	UEPDC	NDZ	0.00	0.00	0.00			ŀ				1	
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00										
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00										
	Reserve Non-Consecutive DID Nos.	, ,		UEPDC	ND6	0.00	0.00	0.00			I					
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00								
	ted DS1 (Interoffice Channel Mileage) -															
	ofor 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port	لــــــا	L'													
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities	, ,	l '								ŀ				1	
	Termination)			UEPDC	1LNO1	75.83	145.98	109.85	19.66	14.99						
		, ,	l '								ŀ				1	
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles		L'	UEPDC	1LNOA	0.3525	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities	, Т	1	1	1				1						1	1
	Termination)		└	UEPDC	1LNO2	0.00	0.00	0.00			 '			ļ	 '	
	Interoffice Channel Mileage - Additional rate per mile - 9-25	, ,	l '	1		1								1	1 '	
	miles		└	UEPDC	1LNOB	0.3525	0.00	0.00			<u> </u>				 '	
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities	, ,	i '	1		1								1	1 '	
	Termination)		└	UEPDC	1LNO3	0.00	0.00	0.00	<u> </u>		<u> </u>			<u> </u>	 '	
	l	, ,	l '	1	1									1	1 '	
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles		——'	UEPDC	1LNOC	0.3525	0.00	0.00	↓		<u> </u>			<u> </u>	 '	
	Local Number Portability, per DS0 Activated		<u> </u>	UEPDC	LNPCP	3.15		0.00			ļ!				ļ'	
	Central Office Termininating Point		└	UEPDC	CTG	0.00								<u> </u>	 '	
	DS1 LOOP WITH CHANNELIZATION WITH PORT		<u> </u>				 				ļ!					
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activ			L			.				ļ				Ļ	
	em can have various rate combinations based on type and nun	nper of	ports i	used	\rightarrow		 		↓		ļ'			 	├ ───	
	S1 Loop		<u></u> '	LIEDMO	1101.50	F7 ===	1 2 2 2				├ ──'			 '	└─ ──'	
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	57.73	0.00	0.00	 		ļ ¹			<u> </u>	 -	
	4-Wire DS1 Loop - UNE Zone 2	-	2	UEPMG	USLDC	75.40		0.00	1		ļ			 	├ ───	
	4-Wire DS1 Loop - UNE Zone 3	20)	3	UEPMG	USLDC	98.59	0.00	0.00						<u> </u>	\vdash	
	SO Channelization Capacities (D4 Channel Bank Configuration	15)	 '	UEPMG	VUM24	131.87	0.00	0.00	1		ļ		30.89	7.00	├ ───	
	24 DSO Channel Capacity - 1 per DS1	-	 '	UEPMG	VUM24 VUM48	131.87 263.74			1		ļ		30.89	7.03	├ ───	
	48 DSO Channel Capacity - 1 per 2 DS1s		 '	UEPMG	VUM48 VUM96	527.48		0.00					30.89	7.03 7.03	\vdash	
	96 DSO Channel Capacity -1per 4 DS1s 144 DS0 Channel Capacity -1 per 6 DS1s			UEPMG UEPMG				0.00	 		\vdash			7.03	\vdash	
	144 DOU Channel Capacity - 1 per 6 DS1s				VUM14	791.42		0.00					30.89			<u> </u>
	102 DC0 Channel Canacity 1 pg-0 DC4-	, —		LIEDMC	1// 18/44.0		0.00	0.00			1		20.00	7.00		
	192 DS0 Channel Capacity -1 per 8 DS1s 240 DS0 Channel Capacity - 1 per 10 DS1s	<u>'</u>	L	UEPMG UEPMG	VUM19 VUM20	827.76 1,318.70	0.00	0.00					30.89 30.89	7.03 7.03		

UNBUNDLE	D NETWORK ELEMENTS - Tennessee				1	1					T -	-		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	384 DS0 Channel Capacity - 1 per 16 DS1s		1	UEPMG UEPMG	VUM38 VUM40	2,109.92	0.00	0.00					30.89 30.89	7.03 7.03		
	480 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity -1 per 24 DS1s		<u> </u>	UEPMG	VUM57	2,637.40 3,164.88	0.00	0.00	-				30.89	7.03		
-	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3,692.36	0.00	0.00	-				30.89	7.03		1
Non-R	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chani	eliztio					0.00					30.03	7.05		
	imum System configuration is One (1) DS1, One (1) D4 Channe						Stem									
	ples of this configuration functioning as one are considered Ac															
	NRC - Conversion (Currently Combined) with or without		1													
	BellSouth Allowed Changes - Top 8 MSAs Only			UEPMG	USAC4	0.00	303.61	15.74					30.89	7.03		
Syster	m Additions Where Currently Combined and New (Not Current	ly Comb	ined)													
In Den	nsity Zone 1 Top 8 MSAs															
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc															
	Fea Activation -			UEPMG	VUMD4	0.00	704.68	441.48	138.36	16.41			30.89	7.03		
Bipola	ar 8 Zero Substitution															
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	590.00								
	Clear Channel Capability Format - Extended Superframe -															
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	590.00								
Altern	ate Mark Inversion (AMI)		<u> </u>				0.00									
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
Fuels	Extended Superframe Format	:41-	D = =4	UEPMG	МСОРО	0.00	0.00	0.00								
	ange Ports Associated with 4-Wire DS1 Loop with Channelization of Ports	on with	Port		+	-										
Excha	inge Ports		<u> </u>						-							
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	14.00	0.00	0.00	0.00	0.00			30.89	7.03		
	Line Side Outward Channelized PBX Trunk Port - Business		1	UEPPX	UEPOX	14.00	0.00	0.00	0.00	0.00			30.89	7.03		
	Line Side Odtward Charmenzed F BX Trunk F Oit - Business			OLI I X	OLI OX	14.00	0.00	0.00	0.00	0.00			30.03	7.03		
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00			30.89	7.03		
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	40.00	0.00	0.00	0.00	0.00			30.89	7.03		
Featu	re Activations - Unbundled Loop Concentration															
	Feature (Service) Activation for each Line Side Port Terminated															
	in D4 Bank			UEPPX	1PQWM	0.66	40.00	20.00	6.00	5.00						
	Feature (Service) Activation for each Trunk Side Port Terminated															
	in D4 Bank			UEPPX	1PQWU	0.66	110.00	30.00	75.00	15.00						
Telepi	hone Number/ Group Establishment Charges for DID Service															
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Numbers	<u> </u>	1	UEPPX	ND6	0.00	0.00	0.00								ļ
1	Reserve DID Numbers	 	 	UEPPX	NDV	0.00	0.00	0.00	1					 	 	ļ
Local	Number Portability	 	1	UEPPX	LNPCP	3.15	0.00	0.00						 	 	
CEATI	Local Number Portability - 1 per port URES - Vertical and Optional	 	1	UEPPA	LINPUP	3.15	0.00	0.00						 	 	
	Switching Features Offered with Line Side Ports Only	├	 	-	+	 	 				-			-	-	1
Local	All Features Available	 	1	UEPPX	UEPVF	0.00	0.00	0.00	+							1
UNRUNDI ED	CENTREX PORT/LOOP COMBINATIONS - COST BASED RATE:	s	 	OLI FA	OLF VI	0.00	0.00	0.00	l l					 	 	
	st Based Rates are applied where BellSouth is required by FCC		State (Commission rule to	provide Unb	undled Local S	witching or Sw	itch Ports								
	tures shall apply to the Unbundled Port/Loop Combination - C								dled Port section	on of this Rate	Exhibit.					1
	d Office and Tandem Switching Usage and Common Transport											oin Port/Lo	op Combinat	ions.		
	e first and additional Port nonrecurring charges apply to Not C												•		Additional N	RCs mav
	also and are categorized accordingly.	,	,				,	3 300				.5	,			
	rket Rates for Unbundled Centrex Port/Loop Combination will	be nead	otiated	on an Individual Ca	ase Basis, un	til further notic	e.		I							
	CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only			1	1	T			İ					İ	İ	i e
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo														1	1
	Port/Loop Combination Rates (Non-Design)	1														1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
	Non-Design	<u> </u>	1	UEP91	<u> </u>	14.18	<u> </u>							<u> </u>		<u></u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design					18.01										
			2	UEP91												•

Version 3Q02: 09/06/02 Page 344 of 362

UNBUNI	DLF	NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Fyhil	oit: C
3.150.11												Svc Order	Svc Order	Incremental		Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
												Elec	Manually		Manual Svc		Manual Svc
CATEGOR	RY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						- (17			per Lor	per LOK	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							_	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design		3	UEP91		23.02										
UN	NE Po	rt/Loop Combination Rates (Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
		Design		1	UEP91		18.26										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Design		2	UEP91		23.33										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Design		3	UEP91		29.98										
UN	NE Lo	op Rate															
		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	12.48							İ	İ	İ	
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	16.31							İ	İ	İ	
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1	21.32							1	1	1	
		2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	16.56										
		2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	21.63	 			1			1	t	1	
		2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	28.28										
UN	NE Po			Ť	02. 0.	02002	20.20										
		es (Except North Carolina and Sout Carolina)															
—	. 0	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
-		2-Wire Voice Grade Port (Centrex 900 termination)Basic Local			02. 0.	02: :/:			10.20	0.10	0.01		00.00	7.00			
		Area			UEP91	UEPYB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
h +		2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			OLI 01	OLI ID	1.70	22.17	10.20	0.40	0.01		00.00	7.00			
		Area			UEP91	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
		2-Wire Voice Grade Port (Centrex from diff Serving Wire			OLI 01	OLI III	1.70	22.17	10.20	0.40	0.01		00.00	7.00			
		Center)2 Basic Local Area			UEP91	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OLI 31	OLI TIVI	1.70	22.14	10.20	0.40	3.31		30.03	7.03			
		Term - Basic Local Area			UEP91	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
		2-Wire Voice Grade Port terminated in on Megalink or equivalent			OLI 31	OLI 12	1.70	22.14	10.20	0.43	3.31		30.03	7.03			
		- Basic Local Area			UEP91	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
		2-Wire Voice Grade Port Terminated on 800 Service Term -			OLI 31	OLI 13	1.70	22.14	10.20	0.43	3.31		30.03	7.03			
		Basic Local Area			UEP91	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
Α.	ΚV	LA, MS, & TN Only			OLF91	ULF 12	1.70	22.14	13.23	0.45	3.91		30.09	7.03			
AL	_, K1,	2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03	-		
-		2-Wire Voice Grade Fort (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03	-		
-					UEP91	UEPQH	1.70	22.14			3.91		30.89	7.03	-		
\vdash		2-Wire Voice Grade Port (Centrex with Caller ID)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire		1	OLF31	ULFUN	1.70	22.14	15.25	8.45	3.91	1	30.09	7.03	 	1	
		2-wire voice Grade Port (Centrex from all Serving wire Center)2			UEP91	UEPQM	1.70	22.14	15.25	8.45	3.91	İ	30.89	7.03	1		
\vdash		2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service		 	OLFSI	UEPQIVI	1.70	22.14	15.25	8.45	3.91		30.89	7.03	 		
		Z-write voice Grade Port, Dill Serving Wire Center - 800 Service Term			UEP91	UEPQZ	1.70	22.14	15.25	8.45	3.91	İ	30.89	7.03	1		
-		TOTAL		 	OLFBI	ULFUZ	1.70	22.14	15.25	0.45	3.91		30.09	1.03		 	
		2 Wire Voice Crade Bort terminated in an Magalists			UEP91	LIEBOO	1.70	22.14	15.05	0.45	2.04	İ	20.00	7.00	1		
\vdash		2-Wire Voice Grade Port terminated in on Megalink or equivalent		 		UEPQ9			15.25	8.45	3.91		30.89	7.03	 		
		2-Wire Voice Grade Port Terminated on 800 Service Term		-	UEP91	UEPQ2	1.70	22.14	15.25	8.45	3.91		30.89	7.03	 		
H_LC		witching Control Intercom Funtionality, per port		 	UEP91	URECS	0.6381			-					 		
, .		Centrex Intercom Funtionality, per port		-	UEF91	UKEUS	0.6381								 		
LC		umber Portability		 	LIED04	LNPCC	0.05			-					 		
-		Local Number Portability (1 per port)		 	UEP91	LINPUU	0.35	 		-					-		
Fe	eature			1	LIED04	LIED\/E	0.00			1			20.00	7.00	 	1	
-		All Standard Features Offered, per port All Select Features Offered, per port		1	UEP91 UEP91	UEPVF UEPVS	0.00	433.78					30.89 30.89	7.03 7.03	 	 	
\vdash				 		UEPVS		433.78		-					 		
		All Centrex Control Features Offered, per port		1	UEP91	UEPVC	0.00						30.89	7.03	 	 	
N/	ARS	Habitandlad Nationals Appear Depleton Constitution		1	LIEDO4	LIADOY	0.00	2.22	0.00	1			00.00	7.00	 	1	
		Unbundled Network Access Register - Combination		<u> </u>	UEP91	UARCX	0.00	0.00	0.00				30.89	7.03	-		
		Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00	ļ			30.89	7.03			
		Unbundled Network Access Register - Outdial		<u> </u>	UEP91	UAROX	0.00	0.00	0.00				30.89	7.03	-		
		aneous Terminations		<u> </u>		1				ļ							
2-1		Trunk Side		<u> </u>	LIEDOA	OFNICO			.=				00.0-		-		
⊢		Trunk Side Terminations, each		<u> </u>	UEP91	CENA6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
In		ice Channel Mileage - 2-Wire			LIEDOA	144000	40.50	00 11	45.55	0 :-	0.01		00.00	7.00			
		Interoffice Channel Facilities Termination - Voice Grade		<u> </u>	UEP91	M1GBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			

NRUNDLE	D NETWORK ELEMENTS - Tennessee			1	<u> </u>						1 -	1 -		ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
							Nonrecurring		Nonrecurring	Disconnect				Rates(\$)		
-						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0174	11131	Addi	11130	Auu	CONTEC	JOHAN	JOWAN	JONAN	JOHIAN	JOHAN
Featur	e Activations (DS0) Centrex Loops on Channelized DS1 Service	е				******										
D4 Ch	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP91	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP91	TPQW7	0.00										
	Different Wire Center			UEP91	1PQWP	0.66										
	Billiotett Wile Genter			OLI OI	11 QVVI	0.00										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.66									1	
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop								1					1		
	Slot			UEP91	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.66										
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex								ļl							
	Conversion - Currently Combined Switch-As-Is with allowed			LIEDOA	110,400		4.00	0.00				20.00	7.00			
	changes, per port New Centrex Standard Common Block			UEP91 UEP91	USAC2 M1ACS	0.00	1.03 658.60	0.29				30.89 30.89	7.03 7.03			
	New Centrex Standard Common Block			UEP91	M1ACC	0.00	658.60					30.89	7.03			
	Secondary Block, per Block			UEP91	M2CC1	0.00	73.55					30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	68.57					30.89	7.03			
UNE-P	CENTREX - 5ESS (Valid in All States)														1	
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE P	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP95		14.18										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95		18.01										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP95		23.02										
UNE P	ort/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo		1	UEP95		18.26										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		-	UEP95	+	18.26										
	Design		2	UEP95		23.33										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			02. 00	1	20.00			1						1	
	Design		3	UEP95		29.98										
UNE L	oop Rate						<u> </u>		<u> </u>					İ		
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	12.48		-		-			·			
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	16.31			ļ					ļ	ļ	
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	21.32										
_	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95 UEP95	UECS2 UECS2	16.56								1	1	
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95 UEP95	UECS2	21.63 28.28	-		-						 	
(INF P	ort Rate		3	OFL:32	UEUSZ	20.28			 					1	 	
All Sta					1				1						1	
510	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03	1	1	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.70	22.14	15.25	8.45	3.91		30.89	7.03	<u> </u>		
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP95	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP95	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area			UEP95	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP95	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			

<u>JNBUND</u> L	ED NETWORK ELEMENTS - Tennessee												Attachi	ment: 2	Exhil	bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Nonrecurring		Nonrecurring	Disconnect				Rates(\$)	DISC 1St	DISC Add I
					+	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
AL. K	(Y, LA, MS, SC, & TN Only							,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	151	,,,,,,	0020					
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent		<u> </u>	UEP95	UEPQ9	1.70	22.14	15.25	8.45	3.91		30.89	7.03	1	-	-
	2-Wire Voice Grade Port Terminated on 800 Service Term	!	<u> </u>	UEP95	UEPQ2	1.70	22.14	15.25	8.45	3.91		30.89	7.03	1	-	
	GA Only	ļ							ļ							
Local	l Switching	ļ	<u> </u>	LIEBAE	LIBEGO				ļ						.	.
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.6381										
Local	l Number Portability	<u> </u>	<u> </u>	ļ											ļ	
	Local Number Portability (1 per port)	<u> </u>	<u> </u>	UEP95	LNPCC	0.35									ļ	
Featu																
	All Standard Features Offered, per port			UEP95	UEPVF	0.00						30.89	7.03			
	All Select Features Offered, per port			UEP95	UEPVS	0.00	433.78					30.89	7.03			
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00						30.89	7.03			
NARS																
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				30.89	7.03			
Misce	ellaneous Terminations															
	re Trunk Side															
	Trunk Side Terminations, each			UEP95	CEND6	8.78	47.75	47.01	9.21	8.47		30.89	7.03			
4-Wir	re Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP95	M1HD1	35.55	75.93	38.15				30.89	7.03			
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	108.67					30.89	7.03			
Interd	office Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP95	MIGBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0174										
Featu	ure Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
	hannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.66										
	·															
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP95	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP95	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.66			İ							
Non-l	Recurring Charges (NRC) Associated with UNE-P Centrex		1				İ		i i			l				
	NRC Conversion Currently Combined Switch-As-Is with allowed						İ		1							
	changes, per port	1		UEP95	USAC2		1.03	0.29				30.89	7.03		I	
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	658.60		1			30.89	7.03			
	New Centrex Customized Common Block		1	UEP95	M1ACC	0.00	658.60		i i			30.89	7.03			
	NAR Establishment Charge, Per Occasion	1	t	UEP95	URECA	0.00	68.57		† †			30.89	7.03		t	
UNE-	P CENTREX - DMS100 (Valid in All States)	1	t		1	2.20			† †				1.30		t	
	re VG Loop/2-Wire Voice Grade Port (Centrex) Combo	1	t	1					† †						t	
	Port/Loop Combination Rates (Non-Design)	1	t	1					† †						t	
J.1L	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo			1							i					1
	S Loop, L S Tolos Sidde i on (Control) i on Control	1	1	UEP9D	1						1	1	1	1	1	1

UNBUND	LED NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Exhib	oit: C
330.10											Svc Order	Svc Order	Incremental		Incremental	Incremental
											Submitted	Submitted		Charge -	Charge -	Charge -
		Intori									Elec	Manually		Manual Svc	Manual Svc	Manual Svc
CATEGOR	Y RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m									per Lore	per Lore	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															DISC 1St	DISC Add I
						Rec	Nonrecurring			g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															í
	Non-Design		2	UEP9D		18.01										l
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	· [ł
	Non-Design		3	UEP9D		23.02										<u> </u>
UN	E Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-														í
	Design		1	UEP9D		18.26										<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	· [ł
	Design		2	UEP9D		23.33										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1														i
$oxed{oxed}$	Design	<u> </u>	3	UEP9D		29.98										1
UN	E Loop Rate													ļ		.
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	12.48										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	16.31										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	21.32										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.56										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	21.63										l
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	28.28										l .
	E Port Rate															<u> </u>
AL	L STATES															<u> </u>
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			<u> </u>
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															ł
	Area			UEP9D	UEPYB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			l
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															ł
	Area			UEP9D	UEPYC	1.70	22.14	15.25	8.45	3.91		30.89	7.03			<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local															ł
	Area			UEP9D	UEPYD	1.70	22.14	15.25	8.45	3.91		30.89	7.03			l
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local															ł
	Area			UEP9D	UEPYE	1.70	22.14	15.25	8.45	3.91		30.89	7.03			<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local															í
	Area			UEP9D	UEPYF	1.70	22.14	15.25	8.45	3.91		30.89	7.03			<u></u>
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local															í
	Area			UEP9D	UEPYG	1.70	22.14	15.25	8.45	3.91		30.89	7.03			<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															ł
	Area			UEP9D	UEPYT	1.70	22.14	15.25	8.45	3.91		30.89	7.03			<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local															ł
	Area			UEP9D	UEPYU	1.70	22.14	15.25	8.45	3.91		30.89	7.03			l .
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local										1	<u> </u>	<u> </u>]		1
	Area			UEP9D	UEPYV	1.70	22.14	15.25	8.45	3.91		30.89	7.03	ļ		.
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local															i
	Area	<u> </u>		UEP9D	UEPY3	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local										1		Ì	Ì		1
	Area	<u> </u>		UEP9D	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp										1		Ì	Ì		1
	Indication))3 Basic Local Area	<u> </u>		UEP9D	UEPYW	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3			l	1											i
$oxed{oxed}$	Basic Local Area	<u> </u>		UEP9D	UEPYJ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			l	1					_	1		_	Ì		1
$oxed{oxed}$	2 Basic Local Area	ļ		UEP9D	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			l							1		Ì	Ì		1
\vdash	Basic Local Area	ļ		UEP9D	UEPYO	1.70	22.14	15.25	8.45	3.91		30.89	7.03	ļ		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3								_							i
$oxed{oxed}$	Basic Local Area	ļ		UEP9D	UEPYP	1.70	22.14	15.25	8.45	3.91		30.89	7.03	ļ		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			l												ł
$oxed{oxed}$	Basic Local Area	ļ	<u> </u>	UEP9D	UEPYQ	1.70	22.14	15.25	8.45	3.91		30.89	7.03	ļ		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			l												ł
$oxed{oxed}$	Basic Local Area	<u> </u>		UEP9D	UEPYR	1.70	22.14	15.25	8.45	3.91		30.89	7.03			1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			l	1				_	_			_			1
	Basic Local Area	1		UEP9D	UEPYS	1.70	22.14	15.25	8.45	3.91	l	30.89	7.03]		

ONBOND	ED NETWORK ELEMENTS - Tennessee			ı	<u> </u>									nent: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3															
	Basic Local Area			UEP9D	UEPY4	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			LIEDOD	LIEDV6	4.70	00.44	45.05	0.45	0.04		00.00	7.00			
	Basic Local Area			UEP9D	UEPY5	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3	1		UEF9D	UEPTO	1.70	22.14	15.25	0.40	3.91		30.09	7.03		1	
	Basic Local Area			UEP9D	UEPY7	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			02. 05	02	0		10.20	0.10	0.01		00.00	7.00			
	Term			UEP9D	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	t														
	Basic Local Area			UEP9D	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic	;														
	Local Area			UEP9D	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
AL,	KY, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3	1		UEP9D	UEPQC	1.70		15.25		3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.70		15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D UEP9D	UEPQE	1.70 1.70		15.25	8.45	3.91 3.91		30.89 30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3 2-Wire Voice Grade Port (Centrex / EBS-M5312)3	1		UEP9D	UEPQF UEPQG	1.70		15.25 15.25	8.45 8.45	3.91	-	30.89	7.03 7.03		-	-
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	1.70		15.25	8.45	3.91	1	30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.70		15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.70		15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication)3			UEP9D	UEPQW	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3	1		UEP9D	UEPQQ	1.70	22.14	15.25	8.45	3.91		30.89	7.03		-	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-14116 VOICE Grade Fort (Centrexvaller SVVC /EDS-IVIST12)2, 3	1	 	OLFBD	ULFUK	1.70	22.14	15.25	0.45	3.91	1	30.09	1.03		 	1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3	1	1	UEP9D	UEPQS	1.70	22.14	15.25	8.45	3.91		30.89	7.03		I	
	2 voice clade i ort (centrewaller ovvo /EDO-140312)2, 3	1	!	021 00	OL: 40	1.70	22.14	10.20	0.40	5.91	<u> </u>	30.03	7.03		I	<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3	1		UEP9D	UEPQ4	1.70	22.14	15.25	8.45	3.91		30.89	7.03		1	
		1			1								1.30		1	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3	1	1	UEP9D	UEPQ5	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3		<u></u>	UEP9D	UEPQ6	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
		1														
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3	<u> </u>	<u> </u>	UEP9D	UEPQ7	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1													1	
	Term	-	<u> </u>	UEP9D	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03		1	
	2 Wire Voice Crade Port terminated in an Magelial as a suital as	.1	1	LIEDOD	LIEBOO	1.70	22.44	15.05	0.45	2.04		20.00	7.00		I	
	2-Wire Voice Grade Port terminated in on Megalink or equivalen 2-Wire Voice Grade Port Terminated on 800 Service Term	4	 	UEP9D UEP9D	UEPQ9 UEPQ2	1.70 1.70	22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91	-	30.89 30.89	7.03 7.03		 	-
1.00	al Switching	1	1	OFLAD	UEFQZ	1.70	22.14	15.25	0.45	3.91		30.89	1.03		+	
LOC	Centrex Intercom Funtionality, per port	 	 	UEP9D	URECS	0.6381			 		-				t	-
Loc	al Number Portability	1	!	021 00	OI LEGO	0.0001			1		<u> </u>				I	<u> </u>
	Local Number Portability (1 per port)	1	†	UEP9D	LNPCC	0.35									1	
Fea	tures	1			1	2.00									1	
1	All Standard Features Offered, per port	1	1	UEP9D	UEPVF	0.00					1	30.89	7.03		1	1

UNB	UNDLE	D NETWORK ELEMENTS - Tennessee				-						T -			ment: 2		bit: C
ATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Increments Charge - Manual Sv Order vs. Electronic
														1st	Add'l	Disc 1st	Disc Add
							Rec	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		All Select Features Offered, per port			UEP9D	UEPVS	0.00	433.78					30.89	7.03			
		All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00						30.89	7.03			
	NARS																1
		Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				30.89	7.03			
		Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				30.89	7.03			
	Minnel	Unbundled Network Access Register - Outdial aneous Terminations			UEP9D	UAROX	0.00	0.00	0.00			1	30.89	7.03			<u> </u>
		Trunk Side				-											
	Z-Wile	Trunk Side Terminations, each			UEP9D	CEND6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
	4-Wire	Digital (1.544 Megabits)			OLF 9D	CLINDO	0.70	22.14	13.23	0.43	3.91		30.09	7.03			+
	7-44116	DS1 Circuit Terminations, each			UEP9D	M1HD1	35.55	75.93	38.15	<u> </u>		 	30.89	7.03	 	1	
	1	DS0 Channels Activiated per Channel	1		UEP9D	M1HDO	0.00	108.67	30.13				30.89	7.03	1		
	Interof	fice Channel Mileage - 2-Wire	1				5.50						30.00		1		
		Interoffice Channel Facilities Termination			UEP9D	MIGBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			
	1	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0174								İ		1
	Featur	Activations (DS0) Centrex Loops on Channelized DS1 Service	e									Ì			1		
		nnel Bank Feature Activations															
		Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.66										1
		Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.66										
		Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.66										
		Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
		Different Wire Center			UEP9D	1PQWP	0.66										
		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.66										
		Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.00										
		Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWQ	0.66 0.66										
	Non D	ecurring Charges (NRC) Associated with UNE-P Centrex		-	UEF9D	IFQWA	0.00										
	NOII-R	NRC Conversion Currently Combined Switch-As-Is with allowed										1					+
		changes, per port			UEP9D	USAC2		1.03	0.29				30.89	7.03			
	+	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	658.60	0.23			1	30.89	7.03			+
		New Centrex Customized Common Block			UEP9D	M1ACC	0.00	658.60					30.89	7.03			+
		NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	68.57					30.89	7.03			t
	UNE-P	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)			02. 02	0112071		00.07					00.00	7.00			
		VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
		ort/Loop Combination Rates (Non-Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															1
		Non-Design		1	UEP9E		14.18										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design		2	UEP9E		18.01										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
		Non-Design		3	UEP9E		23.02										
	UNE P	ort/Loop Combination Rates (Design)															
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1														
		Design		1	UEP9E		18.26										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo		2	UEP9E		23.33										
		Design			UEP9E	-	23.33										
		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	3	UEP9E		29.98								1		
	LINE	Design Dop Rate	1	3	OLFSE		29.98	1		1		}		1	1		
	ONE L	2-Wire Voice Grade Loop (SL 1) - Zone 1	-	1	UEP9E	UECS1	12.48	1		1				-	-	-	
	+	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	1	2	UEP9E	UECS1	16.31					1	-		1		
	+	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP9E	UECS1	21.32	1		<u> </u>		 			 	1	+
	1	2-Wire Voice Grade Loop (SL 2) - Zone 1	1	1	UEP9E	UECS2	16.56	 								1	
	1	2-Wire Voice Grade Loop (SL 2) - Zone 2	1	2	UEP9E	UECS2	21.63								1		
	1	2-Wire Voice Grade Loop (SL 2) - Zone 3	1	3	UEP9E	UECS2	28.28			1				1	 		
	LINE D	ort Rate		1		1		1				1	1		1	1	t

NRONDFI	D NETWORK ELEMENTS - Tennessee		1	ı							_			ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment: Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
AL, F	L, KY, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9E	UEPYB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP9E	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP9E	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP9E	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP9E	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP9E	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
AL. K	Y, LA, MS, & TN Only			OLI OL	OLI 12	1.70	22.17	10.20	0.40	0.01		00.00	7.00			
, n., n	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03	1	1	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9E	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.6381										
Local	Number Portability															
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Featu																
	All Standard Features Offered, per port			UEP9E	UEPVF	0.00	100.70					30.89	7.03			
	All Select Features Offered, per port			UEP9E UEP9E	UEPVS	0.00	433.78					30.89	7.03			
NARS	All Centrex Control Features Offered, per port		1	UEP9E	UEPVC	0.00			-			30.89	7.03		-	
NARS				UEP9E	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial		1	UEP9E	UAR1X	0.00	0.00	0.00			1	30.89	7.03			
	Unbundled Network Access Register - Outdial		1	UEP9E	UAROX	0.00	0.00	0.00				30.89	7.03			
Misce	Ilaneous Terminations			OLI OL	UAITOX	0.00	0.00	0.00				30.03	7.00			
	e Trunk Side															
	Trunk Side Terminations, each			UEP9E	CEND6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
4-Wir	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9E	M1HD1	35.55	75.93	38.15				30.89	7.03			
	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	108.67					30.89	7.03			
Interd	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0174										
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	е														
D4 Cr	annel Bank Feature Activations			LIEDOE	1PQWS	0.00										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E		0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP9E	1PQW6	0.66										
_	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP9E	1PQW7	0.66										
_	Different Wire Center			UEP9E	1PQWP	0.66										-
_	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tije Line/Trunk Loop			UEP9E	1PQWV	0.66									-	<u> </u>
	Slot		1	UEP9E	1PQWQ	0.66										

ONRONDE	ED NETWORK ELEMENTS - Tennessee			1								_		ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)	1	<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.66	11130	Auu i	11130	Auu i	JOHLC	JONAN	JOHAN	JONAN	JOHIAN	JOHAN
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex			OLI 3L	II QWA	0.00										†
14011-	NRC Conversion Currently Combined Switch-As-Is with allowed				+											-
	changes, per port			UEP9E	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	658.60	0.23			1	30.89	7.03			+
	New Centrex Standard Common Block			UEP9E	M1ACC	0.00	658.60		+		-	30.89	7.03		-	-
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	68.57		+		-	30.89	7.03		-	-
LINE				UEF9E	UKECA	0.00	00.37					30.69	7.03			
	P CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)		-		-											
	re VG Loop/2-Wire Voice Grade Port (Centrex) Combo		<u> </u>													
UNE	Port/Loop Combination Rates (Non-Design)		<u> </u>													
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	١.	LIEBOO										l	I	
	Non-Design	 	1	UEP93		14.18	├									
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	l	1	l										Ì	I	
	Non-Design		2	UEP93		18.01										<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	1											<u> </u>		
	Non-Design		3	UEP93		23.02										
UNE	Port/Loop Combination Rates (Design)															ĺ
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															1
	Design		1	UEP93		18.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP93		23.33										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OL1 50		20.00										
	Design		3	UEP93		29.98										
LINE	Loop Rate		3	ULF 93		25.50										
UNE			1	UEP93	UECS1	12.48										
	2-Wire Voice Grade Loop (SL 1) - Zone 1															
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	16.31										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	21.32										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	16.56										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	21.63										L
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	28.28										
	Port Rate															
AL, Ł	(Y, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			1
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															1
	Area			UEP93	UEPYB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															1
	Area	l	1	UEP93	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03	Ì	I	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire	1	1	 	1	0		20	55	0.01	1	30.00	7.00	†	†	†
	Center)2 Basic Local Area	l	1	UEP93	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03	Ì	I	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service		!		J. 1141	1.70	22.17	10.20	0.40	0.01	1	30.00	7.00	 	t	
	Term - Basic Local Area	l	1	UEP93	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03	Ì	I	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1	1	OLF 30	ULFIZ	1.70	22.14	15.25	0.40	3.91	 	30.09	1.03	 	 	+
				LIEDOS	LIEDVO	1.70	22.14	45.05	8.45	3.91		30.89	7.03			
	- Basic Local Area			UEP93	UEPY9	1.70	22.14	15.25	8.45	3.91	ļ	30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP93	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire	1	1											<u> </u>	_	
	Center)2			UEP93	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	l	1													1
	Term	l	1	UEP93	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03	Ì	I	
																1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	l	1	UEP93	UEPQ9	1.70	22.14	15.25	8.45	3.91		30.89	7.03		1	
	2-Wire Voice Grade Port Terminated on 800 Service Term	1	t	UEP93	UEPQ2	1.70	22.14	15.25	8.45	3.91	1	30.89	7.03	1	1	1
Loca	Switching	1	1		J 32	1.70	22.17	10.20	0.40	0.01	1	30.03	7.00	†	†	
Loca	Centrex Intercom Funtionality, per port		!	UEP93	URECS	0.6381	 				1			 	t	
		1	1	OLI 30	UINEUU	0.0501									ļ	
Loca	I Number Portability															

CATEGORY																bit: C
	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted		Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						_	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		
-						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Feature	res						1									
	All Standard Features Offered, per port			UEP93	UEPVF	0.00										
-+	All Centrex Control Features Offered, per port			UEP93	UEPVC	0.00	†				1					1
NARS	7 iii Conii ox Conii or 1 Cataree Cinerca; per peri			02.00	02. 70	0.00	†				1					1
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00				30.89	7.03			
-+	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00		0.00			1	30.89	7.03			1
-+	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00		0.00			1	30.89	7.03			1
Miscel	laneous Terminations			OL1 00	O/ II CO/C	0.00	0.00	0.00				00.00	7.00			
	Trunk Side						1									
2-11116	Trunk Side Terminations, each			UEP93	CEND6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
4-10/:	Digital (1.544 Megabits)			OLF 30	CLINDO	0.78	22.14	15.25	0.40	3.91	 	30.09	1.03	1	1	
4-99176	DS1 Circuit Terminations, each			UEP93	M1HD1	35.55	75.93	38.15	 		 	30.89	7.03	1	1	
-+-	DS0 Channels Activated, Per Channel			UEP93	M1HD0	0.00		30.13	 		 	30.89	7.03	-	-	-
Interef	ffice Channel Mileage - 2-Wire			OLF 30	טטוווואו	0.00	100.07		 			30.09	1.03	-	 	
interor	Interoffice Channel Facilities Termination		-	UEP93	MIGBC	18.58	22.14	15.25	8.45	3.91	 	30.89	7.03			
\longrightarrow	Interoffice Channel Facilities Termination Interoffice Channel mileage, per mile or fraction of mile			UEP93 UEP93	MIGBC	0.0174		15.25	8.45	3.91		30.89	7.03	-	 	
			-	UEP93	IVIIGBIVI	0.0174	 		 							
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	е														
D4 Cha	annel Bank Feature Activations			LIEDOO	40014/0	0.00										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP93	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP93	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop															
	Slot			UEP93	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.66										
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP93	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP93	M1ACS	0.00	658.60					30.89	7.03			
	New Centrex Customized Common Block			UEP93	M1ACC	0.00	658.60					30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP93	URECA		68.57					30.89	7.03			
Note 1	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD						l		İ							
	2 - Requres Interoffice Channel Mileage						1									
Note 3	- Requires Specific Customer Premises Equipment						1					İ				
UNBUNDLED (CENTREX PORT/LOOP COMBINATIONS - MARKET RATES															
	ket Rates are applied where BellSouth is not required by FCC a	and/or \$	State C	ommission rule to p	rovide Unbu	ndled Local S	witching or Swi	tch Ports.								
	urring Charges for all Standard Centrex and Centrex Conrol Fe						l I					İ				
	Office and Tandem Switching Usage and Common Transport					ibit shall appl	y to all combina	tions of loop/	port network el	lements excer	t for UNE C	oin Port/Lo	op Combinat	ions.	İ	1
	first and additional Port nonrecurring charges apply to Not Cu														Additional NR	RCs mav
	also and are categorized accordingly.						,	g goo				.5	. ,			,
	CENTREX - 1AESS - (Valid in AL.FL.GA.KY.LA.MS.&TN only	١					1		1			1	1	1	1	
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo				 		† †				†	 	 	 	 	
	ort/Loop Combination Rates (Non-Design)						†									1
- 1011211	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+ -		 		 		1	 	-		-	
	Non-Design	l	1	UEP91		26.48]			l	Ì	Ì	Ì	1
-+-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			0_1 01		20.40	†									1
1	Non-Design	l	2	UEP91		30.31]				Ì	Ì	Ì	
-+-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OE1 31	1	30.31	 		 		1	1	1	1	1	1
	Non-Design	l	3	UEP91		35.32						l	Ì	Ì	Ì	
			J	OFLAI	+	30.32	+		 				 	 	 	
LIME	ort/Loop Combination Pates (Pesian)										1	1	1			1
UNE P	ort/Loop Combination Rates (Design)						 									
UNE P	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		4	LIED01		20.50										
UNE P			1	UEP91		30.56										

Version 3Q02: 09/06/02 Page 353 of 362

<u>NBUND</u> LE	D NETWORK ELEMENTS - Tennessee												Attach	ment: 2	Exhi	bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
						Rec	Nonrecurring		Nonrecurring	Disconnect		l		Rates(\$)	ı	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		l _													
	Design		3	UEP91		42.28										
UNE L	oop Rate		<u> </u>	LIEDO4	115004	10.10										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91 UEP91	UECS1	12.48										
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1 UECS1	16.31 21.32										+
-	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS1	16.56										+
	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	21.63			-					-	-	+
-	2-Wire Voice Grade Loop (SL 2) - Zone 3			UEP91	UECS2	28.28										+
UNE P				OLI 91	OLCOZ	20.20										+
	ates (Except North Carolina and Sout Carolina)				+											+
All Ste	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03	<u> </u>	<u> </u>	
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local						55.55	.0.00	20.00			30.00		1	1	
	Area			UEP91	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03	1	1	
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local				1 1								1		1	1
	Area			UEP91	UEPYH	14.00	90.00	45.00	20.00	10.00		30.89	7.03	1	1	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															1
	Center)2 Basic Local Area			UEP91	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															1
	Term - Basic Local Area			UEP91	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP91	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP91	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
AL, K۱	Y, LA, MS, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2		<u> </u>	UEP91	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term		<u> </u>	UEP91	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2 Mins Vaiss Conds Boot torreinsted in an Manalink or annivelent			UEP91	UEPQ9	14.00	90.00	45.00	20.00	10.00		20.00	7.03			
_	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91		14.00	90.00	45.00 45.00	20.00	10.00		30.89 30.89	7.03			├
Loos	2-Wire Voice Grade Port Terminated on 800 Service Term Switching	-	 	UEP91	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
Local	Centrex Intercom Funtionality, per port			UEP91	URECS	0.6381			-					-	-	+
Local	Number Portability	-	1	OLF31	UNLUS	0.0301	-		+			-	1	+	 	+
Local	Local Number Portability (1 per port)		1	UEP91	LNPCC	0.35	1				1			1	1	+
Featur			 	02.31	LIVI OO	0.33	1		l l					 	 	+
ı catul	All Standard Features Offered, per port			UEP91	UEPVF	0.00	 					30.89	7.03	t	t	+
_	All Select Features Offered, per port			UEP91	UEPVS	0.00	433.78		 			30.89	7.03	I	I	†
_	All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00	100.70		 			30.89	7.03	I	I	†
NARS	* * *				122.10	0.00						55.50		1	1	
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00	İ			30.89	7.03	1	1	†
	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00	İ			30.89	7.03		1	1
	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				30.89	7.03			
Miscel	llaneous Terminations					-		-								1
	Trunk Side															
	Trunk Side Terminations, each			UEP91	CENA6	8.78	90.00	45.00	20.00	10.00		30.89	7.03			
Interof	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0174				<u> </u>						
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	е														
D4 Ch	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66								ļ	ļ	↓
	1	ľ	1	1	1						l		ı	I	1	1

ONROND	LED NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
ATEGORY	Y RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Sv Order vs. Electronic
													1st	Add'l	Disc 1st	Disc Add'
						Rec	Nonrecurring		Nonrecurring			•		Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP91	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			LIEBOA	4D014/D	0.00										
	Different Wire Center			UEP91	1PQWP	0.66					1					
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tivate Line Loop Slot	+	1	OLI 31	II QVVV	0.00										
	Slot			UEP91	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.66										
Nor	n-Recurring Charges (NRC) Associated with UNE-P Centrex															
	Conversion - Currently Combined Switch-As-Is with allowed															
	changes, per port		<u> </u>	UEP91	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	658.60					30.89	7.03			
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	658.60					30.89	7.03			
	Secondary Block, per Block	1	<u> </u>	UEP91	M2CC1	0.00	73.55				<u> </u>	30.89	7.03		ļ	<u> </u>
	NAR Establishment Charge, Per Occasion			UEP91	URECA		68.57					30.89	7.03			
	E-P CENTREX - 5ESS (Valid in All States) Vire VG Loop/2-Wire Voice Grade Port (Centrex) Combo										1					1
	E Port/Loop Combination Rates (Non-Design)	-			_											
UNI	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo															
	Non-Design		1	UEP95		26.48										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	-	<u> </u>	OL1 30		20.40										
	Non-Design		2	UEP95		30.31										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	-														
	Non-Design		3	UEP95		35.32										
UNI	E Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-														
	Design		1	UEP95		30.56										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	-	_													
	Design Design		2	UEP95		35.63										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	-	2	LIEDOE		40.00										
LIMI	Design E Loop Rate		3	UEP95		42.28										
ON	2-Wire Voice Grade Loop (SL 1) - Zone 1	+	1	UEP95	UECS1	12.48					1					1
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	16.31										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	21.32										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	16.56										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	21.63										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	28.28										
UNI	E Port Rate															
All	States															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			LIEBOE	LIEDVIII	44.00	00.00	45.00	00.00	40.00		00.00	7.00			
	Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire	+	<u> </u>	UEP95	UEPYH	14.00	90.00	45.00	20.00	10.00	1	30.89	7.03		1	1
	Center)2 Basic Local Area			UEP95	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1	 	021 00	OLI IIWI	17.00	30.00	45.00	20.00	10.00	 	30.03	7.03	 	1	
	Term - Basic Local Area			UEP95	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivaler	t				30	22.20									
	- Basic Local Area	1		UEP95	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03	1		
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area		<u> </u>	UEP95	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
AL,	, KY, LA, MS, SC, & TN Only				_			•		•						
	2-Wire Voice Grade Port (Centrex)	1		UEP95	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)	1	<u> </u>	UEP95	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03		ļ	
	2-Wire Voice Grade Port (Centrex with Caller ID)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire	1	<u> </u>	UEP95	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03		ļ	
				1		i i	1		1	i i	•	•		1	1	1

ONRONDL	ED NETWORK ELEMENTS - Tennessee			1										ment: 2		bit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Charge -	Incremental Charge - Manual Svc Order vs. Electronic-	Incrementa Charge - Manual Sv Order vs. Electronic
													1st	Add'l	Disc 1st	Disc Add'
						n	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP95	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
F1 0	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	GA Only I Switching		<u> </u>						-							
LOCA	Centrex Intercom Funtionality, per port			UEP95	URECS	0.6381			-							
Loca	Number Portability			OLF 93	UKLCS	0.0361										
2000	Local Number Portability (1 per port)			UEP95	LNPCC	0.35			1							
Featu				02. 00	2.1. 00	0.00										
	All Standard Features Offered, per port			UEP95	UEPVF	0.00						30.89	7.03			
	All Select Features Offered, per port			UEP95	UEPVS	0.00	433.78					30.89	7.03			
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00						30.89	7.03			
NAR																
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				30.89	7.03			
	ellaneous Terminations															
2-Wi	e Trunk Side															
	Trunk Side Terminations, each		<u> </u>	UEP95	CEND6	8.78	47.75	47.01	9.21	8.47		30.89	7.03			
4-Wii	e Digital (1.544 Megabits) DS1 Circuit Terminations, each		<u> </u>	LIEDOS	141104	05.55	75.00	00.45				00.00	7.00			
				UEP95 UEP95	M1HD1 M1HDO	35.55	75.93	38.15				30.89 30.89	7.03			
Intor	DS0 Channels Activated, each office Channel Mileage - 2-Wire			UEF95	MITHDO	0.00	108.67		-			30.69	7.03			
inter	Interoffice Channel Facilities Termination			UEP95	MIGBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0174	30.00	43.00	20.00	10.00		30.03	7.03			
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e		02. 00	02	0.0111										
	hannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.66										
	·															
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP95	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center		<u> </u>	UEP95	1PQWP	0.66										
	Factors Astination on D.4 Channel Beats Brights Line Long Clat			LIEDOE	40000/	0.00										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop	 	 	UEP95	1PQWV	0.66	 		 				-	1	1	
	Slot	1	1	UEP95	1PQWQ	0.66]						1			
	Feature Activation on D-4 Channel Bank WATS Loop Slot	†		UEP95	1PQWA	0.66	 		† †		<u> </u>		 	1	1	<u> </u>
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex	1			~***	0.00			1				1			
1	NRC Conversion Currently Combined Switch-As-Is with allowed			1	1											
	changes, per port			UEP95	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	658.60					30.89	7.03	<u> </u>	<u> </u>	
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	658.60					30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	68.57					30.89	7.03			
	P CENTREX - DMS100 (Valid in All States)															
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo	ļ		ļ			ļļ		ļ .				ļ			
UNE	Port/Loop Combination Rates (Non-Design)	<u> </u>	<u> </u>						 				 		ļ	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	4	LIEBOD		20.40										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	-	1	UEP9D	+	26.48			1				 	1	1	1
	Non-Design		2	UEP9D		30.31										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	 		OLFBD	+ -	30.31	1		1				1			1
	Non-Design	1	3	UEP9D	1 1	35.32							1			
UNF	Port/Loop Combination Rates (Design)	†	-	02.100	+ -	33.32	 		† †		<u> </u>		 	1	1	
0.12	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1		H	+ -		 		† †		<u> </u>		 	1	1	<u> </u>
	Design	1	1	UEP9D		30.56						1				

ONBONDER	D NETWORK ELEMENTS - Tennessee											001		ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			1	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP9D		35.63										ļ
	Design		3	UEP9D		42.28										
LINE	oop Rate		3	OLFBD	+	42.20					-					-
UNE L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	12.48					-					
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	16.31										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	21.32										4
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.56										4
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	21.63										4
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	28.28			ļ							
	ort Rate		<u> </u>													
ALL S	TATES		<u> </u>	LIEDAD	1									ļ		ļ
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
1	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local	1	1											l		
	Area			UEP9D	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			1
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
	Area			UEP9D	UEPYC	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local															
	Area			UEP9D	UEPYD	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local															1
	Area			UEP9D	UEPYE	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local															
	Area			UEP9D	UEPYF	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local															
	Area			UEP9D	UEPYG	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															
	Area			UEP9D	UEPYT	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
+	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			02. 02	02	1 1.00	00.00	10.00	20.00	10.00	1	00.00	7.00			
	Area			UEP9D	UEPYU	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local			OLI 3D	OLI 10	14.00	30.00	+5.00	20.00	10.00		30.03	7.00			+
	Area			UEP9D	UEPYV	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local			OLFBD	OLFIV	14.00	90.00	45.00	20.00	10.00	-	30.09	7.03			
	Area			UEP9D	UEPY3	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
			<u> </u>	UEP9D	UEFTS	14.00	90.00	45.00	20.00	10.00		30.69	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local			LIEDOD	UEPYH	44.00	00.00	45.00	20.00	40.00		20.00	7.03			
	Area		<u> </u>	UEP9D	UEPTH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			LIEDOD	LIED (M)	44.00	00.00	45.00	00.00	40.00		00.00	7.00			
	Indication))3 Basic Local Area		_	UEP9D	UEPYW	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3															
	Basic Local Area		_	UEP9D	UEPYJ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			4
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2 Basic Local Area			UEP9D	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3															
	Basic Local Area			UEP9D	UEPYO	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3															
	Basic Local Area			UEP9D	UEPYP	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
1	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3	1	1											l		
	Basic Local Area			UEP9D	UEPYQ	14.00	90.00	45.00	20.00	10.00		30.89	7.03	ļ		ļ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3															
	Basic Local Area			UEP9D	UEPYR	14.00	90.00	45.00	20.00	10.00		30.89	7.03	ļ		ļ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3															
	Basic Local Area	<u> </u>	<u></u>	UEP9D	UEPYS	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3															
	Basic Local Area	<u> </u>	<u></u>	UEP9D	UEPY4	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3															
I	Basic Local Area	<u></u>	L	UEP9D	UEPY5	14.00	90.00	45.00	20.00	10.00	<u></u>	30.89	7.03	<u> </u>		<u></u>
ĺ	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3				i				İ							
1	Basic Local Area		I	UEP9D	UEPY6	14.00	90.00	45.00	20.00	10.00		30.89	7.03	ĺ	I	

ONRONDE	ED NETWORK ELEMENTS - Tennessee			ı	, ,							1 -		ment: 2	1	bit: C
											Svc Order Submitted	Svc Order Submitted	Incremental Charge -	Incremental Charge -	Incremental Charge -	Incrementa Charge -
											Elec	Manually	Manual Svc	Manual Svc		Manual Sv
ATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc			RATES(\$)				,				
AILGORI	KATE ELEMENTS	m	Zone	ВСЗ	0300			KAILS(4)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
					1		Nonrecurring		Nonrecurring	Disconnect		l .	OSS	Rates(\$)		<u>. </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3							7.00		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,					00/	
	Basic Local Area			UEP9D	UEPY7	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9D	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	Basic Local Area			UEP9D	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic															
	Local Area			UEP9D	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
AL, K	Y, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	14.00	90.00	45.00	20.00	10.00		30.89	7.03			1
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication)3			UEP9D	UEPQW	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
								<u> </u>								
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service												-			
	Term			UEP9D	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
													-			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	14.00	90.00	45.00	20.00	10.00		30.89	7.03	ļ		ļ
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03	ļ		1
Local	Switching				1											1
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.6381										1
Local	Number Portability															1
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										<u> </u>
Featu				L	<u> </u>							ļ				1
	All Standard Features Offered, per port			UEP9D	UEPVF	0.00						30.89	7.03			1
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	433.78					30.89	7.03			1
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00						30.89	7.03	ļ		ļ
NARS					1											<u> </u>
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00			1	30.89	7.03			
	Unbundled Network Access Register - Outdial		l	UEP9D	UAROX	0.00	0.00	0.00			1	30.89	7.03		1	1

NRONDE	LED NETWORK ELEMENTS - Tennessee			ı							r -			ment: 2		bit: C
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Increment Charge - Manual Sv Order vs Electronic Disc Add
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	cellaneous Terminations															
2-Wi	ire Trunk Side															
	Trunk Side Terminations, each			UEP9D	CEND6	8.78	90.00	45.00	20.00	10.00		30.89	7.03			
4-Wi	ire Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	35.55	75.93	38.15				30.89	7.03			
	DS0 Channels Activiated per Channel		<u> </u>	UEP9D	M1HDO	0.00	108.67					30.89	7.03			
Inter	roffice Channel Mileage - 2-Wire		<u> </u>	LIEDOD	MODO	10.50	00.00	45.00	00.00	10.00		00.00	7.00			
	Interoffice Channel Facilities Termination		<u> </u>	UEP9D	MIGBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0174										
	ture Activations (DS0) Centrex Loops on Channelized DS1 Service	e	-		_											
D4 C	Channel Bank Feature Activations			LIEDOD	4DOMC	0.00										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot		-	UEP9D	1PQWS	0.66								 	 	1
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot		1	UEP9D	1PQWA	0.66										
Non	-Recurring Charges (NRC) Associated with UNE-P Centrex		1	OLI 3D	II QWA	0.00										
11011	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	658.60	0.20				30.89	7.03			
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	658.60					30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP9D	URECA		68.57					30.89	7.03			
UNE	E-P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															
	ire VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Non-Design		1	UEP9E		26.48										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9E		30.31										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9E		35.32										
UNE	E Port/Loop Combination Rates (Design)			OLI SL	+	33.3 <u>Z</u>										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Design		1	UEP9E		30.56										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		Ė	UEP9E												
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2		1	35.63										
<u> </u>	Design		3	UEP9E	+	42.28									-	
UNE	Loop Rate		1	LIEDOE	LIECC4	40.40								 	 	1
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E UEP9E	UECS1 UECS1	12.48 16.31								 	 	1
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3	-	3	UEP9E UEP9E	UECS1	21.32			 					-		
	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1	-	1	UEP9E	UECS1	16.56	ŀ							1	 	-
	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	21.63			 					 	 	
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3	-	3	UEP9E	UECS2	28.28			 					 	t	
UNF	E Port Rate	-		J_1 JL	02002	20.20			 					 	t	
	FL, KY, LA, MS, & TN only	1			1		 		 					 	I	t
1,	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03	İ	1	
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9E	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP9E	UEPYH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			

UNDUNDL	ED NETWORK ELEMENTS - Tennessee			1								_		ment: 2		ibit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Sv Order vs.
							[N		T 81	D'					D130 130	Disc Add I
					+	Rec	Nonrecurring	A -1-111	Nonrecurring		COMEC	COMAN		Rates(\$)	COMAN	COMAN
	2-Wire Voice Grade Port (Centrex from diff Serving Wire		1		+		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Center)2 Basic Local Area			UEP9E	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service		1	UEF9E	UEPTIVI	14.00	90.00	45.00	20.00	10.00		30.69	7.03			+
	Term - Basic Local Area			UEP9E	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent		1	OLI OL	OLI 12	14.00	30.00	40.00	20.00	10.00		00.00	7.00			+
	- Basic Local Area			UEP9E	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term -															1
	Basic Local Area			UEP9E	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
AL, F	(Y, LA, MS, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			LIEDOE	LIEDO:											
	Center)2		1	UEP9E	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03	1	1	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP9E	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
$\longrightarrow \longmapsto$	Term	-	1	UEP9E	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03	1	1	+
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Fort Terminated in 611 Meganin of equivalent			UEP9E	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
Loca	Switching			OLI OL	OLI QZ	14.00	50.00	40.00	20.00	10.00		00.00	7.00			+
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.6381										1
Loca	Number Portability															+
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										1
Featu																1
	All Standard Features Offered, per port			UEP9E	UEPVF	0.00						30.89	7.03			
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	433.78					30.89	7.03			
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00						30.89	7.03			
NARS																
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00				30.89	7.03			
Mina	Unbundled Network Access Register - Outdial		1	UEP9E	UAROX	0.00	0.00	0.00				30.89	7.03			+
	ellaneous Terminations re Trunk Side		1													
2-9911	Trunk Side Terminations, each		1	UEP9E	CEND6	8.78	90.00	45.00	20.00	10.00		30.89	7.03		1	+
4-Wir	re Digital (1.544 Megabits)		1	OLI SL	CLINDO	0.70	30.00	43.00	20.00	10.00		30.03	7.05			+
7 ***	DS1 Circuit Terminations, each			UEP9E	M1HD1	35.55	75.93	38.15				30.89	7.03			1
	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	108.67					30.89	7.03			1
Interd	office Channel Mileage - 2-Wire															1
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0174										
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 C	hannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.66							ļ		ļ	
$\!\!\!\!+\!\!\!\!-$	Feature Activation on D-4 Channel Bank FX line Side Loop Slot		-	UEP9E	1PQW6	0.66							 	1	ļ	+
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9E	1PQW7	0.66										1
-+	Feature Activation on D-4 Channel Bank Centrex Loop Slot -		 	OEF9E	IFQW/	0.06						-	 	1	1	+
	Different Wire Center			UEP9E	1PQWP	0.66						1	1			1
-+	Smotshit this defice		1	02. 02	// 54111	0.00					1			 	1	
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.66						1	1			
- 	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop	1				2.30	i i								İ	1
	Slot			UEP9E	1PQWQ	0.66						1	1			
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.66										
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex					-		-								
T	NRC Conversion Currently Combined Switch-As-Is with allowed		1]			
				LUEDOE	LICACO		4 00	0.29	1		1	30.89	7.03	1	1	
	changes, per port New Centrex Standard Common Block			UEP9E UEP9E	USAC2 M1ACS	0.00	1.03 658.60	0.29				30.89	7.03			

NRONDLE	D NETWORK ELEMENTS - Tennessee				<u> </u>									ment: 2		ibit: C
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge Manual S Order vs
							INIa mana a comina a I		I Name and a series of	Dianamant					D130 130	DISC Add
						Rec	Nonrecurring		Nonrecurring		001150	001111		Rates(\$)	001141	
	NAD Fatablishment Channe Bas Occasion			UEP9E	URECA	0.00	First 68.57	Add'l	First	Add'l	SOMEC	30.89	SOMAN 7.03	SOMAN	SOMAN	SOMAN
LINE	NAR Establishment Charge, Per Occasion CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)		-	UEP9E	URECA	0.00	08.57					30.89	7.03			+
			-													+
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo ort/Loop Combination Rates (Non-Design)		-													+
ONL					+				 							+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP93		26.48										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			ULF 93	+	20.40			1							+
	Non-Design		2	UEP93		30.31										
_	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLF 93	+	30.31			 							+
	Non-Design		3	UEP93		35.32										
LINE	ort/Loop Combination Rates (Design)		3	UEF93	+	33.32										+
UNE	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -			 	1				1		-	1		1	1	+
	Design	1	1	UEP93		30.56					1					
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		-	OLF 93	+	30.30			 							+
	Design	1	2	UEP93		35.63					1					
-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEF93	+	33.03			 							+
	Design	1	3	UEP93		42.28					1					
LINE	oop Rate		3	UEF93		42.20										+
ONE L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP93	UECS1	12.48										+
			2	UEP93	UECS1	16.31										+
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP93		21.32										+
_	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1											+
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1		UECS2	16.56										+
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	21.63										+
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	28.28										+
	ort Rate				-											
AL, K	7, LA, MS, & TN only 2-Wire Voice Grade Port (Centrex) Basic Local Area		-	UEP93	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
			-	UEF93	UEPTA	14.00	90.00	45.00	20.00	10.00		30.09	7.03			+
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP93	LIEDVA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
				UEP93	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			UEP93	UEPYH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
-	Area 2 Wire Voice Crade Bort (Centrey from diff Sening Wire			UEF93	UEPTH	14.00	90.00	45.00	20.00	10.00		30.69	7.03			+
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			UEP93	UEPYM	14.00	00.00	45.00	20.00	10.00		20.00	7.03			
	Center)2 Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service		-	UEF93	UEPTIVI	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
				UEP93	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	Term - Basic Local Area		-	UEP93	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP93	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
_	2-Wire Voice Grade Port Terminated on 800 Service Term -		-	UEF93	UEPT9	14.00	90.00	45.00	20.00	10.00		30.69	7.03			+
	Basic Local Area			UEP93	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
			-	UEP93		14.00	90.00			10.00						+
	2-Wire Voice Grade Port (Centrex)		-	UEP93	UEPQA			45.00	20.00			30.89	7.03			+
_	2-Wire Voice Grade Port (Centrex 800 termination)		-		UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2	1		UEP93	UEPQM	14.00	90.00	45.00	20.00	10.00	1	30.89	7.03			
				UEP93	UEPQIVI	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			LIEDOO	UEDO7	44.00	00.00	45.00	00.00	40.00		00.00	7.00			
	Term			UEP93	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			-
	O Mine Veice Conde Destance instead in an Manufiel and in the	1		LIEDOS	LIEDOO	44.00	00.00	45.00	20.00	40.00	1	20.00	7.00			
-	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	14.00	90.00	45.00	20.00	10.00	ļ	30.89	7.03	-	1	+
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ2	14.00	90.00	45.00	20.00	10.00	ļ	30.89	7.03	-	1	+
Local	Switching		-	LIEDOS	LIDECC	0.0004			 			1		-	1	+
1	Centrex Intercom Funtionality, per port			UEP93	URECS	0.6381										+
Local	Number Portability			LIEDOS	LNCCC	0.05								1		
F4	Local Number Portability (1 per port)			UEP93	LNCCC	0.35			ļ							+
Featu				LIEDOS	HED) "E	2.00			ļ		ļ			-	1	
-	All Standard Features Offered, per port			UEP93	UEPVF	0.00			ļ		ļ			-	1	
NADO	All Centrex Control Features Offered, per port			UEP93	UEPVC	0.00										
NARS	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00	ļ		ļ	30.89	7.03	-	1	

RONDLE	D NETWORK ELEMENTS - Tennessee													ment: 2		bit: C
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Increme
											Submitted	Submitted	Charge -	Charge -	Charge -	Charg
											Elec	Manually	Manual Svc		Manual Svc	Manual
TEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)				per LSR	Order vs.	Order vs.	Order vs.	Order
		m						.,,			per Lor	per Lor	Electronic-	Electronic-	Electronic-	Electro
													1st	Add'l	Disc 1st	Disc A
						Dee	Nonrecurring		Nonrecurring	Disconnect		1	oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00				30.89	7.03			
Miscel	laneous Terminations															
2-Wire	Trunk Side															
	Trunk Side Terminations, each			UEP93	CEND6	8.78	90.00	45.00	20.00	10.00		30.89	7.03			
4-Wire	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP93	M1HD1	35.55	75.93	38.15				30.89	7.03			1
	DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	108.67					30.89	7.03			1
	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP93	MIGBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.0174										
Featur	e Activations (DS0) Centrex Loops on Channelized DS1 Service	-		OL: 00	IVIIODIVI	0.0174	+									
	annel Bank Feature Activations	Ĭ														
D4 0116	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.66										
	realtire Activation on B-4 channel Bank Centrex Ecop Glot			OLI 93	11 QVV0	0.00										
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop		 	OLI 33	II QVV0	0.00	-					1				
	Slot			UEP93	1PQW7	0.66										
_	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			ULF 93	IF Q VV I	0.00	-		-		-					1
	Different Wire Center			UEP93	1PQWP	0.66										
	Different wife Center			UEP93	IFQWF	0.00										1
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.66										
_	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop			UEP93	IPQWV	0.00										<u> </u>
	Slot			UEP93	1PQWQ	0.66										
				UEP93	1PQWQ											
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.66										<u> </u>
	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP93	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block		ļ	UEP93	M1ACS	0.00	658.60					30.89	7.03			ļ
	New Centrex Customized Common Block			UEP93	M1ACC	0.00	658.60					30.89	7.03			ļ
	NAR Establishment Charge, Per Occasion			UEP93	URECA		68.57					30.89	7.03			ļ
	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD	ļ	<u> </u>													<u> </u>
	2 - Requres Interoffice Channel Mileage															
Note 3	- Requires Specific Customer Premises Equipment			I												1

LOCA	I INTE	RCONNECTION - Alabama												Attachment:	2	Exhibit: D	
LOCA	AL IIVI L	RCONNECTION - Alabama	1		1	1	I					Svc Order	Svc Order	Incremental			Incremental
												Submitted			Charge -	Charge -	Charge -
												Elec		Manual Svc			Manual Svc
CATE	GORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RA [*]	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						(+)			per Lak	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
														151	Add I	DISC 1St	DISC Add I
							Rec	Nonre	curring	Nonrecurring	Disconnect				Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL		CONNECTION (CALL TRANSPORT AND TERMINATION)															
	SINGLE	RATE FOR LOCAL AND ISP-BOUND TRAFFIC															
		Single Rate for Local and ISP-bound traffic, per minute of use					0045										i l
	DATES	(1/1/02 - 12/31/02) FOR LOCAL TRANSIT AND MTA TRAFFIC					.0015										
		M SWITCHING															
	IANDL	Tandem Switching Function Per MOU			OHD		0.0005692										
		Multiple Tandem Switching, per MOU (applies to intial tandem			0.15		0.0000002										
		only)			OHD		0.0005692										i l
		Tandem Intermediary Charge, per MOU*			OHD		0.0015										
		harge is applicable only to transit traffic and is applied in ad	dition to	o appli	cable switching and	or interconn	ection charges	i.									i
	TRUNK	CHARGE															ı
		Installation Trunk Side Service - per DS0			OHD	TPP++		333.69	56.91								
		Dedicated End Office Trunk Port Service-per DS0**	<u> </u>		OHD	TDE0P	0.00										igcup
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
	** TL:-	Dedicated Tandem Trunk Port Service-per DS1**	1 : 41	Frad O	OH1 OH1MS	TDW1P	0.00	l mata alamand									
		rate element is recovered on a per MOU basis and is included ON TRANSPORT (Shared)	in the	Ena O	Tice Switching and	l andem Swit	cning, per wo	J rate elements	5								
	COMINI	Common Transport - Per Mile, Per MOU	1		OHD		0.0000026										
		Common Transport - Facilities Termination Per MOU			OHD		0.0003685										
LOCAL	INTER	CONNECTION (DEDICATED TRANSPORT)	1		OTID		0.0000000										
		OFFICE CHANNEL - DEDICATED TRANSPORT															
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
		Per Mile per month			OHL, OHM	1L5NF	0.0101										i l
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -															1
		Facility Termination per month			OHL, OHM	1L5NF	24.15	54.82		13.79							i
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile															i
		per month			OHL, OHM	1L5NK	0.0101										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility															i l
		Termination per month			OHL, OHM	1L5NK	17.28	54.82		13.79							
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			OHL, OHM	1L5NK	0.0101										i l
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility	1		Onl, Onivi	ILSINK	0.0101										
		Termination per month	1		OHL, OHM	1L5NK	17.28	54.82		13.79							ı l
 	 	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per	1		OTTE, OTTIVI	ILOINIX	17.20	54.02		15.75							$\overline{}$
1		month	1		OH1, OH1MS	1L5NL	0.2067										i l
		Interoffice Channel - Dedicated Tranport - DS1 - Facility	†	1	- ,		5.2501										
1		Termination per month	1	1	OH1, OH1MS	1L5NL	68.75	163.61		28.88							1
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per															1
L		month	<u> </u>	<u></u>	OH3, OH3MS	1L5NM	4.67										
1		Interoffice Channel - Dedicated Transport - DS3 - Facility	1	1		I											, 7
L		Termination per month	ļ	<u> </u>	OH3, OH3MS	1L5NM	804.02	325.51		116.91							
<u> </u>	LOCAL	CHANNEL - DEDICATED TRANSPORT		<u> </u>	OLU OLUM	TEEVO	45.00	200.12	00.00	70.00	0.00				-		
—	1	Local Channel - Dedicated - 2-Wire Voice Grade per month Local Channel - Dedicated - 4-Wire Voice Grade per month	1	 	OHL, OHM OHL, OHM	TEFV2 TEFV4	15.96 17.06	386.19 387.06	66.33 67.20	73.28 74.22	6.39 7.33						
—	1	Local Channel - Dedicated - 4-wire voice Grade per month Local Channel - Dedicated - DS1 per month	 	 	OHL, OHM OH1	TEFHG	41.52	387.06 354.94	307.43	74.22 44.38	30.52						
-	1	2004 Chamici Dedicated - DOT per month	-	 	0111	121110	71.32	554.54	307.43	77.30	30.32						$\overline{}$
1		Local Channel - Dedicated - DS3 Facility Termination per month	1	1	ОНЗ	TEFHJ	476.04	903.03	527.87	238.97	167.16						ı l
	LOCAL	INTERCONNECTION MID-SPAN MEET	†	1	1			222.50	02.1.07	200.07	.070						$\overline{}$
		If Access service ride Mid-Span Meet, one-half the tariffed se	rvice Lo	cal Ch	annel rate is applica	ble.											
		Local Channel - Dedicated - DS1 per month	<u></u>		OH1MS	TEFHG	0.00	0.00									
		Local Channel - Dedicated - DS3 per month			OH3MS	TEFHJ	0.00	0.00									
	MULTII	PLEXERS									•						
		Channelization - DS1 to DS0 Channel System	<u> </u>		OH1, OH1MS	SATN1	122.50	182.08	125.14	21.07	19.58						igcup
	<u> </u>	DS3 to DS1 Channel System per month	ļ	<u> </u>	OH3, OH3MS	SATNS	201.37	356.28	187.94	66.51	63.65						
	N-4-	DS3 Interface Unit (DS1 COCI) per month	1	<u> </u>	OH1, OH1MS	SATCO	15.39	13.15	9.43	.,,							
	Notes:	If no rate is identified in the contract, the rates, terms, and co	ondition	is for t	ne specific service o	r tunction w	iii be as set fort	ın ın applicabl	e BellSouth ta	гітт.					l	Page	of 8
	10															i ago	

LOCAL INT	ERCONNECTION - Florida												Attachment:		Exhibit: D	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			TES(\$)					Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonred		Nonrecurring					Rates(\$)		
						-	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
OCAL INTER	CONNECTION (CALL TRANSPORT AND TERMINATION)	-			-											
	E RATE FOR LOCAL AND ISP-BOUND TRAFFIC	1			1											
SINGL	Single Rate for Local and ISP-bound traffic, per minute of use	1			1											
	(1/1/02 - 12/31/02)					.0015										
RATE	S FOR LOCAL TRANSIT AND MTA TRAFFIC				1	.0010										
	EM SWITCHING															
	Tandem Switching Function Per MOU			OHD		0.0005692										
	Multiple Tandem Switching, per MOU (applies to intial tandem															
	only)			OHD		0.0005692										
	Tandem Intermediary Charge, per MOU*			OHD		0.0015										
	charge is applicable only to transit traffic and is applied in ad	dition to	o appli	cable switching and	l/or interconr	nection charges	i.									
TRUN	K CHARGE															
	Installation Trunk Side Service - per DS0			OHD	TPP++		336.43	57.38								
	Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
	Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
	Dedicated Tandem Trunk Port Service-per DS0** Dedicated Tandem Trunk Port Service-per DS1**			OHD OH1 OH1MS	TDW0P TDW1P	0.00										
** This	s rate element is recovered on a per MOU basis and is included	d in the	End O			0.00	I roto olomonte									
	s rate element is recovered on a per MOO basis and is included MON TRANSPORT (Shared)	a in the	Ena U	Tice Switching and	Tandem Swi	ching, per wo	J rate elements	5								
COMIN	Common Transport - Per Mile, Per MOU	1		OHD	1	0.0000026										
	Common Transport - Fer Mile, Fer MOO Common Transport - Facilities Termination Per MOU			OHD	1	0.0003685										
OCAL INTER	RCONNECTION (DEDICATED TRANSPORT)			OTID	1	0.0003003										
	OFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.0091										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -				1											
	Facility Termination per month			OHL, OHM	1L5NF	25.32	31.78		7.03							
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0091										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	18.44	31.78		7.03							
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0091										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility				1											
	Termination per month			OHL, OHM	1L5NK	18.44	31.78		7.03							
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			0114 0114140	41.5511	0.4050										
	month			OH1, OH1MS	1L5NL	0.1856										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			OH1, OH1MS	1L5NL	88.44	98.47		19.05							
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			OHT, OHTIVIS	ILSINL	00.44	90.47		19.05							
	month			OH3, OH3MS	1L5NM	3.87										
	Interoffice Channel - Dedicated Transport - DS3 - Facility			OTIO, OTIONIO	TEGI VIVI	0.01										
	Termination per month			OH3, OH3MS	1L5NM	1,071.00	219.28		70.56							
LOCA	L CHANNEL - DEDICATED TRANSPORT					,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,										
	Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	21.94	265.84	46.97	37.63	4.00						
	Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM	TEFV4	22.81	266.54	47.67	44.22	5.33						
	Local Channel - Dedicated - DS1 per month			OH1	TEFHG	35.28	216.65	183.54	24.30	16.95						
													_			
	Local Channel - Dedicated - DS3 Facility Termination per month			OH3	TEFHJ	531.91	556.37	343.01	139.13	96.84						
	L INTERCONNECTION MID-SPAN MEET	1	L	L	1	ļ									ļ	
NOTE	: If Access service ride Mid-Span Meet, one-half the tariffed se	rvice Lo	cal Ch													
	Local Channel - Dedicated - DS1 per month	1	<u> </u>	OH1MS	TEFHG	0.00	0.00								 	-
	Local Channel - Dedicated - DS3 per month	1	<u> </u>	OH3MS	TEFHJ	0.00	0.00								 	1
MULT	IPLEXERS Channel System DS4 to DS0 Channel System	1	 	OU4 OU4MC	SATN1	440.77	404.40	74.00	44.00	40.40					 	
	Channelization - DS1 to DS0 Channel System	1		OH1, OH1MS	SATINT	146.77	101.42	71.62	11.09	10.49						
				UH3 UH3M6	SATMS	211 10	100.20	110 61	10 21	20.07						
	DS3 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) per month			OH3, OH3MS OH1, OH1MS	SATNS	211.19 13.76	199.28 10.07	118.64 7.08	40.34	39.07						

LOCAL IN	TERCONNECTION - Georgia												Attachment:		Exhibit: D	1
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			TES(\$)					Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec			Disconnect				Rates(\$)		
		<u> </u>			-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
OCAL INTE	RCONNECTION (CALL TRANSPORT AND TERMINATION)	1			+						1					
	GLE RATE FOR LOCAL AND ISP-BOUND TRAFFIC	1			+						1					
Olive	Single Rate for Local and ISP-bound traffic, per minute of use		1		1						1					
	(1/1/02 - 12/31/02)					.0015										
RATI	ES FOR LOCAL TRANSIT AND MTA TRAFFIC															
TANI	DEM SWITCHING															
	Tandem Switching Function Per MOU			OHD		0.0005692										
	Multiple Tandem Switching, per MOU (applies to intial tandem															
	only)			OHD		0.0005692										
	Tandem Intermediary Charge, per MOU*	<u> </u>	<u> </u>	OHD	<u> </u>	0.0015										
	s charge is applicable only to transit traffic and is applied in ad	dition to	appli	cable switching and	or interconr	nection charges	S.									
IRUI	NK CHARGE	ļ	 	OHD	TDD.		200.00	FC C1								├
	Installation Trunk Side Service - per DS0 Dedicated End Office Trunk Port Service-per DS0**	 		OHD OHD	TPP++ TDE0P	0.00	333.28	56.84			 					
	Dedicated End Office Trunk Port Service-per DS0** Dedicated End Office Trunk Port Service-per DS1**	1	1	0H1 OH1MS	TDE1P	0.00				1	 				1	1
	Dedicated End Office Trunk Port Service-per DS1 Dedicated Tandem Trunk Port Service-per DS0**	1		OHD	TDW0P	0.00										-
+	Dedicated Tandem Trunk Port Service-per DS0*		1	OH1 OH1MS	TDW1P	0.00					1					
** Th	is rate element is recovered on a per MOU basis and is included	in the	End O					S								
	MON TRANSPORT (Shared)	1	1			g, p										
	Common Transport - Per Mile, Per MOU			OHD		0.0000026										
	Common Transport - Facilities Termination Per MOU			OHD		0.0003685										
	RCONNECTION (DEDICATED TRANSPORT)															
INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.0222										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month			OHL, OHM	1L5NF	17.07	36.08									
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0222										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	16.45	36.08									
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			OHL, OHM	1L5NK	0.0222										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	16.45	36.08									
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month Interoffice Channel - Dedicated Tranport - DS1 - Facility		1	OH1, OH1MS	1L5NL	0.4523										
	Termination per month			OH1, OH1MS	1L5NL	78.47	111.75									
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per	1		OTTI, OTTINO	TESINE	70.47	111.75									
	month			OH3, OH3MS	1L5NM	2.72										
	Interoffice Channel - Dedicated Transport - DS3 - Facility	1		.,		1				Ì						
	Termination per month	<u> </u>	L	OH3, OH3MS	1L5NM	788.00	330.77		<u></u>	<u> </u>	<u> </u>				<u> </u>	
LOC	AL CHANNEL - DEDICATED TRANSPORT															
	Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	13.91	382.95	62.40								
	Local Channel - Dedicated - 4-Wire Voice Grade per month	<u> </u>		OHL, OHM	TEFV4	14.99	368.44	64.05			ļ					
	Local Channel - Dedicated - DS1 per month	ļ		OH1	TEFHG	38.36	356.15	312.89			ļ					
	Land Olerand Bulliand Book 5 and 7 and 3	1		0110												
1.00	Local Channel - Dedicated - DS3 Facility Termination per month AL INTERCONNECTION MID-SPAN MEET	 		OH3	TEFHJ	515.91	639.50	426.31			 					
	E: If Access service ride Mid-Span Meet, one-half the tariffed se	rvice ! ^	ral Ch	I annel rate is annlica	hle	 					1				-	
NOTI	Local Channel - Dedicated - DS1 per month	. 7108 E0	Jai Ull	IOH1MS	TEFHG	0.00	0.00			1	 					
	Local Channel - Dedicated - DS3 per month	1		OH3MS	TEFHJ	0.00	0.00			1					1	t
MUL	TIPLEXERS	†			1	5.50	2.20			1						
	Channelization - DS1 to DS0 Channel System	1		OH1, OH1MS	SATN1	126.22	198.22	123.59	l	İ					İ	
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	182.04	280.66	195.33								
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	11.02	12.02	8.66								
	s: If no rate is identified in the contract, the rates, terms, and c															

LOCAL INT	FERCONNECTION - Kentucky												Attachment:		Exhibit: D	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			TES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonred		Nonrecurring					Rates(\$)		
			1		+	-	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
OCAL INTE	RCONNECTION (CALL TRANSPORT AND TERMINATION)				-				1							
	ELE RATE FOR LOCAL AND ISP-BOUND TRAFFIC				-				1							
SING	Single Rate for Local and ISP-bound traffic, per minute of use		1		1											
	(1/1/02 - 12/31/02)					.0015										
RATE	ES FOR LOCAL TRANSIT AND MTA TRAFFIC					.0010										
	DEM SWITCHING															
	Tandem Switching Function Per MOU			OHD		0.0005692										
	Multiple Tandem Switching, per MOU (applies to intial tandem															
	only)			OHD		0.0005692										
	Tandem Intermediary Charge, per MOU*			OHD		0.0015										
	s charge is applicable only to transit traffic and is applied in ad	dition to	appli	cable switching and	or interconr	ection charges	3.									
TRU	NK CHARGE	ļ		OUD	TDD											
	Installation Trunk Side Service - per DS0 Dedicated End Office Trunk Port Service-per DS0**	 	<u> </u>	OHD	TPP++	0.00	334.09	57.12						-	-	├
			<u> </u>	OHD OHAMS	TDE0P TDE1P	0.00										
	Dedicated End Office Trunk Port Service-per DS1** Dedicated Tandem Trunk Port Service-per DS0**			0H1 OH1MS OHD	TDW0P	0.00			-							
	Dedicated Tandem Trunk Port Service-per DS0 Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00			-							
** Thi	is rate element is recovered on a per MOU basis and is included	in the	End O				I rate elements	<u> </u>								
	MON TRANSPORT (Shared)	1111111	I O	l	Tunucin Own	lonning, per mo	o rate element	Ĭ								
	Common Transport - Per Mile, Per MOU			OHD		0.0000026										
	Common Transport - Facilities Termination Per MOU			OHD		0.0003685										
LOCAL INTE	RCONNECTION (DEDICATED TRANSPORT)								1							
INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.01										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month			OHL, OHM	1L5NF	29.11	47.34		22.77							
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0115										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	20.97	47.35		22.77							L
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			OHL, OHM	1L5NK	0.0115										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	20.97	47.35		22.77							
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			OH1, OH1MS	1L5NL	0.23										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility			OH1, OH1MS	1L5NL	96.04	105.52		23.09							
	Termination per month Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			OHT, OHTIMS	ILSINL	96.04	105.52		23.09							
	month			OH3, OH3MS	1L5NM	4.97										
	Interoffice Channel - Dedicated Transport - DS3 - Facility			orio, orioino	1201111											
	Termination per month			OH3, OH3MS	1L5NM	1,175.15	335.40		89.57							
LOCA	AL CHANNEL - DEDICATED TRANSPORT															
	Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	18.57	265.78	46.96	46.79	4.98						
	Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM	TEFV4	19.86	266.48	47.65	47.54	5.73						
	Local Channel - Dedicated - DS1 per month			OH1	TEFHG	40.46	209.60	176.51	30.21	21.07						
	Local Channel - Dedicated - DS3 Facility Termination per month	<u> </u>	1	OH3	TEFHJ	576.05	551.38	338.08	173.00	120.42						
	AL INTERCONNECTION MID-SPAN MEET E: If Access service ride Mid-Span Meet, one-half the tariffed se	ndee La	ool Ch	onnel rete io en!:	l blo	 								-	-	
NOTE	Local Channel - Dedicated - DS1 per month	vice LO	cai ch	annei rate is appiica IOH1MS	TEFHG	0.00	0.00							-	-	├ ──
	Local Channel - Dedicated - DS3 per month	1	 	OH3MS	TEFHJ	0.00	0.00		+					1	1	1
MIII -	TIPLEXERS	 		OT IOIVIO	121110	0.00	0.00									
III OL	Channelization - DS1 to DS0 Channel System	1		OH1, OH1MS	SATN1	113.33	101.40	71.60	13.79	13.04						
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	158.20	199.23	118.62	50.16	48.59						
+-	DS3 Interface Unit (DS1 COCI) per month	1	1	OH1, OH1MS	SATCO	11.80	10.07	7.08			1					
	D33 interface onit (D31 COCI) per month			OTTI, OTTINO	0/100	11.00	10.07	7.00								

LOCAL IN	TERCONNECTION - Louisiana												Attachment:		Exhibit: D	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			TES(\$)					Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonred			Disconnect				Rates(\$)		
			1			1	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
OCAL INTE	RCONNECTION (CALL TRANSPORT AND TERMINATION)		1		-	-					1					-
	BLE RATE FOR LOCAL AND ISP-BOUND TRAFFIC		1		-	-					1					-
Olivo	Single Rate for Local and ISP-bound traffic, per minute of use										1					
	(1/1/02 - 12/31/02)					.0015										
RATE	ES FOR LOCAL TRANSIT AND MTA TRAFFIC															
TANI	DEM SWITCHING															
	Tandem Switching Function Per MOU			OHD		0.0005692										
	Multiple Tandem Switching, per MOU (applies to intial tandem															
	only)			OHD		0.0005692										
	Tandem Intermediary Charge, per MOU*	<u> </u>	<u> </u>	OHD		0.0015										
	s charge is applicable only to transit traffic and is applied in ad	dition to	o appli	cable switching and	l/or interconr	ection charges	S.									
IRUI	NK CHARGE Installation Trunk Side Service - per DS0	1	-	OHD	TPP++	 	334.94	56.98			1				 	1
	Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00	334.94	56.98								
	Dedicated End Office Trunk Port Service-per DS0 Dedicated End Office Trunk Port Service-per DS1**		1	0H1 OH1MS	TDE1P	0.00										
	Dedicated Tandem Trunk Port Service-per DS0**		1	OHD	TDW0P	0.00										
	Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00										
** Th	is rate element is recovered on a per MOU basis and is included	d in the	End O					s								
	MON TRANSPORT (Shared)			J		1										
	Common Transport - Per Mile, Per MOU			OHD		0.0000026										
	Common Transport - Facilities Termination Per MOU			OHD		0.0003685										
	RCONNECTION (DEDICATED TRANSPORT)															
INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.013										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			OHL, OHM	1L5NF	22.60	26.62									
	Facility Termination per month Interoffice Channel - Dedicated Transport - 56 kbps - per mile			OHL, OHW	ILDINF	22.60	20.02									
	per month			OHL, OHM	1L5NK	0.013										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility			OTIL, OTIVI	ILOIVIC	0.013										
	Termination per month			OHL, OHM	1L5NK	15.61	26.62									
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.013										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	15.61	26.62									
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			OH1, OH1MS	1L5NL	0.2652										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility															
	Termination per month			OH1, OH1MS	1L5NL	70.47	79.44									
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			OH3, OH3MS	1L5NM	6.04										
	Interoffice Channel - Dedicated Transport - DS3 - Facility	1	1	OI IS, UNSIVIS	ILOINIVI	6.04										+
	Termination per month			OH3, OH3MS	1L5NM	850.45	158.05								1	
LOCA	AL CHANNEL - DEDICATED TRANSPORT	1	1	55, OTIONIO	. 2014101	000.40	100.00									-
	Local Channel - Dedicated - 2-Wire Voice Grade per month	1		OHL, OHM	TEFV2	18.32	187.51	32.21		1					1	t
	Local Channel - Dedicated - 4-Wire Voice Grade per month	1		OHL, OHM	TEFV4	19.41	187.94	32.63								1
	Local Channel - Dedicated - DS1 per month	1		OH1	TEFHG	39.18	172.34	149.27								
	Local Channel - Dedicated - DS3 Facility Termination per month			OH3	TEFHJ	469.44	438.46	256.30			ļ					
	AL INTERCONNECTION MID-SPAN MEET	1	<u> </u>	<u> </u>	1	ļ				ļ	ļ				ļ	
NOTE	E: If Access service ride Mid-Span Meet, one-half the tariffed se	rvice Lo	cal Ch				2.0-				ļ					
	Local Channel - Dedicated - DS1 per month	1	1	OH1MS	TEFHG	0.00	0.00	-		1	<u> </u>				 	
841.11	Local Channel - Dedicated - DS3 per month	1	-	OH3MS	TEFHJ	0.00	0.00				1				 	1
INIUL	TIPLEXERS Channelization - DS1 to DS0 Channel System	1	1	OH1, OH1MS	SATN1	105.09	88.41	60.76		1	 				1	
1	DS3 to DS1 Channel System per month	1		OH3, OH3MS	SATNS	201.48	172.99	91.25							 	1
	200 to 201 ontained by stein per month	1	1			201.70	112.33									
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	11.78	6.39	4.58								

LOCAL INT	FERCONNECTION - Mississippi												Attachment:		Exhibit: D	1
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			ΓES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge - Manual Sv Order vs.
		ļ				Rec	Nonrec		Nonrecurring		001150	001111		Rates(\$)	001441	
			1		+	-	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
OCAL INTE	RCONNECTION (CALL TRANSPORT AND TERMINATION)				-											+
	LE RATE FOR LOCAL AND ISP-BOUND TRAFFIC	1			+										-	+
SiNG	Single Rate for Local and ISP-bound traffic, per minute of use	1														+
	(1/1/02 - 12/31/02)					.0015										
RATE	ES FOR LOCAL TRANSIT AND MTA TRAFFIC															
	DEM SWITCHING															1
	Tandem Switching Function Per MOU			OHD		0.0005692										1
	Multiple Tandem Switching, per MOU (applies to intial tandem															
	only)			OHD		0.0005692										
	Tandem Intermediary Charge, per MOU*			OHD		0.0015										
	s charge is applicable only to transit traffic and is applied in ad	dition to	appli	cable switching and	or interconr	ection charges										
TRUN	NK CHARGE	ļ	<u> </u>	OUD	TDD		004 **	F0 00								↓
	Installation Trunk Side Service - per DS0		<u> </u>	OHD	TPP++	0.00	334.11	56.98							1	+
	Dedicated End Office Trunk Port Service-per DS0**		1	OHD	TDE0P TDE1P	0.00									-	+
	Dedicated End Office Trunk Port Service-per DS1** Dedicated Tandem Trunk Port Service-per DS0**			0H1 OH1MS OHD	TDW0P	0.00										+
	Dedicated Tandem Trunk Port Service-per DS0 Dedicated Tandem Trunk Port Service-per DS1**	1		OH1 OH1MS	TDW1P	0.00									-	+
** Thi	is rate element is recovered on a per MOU basis and is included	l in the	End O				l rate elements									+
	MON TRANSPORT (Shared)	1	I C	lince owncrining and	Tandem own	lerinig, per mo	J rate elements	•								+
00	Common Transport - Per Mile, Per MOU			OHD		0.0000026										+
	Common Transport - Facilities Termination Per MOU			OHD	1	0.0003685									1	1
OCAL INTE	RCONNECTION (DEDICATED TRANSPORT)															1
INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT															1
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.0098										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month			OHL, OHM	1L5NF	22.52	27.57		7.11							
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0098										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month			OHL, OHM	1L5NK	15.68	27.57		7.11							
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile	1		OHL, OHIVI	ILSINK	13.00	21.51		7.11						-	+
	per month			OHL, OHM	1L5NK	0.0098										<u> </u>
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month			OHL, OHM	1L5NK	15.68	27.57		7.11							
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			OH1, OH1MS	1L5NL	0.201										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility			OTTI, OTTIMO	TEGIVE	0.201										+
	Termination per month			OH1, OH1MS	1L5NL	57.33	82.28		14.90							
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			,												1
	month			OH3, OH3MS	1L5NM	4.76										
	Interoffice Channel - Dedicated Transport - DS3 - Facility															
	Termination per month			OH3, OH3MS	1L5NM	641.90	163.70		60.29							
LOCA	AL CHANNEL - DEDICATED TRANSPORT															
	Local Channel - Dedicated - 2-Wire Voice Grade per month	ļ		OHL, OHM	TEFV2	14.91	194.22	33.36	37.79	3.30						
	Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM OH1	TEFV4 TEFHG	15.99	194.66	33.80	38.27	3.78						
	Local Channel - Dedicated - DS1 per month		1	OH1	TEFHG	36.83	178.50	154.61	22.89	15.74					-	+
	Local Channel - Dedicated - DS3 Facility Termination per month	1	1	ОНЗ	TEFHJ	413.87	454.13	264.47	123.23	86.19					I	1
100	AL INTERCONNECTION MID-SPAN MEET	 		0113	i Li i IJ	413.67	404.13	204.47	123.23	00.19					t	+
	E: If Access service ride Mid-Span Meet, one-half the tariffed se	rvice Lo	cal Ch	annel rate is applica	ble.										†	
	Local Channel - Dedicated - DS1 per month	T		OH1MS	TEFHG	0.00	0.00								1	
	Local Channel - Dedicated - DS3 per month	†		OH3MS	TEFHJ	0.00	0.00		1						1	1
MUL	TIPLEXERS				1				1							1
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	102.85	91.57	62.94	10.87	10.10						
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	170.63	179.17	94.52	34.30	32.82						
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	12.96	6.62	4.74					_			
	s: If no rate is identified in the contract, the rates, terms, and co															

LOCAL IN	TERCONNECTION - South Carolina												Attachment:		Exhibit: D	1
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			ΓES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment: Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
					+	-	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
OCAL INT	ERCONNECTION (CALL TRANSPORT AND TERMINATION)	-			-				1							+
	GLE RATE FOR LOCAL AND ISP-BOUND TRAFFIC	1			1											
SIN	Single Rate for Local and ISP-bound traffic, per minute of use	1			1											
	(1/1/02 - 12/31/02)					.0015										
RAT	ES FOR LOCAL TRANSIT AND MTA TRAFFIC					.0010										†
	DEM SWITCHING				1											
	Tandem Switching Function Per MOU			OHD		0.0005692										
	Multiple Tandem Switching, per MOU (applies to intial tandem															
	only)			OHD		0.0005692										
	Tandem Intermediary Charge, per MOU*			OHD		0.0015										
	is charge is applicable only to transit traffic and is applied in ad	dition to	o appli	cable switching and	or interconr	nection charges										
TRU	NK CHARGE							•		•				_		
	Installation Trunk Side Service - per DS0			OHD	TPP++		335.14	57.16								
	Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										ļ
	Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
	Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
	Dedicated Tandem Trunk Port Service-per DS1**		<u> </u>	OH1 OH1MS	TDW1P	0.00										
	nis rate element is recovered on a per MOU basis and is include	d in the	End O	fice Switching and	Tandem Swit	ching, per MOl	J rate elements	3								
CON	MMON TRANSPORT (Shared)			OUD		0.0000000										
	Common Transport - Per Mile, Per MOU			OHD	 	0.0000026										
OCAL INT	Common Transport - Facilities Termination Per MOU ERCONNECTION (DEDICATED TRANSPORT)			OHD	 	0.0003685										
	EROFFICE CHANNEL - DEDICATED TRANSPORT	-			-				1							
INTE	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.0167										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			OTIL, OTIVI	ILOIVI	0.0107										
	Facility Termination per month			OHL, OHM	1L5NF	24.30	40.63		16.77							
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0167										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility			, ,					1							
	Termination per month			OHL, OHM	1L5NK	16.76	40.63		16.77							
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0167										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	16.76	40.63		16.77							
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			OH1, OH1MS	1L5NL	0.3415										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility															
	Termination per month		<u> </u>	OH1, OH1MS	1L5NL	77.14	89.47		16.39							
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			OUS OUSEAC	41 ENINA	0.00										
	month Interoffice Channel - Dedicated Transport - DS3 - Facility	1	-	OH3, OH3MS	1L5NM	8.02									 	
	Termination per month		1	OH3, OH3MS	1L5NM	880.65	279.37		60.33						1	
1.00	CAL CHANNEL - DEDICATED TRANSPORT	1	-	UI IJ, UI IJIVIJ	ILOINIVI	00.00	219.31		00.33						1	
LOC	Local Channel - Dedicated - 2-Wire Voice Grade per month	1		OHL, OHM	TEFV2	15.33	193.53	33.24	36.72	3.21					1	
	Local Channel - Dedicated - 2-Wire Voice Grade per month	1		OHL, OHM	TEFV4	16.54	193.97	33.68	37.19	3.68					 	
	Local Channel - Dedicated - 4-Wire Voice Grade per Horitin	1	<u> </u>	OH1	TEFHG	42.62	177.87	154.06	22.24	15.30					 	
		1		1	1	.2.02		.550		.0.00					1	1
	Local Channel - Dedicated - DS3 Facility Termination per month		1	ОНЗ	TEFHJ	446.00	452.52	264.53	119.75	83.77					1	
	AL INTERCONNECTION MID-SPAN MEET															
NOT	E: If Access service ride Mid-Span Meet, one-half the tariffed se	rvice Lo	cal Ch	annel rate is applica	ble.											
	Local Channel - Dedicated - DS1 per month			OH1MS	TEFHG	0.00	0.00									
	Local Channel - Dedicated - DS3 per month			OH3MS	TEFHJ	0.00	0.00									
MUL	TIPLEXERS															
	Channelization - DS1 to DS0 Channel System	1		OH1, OH1MS	SATN1	107.57	91.24	62.71	10.56	9.81						
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	144.02	178.54	94.18	33.33	31.90					ļ	
	DS3 Interface Unit (DS1 COCI) per month es: If no rate is identified in the contract, the rates, terms, and c	L	<u> </u>	OH1, OH1MS	SATCO	8.64	6.59	4.73	L						ļ	<u> </u>
	no. It was rate in identified in the contract the rates terms and a															

OCAL INTE	RCONNECTION - Tennessee												Attachment:		Exhibit: D	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			ΓES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment: Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
			1		+	1	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
OCAL INTER	CONNECTION (CALL TRANSPORT AND TERMINATION)				-											
	E RATE FOR LOCAL AND ISP-BOUND TRAFFIC	<u> </u>	1													
SINGL	Single Rate for Local and ISP-bound traffic, per minute of use		1		1						1					
	(1/1/02 - 12/31/02)					.0015										
RATES	S FOR LOCAL TRANSIT AND MTA TRAFFIC					.0013										
	EM SWITCHING															
	Tandem Switching Function Per MOU			OHD		0.0005692										
	Multiple Tandem Switching, per MOU (applies to intial tandem															
	only)			OHD		0.0005692										
	Tandem Intermediary Charge, per MOU*			OHD		0.0015										
	charge is applicable only to transit traffic and is applied in ad-	dition to	o appli	cable switching and	or interconr	nection charges	S.									
TRUN	CHARGE															
	Installation Trunk Side Service - per DS0			OHD	TPP++		334.29	57.01								
	Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
	Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
	Dedicated Tandem Trunk Port Service-per DS0**	<u> </u>	<u> </u>	OHD OH1 OH1MS	TDW0P TDW1P	0.00										
** Thio	Dedicated Tandem Trunk Port Service-per DS1** rate element is recovered on a per MOU basis and is included	l in the	End O			0.00	l roto alamanta									
	rate element is recovered on a per MOO basis and is included ION TRANSPORT (Shared)	in the	Ena O	Tice Switching and	l andem Swii	cning, per wo	J rate elements	5								-
COMIN	Common Transport - Per Mile, Per MOU		1	OHD	1	0.0000026					1					
	Common Transport - Facilities Termination Per MOU		1	OHD	1	0.0003685			1							
OCAL INTER	CONNECTION (DEDICATED TRANSPORT)		1	OHD	1	0.0003003			1							
	OFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.0174										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			,												
	Facility Termination per month			OHL, OHM	1L5NF	18.58	17.37		3.51							
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0174										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	17.98	17.37		3.51							
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
	per month			OHL, OHM	1L5NK	0.0174										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	17.98	17.37		3.51							
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			OH1, OH1MS	1L5NL	0.3562										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility		<u> </u>	OHT, OHTMS	ILDINL	0.3562										
	Termination per month			OH1, OH1MS	1L5NL	77.86	76.27		14.99							
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			OTTI, OTTINO	TESINE	77.00	10.21		14.55							
	month			OH3, OH3MS	1L5NM	2.34										
	Interoffice Channel - Dedicated Transport - DS3 - Facility															
	Termination per month			OH3, OH3MS	1L5NM	848.99	176.56		105.91							
LOCAL	CHANNEL - DEDICATED TRANSPORT															
	Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	19.43	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM	TEFV4	20.56	201.53	24.83	55.52	5.51						
	Local Channel - Dedicated - DS1 per month			OH1	TEFHG	40.99	277.35	233.26	33.18	22.30						
			1		1]	
	Local Channel - Dedicated - DS3 Facility Termination per month	ļ		OH3	TEFHJ	611.30	595.37	304.50	215.82	151.15						1
	LINTERCONNECTION MID-SPAN MEET	l	L	L.,,,	<u> </u>											
NOTE:	If Access service ride Mid-Span Meet, one-half the tariffed ser	rvice Lo	cai Ch			0.00	0.00								 	-
	Local Channel - Dedicated - DS1 per month	 	 	OH1MS OH3MS	TEFHG TEFHJ	0.00	0.00								 	!
MIL! T	Local Channel - Dedicated - DS3 per month PLEXERS	 	-	OHSIVIO	IEFHJ	0.00	0.00				-				 	
MULII	Channelization - DS1 to DS0 Channel System	 	1	OH1, OH1MS	SATN1	80.77	141.87	77.11	44.47	42.62					1	
											 				!	1
	DS3 to DS1 Channel System per month			IOH3 OH3MS	ISATNS	222 08	308 03	108 ⊿7	63/1	4 23	1					
	DS3 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) per month			OH3, OH3MS OH1, OH1MS	SATNS	222.98 17.58	308.03 6.07	108.47 4.66	6.34	4.23						

INTERIM SE	RVICE PROVIDER NUMBER PORTABILITY - Alaba	ma											Attachment:	5	Exhibit: E	
		1									Svc Order	Svc Order		Incremental		Incremental
												Submitted		Charge -	Charge -	Charge -
											Elec					
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc		PAT	TES(\$)				-		Manual Svc		
CATEGORI	KATE ELEMENTO	m	Zone	Воо	0000		IVA.	LO(#)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						B	Nonrec	urring	Nonrecurring	Disconnect		L.	oss	Rates(\$)		
						Recurring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTERIM SER	/ICE PROVIDER NUMBER PORTABILITY															
	RCF, per number ported (Business Line)				TNPBL	2.13	0.65		0.07		3.50		19.99	19.99	19.99	19.99
	RCF, per number ported (Residence Line)				TNPRL	2.13	0.65		0.07		3.50		19.99	19.99	19.99	19.99
	RCF, add'l capacity for simultaneous call forwarding, per															
	additional path					0.32										
	RCF, per service order, per location (Business)				TNPBD		1.44	1.44	1.44	1.44	3.50		19.99	19.99	19.99	19.99
	RCF, per service order, per location (Residence)				TNPRD		1.44	1.44	1.44	1.44	3.50		19.99	19.99	19.99	19.99
INTERIM SER	/ICE PROVIDER NUMBER PORTABILITY - DID															
	DID per number ported (Residence)				TNPDR		1.18		1.18		3.50		19.99	19.99	19.99	19.99
	DID per number ported (Business)				TNPDB		1.18		1.18		3.50		19.99	19.99	19.99	19.99
	DID per service order, per location (Residence)				TNPRD		1.44	1.44	1.44	1.44	3.50		19.99	19.99	19.99	19.99
	DID per service order, per location (Business)				TNPBD		1.44	1.44	1.44	1.44	3.50		19.99	19.99	19.99	19.99
	DID, per trunk termination, Initial				TNPT2	11.84	173.73	51.00	50.43	25.00	3.50		19.99	19.99	19.99	19.99
	If no rate is identified in the contract, the rate for the specifi															
NOTE:	Any element that can be ordered electronically will be billed	l accordi	ng to th	e SOMEC rate listed	d. Please ref	er to BellSouth'	s Business Ru	les for Local (Ordering (BBR-	LO) to determ	ine if a prod	duct can be	ordered elect	ronically. For	those eleme	nts that
canno	be ordered electronically at present per the BBR-LO, the list	ted SOM	EC rate	reflects the charge	that would b	e billed to a CL	EC once electr	onic ordering	capabilities co	me on-line fo	that eleme	nt. Otherwi	se, the manua	al ordering ch	arge, SOMAN	l, will be

10/04/02 Page 1 of 8

INTERIM SE	RVICE PROVIDER NUMBER PORTABILITY - Florida	3											Attachment:	5	Exhibit: E	
		-									Svc Order	Svc Order			Incremental	Incremental
												Submitted		Charge -	Charge -	Charge -
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		DAT	TES(\$)			Elec	,			Manual Svc	
CATEGORI	RATE ELEMENTS	m	Zone	ьсэ	0300		KAI	E3(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						B	Nonrec	urring	Nonrecurring	Disconnect		l	oss	Rates(\$)	l	
						Recurring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTERIM SER	/ICE PROVIDER NUMBER PORTABILITY - RCF															
	RCF, per number ported (Business Line)				TNPBL	2.05	0.4145	0.4145	0.0415	0.0415	3.50	11.90			1.83	
	RCF, per number ported (Residence Line)				TNPRL	2.05	0.4145	0.4145	0.0415	0.0415	3.50	11.90			1.83	
	RCF, Per Additional Path					0.7179										
INTERIM SER	/ICE PROVIDER NUMBER PORTABILITY - DID															
	DID per number ported (Residence)				TNPDR		0.6923	0.6923	0.6923	0.6923	3.50	11.90			1.83	
	DID per number ported (Business)				TNPDB		0.6923	0.6923	0.6923	0.6923	3.50	11.90			1.83	
	DID, per trunk termination, Initial				TNPT2	54.95	161.29	80.58	32.73	32.73	3.50	11.90			1.83	
SERVICE PRO	VIDER NUMBER PORTABILITY (RIPH)															
	RIPH, Functionality, Per Rearrangement						20.08	20.08			3.50	11.90			1.83	
	RIPH, Per Number Ported					1.83	0.2165	0.2165	0.0216	0.0216	3.50	11.90			1.83	
	RIPH, Functionality, Per Central Ofc						90.47	90.47	2.54	2.54	3.50	11.90			1.83	
NOTE:	Any element that can be ordered electronically will be billed	accordi	ng to th	e SOMEC rate listed	. Please refe	er to BellSouth	's Business Ru	les for Local (Ordering (BBR-	LO) to determ	ine if a proc	luct can be	ordered elect	ronically. For	those eleme	nts that
canno	be ordered electronically at present per the BBR-LO, the list	ed SOM	EC rate	reflects the charge t	hat would be	e billed to a CL	EC once electr	onic ordering	capabilities co	me on-line for	that eleme	nt. Otherwi	se, the manua	al ordering ch	arge, SOMAN	, will be

04/12/02 Page 2 of 8

INTERIM SE	RVICE PROVIDER NUMBER PORTABILITY - Geor	gia											Attachment:	5	Exhibit: E	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	ES(\$)				Submitted	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
							Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Recurring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTERIM SERV	/ICE PROVIDER NUMBER PORTABILITY - RCF															
	RCF, per number ported (Business Line)				TNPBL	2.03	0.51				3.50		18.94	18.94		
	RCF, per number ported (Residence Line)				TNPRL	2.03	0.51				3.50		18.94	18.94		
	RCF, add'l capacity for simultaneous call forwarding, per additional path					0.2836										
	RCF, per service order, per location (Business)				TNPBD		2.10	2.10			3.50		18.94	18.94		
	RCF, per service order, per location (Residence)				TNPRD		2.10	2.10			3.50		18.94	18.94		
INTERIM SERV	/ICE PROVIDER NUMBER PORTABILITY - DID															
	DID per number ported (Residence)				TNPDR		0.93				3.50		18.94	18.94		
	DID per number ported (Business)				TNPDB		0.93				3.50		18.94	18.94		
	DID per service order, per location (Residence)				TNPRD		2.10	2.10			3.50		18.94	18.94		
	DID per service order, per location (Business)				TNPBD		2.10	2.10			3.50		18.94	18.94		
	DID, per trunk termination, Initial				TNPT2	10.73	135.47	40.00			3.50		18.94	18.94		
Note:	If no rate is identified in the contract, the rate for the specifi	c service	or func	tion will be as set	forth in applic	able BellSouth	tariff or as neg	otiated by the	Parties upon	request by eit	her Party.					
	Any element that can be ordered electronically will be billed to be ordered electronically at present per the BBR-LO, the list															

04/12/02 Page 3 of 8

INTERIM SEI	RVICE PROVIDER NUMBER PORTABILITY - Kentuc	ky											Attachment:	5	Exhibit: E	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC		RA ⁻	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Recurring	Nonred	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Recurring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				•			•	•								
NOTE:	BellSouth and CLEC will each bear their own costs of provid	ling rem	note cal	I forwarding as an ir	terim numb	er portability o	ption.									

04/12/02 Page 4 of 8

INTEDIM SE	RVICE PROVIDER NUMBER PORTABILITY - Louis	iana											Attachment:	<u> </u>	Exhibit: E	
INTERNITOR	NVICE PROVIDER NOWIDER FOR FABILITY - LOUIS	iaiia			1											
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC		RAT	'ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												•	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
															D130 131	DISC Add I
						Recurring	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Recuiring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTERIM SER	VICE PROVIDER NUMBER PORTABILITY - RCF															
	RCF, per number ported (Business Line)				TNPBL	2.91	0.25	0.25			3.50	15.20				
	RCF, per number ported (Residence Line)				TNPRL	2.91	0.25	0.25			3.50	15.20				
	RCF, Per Additional Path					1.24										
INTERIM SER	VICE PROVIDER NUMBER PORTABILITY - DID															
	DID per number ported (Residence)				TNPDR		0.42	0.42			3.50	15.20				
	DID per number ported (Business)				TNPDB		0.42	0.42			3.50	15.20				
	DID, per trunk termination, Initial				TNPT2	68.47	185.13	68.79			3.50	15.20				
SERVICE PRO	VIDER NUMBER PORTABILITY (RIPH)															
	RIPH, Functionality, Per Rearrangement						19.24	19.24			3.50	15.20				
	RIPH, Per Number Ported					1.62	0.19	0.19			3.50	15.20				
	RIPH, Functionality, Per Central Ofc						79.67	79.67			3.50	15.20				
Note:	If no rate is identified in the contract, the rate for the specific	service	or func	tion will be as set for	orth in applic	able BellSouth	tariff or as neg	otiated by the	Parties upon	request by eitl	ner Party.					
	Any element that can be ordered electronically will be billed											luct can be	ordered elect	ronically. For	those element	nts that
canno	t be ordered electronically at present per the BBR-LO, the list	ted SOM	EC rate	reflects the charge	that would b	e billed to a CL	EC once electr	onic orderina	capabilities co	me on-line for	that eleme	nt. Otherwi	se, the manua	al ordering ch	arge. SOMAN	l. will be

04/12/02 Page 5 of 8

INTERIM SE	RVICE PROVIDER NUMBER PORTABILITY - Missis	sippi											Attachment:	5	Exhibit: E	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	ES(\$)			Svc Order Submitted Elec per LSR	Submitted	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						B	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Recurring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	ICE PROVIDER NUMBER PORTABILITY - RCF															
	RCF, per number ported (Business Line)				TNPBL	3.08	0.2596	0.2596	0.0282	0.0282	3.50	15.75				
	RCF, per number ported (Residence Line)				TNPRL	3.08	0.2596	0.2596	0.0282	0.0282	3.50	15.75				
	RCF, Per Additional Path					1.17										
INTERIM SERV	ICE PROVIDER NUMBER PORTABILITY - DID															1
	DID per number ported (Residence)				TNPDR		0.4335	0.4335	0.4701	0.4701	3.50	15.75				i
	DID per number ported (Business)				TNPDB		0.4335	0.4335	0.4701	0.4701	3.50	15.75				
	DID, per trunk termination, Initial				TNPT2	58.41	191.75	71.25	28.94	28.94	3.50	15.75				
SERVICE PRO	/IDER NUMBER PORTABILITY (RIPH)															
	RIPH, Functionality, Per Rearrangement						19.93	19.93			3.50	15.75				
	RIPH, Per Number Ported					1.96	0.1972	0.1972	0.0214	0.0214	3.50	15.75				
	RIPH, Functionality, Per Central Ofc						85.52	85.52	2.51	2.51	3.50	15.75				
NOTE:	Any element that can be ordered electronically will be billed	accordi	ng to th	e SOMEC rate liste	d. Please ref	er to BellSouth'	s Business Ru	les for Local C	Ordering (BBR-I	O) to determi	ne if a prod	uct can be	ordered elect	ronically. For	r those elemei	nts that
cannot	be ordered electronically at present per the BBR-LO, the lists	ed SOM	EC rate	reflects the charge	that would b	e billed to a CL	EC once electr	onic ordering	capabilities co	me on-line for	that eleme	nt. Otherwi	se, the manua	al ordering ch	arge, SOMAN	, will be

04/12/02 Page 6 of 8

INTERIM SE	RVICE PROVIDER NUMBER PORTABILITY - South	Caroli	na										Attachment:	5	Exhibit: E	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
											Elec				Manual Svc	
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RAT	ES(\$)			per LSR		Order vs.	Order vs.	Order vs.	Order vs.
		m						- (.,			per LSK	per LOK	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						B	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Recurring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTERIM OFF	WAS PROVIDED NUMBER PORTABILITY - DOE															
INTERIM SER	VICE PROVIDER NUMBER PORTABILITY - RCF				TNIDDI	0.00	0.00	0.00	0.00	0.00	0.50	45.00				
-	RCF, per number ported (Business Line)				TNPBL	2.68	0.26	0.26	0.03	0.03	3.50	15.69				
	RCF, per number ported (Residence Line)				TNPRL	2.68	0.26	0.26	0.03	0.03	3.50	15.69				 '
	RCF, Per Additional Path					1.04										 '
	RCF, add'l capacity for simultaneous call forwarding, per															1 '
	additional path					0.3854										
	RCF, per service order, per location (Business)				TNPBD		1.37	1.37	44.70	44.70	3.50	15.69				
	RCF, per service order, per location (Residence)				TNPRD		1.37	1.37	44.70	44.70	3.50	15.69				
INTERIM SER	VICE PROVIDER NUMBER PORTABILITY - DID															
	DID per number ported (Residence)				TNPDR		0.43	0.43	0.47	0.47	3.50	15.69				
	DID per number ported (Business)				TNPDB		0.43	0.43	0.47	0.47	3.50	15.69				1
	DID per service order, per location (Residence)				TNPRD		1.37	1.37	44.70	44.70	3.50	15.69				·
	DID per service order, per location (Business)				TNPBD		1.37	1.37	44.70	44.70	3.50	15.69				
	DID, per trunk termination, Initial				TNPT2	73.62	191.07	191.07	28.84	28.84	3.50	15.69				1
	DID, per trunk termination, Subsequent					73.62	71.00	71.00	28.84	28.84	3.50	15.69				()
SERVICE PRO	VIDER NUMBER PORTABILITY (RIPH)															
	RIPH, Functionality, Per Central Ofc						82.23	82.23	2.50	2.50	3.50	15.69				
	RIPH, Functionality, Per Rearrangement						19.86	19.86	1		3.50	15.69				
	RIPH, Per Number Ported					2.02	0.20	0.20	0.02	0.02	3.50	15.69				
Note:	If no rate is identified in the contract, the rate for the specific	service	or func	tion will be as set	forth in applic	cable BellSouth	tariff or as neg	otiated by the	Parties upon r	equest by eit	ner Party.					
	Any element that can be ordered electronically will be billed											luct can be	ordered elect	ronically. Fo	r those eleme	nts that
canno	t be ordered electronically at present per the BBR-LO, the list	ed SOM	EC rate	reflects the charge	that would b	e billed to a CLE	C once electr	onic orderina	capabilities co	me on-line fo	that eleme	nt. Otherwi	se. the manua	al ordering ch	arge, SOMAN	ı. will be

04/12/02 Page 7 of 8

INTE	RIM SE	RVICE PROVIDER NUMBER PORTABILITY - Tenne	ssee											Attachment:	5	Exhibit: E	
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATE	SORY	RATE ELEMENTS	m	Zone	BCS	USOC		RAT	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
	Nonrecurring Nonrecurring Disconnect OSS Rate														Add'l	Disc 1st	Disc Add'l
	Recurring Nonrecurring Disconnect OSS Rates(\$)																
							Recurring	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTER	ERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF																
		RCF, per number ported (Business Line)				TNPBL	1.50										
		RCF, per number ported (Residence Line)				TNPRL	1.25										
		RCF, add'l capacity for simultaneous call forwarding, per															
		additional path					0.50										
		RCF, per service order, per location (Business)				TNPBD		25.00	25.00			3.50		19.99	19.99	19.99	19.99
		RCF, per service order, per location (Residence)	25.00			3.50		19.99	19.99	19.99	19.99						
		f no rate is identified in the contract, the rate for the specific															
	NOTE:	Any element that can be ordered electronically will be billed	accordi	ng to th	e SOMEC rate listed	. Please refe	er to BellSouth	's Business Ru	les for Local (Ordering (BBR-	LO) to determ	ine if a proc	luct can be	ordered elect	ronically. Fo	those eleme	nts that
	cannot	be ordered electronically at present per the BBR-LO, the list	ed SOM	EC rate	reflects the charge t	hat would be	e billed to a Cl	EC once electi	onic ordering	capabilities co	me on-line fo	r that eleme	nt. Otherwi	se, the manua	al ordering ch	arge, SOMAN	l, will be

04/12/02 Page 8 of 8

RATE ELEMENTS Interim Zone BCS USOC RATES(\$) RATE ELEMENTS R	ODUF/ADUF	/CMDS - Alabama												Attachment:	7	Exhibit: F	
RATE ELEMENTS Intering Zone BCS USOC RATES(\$) RATES(\$	ODOLIADOLI	70m20 Alabama		1 1			1					Svc Order	Svc Order				Incremental
CATEGORY RATE ELEMENTS Interi m Zone BCS USOC RATES(\$) RATE S(\$) RADIF LA Add'I First Add'I First Add'I First Add'I SOMEC SOMAN SOMAN																	
RATE ELEMENTS Concept																	
m Rec Nonrecurring Nondeling Nonrecurring Nondeling No	CATEGORY	DATE ELEMENTS	Interi	Zono	pre	HEOC		D.A	TEQ(¢)								1
Some Some	CATEGORI	RATE ELEMENTS	m	Zone	603	0300		NA.	1 L3(\$)			per LSR	per LSR				
Nonrecurring Nonrecurring Nonrecurring Nonrecurring Disconnect OSS Rates(\$)														Electronic-	Electronic-	Electronic-	
CODUF/ADUF/CMDS/EODUF CODUF: Message Processing, per message N/A														1st	Add'l	Disc 1st	Disc Add'l
ODUF/ADUF/CMDS/EODUF ENHANCED OPTIONAL DAILY USAGE FILE (EODUF) EODUF: Message Processing, per message ACCESS DAILY USAGE FILE (ADUF) ADUF: Data Transmission (CONNECT:DIRECT), per message OPTIONAL DAILY USAGE FILE (ODUF) OPTIONAL DAILY USAGE FILE (ODUF) ODUF: Recording, per message N/A ODUF: Message Processing, per message N/A ODUF: Message Processing, per message N/A ODUF: Message Processing, per message N/A ODUF: Message Processing, per message N/A ODUF: Message Processing, per message N/A ODUF: Message Processing, per message N/A ODUF: Message Processing, per message N/A ODUF: Message Processing, per Magnetic Tape provisioned N/A ODUF: Message Processing, per Magnetic Tape provisioned				1			Dee	Nonre	curring	Nonrecurring	Disconnect			oss	Rates(\$)		-
ENHANCED OPTIONAL DAILY USAGE FILE (EODUF) EODUF: Message Processing, per message 0.22206700 EODUF: Message Processing, per message 0.22206700 EODUF: Message Processing, per message N/A 0.00185100 EODUF: Message Processing, per message N/A 0.0011300 EODUF: Recording, per message N/A 0.0001100 EODUF: Message Processing, per message N/A 0.0001100 EODUF: Message Processing, per message N/A 0.00249900 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.00249900 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.00000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.000000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.000000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.0000000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.00000000000000000000000000000000000							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ENHANCED OPTIONAL DAILY USAGE FILE (EODUF) EODUF: Message Processing, per message 0.22206700 EODUF: Message Processing, per message 0.22206700 EODUF: Message Processing, per message N/A 0.00185100 EODUF: Message Processing, per message N/A 0.0011300 EODUF: Recording, per message N/A 0.0001100 EODUF: Message Processing, per message N/A 0.0001100 EODUF: Message Processing, per message N/A 0.00249900 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.00249900 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.00000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.000000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.000000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.0000000000 EODUF: Message Processing, per Magnetic Tape provisioned N/A 0.00000000000000000000000000000000000																	
EODUF: Message Processing, per message 0.22206700																	
ACCESS DAILY USAGE FILE (ADUF)																	
ADUF: Message Processing, per message							0.22206700										
ADUF: Data Transmission (CONNECT:DIRECT), per message	ACCES																
OPTIONAL DAILY USAGE FILE (ODUF) ODUF: Recording, per message N/A 0.00001100 ODUF: Message Processing, per message N/A 0.00249900 ODUF: Message Processing, per Magnetic Tape provisioned N/A 35.76000000		ADUF: Message Processing, per message				N/A	0.00185100										
OPTIONAL DAILY USAGE FILE (ODUF) ODUF: Recording, per message N/A 0.00001100 ODUF: Message Processing, per message N/A 0.00249900 ODUF: Message Processing, per Magnetic Tape provisioned N/A 35.76000000																	1
ODUF: Recording, per message						N/A	0.00011300										
ODUF: Message Processing, per message N/A 0.00249900																	
ODUF: Message Processing, per Magnetic Tape provisioned N/A 35.76000000						N/A											
		ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.76000000										lacksquare
ODUF: Data Transmission (CONNECT:DIRECT), per message N/A 0.00009400		ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00009400										
CENTRALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)	CENTR	ALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
CMDS: Message Processing, per message N/A 0.00400000		CMDS: Message Processing, per message				N/A	0.00400000				•						
CMCC. Part Transfering (COMMISCE PROFEST) and annual Profest annual Profest ann		CMDC: Data Taranasianian (COMMECT-DIDECT) and managed				NI/A	0.00400000										1
CMDS: Data Transmission (CONNECT:DIRECT), per message N/A 0.00100000 Notes: If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff.			L	<u> </u>				L 4 200		1							-

10/04/02 Page 1 of 8

ODUF/ADUF	F/CMDS - Florida												Attachment:	7	Exhibit: F	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted		Charge -	Charge -	Charge -
											Elec					Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						,			per Lor	per Lor	Electronic-	Electronic-		Electronic-
														Add'l		
													1st	Addi	Disc 1st	Disc Add'l
						Rec		curring	Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C	MDS/EODLIE															
	NCED OPTIONAL DAILY USAGE FILE (EODUF)															
LINIA	EODUF: Message Processing, per message					0.080698										
ACCE	SS DAILY USAGE FILE (ADUF)					0.000090										
ACCE	ADUF: Message Processing, per message		1		N/A	0.001656										
-	ADDI : Message Flocessing, per message				IN/A	0.001030										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012450										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000071										
	ODUF: Message Processing, per message				N/A	0.002146										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.91										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010375										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001	•			•			_			
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 2 of 8

ODUF	/ADUF	/CMDS - Georgia												Attachment:	7	Exhibit: F	
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			""										•	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							_ 1	Nonre	curring	Nonrecurring	Disconnect			oss	Rates(\$)	l	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		MDS/EODUF															
		ICED OPTIONAL DAILY USAGE FILE (EODUF)															
		EODUF: Message Processing, per message					0.0034555										
		S DAILY USAGE FILE (ADUF)															
		ADUF: Message Processing, per message				N/A	0.0079506										
		ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0000434										
	OPTIO	NAL DAILY USAGE FILE (ODUF)															
		ODUF: Recording, per message				N/A	0.0000090										
		ODUF: Message Processing, per message				N/A	0.0046462										
		ODUF: Message Processing, per Magnetic Tape provisioned				N/A	28.85										
		ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0000434										
	CENTR	ALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
		CMDS: Message Processing, per message				N/A	0.004										
		CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
	Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 3 of 8

ODUF	/ADUF	CMDS - Kentucky												Attachment:	7	Exhibit: F	
		•										Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS		Zone	BCS	USOC		RA	TES(\$)			per LSR		Order vs.	Order vs.	Order vs.	Order vs.
			m										•	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																D130 131	DISC Add I
							Rec		curring	Nonrecurring					Rates(\$)		
							1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUE	DUE/C	MDS/EODUF		-													
ODUF/				1													
		CED OPTIONAL DAILY USAGE FILE (EODUF)					0.005000										
		EODUF: Message Processing, per message					0.235889										
		S DAILY USAGE FILE (ADUF)															
		ADUF: Message Processing, per message				N/A	0.001857										
		ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0001245										
	OPTIO	IAL DAILY USAGE FILE (ODUF)															
		ODUF: Recording, per message				N/A	0.0000136										
		ODUF: Message Processing, per message				N/A	0.002506										
		ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.90										
		ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010372										
	CENTR	ALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
		CMDS: Message Processing, per message				N/A	0.004										
		CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001	•			•						
		If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appl	icable BellSout	h tariff.									

04/12/02 Page 4 of 8

ODUF/ADUF	-/CMDS - Louisiana												Attachment:	7	Exhibit: F	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC		R.A	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												-	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonre	curring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/C	PMDS/FODUF					-										
	NCED OPTIONAL DAILY USAGE FILE (EODUF)															
Livina	EODUF: Message Processing, per message				+	0.229779										1
ACCES	SS DAILY USAGE FILE (ADUF)				+	0.220110										1
	ADUF: Message Processing, per message				N/A	0.001825										1
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012147										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000117										
	ODUF: Message Processing, per message				N/A	0.002446										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.54										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010122										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appl	icable BellSout	n tariff.									

04/12/02 Page 5 of 8

ODUF/ADU	F/CMDS - Mississippi												Attachment:	7	Exhibit: F	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc		Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC		RA	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												-	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonre	curring	Nonrecurring	Disconnect			oss	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/	CMDS/EODUF															
	NCED OPTIONAL DAILY USAGE FILE (EODUF)															+
	EODUF: Message Processing, per message				1	0.234915										1
ACCE	SS DAILY USAGE FILE (ADUF)					0.20.000										
	ADUF: Message Processing, per message				N/A	0.001861										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012278										
OPTI	ONAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000063										
	ODUF: Message Processing, per message				N/A	0.002509										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.97										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010232										
CENT	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes	s: If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appl	icable BellSout	h tariff.									

04/12/02 Page 6 of 8

ODUF/ADUF/	CMDS - South Carolina												Attachment:	7	Exhibit: F	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RA	TES(\$)				Svc Order Submitted Manually	Incremental Charge -	Incremental Charge -		Incremental Charge - Manual Svc Order vs. Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonre	curring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/CN	ADS/EODLIE															
	CED OPTIONAL DAILY USAGE FILE (EODUF)															
	EODUF: Message Processing, per message		1			0.241298										
	S DAILY USAGE FILE (ADUF)		1			0.241230										
	ADUF: Message Processing, per message				N/A	0.001856										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012515										
OPTION	IAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000216										
	ODUF: Message Processing, per message				N/A	0.002508										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.84										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010429										
	ALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message		<u> </u>		N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 7 of 8

ODUF/ADUI	F/CMDS - Tennessee												Attachment:	7	Exhibit: F	
												Svc Order Submitted		Incremental Charge -	Incremental Charge -	Incremental Charge -
		Interi														Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC		RAT	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-		Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrecurring		Nonrecurring	Disconnect				Rates(\$)	•	•
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/O	MDC/FODUF															
	NCED OPTIONAL DAILY USAGE FILE (EODUF)															
ENHA	EODUF: Message Processing, per message					0.22977900										
ACCE	SS DAILY USAGE FILE (ADUF)					0.22911900										
ACCE	ADUF: Message Processing, per message				N/A	0.00182500										
	Abor : Wessage Frocessing, per message				13/73	0.00102300										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012147										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.00000440										
	ODUF: Message Processing, per message				N/A	0.00244600										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.54000000										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00003390										
CENTI	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific	service	or fun	ction will be as set	forth in appli	icable BellSout	h tariff.									

04/12/02 Page 8 of 8

ATTACHMENT 2, NETWORK ELEMENTS AND OTHER SERVICES

17.7 Port/Loop Combinations

- 17.7.1 Combinations of port and loop unbundled network elements along with switching and transport unbundled network elements provide local exchange service for the origination or termination of calls. Port/loop combinations support the same local calling and feature requirements as described in the Unbundled Local Switching or Port section of this Attachment 2 and the ability to presubscribe to a primary carrier for intraLATA and/or to presubscribe to a primary carrier for interLATA toll service.
- 17.7.2 BellSouth shall make available UNE port/loop combinations, regardless of whether such combinations are Currently Combined, so long as such combinations are ordinarily combined in BellSouth's network.
- 17.7.3 Except as set forth in section 17.7.4 below, BellSouth shall provide UNE port/loop combinations that are ordinarily combined in BellSouth's network, regardless of whether such combinations are Currently Combined at the cost-based rates in Exhibit C.
- 17.7.4 BellSouth is not required to provide combinations of port and loop network elements on an unbundled basis in locations where, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.
- 17.7.4.1 BellSouth shall not be required to provide local circuit switching as an unbundled network element in density Zone 1, as defined in 47 CFR 69.123 as of January 1, 1999 of the Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, MSAs to XMC if XMC's customer has 4 or more DS0 equivalent lines.
- 17.7.4.2 Notwithstanding the foregoing, BellSouth shall provide combinations of port and loop network elements on an unbundled basis where, pursuant to FCC rules, BellSouth is not required to provide local circuit switching as an unbundled network element and shall do so at the market rates in Exhibit C. If a market rate is not set forth in Exhibit C for a UNE port/loop combination, such rate shall be negotiated by the Parties.
- 17.7.5 BellSouth shall make 911 updates in the BellSouth 911 database for XMC's UNE port/loop combinations. BellSouth will not bill XMC for 911 surcharges. XMC is responsible for paying all 911 surcharges to the applicable governmental agency.

17.7.6 **Combination Offerings**

17.7.6.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

- 17.7.6.2 2-wire voice grade Coin port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.6.3 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.6.4 2-wire CENTREX port, voice grade loop, CENTREX intercom functionality, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.6.5 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.6.6 4-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 17.7.6.7 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 4-wire DS1 Loop with normal serving wire center channelization interface, 2-wire voice grade ports (PBX), 2-wire DID ports, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

17.7.7 **Other UNE Combinations**

17.7.7.1 BellSouth shall provide other Currently Combined and Ordinarily Combined and Not Typically Combined UNE Combinations to XSPEDIUS MANAGEMENT CO., LLC in addition to those specifically referenced in this Sections 17.3 and 17.7.6 above, where available. The Parties disagree about whether or not combinations may be connected to BellSouth tariffed services, and either Party may seek Dispute Resolution in accordance with Section 26.1 of the General Terms and Conditions of this Agreement on that specific issue. However, without either Party waiving its rights, the Parties agree that until such dispute is resolved, combinations shall not be connected to BellSouth tariffed services. Nothing in this Section is intended to nullify or amend Section 17.2.5. To the extent XSPEDIUS MANAGEMENT CO., LLC requests a combination for which BellSouth does not have methods and procedures in place to provide such combination, rates and/or methods and procedures for such combination will be developed pursuant to the BFR/NBR process; provided, however, that any combination specifically set forth in Section 17.3 or 17.7.6 will not require use of the BFR/NBR process.

17.7.7.2 **Rates**

17.7.7.3 The rates for Ordinarily Combined UNE Combinations shall be the sum of the recurring rates and nonrecurring rates for the stand-alone network elements as set forth in Exhibit C of this Attachment. The rates for Currently Combined UNE Combinations shall be the sum of the recurring rates for the stand-alone network elements as set forth in Exhibit C, in addition to a nonrecurring charge set forth in Exhibit C. To the extent XSPEDIUS MANAGEMENT CO., LLC requests a Not Typically Combined Combination, or to the extent XSPEDIUS MANAGEMENT CO., LLC requests any combination for which BellSouth has not developed methods and procedures to provide such combination, rates and/or methods and procedures for such combination shall be established pursuant to the BFR/NBR process; provided, however, that any combination specifically set forth in 17.3 or 17.7.6 will not require use of the BFR/NBR process.

17.8 Ordinarily Combined UNE Combinations

BellSouth shall provide Ordinarily Combined UNE Combinations to XMC as new service in all states, where facilities are available, regardless of whether or not such network element combinations are Currently Combined. Ordinarily Combined UNE Combinations, including those listed in Section 17.3, consist of a loop-transport combination, where the transport may consist of an Interoffice Channel, a Local Channel, or a Local Channel and an Interoffice Channel. These combinations may terminate to XMC's collocation; however collocation is not required. The Parties disagree about whether or not combinations may be connected to BellSouth tariffed services, and either Party may seek Dispute Resolution in accordance with Section 26.1 of the General Terms and Conditions of this Agreement on that specific issue. However, without either Party waiving its rights, the Parties agree that until such dispute is resolved, BellSouth will not connect combinations to tariffed services. Nothing in this Section is intended to nullify or amend Section 17.2.5.

17.8.2 Rates

- 17.8.2.1 The rates for Ordinarily Combined UNE Combinations, which replicate the architecture described in Section 17.3 and 17.7.6, shall be the sum of the recurring and non-recurring rates for that combination as set forth in Exhibit C of this Attachment.
- 17.8.2.2 The rates for Ordinarily Combined UNE Combinations, which do not replicate a combination described in Section 17.3 and 17.7.6, shall be the sum of the recurring rates and nonrecurring rates for the stand-alone network elements as set forth in Exhibit C of this Attachment.
- 17.8.3 To the extent that XMC seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, XMC, at its option, may

request that such rates be determined pursuant to the BFR/NBR process set forth in this Agreement.